

Illinois Power Generating Company 1500 Eastport Plaza Dr. Collinsville, IL 62234

October 25, 2021

Illinois Environmental Protection Agency DWPC – Permits MC #15 Attn: Part 845 Coal Combustion Residual Rule Submittal 1021 North Grand Avenue East P.O. Box 19276 Springfield, IL 62794-9276

Re: Coffeen Power Plant Ash Pond No. 2; IEPA ID # W1350150004-02

Dear Mr. LeCrone:

In accordance with 35 I.A.C. § 845.200, Illinois Power Generating Company (IPGC) is submitting an operating permit application for the Coffeen Power Plant Ash Pond No. 2 (IEPA ID # W1350150004-02). One hardcopy and one digital copy are provided with this submittal.

The permit application was prepared in accordance with 35 I.A.C. § 845.230(d)(2) (Existing, Inactive and Inactive Closed CCR Surface Impoundment that have completed an Agency approved closure before July 30, 2021). This submittal includes the completed permit forms as required by § 845.210.

Sincerely,

Cynthin E ubdy

Cynthia Vodopivec SVP-Environmental Health and Safety

Enclosures

Prepared for

Illinois Power Generating Company 1500 Eastport Plaza Drive Collinsville, Illinois 62234

INITIAL OPERATING PERMIT COFFEEN ASH POND 2

Prepared by



425 South Woods Mill Road, Suite 300 St. Louis, MO 63017

October 25, 2021

TABLE OF CONTENTS

1.	Introduction	1	
	1.1. Facility Information	1	
	1.2. Owner Signatures	2	
	1.3. Legal Description	2	
	1.4. Previous Assessments	2	
2.	Operating Permit	4	
	2.1. Initial Operating Permit	4	
	2.2. History of Construction	4	
	2.3. Permanent Markers	4	
	2.4. Slope Maintenance	4	
	2.5. Groundwater Monitoring	4	
	2.6. Initial Post-Closure Care Plan	5	
	2.7. History of Groundwater Exceedances	5	
	2.8. Financial Assurance Requirements	6	
3.	Permit Application7		

TABLE OF CONTENTS

ATTACHMENTS

Attachment A	Legal Description
Attachment B	History of Construction (845.220)
Attachment E	Permanent Markers (845.130)
Attachment H	Hydrogeologic Site Characterization (845.620)
Attachment I	Groundwater Sampling and Analysis Program (845.640)
Attachment J	Slope Maintenance (845.430)
Attachment K	Post Closure Care Plan (845.780)
Attachment M	History of Known Groundwater Exceedances (845.230)
Attachment N	Financial Assurance Requirements (845.900)



1. INTRODUCTION

Illinois Power Generating Company operates the coal-fired Coffeen Power Plant located in Montgomery County, Illinois. The IEPA assigned identification number assigned to this impoundment is: W1350150004-02 for Ash Pond 2. The National Inventory of Dams (NID) number assigned for Ash Pond 2 by the Illinois Department of Natural Resources (IDNR) is IL50714.

This initial operating permit application was developed in accordance with 35 Ill. Admin. Code 845, Standards for the Disposal of Coal Combustion Residuals in Surface Impoundments (Part 845).

This initial operating permit application is for Ash Pond 2.

1.1. Facility Information

<u>Section 845.210(b)(1)</u>: All permit applications must contain the name, address, email address and telephone number of the operator, or duly authorized agent, and the property owner to whom all inquiries and correspondence shall be addressed.

Facility:	Coffeen Ash Pond 2
	Coffeen Power Plant
	134 Cips Lane
	Coffeen, IL 62017
Owner/Operator:	Illinois Power Generating Company 1500 Eastport Plaza Drive Collinsville, Illinois 62234



1.2. <u>Owner Signatures</u>

<u>Section 845.210(b)(2):</u> All permit applications must be signed by the owner, operator or a duly authorized agent of the operator.

The owner of the Coffeen Power Plant is a corporation.

<u>Section 845.210(b)(3)</u>: An application submitted by a corporation must be signed by a principal executive officer of at least the level of vice president, or his or her duly authorized representative, if that representative is responsible for the overall operation of the facility described in the application form.

The signature of Cynthia Vodopivec on behalf of Illinois Power Generating Company can be found in the permit applications located in Section 3.

1.3. Legal Description

<u>Section 845.210(c):</u> All permit applications must contain a legal description of the facility boundary and a description of the boundaries of all units included in the facility.

A legal description has been developed in compliance with Section 845.210(c) and is included in Attachment A.

1.4. Previous Assessments

<u>Section 845.210(d):</u> Previous Assessments, Investigations Plans, and Programs

Previous assessments were performed in accordance with 40 CFR § 257 and are referenced within the permit application and included in the appropriate Attachments.

<u>Section 845.210(d)(1):</u> The Agency may approve the use of any hydrogeologic site investigation or characterization, groundwater monitoring well or system, or groundwater monitoring plan, bearing the seal and signature of an Illinois Licensed Professional Geologist or Licensed Professional Engineer, completed before April 21, 2021 to satisfy the requirements of this Part.

A previous hydrogeologic site investigation or characterization, groundwater monitoring well or system, or groundwater monitoring plan have been completed with a seal from an Illinois Licensed Professional Geologist or Licensed Professional Engineer. However, field investigations have been completed that supplement that work that will be utilized in the following sections of this report.



<u>Section 845.210(d)(2):</u> For existing CCR surface impoundments, the owner or operator of the CCR surface impoundment may use a previously completed location restriction demonstration required by Section 845.300 (Placement Above the Uppermost Aquifer), Section 845.310 (Wetlands), Section 845.320 (Fault Areas), Section 845.330 (Seismic Impact Zones), and Section 845.340 (Unstable Areas) provided that the previously completed assessments meet the applicable requirements of those Sections.

No previous assessments are provided for Section 845.300 (Placement Above the Uppermost Aquifer), Section 845.310 (Wetlands), Section 845.320 (Fault Areas), Section 845.330 (Seismic Impact Zones), and Section 845.340 (Unstable Areas).

<u>Section 845.210(d)(3)</u>: For existing CCR surface impoundments, the owner or operator of the CCR surface impoundment may use a previously completed assessment to serve as the initial assessment required by Section 845.440 (Hazard Potential Classification Assessment), Section 845.450 (Structural Stability Assessment) and Section 845.460 (Safety Factor Assessment) provided that the previously completed assessment: A) Was not completed more than five years ago; and B) Meets the applicable requirements of those Sections.

No previous assessments are provided for Section 845.440 (Hazard Potential Classification Assessment), Section 845.450 (Structural Stability Assessment) and Section 845.460 (Safety Factor Assessment).

<u>Section 845.210(d)(4):</u> For inactive closed CCR surface impoundments, the owner or operator of the CCR surface impoundment may use a post-closure care plan previously approved by the Agency.

No post-closure care plan was previously approved by the Agency.



2. **OPERATING PERMIT**

2.1. Initial Operating Permit

<u>Section 845.230(d)</u>: Initial Operating Permit for Existing, Inactive and Inactive Closed CCR Surface Impoundments

The Coffeen Ash Pond 2 as defined by IEPA is a closed inactive surface impoundment that has not completed post-closure care. Per Part 845, Illinois Power Generating Company is submitting an initial operating permit application to IEPA by October 31, 2021. The following sections contain information or references to documents required for the Operating Permit application (Section 845.230).

2.2. History of Construction

<u>Section 845.230(d)(2)(A):</u> The history of construction specified in Section 845.220(a)(1);

The history of construction prepared in 2016 pursuant to 40 CFR § 257.73(c) is provided in Attachment B.

2.3. Permanent Markers

<u>Section 845.230(d)(2)(E):</u> Evidence of permanent markers required by Section 845.130 have been installed;

Evidence of permanent markers at the Coffeen Ash Pond 2 as required by Section 845.130 is provided in Attachment E.

2.4. Slope Maintenance

<u>Section 845.230(d)(2)(F)</u>: Documentation that the CCR surface impoundment, if not incised, will be operated and maintained with one of the forms of slope protection specified in Section 845.430;

The Coffeen Ash Pond 2 is not incised. Documentation of slope protection as required by Section 845.430 is provided in Attachment J.

2.5. Groundwater Monitoring

<u>Section 845.230(d)(2)(I):</u> Groundwater monitoring information:

The groundwater monitoring information for the Coffeen Ash Pond 2 is described in the following sections.

<u>Section 845.230(d)(2)(I)(i):</u> Hydrogeologic site characterization (see Section 845.620);

Hydrogeologic site characterization for the Coffeen Ash Pond 2 is provided in Attachment H.

<u>Section 845.230(d)(2)(I)(ii):</u> Design and construction plans of a groundwater monitoring system (see Section 845.630);

Design and construction plans of a groundwater monitoring system are provided in Attachment I.

<u>Section 845.230(d)(2)(I)(iii):</u> A groundwater sampling and analysis program that includes selection of the statistical procedures to be used for evaluating groundwater monitoring data (see Section 845.640); and

A groundwater sampling and analysis program that meets the requirements of Section 845.640 is provided in Attachment I.

<u>Section 845.230(d)(2)(I)(iv)</u>: Proposed groundwater monitoring program that includes a minimum of eight independent samples for each background and downgradient well (see Section 845.650(b));

A proposed groundwater monitoring program that meets the requirements of Section 845.650(b) is provided in Attachment I.

2.6. Initial Post-Closure Care Plan

<u>Section 845.230(d)(2)(K):</u> Initial written post-closure care plan, if applicable (see Section 845.780(d));

The Coffeen Ash Pond 2 closure was completed by capping the CCR in place. The initial post closure care plan was developed in accordance with Section 845.780 and is provided in Attachment K.

2.7. History of Groundwater Exceedances

<u>Section 845.230(d)(2)(M)</u>: History of known exceedances of the groundwater protection standards in Section 845.600, and any corrective action taken to remediate the groundwater;

A history of known exceedances and any corrective action taken is provided in Attachment M.



2.8. Financial Assurance Requirements

<u>Section 845.230(d)(2)(N):</u> A certification that the owner or operator meets the financial assurance requirements of Subpart I;

A certification meeting the requirement of Section 845.230(d)(2)(N) stating that the Owner meets the financial assurance requirements of *Subpart I* is provided in Attachment N.



3. PERMIT APPLICATION

All permit applications must be made on the forms prescribed by the Agency and must be mailed or delivered to the address designated by the Agency on the forms. The permit applications (CCR-1 and CCR-2OE) are provided below.

Fo CC	rm R 1	Illinois Environmental Protection Agency CCR Surface Impoundment Permit Application			icy cation	
		Form CC	Form CCR 1 – General Provisions			
Bureau of Water ID Number: For IEPA Use Only						
cc						
Facility Name:						
S	ECTION	1: FACILITY, OPERATOR, AND C	WNER INFO	RMATION (35 III. Adm	n. Code 845.210(b))	
	1.1	Facility Name				
	1.2	Illinois EPA CCR Permit Number (if a	pplicable)			
	1.3	Facility Contact Information				
tion		Name (first and last)	Title		Phone Number	
Informa		Email address				
wner	1.4	Facility Mailing Address				
and O		Street or P.O. box				
oerator,		City or town	State		Zip Code	
ty, Ol	1.5	Facility Location				
Facili		Street, route number, or other specific	c identifier			
		County name	County code	(if known)		
		City or town	State		Zip Code	
	1.6	Name of Owner/Operator	I			

Jfo	1.7	Owner/Operator Contact Information						
ner lı		Name (first and last)	Title	Phone Number				
MO br		Email address						
or, aı								
erat	1.8	Owner/Operator Mailing Address						
lity, Op		Street or P.O. box						
Faci		City or town	State	Zip Code				
		SECTION 2: LEGAL DESCR	IPTION (35 III. Adm. Code 845.21	0(c))				
ion	2.1	Legal Description of the facility bounda	ary					
cript								
I Des								
Lega								
SE	CTION 3	: PUBLICLY ACCESSIBLE INTER	NET SITE REQUIREMENTS (35 II	I. Adm. Code 845.810)				
	3.1							
t Site								
ernei								
Int	32	Is/are the website(s) titled "Illinois CCF	R Rule Compliance Data and Informati	on"				
	0.2	Yes	lo					
		SECTION 4: IMPO	UNDMENT IDENTIFICATION					
ation	4.1	List all the impoundment identification indicate that you have attached a writte	numbers for your facility and check the en description for each impoundment.	e corresponding box to				
tifica			Attached writte	en description				
Iden			Attached writte	en description				
ment			Attached writte	en description				
Ipun			Attached writte	en description				
odu			Attached writte	en description				
			Attached writte	en description				

		Att	ached wri	tten desc	ription	
		Att	ached wri	tten desc	ription	
		Att	ached wri	tten desc	ription	
		Att	ached wri	tten desc	ription	
		SECTION 5: CHECKLIST AND CERTIFICATION S	TATEM	ENT		
5.1 In Column 1 below, mark the sections of Form 1 that you have completed and are submitt application. For each section, specify in Column 2 any attachments that you are enclosin				ubmitting with closing.	n your	
		Column 1			Column 2	
ent		Section 1: Facility, Operator, and Owner Information		w/attacl	nments	
tem		Section 2: Legal Description		w/attacl	nments	
ו Sta		Section 3: Publicly Accessible Internet Site Requirement		w/attacl	nments	
atior		Section 4: Impoundment Identification		w/attacl	nments	
tifica	5.2	Certification Statement				
Checklist and Cei		I certify under penalty of law that this document and all attachment or supervision in accordance with a system designed to assure the and evaluate the information submitted. Based on my inquiry of the system, or those persons directly responsible for gathering the int to the best of my knowledge and belief, true, accurate, and comp significant penalties for submitting false information, including the for knowing violations.	nts were p lat qualifie he persor formation, lete. I am possibility	repared used person or person the inform aware the y of fine a	under my dire nel properly g ns who man mation subm at there are and imprisonr	ection gather age the itted is, nent
0		Name (print or type first and last name) of Owner/Operator			Official Title	9
Signature Cynthin E Wdg				Date Signed		

Fo CCR

Construction History

Fo CR	rm 2OE	Illinois Environmental Protection Agency					
		CCR Surface Impoundme Form CCR 2OE – Initial Operating Pe Surface Impoundment Where a Has Been Completed E	ent Permit Application ermit for Existing or Inactive CCR n Agency-approved Closure Before July 30, 2021				
Bui	reau of	Water ID Number:	For IEPA Use Only				
CC Ini	R Pern tial Per	nit Number: mit					
Fac Co	ility N a offeen F	ame: Power Plant					
SE	CTION	1: CONSTRUCTION HISTORY (35 III. Adm. Cod	e 845.220 and 35 III. Adm. Code 845.230)				
	1.1	CCR surface impoundment name.					
	1.2	Identification number of the CCR surface impoundme	ent (if one has been assigned by the Agency).				
-		W1350150004 - 02					
	1.3	Describe the boundaries of the CCR surface impoundment (35 III. Adm. Code 845.210 (c)).					
-	1.4	State the purpose for which the CCR surface impoun	dment is being used.				
	4 5						
ISIH	1.5	How long has the CCR surface impoundment been in	operation?				
ruction	1.6	List the types of CCR that have been placed in the C	CR surface impoundment.				
CONST							
-	1.7	List the name of the watershed within which the CCR	surface impoundment is located.				
	1.8	What is the size in acres of the watershed within which	ch the CCR surface impoundment is located?				
-	1 9	Check the corresponding boxes to indicate that you h	have attached the following:				
		A description of the physical and engineering materials on which the CCR surface impound	properties of the foundation and abutment dment is constructed.				

		۲ ر	A statement of used in constru	the type, size, range, and physical and engineering properties of the materials cting each zone or stage of the CCR surface impoundment.				
		, S	A statement of surface impour	the method of site preparation and construction of each zone of the CCR dment.				
		, C	A statement of of the CCR sur	the approximate dates of construction of each successive stage of construction face impoundment.				
		[Drawings satist	ying the requirements of 35 III. Adm. Code 845.220(a)(1)(F).				
ح		ļ	A description o	f the type, purpose, and location of existing instrumentation.				
listo		Area Capacity Curves for the CCR Impoundment.						
ction H		A C	A description of calculations use	f each spillway and diversion design features and capacities and provide the ed in their determination.				
onstru		(The constructic CCR surface in	n specifications and provisions for surveillance, maintenance, and repair of the npoundment.				
C	1.10.1	Is there a	any record or k	nowledge of structural instability of the CCR surface impoundment?				
			Yes	No				
	1.10.2	lf you an	swered yes to	Item 1.10.1, provide detailed explanation of the structural instability.				
	See Attachment B							
		SECTION 2: ATTACHMENTS						
	2.1	Check the corresponding boxes to indicate that you have attached the following:						
		E		ig boxes to indicate that you have attached the following.				
ŝ		i	Evidence that t installed.	he permanent markers required by 35 III. Adm. Code 845.130 have been				
chments		i [(Evidence that t installed. Documentation operated and n Code 845.430.	he permanent markers required by 35 III. Adm. Code 845.130 have been demonstrating that the CCR surface impoundment, if not incised, will be naintained with one of the forms of slope protection specified in 35 III. Adm.				
Attachment		i C C E E	Evidence that t installed. Documentation operated and n Code 845.430. Emergency Act 845.520(e).	he permanent markers required by 35 III. Adm. Code 845.130 have been demonstrating that the CCR surface impoundment, if not incised, will be naintained with one of the forms of slope protection specified in 35 III. Adm. ion Plan and accompanying certification required by 35 III. Adm. Code				
Attachment		i C C C C C C C C C C C C C C C C C C C	Evidence that t installed. Documentation operated and n Code 845.430. Emergency Act 845.520(e). Written post-clo	he permanent markers required by 35 III. Adm. Code 845.130 have been demonstrating that the CCR surface impoundment, if not incised, will be naintained with one of the forms of slope protection specified in 35 III. Adm. ion Plan and accompanying certification required by 35 III. Adm. Code osure care plan, if applicable (see 35 III. Adm. Code 845.780(d)).				
Attachment		i C C E E E E E E E E E E E E E E E E E	Evidence that t installed. Documentation operated and n Code 845.430. Emergency Act 845.520(e). Written post-clo History of know 845.600, and a	he permanent markers required by 35 III. Adm. Code 845.130 have been demonstrating that the CCR surface impoundment, if not incised, will be haintained with one of the forms of slope protection specified in 35 III. Adm. ion Plan and accompanying certification required by 35 III. Adm. Code osure care plan, if applicable (see 35 III. Adm. Code 845.780(d)). In exceedances of the groundwater protection standards in 35 III. Adm. Code ny corrective action taken to remediate the groundwater.				
Attachment		i C C C C C C C C C C C C C C C C C C C	Evidence that t installed. Documentation operated and n Code 845.430. Emergency Act 845.520(e). Written post-clo History of know 845.600, and a	he permanent markers required by 35 III. Adm. Code 845.130 have been demonstrating that the CCR surface impoundment, if not incised, will be haintained with one of the forms of slope protection specified in 35 III. Adm. ion Plan and accompanying certification required by 35 III. Adm. Code osure care plan, if applicable (see 35 III. Adm. Code 845.780(d)). In exceedances of the groundwater protection standards in 35 III. Adm. Code ny corrective action taken to remediate the groundwater.				
Attachment	3.1	i Check th monitorin	Evidence that t installed. Documentation operated and n Code 845.430. Emergency Act 845.520(e). Written post-clo History of know 845.600, and a SECT ne correspondir ng information:	he permanent markers required by 35 III. Adm. Code 845.130 have been demonstrating that the CCR surface impoundment, if not incised, will be haintained with one of the forms of slope protection specified in 35 III. Adm. ion Plan and accompanying certification required by 35 III. Adm. Code osure care plan, if applicable (see 35 III. Adm. Code 845.780(d)). I'm exceedances of the groundwater protection standards in 35 III. Adm. Code ny corrective action taken to remediate the groundwater. TION 3: GROUNDWATER MONITORING ng boxes to indicate whether you have attached the following groundwater				

ter		Design and construction plans of a groundwater monitoring system meeting the requirements of 35 III. Adm. Code 845.630.
oundwat		A groundwater sampling and analysis program that includes section of the statistical procedures to be used for evaluating groundwater monitoring data, required by 35 III. Adm. Code 845.640.
อ		Proposed groundwater monitoring program that includes a minimum of eight independent samples for each background and downgradient well, required by 35 III. Adm. Code 845.650(b).

ATTACHMENT A





LEGEND

— - - — SECTION LINE

- **RESTRICTED USE BOUNDARY**
- FOUND SURVEY MARKER AS NOTED
- SET 5/8" IRON REBAR (UNLESS OTHERWISE NOTED)
- BACKGROUND CCR MONITORING WELL
- ┍┛┙ DOWNGRADIENT CCR MONITORING WELL

SURVEY NOTE:

THIS DRAWING AND THE INFORMATION SHOWN HERE ON WAS OBTAINED FROM DATA COLLECTED FROM A FIELD SURVEY MADE BY INGENAE, LLC BETWEEN MARCH 23 THROUGH APRIL 20, 2021. SURVEY COORDINATES, **BEARINGS & DISTANCES ARE REFERENCED TO ILLINOIS** WEST 1202 STATE PLANE COORDINATE SYSTEM NAD 1983.

> Land Description of the Coffeen Power Plant **Closed Ash Pond 2 Restricted Use Area** 78.22 Acres

Part of the Southwest Quarter of Section 11, Township 7 North, Range 3 West of the Third Principal Meridian, Montgomery County, Illinois being more particularly described as

Commencing at the Brass Plug in concrete found at the West Quarter Corner of Section 11, from which bears an Iron Pin at the Northwest corner of Section 11 North 1 degree 10 minutes 50 seconds West a distance of 2613.80 feet; thence from said commencement point at the West Quarter Corner of Section 11, North 89 degrees 53 minutes 06 seconds East, along the South line of the Northwest Quarter of Section 11, being the same as the North line of the Southwest Quarter of section 11 a distance of 2640.27 feet to the 2 inch Iron Pipe at the Center of Section 11, from which bears an Iron Pin at the North Quarter Corner of Section 11 North 1 degree 03 minutes 37 seconds West a distance of 2635.19 feet; thence from said 2 inch Iron Pipe at the Center of Section 11 South 89 degrees 53 minutes 06 seconds West back along the North line of the Southwest Quarter of Section 11 a distance of 299.42 feet; thence South 0 degrees 04 minutes 33 seconds East a distance of 23.65 feet to the Point of Beginning of the Tract described herein; thence continuing South 0 degrees 04 minutes 33 seconds East a distance of 1000.00 feet; thence South 20 degrees 12 minutes 37 seconds West a distance of 539.75 feet; thence South 89 degrees 34 minutes 10 seconds West a distance of 2020.00 feet; thence North 23 degrees 23 minutes 56 seconds West a distance of 220.47 feet; thence North 0 degrees 19 minutes 57 seconds East a distance of 1316.35 feet; thence North 89 degrees 55 minutes 27 seconds East a distance of 2285.00 feet to the Point of Beginning and containing 78.22 Acres.

	IngenAE 502 Earth City Plaza, Suite 120 Earth City, MO 63045 www.ingenae.com
t	Submissions / Revisions: Date: 1
HE V	Copyright © 2021 IngenAE, LLC www.ingenae.com DO NOT SCALE PLANS Copying, Printing, Software and other processes required to produce these prints can stretch or shrink the actual paper or layout. Therefore, scaling of this drawing may be inaccurate. Contact IngenAE with any need for additional dimensions or clarifications. Drawing Name: CLOSED ASH POND 2 RESTRICTED USE
	DOUNDART EXHIBITDate:Project No.9/22/2021Project No.Type:Drawing No.SITEDrawing No.Drawn By:1Approved By:MGScale:AS NOTED

SURVEYOR CERTIFICATE:

THIS IS TO CERTIFY THAT WE, INGENAE, LLC, HAVE AT THE REQUEST OF AND FOR TH EXCLUSIVE USE OF THE OWNERS, PERFORMED A SURVEY OF THE TRACT AS SHOWN HEREON AND THAT THIS IS A TRUE REPRESENTATION OF THAT SURVEY. THIS PLAT AND THE SURVEY FROM WHICH IT IS BASED WERE DONE IN ACCORDANCE WITH THE "MINIMUM STANDARDS OF PRACTICE" FOR LAND SURVEYING IN THE STATE OF

PROFESSIONAL DESIGN FIRM LICENSE NO. 184.007588-0010

MICHAEL J. GRAMINSKI, I.P.L.S. NO. 035.002901 EXPIRES: 11/30/2022



DATE





LEGEND

_____ SECTION LINE

- --- RESTRICTED USE BOUNDARY
- FOUND SURVEY MARKER AS NOTED
- SET 5/8" IRON REBAR (UNLESS OTHERWISE NOTED)
- BACKGROUND CCR MONITORING WELL
- DOWNGRADIENT CCR MONITORING WELL

SURVEY NOTE:

THIS DRAWING AND THE INFORMATION SHOWN HERE ON WAS OBTAINED FROM DATA COLLECTED FROM A FIELD SURVEY MADE BY INGENAE, LLC BETWEEN MARCH 23 THROUGH APRIL 20, 2021. SURVEY COORDINATES, BEARINGS & DISTANCES ARE REFERENCED TO ILLINOIS WEST 1202 STATE PLANE COORDINATE SYSTEM NAD 1983.

IngenAE 502 Earth City Plaza, Suite 120 Earth City, MO 63045 www.ingenae.com
Submissions / Revisions: Date: 1
LuminantProject Name & Location:COFFEENPOWER PLANT134 SIPS LANECOFFEEN, IL62017
Copyright © 2021 Www.figenate.com DO NOT SCALE PLANS Copying, Printing, Software and other processes required to produce these prints can stretch or shrink the actual paper or layout. Therefore, scaling of this drawing may be inaccurate. Contact IngenAE with any need for additional dimensions or clarifications. Drawing Name: CLOSEDD ASH POND 2 RESTRICTED USE BOUNDARY EXHIBIT Date: Project No. 9/22/2021 Project No. Type: Drawing No. SITE Drawn By: CB Approved By: MG Scale: AS NOTED Kenter

ATTACHMENT B



October 2016

Illinois Power Generating Company 134 CIPS Lane Coffeen, IL 62017

RE: History of Construction USEPA Final CCR Rule, 40 CFR § 257.73(c) Coffeen Power Station Coffeen, Illinois

On behalf of Illinois Power Generating Company, AECOM has prepared the following history of construction for Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond at the Coffeen Power Station in accordance with 40 CFR § 257.73(c).

BACKGROUND

40 CFR § 257.73(c)(1) requires the owner or operator of an existing coal combustion residual (CCR) surface impoundment that either (1) has a height of five feet or more and a storage volume of 20 acre-feet or more, or (2) has a height of 20 feet or more to compile a history of construction by October 17, 2016 that contains, to the extent feasible, the information specified in 40 CFR § 257.73(c)(1)(i)–(xii).

The history of construction presented herein was compiled based on existing documentation, to the extent that it is reasonably and readily available (see 80 Fed. Reg. 21302, 21380 [April 17, 2015]), and AECOM's site experience. AECOM's document review included record drawings, geotechnical investigations, construction specifications, etc. for Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond at the Coffeen Power Station.



HISTORY OF CONSTRUCTION

§ 257.73(c)(1)(i): The name and address of the person(s) owning or operating the CCR unit; the name associated with the CCR unit; and the identification number of the CCR unit if one has been assigned by the state.

Owner:	Illinois Power Generating Company
Address:	1500 Eastport Plaza Drive Collinsville, IL 62234
CCR Units:	Ash Pond No. 1 Ash Pond No. 2 GMF Pond, IDNR Dam ID No. IL50579 GMF Recycle Pond, IDNR Dam ID No. IL50578

Ash Pond No. 1 and Ash Pond No. 2 do not have a state assigned identification number.

§ 257.73(c)(1)(ii): The location of the CCR unit identified on the most recent USGS $7^{1}/_{2}$ or 15 minute topographic quadrangle map or a topographic map of equivalent scale if a USGS map is not available.

The locations of Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond have been identified on an USGS 7-1/2 minute topographic quadrangle map in **Appendix A**.

§ 257.73(c)(1)(iii): A statement of the purpose for which the CCR unit is being used.

The following captures the purposes of the CCR units:

- Ash Pond No. 1 is being used to store and dispose of bottom ash and other-non-CCR waste and to clarify recycled process water for plant operations. Ash Pond No. 2 (inactive) was used to store and dispose of bottom ash and fly ash.
- The GMF Pond is being used to store and dispose of gypsum and to clarify recycled process water for plant operations.
- The GMF Recycle Pond was used to store and dispose of gypsum from the plant's scrubber operations prior to the in-service date of the GMF Pond in 2010. The GMF Recycle Pond currently only receives and stores clear process water from the GMF Pond.

Notice of intent to close Ash Pond No. 2 was provided in November, 2015.¹

¹ This history of construction report was prepared on a facility-wide basis for CCR surface impoundments at the Coffeen Power Station. The inclusion of Ash Pond No. 2 in this history of construction report does not concede and should not be construed to concede that Ash Pond No. 2 is subject to the Design Criteria or all Operating Criteria in the CCR Rule.



§ 257.73(c)(1)(iv): The name and size in acres of the watershed where the CCR unit is located.

Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond are located in the Coffeen Lake Watershed with a 12-digit Hydrologic Unit Code (HUC) of 071402030304 and a drainage area of 11,695 acres (USGS, 2016).

§ 257.73(c)(1)(v): A description of the physical and engineering properties of the foundation and abutment materials on which the CCR unit is constructed.

The foundation and abutment materials of Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond consist of native fine-grained soils of wind-blown origin (loess), with some coarse-grained layers, underlain by glacial till. The physical properties of the finegrained soils are described as low- to medium-plasticity silty clay, sandy lean clay, or lean clay with sand, often with trace amounts of gravel; or high plasticity fat clay, often with trace amounts of sand. The clay soils vary from soft to very stiff, moist to wet, and brown to gray. The physical properties of the coarse-grained soils are described as clayey sand, silty sand, or fine to coarse sand, with trace amounts of gravel. The sand is wet and varies from loose to dense and brown to gray. A thin layer of native silty or sandy lean clay is located immediately above the glacial till deposits. The clay is very soft to medium stiff, low to medium plasticity, wet, and orange brown to gray. The physical properties of the glacial till are described as lean clay, or silty to sandy lean clay, with trace amounts of fine gravel, hard, low plasticity, moist to wet, and brown to gray. An available summary of the engineering property typical ranges of the foundation and abutment materials is presented in Table 1 below. The engineering properties are based on previous geotechnical explorations and laboratory testing.

Ash Pond No. 1 and Ash Pond No. 2 are enclosed impoundments with embankments and do not have abutments. The GMF Pond and GMF Recycle Pond were constructed as incised impoundments enclosed by embankments.

Material	Unit Weight	Effective (dr Strength I	rained) Shear Parameters	Total (undrained) Shear Strength Parameters	Post-Earthquake Shear Strength
	(pcf)	Cohesion, c' (psf)	Friction Angle, ¢' (deg)	S _u /p'	S _{ur} /p'
Foundation Clay (Under Embankment)	125	0	32	$S_u/p' = 0.39-0.45,$ Min. $S_u = 700 \text{ psf}$	Peak Undrained
Foundation Clay (Free Field)	125	0	30	$S_u/p' = 0.24-0.28$, Min. $S_u = 450 \text{ psf}$	Peak Undrained
Soft Foundation Clay	125	0	30	$S_u/p' = 0.22-0.28$, Min. $S_u = 275 \text{ psf}$	$S_u/p' = 0.13-0.16,$ Min. $S_{ur} = 200 \text{ psf}$
Glacial Till	135	0	40	$S_u/p' = 0.45-0.64,$ Min. $S_u = 700 \text{ psf}$	Peak Undrained

 Table 1. Summary of Foundation and Abutment Material Engineering Properties



§ 257.73(c)(1)(vi): A statement of the type, size, range, and physical and engineering properties of the materials used in constructing each zone or stage of the CCR unit; the method of site preparation and construction of each zone of the CCR unit; and the approximate dates of construction of each successive stage of construction of the CCR unit.

Physical properties for the embankment construction materials for Ash Pond No. 1, Ash Pond No. 2, GMF Pond, and GMF Recycle Pond are described as silty clay, sandy lean clay, or lean clay with sand, with trace amounts of fine gravel. The fill is soft to very stiff in consistency, low to medium plasticity, moist to wet, and brown to gray. Trace amounts of organic material and ash are present. The embankment fill generally appears to be well-compacted. An available summary of the engineering properties of the embankment construction materials is presented in **Table 2** below. The engineering properties are based on previous geotechnical explorations and laboratory testing.

Material	Unit Weight	F SI	Peak Drained near Strength	Peak Undrained Shear Strength	Post- Earthquake Shear Strength
	(pcf)	Cohesion, c' (psf)	Friction Angle, f ' (deg)	S _u /p'	S _{ur} /p'
Embankment Fill	135	0	31	$S_u/p' = 0.60,$ Min. $S_u = 450 \text{ psf}$	Peak Undrained

Table 2. Summary of	Construction Mate	erial Engineering	Properties for	r Embankments
---------------------	-------------------	-------------------	----------------	---------------

The GMF Pond and GMF Recycle Pond contain liner systems. The liner system within the GMF Pond consists of a 60-mil textured high density polyethylene (HDPE) geomembrane underlain by a 3-foot thick layer of compacted clay. A typical cross section profile of the GMF Pond liner system is shown on drawing C-10206 (sh. 9) presented in **Appendix B**. An available summary of the engineering properties of the GMF Pond liner construction materials from Hanson (2008) is presented in **Table 3** below. The liner system within the GMF Recycle Pond consists of a 60-mil textured HDPE geomembrane underlain by smooth-drum rolled native soil. A typical cross section profile of the GMF Recycle Pond liner system is shown on drawing C-10206 (sh. 20) presented in **Appendix B**.

Material	Unit Weight	Effective (dr. Strength P	ained) Shear Parameters	Total (undrained) Shear Strength Parameters	
	(pcf)	c' (psf)	Φ' (°)	c (psf)	Φ (°)
Clay Liner	121.2	0	28.3	1950	0

The method of site preparation and construction of Ash Pond No. 1 and Ash Pond No. 2 is not reasonably and readily available. Site preparation and construction of the GMF Pond and GMF Recycle Pond were completed in accordance with the applicable construction specification (see § 257.73(c)(1)(xi) below).

The approximate dates of construction of each successive stage of construction of Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond are provided in **Table 4** below.

Table 4. Approximate dates of	construction of ea	ch successive stage of	construction.

Date	Event
1964	Construction of Ash Pond No. 1
1971	Construction of Ash Pond No. 2
1978-1979	Installation of internal embankment and new recycle intake structure in Ash Pond No. 1 and abandonment of existing outfall structure
1984-1985	Closure of Ash Pond No. 2 by installing a clay cover
2000	Installation of a sheet pile wall to facilitate construction of drainage flume along the northeast corner of the Ash Pond No. 1
2009	Installation of well dewatering system in Ash Pond No. 2
2008-2010	Construction of the GMF Pond and the GMF Recycle Pond

§ 257.73(c)(1)(vii): At a scale that details engineering structures and appurtenances relevant to the design, construction, operation, and maintenance of the CCR unit, detailed dimensional drawings of the CCR unit, including a plan view and cross sections of the length and width of the CCR unit, showing all zones, foundation improvements, drainage provisions, spillways, diversion ditches, outlets, instrument locations, and slope protection, in addition to the normal operating pool surface elevation and the maximum pool surface elevation following peak discharge from the inflow design flood, the expected maximum depth of CCR within the CCR surface impoundment, and any identifiable natural or manmade features that could adversely affect operation of the CCR unit due to malfunction or mis-operation.

Drawings that contain items pertaining to the requested information for Ash Pond No. 1, Ash Pond No. 2, the GMF Pond, and the GMF Recycle Pond are listed in **Table 5** below. Items marked as "Not Available" are items not found during a review of the reasonably and readily available record documentation.

	Ash Pond No. 1	Ash Pond No. 2	GMF Pond	GMF Recycle Pond
Dimensional plan view (all zones)	B-35, S-44, S-45	B-560, A1000 (sh. 1)	C-10206 (sh. 4, 9, 10)	C-10206 (sh. 4, 19)
Dimensional cross sections	B-35, S-47 to S-50	B-561	C-10206 (sh. 9)	C-10206 (sh. 20)
Foundation Improvements	Not Applicable	Not Applicable	C-10206 (sh. 10)	C-10206 (sh. 20)
Drainage Provisions	Not Applicable	A1000 (sh. 4)	C-10206 (sh. 15, 16, 20)	C-10206 (sh. 21)
Spillways and Outlets	S-8, S-49	W1008 (sh. 2)	C-10206 (sh. 20)	C-10206 (sh. 22)
Diversion Ditches	Not Applicable	A1000 (sh. 1)	Not Applicable	Not Applicable
Instrument Locations	Plate 2, Figure 2A	Figure 2B	Not Applicable	C-10206 (sh. 19)
Slope Protection	S-49	B-561	C-10206 (sh. 9)	C-10206 (sh. 20)
Normal Operating Pool Elevation	S-8, S-49	Not Applicable	C-10201-25	Not Available
Maximum Pool Elevation	S-8	Not Applicable	C-10201-25	Not Available
Approximate Maximum Depth of CCR in 2016	15 feet	28 feet	16 feet	12 feet

Table 5. List of drawings containing items pertaining to the information requested in § 257.73(c)(1)(vii).

All drawings referenced in Table 5 above can be found in Appendix B and Appendix C.

Based on the review of the drawings listed above, no natural or manmade features that could adversely affect operation of these CCR units due to malfunction or mis-operation were identified.

§ 257.73(c)(1)(viii): A description of the type, purpose, and location of existing instrumentation.

Existing instrumentation at Ash Pond No. 1 and Ash Pond No. 2 include vibrating-wire and open-standpipe piezometers. The purpose of the piezometers is to measure the phreatic surface within and around the impoundments. Two (2) open-standpipe piezometers (AP-P1 and AP-P2) were installed at Ash Pond No. 2 in 2009 and the locations are presented on Figure 2A in **Appendix C**. Two (2) open-standpipe piezometers (B-2 and B-4) were installed at Ash Pond No. 1 in 2010 and the locations are presented on Plate 2 in **Appendix C**. Twelve (12) open-standpipe and vibrating-wire piezometers were installed at Ash Pond No. 1 and Ash Pond No. 2 in 2015 and the locations are presented on Figure 2A in **Appendix C**.

The GMF Pond does not contain existing instrumentation. Existing instrumentation at the GMF Recycle Pond consists of one (1) ultrasonic level transmitter. The purpose of the ultrasonic level transmitter is to measure the water level within the GMF Recycle Pond. The location of the ultrasonic level transmitter is shown on drawing C-10206 (sh. 19) presented in **Appendix B**.

§ 257.73(c)(1)(ix): Area-capacity curves for the CCR unit.

Area-capacity curves for Ash Pond No. 2 and the GMF Recycle Pond are not reasonably and readily available. The area-capacity curves for Ash Pond No. 1 and the GMF Pond are presented in **Figures 1** and **2**, respectively, below. "Area-capacity curves", as defined by 40 CFR § 257.53, "means graphic curves which readily show the reservoir water surface area, in acres, at different elevations from the bottom of the reservoir to the maximum water surface, and the capacity or volume, in acre-feet, of the water contained in the reservoir at various elevations."



Figure 1. Area-capacity curve for Ash Pond No. 1



Figure 2. Area-capacity curve for GMF Pond

The area-capacity curves shown were taken from the pond modeling analysis. Actual pond capacity is limited to the approximate berm elevation listed in **Table 6** below. Any information above berm elevation should be disregarded.

§ 257.73(c)(1)(x): A description of each spillway and diversion design features and capacities and calculations used in their determination.

Ash Pond No. 1 contains a concrete intake structure that drains into a 48-inch diameter (dia.) steel pipe. The steel pipe leads to the recycle pump house. In 2016, the discharge capacity of Ash Pond No. 1 was evaluated using HydroCAD 10 software modeling a 1,000-year, 24-hour rainfall event. The results of the HydroCAD 10 analysis are presented below in **Table 6**.

Ash Pond No. 2 was closed in 1984-1985 by installing a clay cover. Non-contact stormwater is collected in ditches along the clay cover and drain off the pond cover via concrete-lined ditch outlets. CCR-contact stormwater collected within the pond is pumped into the GMF Pond via the well dewatering system at the discretion of plant personnel. The capacity of the diversion ditches and well pumps during a model rainfall event has not been evaluated.

The GMF Pond contains a 14-inch high-density polyethylene (HDPE) pipe culvert for normal flow and a weir-like spillway for high water flow. The GMF Pond also contains a 10-inch dia. HDPE siphon pipe used for dewatering. In 2016, the discharge capacity of the GMF Pond was evaluated using HydroCAD 10 software modeling a 1,000-year, 24-hour rainfall event. The results of the HydroCAD 10 analysis are presented below in **Table 6**.

The GMF Recycle Pond contains a decant structure that drains into two (2) 18-inch dia. HDPE pipes that lead to a pump house. The capacity of the decant structure during a model rainfall event has not been evaluated.

	Ash Pond No. 1	GMF Pond
Approximate Minimum Berm Elevation ¹ (ft)	635.0	631.0
Approximate Emergency Spillway Elevation ¹ (ft)	Not Applicable	624.0
Starting Pool Elevation ¹ (ft)	631.0	621.2
Peak Elevation ¹ (ft)	632.0	623.8
Time to Peak (hr)	24.4	24.1
Surface Area (ac)	20.4	33.4
Storage ² (ac-ft)	19.5	88.3

Table 6. Results of HydroCAD 10 analyses

Note:

1. Elevations are based on NAVD88 datum

2. Storage given is from Starting Pool Elevation to Peak Elevation.

§ 257.73(c)(1)(xi): The construction specifications and provisions for surveillance, maintenance, and repair of the CCR unit.

The construction specifications for Ash Pond No. 1 and Ash Pond No. 2 are not reasonably and readily available. The construction specifications for the GMF Pond and the GMF Recycle Pond are located in *Project Specifications, Gypsum Stack and Recycle Pond Construction* presented in **Appendix D**.

The provisions for surveillance, maintenance, and repair of Ash Pond No. 1 are located in *Operation & Maintenance Manual for #1 Ash Pond* presented in **Appendix E**. The provisions for surveillance, maintenance, and repair of Ash Pond No. 2 are not reasonably and readily available. The provisions for surveillance, maintenance, and repair of the GMF Pond and the GMF Recycle Pond are located in *Operation and Maintenance Manual, Gypsum Management Facility* presented in **Appendix F**.

The operations and maintenance plans for the CCR units identified in this report are currently being revised by Illinois Power Generating Company.

§ 257.73(c)(1)(xii): Any record or knowledge of structural instability of the CCR unit.

In March, 2009, shallow sloughing was observed along the eastern embankment of Ash Pond No. 2. The sloughing was inspected by Hanson Professional Services Inc. A dewatering



system was installed in Ash Pond No. 2 to lower the phreatic surface within the pond. In December, 2015, additional sloughing was observed on the embankment of Ash Pond No. 2 and on the embankment of Ash Pond No. 1. The sloughing was believed to be caused by recent heavy rains and was repaired. Photos of the 2015 sloughing repair are presented in **Appendix G**.

There is no record or knowledge of structural instability at the GMF Pond and the GMF Recycle Pond at Coffeen Power Station.

LIMITATIONS

The signature of AECOM's authorized representative on this document represents that to the best of AECOM's knowledge, information and belief in the exercise of its professional judgment, it is AECOM's professional opinion that the aforementioned information is accurate as of the date of such signature. Any recommendation, opinion or decisions by AECOM are made on the basis of AECOM's experience, qualifications and professional judgment and are not to be construed as warranties or guaranties. In addition, opinions relating to environmental, geologic, and geotechnical conditions or other estimates are based on available data and that actual conditions may vary from those encountered at the times and locations where data are obtained, despite the use of due care.

Sincerely,

Jaudia

Claudia Prado Program Manager

Victor Modeer, P.E., D.GE Senior Project Manager



REFERENCES

Hanson Professional Services Inc. (2008), Support Document for IDNR/OWR Permit Application, Coffeen Power Generating Station Gypsum Management Facility Montgomery County, Illinois

United States Environmental Protection Agency (USEPA). (2015). *Hazardous and Solid Waste Management System; Disposal of Coal Combustion Residuals From Electric Utilities; Final Rule.* 40 CFR Parts 257 and 261, 80 Fed. Reg. 21302, 21380 April 17, 2015.

United States Geological Survey (USGS). (2016). The National Map Viewer. http://viewer.nationalmap.gov/viewer/. USGS data first accessed in March of 2016.

APPENDICES

Appendix A: History of Construction Vicinity Map

Appendix B: Coffeen Power Station Drawings

Appendix C: Coffeen Power Station Boring and Piezometer Locations

Appendix D: Project Specifications, Gypsum Stack and Recycle Pond Construction (Hanson 2008)

Appendix E: Operation & Maintenance Manual for #1 Ash Pond

Appendix F: Operation and Maintenance Manual, Gypsum Management Facility Operation (2015)

Appendix G: Photos of 2015 Sloughing Repair



Appendix A: History of Construction Vicinity Map

Coffeen Power Station – History of Construction §257.73(c)





Appendix B: Coffeen Power Station Drawings

- 1. "Earthwork & Grading Plan", Drawing No. B-35, Revision S, 8 September, 1995, Sargent & Lundy Engineers.
- 2. "Concrete Recycle Pump House Intake Structure", Drawing No. S-8, Revision 6, 23 February, 1996, Stearns-Roger Incorporated.
- 3. "Civil Layout & Grading Plan Sheet 4", Drawing No. S-44, Revision 6, 23 February, 1996, Stearns-Roger Incorporated.
- 4. "Civil Layout & Grading Plan Sheet 5", Drawing No. S-45, Revision 9, 23 February, 1996, Stearns-Roger Incorporated.
- 5. "Civil Miscellaneous Sections and Details, Sheet 2", Drawing No. S-47, Revision 2, 23 February, 1996, Stearns-Roger Incorporated.
- 6. "Civil Ash Pond No 1 Sections and Details", Drawing No. S-48, Revision 2, 23 February, 1996, Stearns-Roger Incorporated.
- 7. "Civil Miscellaneous Sections and Details, Sheet 4", Drawing No. S-49, Revision 4, 23 February, 1996, Stearns-Roger Incorporated.
- 8. "Civil Miscellaneous Sections and Details", Drawing No. S-50, Revision 4, 23 February, 1996, Stearns-Roger Incorporated.
- 9. "Ash Storage Area, Plan", Drawing No. B-560, Revision A, 9 February, 1971, Sargent & Lundy Engineers.
- 10. "Ash Storage Area, Sections & Details", Drawing No. B-561, Revision A, 9 February, 1971, Sargent & Lundy Engineers.
- 11. "Overall Site Plan, Dewatering System, Ash Pond #2", Drawing No. A1000 (sh. 1), Revision A, 12 October, 2009, Ameren Energy Resources Generating.
- 12. "Site Details, Dewatering System, Ash Pond #2", Drawing No. A1000 (sh. 4), Revision A, 12 October, 2009, Ameren Energy Resources Generating.
- 13. "Proposed Site Plan, CCB Management Facility", Drawing No. C-10206 (sh. 4), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 14. "Groundwater Monitoring & Boring Plan, CCB Management Facility", Drawing No. C-10206 (sh. 5), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 15. "Anchor Trench and Liner System, CCB Management Facility", Drawing No. C-10206 (sh. 9), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 16. "Cell G1-Foundation Grade & Control Data, CCB Management Facility", Drawing No. C-10206 (sh. 10), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 17. "Cell G1-Process Water Recovery System, CCB Management Facility", Drawing No. C-10206 (sh. 15), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 18. "Cell G1-PWRS Drain Details, CCB Management Facility", Drawing No. C-10206 (sh. 16), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 19. "Recycle Pond Plan & Control Data, CCB Management Facility", Drawing No. C-10206 (sh. 19), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 20. "Recycle Pond Process Water Transfer Channel Details, CCB Management Facility", Drawing No. C-10206 (sh. 20), Revision 0, 5 January, 2011, Ameren Energy Generating.



Appendix B: Coffeen Power Station Drawings (continued)

- 21. "Recycle Pond–Process Water Decant Sections & Details, CCB Management Facility", Drawing No. C-10206 (sh. 21), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 22. "Recycle Pond Emergency Spillway Sections & Details, CCB Management Facility", Drawing No. C-10206 (sh. 22), Revision 0, 5 January, 2011, Ameren Energy Generating.
- 23. "Ash Pond #2, Drainage Modifications", Drawing No. W1008 (sh. 2), Ameren Energy Generating.






Sec. 1

•									
- 40	REVEKAR		en 'ser		NEVERONCE ORAWINGS	PRINT RECORD	ENG RE	CORD C	MAWIN
2^{n}	GENERAL REMOVE	<u> </u>				1 11 11 11 11 11 11 11 11 11 11 11 11 1	T	100 7040	HUNCO
12	ANALY CO AND CONSTRUCTOR	4/10 187	γ			12 1 7 1 d	ار میں ا	m gap me	makke
¥-	11. Store 10. 11		"——	• •				: 기별 등	A11
12	SOD COUNT 4 AND DEFAN		η			<u> cumpera 4 4 5 1 5 1 5 1 5 1 5 1 5 1 5 1 5 1 5 1</u>	Langer e. 1	<u>3</u>	I.CIDA
- HR	THE REAL POST OF A CONTRACT OF A CONTRACTACT OF A CONTRACTACTACT OF A CO	- CICL - MARY 1 - 200			· · ··· <u> </u>		_ms.s*	Carl Internet	10 3 APP
12	7 (mb # x		· +- ·			4044 06 2012 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	A	100 - 40 16 - 8	Sen 1995
۲h	······································	····÷	and the	→ —			+		ant for (and



_												
-	ACHINA		-	-	841		REFERENCE DRAWINGE	[PRINT RECORD		nécono	ORTANDO I
1÷	2.5MD P 3.	. 530					·	- WEL BIN	10 Mr. 7 Mai 14 1/12 Mai 100, 1/1	A	NYS (MR	• ***
1.	``			ىر -	·		1	SCOULD- N	• <u>•</u> ••••••••••••••••••••••••••••••••••	CHICKLE	153 4470	PHL.HLAT7
1	ACOUNT DURCTION AND 2DV CRD COTE 4. 244	100	· · · ·	10	┍╌┥	14.5.	- ADAMA S AT A SEATIONS	· + rae	- B D F D 2 C D B D D D D D D D D D D D D D D D D D	14424 100	A: 113	Contraction from
12	CALLER CODER A DECKER AND AND SERVICE CALLER	A		<i>.</i>	· ł	01-2-0	CONTRACTOR STORE	Sel-year	·····			CONTINUES ON 1/1
13	1.2780 THERE AND A CALL AND THE A CALL AND A THE			ب	· — I	5 is "	NUCTAR ZATON PACOTES			19482		N-415 6 101-807
17	~~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~	34.	22.1	云	• • • •			000 200	2 3 602 St 3 4 1 4 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1		· <u> </u>	ALE.
1ð	AUCENS & POLK & 104 + 245, 53 TO HE LINE & ATERS	200	170	-	7.4	-	· · · · · · · · · · · · · · · · · · ·	1				
_												













								ŀ																																												ŀ																											
		ſ									ĺ							ŀ		$\left \right $					ĺ	ŀ			k				ŀ																	Í	ĺ									ĺ	ĺ											Í							
																1			1					Ī		ľ		ł	ĥ			•		ŀ	ŀ	ŀ			1		5	Ì							•			ļ		ļ		,			ļ	ļ	ļ		Ņ											ŀ					
																																																	1		ł	ł	ŀ	·		·	ł	╏	ŀ	ł	ŀ	ŀ	ŀ	ľ	:									ŀ	1				
1		ŀ																				ł		ľ	I	ľ		-	-				ŀ	Ī		ĺ		ŀ		2									İ		ŀ	ŀ	-																						ŀ	ľ	İ		
									l																			ŀ					ļ	ŀ			t	ļ												┞	ł	-		à		-																							
							•	1		I					Ì		ľ	ľ	·														I.						1																																				ļ				
		ŀ																															ŀ	ľ								1			i	'	ľ			l																									ŀ				
																										ļ							I,																		ļ				ļ																								
																										ľ	ĺ					k	ľ																	1																													
																										ŀ							ŀ										$\frac{1}{1}$							ĺ	ŀ	ŀ	ŀ			ĺ			1		ŀ																		
												E		:	ł			ŀ	Ņ	ì											ŀ										•	1			ſ											ļ					l			ļ											ľ	ľ			
		ļ							l		ł	·	ľ		ľ	ŀ	ļ	ļ	1	1		ļ														•			:		•						Ì		i		t		ľ			ł	ſ	ŀ		ŀ	ŀ					1				·									
Ī					ĺ	ŀ		1									ľ	ľ					ļ	ļ	ļ		ľ				ľ		ľ												ĺ				4		ľ	ŀ	$\left \right $					ŀ																	ĺ				
	ľ	ſ	ľ	ľ											ł			ľ						ľ	ſ		ŀ		ŀ	ŀ					ŀ																				f																								
																									ŀ		ł	ľ	ĺ		ť	ł		ľ	ľ	ľ	1	İ	ľ	ĺ	P																																						
																									ŀ	I	ŀ			ŀ	ł		ŀ					ŀ												ļ																									ĺ				
																								ŀ		ļ		╞							ŀ											1	1	1		ĺ	ŀ	ŀ					ĺ	ł			ŀ																		
																								ŀ			ſ	ŀ		6	Ē		ſ	Í	ŀ				ļ																																			ŀ	I	ſ	l		
	I					4			ļ																		-		f		ŀ	\mathbf{F}	ŀ	ŀ	ŀ	f	$\left \right $																							ŀ	ŀ	ł																	
									I	1														ŀ	ĺ	ŀ	ŀ		k		ļ																													ſ														ŀ					
+	F	ŧ	t	ľ	F	ľ		ļ																		ĺ	ł	ŀ			ŀ	ŀ				ļ		ł		ŀ																																							
ļ				ŀ		ľ																	ŀ	ł	ŀ				f				Í	ŀ		ľ	ľ									•		ł	ĺ	l			ŀ								ŀ																		
					ļ	ļ			J																																																		ŀ	ļ	ľ																		
f	Ī			t	f	f				t	ĺ												ŀ	t	t	╞	t	ŀ	ŀ	ŀ		ŀ	╞	ŀ.	ł	H	ŀ	t		ŀ	ŀ																				ľ																		
																						ŀ			ł	ŀ		ŧ	ŀ				ſ	f	ŀ		ŀ	P																						ļ							Į									ſ	t		
	ļ	ļ			Ļ		ļ								ł					J			ļ	l		l			ļ				ŀ				ļ														ļ		ŀ			J				ŀ	ŀ	ŀ				4	1	1											
					ŀ	ŀ	ľ				ŀ						ŀ	ŀ	ŀ		;	ł		ſ		ŀ	ŀ		ŀ			ŀ	ŀ	ł	ŀ	ŀ	Í									•	ŀ	l		ł									ł	1																			
ł	ļ	ł		ļ	┞	┞					╞	ŀ	f		ļ	ĺ	ſ	ſ				t				ŀ			ľ							ŀ	ŀ						$\left \right $				-						ŀ			•					ļ	ŀ					I								l				
																			ŀ					ŀ	ł	I.	ŀ		ŀ					ŀ	ł		ŀ	ŀ		$\left \right $						•		ļ																											ļ				
ł		ł								I	ļ	ŀ				ļ	ļ		ł							ŀ				ŀ																			1	ł	ľ	ł	ŀ				1	ĺ	ł			ŀ					1												
h			ŀ	ľ	ŀ																																												1	ł	ļ										ŀ	ŀ																	
	1					2							ŀ				ł	ŀ																									I				2				ŀ			ŀ			1		ŀ	ľ	ľ		•	ł	1						ĺ				ľ				
																		ŀ																																																									ŀ				
																																																																											١				
								·																																																																							
f		f										F												ſ			ļ	ļ	þ		Í	þ	k	ŀ	ŀ			ļ						ļ						ļ	ĺ		ļ	ŀ	1					ĺ		ĺ		h		I	I	ĺ				ſ	ſ						
					f		ŀ		J		ł	f				f	f	f						ļ	ľ	P	ŀ	ļ		ľ			ŀ	ŀ	ŀ								ļ					ļ	ļ	ł	Þ	ŀ	ŀ				1		ļ		ļ					ļ			ļ										
					F	f					F	F			Ŧ		f	F	F							l	╞				h				ļ	•																							ļ				ŀ]	╏				ŀ			
					╞						╞	ŀ					╞	╞						ļ	ľ	ľ								l				F					J								ĺ							ļ	ļ	I	ĺ	ſ	f			ļ					Ĵ				1				
				F	F	F					F	F					F	F	F					ļ							ĺ		ľ	ľ	ŀ	ľ		ł					ł							$\frac{1}{1}$	l					6				ļ			ľ	ľ						Ĵ]		J						
				F		F			1	1	ŀ	F		•	ļ		ļ	F				-		ŧ	İ	ļ		F		f	f	Ē	ł	f	ŧ		Ē	t					ļ		÷					ĺ	ĺ										ľ	ľ		1			1	1		1	I	ĺ	ĺ	1	1				
	- 4	- M.	11	1	e i			1	1	1	1	¢ l	1	1	1	1	£.	£.	e l	r İ	í Í	1	1	1	ſ	É	É	ſ	r i		ſ	ſÎ	É	ť	¢1	e 1	21	. 1	- 1	. 1	- 1	1	- 61	- 6	- 1	- 6		- 6	- 6	~		1 11	F. 1	Γ.			. ~					21		. 1		1.5		. *	- 6		- 64		14		-		н.	4	,



NOTES:

1. THE LOCATION, SIZE AND/OR TYPE OF MATERIAL OF EXISTING UNDERGROUND OR OVERHEAD UTILITIES AS MAY BE INDICATED ON THESE CONSTRUCTION PLANS IS NOT REPRESENTED AS BEING ACCURATE. SUFFICIENT OR COMPLETE. THE OWNER AND THE PROJECT ENGINEER HAVE NOT INDEPENDENTLY VERIFIED THIS INFORMATION AND DO NOT ASSUME ANY RESPONSIBILITY WHATSDEVER IN RESPECT TO THE ACCURACY. SUFFICIENCY OR COMPLETENESS OF THE INFORMATION AND GIVE NO EXPRESSED OR IMPLIED GUARANTEE THAT ANY CONDITIONS INDICATED ARE REPRESENTATIVE OF ACTUAL CONDITIONS TO BE ENCOUNTERED.	_
2. IT SHALL BE THE CONTRACTOR'S RESPONSIBILITY TO DETERMINE THE ACTUAL LOCATION OF ALL SUCH FACILITIES, INCLUDING SERVICE CONNECTIONS TO UNDERGROUND UTLITIES, PRIOR TO CONSTRUCTION. THE CONTRACTOR SHALL NOTIFY ALL UTLITY COMPANIES AND AGENCIES OF HIS CONSTRUCTION PLANS AND SHALL OBTAIN FROM EACH PARTY DETAILED INFORMATION AND ASSISTANCE RELATIVE TO THE LOCATION OF ALL UTLITIES AND THE SCHEDULE OF ANY REMOVALS AND ADJUSTMENTS REQUIRED OF THE UTILITY. THE CONTRACTOR SHALL CONTACT J.U.L.I.E. (1-800-692-0123 OR 811) TO ASSIST IN COMPLETING THIS RESPONSIBILITY.	E
3. THE CONTRACTOR SHALL PROTECT ANY FACILITIES TO THE SATISFACTION OF THE UTILITY OR OWNING-AGENCY WITH THE COST OF ANY REQUIRED PROTECTION TO BE INCIDENTAL TO THE CONTRACT. IN THE EVENT A UTILITY LINE OR SERVICE IS UNEXPECTEDLY ENCOUNTERED DURING CONSTRUCTION. THE CONTRACTOR SHALL IMMEDIATELY NOTIFY THE OWNER AND THE UTILITY COMPANY OR AGENCY OF JURISDICTION. ANY SUCH UTILITIES DISTURBED BY THE CONTRACTOR'S OPERATIONS SHALL BE RESTORED TO SERVICE AT ONCE.	
-	- D
-	С
_	
-	в
LEGEND ELECTRICAL SPLICE BOX ELECPHONE SPLICE BOX GUAROPOST EXISTING FENCE TREE AREA GRAVEL EDGE PAVED OUTLET EDGE	
FM ── FORCE MAIN SS ── SANITARY SEWER UE ── UNDERGROUND ELECTRIC GRAVEL SURFACING PIEZOMETER IRON PIN NOTICE OF LUNITED RESPONSIBILITY 100′, 50′, 0 50′ 100′	А
SCALE IN FEET	
CLASS Amergen Energy A1000 0	





2	0	1	
END			
DN EACH FLIN).			
touris			
			-
TOWERS			
CTWOD			
iii.			
ULT # 7109			- E
f → MORIE, & VUMT,			
ADSI SECTION LINE			
TOUFS			
NETTICES MONTOFICS WELL			
MENTTORINI HELL			
ANT OF			
			- 0
ING THE RESPONSIBILITY			
OWN FOR DEPOSIDETIDAL			
THE ARE BALLED ON AND BE, 15 JUNE EM WEST JOHE, MAYD BE, 45 JUNE OMING, FUIDITS REAL ESTAD, ISAGD JUNE	T FEET		-

- UK	1900101	1199				
-41	REPAR	38.8	0111	CONCILCU.	404,4651	
28	RESIA	28.8	2184	CONCRETE.	RENOVER	
in	非复数认识	18.9	2144	CONCRETE	NONUMERT	
40	REPT	15.4	214.	CONCRETE.	MENUMENT	









	A 41 Taxa	-	70.		-			1000	14.2	17.0			
1.1	10 M	arcar.	290	Said-	1.144.0	11.1.8	122	1000	2444	100	141.00	August -	
	06.1	10.0	121	1 10 1 10	1.100	100	197.0	100	2010	10.215	10.0 IA	44.1	
	-	11 m	1.0.0	Part O	1100	100	100.00	100	2010	10.20	200.00	40. W	
25	5/2	10.0	22	00.0	1123	83	10.0	19.8	2116	005-	0.6	100-	
631	18	00	28	8884	EE 8	10.1	16.3	188	216	216-1	10.8	10.0	L-
621	63	0.1	1991	anger 1	896.8	82	1C.B	198-11	318	898-	11.8	10.8	
221	8.0	18.8	2-96	22264	2153	283	10.0	198	208	235	80	88.8	
384	11.0	Rett	192-1	99934	9982	Eb:	10.0	19800	200	282	10.7	1939-1	
3.5	a	Q.2	199	1111		88.	11.0	199	548 - I	880	10.1	10.0	
	100	6 (m. 140	140		12111.1	10.01	12.5	tion	2000		10.0	10.	
	24	22	144		1111	100.0	44.7	100	2228	100	10.0	40.00	
23	10.10 10.00	10.0	123	1000 A	144	101	-123	1000	1200	201-	23-	20.1	
친구	4.4	12.0	1.00	1000	1000		103	100	12102	200	10.1	40.0	
221			100	2007	1000	100	188	100-	Column 1	2.20	10.4	1828	
000	5	88	100	0000			1000	100	12946	3165	10.0	10.9	10
0.05	88	18.9	284	STOCK	State St.	10.1	18.8	108-	14.944	2048	H.S	10.0	
884	41	抗保一	581	SEL	Sector.	2.2	100	500	2016	938-	20, 21	10.1	
2.2.1		10.2	521	259	1814	10.00	-12.3	5205	222	(12)	RA-	10.1	
23-1	21	122-	22	120.1	111.2	2	10.7	- 625	322	305-	0.7	22.2-	
229	20-	120 m	292	11537	diam'r	201-	12.0	22	2008	208-	30.Y.	A0.1	
68.1	0.8	10.0	280	ein v	1211	56-jr	168	5m-	2010	2000	10.5	11.8	
	10	2018	146	1. m + + +	1111-1	10.5	16.6	200-	Street .	Series.	882-	10.0	
	-	10.0	200		2.413	10	10.0	100	Search (C	8.88	22.0	10.9	
	B.t	12.2	1-1-1			****	10.5	100	Second 1	10.000 A	10.0	40.0	1
221	12	200	- 192	1000		111.01	12.3	100	14450	10-10-10-10-10-10-10-10-10-10-10-10-10-1	#14,00	40.00	Γ.
		104	140		1111-1	100 m	-125-	104	1222-1-	1.2	10.0	10.0	
	10.1	No. or	240	1000 A	1111	1-1-1-1 1-1-1	10.0	121	20100	10.2014	10.0	40.1	
101	10.1	10.0		Solution in the	10-10-0 W	10.1	10.0		2226	1.211	10.0	400.1 100.0	
221	53	18.N	240	000.0-1	1000	10.1	10.0	200-	2223-	2.44	0.0	40.0	
2.0	1.0	10.0	1.000	100.00	1100	10.6	10.1	100.0	14141.4	ACCESSION OF	10.5	AGA - 0	
		14.0	100	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	1 44 4	10.0	100	12	100	Contract of	100.44 101.01	44 M	
381	1	10.0	100	The state of the	-	-	100.00	198-	208.0	213-	10. D	10.0	
100	14.4	18.6	2014	NYAR L	COAR.	10.1	10.0	198	1000	2.2.8	0.6	183.1	
281	G.F.	183	100	Circle 1	TRS	No.	10.2	50	Stat.	1000	14.8	83	- 0
-			Sec	State of		20	180	500	Series (899	338	16.8	1.1
		80	(22) ····	and the second		-	1004	100	THOMAS	610-	10.1	230-	
263	21	80C2			1111	223-	10.2	22-	-512	1000	10. P.	HO.F	
	605	10.00	12	Secol 1	114111	1	10	1000	1000	1000.0	201	10.1	
631		10.9	200	120	1.41		122	108	381	200	17.1	20.0	
	1.1	10.0	100-1	122	120		162	100	28	111.2	14.4		
661	24	10.0	18-1	1998	146		183	198	100	2004.01	and 10	認知	
831	83	10.0	100	940	1000		188	18.	10007	1014	10.0	10.0	
38.	2	11.14	2474	100	9944	10.0	199	19	2010	845.8	14.8	ROCC.	
3.5	10.3	TAX	291	330	141	Sici-	10.8	1924	2F176	2124.8	10.0	10.1	
821	NGA	10.9	241	330	144	585	16.1	100-	2003.7	21/296	394.4 ····	10. U	
	1.1	22		240	100	22	10.0	- 22	232.4	Call?	20.4 	10.0	
839	S.B	12.8	200	C10	12	22	10.0	120	205	10.6	89-	10.1 10.1	
100		108.02	201	210	2116-	201	10.0	100	2251	2161	PCR	BUL	
200	(%)	10.0	100	100-1	12780	10.1	15.0	201	2850C	1917	26.3	10.1	
	64	10.8	59		815 -	10.	108	208-	10,700	1001.0	18.8	10.0	
	2.2	10.0	259	1111	220	10.10	100	198-	1000	828-	第4二	193-1	
-			192	100	1.11	33	188	18	200	1000	10.5	100	
	101 T	1.0.0	125	222	111	100.0	10	100	200.4	1000	12.5	10.0	1-0
	10.4	AND IN COLUMN	107	2444	1772	100.00	12.1	12	10000	1.12.3	10.2	23	
	10. JP.	10.0	122	004	1111	100 G	10.1	104	SPUILA	10-21 N	10.1	40. 0 40. V	
		10.00	155	1916	2110	10.0		12-	101245 CL	1.44	855-	10.9	
1.54	21-	23-	122	410.84	10.00	22.0	ALC: N	100	ATTAC N.	1000	220-	45.8	
131	11.5	10. S	248	160	1.44	8.8.	49.3	285	A 100 R	11-49.4	10.4	45.5	
6.6	0C	26.7	88	298	174	85	198	199-1	Service.	814.8	124	8.9	
663	8-1	16.9	98	946	919	23	12.9	198	Store 1	254.2	88	23-1	
151	10 m	122	9R	Page 1	2118	2.2	12.6	582	Second P.	249.2	S	22.0	
	22-	20.2	544	194	2111	22.2	- 19.6-	1424	Same of	A 44 4	10.0	22	
	11.2	10.0	1000	10.00		And in	10.0	100	10000 (c)	2019-01-F	PTL P	10.0	-
100	(N. 1	19.10		Conter - 1	1.11	14.1	10.1	100	222.1-	244.2	<u>61</u>	10.00	
12	20-	10.5		100 m	1.72	22	12%	100	COMPACT OF COMPANY	(199 X	64-	PO. 8	
530	6.6	10.5	200	Xere -	100	22	100	1981	20100.00	222.2	10.2	Mill I	
121	0.8	12.0	198	100	111		12.4	128	100	1000	12.3	100.01	
	-	-	292	2100-1	199		183	198-0	1993		10.0	183	
221	5	the state	268	Vite 1	12	-	100	198		- 140 8		10.0	
	20	-	100	440	9:14	19.8	14.0	100	10409-0	19441	10.0	1998-1	
1	100	1.4	2681	PEN I	100	1.00	is a	1.4	Contra la		122		
2.4		-	104	260	8954-	10.0	160-	104	Carbon M	111012.0	148.1.		1.5
2.23	21	10.0	<u>20</u>	2500	2122-	221-	- 21-	122.00	3121-	21-15a.7	107.5	#0.4	
2.5-1	21-	10. A	S2	222	2112-	29	- 23	24	Served in	storage of	21-		
2.24	16-	13	52	122	115	20	120	122	Contractor of	Concession in	100.00		
21	×	15.3	120	100	1/12-	10.1	- 19.1	100	Contract of the	120	10.0	100	
630	05.1	10.	292	100	1.41	10.1	15.0	140	200000		10. H		
17-7-1	0.5	10.00	200	200	1.11	10.1	10.0	100	1000		12.2		
	A.8	10.0	100			-	184	22 ···	100.0				
-21		arb. 16	192	100	100	-	101	144			-		
	÷		134	100	222	11	122	24	122.1		15		1
	24	10.4	102	100	1041	10.0	122	123	Taxas -	-	11		-
10.07	10.0	12.0	100	of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local division of the local divisiono	10.00	10.00	12	122	Silling 2	1.16	-	-	
111	111	10.10	12	of Sector	100		12.1	Can .	Cartan Pr	Carles C.	12.0		
0.01		10.0	225	of lates	212	10.1	14.1	-	101-00 U	-	22.2		
30		10.0	200	C'sta	1140	88	And all	58	-	phylase a	10.0		
5.85	18	88	38	Ches.	6188	10.5	1004	580		per l'ande de	-		
		14.7			1.199	10.1	Theorem .	and the second	and a	and the second	- mail		
													1
E In	THE	190	AN ST	(FETS	SER	8.1	EGEN						
	- Coloris	1.1.1.1.1.1.1	1		11. 10. 1	and the second second	NAME OF COLUMN	31	£				1.4
			1	1.500.0	26.27	100	Concept 1 1 and	3					1.1
				4 Start	11-11	She.	501-51	1.007	-				1
		_		4,900.45	1993	100	at min	W MITSEL					-
						-	CEL	- <u>9</u>	T GUNE	ATIO	U URA	DE -	
		- 1			-	=1		100	- GALLENS	11, 11, 14	750		1
					-		1.50	W 1423	ACC: NO	NT 64	C16.17	TT:	4
		-11			1	-1	1119) (D	FERN	The state		1.0	4
		-			1.000		S	-	- Contractions		10.0		1
					A 1 1 1 1 1 1		- A ADDRESS OF TAXABLE PARTY.	ALC: NOT THE OWNER.			and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se		

2487.85.10



















Appendix C: Coffeen Power Station Boring and Piezometer Locations









Appendix D: Project Specifications, Gypsum Stack and Recycle Pond Construction (Hanson 2008)

PROJECT SPECIFICATIONS GYPSUM STACK AND RECYCLE POND CONSTRUCTION GYPSUM MANAGEMENT FACILITY COFFEEN POWER STATION MONTGOMERY COUNTY, ILLINOIS

Prepared For:

AMEREN ENERGY GENERATING COMPANY

Prepared By:

HANSON PROFESSIONAL SERVICES INC. 1525 South Sixth Street Springfield, Illinois 62703

January 2008

GYPSUM STACK AND RECYCLE POND CONSTRUCTION GYPSUM MANAGEMENT FACILITY COFFEEN POWER STATION MONTGOMERY COUNTY, ILLINOIS

TABLE OF CONTENTS

Sheet No.

Division 1 - General Requirements

01356 - Storm Water Pollution Prevention Measures	01356-1 to 01356-3
Division 2 - Site Work	
02100 - Site Preparation	02100-2
02200 - Earthwork	02200-1 to 02200-12
02275 - Riprap	02275-1 to 02275-2
02315 - Granular Drainage Materials	02315-1 to 02315-8
02373 - Geotextiles	02373-1 to 02373-8
02376 - Geosynthetic Clay Liner	02376-1 to 02376-9
Attachment 1 - Bentomat® Certified Properties	End of Section
02640 - HDPE Piping	02640-1 to 02640-4
02800 - HDPE Geomembrane	02800-1 to 02800-12
02936 - Topsoil Seeding and Mulching	02936-1 to 02936-3
03100 - Concrete Formwork	03100-1 to 03100-5
03200 - Concrete Reinforcement	03200-1 to 03200-3
03300 - Cast-In-Place Concrete	03300-1 to 03300-6
03400 - Concrete Embedment Liner	03400-1 to 03400-4

Index-1

DIVISION1-GENERAL REQUIREMENTS Section 01356 – Storm Water Pollution Prevention Measures

PART 1. GENERAL

1.01 DESCRIPTION

A. This section pertains to the construction and maintenance of temporary erosion control systems to control erosion and sediment damage to adjacent properties and water resources, and the removal of erosion control devices when they are no longer required.

1.02 RELATED SECTIONS

The following sections contain items which are related to the work in this section:

02936 - Topsoil, Seeding, and Mulching.

1.03 REFERENCES

The following reference, or cited portions thereof, governs the work:

Illinois Department of Transportation (IDOT): Standard Specifications for Road and Bridge Construction, adopted January 1, 2007.

1.04 SUBMITTALS

- A. Submittals shall follow the provisions of Section 01010.
- B. Preconstruction Submittals: A storm water best management practices (BMP) plan shall be submitted that includes the following items:
 - Site drawing showing anticipated locations of structural erosion controls, areas of disturbed soils, and drainage patterns;
 - 2. Inspection and record-keeping procedures; and
 - Maintenance procedures for erosion controls.

PART 2. PRODUCTS

2.01 EROSION CONTROL SYSTEMS

Materials for erosion control systems shall be in accordance with Article 280.02 of the IDOT Standard Specifications.

PART 3. EXECUTION

3.01 EXAMINATION

The site shall be examined to determine the extent of work required.

3.02 PRECONSTRUCTION JOBSITE INSPECTION

- A. The person who shall be at the jobsite during construction and who shall be responsible for insuring that erosion control work is completed in a timely manner shall be identified at the preconstruction meeting.
- B. A jobsite inspection shall be conducted with the Owner's Representative to review and designate the locations and types of erosion protection to be placed. The inspection shall be scheduled at the preconstruction conference and carried out on the job site before beginning any work that will disturb existing drainage or potentially create erodible conditions.

3.03 CONSTRUCTION

- A. Temporary erosion control systems shall be constructed in accordance with IDOT Standard 280001 and Article 280.04 of the Standard Specifications and as directed by the Owner's Representative. Erosion control devices shall be in place and approved by the Owner's Representative prior to beginning other work.
- B. Incorporate permanent erosion control features into the project at the earliest practicable time to minimize the need for temporary erosion controls.

3.04 MAINTENANCE

- A. Temporary erosion control systems shall be maintained in accordance with Article 280.05 of the Standard Specifications, except that measurement and payment provisions shall not apply.
- B. Temporary erosion control systems for unprotected disturbed areas shall be cleaned of trapped sediment and repaired immediately prior to project close out.
- C. Temporary seeding shall be applied to all disturbed areas except the gypsum stack excavation and the future fill and topsoil stockpiles.

3.05 REMOVAL AND DISPOSAL

When the Owner's Representative deems that temporary erosion control systems are no longer needed, they shall be removed and properly disposed, and silt deposits shall be removed or regarded as directed by the Owner's Representative, and the area seeded. Non-biodegradable temporary erosion control materials shall be disposed of off site. Biodegradable erosion control devices may be disposed of in spoil areas designated by the Owner's Representative. All laws and regulations in disposing of the materials shall be obeyed.

END OF SECTION 01356

1/05jobs/05s3004A\Gypsum Stacking\IDNR Dum Safety Permit Application\Specs\S01356_Storm Water Pollution Prevention Measures.doc

DIVISION 2 - SITEWORK Section 02100 - Site Preparation

PART 1. GENERAL

1.01 DESCRIPTION

This section pertains to stripping of topsoil and vegetation from areas of the site that are to be excavated.

1.02 RELATED SECTIONS

No related sections.

PART 2. PRODUCTS

No products used.

PART 3. EXECUTION

3.01 EXAMINATION

The Contractor shall examine the site to determine the extent of work required.

- 3.02 SITE PREPARATION STRIPPING
 - A. All vegetation and topsoil encountered within the Gypsum Stack grading limits shall be stripped. Topsoil shall be kept clean and free of all foreign material, and stored in separate stockpiles from vegetation and common excavations. Stockpiles shall be located as indicated on the drawings or as directed by the Owner's Representative.
 - B. Payment for stripping shall be based upon removal of 24 inches of topsoil in areas that require stripping.

3.03 DISPOSAL

All materials resulting from site preparation operations shall be stockpiled in the designated spoil area. Contractor shall obey all laws and regulations when disposing of the materials.

END OF SECTION 02100

1:/03jobs/03s5010/Gypsum Stack/Specs/Gypsum Mangement Facility/Pre-Final Specs/S02100_Site Preparation.doc

DIVISION 2 – SITEWORK Section 02200 – Earthwork

PART 1. GENERAL

1.01 DESCRIPTION

This section pertains to excavation, fill, and backfill required for foundation preparation, construction of low-permeability soil layer, anchor trench construction, miscellaneous site grading and berm construction.

1.02 RELATED SECTIONS

- A. The following sections contain items which are related to the work in this section:
 - 1. 01356 Storm Water Pollution Prevention Measures
 - 2. 02100 Site Preparation
 - 3. 02373 Geotextiles
 - 4. 02936 Topsoil, Seeding, and Mulching

1.03 REFERENCES

The following references, or cited portions thereof, govern the work:

- Illinois Department of Transportation (IDOT): Standard Specifications for Road and Bridge Construction, adopted January 1, 2007
- Department of Sustainable Natural Resources, Soil Survey Standard Test Method, Unified Soil Classification System: Field Method (USCS).

1.04 MEASUREMENT AND PAYMENT

- A. The Contractor shall be responsible for estimating the extent of excavation and fill required to complete the work, including, but not limited to, excavation to required elevations; loading, transporting, placing, and compacting low permeability soil; excavation and backfill of anchor trench; and miscellaneous site grading and berm construction. The Contractor shall include the dollar amount associated with all earthwork in his Lump Sum Bid amount.
- B. Removal and replacement of unsuitable foundation material and subgrade stabilization measures directed by the Owner's Representative will be paid for on a time and material basis.

1.06 COORDINATION

Existing utilities or other plant facilities shall not be interrupted, except when permitted in writing by the Owner's Representative and then only after acceptable

temporary services have been provided. A minimum 48-hour notice shall be provided prior to proceeding with an approved temporary interruption.

1.08 SUBMITTALS

A. Materials Handling Plan.

A materials handling plan shall be submitted for construction and protection of the low permeability soil liner. The plan shall describe the following:

- Processing and placement of the low permeability soil type, model number, weight, and critical dimensions of equipment to be used for soil processing, compaction, scarification, and smooth rolling;
- Method of protecting low permeability soil from changes in moisture content and freezing after placement.
- B. Construction Access Ramp Layout.

Layout drawings shall be submitted showing alignment, profile, and typical section of the construction access ramps from the haul road into the bottom of the Gypsum Management Facility excavation. The minimum width of the ramp shall be 50 fL, and the longitudinal grade shall not exceed 8 percent.

PART 2. PRODUCTS

2.01 MATERIALS

A. Earth Fill Material

Earth Fill Material shall consist of a mixture of clay, silt, sand, and gravel-sized particles obtained from previously constructed subsoil stockpiles. These materials can be used separately or mixed as required for best results. When placed, Embankment Material shall have a USCS classification of SM, ML, or CL and shall be uniform. This material shall be free of ice, snow, organic matter, rubbish, and debris. Coarse-grained particles shall be well dispersed to prevent the development of segregated pockets or zones with insufficient fine material to fill the interstices.

B. Soil Liners

The Soil Liner for the Gypsum Management Facility is considered a Clay Liner, and shall be soil classification CL, CL-ML, or CH. The material shall be free of roots, debris, organic or frozen material, and shall have a maximum clod size no greater than the length of the compactor foot for the compaction equipment proposed by the Contractor. When compacted, the material shall have a hydraulic conductivity of less than 1 x 10⁻⁴ cm/sec.
C. Soil Stabilizers and Moisture Conditioning Agents

Additives to accelerate drying or to improve stability and workability of soil shall not be permitted unless approved in writing by the Owner's Representative.

2.02 EQUIPMENT

- A. Compaction Equipment
 - 1. Tamping foot rollers

Compaction equipment shall consist of tamping foot rollers which have a minimum weight of 40,000 pounds. At least one tamping foot shall be provided for each 110 square in. of drum surface. The length of each tamping foot, measured from the outside surface of the drum, shall be at least 1 in. longer than the loose lift thickness.

Steel-Wheeled Rollers

Equipment used to produce a smooth compacted surface shall be a smooth, non-vibratory steel wheeled roller weighing not less than 1,000 lb. per lineal ft.

B. Scarification Equipment

Discs, rotor tillers, or other equipment used to scarify the surface shall be capable of uniformly disturbing the upper 6 in. of surface to provide good bonding between lifts.

C. Mixing and Spreading Equipment

Discs, harrows, and motor graders or other similar equipment shall be available at the site for use in spreading, mixing, and drying Compacted Subsoil Stockpile Material.

PART 3. EXECUTION

- 3.01 PREPARATION
 - A. Control of Work

Benchmarks, monuments, and other reference points shall be maintained throughout the work area.

B. Utility Location

Before starting excavation, the location and extent of underground utilities in the work area shall be established.

3.02 EXCAVATION

A. General

Excavation consists of removal and redistribution of material encountered when establishing required grade and subgrade elevations. The Contractor shall be responsible for dewatering, protection, shoring, and disposal of excavated materials as necessary to complete the excavation.

B. Procedures

Excavation may be accomplished by any method and by use of any equipment that is suitable to the work, except that blasting will not be permitted. Based on previous construction experience at the site, it is recommended that excavation to the foundation grade be completed as far in advance of low permeability soil placement as possible to allow the foundation surface to dry and form a "crust" capable of sustaining compactive effort.

C. Overexcavation

All excavation shall be performed to the lines and grades indicated on the plans. Any overexcavation or excess excavation not requested by the Owner's Representative shall be at the expense of the Contractor.

D. Disposal of Excavated Materials

Contractor shall utilize excavated material as stockpile materials for future use as specified in paragraph E.

- E. Stockpile Requirements
 - Excavated clay and silty clay materials are to be stockpiled in the shortterm subsoil stockpile area.
 - Excess excavated materials are to be stockpile in the areas designated on the drawings.
 - Materials not suitable for use as fill or backfill shall be disposed of onsite in the locations specified by the Owner's Representative.
 - Spread fill material for use by others, topsoil, and low permeability soil are to be stockpiled in layers not to exceed 1 ft loose thickness.
 - Tops of stockpiles are to be graded to ensure positive drainage. Side slopes for stockpiles shall be no steeper than 3H:1V.

 Perimeter ditches are to be excavated to intercept runoff flowing toward stockpile areas and to route it to outlet locations approved by the Owner's Representative.

3.03 SUBGRADE PREPARATION

- A. Areas to receive fill shall be proof rolled under the observation of the Owner's Representative. Soft, loose, weak, or wet materials shall be removed and replaced with compacted fill or stabilized with geotechnical fabric or geogrid as directed by the Owner's Representative. Joints, fractures, and moisture seeps shall be repaired, and local sand deposits, if present at foundation grade, shall be removed and backfilled with compacted fill material as directed by the Owner's Representative.
- B. The Owner's Representative may recommend additional drying time for soft, wet subgrade that has not been exposed long enough to permit "crust" formation. If approved by the Owner's Representative, the Contractor may install, at his own expense, geotechnical fabric or geogrid to stabilize the wet subgrade and expedite construction.
- C. No fill shall be placed until the subgrade has been examined and approved.
- 3.04 GENERAL FILL
 - A. Placement
 - Unless otherwise indicated on the plans, all fill shall be composed of Earth Fill Material.
 - 2. Fill materials used in embankment construction shall normally be placed in lanes parallel to the embankment axis and shall be placed in conformance with the lines, grades, and slopes as indicated on the plans. Placement of fill materials in lanes which are not parallel to the embankment will be allowed only where working room is too restricted for normal placement as determined by the Owner's Representative.
 - 3. Fill shall be spread in approximately flat layers in such a manner as to obtain lifts of relatively uniform thickness without spaces between successively deposited loads. Segregation shall be prevented during placing and spreading. Hauling equipment shall be routed across the fill in such a way as to promote uniform compaction and to prevent the formation of ruts.
 - 4. The maximum compacted thickness of each lift shall not exceed 8 in. where heavy compaction equipment will be used. The maximum compacted thickness shall not exceed 3 in. where power tampers or similar smaller equipment will be used. It may be necessary to reduce the thickness of lifts in order to obtain the required minimum density.
 - Where compacted earth fill is to be placed against existing slopes, each lift shall be keyed into existing slope by removing existing slope material in steps as each new lift is placed.
 - 6. The surface of the fill shall be kept reasonably smooth. The fill surface shall be sloped transverse to the axis of the embankments to allow drainage. If the compacted surface is, in the opinion of the Owner's Representative, too smooth or too dry to bond properly with the

succeeding lift, it shall be roughened by scarifying, light discing, or other acceptable means, and it shall be sprinkled before the succeeding lift is placed thereon. If the surface becomes rutted or uneven subsequent to compaction, it shall be flattened and leveled before placing the next lift. This extra work shall be at the Contractor's expense.

- 7. Fill operations shall be suspended during periods of extended wet weather. Upon resuming operations, all fill materials that are excessively wet or soft shall be reprocessed in place or removed and stockpiled for reprocessing. The removal of soft material shall be carried to such depth as is necessary to expose firm materials. Fill shall not be placed on frozen surfaces.
- 8. When filling operations at any section will be suspended for any period in excess of 12 hours or in wet weather, the surface of the fill shall be rolled smooth to seal it against excessive absorption of moisture and to facilitate runoff. Prior to resuming fill placement and compaction, the fill surface shall be scarified and/or disced and moisture conditioned as required.
- The Contractor will receive no additional compensation for any removal, reprocessing, stockpiling, recompaction, wasting, or similar operation related to suspensions or conditions due to weather or other causes unless caused by the Owner.
- Earth fill access ramps shall not be constructed within the limits of the compacted embankments without prior approval. When such ramps are approved, they shall be constructed of low permeability soil (in-board of the perimeter berm) or compacted fill (out-board of the perimeter berm).
- B. Compaction -
 - Fill materials shall be compacted to a dry density equal to or greater than the following:
 - a. The Gypsum Management Facility: 95 percent of the maximum dry density obtained from the Standard Proctor Test, ASTM D698.

In order to insure uniform coverage and to facilitate construction inspection and control, the compaction of each layer shall proceed in a systematic, orderly, and continuous manner. Rolling shall be parallel to the embankment axis, except where there is insufficient working room for such operations.

- The moisture content of all earth fill materials shall be as uniform as practicable throughout each lift. Fill shall be compacted at a moisture content that is no more than 2 percent below and no more than 2 percent above optimum moisture content.
- 3. Moisture conditioning of fill materials shall be performed by discing, harrowing, plowing, blading, or other suitable means prior to excavation. Moisture conditioning where the fill is placed shall be limited to minor adjustments prior to compaction. Addition of moisture shall be by using a

pressure spray bar mounted in front of or to one side of a water tanker so that water will not collect in the tracks of the truck.

- Compaction of fill materials shall not commence if the moisture content is not within the specified limits. Any materials that are placed but not compacted prior to drying out or becoming too wet shall be removed and replaced or reprocessed at the Contractor's expense.
- No admixtures as drying agents or to improve the workability of the soil will be allowed.

3.05 SOIL LINERS

A. Sources

The Soil Liners for the Gypsum Management Facility shall be constructed from Soil Liner Material as described in paragraph 2.01(B) above.

B. Test Liner

A compacted low permeability soil test liner of the actual full scale liner shall be constructed in accordance with the following requirements:

- Test liner will be constructed from the same soil material sources, to the same design specifications, and with similar equipment and procedures as are proposed for the full scale liner.
- Test liner will be at least four times the width of the widest piece of equipment to be used.
- Test liner will be no less than 100 ft long to allow equipment to reach normal operating speed before reaching a central 40-ft test area.
- Test liner will be constructed with maximum 8-in. compacted lifts for a total liner thickness of 3 ft.
- Test liner will be tested by the Owner's Testing Consultant as described below for each of the following physical properties:
 - a. Multiple two-stage Boutwell permeameter tests will be used on the test liner to determine the hydraulic conductivity. The two-stage field hydraulic conductivity test is a falling head infiltration test conducted in a cased borehole, typically 4 in. in diameter. The test is cited in the U.S. EPA Technical Guidance Document: Quality Assurance and Quality Control for Waste Containment Facilities, September 1993 (EPA/600/R-93/182).
 - Undisturbed samples (Shelby tubes) will be tested in the laboratory for hydraulic conductivity to determine if there is a statistical correlation to the field testing results.
 - c. Other engineering parameters including, but not limited to, particle size analysis, liquid limits, plasticity, water content, and in-place density that are needed to evaluate the full scale liner will be determined.

- Additional test fills will be constructed for each new soil type or for each change in equipment or procedures.
- C. Full Scale Liner Construction:
 - Full scale liner construction shall not be commenced until the results of the in-place compaction testing and Boutwell permeameter tests on the test liner confirm that the construction procedures and specified compaction requirements produce a in-situ hydraulic conductivities as specified in Section 2.01(B) above.
 - 2. The liner shall be constructed according to the placement and compaction requirements for general fill, except the material shall be compacted to a density of no less than 95 percent of maximum dry density at a moisture content between 100 percent and 105 percent of optimum. The same compaction procedures, such as number of passes, speed, and compaction equipment used on construction of the test liner shall be used. Grade stakes shall not be driven into the clay liner.
 - 3. The completed liner shall be smooth rolled to limit moisture loss and promote run-off of surface water. Moisture content shall be maintained within the specified range and erosion or other damage that occurs in the soil liner shall be repaired as directed by the Owner's Representative until the geosynthetic liner is placed.
 - Repair of any rutting or other damage caused by the installation of the geosynthetic liner will be paid for on a time and material basis.
 - 5. Voids created in the clay barrier layer during construction (including, but not limited to, penetrations for test samples, and other penetrations necessary for construction) shall be repaired by removing material that does not meet the requirements for low permeability soil, placing low permeability soil backfill, granular or pelletized bentonite, or a mixture of bentonite and low permeability soil in lifts no thicker than 2 in. and tamping each lift with a steel rod. Each lift shall be tamped a minimum of 25 times altering the location of the rod within the void for each blow. Other ruts and depressions in the surface of the lifts shall be scarified, filled, and then compacted to grade.

3.06 CUSHION DIRT

Cushion Dirt to be placed beneath the upper High Density Polyethylene (HDPE) Geomembrane is to be placed to the specifications for General Fill in Section 3.04 above, except fill materials for Cushion Dirt shall be compacted to a dry density equal to or greater than 90 percent of the maximum dry density obtained from the Standard Proctor Test, ASTM D698.

3.07 ANCHOR TRENCH CONSTRUCTION

- A. Gypsum Management Facility
 - A ledge at the bottom of the anchor trench elevation shall be excavated. Low permeability soil shall be placed and compacted on the ledge as shown on the anchor trench details in the plans.
 - The anchor trench shall be excavated to the depth and width shown on the anchor trench details. The front edge of the trench shall be rounded to eliminate any sharp corners that could cause excessive stress to the geosynthetic liners. Loose soil shall be removed or compacted into the floor of the trench.
 - 3. Subsequent to Geosynthetic Clay Liner (GCL), Bottom HDPE Geomembrane and Geotextile Cushion installation, it shall be verified that the liners cover the entire trench floor, but do not extend up the back of the trench wall. After the liner installation in the trench has been inspected and approved by the Owner's Representative, the trench shall be backfilled with 1 ft. of low permeability soil. The backfill shall be deposited and compacted according to the requirements for general fill in such a manner as to prevent damage to the GCL and liner materials.
 - 4. Subsequent to installation of separation geotextile on top of drainage layer, it shall be verified that the fabric extends across the top of the initial 1 ft layer of trench backfill, but does not extend up the back of the trench wall. After the fabric installation in the trench has been inspected and approved by the Owner's Representative, the trench shall be backfilled with 1 ft of low permeability soil. The backfill shall be deposited and compacted according to the requirements for general fill in such a manner as to prevent damage to the geotextile fabric.
 - 5. Subsequent to installation of the upper HDPE Geomembrane, verify that the liner extends across the top of the initial 1 ft layer of trench backfill, but does not extend up the back of the trench wall. After the liner installation in the trench has been inspected and approved by the Owner's Representative, backfill the remainder of the trench to the top of the low permeability soil layer. Deposit and compact the backfill according to the requirements for general fill in such a manner as to prevent damage to the HDPE Geomembrane.

3.08 TESTING

A. Construction Quality Assurance (CQA) compaction and permeability tests will be made by the Owner's Testing Consultant during the progress of the work as indicated in Appendix 2. The Contractor shall cooperate with the Testing Consultant and allow such tests to be performed.

B. If tests indicate that an area of fill or low permeability soil liner does not meet the specified requirements, additional tests shall be performed to determine the extent of non-compliance. The Contractor shall moisture condition and recompact that area until a passing test result is obtained.

3.08 FINISH GRADING

All excavated and filled areas shall be fine graded and leveled to provide a smooth finish free of debris, foreign matter, objectionable stones, clods, lumps, pockets, or high spots, properly drained and true to indicated elevations. Finish grading shall be only near completion of work or when requested. Any portions of the berm damaged by construction shall be restored. The berm ditch shall be finished to design grade, and the ditch side slopes shaped and trimmed to provide a uniform ditch cross section.

3.09 CONSTRUCTION TOLERANCES

- A. The foundation grade and finished earthwork grades shall be no more than 0.4 ft below and not above plan grade.
- B. The minimum thickness of low permeability soil layer shall be 3 ft.

END OF SECTION 02200

1:05jobs/05s3004A\Gypsum Stacking\IDNR Dam Safety Permit Application\Specs\S02200_Earthwork.doc

DIVISION 2 - SITE WORK Section 02275 - Riprap

PART 1. GENERAL

1.01 DESCRIPTION

This section pertains to the placement of riprap for erosion control.

1.02 RELATED SECTIONS

The following section contains items which are related to the work in this section:

02200 - Earthwork

1.03 REFERENCES

Specified references or cited portions thereof, current at date of bidding documents unless otherwise specified, govern the work.

A. Illinois Department of Transportation (IDOT): Standard Specifications for Road and Bridge Construction, adopted January 1, 2007.

1.04 SUBMITTALS

Product Data: Provide quarry name and material type prior to delivery.

PART 2. PRODUCTS

2.01 MATERIALS

- A. Stone Riprap and Bedding materials according to Article 1005.01 of the Illinois Standard Specifications for Road and Bridge Construction.
- B. Filter Fabric material for Stone Riprap according to Article 1080.03, with an AOS (Apparent Opening Size) as indicated on the plans.
- C. Supplier shall be listed on the current IDOT Approved Aggregate Source List.
- D. Gradation as indicated in the drawings. Quality shall be Class A.

PART 3. EXECUTION

3.01 CONSTRUCTION REQUIREMENTS

- A. Stone Riprap and Bedding shall be installed in accordance with Section 281 of the Illinois Standard Specifications for Road and Bridge Construction for the placement of Stone Riprap. Measurement and payment provisions of Section 281 shall not apply.
- B. Filter Fabric for Stone Riprap shall be installed in accordance with Section 282 of the Illinois Standard Specifications for Road and Bridge Construction.
- C. The Owner's Representative shall be allowed to visually inspect Riprap for compliance with specifications prior to placement.

END OF SECTION 02275

DIVISION 2 – SITEWORK Section 02315 – Granular Materials

PART 1, GENERAL

1.01 DESCRIPTION

A. Gypsum Management Facility

This section pertains to the following:

- Furnishing and placing granular drainage materials for the drainage layer and leachate collection system.
- Furnishing and placing coarse aggregate for encasement of the ring drain collection piping.
- Furnishing and installing materials for roadbed construction related to the Gypsum Management Facility access roads and the McKinley Road relocation.
- Recycle Pond Drain.

1.02 RELATED SECTIONS

The following sections contain items which are related to the work in this section:

- 1. 02300 Earthwork
- 02373 Geotextiles
- 02640 HDPE Piping

1.03 REFERENCES

The following references, or cited portions thereof, govern the work:

- Illinois Department of Transportation (IDOT): Standard Specifications for Road and Bridge Construction, adopted January 1, 2007.
- American Society for Testing and Materials (ASTM):
 - ASTM D 75 (2003) Practice for Sampling Aggregates.
 - b. ASTM D 422 (1963; R 2002) Test Method for Particle-Size Analysis of Soils.
 - ASTM D 2434 (1968, R 2000) Test Method for Permeability of Granular Soils (Constant Head).
 - d. ASTM D 3042 (2003) Test Method for Insoluble Residue in Carbonate Aggregates.
 - ASTM C 1260 (2005) Standard Test Method for Potential Alkali Reactivity of Aggregates (Mortar-Bar Method).

 American Geological Institute (AGI). Geoscience Handbook AGI Data Sheets, 4th Edition.

1.04 MEASUREMENT AND PAYMENT

The Contractor shall be responsible for estimating the extent of granular materials required to complete the work including, but not limited to, construction of drainage layer, encasement of leachate collection piping, and road-bed construction. The Contractor shall include the dollar amount associated with furnishing and placing all granular materials in his Lump Sum Bid amount.

1.05 COORDINATION

A. The geosynthetic liner shall be covered with granular materials as soon as practicable after a section of liner has been approved by the Owner's Representative.

1.06 SUBMITTALS

- A. Product Data:
 - Aggregate source list: Submit a list of proposed aggregate sources.
 - Shipping Tickets: Submit shipping tickets for the granular materials delivered to the site. Shipping tickets shall be according to paragraph 1004.01f of the IDOT Standard Specifications.

B. Test Reports.

- Submit results of grain size analysis (ASTM D422) and hydraulic conductivity testing (ASTM D2434) for gradations established by the Contractor that provide the specified hydraulic conductivity. Test results are required for each proposed source and gradation. Submit test results for each source demonstrating compliance with reactivity, soundness, and abrasion requirements specified herein.
- C. Samples:
 - Submit one sample per source for each gradation proposed for use on the project. Samples shall be at least one pound and shall be obtained and shipped according to ASTM D75. Submit samples at least 15 days prior to starting construction of the drainage layer and coarse aggregate encasement for leachate piping.

1.07 STORAGE AND HANDLING

A. Storage and handling of granular materials shall be according to paragraph 1004.01e of the IDOT Standard Specifications.

1.01 PART 2. PRODUCTS

2.01 MATERIALS

- A. General
 - Unless otherwise approved by the Owner's Representative, granular materials shall be obtained from sources listed on the current IDOT Approved Aggregate Source List (www.dot.il.gov/materials/ approvedaggregatesources.pdf).
 - Coarse Granular materials shall be meet the Description of Gravel, as described in Section 1004.01(a)(1) of the IDOT Standard Specifications, and shwll be spherical to sub-discoidal, sub-rounded to well rounded particles as defined by AGI Data Sheet, 4th Edition, Sheet 8.4 – Comparison Charts for Estimating Roundness and Sphericity.
 - Granular materials shall experience no more than 15 percent carbonate loss per ASTM D3042.
 - Granular materials shall be free of deleterious material, and shall meet the Na₂SO₄ soundness and Los Angeles Abrasion Specifications for Class B quality aggregate per paragraph 1004.01 of the IDOT Standard Specifications.
 - All material shall pass the 2 in. sieve, and no greater than 5 percent shall be retained on the No. 200 sieve.
 - Granular materials shall be innocuous to alkali-silica reactivity, and shall exhibit internal expansions of less than 0.10 percent at 16 days after casting as determined by ASTM C 1260.

- B. Gypsum Management Facility Granular Materials
 - Granular Materials for Drainage Layer

Gradation for granular material for drainage layer shall be as required to provide a minimum hydraulic conductivity (ASTM D2434) of 1 x 10⁻³ cm/sec.

Coarse Aggregate around Ring Drain Collection Piping

Coarse Aggregate used to encase the ring drain collection piping shall be IDOT Gradation CA 7 material as outlined in Article 1004.01 of the IDOT Standard Specifications for Road and Bridge Construction.

Filter Sand

Filter Sand used for protective cover over the ring drain collection system shall be IDOT Gradation FA 1, Class B or better according to Article 1003 of the IDOT Standard Specifications for Road and Bridge Construction.

Aggregate Base Course, Type B

Aggregate Base Course, Type B used for base material for all new access roads and shall be IDOT Gradation CA 2, in accordance with Section 1004.04 of the IDOT Standard Specifications for Road and Bridge Construction. The material shall originate from an IDOT approved source. The Na₂SO₄ soundness and Los Angeles Abrasion Specifications for Class B quality aggregate per paragraph 1004.01 of the IDOT Standard Specifications shall not apply.

Aggregate Surface Course, Type B

Aggregate Surface Course, Type B used for surface material for all new access roads and the McKinley Road relocation shall be IDOT Gradation CA 6, in accordance with Section 1004.04 of the IDOT Standard Specifications for Road and Bridge Construction. The material shall originate from an IDOT approved source. The Na₂SO₄ soundness and Los Angeles Abrasion Specifications for Class B quality aggregate per paragraph 1004.01 of the IDOT Standard Specifications shall not apply.

2.02 EQUIPMENT

Equipment for spreading and compacting granular materials shall be low ground pressure equipment to prevent damage to the underlying geosynthetic liners.

PART 3. EXECUTION

3.01 PROTECTION OF GEOSYNTHETICS

- A. Protection of the geosynthetic liners is critically important. Approved geosynthetic liner shall be covered by granular material as soon as practicable. Granular material shall be placed to a minimum thickness of 1 ft before any heavy equipment or loaded trucks are allowed on the lined area.
- B. No equipment will be permitted directly on the geosynthetic liner.
- C. Any damage to the geosynthetic liner system shall be repaired, as directed by the Owner's Representative, at the expense of the Contractor.

3.02 GRANULAR DRAINAGE LAYER (GYPSUM MANAGEMENT FACILITY)

- A. Placement on Cell Floor
 - The granular material shall be back-dumped on the geotextile cushion fabric in a sequence of operations beginning at the perimeter of the liner on the cell floor.
 - 2. Placement of material on the fabric shall be accomplished by spreading dumped material off of previously placed material with a bulldozer blade or endloader, in such a manner as to prevent tearing or shoving of the cloth. Dumping of material directly on the fabric will only be permitted to establish an initial working platform. No vehicles or construction equipment shall be allowed on the fabric prior to placement of the granular blanket to a minimum thickness of 1 ft.
- B. Placement on Cell Side Slopes
 - Placement of granular material on cell side slopes shall be accomplished using methods and equipment similar to that specified for placement of material on cell floor.
 - The Contractor may place gypsum underlain with separation geotextile fabric to buttress the granular material on the slope:
 - a. The Construction Quality Assurance (CQA) survey to certify thickness of drainage material shall be completed within the footprint of the gypsum stack before gypsum placement.

Separation geotextile fabric shall extend beyond the toe of gypsum buttress a sufficient distance to prevent contamination of the granular drainage layer. See Sections 02373 and 02320 for construction of separation geotextile fabric and gypsum, respectively.

3.04 COARSE AGGREGATE FOR ENCASEMENT OF RING DRAIN COLLECTION PIPING (GYPSUM MANAGEMENT FACILITY)

- A. The geotextile filter fabric for encasement of leachate collection piping shall be placed on the approved cushion geotextile fabric according to Section 02373 – Geotextiles.
- B. The coarse aggregate shall be placed on the encasement fabric to the width shown on the plans to the level of the bottom of the ring drain collection piping.
- C. Course aggregate shall be placed and tamped along the pipe during pipe installation. The coarse aggregate shall be placed longitudinally along the pipe in lifts not to exceed 8 in. thick to a height of at least the center of the pipe. The aggregate shall be maintained at equal elevation on each side of the pipe, and the first lift of material shall be mechanically tamped to ensure that the space under the pipe is completely filled. The top of pipe shall not be covered until the CQA survey certifies leachate piping grade has been completed.
- C. After the CQA survey has been completed, coarse aggregate material shall continue to be placed in lifts not to exceed 8 in. thick, as specified in the previous paragraph until the minimum cover height shown in the plans is attained.
- D. The running of trucks or heavy equipment over leachate piping shall be avoided until there is at least a 12 in. cover of Filter Sand over the completed geotextile envelop. Temporary ramps no steeper than 10H:1V transverse to the piping shall be provided for temporary equipment crossings until the first lift of gypsum is placed.

3.03 ROADWAY CONSTRUCTION

- Prepare the roadway subgrade as shown on the plans, in accordance with Section 02200 – Earthwork.
- B. Furnish Geotechnical Fabric for Ground Stabilization in accordance with Section 02373 – Geotextiles.
- C. Furnish Aggregate Base Course, Type B in accordance with Article 351 of the IDOT Standard Specifications for Road and Bridge Construction.

02315-6

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02315_Granular Materials.doc

D. Furnish Aggregate Surface Course, Type B in accordance with Article 402 of the IDOT Standard Specifications for Road and Bridge Construction.

3.04 TESTING

- A. CQA gradation and permeability tests will be made by the Owner's Testing Consultant during the progress of the work as indicated in Appendix 2. The Contractor shall cooperate with the Testing Consultant and allow such tests to be performed.
- B. If tests indicate that an area of granular material or coarse aggregate does not meet the specified requirements, then the Contractor shall remove the material and replace it with suitable material.

3.05 FINISH GRADING

The granular drainage layer shall be fine graded to provide a smooth finish before a CQA survey of the completed portion of the drainage layer is requested. Ruts or erosion damage shall be repaired before placement of the separation geotextile fabric.

3.06 CONSTRUCTION TOLERANCES

The minimum thickness of drainage layer shall be 1 ft.

END OF SECTION 02315

Division 2 – SITEWORK Section 02373 – Geotextiles

PART 1. GENERAL

1.01 DESCRIPTION

This section pertains to furnishing and installing geotextile fabrics on prepared surfaces.

1.02 RELATED SECTIONS

The following sections contain items which are related to the work in this section:

- 1. 02300 Earthwork
- 2. 02315 Granular Materials
- 02800 HDPE Geomembrane

1.03 REFERENCES

The following references, or cited portions thereof, govern the work:

- Illinois Department of Transportation (IDOT): Standard Specifications for Road and Bridge Construction, adopted January 1, 2007.
- American Society for Testing and Materials (ASTM):
 - ASTM 3776 (1996; R 2002) Standard Test Method for Mass per Unit Area (Weight) of Fabric;
 - ASTM D 3786 (2001) Test Method for Hydraulic Bursting Strength of Textile Fabrics – Diaphragm Bursting Strength Tester Method;
 - ASTM D 4533 (2004) Test Method for Trapezoid Tearing Strength of Geotextiles;
 - d. ASTM D 4632 (1991; R 2003) Test Method for Grab Breaking Load and Elongation of Geotextiles;
 - ASTM D 4751 (2004) Test Method for Determining Apparent Opening Size of Geotextile;
 - f. ASTM D 4833 (2000) Test Method for Index Puncture Resistance of Geotextiles, Geomembranes, and Related Products;
 - g. ASTM D 4873 (2002) Guide for Identification, Storage, and Handling of Geosynthetic Rolls;
 - h. ASTM D 4884 (1996; R 2003) Test Method for Strength of Sewn or Thermally Bonded Seams of Geotextiles;
 - ASTM D5261-92(2003) Standard Test Method for Measuring Mass per Unit Area of Geotextiles

j. ASTM D6241-04 Standard Test Method for the Static Puncture Strength of Geotextiles and Geotextile-Related Products Using a 50-mm Probe

1.04 MEASUREMENT AND PAYMENT

- A. The Contractor shall be responsible for estimating the extent of geotextile fabric required to complete the work including fabric for laps, anchorage, repairs, and samples for Construction Quality Assurance (CQA) testing. The Contractor shall include the dollar amount associated with all geotextile construction in his Lump Sum Bid amount, except as specified in paragraph B.
- B. Geotextile fabric for ground stabilization, when directed by the Owner's Representative, will be paid for on a time and materials basis.
- C. No additional payment will be made for geotextile fabric for ground stabilization installed at the Contractor's discretion.

1.05 SUBMITTALS

- A. Product Data
 - The manufacturer's list of guaranteed properties for each geotextile fabric or geogrid proposed for use on the project shall be submitted.
 - 2. The manufacturer's installation guidelines shall be submitted.
- B. Samples

Samples of geotextile fabrics shall be submitted for CQA prequalification testing. Sample size and sampling frequency are specified in Appendix 2.

C. Inventory

A copy of the roll inventory that identifies, as a minimum, manufacturer or supplier, product or style number, roll number, width, and length of roll as identified on the roll label shall be submitted.

1.06 STORAGE AND HANDLING

Geotextiles shall be stored and handled according to ASTM D4873.

02373-2

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02373_Geotextiles.doc

PART 2. PRODUCTS

2.01 MATERIALS

A. Geotextile Fabric for Liner System

Geotextile fabrics for use in the cell liner system shall consist of non-woven filaments of polypropylene, polyester, or polyethylene. Stabilizers and/or inhibitors shall be added to the base polymer if necessary to make the filaments resistant to deterioration caused by ultraviolet light and heat exposure. Reclaimed or recycled fibers or polymer shall not be added to the formulation. Non-woven fabric may be needle-punched, heat-bonded, or a combination thereof. The filaments shall be dimensionally stable (i.e., filaments shall maintain their relative position with respect to each other) and resistant to delamination. The edges of the geotextile shall be finished to prevent the outer fiber from pulling away from the geotextile. The filaments shall be free from any chemical treatment or coating that might significantly reduce porosity and permeability.

Physical Properties ⁽¹⁾	4 oz. (Separation)	6 oz. (PWRS)	16 oz. (CA Envelope)
Mass/Unit Area (oz/yd ²) ASTM D5261	4.0	6.0	16.0
Grab Tensile Strength (lb.) ASTM D4632	115	160	380
Grab Elongation (%) ASTM D4632	50	50	50
Puncture Strength (lb.) ASTM D4833	65	85	240
Puncture (CBR) Strength (lb.) ASTM D6241	310	410	1025
Mullen Burst Strength (psi) ASTM D3786	210	280	750
Trapezoidal Tear Strength (lb.) ASTM D4533	50	60	150
Width (ft.)	15	15	15
Apparent Opening Size (AOS) Max. US Std. Sieve No. ASTM D4751	70	70	100
UV Resistance ⁽²⁾ (%) ASTM D4355	70	70	70
Roll Width (ft.)	15	15	15

Fabric shall have the following physical properties:

Notes:

 All Values listed are Minimum Average Roll Values (MARV) unless otherwise noted, calculated as the typical minus two standard deviations.

(2) UV Resistance is a minimum value and not a MARV. Evaluation to be on 2.0 inch strip tensile specimens after 500 hours exposure. A. Cushion Geotextile Fabric.

Cushion geotextile fabric shall consist of non-woven filaments of polypropylene, polyester, or polyethylene. Stabilizers and/or inhibitors shall be added to the base polymer if necessary to make the filaments resistant to deterioration caused by ultraviolet light and heat exposure. Reclaimed or recycled fibers or polymer shall not be added to the formulation. Non-woven fabric may be needle-punched, heatbonded, or a combination thereof. The filaments shall be dimensionally stable (i.e., filaments shall maintain their relative position with respect to each other) and resistant to delamination. The edges of the geotextile shall be finished to prevent the outer fiber from pulling away from the geotextile. The filaments shall be free from any chemical treatment or coating that might significantly reduce porosity and permeability.

Physical Properties ⁽¹⁾	10 oz. (Cushion)	
Mass per unit area (oz/yd ²) ASTM D5261	10	
Grab Tensile Strength (lb.) ASTM D4632	230	
Grab Tensile Elongation (%) ASTM D4632	50	
Trapezoidal Tear Strength (lb.) ASTM D4533	95	
Puncture (CBR) Strength (lb.) ASTM D6241	700	
Puncture (CBR) Elongation (in.) ASTM D6241	1.5	
UV Resistance ⁽²⁾ (%) ASTM D4355	70	
Apparent Opening Size (Max.) (AOS) Sieve No ASTM D4751	 .	
Roll Width (ft.)	15	

Fabric shall have the following physical properties:

Notes:

 All Values listed are Minimum Average Roll Values (MARV) unless otherwise noted, calculated as the typical minus two standard deviations.

(2) UV Resistance is a minimum value and not a MARV. Evaluation to be on 2.0 inclustrip tensile specimena after 500 hours exposure.

B. Geotechnical Fabric for Ground Stabilization

Geotechnical fabric for ground stabilization shall conform to Article 1080.02 of the IDOT Standard Specifications for Road and Bridge Construction. C. Thread for Seams

High strength thread should be used such that seam test should conform to ASTM D4884. The thread shall meet the chemical, ultraviolet, and physical requirements of the geotextile, and the color shall be different from that of the geotextile.

D. Securing Devices

Pins, staples, and other devices that project through the geotextile fabric are not permitted for fabrics installed above the geomembrane. Sandbags, stone, or other appropriate means approved by the Owner's Representative shall be used to prevent movement of the geotextile.

2.02 EQUIPMENT

A. Equipment for spreading and compacting granular materials shall be low ground pressure equipment to prevent damage to the underlying geosynthetic liners.

PART 3. EXECUTION

- 3.01 SAMPLES FOR CQA TESTING
 - A. Geotextile fabric samples shall be obtained, identified and packaged from rolls designated by the Owner's Representative according to ASTM D4873.
 - B. Samples shall be 3 ft. wide by the full roll width.

3.02 BASE PREPARATION

- A. Surface on which the geotextile will be placed shall be prepared to a relatively smooth surface condition, and shall be free from obstruction, debris, depressions, erosion features, or any irregularities that would prevent continuous, intimate contact of the geotextile with the entire surface. Rills, gullies, and ruts must be graded out of the surface before geotextile placement. Areas on which geotextile are to be placed shall be graded and/or dressed in accordance with Section 02200 Earthwork and Section 02315 Granular Drainage Materials. Immediately prior to placing the geotextile, the prepared base will be inspected by the Owner's Representative, and no material shall be placed thereon until that area has been approved.
- B. Geotextile cushion fabric will be installed directly on the geosynthetic liner. Jointly inspect the liner with the Owner's Representative before commencing fabric installation each day. Notify the Owner's Representative promptly of any damage or defects observed in the liner as fabric installation progresses. Do not place fabric in the damaged or defective area until the liner has been repaired and

^{1/05}jobs/05s3004A/Gypsum Stacking/JDNR Dam Safety Permit Application/Specs/S02373_Geotextiles.doc

approved by the Owner's Representative. Submit a daily inspection report identifying the area of fabric placement and certifying that there were no visible defects in the area of fabric placement.

C. Do not run heavy vehicle traffic directly on the geosynthetic liner or cushion geotextile. Use vehicles and equipment as specified in paragraph 2.02 to transport and deploy fabric on the liner. Operate the equipment with care, and place protective cover over the geomembrane, if necessary, to avoid damaging the liner. Route traffic and personnel over installed cushion fabric and use the installed fabric as a working platform to the greatest extent possible.

3.03 INSTALLATION

- A. General Requirements:
 - Geotextile fabric shall be unrolled and laid out following these requirements to the greatest extent practical:
 - Orient panels with the longest dimension parallel to the slope.
 - b. Minimize the number of seams in corners and odd-shaped areas.
 - Extend panels on slopes a minimum of 5 ft onto a horizontal surface.

Geotextile panels shall be unrolled using methods that will not damage the fabric and will protect underlying surface from damage. While unrolling, the geotextile fabric shall be visually inspected for imperfections and faulty or suspect areas marked. Ballast shall be placed on fabric to prevent wind uplift. Expansion and contraction should be allowed for by leaving slack.

Heavy vehicle traffic shall not be run directly on geotextile fabric. Fabric in areas of heavy traffic shall be protected with protective cover over the fabric.

2. Laps

Individual panels of geotextile fabric shall be lapped according to manufacturer's instructions and as specified herein. Provide a minimum overlap of 3 in. unless otherwise specified herein or in the plans. Shingle overlaps so that water or other material cannot run down the slope between the two layers of fabric. 3. Field Seams

Continuously sew all laps on slopes steeper than 10H:1V. This requirement does not apply to the heavy geotextile fabric for envelopment of coarse aggregate around leachate piping.

4. Defects and Repairs

Examine the installed geotextile fabric for defects, holes discontinuous seams, puckered or separated laps, etc. Repair defective laps and seams. Patch holes and defects according to manufacturer's recommendations and as directed by the Owner's Representative. Do not cover suspect or patched areas until they have been inspected and approved by the Owner's Representative.

- B. Geotextile Fabric for Separation
 - Use low ground pressure equipment to avoid rutting the granular material.
 - Horizontal seams (parallel to top of slope) will be permitted on cell side slopes to facilitate staged construction of the drainage layer on the side slope.
 - Extend separation geotextile fabric into and across the bottom of the anchor trench and complete backfill of the trench according to Section 02200.
- C. Geotextile Fabric for Coarse Aggregate Envelope
 - Geotextile for coarse aggregate envelope will be installed directly on the cushion fabric. Remove any foreign materials from the cushion fabric within the footprint of the coarse aggregate leachate piping encasement before installing the geotextile envelope. Place sufficient width to completely envelop the coarse aggregate and provide a longitudinal lap of at least 6 in.
 - After the coarse aggregate encasement has been completed, according to Section 02315, wrap the geotextile around the mounded aggregate, and cover the lap with at least 6 in. of material before permitting vehicle or equipment on the fabric.
 - Any ballast material other than coarse aggregate, according to Section 02315, that is placed within the envelope will require removal during coarse aggregate construction.
- D. Geotechnical Fabric for Ground Stabilization
 - Install Geotechnical Fabric for Ground Stabilization in accordance with Section 210 of the IDOT Standard Specifications for Road and Bridge Construction.

- If approved by the Owner's Representative, the Contractor may, at his own expense, install geotextile or geogrid for ground stabilization outside the limits designated by the Owner's Representative.
- Submit as-built drawings that clearly delineate limits and type of ground stabilization.

3.04 PROTECTION

- A. Protect installed fabric until it is covered by at least 1 ft. of overlying material.
- B. Any damage to the geotextile during its installation or during placement of overlying materials shall be replaced by the Contractor at no cost to the Owner. Unless otherwise noted, the work shall be scheduled so that the covering of the geotextile with a layer of the specified material is accomplished within 14 calendar days after placement of the geotextile. Failure to comply shall require replacement of geotextile. The geotextile shall be protected from damage prior to and during the placement of overlying materials. Before placement of overlying materials, the Contractor shall demonstrate that the placement technique will not cause damage to the geotextile.

3.05 TESTING AND INSPECTION

A. Prequalification Testing

Geotextiles are subject to CQA testing by the Owner's Testing Consultant to verify conformance with the manufacturer's list of guaranteed properties according to Appendix 2. The Contractor shall provide samples as specified herein. If tests indicate nonconformance to the list of guaranteed properties, provide additional samples as directed by the Owner's Representative to determine the extent of the non-conformance. Any fabric that does not conform to the list of guaranteed properties shall be removed from the site.

- B. Installed fabric shall be inspected by the Owner's Representative. No material shall be placed on the fabric, other than ballast, until the installation has been approved by the Owner's Representative. Ballast shall not obscure seams or significant length of unseamed laps. The Owner's Representative may require removal of ballast to inspect suspect areas.
- C. If the Owner's Representative suspects that completed work has been damaged by construction methods that do not conform to the specifications, he may require removal of completed work to verify the integrity of the underlying materials. The Contractor shall bear the cost of removal and subsequent repair as directed by the Owner's Representative.

END OF SECTION 02373

DIVISION 2 - SITEWORK Section 02376 - Geosynthetic Clay Liner

PART 1. GENERAL

1.01 DESCRIPTION

- A. This section covers furnishing and installation of a reinforced needlepunched Geosynthetic Clay Liner (GCL) at the Gypsum Management Facility and the CCB Management Facility.
- B. The work includes furnishing all equipment and materials, providing all labor, supervision, administration and management necessary to perform the work as specified herein and as shown on the plans.

1.02 RELATED SECTIONS

None.

1.03 REFERENCES

The following references, or cited portions thereof, govern the work

- American Society for Testing and Materials (ASTM):
 - ASTM D 4632 (1991; R 2003), Standard Test Method for Grab Breaking Load and Elongation of Geotextiles;
 - ASTM D 4643 (2000), Test Method for Determination of Water (Moisture) Content of Soil by the Microwave Oven Method;
 - c. ASTM D 5084 (2003), Test Method for Measurement of Hydraulic Conductivity of Saturated Porous Materials Using a Flexible Wall Permeameter;
 - ASTM D 5261 (1992; R 2003), Test Method for Measuring Mass Per Unit Area of Geotextiles;
 - ASTM D 5321 (2002), Test Method for Determining the Coefficient of Soil and Geosynthetic or Geosynthetic and Geosynthetic Friction by the Direct Shear Method;
 - f. ASTM D 5887 (2004), Test Method for Measurement of Index Flux Through Saturated Geosynthetic Clay Liner Specimens Using a Flexible Wall Permeameter;
 - g. ASTM D 5888 (1995; R 2002), Practice for Storage and Handling of Geosynthetic Clay Liners;
 - h. ASTM D 5889 (1997; R 2003), Practice for Quality Control of Geosynthetic Clay Liners;

- ASTM D 5890 (2002), Test Method for Swell Index of Clay Mineral Component of Geosynthetic Clay Liners;
- ASTM D 5891 (2002), Test Method for Fluid Loss of Clay Component of Geosynthetic Clay Liners.

1.04 SUBMITTALS

- A. With the bid, the Contractor shall furnish the following information:
 - Conceptual description of the proposed plan for placement of the GCL panels over the areas of installation.
 - GCL Manufacturer's Quality Control (MQC) Plan for documenting compliance with Sections 2.01 and 2.02 of these specifications.
 - GCL manufacturer's historical data for reinforced GCL of a) 10,000-hour creep shear testing per Section 2.01 D, and b) seam flow data at 2 psi confining pressure per Section 2.01 E.
 - A copy of GCL manufacturer's International Standards Organization (ISO) Quality Certificate of Registration.
 - 5. Statement of experience from the proposed GCL supplier.
 - 6. Statement of experience from the proposed GCL installer.
- B. At the Owner Representative's or Owner's request, the Contractor shall furnish:
 - 1. A representative sample of the GCLs.
 - A project reference list for the GCL(s) consisting of the principal details of at least ten projects totaling at least 10 million sq. ft (100,000 sq. meters) in size.
- C. Upon shipment, the Contractor shall furnish:
 - The GCL manufacturer's Quality Assurance/Quality Control (QA/QC) certifications to verify that the materials supplied for the project are in accordance with the requirements of this specification.
 - Inventory of materials received.
- D. As installation proceeds, the Contractor shall submit certificates of subgrade acceptance, signed by the Contractor and Construction Quality Assurance (CQA) Inspector (see Sections 1.06 and 3.03) for each area that is covered by the GCL.
- E. Warranty

After construction, the contractor shall submit material and installation warranty certificates.

1.05 QUALIFICATIONS

- A. GCL Manufacturer must have produced at least 10 million sq. ft. (1 million sq. meters) of GCL, with at least 8 million sq. ft. (800,000 sq. meters) installed.
- B. The GCL Installer must either have installed at least 1 million sq. ft. (100,000 sq. meters) of GCL, or must provide to the Engineer satisfactory evidence, through similar experience in the installation of other types of geosynthetics, that the GCL will be installed in a competent, professional manner.

1.06 CONSTRUCTION QUALITY ASSURANCE (CQA)

- A. The Owner shall provide a third-party inspector for CQA of the GCL installation. The inspector shall be an individual or company who is independent from the manufacturer and installer and who shall be responsible for monitoring and documenting activities, related to the CQA of the GCL throughout installation.
- B. Testing of the GCL as necessary to support the CQA effort shall be performed by a third party laboratory retained by the Owner and independent from the GCL manufacturer and installer.

WARRANTY

The geomembrane material shall be warranted, on a pro-rata basis against manufacturer's defects for a period of five (5) years from the date of liner installation. The installation shall be warranted against defects in workmanship for a period of (1) year from the date of liner completion.

PART 2. PRODUCTS

2.01 MATERIALS

- A. Acceptable products for the GCL are GCL Bentomat[®] SDN, as manufactured by CETCO, 1350 West Shure Drive, Arlington Heights, Illinois 60004 USA (847-392-5800), or an engineer-approved reinforced needlepunched GCL material equal to Bentomat SDN.
- B. The delineation of areas to receive GCL shall be agreed by the Installer and the Engineer prior to installation.
- C. The GCL and its components shall have the properties shown in the GCL Certified Properties table at the end of this section.
- D. The reinforced GCL shall have 10,000 hour test data for large-scale constant-load (creep) shear testing for related products under hydrated conditions. The

^{1:05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02376_GCL.doc

displacement shall be 0.13 in. (3.3 mm) or less at a constant shear load of 250 psf (12 kPa) and a normal load of 500 psf (24 kPa).

- E. The reinforced GCL shall have seam test data from an independent laboratory showing that the seam flow with a grooved cut in one of the nonwoven geotextiles is less than 1 x 10⁻⁸ m³/m²/s at 2 psi hydraulic pressure.
- F. The minimum acceptable dimensions of full-size GCL panels shall be 150 ft. (45.7 m) in length. Short rolls [(those manufactured to a length greater than 70 ft. (21 meters) but less than a full-length roll)] may be supplied at a rate no greater than three (3) per truckload or three (3) rolls every 36,000 sq. ft. (3,500 sq. meters) of GCL, whichever is less.
- G. A 6-inch (150 mm) overlap guideline shall be imprinted on both edges of the upper geotextile component of the GCL as a means for providing quality assurance of the overlap dimension. Lines shall be printed in easily visible, nontoxic ink.

2.02 PRODUCT QUALITY DOCUMENTATION

The GCL manufacturer shall provide the Contractor or other designated party with manufacturing QA/QC certifications for each shipment of GCL. The certifications shall be signed by a responsible party employed by the GCL manufacturer and shall include:

- A. Certificates of analysis for the bentonite clay used in GCL production demonstrating compliance with the swell index and fluid loss parameters shown in the GCL Certified Properties tables.
- B. Manufacturer's test data for finished GCL product(s) of bentonite mass/area, GCL tensile strength and GCL peel strength (reinforced only) demonstrating compliance with the index parameters shown in the GCL Certified Properties tables.
- C. GCL lot and roll numbers supplied for the project (with corresponding shipping information).

2.03 PRODUCT LABELING

- A. Prior to shipment, the GCL manufacturer shall label each roll, identifying:
 - Product identification information (Manufacturer's name and address, brand product code).
 - 2. Lot number and roll number.
 - 3. Roll length, width and weight.

2.04 PACKAGING

- A. The GCL shall be wound around a rigid core whose diameter is sufficient to facilitate handling. The core is not necessarily intended to support the roll for lifting but should be sufficiently strong to prevent collapse during transit.
- B. All rolls shall be labeled and bagged in packaging that is resistant to photodegradation by ultraviolet (UV) light.

2.05 ACCESSORY BENTONITE

A. The granular bentonite sealing clay used for overlap seaming, penetration sealing and repairs shall be made from the same natural sodium bentonite as used in the GCL and shall be as recommended by the GCL manufacturer. Seaming of GCLs shall be conducted in accordance with the manufacturer's guidelines for each particular GCL. Please refer to the installation guidelines for Bentomat /Claymax GCLs.

PART 3. EXECUTION

3.01 SHIPPING AND HANDLING

- A. The rolls of GCL shall be packaged and shipped by appropriate means to prevent damage to the material and to facilitate off-loading.
- B. The Installation Supervisor shall be present during delivery and unloading of the GCL. A visual inspection of each roll should be made during unloading to identify if any packaging has been damaged. Rolls with damaged packaging should be marked and set aside for further inspection. The packaging should be repaired prior to being placed in storage. The Installation Supervisor shall prepare and submit an inventory that includes lot and roll number for materials received.
- C. The Installer is responsible for unloading the GCL. The Owner will make available equipment and operators employed at the site to assist with unloading. The Installer shall coordinate with the Owner to determine equipment availability and should contact the Manufacturer prior to shipment to ascertain the appropriateness of the proposed unloading methods and equipment.

3.02 STORAGE

A. Storage of the GCL rolls shall be the responsibility of the Installer. A dedicated storage area shall be provided by the Owner at the job site. Submit storage area requirements (size and preferred location) with bid documents.

- B. Rolls should be stored in a manner that prevents sliding or rolling from the stacks and may be accomplished by the use of chock blocks. Rolls should be stacked at a height no higher than that at which the lifting apparatus can be safely handled (typically no higher than four).
- C. All stored GCL materials and the accessory bentonite must be covered with a plastic sheet or tarpaulin until their installation.
- D. The integrity and legibility of the labels shall be preserved during storage.

3.03 EARTHWORK

- A. The low permeability soil layer upon which the GCL is installed shall be prepared and compacted prior to installation. The surface shall be smooth, firm, and unyielding, and free of:
 - 1. Vegetation.
 - Construction debris.
 - 3. Sticks.
 - 4. Sharp rocks.
 - 5. Void spaces.
 - 6. Ice.
 - Abrupt elevation changes.
 - 8. Standing water.
 - 9. Cracks larger than 0.25 in. (6 mm) in width.
 - 10. Any other foreign matter that could contact the GCL.
- B. Immediately prior to GCL deployment, the low permeability soil layer shall be final-graded by the contractor to fill in all voids or cracks and then smooth-rolled to provide the best practicable surface for the GCL. At completion of this activity, no wheel ruts, footprints or other irregularities shall exist in the subgrade. Furthermore, all protrusions extending more than 0.5 in. (12 mm) from the surface shall either be removed, crushed or pushed into the surface with a smoothdrum compactor.
- C. On a continuing basis, the project CQA inspector shall certify acceptance of the subgrade before GCL placement.
- D. It shall be the Installer's responsibility thereafter to indicate to the Owner's Representative any change in the condition of the low permeability soil layer that could cause the subgrade to be out of compliance with any of the requirements listed in this Section. The Installation Supervisor shall certify in the daily report that no GCL was placed over visibly defective low permeability soil surface.
- E. At the top of sloped areas of the job site, an anchor trench for the GCL shall be excavated by the contractor in accordance with the project plans. The trench shall

^{1:05}jobs/05s3004A/Gypsum Stacking/JDNR Dam Safety Permit Application/Specs/S02376_GCL.doc

be excavated and approved by the CQA Inspector prior to GCL placement. No loose soil shall be allowed at the bottom of the trench and no sharp corners or protrusions shall exist anywhere within the trench.

3.04 GCL PLACEMENT

- A. GCL rolls shall be delivered to the working area of the site in their original packaging. Immediately prior to deployment, the packaging shall be carefully removed without damaging the GCL. The orientation of the GCL (i.e., which side faces up) shall be in accordance with the Owner Representative's recommendations.
- B. Equipment which could damage the GCL, shall not be allowed to travel directly on it. If the installation equipment causes rutting of the subgrade, the subgrade must be restored to its originally accepted condition before placement continues.
- C. Care must be taken to minimize the extent to which the GCL is dragged across the subgrade in order to avoid damage to the bottom surface of the GCL. A temporary geosynthetic subgrade covering commonly known as a slip sheet or rub sheet may be used to reduce friction damage during placement.
- D. The GCL panels shall be placed parallel to the direction of the slope.
- E. All GCL panels shall lie flat on the underlying surface, with no wrinkles or folds, especially at the exposed edges of the panels.
- F. Only as much GCL shall be deployed as can be covered at the end of the working day with soil, a geomembrane, or a temporary waterproof tarpaulin. The GCL shall not be left uncovered overnight. If the GCL is hydrated when no confining stress is present, the Installer shall remove and replace the hydrated material as directed by the Owner Representative.

3.05 ANCHORAGE

A. As directed by the project drawings and specifications, the end of the GCL roll shall be placed in an anchor trench at the top of the slope. The front edge of the trench shall be rounded so as to eliminate any sharp corners. Loose soil shall be removed from the floor of the trench. The GCL shall cover the entire trench floor, but shall not extend up the rear trench wall.

3.06 SEAMING

A. The GCL seams shall be constructed by overlapping their adjacent edges according to the manufacturer's recommendations. Care should be taken to ensure that the overlap zone is not contaminated with loose soil or other debris.

- B. The minimum dimension of the longitudinal overlap should be 6 in. (150 mm) for Bentomat SDN. If the GCL is manufactured with a grooved cut in the nonwoven geotextile that allows bentonite to freely extrude into the longitudinal overlap then no bentonite-enhanced seam is required for this overlap. If the GCL does not have a grooved cut in one of the nonwoven geotextiles in the longitudinal overlap, then bentonite-enhanced seams are required as described below.
- C. End-of-roll overlapped seams shall be constructed with a minimum overlap of 24 in. (600 mm) for Bentomat SDN. Seams at the ends of the panels should be constructed such that they are shingled in the direction of the grade to prevent the potential for runoff flow to enter the overlap zone. End-of-roll overlapped seams for all reinforced GCL seams require bentonite-enhanced seams as described below.
- D. Bentonite-enhanced seams shall be constructed between the overlapping adjacent panels as follows. The underlying edge of the longitudinal overlap is exposed and then a continuous bead of granular sodium bentonite is applied along a zone defined by the edge of the underlying panel and the 6-inch (150 mm) line. The granular bentonite shall be applied at a minimum application rate of one quarter pound per lineal ft. (0.4 kg/m). A similar bead of granular sodium bentonite is applied at the end-of-roll overlap.

3.07 DETAIL WORK

- A. There shall be no penetrations through the GCL.
- B. Cutting the GCL should be performed using a sharp utility knife. Frequent blade changes are recommended to avoid damage to the geotextile components of the GCL during the cutting process.

3.08 DAMAGE REPAIR

A. If the GCL is damaged (torn, punctured, perforated, etc.) during installation, it may be possible, if approved by the Owner's Representative, to repair it by cutting a patch to fit over the damaged area. The patch shall be obtained from a new GCL roll and shall be cut to size such that a minimum overlap of 12 in. (300 mm) is achieved around all of the damaged area. Granular bentonite or bentonite mastic shall be applied around the damaged area prior to placement of the patch. It may be desirable to use an adhesive to affix the patch in place so that it is not displaced during cover placement. Patching shall be observed and approved by the Owner's Representative.

GCL CERTIFIED PROPERTIES

MATERIAL PROPERTY	TEST METHOD	TEST FREQUENCY ft ² (m ²)	REQUIRED VALUES
Bentonite Swell Index ¹	ASTM D 5890	1 per 50 tonnes	24 mL/2g min.
Bentonite Fluid Loss ¹	ASTM D 5891	1 per 50 tonnes	18 mL max.
Bentonite Mass/Area ²	ASTM D 5993	40,000 ft ² (4,000 m ²)	0.75 lb/ft² (3.6 kg/m²) min
GCL Grab Strength ³	ASTM D 6768	200,000 ft ² (20,000 m ²)	30 lbs/in (53 N/cm) MARV
GCL Peel Strength ³	ASTM D 6496	40,000 ft ² (4,000 m ²)	2.5 lbs/in (4.4 N/cm) min
GCL Index Flux ⁴	ASTM D 5887	Weekly	1 x 10 ⁻⁸ m ³ /m ² /sec max
GCL Hydraulic Conductivity4	ASTM D 5887	Weekly	5 x 10 ⁻⁹ cm/sec max
GCL Hydrated Internal Shear Strength ⁵	ASTM D 5321 ASTM D 6243	Periodic	500 psf (24 kPa) typ @ 200 psf

Notes

Bentonite property tests performed at a bentonite processing facility before shipment the manufacturer's production facilities.

² Bentonite mass/area reported at 0 percent moisture content.

¹ All tensile strength testing is performed in the machine direction using ASTM D 6768. All peel strength testing is performed using ASTM D 6496. Upon request, tensile and peel results can be reported per modified ASTM D 4632 using 4 inch grips.

⁶ Index flux and permeability testing with deaired distilled/deionized water at 80 psi (551kPa) cell pressure, 77 psi (531 kPa) headwater pressure and 75 psi (517 kPa) tailwater pressure. Reported value is equivalent to 925 gal/acre/day. This flux value is equivalent to a permeability of 5x10⁹ cm/sec for typical GCL thickness. Actual flux values vary with field condition pressures. The last 20 weekly values prior the end of the production date of the supplied GCL may be provided.

¹ Peak values measured at 200 psf (10 kPa) normal stress for a specimen hydrated for 48 hours. Site-specific materials, GCL products, and test conditions must be used to verify internal and interface strength of the proposed design.

END OF SECTION 02376

02376-9

1:05jobs/05s3004A/Gypourn Stacking/IDNR Dam Safety Permit Application/Specs/S02376_GCL.doc

DIVISION 2 - SITE WORK Section 02640 - HDPE Piping

PART 1. GENERAL

1.01 DESCRIPTION

This section pertains to construction of the HDPE (High Density Polyethylene) Piping at the Gypsum Management Facility and the CCB Management Facility.

1.02 RELATED SECTIONS

None.

1.03 REFERENCES

The following references, or cited portions thereof, govern the work

- A. American Society of Testing and Materials:
 - ASTM D 2683 (2004); Specification for Socket-Type Polyethylene Fittings for Outside Diameter-Controlled Polyethylene Pipe and Tubing.
 - ASTM D 3261 (2003); Specification for Butt Heat Fusion Polyethylene (PE) Plastic Fittings for Polyethylene (PE) Plastic Pipe and Tubing.
 - ASTM D 3350 (2005); Specification for Polyethylene Plastics Pipe and Fittings Materials.
 - 4. ASTM F 412 (2001a); Terminology Relating to Plastic Piping System.
 - ASTM F 1055 (1998); Specification for Electrofusion Type Polyethylene Fittings for Outside Diameter Controlled Polyethylene Pipe and Tubing.
 - ASTM F 1056 (2004); Specification for Socket Fusion Tools for Use in Socket Fusion Joining Polyethylene Pipe or Tubing and Fittings.

1.04 SUBMITTALS

A. Qualifications

Submit qualifications of the Welding Supervisor who will be responsible for construction quality control of the pipe joining process.

B. Material Certifications

Submit manufacturer certifications that the pipe provided complies with the requirements herein.

- C. Product Data
 - 1. Submit product data and operating instructions for pipe joining equipment.
 - Submit pipe manufacturer's recommended procedures for storing, handling, and installing pipe and fittings.

1.05 QUALIFICATIONS

A. The Contractor or Subcontractor performing the work under this section shall have in his employ a Welding Supervisor who has completed a minimum of 1,000 ft of pipe joining work using the type of equipment proposed for use in this work. The Welding Supervisor shall be on site at all times during pipe line installation, and shall provide direct supervision over other employees.

1.06 WARRANTY

A. The pipe and fittings shall be warranted, on a pro-rata basis, against manufacturer's defects for a period of five (5) years from the date of pipe installation. The installation shall be warranted against defects in workmanship for a period of one (1) year from the date of completion of the leachate collection piping system.

PART 2. PRODUCTS

2.01 MATERIALS

- A. Pipe
 - Pipe material shall be High Density Polyethylene (HDPE) PE 3408, according to ASTM F412, with a cell class designation of 345464C, according to ASTM D3350. Iron pipe size (IPS) and standard dimension ratio (SDR) shall be as indicated in the plans.
 - 2. Size and spacing of holes in perforated pipe shall be as indicated in the plans.
- B. Fittings
 - Fittings shall be made of the same material, and shall have a pressure rating no less than 160 psi. Butt fusion, socket, or electrofusion fittings, according to ASTM D3261, ASTM D2683, and ASTM F1055, respectively, are acceptable.

2.02 EQUIPMENT

02640-2

1:05jobi/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02640_HDPE Piping.doc
A. Butt Fusion Machine

The butt fusion machine shall include the following features:

- 1. Facer with rotating planer block design.
- Heater faces coated by the manufacturer to prevent molten plastic from adhering to the heater face.
- Hydraulic-operated jaws suitable for use with the pipe sizes indicated in the plans.
- B. Socket Fusion Equipment

Socket fusion heating tools and depth gauges shall be of the same manufacturer, unless they are all marked F1056, indicating compliance with ASTM F1056.

C. All equipment shall conform to any requirements specified in the pipe and socket manufacturer's installation instructions, and shall be approved by the Owner's Representative.

PART 3. EXECUTION

- 3.01 MATERIAL DELIVERY, STORAGE, AND HANDLING
 - A. HDPE pipe and fittings shall be packaged and shipped by appropriate means to prevent damage to the material and to facilitate off-loading. The Owner will provide an on-site storage site. Storage site requirements (size and preferred location) shall be submitted with the bid documents.
 - B. Storage and handling shall be according to manufacturer's recommendations.

3.02 BASE PREPARATION

All HDPE piping shall be installed on a layer of coarse aggregate placed by the Contractor in accordance with the plans. The grade of the coarse aggregate base shall be verified before installing the piping.

3.03 INSTALLATION

All pipe and fittings shall be installed according to the manufacturer's recommendations. Removal of weld beads is not required. Contractor shall place coarse aggregate along the pipe to provide lateral stability. Welds shall not be obscured until they have been approved by the Owner's Representative, the top of pipe shall not be covered until the Construction Quality Assurance (CQA) survey has been completed to verify conformance with specified tolerances.

3.04 INSPECTIONS

- A. The Owner's Representative shall shall be visually inspect pipe materials to verify that each pipe material is properly stamped (by the manufacturer) for ASTM acceptance before installation. Defective or damaged materials shall be removed from the site.
- B. Each weld and connection shall be visually inspected by the Owner's Representative. Defective welds shall be repaired as directed by the Owner's Representative and according to manufacturer's recommendations. Welds and connections shall not be covered until they have been approved by the Owner's Representative.

3.05 TOLERANCES

A. HDPE piping shall be located within 0.5 ft. of plan location, and elevation shall be within 0.1 ft. of plan elevation with no adverse slopes.

END OF SECTION 02936

DIVISION 2 - SITEWORK Section 02800 - HDPE Geomembrane

PART 1. GENERAL

1.01 DESCRIPTION

- A. This section includes manufacturing, furnishing, and installing High Density Polyethylene (HDPE) Geomembranes for the Gypsum Management Facility and the Gypsum Management Facility Recycle Pond.
- B. The work includes furnishing all equipment and materials and providing all labor, supervision, administration and management necessary to perform the work as shown on the plans.

1.02 RELATED SECTIONS

- A. The following sections contain items which are related to the work in this section:
 - 1. 02373 Geotextiles
 - 02376 Geosynthetic Clay Liner

1.03 REFERENCES

- A. The following references, or cited portions thereof, govern the work:
 - American Society for Testing and Materials (ASTM):
 - a. D 638, Standard Test Method for Tensile Properties of Plastics.
 - b. D 751, Standard Test Methods for Coated Fabrics.
 - c. D 792, Standard Test Methods for Density and Specific Gravity (Relative Density) of Plastics by Displacement.
 - D 1004, Standard Test Method for Initial Tear Resistance of Plastic Film and Sheeting.
 - D 1204, Standard Test Method for Linear Dimensional Changes of Non Rigid Thermoplastic Sheeting or Film at Elevated Temperature.
 - D 1238, Standard Test Method for Flow Rates of Thermoplastics by Extrusion Plastometer.
 - g. D 1505, Standard Test Method for Density of Plastics by Density-Gradient Technique.
 - h. D 1603, Standard Test Method for Carbon Black in Olefin Plastics.
 - D 3895, Test Method for Oxidative Induction Time of Polyolefins by Thermal Analysis.
 - D 4218, Test Method for Determination of Carbon Black Content in Polyethylene Compounds by the Muffle-Furnace Technique.

- k. D 4437, Standard Practice for Determining the Integrity of Field Seams Used in Joining Flexible Polymeric Sheet Geomembranes.
- D 4833, Test Method for Index Puncture Resistance of Geotextiles, Geomembranes and Related Products.
- D 5199, Standard Test Method for Measuring Nominal Thickness of Smooth Geomembranes.
- D 5397, Standard Test Method for Evaluation of Stress Crack Resistance of Polyolefins using Notched Constant Tensile Load Test.
- D 5596, Standard Practice for Microscopical Examination of Pigment Dispersion in Plastic Compounds.
- p. D 5641, Standard Practice for Geomembrane Seam Evaluation by Vacuum Chamber.
- q. D 5721, Practice for Air-Oven Aging of Polyolefin Geomembranes.
- r. D 5820, Test Method for Air Testing.
- D 5885, Test Method for Oxidative Induction Time of Polyolefin Geosynthetics by High Pressure Differential Scanning Calorimetry.
- D 5994, Standard Test Method for Measuring Nominal Thickness of Textured Geomembranes
- D 6365, Standard Practice for the Nondestructive Testing of Geomembrane Seams using The Spark Test
- Geosynthetic Research Institute (GRI):
 - GRI GM 6, Pressurized Air Channel Test for Dual Seamed Geomembranes
 - b. GRI GM 9, Cold Weather Seaming of Geomembranes
 - GRI GM 10, Specification for Stress Crack Resistance of HDPE Geomembrane Sheet
 - GRI GM 13, Test Properties, Testing Frequency and Recommended Warranty for High Density Polyethylene (HDPE) Smooth and Textured Geomembranes
 - e. GRI GM 14, Test Frequencies for Destructive SeamTesting

1.04 SUBMITTALS

- A. Submit the following to the Engineer or Owner, for review and approval, within a reasonable time so as to expedite shipment or installation of the Geomembrane:
 - Documentation of manufacturer's qualifications as specified in subsection 1.05A of this Section.
 - Manufacturer's Quality Control program manual or descriptive documentation.
 - A material properties sheet, including at a minimum all properties specified in GRI GM 13, including test methods used.
 - Sample of the material.

^{1:05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

- Documentation of Installer's qualifications, as specified below and in subsection 1.05B of this Section.
 - a. Submit a list of at least ten completed facilities. For each name and type of facility; its location; the date of installation; number of contact at the facility; type and thickness of geomembrane and; surface area of the installed geomembrane.
 - Submit resumes or qualifications of the Installation Supervisor, Master Seamer and Technicians to be assigned to this project.
 - c. Quality Control Program.
- Example Material Warranty and Liner Installation Warranty complying with subsections 1.07 and 1.08 of this Section.
- Resin Supplier's name, resin production plant identification, resin brand name and number, production date of the resin, resin Manufacturer's quality control certificates, and certification that the properties of the resin meet the requirements
- B. Shop Drawings
 - Submit copies of shop drawings for engineer's approval within a reasonable time so as not to delay the start of geomembrane installation. Shop drawings shall show the proposed panel layout identifying seams and details. Seams should generally follow direction of the slope. Butt seams or roll-end seams should not occur on a slope unless approved by the Owner's Representative. Butt seams on a slope, if allowed, should be staggered.
 - Placement of geomembrane will not be allowed to proceed until Owner's Representative has received and approved the shop drawings.
- C. Additional Submittals (In-Progress and at Completion)
 - Manufacturer's warranty (refer to subsection 1.08).
 - Geomembrane installation warranty (refer to subsection 1.09).
 - Daily written acceptance of subgrade surface (refer to subsection 3.01.C).
 - Low-temperature seaming procedures if applicable (refer to subsection 3.03.A)
 - Prequalification test seam samples (refer to subsection 3.05.A.6).
 - Field seam non-destructive test results (refer to subsection 3.05.B.1).
 - Field seam destructive test results (refer to subsection 3.05.C.6).
 - Daily field installation reports (refer to subsection 3.05.G).
 - 9. Installation record drawing, as discussed in subsection 3.05.G).

1.05 QUALITY CONTOL

A. Manufacturer's Qualifications:

The manufacturer of geomembrane of the type specified or similar product shall have at least five years experience in the manufacture of such geomembrane. In addition, the geomembrane manufacturer shall have manufactured at least

10,000,000 sq. ft. of the specified type of geomembrane or similar product during the last five years.

- B. Installer's Qualifications:
 - 1 The Geomembrane Installer shall be the Manufacturer, approved Manufacturer's Installer or a contractor approved by the Owner's Representative to install the geomembrane.
 - 2 The Geomembrane Installer shall have at least three years experience in the installation of the specified geomembrane or similar. The Geomembrane Installer shall have installed at least 10 projects involving a total of 5,000,000 sq. ft. of the specified type of geomembrane or similar during the last three years.
 - 3 Installation shall be performed under the direction of a field Installation Supervisor who shall be responsible throughout the geomembrane installation, for geomembrane panel layout, seaming, patching, testing, repairs, and all other activities of the Geomembrane Installer. The Field Installation Supervisor shall have installed or supervised the installation and seaming of a minimum of 10 projects involving a total of 5,000,000 sq. ft. of geomembrane of the type specified or similar product.
 - 4 Seaming shall be performed under the direction of a Master Seamer (who may also be the Field Installation Supervisor or Crew Foreman) who has seamed a minimum of 3,000,000 sq. ft. of geomembrane of the type specified or similar product, using the same type of seaming apparatus to be used in the current project. The Field Installation Supervisor and/or Master Seamer shall be present whenever seaming, patching, other welding operations, and testing is performed.
 - 5 All seaming, patching, other welding operations, and testing shall be performed by qualified technicians employed by the Geomembrane Installer.

1.06 DELIVERY, STORAGE AND HANDLING

- A. Each roll of geomembrane delivered to the site shall be labeled by the manufacturer. The label shall be firmly affixed and shall clearly state the manufacturer's name, product identification, material thickness, roll number, roll dimensions and roll weight.
- B. Geomembrane shall be protected from mud, dirt, dust, puncture, cutting or any other damaging or deleterious conditions.
- C. Rolls shall be stored away from high traffic areas. Continuously and uniformly support rolls on a smooth, level prepared surface.
- D. Rolls shall not be stacked more than three high.

1.07 PROJECT CONDITIONS

Geomembrane shall not be installed in the presence of standing water, while precipitation is occurring, during excessive winds, or when material temperatures are outside the limits specified in Section 3.03.

^{1/05}jobs/05s3004A/Gypsum Stacking/JDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

1.08 MATERIAL WARRANTY

As required by specification, or as required in GRI GM 13 (attachment A)

1.09 GEOMEMBRANE INSTALLATION WARRANTY

The Geomembrane Installer shall guarantee the geomembrane installation against defects in the installation and workmanship for 1 year commencing with the date of final acceptance.

1.10 GEOMEMBRANE PRE-CONSTRUCTION MEETING

- A. Geomembrane Pre-Construction Meeting shall be held at the site prior to installation of the geomembrane. At a minimum, the meeting shall be attended by the Geomembrane Installer, Owner, Owner's representative (Engineer and/or CQA Firm), and the General Contractor.
- B. Topics for this meeting shall include:
 - 1. Responsibilities of each party.
 - Lines of authority and communication. Resolution of any project document ambiguity.
 - Methods for documenting, reporting and distributing documents and reports.
 - Procedures for packaging and storing archive samples.
 - Review of time schedule for all installation and testing.
 - Review of panel layout and numbering systems for panels and seams including details for marking on geomembrane.
 - Procedures and responsibilities for preparation and submission of as-built panel and seam drawings.
 - Temperature and weather limitations. Installation procedures for adverse weather conditions. Defining acceptable subgrade, geomembrane, or ambient moisture and temperature conditions for working during liner installation.
 - Subgrade conditions, dewatering responsibilities and subgrade maintenance plan.
 - Deployment techniques including allowable subgrade for the geomembrane.
 - Plan for controlling expansion/contraction and wrinkling of the geomembrane.
 - 12. Covering of the geomembrane and cover soil placement.
 - Measurement and payment schedules.
 - Health and safety.
- C. The meeting shall be documented by the Owner's Representative and minutes shall be transmitted to all parties.

E/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

PART 2. PRODUCTS

2.01 SOURCE QUALITY CONTROL

Manufacturing Quality Control

- A. The test methods and frequencies used by the manufacturer for quality control/quality assurance of the above geomembrane prior to delivery, shall be in accordance with GRI GM 13, or modified as required for project specific conditions.
- B. The manufacturer's geomembrane quality control certifications, including results of quality control testing of the products, as specified in subsection 2.01.C of this Section, must be supplied to the Owner's Representative. The certification shall be signed by a responsible party employed by the manufacturer, such as the QA/QC Manager, Production Manager, or Technical Services Manager. Certifications shall include lot and roll numbers and corresponding shipping information.
- C. The Manufacturer will provide Certification that the geomembrane and welding rod supplied for the project have the same base resin and material properties.

2.02 GEOMEMBRANE

- A. The geomembrane shall consist of new, first quality products designed and manufactured specifically for the purpose of this work which shall have been satisfactorily demonstrated by prior testing to be suitable and durable for such purposes. The geomembrane rolls shall be seamless, high density polyethylene (HDPE- Density >0.94g/cm) containing no plasticizers, fillers or extenders and shall be free of holes, blisters or contaminants, and leak free verified by 100% in line spark or equivalent testing. The geomembrane shall be supplied as a continuous sheet with no factory seams in rolls. The geomembrane will meet the property requirements as shown in Table A. (GRI GM 13)
- B. Material shall be reviewed for conformance to the project specifications by the Owner's Representative
- C. The geomembrane seams shall meet the property requirements as shown in Table 2, (Attachment B).

PART 3. EXECUTION

3.01 SUBGRADE PREPARATION

A. Geomembrane installed over geosynthetic clay liner (GCL).

The area of GCL to be covered with geomembrane shall be jointly inspected daily with the Owner's Representative before commencing geomembrane installation for the day, and the condition of the GCL shall be continuously observed as geomembrane installation progresses. Rocks, stones, sticks, sharp objects and debris of any kind shall be removed from the surface of the GCL. The Owner's Representative shall be notified of any discontinuities, premature hydration, or

otherwise defective GCL. Geomembrane shall not be placed over suspect areas until they have been repaired to the satisfaction of the Owner's Representative. The Installation Supervisor shall certify daily in writing that the GCL surface was acceptable at the time of geomembrane installation.

B. Geomembrane installed over cushion dirt.

The area of cushion dirt to be covered with geomembrane shall be prepared in accordance with the Section 02200 – Earthwork. The surface shall be smooth and free of ruts and holes, rocks, stones, sticks, sharp objects and debris of any kind.

- C. The Geomembrane installer shall provide daily written acceptance for the surface to be covered by the geomembrane in that day's operations. The surface shall be maintained in a manner, during geomembrane installation, to ensure subgrade suitability.
- D. All subgrade damaged by construction equipment and deemed unsuitable by the Owner's Representative for geomembrane deployment shall be repaired prior to placement of the geomembrane. All repairs shall be reviewed by the Owner's Representative and approved by the Geomembrane Installer. This damage, repair, and the responsibilities of the contractor and Geomembrane Installer shall be defined in the preconstruction meeting.

3.02 GEOMEMBRANE PLACEMENT

- A. No geomembrane shall be deployed until the applicable certifications and quality control certificates listed in subsection 1.04 of this Section are submitted to and approved by the Owner's Representative. Should geomembrane material be deployed prior to approval by the Owner's Representative it will be at the sole risk of the Geomembrane Installer and/or Contractor. If the material does not meet project specifications it shall be removed from the work area at no cost to the owner.
- B. The geomembrane shall be installed to the limits shown on the project drawings and essentially as shown on approved panel layout drawings.
- C. No geomembrane material shall be unrolled and deployed if the material temperatures are lower than 0 degrees C (32 degrees F). Temperature limitations should be defined in the preconstruction meeting. Typically, only the quantity of geomembrane that will be anchored and seamed together in one day should be deployed.
- D. No vehicular traffic shall travel on the geomembrane other than an approved low ground pressure All Terrain Vehicle or equivalent.
- E. Sand bags or equivalent ballast shall be used as necessary to temporarily hold the geomembrane material in position under the foreseeable and reasonably - expected wind conditions. Sand bag material shall be sufficiently close- knit to prevent soil fines from working through the bags and discharging on the geomembrane.
- F. Geomembrane placement shall not be done if moisture prevents proper subgrade preparation, panel placement, or panel seaming. Moisture limitations 02800-7

^{1/05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Speci/S02800_HDPE Geomembrane.doc

should be defined in the preconstruction meeting.

- G. Damaged panels or portions of the damaged panels which have been rejected shall be marked and their removal from the work area recorded.
- H. The geomembrane shall not be allowed to "bridge over" voids or low areas in the subgrade. In these areas, the subgrade shall be prepared to allow the geomembrane to rest in intimate contact with the subgrade.
- Wrinkles caused by panel placement or thermal expansion should be minimized in accordance with section 1.10 B. 11.
- J. Considerations on Site Geometry: In general, seams shall be oriented parallel to the line of the maximum slope. In corners and odd shaped geometric locations, the total length of field seams shall be minimized. Seams shall not be located at low points in the subgrade.
- K. Overlapping: The panels shall be overlapped prior to seaming to whatever extent is necessary to effect a good weld and allow for proper testing. In no case shall this overlap be less than 75mm (3 in.).

3.03 SEAMING PROCEDURES

- A. Cold weather installations should follow guidelines as outlined in GRI GM9.
- B. No geomembrane material shall be seamed when liner temperatures are less than 0 degrees C (32 degrees F).
- C. No geomembrane material shall be seamed when the sheet temperature is above 75 degrees C (170 degrees F) as measured by an infrared thermometer or surface thermocouple.
- D. Seaming shall primarily be performed using automatic fusion welding equipment and techniques. Extrusion welding shall be used where fusion welding is not possible such as at pipe penetrations, patches, repairs and short (less than a roll width) runs of seams.
- E. Fishmouths or excessive wrinkles at the seam overlaps, shall be minimized and when necessary cut along the ridge of the wrinkles back into the panel so as to effect a flat overlap. The cut shall be terminated with a keyhole cut (nominal 10 mm (1/2 in) diameter hole) so as to minimize crack/tear propagation. The overlay shall subsequently be seamed. The key hole cut shall be patched with an oval or round patch of the same base geomembrane material extending a minimum of 150 mm (6 in.) beyond the cut in all directions.

3.04 PIPE AND STRUCTURE PENETRATION SEALING SYSTEM

- Provide penetration sealing system as shown in the Project Drawings.
- B. Penetrations shall be constructed from the base geomembrane material, flat stock, prefabricated boots and accessories as shown on the Project Drawings. The pre-fabricated or field fabricated assembly shall be field welded to the geomembrane as shown on the Project Drawings so as to prevent leakage. This assembly shall be tested as outlined in section 3.05.B. Alternatively, where field non destructive testing can not be performed, attachments will be field spark tested by standard holiday leak detectors in accordance with ASTM 6365 Spark testing should be done in areas where both air pressure testing and vacuum testing are not possible.
 - Equipment for Spark testing shall be comprised of but not limited to: A 02800-8

^{1/05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE_Geomembrane.doc

hand held holiday spark tester and conductive wand that generates a high voltage.

- 2. The testing activities shall be performed by the Geomembrane Installer by placing an electrically conductive tape or wire beneath the seam prior to welding. A trial seam containing a non welded segment shall be subject to a calibration test to ensure that such a defect (non welded segment) will be identified under the planned machine settings and procedures. Upon completion of the weld, enable the spark tester and hold approximately 25mm (1 in) above the weld moving slowly over the entire length of the weld in accordance with ASTM 6365. If there is no spark the weld is considered to be leak free.
- A spark indicates a hole in the seam. The faulty area shall be located, repaired and retested by the Geomembrane Installer.
- Care should be taken if flammable gases are present in the area to be tested.

3.05 FIELD QUALITY CONTROL

The Owner's Representative shall be notified prior to all pre qualification and production welding and testing, or as agreed upon in the pre construction meeting.

- A. Prequalification Test Seams
 - Test seams shall be prepared and tested by the Geomembrane Installer to verify that seaming parameters (speed, temperature and pressure of welding equipment) are adequate.
 - 2. Test seams shall be made by each welding technician and tested in accordance with ASTM D 4437 at the beginning of each seaming period. Test seaming shall be performed under the same conditions and with the same equipment and operator combination as production seaming. The test seam shall be approximately 3.3 meters (10 feet) long for fusion welding and 1 meter (3 feet) long for extrusion welding with the seam centered lengthwise. At a minimum, tests seams should be made by each technician 1 time every 4–6 hours; additional tests may be required with changes in environmental conditions.
 - 3. Two 25 mm (1 in) wide specimens shall be die-cut by the Geomembrane Installer from each end of the test seam. These specimens shall be tested by the Geomembrane Installer using a field tensiometer testing both tracks for peel strength and also for shear strength. Each specimen shall fail in the parent material and not in the weld, "Film Tear Bond"(F.T.D. failure). Seam separation equal to or greater than 10% of the track width shall be considered a failing test.
 - The minimum acceptable seam strength values to be obtained for all specimens tested are listed in Subsection 3.05.C.4 of this Section. All four specimens shall pass for the test seam to be a passing seam.
 - 5. If a test seam fails, an additional test seam shall be immediately conducted. If the additional test seam fails, the seaming apparatus shall be rejected and not used for production seaming until the deficiencies are

^{1/05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

corrected and a successful test seam can be produced.

- 6. A sample from each test seam shall be labeled. The label shall indicate the date, geomembrane temperature, number of the seaming unit, technician performing the test seam and pass or fail description. The sample shall then be given to the Owner's Representative for archiving.
- B. Field Seam Non-destructive Testing
 - All field seams shall be non-destructively tested by the Geomembrane Installer over the full seam length before the seams are covered. Each seam shall be numbered or otherwise designated. The location, date, test unit, name of tester and outcome of all non-destructive testing shall be recorded and submitted to the Owner's Representative.
 - Testing should be done as the seaming work progresses, not at the completion of all field seaming. All defects found during testing shall be numbered and marked immediately after detection. All defects found should be repaired, retested and remarked to indicate acceptable completion of the repair.
 - Non-destructive testing shall be performed using vacuum box, air pressure or spark testing equipment.
 - Non-destructive tests shall be performed by experienced technicians familiar with the specified test methods. The Geomembrane Installer shall demonstrate to the Owner's Representative all test methods to verify the test procedures are valid.
 - Extrusion seams shall be vacuum box tested by the Geomembrane Installer in accordance with ASTM D 4437 and ASTM D 5641 with the following equipment and procedures:
 - a. Equipment for testing extrusion seams shall be comprised of but not limited to: a vacuum box assembly consisting of a rigid housing, a transparent viewing window, a soft rubber gasket attached to the base, port hole or valve assembly and a vacuum gauge; a vacuum pump assembly equipped with a pressure controller and pipe connections; a rubber pressure/vacuum hose with fittings and connections; a plastic bucket; wide paint brush or mop; and a soapy solution.
 - The vacuum pump shall be charged and the tank pressure adjusted to approximately 35 kPa (5 psig).
 - c. The Geomembrane Installer shall create a leak tight seal between the gasket and geomembrane interface by wetting a strip of geomembrane approximately 0.3m (12 in) by 1.2m (48 in) (length and width of box) with a soapy solution, placing the box over the wetted area, and then compressing the box against the geomembrane. The Geomembrane Installer shall then close the bleed valve, open the vacuum valve, maintain initial pressure of approximately 35 kPa (5 psig) for approximately 5 seconds. The geomembrane should be continuously examined through the viewing window for the presence of soap bubbles, indicating a leak. If no bubbles appear after 5 seconds, the area shall be 02800-10

E/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800 HDPE Geomembrane.doc

considered leak free. The box shall be depressurized and moved over the next adjoining area with an appropriate overlap and the process repeated.

- All areas where soap bubbles appear shall be marked, repaired and then retested.
- e. At locations where seams cannot be non destructively tested, such as pipe penetrations, alternate nondestructive spark testing (as outlined in section 3.04.B) or equivalent should be substituted.
- f. All seams that are vacuum tested shall be marked with the date tested, the name of the technician performing the test and the results of the test.
- Double Fusion seams with an enclosed channel shall be air pressure tested by the Geomembrane Installer in accordance with ASTM D 5820 and ASTM D 4437 and the following equipment and procedures:
 - a. Equipment for testing double fusion seams shall be comprised of but not limited to: an air pump equipped with a pressure gauge capable of generating and sustaining a pressure of 210 kPa (30 psig), mounted on a cushion to protect the geomembrane; and a manometer equipped with a sharp hollow needle or other approved pressure feed device.
 - b. The Testing activities shall be performed by the Geomembrane Installer. Both ends of the seam to be tested shall be sealed and a needle or other approved pressure feed device inserted into the tunnel created by the double wedge fusion weld. The air pump shall be adjusted to a pressure of 210 kPa (30 psig), and the valve closed,. Allow 2 minutes for the injected air to come to equilibrium in the channel, and sustain pressure for 5 minutes. If pressure loss does not exceed 28 kPa (4 psig) after this five minute period the seam shall be considered leak tight. Release pressure from the opposite end verifying pressure drop on needle to ensure testing of the entire seam. The needle or other approved pressure feed device shall be removed and the feed hole sealed.
 - c. If loss of pressure exceeds 28 kPa (4 psig) during the testing period or pressure does not stabilize, the faulty area shall be located, repaired and retested by the Geomembrane Installer.
 - Results of the pressure testing shall be recorded on the liner at the seam tested and on a pressure testing record.
- C. Destructive Field Seam Testing
 - 1. One destructive test sample per 150 linear m (500 linear ft) seam length or another predetermined length in accordance with GRI GM 14 shall be taken by the Geomembrane Installer from a location specified by the Owner's Representative. The Geomembrane Installer shall not be informed in advance of the sample location. In order to obtain test results prior to completion of geomembrane installation, samples shall be cut by the Geomembrane Installer as directed by the Owner's Representative as seaming progresses.

02800-11

í

^{1/05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

- 2. All field samples shall be marked with their sample number and seam number. The sample number, date, time, location, and seam number shall be recorded. The Geomembrane Installer shall repair all holes in the geomembrane resulting from obtaining the seam samples. All patches shall be vacuum box tested or spark tested. If a patch cannot be permanently installed over the test location the same day of sample collection, a temporary patch shall be tack welded or hot air welded over the opening until a permanent patch can be affixed.
- 3. The destructive sample size shall be 300 mm (12 in) wide by I m (36 in) long with the seam centered lengthwise. The sample shall be cut into three equal sections and distributed as follows: one section given to the Owner's Representative as an archive sample; one section given to the Owner's Representative for laboratory testing as specified in paragraph 5 below; and one section retained by the Geomembrane Installer for field testing as specified in paragraph 4 below.
- 4. For field testing, the Geomembrane Installer shall cut 10 identical 25 mm (1 in) wide replicate specimens from his sample. The Geomembrane Installer shall test five specimens for seam shear strength and five for peel strength. Peel tests will be performed on both inside and outside weld tracks. To be acceptable, 4 of 5 test specimens must pass the stated criteria in section 2.02 with less than 10% separation. If 4 of 5 specimens pass, the sample qualifies for testing by the testing laboratory if required.
- If independent seam testing is required by the specifications it shall be conducted in accordance with ASTM 5820 or ASTM D4437 or GRI GM 6.
- Reports of the results of examinations and testing shall be prepared and submitted to the Owner's Representative.
- 7. For field seams, if a laboratory test fails, that shall be considered as an indicator of the possible inadequacy of the entire seamed length corresponding to the test sample. Additional destructive test portions shall then be taken by the Geomembrane Installer at locations indicated by the Engineer, typically 3 m (10 ft) on either side of the failed sample and laboratory seem tests shall be performed. Passing tests shall be an indicator of non-adequate seams and all seams represented by the destructive test location shall be repaired with a cap-strip extrusion welded to all sides of the capped area. All cap-strip seams shall be non-destructively vacuum box tested until adequacy of the seams is achieved. Cap strip seams exceeding 50 M in length (150 FT) shall be destructively tested.
- D. Identification of Defects
 - Panels and seams shall be inspected by the Installer and Owner's Representative during and after panel deployment to identify all defects, including holes, blisters, undispersed raw materials and signs of contamination by foreign matter.
- E. Evaluation of Defects: Each suspect location on the liner (both in geomembrane 02800-12

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

seam and non-seam areas) shall be non-destructively tested using one of the methods described in Section 3.05.B. Each location which fails non-destructive testing shall be marked, numbered, measured and posted on the daily "installation" drawings and subsequently repaired.

- If a destructive sample fails the field or laboratory test, the Geomembrane Installer shall repair the seam between the two nearest passed locations on both sides of the failed destructive sample location.
- Defective seams, tears or holes shall be repaired by reseaming or applying a extrusion welded cap strip.
- Reseaming may consist of either:
 - Removing the defective weld area and rewelding the parent material using the original welding equipment; or
 - Reseaming by extrusion welding along the overlap at the outside seam edge left by the fusion welding process.
- Blisters, larger holes, and contamination by foreign matter shall be repaired by patches and/or extrusion weld beads as required. Each patch shall extend a minimum of 150 mm (6 in) beyond all edges of the defects.
- All repairs shall be measured, located and recorded.
- F. Verification of Repairs on Seams: Each repair shall be non-destructively tested using either vacuum box or spark testing methods. Tests which pass the nondestructive test shall be taken as an indication of a successful repair. Failed tests shall be reseamed and retested until a passing test results. The number, date, location, technician and test outcome of each patch shall be recorded.
- G. Daily Field Installation Reports: At the beginning of each day's work, the Installer shall provide the Engineer with daily reports for all work accomplished on the previous work day. Reports shall include the following:
 - 1. Total amount and location of geomembrane placed;
 - Total length and location of seams completed, name of technicians doing seaming and welding unit numbers;
 - Drawings of the previous day's installed geomembrane showing panel numbers, seam numbers and locations of non-destructive and destructive testing;
 - Results of pre-qualification test seams;
 - 5. Results of non-destructive testing; and
 - 6. Results of vacuum testing of repairs.
- H. Destructive test results shall be reported prior to covering of liner or within 48 hours.

3.06 LINER ACCEPTANCE

- A. Geomembrane liner will be accepted by the Owner's Representative when:
 - The entire installation is finished or an agreed upon subsection of the installation is finished;

^{1/05}jobi/05s3004A/Gypuum Stacking/IDNR Dam Safety Permit Application/Specs/S02800_HDPE Geomembrane.doc

- 2. All Installer's QC documentation is completed and submitted to the owner
- Verification of the adequacy of all field seams and repairs and associated geomembrane testing is complete.

3.07 ANCHOR TRENCH

Construct as specified on the project drawings.

3.08 DISPOSAL OF SCRAP MATERIALS

A. On completion of installation, the Geomembrane Installer shall dispose of all trash and scrap material in a location approved by the Owner, remove equipment used in connection with the work herein, and shall leave the premises in a neat acceptable manner. No scrap material shall be allowed to remain on the geomembrane surface.

PART 4. GRI GM13 SPECIFICATIONS

Geosynthetics Research Institute (GRI) Test Method GM13 – "Test Methods, Test Properties and Testing Frequency for High Density Polyethylene (HDPE) Smooth and Textured Geomembranes", Revision 8, Dated July 10, 2006.

ATTACHMENT A:

Minimum Average Weld Properties for Smooth and Textured HDPE Geomembranes (English units)								
Property	Test Method	30 mil	40 mil	50 mil	60 mil	80 mil	100 mil	120 mil
Peel strength (fusion & extrusion) lb/in. Shear strength (fusion & extrusion) lb/in.	ASTM 4437 ASTM 4437	39 60	52 80	65 100	78 120	104 160	130 200	156 239

END OF SECTION 02800

DIVISION 2 - SITE WORK Section 02936 - Topsoil, Seeding, and Mulching

PART 1. GENERAL

1.01 DESCRIPTION

This section pertains to seeding and placing mulch or erosion control blanket over seeded areas.

1.02 RELATED SECTIONS

- A. Specified elsewhere:
 - 1. 02200 Earthwork

1.03 REFERENCES

The following reference or cited portions thereof, current at date of bidding documents unless otherwise specified, governs the work.

A. Illinois Department of Transportation (IDOT): Standard Specifications for Road and Bridge Construction, adopted January 1, 2007.

1.04 SPECIFICATIONS

- A. Work shall conform to the applicable requirements of Sections 250 and 251 of Standard Specifications for Road and Bridge Construction and to the requirements hereinafter specified.
- Exceptions: All references in the IDOT specifications to methods of measurement and payment shall not apply.

1.05 WARRANTY

A. Warranty for one (1) year plus one growing season from date of substantial completion shall be provided.

PART 2. PRODUCTS

2.01 MATERIALS

A. Seed: Seed shall conform to Article 1081.04 of the IDOT Standard Specifications. The composition of the Ameren Energy Resources Generating hay seeding mix shall

^{1:05}jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02936_Seeding and Mulching.doc

be as follows:

Seed Type	Pounds/Acre				
Vernal Alfalfa	12				
Wrangler Alfalfa	8				
Medium Red Clover	6				
Timothy	4				

- B. Mulch Material and Erosion Control Blanket: Mulch material shall conform to Article 1081.06 and the excelsior blanket/knitted straw mat shall conform to Article 1081.10 of the IDOT Standard Specifications.
- C. Fertilizer and agricultural ground limestone will not be permitted.

PART 3. EXECUTION

3.01 CONSTRUCTION

- A. Seed bed preparation and seeding methods shall conform to Section 250 of the IDOT Specifications. Seeding of areas disturbed by construction activities after September 30, 2008 may be deferred until Spring 2009.
- B. Seed shall be applied to the perimeter berm ditch, to disturbed portions of the perimeter berm, and to all disturbed earth surfaces outside of the existing perimeter berm. IDOT seeding mixture 7 shall be used on stockpiles. IDOT seeding mixture 1A shall be used on the gypsum stack perimeter earthen berm, the recycle pond dam embankment and on slopes that are 4H:1V or steeper. The Ameren hay seed mix shall be used on slopes flatter that 4H:1V.
- C. Application rates for IDOT seed mixtures shall be as specified in Section 250 of the IDOT Specifications. The application rate for the Ameren Energy Resources Generating's seed mix shall be as specified in the Ameren Energy Resources Generating's hay seeding mix.
- D. Seeded areas shall be mulched in accordance with Article 251.03. The Contractor may use either Method 2 or Method 3.

3.02 MAINTENANCE OF COMPLETED WORK

A. All areas seeded by the Contractor shall be maintained by the Contractor during the period between completion of such work and final completion and acceptance of the Contractor's work by the Owner. This maintenance shall be such that the completed work, at time of acceptance, complies in all respects with the requirements herein specified.

02936-2

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S02936_Seeding and Mulching.doc

B. The areas seeded will be required to germinate. If the seed does not germinate, the Contractor will be required to regrade and reseed at no additional cost.

END OF SECTION 02936

DIVISION 3 - CONCRETE Section 03100 - Concrete Formwork

PART 1. GENERAL

1.01 WORK INCLUDES

A. The complete installation of the formwork for cast-in-place concrete, with shoring, bracing and anchorage, openings for other work, form accessories, form stripping.

1.02 RELATED SECTIONS

- A. Section 03200 Concrete Reinforcement.
- B. Section 03300 Cast-In-Place Concrete.
- C. Section 03400 Concrete Embedment Liner.

1.03 REFERENCES

- A. ACI 347 Recommended Practice For Concrete Formwork.
- B. ACI 301 Specifications For Structural Concrete For Buildings.

1.04 DESIGN REQUIREMENTS

A. Design, engineer and construct formwork, shoring and bracing to conform to design and code requirements; resultant concrete to conform to required shape, line and dimension.

1.05 QUALITY ASSURANCE

Perform Work in accordance with ACI 347 and 301.

1.06 REGULATORY REQUIREMENTS

 Conform to applicable code for design, fabrication, erection and removal of formwork.

1.07 DELIVERY, STORAGE, AND HANDLING

 Store off ground in ventilated and protected manner to prevent deterioration from moisture.

1.08 COORDINATION

- A. Coordinate this Section with other Sections of work which require attachment of components of formwork.
- B. If formwork is placed which results in insufficient concrete cover over reinforcement, request instructions from Owner's Representative before proceeding.

PART 2. PRODUCTS

2.01 WOOD FORM MATERIALS

- A. Softwood Plywood: 3/4 in. PS 1-83 "B-B" (concrete form) plywood, Class I, exterior grade or better, mill-oiled and edge sealed with each piece bearing legible inspection trademark.
- B. Architectural Plywood: 3/4 in. PS 1-83 "B-B" plyform, Class I, with High Density smooth overlay, 1 surface, edge sealed with each piece bearing legible inspection trademark.

2.02 MANUFACTURERS - PREFABRICATED FORMS

- Weyerhauser Concrete Form.
- B. Georgia Pacific, G-P Exterior Soft Wood Plywood Product.
- C. Plywood and Door Corporation's Finn-Form.

2.03 PREFABRICATED FORMS

- A. Preformed Steel Forms: Minimum 16 gage matched, tight fitting, stiffened to support weight of concrete without deflection detrimental to tolerances and appearance of finished surfaces.
- B. Glass Fiber Fabric Reinforced Plastic Forms: Matched, tight fitting, stiffened to support weight of concrete without deflection detrimental to tolerances and appearance of finished concrete surfaces.

2.04 FORMWORK ACCESSORIES

A. Form Ties: Snap-off type, galvanized metal, adjustable length, 1 in. back break dimension, free of defects that could leave holes larger than 1 in. in concrete surface; Dayton-Sure Grip snap-in-form ties, as manufactured by Dayton Superior

Corp., Symons Ties as manufactured by Symons Corporation, Snap-Tys as manufactured by Richmond Corporation. Ties shall be removed after forms are removed, and holes filled with mortar that matches the adjacent surfaces.

- B. Form Release Agent: Colorless mineral oil which will not stain concrete, or absorb moisture; by Magic Kote manufactured by Symons Manufacturing Co., Form Coat manufactured by Concrete Services Co., Formcel manufactured by Lambert Corp.
- C. Corners: Chamfered, wood strip type; 3/4 x 3/4 in. size on all exterior corners, 3 x 3 in. size where shown on the drawings; maximum possible lengths.
- D. Nails, Spikes, Lag Bolts, Through Bolts, Anchorages: Sized as required, of sufficient strength and character to maintain formwork in place while placing concrete.
- E. Concrete Embedment Liner, where required, shall be installed in accordance with Section 03400 – Concrete Embedment Liner.

PART 3. EXECUTION

3.01 EXAMINATION

A. Verify lines, levels and centers before proceeding with formwork. Ensure that dimensions agree with drawings.

3.02 EARTH FORMS

Earth forms are not permitted, except for footings.

3.03 ERECTION - FORMWORK

- A. Erect formwork, shoring and bracing to achieve design requirements, in accordance with requirements of ACI 301. Metal forms shall be installed in strict accordance with manufacturer's directions and specifications.
- B. Provide bracing to ensure stability of formwork. Shore or strengthen formwork subject to overstressing by construction loads.
- C. Arrange and assemble formwork to permit dismantling and stripping. Do not damage concrete during stripping. Permit removal of remaining principal shores.
- D. Align joints and make watertight. Keep form joints to a minimum.
- E. Obtain approval before framing openings in structural members which are not indicated on drawings.

^{1:05}job/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S03100_Concrete Formwork.doc

3.04 APPLICATION - FORM RELEASE AGENT

- A. Apply form release agent on formwork in accordance with manufacturer's recommendations.
- B. Apply prior to placement of reinforcing steel, anchoring devices, and embedded items.
- C. Do not apply form release agent where concrete surfaces will receive special finishes or applied coverings which are affected by agent.

3.05 INSERTS, EMBEDDED PARTS, AND OPENINGS

- A. Provide formed openings where required for items to be embedded in or passing through concrete work.
- B. Locate and set in place items which will be cast directly into concrete.
- C. Coordinate work of other Sections in forming and placing openings, slots, reglets, recesses, chases, sleeves, bolts, anchors, and other inserts.
- D. Install accessories in accordance with manufacturer's instructions, straight, level, and plumb. Ensure items are not disturbed during concrete placement.
- E. Provide temporary ports or openings in formwork where required to facilitate cleaning and inspection. Locate openings at bottom of forms to allow flushing water to drain.
- F. Close temporary openings with tight fitting panels, flush with inside face of forms, and neatly fitted so joints will not be apparent in exposed concrete surfaces.

3.06 FORM CLEANING

- Clean and remove foreign matter within forms as erection proceeds.
- B. Clean formed cavities of debris prior to placing concrete.
- C. Flush with water or use compressed air to remove remaining foreign matter. Ensure that water and debris drain to exterior through clean-out ports.
- D. During cold weather, remove ice and snow from within forms. Do not use de-icing salts or water to clean out forms. Use compressed air or other means to remove foreign matter.

3.07 FORMWORK TOLERANCES

Construct formwork to maintain tolerances required by ACI 301.

3.08 FIELD QUALITY CONTROL

- A. Inspect erected formwork, shoring, and bracing to ensure that work is in accordance with formwork design, and that supports, fastenings, wedges, ties, and items are secure.
- B. Do not reuse wood formwork more than three times for concrete surfaces to be exposed to view. Do no patch formwork.

3.09 FORM REMOVAL

- A. Do not remove forms or bracing until concrete has gained sufficient strength to carry its own weight and imposed loads.
- B. Loosen forms carefully. Do not wedge pry bars, hammers, or tools against finished concrete surfaces scheduled for exposure to view.
- C. Store removed forms in manner that surfaces to be in contact with fresh concrete will not be damaged. Discard damaged forms.

END OF SECTION 03100

DIVISION 3 - CONCRETE Section 03200 - Concrete Reinforcement

PART 1. GENERAL

- 1.01 WORK INCLUDES
 - A. The complete installation of the reinforcing steel bars and accessories for cast-in-place concrete.

1.02 RELATED SECTIONS

- A. Section 03100 Concrete Formwork.
- B. Section 03300 Cast-in-Place Concrete.

1.03 REFERENCES

- A. ACI 301 Structural Concrete for Buildings.
- B. ACI 318 Building Code Requirements For Reinforced Concrete.
- C. ACI SP-66 American Concrete Institute Detailing Manual.
- D. ASTM A615 Deformed and Plain Billet Steel Bars for Concrete Reinforcement.
- E. CRSI Concrete Reinforcing Steel Institute Manual of Practice.

1.04 SUBMITTALS

- Submit under provisions of Section 01010.
- B. Shop Drawings: Indicate bar sizes, spacings, locations, and quantities of reinforcing steel, and bending and cutting schedules. Contract drawings shall not be reproduced as the basis for shop drawings.
- C. Manufacturer's Certificate: Certify that products meet or exceed specified requirements.

1.05 QUALITY ASSURANCE

- A. Perform Work in accordance with CRSI Manual of Standard Practice.
- B. Submit certified copies of mill test report of reinforcement materials analysis.

1.06 COORDINATION

A. Coordinate with placement of formwork, formed openings and other work.

PART 2. PRODUCTS

2.01 REINFORCEMENT

A. Reinforcing Steel: ASTM A615, 60 ksi yield grade; deformed billet steel bars.

2.02 ACCESSORY MATERIALS

- A. Tie Wire: Minimum 16 gage, annealed steel wire, epoxy coated when used with epoxy-coated reinforcement.
- B. Chairs, Bolsters, Bar Supports, Spacers: Sized and shaped for strength and support of reinforcement during concrete placement conditions.
- C. Special Chairs, Bolsters, Bar Supports, Spacers Adjacent to Weather Exposed Concrete Surfaces: Plastic coated steel type; size and shape as required.

2.03 FABRICATION

- A. Fabricate concrete reinforcing in accordance with CRSI Manual of Standard Practice and ACI SP-66.
- B. Splice reinforcement on at locations indicated on drawings. Indicate location of splices on shop drawings.

PART 3. EXECUTION

3.01 PLACEMENT

- A. Place, support and secure reinforcement against displacement. Do not deviate from required position. Clean reinforcement of foreign particles or coatings.
- B. Accommodate placement of formed openings.
- C. Conform to ACI 318 code for concrete cover over reinforcement.

3.02 FIELD QUALITY CONTROL

A. Contractor shall notify the Owner's Representative at least 24 hrs. in advance of concrete placement. Placement of reinforcing shall occur in such sequence that the Owner's Representative has sufficient time to inspect the correctness of the reinforcing within the placement area. The Owner's Representative retains the right to require necessary revisions be made before concrete is placed.

END OF SECTION 03200

DIVISION 3 - CONCRETE Section 03300 - Cast-In-Place Concrete

PART 1. GENERAL

1.01 WORK INCLUDES

A. The complete installation of cast-in-place concrete structures, including joint sealants.

1.02 RELATED SECTIONS

- A. Section 03100 Concrete Formwork: Formwork and accessories.
- B Section 03200 Concrete Reinforcement.
- C. Section 03400 Concrete Embedment Liner

1.03 REFERENCES

- A. ACI 301 Structural Concrete for Buildings.
- B. ACI 302 Guide for Concrete Floor and Slab Construction.
- C. ACI 304 Recommended Practice for Measuring, Mixing, Transporting and Placing Concrete.
- D. ACI 305R Hot Weather Concreting.
- E. ACI 306R Cold Weather Concreting.
- F. ACI 308 Standard Practice for Curing Concrete.
- G. ACI 318 Building Code Requirements for Reinforced Concrete.
- H. ASTM C31 Concrete Test Specimens.
- I. ASTM C33 Concrete Aggregates.
- J. ASTM C94 Ready-Mixed Concrete.
- K. ASTM C150 Portland Cement.
- L. ASTM C260 Air Entraining Admixtures for Concrete.

03300-1

1/05jobi/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S03300_Cast-In-Place Concrete.doc

M. ASTM C494 - Chemical Admixtures for Concrete.

1.04 SUBMITTALS

A. Product Data: Provide data on joint devices, attachment accessories, admixtures.

1.05 QUALITY ASSURANCE

- Perform Work in accordance with ACI 301.
- B. Acquire cement and aggregate from same source for all work.
- C. Conform to ACI 305R when concreting during hot weather.
- D. Conform to ACI 306R when concreting during cold weather.

1.06 COORDINATION

A. Coordinate this Section with other Sections which require embedment of components in cast-in-place concrete.

1.07 PRODUCT DATA

- A. Submit proposed mix design to Owner's Representative for review prior to commencement of work. Identify source and provide material certificates for cement, fine and coarse aggregates. Provide recent laboratory gradation for fine and coarse aggregates and mix design information in accordance with ACI 301.
- B. Submit Construction joint plan.

PART 2. PRODUCTS

2.01 CONCRETE MATERIALS

- A. Cement: ASTM C150, Type I Normal Portland Type, Gray Color.
- B. Fine and Coarse Aggregates: ASTM C33.
- C. Water: Potable.

2.02 ADMIXTURES

A. Air Entrainment: ASTM C260.

03300-2

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S03300_Cast-In-Place Concrete.doc

- B. Chemical: ASTM C494. Maximum 0.05% Chloride Ion Contents.
- C. The use of calcium chloride in any concrete is not permitted.

2.03 ACCESSORIES

- A. Non-Shrink Grout: Premixed compound consisting of non-metallic aggregate, cement, water reducing and plasticizing agents; capable of developing minimum compressive strength of 2,400 psi in 48 hours and 7,000 psi in 28 days.
- B. Curing Compound: Dress and Seal No. 18 by L&M Construction Chemicals, MB-429 by Master Builders, or Sikagard Cure/Hard by the Sika Corporation.
- C. Epoxy Grouted Adhesive Anchors: Hilti, Red Head, Simpson, or Rawl.

2.04 CONCRETE MIX

- A. Mix concrete in accordance with ACI 304. Deliver concrete in accordance with ASTM C94.
- B. Select proportions for normal weight concrete in accordance with ACI 301.
- C. Provide normal weight concrete of the following characteristics:
 - Compressive strength at 28 days: 4,000 psi.
 - Slump: 4 in. A tolerance of up to 1 in. above the maximum shall be allowed for one batch in any five consecutive batches tested.
 - Water/cement ratios: 0.4 (max).
- D. Use accelerating admixtures in cold weather only when approved by Owner's Representative. Use of admixtures will not relax cold weather placement requirements.
- E. Use set-retarding admixtures during hot weather only when approved by Owner's Representative.
- F. Water-reducing admixtures may be used in all concrete except footings and in strict compliance with the manufacturer's directions.
- G. Add air-entraining agent to concrete mix for air content of 6% (± 1%).

PART 3. EXECUTION

3.01 EXAMINATION

- Verify requirements for concrete cover over reinforcement.
- B. Verify that anchors, seats, plates, reinforcement and other items to be cast into concrete are accurately placed, positioned securely, and will not cause hardship in placing concrete.

3.02 PLACING CONCRETE

- Place concrete in accordance with ACI 301.
- B. Notify Owner's Representative minimum of 24 hours prior to commencement of operations.
- C. Ensure reinforcement, inserts, and embedded parts are not disturbed during concrete placement.
- D. Maintain records of concrete placement. Record date, location, quantity, air temperature, and test samples taken.
- E. Place concrete continuously between predetermined expansion, control, and construction joints.
- F. When air temperature is between 80°F and 90°F, reduce the mixing and delivery time specified in ASTM C94 from 1-1/2 hours to 75 minutes. When the air temperature is above 90°F, reduce the mixing and delivery time to 60 minutes.
- G. Cold weather concreting. Comply with ACI 306 except as follows:
 - In freezing weather, provide suitable means for maintaining concrete temperature at a minimum of 70°F for three days, or 50°F for five days after placing.
 - Cooling of concrete to outside temperature: Not faster than 1° per hour for first day and 2° per hour thereafter until outside temperature is reached.
 - Maximum temperature of concrete produced with heated aggregated, heated water, or both, at any time during its production or transportation: 90°F.
 - Do not mix chemicals or other foreign materials in concrete to prevent freezing or to accelerate hardening of concrete, unless approved in writing by Owner's Representative.

03300-4

E/05jobs/05s3004A/Gypsum Stacking/JDNR Dam Safety Permit Application/Specs/S03300_Cast-In-Place Concrete.doc

- H. Hot weather concreting. Comply with ACI 305R.
 - ACI recommendations shall be observed when any combination of high air temperature, low relative humidity and wind velocity tend to impair the quality of fresh or hardened concrete.
 - Retarding and water reducing admixtures shall be approved in writing for each concrete mix design prior to placement.

3.03 CONCRETE FINISHING

- A. Provide exterior concrete formed surfaces to be left exposed with smooth rubbed finish in accord with ACI 301. All other formed surfaces shall have fins, projections and offsets removed.
- B. Provide Class A tolerances to exterior concrete slabs according to ACI 301.
 - 1. Broom finish all exterior slabs. Broom out all tool marks.
- C. Pitch slabs to drain.

3.04 CURING AND PROTECTION

- Immediately after placement, protect concrete from premature drying, excessively hot or cold temperatures, and mechanical injury.
- B. Maintain concrete with minimal moisture loss at relatively constant temperature for a period necessary for hydration of cement and hardening of concrete in accordance with ACI 308.
- C. Cure and protect finished concrete slabs in accordance with ACI 308.

3.05 FIELD QUALITY CONTROL

- A. Field inspection and testing will be performed in accordance with ACI 301 and under provisions of Section 01010, paragraph 8.0.
- B. Testing firm will take cylinders, perform slump and air entrainment tests in accordance with ACI 301.
- C. Provide free access to Work and cooperate with appointed firm.
- D. Submit proposed concrete mix design to Owner's Representative firm for review 14 days prior to commencement of Work.

E. Testing frequency shall be as specified in Section 01010, paragraph 8, except that one additional test cylinder will be taken during cold weather concreting, cured on job site under same conditions as concrete it represents.

3.06 PATCHING

- A. Defective Concrete: Concrete not conforming to required lines, details, dimensions, tolerances or specified requirements.
- B. Repair or replacement of defective concrete will be determined by Owner's Representative and performed by the Contractor at no additional cost to the project.
- C. Do not patch, fill, touch-up, repair, or replace exposed concrete except upon express direction of Owner's Representative for each individual area.

END OF SECTION 03300

DIVISION 3 - CONCRETE Section 03400 - Concrete Embedment Liner

PART 1. GENERAL

1.01 WORK INCLUDES

A. Specifications and guidelines for manufacturing and installing high-density polyethylene embedment liners.

1.02 RELATED SECTIONS

- A. Section 03100 Concrete Formwork.
- B. Section 03300 Cast-in-Place Concrete.

1.03 REFERENCES

A. American Society for Testing and Materials (ASTM)

- D 1505 Test Method for Density of Plastics by the Density-Gradient Technique
- 2. D 1603 Test Method for Carbon Black in Olefin Plastics
- D 5199 <u>Standard Test Method for Measuring Nominal Thickness of</u> <u>Geotextiles and Geomembranes</u>
- D 5596 Standard Test Method for Microscopic Evaluation of the Dispersion of Carbon Black in Polyolefin Geosynthetics
- D 6693 Standard Test Method for Determining Tensile Properties of Nonreinforced Polyethylene and Nonreinforced Flexible Polypropylene Geomembranes
- D 1204 Standard Test Method for Linear Dimensional Changes of Nongrid Thermoplastic Sheeting or Film at Elevated Temperature
- D 696 Standard Test Method for Coefficient of Linear Thermal Expansion of Plastics Between -30°C and 30°C With a Vitreous Silica Dilatometer
- D 746 Standard Test Method for Brittleness Temperature of Plastics and Elastomers by Impact
- D 570 Standard Test Method for Water Absorption of Plastics
- 10. E 96 Standard Test Method for Water Vapor Transmission of Material

1.04 SUBMITTALS

A. All work for and in connection with the installation of the lining, field seaming and welding joints shall be completed in strict conformity with all applicable instructions and recommendations of the liner manufacturer.

03200-1

E05jobi/05s3004A/Gyptum Stacking/IDNR Dum Safety Permit Application/Specs/S03400_Concrete Embedment Liner.doc

B. Included with the shipment of liner, submit certified test reports that the liner and material are manufactured in accordance with standards specified herein.

1.05 QUALIFICATIONS

- A. The HDPE liner specified in this section shall be furnished by a manufacturer who is fully experienced, reputable and qualified in the manufacturing of the materials. The manufacturer must at least 10 years of manufacturing experience.
- B. Locking devices must be extruded to the sheet as a one step process.
- C. Liner shall be GSE StudLiner as manufactured by GSE Lining Technology, Inc.
- D. Liner shall be 8 feet in width.
- E. Liner shall demonstrate a minimum pull-out strength of 14,000 psf.

1.06 COORDINATION

A. Coordinate with placement of formwork, formed openings and other work.

PART 2. PRODUCTS

2.01 ROLL DIMENSIONS

- A. Embedment sheets shall be produced in rolls that are 8.0 ft (2.4 m) in width and a thickness range of 80 mils (2.0 mm) to 200 mils (5.0 mm) in thickness. Roll lengths vary according to thickness.
- B. Locking studs of the same material as that of the liner shall be integrally extruded with the sheet. Stud spacing shall be on approximate 1.25 in (30 mm) centers, such that there are approximately 110 studs per square foot (1200 per square meter).

2.02 MATERIAL PROPERTIES

- A. The material used in the embedment liner and in all welding strips shall be made from 97-98% virgin high density polyethylene and 1.5-3% carbon black or pigmentation for the purpose of an otherwise specified color.
- B. Plasticizer shall not be added to the resin formulation.
- C. Embedment sheet and welding strips shall be free of holes, pinholes, bubbles, blisters, excessive contamination by foreign matter, and nicks and cuts on roll edges.
- D. The HDPE cap strips shall be made from HDPE, have good impact resistance and have an elongation sufficient to bridge up to 1/4 inch settling cracks.
- E. Cap strips shall be approximately 4 inches wide or greater and shall be equivalent to that of the liner.
- F. Material shall maintain a repairable state through it's lifecycle by methods approved and recommended by the manufacturer.
- G. Embedment sheets shall have the following physical properties when tested in accordance with Table 1.
- H. Raw resin shall have the following properties when tested in accordance with Table 2.

Property	Test Method		Nominal Value			Testing Frequency
Thickness, mm (mil)	ASTM D 5199	2.00 (80)	3.00 (120)	4.00 (160)	5.00 (200)	Every 5th roll
Density, g/cm3	ASTM D 1505	0.94	0.94	0.94	0.94	1/100,000 ft2
Tensile Properties Strength@Yield,lb/in ² (MPa) Elongation @ Break, %	ASTM D 6693 Type IV, Dumbell G.L.= 2.0in.	2,200 (14.5) 500	2,200 (14.5) 500	2,200 (14.5) 500	2,200 (14.5) 500	1/100,000 ft ²
Stud Pull-Out Strength ¹ , 1b/ft ² (kN/m ²)		>14,000 (669.89)	>14,000 (669.89)	>14,000 (669.89)	>14,000 (669.89)	1/ product
Carbon Black Content/ Pigment Content, % Black Liner Gray Liner	ASTM D 1603, mod. ASTM D 5630, mod.	2-3 1.5-2.5	2-3 1,5-2.5	2-3 1.5 - 2.5	2-3 1.5 - 2.5	1/100,000 ft ²
Carbon Black Dispersion ²	ASTM D 5596	Note 2	Note 2	Note 2	Note 2	1/100,000 ft ²
Notched Constant Tensile Load, hours	ASTM D 5397	400	400	400	400	1/ formulation
Coefficient of Linear Thermal Expansion, per ^a C	ASTM D 696	1.20E-04	1.20E-04	1.20E-04	1.20E-04	I/ product
Low Temperature Brittleness, °C	ASTM D 746	-77	-77	-77	-77	1/ product
Dimensional Stability, % (each direction)	ASTM D 1204	<u>±</u> 1.0	± 1.0	± 1.0	± 1.0	1/ product
Water Absorption, %	ASTM D 570	0.1	0.1	0.1	0.1	1/ product
Water Vapor Transmission, (g/m ² /day)	ASTM E 96	<0.01	<0.01	<0.01	<0.01	1/ product

Table 1: Material Properties

03200-3

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S03400_Concrete Embedment Liner.doc

¹Note: Concrete must have a compressive strength of at least 5,000 lb/in¹ (34,500 kPa). ²Note: Dispersion only applies to near spherical agglomerates. 9 of 10 views shall be Category 1 or 2. No more than 1 view for category 3.

03200-4

Table 2: Raw Material Properties

Property	Test Method	Value	Testing Frequency
Density, g/cm3	ASTM D 1505	0.932	1/ resin lot
Melt Flow, g/10 min	ASTM D 1238 (190/2.16)	≤1.0	1/ resin lot
OIT, minutes	ASTM D 3895 (latm/200°C)	100	1/ formulation

2.03 MATERIAL SUPPLY

- A. Embedment sheets shall be supplied in roll form, sheets, pre-fabricated tubes or panels.
- B. Cap strips shall be supplied in 4 inch widths or greater.

PART 3. EXECUTION

3.01 PLACEMENT

- A. Place, support and secure reinforcement against displacement. Do not deviate from required position. Clean reinforcement of foreign particles or coatings.
- B. Accommodate placement of formed openings.
- C. Conform to ACI 318 code for concrete cover over reinforcement.

3.02 FIELD QUALITY CONTROL

A. Contractor shall notify the Owner's Representative at least 24 hrs. in advance of concrete placement. Placement of the Concrete Embedment Liner shall occur in such sequence that the Owner's Representative has sufficient time to inspect the correctness of the placement within the concrete formwork area. The Owner's Representative retains the right to require necessary revisions be made before concrete is placed.

END OF SECTION 03200

03200-5

1/05jobs/05s3004A/Gypsum Stacking/IDNR Dam Safety Permit Application/Specs/S03400_Concrete Embedment Liner.doc



Appendix E: Operation & Maintenance Manual for #1 Ash Pond



Coffeen Power Station

Operational Procedure

x-xxx-xxxx--xxx

Operation & Maintenance Manual for #1 Ash Pond

(Bottom Ash Recycle Pond)

Effective Date: xx/xx/xxxx

Reason for Change: New Procedure

Approved By:	X	Date:	xx/xx/xxxx
	Х		
	John Romang		

Responsible Department: Coffeen Power Station, Technical Services Department

Table of Contents

<u>Section</u>

Page Number

1.0	Purpose	1
2.0	Scope	1
3.0	Responsibilities	1
4.0	Historical Information	1
5.0	Water Supply	Error! Bookmark not defined.
6.0	Operations Requirements	2
7.0	Dam Safety Requirements	
8.0	Maintenance Logs	
9.0	Contact Numbers	
10.0	References	

1.0 Purpose

- 1.1 This procedure is intended to ensure the safe and environmentally responsible operation and use of the #1 Ash Pond (Bottom Ash Recycle Pond) at the Coffeen Power Station. The primary purpose of the #1 Ash Pond is for the removal of bottom ash by settling and the recirculation of slag tank water. The pond is used to supply water to the Unit 1 and Unit 2 ash handling systems via the recycle pumps.
- 2.0 Scope
- 2.1 This procedure applies to all onsite personnel and the Dam Safety Group staff.
- 3.0 Responsibilities
- 3.1 Outside Unit Operator Checks the pond level and screens once a shift. Operates the facilities as described in this Operational Procedure. Reports any conditions noted during routine activities to the Shift Supervisor and Chemistry Department. Writes job requests if a problem is identified.
- 3.2 Shift Supervisor (SS) Calls the Chemistry Department when structural concerns or overflow conditions are reported. Make entries into the shift electronic log book (e-log) indicating the concern and actions taken.
- 3.3 Dam Safety Inspector Conducts weekly detailed dam safety inspections and provides a report with findings and recommendations. Make entries in e-log indicating the concern and actions taken.
- 4.0 Historical Information
- 4.1 The #1 Ash Pond was initially constructed to be a mixed ash deposition pond and was put in service in the mid-1960's. It is located east of the Main Building. It is a 23 acre pond with a maximum outer berm height of 41.5 feet above ground surface level (approximately elevation 637.5'). The pond overflow was located on the north east corner of the pond and discharged into the flume.
- 4.2 The #1 Ash Pond was converted to act as a closed loop system in the late 1970's when the dewatering bins were installed. The mixed ash was removed and deposited into the #2 ash pond during the closure of #2 pond. The #1 Ash Pond berms were modified and an inner berms was added to the pond to aid in dropping out bottom ash solids. Exterior berm elevation is approximately 637.5 feet.

- 4.3 The #1 Ash Pond was equipped with an emergency overflow at the outlet structure. When the pond level reaches approximately 6.5 feet from the top of the berm, it will overflow into the flume. Overflow will be reported to the EPA. In 2011 there was an assessment of the overflow pipe which showed no obstructions or damage.
- 4.4 In 2006, the bottom ash system was modified to directly sluice bottom ash into the pond, bypassing the retired dewatering bins. Bottom Ash is removed from the pond via an outside contractor on an as needed (typically daily) basis.

5.0 Water Supply

All water inlets to the pond are located on the west side of the pond.

The ash sluice lines (from the valve house) discharge to the pond. These lines are used to convey ash from the slag handling system to the #1 Ash Pond. These lines are the southern most of the pond inlets. HPSW system is routed to the pond (valve house sparger valves, floor drains at Unit 1 cyclone level).

The Slag Tank Overflow sump pumps discharge into the pond at the concrete culvert located directly east of the of the lime/soda ash silo. Also in this area, a small stainless line extends thru the concrete. This is the discharge of the sludge pumps at the Waste Treatment System in the Recycle Pump House building.

The recycle pump flow control valves discharge to the pond through a line located at the northwest corner of the pond. Also in the vicinity of this line is the discharge pipe of the recycle pump house sump pumps.

Water from the Unit 1 and Unit 2 oil water separators are typically routed to the pond via the Slag Tank Overflow Pump (STOP) House sumps. Water entering these sumps are floor and roof drains in the plant and the yard area immediately to the north of the main building.

6.0 Operations Requirements

Pond Level - Plant personnel shall monitor the level of the #1 Ash Pond on a daily basis. Pond level is maintained at approximately 1.0' to 1.5' at the water level staff gauge located on the pond side of the screens. The staff gauge has elevation 629.0' as the 0 elevation.

At 2.0' water level (elevation 631.0 feet), the pond overflows resulting in a sampling and analysis requirement for Total Suspended Solids and Oil and Grease with reporting of the results to the IEPA. If the pond is found at or above 2.0' on the pump side staff gauge, contact Chemistry immediately.

Water can be added to the pond from either the Unit 1 or Unit 2 Low Pressure Service Water (LPSW) headers via piping that discharges to the slag tank overflow trench.

Water can be drained from the pond via the water supply pipe to the dewatering bins. Opening this valve drains water from the recycle header which will remove water from the #1 Ash Pond.

Recycle Pump Intake Structure – Suction to the recycle pumps is supplied from the intake structure located at the west end of the north leg of the pond. This is the only water discharge point from the pond. Water level staff gauges are located upstream and downstream of the trash screens for determination of the screen differential. At 0.5' differential, the screens should be cleaned. Level sensors are also installed upstream and downstream of the screens. Digital displays of the upstream and downstream levels are located along the north side of the catwalk leading out to the screen enclosure. These level sensors will generate a high screen differential alarm in the Control Room DCS. Check screen differential (should clean screens at 6 inches differential.) When the screens become plugged, suction to the recycle pumps is reduced. Call shift supervisor to report if screens needs to be cleaned.

Oil Boom – Plant personnel shall monitor the oil boom that is provided upstream of the intake structure. Check condition of oil booms across pond, at discharge, and across pond inlet. Booms should be replaced when they become oil saturated or damaged. Also check that booms have not come unattached from one another. Write JR to change out booms or to reconnect booms when required.

Emergency Conditions – If a condition arises where there is a possibility of an embankment failure, then the following procedures will be followed:

- 1. Notify the Supervising Engineer Dam Safety immediately.
- 7.0 Dam Safety Requirements
- 7.1 Dam Safety Inspections The plant's impoundment and flood prevention structures shall be inspected and maintained in a manner to ensure safe and environmentally responsible operations. A regular maintenance program shall be performed and shall consist of the following inspection items:

- 1. Earth embankments: Walk the crest, side slopes, and downstream toe of the dam concentrating on surface erosion, seepage, cracks, settlement, slumps, slides, and animal burrows. Frequency of inspection: Weekly.
- 2. Vegetation: Grass should be a thick vigorous growth to stabilize the earth embankment soils and prevent erosion form occurring. There should be NO trees on the earth embankment and none within a minimum of 20 feet of the embankment toe or other structures. Mowing frequency: Semiannually.
- 3. Well Readings: Record level of wells on the crest and toe of the berm. Frequency: Quarterly.
- 4. Special Inspections Special inspections of the levees and ash pond berms shall be performed after earthquakes, floods, water level exceedance in the ponds, or heavy rainfall events. Inspection and report shall be equal to an annual inspection level of detail. Water level in the pond should be noted after a heavy rainfall. Dam Safety staff shall accompany plant personnel on special inspections. Frequency: As required.
- 8.0 Maintenance Log
- 8.1 Dam & Berm Inspector shall enter on e-log under the Dam Safety tab all weekly inspections, any usual occurrences, and maintenance performed.
- 9.0 Contact Numbers

Plant Environmental Supervisor: John Romang / 217-534-7629 Plant Dam & Berm Inspector: Vito Passariello/ 217-534-7664 Plant Control Room: 217-534-7668 / 217-534-7669 Supervising Engineer Dam Safety: Steve Bluemner / 314-554-6298 Dam Safety Staff Contact: Mike Wagstaff / 314-554-6296

10.0 References



Appendix F: Operation and Maintenance Manual, Gypsum Management FacilityOperation (2015)

Operation and Maintenance Manual Coffeen Energy Center Gypsum Management Facility Montgomery County, Illinois

IDNR Permit # DS2014019 Dam Permit # IL50578 & IL50579

Prepared For:

ILLINOIS POWER GENERATING COMPANY Coffeen Energy Center 134 CIPS Lane Coffeen, Illinois 62017

Prepared By:

HANSON PROFESSIONAL SERVICES INC. 1525 South Sixth Street Springfield, Illinois 62703

Amended By:

DYNEGY OPERATING COMPANY 1500 Eastport Plaza Drive Collinsville, Illinois 62234

> Original: February 2008 Amended: March 2015

OPERATION AND MAINTENANCE MANUAL COFFEEN ENERGY CENTER GYPSUM MANAGEMENT FACILITY MONTGOMERY COUNTY, ILLINOIS

TABLE OF CONTENTS

<u>Section</u> <u>Pa</u>	ıge
SECTION 1.0 General	1
1.1 Reasons For Development And Dissemination Of The O&M Manual	1
1.2 General Responsibilities Concerning Dams	1
SECTION 2.0 Definitions	2
SECTION 3.0 Information About The Dams	4
3.1 Location	4
3.2 Description Of Dam And Appurtenances	4
3.3 Size and Hazard Classification	5
3.4 Purpose Of The Dams	5
3.5 Pertinent Data	6
SECTION 4.0 Operations Activities	8
4.1 Introduction	8
4.2 Site Operations and Personnel	8
4.2.1 Site Operations	8
4.2.2 Personnel	8
4.3 Gypsum Management Facility Startup	8
4.4 Water Balance	9
4.5 Gypsum Management Facility Operation	9
4.5.1 Routine Operations	9
4.5.2 Piezometer Installation and Monitoring	10
4.6 Dam Inspections	12
4.6.1 Operation and Maintenance Inspection	12
4.6.2 Engineering Inspection	15
4.6.3 Review of Emergency Action Plan	16
SECTION 5.0 Maintenance Activities	17

LIST OF TABLES

Table 3-1	Pertinent Data for the Gypsum Stack Earthen Dam	.6
Table 3-2	Pertinent Data for the Recycle Pond Dam	.7

LIST OF FIGURES

Figure 4-1	Anticipated Phreatic	Surface in Gypsum Stack11	1
0	T	✓ 1	

LIST OF APPENDICES

APPENDIX A –	Location Map
APPENDIX B –	Operation and Maintenance Inspection Checklist
APPENDIX C –	Engineering Inspection Form
APPENDIX D –	Herbicides
APPENDIX E –	Construction Drawings

SECTION 1.0 GENERAL

This operation and maintenance (O&M) manual outlines objectives, proposed policies, responsibilities, and procedures for Coffeen Energy Center personnel who are responsible for the management of the Coffeen Energy Center Gypsum Management Facility (GMF). The GMF incorporates two reservoirs, the Gypsum Pond and the Recycle Pond, for processing and storing gypsum.

1.1 REASONS FOR DEVELOPMENT AND DISSEMINATION OF THE O&M MANUAL

The State of Illinois Rivers, Lakes and Streams Act, (615 ILCS 5) Paragraph 23a includes the statement "The Department is authorized to carry out inspections of any dam within the State, and to establish standards and issue permits for the safe construction of new dams and the reconstruction, repair, <u>operation and maintenance</u> of all existing dams." (emphasis added).

Part 3702 of Section 17 of the Illinois Administrative Code, Chapter I entitled the "Construction and Maintenance of Dams" details the requirements to obtain a permit for the construction, operation, and maintenance of a dam. Section 3702.40 b) includes the following statements:

"4) An applicant for a Class I or II dam shall submit an operational plan specifying the method and schedule for the operation of the dam and the routine operating procedures to keep the dam in good working order, including an emergency warning plan." and

"5) As a condition of each permit, the dam owner shall submit a maintenance plan detailing the procedures and schedules to be followed to maintain the dam and its appurtenances in a reasonable state of repair."

Thus, it is a requirement of all dam owners who have dams which fall under the jurisdiction of the Illinois Department of Natural Resources Office of Water Resources (IDNR-OWR) to operate and maintain them safely.

As a dam owner, Illinois Power Generating Company (IPGC) Coffeen Energy Center is responsible for the safety of the public and for maintaining the structures at the facility for both safety and economy. The overall public interest is served by providing a document to serve as a basis for the safe and economical operation and maintenance of the dam during both emergency and day-to-day conditions.

<u>1.2 GENERAL RESPONSIBILITIES CONCERNING DAMS</u>

IPGC is responsible for the operation and maintenance of the Gypsum Pond Dam and the Recycle Pond Dam. These responsibilities include general maintenance (mowing, removing debris from decants, placing riprap where needed, etc.), operation, inspection and emergency action decisions.

SECTION 2.0 DEFINITIONS

Appurtenant Works - The structures or machinery auxiliary to dams which are built to operate and maintain dams; such as outlet works, spillways, gates, valves, channels, etc.

Boil - A stream of water discharging from the ground surface downstream of the dam carrying with it a volume of soil which is distributed around the hole formed by the discharging water.

Berm - A horizontal step or bench in the sloping profile of an embankment dam.

Breach - A break, gap, or opening (failure) in a dam which releases impoundment water.

Dam - A barrier built for impounding or diverting the flow of water.

Dike (Levee) - An embankment, usually applied to embankments or structures built to protect land from flooding.

Drain, Layer or Blanket - A layer of pervious material in a dam to facilitate the drainage of the embankment including such items as a toe drain, a weephole, and a chimney drain.

Drawdown - The resultant lowering of the water surface level due to the release of water from the impoundment.

Embankment - Fill material, usually rock or earth, placed with sloping sides.

Earthen Dam - Any dam constructed of excavated natural materials.

Emergency Action Plan - A predetermined plan of action to be taken to reduce the potential for property damage and loss of lives.

Failure - An incident resulting in the uncontrolled release of water from the dam.

Freeboard - The vertical distance between a stated water level and the top of the dam.

Gate or Valve - In general, a device in which a leaf or member is moved across the waterway to control or stop the flow.

Groin - The junction of the upstream or downstream face of the dam with the valley wall.

Maintenance - The upkeep, involving labor and materials, necessary for efficient operation of dams and their appurtenant works.

Operation - The administration, management, and performance needed to operate the dam and appurtenant works.

Operation and Maintenance Inspection - Inspections conducted by the dam operator. These inspections are frequent visual "Walk-around" inspections of the dam surface and appurtenant works.

Outlet - An opening through which water can freely discharge for a particular purpose from an impoundment.

Phreatic Surface - The upper surface of saturation in an embankment.

Piping - The progressive development of internal erosion by seepage, appearing downstream as a hole or seam, discharging water that contains soil particles.

Riprap - A layer of large stones, broken rock or precast blocks placed in a random fashion usually on the upstream slope of an embankment dam, on a reservoir shore, or on the sides of a channel as a protection against wave and ice action.

Silt/Sediment - Soil particles and debris in an impoundment.

Slump/Slide Area - A portion of earth embankment which moves downslope, sometimes suddenly, often with cracks developing.

Spillway System - A structure or structures over or through which flows are discharged. If the flow is controlled by gates, it is considered a controlled spillway. If the elevation of the spillway crest is the only control of the flows, it is considered an uncontrolled spillway.

Emergency Spillway - A spillway designed to operate very infrequently, only during exceptionally large floods, usually constructed of materials expected to erode slowly.

Principal Spillway - The main spillway which controls both normal and flood flows and is usually constructed of non-erodable materials.

Auxiliary Spillway - A spillway which works in conjunction with the principal spillway to control flood flows and is usually constructed of non-erodable materials.

Stilling Basin - A basin constructed to dissipate the energy of fast flowing water, such as from a spillway, and to protect the streambed from erosion.

Toe of Embankment - The junction of the face of the dam with the ground surface in the floodplain upstream or downstream of the dam.

SECTION 3.0 INFORMATION ABOUT THE DAMS

3.1 LOCATION

The Gypsum Pond Dam and Recycle Pond Dam are located in the NW 1/4 of Section 11, Township 7 North, Range 3 West of the Third Principal Meridian in Montgomery County, Illinois. More specifically, the dams are located approximately 1.5 miles south of Coffeen, Illinois. A map showing the location of the dams is included in Appendix A.

3.2 DESCRIPTION OF DAM AND APPURTENANCES

The gypsum pond perimeter earthen dam, the gypsum pond "gypsum" dam, and the recycle pond dam will all be regulated in accordance with 17 Illinois Administrative Code (IAC) Part 3702, Construction and Maintenance of Dams. The gypsum pond perimeter earthen dam, which will be lined with a dual high density polyethylene (HDPE) geomembrane system, will have a maximum embankment height of 13 ft and a maximum impounding capacity of 442 acre-ft (measured at the top of earthen dam elevation 632 ft). There will be an additional 123 acre-ft of incised storage. The total volume of gypsum stored within the completed gypsum pond dams will be approximately 2,478 acre-ft.

The dam for the recycle pond, which will be lined with a 60 mil HDPE geomembrane, will have a maximum embankment height of 16 ft and a maximum impounding capacity of 243 acre-ft (measured at the top of dam elevation 629 ft). There will be an additional 99 acre-ft of incised storage.

The gypsum pond will be divided into two sub-cells for the containment of scrubber sludge (gypsum). Discharges to the site will switch back and forth between the two sub-cells so that one sub-cell can be dewatered and raised while the other is in use. There will be two fixed decant pipes constructed in the gypsum stack – one for each sub-cell - which will discharge to stilling wells located adjacent to the perimeter ditches. The control elevation on the decant pipes will be maintained 5.0 ft below the lowest point on the stack cell crest. The decant pipes will enable the cells to be dewatered after storm events so that a minimum of 5.0 ft of freeboard will be maintained in each cell. A minimum of 4.7 ft of freeboard is required above the decant inlet to contain the Probable Maximum Flood (PMF) storm event in addition to peak wind generated waves.

The gypsum pond dam perimeter ditches will be located on the interior sides of the earthen dam. Runoff from the stack will be conveyed through the ditches to a transfer channel which will discharge into the recycle pond. The ditches will be trapezoidal in shape with a 15 ft bottom width, a maximum depth of 9 ft and a longitudinal slope of 0.0005 ft/ft. Side slopes will be 3H:1V. During operation, the ditches will be monitored for erosion. If erosion of the designed ditch geometry occurs, a geogrid will be used for stabilization.

The transfer channel between the gypsum pond dam and the recycle pond have a trapezoidal cross-section with 3H:1V side slopes will be lined with HDPE. The 500 ft long transfer channel will transition from a 32-ft bottom width at an invert elevation of 623.0 ft at the upstream end to a 60-ft bottom width at an invert elevation of 622.0 ft at the downstream end. The transfer channel will be fitted with stop logs capable of raising the discharge control elevation to 625.0 ft. To prevent degradation of the HDPE liner due to flow velocities, the transfer channel and a portion of the recycle pond dam will incorporate an additional sacrificial layer of HDPE.

The emergency spillway for the recycle pond will consist of three 6 ft by 6 ft precast reinforced concrete risers (drop inlets) with a top elevation of 624 ft (5 ft below the top of the dam). The recycle pond's HDPE liner will attach to the exterior sides of each riser. A 4-ft diameter HDPE outlet conduit will be constructed at each riser with an upstream invert of 615.0 ft and a downstream invert of 613.0 ft. Assuming a normal pool elevation of 624 ft (control elevation of the risers), the emergency spillway has been designed to pass the 24-hour PMF storm event with adequate freeboard to prevent overtopping of the recycle pond crest by wind generated waves. The emergency spillway has been provided in the event of accident or catastrophic rainfall only. It is not expected to be activated during the life of the facility. As designed, all discharges from the system will be through the pump house located on the southeast corner of the recycle pond.

3.3 SIZE AND HAZARD CLASSIFICATION

If a worst case failure of the gypsum pond dam were to occur, and the entire volume of the stack is released easterly into Coffeen Lake, the Coffeen Lake reservoir has adequate freeboard to accept this additional volume without overtopping the dam during flood events up to and including the 60 percent PMF. However, the power plant and several residences could potentially be impacted if the gypsum stack dam were to fail in a westerly direction. Considering the regulatory criteria established in Part 3702, the gypsum stack perimeter earthen dam and the gypsum stack "gypsum" dam are classified as intermediate-size Class I (high hazard potential) dams.

A failure of the recycle pond dam would discharge water to Coffeen Lake but it is not anticipated to result in loss of life or any significant economic damage. Breach analyses indicate that a failure of the recycle pond dam during a PMF event would be expected to result in an increase in the Coffeen Lake water surface elevation of not more than ½ inch. Accordingly, the recycle pond dam is classified as a small-size Class III (low hazard potential) dam.

<u>3.4 PURPOSE OF THE DAMS</u>

The dams will be used to dewater, store and dispose of flue gas desulphurization sludge (gypsum) from the Coffeen Power Station (the Plant). Gypsum will be transported to the Gypsum Pond Dam in slurry form (approximately 20 percent solids) and allowed to settle. Clarified process water will then be decanted to the recycle pond and returned to the Plant for reuse via a pipeline.

3.5 PERTINENT DATA

Pertinent data about the dams, appurtenant works, and reservoirs are presented in Table 3-1 and Table 3-2.

Table 3-1 Pertinent Data for the Gypsum Pond Earthen Dar	m
(Based on the Construction of 2 Gypsum Cells)	

Perimeter Ditches			Transfer Channel		
Bottom Width	15.00	feet	Bottom Width	32.00	feet
Top Width	73.50	feet	Top Width	86.00	feet
Depth	9.00	feet	Depth	9.00	feet
Outer Side Slope	3:1	H:V	Upstream Invert	623.00	feet
Inner Side Slope	3:1	H:V	Downstream Invert	622.00	feet
Upstream Invert	624.85	feet	Weir Elevation	625.00	feet
Downstream Invert	623.00	feet	Weir Length (at 2 ft height)	44.00	feet
Ditch slope	0.00050	ft/ft			
Bank Full Cross-sectional Area	378.00	sf	Dam		
Length of Each Ditch (Centerline)	3710.00	feet	Top of Dam Elevation	632	feet
Bank Full Volume of Each Ditch	32.19	acre-ft	Reservoir Surface Area	77.29	acres
Total Ditch length (Centerline)	7420.00	feet	Total Watershed Area	77.29	acres
Total Ditch Bank Full Volume	64.39	acre-ft	Dam Length	7720	feet
			Dam Height	13	feet
1.0 PMF Storm Event			0.5 PMF Storm Event		
Storm Duration	24	hours	Storm Duration	24	hours
Peak Outflow Discharge	1100.7	cfs	Peak Outflow Discharge	541.1	cfs
Total Discharge Volume	228.83	acre-ft	Total Discharge Volume	122.41	acre-ft
Peak WSEL in Perimeter Ditches	629.89	feet	Peak WSEL in Perimeter Ditches	628.23	feet
Freeboard over Max WSEL	2.11	feet	Freeboard over Max WSEL	3.77	feet
Wave Runup/Wind Setup	2.06	feet	Wave Runup/Wind Setup	2.06	feet
Adequate Freeboard?	YES		Adequate Freeboard?	YES	
100-yr Storm Event - Critical Dur	ation		100-yr Storm Event - 24 Hour Du	ration	
Storm Duration	12	hours	Storm Duration	24	hours
Peak Outflow Discharge	92.6	cfs	Peak Outflow Discharge	62.9	cfs
Total Discharge Volume	50.91	acre-ft	Total Discharge Volume	57.01	acre-ft
Peak WSEL in Perimeter Ditches	626.07	feet	Peak WSEL in Perimeter Ditches	625.84	feet
Freeboard over Max WSEL	5.93	feet	Freeboard over Max WSEL	6.16	feet
Wave Runup/Wind Setup	2.06	feet	Wave Runup/Wind Setup	2.06	feet
Adequate Freeboard?	YES		Adequate Freeboard?	YES	

Note: The Critical Storm Duration is the duration of the rainfall event which produces the highest reservoir water surface elevation in the Gypsum Stack Perimeter Ditches for the given storm frequency. In each case, the starting normal pool elevation of the Recycle Pond is considered to be at elevation 624 ft.

Table 3-2 Pertinent Data for the Recycle Pond	Dam
(Based on the Construction of 2 Gypsum Ce	lls)

Dam			3 Spillways- 6ft x 6ft inlet w/ 4ft d	ia outlet pip	e
Top of Dam Elevation	629	feet	Weir Length	22	feet
Invert of Reservoir Elevation	605	feet	Weir Elevation	624.00	feet
Reservoir Area at Invert	11.55	acres	Outlet Conduit Length	120	feet
Reservoir Area at Top of Dam	17.07	acres	Outlet Conduit Diameter (Inside)	48	inch
Total Reservoir Volume	341.91	acre-ft	Upstream Invert	615	feet
Volume at Elevation 624 ft	259.60	acre-ft	Downstream Invert	614	feet
Total Watershed Area	94.36	acres	Outlet Conduit Slope	0.00833	
Dam Length	3600	feet			
Dam Height	16	feet			
1.0 PMF Storm Event - Normal	Pool at El	ev. 624 ft	1.0 PMF Storm Event - Normal	Pool at Elev	. 609 ft
Storm Duration	24	hours	Critical Storm Duration	24	hours
Peak Inflow	1261.6	cfs	Peak Inflow	1261.6	cfs
Peak Outflow	586.9	cfs	Peak Outflow	289.7	cfs
Peak Storage	315.47	acre-ft	Peak Storage	280.65	acre-ft
Peak WSEL (HEC-HMS)	627.45	feet	Peak WSEL (HEC-HMS)	625.34	feet
Freeboard over Peak WSEL	1.55	feet	Freeboard over Peak WSEL	3.66	feet
Wave Runup/Wind Setup	1.20	feet	Wave Runup/Wind Setup	1.20	feet
Adequate Freeboard?	YES		Water Released from Dam?	YES	
0.5 PMF Storm Event - Normal	Pool at El	ev. 624 ft	0.5 PMF Storm Event - Normal	Pool at Elev	. 613 ft
Storm Duration	24	hours	Critical Storm Duration	24	hours
Peak Inflow	608.4	cfs	Peak Inflow	608.4	cfs
Peak Outflow	413.6	cfs	Peak Outflow	0	cfs
Peak Storage	286.48	acre-ft	Peak Storage	255.83	acre-ft
Peak WSEL (HEC-HMS)	625.69	feet	Peak WSEL (HEC-HMS)	623.75	feet
Freeboard over Peak WSEL	3.31	feet	Freeboard over Peak WSEL	5.25	feet
Wave Runup/Wind Setup	1.20	feet	Wave Runup/Wind Setup	1.20	feet
Adequate Freeboard?	YES		Water Released from Dam?	NO	
100-vr Storm Event - Normal	Pool at Fl	≥v 624 ft	100-vr Storm Event - Normal I	2001 at Flev	619 ft

Critical Storm Duration	12	hours Critical Storm Duration		24	hours	
Peak Inflow	113.2	cfs	Peak Inflow	76.6	cfs	
Peak Outflow	95.8	cfs	Peak Outflow		cfs	
Peak Storage	269.36	acre-ft	Peak Storage	258.48	acre-ft	
Peak WSEL (HEC-HMS)	624.63	feet	Peak WSEL (HEC-HMS)	623.94	feet	
Freeboard over Peak WSEL	4.37	feet	Freeboard over Peak WSEL	5.06	feet	
Wave Runup/Wind Setup	1.20	feet	Wave Runup/Wind Setup	1.20	feet	
Adequate Freeboard?	YES		Water Released from Dam?	NO		

Note: The above variation in normal pool elevations for the Recycle Pond is for the purpose of documenting the water surface elevation which must be maintained in the recycle pond in order to prevent the release of water from the GMF for the above described storm events.

SECTION 4.0 OPERATIONS ACTIVITIES

4.1 INTRODUCTION

The operations plan describes the proposed operation of the Coffeen Gypsum Management Facility (GMF) which includes the gypsum pond and the recycle pond.

4.2 SITE OPERATIONS AND PERSONNEL

4.2.1 Site Operations

The GMF will receive gypsum slurry 24 hours per day, seven days per week. Routine facility maintenance and construction activities will generally be conducted during day shift hours. The crest widths for both the gypsum stack earthen dam and the recycle pond dam are 20 ft. In addition, multi-directional ramps are being provided for both structures so that they are readily accessible by inspection, maintenance and gypsum recovery equipment.

The Plant is a restricted access location. Additional fencing around the perimeter of the active sedimentation cells of the gypsum stack and the recycle pond will be erected to prevent unauthorized access to the GMF, which is also under surveillance by security personnel.

4.2.2 Personnel

The proposed GMF will be owned and operated by Ameren Energy Generating Company (Ameren). Corporate offices are located in St. Louis, Missouri. Overall responsibility for the GMF operation lies with Ameren management personnel.

4.3 GYPSUM MANAGEMENT FACILITY STARTUP

The major components of the proposed GMF consist of:

- The gypsum stack dam/impoundment;
- The recycle pond;
- The earthen transfer channel that connects the two structures and through which process water will be decanted from the gypsum stack into the recycle pond; and
- The recycle pond decant and pumphouse through which process water will be returned to the Plant for reuse.

Both the recycle pond and the gypsum stack dam will be constructed before gypsum is placed within the gypsum stack dam/impoundment.

Upon startup, it is likely that the gypsum stack impoundment will have no more than a few feet of water in the bottom to prevent the high density polyethylene (HDPE) geomembrane from moving. The gypsum slurry (approximately 20 percent solids) will be pumped from the

Plant to the gypsum stack via piping. The piping will be HDPE with a suitable pressure rating for the intended hydraulic and static head. The HDPE pipe will discharge the slurry into the impoundment, and gypsum will settle by gravity.

It will take approximately 10 months before the gypsum stack impoundment is filled to elevation 623 ft, the point where process water may begin flowing into the recycle pond via the HDPE-lined earthen channel connecting the two structures. As soon as water begins to fill the recycle pond, it will be pumped back to the Plant for reuse.

4.4 WATER BALANCE

The capacity of the recycle pond has been designed to accommodate all precipitation runoff from the entire gypsum pond/recycle pond area during a 2-week complete maintenance outage at the Coffeen Power Station (the Plant) followed by a 12-week outage of one of the two units. The runoff and excess water accumulated during this time can be stored within the recycle pond without discharging. The design is based on the maximum 3.5 month precipitation that has occurred in the area since 1950. This occurred in April, May, June and half of July, 1957 and consisted of 28.83 inches of rainfall.

The water balance has been carried out for the expected life of the Site. During the first nine or ten months of operation, the water balance is positive, meaning that there is more water entering the gypsum stack/recycle pond system through process water and precipitation than is leaving the system through process water return and evaporation. However, there is 15 ft of freeboard between the pump discharge and the emergency spillway. With proper water-level management, the water surface will remain well below discharge elevation. After this initial startup period, the water balance is negative, meaning that other water sources will need to be continually added to the process water makeup stream to maintain the volume necessary for transport of the gypsum slurry.

The water balance is of particular concern since the entire system is designed to be a closed loop with no discharges. (As previously noted, the recycle pond has been designed with an emergency spillway, but this is only to protect the structures in the event of an unforeseen accident or catastrophic rainfall event.) Table 3.5-2 lists the maximum water surface elevation allowed in the recycle pond in order to prevent the discharge of water for the 100-year storm event and the 0.5 PMF storm event.

4.5 GYPSUM MANAGEMENT FACILITY OPERATION

4.5.1 Routine Operations

Gypsum slurry will initially be discharged at the southwestern corner of the gypsum pond impoundment. Settled gypsum will gradually create a plane of material sloping gently towards the north end of the impoundment. Depending on the slope of the settled gypsum, the discharge pipe may be moved to other corners of the impoundment to evenly distribute the material. Care must be taken during the initial filling period so to ensure that the sand layer covering the ring drains is not disturbed. If necessary, the sand may be armored with larger washed aggregate or the impoundment may be gradually filled with water to cover the sand prior to the discharge of gypsum slurry into the impoundment.

Once the gypsum plane reaches approximately elevation 627 ft (5 ft below the earthen dam crest), a track excavator or similar piece of equipment will be used to create the first gypsum berm and to form the perimeter ditch. Each gypsum berm will be approximately 10 ft in height and will effectively create a two-compartment impoundment within its perimeter. Gypsum for construction of the gypsum berm will be obtained from the settled material on the inside of the berm, creating an inner ditch. Gypsum slurry will then be discharged alternately into the inner ditch of each compartment. Gypsum will settle out into the inner ditch and clarified process water will flood the compartment to a depth of several feet. This water will be decanted to the perimeter ditch by way of an HDPE decant pipe which will discharge to a stilling well located at the toe of the gypsum stack.

As each compartment fills with settled gypsum, the discharge piping will be moved to the alternate compartment. The compartment, or sub-cell, that is not in service will be allowed to dewater and another gypsum berm will be constructed on top of the previous gypsum berm, effectively raising the gypsum stack another 10 ft. This alternating cycle of gypsum discharge, compartment dewatering and berm construction will continue. Gypsum will be deposited in the stack with an average dry density of approximately 74 lb/ft^{3.} Drawing No. C-10201-25 provides a visual description of this process.

4.5.2 Piezometer Installation and Monitoring

The side slopes of the gypsum pond will be constructed with 3:1 side slopes. After consolidation of the settled gypsum over time, the final slopes should approach 3.75:1. The stability of each gypsum pond slope is critically dependent on the location of the phreatic surface which is anticipated to develop within the stack. Ring drains are intended to lower the phreatic surface so that it is located an adequate distance from the surface of the slope in order to maintain slope stability. In order to monitor the phreatic surface within the stack, piezometers will be installed on each side of the gypsum pond. The piezometers will be installed every 15 vertical feet up the slope (45 horizontal feet based on 3:1 side slopes) and will extend to a depth of at least 15 feet below the anticipated phreatic water surface elevation as shown in Figure 4-1. At the time of installation, each piezometer will be labeled with the "critical elevation" corresponding to the anticipated phreatic surface elevation at that location. The anticipated phreatic surface elevation is the water surface elevation which was used in the slope stability analysis of the gypsum pond. The water level in each piezometer will be read and recorded on a monthly basis. If at any time a reading is recorded higher than "critical elevation" for that specific peizometer, the design engineer must be contacted immediately for evaluation of the reading. Any readings above the "critical elevation" may be indicative of improper ring drain function and/or slope instability which could lead to a failure of the gypsum stack. Therefore, it is critical that the piezometers are installed in accordance with the construction plans and specifications and monitored in accordance with this manual. It may be necessary to install additional subdrainage to maintain the phreatic surface at the desired level within the gypsum stack.

Figure 4-1 Anticipated Phreatic Surface in Gypsum Pond

Refer to figure at the end of the report text.

C:\Users\nanakm\Desktop\Coffeen GMF O&M Manual REDLINE Draft Revision 3-2015.doc

4.6 DAM INSPECTIONS

The inspection program includes two types of dam inspections. The first is regularly conducted by the dam operator and is referred to as an Operation and Maintenance Inspection. The second type of inspection, referred to as the Engineering Inspection, is conducted by a qualified engineer approved by IPGC. All engineering inspection reports must be signed and sealed by an Illinois Registered Professional Engineer.

The dam operator will perform monthly Operation and Maintenance Inspections of the gypsum pond perimeter earthen dam <u>and the gypsum berms and side slopes</u> during the operating life of the structure. During these inspections, the gypsum stack ditches and the transfer channel will also be examined for signs of erosion and liner degradation. The "operating life of the structure" will be considered to cease upon covering of the gypsum with an HDPE/soil cover. Engineering Inspections will be conducted on an annual basis during the operating life of the structure and will continue after covering of the gypsum pond until authorization to abandon the structure is received from IDNR/OWR.

4.6.1 Operation and Maintenance Inspection

Occasional "walk-around" inspections of the dams and appurtenant works are to be made by the dam operator. During these inspections, a checklist of items to be maintained and items to be observed should be recorded. Appendix A provides an example of the Operation and Maintenance Inspection Checklist to be utilized for these inspections. If any of the following items are found to be unusual or are cause for concern, the Shift Supervisor should be notified and the Emergency Action Plan should be immediately consulted for guidance on an appropriate course of action.

Frequency: Operation and maintenance inspections will be performed by the dam operator on a monthly basis and also during and after unusual events such as heavy rainfall or an earthquake.

Inspection Items: During each inspection the following items should be noted in particular.

- 1. Water Level Maximum reservoir levels as a result of heavy rainfall should be recorded.
- 2. Earth Embankment Walk the crest, side slopes and downstream toe of the dam concentrating on surface erosion, seepage, cracks, settlement, slumps, slides, and animal burrows. These are described as follows:
 - Surface Erosion Removal of vegetative cover by water action or pedestrian or vehicle usage forming deep ruts or gullies.
 - Seepage The passage of water through and/or underneath the earth embankment abutment and natural groundline or at the contact between the embankment and

outlet works. It can be indicated by cattails or other wet environmental vegetation, erosion, channelization, or slumping on the embankment face.

- Cracks Deep cracks usually indicate the movement of the dam and/or the foundation and can be in either the longitudinal (along the length of the dam) or transverse (across the dam) directions. Cracking can be an indicator of the beginning of slumps. Shallow cracks may develop during the summer when the surface soils of the embankment become severely dried and are usually of no concern in regard to the safety of the dam.
- Settlement Settlement is indicated by depressions or low spots and can be signs of consolidation of the dam or foundation or the loss of material beneath the settlement area.
- Slumps/Slides A slow or sudden movement of the earth embankment slope on either face toward the toe of the dam.
- If seepage indicates the presence of soil particles, or if deep cracks, settlement, slumps, or slides are noticed, a qualified engineer should be contacted immediately for consultation.
- Animal Burrows Animal burrows result in a loss of earth embankment material and can provide seepage paths for water through the embankment.
- 3. Gypsum Embankment Walk the crest, side slopes and downstream toe of the dam concentrating on surface erosion, seepage, cracks, settlement, slumps, slides and animal burrows. The descriptions for these are the same as for earth embankment.
- 4. Vegetation Grass should be a thick vigorous growth to stabilize the earth embankment soils and prevent erosion from occurring. Note the height of the grass; if greater than 1 foot a mowing of the area should be scheduled before the next inspection. There should be NO trees on the earth embankment and NONE within a minimum of 20 feet of the embankment toes or other structures. The gypsum embankment will not be seeded and is not expected to have any vegetation.
- 5. Gypsum Stack piezometers should be inspected for any damage or loss of function. Damaged piezometers must be promptly repaired or replaced since their function is critical to ensuring stability of the gypsum stack.
- 6. The water level in each Gypsum Stack piezometer must be measured and recorded during each monthly inspection. If the water level in any piezometer is above the "critical elevation" as discussed in Section 4.5.2 of this plan, the Ameren Technical Services Superintendent should be notified and the Emergency Action Plan should be immediately consulted for guidance on an appropriate course of action.

- 7. Gypsum Pond LD/LCRS Drains The change in location or amount of flows discharging from the Leak Detection/Leachate Collection Recovery System (LD/LCRS) should be recorded. If a significant change has occurred, a qualified engineer should be contacted for consultation.
- 8. Gypsum Stack Ring Drains The change in location or amount of flows discharging from the Ring Drains should be recorded. If a significant change has occurred, a qualified engineer should be contacted for consultation.
- 9. Gypsum Stack Fixed Decant Check the alignment and supports for the pipe. Record the amount of flows discharging from the pipe and any erosion or scour around the discharge point.
- 10. Gypsum Stack Perimeter Ditch The perimeter ditch should have a consistent prismatic shape for the entire length. Inspect the perimeter ditch for evidence of erosion, sediment deposition and irregularity in channel geometry, especially in the vicinity of siphon, decant or ring drain outfall structures. If irregularities are noted, repairs should be scheduled and completed.
- 11. Drawdown Facilities Check to make sure that the drawdown stop logs in the transfer ditch are undamaged, operating well and allowing for the free flow of water over them. Confirm during inspections the valves are opened and closed at least quarterly.
- 12. Transfer Channel Check for any debris or other obstructions which may block or restrict the free flow of water. Check for any pools or undulation of the floor of the channel.
- 13. Recycle Pond Decant Check for any debris or other obstructions around the Recycle Pond decant which may block or restrict the free flow of water. The emergency dewatering valve should be lubricated. If there is no return water in the pipe, the emergency dewatering valve should be exercised. Record the physical and operating conditions of the system.
- 14. Recycle Pond Drop Inlet Spillways Check for any debris or other obstructions around the inlet crest and at the bottom of the drop inlet which may block or restrict the free flow of water. Check for the development of any rusty areas on the concrete, and seepage, cracking, breaking, or spalling of the concrete. Check for settlement or cracking of the crest. Check for any debris in the pipes which may restrict the flow of water. Check for any tears or leaks in the HDPE liner covering the concrete.
- 15. Recycle Pond Rip Rap Basin Check for any debris or other obstructions in the riprap basin which may block or restrict the free flow of water. Check to make sure that the rip rap is remaining in a uniform position. Freeze/thaw action or flow over the rip rap may tend to lift or fracture, thus requiring replacement or leveling to maintain the necessary level of protection. NO trees or woody vegetation should be growing through the rip rap.

- 16. Fences Check for damage, accumulated debris, operation of gates and locks, and adequacy of locations (this may change with time as people access the area or development occurs in the area).
- 17. Perimeter Check the perimeter of the dams for a distance of at least 100 feet beyond the toe for signs of seepage or boils.
- 18. HDPE Liner Wherever exposed, the HDPE Liner should be inspected for tears, gouges, protrusions under the liner and abrasion.

Records: A log book of activities occurring at the dam is to be kept current by the dam operator. The log book should be reviewed during the Engineering Inspection. This book should contain at the least the following documentation:

- 1. Completed operation and maintenance inspection checklists
- 2. Readings from all piezometers on the Gypsum Stack
- 3. Additional visual observations
- 4. A list of maintenance performed
- 5. A list of any unusual occurrences at the dam
- 6. Copies of the engineering inspection reports

4.6.2 Engineering Inspection

The engineering inspection is to be conducted by a qualified engineer approved by Ameren. The inspection will provide a thorough evaluation of the dam condition and appurtenances. Appendix B is an example of the inspection report form which is to be utilized for these inspections.

Frequency: The Gypsum Pond Dam is a Class I, High Hazard Potential dam and is to be inspected by an Illinois Registered Professional Engineer at least once per year. The Recycle Pond Dam is classified as a Class III, Low Hazard Potential dams and is to be inspected by an Illinois Registered Professional Engineer at least once every five years.

Inspection Items: The engineer will thoroughly inspect all of the items noted in Section 4.6.1 Operation and Maintenance Inspection.

Records: The Dam Inspection Report form from IDNR-OWR "Guidelines and Forms for Inspection of Illinois Dams" (a copy of which is included in Appendix B), will be completed by the inspecting engineer and will be signed and sealed by an Illinois Registered Professional Engineer. This report will document problem areas and deficiencies; recommend remedial actions for problem areas; and establish time requirements for dealing with the problems. The original report will be retained in Dynegy Operating Company (DOC) files, and a copy of the report will be submitted to the Illinois Department of Natural Resources, Office of Water Resources.

4.6.3 Review of Emergency Action Plan

The emergency action plan should be reviewed annually to assure that all contacts, addresses and telephone numbers are current. Changes in the adjacent land use should also be noted and may dictate the need for revisions to the plan. Changes to the plan should be made as appropriate but only with the concurrence of the Montgomery County Emergency Services and Disaster Agency and of the Illinois Department of Natural Resources, Office of Water Resources. Copies of any revisions should also be forwarded to all personnel and known emergency responders that possess previous versions the plan.

SECTION 5.0 MAINTENANCE ACTIVITIES

Timely repairs are a must after problem areas have been identified. The dam operator is to perform the work required to correct items noted in the operation and maintenance inspections and engineering inspections. Such items include repairing erosion of the gypsum slopes, mowing, seeding, tree and brush removal, replacing rip rap, repairing fences and locks, clearing debris, etc. The maintenance activities specified in the following sections are minimum requirements. NOTE: NO alterations or repairs to structural elements should be made without the assistance of the Ameren Chief Dam Safety Engineer and the concurrence of the Illinois Department of Natural Resources, Office of Water Resources.

Debris: Remove all trash, logs and other debris which may obstruct flow into the principal spillway pipes and drop inlets, or block passage from their discharge channels.

Rip Rap: Replenish rip rap as needed to provide adequate protection against erosion.

Vegetation Control

1. Maintain a good grass cover on the embankment by seeding, fertilizing and mulching areas which are refilled, barren, or thinly vegetated. Seeding mixtures used for maintenance reseeding shall result in a cover compatible with adjacent cover. The seeding mixture specified at the time of the dam's construction was IDOT Standard Specifications Class 1A (Salt Tolerant Lawn Mixture) as follows:

- 2. Grassed areas such as the embankment and the areas beyond the embankment toes for a distance of at least 20 feet should be mowed at least twice annually or at any time the height of the grass exceeds 1 foot.
- 3. All erosion areas will be filled and compacted, reseeded, fertilized and mulched to establish a thick erosion resistant cover.
- 4. Remove all trees and brush growing on the dam embankment to prevent development of a root system which could provide seepage paths. Herbicides utilized for tree and brush control are discussed in Appendix D.
- 5. Keep the riprap basin clear of weeds, brush, and trees.

6. Clear all brush and trees for a distance of approximately 20 feet beyond the toe of each dam.

Animal Damage: Fill rodent holes and other animal burrows with compacted clayey soil and reseed. If rodents become a nuisance, an effective rodent control program as approved by the Illinois Department of Natural Resources District Wildlife Biologist should be implemented.

Signs: All warning signs shall be maintained (repaired, painted, or replaced) as needed.

Gypsum Slopes: Erosion of the gypsum slopes will be evident with the presence of erosion rills. Erosion rills should be filled with additional gypsum material and graded to conform with the design slope.

Piezometers: All piezometers on the gypsum stack shall be inspected for signs of damage or displacement. Non-functioning piezometers shall immediately be replaced.

APPENDIX A LOCATION MAP APPENDIX B OPERATION AND MAINTENANCE INSPECTION CHECKLIST

OPERATION AND MAINTENANCE INSPECTION CHECKLIST

Dam Name (circle one): Gypsu	ım Pond D	am	Recycle Pond Dam
Date:	Ti	me: _	
Name of Inspector:			
Reservoir Elevation:	fe	et	
ITEM	<u>NO</u>	<u>YES</u>	<u>IF YES</u>
Record Piezometer Readings for Gypsum Stack. Are any readings above the critical level? (see section 4.5.2 of O&M Manual)	a		Contact Manager, Environment & Chemistry and notify Hanson Professional Services
Gypsum Stack. Any damage?			Chemistry
Deep Surface Cracks			Contact Manager, Environment & Chemistry
Slump or Slide on the upstream or downstream face			Contact Manager, Environment & Chemistry
Erosion from runoff, wave action or traffic			Repair and stabilize
Embankment, abutment or spillway seepage	e		Contact Manager, Environment & Chemistry
Seepage or flows of muddy water			Contact Manager, Environment & Chemistry
Uneven settlement			Contact Manager, Environment & Chemistry
Trees, brush or burrow holes on the embankment or in the riprap basin			Remove trees and brush, fill holes
Transfer channel or Spillway pipes blocked			Clear immediately
Damage to stop logs			Repair or replace
Damage to HDPE Liner			Repair and schedule engineer inspection
Settlement or displacement of Gypsum Pone fixed decant pipes or outlets	d		Schedule engineer inspection
Discharge from Gypsum Pond LD/LCRS Drains?			Record discharge rate for each outlet (time to fill bucket)
Discharge from Gypsum Pond Ring Drains	?		Record discharge rate for each outlet (time to fill bucket)
Gypsum Stack Perimeter Ditch erosion			Schedule repair
Problems with Recycle Pond spillways			Contact Manager, Environment & Chemistry
Problems with Recycle Pond decant			Contact Manager, Environment & Chemistry
Height of grass (inches)	i	inches	If more than 1 foot, schedule mowing
Damage to fencing, gates and locks or other access restriction measures			Contact Manager, Environment & Chemistry
Confirm drawdown facilities are opened and	d l		Contact Manager, Environment &
closed at least quarterly.			Chemistry

Comments:

APPENDIX C ENGINEERING INSPECTION FORMS
Dam Inspection Report

Name of Dam	Dam Identification Number		
Permit Number	Class of Dam		
Location <u>NW 1/4</u> Section <u>11</u>	Township <u>7N</u> Range <u>3W 3rd</u> P.M.		
Owner			
Name	Telephone Number (Day)		
Street	Telephone Number (Night)		
	County Montgomery		
City Zip Co	de		
Type of Dam			
Type of Spillway			
Date(s) Inspected			
Weather When Inspected			
Temperature When Inspected			
Pool Elevation When Inspected			
Tailwater Elevation When Inspected			
	Inspection Personnel:		
	Nome		
	name		
	Name Title		
	Namo		
	Name Title		

Professional Engineer's Seal

The Department of Natural Resources is requesting information that is necessary to accomplish the statutory purpose as outlined under the River, Lakes and Streams Act, 615 ILCS 5 (1994 State Bar Edition). Submittal of this information is REQUIRED. Failure to provide the required information could result in the initiation of non-compliance procedures as outlined in Section 702.160 of the "Rules for Construction and Maintenance of Dams". This form has been approved by the State Forms Management Center.

CONDITION CODES

<u>EC</u>	=	Emergency Condition. A serious dam safety condition exists that needs immediate action. Emergency measures implemented as instructed by Chief Dam Safety Engineer; such as, pool draw down, work stoppage, plant stoppage.
<u>NE</u>	=	No evidence of a problem
<u>GC</u>	=	Good condition
MM	=	Item needing minor maintenance and/or repairs within the year, the safety or integrity of the item is not yet imperiled
<u>IM</u>	=	Item needing immediate maintenance to restore or ensure its safety or integrity. Remediation should be completed within <u>1 month.</u>
<u>EC</u>	=	Emergency condition which if not immediately repaired or other appropriate measures taken could lead to failure of the <u>dam</u>
<u>OB</u>	=	Condition requires regular observation to ensure that the condition does not become worse
<u>NA</u>	=	Not applicable to this dam
<u>NI</u>	=	Not inspected - list the reason for non-inspection under deficiencies
<u>EC</u>	:	Emergency Condition. A serious dam safety condition exists that needs immediate action. Emergency measures implemented as instructed by Chief Dam Safety Engineer; such as, pool draw down, work stoppage, plant stoppage.

GYPSUM STACK - EARTH EMBANKMENT

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Surface Cracks			
Vertical and Horizontal Alignment of Crest			
Unusual movement or Cracking at or Beyond Toe			
Sloughing or Erosion of Outer Embankment Slopes			
Upstream Face Slope Protection (HDPE Liner)			
Seepage			
Animal Damage			

GYPSUM STACK - EARTH EMBANKMENT

(Continued)

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Vegetative Cover			

GYPSUM STACK - GYPSUM EMBANKMENT

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Surface Cracks			
Vertical and Horizontal Alignment of Crest			
Unusual movement or Cracking at or Beyond Toe			
Sloughing or Erosion of Outside Embankment Slopes			
Sloughing or Erosion of Inside Embankment Slopes			
Seepage			
Animal Damage			

GYPSUM STACK - GYPSUM EMBANKMENT (Continued)

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Condition of Piezometers on Gypsum Stack			
Piezometer Readings on Gypsum Stack Above Critical Level?			

<u>GYPSUM STACK – PERIMETER DITCH</u>

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Ditch Geometry (15 ft bottom width, 3:1 slopes, 8-9 ft depth)			
Concrete Apron at ring drain outlets			
Ring Drain Discharge Pipes			
Stilling Wells for Fixed Decants			

TRANSFER CHANNEL - (between gypsum stack and recycle pond)

Drop Inlet	Structure	X Overflow Spillway Structure	Gated
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Debris			
Side Slope Stability			
HPDE Liner			
HDPE Liner Welds			
Stop Logs			
Differential Settlement			

RECYCLE POND - EMBANKMENT

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Surface Cracks			
Vertical and Horizontal Alignment of Crest			
Unusual movement or Cracking at or Beyond Toe			
Sloughing or Erosion of Outer Embankment Slopes			
Upstream Face Slope Protection (HDPE Liner)			
Seepage			
Animal Damage			

RECYCLE POND - EMBANKMENT

(Continued)

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Vegetative Cover			

RECYCLE POND - PRINCIPAL SPILLWAY (Left, Looking Downstream)

X Drop Inlet	Structure	Overflow Spillway Structure	Gated
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Alignment of Structure Walls			
Construction Joints			
Differential Settlement			
Erosion, Spalling, Cavitation			
Joint Separation			
Seepage Around or into Conduit			
Surface Cracks			

<u>RECYCLE POND - PRINCIPAL SPILLWAY (Left, Looking Downstream)</u>

(Continued)

X Drop Inlet	Structure	Overflow Spillway Structure	Gated
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Structural Cracks			

RECYCLE POND - PRINCIPAL SPILLWAY (Center)

X Drop Inlet Structure		Overflow Spillway Structure	Gated
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Alignment of Structure Walls			
Construction Joints			
Differential Settlement			
Erosion, Spalling, Cavitation			
Joint Separation			
Seepage Around or into Conduit			
Surface Cracks			

RECYCLE POND - PRINCIPAL SPILLWAY (Center)

(Continued)

X Drop Inlet Structure		Overflow Spillway Structure	Gated
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Structural Cracks			

RECYCLE POND - PRINCIPAL SPILLWAY (Right, Looking Downstream)

X Drop Inlet	Structure	Overflow Spillway Structure	Gated	
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE	
Alignment of Structure Walls				
Construction Joints				
Differential Settlement				
Erosion, Spalling, Cavitation				
Joint Separation				
Seepage Around or into Conduit				
Surface Cracks				

RECYCLE POND - PRINCIPAL SPILLWAY (Right, Looking Downstream) (Continued)

X Drop Inlet Structure		Overflow Spillway Structure	Gated
ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Structural Cracks			

RECYCLE POND - ENERGY DISSIPATOR

FHWA HEC-14, Riprap Basin

Type:

Outlet Works

X Principal Spillway

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Riprap			
Outlet Channel			
Debris			

RECYCLE POND - DECANT STRUCTURE

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Alignment			
Connection to Bollard			
Debris in Inlets			
Condition of Pipe			
Condition of Liner Beneath Pipe			
Connection to Ballast			
Connection of Pipe Boot to Liner			

RECYCLE POND - DECANT STRUCTURE

(continued)

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE
Seepage Around or into Conduit			

RECYCLE POND – WATER LEVEL GAGE STRUCTURE

ITEM	CONDITION	DEFICIENCIES	RECOMMENDED REMEDIAL MEASURES & SCHEDULE

APPENDIX D HERBICIDES

HERBICIDES

Site personnel should check with the Illinois Department of Natural Resources, Regional Fisheries Biologist and the Regional Wildlife Biologist before using any herbicide. Read the product label prior to use and follow the use directions and precautions accordingly.

On March 1, 1979 the U.S. Environmental Protection Agency (U.S.E.P.A.) halted the use of the herbicide 2, 4, 5-T in parks and recreation areas. The use of silvex (2, 4, 5-TP) around water has also been banned.

The Agronomy Department at the University of Illinois and the Aquatic Biology Section of the Department of Natural Resources, Office of Scientific Research and Analysis indicate that the herbicides containing the 2, 4-D or 2, 4-DP are legal for use in parks and recreation areas and effective for controlling brush and woody growth. Some examples of approved herbicides are:

- 1. Tordon RTU by DOW Chemical. (Can be obtained with blue dye.)
- 2. WEEDONE 170 by Union Carbide
- 3. WEEDONE, 2, 4-DP by Union Carbide
- 4. A 1% to 2% solution of ROUNDUP
- 5. Garlon by DOW Chemical
- 6. Banvel by Sandoz

Your distributor may carry brand name herbicides other than those listed above. Be certain that the product does not contain the ingredients 2, 4, 5-T or 2, 4, 5-TP. An example of an unacceptable product is ESTERON 2, 4, 5 by DOW Chemical.

APPENDIX E CONSTRUCTION DRAWINGS



Appendix G: Photos of 2015 Sloughing Repairs

AECOM



Figure G.1. Photo of 2015 sloughing prior to repairs.



Figure G.2. Photo of 2015 sloughing prior to repairs.

AECOM



Figure G.3. Photo of 2015 sloughing area after repairs.



Figure G.4. Photo of 2015 sloughing area after repairs.

ATTACHMENT E



ATTACHMENT H





OBG | There's a Way

November 21, 2017

Mr. Rick Diericx Managing Director – Environmental Compliance Group Dynegy Operating Company 1500 Eastport Plaza Drive Collinsville, IL 62234-6135

Subject:Response to IEPA Comments – Coffeen Station Inactive Ash Pond No. 2
Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2
NRT Project No. 2380

Dear Mr. Diericx:

Natural Resource Technology, Inc., an OBG Company (NRT) is providing this letter to Dynegy Operating Company (Dynegy) in response to comments received from the Illinois Environmental Protection Agency (IEPA) dated October 27, 2017 regarding the *Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2* (Closure Plan; AECOM, January, 2017) at Illinois Power Generating Company Coffeen Power Station, in Coffeen, IL.

This Response to Comments will serve as Addendum 1 to the Closure Plan dated January 2017. For ease of review, IEPA comments are presented below in italics, followed by responses. Supplemental information to support the responses, when required, is included as Attachments 1-3. This document provides responses to all IEPA comments numbered 1 - 8.

Comment 1

It is mentioned multiple times in the closure and post-closure care plans that there are coal mines in the vicinity at depth. Please provide additional information on the locations and depths of the coal mines in the vicinity of Ash Pond No. 2.

<u>Response:</u> The Truax-Traer Coal Company and the Consolidation Coal Company extracted coal underlying Ash Pond No. 2 from 1964 to 1983 (Attachment 1 – Mine Index 871). The mine was originally known as the "Hillsboro" Mine and after Consolidation Coal Company took over the mining operation it was renamed "Consolidation No. 63, Hillsboro". Herrin No. 6 Coal was mined from depths of 500-510 feet below ground surface. The coal seam was 5.8-7.1 feet thick and an estimated 26,800,000 tons of coal were removed from the mine during the operational period (see Reference 1) with an extraction ratio of approximately 25% based on an estimate of volume removed from the recorded mine maps. A Mine Workings Map dated 1969 is overlaid on an aerial of Ash Pond No. 2 and included in Attachment 1 (Figure 1 – Overlay of Historic Mine). Comparison of the mine extents from 1969 with the extent of the mine included in Reference 1 and shown on the Coal Mines in Illinois, Coffeen Quadrangle map included in Attachment 1 indicates there was no additional mining below Ash Pond No. 2 between 1969 and 1983 when the mine closed. As stated in Mine Index 871 (Attachment 1) the Coal Section of the Illinois State Geologic Survey (ISGS) has been assured that the extents included in Reference and Attachment 1 are final and complete.

 Image: Constraint of the street, sifth Floor
 p 414-837-3607
 NRT | AN OBG COMPANY obg.com/nrt

 Milwaukee, WI 53204
 f 414-837-3608
 obg.com/nrt

Comment 2

The Agency requests the addition of a groundwater monitoring well on the east side of Inactive Ash Pond No. 2.

<u>Response:</u> NRT has evaluated the site conditions in the area east of Ash Pond No. 2 and has determined that there is no feasible access to install a monitoring well outside of the embankment because of steep slopes, heavy vegetation, and the presence of wetlands. In addition, several borings along the east side of Ash Pond No. 2 (B403A, P010, and G402) did not encounter the uppermost aquifer during drilling. Given significant access concerns and the limited nature of the aquifer NRT does not recommend installation of a monitoring well.

However, if required or deemed necessary in the future by the IEPA, a boring could be advanced through the berm following completion of closure construction activities to determine if the uppermost aquifer (Hagarstown) is present. Assuming the uppermost aquifer is present, a well could be installed through the berm if requested.

Comment 3

The two new groundwater monitoring wells to the west of the Inactive Ash Pond No. 2 are approved.

<u>Response:</u> A schedule for installation of the wells will be developed following approval of this Closure Plan. Appropriate documentation will be submitted to the IEPA upon completion of installation.

Comment 4

Will changes need to be made to the NPDES permit as part of the implementation of the closure plan?

<u>Response:</u> Yes, an application to modify the Coffeen NPDES permit (IL0000108) is being prepared to authorize the discharge of "dewatering" wastewaters from Ash Pond No. 2 to Coffeen Lake. The application will include an Antidegradation Assessment for Coffeen Lake in support of that modification request.

Comment 5

Calibration of the MODFLOW model was completed using only November 2016 groundwater elevations. Further calibration using more groundwater elevation data over a larger span of time should be completed.

<u>Response</u>: November 2016 groundwater elevations were used to calibrate the MODFLOW model because this was the first and only complete data set available for model development that included groundwater elevations from monitoring wells set in the uppermost aquifer, including wells G406 and G407 installed in August 2016. In order to address IEPA concerns about the efficacy of the groundwater elevation data range used for the model calibration, a comparison of observed versus predicted groundwater elevation values and groundwater flow directions will be provided in post-closure annual reports to assess model performance. If the predictive model does not adequately represent groundwater elevation and flow directions, the model will be recalibrated using available groundwater elevation data collected after November 2016. Similarly, post-closure groundwater quality data will be compared to transport model predictions to assess model performance in post-closure annual reports.



Comment 6

In the Application for a GMZ, it is noted that the problem with groundwater was identified via sampling in 2015. The Agency notes that the Coffeen Power Station received a Violation Notice in 2012 for groundwater standards violations in groundwater around Inactive Ash Pond No. 2.

<u>Response:</u> The GMZ application was revised to include reference to Violation Notice W-2012-00064 (Attachment 2).

Comment 7

Also in the GMZ application, no other remedies to groundwater violations are considered other than to state they are not deemed practical or cost effective. Please discuss other remedies considered and why they are not practical for mitigation relative to the nature of the subsurface or cost-effective.

Response: The GMZ Application was revised to include the following text (Attachment 2):

"Previous experience at similar sites developing and evaluating alternative remedial options and determining costs indicates capping is often the most cost-effective and cost-efficient remedy. Therefore, dewatering and capping were initially evaluated. Based on the results of the evaluation and predictive modeling, the selected remedy successfully mitigates groundwater impacts. Groundwater monitoring will continue to be performed to evaluate the effectiveness of the remedy. If the selected remedy is not demonstrated as successful through collection of data and comparison to predictive values and applicable groundwater quality standards, then other remedial options will be evaluated."

Comment 8

All monitoring wells must be sampled for the parameters listed in 35 IAC 620.410 (a) and (d), with the exception of perchlorate. Statistical analysis for each well's parameters using approved methods listed in 40 CFR 257 should be included.

<u>Response:</u> Tables 2 - 5 of the Groundwater Monitoring Plan, which was included as Appendix B of the Closure Plan, have been revised to include all parameters of 35 IAC 620.410 (a) and (d) with the exception of perchlorate. Revised tables are included in Attachment 3. Note these tables also include aluminum and proposed changes to groundwater standards which are included in IEPA's Proposed Changes to 35 IAC Part 620.

Please don't hesitate to contact us if you have any questions regarding these responses to comments and associated attachments provided herein.

Sincerely, NRT | An OBG Company

stand R Kellen

Nathaniel R. Keller, PG Hydrogeologist

tuit ! hade

Stu J. Cravens, PG Principal Hydrogeologist



Attachments:

Attachment 1: Comment 1 – Historic Mine Documentation
Attachment 2: Comment 2 – Revised GMZ Application
Attachment 3: Comment 8 – Revised Tables 2,3,4, and 5 from Appendix B: Groundwater Monitoring Plan

cc: Ms. Amy Zimmer – IEPA, Hydrogeology and Compliance Unit Mr. Tom Davis, PE - Dynegy, Inc.
Mr. Jason Frierdich, PE – Dynegy, Inc.
Mr. Matt Ballance, PE – Dynegy, Inc.
Mr. Vic Modeer, PE – Dynegy, Inc.
Mr. John Romang, - Illinois Power Generating Company

References:

1 Obrad, Jennifer M., 2011. Directory of Coal Mines in 7.5-Minute Quadrangle Series, Coffeen Quadrangle Montgomery and Bond Counties. The Board of Trustees of the University of Illinois.





Attachment 1

Comment 1-Historic Mine Documentation



Mine Index 871 Consolidation Coal Company, Hillsboro Mine (Consolidation No. 63 Mine)

Type: Underground Total mined-out acreage shown: 4,841

SHAFT, SLOPE, DRIFT or TIPPLE LOCATIONS

Туре	County	Township-Range	Section	Quarters-Footage	
Man shaft	Montgomery	7N 3W	14	SW NE NW	
Air shaft	Montgomery	7N 2W	18	SE NE NE	
Hoist & air shaft	Montgomery	7N 3W	14	NE NW NW	

GEOLOGY

		Th	ickness (1	ft)	Mining	
Seam(s) Mined	Depth (ft)	Min	Max	Ave	Method	
Herrin	500-510			5.83-7.17	RPP	

<u>Geologic Problems Reported</u>: Roof problems were widespread, the sites characterized by slickensided fault planes that cut irregularly through the roof shales and claystones. Small clay dikes were also associated with this small-scale faulting. Floor heaving was slight, but had been a larger problem in the past.

PRODUCTION HISTORY

Company	Mine Name	Years	(tons)	
Truax-Traer Coal Company	Hillsboro	1964-1970	5,605,812	
Consolidation Coal Company	Consolidation No. 63, Hillsboro	1971-1983	21,173,542	
			26,779,354	

. .

Last reported production: July 1983

SOURCES OF DATA

		Original	Digitized		
Source Map	Date	Scale	Scale	Мар Туре	
Company, Coal Section files	2-1-1983	1:12000	1:2170	Final *	

* The map date is before mine closure, but the Coal Section has been assured that the workings shown on the map are indeed final. The mined area shown on the accompanying map is the approximate size expected for the reported production. This suggests that the mine outline is complete.

Annotated Bibliography (data source, brief description of information)

Coal Reports - Production, ownership, years of operation, mine type, depth, thickness. Directory of Illinois Coal Mines (Montgomery County) - Mine names, mine index, ownership, years of operation. Mine notes (Montgomery County) - Shaft location, seam, depth, thickness, geologic problems. Company map, Coal Section files, 1983 Line Project - Shaft locations, mine outline, mining method.




COFFEEN ASH POND NO. 2 | RESPONSE TO IEPA COMMENTS



Attachment 2

Comment 2 – Revised GMZ Application



Title 35, Illinois Admin. Code, Part 620 – APPENDIX D Confirmation of an Adequate Corrective Action Pursuant to 35 Ill. Adm. Code 620.250(a)(2)

Pursuant to 35 Ill. Adm. Code 620.250(a) if an owner or operator provides a written confirmation to the Agency that an adequate corrective action, equivalent to a corrective action process approved by the Agency, is being undertaken in a timely and appropriate manner, then a groundwater management zone may be established as a threedimensional region containing groundwater being managed to mitigate impairment caused by the release of contaminants from a site. This document provides the form in which the written confirmation is to be submitted to the Agency.

- Note 1.Parts I and II are to be submitted to IEPA at the time that the facility claims the alternative
groundwater standards. Part III is to be submitted at the completion of the site investigation.
At the completion of the corrective process, a final report is to be filed which includes the
confirmation statement included in Part IV.
- Note 2. The issuance of a permit by IEPA's Division of Air Pollution Control or Water Pollution Control for a treatment system does not imply that the Agency has approved the corrective action process.
- Note 3. If the facility is conducting a cleanup of a unit which is subject to the requirements of the Resource Conservation and Recovery Act (RCRA) or the 35 Ill. Adm. Code 731 regulations for Underground Storage Tanks, this confirmation process is not applicable and cannot be used.
- Note 4. If the answers to any of these questions require explanation or clarification, provide such in an attachment to this document.

Information provided in the following technical documents is referenced within this form:

- AECOM, 2016c. Revised 30% Closure Design Package for Coffeen Power Station Ash Pond No. 2. April 16, 2016.
- Natural Resource Technology, Inc., 2016a. Hydrogeologic Characterization Report. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.
- Natural Resource Technology, Inc., 2016b. Groundwater Monitoring Plan. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.
- Natural Resource Technology, Inc., 2016c. Groundwater Model Report. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.
- Natural Resource Technology, Inc., 2016d. Hydrostatic Modeling Report. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.

A legal description and map of the proposed GMZ is provided in Appendix A of this GMZ Application. The GMZ will extend vertically through all water-bearing strata through the Hagarstown Member and upper weathered portions of the Vandalia Till Member at an estimated elevation ranging from approximately 604 to 608 ft MSL.



Part I. Facility Information

Facility Name	Coffeen Power Station
Facility Address	134 CIPS Lane, Coffeen, IL 62017
County Montg	omery
Standard Industri	al Code (SIC) 4911

- 1. Provide a general description of the type of industry, products manufactured, raw materials used, location and size of the facility. *Electric power generation and coal combustion residual (CCR) disposal. Ash Pond 2 is located within the Coffeen Power Station which encompasses approximately 4,000 acres including a 1,100-acre lake.*
- 2. What specific units (operating or closed) are present at the facility which are or were used to manage waste, hazardous waste, hazardous substances or petroleum?

	YES	<u>NO</u>
Landfill	Х	
Surface Impoundment	X	
Land Treatment		Х
Spray Irrigation		Х
Waste Pile		Х
Incinerator		Х
Storage Tank (above ground)	X	
Storage Tank (underground)	X	
Container Storage Area	X	0
Injection Well	X	
Water Treatment Units	X	
Septic Tanks	X	
French Drains	X	
Transfer Station		Х
Other Units (please describe)		2

Provide an extract from a USGS topographic or county map showing the location of the site and a more detailed scaled map of the facility with each waste management unit identified in Question 2 or known/suspected source clearly identified. Map scale must be specified and the location of the facility must be provided with respect to Township, Range and Section. *Facility is located in Sections 10 and 11, Tier 7 N, Range 3 W, of the 3rd PM. Figure 1 has the facility located on a USGS topographic map (7½ minute).*



- 4. Has the facility ever conducted operations which involved the generation, manufacture, processing, transportation, treatment, storage or handling of "hazardous substances" as defined by the Illinois Environmental Protection Act? Yes ⊠ No □
 If the answer to this question is "yes" generally describe these operations. *Storage and handling of anhydrous ammonia, sulfuric acid, 50% sodium hydroxide, and chlorine gas.*
- Has the facility generated, stored or treated hazardous waste as defined by the Resource Conservation and Recovery Act? Yes ⊠ No □
 If the answer to this question is "yes" generally describe these operations. *Small quantity TSD.*
- 6. Has the facility conducted operations which involved the processing, storage or handling of petroleum? Yes ⊠ No □ If the answer to this question is "yes" generally describe these operations. *Store, load, and unload diesel fuel and kerosene.*
- 7. Has the facility ever held any of the following permits?
 - a. Permits for any waste storage, waste treatment or waste disposal operation. Yes ⊠ No □ If the answer to this question is "yes", identify the IEPA permit numbers. *IL0000108 and 1998-289- UIC.*
 - b. Interim Status under the Resources Conservation and Recovery Act (filing of a RCRA Part A application).
 Yes □ No ⊠
 If the answer to this question is "yes", attach a copy of the last approved Part A application.
 - c. RCRA Part B Permits. Yes □ No ⊠
 If the answer to this question is "yes", identify the permit log number.
- 8. Has the facility ever conducted the closure of a RCRA hazardous waste management unit? Yes 🗆 No 🗵
- 9. Have any of the following State or federal government actions taken place for a release at the facility?
 - a. Written notification regarding known, suspected or alleged contamination on or emanating from the property (e.g., a Notice pursuant to Section 4(q) of the Environment Protection Act)? Yes ⊠ No □ If the to this question is "yes", identify the caption and date of issuance. *Violation Notice No. W-2012-00064 was issued on June 27, 2012 for boron, manganese, sulfate, and total dissolved solids concentrations which exceeded Class I GW Standards at APW-2/G402.*
 - b. Consent Decree or Order under RCRA, CERCLA, EPAct Section 22.2 (State Superfund), or EPAct Section 21(f) (State RCRA). Yes □ No ⊠
 - c. If either of Items a. or b. were answered by checking "yes", is the notice, order or decree still in effect? Yes ⊠ No □ *Concentrations remain above Class I GW Standards which is why this GMZ is being requested.*
- 10. What groundwater classification will the facility be subject to at the completion of the remediation?

Class I 🛛 Class II 🗆 Class III 🗆 Class IV 🗆

If more than one Class applies, please explain.

11. Describe the circumstances which the release to groundwater was identified. Groundwater sampling at Ash Pond 2 was initiated in 2015. Exceedances of Class I groundwater quality standards in monitoring wells associated with Ash Pond 2 include the parameters arsenic, boron, lead, manganese, sulfate, and total dissolved solids.



Based on my inquiry of those persons directly responsible for gathering the information, I certify that the information submitted is, to the best of my knowledge and belief, true and accurate.

Coffeen Power Station

Facility Name

134 CIPS Lane, Coffeen, IL 62017

Location of Facility

1358030005

Illinois EPA Identification Number

Signature of Owner/Operator

Illinois Power Generating Company Name of Owner/Operator

Date



PART II: Release Information

1. Identify the chemical constituents release to the groundwater. Attach additional documents as necessary.

Chemical Description	Chemical Abstract No.
Arsenic	7440-38-2
Boron	7440-42-8
Lead	7439-92-1
Manganese	7439-96-5
Sulfate	14808-79-8
Total Dissolved Solids	10052

- 2. Describe how the site will be investigated to determine the source or sources of the release. Ash Pond 2 has been investigated as described in the Hydrogeologic Characterization Report (Natural Resource Technology, Inc. [NRT], 2016a).
- 3. Describe how groundwater will be monitored to determine the rate and extent of the release. The monitoring network to monitor the rate and extent of the release is described in the Groundwater Monitoring Plan (NRT, 2016b).
- 4. Has the release been contained on-site at the facility? *The release is contained within the facility boundary. Migration of CCR constituents is limited by Coffeen Lake, which acts as a groundwater discharge area and hydraulic barrier.*
- 5. Describe the groundwater monitoring network and groundwater and soil sampling protocols in place at the facility. *The groundwater monitoring network and sampling protocols are described in the Groundwater Monitoring Plan (NRT, 2016b).*
- 6. Provide the schedule for investigation and monitoring. *The site investigation is complete and groundwater monitoring will continue for the required/permitted frequency and monitoring period as described in the Groundwater Monitoring Plan Section 4.2: Sampling Schedule (NRT, 2016b).*
- 7. Describe the laboratory quality assurance program utilized for the investigation. Laboratory quality assurance is described in the Groundwater Monitoring Plan Sections 4.4: Laboratory Analysis and 4.5: Quality Assurance (NRT, 2016b). The quality assurance/quality control procedures described in the Groundwater Monitoring Plan will be supplemented by the selected Illinois EPA-approved laboratory's QA Manual.
- 8. Provide a summary of the results of available soil testing and groundwater monitoring associated with the release at the facility. The summary or results should provide the following information: dates of sampling; types of samples taken (soil or water); locations and depths of samples; sampling and analytical methods; analytical laboratories used; chemical constituents for which analyses were performed; analytical detection limits; and concentrations of chemical constituents in ppm (levels below detection should be identified as "ND"). A narrative summary of the results of groundwater monitoring is discussed in the Hydrogeologic Characterization Report Section 3: Groundwater Quality (NRT, 2016a). Analytical data summary tables and graphs are available in the Hydrogeologic Characterization Report Appendix F: Groundwater Quality Data and Appendix G: Water Quality Trend Graphs (NRT, 2016a). Lab reports for all monitoring events have previously been submitted to the Agency.





PART II: Release Information (Continued)

Based on my inquiry of those persons directly responsible for gathering the information, I certify that the information submitted is, to the best of knowledge and belief, true and accurate and confirm that the actions identified herein will be undertaken in accordance with the schedule set forth herein.

Coffeen Power Station		
Facility Name	Signature of Owner/Operator	
134 CIPS Lane, Coffeen, IL 62017	Illinois Power Generating Company	
Location of Facility	Name of Owner/Operator	
1358030005		
Illinois EPA Identification Number	Date	





Part III: Remedy Selection Information

- 1. Describe the selected remedy. The remedy includes ash dewatering, relocating/reshaping the CCR within Ash Pond 2 to achieve acceptable grades, construction of a geomembrane cover system and establishing a vegetative cover to minimize long-term erosion (AECOM, 2016).
- 2. Describe other remedies which were considered and why they were rejected. *Previous experience at similar* sites developing and evaluating remedial alternatives and costs indicate capping is often the most cost effective and cost-efficient. Therefore, dewatering and capping were initially evaluated. Based on the results of the evaluation and modeling, the selected remedy successfully mitigates groundwater impacts. If the selected remedy is not shown successful through collection of data and comparison to predictive values, then other remedial options will be evaluated.
- Will waste, contaminated soil or contaminated groundwater be removed from the site in the course of this remediation? Yes □ No ⊠
 If the answer to this question is "yes", where will the contaminated material be taken?
- 4. Describe how the selected remedy will accomplish the maximum practical restoration of beneficial use of groundwater. The dewatering and installation of a geomembrane cover system will control the potential for water infiltration into the closed CCR unit and will allow drainage of surface water off of the cover system. These actions will reduce leachate generation and migration and groundwater quality will improve over time, as described in the Groundwater Model Report (NRT, 2016c).
- 5. Describe how the selected remedy will minimize any threat to public health or the environment. The currently defined extent of the release does not threaten public health. As discussed in the Hydrogeologic Characterization Report Section 2.5 (NRT, 2016a), there are currently no impairments to groundwater usage on the Coffeen Power Station property or surrounding properties associated with Ash Pond 2. No impairments to groundwater usage resulting from establishment of the proposed GMZ are anticipated. CCR dewatering and the geomembrane cover system will reduce leachate generation and migration from Ash Pond 2 and minimize CCR constituents entering the environment, as described in the Groundwater Model Report (NRT, 2016c).
- 6. Describe how the selected remedy will result in compliance with the applicable groundwater standards. *The in place closure of Ash Pond 2, as proposed, will result in a reduction of leachate production, decreasing CCR constituent concentrations and contraction of the groundwater plume. A Groundwater Model Report (NRT, 2016c), included in Appendix D of AECOM 2016, suggests that the geosynthetic cover system will control recharge and subsequent leachate generation within the limits of the Site and reduce concentrations of boron below Class I standards. Concentration reductions are expected to begin approximately one year after completion of the cover system.*
- 7. Provide a schedule for design, construction and operation of the remedy, including dates for the start and completion. *A schedule for implementing the remedies is included in Section 1.3 in AECOM, 2016.*
- 8. Describe how the remedy will be operated and maintained. *The operation and maintenance of the remedy is described in Section 3: Post-Closure Care Plan (AECOM, 2016).*
- 9. Have any of the following permits been issued for the remediation?
 - a. Construction or Operating permit from the Division of Water Pollution Control. Yes \Box No igtiade
 - b. Land treatment permit from the Division of Water Pollution Control. Yes □ No ⊠ If the answer to this question is "yes", identify the permit number.
 - c. Construction or Operating permit from the Division of Air Pollution Control. Yes \Box No \boxtimes



If the answer to this question is "yes", identify the permit number.

10. How will groundwater at the facility be monitored following completion of the remedy to ensure that the groundwater standards have been attained? *Groundwater monitoring procedures are described in Section 4 of the Groundwater Monitoring Plan (NRT, 2016b).*

Based on my inquiry of those persons directly responsible for gathering the information, I certify that the information submitted is, to the best of my knowledge and belief, true and accurate and confirm that the actions identified herein will be undertaken in accordance with the schedule set forth herein.

Coffeen Power Station	
Facility Name	Signature of Owner/Operator
134 CIPS Lane, Coffeen, IL 62017	Illinois Power Generating Company
Location of Facility	Name of Owner/Operator
1358030005	
Illinois EPA Identification Number	Date



COFFEEN ASH POND NO. 2 | RESPONSE TO IEPA COMMENTS



Attachment 3

Comment 8 – Revised Tables 2,3,4, and 5 from Appendix B: Groundwater Monitoring Plan



Table 2. Proposed Monitoring Well Network and Analyses Coffeen Power Station - Ash Pond 2 **Groundwater Monitoring Plan**

	Other Analyses (USEPA	CCR Rule or IEPA)		40 CER 257 - Annendix III	and Amendia IV	Barameters Groundwater		LIEVAUOI			None			IEPA Approved Parameters	40 CFR 257 - Appendix III and Appendix IV	Parameters, Groundwater Elevation
	Additional Monitoring	Programs Performed at Well	Ash Pond 2 & GRP - CCR								None	BION		SW Pond - IEPA		GRP - IEPA and CCR
Proposed Analyses for	IEPA	Monitoring								IEPA 620.410	(a), and (d), no	nerchlorate ³				
	Screen	Bottom	20.9	22.7	21.3	22.8	20.4	13.7	16.5	18.4	21.6	TBD	TBD	21.6	29.6	20.5
	Screen	Тор	16.1	18.1	16.9	12.8	15.8	9.0	11.7	13.6	16.8	TBD	TBD	17.1	25.3	15.7
	Screen	length	5	5	4	10	5	5	5	5	5	TBD	TBD	5	4	5
Bottom of	Screen	Elevation	605	603.66	604.24	590.6	606.03	601.93	607.14	603.49	599.74	TBD	TBD	604.76	602.40	605.32
Top of	Screen	Elevation	609.79	608.31	608.67	600.60	610.70	606.68	611.89	608.30	604.57	TBD	TBD	609.26	606.79	610.16
Measuring Point	Elevation	(2015)	625.92	626.36	625.57	613.37	626.47	615.67	623.63	621.86	621.32	TBD	TBD	626.35	632.04	625.85
Ground Surface	at Time	of Install	622.92	623.82	623.03	610.56	623.81	613.10	620.90	621.86	618.35	TBD	TBD	623.52	629.19	622.95
	Boring/	Well ID	G270	G281	G401	G402	G403	G404	G405	G406	G407	G410 ¹	G411 ¹	G154	G279	G280

Notes:

¹ Proposed wells to be installed upon approval of Closure Plan and GMZ application

² Field parameters include: pH, oxidation -reduction potential, specific conductance, temperature, and dissolved oxygen

³ Groundwater samples collected for metals analyses will be field filtered. Groundwater quality analyses including methods and sampling details are included in Table 4.



Table 3. Background Groundwater Quality and Applicable Groundwater Quality Standards Groundwater Monitoring Plan

Coffeen Power Station - Ash Pond 2

			Backaround	Applicable Groundwater		
	Sampling	IL Class I Std ¹	Concentration ²	Standard ³ for	Maximum ⁵	Minimum ⁵
Parameters (totals)	Program	(mg/L)	tor IEPA (mg/L)	IEPA (mg/L)	(mg/L)	(mg/L)
Aluminum (d)	IEPA	3.5	toq	tDQ	1.0	c00.0>
Antimony	CCR, IEPA	0.006	tbd	tbd	<0.003	<0.003
Arsenic	CCR, IEPA	0.01	tbd	tbd	0.25	<0.001
Barium	CCR, IEPA	2.0	tbd	tbd	0.24	0.014
Beryllium	CCR, IEPA	0.004	tbd	tbd	0.0018	<0.001
Boron	CCR, IEPA	2.0	tbd	tbd	11	<0.01
Calcium	CCR	NS	tbd	tbd	450	<0.1
Cadmium	CCR, IEPA	0.005	tbd	tbd	0.008	<0.001
Chloride	CCR, IEPA	200	tbd	tbd	160	1.5
Chromium	CCR, IEPA	0.1	tbd	tbd	0.034	<0.004
Cobalt	CCR, IEPA	1, 0.002 ⁴	tbd	tbd	0.28	<0.001
Copper (d)	IEPA	0.65, 0.2 ⁴	tbd	tbd	0.021	<0.001
Cyanide	IEPA	0.2	tbd	tbd	<0.005	<0.003
Fluoride	CCR, IEPA	4	tbd	tbd	1.06	0.031
Iron (d)	IEPA	5	tbd	tbd	13	<0.005
Lead	CCR, IEPA	0.0075	tbd	tbd	0.220	<0.001
Lithium	CCR	NS	tbd	tbd	0.057	<0.01
Manganese(d)	IEPA	0.15	tbd	tbd	1.02	<0.018
Mercury	CCR, IEPA	0.002	tbd	tbd	0.00093	<0.0002
Molybdenum	CCR, IEPA	NS	tbd	tbd	0.043	<0.001
Nickel (d)	IEPA	0.1	tbd	tbd	0.035	<0.003
Nitrate-N	IEPA	10	tbd	tbd	8.8	<0.01
Selenium	CCR, IEPA	0.05	tbd	tbd	0.027	<0.001
Silver (d)	IEPA	0.05	tbd	tbd	<0.005	<0.003
Sulfate	CCR, IEPA	400	tbd	tbd	2,500	2.3
Thallium	CCR, IEPA	0.002	tbd	tbd	0.0013	<0.001
TDS (d)	CCR, IEPA	1,200	tbd	tbd	3,900	320
Vanadium (d)	IEPA	0.049, 0.00049 ⁴	tbd	tbd	0.025	<0.003
Zinc (d)	IEPA	5	tbd	tbd	0.59	<0.002
Field pH	CCR, IEPA	6.5 - 9.0	tbd	tbd	8.03	5.80
Radium 226/228	CCR. IEPA	20/20.54	tbd	tbd	4.46	0.185

Notes:

All parameters are totals unless noted. Standards apply to dissolved or total concentrations (d) Dissolved **tbd** = To Be Determined for Illinois EPA monitoring program; CCR Appendix III and IV parameters based on future monitoring , started in November 2015

Bold = Background Concentration exceeds Class I Groundwater Standard

Red = Exceeds Applicable Groundwater Standard NS = No Class II Groundwater Standard

USEPA (t) = background concentration for parameter [total] required under USEPA program (40 CFR Part 257)

² Background Concentration to be calculated following 8 sampling events at all wells in accordance with 40 CFR 257 ¹ IPCB 620 Class I: Potable Resource Groundwater Standard

³ Applicable Groundwater Standard is the higher of the Background Concentration and the Class I Groundwater Standard (or the lower if compared to the pH lower limit)

⁴ Standards listed are proposed changes to 35 III. Adm. Code Part 620 by Illinois EPA

⁵ Groundwater concentrations based on historical results for wells in the proposed sampling program ³ Groundwater samples collected for metals analysis as required by IL620.410 will be field filtered



Table 4. Sampling and Analysis Summary Coffeen Power Station - Ash Pond 2 **Groundwater Monitoring Plan**

Parameter	Analytical Method ⁵	Number of Samples	Field Duplicates ¹	Field Blanks ²	Equipment Blanks ²	MS/MSD ³	Total	Container Type	Minimum Volume ⁴	Preservation (Cool to 4 °C for all samples)	Sample Hold Time from Collection Date
Metals - Appendix III ⁽¹⁾											
Boron (total and dissolved)	6020	14	2	0	0	-	17	plastic	600 mL	HNO ₃ to pH<2	6 months
Calcium	6020	7	1	0	0	1	9	plastic	600 mL	HNO ₃ to pH<3	6 months
Metals - Appendix IV ⁽²⁾ and Add	litional Metals										
Other Metals ⁽³⁾ (total and discolved)	6020	14	2	0	0	٢	17	plastic	600 mL	HNO3 to pH<2	6 months
Manganese (d)	6020	14	2	0	0	-	17	plastic	600 mL	HNO3 to pH<2	6 months
Lithium	6020	7	-	0	0	۲	თ	plastic	600 mL	HNO3 to pH<2	6 months
Mercury	7470A or 6020	14	2	0	0	-	17	plastic	400 mL	HNO ₃ to pH<2	28 days
norganic Parameters - Append	ix III (1) and Othe	er Inorganic P	arameters								
	SM 4500-CN										
Cyanide	or C – EPA	14	2	0	0	-	17	amber	50 mL	NaOH, Cool to 4 °C	14 days
Elinorido	335.4 0214	11	ç	c	c	•	17	nlactic	300 ml	Cont to 4 °C	28 dave
Chloride	9251	4	10	0	0 0	• •	17	plastic	100 mL	Cool to 4 °C	28 days
Nitrate-N	EPA 300.0	14	0	0	0	~	17	plastic	10 mL	Cool to 4 °C	48 hours
Sulfate	9036	14	2	0	0	-	17	plastic	50 mL	Cool to 4 °C	28 days
Total Dissolved Solids	SM 2540 C	14	2	0	0	1	17	plastic	200 mL	Cool to 4 °C	7 days
Radium - Appendix IV ⁽²⁾											
Radium 226	9315 or EPA 903	14	0	0	0	Ļ	15	plastic	1000 mL	HNO3 to pH<2	6 months
Radium 228	9320 or EPA 904	14	0	0	0	۲	15	plastic	1000 mL	HNO ₃ to pH<2	6 months
Field Parameters											
(1) Hd	SM 4500-H+ B	14	NA	NA	NA	AN	14	flow-through cell	NA	none	immediately
Dissolved Oxygen	SM 4500- 0/405.1	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Temperature	SM 2550	14	NA	AN	NA	NA	14	flow-through cell	NA	none	immediately
Oxidation/Reduction	SM 2580 B	14	AN	NA	NA	NA	14	flow-through cell	AN	none	immediately
Specific Conductivity	SM 2510 B	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Turbidity ⁽⁴⁾	SM 2130 B	14	NA	NA	NA	NA	14	flow-through cell or hand- held turbidity meter	NA	none	immediately

Notes: ⁽¹⁾ USEPA Appendix III Parameters (boron, calcium, chloride, fluoride, pH, sulfate, total dissolved solids (TDS))

⁽²⁾ USEPA Appendix IV Parameters

(antimony, arsenic, barium, beryllium, cadmium, chromium, cobalt, fluoride, lead, lithium, mercury, molybdenum, selenium, thallium, radium 226 and 228 combined)

(3) Other Metals = aluminum, antimony, arsenic, barium, beryllium, cadmium, chromium, cobalt, copper, iron, lead, molybdenum, nickel, silver, selenium, thallium, vanadium, zinc

(4) If turbidity exceeds 10 NTUs, a duplicate sample filtered through a .45 micron filter may be collected for metals analysis in addition to the unfiltered sample. Both samples would be submitted for analysis. NA = not applicable HNO₃ = nitric acid

°C = degrees Celsius

mL = milliliter

1. Field duplicates will be collected at a frequency of one per group of 10 or fewer investigative water sample. Field duplicates will not be collected for radium analysis.

2. Field blanks will be collected at the discretion of the project manager; Equipment blanks will be collected at a rate of 1 per sampling event if non-dedicated equipment is used.

3. Matrix Spike/Matrix Spike Duplicate (MS/MSD) samples will be collected at a frequency of one per group of 20 or fewer investigative water samples per CCR unit/multi-unit. Additional volume to be determined by laboratory.

Sample volume is estimated and will be determined by the laboratory.
 Analytical method numbers are from SW-846 unless otherwise indicated. Analytical methods may be updated with more recent versions as apropriate.



Table 5. Summary of Detection Limits for Proposed Monitoring Program Class I Groundwater StandardsGroundwater Monitoring PlanCoffeen Power Station - Ash Pond 2

Constituent	Unit	Analytical Methods ¹	USEPA MCL ² (ug/L)	IL Class I Std ⁷ (ug/L)	RL ⁴ (ug/L)	MDL ⁴ (ug/L)
Metals ⁸						
Aluminum (d)	µg/L	6020	200 ³	3500	10	0.85
Antimony	µg/L	6020	6	6	1	0.25
Arsenic	µg/L	6020	10	10	1	0.25
Barium	µg/L	6020	2000	2000	1	0.4
Beryllium	µg/L	6020	4	4	1	0.5
Boron	µg/L	6020	NS	2000	25	10
Boron(d)	µg/L	6020	NS	2000	2.3	10
Cadmium	µg/L	6020	5	5	1	0.25
Calcium	µg/L	6020	NS	NS	125	100
Chromium	µg/L	6020	100	100	1	0.3
Cobalt	µg/L	6020	NS	1000, 2	1	0.25
Copper (d)	µg/L	6020	NS	650, 200	3	0.025
Cyanide	µg/L	4500	200	200	5	0.85
Iron (d)	µg/L	6010	300 ³	5,000	10	0.88
Lead	µg/L	6020	NS	7.5	1	0.25
Lithium	µg/L	6020	NS	NS	1	0.5
Manganese (d)	µg/L	6020	50 ³	150	1	0.055
Mercury	µg/L	6020 or 7470A	2	2	0.2	0.051
Molybdenum	µg/L	6020	NS	NS	1	0.25
Nickel (d)	µg/L	6020	NS	100	5	0.075
Nitrate- N	µg/L	300	10000	10000	30	8
Selenium	µg/L	6020	50	50	1	0.9
Silver (d)	µg/L	6020	100 ³	50	55	0.028
Thallium	µg/L	6020	2	2	1	0.25
Vanadium (d)	µg/L	6020	NS	49, 0.49	5	0.27
Zinc (d)	µg/L	6020	5000 ³	5	6	0.495
Inorganics	States and	the state of the				
Fluoride	mg/L	9214	4	4	0.1	0.05
Chloride	mg/L	9251	250 ³	200	5	1
Sulfate	mg/L	9036	250 ³	400	10	5
Total Dissolved Solids	mg/L	SM 2540 C	500 ³	1200	20	10
Other			- I State		The Contraction	
Combined Radium 226/228	pCi/L	9315/9320 or EPA 903/904	5	20/20, 5	5	6
Field			120123			
pН	SU	SM 4500-H+ B	NS	6.5-9.0	NA	NA
Oxidation/Reduction Potential	mV	SM 2580 B	NS	NS	NA	NA
Dissolved Oxygen	mg/L	SM 4500-0/405.1	NS	NS	NA	NA
Temperature	°C	SM 2550	NS	NS	NA	NA
Specific Conductivity	µS/cm	SM 2510 B	NS	NS	NA	NA
Turbidity	NTU	SM 2130 B	NS	NS	NA	NA

Notes:

NS = No standard

RL = Reporting limit as established by the laboratory MDL = Method detection limit as established by the laboratory

SM = Standard Methods for the Examination of Water and Wastewater

ug/L = micrograms per liter

mg/L = milligrams per liter

pCi/L = picoCuries per liter

µS/cm = microSiemens per centimeter NTU = nephelometric turbidity unit

(d) = dissolved analysis

1. Analytical method numbers are from SW-846 unless otherwise indicated.

2. USEPA MCL = United States Environmental Protection Agency Maximum Contaminant Level.

3. USEPA SMCL = United States Environmental Protection Agency Secondary Maximum Contaminant Level.

4. Reporting limits and method detection limits will vary depending on the laboratory performing the work.

5. All radium results will be reported (values may be positive or negative) and will include uncertainty and the calculated MDC.

6. Laboratories calculate a minimum detectable concentration (MDC) based on the sample.

7. 35 IAC 620.410 standards are listed including proposed changes submitted to the IL Pollution Control Board



SMARTER SOLUTIONS

EXCEPTIONAL SERVICE

VALUE

HYDROGEOLOGIC SITE CHARACTERIZATION REPORT

Ash Pond 2 Coffeen Power Station Coffeen, Illinois

January 24, 2017



ENVIRONMENTAL CONSULTANTS



ENVIRONMENTAL CONSULTANTS

234 W. Florida Street, Fifth Floor Milwaukee, Wisconsin 53204 (P) 414.837.3607 (F) 414.837.3608

HYDROGEOLOGIC SITE CHARACTERIZATION REPORT

ASH POND 2 COFFEEN POWER STATION COFFEEN, ILLINOIS

Project No. 2380

Prepared For:

Illinois Power Generating Company Coffeen Power Station 134 Cips Lane Coffeen, IL 62107

Prepared By:

Natural Resource Technology, Inc. 234 W. Florida Street, Fifth Floor Milwaukee, Wisconsin 53204

January 24, 2017

hadi

Stuart J. Cravens, PG Principal Hydrogeologist



I R Kellen

Nathaniel R. Keller, PG Hydrogeologist

WWW.NATURALRT.COM

TABLE OF CONTENTS

1	INTR	ористі	ON	1-1
	1.1	Overvie	ew	1-1
	1.2	Site Lo	cation and Background	1-2
	1.3	Site His	story	1-3
2	GEO		ND HYDROGEOLOGY	2-1
	2.1	Region	al Geology	2-1
		2.1.1	Unlithified Deposits	2-1
		2.1.2	Bedrock	2-2
	2.2	Site Ge	eology	2-2
		2.2.1	Ash/Fill	2-2
		2.2.2	Roxana Silt / Peoria Silt (Loess Unit)	2-3
		2.2.3	Hagarstown Member	2-3
		2.2.4	Vandalia Member	2-4
		2.2.5	Mulberry Grove Member	2-5
		2.2.6	Smithboro Member	2-5
		2.2.7	Yarmouth Soil	2-5
		2.2.8	Lierle Clay Member	2-6
		2.2.9	Bedrock	2-6
	2.3	Hydrog	jeology	2-6
		2.3.1	Groundwater Occurrence and Elevations	2-7
		2.3.2	Groundwater Flow	2-8
		2.3.3	Vertical Groundwater Gradients	2-8
		2.3.4	Groundwater Discharge to Coffeen Lake and the Unnamed Creek	2-9
	2.4	Conce	ptual Site Model	2-9
	2.5	Potable	e Water Well Inventory	2-10
3	GRO	UNDWA	TER QUALITY	3-1
	3.1	Summa	ary of Groundwater Monitoring	3-1
	3.2	Ground	dwater Monitoring Results and Analysis	3-1
	3.3	Impact	ed Groundwater Discharge to Surface Water	3-5
4	CON	CLUSIO	NS	4-1
5	REFE	RENCE	S	5-1

FIGURES

Figure 1	Site Location
Figure 2	CCR Management Units
Figure 3	Monitoring Well Location Map
Figure 4	Thickness of Loess / Clay Below Ash Pond 2
Figure 5	Structural Contour Map – Top of Hagarstown Member

2380 Hydrogeologic Characterization Report 170124



- Figure 6 Thickness of Hagarstown Member Sand Unit
- Figure 7Structural Contour Map Top of Vandalia Till
- Figure 8 Structural Contour Map Top of Smithboro Till
- Figure 9 Cross-section A-A'
- Figure 10 Cross-section B-B'
- Figure 11 Cross-section C-C'
- Figure 12Potentiometric Surface Map May 2016
- Figure 13 Potentiometric Surface Map November 2016
- Figure 14 Potential Source Water and Surface Water Sampling Locations

TABLES

Table 1	Summary of Existing Monitoring Well Network
Table 2	Summary of Laboratory Hydraulic Conductivity in Confining Units
Table 3	Summary of Field Hydraulic Conductivity Tests
Table 4	Summary of AECOM Piezometer Construction
Table 5	Groundwater Elevations in Monitoring Wells: 2015 – 2016
Table 6	Groundwater Elevations in Piezometers: 2015 – 2016
Table 7	Ash Pond 2 Leachate Elevations in 2009
Table 8	Summary of Vertical Hydraulic Gradients
Table 9	Statistical Summary of Groundwater Monitoring Parameters and Class I Exceedances: 2015 – 2016
Table 10	Potential Source Water and Surface Water Sampling Results

APPENDICES

Boring Logs and Well Construction Details			
Appendix A1	Ash Pond 2		
Appendix A2	GMF Recycle Pond		
Appendix A3	GMF Gypsum Stack Pond		
Appendix A4	Landfill		
Appendix A5	Southwest Pond		
Appendix A6	Ash Pond 1		
Appendix A7	Historical Boring Logs		
Appendix A8	AECOM Boring Logs and Piezometer Construction Documentation		
Grain Size Analyses and Hydraulic Conductivity Test Results			
Appendix B1	Hydraulic Conductivity Analyses (Wells)		
Appendix B2	Hydraulic Conductivity Analyses (Ash Pond Piezometers)		
Appendix B3	Laboratory Hydraulic Conductivity Tests		
Monitoring Well Hydrographs (2015 - 2016)			
Discharge to Surface Water Calculations			
Water Well Locations and Records within 2,500-Foot Radius of Coffeen Ash Pond 2			
Groundwater Quality Data			
Water Quality Trend Graphs			
	Boring Logs and Appendix A1 Appendix A2 Appendix A3 Appendix A4 Appendix A5 Appendix A5 Appendix A7 Appendix A7 Appendix B1 Appendix B1 Appendix B2 Appendix B3 Monitoring Well Discharge to Su Water Well Loca Groundwater Q		



1 INTRODUCTION

1.1 Overview

This Hydrogeologic Site Characterization Report was prepared by Natural Resource Technology, Inc. (NRT) in support of a Closure Plan for Ash Pond 2 located at the Coffeen Power Station (CPS, Site) which is owned by Illinois Power Generating Company (IPGC). This report and the Closure Plan will apply specifically to this Coal Combustion Residuals (CCR) surface impoundment (Ash Pond 2), and not to any of the other impoundments present on the Site which include the following: Ash Pond 1, Ash Landfill, Gypsum Stack Pond, and Recycle Pond. However, information gathered to evaluate these other CCR units regarding geology, hydrogeology, and groundwater quality is included, where appropriate.

Numerous hydrogeologic investigations have been performed concerning the CCR Units located at the Site. The information presented in this site characterization report includes recent data collected to comply with the Federal CCR Rule (40 CFR Part 257) as well as comprehensive data collection and evaluations from prior hydrogeologic investigation reports (recent to oldest), including, but not limited to, the following:

- Corrective Action Plan. Hanson, April 16, 2016. A plan to remediate groundwater exceedances around Ash Pond 2 and other units. Proposed plan includes reduction in leachate within Ash Pond 2, enhanced cover system on Ash Pond 2, and a Groundwater Management Zone (GMZ).
- Revised 30% Closure Design Package for Coffeen Power Station Ash Pond No. 2. AECOM, April 2016. A 30% design package for closure of Ash Pond 2 including the design basis and summary in addition to preliminary construction costs and schedule.
- Uppermost Aquifer Considerations. Hanson, April 2016. A discussion of the construction of the gyspum pond and relation to the uppermost aquifer in the vicinity of the site.
- Addendum to the 30% Design Data Package for Dynegy Coffeen Power Station; GMF Gypsum Stack Pond. AECOM, January 27, 2016. A geotechnical program consisting of CPT soundings to obtain information for compliance with requirements of the federal CCR Rule.
- 30% Design Data Package for Dynegy Coffeen Power Station; Ash Ponds Nos. 1 and 2 CCR Units. AECOM, January 14, 2016. A geotechnical program consisting of installation of auger borings, CPT soundings, piezometers and soil testing to obtain information for compliance with requirements of the federal CCR Rule.
- G153 Assessment. Hanson, January 15, 2015. An investigation and evaluation of elevated concentrations of sulfate and manganese near the Southwest Storm Water Detention Pond (SW Pond).



- Phase 1 Hydrogeological Assessment Report, Coffeen Energy Center, Montgomery County, IL. Natural Resource Technology, inc., March 2013. An investigation and assessment of the potential for impacts to water quality from unlined impoundments at the Coffeen Power Station. Included a survey to identify wells within 2,500 feet of the Site.
- Section_3 Hydrogeologic Report. Hanson, August, 2009, A summary of the geology and hydrogeology in the area of the proposed Ash Landfill, Gypsum Stack Pond, and Recycle Pond.

In conjunction with this report, a Groundwater Monitoring Plan and a Groundwater Management Zone Application are being prepared to support the closure of Ash Pond 2. In addition, a groundwater flow and transport model was developed to evaluate the effect of various ash pond closure scenarios on groundwater quality and to predict the fate and transport of CCR leachate components. Modeling has also been conducted to enable estimation of the time required for hydrostatic equilibrium of groundwater to be achieved beneath Ash Pond 2.

1.2 Site Location and Background

Ash Pond 2 is one of five CCR units at the CPS, located approximately 2 miles south of the City of Coffeen in Montgomery County, Illinois (Figure 1). The power plant and the CCR Units are situated on a peninsula between two lobes of Coffeen Lake which was created in 1963 by damming a portion of the East Fork of Shoal Creek (IDNR, 2014). The lake covers approximately 1,100 acres and provides cooling water for the CPS.

Ash Pond 2 is located within Section 11 Township 7 North and Range 7 East. The city of Coffeen is approximately 2 miles north of the CPS and the city of Hillsboro, IL is about 8 miles to the northwest. The CPS is located in an agricultural area. Historically, several coal mines were operated at depth in the vicinity of the site as well as a US Minerals processing facility located to the north. The CPS property is bordered by Coffeen Lake on the west, east, and south, and by agricultural land to the north.

Ash Pond 2 was first investigated in 2010, as requested by the Illinois Environmental Protection Agency(Illinois EPA). Results of the investigation (NRT, 2013) indicated the presence of CCR constituents in groundwater in the vicinity of Ash Pond 1 and 2 as well as exceedances of Class I Groundwater Standards for arsenic, boron, lead, sulfate, manganese, iron,sulfate, total dissolved solids, and pH. Additional wells were installed in 2015 to comply with the Federal CCR Rule (40 CFR Part 257), and define the extent of Class I exceedances associated with Ash Pond 2. Based on the groundwater results of wells installed in 2015, Hanson submitted a Corrective Action Plan (CAP, Hanson, 2016a) to define the proposed remedy. Ash Pond 2 will be closed by leaving CCR in place using an alternative geomembrane cover system, following partial dewatering of the pond. This design will control the potential for slope failure and water infiltration into the closed CCR unit and will allow for drainage of surface water off of the cover system (AECOM, 2016a). Illinois EPA responded to Dynegy regarding the CAP with a draft letter including comments and Dynegy and Illinois EPA met on July 20, 2016 to discuss the CAP and proposed remedy. Following the discussion, Illinois EPA in a letter dated August 9, 2016 provided the following comments (summarized):

- 1. Investigation may be required to define the source and extent of exceedances from Ash Pond 2.
- 2. A vertical component is required for the GMZ.
- 3. The GMZ contains portions of CCR units not proposed to close, GMZ must be revised to include only areas where CAP for Ash Pond 2 will mitigate impacts.
- 4. Hydraulic conductivity of foundation layer needs to be evaluated to determine if CAP is appropriate.
- 5. A groundwater monitoring plan, specific to Ash Pond 2 must be submitted with CAP.
- 6. Modeling is required to demonstrate corrective action will be successful, and at what point in the future.
- 7. Evaluate the impacts of the corrective action on Coffeen Lake, with respect to applicable surface water standards.

The Closure Plan, to which this Hydrogeologic Characterization Report, the Groundwater Monitoring Plan, Groundwater Model Report, and Hydrostatic Model Report are attached, provide the information necessary to address these comments and justify the selected corrective action.

1.3 Site History

CPS began operation in 1972 and CCR from the coal fired units was disposed of in Ash Pond 1. Ash Pond 2 was also utilized in the early 1970's and Ash Pond 1 was reconstructed in 1978. Both of these units were used until the mid-1980's. Currently, two coal fired units at Coffeen generate 945 MW of electricity with CCR being handled and filled in Ash Pond 1, the Ash Landfill, the Gypsum Stack Pond, and the Recycle Pond.

Ash Pond 2, which is the sublect of this closure, has a surface area of approximately 60 acres with berms up to 47 feet higher than the surrounding land surface. Ash Pond 2 was removed from service and capped in the mid 1980's. Prior to capping, this pond was identified as Outfall 004 in the facility NPDES operating permit, IL0000108. A 2-foot clay and soil cap was placed on the surface of the pond with contouring and drainage provided to direct storm water to four engineered revetment down drain structures (NRT, 2013).

Other CCR units (not the subject of this Closure Plan) at CPS include the following:

Ash Pond 1 (active unlined impoundment) covers an area of approximately 23 acres, has berms up to 41 feet above the surrounding land surface, and a volume of 300 acre-feet. It primarily receives bottom ash and low volume wastes from floor drains in the main power block building. Several years ago, air heater wash and boiler chemical cleaning wastes were directed to Ash Pond 1 but this practice was discontinued. This impoundment (also known as



the Bottom Ash/ Recycle Pond) is a reclaimed ash pond that was reconstructed utilizing the existing earthen berms with reinforcement, as provided by Water Pollution Control Permit 1978-EA-389 issued by the Agency on May 26, 1978. The bottom ash is periodically removed for beneficial uses by a third-party contractor.

- A Gypsum Management Facility (GMF), consisting of a 77-acre Gypsum Stack Pond and 17-acre Recycle Pond, receives blowdown from the air emission scrubbers and has been in operation since 2010. Construction of the GMF was in accordance with Water Pollution Control Permit 2008-EA-4661 and features a composite HDPE liner with 3-feet of recompacted soil at 1 x 10⁻⁷ cm/s. Both GMF ponds have a groundwater underdrain system. The Gypsum Stack Pond system was actively pumped during construction, but is currently not used. The Recycle Pond underdrain is a passive, gravity drained system.
- Fly ash is managed in a composite lined landfill constructed in 2010. The Ash Landfill has an active groundwater underdrain system that is currently being actively pumped. Additionally, the Ash Landfill leachate collection system is restricted by rule to no more than 1 foot of leachate on the composite liner.
- An Illinois EPA groundwater monitoring program is in effect for the GMF (under Bureau of Water) and Ash Landfill (under Bureau of Land).
- The SW Pond is not a CCR unit; it manages stormwater from around the Ash Landfill. It is managed under NPDES Permit #2011-EB-1289.

2 GEOLOGY AND HYDROGEOLOGY

Since 2010, several site investigations have been completed in the vicinity of the CCR units. The initial site investigation near Ash Pond 2 (Geotechnology, 2010) included the installation of one well near the southeast corner of the unit. Significant investigation has been completed north of the unit prior to permitting of the Ash Landfill, Gypsum Stack Pond, and Recycle Pond. Following promulgation of the Federal CCR Rule (40CFR 257), additional investigation was completed by AECOM (2016a, 2016b)) to obtain geotechnical information and NRT (2013) to install additional wells for groundwater monitoring. The most recent investigation was completed in 2016 by NRT to better define the vertical and horizontal extent of impacts and refine the GMZ proposed in the CAP. The data collected from recent investigations as well as historical work is incorporated in this report to develop an accurate site conceptual model.

2.1 Regional Geology

2.1.1 Unlithified Deposits

Pleistocene deposits of unlithified glacial diamictons, lacustrine/alluvial deposits, and windblown loess overlie Pennsylvanian-age bedrock throughout central Illinois. The most extensive glacial deposits are those from the Illinoian Stage which cover much of the state and are present at the Site. Diamictons deposited by glacial activity during the more recent Wisconsian Stage does not extend into the Coffeen area. Windblown (aeolian) deposits, the Peoria and Roxana Silts, cover the glacial deposits over a majority of the state. These units are fine-grained deposits blown from river valleys by prevailing winds. The Quaternary units are described briefly below (Hensel and Johnson, 1996), and in detail later in this section.

The Illinois State Geologic Survey has mapped loess thickness at the CPS at less than 5 ft. The Peoria Silt is generally classified and described as light yellow-tan to gray, fine sandy silt. The Roxana Silt is predominately silt-sized material, but can be sandier in localized areas and the base of this unit is often colluvium of silt, and sand (Hensel and Johnson, 1996).

Till members of the Glasford Formation include the Smithboro Member, the Mulberry Grove Member, the Vandalia Member, and the Hagarstown Member (oldest to youngest). The Smithboro Member is described as a gray, compact, silty till. The Smithboro is bounded below by the Yarmouth Soil. The Mulberry Grove Member is intermittent at the Site, and is described as a calcareous gray silt and fine sand containing some fossil mollusks. The Vandalia Member is described as a sandy till with thin lenticular bodies of silt, sand, and gravel. It is calcareous, except where weathered, generally gray, and moderately compact. The Hagarstown Member is bounded at the top by the Sangamon Soil. The member consists of gravelly till, poorly sorted gravel, well sorted gravel, and sand (Willman and Frey, 1970).

2380 Hydrogeologic Characterization Report 170124



2.1.2 Bedrock

The site and surrounding areas are underlain by rocks belonging to the Pennsylvanian Bond Formation (Kolata, 2005). Detailed descriptions of the Pennsylvanian strata of Illinois were published by Willman et al. (1975). The Bond Formation includes all strata from the base of the Shoal Creek Limestone Member or the LaSalle Limestone Member to the top of the Millersville Limestone Member or the Livingstone Limestone Member. It is overlain by the Mattoon Formation and underlain by the Modesto Formation. It varies from less than 150 feet thick in eastern Illinois to over 300 feet thick in southeastern Illinois, averaging about 250 feet. The Bond Formation is characterized by a high percentage of limestone and calcareous clays and shales. It is bound by thick limestone members (up to 50 feet), the thickest and purest limestones in the Pennsylvanian System of Illinois. Gray shales constitute the greatest part of the formation, although thick channel sandstones are developed locally.

The elevation of the bedrock surface in the area ranges from 450 to 500 feet above mean sea level (Herzog et al., 1994). The bedrock surface slopes gently towards the west into a minor bedrock valley that runs north-south. Well logs indicate that the lithology of the uppermost bedrock is predominantly shale (Zeizel, 1959).

2.2 Site Geology

Quaternary deposits in the Coffeen area consist mainly of glacial diamictons and outwash deposits that were deposited during Illinoian and Pre-Illinoian glaciations. A hydrogeologic investigation was conducted in 2006 and 2007 by Hanson to characterize the site geology as part of an application for an Initial Facility Report and Operating (Subpart B) Permit, and Hanson issued a Hydrogeologic Report as part of those applications in 2009. The following geologic descriptions are based on the findings of these investigations and supplemented with data near Ash Pond 2 collected from recent investigations performed to meet requirements of the CCR Rule.

Figure 3 and Table 1 display and summarize the monitoring wells present on site, and Appendix A includes boring logs and well construction logs used for this report. Appendix B includes reports for both laboratory and field hydraulic conductivity tests for recent testing completed in 2015 and 2016 in addition to test results from piezometers within the Ash Pond. The major unconsolidated materials present at the site include the following (listed from ground surface down).

2.2.1 Ash/Fill

Ash is present within Ash Pond 2 at thicknesses up to approximately 36 feet as measured in OW-4, but ash is generally only 24 to 29 feet thick as shown in the boring logs for OW1, OW-2, OW-3, and OW-5 (Appendix A). A majority of Ash Pond 2 overlies the Loess/ Silt unit which comprises the Upper Confining Unit (next Subsection). The remaining areas of Ash Pond 2 overlie the Vandalia Till where former



drainage features were present prior to construction and filling (Figure 4). Field hydraulic conductivity tests from piezometers screened within the ash (Appendix B2) resulted in a geometric mean of 1.5×10^{-3} cm/sec indicating a relatively permeable ash unit.

2.2.2 Roxana Silt / Peoria Silt (Loess Unit)

The Roxana Silt and Peoria Silt are stratigraphically distinct units, but are difficult to differentiate in the area of the Coffeen Power Station. In this report, the combined silts will be called the Loess Unit. The Loess Unit extends from beneath the topsoil, which is derived from the loess, to the the top of the Hagarstown (Beds) Member. Thicknesses range from a minimum of <1.0 ft as observed in G401 and G403 located adjacent to and west/southwest of Ash Pond 2, to a maximum of <6 feet as measured in a boring advanced within the footprint of the landfill prior to construction (MW14S). Actual thickness of the Loess Unit is difficult to evaluate due to the illuviation of the model soil horizon. Around Ash Pond 2, the thickness is presumed to be less than four feet. The loess has been variously classified as silt or clayey silt, with minor amounts of sand. The loess often exhibits mottling, concretions, and some fracturing and may be saturated, dependent on the seasonal variation in groundwater levels.

The Loess Unit is generally considered unsaturated and the upper-most aquifer is recharged by precipitation that percolates through this unit. The laboratory tests from recent geotechnical analysis reported vertical hydraulic conductivity values ranging from 1.3×10^{-8} to 5.0×10^{-7} cm/sec, with a geometric mean of 1.0×10^{-7} cm/sec (Table 2). This unit was likely removed from within the footprint of Ash Pond 2 either through erosion or based on ash pond design grades of 616.5 ft. Construction of the landfill and GMF units required the excavation and removal of this layer.

2.2.3 Hagarstown Member

The Hagarstown Member (consisting of gravelly clay till and sandy materials in contact with the Vandalia, (also referred to as Hagarstown Beds) has been separated into two units for this discussion; the first unit, consisting of the gravelly clay till and the second consisting of sandy material overlying the Vandalia Member. The clay till portion had varying thicknesses ranging from 1.9 feet (G404) to over 12 feet as observed in the borings for G401 and G403, located adjacent to, south and west of the Pond. This unit underlies 95% of the pond and in general the clayey portion is at least 9 feet thick (Figure 4). The thickness of the sandy portion of the Hagarstown is generally less than 3 feet, and near Ash Pond 2 it is generally 1 to 2 feet thick as seen in borings for G401, G404, and G405. The composition of the sandy portion of the Hagarstown unit varies across the site and was classified as gravelly till, poorly sorted gravel, well sorted gravel, sand and silty sand. The elevation of the top of the Hagarstown is mapped in Figure 5 and the general thicknesses are illustrated on Figure 6. As displayed in Figure 5, the elevation of the top of the Hagarstown generally declines as the unit approaches Coffeen Lake or other topographic drainage features.



The elevation of the bottom of the discharge channel south of Ash Pond 2 is approximately 600 ft, based on the Ash Pond design drawing. This implies that the channel intersects and cuts through the Hagarstown along the southern side of the Ash Pond. Along the eastern edge, the creek is at an elevation of ~594 to ~590, indicating that the Hagarstown is also exposed. along the ravine slope.

During construction of the Ash Landfill, the Gypsum Pond, and the Recycle Pond, the Loess Unit and the Hagarstown Beds were excavated to facilitate construction by limiting groundwater flow into excavations. The excavations were backfilled with structural fill and an underdrain system was installed to mitigate inward hydraulic pressure and potential liner uplift damage before the CCR units were filled. The Ash Landfill underdrain system remains active. The Recycle Pond is a gravity drain system, and the Gypsum Pond remains in place, but the system is inactive.

The hydraulic characteristics of the Hagarstown Member indicate the unit has a moderate hydraulic conductivity. The results of single-well field permeability tests have hydraulic conductivity values ranging from 3.1×10^{-5} to 1.6×10^{-3} cm/sec, with a geometric mean of 2.9×10^{-4} cm/sec (Table 3). In several locations (i.e., G402, G403, and G404) this unit is very thin, or was not observed in the boring as shown on the boring logs (Appendix A); however, field hydraulic conductivities are similar to those measured in wells screened across the Hagarstown Beds, indicating that at these locations the upper Vandalia is hydraulically similar to the Hagarstown Beds and is likely connected to and continuous with the Hagarstown. The hydraulic conductivity values measured within wells screened across the Hagarstown Beds are significantly higher than both the overlying Roxana/Peoria Silts and lower Vandalia (Till) Member.

2.2.4 Vandalia Member

The Vandalia (Till) Member is a sandy/silty till with thin, discontinuous lenses of silt, sand, and gravel. The Vandalia Till was encountered in all borings advanced at the site. The Vandalia Till ranged in thickness from 11.7 ft in SB-10 (drilled near MW10S) in the northern portion of the CPS, to 31.0 ft in SB-12 (near MW12S between the Gypsum Stack Pond and the Recycle Pond). Based on Borings G45D and G46D, the Vandalia Till is approximately 7 feet thick in the southeast corner (G46D) to 18 feet thick along the northern edge of the Ash Pond 2 (G45D). The elevation of the top of the Vandalia Till is depicted in Figure 7. Below Ash Pond 2 the elevation ranges from approximately 608.5 to 598 ft. Similar to the observed top elevation of the Hagarstown Beds, the top of the Vandalia unit declines in elevation near Coffeen Lake and topographic drainage features. This unit is relatively thick throughout the site, with an average thickness of of over 15 feet (Hanson 2009).

Results from laboratory tests completed for vertical hydraulic conductivity indicate the Vandalia unit has a very low vertical hydraulic conductivity. The laboratory tests reported hydraulic conductivity values ranging from 6.8×10^{-9} to 4.5×10^{-6} cm/sec, with a geometric mean of 4.9×10^{-8} cm/sec (Table 2). Field



hydraulic conductivity tests completed in temporary piezometers (T408 and T409) indicate horizontal conductivities of 9.0×10^{-7} and 3.4×10^{-5} cm/sec, respectively (Table 3). The maximum value was measured in a sand seam within the Vandalia Till, but likely is not representative of the diamicton because sand seams are infrequent and discontinuous.

2.2.5 Mulberry Grove Member

The Mulberry Grove (Silt) Member typically consists of a thin, lenticular unit of gray sandy silt (Willman et al., 1975). It represents the interval between the retreat of the glacier that deposited the Smithboro Member and the advance of the glacier that deposited the Vandalia Member. At the site, the Mulberry Grove Member is represented by pockets (generally less than 2 ft thick) of gray sandy silt. This unit was absent in many borings through the central portion of the site from south to north. Where sampled, the Mulberry Grove Member ranged in thickness from 0.5 ft in Borings SB-12 (MW12D) to 4.9 ft in Boring SB-11 (MW11D), both of which are located near the Gypsum Pond (Hanson 2009). The Mulberry Grove Silt was not encountered in the two deep borings, G45D and G46D, advanced near Ash Pond 2.

The laboratory tests reported vertical hydraulic conductivity values of 1.6×10^{-6} and 1.9×10^{-6} cm/sec, with a geometric mean of 1.7×10^{-6} cm/sec. These silts appear to be deposited in depressions found in the surface of the underlying Smithboro Member.

2.2.6 Smithboro Member

The Smithboro (Till) Member is described as a gray, compact, silty, clayey diamicton. The Smithboro Member ranges in thickness from 6.7 ft in Boring SB-05 (near MW05S) to 21.2 ft in Boring SB-02 (near MW02S) located northwest of the landfill. Figure 8 displays the top elevation of the Smithboro Till. As shown on the figure the top of the unit ranges in elevation from 596 to 590 ft below Ash Pond 2. Borings advanced in the vicinity of Ash Pond 2 did not did not penetrate the full thickness of the Smithboro Till.

Laboratory and field conductivity testing indicate the Smithboro Member has a low hydraulic conductivity. Laboratory test reported vertical hydraulic conductivity values ranging from 1.1×10^{-9} to 1.0×10^{-7} cm/sec with a geometric mean of 1.3×10^{-8} cm/sec (Table 2). Horizontal hydraulic conductivities calculated from single well tests performed in wells G45D and G46D were 4.0×10^{-8} and 4.9×10^{-7} cm/sec, respectively (Table 3).

2.2.7 Yarmouth Soil

The Yarmouth was not encountered in the area of Ash Pond 2 as borings did not extend through the Smithboro. Historical borings in the northern portion of the site which encountered the Yarmouth were summarized previously (Hanson 2009). The Yarmouth Soil is described as the weathered zone on the



Kansan drift, but in some places it consists of accretionary deposits of fine sediment and organic material that accumulated in poorly drained areas on the surface of the Kansan deposits. The Yarmouth Soil was absent in Borings SB-18 and SB-04. Where encountered, the Yarmouth Soil ranged in thickness from 0.8 ft in Boring SB-13 to 5.1 ft in Boring SB-08.

The Yarmouth Soil (considered the deep water-bearing zone) possesses a moderate to moderately low hydraulic conductivity. The single-well permeability tests conducted by Hanson (2009) reported moderate hydraulic conductivity values ranging from 1.3×10^{-4} to 1.7×10^{-3} cm/sec, with a geometric mean of 4.4×10^{-4} cm/sec.

2.2.8 Lierle Clay Member

The Lierle Clay was not encountered in the areas of Ash Pond 2 because borings were terminated in the Smithboro Till. Historical borings in other portions of the CPS which encountered the Lierle were summarized previously (Hanson 2009). The Lierle Clay Member is the upper-most member of the Kansan Stage Banner Formation. It is described as an accretion-gley with clay, silt and some sand. It was encountered by Hanson (2009) in all but a few borings on site (i.e., SB-02 and SB-11), and was the target formation for completion ofsome boreholes. No boring penetrated the full thickness of the Lierle Clay.

The Lierle Clay has a very low hydraulic conductivity. The laboratory tests reported very low vertical hydraulic conductivity values ranging from 3.4×10^{-9} to 1.3×10^{-8} cm/sec, with a geometric mean of 3.6×10^{-9} cm/sec.

2.2.9 Bedrock

The site is underlain by the Pennsylvanian-age Bond Formation, which is characterized by Willman et al., (1975) as having a high percentage of limestone and calcareous clays and shales. Bedrock was not encountered in any borings advanced onsite so site specific information is not available.

2.3 Hydrogeology

One monitoring well was initially installed in 2010 near Ash Pond 2: however, a significant number of wells were installed around the CCR Units prior to and following their construction. In 2015 additional wells and piezometers were installed within and around Ash Pond 2 to meet requirements of the CCR Rule. This discussion focuses on Ash Pond 2 but utilizes historical results from wells at surrounding units, where appropriate, to evaluate the sitewide hydrogeology.

A summary of the current monitoring well networks and construction details is included in Table 1, and recent AECOM piezometers are summarized in Table 4. Since 2006 the hydrogeology of the site has been characterized and described through multiple investigations. This section discusses recently



collected information, focusing on the existing well network and piezometers installed in 2015 and 2016 around Ash Pond 2 as well as appropriate historical data from other units onsite.

2.3.1 Groundwater Occurrence and Elevations

Leachate is present within the ash, and groundwater is present within all geologic units at the CPS; however, the Hagarstown Beds are considered the uppermost aquifer (groundwater monitoring unit or Illinois EPA aquifer) at the site. Hydraulic conductivities of the Hagarstown Beds are generally one to two orders of magnitude higher that the surrounding materials and support the designation of the Hagarstown Beds (and in areas the Upper Vandalia Till) as the Uppermost Aquifer (groundwater monitoring unit). Near Ash Pond 2 (i.e., G402 and G403) the Hagerstown was either very thin or not observed in the boring; however, hydraulic conductivity tests in these wells indicate similar hydraulic conductivities to the Hagarstown Member, indicating that the weathered Upper Vandalia Member is likely part of the Uppermost Aquifer and hydraulicly connected in these areas.

Measured groundwater elevations typically range from about 602 ft near discharge zones along the edges of Coffeen Lake and the Unnamed Creek, to 623 ft in upgradient wells along an approximate center line of the site between the two lobes of the lake (i.e., groundwater divide). Based on the elevation of Coffeen Lake (normal pool elevaton of 590 ft), Unnamed Creek (~590 ft), southern discharge channel (~604 ft) and lowest observed elevation of the Hagarstown Beds (602.6 ft) it is likely that groundwater seeps exist along the shoreline of the lake and slopes near these drainage features. Groundwater elevations are generally consistent and appear to be controlled by discharge as seeps along the shoreline of Coffeen Lake. Geologic cross-sections illustrate the occurrence of the Hagarstown Beds in Figures 9 to 11.

A summary of groundwater elevations from existing wells is included in Table 5 and hydrographs for representative well locations are included in Appendix C. Table 6 summarizes water elevations from piezometers located within and adjacent to Ash Pond 2 and Ash Pond 1. Table 6 also includes elevations from AECOM piezometer (P012) which is screened below the impoundment in the Hagarstown Beds.

Ash leachate is present within Ash Pond 2 and the elevations within the impoundment are significantly above groundwater elevations measured outside the impoundment in the Hagarstown Beds (i.e., observation wells [OW] versus monitoring wells G401 to G405. Within the impoundment, measurements collected from OW-1 through OW-5 indicate the CCR porewater elevation ranges between 635 and 637 (Table 7), while wells outside and below the impoundment range from approximately 604 ft (G402) to 625 ft (P012), indicating the groundwater elevation in the Hagarstown Beds is generally 10 to 15 feet lower than those measured within the impoundment. The leachate appears to be perched above the lower permeability Loess Unit and/or Hagarstown Member gravelly clay till, which locally acts as an aquitard. Leachate head elevations decrease significantly across the impoundment berms and seeps

observed on the berms are likely derived from flow along the ash/aquitard interface. These seeps have been observed adjacent to well locations G404 and G405 along the north impoundment berm and G406 at the southwest corner of Ash Pond 2. The presence of a surface water control channel adjacent to and along the north berm, the ravine to the east, and the southern discharge channel all limit the lateral extent of ash leachate movement.

2.3.2 Groundwater Flow

Potentiometric surface maps prepared from elevation data measured in monitoring wells indicate groundwater elevations can be variable but flow directions are generally consistent. Groundwater generally flows from the center of the site both to the east to the Unnamed Creek, and westward towards Coffeen Lake. Near Ash Pond 1 flow appears to be radial based on mounding created by that unit. In the vicinity of Ash Pond 2 there appears to be a component of radial flow due to water elevations within the ash pond. However, flow within the underlying Hagarstown Beds appears to be east to the Unnamed Creek and south to the discharge flume (water elevation of approximately 604 ft). As shown in Figure 4, ash is in contact with the Hagarstown Member in two areas on the eastern side of the impoundement. Given the elevated heads measured in Ash Pond 2, the predominant flow of water from the ash into the Hagarstown Member likely occurs in these former ravines and migrates within the Hagarstown eastward, beneath the constructed berms, and discharges through seeps into the Unnamed Creek. As previously noted, seeps were also observed along the north and south sides of Ash Pond 2. The horizontal gradient between wells G405 and G404, as measured in May 2016 and November 2016, is 0.007. Representative potentiometric surface maps are shown in Figure 12 and Figure 13.

Construction of the Ash Landfill, Gypsum Stack Pond, and Recycle Pond required removal of the Hagarstown Beds, in effect removing the aquifer beneath the footprint of these units (Hanson 2016b). The groundwater flow dynamics beneath/around the Ash Landfill and GMF Units is affected by several factors, including: removal of the Hagarstown Member from beneath the Units; presence of the construction dewatering systems around the units; and the lateral variability of lithology within the Hagarstown Member. It is uncertain whether these constructed units significantly limit lateral groundwater flow, either by creating no flow zones or by capturing groundwater via their dewatering systems. Potentiometric contours included in Figures 12 and 13 project flow beneath the units.

2.3.3 Vertical Groundwater Gradients

Nested monitoring wells G405D / T408 / G45D and G406 / T409 / G46D were installed in August 2016 to determine the vertical extent of impacts near Ash Pond 2. Based on the water elevations measured in these wells in November 2016, vertical gradients are upward between the Vandalia and Hagarstown Beds and range from 0.009 to 0.42 ft/ft. Between the Vandalia and Smithboro Tills vertical gradients are



significantly downward, ranging from 1.46 to 2.44 ft/ft. A summary of vertical gradients is included in Table 8.

2.3.4 Groundwater Discharge to Coffeen Lake and the Unnamed Creek

The flow of groundwater to both the Unnamed Creek and Coffeen Lake (via the discharge flume) was estimated using a number of general conservative assumptions regarding creek flow and lake levels. Calculations are included in Appendix D, and the results are summarized as follows:

- Flow from Ash Pond 2 through the Hagarstown along the eastern edge of the impoundment is approximately 0.034 cubic feet per second (cfs), or 2 cubic feet per minute (cfm); and estimated flow in the Unnamed Creek based on its geometry is approximately 43.3 cfs (2,600 cfm).
- Flow from Ash Pond 2 from the Hagarstown into the discharge channel along the southern edge is approximately 0.032 cfs (1.9 cfm), and flow in the discharge channel based on average daily flow (based on the NPDES permit) is approximately 773 cfs (46,400 cfm).

2.4 Conceptual Site Model

The geology and hydrogeology described above is summarized and grouped into the following hydrostratigraphic units for the remaining discussion in this report:

- Ash Unit Saturated CCR within the various CCR Units
- Upper Confining Unit Low permeability clays and silts, including the Loess Unit and the upper clayey till portion of the Hagarstown Member
- Uppermost Aquifer (Groundwater Monitoring Zone) Thin (generally less than 3 feet), moderate permeability sand, silty sand, and sandy silt/clay units which include the Hagarstown Member and the upper Vandalia Member (where weathered).
- Lower Confining Unit Thick (generally greater that 15 feet), very low permeability sandy silt till or clay till that includes the unweathered lower Vandalia Member, Mulberry Grove Member (discontinous), and Smithboro Member.

Mounding of water within saturated ash in the impoundment creates a component of radial flow. The extent of this groundwater movement appears to be limited, as the elevated heads overlying the Upper Confining Unit dissipate across the Ash Pond 2 berms. Potentially impacted water from the seeps observed along the berms may partially infiltrate through the Upper Confining Unit and/or run off toward the Lake or Unnamed Creek.

The Uppermost Aquifer underlying Ash Pond 2 consists of the Hagarstown Member and the weathered (upper) portions of the Vandalia Member, which is being monitored to define the extent of CCR constituents derived from Ash Pond 2. The Uppermost Aquifer is confined except where the Hagarstown Member is exposed along the eastern side of the impoundment within the former ravine (Figure 4). Based



on hydraulic conductivity values (10⁻³ to 10⁻⁴ cm/sec) measured in the monitoring wells screened in the Hagarstown Member, groundwater at CPS has previously been classified as Class I in accordance with 35 IAC 620 (Hanson, 2009).

CCR within Ash Pond 2 is underlain by the Upper Confining Unit beneath the majority of the impoundment footprint. However, in former drainage features present prior to construction the saturated ash is in contact with the Hagarstown Beds and underlain by the Vandalia Member. Given the relatively high permeability of the Hagarstown Beds, leachate from Ash Pond 2 infiltrates into this unit, migrates through/under the eastern berm (where the Hagarstown Member is continuous below the berm) and discharges along the slope, as evidenced by the observation of seeps (Figure 4).

Groundwater within the Hagarstown Beds beyond the boundary of Ash Pond 2 flows predominantly to the east and south. Both the southern discharge flume and the Unnamed Creek intersect and cutoff the Hagarstown unit, eliminating further migration of potentially impacted groundwater. Impacted groundwater may also migrate to the north and northwest in the Hagarstown, potentially under the influence of the passive (gravity drain) underdrain system associated with the Recycle Pond and active underdrain system associated with the Landfill.

2.5 Potable Water Well Inventory

A potable water well inventory was completed in 2013 during the Phase I Hydrogeological Assessment (NRT, 2013). Public records were searched to identify water supply wells located within 2,500 feet of the unlined impoundments. The Coffeen property boundary is located in Township 7 North, Range 8 West, and the unlined impoundments are located in the southwest quarter of Section 11. The 2,500 foot boundary spans across Sections 10, 11, 14, and 15. All wells within Sections 10, 11, 14, and 15 are shown on Figure E1 and tabulated in Appendix E.

The following sources of information were queried to identify well locations.

- Illinois State Geological Survey's Illinois Water Well (ILWATER) Internet Map Service
- Illinois State Water Survey Domestic Well Database
- Illinois EPA's web-based Geographic Information System (GIS) files

Twenty-six water well records were identified within the four sections surrounding the unlined impoundments, and are numbered 1 through 26 on Figure E1. Based on state records there are two non-community water supply (NCWS) wells, one industrial/commercial well, ten monitoring wells (associated with the White and Brewer facility), and 13 farm/domestic water wells within the four section search area (Appendix E). The Coffeen Power Station does not have a water supply well. The two NCWS wells, points 25 and 26, are located within Sections 10 and 15 and outside of the 2,500 feet boundary of



the impoundments (Figure E1 and Appendix E). There are no maximum setback zones (as defined by 35 IAC 671) for these two NCWS wells.

All except one of the wells identified in the well search are east or west of Coffeen Lake, which is a hydraulic divide for potentially impacted groundwater. The only water well located between the east and west branches of the lake is well 22, which was reportedly installed in 1981 and completed in sand and gravel at a depth of 39 feet. This well was observed during the original Hydrogeologic Investigation and removed by the Contractor during construction of the Recycle Pond (Rhonald Hasenyager, Hanson Professional Services, Inc.; personal communication).

Public water supply (PWS) wells within a ten mile radius of the Coffeen CCP impoundments were identified via a search of the Illinois State Water Survey's Illinois Water Inventory Program (IWIP) database (not available on-line) by RAPPS (2009). Three wells belonging to the Village of Fillmore are located within the search radius, the closest one is approximately eight miles northeast of the impoundments.



3 GROUNDWATER QUALITY

3.1 Summary of Groundwater Monitoring

Groundwater monitoring at the Coffeen Power Station was initiated in 2006 prior to construction of the Gypsum Management Facilty and Landfill; however, consistent data collection near Ash Pond 2 began in November 2015 to comply with the CCR Rule. There are currently no IEPA monitoring requirements for Ash Pond 2. The following discussion presents an analysis of historical data collected from 2010 to 2012 (in well G402, formerly APW-2) and recent CCR data from November 2015 to 2016. Limited data from monitoring wells near the Landfill, Gypsum Stack Pond and Gypsum Recycle Pond are also included to evaluate historical trends. Groundwater data from monitoring wells near Ash Pond 1 are not included in this analysis.

Groundwater monitoring around Ash Pond 2 that was initiated in November 2015 included seven well locations. Sampling is conducted quarterly at 2 background wells and 5 downgradient wells for the following parameters:

Metals (totals)				
Antimony	Boron	Cobalt	Molybdenum	
Arsenic	Cadmium	Lead	Selenium	
Barium	Calcium	Lithium	Thallium	
Beryllium	Chromium	Mercury		
Inorganics (totals)				
Fluoride	Chloride	Sulfate	Total Dissolved Solids	
Field				
рН	Dissolved Oxygen	Specific Conductivity	Turbidity	
Oxidation/Reduction Potential	Temperature			

In August 2016, four additional wells (G406, G407, G45D, G46D) were installed to further define the extent of impacts from Ash Pond 2 and refine the GMZ proposed in the CAP. Groundwater samples were collected in August and November, 2016 and results are included in this analysis. Data for the expanded parameter list for the federal CCR sampling will be reported in accordance with the groundwater monitoring plan included with the Closure Plan for Ash Pond 2.

3.2 Groundwater Monitoring Results and Analysis

Analytical results for Ash Pond 2 wells from 2010 (APW-2) through October 2016, and from 2008 to 2016 for select wells around the SW Pond and Recycle Pond are summarized in Appendix F. Statistics



showing the minimum and maximum concentrations detected in the groundwater samples is included for each well in Table 9. Table 9 also includes a comparison of groundwater data from wells to the Groundwater Quality Standards for Class I: Potable Groundwater. The well locations are shown on Figure 3. Note these results include both dissolved and total analytes, and are compared to 35 IAC 620 Class I Standards. Where possible, emphasis is placed on dissolved analyses, but totals are used when no dissolved analysis was available.

Parameters that have been detected in groundwater at concentrations exceeding the Class I groundwater quality standards include the following: arsenic, boron, lead, manganese, sulfate, and total dissolved solids (reported as residue on evapoaration or ROE). A summary of recent exceedances in 2015 and 2016 follows for the parameters of concern. Time-series graphs for each of the groundwater parameters at the 7 monitoring wells around Ash Pond 2 and select wells around the SW Pond and Recycle Pond are included in Appendix F including all historical results for these locations.

<u>Arsenic</u>

Arsenic exceeded the Class I standard (0.010 mg/L) in well G405 (November 2015), and G402 (November 2015, and February and May 2016) which are located at the toe of Ash Pond 2 on the north and southeast side, respectively. Results from other wells on-site are either non-detect or at concentrations well below the standard. However, arsenic is naturally occurring in Illinois groundwater. State agencies (IDNR, IDPH & IEPA, 2001) have noted that "arsenic concentrations that vary dramatically over a relatively short distance indicates that local conditions dictate arsenic concentration in groundwater, and it is difficult to make regional generalizations." Therefore, it is equally dificcult to make large-scale source determinations because of the concentration variations over these reported short distances.

<u>Boron</u>

Boron exceeded the 2 mg/L Class I standard during all events at wells G401 and G405, located south and north of Ash Pond 2, respectively. Exceedances were also measured in G402, southeast of the unit and G404 and the nearby Recycle Pond well G275, located northeast of the unit. Recently installed deep wells (G45D and G46D) did not exceed the Class I standard, indicating the impacts do not extend vertically through the Vandalia Till. West of the unit, G403 exhibits very low concentrations of CCR parameters which may be related to the relatively thin Hagarstown unit at this location (~0.2 ft). The groundwater elevation at G403 is also the highest measured value around Ash Pond 2, implying the impacted water does not flow toward this location. The highest concentrations of boron in groundwater have been detected in G405 and range from 15 to 17 mg/L. This concentration is significantly higher than other wells located adjacent to the Pond, and may be related to the shallow depth and increased


thickness of the Hagarstown Beds observed at this location. As discussed previously it is unclear if the seeps along the northern berm are infiltrating and resulting in localized increases in concentration.

Ash Pond 2 was closed/capped in the mid-1980's and it is expected that groundwater has reached equilibrium with respect to Ash Pond 2 (both concentrations and heads). Long term data collection at G402 (APW-2) supports this conclusion as boron concentrations have remained relatively stable (Appendix F). Results from sampling at G275 indicate highly variable concentrations of boron which may be related to the well's proximity to the Unnamed Creek, which exhibits significant seasonal fluctuations in discharge. Increasing trends are visible in results from G271, G273, and G274 (located north of the unit between the Recycle Pond and Ash Pond 2), although concentrations are significantly below the standard. The slight increases in concentration at these locations may be attributed to the construction of the Recycle Pond. As discussed in Subsection 2.3.2, construction of the Recycle Pond has likely impacted the groundwater flow dynamics in the area. The removal of the Hagarstown Member and the installation of the passive dewatering system may be allowing the northerly migration of contaminants toward the Recycle Pond monitoring system wells. Additional data is necessary to establish long term trends in boron concetrations at other monitoring wells near Ash Pond 2.

Lead

Lead was detected above standard (0.0075 mg/L) adjacent to Ash Pond 2 in G402 and G405. Concentrations above standard were also reported in G275, G276, G278, and G280 near the GRP. However, with the exception of G402, additional samples collected at these locations did not contain elevated lead concentrations, suggesting exceedances at these locations were anomalous and likely related to turbidity.

Sulfate

Sulfate, like boron, is a primary indicator of CCR leachate, and exceeded the 400 mg/L Class I standard at Ash Pond 2 wells G401, G402, G405, G406 (located around the perimeter of the Pond) and G407 (located approximately 1,000 feet west of the Pond). It also exceeds the standard at G271, G273, G275 and G279 adjacent to the Recycle Pond and north of Ash Pond 2, and G153 adjacent to the SW Pond and northwest of Ash Pond 2. At well G274 exceedances occurred in December 2012 and January 2013, but recent results indicate concentrations are below the standard. Sulfate, when present with elevated boron concentrations, is presumed to be indicative of potential impacts from a CCR unit. The highest concentrations of sulfate are detected in G401, which is located on the south side of Ash Pond 2. Sulfate concentrations generally correlate with boron concentrations with the exception of G271, G273, G279, and G407 where elevated sulfate is present without a significant boron presence. Increasing trends of boron in G71, G273, and G279 suggest that sulfate concentrations may be partially in response to effects from Ash Pond 2, but limited data from G407 (installed in August 2016) does not allow a similar analysis.



Current data does not suggest wells G153 and G407 are impacted by Ash Pond 2; however, without more data a definitive source of the sulfate in G153 and G407 cannot be determined.

Increasing trends are apparent at wells G153, G271, and G272. As discussed previously, increasing concentrations in G271 and G272 may be naturally occurring background in conjunction with changes in groundwater flow and geochemical conditions following construction of the Recycle Pond. It is unclear why sulfate concentrations are increasing in G153; it may be due to similar effects following construction of the Landfill and SW Pond. Sulfate concentrations in this well have been elevated since it was installed in 2011. Background well G270 had increased concentrations following installation in 2008 until December 2014; however, since then the concentration has stabilized and declined. Remaining wells exhibit stable concentration trends, although more data is needed for CCR wells installed in 2015 (G401 to G405).

<u>Manganese</u>

Manganese concentrations exceeded the Class I standard of 0.15 mg/L at monitoring wells G402, G404, G405, and G406 during 2015 and 2016. Wells G401, G402, and G403 are part of the CCR Rule monitoring program and have not been sampled for manganese (not an Appendix III or Appendix IV parameter). Wells around the SW Pond G151, G152, and G153 have exceeded the standard at least once during 2015-2016 and concentrations of manganese around the Recycle Pond are highly variable, however only G270 has exceeded the standard during 2015-2016. In locations where elevated manganese concentrations are present with both elevated boron and sulfate, it is likely related to effects of the impoundment(i.e. changes in geochemical conditions) but not necessarily directly from the CCR leachate. However, in some cases (i.e. G151, G152, G153, G270, and G407) the lack of elevated concentrations of boron (and in some cases sulfate, i.e. G151, G152, and G270) in groundwater indicates that the elevated manganese concentrations may be primarily associated with with changing geochemical factors (following construction of nearby CCR units and only secondarily related to the impoundment. Time concentration plots indicate decreasing manganese trends in groundwater from G270 and G273, no other trends are apparent in monitoring wells evaluated for this analysis.

Total Dissolved Solids

TDS concentrations regularly exceed the Class I standard of 1,200 mg/L at wells G401, G402, G405, G406, and G407 around Ash Pond 2 and also adjacent to the SW Pond (G153) and the GRP (G273 and G275). TDS reflects concentrations of major ions in groundwater and in the wells listed above sulfate concentrations make up the largest component of the TDS exceedances. TDS concentrations mirror trends observed in sulfate concentrations, with slightly increasing trends observed in G153, G271, and G272.



<u>рН</u>

The pH measured in G401 (south of the unit) fell below the lower limit (6.5 standard units, SU) during sampling in 2015 and 2016. Groundwater in this well appears to be impacted (based on boron and sulfate concentrations) and results from recent surface and source water sampling indicates the CCR leachate is slightly acidic, and is likely the source of acidity causing low pH values in groundwater. Monitoring well G402, also located south of the unit, exhibits simalr pH measurements but does not exceed the lower pH standard.

3.3 Impacted Groundwater Discharge to Surface Water

Based on the groundwater discharge calculations discussed in 2.3.4, and maximum concentrations of boron and sulfate detected in groundwater, the concentration of boron and sulfate loading into Coffeen Lake and the Unnamed Creek were calculated to determine if they exceed surface water standards. Calculations are included in Appendix D, and indicate that approximately 12.5 kg of boron are discharged to the creek each day, resulting in a calculated concentration of 0.115 mg/L. The estimated sulfate concentration is 1.9 mg/L.

In the discharge channel to the south, approximately 1.6 kg of boron is discharged via groundwater baseflow each day, resulting in a calculated concentration of 0.01 mg/L; the concentration of sulfate in the discharge flume is estimated at 50.1 mg/L. In both discharge areas the resulting concentrations are significantly below standards. Recent surface water sampling (Table 10) also confirms that Coffeen Lake is not impacted by Ash Pond 2 because measured concentrations of sulfate (~55 mg/L) and boron (≤280 ug/L) are well below standards and similar to background groundwater concentrations measured elsewhere onsite (i.e., at background wells G200 and R201).



4 CONCLUSIONS

Based on extensive site investigation and monitoring, the site has been characterized and a detailed site conceptual model has been developed. In conjunction with the hydrogeologic investigation, a groundwater model has also been developed to predict the effect of ash pond closure via capping on groundwater quality. The groundwater model report is being submitted under separate cover.

CPS and Ash Pond 2 overlie a significant thickness of glacially-derived deposits. The unlithified deposits which occur at Ash Pond 2 include the following hydrstratigraphic/geologic units:

- Ash Unit Saturated CCR within the various CCR Units
- Upper Confining Unit Low permeability clays and silts, including the Loess Unit and the upper clayey till portion of the Hagarstown Member
- Uppermost Aquifer (Groundwater Monitoring Zone) Thin (generally less than 3 feet), moderate to high permeability sand, silty sand, and sandy silt/clay units which include the Hagarstown Member and the upper Vandalia Member (where weathered).
- Lower Confining Unit Thick (generally greater that 15 feet), very low permeability sandy, silt till, or clay till that include the unweathered Vandalia Member, Mulberry Grove Member (discontinous), and Smithboro Member.

CCR is underlain by the Upper Confining Unit in the majority of the Ash Pond 2 footprint and mounding of water within saturated ash in the impoundment creates a component of radial flow. However, the extent of this groundwater movement appears to be limited, as the elevated heads overlying the Upper Confining Unit dissipate across the Ash Pond 2 berms. Leachate seeps observed along the berms may partially infiltrate through the Upper Confining Unit and/or run off with surface water toward the Lake or Unnamed Creek. In localized areas on the east side of Ash Pond 2, former drainage features that ocurr within the pond footprint eroded through the Upper Confining Unit such that the CCR is in contact with the Hagarstown Beds. In these areas, CCR leachate appears to migrate in the Hagarstown Beds and discharge through seeps along the ravine into the Unnamed Creek to the east.

The uppermost aquifer underlying Ash Pond 2 consists of the Hagarstown Member and the weathered portions of the Vandalia Member, which is being monitored to define the extent of CCR constituents derived from Ash Pond 2. Groundwater at CPS has previously been classified as Class I Potable Resource Groundwater. The uppermost aquifer is confined except where the Hagarstown Member is exposed along the eastern-side of the impoundment within the former ravine.

Groundwater within the Hagarstown Beds beyond Ash Pond 2 flows predominantly to the east and south. There is limited groundwater flow from Ash Pond 2 toward the west due to a thinning or lower hydraulic



conductivity of the Hagarstown Beds. Based on hydraulic conductivities and vertical gradients, horizontal groundwater flow in the overlying clays and underlying tills are negligible. Groundwater flow occurs primarily in the more permeable zones within the Hagarstown Beds. However, migration of impacts is limited to the east and south by the presence of the Unnamed Creek, Coffeen Lake and a surface water discharge flume, all three of which are areas of groundwater discharge and act as hydraulic barriers and/or groundwater divides.

Exceedances of Class I groundwater quality standards are present in monitoring wells at several locations around Ash Pond 2 for arsenic. boron, lead, manganese, sulfate, and total dissolved solids. The exceedances of Class I groundwater quality standards for arsenic, lead, manganese, sulfate, and TDS at several locations may be only partially attributed to Ash Pond 2, and are primarily a result of natural geochemical variations or background concentrations, specifically in wells where boron concentrations are low. The concurrent detection of elevated boron and sulfate concentrations are key indicators of the presence of CCR constituents related to Ash Pond 2.

Boron concentrations are generally stable with the exception of wells G271, G272, and G274 (located between Ash Pond 2 and the Recycle Pond), which have shown slight increases. However, concentrations remain significantly below the standard (2 mg/L). Increasing trends measured at these wells are attributed to the construction of the GRP north of the wells and the reduction of unimpacted groundwater flow towards these wells, or increased flow from Ash Pond 2, due to the underlying drainage system at the GRP.

Given the current groundwater data and Site information, groundwater quality is expected to improve following closure, as capping will reduce the infiltration of water and leachate generation from Ash Pond 2. Because CCR will remain in Ash Pond 2, a groundwater monitoring plan is being submitted with this Closure and Post-closure Care Plan. I.



5 REFERENCES

- AECOM, 2016a. Revised 30% Closure Design Package for Coffeen Power Station, GMF Gypsum Stack Pond. 1/27/2016.
- AECOM, 2016b. 30% Design Data Report for Dynegy Coffeen Power Station; Ash Ponds Nos. 1 and 2 CCR Units.1/12/2016.
- AECOM, 2016c. Revised 30% Closure Design Package for Coffeen Power Station Ash Pond No. 2. April 16, 2016.
- Hanson, 2016a. Corrective Action Plan. Coffeen Power Station, Illinois Power Generating Company, Coffeen, Montgomery County, IL. April 19, 2016.
- Hanson 2016b. 40 CFR 257.60 Uppermost Aquifer Considerations, Gypsum Pond, Illinois Power Holdings LLC, Coffeen Energy Center, Montgomery County, Illinois, April 2016.
- Hanson, 2015. Ash Landfill Storm Water Runoff Pond G153 Assessment. Hanson Professional Services, Inc. January 15, 2015.
- Hanson, 2014. G153 Assessment. WPC Permit No. 2011-EB-1289, Coffeen Power Station, Illinois Power Generating Company, Coffeen, Montgomery County, Illinois, December 19, 2014.
- Hanson, 2009. Section 3: Hydrogeologic Investigation. Initial Facility Report, Hanson Professional Services, Inc. August 2009.
- Herzog, B.L., B.J. Stiff, C.A. Chenoweth, K.L. Warner, J.B. Sieverling, and C. Avery, 1994, Buried Bedrock Surface of Illinois: Illinois State Geological Survey map, scale 1:500,000.
- IDNR, 2014. Coffeen Lake State Fish and Wildlife Area Park Brochure. DNR 06/14 IOCI 0625-14
- Kolata, D.R., 2005, Bedrock Geology of Illinois: Illinois State Geological Survey map, scale 1:500,000.
- Natural Resource Technology, Inc., 2013. Phase I Hydrogeological Assessment Report. CoalCombustion Product Impoundments, Coffeen Energy Center, Montgomery County, IL.
- Rapps Engineering & Applied Science, November, 2009. Site Characterization and Groundwater Monitoring Plan For CCP Impoundments, Ameren Energy Generating Company, Coffeen Power Station, Montgomery County, Illinois.
- Willman, H.B., and J.C. Frye, 1970, Pleistocene Stratigraphy of Illinois: Illinois State Geological Survey, Bulletin 94, 204 p.
- Willman, H.B., J.C. Frye, J.A. Simon, K.E. Clegg, D.H. Swann, E. Atherton, C. Collinson, J.A. Lineback, T.C. Buschbach, and H.B. Willman, 1967, Geologic Map of Illinois: Illinois State Geological Survey map, scale 1:500,000.
- Willman, H.B., E. Atherton, T.C. Buschbach, C. Collinson, J.C. Frye, M.E. Hopkins, J.A. Lineback, and J.A. Simon, 1975, Handbook of Illinois Stratigraphy: Illinois State Geological Survey, Bulletin 95, 261 p.
- Zeizel, A.J., 1959, Geologic Report on the Ground-Water Geology at Fillmore, Sections 23 and 26, T8N., R2W., Montgomery County: Illinois State Geological Survey, Open File Report



FIGURES











20/2016_3-27-16 PM





- Background
- Compliance
- Detection

🖶 GMZ Study

🖶 Temporary

CCR MONITORED UNIT

DRAWN BY/DATE: MDM 10/11/16 REVIEWED BY/DATE:

JJW 10/13/16 APPROVED BY/DATE: SJC 12/20/16 HYDROGEOLOGIC CHARACTERIZATION REPORT ASH POND 2 COFFEEN POWER STATION COFFEEN, IL

THICKNESS OF HAGARSTOWN MEMBER SAND UNIT

G305

5 1.5

G306 5 1.9

> PROJECT NO: 2285 FIGURE NO: 6

300

SCALE IN FEET

0

600

Ν





2016 3:30:52 PN



2/20/2016. 3:31:20 PM









Dec 21, 2016 12:11pm PLOTTED BY: amilispaugh SAVED BY: acawrse Y:\Mapping\Projects\23\2380\CAD\2-0\Figure 10_Geologic Cross-Section B-B'.dwg Layout1 MAGES: MAGES:

B-B' CROSS-SECTION TRANSECT CLAY SOIL SAND SILT	DRAWN BY: AGC DATE: 11/30/16	CHECKED BY: NRK DATE: 12/20/16	APPROVED BY: SJC DATE: 12/20/16	DRAWING NO: Fig 10_Geologic Cross-Section B-B'	REFERENCE: SEE INFO BLOCK
WELL SCREEN WELL SCREEN G276 G276 G275 C274 G404 - B=1(d) - B=2(s) - COF-P010				HYDROGEOLOGIC CHARACI ERIZATION REPORT ASH POND 2	COFFEEN POWER STATION COFFEEN, IL
F-P007 B' G302 COF-P0 3-P005		PF	N R T ROJE	IATU ESOI ECH	IRAL JRCE NOLOGY I O.
SCALE IN FEET		F	IGUF 1	RE NO 10	D.



spaugh ure 11_ TED BY: (CAD\2-23 23 /Map Ne Cirk









ations mud - Author: touchman: Data/Itma: 19/99/9018-2-60-50 DM

TABLES

Table 1Summary of Existing Monitoring Well NetworksHydrogeologic Characterization ReportCoffeen Energy Center - Ash Pond 2

			Surface	ТОС				Top of	Bottom of		
			Elevation (ft	Elevation (ft	Predecessor or			Screen	Screen		
Well ID	Easting	Northing	MSL)	MSL)	prior name	Unit	Well Type	Elevation	Elevation	Interpreted Screened Unit	Comments
G301	2,515,582.97	872,234.82	620.27	622.65			MW	609.0	604.3	Upper Vandalia Till	
G302	2,516,214.19	872,252.95	617.95	620.04			MW	604.7	600.1	Upper Vandalia Till	
G303	2,516,639.65	8/1,382.14	619.10	622.02	AP-3		MVV	609.1	599.1	Hagarstown/Vandalia Till Contact	No continuous soil sampling
G304	2,515,519.74	871,397.69	623.46	626.72	AP-4	Ash Pond 1	MVV	613.5	603.5	Hagarstown Beds	
G305	2,515,199.36	871,156.33	622.54	625.55			MVV NAVA	609.1	604.3	Hagarstown Beds	
G306	2,516,120.41	871,140.98	622.84	625.72			IVIVV NAVA/	609.8	605.2	Hagarstown Beds	
G307	2,515,553.26	871,398.55	622.08	624.47			IVIVV	609.1	604.3	Hagarstown Beds	
G281	2,514,455.48	874,375.37	623.82	626.36			IVIVV NAVA/	608.31	603.66	Hagarstown Beds	
G401	2,515,014.04	872,510.57	610 56	642.37				600.67	500 G	Hagarstown beus	
G402	2,510,032.40	972,500.49	622.91	626.47	AP-2			600.6 610.7	590.0		
G403	2,514,010.03	873,000,01	613 10	615.67			IVIVV N/I\\/	606.68	601.03	Hagarstown Beds	
G404	2,515,335,67	873 996 63	620.90	623.63			Ν/\Λ/	611.89	607.14	Hagarstown Beds	
G406	2 514 702 38	872 521 34	621.86	621.86			Μ\Δ/	608.3	603.49	Hagarstown Beds	
G407	2 513 705 87	872 973 39	618.35	621.00			M\A/	604 57	599 74	Hagarstown Beds	
0407	2,010,100.01	072,070.00	010.00	021.02			PZ (Hvd.	001.07	000.14		
T408	2,515,314.91	873,999.36	621.09	624.08			conductivity only)	600.43	595.6	Vandalia Till	
	2 514 602 90	972 517 70	601.95			Ash Pond 2	PZ (Hyd.				
T409	2,514,095.09	072,517.79	021.00	625.01			conductivity only)	600.06	595.26	Vandalia Till (Sand Seam)	
G45D	2,515,322.23	873,998.03	620.94	623.81	G405D		PZ	589.06	579.42	Smithboro Till	
G46D	2,514,697.78	872,519.70	621.91	625.24	G406D		PZ	580.3	570.65	Smithboro Till	
OW-1	2,514,754.56	872,875.48	639.80	641.14			Ash Pond PZ	634.74	619.74	Ash	
OW-2	2,515,525.87	873,761.43	639.70	641.34			Ash Pond PZ	629.84	614.84	Ash	
OW-3	2,515,824.43	873,147.40	640.40	641.78			Ash Pond PZ	631.28	616.28	Ash	
OW-4	2,516,277.90	873,535.97	644.70	damaged			Ash Pond PZ	629.7	604.7	Ash	
OW-5	2,516,466.21	872,770.83	637.70	638.8			Ash Pond PZ	627.7	602.7	Ash	
B-2(s)	2,516,615.41	873,650.85		639.44			PZ			Fill	in east-side berm
B-1(d)	2,516,615.78	873,647.93		638.7			PZ			Fill	in east-side berm
G101	2,514,215.00	876,575.00	625.27	627.6			MW	609.59	604.95	Hagarstown Beds	
G102	2,514,537.00	876,554.00	625.70	630.96	MW03S		MVV	613.68	608.92	Hagarstown Beds	
G103	2,514,500.93	876,200.00	630.99	633.8	0101		IVIVV NAVA/	615.11	610.32	Hagarstown Beds	
R104	2,514,503.40	875,857.80	629.03	632.84	G104			614.44	609.71	Hagarstown Beds	
G105	2,514,509.03	075,500.00 975 150 00	629.20	632.08				614.02	600.42	Hagarstown Beds	
G100	2,514,515.00	874 004 07	020.39	620.22				612.02	609.43	Hagarstown Bods	
G107	2,514,558.50	874,994.07	627.50	630.23			IVIVV N/I\/	610.68	606	Hagarstown Beds (Silt)	
G100	2,514,240.40	874 970 13	627.20	620.76			Ν/\Δ/	611.81	607.27	Hagarstown Beds	
G110	2,514,157.00	875 015 38	627.02	629.65			Μ\Δ/	611.01	607.43	Hagarstown Beds	
G110	2,513,981,89	875 058 47	627.24	629.9			M\\\/	612.63	608.09	Hagarstown Beds	
G119	2,513,910.00	875.675.00	628.85	631 55			MW	611.56	607.02	Hagarstown Beds	
G120	2,513,905.81	875.845.00	629.30	631.87			MW	614.2	609.68	Hagarstown Beds	
G121	2.513.910.00	875.965.00	629.57	632.83		Landfill	MW	612.78	608.1	Hagarstown Beds	
G122	2.513.905.00	876.080.00	629.86	632.69			MW	613.35	608.81	Hagarstown Beds	
G123	2,513,905.00	876,190.00	630.13	632.96			MW	609.19	604.67	Hagarstown Beds	
	, ,	,									
G124	2,513,900.48	876,305.00	630.42	633.39			MW	614.44	609.91	Hagarstown (Silt)/Vandalia Till Contact	
G125	2,513,900.00	876,410.00	630.68	633.51			MW	613.65	609.12	Hagarstown Beds	
G126	2,513,878.29	875,049.31	622.96	625.39			MW	610.07	605.53	Hagarstown Beds (Sandy clay)	
T127	2,513,915.00	875,360.00	628.07	630.96			MW	610.54	606	Hagarstown Beds	
T128	2,513,915.00	875,510.00	628.44	630.93			MW	611.91	607.4	Hagarstown Beds	
TA31	2,513,856.77	876,542.25	623.89	626.55			MW	608.8	604.32	Hagarstown Beds	
TA32	2,513,605.19	877,532.57	618.93	621.42			MW	607.62	603.25	Hagarstown Beds	
TA33	2,513,248.73	876,605.45	622.51	625.27			MW	610.28	605.62	Hagarstown Beds	
TA34	2,513,466.73	875,906.10	624.10	626.52			MW	613.18	608.69	Hagarstown Beds	



G151	2,513,805.90	875,023.70	622.82	625.93		MW	607.48	602.98	Hagarstown Beds	
G152	2,513,894.50	874,687.50	623.06	626.52		MW	609.47	604.97	Hagarstown Beds	
G153	2,513,532.70	874,532.70	623.30	626.35	SW Pond	MW	607.4	602.96	Hagarstown Beds	
G154	2,513,243.10	874,978.40	623.52	626.35		MW	609.26	604.76	Hagarstown Beds	
G155	2,513,501.80	875,127.70	622.89	625.86		MW	607.8	603.31	Hagarstown Beds	
G200	2,515,650.00	877,930.60	624.20	625.94		MW	612.01	607.22	Hagarstown Beds	
R201	2,514,842.00	877,925.30	624.02	626.34	G201	MW	611.75	607.36	Hagarstown Beds	
T202	2,514,895.00	876,699.40	626.22	628.63		MW	613.95	609.57	Hagarstown Beds	
G205	2,515,914.90	875,550.20	622.15	624,45		MW	612.11	607.62	Hagarstown Beds	
G206	2.514.669.20	875,103,90	630,54	632.82		MW	613.03	608.62	Hagarstown Beds	
G207	2.514.837.90	875,166,40	630.61	633.21		MW	612.37	607.84	Hagarstown Beds	
G208	2.514.993.60	875.231.50	630.57	633.16		MW	613.04	608.51	Hagarstown Beds	
G209	2.515.149.60	875.298.20	630.57	632.91		MW	612.83	608.29	Hagarstown Beds	Low recovery in interval with Hagarstown
G210	2.515.299.00	875,359,70	630.48	632.99		MW	611.09	606.55	Hagarstown Beds	
0210	2,010,200.00	010,000110	000110	002.00	Gypsum Stack		011.00	000.00	Hagarstown (Sandy Clay)//andalia Till	
G211	2,515,449.10	875,424.50	630.31	632.64		Ν/\\\/	612.97	608 43	Contact	
G217	2 515 583 00	875 486 50	630 50	632.80			613.85	600.40	Hagarstown Beds	
G212	2,515,505.00	875 544 40	630.34	622.09			613.50	609.05	Hagarstown Bods	
G213	2,515,725.50	875 668 00	620.34	622.01			612.64	609.05	Hagarstown Bods	
G214	2,515,900.80	875,000.00 875,910,20	620.49	032.00			611.07	606.69	Hagarstown Boda	
G215	2,515,971.60	075,010.20 975,076,10	630.46	633.06			610.24	606.66 605.86	Hagarstown Beda	
G216	2,515,968.50	875,976.10	630.28	632.76			610.24	605.86	Hagarstown Beds	
G217	2,515,963.00	876,185.60	630.67	633.1		IVIVV NAVA	610.18	605.79	Hagarstown Beds	
G218	2,515,962.20	876,380.90	630.64	633.11		MVV	610.31	605.87	Hagarstown Beds	
MW02S	2,513,210.04	876,408.86	624.10	sealed		MW	613.76	608.98		paired with deep well
MW04S	2,514,450.58	877,999.73	622.40	625.79		MW	612.57	608.14	Hagarstown Beds	
MW05S	2,513,285.49	878,175.59	622.60	625.8		MW	609.94	605.19	Hagarstown Beds	paired with deep well
MW06S	2,513,189.40	879,021.15	623.10	626.14		MW	612.06	607.48	Hagarstown Beds	paired with deep well
MW07S	2,514,397.54	879,181.12	624.50	627.54		MW	614.59	610.71	Hagarstown Beds	paired with deep well
MW08S	2,514,478.83	879,776.62	624.70	627.9		MW	613.19	608.7	Hagarstown Beds	paired with deep well
MW09S	2,515,666.24	879,684.90	624.60	627.46		MW	613.39	608.98	Hagarstown Beds	paired with deep well
MW10S	2,515,914.37	878,250.50	621.20	624.22	Hydrogeologic	MW	609.92	605.44	Hagarstown Beds	paired with deep well
MW11S	2,515,971.16	876,749.44	622.00	625.08	Characterization	MW	613.11	608.37	Hagarstown Beds	paired with deep well
MW12S	2,515,900.54	875,520.08	622.20	625.21	Report (historical)	MW	611.59	607.02	Hagarstown Beds	paired with deep well
MW13S	2,513,925.29	874,695.66	622.70	625.89		MW	611.27	606.47	Hagarstown Beds	paired with deep well
MW14S	2,514,125.95	875,737.78	624.60	sealed		MW	612.34	607.58	Hagarstown Beds	
MW15S	2,515,076.27	875,971.13	623.80	sealed		MW	609.39	604.64	Hagarstown Beds	paired with deep well
MW16S	2,515,087.98	877,355.14	626.10	629.37		MW	611.51	606.69	Hagarstown Beds	paired with deep well
MW17S	2,515,084.76	878,658.54	627.10	630.47		MW	613.08	603.54	Hagarstown Beds	paired with deep well
MW18S	2,513,745.20	878,604.67	625.60	628.66		MW	614.29	609.81	Hagarstown Beds	
MW20S	2,515,876.50	874,228.00		622.84		MW				
G270	2,514,996.84	874,801.92	622.92	625.92		MW	609.79	605	Hagarstown Beds	
G271	2,515,517.12	874,239.38	622.89	625.57		MW	612.93	608.58	Hagarstown Beds	
G272	2,515,744.99	874,234.83	620.72	623.81		MW	611.61	606.74	Hagarstown Beds (Silt)	
G273	2,515,975.49	874,235.24	620.17	623.02		MW	611.09	605.61	Hagarstown Beds	
G274	2,516,195.60	874,239.25	621.67	624.04		MW	608.77	604	Hagarstown Beds	
G275	2,516,375.86	874,298.94	616.14	618.26	Recycle Pond	MW	607.92	603.52	Hagarstown Beds	
G276	2,516,358.83	874,438.60	629.14	632.		MW	606.73	601.92	Hagarstown Beds	Very thin sand
G277	2,516,370.51	874,581.80	620.79	623.08		MW	606.5	602.02	Hagarstown Beds	
G278	2,516,200.66	874,875.37	628.85	631.19		MW	609.92	605.15	Hagarstown Beds	
G279	2,516.245.60	875.028.06	629.19	632.04		MW	606.79	602.4	Hagarstown Beds	
G280	2,515.679.48	875.045.11	622.95	625.85		MW	610.16	605.32	Hagarstown Beds	
	, .,	-,	2.5							

Notes:

Ground surface elevations based on information included on boring logs when the well was installed, and top of PVC elevations based on survey completed in Fall 2015.

[O:RJH 9/2016, C: NRK 10/10/2016, O: NRK 11/8/2016, C: JJW 11/10/2016]



Table 2

Summary of Laboratory Hydraulic Conductivity in Confining Units Hydrogeologic Characterization Report Coffeen Energy Center - Ash Pond 2

Laboratory Tests											
	Approximate	Hydraulic									
	Sample	Conductivity									
Well/ Soil Boring ID	Elevation (ft)	(cm/sec)	Interpreted Unit								
COF-B001	613.0	1.3E-08									
COF-B003	606.5	2.2E-07	Loess - Upper Confining								
COF-B004	610.5	5.0E-07	Unit								
COF-B007	615.0	7.0E-08									
G	eometric Mean	1.0E-07									
G46D	599.2	4.5E-06									
T408	597.6	1.5E-07									
SB-12	577.7-572.7	6.8E-09	Vandalia Till								
SB-13	598-593	7.0E-09									
SB-18	603.5-603	8.8E-09									
G	eometric Mean	4.9E-08									
SB-09	598.5-598	1.9E-06	Mulborny Crove Silt								
SB-16	589-588.5	1.6E-06	Muberry Grove Silt								
G	eometric Mean	1.7E-06									
G45D	586.4	1.0E-07									
G46D	578.9	2.1E-08	Smithboro Till								
SB-07	572-571.5	1.1E-09									
G	eometric Mean	1.3E-08									
SB-19	569-564	3.4E-09	Doop Confining Unit								
SB-16	548-547.5	1.3E-08									
G	eometric Mean	6.6E-09									

[O:NRK 10/26/2016]



Table 3Summary of Field Hydraulic Conductivity TestsHydrogeologic Characterization ReportCoffeen Energy Center - Ash Pond 2

		Method	Method			Well	Approximate	
Well ID	Unit	(fb)	(rb)	K (fh)	K (rh)	Geometric	Screened	Interpreted Unit
		(11)	(11)			Mean	Elevation (ft)	
					Upper-mo	ost Aquifer		
R104		KGS	B-R	7.0E-05	2.8E-04	1.4E-04	614.4-609.7	
G105		KGS	KGS	1.5E-04	5.7E-05	9.2E-05	613.2-608.4	
G106		B-R	B-R	4.0E-05	7.4E-04	1.7E-04	614.0-609.4	
G107	II	KGS	KGS	6.3E-05	8.9E-05	7.5E-05	613.9-609.3	
G110	ndf	KGS	KGS	4.7E-05	2.0E-05	3.1E-05	612.0-607.4	Hagarstown Beds
G119	La	KGS	KGS	8.6E-05	8.2E-05	8.4E-05	611.6-607	
G120		low water e	levation; no t	est conduct	ed		614.2-609.7	
G125		KGS	KGS	4.8E-05	4.1E-05	4.4E-05	613.7-609.1	
T127		KGS	KGS	1.2E-03	1.7E-05	1.4E-04	610.5-606	
				Unit Geom	netric Mean	8.5E-05		
T202		KGS	KGS	4.5E-04	5.5E-04	5.0E-04	614.0-609.6	
G206	7	B-R	KGS	3.0E-04	1.6E-04	2.2E-04	613.0-608.6	
G208	one	KGS	KGS	6.0E-05	2.1E-05	3.5E-05	613.0-608.5	
G209	L P	KGS	KGS	2.0E-04	1.6E-04	1.8E-04	612.8-608.3	
G210	sur	KGS	KGS	5.0E-04	4.8E-04	4.9E-04	611.1-606.6	Hagarstown Beds
G212	jyp.	KGS	KGS	1.3E-04	1.8E-04	1.5E-04	613.9-609.3	
G215	U	KGS	KGS	5.0E-04	3.5E-04	4.2E-04	611.1-606.7	
G218		KGS	KGS	4.1E-04	4.1E-04	4.1E-04	610.3-605.9	
				Unit Geom	netric Mean	2.3E-04		
G270		KGS	KGS	5.5E-04	4.8E-04	5.1E-04	609.8-605.0	
G271	pu	KGS	KGS	1.6E-04	1.1E-03	4.2E-04	612.9-608.6	Hagarstown Beds
G273	Ро	KGS	KGS	1.0E-03	8.3E-04	9.1E-04	611.1-605.6	
G276	cle	low v	water				606.7-601.9	Hagarstown Beds, v. thin
G279	ecy	KGS	KGS	1.7E-03	1.5E-03	1.6E-03	606.8-602.4	
G280	Å	KGS	KGS	1.3E-03	1.3E-03	1.3E-03	610.2-605.3	Hagarstown Beds
G281		KGS	KGS	2.1E-03	8.9E-04	1.4E-03	608.3-603.7	
	_			Unit Geom	netric Mean	9.0E-04		
G301	1 pr	KGS	KGS	2.7E-04	5.0E-04	3.7E-04	609-604.3	Upper Vandalia Till
G302	Por	KGS	KGS	4.9E-04	6.3E-04	5.6E-04	604.7-600.1	
G303	sh	KGS	KGS	5.6E-05	3.1E-05	4.2E-05	609.1-599.1	Hagarstown/Vandalia Till Contact
G304	A	KGS	KGS	8.9E-04	1.0E-03	9.4E-04	613.5-603.5	Hagarstown Beds
0.404						3.0E-04	COO 7 COO 7	
G401	q 2	B-R	B-R	1.8E-04	2.8E-04	2.2E-04	608.7-603.7	Hagarstown Beds
G402	ouc	KGS	KGS	4.5E-04	1.9E-04	2.9E-04	600.6-590.6	Upper Vandalla Till
G403	ЧЬ	KGS	KGS	4.3E-03	7.2E-03	5.6E-05	610.7-606.0	Hagaistown Beds, v. thin
G404	As	KGS	KGS		3.0E-04	4.0E-04	611.0.607.1	Hagarstown Beds
6405		KG3	KG3	9.0E-04	9.7 E-04	9.7E-04	011.9-007.1	
G153	SW Pond	KGS	KGS	2 5E-04		2.7E-04	607 5-603 0	Hagarstown Beds
0100		Ree	1.00	Unit Geor	netric Mean	3.7E-04	007.0 000.0	Hagaistown Deas
MW03S	st.	B-R	B-R	6.0E-04	1.1E-03	8.1E-04	613.7-608.6	
MW04S	ves	B-R	B-R	1.3E-03	8.0E-04	1.0E-03	612.6-607.6	
MW10S	L .	B-R	B-R	8.0E-04	8.0E-04	8.0E-04	610.9-604.9	
MW13S	geo	B-R	B-R	1.0E-03	2.0E-04	4.5E-04	611.3-606.1	
MW14S	dro	B-R	B-R	1.0E-03	5.0E-04	7.1E-04	612.4-607.2	Hagarstown Beds

				Unit Geome	tric Mean	5.4E-04			
MW17S	20	B-R	B-R	5.8E-04	5.5E-04	5.6E-04	613.1-603		
MW16S	60	B-R	B-R	6.0E-04	4.5E-04	5.2E-04	611.5-606.3		
MW15S	НУ	B-R	B-R	1.5E-04	8.1E-05	1.1E-04	609.3-604.2		

	Lower Confining Unit (Vandalia and Smithboro Till)												
T408	12	KGS	KGS	2.15E-06 7.50E-08	9.02E-07	600.4-595.2	Vandalia Till						
T409	onc	KGS	KGS	3.6E-05 3.20E-05	3.41E-05	600.1-594.9	Vandalia Till (sand seam)						
G405D	ЧЧ	KGS	KGS		4.90E-07	589.1-579	Smithboro Till						
G406D	As	KGS	KGS		4.00E-08	580.3-570.3							
	Unit Geometric Mean 5.55E-06												

Notes:

[O:RJH 9/2016, C: KLT 12/2016)

fh = Falling head test

rh = Rising head test

Hydraulic Conductivity tests analyzed using Aqtesolv® Pro version 4.50 (HydroSOLVE, Inc.)

Test Methods

- B-R Bouwer and Rice, 1976. "A Slug Test for Determining Hydraulic Conductivity of Unconfined Aquifer with Completely or Partially Penetrating Wells", Water Resources Research v.12, no. 3. American Geophysical Union, Washington, DC. pp. 423-428.
- KGS Hyder, Z., J.J. Butler, C.D. McElwee, and W. Liu, 1974. "Slug tests in partially penetrating wells", Water Resources Research, v. 30, no. 11. American Geophysical Union, Washington, DC. pp. 2945-2957.



Table 4Summary of AECOM Piezometer ConstructionHydrogeologic Characterization ReportCoffeen Energy Center - Ash Pond 2

Well ID	Unit	Coord	linates	Ground	Location	Well	Top of PVC	Total Depth	Top of Screened	Bottom of	
weirid	Unit	Northing	Easting	Elevation (ft)	Location	Туре	(ft)	(ft)	Inteval (ft bgs)	Interval (ft bgs)	Screened Interval
COF-P000		871590	2516694	634.9	Crest	PZ	639.3	27.9	619.9	609.9	Fill and Loess/Clay
COF-P001		871488	2516735	615.6	Toe	VWP	VWP	12.0	603.6	603.6	Clay/ Silt
COF-P002		871459	2516042	635.4	Crest	ΡZ	639.2	39.3	609.9	599.9	Hagarstown Beds
COF-P003	Ach Dond 1	871786	2515150	635.7	Crest	ΡZ	640.3	48.0	607.2	592.2	Hagarstown Beds
COF-P005	ASII FUIIU I	872198	2516077	635.0	Crest	ΡZ	639.6	23.2	620.0	615.0	Fill/ Clay Contact
COF-P006		872106	2516696	635.1	Crest	ΡZ	639.7	47.4	605.1	590.1	Hagarstown Beds
COF-P007		872235	2516081	617.5	Toe	VWP	VWP	15.0	602.5	602.5	Hagerstown Beds
COF-P008		871814	2515103	625.1	Toe	VWP	VWP	19.0	606.1	606.1	Hagerstow Beds
COF-P009		873894	2515111	638.0	Crest	ΡZ	642.6	33.0	618.0	608.0	Clay/ Hagarstown Beds
COF-P010	Ach Dond 2	873505	2516621	635.2	Crest	ΡZ	639.1	41.5	611.7	596.7	Clay/ Hagarstown Beds
COF-P012	ASH POHd 2	872691	2515086	642.2	Interior (ash)	ΡZ	646.4	43.3	609.7	599.7	Hagarstown Beds
COF-P014		872538	2515412	635.0	Crest	PZ	639.5	45.4	606.5	596.5	Hagarstown Beds

[O:AECOM, 2016]

Notes:

1. For standpipe piezometers, stickup and WL measurement referenced to top edge of yellow protector cover with hinged cap open.

2. Total Depth = Approx. bottom of screen for standpipe piezometers, or installed depth for VWPs.

3. VWP = vibrating wire peizometer installed at locations not accessible with drill rig.

4. Piezometers COF-P004, -P011, and -P013 were planned but not installed due to access issues.



Table 5 Groundwater Elevations in Monitoring Wells: 2015 - 2016 Hydrogeologic Characterization Report Coffeen Energy Center - Ash Pond 2

Date	G101	G102	G103	G105	G106	G107	G108	G109	G110	G111	G119	G120	G121	G122	G123	G124	G125	G126	G151	G152
January-15	614.48	619.18	620.82	621.95	620.45	619.23	618.42	617.78	616.76	615.93	615.64	612.75	613.63	610.79	610.84	615.27	617.83	615.22	NM	NM
April-15	618.87	622.06	622.58	623.73	622.19	620.85	620.31	619.71	618.60	617.48	615.86	613.43	614.63	615.94	612.41	617.85	620.45	616.45	NM	NM
July-15	618.53	622.12	621.70	622.72	621.43	620.15	621.22	620.41	619.55	618.03	616.55	613.47	614.09	615.26	612.76	618.25	620.71	616.34	615.43	616.47
October-15	617.15	622.02	620.69	621.65	620.50	619.10	618.92	618.12	617.70	616.79	615.31	612.94	613.31	614.39	611.89	617.27	619.66	614.13	614.86	614.06
November-15	612.95	618.96	NM	NM	619.32	NM	NM	NM	616.55	NM	NM	612.37	NM	NM	NM	NM	614.60	NM	NM	NM
February-16	618.46	624.04	NM	NM	621.55	NM	NM	NM	617.88	NM	NM	613.06	NM	NM	NM	NM	619.95	NM	614.88	616.61
February-16	617.62	623.35	621.29	622.73	621.68	620.70	619.16	618.58	617.25	616.62	616.00	612.89	613.65	615.17	611.58	616.22	619.30	615.80	NM	NM
May-16	618.89	625.34	NM	NM	622.11	NM	NM	NM	618.53	NM	NM	613.37	NM	NM	NM	NM	620.22	NM	615.58	617.37
May-16	618.79	625.33	623.42	623.62	622.00	620.74	620.15	619.58	618.57	617.44	615.85	613.17	614.86	616.82	611.76	616.78	620.21	616.54	NM	NM
June-16	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
July-16	618.44	623.92	NM	NM	620.62	NM	NM	NM	617.64	NM	NM	612.87	NM	NM	NM	NM	621.53	NM	NM	NM
Date	G153	G154	G155	G200	G205	G206	G207	G208	G209	G210	G211	G212	G213	G214	G215	G216	G217	G218	G270	G271
January-15	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
April-15	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
July-15	615.93	614.85	614.45	623.35	NM	621.87	622.31	622.76	622.37	621.88	621.79	621.77	621.55	618.83	618.43	NM	618.10	618.82	618.88	616.57
October-15	614.45	612.24	613.51	621.05	NM	620.69	620.72	620.62	620.56	619.83	619.00	620.76	620.21	617.56	616.56	NM	616.71	616.93	616.07	614.12
November-15	NM	NM	NM	621.66	NM	619.27	NM	NM	620.06	NM	NM	618.54	NM	NM	616.38	NM	NM	617.11	621.06	613.77
February-16	615.93	615.65	614.77	623.29	NM	621.92	622.18	622.06	622.26	621.68	622.03	621.99	621.21	617.95	618.31	618.61	618.20	619.05	622.94	615.87
February-16	NM	NM	NM	622.93	620.32	622.36	NM	NM	622.44	NM	NM	621.85	NM	NM	618.08	NM	NM	619.15	622.67	615.49
May-16	616.90	617.11	615.79	622.52	620.48	622.30	622.55	622.63	622.74	622.41	622.39	622.04	621.59	619.25	619.45	619.62	619.03	620.10	622.77	616.05
May-16	NM	NM	NM	623.13	NM	NM	NM	NM	622.69	NM	NM	622.08	NM	NM	619.27	NM	NM	620.01	623.02	616.19
June-16	NM	NM	NM	NM	NM	620.51	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
July-16	NM	NM	NM	622.82	NM	621.71	NM	NM	621.52	NM	NM	620.89	NM	NM	617.10	NM	NM	618.01	617.73	616.62
Date	G272	G273	G274	G275	G276	G277	G278	G279	G280	G281	G301	G302	G303	G304	G305	G306	G307	G401	G402	G403
January-15	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
April-15	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
July-15	614.62	611.18	611.08	604.26	604.00	NM	NM	607.48	618.68	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
October-15	612.56	610.41	610.06	NM	NM	NM	NM	608.14	614.54	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
November-15		611.82		NM	603.25	NM	NM	607.80	618.45	619.56	616.51	610.74	616.70	623.78	NM	NM	NM	607.82	604.02	621.81
February-16	614.68	613.26	610.14	604.66	603.71	NM	606.40	609.16	621.37	621.21	NM	NM	NM	NM	NM	NM	NM	608.14	604.90	621.78
February-16		612.59		NM	603.77	NM	NM	609.01	621.15	NM	617.21	613.14	617.87	624.07	618.35	619.55	NM	NM	NM	NM
May-16	615.21	612.83	610.49	605.12	604.71	603.84	607.02	610.17	621.94	620.93	616.75	614.60	617.97	623.91	NM	NM	NM	608.00	605.18	621.76
May-16	NM	613.18	NM	NM	604.72	NM	NM	610.21	621.95	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM
June-16	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	NM	616.28	615.11	NM	NM	NM	NM
July-16	NM	611.27	NM	NM	604.92	NM	NM	606.94	618.21	620.30	614.65	608.16	614.92	626.72	618.12	619.07	624.42	608.47	604.33	622.16
							-													
Date	G404	G405	R104	R201	T127	T128														
January-15	NM	NM	623.03	NM	615.65	614.73														
April-15	NM	NM	624.77	NM	616.04	614.89														
1 1 4 5	N 1 N 4	N 18 4	004.00	000.07	040.04	045 40														

April-15	INIVI	NIVI	624.77	NIM	616.04	614.89
July-15	NM	NM	624.06	623.27	616.04	615.40
October-15	NM	NM	621.69	619.94	615.66	614.67
November-15	611.67	618.85	621.34	622.44	615.91	NM
February-16	611.58	618.90	624.11	623.40	616.04	614.91
February-16	NM	NM	622.37	623.14	615.96	NM
May-16	611.46	618.99	624.89	622.81	616.15	NM
May-16	NM	NM	623.95	623.39	616.12	614.99
June-16	NM	NM	NM	NM	NM	NM
July-16	611.67	618.51	623.65	622.36	615.96	NM

<u>Notes:</u> NM= Not measured



Table 6Groundwater Elevations in Piezometers: 2015 - 2016Hydrogeologic Characterization ReportCoffeen Energy Center - Ash Pond 2

Water Elevation (feet)												
PZ ID	COF-P000	COF-P001	COF-P002	COF-P003	COF-P005	COF-P006	COF-P007	COF-P008	COF-P009	COF-P010	COF-P012	COF-P014
Date				Ash F	ond 1			Ash P	ond 2			
8/29/2015	616.97	614.30	624.90	620.38	620.62	612.94	611.40	622.55	621.50	605.62	625.43	613.57
10/5/2015	618.17		624.69	619.93	621.42	612.90			621.40	607.22	624.91	613.29
10/30/2015	617.95	613.29	624.68	619.82	622.15	612.72	610.32	622.11	620.73	607.99	624.53	613.07
11/23/2015	619.15	614.66	625.80	621.06	622.60	612.81	612.14	623.32	623.02	608.98	624.95	613.31
12/23/2015	619.57	614.77	626.19	622.30	622.50	613.22	614.21	623.80	624.31	609.18	625.82	614.08
1/18/2016	619.64	615.03	625.89	620.74	621.89	612.56	614.60	623.27	622.76	610.59	626.04	613.86

Notes:

-- No data available

[O:AECOM, 2016]



Table 7Ash Pond 2 Leachate Elevations in 2009Hydrogeologic Characterization ReportCoffeen Energy Center - Ash Pond 2

		Leachate Elevations (ft)											
	Date												
Well ID	4/1/2009 ¹	5/29/2009	6/6/2009	8/6/2009	9/21/2009	11/9/2009							
OW-1	619.83	636.87	636.59	635.48	635.10	636.15							
OW-2	615.14	636.62	636.69	635.24	634.92	635.98							
OW-3	616.68	636.57	636.43	635.26	634.90	635.90							
OW-4		638.47											
OW-5	609.05	636.48	635.81	635.18	634.85	635.77							

Notes:

[O:Hanson, 2016]

¹ Measurement collected after pumping of Ash Pond 2.

-- No leachate elevation data



Table 8Summary of Vertical Hydraulic GradientsHydrogeologic Characterization ReportCoffeen Energy Center - Ash Pond 2

		Screen		Groundwater	Vertical
Well ID	Date	Elev. (ft) ¹	Formation	Elevation (ft MSL)	Gradient ²
G405	11/12/2016	609.515	Hagarstown Beds	618.48	-0.000
T408	11/12/2016	598.015	Vandalia Till	618.58	-0.009
T408	11/12/2016	598.015	Vandalia Till	618.58	2.44
G45D	11/12/2016	584.24	Smithboro Till	584.91	2.44
G406	11/12/2016	605.895	Hagarstown Beds	612.51	0.42
T409	11/12/2016	597.66	Vandalia Till	615.98	-0.42
T409	11/12/2016	597.66	Vandalia Till	615.98	1 46
G46D	11/12/2016	575.475	Smithboro Till	583.59	1.40

Notes:

1. Center of screen

[O:NRK, 12/2016, C:JJW 12/2016]

2. Based on dates when both wells were sampled, **negative** vaues indicate upward gradients while **positive** indicate downward gradients



Table 9Statistical Summary of Groundwater Monitoring ParametersHydrogeologic Characterization ReportCoffeen Energy Center

BORON (dissolved - mg/L)

Monitoring	Number of	nber of				Standard	Percent	Sen Slope
Well Number	Data Points	ata Points Mean		Maximum	Minimum	Deviation	Non-Detects	Trend
APW-2 (G402)	8	7.245	7.18	8.10	6.30	0.626	0.0%	0.27
G151	16	0.016	0.012	0.052	0.003	0.013	50.0%	0
G152	16	0.047	0.041	0.11	0.015	0.025	0.0%	0.01
G153	16	0.028	0.027	0.058	0.012	0.012	0.0%	-0.01
G154	16	0.036	0.037	0.045	0.021	0.008	0.0%	0
G270	34	0.029	0.012	0.125	0.005	0.034	47.1%	0
G271	29	0.281	0.260	0.610	0.130	0.128	0.0%	0.05
G272	29	0.019	0.005	0.27	0.005	0.050	86.2%	0
G273	29	0.224	0.18	0.62	0.019	0.176	6.9%	0.06
G274	29	0.295	0.18	0.97	0.005	0.302	27.6%	0.13
G275	25	3.176	3.50	4.60	0.91	1.04	0.0%	0.1

BORON (total - mg/L)

Monitoring	Number of					Standard	Percent	Sen Slope	
Well Number	Data Points	Mean	Median	Maximum	Minimum	Deviation	Non-Detects	Trend	
G151	3	0.0183	0.019	0.026	0.010	0.008	0%	-0.03	
G152	3	0.038	0.036	0.047	0.032	0.008	0%	-0.03	
G153	3	0.044	0.036	0.064	0.031	0.018	0%	-0.07	
G154	3	0.030	0.031	0.038	0.021	0.009	0%	0.01	
G270	38	0.021	0.005	0.125	0.005	0.026	68%	0.00	
G271	33	0.309	0.240	0.980	0.110	0.190	0%	0.03	
G272	29	0.014	0.005	0.060	0.005	0.017	69%	0.00	
G273	33	0.208	0.200	0.480	0.016	0.137	9%	0.05	
G274	29	0.264	0.160	0.740	0.005	0.253	24%	0.11	
G275	25	3.336	3.600	4.600	1.200	0.965	0%	0.04	
G401	4	3.575	3.450	4.100	3.300	0.359	0%	0.79	
G402	4	6.500	6.450	7.400	5.700	0.707	0%	1.83	
G403	4	0.036	0.033	0.064	0.014	0.021	0%	-0.03	
G404	4	2.075	1.850	3.200	1.400	0.806	0%	0.40	
G405	4	16.25	16.500	17.000	15.000	0.957	0%	-1.96	

MANGANESE (dissolved - mg/L)

Monitoring	Number of	umber of				Standard	Percent	Sen Slope
Well Number	Data Points	Mean	Median	Maximum	Minimum	Deviation	Non-Detects	Trend
APW-2 (G402)	8	0.432	0.414	0.730	0.130	0.166	0%	0.01
G151	16	0.131	0.092	0.560	0.011	0.131	0%	0.00
G152	16	0.294	0.180	0.927	0.034	0.277	0%	0.07
G153	16	0.235	0.175	1.020	0.024	0.249	0%	-0.12
G154	16	0.062	0.035	0.462	0.003	0.112	0%	0.00
G270	34	0.094	0.045	0.470	0.003	0.108	0%	-0.02
G271	29	0.015	0.001	0.220	0.001	0.045	62%	0.00
G272	29	0.011	0.001	0.230	0.001	0.043	76%	0.00
G273	29	0.042	0.028	0.330	0.001	0.059	0%	-0.01
G274	29	0.003	0.001	0.046	0.001	0.009	69%	0.00
G275	25	0.051	0.019	0.340	0.001	0.086	8%	0.00

Notes:

Sen Slope Trend is in milligrams per Liter per year; negative value (-) is downward trend; positive value is upward trend, BOLD indicates statistical significance (Mann Kendall 95%).Mann Kendall tests not performed on wells with less than 8 data ponts.
Sample results below the method detection limit (MDL) for that parameter have been replaced by the detection limit.

Wells with groundwater exceeding Class 1 groundwater standard for the given parameter in 2015 or 2016.



Table 9Statistical Summary of Groundwater Monitoring ParametersHydrogeologic Characterization ReportCoffeen Energy Center

MANGANESE (total - mg/L)

Monitoring	Number of					Standard	Percent	Sen Slope	
Well Number	Data Points	Data Points Mean		Maximum	Minimum	Deviation	Non-Detects	Trend	
G151	3	0.225	0.23	0.36	0.086	0.137	0%	0.56	
G152	3	0.32	0.31	0.37	0.27	0.050	0%	0.12	
G153	3	0.67	0.51	1.30	0.21	0.563	0%	-2.21	
G154	3	0.16	0.16	0.16	0.15	0.006	0%	0.02	
G270	34	0.146	0.083	0.62	0.021	0.134	0%	-0.03	
G271	29	0.058	0.021	0.27	0.001	0.075	0%	-0.01	
G272	29	0.029	0.011	0.15	0.002	0.040	0%	0.00	
G273	29	0.06	0.043	0.19	0.019	0.043	0%	-0.01	
G274	29	0.05	0.02	0.37	0.003	0.087	0%	0.00	
G275	25	0.17	0.09	0.78	0.01	0.203	0%	0.01	

SULFATE (dissolved - mg/L)

Monitoring	Number of					Standard	Percent	Sen Slope
Well Number	Data Points	Mean	Median	Maximum	Minimum	Deviation	Non-Detects	Trend
APW-2 (G402)	8	805.375	840	1100	500	173.11	13%	5.17
G151	16	112	110	140	100	11.21	0%	4.8
G152	16	161	135	300	107.0	61.15	0%	26.3
G153	16	1824	1800	2100	1500	206.55	0%	184.5
G154	16	110	105	175	82	23.60	0%	-12.0
G270	34	54	43	140	2	50.37	0%	15.7
G271	29	354	360	520	220	87.45	0%	35.8
G272	29	264	270	380	140	63.00	0%	28.1
G273	29	524	520	700	46	131.06	0%	9.0
G274	29	340	350	400	240	41.75	0%	0.0
G275	25	678	720	920	280	183.63	0	30.3

SULFATE (total - mg/L)

Monitoring	Number of					Standard	Percent	Sen Slope	
Well Number	Data Points	Mean	Median	Maximum	Minimum	Deviation	Non-Detects	Trend	
G151	3	104.333	98	120	95	13.65	0%	6.09	
G152	3	120	120	130	110	10	0%	20.3	
G153	3	1633	1500	1900	1500.0	231	0%	0.0	
G154	3	99	93	110	93	10	0%	34.5	
G270	38	59	65	140	2	50	0%	15.6	
G271	33	357	350	540	230	87	0%	28.3	
G272	29	253	250	380	120	69	0%	30.5	
G273	33	530	520	750	340	106	0%	6.5	
G274	29	337	330	460	230	55	0%	6.7	
G275	25	684	720	990	310	213	0%	0.8	
G281	4	330	325	370	300	32	0%	68.7	
G401	4	2275	2250	2500	2100	171	0%	-390.6	
G402	4	1013	980	1200	890	133	0%	-392.5	
G403	4	18	14	35	10	12	0%	-30.9	
G404	4	165	165	190	140.0	23.8	0%	-12.3	
G405	4	1700	1700	1800	1600.0	81.7	0%	-71.6	

Notes:

Sen Slope Trend is in milligrams per Liter per year; negative value (-) is downward trend; positive value is upward trend, BOLD indicates statistical significance (Mann Kendall 95%).Mann Kendall tests not performed on wells with less than 8 data ponts.
Sample results below the method detection limit (MDL) for that parameter have been replaced by the detection limit.

Wells with groundwater exceeding Class 1 groundwater standard for the given parameter in 2015 or 2016.



Table 9Statistical Summary of Groundwater Monitoring ParametersHydrogeologic Characterization ReportCoffeen Energy Center

TOTAL DISSOLVED SOLIDS (mg/L)

Monitoring	Number of					Standard	Percent	Sen Slope	
Well Number	Data Points	Mean	Median	Maximum	Minimum	Deviation	Non-Detects	Trend	
APW-2 (G402)	8	1651.25	1600	1810	1600	78.456	0%	0	
G151	17	551	560	660	500	44	0%	0.0	
G152	17	677	620	1,100	494	156	0%	73	
G153	17	3,362	3,300	3,900	2,800	341	0%	42.5	
G154	17	476	460	624	410	52	0%	17	
G270	37	454	460	550	340	53	0%	11	
G271	31	862	860	1,000	710	96	0%	29	
G272	27	703	680	840	570	69	0%	25.5	
G273	31	1,085	1,100	1,300	840	140	0%	0	
G274	27	863	870	980	770	64	0%	-16.1	
G275	24	1,388	1,500	2,000	840	270	0%	0.0	
G281	4	770	760	820	740	38	0%	-28.7	
G401	4	2,925	2,950	3,000	2,800	96	0%	-185.3	
G402	4	1,600	1,600	1,700	1,500	115	0%	-346	
G403	4	325	320	340	320	10	0%	0.0	
G404	4	555	570	620	460	68	0%	-14	
G405	4	2,325	2,300	2,500	2,200	150	0%	-347	

Notes:

[O: NRK 12/2016, C: SJC 12/2016]

Sen Slope Trend is in milligrams per Liter per year; negative value (-) is downward trend; positive value is upward trend, BOLD indicates statistical significance (Mann Kendall 95%). Mann Kendall tests not performed on wells with less than 8 data ponts.
Sample results below the method detection limit (MDL) for that parameter have been replaced by the detection limit.

Wells with groundwater exceeding Class 1 groundwater standard for the given parameter in 2015 or 2016.



Table 10 Potential Source Water and Surface Water Sampling Results - October 24, 2016 Hydrogeologic Characterization Report Coffeen Energy Center

Sample ID	Sample Location	Alkalinity, bicarbonate mg/L	Alkalinity, carbonate mg/L	Antimony, Total ug/L	Arsenic, Total ug/L	Barium, Total ug/L	Beryllium, Total ug/L	Boron, Total ug/L	Cadmium, Total ug/L	Calcium, Total mg/l	Chloride, Total mg/L	Chromium, Total ug/L	Cobalt, Total ug/L	Fluoride, Total mg/L	Lithium, Total ug/L	Magnesium, Total mg/L	Mercury, Total ug/L	Molybdenum , Total ug/L	рН	Potassium, Total mg/L	Selenium, Total ug/L	Sodium, Total mg/l	Solids - total dissolved solids (TDS) mg/L	Sulfate, Total mg/L	Thallium, Total ug/L
AP1a		90	< 2.0	3.1	7.2	130	< 1.0	3800	< 1.0	380	3.2	< 4.0	< 2.0	< 0.250	39	67	< 0.20	47	6.99	7.3	3.0	33	1800	1500	< 1.0
AP1b		120	< 2.0	3.3	17	100	< 1.0	3100	< 1.0	320	13	< 4.0	< 2.0	0.977	47	52	< 0.20	100	7.01	18	6.9	53	1600	1300	< 1.0
AP1c		80	< 2.0	< 3.0	18	130	< 1.0	2900	< 1.0	390	13	< 4.0	< 2.0	0.614	51	49	< 0.20	81	7.05	18	3.1	59	1900	1600	< 1.0
AP1d	Ash Pond 1	90	< 20	< 3.0	2.0	200	< 1.0	2000	< 1.0	210	18	< 4.0	< 2.0	1.08	55	35	< 0.20	31	7.21	26	1.5	83	980	1000	< 1.0
AP1e	ASITIONUT	80	< 2.0	< 3.0	1.4	160	< 1.0	2100	< 1.0	200	18	< 4.0	< 2.0	1.00	55	34	< 0.20	30	7.12	26	1.2	80	1200	960	< 1.0
AP1f		110	< 20	< 3.0	1.5	150	< 1.0	2100	< 1.0	200	17	< 4.0	< 2.0	1.00	57	35	< 0.20	31	7.20	26	1.2	82	1200	1000	< 1.0
AP1g		75	< 2.0	< 3.0	1.3	140	< 1.0	2100	< 1.0	210	17	< 4.0	< 2.0	1.00	56	32	< 0.20	29	7.21	26	1.4	100	1200	970	< 1.0
AP1h		90	< 20	< 3.0	1.5	180	< 1.0	2200	< 1.0	230	17	< 4.0	< 2.0	0.980	55	30	< 0.20	31	7.41	27	1.2	110	1200	1000	< 1.0
AP2e		55	< 20	< 3.0	23	26	< 1.0	5300	< 1.0	210	< 5.0	< 4.0	< 2.0	0.438	190	40	< 0.20	90	6.49	27	< 1.0	25	1700	1500	< 1.0
AP2f	Ash Pond 2	100	< 20	< 3.0	1.2	22	< 2.0	2000	< 1.0	170	< 5.0	< 4.0	< 2.0	0.398	130	33	< 0.20	3.2	6.42	21	< 1.0	14	1700	1500	< 1.0
AP2g		4.0	< 2.0	< 3.0	5.5	20	< 2.0	4300	4.6	410	< 5.0	< 4.0	< 2.0	0.506	180	51	< 0.20	41	6.46	29	< 1.0	27	2400	2300	< 1.0
AP2h		140	< 2.0	< 3.0	75	23	< 1.0	14000	< 1.0	310	1.7	< 4.0	< 2.0	0.406	120	29	< 0.20	570	7.17	40	< 1.0	39	1500	1300	< 1.0
CLa		80	< 2.0	< 3.0	1.8	54	< 1.0	270	< 1.0	23	23	< 4.0	< 2.0	0.443	< 10	12	< 0.20	5.0	7.22	7.4	< 1.0	19	190	55	< 1.0
CLb	Coffeen Lake	80	< 2.0	< 3.0	1.8	52	< 1.0	280	< 1.0	23	22	< 4.0	< 2.0	0.425	< 10	11	< 0.20	4.9	7.52	7.8	< 1.0	19	180	56	< 1.0
CLc	OUNCER Lake	75	< 2.0	< 3.0	1.8	56	< 1.0	280	< 1.0	23	22	< 4.0	< 2.0	0.426	< 10	12	< 0.20	4.7	7.62	7.4	< 1.0	19	160	54	< 1.0
CLd		80	< 2.0	< 3.0	1.8	54	< 1.0	270	< 1.0	23	23	< 4.0	< 2.0	0.421	< 10	12	< 0.20	4.8	7.30	7.5	< 1.0	20	170	54	< 1.0
GPa		4.0	< 2.0	< 3.0	4.7	120	2.6	59000	40	450	1900	< 4.0	52	42.7	300	1500	< 0.80	130	7.16	210	890	620	17000	17000	< 1.0
GPb	Gypsum Stack	10	< 2.0	8.6	92	1100	< 20	97000	67	1400	2600	150	110	69.4	480	2500	27	140	6.65	360	1500	1000	28000	27000	< 4.0
GPc	Pond	5.5	< 2.0	< 6.0	4.4	110	2.6	72000	41	570	1800	< 4.0	54	49.2	300	1500	< 0.80	120	6.73	260	890	650	17000	20000	< 1.0
GPd		6.0	< 2.0	< 3.0	4.2	110	< 20	66000	38	560	1900	< 4.0	52	51.3	300	1500	< 0.80	120	6.73	230	800	660	16000	19000	< 1.0
RPa		< 2.0	< 2.0	< 3.0	3.3	89	2.3	60000	37	380	1600	< 4.0	44	47.0	310	1200	< 0.20	63	6.70	190	840	450	16000	17000	< 1.0
RPb	Gypsum	< 2.0	< 2.0	< 3.0	2.8	90	2.1	56000	33	380	1800	< 4.0	45	44.4	310	1200	< 0.40	64	6.49	210	780	470	16000	18000	< 1.0
RPc	Recycle Pond	< 2.0	< 2.0	< 3.0	3.1	88	2.2	59000	37	380	2000	< 4.0	43	45.0	310	1200	< 0.20	64	6.32	200	830	440	16000	20000	< 1.0
RPd		< 2.0	< 2.0	< 3.0	3.4	89	2.4	59000	36	400	1600	< 4.0	45	51.2	310	1300	< 0.20	66	6.32	200	860	460	17000	16000	< 1.0

[O:MDM 12/2016, NRK 12/2016]



APPENDIX A

BORING LOGS AND WELL CONSTRUCTION DETAILS
APPENDIX A1

ASH POND 2

FI	EL CLIEN Sit Locatio Projec DATE	D] T: N te: C n: C n: C s: S S: S F ir R: S	BOR atural Re offeen En offeen, Ill 5E0030 cart: 9/1- nish: 9/1 unny, hi 6	sour ergy linois 4/20 4/20 50's	ce Te Cent S 15	G L(echnolo ter	DG gy, Inc.	CONTRACTOR: Ramsey Geotechnical Engine Rig mfg/model: D-50 Turbo Tracked MST 8 Drilling Method: Hollow Stem Auger (3¼"ove FIELD STAFF: Driller: D. Crump Helper: D. Groves Eng/Geo: R. Hasenyager	eering, LLC 00ATV erdrill / 4¼") BOREHOLE ID: G401 Well ID: G401 Surface Elev: 623.03 ft. MSL Completion: 19.30 ft. BGS Station: 2,515,614.84N 872,510.57E
S	SAMPL	E	Т	EST	TINC	Ĵ	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	al (in		~	(0)	b/\hat{H}^3	o (tsf) e	Quad Town	rangle: Coffeen, IL ship: East Fork	Ψ = Dry - During Drilling Ψ =
er	/ Tot		/6 <i>ir</i> alue	ure (9	en. (l	$^{\rm f)}_{\rm Typ}$	Sectio	n 11, Tier 7N; Range 3W	<u> </u>
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	16/24 67%	ss	2-2 3-7 N=5	17			2	Dark grayish brown (10YR4/2), moist, soft, CLAY wit little silt and trace very fine- to fine-grained sand - FILI 	h
2A	21/24 88%	ss	8-11 8-9	17		1.80		Dark gray (10YR4/1), moist, stiff, SILT with little clay a trace very fine-grained sand.	und 620
2B		N	N=19	25				Yellowish brown (10YR5/6), moist, stiff, CLAY with so silt and trace very fine- to fine-grained sand.	me
3A	23/24 96%	ss	3-4 7-8 N=11	23		2.50		Gray (10YR5/1) with 20% yellowish brown (10YR5/6 mottles, moist, medium, CLAY with some silt and trac very fine- to fine-grained sand.	
4A	24/24 100%	ss	8-9 12-14 N=21	21		3.30			616
4B		()		19		2.80	8	Gray (10YR5/1) with 30% yellowish brown (10YR5/8 mottles, moist, stiff, SILT and very fine-grained SAND w trace clay.	b) vith
5A	24/24 100%	ss	2-3 4-5 N=7	21		1.30	10 -		614
6A	24/24 100%	ss	2-4 5-6 N=9	17		2.50	12	Gray (10YR5/1) with 30% yellowish brown (10YR5/6 mottles, moist, medium, CLAY with some silt and trac very fine- to fine-grained sand.	612 e
7A	24/24 100%	ss	9-7 8-9 N=15	21		1.40	14		610
8A 8B	24/24 100%	ss	2-3 2-4 N=5	17 19		1.30	16	Gray (10YR6/1), moist soft, CLAY with very fine- to fine-grained sand and little silt. Yellowish brown (10YR5/6), wet, loose, very fine- to	
9A	20/24 83%	ss	5-4 5-10 N=9	21			1 mpm	fine-grained SAND with trace silt. Yellowish brown (10YR5/6), wet medium, SILT with so very fine-grained sand and little clay. Yellowish brown (10YR5/6), wet, loose, very fine- to	me 606
9B	12/16		72 11	16			18	medium-grained SAND with trace silt.	
10A	75%	Å ss	50/4''	6		4.50		little very fine- to very coarse sand.	
								End of boling - 19.5 feet	

NOTE(S): G401 installed in borehole.

	Surf	ace Elevation: <u>610.56</u> Datum <u>msl</u>	Completion Date: N	8/27/10 lorthing: 872502.26 Easting: 2516632.59	0010	IGHT (pct) COUNTS ERY/ROD	.ES	V	VELL DI	AGRAM	i	
	DEPTH IN FEET	DESCR		TERIAL	GRAPHIC	DRY UNIT WE SPT BLOW (CORE RECOV	SAMPL	Dian 7	Stickup neter: 6 in	nches	-2.8	Lever Lever Lever
		Medium stiff, brown	, silty CLAY - CL			6-3-3	SS1			Concrete	1.0	609.6
ļ	 5	Modium atiff const.				1-2-4	SS2	2" sch 40 PVC		Bentonite		
		iwedium sun, gray, s	Sinty CLAF, trace sand	1- CL	Ţ	1-2-5	SS3					
IL TYPES ONLY.	— 10-	Hard to stiff, brown CL (TILL)	to gray, silty CLAY, tr	ace sand seams -		9-22-28	SS4			T 	8.0 10.0	602.6 600.6
TWEEN SO URPOSES												
DARIES BE TRATION P	- 15-					1-6-12	SS5	2° sch 40 PVC 0.10 slotted		Filter sand		
ATE BOUN FOR ILLUS		- - -										
APPROXIM APHIC LOG	- 20-	Boring terminated a	t 20 feet.			3-4-9	SS6	Bottom cap		<u> </u>	20,0 20,4	590.6 590.2
SENT THE DUAL. GRV												
AV BE GRA	- 25-											
CATION LIN NSITION M												
D THE TRATIFI	- 30-	-										
310 NOTE	·											ĺ
1.6PJ 12/1	- 35-											
INC 063830												
PJ GF		GROUNDWATER D	<u>ATA</u>		<u>DATA</u>			Drawn by: KSA Date: 9/10/10	Checked t	-4-11 Dat	p'vd. by; te: 1/4	4/11
01- COFFEEN.G	EN	COUNTERED AT 7 F	EET \$	AUGER <u>4 1/4*</u> F WASHBORING FR <u>MVU</u> DRILLER <u>S</u>	ЮLLO ¹ ОМ <u>WG_</u> L(W STEM FEET DGGER		G	GEOTE	CHNOL FROM TO	OGY HE GROUN	
WL J017150.(<u>CME 55TRK</u> D HAMMER TYP	RILL R E <u>Auto</u>	1G 2		Ameren-Co	offeen Ast	1 Pond Ev	aluati	on
BORING 2002	RE	MARKS:						LOC	GOF BOR	ING: APW G402	V-2	
LOG OF							_	Pro	ect No.	J017150.	01	

F	[EL]	D]	BOR	I	NG	G L(DG			6		
]	CLIEN Sit Locatio Projec DATE	T: N te: C on: C ct: 1: S: Si	atural Re offeen En offeen, Ill 5E0030 tart: 9/1	esour hergy linois 1/20	ce Te Cent 5	echnolo <u></u> ter	gy, Inc.	CONTRACTOR: Ramsey Geotechnical Engin Rig mfg/model: D-50 Turbo Tracked MST & Drilling Method: Hollow Stem Auger (3¼"ov FIELD STAFF: Driller: D. Crump	eering, LLC 800ATV erdrill / 4¼")	BOI	REHOLE ID Well ID Surface Elev Completion	: G403 : G403 : 623.81 ft. MSL : 18.15 ft. BGS
WE	ATHE	Fir R: R	1ish: 9/1 aining, hi	1/20 60's	15			Helper: D. Groves Eng/Geo: K. Theesfeld			Station	2,514,616.63N 873,561.34E
5	SAMPL	Æ	Т	EST	INC	J	TOPOGI	RAPHIC MAP INFORMATION:	WATER LE	EVEL	INFORMA	FION:
	al (in			()	o∕fî³)	e (tsf) e	Quadi	rangle: Coffeen, IL shin: East Fork	$\mathbf{\Psi} = 15$ $\mathbf{\Psi} =$	5.00 -	During Drillin	ng
ы	/ Tot overy		/ 6 in lue	lre (%	en. (Il	$O_{\rm Type}$	Sectio	n 11, Tier 7N; Range 3W	<u> </u>			
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Bore De	ehole etail	Elevation ft. MSL	Remarks
1A	19/24 79%	ss	2-2 2-2 N=4	25		0.80	,	Very dark brown (10YR2/2) grading to dark grayish brown (10YR4/2), moist, medium, SILT with some clay, trac roots and grass.	$\frac{\sqrt{k_{l}}}{\sqrt{k_{l}}}$			
2A $22/24$ 35 $3-4$ 29 Very dark brown (10YR2/2) grading to dark grayish brown (10YR4/2), moist, stiff, SILT with some clay, trace wood.												
2A 2B	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$											
3A	8/24 <i>33%</i>	ss	2-3 4-4 N=7	25		1.50		Yellowish brown (10YR5/4), moist, stiff, CLAY with liss silt and trace very fine-grained sand.	ittle			
4A	21/24 88%	ss	8-7 8-7 N=15	20		1.30		Grayish brown (10YR5/2) with 15% yellowish brown (10YR5/6) and 5% very dark grayish brown (10YR3/ mottles, moist, stiff, CLAY with little silt and trace ve fine-grained sand.			616	
5A	20/24 83%	ss	2-2 3-3 N=5	22		0.70	10	Grayish brown (10YR5/2) with 5% yellowish brown (10YR5/6) mottles, moist, medium, CLAY with little s and trace very fine-grained sand.	ı silt		614	
6A	24/24 100%	ss	2-2 3-4 N=5	23		1.40		Grayish brown (10YR5/2) with 30% yellowish brow (10YR5/6) mottles, moist, very stiff, CLAY with little s few very fine- to medium-grained sand, and trace grave	n silt, el.			
7A	21/24 88%	ss	5-5 6-5 N=11	20		0.90	12	Grayish brown (10YR5/3) with 45% yellowish brow (10YR5/6) and 5% dark brown (10YR3/3) mottles, mo stiff, SILT with some clay, few very fine- to coarse-grai sand, and trace gravel.	n list, lined			
8A	24/24 100%	ss	3-2 3-6 N=5	17			14 ¶πſππηπſſſπ	Grayish brown (10YR5/3) with 40% yellowish brown (10YR5/6) mottles, moist, medium, SILT with little cl. few very fine- to coarse-grained sand, and trace grave Yellowish brown (10YR5/6) with 30% grayish brown (10YR5/2) mottles, moist, medium, SILT with little cl. few very fine- to coarse-grained sand, and trace grave	n — — — — — — — — — — — — — — — — — — —			
9A	$9A \begin{vmatrix} 19/24 \\ 79\% \\ 0/2 \\ 0\% \end{vmatrix} BD \begin{vmatrix} 8 \\ 8 \\ 8 \\ 8 \\ 8 \\ 8 \\ 8 \\ 8 \\ 8 \\ 8$											
								End of boring - 16.15 feet				
)TE(S)•	G40	3 installe	d in i	horeł	nole						

F	[EL]	D 1	BOR	(II)	١G	G L(DG		0	
WE	CLIEN Sit Locatio Projec DATE	T: N te: C n: C ct: 1: S: St S: St Fin R: Pa	atural Re offeen En offeen, Ill 5E0030 tart: 5/1 nish: 5/1 artly sunr	sourd ergy linois /200 /200 ny, w	ces T Cent 3 7 7 7 7 7 arm	`echnole ter	ogy, Inc.	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4¼" Hollow stem auger with sampler FIELD STAFF: Driller: A. Rachford Helper: M. Brown Eng/Geo: R. Hasenyager	BOR split spoon S	EHOLE ID: SB22 Well ID: G404 urface Elev: 613.10 ft. MSL Completion: 12.00 ft. BGS Station: 873,999.77N 2,516,397.85E
5	SAMPL	E	T	EST	TING	<u>;</u>	TOPOGRA	APHIC MAP INFORMATION:	WATER LEVEL I	NFORMATION:
ber	v / Total (in) covery		s / 6 in 'alue	ture (%)	Den. (lb/ft ³)	sf) Qp (tsf) re Type	Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 11, Tier 7N; Range 3W		5/10/2007
Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Detail	Elevation ft. MSL Remarks
1A 1B	19/24 79%	ss		26 26		0.85 B		Black (10YR2/1), moist, firm, clayey SILT (TOPSOI		612
2A	19/24 79%	ss		16		2.47 B	¥ 2	Gray (10YR5/1) with 35% dark yellowish brown (10YR4/6) mottles, moist, firm, clayey SILT with sand trace gravel	and	610
3A	18/24	V ss		19		2.18 B	4			
3B	/3%	\wedge		18		2.33 B	6	Gray (10YR6/1) with 20% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with sand and trace gra	8) vel	008
4A	24/24	M		23		0.58 B		Yellowish brown (10YR5/8), moist, soft, sandy CLAY slight trace gravel	vith	
4B	100%	A ss		18			8-	Yellowish brown (10YR5/8), very moist to wet, very so clayey, very fine- to medium-grained SAND with trac gravel	oft, e	606
5A	23/24 96%	ss		10				Yellowish brown (10YR5/4) with 30% yellowish brow (10YR5/8) mottles, moist, hard, clayey SILT with sand trace gravel	/n and	604
5B	19/24			19			10	Yellowish brown (10YR5/6), wet, loose, very fine- to medium-grained SAND with coarse-grained sand and sl trace gravel	ght	
6B	79%	∬ ^{ss}		19			12	Dark yellowish brown (10YR4/6) with 40% yellowis brown (10YR5/8) mottles, moist, very hard, very silt CLAY with sand and gravel End of Boring = 12.0 ft. BGS	r n 7	602
								0		

NOTE(S):

F	TIELD BORING LOG												
WE	CLIEN Sit Locatio Projec DATE	T: Na ae: Co n: Co xt: 15 S: St Fin R: Pa	atural Resolution offeen En offeen, Ill EE0030 art: 5/1/ ish: 5/1/ rtly sunn	sourc ergy inois /200 /200 y, wa	ces T Cent 7 7 arm	echnolo	ogy, Inc.	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4 ¹ / ₄ " Hollow stem auger with sampler FIELD STAFF: Driller: A. Rachford Helper: M. Brown Eng/Geo: R. Hasenyager	split spoon	BOREHOLE ID: SB21 Well ID: G405 Surface Elev: 620.90 ft. MSL Completion: 14.21 ft. BGS Station: 873,996.79N 2,515,335.70E			
	SAMPL	E	T	EST	INC	; 	TOPOGRA	PHIC MAP INFORMATION:	WATER L	EVEL INFORMATION:			
ber	v / Total (in ecovery		s / 6 in /alue	ture (%)	Den. (lb/ft^3)	sf) Qp (tsf) re Type	Quadraı Townshi Section 1	ngle: Coffeen, IL ip: East Fork 11, Tier 7N; Range 3W	$\underline{\bar{\Psi}} = \underline{\bar{\Psi}} = \bar{$	1.28 - 05/10/2007			
Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Bor De	rehole Elevation etail ft. MSL Remarks			
1A	19/24 79%	ss		43		0.78 B	¥	Black (10YR2/1), moist, soft, clayey SILT (TOPSOI					
1B 2A	12/12 100%	Ss Ss		26 27		1.94 B 2.52 BSh		Gray (10YR6/1) with 30% yellowish brown (10YR5/	3)	618			
3A	24/24	SS SS		24		3.92 BSh	4	moures, moist, min, sity CLA r		616			
3B	100%	$\left \right\rangle$		24		2.33 BSh	6	Gray (10YR6/1), moist, firm, silty CLAY slight trace s Gray (10YR5/1), very moist, soft, clayey, very fine- t fine-grained SAND	und				
4A	24/24 100%	ss		20		2.33 BSh	8	Gray (10YR6/1) with 25% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with sand and trace gra	8) vel	- 614			
5A	24/24 100%	ss		24		1.55 BSh	10	Yellowish brown (10YR5/8) with 40% gray (10YR6/ mottles, moist, firm, silty CLAY with sand and trace gra	l) vel	-612			
6A 6B	24/24 100%	ss		19 18				Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Dark brown (10YR3/3), very moist, soft, clayey, fine- very coarse-grained SAND with slight trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to	to	610			
7A	24/24 100%	ss		9		7.42 BSh		fine-grained SAND Dark yellowish brown (10YR4/4), moist, soft, sandy SI with trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Gray (10YR5/1), moist, very hard, very silty CLAY w sand and gravel	LT	608			
							· ·	End of Boring = 14.2 ft. BGS					

WE	CLIEN Sit Location Projec DATE ATHEI	Γ: Να e: Co n: 13 ct: 16 S: St Fin R: Cl E	atural Re offeen Po 34 CIPS I 5E0080 (aart: 8/1) hish: 8/1 loudy, rai	sourc wer S Lane, 6/20 7/20 n, (h	xes T Static , Coff 16 16 i-70s	echnolo n - Ash èeen, IL	ogy, Inc. 1 Pond 2 . 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld	BOREHOLE ID: G405D Well ID: G45D Surface Elev: 620.94 ft. MSL Completion: 42.00 ft. BGS Station: 873,998.03N 2,515,322.23E
er	/ Total (in)	E	/ 6 in Ilue	are (%)	en. (lb/ft ³)	$\int_{a} \frac{Qp}{T \text{ype}} (\text{tsf})$	TOPOGR Quadra Townsl Section	APHIC MAP INFORMATION: angle: Coffeen hip: East Fork 11, Tier 7 N.; Range 3 W.	WATER LEVEL INFORMATION: $\underline{\Psi} = 16.00$ - During Drilling $\underline{\Psi} = $ $\underline{\Psi} = $
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
	0/60						2	Black (10YR2/1), moist, soft, clayey SILT (TOPSOII	
	0%	BD					4	Gray (10YR6/1) with 30% yellowish brown (10YR5/8 mottles, moist, firm, silty CLAY	8)
								Gray (10YR6/1), moist, firm, silty CLAY slight trace sa	and 616
							6	Gray (10YR5/1), very moist, soft, clayey, very fine- to fine-grained SAND	
	0/60 <i>0%</i>	BD					п	Gray (10YR6/1) with 25% yellowish brown (10YR5/8 mottles, moist, firm, silty CLAY with sand and trace gra	8) vel 614
							8	Yellowish brown (10YR5/8) with 40% gray (10YR6/ mottles, moist, firm, silty CLAY with sand and trace gra	612 l) vvel
	0/48 <i>0%</i>	BD					12	Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Dark brown (10YR3/3), very moist, soft, clayey, fine- very coarse-grained SAND with slight trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Dark yellowish brown (10YR4/4), moist, soft, sandy SI	610 to
								Vellowish brown (10YR5/4), wet, loose, very fine- to fine-orained SAND	
1.4	24/24	\mathbb{V}	6-23	1.		1.50	14	Gray (10YR5/1), moist, very hard, very silty CLAY wi sand and gravel	
IA	100%	ss	37-44 N=60			1.50	¥ 16 -	Dark gray (10YR4/1), moist, hard, SILT with few fine-	. to
2A	23/24 96%	ss	14-32 41-45 N=73	7		4.50	10	coarse-grained sand, little clay and trace small gravel.	
3A	14/17 82%	ss	16-47 50/5"	7		4.50		Dark gray (10YR4/1), dry, hard, SILT with few fine- t coarse-grained sand, little clay and trace small gravel.	to

F	EL	DI	BOR	I	NG	L C	JG				<		ANSON
	CLIENT: Natural Resources Technology, Inc. CONTRACTOR: Bulldog Drilling, Inc. Site: Coffeen Power Station - Ash Pond 2 Rig mfg/model: CME-750 ATV Drill Location: 134 CIPS Lane, Coffeen, IL 62017 Drilling Method: 4 1/4" Hollow Stem Auger Project: 16E0080 Surface Elev: 620.94 ft. MSL DATES: Start: 8/16/2016 FIELD STAFF: Driller: J. Dittmaier												
WE	DATES ATHEF	S: St Fin R: Cl	art: 8/1 ish: 8/1 oudy, rai	6/20 7/20 in, (h	16 16 i-70s	5)		FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld				Completion Station	a: 42.00 ft. BGS a: 873,998.03N 2,515,322.23E
s	AMPL	E	Т	EST	ING	}	TOPOGR	APHIC MAP INFORMATION:	WA	TER	LEVEI	INFORMA	ΓΙΟΝ:
	(in)				ft ³)	tsf)	Quadr	angle: Coffeen		Y =	16.00 -	During Drilli	ng
	Fotal <i>ery</i>		í in e	(%)	(lb/	$\tilde{O}^{p}($	Towns Section	hip: East Fork	-	= <u>V</u>			
aber	, / AC	Э	vs/(Valu D	sture	Den	(tsf) ure T	Denth	L'detect		<u>*</u>		El. dia	
Nun	Rec % R	Typ	Blor N - RQ	Moi	Dry	Qu Fail	ft. BGS	Description		1	Detail	ft. MSL	Remarks
		$\langle $						Dark gray (10YR4/1), dry, hard, SILT with few fine- coarse-grained sand, little clay and little small gravel	to /				
	19/24 70%	ss	15-36 32-36					Dark gray (10YR4/1), moist, dense, fine- to coarse-grai SAND.	ned /	\square		600	
	1970	A	N=68						/				
4A				8		4.50	22						
	19/24	VI	10-18									508	
5A	79%	$\int \int ds$	30-33 N=48	8		4.50		Dark gray (10YR4/1), dry, hard, SILT with few fine- coarse-grained sand little clay and little small gravel	to			598	
	Į						24	course granica sana, nuce only and nuce share graver	•				
		M	22.42										
6A	11/24 46%	ss	34-36	8		4.50						- 596	
		$\langle \rangle$	N=/6										
		1					26			- + -			
7A	11/24	$\mathbb{V}_{\mathbb{R}}$	28-26	11		4 50		Dark gray (10YR4/1) with 5% light brownish gray				594	
,	46%	\bigwedge	23-20 N=49					coarse-grained sand, little clay and little small gravel)				
	Į						28			_	-1.1.1.		
		M	7 11					Dark gray (10YR4/1) with 5% light brownish gray				E	
	23/24 96%	ss	16-25 N=27					(10YR6/2) and dark greenish gray (10YR4/2) mottles, hard, SILT with few fine- to coarse-grained sand, little c	dry, clay		물리	592	
8A		/\	11-27	10		4.50		and little small gravel.					
9A	İ			16				Dark gray (10YR4/1), dry, hard, SILT with few fine- coarse-grained sand little clay and little small gravel	to				
9B	17/24	V ss	7-14 12-12	14				Dark gray (10YR4/1), dry, very stiff, SILT with some v	very			590	
9C	71%	A	N=26	9				Very dark gray (10YR3/1), moist, very stiff, CLAY w	ith				
	-						32	few silt and little medium-grained sand.		-4		£ I	
	24/24	VI	2-5										
10.4	100%	∬ ss	<i>8-13</i> N=13	15		2.25						588	
10A		/\		15		5.25		Very dark gray (10YR3/1), moist, very stiff, CLAY w	ith			E I	
11-1							34	little silt and little medium-grained sand.					
11-2	22/24	SH		15								586	
11-3	9270											E I	
11-4							36			- //			
	24/24	V	2-5					Vary dark arou (10VP2/1) maint wary stiff CLAV	ith				
12.4	100%	∬ ss	<i>8-10</i> N=13	16		2.00		little silt, little medium-grained sand, and trace small gra	ivel.			584	
12A				10		2.00						E I	
	Ī						38						
	22/24	ss	2-5 7-8					Very dark gray (10YR3/1) with 5% black (10YR2/1 mottles, moist, very stiff, CLAY with little silt. little)			582	
13A	9270	\mathbb{N}	N=12	16		2.00		medium-grained sand, and trace small gravel.				E I	
NO	TE(S)	 G45	 D install	 ad in	 hori-	 10	40 ∃			///	/IEV//		
	11(3);	Bori	ng was b	lind	drille	d to 14	.0 feet bgs. I	Blind drill lithologies from boring G405.					
1													Page 2 of 3

F	IEL] clien'	DI T: Na	BOR atural Re	sour	√G ces 7	LC Technol	DG	CONTRACTOR: Bulldog Drilling, Inc.	<	€ H	ANSON
	Sit Locatio Proio	te: Co n: 13	offeen Po 4 CIPS I	wer l	Statio , Cof	on - Asl feen, II	n Pond 2 2 62017	Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger	BOF	REHOLE ID: Well ID:	G405D G45D 620.94 # MSI
WI	DATE	S: St Fin R: Cl	art: 8/1 hish: 8/1 oudy, rai	6/20 7/20 in, (h	16 16 1-70:	s)		FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld		Completion: Station:	42.00 ft. BGS 873,998.03N 2,515,322.23E
	Total (in) SAMPL	E.	6 in L	e (%)	1. (lb/ft ³)	$\frac{Qp}{\Gamma \text{ype}}$	TOPOGF Quadr Towns Section	& APHIC MAP INFORMATION: *angle: Coffeen ship: East Fork n 11, Tier 7 N.; Range 3 W.	WATER LEVEL $\underline{\Psi} = 16.00 - 1$ $\underline{\Psi} = $ $\overline{\nabla} =$	INFORMAT During Drilling	ION:
Number	Recov / % Reco	Type	Blows / N - Vali RQD	Moistur	Dry Dei	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	- Borehole Detail	Elevation ft. MSL	Remarks
14A	23/24 96%	ss	<i>1-3</i> 7-7 N=10	16		3.50	42	Very dark gray (10YR3/1) with 5% black (10YR2/1) mottles, moist, very stiff, CLAY with little silt, little medium-grained sand, and trace small gravel. [Continued from previous page]		580	
							42	End of Boring = 42.0 ft. BGS			
		045	D' / "	1.	, ·						

F	EL	D]	BOR	IN	IG	L C)G					ANSON
WE	CLIEN Si Locatio Proje DATE CATHE	T: N te: C n: 13 ct: 10 S: St Fin R: St	atural Res offeen Pov 34 CIPS L 5E0080 cart: 8/19 hish: 8/19 unny, (mid	sourd wer \$ Lane, 9/20 9/20 d-70	ces T Statio , Cof 16 16 s)	echnolo on - Asl feen, II	ogy, Inc. h Pond 2 . 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld		во	REHOLE I Well I Surface Ele Completio Statio	D: G406 D: G406 v: 621.86 ft. MSL n: 18.75 ft. BGS n: 872,521.34N 2,514,702.38E
5	SAMPL	E.	T	EST	INC	Ĵ	TOPOGRA	APHIC MAP INFORMATION:	WATI	ER LEVEL	INFORMA	ATION:
	l (in				/ft ³)	(tsf)	Quadra	ngle: Coffeen	Ţ	= Dry -	During Drill	ing
	Tota very		6 in 1e	e (%	т (lp	$[Q_p]$	I ownsh Section	ip: East Fork 11. Tier 7 N.: Range 3 W.	⊥ ∑	. =		
nber	00 / V0	e	ws/ Valı D	istur	Der	(tsf) ure	Denth	Lithologic		Borehole	Elevation	
INN	Rec % I	Typ	Blo N - RQ	Mo	Dry	Qu Fail	ft. BGS	Description		Detail	ft. MSL	Remarks
								Dark brown (10YR3/3), moist, stiff, SILT with few cla and trace organics.	iy		E	
							2	and trace organics. Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay and trace small gravel. Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay. Grayish brown (10YR5/2) with 5% dark yellowish bro (10YR4/6) mottles, moist, stiff, CLAY with few silt at little fine-grained sand. Very pale brown (10YR7/4) with 25% yellowish brow (10YR5/6) mottles, moist, medium, CLAY with trace s Gray (10YR5/1) with 10% yellowish brown (10YR5/6) mottles, moist, medium, CLAY with trace s Gray (10YR5/1) with 10% yellowish brown (10YR5/and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt, and trace fine- to medium-grained sand. Gray (10YR5/1) with 10% yellowish brown (10YR5/and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt, little fine- to medium-grained sand and trace small gravel. Gray (10YR5/1) with 10% yellowish brown (10YR5/and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with few fine- to medium-grained sand and trace small gravel. Gray (10YR5/1) with 10% yellowish brown (10YR5/and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little si and trace small gravel. Gray (10YR6/1) with 25% brownish yellow (10YR6/and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little si and trace small gravel.	-		- 620 - 618 - 616 - 614 - 612 - 610 - 608	
								Gray (10YR6/1) with 10% brownish yellow (10YR6/ and 5% strong brown (7.5YR4/6) mottles, moist, stifl CLAY with some fine- to medium-grained sand, little s and trace small gravel. Grav (10YR6/1) with 5% brownish vellow (10YR6/6	 8) ilt,			
))) ıy.			
							18-	Gray (10YR5/1), dry, hard, SILT with few clay, few fin to coarse-grained sand and trace small gravel.	ne-		604	
							∄_	End of Boring = 18 75 ft RCS			∐–	
								Lie of boring 10.70 it boo				
NC)TE(S):	G40 Bori	6 installed ng was bl	d in l ind o	borin drille	ıg. d adjac	ent to G406D	l.				

F	EL	D 1	BOR	IN	NG	G L(DG		C HANSON
	CLIEN Sit Locatio Projec DATE	T: N te: C n: 13 xt: 16 S: St Fir	atural Re offeen Po 84 CIPS 1 5E0080 cart: 8/1 hish: 8/1	source wer S Lane, 9/20 9/20	ces T Statio , Cof 16 16	echnolo on - Asl feen, II	ogy, Inc. n Pond 2 L 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill	BOREHOLE ID: G406D Well ID: G46D Surface Elev: 621.91 ft. MSL Completion: 52.00 ft. BGS Station: 872,519.70N 2514 (07.785
WE S	SAMPL	к: 51 Е	inny, (mi T	a-70 EST	s) TINC	7		Eng/Geo: K. Theesteld	2,514,697.78E
nber	ov / Total (in) lecovery	e	<i>ws / 6 in</i> Value D	isture (%)	Den. (lb/ft ³)	(tsf) <i>Qp</i> (tsf) ure Type	TOPOGR Quadr Towns Section	APHIC MAP INFORMATION: angle: Coffeen hip: East Fork 11, Tier 7 N.; Range 3 W. Lithelogie	WATER LEVEL INFORMATION: $\underline{\Psi} = Dry - During Drilling$ $\underline{\Psi} = \sum_{\underline{\nabla}} =$
Nur	Rec % h	Typ	Blo N - RQ	Moj	Dry	Qu Fail	ft. BGS	Description	Detail ft. MSL Remarks
1A	10/24	M	4-3	11		1.50		Dark brown (10YR3/3), moist, stiff, SILT with few cla	
1B	75%	ss ss	4-6 N=7	15		3.00	2	Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay and trace small gravel.	620
2A	12/24 50%	ss	4-6 3-3 N=0	12		2.50			
2B 3A	3/24 13%	SS SS	3-3 4-7 N=7	18 24		2.00		Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay. Grayish brown (10YR5/2) with 5% dark yellowish brow (10YR4/6) mottles, moist, stiff, CLAY with few silt an little fine-grained sand. Very pale brown (10YR7/4) with 25% yellowish brow (10YR5/6) mottles, moist, medium, CLAY with trace si	
4A	20/24 83%	ss	2-3 4-5 N=7	21		1.25	6 8	Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stift CLAY with little silt and trace fine- to medium-grained sand.	616 5) 6 614
5A	19/24 79%	ss	1-3 4-6 N=7	18		1.75		Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt, little fine- to medium-grained san and trace small gravel.	
6A	23/24 96%	ss	2-2 4-5 N=6	18		2.50		Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little sil	
7A	21/24 88%	ss	1-3 4-5 N=7	16		1.00		and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little sil and trace small gravel.	
8A	23/24 96%	ss	1-2 2-2 N=4	18		0.75		Gray (10YR6/1) with 10% brownish yellow (10YR6/8 and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with some fine- to medium-grained sand, little si and trace small gravel.	3) Alt,
8B		()		17		0.75		Gray (10YR6/1) with 5% brownish yellow (10YR6/6 mottles, wet, loose, fine-grained SAND with some clay	606
9A	22/24 92%	ss	4-13 27-23	8				Gray (10YR6/1) with 5% brownish yellow (10YR6/6 mottles, moist, loose, fine-grained SAND with some cla) y.
9B		<u> </u>	11-40	8			18		
10A	17/24 71%	ss	13-31 33-42 N=64	7		4.50		Gray (10YR5/1), dry, hard, SILT with few clay, few fir to coarse-grained sand and trace small gravel.	
NC)TE(S):	G46	D installe	ed in	borii	ng.	20 —		11111002 1

F	EL	DI	BOR	IN	NG	G L(DG		HANSON
WE	CLIENT Sit Location Projec DATES	Γ: Ν e: C n: 13 ct: 16 S: St Fir R: St	atural Re offeen Po 34 CIPS I 5E0080 tart: 8/1 hish: 8/1 unny, (mi	sourd wer \$ Lane, 9/20 9/20 d-70	ces T Statio , Cof 16 16 s)	echnolo on - Asl feen, II	ogy, Inc. n Pond 2 . 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld	BOREHOLE ID: G406D Well ID: G46D Surface Elev: 621.91 ft. MSL Completion: 52.00 ft. BGS Station: 872,519.70N 2,514,697.78E
S	SAMPL	E	Т	EST	rinc	3	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
ber	v / Total (in) covery		s / 6 in 'alue	ture (%)	Den. (lb/ft ³)	sf) Qp (tsf) re Type	Quadra Townsl Section	angle: Coffeen nip: East Fork 11, Tier 7 N.; Range 3 W.	$\underline{\Psi} = Dry - During Drilling$ $\underline{\Psi} = $ $\underline{\nabla} =$
Numl	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
11-1	3/3 100%	SH							
12-1 13-1 13-2 13-3 13-4	8/24 33% 24/24 100%	DS DS					22	Gray (10YR5/1), dry, hard, SILT with few clay, few fi to coarse-grained sand and trace small gravel. [Continued from previous page]	ne- 600
14A 14B 14C	21/21 100%	ss	7-30 32-20 N=62	19 7 8		4.50	26	 <u>Pale brown (10YR6/3), wet, loose, fine-grained SAND</u> Pale brown (10YR6/3), wet, loose, fine-grained SAND small GRAVEL. Gray (10YR5/1), dry, hard, SILT with some clay, tra fine- to coarse-grained sand and trace small gravel. 	2 and 0 22 596
15A	12/24 50%	ss	15-16 10-11 N=26	10		4.50	28	Dark gray (10YR4/1), dry, hard, CLAY with some si trace fine- to coarse-grained sand and trace small grav	lt, el. 594
16A	17/24 71%	ss	2-5 7-11 N=12	16		3.00	30	Dark gray (10YR4/1), moist, very stiff, CLAY with for silt, trace fine- to coarse-grained sand and trace smal gravel.	W 1 592
17A	17/24 71%	ss	2-5 7-12 N=12	15		2.50	32	Dark gray (10YR4/1), moist, very stiff, CLAY with for silt, trace fine- to coarse-grained sand and trace small	w 590
18A	12/24 50%	ss	3-7 12-20 N=19	17		2.50	34	Dark gray (10YR4/1), moist, very stiff, CLAY with ft silt, trace fine- to coarse-grained sand and trace smal gravel.	
19A	23/24 96%	ss	4-4 6-9 N=10	25		1.00		Dark gray (10YR4/1) with 5% very dark gray (10YR5/3) mottles, moist, very stiff, CLAY few silt, trace fine- to coarse-grained sand and trace sm gravel.	/1) with all nd
20A	22/24 92%	ss	2-2 5-7 N=7	17		1.00	30	brown (10YR5/3) mottles, moist, very stiff, CLAY with silt, trace fine- to coarse-grained sand and trace smal gravel Dark gray (10YR4/1), moist, very stiff, CLAY with ft silt, trace fine- to coarse-grained sand and trace smal	few / 1 500
21A	24/24 100%	ss	2-3 4-6 N=7	17		1.00	40	graver.	582
NO	DTE(S):	G46	D installe	ed in	borii	ng.			

ſ

F	CLIENT: Natural Resources Technology, Inc.													
	CLIEN Sit Location	Γ: Ν e: Co n: 13	atural Re offeen Po 34 CIPS I	soure wer S Lane	ces T Statio , Cof	echnolo on - Asl feen, II	ogy, Inc. 1 Pond 2 2 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger	BOREHOLE ID: G406D Well ID: G46D					
WF	Projec DATES CATHEF	rt: 16 S: St Fin R: Su	E0080 art: 8/1 nish: 8/1 nny, (mi	9/20 9/20 d-70	16 16 s)			FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld	Surface Elev: 621.91 ft. MSL Completion: 52.00 ft. BGS Station: 872,519.70N 2,514,697.78E					
5	SAMPL	Е	Т	EST		; 0	TOPOGI	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:					
	otal (i ery		in	(%)	(lb/ft ³	<i>Qp</i> (tsi ype	Towns	hip: East Fork	$\underline{\mathbf{Y}} = \mathbf{D}\mathbf{Y} + \mathbf{D}\mathbf{u}$ ing Diffing $\underline{\mathbf{Y}} = \mathbf{\nabla}$					
mber	соv / Т <i>Recov</i> a	be	ws/6 Value D	isture	y Den.	(tsf) lure T	Depth	Lithologic	$\underline{Y} =$ Borehole Elevation					
Nu	Re %	Ty	Blc N	Ŭ	D.	Qu Fai	ft. BGS	Description	Detail ft. MSL Remarks					
22A	24/24 100%	ss	1-3 5-6 N=8	17		1.50	42		580					
23-1 23-2 23-3 23-4	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$													
24A	$\begin{array}{c c c c c c c c c c c c c c c c c c c $													
25A	24/24 100%	ss	2-2 5-7 N=7	18		1.25	48	[Continued from previous page]	574					
26A	24/24 100%	ss	2-5 6-8 N=11	17		1.75	50		572					
27A	23/24 96%	ss	2-3 9-12 N=12	15		3.50	52		570					
								End of Boring = 52.0 ft . BGS						
NO														

F	[EL]	DI	BOR	IN	NG	L(
	CLIEN Sit Location Project	F: N e: C n: 13	atural Re offeen Po 34 CIPS 1	sour wer S Lane	ces T Statio , Cof	echnolo on - Asl feen, IL	ogy, Inc. CONTRACTOR: Bulldog Drilling, Inc. h Pond 2 Rig mfg/model: CME-750 ATV Drill L 62017 Drilling Method: 4 1/4" Hollow Stem Auger Well ID: G407 Surface Flow: 618 35 ft MSL		
WE	DATE	S: St Fir R: R	tart: 8/1 nish: 8/1 ain, (mid-	6/20 6/20 -70s)	16 16		FIELD STAFF: Driller: J. Dittmaier Completion: 20.00 ft. BGS Helper: M. Hill Station: 872,973.39N Eng/Geo: K. Theesfeld 2,513,705.87E	5	
	SAMPL	E	T	TEST		J _	TOPOGRAPHIC MAP INFORMATION: WATER LEVEL INFORMATION:		
	tal (i v		u	(%)	lb/ft ³	<i>p</i> (tsf be	Quadrangle: Coffeen $\underline{Y} = 16.00$ - During DrillingTownship: East Fork $\underline{Y} =$		
er	/ / Tc		s / 6 i alue	ure (en. ($f_{\rm P} O$	Section 10, Tier 7 N.; Range 3 W. ∑ =		
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	DepthLithologicBoreholeElevationft. BGSDescriptionDetailft. MSLRemarks		
1A	12/24 50%	ss	4-3 3-3 N=6	14		3.50	Very dark gray (10YR3/1), wet, medium, SILT with some organics. [Fill] Gray (10YR6/1), wet, loose, SAND with some gravel and little clay.		
2A	20/24 83%	ss	2-2 4-4 N=6	18		1.50	2 Yellowish brown (10YR5/6) with 5% dark yellowish brown (10YR3/6) mottles, moist, very stiff, SILT with <u>some clay and trace very fine- to fine-grained sand.</u> Brown (10YR5/3) with 10% yellowish brown (10YR5/6) mottles, moist, stiff, SILT with / <u>Brown (10YR5/3) with 10% yellowish brown (10YR5/6)</u> mottles moist, stiff, SILT with fine to		
3A	23/24 96%	ss	<i>1-2</i> <i>3-4</i> N=5	19		1.75	4 <u>coarse-grained sand, and trace small gravel.</u> Brown (10YR5/3) with 25% yellowish brown (10YR5/6) mottles, moist, stiff, CLAY with some silt, trace fine-grained sand and trace small gravel.		
4A	24/24 100%	ss	<i>1-3</i> <i>3-5</i> N=6	19		1.50	6 612 Brown (10YR5/3) with 10% yellowish brown (10YR5/6) mottles, moist, stiff, CLAY with some silt, little fine- to coarse-grained sand and trace small gravel.		
5A	21/24 88%	ss	1-2 4-4 N=6	19		0.50	Yellowish brown (10YR5/6) with 25% brown (10YR5/3) mottles, moist, medium, CLAY with few silt, few fine-grained sand, and trace small gravel.		
6A	22/24 92%	ss	1-2 2-1 N=4	17			Yellowish brown (10YR5/8) with 5% gray (10YR5/1) mottles, moist, very loose, fine-grained SAND with some clay and trace small gravel.		
7A	24/24 100%	ss	7-29 33-17 N=62	8			Gray (10YR5/1) with 25% yellowish brown (10YR5/8) mottles, moist, very dense, fine-grained SAND		
							14 Little fine- to coarse-grained sand 604		
8A	24/24 100%	ss	3-7 12-17 N=19	12		4.50	Yellowish brown (10YR5/4) with 5% yellowish brown (10YR5/6) and 5% black (10YR2/1) mottles, SILT with some clay and little fine- to coarse-grained sand.		
9A	24/24 100%	ss	4-9 14-20 N=23	13		4.00	Yellowish brown (10YR5/4) with 5% yellowish brown (10YR5/6), 5% dark gray (10YR4/1) and 5% black (10YR2/1) mottles, moist, hard, SILT with little fine- to coarse-grained sand and trace small gravel.		
10A	24/24 100%	ss	2-8 14-19 N=22	14		4.50	Yellowish brown (10YR5/4) with 5% yellowish brown (10YR5/6), 5% dark gray (10YR4/1) and 5% black (10YR2/1) mottles, wet, stiff, SILT with little fine- to coarse-grained sand and little small gravel. Dark grayish brown (10YR4/2) with 10% dark yellowish		
	brown (10YR3/6) mottles, moist, hard, CLAY with some silt, little fine- to coarse-grained sand and trace small gravel.								
	End of Borning = 20.0 R. BGS								
NO	NOTE(S): G407 installed in boring.								
1									

FIELD BORING LOG CLIENT: Natural Resources Technology, Inc. Site: Coffeen Power Station - Ash Pond 2 Location: 134 CIPS Lane, Coffeen, IL 62017 Project: 16E0080 DATES: Start: 8/17/2016 Finish: 8/17/2016 WEATHER: Hazy, (low-80s)					Ces T Statio , Cof 16 16)	GLC Technolo on - Asl Teen, II	DG ogy, Inc. n Pond 2 . 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld	BOREHOLE ID: T408 Well ID: T408 Surface Elev: 621.09 ft. MSL Completion: 25.92 ft. BGS Station: 873,999.36N 2,515,314.91E
r	(Total (in)		d in ue	re (%) a	n. (lb/ft ³)	$\frac{Op}{Type}$ (tsf)	TOPOGR Quadr Towns Section	APHIC MAP INFORMATION: angle: Coffeen hip: East Fork 111, Tier 7 N.; Range 3 W.	WATER LEVEL INFORMATION: $\underline{\Psi} = $ Dry - During Drilling $\underline{\Psi} = $ $\underline{\nabla} = $
Numbe	Recov . % Recc	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
	0/60						2	Black (10YR2/1), moist, soft, clayey SILT (TOPSOII	
	0%	BD					4	Gray (10YR6/1) with 30% yellowish brown (10YR5/8 mottles, moist, firm, silty CLAY	3)
	-							Gray (10YR6/1), moist, firm, silty CLAY slight trace sa	und 616
							6	Gray (10YR5/1), very moist, soft, clayey, very fine- to fine-grained SAND	
	0/60	BD						Gray (10YR6/1) with 25% yellowish brown (10YR5/8 mottles, moist, firm, silty CLAY with sand and trace gra	8) vel 614
							8	Yellowish brown (10YR5/8) with 40% gray (10YR6/1 mottles, moist, firm, silty CLAY with sand and trace gra	612 vel
	0/60							Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Dark brown (10YR3/3), very moist, soft, clayey, fine- very coarse-grained SAND with slight trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND	610
	0%	Б						Dark yellowish brown (10YR4/4), moist, soft, sandy SI with trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Gray (10YR5/1), moist, very hard, very silty CLAY wi sand and gravel	LT 608
4A	22/24 92%	ss	8-16 33-35 N=49			4.50	16	Dark gray (10YR4/1), moist, hard, SILT with few fine- coarse-grained sand, little clay and trace small gravel.	to
	0/36 <i>0%</i>	BD						Dark gray (10YR4/1), dry, hard, SILT with few fine- t coarse-grained sand, little clay and trace small gravel.	
NC)TE(S):	T40 Bori	8 installe ng was b	d in lind	borin drille	g. ed to 26	.0 feet bgs. E	Blind drill lithologies from boring G405 and G405D.	Page 1 of 2

F	[EL]	D	BOR	IN	IG	G L(DG			C		
	CLIEN Sit	T: Na ae: Co	atural Re offeen Po	soure wer S	ces T Statio	echnolo on - Asl	ogy, Inc. h Pond 2	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill		BOR	BOREHOLE ID: T408	
	Location: 134 CIPS Lane, Cotteen, IL 6201/				Cof	feen, II	62017	Drilling Method: 4 1/4" Hollow Stem Auger			Well II	D: T408
	Projec	et: 16	E0080	7/20	17					2	Surface Elev	v: 621.09 ft. MSL
	DATE	5: St Fin	art: 8/1	//20 7/20	16 16			FIELD STAFF: Driller: J. Dittmater			Completion	n: 25.92 ft. BGS
WF	ATHE	R: Ha	azv. (low-	-80s)	10			Eng/Geo: K. Theesfeld			Statio	2.515.314.91E
	SAMPL	E	T	EST	TING	2		g,				_,
					1.0		TOPOGR	APHIC MAP INFORMATION:	WATE	R LEVEL	INFORMA	TION:
	ul (i)				/H ³)	(tsf	Quadr	angle: Cotteen		= Dry - I	During Drilli	ng
	Fota ery		e in e	%	(lb	ype 0	I owns Section	hip: East Fork		_		
er.	L / 7		s/6 alu	ture	Jen.	sf) re T	Section	11, Her / N.; Kange 5 W.	<u>¥</u> -	_		
Numb	Recov % Re	Type	<i>вlo</i> ws N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		Borehole Detail	Elevation ft. MSL	Remarks
7-1 7-2 7-3 7-4	0/36 0% 20/24 83% 0/12 0%	BD DS BD					22	Dark gray (10YR4/1), dry, hard, SILT with few fine- coarse-grained sand, little clay and little small grave Dark gray (10YR4/1), moist, dense, fine- to coarse-gra SAND. Dark gray (10YR4/1), dry, hard, SILT with few fine- coarse-grained sand, little clay and little small grave End of Boring = 25.92 ft. BGS	to l. to l.			

CLIENT: Natural Resources Technology, Inc. CONTRACTOR: Bulldog Drilling, Inc. Site: Coffeen Power Station - Ash Pond 2 Rig mfg/model: CME-750 ATV Drill Location: 134 CIPS Lane, Coffeen, IL 62017 Drilling Method: 4 1/4" Hollow Stem Auger Project: 16E0080 Sun DATES: Start: 8/19/2016 FIELD STAFF: Driller: J. Dittmaier Finish: 8/19/2016 Helper: M. Hill WEATHER: Cloudy, (70s) Eng/Geo: K. Theesfeld								BOREHOLE ID: T409 Well ID: T409 Surface Elev: 621.85 ft. MSL Completion: 26.99 ft. BGS Station: 872,517.79N 2,514,693.89E
er ,	/ Total (in)		/ 6 in ulue	en (lb/ft ³)	$(1) \frac{Qp}{Dp} (tsf)$	TOPOGRA Quadra Townsh Section	APHIC MAP INFORMATION: ngle: Coffeen nip: East Fork 11, Tier 7 N.; Range 3 W.	WATER LEVEL INFORMATION: $\underline{\Psi} = Dry - During Drilling$ $\underline{\Psi} = $ $\underline{\nabla} =$
Numb	Recov % Rec	Type	Blows N - V ₈ RQD	Drv D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
	0/60 <i>0%</i>	BD				2 4	Dark brown (10YR3/3), moist, stiff, SILT with few cla and trace organics. Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay and trace small gravel. Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay. Gravish brown (10YR5/2) with 5% dark vellowish brown	ny 620 618
	0/60 <i>0%</i>	BD					(10YR4/6) mottles, moist, stiff, CLAY with few silt an little fine-grained sand. Very pale brown (10YR7/4) with 25% yellowish brow (10YR5/6) mottles, moist, medium, CLAY with trace si Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt and trace fine- to medium-grained sand. Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt, little fine- to medium-grained san and trace small gravel.	$\frac{d}{d} = $
	0/60 <i>0%</i>	BD					Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little sil and trace small gravel. Gray (10YR6/1) with 25% brownish yellow (10YR6/8 and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little sil and trace small gravel. Gray (10YR6/1) with 10% brownish yellow (10YR6/8 and 5% strong brown (7.5YR4/6) mottles, moist, stiff	$\frac{1}{2}$
	0/60 <i>0%</i>	BD					CLAY with some fine- to medium-grained sand, little si and trace small gravel. Gray (10YR6/1) with 5% brownish yellow (10YR6/6 <u>mottles, wet, loose, fine-grained SAND with some clay</u> Gray (10YR6/1) with 5% brownish yellow (10YR6/6 mottles, moist, loose, fine-grained SAND with some cla Gray (10YR5/1), dry, hard, SILT with few clay, few fir to coarse-grained sand and trace small gravel.	lit,) /



Cano	onie Co	nstru	ction Co.	Date Feb. 8	, 1985	
anonie c	anonie Te	st Bori	ng Services	Total Footage 5	0'0"	
milton Lakes 500 Park Boulevard Suite	1212 Itacca Illin	ois 60142	212 772 4877	Foreman W. Ho	<u>11oman</u>	
		0/3 00 143,	0	Classification by	Foreman	
t Hanson Engineers, Inc.	· 1		Geographic Location	Coffeen, Illi	nois	
			DOWING TWO UW-2		t.L.	
CONTINUES				Coordinates		
Ground Surface			Grou	nd Surface		0
very hard brown silty sandy CLAY, with small to medium gravel		-510"	very ha sandy C to larg	rd brown silty LAY with small e gravel		-3
COAL & CINDERS		5.0	COAL & C	CINDERS		
medium brown and gray silty CLAY	PUSHED	-20'0" -22'0"				
screen to a depth of 22	of 2" PVC "0".		medium E silty CL	prown AY	-PUSHED	- 26 - 28
			Installe screen t	d well with 15 o a depth of 2	'0" of 2" PV 8'0".	

D....Figures in right hand column indicate number of blows required to drive 2" O.D. sampling spoon (6" u.n.o.), using a 140 lb. weight falling 30 inches.

Sheet 1 of 4



- A.... All borings are plotted to a scale of 1"=....8.. ft., using ground..surface... as a fixed datum.
- B....Classifications are made from visual inspection of samples and are our opinion thereof.
- C....Water Levels (WL). Figure indicates time of reading (hours) after completion of boring. Water levels indicated are those observed when borings were made, or as noted. Porosity of the soil strata, variations of rainfall, site topography, etc., may cause changes in these levels.
- D....Figures in right hand column indicate number of blows required to drive 2" O.D. sampling spoon (6" u.n.o.), using a 140 lb. weight falling 30 inches.

C Ham Client	Cance anonie C ilton Lakes, 500 Park Boulevard, Suite Hanson Engineers, Inc.	anonie Test	Co. vices	Date Feb. 8, 1985 Total Footage 35'0" Foreman W. Holloman Classification By Foreman Proceeding of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s				
Boring	No. 0W-5 0.G. E			Boring No	•		Q.G. EL.	
	Coordinates		<u>,</u>		•	Coordin	ates	
	Ground Surface	· · ·			• • •	<u> </u>		
	very hard brown silty sandy CLAY with small to_medium_gravel		0'0" 4'0"					
• • • •2						· . · ·		
								•
	COAL & CINDERS mixed with lime				All wel accorda attache "Typica Well Co	ls installe nce with th d "Exhibit l Observati nstruction	ed in ne 2", ion :.	
			00104				•	
	soft brown and gray CLAY Installed well with 25 PVC screen to a depth	 'O" of 2" of 35'O".	33'0" 35'0"					

A All borings are plotted to a scale of 1"= ... R... ft., using ... graund ... surface as a fixed datum.

B....Classifications are made from visual inspection of samples and are our opinion thereof.

C....Water Levels (WL). Figure indicates time of reading (hours) after completion of boring. Water levels indicated are those observed when borings were made, or as noted. Porosity of the soil strata, variations of rainfall, site topography, etc., may cause changes in these levels.

D....Figures in right hand column indicate number of blows required to drive 2" O.D. sampling spoon (6" u.n.o.), using a 140 lb. weight falling 30 inches.



Illinois Environmental Protection Agen	cy		Well	Completion	Report
Site #: County:	Montgomery		W	/ell #:G4	-01
Site Name: Coffeen Power Station			В	orehole #:	G401
State Plane Coordinate: X 872,510.6 Y 2,515,614.8 (or) Latit	tude:		Longitud	e:	
Surveyed By: Gary C. Rogers	IL Registrat	tion #: <u>035-00</u>	02957		
Drilling Contractor: <u>Ramsey</u>	Driller: <u>I</u>	D. Crump			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:	Rhonald W.	Hasenyager	r, LPG #196-0002	246
Drilling Method: Hollow stem auger	Drilling Flu	id (Type): <u>no</u>	ne		
Logged By: <u>Rhonald W. Hasenyager</u>	Date Started	d: <u> </u>	15 Date	e Finished: <u>9/1</u>	4/2015
Report Form Completed By: _ Suzanna L. Keim	Date:	10/7/2015			
ANNULAR SPACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
-		625.84	-2.81	Top of Protective	Casing
L		625.57	-2.54	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>		623.03	0.00	Ground Surface	
Turna of Annular Scalanty - Dontonita China		621.33	1.70	Top of Annular S	ealant
Installation Mathada Cravity					
Setting Time:24 hours	Σ			Static Water Leve	el
Type of Bentonite Seal (Granular) Bellet Shurry				(Anter Completion)	
(choose one)					
Installation Method: <u>Gravity</u>		<u>n/a</u>	<u></u>	Top of Seal	
Setting Time: 25 minutes		610.12	12.91	Top of Sand Pack	
Type of Sand Pack: <u>Quartz Sand</u>					
Grain Size: <u>10-20</u> (sieve size)		_608.67_	14.36	Top of Screen	
Installation Method: <u>Gravity</u>		(04.04	10.70	D	
Type of Backfill Material: <u>n/a</u>		<u>604.24</u> 603.74	18.79	Bottom of Screen Bottom of Well	
(if applicable)		603 73	10.30	Pottom of Poreho	ام
		* Referenced to a	National Geodet	ic Datum	
		CAS	ING MEAS	SUREMENTS	
	Dia	ameter of Boreho	le	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID	of Riser Pipe		(inches)	2.0
	Pro	ptective Casing L	ength	(feet)	5.0
Destastive Casing SS204 SS214 DITE NVG or	HED. Storl	er Pipe Length		(feet)	16.70
Riser Pine Above W T SS304 SS316 PTFE PVC OT	HER. DICE	ttom of Screen to	End Cap	(feet)	0.50
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTT	HER: Scr	tal Length (1s	t slot to last slo	t) (feet)	4.03
Screen SS304 SS316 PTFE (PVC) OT	HER: Scr	een Slot Size **	·····5	(inches)	0.010

Site #: County: Montgomery Well #: G403 Site Anne: County: Montgomery Well #: G403 Site Phote Conditate: X:: Site Conditate: Site Conditate: G403 Site Phote Conditate: X:: Site Conditate: Integritation Headerstand Site Conditions: X:: Site Conditions: Integritation #: 035:002957 Drilling Contractor: Ratreey Driller: D. Cuurp Consulting Finar: Hasenyager, IPG #196:000246 Drilling Method: Holdow stem auger Drilling Fluid (Type): Date Laged By: Kristen L. Theesfeld Date Starred: 9(11/2015) Date Finashed: 9(11/2015) Raport Form Completed By: Starrant I. Keim Date 107/2015 Integritation Method: 9(11/2015) ANNULAR SPACE DETAILS Iterations Depths (0.01 ft.) (MS1) (MS1) Statie Water I evel (MS1) (MS1) (MS1) (MS1) (MS1) Type of Sunface Seal: Constart Plan Date Statie Water I evel (Marc Cooptaixe) Type of Sunface Seal: Garant Plan Number of Staren G4032 G12.64 11.17 Top of Scal <t< th=""><th>Illinois Environ</th><th>mental Protection</th><th>n Agency</th><th></th><th></th><th></th><th>Wel</th><th>l Completion</th><th>Report</th></t<>	Illinois Environ	mental Protection	n Agency				Wel	l Completion	Report
Site Name: Coffeen Power Station Borchole #: G403 Site Site Longitude:	Site #:		County: <u>Mor</u>	ntgomery	/		v	Vell #:64	03
State Interaction statute Interaction Interaction Surveyed By: Gary C. Rögers II. Registration #: Interaction Dilling Contractor: Ramsey Dilling Contractor: Registration #: 055:0022957 Dilling Contractor: Ramsey Dilling Contractor: Registration #: 055:0022957 Dilling Method: Hollow stem auger Dilling Fluid (Type): none Logged By: Kristen L. Theesfeld Date Survey: 9/11/2015 Date Finished: 9/11/2015 Report Form Completed By: Suzanna L. Keim Date: 10/7:2015 Date Finished: 9/11/2015 ANNULAR SPACE DETAILS Elevations Depths (0.01 ft.) (MSL)? (MSL)? (MSL)? (B(S) Type of Surface Seal: Concrete	Site Name: <u>Coffeen Power Sta</u>	ation					E	Borehole #:	G403
Surveyed By: <u>Gary C. Rogers</u> II. Registration #: <u>035-002957</u> Dolling Contractor: <u>Ranksey</u> Driller: <u>D. Crump</u> Consulting Firm: <u>Hanson Professional Services Inc.</u> Geologist: <u>Rhonald W. Hasenyager, I.PG #196-000246</u> Dolling Method: <u>Hollow stem auger</u> Drilling Fluid (Type): <u>none</u> Logged By: <u>Stristen L. Theosfeld</u> Date Started: <u>9/11/2015</u> ANNULAR SPACE DETAILS Date: <u>10/7/2015</u> Very of Surface Seal: <u>Concrete</u> <u>626.47</u> Type of Surface Seal: <u>Concrete</u> <u>621.81</u> Type of Manular Sealant: <u>Bentonite Chips</u> <u>621.81</u> Installation Method: <u>Gravity</u> Suitic Water Level (Atter Coupletion) Type of Sand Pack: <u>Outar's Sand</u> <u>610.70</u> Type of Sand Pack: <u>Outar's Sand</u> <u>610.70</u> Installation Method: <u>Gravity</u> <u>506.61</u> Type of Sand Pack: <u>Outar's Sand</u> <u>610.70</u> Installation Method: <u>Gravity</u> <u>506.66</u> Installation Method: <u>Gravity</u> <u>506.66</u> Installation Method: <u>Gravity</u> <u>506.66</u> Installation Method: <u>Gravity</u> <u>506.66</u> Installation Method: <u>Gravity</u> <u>506.66</u> Installation Method: <u>Gravity</u> <u>507.72</u> Determed Constructin the sach areal	State Plane Coordinate: X 873,561	1.3 Y 2,514,616.6	(or) Latitude:				Longitud	le:	
Deilling Contractor: Ransen Professional Services Inc. Geologist: Rhonald W. Hasenyager, LPG #196-000246 Drilling Method: Hollow stem auger Drilling Muthod: Rhonald W. Hasenyager, LPG #196-000246 Drilling Method: Hollow stem auger Drilling Fluid (Type): none Logged By: Kristen L. Theeseleld Date: 107/2015 ANNULAR SPACE DETAILS Fleevations Depths (0,01 ft.) (MSL)* (BGSS) (0,01 ft.) (MSL)* Type of Surface Seal: Concrete 623.81 0.00 Ground Surface Type of Annular Sealant: Bentonite Chips 621.81 2.00 Top of Annular Sealant Installation Method: _na n/a Top of Saud Pack. 610.70 13.11 Top of Saud Pack. Type of Banchill Material: _nfa mfa 610.60 18.15 Battom of Screen a055.66 18.15 Battom of Screen 606.63 17.78 Battom of Screen a055.66 18.15 Battom of Screen 605.66 18.15 Battom of Screen a055.66 18.15 Battom of Screen 605.66 18.15	Surveyed By: Gary C. Rogers			IL Regi	istrat	ion #: <u>035-0</u>	02957		
Consulting Firm: Ianson Professional Services Inc. Geologist: Rehnald W. Hasenyager, IPG #196-000246 Dailing Heidot: Hollow stem auger Drilling Fluid (Type):	Drilling Contractor: Ramsey			Driller:	E). Crump			
Drilling Method: Hollow stem auger Drilling Fluid (Type):none Logged By: Kristen L. Theesfeld Date Started:	Consulting Firm: Hanson Profe	essional Services Inc.		Geolog	ist:	Rhonald W.	Hasenyage	r, LPG #196-000	246
Logged By: Kristen L. Thecsfield Date Started: 9/11/2015 Date Finished: 9/11/2015 Report Form Completed By: Suzanna I. Keim Date: 10/7/2015 ANNULAR SPACE DETAILS Flevations Depths (0.01 ft.) (MSL)* Depths (0.01 ft.) Type of Surface Seal: Concrete 626.72 -2.91 Top of Protective Casing Type of Annular Sealant: Bentonite Chips 0.00 Ground Surface Installation Method: Gravity Static Water Level (Atter Conference) Type of Sand Pack: Quartz Sand Static Water Level (Atter Conference) Installation Method: Gravity Static Water Level (Atter Conference) Type of Sand Pack: Quartz Sand (drapplaable) 610.70 13.11 Top of Sand Pack Type of Bantonite Gravity Installation Method: Gravity Gravity Bottom of Screen for pof HackHill Material: In/a (drapplaable) 18.15 Bottom of Screen histallation Method: Gravity Gravity Do filser Pipe 005.66 18.15 Bottom of Screen <t< td=""><td>Drilling Method: Hollow stem</td><td>auger</td><td></td><td>Drilling</td><td>g Flui</td><td>d (Type): no</td><td>ne</td><td></td><td></td></t<>	Drilling Method: Hollow stem	auger		Drilling	g Flui	d (Type): no	ne		
Report Form Completed By: Suzanna L. Keim Date: 10/7/2015	Logged By: Kristen L. Theesf	eld		Date St	tarted	l: 9/11/20	15 Dat	te Finished: 9/1	1/2015
ANNULAR SPACE DETAILS Elevations Depths (MSL)* (0.01 ft.) Type of Surface Seal: Concrete 626.47 -2.66 Top of Protective Casing Type of Surface Seal: Concrete 623.81 0.00 Ground Surface Type of Annular Sealant: Bentonite Chips 621.81 2.00 Top of Annular Sealant Installation Method: Gravity Static Water Level (Atter Completion) 53tatic Water Level (Atter Completion) Type of Bentonite Seal - Granular Pellet Starry (choose one) Static Top of Seal Setting Time: n/a Static Water Level (Atter Completion) Static Water Level (Atter Completion) Type of Sand Puck: Quartz Sand Type of Sand Puck: Quartz Sand Type of Backfill Material: n/a Type of Backfill Material: Type of Sand Puck: <td>Report Form Completed By: Su</td> <td>zanna L. Keim</td> <td></td> <td>Date:</td> <td></td> <td>10/7/2015</td> <td></td> <td></td> <td></td>	Report Form Completed By: Su	zanna L. Keim		Date:		10/7/2015			
Image: Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section of Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Section Sectin Sectin Section Sectin Section Section Section Secti	ANNIILAR SPA	CE DETAILS		-		Elevations	Denths	(0.01 ft)	
526.72 -2.91 Top of Protective Casing Type of Surface Scal: Concrete 626.47 -2.66 Top of Riser Pipe Type of Annular Scalant: Bentonite Chips 621.81 0.00 Ground Surface Installation Method: Gravity 621.81 2.00 Top of Annular Scalant Installation Method: Gravity Static Water Level (After Completion) (After Completion) Type of Bentonite Scal - Granular Pellet Slurry (above one) Static Water Level (After Completion) Type of Sand Pack: Quartz Sand 612.64 11.17 Top of Sand Pack Type of Backfill Material: n/a (drapheable) 605.66 18.15 Bottom of Screen Installation Method:						(MSL)*	(BGS)	(0.01 10.)	
Setting Time:						626.72	-2.91	Top of Protective	Casing
Type of Surface Seal: Concrete Type of Annular Sealant: Bentonite Chips Installation Method: Gravity Setting Time: >24 hours Type of Bentonite Seal - Granular Pellet Slurry (choose one) Installation Method: n/a Type of Bentonite Seal - Granular Pellet Slurry (choose one) Installation Method: n/a Type of Sand Pack: Quartz Sand Grain Size: 10-20 (store size) 610.70 Installation Method: Gravity Type of Backfill Material: n/a (frappleable) (frappleable) Installation Method: (frappleable) Installation Method: Gravity Choose one type of material for each area) CASING MEASUREMENTS Number of Borehole * Referenced to a National Genderic Datum CASING MEASUREMENTS Diameter of Borehole * Referenced to a National Genderic Datum CASING MEASUREMENTS Diameter of Borehole (media Store) Riser Pipe Above W.T. Stole PTF Riser Pipe						626.47	-2.66	Top of Riser Pipe	
Type of Annular Sealant: Bentonite Chips Installation Method: Gravity Setting Time: >24 hours Type of Bentonite Seal – Granular Pellet Slurry Installation Method: n/a Type of Bentonite Seal – Granular Pellet Slurry Installation Method: n/a Installation Method: n/a Oran Pack: Quartz Sand Grain Size: 10-20 Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (if applicable) Installation Method: (if applicable) Installation Method: CASING MEASUREMENTS VELL CONSTRUCTION MATERIALS (Choose one type of material for each area) CASING MEASUREMENTS VELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Diameter of Borehole (nees) 8.0 ID of Riser Pipe (nees) 2.0 Protective Casing \$334< \$3316	Type of Surface Seal: <u>Concrete</u>					623 81	0.00	Ground Surface	
Type of Annular Sealant: Bentonite Chips Installation Method: Gravity Setting Time: >24 hours Static Water Level (After Completion) Type of Bentonite Seal - Granular Pellet Slurry (choose one) Installation Method: $n'a$ Top of Seal Setting Time: n/a Top of Seal Setting Time: n/a Top of Seal Setting Time: n/a Top of Seal Grain Size: 10-20 (siew size) Installation Method: Gravity Gold Second Type of Backfill Material: n/a n/a (rf applicable) (rf applicable) Edition Installation Method: m/a 17.78 Bottom of Screen 605.66 18.15 Bottom of Screen 605.66 18.15 Bottom of Borehole * * Referenced to a National Geodette Datam CASING MEASUREMENTS CASING MEASUREMENTS Diameter of Borehole (mekes) 2.0 Protective Casing S304 S316 PTFE PVC Riser Pipe Above W.T. S334 S316					7	621.81	2 00	Top of Annular S	ealant
Installation Method: Gravity Setting Time: ≥ 24 hours Type of Bentonite Seal – Granular Pellet Slurry Installation Method: n/a Top of Seal Setting Time: n/a Top of Seal Setting Time: n/a Top of Seal Setting Time: n/a Top of Seal Grain Size: 10-20 (siew size) Installation Method: Gravity Gravity Type of Backfill Material: n/a for of Screen Grain Size: 10.20 (siew size) Installation Method: n/a for of Screen Grain Size: 10.20 (siew size) Installation Method: n/a for of Screen Grain Size: 10.20 (siew size) Installation Method: n/a for of Screen Method: n/a n/a for of Screen Method: n/a for of Screen for of Screen Installation Method: n/a for of Screen for of Screen (Choose one type of materal for each area) 000 </td <td>Type of Annular Sealant: <u>Benton</u></td> <td>nite Chips</td> <td>- 1</td> <td>- FP</td> <td></td> <td></td> <td></td> <td></td> <td>calant</td>	Type of Annular Sealant: <u>Benton</u>	nite Chips	- 1	- FP					calant
Setting Time: $_>24$ hours \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare \blacksquare	Installation Method: <u>Gravit</u>	у							
Type of Bentonite Seal - Granular Pellet Slurry Installation Method: n/a Top of Seal Setting Time: n/a n/a Top of Seal Grain Size: 10-20 (sieve size) 610.70 13.11 Top of Screen Installation Method: Gravity 606.03 17.78 Bottom of Screen Type of Backfill Material: n/a 605.66 18.15 Bottom of Screen Installation Method: (if applicable) 605.66 18.15 Bottom of Borehole Installation Method:	Setting Time: <u>>24 hours</u>		<u> </u>	∠				Static Water Leve (After Completion)	el
(choose one)	Type of Bentonite Seal Gran	ular Pellet Slurry							
Installation Nethod: Ind	Installation Method: n/a	(choose one)		Ŭ.		n/a	n/a	Top of Seal	
Setting Time. Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image	Setting Time: n/a					<u> </u>	u	Top of Sear	
Type of Sand Pack: Quartz Sand Grain Size: 10-20 Installation Method: Gravity Type of Backfill Material: n/a (if applicable) 606.03 Installation Method: (if applicable) MELL CONSTRUCTION MATERIALS Diameter of Borehole (inches) 2.0 Protective Casin				X		612.64	11.17	Top of Sand Pack	ζ.
Grain Size: 10-20 (sieve size) Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (if applicable) Installation Method: (if applicable) Installation Method: 605.66 18.15 Bottom of Screen 605.66 18.15 Bottom of Borehole 605.66 * Referenced to a National Geodetic Datum CASING MEASUREMENTS WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Protective Casing \$\$304 \$\$316 PIFE PVC OTHER: Steel Riser Pipe Above W.T. \$\$304 \$\$316 PIFE PVC OTHER: Screen Length (1st slot to last slot) (feet) 0.37 Screen Length of Casing (feet) 20.81 10 of Casing (feet) 20.81	Type of Sand Pack: <u>Quartz San</u>	d							
Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (if applicable) Installation Method: (if applicable) Installation Method: (if applicable) WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) (if PTFE PVC OTHER: Steel Protective Casing \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Above W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Above W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$316 \$PTFE \$PVC OTHER: Steel Riser Pipe Below W.T. \$\$304 \$\$\$316 \$PTFE \$PVC OTHER: St	Grain Size: 10-20 (sie	ve size)				610.70		Top of Screen	
Type of Backfill Material: n/a Image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen in the image: Construction of Screen	Installation Method: <u>Gravit</u>	у	- $ $			606.02	1770	D-#	
(if applicable) Installation Method:	Type of Backfill Material: <u>n/a</u>					605.66	17.78	Bottom of Screen Bottom of Well	
Installation Method:		(if applicable)							
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Diameter of Borehole (inches) 8.0 Do f Riser Pipe (inches) 2.0 Protective Casing \$\$304 \$\$316 PTFE PVC OTHER: Steel 15.77 Riser Pipe Above W.T. \$\$304 \$\$316 PTFE PVC OTHER: Bottom of Screen to End Cap (feet) 0.37 Riser Pipe Below W.T. \$\$304 \$\$316 PTFE PVC OTHER: Total Length of Casing (feet) 4.67	Installation Method:					605.66 * Referenced to a	18.15 National Geode	Bottom of Boreho	ble
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Diameter of Borehole (inches) 8.0 ID of Riser Pipe (inches) 2.0 Protective Casing SS304 SS316 PTFE PVC OTHER: Steel Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Quarter of Borehole (inches) 2.0 Riser Dipe Length (feet) 5.0 Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Quarter of Borehole (feet) 20.81 Quarter of Borehole (feet) 20.81						CAS		SUDEMENTS	
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) ID of Riser Pipe (inches) 2.0 ID of Riser Pipe (inches) 2.0 Protective Casing SS304 SS316 PTFE PVC OTHER: Steel Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Screen Length (1st slot to last slot) (feet) 4.67 Total Length of Casing (feet) 20.81					Dia	meter of Boreho		(inches)	8.0
Protective Casing SS304 SS316 PTFE PVC OTHER: Sereen to End Cap (feet) 5.0 Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Screen to End Cap (feet) 0.37 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Screen Length (feet) 4.67 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Total Length of Casing (feet) 20.81	WELL CONS	TRUCTION MATERIA	LS			of Riser Pipe		(inches)	2.0
Protective Casing SS304 SS316 PTFE PVC OTHER: Steel Riser Pipe Length (feet) 15.77 Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Bottom of Screen to End Cap (feet) 0.37 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Screen Length (1st slot to last slot) (feet) 4.67 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Total Length of Casing (feet) 20.81	(choose on	, pe or material for each area)			Pro	tective Casing L	ength	(feet)	5.0
Protective CasingSS304SS316PTFEPVCOTHER: SteelBottom of Screen to End Cap(feet)0.37Riser Pipe Above W.T.SS304SS316PTFEPVCOTHER:Screen Length(1st slot to last slot)(feet)4.67Riser Pipe Below W.T.SS304SS316PTFEPVCOTHER:Total Length of Casing(feet)20.81	[1]	Ris	er Pipe Length		(feet)	15.77
Kiser Pipe Above W.1. SS304 SS316 PIFE PVC OTHER: Screen Length (1st slot to last slot) (feet) 4.67 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Total Length of Casing (feet) 20.81	Protective Casing	SS304 SS316 PTFE	PVC OTHER: S	Steel	Bot	tom of Screen to	o End Cap	(feet)	0.37
Total Length of Casing (feet) 20.81	KISET PIPE Above W.T. Riser Pipe Below W.T.	55304 55316 PIFE	PVC OTHER:		Scr	een Length (1s	st slot to last slo	ot) (feet)	4.67
Screen SS304 SS316 P1FE PVC OTHER: Screen Slot Size ** (index) 0.010	Screen	SS304 SS316 PTFE (PVC OTHER:		Scr	ai Length of Ca een Slot Size **	sing	(inches)	20.81

Illinois Environ	mental Protection	Agency			Well	Completion	1 Report
Site #:	(County: <u>Mor</u>	ntgomery	r	W	Vell #:G	404
Site Name: Coffeen Energy Co	enter				В	orehole #:	SB22
State Plane Coordinate: X 2,516,397	<u>7.9</u> Y <u>873,999.8</u> ((or) Latitude:			Longitud	e:	
Surveyed By: <u>Darren E. Forgy</u>	,		IL Regi	stration #: <u>035-</u>	003637		
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.		Driller:	A. Rachford			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.		Geolog	ist:, LPG #			
Drilling Method: <u>Hollow stem</u>	auger		Drilling	gFluid (Type): <u>n</u>	one		
Logged By: <u>Rhonald W. Hase</u>	nyager		Date St	arted: <u>5/1/20</u>	007 Dat	e Finished: <u>5</u>	/1/2007
Report Form Completed By:Rh	onald W. Hasenyager		Date:	5/2/2007			
ANNULAR SPA	CE DETAILS			Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
				616.02	-2.92	Top of Protective	e Casing
				615.77	-2.67	Top of Riser Pip	e
Type of Surface Seal: <u>Concrete</u>				<u> </u>	0.00	Ground Surface	
Type of Annular Sealant:	nite chips			613.10	0.00	Top of Annular S	Sealant
Installation Method:gravity	,						
Setting Time: <u>>12 hours</u>			Z	611.03	2.07	Static Water Lev (After Completion)	el 5/10/2007
Type of Bentonite Seal Gran	ular Pellet Slurry						
Installation Method:	(choose one)	— K	x x	n/a	n/a	Top of Seal	
Setting Time:		-	×	608.05	5.05	Top of Sand Pac	k
Type of Sand Pack:Quartz sand	l						
Grain Size: 10/20 (sie	ve size)			606.68	6.42	Top of Screen	
Installation Method: gravity	,	-		601.02	11 17	Dattam of Saraa	
Type of Backfill Material:	ation sand (if applicable)			601.48	11.62	Bottom of Well	1
Installation Method: <u>slough</u>				<u>601.10</u> * Referenced to	12.00	Bottom of Boreh	ole
				CA Diamatar of Paral	SING MEA	SUREMENTS	8.0
WELL CONS	TRUCTION MATERIAL	LS		Diameter of Borel	1018	(inches)	2.0
(Choose on	e type of material for each area)			Protective Casing	Length	(feet)	5.0
				Riser Pipe Length		(feet)	9.09
Protective Casing	SS304 SS316 PTFE	PVC OTHER:	Steel	Bottom of Screen	to End Cap	(feet)	0.45
Riser Pipe Above W.T.	SS304 SS316 PTFE	PVC OTHER:		Screen Length (1st slot to last slo	t) (feet)	4.75
Riser Pipe Below W.T.	SS304 SS316 PTFE	PVC) OTHER:		Total Length of C	asing	(feet)	14.29

Screen

SS304

SS316

PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Environmental Prot	tection Agency			Well	Completior	n Report
Site #:	County: <u>Mo</u>	ntgomery		W	/ell #:G4	405
Site Name: Coffeen Energy Center				В	orehole #:	SB21
State Plane Coordinate: X 2,515,335.7 Y 873,9	096.8 (or) Latitude:			Longitud	e:	
Surveyed By: <u>Darren E. Forgy</u>		IL Regis	stration #: <u>035-0</u>	03637		
Drilling Contractor: <u>Reynolds Drilling Corp.</u>		Driller:	A. Rachford			
Consulting Firm: <u>Hanson Professional Services</u>	Inc.	Geologis	st:, LPG #			
Drilling Method: Hollow stem auger		Drilling	Fluid (Type): <u>no</u>	ne		
Logged By:Rhonald W. Hasenyager		Date Sta	urted: <u>5/1/200</u>	07 Dat	e Finished: <u>5</u> /	/1/2007
Report Form Completed By: <u>Rhonald W. Hasen</u>	yager	Date:	5/2/2007			
ANNULAR SPACE DETAILS			Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
			624.04	3.14	Top of Protective	e Casing
			623.78	2.88_	Top of Riser Pipe	e
Type of Surface Seal: Concrete			620.90	0.00	Ground Surface	
Type of Annular Sealant Bentonite chins			620.90	0.00	Top of Annular S	Sealant
Installation Method: aravity	¥					
Setting Time: _ >12 hours		⊻	619.67	1.23	Static Water Leve (After Completion)	el 5/10/2007
Type of Bentonite Seal Granular Pellet (choose one)	Slurry					
Installation Method:	×	x x	n/a	n/a	Top of Seal	
Setting Time:	×	×	613.19	7.71	Top of Sand Pac	k
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)			611.89	9.01	Top of Screen	
Installation Method: gravity			607.14	13.76	Bottom of Screen	1
Type of Backfill Material:	>)		606.69	14.21	Bottom of Well	
Installation Method:			606.69 * Referenced to a	14.21 National Geodet	Bottom of Boreh	ole
		Γ	Diameter of Boreho	ING MEA	SUREMENTS (inches)	8.0
WELL CONSTRUCTION MA (Choose one type of material for eac	ATERIALS h area)		ID of Riser Pipe		(inches)	2.0
		-	Protective Casing L	ength	(feet)	5.0
Destanting Conting	DEEE DVG omme		Riser Pipe Length		(feet)	11.89
Protective Casing SS304 SS316 Riser Pipe Above W.T. SS204 SS216	PIFE PVC OTHER:	steel	Bottom of Screen to	End Cap	(feet)	0.45
Riser Pine Below W T SS304 SS316	PTFE PVC OTHER:		Screen Length (1s	t slot to last slo	t) (feet)	4.75
Screen SS304 SS316	PTFE PVC OTHER:		Screen Slot Size **	sing	(inches)	0.010

Illinois Environmental Protection Agency	Well Completion	n Report
Site #: County: _ Mon	ntgomery Well #:	645D
Site Name: <u>Coffeen Power Station - Ash Pond 2</u>	Borehole #:	G405D
State Plane Coordinate: X_2,515,322.2 Y_873,998.0 (or) Latitude:	<u>39° 3' 51.657"</u> Longitude: <u>-89°</u>	<u>23' 46.612"</u>
Surveyed By:Gary C. Rogers	IL Registration #:035-002957	
Drilling Contractor: Bulldog Drilling, Inc.	Driller: J. Dittmaier	
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:	0246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):	
Logged By: Kristen L. Theesfeld	Date Started:8/16/2016 Date Finished:8	8/17/2016
Report Form Completed By:Suzanna L. Keim	Date:8/24/2016	
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft (MSL)* (BGS)	.)
	<u>624.16</u> <u>-3.22</u> Top of Protectin	ve Casing
		pe
Type of Surface Seal: Concrete	<u>620.94</u> <u>0.00</u> Ground Surface	2
Type of Annular Sealant: <u>High-solids bentonite</u>	<u>618.94</u> <u>2.00</u> Top of Annular	Sealant
Installation Method:		
Setting Time: >24 hours	∑	wel) 8/29/2016
Type of Bentonite Seal Granular Pellet Slurry		
Installation Method: <u>Gravity</u>	<u>590.59</u> <u>30.35</u> Top of Seal	
Setting Time: <u>38 minutes</u>	<u>589.62</u> <u>31.32</u> Top of Sand Pa	ck
Type of Sand Pack: <u>Quartz Sand</u>	589.06 31.88 Top of Screen	
Grain Size: <u>10-20</u> (sieve size)		
	<u>579.42</u> <u>41.52</u> Bottom of Scree	en
Type of Backfill Material:	<u>579.02</u> <u>41.92</u> Bottom of Well	
Installation Method:	578.94 42.00 Bottom of Bore * Referenced to a National Geodetic Datum	hole
	Diameter of Borehole (inches	80
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches	b) 2.0
	Protective Casing Length (fee	5.0
Protective Casing SS304 SS316 PTEE PV/C OTHER R	Riser Pipe Length (fee	34.75
Riser Pipe Above W.T. SS304 SS316 PTFE (PVC) OTHER:	Bottom of Screen to End Cap (fee	0.40 0.64
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (fee	44 79
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:Mon	ontgomery Well #: G406
Site Name: <u>Coffeen Power Station - Ash Pond 2</u>	Borehole #: G406
State Plane Coordinate: X_2,514,702.4 Y_872,521.3 (or) Latitude:	: <u>39°</u> <u>3'</u> <u>37.114"</u> Longitude: <u>-89°</u> <u>23'</u> <u>54.628"</u>
Surveyed By: <u>Gary C. Rogers</u>	IL Registration #:035-002957
Drilling Contractor: Bulldog Drilling, Inc.	Driller: J. Dittmaier
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):
Logged By: Kristen L. Theesfeld	Date Started:
Report Form Completed By:Suzanna L. Keim	Date:8/24/2016
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
	<u></u>
Type of Surface Seal: Concrete	<u>621.86</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	619.862.00 Top of Annular Sealant
Installation Method: Tremie	
Setting Time: _>24 hours	∑ Static Water Level (After Completion)
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	<u>610.74</u> <u></u>
Setting Time: <u>30 minutes</u>	609.65 <u>12.21</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz Sand</u> Grain Size: <u>10-20</u> (sieve size)	
Installation Method: <u>Gravity</u>	<u></u>
Type of Backfill Material:	603.11 18.75 Bottom of Well
Installation Method:	<u>603.11</u> <u>18.75</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTFF PVC OTHER.	Riser Pipe Length (feet) 17.06 Steel Detterm of Server (F. L.C.) Detterm of Server (F. L.C.) Detterm of Server (F. L.C.)
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER	Bottom of Screen to End Cap (feet) 0.38 Screen Length (latelate last slat) (feet) 4.91
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 22.25
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency		Well Completion Report		
Site #: County:	itgomery	Well #: G46D		
Site Name: Coffeen Power Station - Ash Pond 2		Borehole #:G406D		
State Plane Coordinate: X_2,514,697.8 Y_872,519.7 (or) Latitude:	<u>39° 3' 37.098"</u>	Longitude: <u>-89°</u> <u>23'</u> <u>54.687"</u>		
Surveyed By:Gary C. Rogers	IL Registration #:035-002	2957		
Drilling Contractor:Bulldog Drilling, Inc	Driller: J. Dittmaier			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W. H</u>	lasenyager, LPG #196-000246		
Drilling Method: Hollow stem auger	Drilling Fluid (Type): <u>non</u>	e		
Logged By: Kristen L. Theesfeld	Date Started: <u>8/19/201</u>	6 Date Finished:8/19/2016		
Report Form Completed By: <u>Suzanna L. Keim</u>	Date: <u>8/24/2016</u>			
ANNULAR SPACE DETAILS	Elevations	Depths (0.01 ft.)		
	625.53	-3.62 Top of Protective Casing		
		-3.33 Top of Riser Pipe		
Type of Surface Seal: Concrete	<u>621.91</u>	0.00 Ground Surface		
Type of Annular Sealant: High-solids bentonite	619.91	2.00 Top of Annular Sealant		
Installation Method: Tremie				
Setting Time: _>24 hours	<u>7</u> <u>589.60</u> _	32.31 Static Water Level (After Completion) 8/29/2016		
Type of Bentonite Seal Granular Pellet Slurry (choose one)				
Installation Method: <u>Gravity</u>	582.34	39.57 Top of Seal		
Setting Time: <u>>12 hours</u>				
Type of Sand Pack: <u>Quartz Sand</u> Grain Size: <u>10-20</u> (sieve size)		41.61 Top of Screen		
	570.65	51.26 Bottom of Screen		
Type of Backfill Material:	570.26	51.65 Bottom of Well		
Installation Method:	<u>569.91</u> * Referenced to a N	52.00 Bottom of Borehole ational Geodetic Datum		
	CASE			
	Diameter of Borehole	e (inches) 80		
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	(inches) 2.0		
	Protective Casing Let	ngth (feet) 5.0		
Destacting Optime South South NEED IN 10	Riser Pipe Length	(feet) 44.94		
Protective Casing SS304 SS316 PTFE PVC OTHER: [S Riser Pine Above W T SS304 SS316 PTFE PVC OTHER: [S	Bottom of Screen to	End Cap (feet) 0.39		
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casis	slot to last slot) (feet) 9.65		
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **	(inches) 0.010		

Illinois Environ	mental Protection	Agency				Well	Completion	n Report
Site #:	Co	ounty: <u>Mon</u>	itgomery	y		W	/ell #:G	407
Site Name: <u>Coffeen Power Sta</u>	ation - Ash Pond 2					В	orehole #:	G407
State Plane Coordinate: X_2,513,705	5.9 Y <u>872,973.4</u> (or	r) Latitude:	39°	3	<u> </u>	Longitud	e: <u>-89°</u> 2	<u>4' 7.213"</u>
Surveyed By: <u>Gary C. Rogers</u>			IL Reg	istrati	on #: <u>035-0</u>	02957		
Drilling Contractor: <u>Bulldog D</u>	rilling, Inc.		Driller:	: <u>J.</u>	Dittmaier			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.	Geolog	gist: _	Rhonald W.	Hasenyager	r, LPG #196-000)246	
Drilling Method: <u>Hollow stem</u>	auger		Drilling	g Fluid	d (Type): <u>nc</u>	one		
Logged By: <u>Kristen L. Theesf</u>	eld		Date St	tarted	:8/16/20) <u>16</u> Date	e Finished: <u>8</u> /	16/2016
Report Form Completed By: <u>Su</u>	zanna L. Keim		Date:		8/24/2016			
ANNULAR SPA	CE DETAILS]	Elevations	Depths	(0.01 ft.)
					621.70	-3.35	Top of Protective	e Casing
					621.32	-2.97	Top of Riser Pip	e
Type of Surface Seal: <u>Concrete</u>				:	618.35	0.00	Ground Surface	
					616.35	2.00	Top of Annular	Sealant
Type of Annular Sealant: <u>High-s</u>	solids bentonite	- 🏹	TP -				1	
Installation Method: <u>Tremie</u>	2	- _	_					1
Setting Time: <u>>24 hours</u>		-					(After Completion)	'el
Type of Bentonite Seal Gram	ular Pellet Slurry (choose one)			-				
Installation Method:	у	- **	x x		607.50	10.85	Top of Seal	
Setting Time: <u>15 minutes</u>		- 🕅	×		605.50	12.85	Top of Sand Pac	k
Type of Sand Pack:Quartz Sand	d							
Grain Size: 10-20 (sie	ve size)		╡		_604.57_	13.78	Top of Screen	
Installation Method:Gravit	У	-						
Type of Backfill Material:Quart	tz Sand				<u>599.74</u> 599.31	<u>18.61</u> <u>19.04</u>	Bottom of Screen Bottom of Well	n
In the life in Matheda Constit	(if applicable)				508 25	20.00	Dettern of Densh	-1-
Installation Method: Gravit	y	_ L			* Referenced to a	National Geodet	ic Datum	lole
					CAS	SING MEAS	SUREMENTS	
		~		Diar	meter of Boreho	ole	(inches)	8.0
WELL CONS (Choose on	EXAMPLE 11 IN MATERIALS (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	8		ID c	of Riser Pipe		(inches)	2.0
				Prot	ective Casing I	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTFF P	VC OTHER G	teel	Rise	er Pipe Length	F 1 C	(feet)	16.75
Riser Pipe Above W.T.	SS304 SS316 PTFE P	VC) OTHER:		Bott Sore	tom of Screen t	o End Cap	(feet)	0.43
Riser Pipe Below W.T.	SS304 SS316 PTFE P	VC OTHER:		Tota	al Length of Ca	sing	(feet)	22.01

SS304

Screen

Well Completion Form (revised 02/06/02)

SS316

PTFE PVC OTHER:

Screen Slot Size **

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Environmental Protection Agency		Well Completion	on Report	
Site #: County:	ntgomery	Well #:	T408	
Site Name: Coffeen Power Station - Ash Pond 2		Borehole #:	T408	
State Plane Coordinate: X 2,515,314.9 Y 873,999.4 (or) Latitude:	<u>39° 3' 51.671"</u>	Longitude: <u>-89°</u>	<u>23' 46.704"</u>	
Surveyed By:Gary C. Rogers	IL Registration #:035-002	2957		
Drilling Contractor: Bulldog Drilling, Inc.	Driller: J. Dittmaier			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W. H</u>	Hasenyager, LPG #196-00	00246	
Drilling Method: Hollow stem auger	Drilling Fluid (Type): <u>non</u>	e		
Logged By: Kristen L. Theesfeld	Date Started: <u>8/17/201</u>	6 Date Finished:	8/17/2016	
Report Form Completed By:Suzanna L. Keim	Date: <u>8/24/2016</u>			
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (0.01 f	t.)	
		<u>-3.35</u> Top of Protecti	ve Casing	
		-2.99 Top of Riser P	ipe	
Type of Surface Seal: Concrete	<u>621.09</u>	0.00 Ground Surfac	e	
Type of Annular Sealant: High solids bentonite	619.09	2.00 Top of Annula	r Sealant	
Installation Method: Tremie				
Setting Time: _>24 hours	<u></u>	2.88 Static Water L (After Completio	evel n) 9/1/2016	
Type of Bentonite Seal Granular Pellet Slurry				
Installation Method: <u>Gravity</u>	602.99	18.10 Top of Seal		
Setting Time: <u>30 minutes</u>	602.04 <u>19.05</u> Top of Sand Pack			
Type of Sand Pack: <u>Quartz Sand</u> Grain Size: <u>10-20</u> (sieve size) Installation Method: Gravity		20.66 Top of Screen		
		25.49 Bottom of Scre	en	
Type of Backfill Material:		25.92 Bottom of Wel	1	
Installation Method:	<u>595.17</u> * Referenced to a N	25.92 Bottom of Bord	ehole	
	CASI	NG MEASUREMENTS		
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	(inche	s) 2.0	
() more one offer on and a constrainty	Protective Casing Le	ength (fee	et) 5.0	
	Riser Pipe Length	(fee	et) 23.65	
Protective Casing SS304 SS316 PTFE PVC OTHER: [\$ Riser Pine Above W T SS304 SS316 PTEE PVC OTHER: [\$	Bottom of Screen to	End Cap (fee	t) 0.43	
Riser Pipe Below W.T. SS304 SS310 PTE PVC OTHER:	Total Length (1st)	slot to last slot) (fee	$\frac{4.83}{28.01}$	
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **	(inche	$\frac{20.91}{0.010}$	

Illinois Environ	mental Protection	Agency				Well	l Completi	on Report	
Site #:	Co	ounty: <u>Mon</u>	itgomery	7		W	Vell #:	T409	
Site Name: <u>Coffeen Power Sta</u>	ation - Ash Pond 2					В	orehole #:	T409	
State Plane Coordinate: X_2,514,693	<u>3.9</u> Y <u>872,517.8</u> (or) Latitude:	<u>39°_</u>	3	<u>' 37.079"</u>	Longitud	e: <u>-89°</u>	23' <u>54.736''</u>	
Surveyed By: <u>Gary C. Rogers</u>			IL Regi	stratio	on #: <u>035-0</u>	02957			
Drilling Contractor: <u>Bulldog D</u>	Drilling, Inc.	Driller: J. Dittmaier							
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologi	ist: _	Rhonald W.	Hasenyager	r, LPG #196-(000246		
Drilling Method: <u>Hollow stem</u>	auger		Drilling	g Fluic	l (Type): <u>no</u>	ne			
Logged By: <u>Kristen L. Theesf</u>	èld		Date St	arted:	8/19/20	0 <u>16</u> Dat	e Finished:	8/19/2016	
Report Form Completed By:	izanna L. Keim		Date: _		8/24/2016				
ANNULAR SPA	CE DETAILS]	E levations (MSL)*	Depths (BGS)	(0.01	ft.)	
					625.35	3.50	Top of Protec	tive Casing	
					625.01	3.16	Top of Riser l	Pipe	
Type of Surface Seal: <u>Concrete</u>			Y D	>—	621.85	0.00	Ground Surfa	ce	
Type of Annular Sealant: High-	solids bentonite				619.85	2.00	_ Top of Annular Sealant		
Installation Method: Tremie	e								
Setting Time:		_ <u> </u>	Z		615.80	6.05	Static Water I (After Completi	Level on) 8/31/2016	
Type of Bentonite Seal Gran	ular Pellet Slurry								
Installation Method:	y		××		602.65	19.20	Top of Seal		
Setting Time: <u>30 minutes</u>		-	<u>601.30</u> <u>20.55</u> Top of Sand Pack				Pack		
Type of Sand Pack: <u>Quartz San</u> Grain Size: <u>10-20</u> (sie	d eve size)				600.06	21.79	Top of Screen	i.	
Installation Method: <u>Gravit</u>	У	-			595.26	26.59	Bottom of Sci	een	
Type of Backfill Material:	(if applicable)	_ _				_20.99_	Bottom of We		
Installation Method:		_			594.86 * Referenced to a	 National Geodet	Bottom of Bo tic Datum	rehole	
					CAS		SUBEMENIT	2	
				Dian	neter of Boreho	ole	(incl	(es) 8.0	
WELL CONS (Choose on	STRUCTION MATERIALS the type of material for each area)	5		ID o	f Riser Pipe		(incl	les) 2.0	
				Prot	ective Casing L	ength	(f	eet) 5.0	
Protective Casing	SS304 SS316 PTFF PV	C OTHER G	iteel	Rise	r Pipe Length	- E- 1C	(f	eet) 24.95	
Riser Pipe Above W.T.	SS304 SS316 PTFE PV	C OTHER:		Scre	om of Screen to	UEnd Cap	(f	eet) 0.40	
Riser Pipe Below W.T.	SS304 SS316 PTFE P	C OTHER:		Tota	l Length of Cas	sing	(f	eet) 30.15	
Screen	SS304 SS316 PTFE PV	C OTHER:		Scre	en Slot Size **		(incl	ues) 0.010	

APPENDIX A2

GMF RECYCLE POND

F	ELI		BOR	I	NG	G L(DG		C HANSON
CLIENT: AEG Coffeen Power Statio Site: CCB Management Facility						Station acility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G271
Location: Coffeen, Illinois								Drilling Method: 4 ¹ /4" HSA w/SS & CME sam	plers Well ID: G271 Surface Flay: 622.89 ft MSI
DATES: Start: 9/9/2009					9			FIELD STAFF: Driller: G. Mills	Completion: 16.00 ft. BGS
WE	ATHEF	Fin R: Su	i sh: 9/1 inny, wai	0/20 m (7	09 70's)			Helper: J. Twellman Eng/Geo: R. Hasenvager	Station: 874,239.38N 2,515,517.12E
s	SAMPLI	E	Т	EST	FINC	Ĵ	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	1 (in)				(ft ³)	(tsf)	Quadr	angle: Coffeen, IL	$\mathbf{Y} = 12.70$ - While drilling
	. Tota very		<i>6 in</i> ue	e (%	n. (lb.	$_{\rm Type}^{Qp}$	Towns Section	11, Tier 7N; Range 3W	$\underline{\Psi} = 12.50 - 9/21/09$ $\underline{\nabla} =$
Iumbei	ecov / 6 Reco	ype	lows/ I - Val tQD	loistur	Ty De	bu (tsf) ailure	Depth	Lithologic	Borehole Elevation
z	R %	T	B N B	2		Ощ		FILL - Yellowish brown (10YR5/4), moist, firm, silty	
	20/24	\langle	2-5					CLAY with trace sand. Grayish brown (10YR5/2), dry, friable, clayey SILT.	622
1A	83%	A ss	5-6 N=10	27					
							2	Yellowish brown (10YR5/6) with 10% gray (10YR6/	
	24/24	\langle	2-6					trace gravel.	
24	100%	ss	5-5 N=11	23					620
211								Gray (10YR6/1), moist, firm, very silty CLAY with slig	
			15						
2.4	23/24 96%	ss	4-5 N=9	10					618
3A			11 9	18					
								One (10005/1) id 200/ here with all (10006/	
	24/24 100%	ss	2-4 4-5					mottles, moist, firm, silty CLAY with sand and trace gra	⁵⁾ vel. 616
4A		$\langle \rangle$	IN=8	17					
	Ī						8		
5A	24/24	ss	2-4 4-6	20					-614
	10070	\mathbb{A}	N=8					Very dark gray (10YR3/1), organic-rich (PEAT), silt CLAY and trace sand.	
							10	Brownish yellow (10YR6/8) with 20% gray (10YR5/2	
	24/24	ss	2-4 4-5					mottles, moist, firm, silty CLAY with sand and trace gra	vel612
6A	100%	\mathbb{N}	N=8	22					
6B				20			12	mottles, very moist, soft, sandy CLAY with silt and slig	ht
//	20/24	ss	2-2 3-7	20				Brownish yellow (10YR6/6), very moist to wet, soft, sar	ndy - 610
7B	83%	\mathbb{N}	N=5	19				Gray (10YR6/1), wet, loose, very fine to medium SAN	
								with silt.	
	24/24	ss	10-19 30-33					Gray (10YR5/1), slightly moist, hard, very silty CLA	й — 608
8A	100%	$\langle \rangle$	N=49	7				with sand and gravel.	
	I L				I	I	16 =	End of Boring = 16.0 ft. BGS	

NOTE(S):

F	FIELD BORING LOG								
	CLIEN' Sit Locatio Projec DATE	T: Al te: C0 n: C0 ct: 05 S: St Fin	EG Coffe CB Mana offeen, III 5S3004A art: 9/1 iish: 9/1	en P igem inois 0/20 0/20	ower ent F 5 09 09	Station acility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME san FIELD STAFF: Driller: G. Mills Helper: J. Twellman	BOREHOLE ID: G272 well ID: G272 Surface Elev: 620.72 ft. MSL Completion: 14.32 ft. BGS Station: 874,234.83N
WE	ATHE	R: Su	inny, wai	m (7	'0's)			Eng/Geo: R. Hasenyager	2,515,744.99E
	SAMPL	E	T	EST	INC	;	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
ber	v / Total (i) covery		s / 6 in ⁷ alue	ture (%)	Den. (lb/ft ³)	sf) <i>Qp</i> (tsf re Type	Quadr Towns Section	angle: Coffeen, IL hip: East Fork n 11, Tier 7N; Range 3W	$\underline{\Psi}$ = 13.00 - While drilling $\underline{\Psi}$ = 9.49 - 9/21/09 $\underline{\nabla}$ =
Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	20/24 83%	ss	3-2 3-4 N=5	18				FILL - Brownish yellow (10YR6/6) with 30% gray (10YR5/1) mottles, moist, firm, silty CLAY with trace s and slight trace gravel.	sand
2A	24/24 100%	ss	4-5 6-7 N=11	25			2	Brownish yellow (10YR6/6) with 20% brownish yello (10YR6/8) and 20% gray (10YR5/1) mottles, moist, fi silty CLAY with trace sand and slight trace gravel.	w m,
3A	23/24 96%	ss	2-4 4-6 N=8	18					616
3B		$\left(\right)$		17			6		
4A	24/24 100%	ss	3-4 4-4 N=8	20			8	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with sand and trace gra	6) ivel.
5A	23/24 96%	ss	2-3 3-5 N=6	21			⊻ 10	Yellowish brown (10YR5/8) with 15% gray (10YR6/ mottles, moist, soft, silty CLAY with sand and slight tra gravel.	
6A	22/24 92%	ss	2-3 3-3 N=6	23				Gray (10YR6/1) with 30% brownish yellow (10YR6/ mottles, very moist, soft, silty CLAY with sand and slig trace gravel.	6) ght 610
7A	18/24	ss	2-9 15-21	14			12 <u>−</u>	Gray (10YR6/1), very moist, loose, SILT and very fin sand.	$\begin{array}{c c} \mathbf{n} \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ \mathbf{e} \\ $
7B		\wedge	N=24	10			14	Gray (10YR6/1), wet, loose, SILT and very fine sand Yellowish brown (10YR5/6), moist, hard, very silty CL with sand and trace gravel.	AY
								End of Boring = 14.3 ft. BGS	·

FIELD BORING LOG									
CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A				ower ent F	Station Facility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME sam	plers BOREHOLE ID: G273 Well ID: G273 Surface Elev: 620.17 ft. MSL	
DATES: Start: 9/10/2009 Finish: 9/10/2009 WEATHER: Sunny, warm (70's)					09 09 70's)			FIELD STAFF: Driner: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	Completion: 16.00 ft. BGS Station: 874,235.24N 2,515,975.49E
5	SAMPL	AMPLE TESTING TOPOGRAPHIC MAP INFORMATION: WAT							WATER LEVEL INFORMATION:
	al (ir			0	b/ft ³)	e (tsf) e	Quadr Towns	angle: Coffeen, IL hin: East Fork	$\Psi = 13.50 - \text{While drilling}$ $\Psi = 9.89 - 9/21/09$
н	/ Tot		/ 6 in lue	re (%	sn. (ll	$Q_{\rm qy}^{\rm L}$	Section	11, Tier 7N; Range 3W	<u> </u>
Numbe	Recov % Reco	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	16/24 67%	ss	3-3 3-3 N=6	24				FILL -Dark yellowish brown (10YR4/6), moist, firm, sil CLAY with slight trace sand.	lty
2A	24/24 100%	ss	3-4 5-7 N=9	28				FILL - Gray (10YR5/1) with 20% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with sligh trace sand and gravel.	m t
3A 3B	24/24 100%	ss	3-5 6-8 N=11	18 25				Brownish yellow (10YR6/8) with 40% gray (10YR5/1 mottles, moist, firm, silty CLAY with trace sand and slig trace gravel.) ht
4A	24/24		3-5	19			6	Gray (10YR6/1) with 10% brownish yellow(10YR6/6 mottles, moist, firm, silty CLAY with trace sand and slig trace gravel.	614
4B	100%	ss	5-6 N=10	16			8	Gray (10YR5/1), moist, firm, silty CLAY with sand an trace gravel.	d 612
5A	23/24 96%	ss	2-4 5-4 N=9	19				Vellowish brown (10VR5/8) with 30% oray (10VR6/1	
5B				21			⊻ ₁₀	mottles, moist, soft, sandy CLAY with silty and slight tra	nce
6A	24/24 100%	ss	1-2 3-4 N=5	19			12	Brownish yellow (10YR6/8) with 10% gray (10YR6/1 mottles, very moist, soft, sandy CLAY with silt and slig trace gravel.) ht
7A	24/24 100%	ss	4-8 17-24 N=25	11				Gray (10YR6/1), moist, hard, very silty CLAY with sar and trace gravel.	nd 608
7B				11				Light yellowish brown (10YR6/4), wet, loose, very fine- <u>very coarse-grained SAND with trace silt.</u> Light yellowish brown (10YR6/4), wet, dense, sandy, sil	to ty
8A	22/24 92%	ss	9-22 22-23 N=44	8				CLAY. Gray (10YR6/1), moist, hard, very silty CLAY with sar and trace gravel.	nd
	End of Boring = 16.0 ft. BGS								

NOTE(S):
F	EL	DI	BOR	I	NG	L(DG		C HANSON		
	CLIEN Si Locatio Proje	T: A te: C on: C ct: 05	EG Coffe CB Mana offeen, Ill 5S3004A	een P agem linois	'ower ient F s	Statior acility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME san	nplers BOREHOLE ID: G274 Well ID: G274 Surface Elev: 621.67 ft. MSL		
	DATE	S: St Fir	art: 9/1	6/20 6/20	09			FIELD STAFF: Driller: G. Mills Helner: I. Twellman	Completion: 18.06 ft. BGS Station: 874 239 25N		
WE	EATHE	R: Su	inny, wai	rm (8	30's)			Eng/Geo: R. Hasenyager	2,516,195.60E		
	SAMPL	E.	Т	EST	FINC	;	TOPOGE	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:		
	tal (i) y		u	(%)	lb/ft ³)	<i>p</i> (tsf be	Quadı Towns	angle: Coffeen, IL hip: East Fork	$\underline{\Psi} = 16.00$ - While drilling $\underline{\Psi} = 13.12 - 9/21/09$		
ber	v / To cover		s/6i alue	ture ('	Den. ($f_{\rm T} O$	Section	n 11, Tier 7N; Range 3W	$\underline{\nabla}$ =		
Numł	Recov % Re	Type	Blows N - V RQD	Moist	Dry I	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks		
1A	20/24 83%	ss	2-3 3-4 N=6	17					620		
2A	24/24 100%	ss	4-6 7-9 N=13	25				FILL - Gray (10YR5/1) with 30% yellowish brown (10YR5/6) mottles, moist, firm, silty CLAY with trace s and slight trace gravel.	sand		
3A	$3A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} \\ SS \begin{vmatrix} 3-4 \\ 6-9 \\ N=10 \end{vmatrix} \\ 21 \end{vmatrix}$										
3B		H		21			6-		010		
4A	16/24 67%	ss	3-6 6-8 N=12	24			8	Gray (10YR5/1) with 20% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with slight trace sand a gravel.	(6) and 614		
5A	24/24 100%	ss	2-4 4-6 N=8	20				Dark gray (10YR4/1) with 15% yellowish brown (10YR5/8) mottles, moist, soft, silty CLAY with trace s and slight trace gravel.	and 612		
6A	22/24 92%	ss	1-3 4-6 N=7	19			12				
7A	23/24 96%	ss	1-2 4-4 N=6	21			⊻ 14-	Gray (10YR5/1) with 30% yellowish brown (10YR5/ mottles, moist, soft, silty CLAY with sand and trace gra	8) vel. 608		
8A	22/24 92%	ss	1-3 3-6 N=6	17			¥ 16 ±	Yellowish brown (10YR5/8), very moist, soft, silty CL. with sand and trace gravel.	ĀY 606		
0.4	14/24	M	wor-4	12				Brownish yellow (10YR6/6), wet, loose, very fine- to v coarse-grained SAND.	ery		
9A 9B	58%	∬ ^{ss}	9-11 N=13	13				Brownish yellow (10YR6/6), moist, firm, very silty CL with sand and gravel. Gray (10YR6/1), moist, hard, very silty CLAY with sa	AY and 604		
								End of Boring = 18.1 ft. BGS			
NC)TE(S):										

FIELD BORING LOG										
U U U U U U U U U U U U U U U U U U U	CLIEN Sit Location Projec DATE ATHEI	F: A e: C n: C f: 0: f: 0: S: St S: St Fir R: Su	EG Coffe CB Mana offeen, III S3004A cart: 9/1 hish: 9/1 unny, wan	een P agem inois 6/20 6/20 m (8	ower ent F o 09 09 60's)	Station	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME sam FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	BOREHOLE ID: G275 plers Well ID: G275 Surface Elev: 616.14 ft. MSL Completion: 13.19 ft. BGS Station: 874,298.94N 2,516,375.86E	
SAMPLE TESTING							TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:	
umber	scov / Total (i <i>Recovery</i>	/pe	<i>ows / 6 in</i> - Value Q D	oisture (%)	y Den. (lb/ft ³	ı (tsf) <i>Qp</i> (tsf ilure Type	Quad Town Sectio	ship: East Fork n 11, Tier 7N; Range 3W Lithologic	$\Psi = 10.90$ - while drilling $\Psi = 10.25 - 9/21/09$ $\overline{\Psi} =$ Borehole Elevation	
Ñ	Re Re	Ĥ	R N BI	Ž	D	ЪQ	ft. BGS	Description	Detail ft. MSL Remarks	
1A	22/24 92%	ss	2-4 4-6 N=8	22			2	FILL - Gray (10YR5/1) with 30% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY with trace se and slight trace gravel.	und 614	
2A	18/24 75%	ss	3-6 7-7 N=13	19			4	Dark gray (10YR4/1) with 20% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY with sligh trace sand and gravel.	t612	
3A	24/24 100%	ss	2-4 4-5 N=8	22			6	Gray (10YR6/1), moist, soft, silty CLAY with sand an slight trace gravel.	d 610	
4A	24/24 100%	ss	2-3 4-6 N=7	17			8	Gray (10YR6/1) with 30% yellowish brown (10YR5/8 mottles, moist, soft, silty CLAY with sand and slight tra gravel.		
5A	24/24 100%	ss	<i>1-2</i> <i>3-3</i> N=5	18			milini			
5B	-			21			₩ 10-	Yellowish brown (10YR5/8) with 15% gray (10YR6/1 mottles, very moist, soft, silty CLAY with sand and trac		
6A	20/24	V	woh-4	16			* ▼ =	gravel.		
6B	83%	ss	5-2 N=9	13				mottles, very moist, very soft, silty CLAY with sand an trace gravel.	a / The second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second sec	
6C	12/12	V ss	10-35	15			12	Gray (10YR6/1), wet, loose, very fine- to very coarse-grained SAND with trace gravel. Yellowish brown (10YR5/6), moist, hard, very silty CL/		
7A	100%	\wedge	10.55	8				with sand and gravel.		
								ели от вогид = 13.2 tt. вG8		

F	FIELD BORING LOG												
	CLIENT Sit Location Projec	Γ: Α΄ α: Ο΄ n: Ο΄	EG Coffe CB Mana offeen, Ill 553004 A	een P agem linois	ower ent F S	Station acility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME san	nplers	BOF	REHOLE II Well II	D: G276 D: G276 v: 629 14 ft MSL	
WE	DATES	S: S1 Fir R: S1	tart: 9/1 nish: 9/1 nny, mil	6/20 6/20 d (70	09 09)'S)			FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager		Completion: 28.00 ft. BGS Station: 874,438.60N 2,516,358.83E			
	otal (in)	E	T 	EST	(lb/ft ³)	$\frac{2p \text{ (tsf)}}{\text{ype}}$	TOPOGRAPHIC MAP INFORMATION:WATEQuadrangle: Coffeen, ILImage: Coffeen, ILTownship: East ForkImage: Coffeen, IL			ER LEVEL INFORMATION: = Dry - While drilling = 25.55 - 9/21/09			
lumber	lecov / T 6 Recove	ype	<i>lows / 6</i> 1 - Value tQD	Moisture Dry Den. Qu (tsf) Failure T			Depth	Lithologic	⊻ = Bord	ehole	Elevation	Remarks	
14	17/24 71%		5-8 9-10 N=17	10		Or		Description			628	Kellaiks	
2A	19/24 79%	ss	7-7 10-14 N=17	15			2	FILL - Yellowish brown (10YR5/4) with 20% gray (10YR5/1) mottles, moist, hard, silty CLAY with trace s and slight trace gravel.	sand		626		
3A	11/24 46%	ss	5-10 14-27 N=24	14			4				624	Rock fragment in split spoon shoe	
4A	24/24 100%	ss	5-9 10-14 N=19	8			6	FILL - Yellowish brown (10YR5/4) with 20% gray (10YR5/1) mottles, slightly moist, hard, silty CLAY w trace sand and slight trace gravel. FILL - Yellowish brown (10YR5/4) with 10% gray (10YR5/1) mottles, slightly moist hard friable clave			622		
4B 5A	17/24 71%	ss	4-4 8-19 N=12	5 22			8	SILT with sand and trace gravel.			620		
6A	17/24 71%	ss	4-5 8-14 N=13	14			10	FILL - Yellowish brown (10YR5/4) with 25% gray (10YR5/1) mottles, slightly moist, firm, silty CLAY w slight trace sand and gravel.	ith	, C, C, C, C, C, C, C,	618		
7A	16/24 67%	ss	6-7 2-4 N=9	20			14				616		
8A	20/24 83%	ss	2-4 6-6 N=10	21			16	Gray (10YR6/1) with 10% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with slight trace sand a gravel.	8) and		614		
9A 9B	22/24 92%	ss	1-4 5-7 N=9	17 13				Gray (10YR6/1) with 20% yellowish brown (10YR5/ mottles, moist, soft, sandy CLAY with silt and slight tra gravel.			612		
10A	23/24 96%	ss	2-3 8-12 N=11	20				Gray (10YR6/1) with 30% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with sand and slight tr gravel.	 8) ace		610		
NC	DTE(S):			'			20		v / / / 68	× */ / / /	. 1		

F	FIELD BORING LOG											
WI	CLIEN Sit Locatio Projec DATE	Г: А n: С n: С ct: 05 S: St Fin R: Su	EG Coffe CB Mana offeen, Ill 5S3004A cart: 9/1 iish: 9/1 inny, mile	een P agem linois 6/20 6/20 d (70	lower ient F s 09 09 09)'S)	Station Facility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME san FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	EHOLE ID: G2 Well ID: G2 Surface Elev: 62 Completion: 2 Station: 8 2,5	276 276 29.14 ft. MSL 28.00 ft. BGS 574,438.60N 516,358.83E		
	SAMPL	E	Т	EST	TINC	3	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL	INFORMATIO	N:	
er	ber v / Total (in) ecovery reaction (in) reaction			en. (lb/ft ³)	f) Qp (tsf) e Type	Quadr Towns Section	Trangle: Coffeen, IL $\Psi = Dry - While drilling(ship: East Fork\Psi = 25.55 - 9/21/09(on 11, Tier 7N; Range 3W)\Psi = 25.55 - 9/21/09$					
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Detail	Elevation ft. MSL	Remarks	
11A	24/24 100%	ss	1-3 5-7 N=8	21			22	Gray (10YR6/1) with 30% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with sand and slight tr gravel. [Continued from previous page]	8) ace	608		
12A	100%	ss	6-7 N=10	16			24	Yellowish brown (10YR5/8), moist, firm, silty CLAY v sand and slight trace gravel.				
13A 14A	24/24 100% 24/24 100%	ss ss	2-3 4-6 N=7 1-5 15-29	16 16			₹ 26	Gray (10YR6/1) with 25% yellowish brown (10YR5/ mottles, very moist, soft, silty CLAY with sand and tragravel. Gray (10YR6/1), very moist, loose, very fine- to fine-grained, SAND Gray (10YR6/1) with 25% yellowish brown (10YR5/ mottles, very moist, soft, silty CLAY with sand and tragravel. Gray (10YR6/1), very moist, firm, clayey SILT with tr wrut firm, provided and	6) ce 6) ce ace ce ce ce ce ce ce ce ce ce	604		
		/\	N≕20				28	Gray (10YR6/1) with 40% yellowish brown (10YR5/ mottles, moist, hard, very silty CLAY with sand and tra gravel. Yellowish brown (10YR5/4), moist, hard, very silty CL with sand and trace gravel. End of Boring = 28.0 ft. BGS	4) / / / / / / / / / / / / / / / / / / /			

FI			BOR	I	NG		DG	CONTRACTOR	HANSON
	ULIEN Sit Locatio Proied	1: A e: C n: C n: C	EG Coffe CB Mana offeen, Ill 5S3004A	en P agem linois	ower ent F	Station	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME sar	mplers Well ID: G277 Surface Elev: 620.79 ft. MSL
WE	DATE	S: St Fin R: St	art: 9/1 iish: 9/1 inny, mil	4/20 4/20 d (70	09 09)'S)			FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	Completion: 20.00 ft. BGS Station: 874,581.80N 2,516,370.51E
S	SAMPL	E	Т	EST		i O	TOPOGRA	PHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	y y		и	(%)	lb/ft ³	<i>p</i> (tsf pe	Quadrar Townshi	igle: Coffeen, IL p: East Fork	$\Psi = 16.40$ - While drilling $\Psi = 18.23 - 9/21/09$
er	/ / To		s / 6 i alue	ure ('	en. ($O_{\rm Typ}^{\rm cl}$	Section 1	1, Tier 7N; Range 3W	<u> </u>
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	23/24 96%	ss	2-2 4-6 N=6	21				FILL - Yellowish brown (10YR5/4) with 10% yellow brown (10YR5/8) mottles, very moist, soft, silty CLA with slight trace sand and gravel.	ish YY
	24/24	N SS	2-4				2	FILL - Yellowish brown (10YR5/4) with 20% yellow brown (10YR5/8) mottles, moist, firm, silty CLAY w slight trace sand and gravel.	ish ith — 618
2A 2B	100%		N=10	22 16			4	Yellowish brown (10YR5/4) with 20% gray (10YR5, and 10% yellowish brown (10YR5/8) mottles, moist, fi silty CLAY with slight trace sand. Light brownish gray (10YR6/2), dry. friable, clayey SI	
3A	17/24	\mathbb{V}_{ss}	4-7					. <u></u>	
3B	71%		N=16	22			6	Light brownish gray (10YR6/2) with 25% yellowish br (10YR5/8) mottles, moist, firm, clayey SILT with slig trace sand.	own ght
4A	18/24	W _{ss}	4-8 8-6	13					614
	75%		N=16				8	Gray (10YR6/1) with 50% very dark grayish brown (10YR3/2) mottles, moist, hard, clayey SILT with slig trace sand and gravel.	
5A	19/24 79%	ss	3-7 8-9 N=15	12			10	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, hard, very silty CLAY with sand and sli trace gravel.	/6) ight 612
6A	22/24 92%	ss	3-9 9-10 N=18	14					610
6B				16			12		
7A	18/24 75%	ss	3-5 7-9 N=12	17				Yellowish brown (10YR5/8) with 30% gray (10YR6, mottles, moist, hard, silty CLAY with sand and trace gra	/1) avel.
	22/24 92%	ss	2-5 5-8				14		606
8A		\wedge	N=10	12			16	Gray (10YR6/1), moist, slightly dense, silty, very fine- very coarse-grained SAND.	- to
9A	22/24 92%	ss	<i>1-2</i> <i>3-3</i> N=5	14			₽. Introduction	Gray (10YR6/1) with 30% yellowish brown (10YR5, mottles, moist, firm, silty CLAY with sand and trace gra Gray (10YR6/1) with 20% brownish yellow (10YR6, mottles, wet, very soft, silty, very fine- to fine-graine SAND with trace gravel.	/8) avel. d
10A				11			⊉ ¹⁸ –	Gray (10YR6/1) with 30% grayish brown (10YR5/2 mottles, moist, soft, sandy CLAY with slight trace gray	2) vel.
10B	23/24 96%	ss	1-3 19-47 N=22	9				Brownish yellow (10YR6/6) with 25% gray (10YR6/ mottles, moist, hard, very silty CLAY with sand and tr gravel.	(1) ace 602
10	I .		I	I	I	I	20	End of Boring = 20.0 ft. BGS	
NO)TE(S):								

F	TELD BORING LOG CONTRACTOR: Layne-Western Co CLIENT: AEG Coffeen Power Station CONTRACTOR: Layne-Western Co												
	CLIENT Sit Location Project	F: Al e: C n: C n: C	EG Coffe CB Mana offeen, Ill	een P agem linois	ower ent F	Station acility	1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME sat	mplers BOREHOLE ID: G278 Well ID: G278 Surface Flay: 628 85 ft MSI				
WE	DATES	S: St Fin R: St	art: 9/1 iish: 9/1 inny, wai	1/20 1/20 rm (7	09 09 70's)			FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	Completion: 24.06 ft. BGS Station: 874,875.37N 2,516,200.66E				
r 7 Total (in) 7 for 7 for 7 for 7 for 7 for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for 10^{10} for						$\frac{Qp \text{ (tsf)}}{\text{Type}}$	TOPOGR Quadra Townsl Section	APHIC MAP INFORMATION: angle: Coffeen, IL nip: East Fork 11, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = Dry$ - While drilling $\Psi = 23.98 - 9/21/09$ $\overline{\nabla} =$				
Number	Recov / % Reco	Type	Blows / N - Vali RQD	Moistur	Dry De	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
1A	16/24 67%	ss	3-4 6-7 N=10	19			2		628				
2A	24/24 100%	ss	4-7 8-11 N=15	21			4		626				
3A	22/24 92%	ss	9-10 9-35 N=19	10			6		624 Rock fragment in split spoon shoe				
4A	4/24 17%	ss	20-7 10-8 N=17				8-	FILL - Yellowish brown (10YR5/4) with 15% yellow brown(10YR5/8) mottles, moist, hard, silty CLAY w trace sand and slight trace gravel.	rish ith - 622 Rock fragment in split spoon shoe				
5A	14/24 58%	ss	11-6 8-8 N=14	15			10		620 Rock fragment in split spoon shoe				
6A	20/24 83%	ss	6-4 8-9 N=12	26			12		618				
7A	24/24 100%	ss	2-4 8-11 N=12	18			14	Gray (10YR6/1) with 30% brownish yellow (10YR6 	/6) nd				
8A	20/24 83%	ss	4-7 10-11 N=17	12				Very dark gray (10YR3/1), moist, firm, clayey SILT v slight trace sand and trace roots.	with614				
8B			11-17	22			16-	Gray (10YR5/1) with 30% yellowish brown (10YR5, mottles, moist, firm, silty CLAY with sand and trace gradering and trace gradering of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure of the sand structure o	/8) avel.				
9A	22/24 92%	ss	4-6 6-9 N=12	17				Gray (10YR6/1) with 10% yellowish brown (10YR5, mottles, moist, firm, silty CLAY with sand and trace gra	/8) avel.				
10A	20/24 83%	ss S	2-4 5-8 N=9	21			20-	Yellowish brown (10YR5/8) with 20% gray (10YR5/ mottles, moist, firm, silty CLAY with sand and trace gra					
	JIE(S):												

FI	FIELD BORING LOG CRE HANSON										
	CLIEN'	T: A	EG Coffe	en P	ower	Station	ı	CONTRACTOR: Layne-Western Co			
	Sit	te: C	CB Mana	agem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G278		
1	Locatio	n: Co	offeen, Il	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME sar	mplers Well ID: G278		
	Proje	ct: 05	5S3004A						Surface Elev: 628.85 ft. MSL		
	DATE	S: St	tart: 9/1	1/20	09			FIELD STAFF: Driller: G. Mills Completion: 24.06 ft			
		Fir	nish: 9/1	1/20	09			Helper:J. TwellmanStation:874,875.37N			
WE	ATHE	R: Si	unny, wa	rm (7	'0's)			Eng/Geo: R. Hasenyager 2,516,200.66E			
S	SAMPL	Æ	Г	TEST	TIN	3	TOPOGR	APHIC MAP INFORMATION.	WATER LEVEL INFORMATION.		
5r	/ Total (in) <i>wery</i>		/ 6 in lue	re (%)	en. (lb/ft ³)	Qu (tsf) <i>Qp</i> (tsf) Failure Type	Quadra Townsh Section	ingle: Coffeen, IL iip: East Fork 11, Tier 7N; Range 3W	$\Psi = Dry - While drilling$ $\Psi = 23.98 - 9/21/09$ $\overline{\Psi} =$		
Numbe	Recov % Reco	Type	Blows N - Va RQD	Moistu	Dry De		Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks		
11A	18/24 75%	ss	2-4 7-7	19				Yellowish brown (10YR5/8) with 20% gray (10YR5. mottles, moist, firm, silty CLAY with sand and trace gra [Continued from previous page]	(1) avel. 608		
11B							22	Yellowish brown (10YR5/6), very moist, soft, silty, very fine- to medium-grained SAND.			
$ \begin{array}{c c} 12A \\ 12B \\ 12B \\ \end{array} \begin{array}{c} 20/24 \\ 83\% \\ 83\% \\ \end{array} \begin{array}{c} I \\ ss \\ ss \\ N=15 \\ 10 \\ 10 \\ 10 \\ 10 \\ 10 \\ 10 \\ 10 \\ $							₽ 24	Yellowish brown (10YR5/4), moist, hard, very silty CL with sand and trace gravel.	AY 606		
	End of Boring = 24.1 ft. BGS										

F'I WE	CLIENT Site Location Projec DATES	D F : A e : Co n : Co i t: 05 s : St F i t s : St r r r	EG Coffe CB Mana offeen, III 5S3004A tart: 9/1 nish: 9/1 unny, wan	cen P agem linois 0/20 0/20 rm (8	ower ent F s 09 09 80's)	Station Sacility	JG	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS & CME sau FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	mplers BOREHOLE ID: G279 Well ID: G279 Surface Elev: 629.19 ft. MSL Completion: 28.00 ft. BGS Station: 875,028.06N 2,516,245.60E				
er	/ Total (in)		1/6 in alue	ure (%)	en. (lb/ft ³)	f_{e} Type	TOPOGR Quadra Towns Section	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork 11, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 23.60$ - While drilling $\Psi = 24.68 - 9/21/09$ $\overline{\Psi} =$				
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
1A	24/24 100%	ss	3-3 5-6 N=8	18			2	FILL - Brown (10YR4/3) with 30% yellowish brow (10YR5/6) mottles, moist, firm, silty CLAY with sand trace gravel.	n and				
2A	24/24 100%	ss	5-9 10-11 N=19	14			4		626				
3A	24/24 100%	ss	5-9 9-10 N=18	17			6		624				
4A	24/24 100%	ss	4-5 7-6 N=12	21					622				
5A	24/24 100%	ss	3-3 5-7 N=8	19			8 10	FILL - dark gray (10YR4/1) with 10% brownish yell (10YR6/6) mottles, moist, hard, silty CLAY with sand trace gravel.	ow and 620				
6A	24/24 100%	ss	3-4 6-9 N=10	17			12						
7A	23/24 96%	ss	2-5 5-6 N=10	23			14						
84	24/24 100%	ss	2-3 7-6 N=10	23				Brownish yellow (10YR6/8) with 30% gray (10YR5 mottles, moist, firm, silty CLAY with slight trace sand gravel.	/1) and 614				
9A	18/24 75%	ss	4-7 8-9 N=15	25			10	Yellowish brown (10YR5/8) with 20% gray (10YR6 mottles, moist, firm, silty CLAY with slight trace sand gravel.	/1) and 612				
10A	24/24 100%	ss	3-6 7-10 N=13	17			20	Gray (10YR6/1) with 25% yellowish brown (10YR5 mottles, moist, firm, silty CLAY with sand and trace gr	 5/8) ravel.				
NC)TE(S):						20		Page 1 of 2				

F	FIELD BORING LOG CREATER HANSON															
	CLIEN	Г: А	EG Coffe	en P	ower	Station	n	CONTRACTOR: Layne-Western Co								
	Sit	e: C	CB Mana	gem	ent F	acility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G279							
	Locatio	n: C	offeen, Ill	inois	5			Drilling Method: 4¼" HSA w/SS & CME samp	lers Well ID: G279							
	Projec	et: 05	5S3004A						Surface Elev: 629.19 ft. MSL							
	DATE	S: St	art: 9/10	0/20	09			FIELD STAFF: Driller: G. Mills	Completion: 28.00 ft. BGS							
		Fir	nish: 9/1	0/20	09			Helper: J. Twellman Station: 875,028.								
WE	WEATHER: Sunny, warm (80's)							Eng/Geo: R. Hasenyager	2,516,245.60E							
SAMPLE TESTING							TOPOGI	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:							
(in) (in) (in) (in) (in) (in) (in) (in)					£	sf)	Quad	angle: Coffeen, IL	$\mathbf{\Psi} = 23.60$ - While drilling							
er / Total <i>overy</i> / 6 in lue			(%	lb/f	$pe^{(1)}$	Town	hip: East Fork	$\Psi = 24.68 - 9/21/09$								
			6 i lue	Le (u.	$ _{\mathcal{O}_{\mathcal{V}}^{\Gamma}}$	Sectio	n 11, Tier 7N; Range 3W	$\underline{\nabla}$ =							
mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber mber				v De	(tsf lure	Depth	Lithologic	Borehole Elevation								
Nu	%	$\mathbf{T}_{\mathbf{y}}$	R. N.	Ĭ	۲.	Fai Fai	ft. BGS	Description	Detail ft. MSL Remarks							
11A	23/24 96%	ss	2-4 5-7 N=9	18				Gray (10YR6/1) with 25% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY with sand and trace grave [Continued from previous page]	el. 608							
12A	19/24 79%	ss	4-9 8-9 N=17	13			 	Yellowish brown (10YR5/8), moist, firm, clayey SILT ar very fine-grained SAND with slight trace gravel.	d 606							
12B				12			24									
13A 14A	17/24 71%	ss	<i>1-5</i> <i>5-7</i> N=10	18			⊻ 26-	Light brownish gray (10YR6/2), wet, loose, very fine- to coarse-grained SAND.	, 							
	24/24	V ss	10-10					Brownish yellow (10YR6/6), moist, hard, very silty CLA	Y							
14B	14B		ss 18-18 N=28	14	14			Gray (10YR6/1), moist, hard, very silty CLAY with sand and trace gravel.	n - 602							
					•	End of Boring = 28.0 ft. BGS										

Illinois Environ	mental Protection Agency			Well	Completion	Report
Site #:	County:	ntgomery		W	/ell #:G2	271
Site Name: AEG Coffeen Pow	ver Station CCB Management Facility			В	orehole #:	G271
State- Plant Plane Coordinate: X_2,515,517	<u>.1</u> Y <u>874,239.4</u> (or) Latitude:	0		Longitud	e:°	
Surveyed By:Jeffrey D. Emric	:k	IL Regist	ration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-Wes</u>	stern Co	Driller: _	G. Mills			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.	Geologist	: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	0246
Drilling Method: <u>Hollow stem</u>	auger	Drilling F	Fluid (Type):			
Logged By: <u>Rhonald W. Hase</u>	nyager	Date Star	ted: <u>9/9/200</u>	<u>)9</u> Dat	e Finished: <u>9/</u>	10/2009
Report Form Completed By:	zanna L. Simpson	Date:	10/7/2009			
ANNULAR SPAC	CE DETAILS		Elevations	Depths	(0.01 ft.))
			(MSL)*	(BGS)		a .
			625.88	-2.99	Top of Protective	Casing
			625.57	-2.68	Top of Riser Pipe	e
Type of Surface Seal: <u>Concrete</u>			622.89	0.00	Ground Surface	
True of Annular Socients - High a	olida hantanita		619.89	3.00	Top of Annular S	Sealant
Installation Method: Tramia						
Setting Time: >24 hr			610 39	12 50	Static Water Leve	4
5etting Time. <u>- 24 m.</u>		¥-			(After Completion)	9/21/2009
Type of Bentonite Seal Granu	Ilar Pellet Slurry					
Installation Method: <u>Gravity</u>	· · · · · · · · · · · · · · · · · · ·	x x	616.16	6.73	Top of Seal	
Setting Time: <u>10 min</u>	X	X	613.87	9.02	Top of Sand Pacl	K
Type of Sand Pack Operator and						
Grain Size: 10/20 (siet	ve size)		612.93	9.96	Top of Screen	
Installation Method: Gravity						
			608.58	14.31	Bottom of Screen	L
Type of Backfill Material: <u>Quart</u>	z sand (if applicable)		608.10	14.79	Bottom of Well	
Installation Method: <u>Gravity</u>			606.89 * Referenced to a	16.00	Bottom of Boreh	ole
			Kelefenced to a	National Geoder	le Datum	
		Г	CAS	SING MEA	SUREMENTS	
WELL CONS	TRUCTION MATERIALS	I	Diameter of Boreho	ole	(inches)	8.0
(Choose one	e type of material for each area)		D of Riser Pipe	anath	(inches)	2.0
		T	Siser Pine Length	Jength	(feet)	12.64
Protective Casing	SS304 SS316 PTFE PVC OTHER:	Steel	Bottom of Screen to	o End Cap	(feet)	0.48
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER:	5	Screen Length (1	st slot to last slo	t) (feet)	4.35
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER:] [7	Fotal Length of Ca	sing	(faat)	17 47

SS304

SS316

PTFE PVC OTHER:

Screen

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

(inches)

Illinois Environmental Protection Agency		Well	Completion	Report
Site #: County:	tgomery	W	Tell #: G2	272
Site Name: AEG Coffeen Power Station CCB Management Facility		В	orehole #:	G272
State- Plant Plane Coordinate: X_2,515,745.0 Y_874,234.8 (or) Latitude:	o	Longitude	e:°	
Surveyed By:Jeffrey D. Emrick	IL Registration #:	35-003507		
Drilling Contractor: Layne-Western Co	Driller: <u>G. Mills</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonalc</u>	l W. Hasenyager	r, LPG #196-000	0246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):			
Logged By: <u>Rhonald W. Hasenyager</u>	Date Started:9/1	<u>0/2009</u> Date	e Finished:9/2	10/2009
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date: 10/7/20	09		
ANNULAR SPACE DETAILS	Elevatio (MSL)	ons Depths * (BGS)	(0.01 ft.))
	<u></u>	-3.39	Top of Protective	Casing
	623.81	3.09	Top of Riser Pipe	2
Type of Surface Seal: Concrete	620 72	2 0.00	Ground Surface	
	617.72	2 3.00	Top of Annular S	ealant
Type of Annular Sealant: <u>High-solids bentonite</u>		<u> </u>	Top of Annual C	culuitt
Installation Method:				
Setting Time: $_>24 \text{ hr.}$	611.23	9.49	Static Water Leve (After Completion)	el 9/21/2009
Type of Bentonite Seal Granular Pellet Slurry				
(choose one)	614 55	6 17	Top of Seal	
Setting Time: 10 min				
	612.74	<u> </u>	Top of Sand Pack	ζ.
Type of Sand Pack:Quartz sand	611.61	0.11	T	
Grain Size: <u>10/20</u> (sieve size)		9.11	Top of Screen	
Installation Method:	606 74	13 98	Bottom of Screen	
Type of Backfill Material:	606.40	14.32	Bottom of Well	
(it applicable)	606.40) 14.32	Bottom of Boreh	ale
	* Reference	ed to a National Geodetic	c Datum	
		CASING MEAS	SUREMENTS	
	Diameter of B	orehole	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pi	pe	(inches)	2.0
	Protective Cas	ing Length	(feet)	5.0
Protective Casing SS304 SS316 DTEE DVC OTHER	Steel Riser Pipe Ler	ngth	(feet)	12.20
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Scr	een to End Cap	(feet)	0.34
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length	of Casing	(feet)	+.0/ 17.41
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Si	ze **	(inches)	0.010

**Hand-Slotted Well Screens Are Unacceptable

Illinois Environ	nmental Protection Agency			Well (Completion	Report
Site #:	County: <u>Mo</u>	ntgomery		We	11 #: <u> </u>	273
Site Name: <u>AEG Coffeen Pov</u>	ver Station CCB Management Facility			Bor	ehole #:	G273
State- Plant Plane Coordinate: X_2,515,975	5.5 Y <u>874,235.2</u> (or) Latitude:			Longitude:	°	
Surveyed By:Jeffrey D. Emri	ck	IL Regis	tration #: <u>035-0035</u>	507		
Drilling Contractor: <u>Layne-We</u>	stern Co	Driller:	G. Mills			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologis	t: <u>Rhonald W. Ha</u>	senyager,	LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling 1	Fluid (Type):			
Logged By: Rhonald W. Hase	nvager	Date Sta	rted: 9/10/2009	Date 1	Finished: 9/	0/2009
Report Form Completed By: Su	zanna I. Simnson	Date:	10/7/2009			
ANNULAR SPA	CE DETAILS		Elevations D	 Depths	(0.01 ft.)	
			(MSL)* ((BGS)	, ,	
			623.33	-3.16	Top of Protective	Casing
			623.02	-2.85	Top of Riser Pipe	•
Type of Surface Seal: <u>Concrete</u>			620.17	0.00	Ground Surface	
Type of Annular Sealant: <u>High-s</u>	olids bentonite		617.17	3.00	Top of Annular S	ealant
Installation Method:	;					
Setting Time:24 hr.		\mathbf{Z}	610.28	9.89	Static Water Leve	
					(After Completion)	9/21/2009
Type of Bentonite Seal Gran	ular Pellet Slurry (choose one)	Ϋ́Γ				
Installation Method: <u>Gravity</u>	/	×	614.07	6.10	Top of Seal	
Setting Time: <u>10 min</u>	X		612.45	7.72	Top of Sand Pacl	C
Type of Sand Pack: <u>Quartz sand</u>						
Grain Size: 10/20 (sie	ve size)		611.09	9.08	Top of Screen	
Installation Method:Gravity	/					
Type of Backfill Material Ouar	z sand		<u>605.61</u> <u>605.07</u>	<u>14.56</u>	Bottom of Screen Bottom of Well	
	(if applicable)			<u>10.10</u>	Doublin of wen	
Installation Method: <u>Gravity</u>	1		604.17 * Referenced to a Nation	16.00	Bottom of Boreho	ole
			CASIN	G MEASU	UREMENTS	
			Diameter of Borehole		(inches)	8.0
WELL CONS (Choose on	I KUCTION MATERIALS e type of material for each area)		ID of Riser Pipe		(inches)	2.0
		-	Protective Casing Leng	gth	(feet)	5.0
Protective Casing	SS304 SS316 PTFE PVC OTHER:	<u>Steel</u>	Kiser Pipe Length	nd Can	(feet)	0.54
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER:		Screen Length (1st slo	ot to last slot)	(feet)	5.48
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER:		Total Length of Casing	2	(feet)	17.95

SS304

SS316

PTFE

PVC OTHER:

Screen

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

(inches)

Illinois Environmental Protection Agency	Well Completion Repor
Site #: County:	tgomery Well #: G274
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #: G274
State- Plant Plane Coordinate: X_2,516,195.6 Y_874,239.2 (or) Latitude:	°' Longitude:°'
Surveyed By:Jeffrey D. Emrick	IL Registration #: 035-003507
Drilling Contractor: Layne-Western Co	Driller: <u>G. Mills</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W. Hasenyager, LPG #196-000246</u>
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):
Logged By: <u>Rhonald W. Hasenyager</u>	Date Started:9/16/2009 Date Finished:9/16/2009
Report Form Completed By: Suzanna L. Simpson	Date:10/7/2009
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	624.32 -2.65 Top of Protective Casing
	624.04 Top of Riser Pipe
Type of Surface Seal: Concrete	621.67 0.00 Ground Surface
	618.67 3.00 Top of Annular Sealant
Type of Annular Sealant: <u>High-solids bentonite</u>	
Installation Method:	
Setting Time: ≥ 24 hr.	Z 608.55 13.12 Static Water Level (After Completion) 9/21/2009
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	<u>611.93</u> <u>9.74</u> Top of Seal
Setting Time: <u>15 min</u>	
Type of Sand Pack:	
Grain Size: <u>10/20</u> (sieve size)	$= \frac{608.77}{12.90}$ Top of Screen
Installation Method:	
Type of Backfill Material: n/a	$\boxed{\begin{array}{c} \underline{604.00} \\ 603.61 \end{array}} \begin{array}{c} \underline{17.67} \\ 18.06 \end{array}$ Bottom of Screen
(if applicable)	
Installation Method: <u>n/a</u>	<u>603.61</u> <u>18.06</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	Diameter of Borehole (inches) 8 0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
,	Protective Casing Length (feet) 5.0
Destastiva Casing SS204 SS216 DEED DVG STUD	Riser Pipe Length (feet) 15.27
Florective casing SS304 SS316 PTFE PVC OTHER: Riser Pine Above W T SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.39
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (fact) 20.43
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

**Hand-Slotted Well Screens Are Unacceptable

Illinois Environ	mental Protection Agen	icy		Well	Completion	Report
Site #:	County:!	Montgomery		W	Vell #: G2	275
Site Name: AEG Coffeen Powe	er Station CCB Management Faci	lity		В	orehole #: 0	G275
State- Plant Plane Coordinate: X 2,516,375.9	9 Y <u>874,298.9</u> (or) Latitu	ıde:°	''	Longitude	e:°	' <u> </u>
Surveyed By: <u>Jeffrey D. Emrick</u>	ζ	IL Regis	stration #:035-00	03507		
Drilling Contractor: <u>Layne-West</u>	tern Co	Driller:	G. Mills			
Consulting Firm: <u>Hanson Profes</u>	ssional Services Inc.	Geologi	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem a</u>	uger	Drilling	Fluid (Type):			
Logged By: <u>Rhonald W. Hasen</u>	yager	Date Sta	arted: <u>9/16/20</u>	09 Date	e Finished: <u>9/1</u>	6/2009
Report Form Completed By:	anna L. Simpson	Date:	10/7/2009			
ANNULAR SPAC	E DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			618.53	-2.39	Top of Protective	Casing
			_618.26	-2.12	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>			616.14	0.00	Ground Surface	
True of Annular Scalant, High gal	lida harrtanita		613.14	3.00	Top of Annular S	ealant
Lustellation Methods Transis						
Sotting Times 24 hr			605 80	10.25	Statio Water Leve	1
Setting Time. <u>~24 m.</u>		\ <u>⊻</u>			(After Completion)	9/21/2009
Type of Bentonite Seal Granula	ar Pellet Slurry (choose one)					
Installation Method: <u>Gravity</u>	×		610.42	5.72	Top of Seal	
Setting Time: <u>15 min</u>	Z		609.16 <u>6.98</u> Top of Sand Pack			
Type of Sand Pack: Quartz sand						
Grain Size: <u>10/20</u> (sieve	e size)		607.92	8.22	Top of Screen	
Installation Method: <u>Gravity</u>						
Type of Backfill Material n/a			<u>603.52</u> 602.95	12.62	Bottom of Screen Bottom of Well	
Type of Backfill Material. <u>Iva</u>	(if applicable)				Bottom of wen	
Installation Method: <u>n/a</u>			602.95 * Referenced to a 1	13.19 National Geodeti	Bottom of Boreho c Datum	ble
		Γ	CAS	ING MEAS	SUREMENTS	8.0
WELL CONST	RUCTION MATERIALS	-	ID of Riser Pine	ne	(inches)	2.0
(Choose one t	.jpo or material for each areas		Protective Casing L	ength	(feet)	5.0
			Riser Pipe Length		(feet)	10.34
Protective Casing	SS304 SS316 PTFE PVC OTH	ER: Steel	Bottom of Screen to	End Cap	(feet)	0.52
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTH	ER:	Screen Length (1s	t slot to last slo	t) (feet)	4.45
Screen	SS304 SS316 PTFE (PVC) OTH	ER:	Screen Slot Size **	sing	(inches)	0.010

Well Completion Form (r	revised 02/06/02)
-------------------------	-------------------

Screen

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Enviro	nmental Protection Agenc	y		Well	Completion	Report
Site #:	County: <u>M</u>	ontgomery		W	/ell #: G2	276
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	y		В	orehole #:	G276
State- Plant Plane Coordinate: X_2,516,355	8.8 Y 874,438.6 (or) Latitude	e:°	<u>'</u> "	Longitud	e:°	
Surveyed By:Jeffrey D. Emri	ck	IL Registr	ration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	G. Mills			
Consulting Firm: <u>Hanson Prof</u>	fessional Services Inc.	Geologist:	Rhonald W.	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	_ Drilling F	luid (Type):			
Logged By: Rhonald W. Hase	envager	Date Start	ted: 9/16/20	09 Dat	e Finished: 9/1	6/2009
Report Form Completed By: Su	izanna L. Simpson	- Date:	10/7/2009			
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)	
			(MSL)*	(BGS)		G .
	T		632.40	-3.26	Top of Protective	Casing
	Į		632.00	-2.86	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>			629.14	0.00	Ground Surface	
			626.14	3.00	Top of Annular S	ealant
Type of Annular Sealant: <u>High-</u>	solids bentonite	T P			-	
Installation Method:	2		(02.50	05.55		
Setting Time: <u>>24 hr.</u>		¥	603.59	25.55	(After Completion)	el 9/21/2009
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method: Gravit	(choose one)		610.06	19.08	Top of Seal	
Setting Time: 15 min		\square				
0	×		_608.11_		Top of Sand Pack	
Type of Sand Pack:Quartz sand	<u>i</u>		(0(7)	22.41	T. 60	
Grain Size: <u>10/20</u> (sig	eve size)		000.73		Top of Screen	
Installation Method:Gravit	y		601 92	27 22	Bottom of Screen	
Type of Backfill Material:Quar	tz sand		601.49	27.65	Bottom of Well	
Installation Method: Gravit	(if applicable)		601 14	28.00	Bottom of Boreho	ole
			* Referenced to a	National Geodeti	c Datum	
			CAS	SING MEA	SUREMENTS	
WELL CONS	TRUCTION MATERIALS		Diameter of Boreho	ole	(inches)	8.0
WELL CONS (Choose or	ne type of material for each area)		D of Riser Pipe	an at 1-	(inches)	2.0
			Totective Casing L	Length	(feet)	25.0
Protective Casing	SS304 SS316 PTFE PVC OTHER	: Steel F	Bottom of Screen to	o End Can	(feet)	0.43
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	: S	creen Length (1	st slot to last slo	t) (feet)	4.81
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	: T	otal Length of Ca	sing	(feet)	30.51

SS304

SS316

PTFE

PVC OTHER:

Screen

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Environ	nmental Protection Age	ncy		Well	Completion	Report
Site #:	County:	Montgomery	r	W	/ell #: G2	277
Site Name: <u>AEG Coffeen Pov</u>	ver Station CCB Management Fac	cility		В	orehole #:	G277
State- Plant Plane Coordinate: X_2,516,370).5 Y 874,581.8 (or) Lati	tude:°	""	Longitud	e:°	""
Surveyed By:Jeffrey D. Emri	ck	IL Regi	stration #: <u>035-00</u>	03507		
Drilling Contractor: <u>Layne-We</u>	stern Co	Driller:	G. Mills			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologi	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	Fluid (Type):			
Logged By: Rhonald W. Hase	envager	Date St	arted: 9/14/20	09 Date	e Finished: 9/2	14/2009
Report Form Completed By: Su	zanna L. Simpson	Date:	10/7/2009			
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)	,
			(MSL)*	(BGS)	Top of Protoctive	Casing
	C		025.55	-2.30	Top of Protective	Casing
			623.08	-2.29	Top of Riser Pipe	•
Type of Surface Seal: <u>Concrete</u>			620.79	0.00	Ground Surface	
			617.79	3.00	Top of Annular S	ealant
Type of Annular Sealant: <u>High-s</u>	solids bentonite	T IP			1	
Installation Method:	;					
Setting Time:24 hr.			602.56	18.23	Static Water Leve (After Completion)	el 9/21/2009
Type of Bentonite Seal Gran	ular Pellet Slurry —					
Installation Mathed: Gravity	(choose one)		608.00	12 70	Top of Seel	
Sotting Time: 18 min	Y		_008.00_	12.79	Top of Sear	
Setting Time. <u>18 min</u>			607.00	13.79	Top of Sand Pack	ζ.
Type of Sand Pack:Quartz sand	l					
Grain Size: <u>10/20</u> (sid	we size)		606.50	14.29	Top of Screen	
Installation Method:Gravity	/					
Type of Backfill Material: Ouar	tz sand		<u>602.02</u> 601.55	<u>18.77</u> 19.24	Bottom of Screen Bottom of Well	l
	(if applicable)					
Installation Method:	/		600.79 * Referenced to a 1	20.00 National Geodeti	Bottom of Boreho c Datum	ble
			~ . ~			
			CAS	ING MEAS	SUREMENTS	80
WELL CONS	TRUCTION MATERIALS		ID of Riser Pipe		(inches)	2.0
	vi		Protective Casing L	ength	(feet)	5.0
		1	Riser Pipe Length		(feet)	16.58
Protective Casing	SS304 SS316 PTFE PVC OT	HER: <u>Steel</u>	Bottom of Screen to	End Cap	(feet)	0.47
Riser Pipe Relow W T	SS304 SS316 PTFF PVC OT	HER:	Screen Length (1s	st slot to last slo	t) (feet)	4.48
riber i pe below w.i.			1 otal Length of Cas	sing	(feet)	21.33

SS304

SS316

PTFE

PVC OTHER:

Screen

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:Mo	ntgomery Well #:G278
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #: G278
State- Plant Plane Coordinate: X_2,516,200.7 Y_874,875.4 (or) Latitude:	°' Longitude:°'
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: Layne-Western Co	Driller: <u>G. Mills</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type):
Logged By: Rhonald W. Hasenyager	Date Started:
Report Form Completed By:Suzanna L. Simpson	Date:10/7/2009
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>631.49</u> <u>2.64</u> Top of Protective Casing
	<u>631.17</u> <u>-2.32</u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>628.85</u> <u>0.00</u> Ground Surface
	625.853.00 Top of Annular Sealant
Type of Annular Sealant: <u>High-solids bentonite</u>	
Installation Method: <u>Tremie</u>	
Setting Time: _>24 hr.	
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	$\underline{-613.74} \underline{15.11} \text{Top of Seal}$
Setting Time: 22 min	611.90 <u>16.95</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	<u>609.92</u> <u>18.93</u> Top of Screen
Installation Method:	
Type of Backfill Material: n/a	$= \frac{605.15}{604.79} \frac{23.70}{24.06} \text{Bottom of Screen}$
(if applicable)	
Installation Method:	604.79 24.06 Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUDEMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Destastiva Casing 20104 0014 DEED DVG STORE	Riser Pipe Length (feet) 21.25
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.36
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 26.38
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inchec) 0.010

Well Comple	etion Form	(revised	02/06/02)
-------------	------------	----------	-----------

Screen

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Enviror	nmental Protectio	n Agency				Well	Completio	n Report
Site #:	(County: <u>Mon</u>	tgomery			W	/ell #:G	279
Site Name: <u>AEG Coffeen Pov</u>	ver Station CCB Manage	ment Facility				В	orehole #:	G279
State- Plant Plane Coordinate: X_2,516,245	5.6 Y <u>875,028.1</u> (or) Latitude:		'		Longitud	e:°	
Surveyed By:Jeffrey D. Emrid	ck		IL Regi	stration #:	035-0	03507		
Drilling Contractor: <u>Layne-We</u>	stern Co		Driller:	G. Mill	S			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.		Geologi	st: <u>Rho</u> i	nald W.	Hasenyage	r, LPG #196-00	0246
Drilling Method: Hollow stem	auger		Drilling	Fluid (Typ	e):			
Logged By: Rhonald W Hase	envager		Date St	arted.	9/10/20)09 Dat	e Finished [.] 9	/10/2009
Report Form Completed By: Su	zanna I. Simpson		Date:	10/7	//2009	<u></u> Dut		10/2009
ANNULAR SPACE	CE DETAILS		Date	Elev	ations	Depths	(0.01 ft.)
				(M	SL)*	(BGS)	(,
				632	2.33	-3.14	Top of Protectiv	e Casing
				632	2.04_	-2.85	Top of Riser Pip	e
Type of Surface Seal: <u>Concrete</u>			Y III	<u> </u>	9.19	0.00	Ground Surface	
				/62	5.19	3.00	Top of Annular	Sealant
Type of Annular Sealant: <u>High-s</u>	olids bentonite	- 🏹						
Installation Method: <u>Tremie</u>	:	- _	_	(0)	1.66	07.52		
Setting Time: <u>>24 hr.</u>		— <u>`</u>	4	_60	1.66		(After Completion	/el) 9/21/2009
Type of Bentonite Seal Gram	ular Pellet Slurry	\pm						
Installation Method: Gravity	(choose one)			61	0.45	18.74	Top of Seal	
Setting Time: 18 min							- op	
		- 🕅	×	_603	8.77_	20.42	Top of Sand Pac	k
Type of Sand Pack: <u>Quartz sand</u>		_						
Grain Size: <u>10/20</u> (sie	ve size)			_60	5.79_	22.40	Top of Screen	
Installation Method: <u>Gravity</u>	I	-		(0)	2 40	2(7)		
Type of Backfill Material:Quart	z Sand			<u> 60</u> . <u> 60</u> .	<u>2.40</u> 4.51	26.79	Bottom of Scree Bottom of Well	n
	(if applicable)			(0	1 10	29.00		
Installation Method: <u>Gravity</u>	/	[<u> </u>	erenced to a	National Geodeti	Bottom of Borer	iole
					CAS	SING MEA	SUREMENTS	
			[Diameter	of Boreho	ole	(inches)	8.0
WELL CONS (Choose on	TRUCTION MATERIAL e type of material for each area)	LS		ID of Rise	er Pipe		(inches)	2.0
				Protective	Casing I	Length	(feet)	5.0
Protective Casing	\$\$304 \$\$316 DTEE	DVC OTHED.	Steel	Riser Pipe	Length		(feet)	25.25
Riser Pine Above W T	SS304 SS316 PTFE	PVC OTHER		Bottom of	Screen to	o End Cap	(feet)	0.53
Riser Pipe Below W.T.	SS304 SS316 PTFE	PVC OTHER:		Total Len	ngun (1) gth of Ca	si siot to last slo sing	(feet)	30.17

Screen

SS304

SS316

PTFE

PVC OTHER:

Screen Slot Size **
**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

APPENDIX A3

GMF GYPSUM STACK POND

CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 2/25/2008 Finish: 2/25/2008 WEATHER: Overcast, cold							on /	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ /4" HSA w/SS & CME sample FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: S. Simpson	BOREHOLE ID: G200 Burface Elev: 624.2 ft. MSL Completion: 18.0 ft. BGS Station: 877,930.59N 2,515,649.96E
er	/ Total (in)	E	/ 6 in due	rre (%)	en. (lb/ft ³)	f) e Type	TOPOGR Quadra Townsl Section	APHIC MAP INFORMATION: angle: Coffeen, IL nip: East Fork 2, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\underline{\Psi} = 13.50$ - While drilling $\underline{\Psi} = 2.75 - 3/12/08$ $\underline{\nabla} =$
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moistı	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A 2A	24/24 100%	ss	3-2 3-3 N=5	31 26		1.36 <i>B</i> 1.94	2	Very dark grayish brown (10YR3/2), moist, firm, fr clayey SILT Dark gray (10YR4/1) with 5% yellowish brown (10Y mottles, moist, firm, silty CLAY	$\frac{3\sqrt{6}}{\sqrt{6}} = 624$ $\frac{3\sqrt{6}}{\sqrt{6}} = 622$
20	19/24 79%	ss	3-3 6-6 N=9	26		BSh	<u>¥</u>	Dark gray (10YR4/1) with 70% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY	
2В	10/24	\forall	3-3	26		2.33 Sh	4	Dark gray (10YR4/1) with 70% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY, slight sand	trace
3A 3B	19/24 79%	ss	4-5 N=7	26		1.59 B	, dumini	Very dark gray (10YR3/1), moist, firm, silty CLAY, trace sand	, slight
4A	22/24 92%	ss	5-5 5-5 N=10	29		0.31 B		Dark gray (10YR4/1) with 10% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY, trace c sand	n coarse
5A	20/24 83%	ss	2-2 3-5 N=5	25		1.09 B	8	Dark gray (10VP4/1) with 10% vollowich brow	 616
6A	22/24 92%	ss	1-3 2-3 N=5	22		1.01	10	(10YR5/8) mottles, moist, firm, silty CLAY, sand slight trace gravel	and - 614
7A 7B	24/24 100%	ss	3-3 5-6 N=8	15 18		0.50 B	⊻	Yellowish brown (10YR5/8), moist, soft, sandy CL Gray (10YR5/1), wet, soft, fine- to coarse-grained S	LAY
8В	19/24 79%	ss	0-3 5-8 N=8	17		0.27 B	16	Yellowish brown (10YR5/4), wet, soft, fine- to coarse-grained SAND, trace gravel	
9A 0P	24/24 100%	ss	8-15 30-50 N=45	0				Gray (10YR5/1), moist, hard, silty CLAY, trace san gravel	d and
ועל	. 1	<u> </u>		1 0			18	End of Boring = 18.0 ft. BGS	
	NOTE(S):							

FIELD BORING LC CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 10/15/2010						Station	DG	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA (blind drill)	BOREHOLE ID: R201 Well ID: R201 Surface Elev: 624.02 ft. MSL Completion: 17.22 ft BGS			
DATES: Start: 10/15/2010 Finish: 10/15/2010 WEATHER: Sunny (mid-50's)					010 010 s)			Helper: J. Litsch/D. Smail Eng/Geo: R. Hasenyager		Station	: 17.22 ft. BGS : 877,925.26N 2,514,841.96E	
aber 5	ov / Total (in) H	Æ	vs / 6 in Value D	sture (%) IS	Den. (lb/ft ³) Z	$\left(\frac{\operatorname{tsf}}{\operatorname{ure}} \frac{Qp}{Qp} \left(\operatorname{tsf} \right) \right)^{4/3}$	TOPOGRA Quadra Townsh Section	APHIC MAP INFORMATION: ngle: Coffeen, IL ip: East Fork 2, Tier 7N; Range 3W	Deadad	El seter		
Nun	Rec % R	Typ	Blov N - N	Moi	Dry	Qu (Failt	ft. BGS	Description	Detail	ft. MSL	Remarks	
							2	Very dark grayish brown (10YR3/2), moist, soft, friable, clayey SILT, slight trace sand and gravel 		622		
								Gray (10YR5/1) with 5% yellowish brown (10YR5/6) mottles, moist, firm, silty CLAY, sand and trace gravel		620		
							8	Gray (10YR5/1), moist, firm, sandy CLAY, trace silt and slight trace gravel		616		
							10	Yellowish brown (10YR5/8) with 10% gray (10YR5/1) mottles, moist, firm, sandy CLAY, trace gravel		614		
							12	Yellowish brown (10YR5/8), wet, soft, silty SAND, trace gravel		612		
								Yellowish brown (10YR5/8), moist, firm, clayey SILT				
							14	Greenish gray (5GY6/1), moist, firm, interbedded clayey SILT and SILT		610		
							16	Yellowish brown (10YR5/8), wet, soft, fine- to coarse-grained SAND, slight trace gravel 		608		
								and gravel End of Boring = 17.22 ft. BGS]			
NC)TE(S):	: R20	1 blind dr	rilled	l in b	orehole	approximate	ly 8 ft. west of G201. Lithology taken from G201.				
											Page 1 of 1	

F	[EL]	D 1	BOR	RII	NG	L	OG		
CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A								CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS & CME sample	BOREHOLE ID: G201 Well ID: G201 Surface Elev: 623.9 ft. MSL
	DATE	S: St Fin	art: 2/2 nish: 2/2	5/20 5/20	08 08			FIELD STAFF: Driller: B. Williamson Helper: R. Keedy	Completion: 18.2 ft. BGS Station: 877,924.94N
WE	EATHE	R: 0 [,]	vercast, o	cold				Eng/Geo: S. Simpson	2,514,849.47E
	SAMPL	E	Г	TEST	TING	÷	TOPOGE	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	v v		ı	(%	lb/ft ³	ы	Quadr Towns	angle: Coffeen, IL hip: East Fork	$\underline{\Psi} = 10.20$ - While drilling $\underline{\Psi} = 2.17 - 3/12/08$
er	//To		√6 ii alue	ure (en. ($\stackrel{(f)}{=} T_{y_l}$	Section	a 2, Tier 7N; Range 3W	<u>\u03c7</u> =
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failun	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	22/24 92%	ss	3-3 2-3 N=5	23				Very dark grayish brown (10YR3/2), moist, soft, fr clayey SILT, slight trace sand and gravel	iable, $\frac{\sqrt{b}}{\sqrt{b}}$
1B		Δ		33		1.16		Dark brown (10YR3/3), moist, soft, silty CLAY	Y622
	22/24 92%	ss	2-3 5-6				¥ -	Yellowish brown (10YR5/8) with 20% gray (10YF mottles moist firm silty CLAY trace sand slight	85/I)
2A	($\left(\right)$	IN=8	26		2.33 BSh	4-	gravel	
3A	20/24 83%	ss	2-4 5-5 N=9	15		1.94 B	6	Gray (10YR5/1) with 5% yellowish brown (10YR mottles, moist, firm, silty CLAY, sand and trace gr	5/6) ravel - 618
4A	24/24 100%	ss	7-7 7-6 N=14	19		1.24 B	8	Gray (10YR5/1), moist, firm, sandy CLAY, trace si slight trace gravel	It and616
5A	24/24 100%	ss	1-2 3-3 N=5	23		1.16 B	▼ 10	Yellowish brown (10YR5/8) with 10% gray (10YF mottles, moist, firm, sandy CLAY, trace grave	25/1) 1 - 614
6A	23/24 96%	ss	0-1 1-2 N=2	20			12	Yellowish brown (10YR5/8), wet, soft, silty SAND gravel	, trace
		\bigvee	3.6					Yellowish brown (10YR5/8), moist, firm, clayey S	
7A	20/24 83%	ss	6-12 N=12	22		2.72 BSh	14	Greenish gray (5GY6/1), moist, firm, interbedded c SILT and SILT	layey
8A	24/24 100%	ss	4-7 7-10 N=14	23		1.59 Sh			
8B		\square		19			16	Yellowish brown (IUYR5/8), wet, soft, fine- to coarse-grained SAND, slight trace gravel	
9A		V	7-12	15				Yellowish brown (10YR5/8), wet, firm, very fine	
9B	24/24 100%	ss	25-30 N=37	20				Gray (10YR5/1), wet, soft, SILT	
90		\square					18	Gray (101 K5/1), wet, son, line- to coarse-grained S slight trace gravel	
			I	1	I	I	10-=	Gray (101 K5/1), moist, nard, silty CLAY, trace san gravel End of Poring = 19,15 ft, DCS	
								End of boring = 18.15 ft, BG8	
	NOTE(S):							

FI	(EL) CLIEN Sit Locatio Projec DATE	D] T : All ae: Co n : Co ct: 05 S : St Fin R : O	EG Coffe CB Mana offeen, II S3004A art: 2/2 iish: 2/2 vercast, o	een H agem linoi 1/20 1/20 cold	Powe nent I is 08 08	G L or Statio Facility	OG m	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS & CME samplers FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: S. Simpson	BOREHOLE ID: G205 Well ID: G205 Surface Elev: 622.2 ft. MSL Completion: 16.0 ft. BGS Station: 875,550.19N 2,515,914.87E			
5		E	1	ES]	ring 	j	TOPOGRA	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
lber	ov / Total (in 2covery		∕s∕6 in Value	sture (%)	Den. (lb/ft ³)	tsf) tre Type	Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 11, Tier 7N; Range 3W	$\Psi = 11.10 - While drilling$ $\Psi = 5.06 - 3/12/08$ $\overline{\Psi} = $			
Num	Reco % Re	Type	Blow N - V RQI	Mois	Dry J	Qu (t Failt	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A 1B	24/24 100%	ss	4-3 4-4 N=7	22 30		1.67	2	Very dark grayish brown (10YR3/2), moist, firm, cla SILT Yellowish brown (10YR5/4) with 30% yellowish bro (10YR5/8) mottles, moist, firm, silty CLAY, slight tr	yey <u> <u> <u> </u> <u> </u> <u> </u> <u> </u> <u> </u> <u> </u> <u> </u> <u> </u> <u> </u></u></u>			
2A	19/24 79%	ss	3-5 6-5 N=11	24		Sh 1.86 B		Yellowish brown (10YR5/4) with 30% yellowish bro (10YR5/8) mottles, moist, hard, silty CLAY, slight tr	own ace 618			
3A	20/24 83%	ss	2-2 5-5 N=7	19		1.55 B	𝕎 6−					
4A	19/24 79%	ss	5-6 7-8 N=13	20		1.12 B	8-	Gray (10YR6/1) with 30% yellowish brown (10YR5 mottles, moist, firm, silty CLAY with sand, slight tra gravel	5/8) ace 614			
5A	16/24 67%	ss	2-2 3-5 N=5	20		0.62 BSh	10	Yellowish brown (10YR5/6), moist, firm, silty CLAY sand, trace gravel	with612			
6A	22/24 92%	ss	1-2 3-3 N-5	17		0.62	⊻					
6B 7A	($\left(\right)$	11-3	17 25		וטנו	12-	Yellowish brown (10YR5/6), moist, soft, sandy SIL slight trace gravel	T,610			
7B	23/24 96%	ss	3-3 5-6 N=8	17		1.36 BSh	14	Yellowish brown (10YR 5/4), moist, soft, silty SAN trace gravel	D,			
8A 8B	17/24 71%	ss	5-19 26-35 N=45	15 10				Gray (10YR5/1), moist, hard, silty CLAY, trace sand gravel	and			
	□ □ □ □ □ 16 □ End of Boring = 16.0 ft. BGS											

F	FIELD BORING LOG											
	CLIEN Sit Locatio Projec	T: A te: C n: C ct: 05	EG Coffe CB Mana offeen, Ill 5S3004A	en P Igem linois	ower ient F s	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G206 Well ID: G206 Surface Elev: 630.54 ft. MSL			
WI	DATE EATHEI	S: Si Fir R: Si	tart: 10/3 nish: 10/3 unny, war	14/2(14/2(m, b	010 010 preez	y (lo-7()'s)	FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Completion: 24.00 ft. BGS Station: 875,103.91N 2,514,669.16E			
5	SAMPL	E	Т	EST	ING	Ì	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	tal (in) v		ı	(%)	lb/ft³)	ы	Quadr Towns	angle: Coffeen, IL hip: East Fork	$\mathbf{Y} = 22.00$ - While drilling $\mathbf{Y} = 21.54$ - Upon completion			
er	/ To		/6 i alue	ure ('	en. (f) е Ту _ї	Sectior	11, Tier 7N; Range 3W	<u> </u>			
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	12/24 50%	ss	2-2 3-2 N=5	18				FILL - Grayish brown (10YR5/2), moist, firm, silty CL with trace sand and gravel.	AY			
2A	20/24 83%	ss	2-2 3-5 N=5	16					628			
3A	20/24 83%	ss	4-9 6-8 N=15	19				FILL - Dark gray (10YR4/1) with 30% dark yellowis brown (10YR4/6) mottles, moist, firm, silty CLAY w trace sand and gravel.	sh ith			
4A	19/24 79%	ss	2-4 5-6 N=9	20					624			
5A	17/24	V ss	2-3	30			8	Very dark gray (10YR3/1) with 20% dark yellowish bro (10YR4/6) mottles, moist, firm, silty CLAY with trac sand, trace roots.	own ce 			
	71%		N=7				10	Dark grayish brown (10YR4/2) with 35% dark yellow brown (10YR4/6) mottles, moist, firm, silty CLAY w trace sand.	ish ith			
6A	22/24 92%	ss	2-3 4-6 N=7	19			12	Gray (10YR5/1) with 20% dark yellowish brown (10YR4/6) mottles, moist, firm, clayey SILT with tra sand and gravel.	ce			
7A	23/24 96%	ss	1-2 3-4 N=5	23				Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, firm, silty CLAY with trac sand and gravel.	ce 618			
8A	22/24 92%	ss	<i>1-1</i> <i>3-3</i> N=4	22				Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace s and gravel.	sand			
9A	24/24 100%	ss	1-1 2-2 N=3	21				Dark yellowish brown (10YR4/6) with 30% gray (10YR5/1) mottles, moist, soft, silty CLAY with trace s and gravel. Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace s and gravel	614 sand / 614 sand			
10A	24/24 100%	ss	woh-woł 1-5	25			18	Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, very soft, silty CLAY with t <u>sand and gravel.</u> Gray (10YR5/1), moist, very soft, very fine- to fine-grained sandy CLAY with trace gravel.	race 612			
NC) DTE(S):	∐ G20	 6 installee	 1 in l	 boreł	nole.	20	Gray (10YR5/1), moist, firm, very fine- to fine-graine	$\overline{ed} = -$			



CUTENT: ASC Ceffice Power Station Site: CBA Management Taking Taking informatic CME/250 ATV Delt Taking informatic CME/250 ATV Delt Data Mathews 40° 180 ASS amples. The protein CSS001A DATES: Nark: 108/2010 FIELD STATE: Delta: Data in Statistic 108/2010 FIELD STATE: Delta: Data in Statistic 108/2010 FIELD STATE: Delta: Data in Statistic 108/2010 FIELD STATE: Delta: Data in Statistic 108/2010 FIELD STATE: Delta: Data in Statistic 108/2010 FIELD STATE: Delta: Data in Statistic 108/2010 FIELD Statistic 108/2010 FIE	F	FIELD BORING LOG											
Lacency Coller, Humon Dolling Multiple 42° H5A wSS samples We (2017) Project: 0550040 FIELD SLIFE: Deflict: D. Malaria Completion: 2:04.01 #.055 WEATURE: Nam; 108/2010 FIELD SLIFE: Deflict: D. Malaria Section: 8:75.10.63.N0 WEATURE: Start; 108/2010 TOSING Engress: 1:1.88:200 Section: 8:75.10.63.N0 WEATURE: Start; 108/2010 TOSING Engress: 1:1.88:200 NATEL EXENT: NORMATION: WEATURE: Start; 108/2010 TOSING Toronspic: 1:6:1.80:1.1. Toronspic: 1:6:1.80:1. Visition: 11, Tior 7N; Range: JW Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. 10. Section: 11, Tior 7N; Range: JW Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. 20.4 Section: 11, Tior 7N; Range: JW Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. 20.4 Section: 11, Tior 7N; Range: JW Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. 20.4 Section: 11, Tior 7N; Range: JW Toronspic: 1:6:1.90:1. Toronspic: 1:6:1.90:1. Toronspi		CLIEN Sit	T: A te: C	EG Coffe CB Mana	een P igem	ower ent F	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G207			
Data PRLD STAFE Delta NTAFE Comparison Comparison </td <td></td> <td>Locatio</td> <td>n: Co</td> <td>offeen, Ill</td> <td>linois</td> <td>5</td> <td></td> <td></td> <td>Drilling Method: 4¹/₄" HSA w/SS samplers</td> <td>Well ID: G207 Surface Floy: 630.61 ft MSI</td>		Locatio	n: Co	offeen, Ill	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G207 Surface Floy: 630.61 ft MSI			
High: US200 Height: Lake/D Statu: Statu:<		DATE	S: S1	tart: 10/8	8/201	10			FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS			
TextureTargetsTextureSAMPLETextureTextureTopOcRAPHIC MAP INFORMATION: \underline{y} Texture \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} \underline{y} <td>W/F</td> <td>атнғ</td> <td>Fir D. S.</td> <td>nish: 10/8</td> <td>8/20 d</td> <td>10</td> <td></td> <td></td> <td>Helper: J. Litsch/D. Smail</td> <td>Station: 875,166.36N</td>	W/F	атнғ	Fir D. S.	nish: 10/8	8/20 d	10			Helper: J. Litsch/D. Smail	Station: 875,166.36N			
WILK LEVEL INFORMATION:WILK LEVEL INFORMATION:Ourdarage: Configure, IIOurdarage: Configure, IIOurdarage: Configure, IITo conside: Ease Tool.WILK LEVEL INFORMATION:Ourdarage: Configure, IITo conside: Ease Tool.WILK LEVEL INFORMATION:WILK LEVEL INFORMATION:Ourdarage: Configure, IITo conside: Ease Tool.WILK LEVEL INFORMATION:Understanding to the tool.Tool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Dy: Upon completionTool Section:Tool Section:Tool Section:Tool Section:Tool Section: <th co<="" th=""><th></th><th>SAMPL</th><th>к. э. Е</th><th>T</th><th>u EST</th><th>INC</th><th>Ţ</th><th>TODOG</th><th></th><th>2,314,037.74E</th></th>	<th></th> <th>SAMPL</th> <th>к. э. Е</th> <th>T</th> <th>u EST</th> <th>INC</th> <th>Ţ</th> <th>TODOG</th> <th></th> <th>2,314,037.74E</th>		SAMPL	к. э. Е	T	u EST	INC	Ţ	TODOG		2,314,037.74E		
$\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$ $\frac{1}{20}$		(in)				()		TOPOGR Quadr	angle: Coffeen, IL	WATER LEVEL INFORMATION: $\mathbf{\nabla} = $ Dry - While drilling			
a a ab a a ac a a bc c b bc c b bc c b bc c b b b b b b b b b b b b b b b c b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b b 		otal (<i>ry</i>		in	(%)	(lb/fi	ədı	Towns	hip: East Fork	Ψ = Dry - Upon completion			
$\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ $\overline{2}$ 2	ber	v / T		s/6 /alue	ture	Den.	sf) ure Tj	Section	1 11, Tier 7N; Range 3W	<u>¥</u> =			
1A $\frac{23}{90\%}$ 88 $\frac{6}{5}$ $\frac{5}{5}$ $\frac{6}{5}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ $\frac{6}{10}$ <td>Num</td> <td>Reco % Re</td> <td>Type</td> <td>Blow N - V RQD</td> <td>Mois</td> <td>Dry I</td> <td>Qu (t Failh</td> <td>Depth ft. BGS</td> <td>Lithologic Description</td> <td>Borehole Elevation Detail ft. MSL Remarks</td>	Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failh	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
$1A$ $\frac{2324}{2075}$ $8S$ $\frac{2}{3}$ $\frac{2}{3}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$ $\frac{1}{10}$			V	(5					FILL Vallowich brown (10VP5/6) with 10% grav	-630			
1A A N=10 15 Instantial and sign function of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the solution of the s		23/24 96%	ss	6-5 5-8					(10YR6/1) and 5% black (10YR2/1) mottles, slightl	y 🛞 🛸			
2A 24.74 100% N = 11 15 2 - TIL - Gray (10YR5/1) with 20% gray (10YR5/1) with 20% gray (10YR5/3) motiles, most, hard, silty CLAY with sand and trace gravel.	1A		\wedge	N=10	15				gravel.				
2A $24/24$ 100% / 8 $4/7N=11$ 15 (10VR58) motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, motils, m			()					2	FILL - Gray (10XR5/1) with 20% vellowish brown				
100% 8 $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$ $100%$	24	24/24	V	4-4	15				(10YR5/8) mottles, moist, hard, silty CLAY with sand	and 628			
3A $\frac{1}{23/24}$ $\frac{3}{8}$ $\frac{3}{7.9}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ $\frac{4}{100}$ <t< td=""><td>21</td><td>100%</td><td>Å ss</td><td>7-8 N=11</td><td>15</td><td></td><td></td><td></td><td></td><td></td></t<>	21	100%	Å ss	7-8 N=11	15								
3A $23/24 \\ 90\% \\ 90\% \\ 80\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 100\% \\ 10$			Δ					4					
3A $23/24$ 8 $3-4$ $6-29$ $4A$ $24/24$ 8 $3-4$ 16 FILL - Dark yellowish brown (10YR54), with 20% yellowish brown (10YR54) $6-24$ $5A$ $24/24$ 8 $2-2$ 22 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74 74			V						FILL - Yellowish brown (10YR5/8) with 20% gray				
3A 20.5° AN=1317 1764A 24.24 100% ss $\frac{3.4}{6}$ 165A 24.24 100% ss $\frac{2.2}{5.4}$ 225B $\frac{2.2}{5.4}$ 2275B $\frac{2.2}{5.4}$ 24106A $\frac{2.2.2}{2.5}$ 2777A $\frac{24.24}{100\%}$ ss $\frac{2.2}{5.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{5.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{5.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{5.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{5.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{5.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{3.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{3.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{3.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{3.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{3.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{2.2}{3.3}$ 277A $\frac{24.24}{100\%}$ ss $\frac{3.4}{3.43}$ 258A $\frac{2.2.2}{2.3}$ $\frac{2.2}{3.43}$ 259A $\frac{2.2.2}{2.3}$ $\frac{2.2}{10}$ $\frac{2.2}{10}$ 9A $\frac{2.2.2}{2.3}$ $\frac{2.2}{3.43}$ 219A $\frac{2.2.2}{2.3}$ $\frac{2.2}{10}$ $\frac{2.2}{10}$ 9A $\frac{2.2.2}{2.3}$ $\frac{2.2}{10}$ $\frac{2.2}{10}$ <td< td=""><td></td><td>23/24</td><td>ss</td><td>3-6 7-9</td><td></td><td></td><td></td><td></td><td>(10 Y R6/1) motules, moist, nard, sitty CLA Y with sand trace gravel.</td><td></td></td<>		23/24	ss	3-6 7-9					(10 Y R6/1) motules, moist, nard, sitty CLA Y with sand trace gravel.				
4A $\frac{24/24}{100\%}$ ss $\frac{3.4}{6.7}$ 166a5A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.4}$ 225B $\frac{2.2}{100\%}$ ss $\frac{2.2}{3.4}$ 226A $\frac{22.2}{2.3}$ $\frac{2.2}{100\%}$ $\frac{2.2}{100\%}$ $\frac{2.2}{100\%}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 8A $\frac{24/24}{100\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{2.7}{10}$ 9A $\frac{23/24}{20\%}$ ss $\frac{2.2}{3.3}$ $\frac{2.7}{10}$ $\frac{14}{14}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.3}$ $\frac{14}{14}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.3}$ $\frac{10}{10}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.4}$ $\frac{10}{10}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.4}$ $\frac{10}{10}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.4}$ $\frac{10}{10}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.4}$ $\frac{10}{10}$ 9A $\frac{23/24}{20\%}$ ss $\frac{10}{3.4}$ $\frac{10}{10}$ 9A	3A	90%	Λ	N=13	17								
4A $24/24$ 100% ss $\frac{3-4}{8}$ N=1016FIL - Dark yellowish brown (10/R5/4), moist, hand gravel. CLAY with slight traces and and gravel.6245A $24/24$ 100% ss $\frac{2-2}{3-4}$ 2222FIL - Gray (10/R5/1) with 20% yellowish brown (10/R5/8) mottles, moist, firm, slip CLAY with slight traces and and gravel.6225B2410Gray (10/R5/1) with 15% yellowish frown (10/R5/8) mottles, most, firm, slip CLAY with slight traces and and gravel.6225B2410Gray (10/R5/1) with 15% yellowish frown (10/R5/8) mottles, most, firm, slip CLAY with slight traces and and gravel.6206A $22/24$ 922% ss $\frac{2-2}{2-3}$ $2-3$ 277A $24/24$ 100% ss $\frac{2-2}{2-3}$ $2-3$ 6308A $24/24$ 100% ss $\frac{2-2}{2-3}$ $2-3$ 277A $24/24$ 100% ss $\frac{2-2}{2-3}$ $2-3$ 278A $24/24$ 100% ss $\frac{2-2}{2-3}$ $2-3$ 6189A $23/24$ 50% ss $\frac{2-2}{2-3}$ $2-3$ 259A $23/24$ 50% ss $\frac{woh-2}{2-3}$ 19 259A $23/24$ 50% ss $\frac{woh-2}{2-3}$ $2-3$ 259A $23/24$ 50% ss $\frac{woh-2}{2-3}$ $2-3$ 259A $23/24$ 50% ss $\frac{woh-2}{2-3}$ $2-3$ 269A $23/24$ 50% ss $\frac{woh-2}{2-3}$ $2-3$ 269A $23/24$ 50% 19109A </td <td></td> <td></td> <td>H</td> <td></td> <td></td> <td></td> <td></td> <td>6</td> <td></td> <td></td>			H					6					
4A $24/24$ ss 6.7 16 5A $24/24$ ss $\frac{2}{3.4}$ 22 5A $24/24$ ss $\frac{2}{3.4}$ 22 5B 24 10 Yellowish brown (10YR5/0) with 25% gray (10YR5/1) mottles, moist, firm, silty CLAY with slight trace sand and gravel. 622 5B 24 10 Gray (10YR5/1) with 25% gray (10YR5/1) mottles, moist, firm, silty CLAY with slight trace sand and gravel. 620 6A $22/24$ ss $\frac{2}{2.5}$ 27 620 7A $24/24$ ss $\frac{2}{2.5}$ 27 618 7A $24/24$ ss $\frac{2}{2.2}$ 27 618 9A $23/24$ ss $\frac{2}{2.3}$ 27 7 9A $23/24$ ss $\frac{2}{2.3}$ 27 7 9A $23/24$ ss $\frac{woh-1}{2.3}$ 25 616 9A $23/24$ ss $\frac{woh-2}{2.3}$ 22 7 614 100% ss $\frac{woh-2}{2.3}$ 22 22 614 616 10A $24/24$		24/24	V	3-4					FILL - Dark yellowish brown (10YR4/4), moist, hard, CLAY with slight trace sand and gravel.	silty 624			
SA $24/24$ 100% Sa 2^{-2}_{3-4} 2^{-2}_{3-4} Carav (107R5/8) motiles, most, fam, sity CLAY with slight trace sand and gravel.SA $24/24$ 	4A	100%	ss	6-7 N=10	16								
Index stand and gravel.5A $\frac{24/24}{100\%}$ ss $\frac{2}{3.4}$ 225B2410 $\frac{1}{300\%}$ $\frac{1}{300\%}$ $\frac{1}{300\%}$ 6A $\frac{22/24}{92\%}$ ss $\frac{2}{2.5}$ 27 $\frac{1}{300\%}$ 6A $\frac{22/24}{92\%}$ ss $\frac{2}{2.5}$ 27 $\frac{1}{300\%}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{2.5}$ 27 $\frac{1}{300\%}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{2.5}$ 27 $\frac{1}{300\%}$ 8A $\frac{24/24}{100\%}$ ss $\frac{2.2}{2.3}$ 27 $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{2.2}{2.3}$ 22 $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A $\frac{23/24}{90\%}$ ss $\frac{1}{2.3}$ $\frac{1}{10}$ 9A			((10YR5/8) mottles, moist, hard, silty CLAY with slig	ht ht			
Telewish forwin (10/R5/6) with 25% gray (10/R5/1)5A $\frac{24/24}{100\%}$ $\frac{32}{8}$ $\frac{22}{2}$ $\frac{22}{2}$ 5B2410 $\frac{10}{6}$ $\frac{10}{9}$ $\frac{10}{9}$ $\frac{10}{6}$ 6A $\frac{22/24}{22/24}$ ss $\frac{2-2}{2}$ $\frac{2}{2}$ $\frac{2}{2}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2-2}{2}$ $\frac{2}{27}$ $\frac{10}{9}$ $\frac{10}{9}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2-2}{2}$ $\frac{2}{27}$ $\frac{10}{9}$ $\frac{10}{9}$ $\frac{10}{9}$ 7A $\frac{24/24}{100\%}$ ss $\frac{2-2}{2}$ $\frac{2}{27}$ $\frac{11}{10}$ $\frac{11}{9}$ $\frac{11}{9}$ $\frac{11}{9}$ 8A $\frac{24/24}{100\%}$ ss $\frac{2-2}{2-3}$ $\frac{2}{27}$ $\frac{11}{10}$ $\frac{11}{10}$ $\frac{11}{9}$ $\frac{11}{9}$ $\frac{11}{9}$ $\frac{11}{9}$ 9A $\frac{23/24}{20\%}$ ss $\frac{whr-2}{2-3}$ 22 $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ 9A $\frac{23/24}{96\%}$ ss $\frac{whr-2}{2-3}$ 22 $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ 9A $\frac{23/24}{96\%}$ ss $\frac{whr-2}{2-3}$ 22 $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ 9A $\frac{23/24}{96\%}$ ss $\frac{whr-2}{2-3}$ $\frac{22}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ $\frac{10}{10}$ 9A $\frac{23/24}{96\%}$ ss whr			∇						Vallaugich brown (10VD5/6) with 25% array (10VD5				
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5A	24/24	ss	2-2 3-4	22				mottles, moist, firm, silty CLAY with slight trace sand	and 622			
5Bmottles, moist, firm, sity CLAY with slight trace sand and gravel.6A $22/24$ 92% ss 2.2 2.5 27Gray (10YR5/1) with 40% yellowish brown (10YR5/8) mottles, very moist, firm, sity CLAY with slight trace sand and gravel.6207A $24/24$ 100% ss 2.2 		100%	Λ	N=5					Gray (10YR5/1) with 15% yellowish brown (10YR5/	8)			
$6A$ $22/24$ 92% ss $2-2$ $2-5$ 27 $N=4$ $Gray (10YR5/1)$ with 40% yellowish brown (10YR5/8) mottles, very moist, firm, sity CLAY with slight trace sand and gravel. $7A$ $24/24$ 100% ss $2-22-327N=412127A24/24100\%ss2-22-327N=4618148A24/24100\%ss\frac{2-2}{2-3}N=325N=31414-Gray (10YR6/1) with 30% gray (10YR6/1)mottles, very moist, soft, sithy CLAY with sand and slighttrace gravel.6166169A23/2496\%sswoh-22-322N=41616-Teal products, very moist, very soft, sandy, sithy CLAY withsandy, sithy CLAY withrace gravel.61461410A24/2410\%sswoh-woh2-31916101818206127272/2273/2273/2273/2273/2273/2210A24/2410\%2074/241974/242074/24100\%74/24100\%74/24100\%74/2473/2273/2273/2273/2273/2273/2274/2474/24100\%74/24100\%74/24100\%74/24100\%74/24100\%74/2473/2273/2274/241974/241974/241974/24100\%74/2474/24$	5B		H		24			10	mottles, moist, firm, silty CLAY with slight trace sand gravel.	and			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		22/24	V	2-2	07				Grav (10VR5/1) with 40% vellowish brown (10VR5/				
7A $24/24$ 100% 8s $2 \cdot 2$ $2 \cdot 3$ 271246188A $24/24$ 100% 8s $woh-1$ $2 \cdot 3$ 2576189A $24/24$ 96% 8s $woh-2$ $2 \cdot 3$ 226169A $23/24$ 96% 8s $woh-2$ $2 \cdot 3$ 2261410A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 191610A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 8s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 9s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 9s $woh-woh$ $2 \cdot 3$ 1961210A $24/24$ 100% 9s $woh-woh$ 19 1961210A $26/24$ $2 \cdot 3$ 191061210A $26/24$ $2 \cdot 3$ 191061210A $26/24$ $2 \cdot 3$ 19101010A $26/24$ $2 \cdot 3$ 19101010A $26/24$ $2 \cdot 3$ 19101010A $26/24$ $2 \cdot 3$	6A	92%	ss	2-5 N=4	21				mottles, very moist, firm, silty CLAY with slight trad	ze			
7A $\frac{24/24}{100\%}$ ss $\frac{2.2}{2.3}$ 27128A $\frac{24/24}{100\%}$ ss $\frac{woh-1}{2.3}$ 25 $\frac{14}{14}$ $\frac{14}{14}$ $\frac{14}{14}$ 9A $\frac{24/24}{96\%}$ ss $\frac{woh-2}{2.3}$ 25 $\frac{14}{16}$ $\frac{14}{16}$ $\frac{14}{16}$ 9A $\frac{23/24}{96\%}$ ss $\frac{woh-2}{N=4}$ 22 $\frac{16}{16}$ $\frac{16}{16}$ 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2.3}$ 19 $\frac{16}{18}$ 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2.3}$ 19 $\frac{16}{18}$ 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2.3}$ 19 $\frac{16}{20}$ 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2.3}$ 19 $\frac{16}{20}$ 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2.3}$ 19 $\frac{16}{20}$ 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2.3}$ 19 $\frac{16}{20}$ 10A $\frac{26}{2.5}$ $\frac{19}{2.0}$ $\frac{10}{20}$ $\frac{10}{20}$ $\frac{10}{20}$ 10A $\frac{26}{2.5}$ $\frac{19}{2.5}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ 10A $\frac{26}{2.5}$ $\frac{19}{2.5}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ 10A $\frac{26}{2.5}$ $\frac{10}{2.5}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ 10A $\frac{26}{2.5}$ $\frac{10}{2.5}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ $\frac{10}{2.0}$ 10A $\frac{26}{2.5}$ $\frac{10}{2.5}$ $\frac{10}{2.5}$ $\frac{10}{2.5}$ $\frac{10}{2.5}$ 10A $\frac{10}{2.5}$ $\frac{10}{2.5}$ $\frac{10}{$			()					12	sand and graver.				
7A $\frac{24/24}{100\%}$ ss $\frac{2-2}{2\cdot3}$ 276188A $\frac{24/24}{100\%}$ ss $\frac{woh-1}{2\cdot3}$ 2514 $\frac{14}{10}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%}$ $\frac{14}{100\%$			∇										
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	7A	24/24	ss	2-2 2-3	27					618			
8A $\frac{24/24}{100\%}$ ss $\frac{woh-1}{2\cdot3}$ 25 9A $\frac{23/24}{96\%}$ ss $\frac{woh-2}{N=4}$ 22 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2\cdot3}$ 19 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2\cdot3}$ 19 NOTE(S): G207 installed in borehole.		100%	Λ	N=4					Yellowish brown (10YR5/8) with 30% gray (10YR6/ mottles, very moist, soft, silty CLAY with trace sand a	(1) and			
8A $\frac{24/24}{100\%}$ ss $\frac{woh-1}{2\cdot3}$ 25 9A $\frac{23/24}{96\%}$ ss $\frac{woh-2}{N=3}$ 22 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2\cdot3}$ 19 10A $\frac{24/24}{100\%}$ ss $\frac{woh-woh}{2\cdot3}$ 19 NOTE(S): G207 installed in borehole.			H					14 -	gravel.				
8A $\frac{24/24}{100\%}$ ss $\frac{2\cdot3}{N=3}$ 25 N=3 $\frac{2\cdot3}{N=3}$ 25 9A $\frac{23/24}{96\%}$ ss $\frac{woh-2}{2\cdot3}$ 22 16 $\frac{16}{16}$ Yellowish brown (10YR5/6) with 25% gray (10YR6/1) mottles, very moist, very soft, sandy, silty CLAY with trace gravel. $\frac{614}{100\%}$ 614 18 $\frac{18}{2\cdot3}$ $\frac{19}{2\cdot3}$ 19 $\frac{18}{2\cdot3}$ $\frac{19}{2\cdot3}$ 19 $\frac{18}{2\cdot3}$ $\frac{19}{2\cdot3}$ 19 $\frac{18}{2\cdot3}$ $\frac{18}{2\cdot3}$ $\frac{19}{2\cdot3}$ 1		24/24	V	woh-1						616			
9A $23/24$ 96% ss $woh-2$ $2-3$ $N=4$ 2216 16 $trace gravel.$ $Trace gravel.$ 61410A $24/24$ 100% ss $woh-woh$ $2-3$ 19 18 20 $Trace gravel.$ $Trace gravel.$ 612NOTE(S): G207 installed in borehole.	8A	24/24	ss	2-3 N=3	25				Gray (10YR6/1) with 30% yellowish brown (10YR5/				
9A $\begin{bmatrix} 23/24\\ 96\%\\ 96\%\\ 8s \end{bmatrix} \begin{bmatrix} woh-2\\ 2-3\\ N=4 \end{bmatrix}$ 22 $\begin{bmatrix} 16\\ -\\ 18\\ -\\ 18\\ -\\ 20\end{bmatrix}$ Yellowish brown (10YR5/6) with 25% gray (10YR6/1) mottles, very moist, very soft, sandy, silty CLAY with trace gravel. 10A $\begin{bmatrix} 24/24\\ 100\%\\ 2-3 \end{bmatrix}$ 19 $\begin{bmatrix} 16\\ -\\ 18\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\ -\\$			()						trace gravel.				
9A $23/24$ 96% ss $woh-2$ $2-3$ 22Yellowish brown (10YR5/6) with 25% gray (10YR6/1) mottles, very moist, very soft, sandy, silty CLAY with trace gravel.10A $24/24$ 100% ss $woh-woh$ $2-3$ 1918			\forall										
$10A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} x \begin{vmatrix} x - y \\ 2 - 3 \end{vmatrix} x \begin{vmatrix} x - y \\ 2 - 3 \end{vmatrix} x \begin{vmatrix} x - y \\ 19 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 18 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \begin{vmatrix} x - y \\ 10 \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} x \end{vmatrix} $	9A	23/24	V	woh-2	22				Vellowish brown (10VR5/6) with 25% area (10VP6				
$10A \begin{vmatrix} 24/24 \\ 100\% \\ ss \end{vmatrix} woh-woh \\ 2-3 \end{vmatrix} 19 \begin{vmatrix} 18 \\ -2-3 \\ 20 \end{vmatrix} = \begin{bmatrix} 18 \\ -4 \\ -4 \\ -4 \\ -4 \\ -4 \\ -4 \\ -4 \\ -$		96%		N=4					mottles, very moist, very soft, sandy, silty CLAY wi	ĥ E			
$10A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} ss \begin{vmatrix} woh-woh \\ 2-3 \end{vmatrix} 19 \begin{vmatrix} 19 \\ 20 \end{vmatrix} - \frac{1}{Yellowish brown (10YR5/6) with 30\% gray (10YR6/1) \\ mottles, very moist, very soft, sandy CLAY with silt and trace gravel.} 612$ $NOTE(S): G207 installed in borehole.$			Д					18	trace gravel.				
$10A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} \text{ ss} \begin{vmatrix} w0n-w0n \\ 2-3 \end{vmatrix} \begin{vmatrix} 19 \\ 2-3 \end{vmatrix} 19 \end{vmatrix}$ Yellowish brown (10YR5/6) with 30% gray (10YR6/1) mottles, very moist, very soft, sandy CLAY with silt and trace gravel.			V	woh						612			
Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image: Image:	10A	24/24 100%	ss	2-3	19				Yellowish brown (10YR5/6) with 30% gray (10YR6/				
NOTE(S): G207 installed in borehole.			\wedge						mottles, very moist, very soft, sandy CLAY with silt a trace gravel.	nd			
	NC))TE(S):	G20	7 installed	d in l	boreł	nole.	20 -					

F	FIELD BORING LOG												
	CLIEN	T: A	EG Coffe	een P	ower	r Statio	n	CONTRACTOR: Layne-Western Co					
	Sit	te: Co	CB Mana	igem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G207				
	Locatio	n: Co	offeen, Ill	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G207				
	Projec	et: 05	5S3004A						Surface Elev: 630.61 ft. MSL				
	DATE	S: St	t art: 10/8	8/201	10			FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS				
		Fir	nish: 10/3	8/20	10			Helper: J. Litsch/D. Smail	Station: 875,166.36N				
WE	WEATHER: Sunny, mildEng/Geo: R. Hasenyager2,514,837.94E												
SAMPLE TESTING TOPOGRAPHIC MAP INFORMATION: WATER LEVEL INFORMATIC													
er	/ Total (in) overy		/ 6 in llue	ıre (%)	en. (lb/ft ³)	f) e Type	Quadran Township Section 1	gle: Coffeen, IL 5: East Fork 1, Tier 7N; Range 3W	$\Psi = Dry - While drilling$ $\Psi = Dry - Upon completion$ $\Psi =$				
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry Do	Qu (ts1 Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
11A 12A	22/24 92% 24/24 100%	SS SS SS	woh-1 2-2 N=3 10-24 26-30 N=50	18			22	Yellowish brown (10YR5/6) with 30% gray (10YR6/ mottles, very moist, very soft, sandy CLAY with silt a trace gravel. [Continued from previous page] Gray (10YR6/1) with 40% yellowish brown (10YR5/ mottles, very moist to wet, loose, silty, very fine- to fine-grained SAND with slight trace gravel. Gray (10YR5/1), slightly moist, hard, very silty CLA with trace sand and slight trace gravel.	1) nd 6) Y Y				
	End of Boring = 24.0 ft. BGS												

F	FIELD BORING LOG												
	CLIEN Sit Location Projec DATE	Γ: Α α: C n: C n: C st: 05 S: St	EG Coffe CB Mana offeen, Ill S3004A art: 10/	een P igem linois 7/2.01	Power ient F s	Statio acility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: D Mahurin	BOREHOLE ID: G208 Well ID: G208 Surface Elev: 630.57 ft. MSL Completion: 24 00 ft BGS				
WF	THE	Fin Fin	ish: 10/	7/20 m (1	10	c)		Helper: J. Litsch/D. Smail	Station: 875,231.46N				
	SAMPL	к. эт Е	T	EST	CINC	s) i	TOPOCE		2,314,775.37E				
er	/ Total (in) overy		/ 6 in Ilue	ire (%)	en. (lb/ft ³)	f) e Type	TOPOGE Quadr Towns Section	applie: Coffeen, IL hip: East Fork h 11, Tier 7N; Range 3W	water Level information: $\Psi = Dry$ - While drilling $\Psi = 23.92$ - Upon completion $\overline{\Psi} =$				
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (ts) Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
1A	19/24 79%	ss	4-3 3-5 N=6	23			2	FILL - Brown (10YR4/3) with 5% dark gray (10YR4 and 5% dark yellowish brown (10YR4/6) mottles, mo firm, silty CLAY with trace sand and gravel.	/1) ist, 				
2A	24/24 100%	ss	2-3 4-6 N=7	14			4						
3A	24/24 100%	ss	2-4 4-7 N=8	21			6	FILL - Brown (10YR4/3) with 15% dark gray (10YR4 and 5% dark yellowish brown (10YR4/6) mottles, mo firm, silty CLAY with trace sand and gravel.	4/1) vist,				
4A	24/24 100%	ss	2-4 6-8 N=10	17			8		- 624				
	20/24	V	2-2					Very dark gray (10YR3/1), moist, firm, silty CLAY v	vith 622				
5A	83%	ss	4-5 N=6	24				trace sand and gravel, trace roots. Dark gray (10YR4/1) with 10% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with tra 	vn ce				
6A	23/24 96%	ss	1-2 4-4 N=6	26			1	Dark gray (10YR4/1) with 15% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with tra sand, trace roots.	vn ce 620				
7A	19/24 79%	ss	<i>1-2</i> <i>2-3</i> N=4	23			12	Dark gray (10YR4/1) with 15% dark yellowish brow (10YR4/6) mottles, moist, soft, silty CLAY with tra	vn ce				
8A	22/24 92%	ss	<i>1-1</i> <i>2-3</i> N=3	24			16	sand, trace roots.	616				
9A	24/24 100%	ss	<i>1-1</i> 2-3 N=3	24				Dark gray (10YR4/1) with 15% dark yellowish brow (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel, trace roots.	vn sand 614				
9B		$\langle \rangle$	wah wal	20			18 -	Gray (10YR5/1) with 20% dark yellowish brown (10YR4/6) mottles, very moist, soft, silty, very fine-fine-grained sandy CLAY with trace gravel.	to 612				
10A 10B	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$						20	Dark yellowish brown (10YR4/6) with 15% gray (10YR5/1) mottles, very moist to wet, soft, clayey, very fine- to medium-grained SAND with trace gravel.	ery				
NC	DTE(S):	G208	8 installed	d in l	boreł	ole.	20						

FI	EL	DI	BOR	IN	G	L	DG		HANSON	
	CLIEN	T : A	EG Coffe	en P	owe	r Statio	n	CONTRACTOR: Layne-Western Co		
	Sit	e: Co	CB Mana	igem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G208	
	Locatio	n: Co	offeen, Ill	inois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G208	
	Projec	et: 05	S3004A						Surface Elev: 630.57 ft. MSL	
	DATE	S: St	art: 10/7	7/201	10			FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS	
		Fin	ish: 10/	7/20	10			Helper: J. Litsch/D. Smail Statio		
WE	ATHE	R: Su	inny, war	m (l	o-70	's)		Eng/Geo: S. Simpson	2,514,993.57E	
S	SAMPL	E	Т	EST	TING	Ĵ	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION.	
sr	/ Total (in) overy		/ 6 in lue	ire (%)	en. (lb/ft ³)	t) e Type	Quad Town Sectio	rangle: Coffeen, IL ship: East Fork n 11, Tier 7N; Range 3W	Ψ = Dry - While drilling Ψ = 23.92 - Upon completion $\overline{\Psi}$ =	
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry Do	Qu (ts1 Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks	
11A	20/24 83%	ss	woh-2 5-12 N=7	13			22	Yellowish brown (10YR5/4), moist, firm, very silty CI with sand and gravel.	AY	
12A	$12A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} \int ss \begin{vmatrix} 6-11 \\ 18-24 \\ N=29 \end{vmatrix} 9$							Dark gray (10YR4/1), slightly moist, hard, very silty CLAY with sand and gravel.		
							21	End of Boring = 24.0 ft. BGS		

F	FIELD BORING LOG												
	CLIEN Sit Location Projec DATE	Г: А) е: Со n: Со et: 05 S: St Fin	EG Coffe CB Mana offeen, III S3004A art: 10/ ish: 10/	een P agem linois 7/201 7/201	Power ent F s 10 10	Statio Cacility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail	BOREHOLE ID: G209 Well ID: G209 Surface Elev: 630.57 ft. MSL Completion: 24.00 ft. BGS Station: 875,298.23N				
wr S	SAMPLI	K: SL E	inny, coo T	TEST	-50 s) FING) 		Eng/Geo: S. Simpson	2,515,149.50E				
ber	/ / Total (in) covery		alue	ure (%))en. (lb/ft ³)	st) ve Type	TOPOGR Quadr Towns Sectior	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork 111, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = Dry$ - While drilling $\Psi = 22.40$ - Upon completion $\overline{\Sigma} =$				
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failun	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
1A	24/24 100%	ss	4-4 4-6 N=8	21			2		630				
2A	24/24 100%	ss	3-4 6-6 N=10	13			4	FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a 5% dark yellowish brown (10YR4/6) mottles, moist, fi silty CLAY with trace sand and gravel.	nd rm,				
3A	24/24 100%	ss	2-3 6-8 N=9	19									
4A	22/24 92%	ss	2-3 6-8 N=9	17			6		624				
	(8	Grayish brown (10YR5/2), moist, firm, clayey SILT w trace sand and gravel.					
5A	18/24 75%	ss	2-3 3-5 N=6	20			10	Dark gray (10YR4/1) with 15% dark vellowish brow	n 622				
6A	24/24 100%	ss	1-2 2-5 N=4	26			12	(10YR4/6) mottles, moist, firm, silty CLAY with trac sand.	620				
7A	22/24 92%	ss	<i>1-3</i> <i>4-4</i> N=7	22				Dark gray (10YR4/1) with 25% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with trac sand.	n ce 618				
8A	24/24 100%	ss	<i>woh-1</i> 2-3 N=3	25			14		616				
9A	19/24 79%	ss	<i>woh-1</i> 2-3 N=3	24			18	Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace s and gravel.	sand				
10A	14/24 58%	ss	<i>woh-2</i> 3-3 N=5	20					612				
NC	NOTE(S): G209 installed in borehole.												

I

FI	FIELD BORING LOG											
	CLIEN	T: A	EG Coffe	een P	owe	r Statio	n	CONTRACTOR: Layne-Western Co				
	Sit	e: Co	CB Mana	igem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G209			
	Locatio	n: Co	offeen, Ill	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G209			
	Projec	et: 05	5S3004A						Surface Elev: 630.57 ft. MSL			
	DATE	S: St	art: 10/	7/201	10			FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS			
		Fin	nish: 10/	7/20	10			Helper: J. Litsch/D. Smail	Station: 875,298.23N			
WE	ATHE	R: Si	unny, coo	l (lo-	-50's))		Eng/Geo: S. Simpson	2,515,149.56E			
S	SAMPL	E	Т	EST	TING	3	TOPOGE	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
ST	/ Total (in) overy	Very V6 in lue re (%) $r. (lb/f^3)$ True					Quadi Town Sectio	rangle: Coffeen, IL ship: East Fork n 11, Tier 7N; Range 3W	Ψ = Dry - While drilling Ψ = 22.40 - Upon completion $\overline{\Psi}$ =			
Numbe	Recov % Reco	Type	Blows N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
11A	1/24 <i>4%</i>	ss	woh-1 1-1 N=2	21			22	Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel. [Continued from previous page]	sand			
12A	20/24 83%	ss	9-16 17-26 N=33	7			✓ Dark gray (10YR4/1), slightly moist, hard, very silty 608 24 CLAY with sand and gravel.					
	End of Boring = 24.0 ft. BGS											

F	[EL]	DI	BOR	IN	IG	L(DG		C HANSON
	CLIEN Sit Locatio Projec	T: A te: C on: C ct: 05	EG Coffe CB Mana offeen, Il 5S3004A	een P agem linois	Power lent F s	Statio acility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	BOREHOLE ID: G210 Well ID: G210 Surface Elev: 630.48 ft. MSL
WI	DATE EATHE	S: St Fir R: St	t art: 10/ nish: 10/ unny, wai	6/20 6/20 rm (1	10 10 nid-7	70's)		FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Completion: 25.00 ft. BGS Station: 875,359.71N 2,515,298.97E
5	SAMPL	E.	Г	TEST	TING	r	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	y (ir		и	(%)	lb/ft ³)	ре	Quadr Towns	hip: East Fork	$\Psi = 20.00$ - While drilling $\Psi = 19.90$ - Upon completion
ber	v / Tc cover		s/6i alue	ture (Den. (sf) re T _{y,}	Section	11, Tier 7N; Range 3W	<u>\[\]</u> =
Numl	Reco % Re	Type	<i>Вlow</i> N - V RQD	Moist	Dry I	Qu (t) Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	20/24 83%	ss	4-3 5-6 N=8	16			2		
2A	24/24 100%	ss	3-4 7-7 N=11	15			4	FILL - Brown (10YR4/3) with 20% dark gray (N5/1) mottles, moist, firm, silty CLAY with trace sand and gravel.	628
3A	24/24 100%	ss	3-5 10-9 N=15	19			6		
4A	24/24 100%	ss	3-6 9-11 N=15	17				FILL - Dark grayish brown (10YR4/2), slightly moist firm, clayey SILT with trace sand and gravel.	624
		4					8	Gray (10YR5/1) with 10% dark grayish brown (10YR4. mottles, moist, firm, silty CLAY with trace sand.	
5A	24/24	V ss	3-4 5-7	15				Dark gray (10YR4/1), moist, firm, silty CLAY with tra sand, trace roots.	ce 622
	100%	A	N=9				10	Gray (10YR5/1) with 15% yellowish brown (10YR5/6 mottles, moist, firm, silty CLAY with trace sand.	
6A	22/24 92%	ss	2-2 4-6 N=6	26			12	Gray (10YR5/1) with 15% yellowish brown (10YR5/6 mottles, moist, firm, silty CLAY with trace sand and gravel.	
7A	19/24 79%	ss	<i>1-3</i> <i>3-5</i> N=6	23			14	Gray (10YR5/1) with 5% dark yellowish brown (10YR4 mottles, moist to very moist, firm, silty CLAY with san	
8A	24/24 100%	ss	2-2 2-4 N=4	26			16	Gray (10YR5/1) with 5% dark yellowish brown (10YR4 mottles, moist, soft, silty CLAY with sand and trace gravel.	H/6) 616
9A	24/24 100%	ss	1-1 2-3 N=3	24			18	Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with sand a trace gravel.	nd
10A	20/24 83%	ss	1-1 2-2 N=3	24				Gray (N6/1), very moist, very soft, silty, very fine- to	
10B		\square		31			¥ ₂₀ ∃	fine-grained sandy CLAY with trace gravel.	
NC	DIE(S):	G210	U installe	d in l	boreł	iole.			

FI	EL	DI	BOR	IN	IG	L	DG		HANSON	
	CLIEN	T: A	EG Coffe	en P	ower	r Statio	n	CONTRACTOR: Layne-Western Co		
	Sit	te: C	CB Mana	gem	ent F	acility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G210	
]	Locatio	n: Co	offeen, Ill	inois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G210	
	Proje	et: 05	5S3004A						Surface Elev: 630.48 ft. MSL	
	DATE	S: St	tart: 10/6	5/201	10			FIELD STAFF: Driller: D. Mahurin	Completion: 25.00 ft. BGS	
		Fin	nish: 10/6	5/20	10			Helper: J. Litsch/D. Smail	Station: 875,359.71N	
WE	ATHE	R: Si	unny, war	m (r	nid-7	70's)		Eng/Geo: S. Simpson	2,515,298.97E	
S	AMPL	E	Т	EST	TING	Ì	TOPOCR	APHIC MAP INFORMATION.	WATER LEVEL INFORMATION:	
er	/ Total (in) overy		/ 6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	Quadra Quadra Towns Section	angle: Coffeen, IL hip: East Fork 11, Tier 7N; Range 3W	$\Psi = 20.00 - While drilling$ $\Psi = 19.90 - Upon completion$ $\Psi = 19.90 - Upon completion$	
quinN	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks	
11A	14/24 58%	ss	woh-5 6-18 N=11	12				Yellowish brown (10YR5/4), moist, medium dense, si fine- to coarse-grained SAND with trace gravel. Gray (N6/1), moist, medium dense, silty, very fine - fine-grained SAND. Yellowish brown (10YR5/4), slightly moist, hard, clay	ilty, to vev	
12A	19/24 79%	ss	7-18 26-36 N=44	14				SILT with sand and gravel. Brown (10YR5/3), very moist, dense, very fine- to fine-grained sandy SILT with trace gravel.	608	
13A	$13A \begin{vmatrix} 10/12 \\ 83\% \end{vmatrix} \times ss \begin{vmatrix} 27-50/6'' \\ 9 \end{vmatrix}$						24	ith 606		
	End of Boring = 25.0 ft. BGS									

NOTE(S): G210 installed in borehole.

Page 2 of 2

F	[EL]	DI	BOR	IN	IG	L(DG				
	CLIEN Sit Location Project	T: A ne: C n: C	EG Coffe CB Mana offeen, II	een P agem linois	ower ent F	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G211 Well ID: G211 Surface Elev: G30 31 ft MSI		
wi	DATE	S: St Fir R: St	tart: 10/ hish: 10/ hiny, war	11/20 11/20 rm (1	010 010 0-70	's)		FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Completion: 24.00 ft. BGS Station: 875,424.49N 2,515,449.06E		
	SAMPL	E	Г	TEST		; 	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:		
	otal (i 7		in	(%)	(lb/ft ³	be	Towns	hip: East Fork	$\mathbf{Y} = 20.00$ - while drilling $\mathbf{Y} = 20.60$ - Upon completion		
ber	v / Td		s / 6 i /alue	ture (Den. (sf_{Ty}	Section	11, Tier 7N; Range 3W	<u>\Z</u> =		
Num	Recc % Re	Type	Blow N - V RQI	Mois	Dry]	Qu (1 Failt	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks		
1A	20/24 83%	ss	5-3 4-5 N=7	23			2	FILL - Brown (10YR4/3) with 20% dark gray (10YR4 mottles, moist, firm, silty CLAY with trace sand and gravel.	V(1) d - 630 - 630 - 630 - 630 - 630 - 630		
2A	20/24 83%	ss	3-5 5-8 N=10	17			4				
3A	24/24 100%	ss	2-3 5-7 N=8	24			6	FILL - Dark gray (10YR4/1) with 20% brown (10YR4 and 5% dark yellowish brown (10YR4/6) mottles, mo firm, silty CLAY with trace sand and gravel.	V(3) ist, 626		
4A	24/24 100%	ss	3-5 7-9 N=12	29			•				
5A	24/24 100%	ss	2-2 3-5 N=5	31			0	Dark gray (10YR4/1) with 20% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY, slight tra roots.	n ace 622		
6A	17/24 71%	ss	1-2 4-4 N=6	19			10	Dark gray (10YR4/1) with 20% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with tra- sand.	620 m ce		
7A	24/24 100%	ss	1-2 2-4 N=4	22			14	Dark gray (10YR4/1) with 20% dark yellowish brow (10YR4/6) mottles moist soft silty CLAY with trace	n sand		
8A	24/24 100%	ss	<i>1-1</i> <i>3-2</i> N=4	28			16	and gravel.	614		
9A	22/24 92%	ss	1-1 1-2 N=2	19			18	Dark yellowish brown (10YR4/6) with 30% gray (10YR5/1) mottles, very moist, soft, very fine- to fine-grained sandy CLAY with trace gravel.	612		
10A	19/24 79%	ss	1-4 5-11 N=9	13			¥ 20	Dark yellowish brown (10YR4/4) with 15% grayish br (10YR5/2) mottles, moist, firm, clayey SILT with sand gravel.	own and i i i i i i i i i i i i i i i i i i i		
NC	NOTE(S): G211 installed in borehole.										

FI	EL	DI	BOR	IN	IG	L	DG		HANSON
	CLIEN	Г: А	EG Coffe	een P	ower	r Statio	n	CONTRACTOR: Layne-Western Co	
	Sit	e: C	CB Mana	igem	ent F	acility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G211
	Locatio	n: Co	offeen, Il	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G211
	Projec	t: 05	5S3004A						Surface Elev: 630.31 ft. MSL
	DATES	S: St	tart: 10/	11/20	010			FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS
		Fin	nish: 10/	11/20	010			Helper: J. Litsch/D. Smail	Station: 875,424.49N
WE	ATHEF	R: Si	unny, wai	rm (l	o-70'	s)		Eng/Geo: S. Simpson	2,515,449.06E
S	SAMPL	E	T	EST	ING	ł	TOPOGR	APHIC MAP INFORMATION.	WATER LEVEL INFORMATION.
er	/ Total (in) overy		alue alue ure (%) en. ($1b/ft^3$) f) f) f)				Quadr Quadr Towns Section	angle: Coffeen, IL ship: East Fork a 11, Tier 7N; Range 3W	Ψ = 20.00 - While drilling Ψ = 20.60 - Upon completion $\overline{\Psi}$ =
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry Do	Qu (ts1 Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
11A	18/24 75%	ss	7-17 23-21 N=40	10			¥	Grayish brown (10YR5/2), slightly moist, hard, very s CLAY with sand and gravel.	ilty
12A	24/24 100%	ss	4-14 15-17 N=29	13			22	Dark gray (10YR4/1), slightly moist, hard, very silt CLAY with sand and gravel.	608
1								End of Borling – 24.0 It. BGS	

F	EL	DI	BOR	IN	IG	L() G						
CLIENT: AEG Coffeen Power Statio Site: CCB Management Facility Location: Coffeen, Illinois							n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G212 Well ID: G212 Surface Flay: 630 59 ft MSI				
Project: 0535004A DATES: Start: 10/11/2010 Finish: 10/11/2010 WEATHER: Sunny, warm (lo-80's)						's)		FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Surface Efect 050.35 ft MSL Completion: 24.00 ft BGS Il Station: 875,486.50N 2,515,583.03E				
SAMPLE TESTING						; 	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:				
	al (ir		-	(0)	b/ft³)	e	Quadr Towns	angle: Coffeen, IL hip: East Fork	$\mathbf{Y} = 19.00$ - While drilling $\mathbf{Y} = 20.72$ - Upon completion				
er	Tor Tover		/6 ii alue	ure (⁰	en. (e^{Ty_l}	Section 11, Tier 7N; Range 3W		$\overline{\Sigma} =$				
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
1A	24/24 100%	ss	4-3 3-6 N=6	17				FILL - Brown (10YR4/3), slightly moist, firm, silty CI with trace sand and gravel.	LAY 630				
2A	24/24 100%	ss	2-3 4-5 N=7	21			2	FILL - Dark gray (10YR4/1) with 20% brown (10YR4 and 10% dark yellowish brown (10YR4/6) mottles, m firm, silty CLAY with trace sand.	t4/3) noist,628				
3A	24/24 100%	ss	2-5 6-7 N=11	13			4 	FILL - Brown (10YR4/3) with 15% dark gray (10YR4 and 10% dark yellowish brown (10YR4/6) mottles, m firm, silty CLAY with trace sand and gravel.	24/1) noist,				
4A	24/24 100%	ss	2-5 7-10 N=12	15			8		624				
5A	24/24 100%	ss	2-2 4-7 N=6	29			10	Dark gray (10YR4/1) with 10% dark yellowish brov (10YR4/6) mottles, moist, firm, silty CLAY, slight tr roots.	wn race 622				
6A	18/24 75%	ss	2-3 4-6 N=7	23				Dark gray (10YR4/1) with 10% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with tra sand.	wn ace 620				
7A	17/24 71%	ss	1-2 2-2 N=4	25			12		618				
8A	24/24 100%	ss	woh-1 2-3 N=3	27			16	Dark gray (10YR4/1) with 15% dark yellowish brow (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel.	wn e sand				
9A	22/24 92%	ss	1-1 2-2 N=3	25			10		614				
10A	24/24 100%	ss	woh-woh 1-2	19			¥	Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, very moist, very soft, silty, very f to fine-grained sandy CLAY with trace gravel.	fine- 612				
10B NC) DTE(S):	G212	 2 installed	22 1 in l	 boreł	nole.	20	Gray (10YK5/1), loose, wet, silty, very fine- to					
FI	FIELD BORING LOG												
-------	------------------------------	--------	------------------------	---------	---------------------------	-------------------	----------------------------------------------------------------------------------------------------	--------------------------------------------------------------------------------------------	--	--	--	--	--
	CLIEN	T: A	EG Coffe	een P	owe	r Statio	n CONTRACTOR: Layne-Western Co						
	Sit	te: Co	CB Mana	igem	ent F	acility	Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G212					
	Locatio	n: Co	offeen, Ill	linois	s		Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G212					
	Proje	et: 05	5S3004A					Surface Elev: 630.59 ft. MSL					
	DATE	S: St	art: 10/	11/20	010		FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS					
		Fin	ish: 10/	11/20	010		Helper: J. Litsch/D. Smail	Station: 875,486.50N					
WE	ATHE	R: Su	inny, wai	m (l	0-80	s)	Eng/Geo: S. Simpson	2,515,583.03E					
S	SAMPL	E	Т	EST	TING	j I	TOPOGRAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:					
er	/ Total (in) overy		/ 6 in lue	ire (%)	en. (lb/ft ³)	e Type	Quadrangle: Coffeen, IL Township: East Fork Section 11, Tier 7N; Range 3W	Ψ = 19.00 - While drilling Ψ = 20.72 - Upon completion $\overline{\Psi}$ =					
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry De	Qu (ts1 Failur	Depth Lithologic ft. BGS Description	Borehole Elevation Detail ft. MSL Remarks					
11A	7/24 29%	ss	1-6 10-22 N=16	19			■ Medium-grained SAND. Brown (10YR5/3), moist, medium dense, SILT with transmission and gravel.	ace 610					
12A	20/24 83%	ss	5-21 18-27 N=39	12			22 Dark gray (10YR4/1), slightly moist, hard, very silty CLAY with sand and gravel.	608					
	End of Boring = 24.0 ft. BGS												

F	[EL]	DI	BOR	IN	IG	L(DG					
	CLIEN Sit Locatio Projec	T: A te: C on: C ct: 05	EG Coffe CB Mana offeen, Ill 5S3004A	en P Igem linois	ower ent F	r Statio Facility	n	CONTRACTOR: Layne-Western CoRig mfg/model: CME-750 ATV DrillDrilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G213 Well ID: G213 Surface Elev: 630.34 ft. MSL			
WI	DATE EATHE	S: St Fir R: Pa	tart: 10/2 nish: 10/2 artly cloud	12/20 12/20 dy, n	010 010 nild ((mid-50)'s)	FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Completion: 24.00 ft. BGS Station: 875,544.37N 2,515,723.51E			
5	SAMPL	E	T	EST	INC	Ĵ	TOPOGR	APHIC MAP INFORMATION.	WATED I EVEL INFORMATION:			
	(in)	(III)					Quadra	ngle: Coffeen, IL	$\mathbf{\Psi} = 20.00$ - While drilling			
	Fotal <i>ery</i>		e e	(%)	(lb/	Jype	Townsl	hip: East Fork	$\underline{\Psi}$ = 19.92 - Upon completion ∇ =			
lber	_ / AC	0	vs/6 Valu D	sture	Den	tsf) ure I	D		<u>*</u>			
Nun	Reco % R	Typ	Blov N - J RQI	Moi	Dry	Qu (Fail	ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	23/24 96%	ss	4-3 4-5 N-7	15					630			
		$\left(\right)$	11-7				2-		628			
2A	22/24 92%	ss	2-4 6-8 N=10	21								
		(4		626			
3A	22/24 92%	ss	2-4 7-8 N=11	17				FILL - Brown (10YR4/3) with 10% dark gray (10YR4 and 5% dark yellowish brown (10YR4/6) mottles, mo silty CLAY with trace sand and gravel	/1) ist,			
		A					6-					
4A	22/24	V	2-4	16								
-12 1	92%	A ss	4-8 N=8									
		$\left(\right)$					8-					
5A	20/24	V ss	1-3 6-6	12								
	83%	\bigwedge	N=9									
		(10					
6A	20/24	ss	2-2 5-7	24				(10YR4/6) mottles, moist, firm, silty CLAY with trac sand trace roots	n ce			
	0570	\wedge	N=7					Gray (10YR5/1) with 10% dark yellowish brown				
		\forall					12	(10 Y K4/6) motues, moist, firm, clayey SL1, trace roo	618			
7A	20/24 83%	ss	2-3 3-5	19				Dark gray (10YR4/1) with 3% dark yellowish brown (10YR4/6) mottles, moist, firm, silty CLAY with trade	n ce			
		\bigwedge	IN=0				14	sand and gravel.				
		∇							616			
8A	22/24 92%	ss	1-2 2-3 N=4	24				Dark gray (10YR4/1) with 10% dark yellowish brow	n .			
		\square					16	(10YK4/6) mottles, moist, soft, silty CLAY with trace s and gravel.	sand			
		V	woh I						614			
9A	24/24 100%	ss	2-2 N=3	24				Gray (10YR5/1) with 25% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace s and gravel.	sand			
10.1	10/24	\forall	woh-woł				18	Gray (10YR5/1) with 20% dark yellowish brown (10YR4/6) mottles, moist, soft, very fine- to fine-grain sandy CLAY with trace gravel.	ned 612			
10A	75%	ss	1-2	24				Dark yellowish brown (10YR4/6), very moist, soft, san	ndy			
_		\square					I¥ ₂₀ ∃	CLAT with trace gravel.				
NC	DTE(S):	G21	3 installed	1 in l	boreł	nole.						

FI	EL	DI	BOR	IN	IG	L(DG		HANSON	
	CLIEN	T: A	EG Coffe	en P	ower	r Statio	n	CONTRACTOR: Layne-Western Co		
	Sit	te: CO	CB Mana	igem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G213	
	Locatio	n: Co	offeen, Ill	linoi	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G213	
	Proje	et: 05	5S3004A						Surface Elev: 630.34 ft. MSL	
	DATE	S: St	art: 10/	12/20	010		FIELD STAFF: Driller: D. Mahurin		Completion: 24.00 ft. BGS	
		Fin	ish: 10/	12/2	010			Helper: J. Litsch/D. Smail	Station: 875,544.37N	
WE	ATHE	R: Pa	artly cloud	dy, n	nild ((mid-50)'s)	Eng/Geo: S. Simpson	2,515,723.51E	
S	SAMPL	E	Т	EST	ING	Ĵ	TOPOGRA	PHIC MAP INFORMATION.	WATER LEVEL INFORMATION.	
ır	/ Total (in) wery		/ 6 in lue	re (%)	en. (lb/ft ³)) e Type	Quadrar Townshi Section	ngle: Coffeen, IL p: East Fork 11, Tier 7N; Range 3W	$\underline{\Psi} = 20.00$ - While drilling $\underline{\Psi} = 19.92$ - Upon completion $\underline{\nabla} =$	
Numbe	Recov % Reco	Type	Blows / N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks	
11A	18/24 75%	ss	<i>1-1</i> <i>1-2</i> N=2	18				Dark yellowish brown (10YR4/6), moist, soft, clayey, fine- to coarse-grained SAND with trace gravel. Dark vellowish brown (10YR4/6), moist, firm, very si	very 610	
12A	22/24 92%	ss	10-13 18-22 N=31	11			22	Dark gray (10YR4/1), slightly moist, hard, very silt CLAY with sand and gravel.	608 y	
	End of Boring = 24.0 ft. BGS									

F	FIELD BORING LOG											
	CLIEN Sit Location	Γ: Α α: C n: C n: C	EG Coffe CB Mana offeen, II	een P igem linois	Power ient F s	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G214 Well ID: G214			
WI	DATE:	S: St Fin R: St	tart: 10/ nish: 10/ nny, coo	14/20 14/20 d (lo-	010 010 -40's))		FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Completion: 24.00 ft. BGS Station: 875,668.02N 2,515,960.84E			
5	SAMPL	E	Т	EST	rine 	Ì	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	tal (in		~	(0)	b/ft ³)	e	Quadra Townsł	ingle: Coffeen, IL iip: East Fork	$\mathbf{\Psi} = $ Dry - While drilling $\mathbf{\Psi} = $ Dry - Upon completion			
er	/ To		/6 in alue	ure (9	en. (]	f) e Typ	Section	11, Tier 7N; Range 3W	$\overline{\Sigma} =$			
Numb	Recov % Rec	Type	Blows N - V; RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	24/24 100%	ss	6-7 7-9 N=14	15			2	FILL - Brown (10YR4/3) with 10% dark gray (10YR4 and 10% dark yellowish brown (10YR4/6) mottles, mo firm, silty CLAY with trace sand and gravel.	/1) pist, 630 - 628			
2A	24/24 100%	ss	3-3 6-5 N=9	22			4	FILL - Dark gray (10YR4/1) with 15% dark yellowis brown (10YR4/6) mottles, moist, firm, silty CLAY w trace sand and gravel.				
3A	24/24 100%	ss	3-4 6-8 N=10	18								
4A	24/24 100%	ss	3-4 7-10 N=11	17			6 8	FILL - Brown (10YR4/3) with 10% dark yellowish bro (10YR4/6) and 5% dark gray (10YR4/1) mottles, mo firm, silty CLAY with trace sand and gravel.	wn ist, 624			
5A	24/24 100%	ss	3-2 4-5 N=6	19			10	Grayish brown (10YR5/2) with 15% dark gray (10YR- and 10% dark yellowish brown (10YR4/6) mottles, mo firm, clayey SILT with trace sand and gravel.	4/1) bist, 622 622 622			
6A	24/24 100%	ss	2-3 4-7 N=7	24			12	Brown (10YR4/3) with 15% dark gray (10YR4/1) ar 10% dark yellowish brown (10YR4/6) mottles, mois firm, silty CLAY with trace sand and gravel.	nd t,			
7A 7B	24/24 100%	ss	2-3 4-6 N=7	22 16				Gray (10YR6/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY.	618			
8A	24/24 100%	ss	woh-2 3-4 N=5	22			14	Dark gray (10YR4/1) with 10% dark yellowish brow (10YR4/6) mottles, moist, firm, silty CLAY with tra- sand and gravel.	n 616			
9A	22/24 92%	ss	<i>1-2</i> <i>2-3</i> N=4	21			16	Dark gray (10YR4/1) with 10% dark yellowish brow (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel. Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel	n 614 sand $ 614$			
10A	22/24 92%	ss	woh-2 2-2 N=4	15				und graver.				
10B			 1 in 11	21			20					
)1E(S):	G214	+ installe	unt	oorer	ioie.						

FIELD	FIELD BORING LOG												
CLIENT: A	AEG Coffe	en P	owe	r Statio	n CONTRACTOR: Layne-Western Co								
Site: (CCB Mana	igem	ent I	Facility	Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G214							
Location: (Coffeen, Ill	inoi	s		Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G214							
Project: (5S3004A					Surface Elev: 630.39 ft. MSL							
DATES: S	start: 10/1	14/20	010		FIELD STAFF: Driller: D. Mahurin	Completion: 24.00 ft. BGS							
Fi	nish: 10/	14/2	010		Helper: J. Litsch/D. Smail	il Station: 875,668.02N							
WEATHER: S	Sunny, coo	l (lo	-40's)	Eng/Geo: S. Simpson 2,515,960.								
SAMPLE	Т	EST	TINC	Ĵ	TOPOGRAPHIC MAP INFORMATION.	WATER LEVEL INFORMATION							
er / Total (in) overy	/6 in Ilue	ıre (%)	en. (lb/ft ³)	f) e Type	Quadrangle: Coffeen, IL Township: East Fork Section 11, Tier 7N; Range 3W	$\Psi = \text{Dry - While drilling}$ $\Psi = \text{Dry - Upon completion}$ $\Psi = $							
Numbe Recov % Rec Type	Blows N - Va RQD	Moistu	Dry Do	Qu (ts1 Failur	Depth Lithologic ft. BGS Description	Borehole Elevation Detail ft. MSL Remarks							
11A 20/24 11B 23%	woh-woh 3-12	24 14			Gray (10YR5/1), wet, loose, silty, very fine- to medium-grained SAND with trace gravel and clay seams. [Continued from previous page] Dark yellowish brown (10YR4/6), moist, firm, very 22CLAY with trace sand and gravel	ey silty							
$12A \begin{array}{ c c c c c c c c c c c c c c c c c c c$					24 Dark gray (10YR4/1), slightly moist, hard, very si CLAY with trace sand and gravel.	lty							

FI	FIELD BORING LOG											
	CLIEN Sit Location Projec	T: Al ne: Co n: Co n: Co	EG Coffe CB Mana offeen, Il 5S3004A	een P agem linois	Power Ient F s	r Statio Facility	n	CONTRACTOR: Layne-Western CoRig mfg/model: CME-750 ATV DrillDrilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G215 Well ID: G215 Surface Elev: 630.48 ft. MSL			
	DATE	S: St	tart: 10/	13/20	010			FIELD STAFF: Driller: D. Mahurin	Completion: 24.31 ft. BGS			
WF	ATHEI	R: Si	unny, wai	rm, v	vindy	/ (hi-60	's)	Eng/Geo: S. Simpson	2,515,971.55E			
S	AMPL	E	Г	EST	ring	r I	TOPOGRA	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	al (in			0	b/ft ³)	6	Quadra Townsh	ngle: Coffeen, IL in: East Fork	\mathbf{Y} = Dry - While drilling \mathbf{V} = Dry - Upon completion			
er	/ Tot overy		/ 6 in lue	ure (%	en. (Il	f) e Typ	Section	11, Tier 7N; Range 3W	$\underline{\nabla} = 22.52 - 10/14/10$			
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (ts) Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
		$\langle $	5.2						630			
1A	23/24 96%	ss	3-3 3-5 N=6	18								
	l	Δ					2					
		\bigvee	2.2					EII I Drown (10VD 4/2) with 200/ dark grov (10VD)	628			
2A	19/24 79%	ss	5-6 N=8	17				and 10% dark yellowish brown (10YR4/6) mottles, mo	pist,			
		(inin, sity CLAT with trace said and graver.				
	ſ	$\overline{\mathbf{V}}$					4		626			
3A	20/24 83%	ss	2-3 7-7	13								
		(N-10									
	Ţ	$\overline{\mathbf{V}}$					0	FILL - Dark gravish brown (10VR4/2) moist firm si	 hy 624			
4A	23/24 96%	ss	3-6 6-7 N=12	16				CLAY with trace sand and gravel.				
4B	l	\square	IN-12	27			8	FILL - Gray (10YR5/1) with 15% dark yellowish bro (10YR4/6) mottles moist firm silty CLAY with tra	wn — — — — — — — — — — — — — — — — — — —			
		\bigvee	3.3					sand. Very dark gray (10YR3/1), moist, firm, silty CLAY w	ith / // 622			
5A	20/24 83%	ss	3-5 N=6	20				trace sand, trace roots.				
	ļ	Δ	1.0				10					
	1	\bigvee	2.2					Dark gray (10YR4/1) with 30% dark yellowish brow (10YR4/6) moist, firm, silty CLAY with trace sand				
6A	13/24 54%	ss	3-5 N=5	24								
	l	\square					12					
		\backslash						Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles moiet firm clayer SII T with tra	618			
7A	19/24 79%	ss	$\begin{array}{c} 2-3 \\ 4-6 \\ N=7 \end{array}$	17								
		\wedge	1, 7				14	Dark gray (10VP 4/1) moist firm aloury SILT with the				
		$\langle \rangle$					14	sand.				
8A	20/24 <i>83%</i>	ss	$\begin{array}{c c} 2-3 \\ 4-5 \\ N-7 \end{array}$	19								
		(19-7									
	ţ	$\overline{\mathbf{V}}$					16	Dark gray (10YR4/1) with 30% Dark vellowish brow	/n 614			
9A	22/24 92%	ss	$\begin{vmatrix} 1-3\\ 3-4 \end{vmatrix}$	19				(10YR4/6) mottles, moist, firm, silty ĆLAY with tra sand and gravel.	ce			
		\wedge	N=0									
		$\overline{\mathbf{V}}$					18		612			
	24/24 100%	ss	$ woh-1 \\ 2-2 \\ N=2$					Dark gray (10YR4/1) with 30% Dark yellowish brow	m l			
10A		\wedge	N=3	17				(10Y K4/0) mottles, moist, soft, sandy CLAY with tra gravel.				
NO	TE(S):	G21:	5 installe	d in l	boreh	iole.	20 —					

FI	ELI	DI	BOR	IN	JG	L C	DG		HANSON
	CLIEN	Г: А	EG Coffe	een P	ower	r Statio	n	CONTRACTOR: Layne-Western Co	
	Sit	e: C	CB Mana	igem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G215
	Locatio	n: Co	offeen, Ill	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G215
	Projec	et: 05	5S3004A						Surface Elev: 630.48 ft. MSL
	DATE	S: St	art: 10/	13/20	010			FIELD STAFF: Driller: D. Mahurin	Completion: 24.31 ft. BGS
		Fin	nish: 10/	13/20	010			Helper: J. Litsch/D. Smail	Station: 875,810.19N
WE	ATHE	R: Si	unny, wai	m, v	vindy	/ (hi-60)'s)	Eng/Geo: S. Simpson	2,515,971.55E
S	AMPL	Е	Т	EST	TING	Ť	TOPOGRA	PHIC MAP INFORMATION.	WATER LEVEL INFORMATION.
	in)				<u>_</u>		Ouadrai	ngle: Coffeen II	$\mathbf{\nabla} = \mathbf{D}\mathbf{r}\mathbf{v}$ - While drilling
	al			0	b/ft	e	Townshi	n: East Fork	$\nabla = Dry - Upon completion$
	rot		6 in Ie	6	[]	T_{YD}	Section	1. Tier 7N: Range 3W	$\nabla = 22.52 - 10/14/10$
ber	v /	•	's∕ ∕alı	tur	Der	tsf) ure		, , , ,	-
Ium	é Ré	ype	10 10	lois	_V	ou (aih	Depth ft BCS	Lithologic	Borehole Elevation Detail ft MSI Remarks
Z	N %	<u> </u>	R Z R	2		0r	n. 503	Description	
		$\backslash /$						Dark yellowish brown (10YR4/6), moist, medium der	ise, 610
11A	20/24	V	2-4	17			=	clayey SILT with sand and trace gravel.	
	83%	∫ ^{ss}	4-4 N=8	1,				very fine- to fine-grained SAND.	
		Λ						Dark yellowish brown (10YR4/6) with 30% dark gra	IV III
	ŧ	\square					22	(10YR4/1) mottles, moist, firm, sandy CLAY with tr	ace
		$\backslash /$						gravel. Gravish brown (10VP5/2) slightly moist vory firm y	- 608
12A	24/24	V	7-11	11				silty CLAY with sand and gravel.	
	100%	1 55	N=28					, , , , , , , , , , , , , , , , , , , ,	
		/\						Dark grav (10YR4/1), slightly moist, hard very silt	
12B	0/4			9			24 –	CLAY with sand and gravel.	
	0%		I	I	1	I		End of Boring = 24.3 ft. BGS	

F]	FIELD BORING LOG											
	CLIEN Sif	T: A	EG Coffe CB Mana	en P gem	ower ent F	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G216			
	Locatio	n: Co	offeen, Ill	linois	5	activity		Drilling Method: 4¼" HSA w/SS samplers	Well ID: G216			
	Projec DATE	st: 05 S: St	83004A art: 10/1	13/20	010			FIELD STAFF: Driller: D. Mahurin	Completion: 26.00 ft. BGS			
		Fin	ish: 10/1	13/2	010			Helper: J. Litsch/D. Smail	Station: 875,976.05N			
WI	EATHEI SAMPL	к: Ра Е	rtly cloud	iy, n EST	nild,	windy	(lo-60's)	Eng/Geo: S. Simpson	2,515,968.53E			
	(ii						TOPOGR Ouadr	APHIC MAP INFORMATION: angle: Coffeen II.	WATER LEVEL INFORMATION: $\mathbf{\nabla} = 21.00$ - While drilling			
	otal (<i>ry</i>		in	(%)	(lb/fi	pe	Towns	hip: East Fork	$\underline{\Psi}$ = Dry - Upon completion			
ber	v / T		ss / 6 /alue	ture	Den.	tsf_{T}	Section	1 11, Tier 7N; Range 3W	<u>V</u> =			
Num	Recc % Re	Type	<i>Вlo</i> и N - V RQI	Mois	Dry	Qu (Faili	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
		\bigvee	3-2						630			
1.	24/24	ss	2-4 N=4	22								
IA		()		22								
		∇						FILL - Brown (10YR4/3) with 20% dark gray (10YR4	(6) - 628			
2A	19/24	ss	2-2 5-4	28				firm, silty CLAY with trace sand and gravel.				
	/ 3/0	Λ	N=7									
		$\left(\right)$					4		626			
	24/24	V	2-2									
3A	100%	Å ss	4-5 N=6	19				- — — — — — — — — — — — — — — — — — — —				
		Д					6	and 10% dark yellowish brown (10YR4/6) mottles, mo	pist,			
		V	24						624			
	22/24 92%	ss	6-6					FILL - Dark grayish brown (10YR4/2) with 10% dar	k 🛞 🖳			
4A		\bigwedge	IN-10	19				yellowish brown (10YR4/6) mottles, moist, firm, silt CLAY with trace sand and gravel.				
		∇					8		622			
5A	20/24	ss	2-3 3-6	18								
	83%	Λ	N=6									
		$\left(\right)$					10-	Dark gray (10YR4/1) with 40% gray (10YR6/1) and 1 dark vellowish brown (10YR4/6) mottles, moist, firm	0%			
64	18/24	V	2-3	17				clayey SILT with trace sand.				
0A	75%	Å ss	3-4 N=6	1/								
		Д					12					
		V	1.2						- 618			
7A	16/24 67%	ss	3-4 N-5	20				Gray (10YR5/1) with 20% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace s	sand			
		\bigwedge	IN-3					and gravel.				
		\forall							616			
8A	22/24	V ss	2-2 5-5	20				Very dark gray (10YR3/1), moist, firm, silty CLAY w trace sand.	rith			
	92%	Λ	N=7									
		$\left(\right)$					16-					
	23/24	V	woh-2	10				Gray (10YR5/1) with 10% dark yellowish brown (10YR4/6) motiles maint firm aity CLAN with tra				
УA	96%	ss	3-3 N=5	18				sand and gravel.				
		Д										
		\bigvee						Gray (10YR5/1) with 10% dark yellowish brown	612			
10A	24/24 100%	ss	won-woh 1-2	17				(10YR4/6) mottles, moist, firm, sandy CLAY with tra gravel.	ice			
		\wedge										
NC) DTE(S):	山 G210	i 6 installec	l in l	ı boreh	nole.	20-		KYYYZEI EKYYY			
I I												

1

FI	FIELD BORING LOG											
	CLIEN	T: Al	EG Coffe	en P	ower	r Statio	n	CONTRACTOR: Layne-Western Co				
	Sit	e: CO	CB Mana	igem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G216			
	Locatio	n: Co	offeen, Ill	linois	5			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G216			
	Projec	et: 05	S3004A						Surface Elev: 630.28 ft. MSL			
	DATE	S: St	art: 10/	13/20	010			FIELD STAFF: Driller: D. Mahurin	Completion: 26.00 ft. BGS			
		Fin	ish: 10/	13/20	010			Helper: J. Litsch/D. Smail	Station: 875,976.05N			
WE	EATHEI	R: Pa	rtly cloud	dy, n	nild,	windy	(lo-60's)	Eng/Geo: S. Simpson	2,515,968.53E			
S	SAMPL	E	Т	EST	ING	Ĵ	TOPOCDAL		WATED LEVEL INFORMATION			
							TOPOGRA	PHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	1 (i				(H3		Quadran	gie: Coffeen, IL	$\underline{\Psi} = 21.00$ - while drilling			
	ota		in	8	e e	ype	I ownsnij	D: East Fork	$\underline{\Psi}$ = Dry - Upon completion			
er	1 / J		/ 6 alue	ure	en.	f) .e T	Section 1	1, Her /N; Range 3W	<u> </u>			
qui	Rec	pe	SWC D	oist	^Z D	t (ts ilun	Depth	Lithologic	Borehole Elevation			
N	Re %	Ту	\mathbf{R}^{Bl}	Ž	D	Qu	ft. BGS	Description	Detail ft. MSL Remarks			
		$\langle $						Dark yellowish brown (10YR4/6) and 40% gray (10YR5/1) mottles, moist, soft, sandy CLAY with tra	ace 610			
11A	20/24	V	2-9 9-7	14				gravel.				
	83%	Λ	N=18					Light brownish gray (10YR6/2) wet medium dense y				
		()						fine- to fine-grained sandy SILT with trace gravel.				
		\rightarrow						Yellowish brown (10YR5/6), wet, medium dense, sil	ty, / + + = + +			
		\mathbf{V}					<u>∃</u> ``-	very fine-to medium-grained SAND.				
12A	22/24	V 55	3-10 21-25	19				fine-grained sandy SILT				
12B	92%	Λ	N=31	16			l 1,	Gray (10YR5/1), wet, medium dense, very fine- to	/討保護目標投資→-			
120		()		10			= `_	fine-grained sandy SILT				
		\rightarrow					24	Dark gray (10Y R4/1), wet, dense, silty, fine- to				
		\mathbb{N}	1 / 25					Coalse-granicu SAND with graver.				
13A	20/24	V ss	14-25	9				Dark gray (10YR4/1), slightly moist, hard, very silt	y L			
	83% N=							CLAY with sand and gravel.				
		(
	End of Boring = 26.0 ft. BGS											

NOTE(S): G216 installed in borehole.

F	FIELD BORING LOG											
	CLIEN Sit Locatio Proje	T: A te: C on: C	EG Coffe CB Mana offeen, II	een P agem linois	owe ent F	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	BOREHOLE ID: G217 Well ID: G217 Surface Flay: G30 67 ft MSI			
WI	DATE	S: Si Fir R: Si	tart: 10/ nish: 10/ unny, war	12/20 12/20 rm (1	010 010 0-80'	's)		FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: S. Simpson	Completion: 26.00 ft. MSL Station: 876,185.57N 2,515,963.02E			
	SAMPL	.E	Г	TEST			TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	$\frac{\gamma}{n}$ (ir $\frac{\gamma}{n}$ (b/ft ³)				lb/ft ³	ьe	Quadr Towns	hip: East Fork	$\underline{\Psi} = 23.00$ - while drilling $\underline{\Psi} = 24.82$ - Upon completion			
ber	v / To		s / 6 i /alue	ture (Den. (sf_{Ty}	Section	11, Tier 7N; Range 3W	$\overline{\Sigma} = 23.98 - 10/13/10$			
Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry J	Qu (t Failh	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	20/24 83%	ss	5-2 3-4 N=5	21			2	FILL - Brown (10YR4/3) with 10% dark gray (10YR4/3)	4/1)			
2A	19/24 79%	ss	2-3 5-6 N=8	28			4	and 5% dark yellowish brown (10YR4/6) mottles, mo firm, silty CLAY with trace sand and gravel.	oist, 628			
3A	19/24 79%	ss	2-3 6-7 N=9	14				FILL - Dark gray (10YR4/1) with 25% dark yellowis brown (10YR4/6) mottles, moist, firm, silty CLAY w trace sand and gravel.	sh hith			
4A	23/24 96%	ss	5-6 7-8 N=13	15			0	FILL - Brown (10YR4/3) with 10% dark gray (10YR4 and 10% dark yellowish brown (10YR4/6) mottles, mo firm, silty CLAY with trace sand and gravel.	4/1) oist, 624			
5A	20/24 83%	ss	3-5 7-6 N=12	13			8	FILL - Dark grayish brown (10YR4/2) with 5% dar yellowish brown (10YR4/6) slightly moist, firm, clay SILT with trace sand and gravel.	k ey 622			
6A	19/24 79%	SS SS	3-3 4-5 N=7	27			10	FILL - Very dark gray (10YR3/1), moist, firm, silty CI with trace sand and gravel. Dark gray (10YR4/1), moist, firm, silty CLAY with tra- sand and gravel. Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles moist firm silty CLAY with tra-	AY ace 			
7A	18/24 75%	ss	3-4 6-8 N=10	28			12	Dark grayish brown (10YR4/2) with 10% dark yellow brown (10YR4/6) mottles, moist, firm, silty CLAY tra-	rish ace 618			
8A	20/24 83%	ss	2-4 6-8 N=10	16			14	Dark gray (10YR4/1), moist, firm, silty CLAY with tr sand and gravel.	ace 616			
9A	19/24 79%	ss	2-3 4-5 N=7	26				Dark gray (10YR4/1) with 10% yellowish brown (10YR5/6) mottles, moist, firm, silty CLAY with tra- sand and gravel.	ce 614			
10A	19/24 79%		1-2 2-3 N=4	18			18- 20-	Gray (10YR5/1) with 5% dark yellowish brown (10YR mottles, moist, soft, silty CLAY with sand and trace gravel.	44/6) e 612			
	JIE(S):	G21	/ installe	u in l	oreh	iole.						

FIE	LD I	BOR	IN	JG	L)G				
CLI	ENT: A	EG Coffe	en P	ower	r Statio	n	CONTRACTOR: Layne-Western Co			
	Site: Co	CB Mana	igem	ent F	acility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G217		
Loca	tion: Co	offeen, Ill	linois	3			Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G217		
Pro	oject: 05	5S3004A						Surface Elev: 630.67 ft. MSL		
DA	TES: St	tart: 10/1	12/20)10			FIELD STAFF: Driller: D. Mahurin	Completion: 26.00 ft. BGS		
	Fir	nish: 10/	12/20	010			Helper:J. Litsch/D. SmailStation:876,185.57N			
WEATI	HER: SI	unny, war	rm (k	o-80'	's)		Eng/Geo: S. Simpson	2,515,963.02E		
SAM	PLE	Т	EST	ING	7	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:		
er / Total (in)	very	/ 6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	Quad Town Sectio	rangle: Coffeen, IL sship: East Fork on 11, Tier 7N; Range 3W	$\Psi = 23.00 - While drilling \Psi = 24.82 - Upon completion \Psi = 23.98 - 10/13/10$		
Numb Recov	% Kec Type	Blows N - Va RQD	Moistu	Dry D	Qu (tsi Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks		
22/2 92	24 % ss	woh-woh 1-2	18			22	Gray (10YR5/1) with 5% dark yellowish brown (10YR mottles, moist, soft, silty CLAY with sand and trace gravel. <u>[Continued from previous page]</u> Gray (10YR5/1) with 20% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with sand a trace gravel.	4/6) 		
10/2 42	24 1% ss	4-6 7-10 N=13	15			⊻ _	Yellowish brown (10 YR5/4), wet, medium dense, sil very fine- to coarse-grained SAND with trace gravel	ty,		
13A 22/2 92	24 19% ss	8-18 17-17 N=35	12			- 24 ▼	Dark gray (10YR4/1), slightly moist, hard, very silty CLAY with sand and gravel.	y 606		
						20	End of Boring = 26.0 ft BCS			

End of Boring = 26.0 ft. BGS

F	EL	DI	BOR	IN	IG	L(DG		HANSON			
	CLIEN Sit Locatio Projec DATE	Γ: Α n: C n: C st: 05 S: St	EG Coffe CB Mana offeen, Il 5S3004A tart: 10/	een P agem linois 12/20	Power ient F s 010	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: D. Mahurin	BOREHOLE ID: G218 Well ID: G218 Surface Elev: 630.64 ft. MSL Completion: 26.00 ft. BGS			
w	ATHE	Fir	nish: 10/	12/20	010	(1 - 70)	2)	Helper: J. Litsch/D. Smail	Station: 876,380.92N			
	SAMPL	к: га Е		uy, w TEST	ring	(10-70 G	s)		2,313,902.10E			
	(in)				t³)		TOPOGR Quadr	angle: Coffeen, IL	WATER LEVEL INFORMATION: $\mathbf{\Psi} = 24.00$ - While drilling			
	otal		in .	(%)	(lb/f	ype	Towns	hip: East Fork	Ψ = 24.76 - Upon completion			
lber	ло <i>со</i> риссе	Ð	vs / 6 Value D	sture	Den.	tsf) ure T	Section					
Nun	Reco % R	Typ	Blov N - J	Moi	Dry	Qu (Fail	ft. BGS	Description	Detail ft. MSL Remarks			
1A	18/24 75%	ss	4-1 2-1 N=3	20			2	FILL - Brown (10YR4/3) with 15% dark gray (10YR4 and 5% dark yellowish brown (10YR4/6) mottles, mo soft, silty CLAY with trace sand and gravel.	4/1) vist, 630			
2A	22/24 92%	ss	2-2 3-5 N=5	20								
3A	19/24 79%	ss	2-3 4-8 N=7	17			4-	FILL - Dark gray (10YR4/1) with 30% brown (10YR4 and 10% dark yellowish brown (10YR4/6) mottles, me firm, silty CLAY with trace sand and gravel.	4/3) pist, 626			
4A	22/24 92%	ss	2-5 6-8 N=11	14			6 1 8	FILL - Brown (10YR5/3) with 10% dark gray (10YR4 mottles, slightly moist, firm, clayey SILT with trace sa and gravel.	4/1) and			
5A	20/24 83%	ss	3-4 8-7 N=12	17				Dark gravish brown (10YR4/2) with 5% dark vellow	- 622			
6A	19/24	ss	2-2 3-5	19			10	brown (10YR4/6) mottles, moist, firm, clayey SILT w trace sand.	620			
6B	/ / / /	\bigwedge	N=5	25			12 -	Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, firm, silty CLAY with tra sand, slight trace roots.	ce			
7A	22/24 92%	ss	2-3 5-7 N=8	22				Dark gray (10VP4/1) with 15% dark vallowish brow				
8A	18/24 75%	ss	2-3 4-5 N=7	19				(10YR4/6) mottles, moist, firm, clayey SILT with tra sand.	616			
9A	24/24 100%	ss	2-2 2-4 N=4	19			18	Gray (10YR5/1) with 10% dark yellowish brown (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel.	sand			
10A	24/24 100%	ss	1-2 2-3 N=4	18			20		612			
NC	DTE(S):	G21	8 installe	d in l	boreh	nole.						

FI	ELI) I	BOR	IN	JG	L)G		C		NSON
	CLIENT	: Al	EG Coffe	en P	ower	Statio	n	CONTRACTOR: Layne-Western Co			
	Site	e: CO	CB Mana	gem	ent F	acility		Rig mfg/model: CME-750 ATV Drill	BOR	REHOLE ID: G2	18
	Location: Coffeen, Illinois					Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	Well ID: G218				
	Projec	t: 05	S3004A						S	Surface Elev: 63	0.64 ft. MSL
	DATES	S: St	art: 10/1	2/20	010			FIELD STAFF: Driller: D. Mahurin		Completion: 2	6.00 ft. BGS
		Fin	ish: 10/1	12/20	010			Helper: J. Litsch/D. Sma	il	Station: 8	76,380.92N
WE	ATHEF	R: Pa	rtly cloud	ły, w	/arm	(lo-70'	s)	Eng/Geo: S. Simpson		2,5	15,962.16E
S	SAMPLI	£	Т	EST	ING		TOPOGE	RAPHIC MAP INFORMATION:	WATER LEVEL I	NFORMATION	
	(iii) (iii)				Ouadı	angle: Coffeen IL	$\nabla = 24.00 - V$	While drilling			
$\frac{1}{2e}$ $\frac{1}{2e}$ $\frac{1}{2e}$ $\frac{1}{2e}$			е	Towns	ship: East Fork	$\mathbf{V} = 24.76$ - Upon completion					
	ver		6 ir ue	.e ()	ы. С	T_{YF}	Section	n 11, Tier 7N; Range 3W	$\overline{\Sigma} =$		
ıbeı	00 /	е	Val Val	stur	Dei	tsf) ure	Durth	T 24 - 1 - 2	Develop	El. d'an	
Nun	Rec % R	Typ	RQ	Moi	Dry	Qu (Fail	ft. BGS	Description	Detail	ft. MSL	Remarks
		T					=			F I	
11A	22/24 92%	ss	woh-woh woh-woh	16			22	Gray (10YR5/1) with 10% dark yellowish brow (10YR4/6) mottles, moist, very soft, clayey, very fin coarse-grained SAND with trace gravel.	n ne- to	610	
12A 12B	24/24 100%	ss	1-1 1-3 N=2	10 16				Yellowish brown, wet, loose, silty, very fine- to coarse-grained SAND with trace gravel. Dark gray (10YR4/1) with 10% dark yellowish bro (10YR4/6) mottles, moist, soft, silty CLAY with trace and gravel.	wn e sand	608	
13A	24/24 100%	ss	<i>1-5</i> <i>9-13</i> N=14	20			¥ 24	Gray (10YR5/1), wet, loose, silty, very fine- to coarse-grained SAND with trace gravel. Dark gray (10YR4/1), slightly moist, very firm, very CLAY with and and gravel	/ silty	606	
13B	$ _{13B} $ \square $ _{17} $ $ _{17} $						26	End of Boring = 26.0 ft. BGS			

F	EL CLIEN Sit	D I Г: А се: С	BOR EG Coffe CB Mana	een F	NG Power	F L (r Statio Facility	DG n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: T202					
Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 10/15/2010					Drilling Method: 4'4" HSA w/SS samplers Well ID: T202 Surface Elev: 626.22 ft. MSL FIELD STAFF: Driller: D. Mahurin Helner: L titsch/D. Smail Station:									
WB	EATHEI	Fin R: Si	usn: 10/ unny (mie	15/2 d-50'	010 's)			Eng/Geo: R. Hasenyager 2,514,895.01						
5	SAMPL	E	Г	TEST	TINC	Ĵ	TOPOGE	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:					
	al (in))/ft ³)	0)	Quadu	angle: Coffeen, IL hin: Fast Fork	$\mathbf{\Psi} = $ Dry - While drilling $\mathbf{\Psi} = 15.42$ - Upon completion					
5	/ Tota wery		6 in lue	re (%	n. (lt	Type	Section	1 11, Tier 7N; Range 3W	$\underline{\nabla}$ =					
Numbe	Recov	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks					
		\backslash						FILL - Yellowish brown (10YR5/6), slightly moist, has silty CLAY with trace sand and gravel.	ard, 626					
1A	23/24 96%	ss	5-7 N=9	27			2	Yellowish brown (10YR5/6) with 30% gray (10YR5 mottles, moist, hard, silty CLAY with slight trace sand gravel.	/1) and -624					
2A	24/24 100%	ss	3-4 6-6 N=10	25										
2B				26			4		622					
3A	22/24 92%	ss	2-4 5-6 N=9	22				Yellowish brown (10YR5/8) with 35% gray (10YR5 and 15% very dark brown (10YR2/2) mottles, moist, h silty CLAY with trace sand and slight trace gravel.	/1) hard,					
4A	23/24 96%	ss	<i>1-2</i> <i>3-3</i> N=5	21			8	Gray (10YR6/1) with 30% yellowish brown (10YR5 mottles, moist, soft, silty CLAY with trace sand and sh	/8) light 618					
5A	23/24 96%	ss	<i>1-1</i> <i>3-2</i> N=4	22			10	trace gravel.						
6A	24/24 100%	ss	<i>woh-1</i> 3-3 N=4	23				Gray (10YR6/1) with 15% yellowish brown (10YR5 mottles, very moist, soft, silty CLAY with sand and sl	/8) ight 					
6B		$\left(\right)$		21			12	Yellowish brown (10YR5/8) with 10% gray (10YR6 mottles, very moist, firm, silty CLAY with sand and the gravel.	/1) race 614					
7A	24/24 100%	ss	1-2 1-2	21				-						
7B		$\left(\right)$	IN=3	25			14	Gray (10YR6/1) with 10% brownish yellow (10YR6 mottles, very moist, soft, sandy CLAY with silt and sl trace gravel.	/6) ight = 612					
8A	24/24 100%	ss	woh-2 1-8 N=3	17			¥ 16	Gray (10YR5/1), wet, loose, very fine- to medium-gra SAND with trace gravel.	ined					
9A	24/24 100%	ss	6-15 20-24 N=35	9				Gray (10YR6/1), moist, hard, very silty CLAY with s and trace gravel.	and 610					
	1	<u> </u>	I	I	1	ļ	18 -	End of Boring = 18.0 ft. BGS						
NC	DTE(S):	T202	2 installed	d in l	ooreh	nole.								

FI	FIELD BORING L CLIENT: AEG Coffeen Power Stati Site: CCB Management Facilit Location: Coffeen, Illinois Project: 05S3004A DATES: Stort: 2/26/2008						OG on	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS & CME samplers EIEL D STAFE: Driller: P. Williemerer						G270 : G270 : 622.9 ft. MSL
	DATES: Start: 2/26/2008 Finish: 2/26/2008							Helper: R. Keedy					Completion Station	: 18.3 ft. BGS : 874,801.92N
WI	EATHE	R: 0	vercast, o	cold		<u>, </u>		Eng/Geo: S. Simpson	1					2,514,996.84E
ber 5	v / Total (in)	<u>E</u>	s/6 in alue	ure (%)	Den. (lb/ft ³)	sf) re Type	TOPOGR Quadr Towns Sectior	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork 11, Tier 7N; Range 3W	INFORMAT While drilling 8/12/08	ION:				
lmuN	Reco % Re	Type	Blow: N - V RQD	Moist	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description		Bo D	reh etai	ole 1	Elevation ft. MSL	Remarks
	20/24 83%	ss	2-2 2-4 N=4					Dark grayish brown (10YR4/2), moist, firm, clayey	SILT		X///X////		622	
1A		$\left(\right)$		24			2	 Dark grayish brown (10YR4/2), moist, firm, silty C Dark grayish brown (10YR4/2) with 5% yellowish t (10YR5/8) mottles moist firm silty CLAY slight 	LAY					
2A	19/24 79%	ss	3-4 5-9 N=9	22		2.33 B		Gray (10YR5/1) with 70% yellowish brown (10YR mottles moint firm city CLAY slight trace and	25/8)				620	
2B		$\overline{\langle}$	14-5	20		5.04 Sh	4	gravel						
3A	20/24 83%	ss	7-8 N=12	17		2.52 Sh	⊻ 6	Dark gray (10YR4/1) with 5% yellowish brown (10Y mottles, moist, firm, silty CLAY, trace sand, slight gravel	(R5/8) trace				618	
4A	24/24 100%	ss	8-6 7-5 N=13	21		1.24 BSh							616	
4B	22/24 92%	ss	2-3 4-4	21		1.20 B	8-	Gray (10YR5/1) with 10% yellowish brown (10YR mottles, moist, firm, silty CLAY, trace sand, slight gravel	85/8) trace				614	
5A		$\overline{\mathbb{N}}$	N=/	21		1.36 <i>B</i>	10	Gray (10YR5/1) with 60% yellowish brown (10YR mottles, moist, firm, silty CLAY, trace sand, slight	85/8) trace					
6A	24/24 100%	ss	2-3 N=4	21		0.74 BSh		gravel Gray (10YR5/1), moist, soft, sandy CLAY					612	
ОD	17/24	V.	2-2	24		B	12	Grav (10YR5/1), moist, soft, sandy CLAY, trace g	ravel				610	
7A	71%		N=4	21			14	Gray (10YR5/1), moist, soft, fine- to coarse-grain	ied					
8A	19/24 79%	ss	1-3 5-6	20				<u>SAND, trace gravel</u> Dark yellowish brown (10YR4/4), moist, soft, sar <u>CLAY</u> <u>Crow (10YPS/1) with 10% releasible brown (10YPS</u>	\overline{dy}				608	
8B		$\left(\right)$	N=8	17		4.46 Sh	⊻ 16	mottles, moist, firm, silty CLAY, slight trace sand	and					
9A	24/24 100%	ss	6-8 30-35 N-38	20				Yellowish brown (10YR5/4), wet, soft, fine to coa SAND	arse			· · · · · · · · · · · · · · · · · · ·	606	
9B		\square		8			18	Gray (10YR5/1), moist, hard, silty CLAY, trace san gravel End of Boring – 18 27 ft BCS	d and					
	NOTE(S):												

F		D 1	BOF	RII	NC	L	OG		
CLIENT: AEG Coffeen Power Stat Site: CCB Management Facilit Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 2/26/2008 Finish: 2/26/2008				Powe nent l is 08 08	r Statio Facility	on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS & CME sample FIELD STAFF: Driller: B. Williamson Helper: R. Keedy	BOREHOLE ID: G280 Well ID: G280 Surface Elev: 623.0 ft. MSL Completion: 18.0 ft. BGS Station: 875.045.11N	
WE	EATHE	R: 0	vercast, o	cold				Eng/Geo: S. Simpson	2,515,679.48E
5	SAMPL	Е	1	TEST	INC	j –	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	l (in				0/ft ³)		Quad	rangle: Coffeen, IL	$\mathbf{V} = 15.60$ - While drilling $\mathbf{V} = 4.34 + 3/12/08$
	. Tota very		6 in ue	e (%	n. (lb	Type	Sectio	n 11, Tier 7N; Range 3W	$\overline{\underline{\nabla}} = +.54 + 5/12/08$ $\overline{\underline{\nabla}} = -$
mbe	cov / Reco	pe	2D	oistur	y De	ı (tsf) ilure	Depth	Lithologic	Borehole Elevation
ź	Re %	-É	BI' N N	Ž	Đ	9 Pa	ft. BGS	Description	Detail ft. MSL Remarks
1A	24/24 100%	ss	5-3 4-4	23				Dark grayish brown (10YR4/2), moist, firm, clayey	SILT 52
1D		\wedge	N=7	0		0.00		Brown (10YR4/3) with 20% yellowish brown (10Y mottles, moist, firm, silty CLAY	R5/8)
IB	(26		2.33 B	2	Dark yellowish brown (10YR4/4), moist, firm, silty	CLAY
2A	24/24	V ss	3-4 1-6	30		1.28 BSh			
	100%	Λ	N=8					Dark gray (10YR4/1) with 40% yellowish brow	n
2B	($\left(\right)$		25			v 4 −	(10YR5/8) mottles, moist, firm, silty CLAY	
	10/24	V	3-4						
34	79%	Å ss	6-6 N=10	14		3 10			618
511						Sh	6	Dark gray (10YR4/1) with 40% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY, slight	n trace
		\backslash						sand	
	22/24 92%	ss	9-11 10-8					Dark gray (10XR4/1) with 40% vellowish brow	616
4A	4A 270		N=21	18		1.67 BSh		(10YR5/8) mottles, moist, firm, silty CLAY, trace	sand,
	(8		
5A	19/24	V	2-2	20		1.47			614
	79%	Λ	N=6			В		(10YR5/8) mottles, moist, firm, silty CLAY, sand,	n trace
5B	((21		1.28 B	10-	gravel	
	22/24	V	2-3			Б			
	92%	ss	<i>3-3</i> N=6					V-II	- 612
6A		\square		20			12	(10YR6/1) mottles, moist, soft, sandy CLAY	
		\backslash						Yellowish brown (10YR5/8), moist, soft, fine to co	arse 1777
	23/24	ss	3-14 23-21					SAND, trace gravel Yellowich brown (10YR5/8) moist firm sandy CI	\overline{A}
		Λ	N=37						
7A	(13			14-	Yellowish brown (10YR5/4), moist, firm, clayey S	
	23/24	V	12-17					trace sand and gravel	
8A	96%	\bigwedge	N=41	9			▼ ∃		
8B		\square		15			16-	Yellowish brown (10YR5/4), wet, soft, fine- to	
9A	0.4/0.1	V	11-27	26				coarse-grained SAND, trace gravel	
	24/24	ss	54-43 N=81					Gray (10YR5/1), moist, hard, silty CLAY, trace san	d and
9B				7				gravel	
				-			_	End of Boring = 17.98 ft. BGS	
	NOTE	S):							
		5).							

Illinois Environmental Protection Agen	cy Well Completion Repor
Site #: County:	fontgomery Well #: G200
Site Name: _ AEG Coffeen Power Station CCB Management Faci	ility Borehole #: <u>G200</u>
State- Plant Plane Coordinate: X 877,930.6 Y 2,515,650.0 (or) Latitud	le:°' Longitude:°'
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W Hasenyager, LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type):
Logged By: <u>Suzanna L Simpson</u>	Date Started:2/25/2008 Date Finished:2/25/2008
Report Form Completed By: Suzanna L Simpson	Date:2/29/2008
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
	(MSL) ^{**} (BGS) 626.54 -2.34 Top of Protective Casing
T.	625.94 -1.74 Top of Riser Pipe
Type of Surface Seal: Concrete	<u>624.20</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: Bentonite chips	<u>620.70</u> <u>3.50</u> Top of Annular Sealant
Installation Method: <u>Gravity</u>	
Setting Time: >24 hr.	$\underline{\nabla}$ <u>621.45</u> <u>2.75</u> Static Water Level (After Completion) 3/12/2008
Type of Bentonite Seal Granular Pellet Slurry	
(choose one)	
Sotting Time: > 24 br	$\phantom{00000000000000000000000000000000000$
Setting Time. <u>>24 m.</u>	<u></u>
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	<u>612.01</u> <u>12.19</u> Top of Screen
Installation Method: <u>Gravity</u>	607.22 16.98 Bottom of Screen
Type of Backfill Material: Formation Sand	<u>606.84</u> <u>17.36</u> Bottom of Well
Installation Method: <u>Slough</u>	606.20 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0 Protective Casing Length (fact) 5.0
	Riser Pipe Length (feet) 13.93
Protective Casing SS304 SS316 PTFE PVC OTHER	R: Steel Bottom of Screen to End Cap (feet) 0.38
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER	R: Screen Length (1st slot to last slot) (feet) 4.79

PTFE PVC OTHER:

PTFE PVC OTHER:

Well Com	pletion For	m (revised	02/06/02)

SS304

SS304

SS316

SS316

Riser Pipe Below W.T.

Screen

**Hand-Slotted Well Screens Are Unacceptable

Total Length of Casing

Screen Slot Size **

19.10

0.010

(feet)

(inches)

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: _Mor	ntgomery Well #: <u>R201</u>
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #: R201
State Plane Coordinate: X_2,514,842.0 Y_877,925.3 (or) Latitude:	<u>39°</u> <u>4'</u> <u>30.5 "</u> Longitude: <u>-89°</u> <u>23 '</u> <u>52.3 "</u>
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W. Hasenyager, LPG #196-000246
Drilling Method:Hollow stem auger	Drilling Fluid (Type):n/a
Logged By:Rhonald W. Hasenyager	Date Started:
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
	<u>626.51</u> Top of Protective Casing
Type of Surface Seal: Concrete	<u>624.02</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	<u>621.52</u> <u>2.50</u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time: _>24 hr.	$ \underline{\nabla} \qquad \underline{618.70} \qquad \underline{5.32} \qquad \underline{\text{Static Water Level}} \\ (After Completion) 11/15/2010 $
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	614.479.55 Top of Seal
Setting Time: 48 min	612.90 <u>11.12</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	<u></u>
Installation Method: <u>Gravity</u>	607.3616.66Bottom of Screen
Type of Backfill Material: <u>n/a</u> (if applicable)	<u>606.80</u> <u>17.22</u> Bottom of Well
Installation Method:/a	<u>606.80</u> <u>17.22</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
(Protective Casing Length (feet) 5.0
Protection Occimentation (2004) 2021/ DEED DUG COMPA	Riser Pipe Length (feet) 14.59
Protective Casing SS304 SS316 PTFE PVC OTHER: Riser Pine Above W T SS304 SS316 PTEE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.56
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER.	Screen Length (1st slot to last slot) (feet) 4.39 Total Length Creing (f. a) 10.54
Screen SS304 SS316 PTFE PVC OTHER:	Item Length of Casing (ieet) 19.34 Screen Slot Size ** (inches) 0.010

Illinois Enviro	7	Well Completion Report							
Site #:	C	ounty: <u>Mon</u>	tgomer	y		W	/ell #:	G201	
Site Name: AEG Coffeen Po	wer Station CCB Manage	ment Facility	v				orehole #·	G201	
State- Plant Plane Coordinate: X 877,924	4.9 Y <u>2,514,849.5</u> (o	r) Latitude:) <u> </u>		Longitud	e:°		
Surveyed By: <u>Jeffrey D. Emri</u>	ick		IL Reg	istration #:	035-0	03507			
Drilling Contractor: <u>Testing Sec</u>	ervice Corporation		Driller	B. Wil	liamson				
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.		Geolog	ogist: <u>Rhonald W Hasenyager</u> , LPG #196-000246					
Drilling Method: <u>Hollow stem</u>	auger		Drillin	g Fluid (Ty	/pe):				
Logged By: <u>Suzanna L Simp</u>	son		Date S	e Started: <u>2/25/2008</u> Date Finished: <u>2/25/2008</u>					
Report Form Completed By: Su	zanna L Simpson		Date:	2/29	9/2008				
ANNULAR SPA	CE DETAILS		-	Elev	ations	Depths (BGS)	(0.01	ft.)	
				_62	7.66	-3.76	Top of Protec	tive Casing	
				62	7.12	-3.22	Top of Riser	Pipe	
Type of Surface Seal: Concrete				>62	3.90	0.00	Ground Surfa	ce	
Type of Annular Sealant: Bento	nite chips			62	0.60	3.30	Top of Annul	ar Sealant	
Installation Method: Gravit	v	- 4							
Setting Time: <u>>24 hr.</u>	<u>,</u>	_ <u>\</u>	Z	_62	1.73	2.17	Static Water I	Level	
Type of Bentonite Seal Gran	ular Pellet Slurry						(Aner Complet	01) 3/12/2008	
Installation Method: <u>Gravit</u>	(choose one) Y			62	0.60	3.30	Top of Seal		
Setting Time: <u>>24 hr.</u>		- 🕅	×	61	1.80	12.10	Top of Sand I	Pack	
Type of Sand Pack: Quartz sand	1								
Grain Size: 10/20 (sie	eve size)			61	0.89	13.01	Top of Screen	L	
Installation Method: <u>Gravit</u>	у	-							
Type of Backfill Material: <u>n/a</u>	(if applicable)			<u>_60</u> _60	<u>6.10</u> 5.75	17.80 18.15	Bottom of Sci Bottom of We	een ell	
Installation Method: <u>n/a</u>	(_		<u>60</u>	<u>5.75</u>	<u>18.15</u>	Bottom of Bo	rehole	
				* Ke	lerenced to a	a National Geode	are Datum		
				[CAS	SING MEA	SUREMENT	S	
WELL CONS	STRUCTION MATERIA	LS		Diameter	of Boreh	nole	(inch	es) 8.0	
(Choose or	ne type of material for each area)			Protective	er Pipe	Length	(inch	$\frac{2.0}{2.0}$	
				Riser Pip	e Length	Longui	(fe	et) 16.23	
Protective Casing	SS304 SS316 PTFE P	VC OTHER: (Steel	Bottom o	f Screen	to End Cap	(fe	et) 0.35	
Riser Pipe Above W.T.	SS304 SS316 PTFE P	VC OTHER:		Screen Le	ength (1	st slot to last slo	ot) (fe	et) 4.79	
Riser Pipe Below W.T.	SS304 SS316 PTFE P	VC OTHER:		Total Ler	ngth of Ca	asing	(fe	et) 21.37	

Screen

SS304 SS316 PTFE PVC OTHER:

 Total Length of Casing
 (feet)

 Screen Slot Size **
 (inches)

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Enviro	nmental Protection Agen	cy		Well	Completior	ı Report
Site #:	County: <u>N</u>	Iontgomery	1	W	/ell #:G2	205
Site Name: <u>AEG Coffeen Po</u>	wer Station CCB Management Faci	lity		В	orehole #:(<u>3205</u>
State- Plant Plane Coordinate: X 875,550	0.2 Y 2,515,914.9 (or) Latitud	le:°	· · · · · · · · · · · · · · · · · · ·	Longitud	le:°	
Surveyed By: <u>Jeffrey D. Emri</u>	ick	IL Regi	istration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Testing Sec</u>	ervice Corporation	_ Driller:	B. Williamson			
Consulting Firm: <u>Hanson Prof</u>	fessional Services Inc.	_ Geolog	ist: <u>Rhonald W</u>	Hasenyage	r, LPG #196-00	0246
Drilling Method: <u>Hollow sterr</u>	1 auger	_ Drilling	g Fluid (Type):			
Logged By: <u>Suzanna L Simp</u>	son	_ Date St	arted: <u>2/21/20</u>	08 Date	e Finished: <u>2/2</u>	21/2008
Report Form Completed By: <u>Su</u>	zanna L Simpson	_ Date: _	2/29/2008			
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.))
	_		624.87	2.72	Top of Protective	e Casing
			624.45	-2.30	Top of Riser Pip	e
Type of Surface Seal: Concrete			<u> 622.15 </u>	0.00	Ground Surface	
Type of Annular Sealant: <u>Benton</u>	nite chips		619.95	2.20	Top of Annular S	Sealant
Installation Method: <u>Gravit</u>	<u>y</u>					
Setting Time: <u>>24 hr.</u>		Σ	617.09	5.06	Static Water Lev (After Completion)	el 3/12/2008
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method: <u>Gravit</u>	(choose one)		619.95	2.20	Top of Seal	
Setting Time: <u>>24 hr.</u>	X		613.35	8.80	Top of Sand Pac	k
Type of Sand Pack: <u>Quartz sand</u>	d					
Grain Size: 10/20 (sie	eve size)		612.11	10.04	Top of Screen	
Installation Method: <u>Gravit</u>	у					
Type of Backfill Material: Form	nation Sand (if applicable)		<u>607.62</u> <u>607.08</u>	<u>14.53</u> <u>15.07</u>	Bottom of Screen Bottom of Well	a
Installation Method: <u>Slough</u>	<u>n</u>		<u>606.15</u> * Referenced to a		Bottom of Boreh	ole
			CAS Diameter of Boreh	ING MEA	SUREMENTS	8.0
WELL CONS (Choose or	STRUCTION MATERIALS ne type of material for each area)		ID of Riser Pipe		(inches)	2.0
			Protective Casing	Length	(feet)	5.0
Protoctive Cooing	SS204 SS216 DTEE DVC OTHE	D. Staal	Riser Pipe Length		(feet)	12.34
Riser Pipe Above W.T.	SS304 SS316 PTFE (PVC) OTHE	R:	Bottom of Screen f	to End Cap	(feet)	0.54

PTFE PVC OTHER:

PTFE PVC OTHER:

Well Com	pletion	Form	(revised)	02/06/02

SS304

SS304

SS316

SS316

Riser Pipe Below W.T.

Screen

**Hand-Slotted Well Screens Are Unacceptable

Total Length of Casing

Screen Slot Size **

17.37

0.010

(feet)

(inches)

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: Mor	ntgomery Well #: <u>G206</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: G206
State Plane Coordinate: X_2,514,669.2 Y_875,103.9 (or) Latitude:	<u>39° 4' 2.6"</u> Longitude: <u>-89° 23' 54.8"</u>
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: Layne-Western Co	Driller: D. Mahurin
Consulting Firm: Hanson Professional Services Inc.	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: <u>Suzanna L. Simpson</u>	Date Started: <u>10/14/2010</u> Date Finished: <u>10/14/2010</u>
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/15/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
Type of Surface Seal:	<u>630.54</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	627.84 2.70 Top of Annular Sealant
Installation Method: Tremie	
Setting Time: _>24 hr.	$\underline{\nabla} \qquad \underline{611.96} \qquad \underline{18.58} \qquad \text{Static Water Level} $
Type of Bentonite Seal Granular Pallet Shurny	(Aner Completion) 11/15/2010
(choose one)	
Installation Method: <u>Gravity</u>	-616.24 _ 14.30 Top of Seal
Setting Time: <u>15 min</u>	615.04 <u>15.50</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	$\underbrace{\begin{array}{c} \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{} \\ \underline{}$
Installation Method: <u>Gravity</u>	608 62 21 92 Bottom of Screen
Type of Backfill Material: <u>n/a</u>	$\underline{\begin{array}{c} \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline$
Installation Method:n/a	606.54Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0 Piser Pine Length (feet) 10.70
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Can (feet) 0.50
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.41
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 24.70
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Enviro	nmental Protection Agency	y	Well Completion Report					
Site #:	County:	ontgomery	/		W	/ell #:G	207	
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	у			В	orehole #:	G207	
State Plane Coordinate: X 2,514,83	7.9 Y <u>875,166.4</u> (or) Latitude	: <u>39</u>	24	<u>' 3.2 "</u>	Longitud	e: <u>-89 ° 2</u>	<u>3 ' 52.6 "</u>	
Surveyed By:Jeffrey D. Emri	ck	IL Reg	istration	#:	03507			
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	<u> </u>	Iahurin				
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geolog	ist: <u>R</u>	honald W.	Hasenyage	r, LPG #196-000	0246	
Drilling Method: <u>Hollow stem</u>	auger	Drilling	g Fluid (Гуре): <u>n/a</u>	a			
Logged By: <u>Suzanna L. Simp</u>	son	_ Date St	arted:	10/8/20	0 <u>10</u> Date	e Finished: <u>10</u>	/8/2010	
Report Form Completed By: <u>Su</u>	zanna L. Simpson	Date:	1	0/8/2010				
ANNULAR SPA	CE DETAILS		El (evations MSL)*	Depths (BGS)	(0.01 ft.)		
			_	633.37	2.76	Top of Protective	Casing	
	F		_	633.21	-2.60	Top of Riser Pipe	e	
Type of Surface Seal: <u>Concrete</u>			>	630.61	0.00	Ground Surface		
Type of Annular Sealant: High-	solids bentonite		_	628.61	2.00	Top of Annular S	ealant	
Installation Method: Tremie	· · · · · · · · · · · · · · · · · · ·							
Setting Time:24 hr.		∑		612.86	_17.75_	Static Water Leve (After Completion)	el 11/15/2010	
Type of Bentonite Seal Gran	ular Pellet Slurry					(i nei completion)	11,10,2010	
Installation Method: Gravit	(choose one)	Ŭ.		614.76	15.85	Top of Seal		
Setting Time: <u>15 min</u>			_	613 63	16.08	Top of Sand Paol	<i>r</i>	
	× ×		_	015.05			Υ.	
Type of Sand Pack: <u>Quartz sanc</u>				612.37		Top of Screen		
Installation Method: Gravit	ve size)							
			_	<u>607.84</u>	22.77	Bottom of Screen	l	
Type of Backfill Material: <u>n/a</u>	(if applicable)			607.31		Bottom of Well		
Installation Method: <u>n/a</u>			*	606.61 Referenced to a	24.00 National Geodeti	Bottom of Boreh	ole	
				G + 0				
			Diama	CAS	SING MEA	SUREMENTS	8.0	
WELL CONS	TRUCTION MATERIALS		ID of F	Riser Pipe	JIE	(inches)	2.0	
	a gro or material for each area)		Protect	ive Casing I	ength	(feet)	5.0	
		1	Riser F	ipe Length		(feet)	20.84	
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Bottom	n of Screen to	o End Cap	(feet)	0.53	
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER		Screen	Length (1	st slot to last slo	t) (feet)	4.53	
Screen	SS304 SS316 PTFE PVC OTHER		Screen	ength of Ca Slot Size **	sing	(feet)	25.90	
L	1		Louion	5101 5120		(menes)	0.010	

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:	ntgomery Well #: <u>G208</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: <u>G208</u>
State Plane Coordinate: X 2,514,993.6 Y 875,231.5 (or) Latitude:	<u>39° 4' 3.9"</u> Longitude: <u>-89° 23' 50.6"</u>
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By:Suzanna L. Simpson	Date Started: Date Finished:
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/8/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u></u>
	<u>633.16</u> 2.59 Top of Riser Pipe
Type of Surface Seal:	<u>630.57</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High solids bentonite	<u>627.77</u> <u>2.80</u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time: _>24 hr.	$\underline{\nabla} \qquad \underline{614.76} \qquad \underline{15.81} \qquad \text{Static Water Level} $
Type of Bentonite Seal Granular Pellet Slurny	(Anter Completion) 11/15/2010
(choose one)	616.07 14.50 Tax afficial
Sotting Time: 0 min	$\frac{616.07}{14.50}$ 10p of seal
	<u>613.90</u> <u>16.67</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	$\underbrace{-613.04}_{1/.53}$ Top of Screen
Installation Method: <u>Gravity</u>	608 51 22 06 Bottom of Screen
Type of Backfill Material: <u>n/a</u>	<u>607.97</u> <u>22.60</u> Bottom of Well
Installation Method: n/a	606.57 24.00 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTFE PVC OTHER:	Kisci Fipe Lengin (feet) 20.12 Steel Bottom of Screen to End Can 0.54
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4 53
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 25.19
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: _Mor	ntgomery Well #: <u>G209</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: <u>G209</u>
State Plane Coordinate: X 2,515,149.6 Y 875,298.2 (or) Latitude:	<u>39° 4' 4.5"</u> Longitude: <u>-89° 23' 48.7"</u>
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By:Suzanna L. Simpson	Date Started: 10/7/2010 Date Finished: 10/7/2010
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/8/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>633.16</u> Top of Protective Casing
	<u></u>
Type of Surface Seal: Concrete	<u>630.57</u> <u>0.00</u> Ground Surface
Ture of Annulus Coslants . Useh colide houterite	<u>627.57</u> <u>3.00</u> Top of Annular Sealant
Installation Mathod: Tramia	
Setting Time: >24 hr.	\bigtriangledown 615.52 15.05 Static Water Level
	(After Completion) 11/15/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	$\underbrace{616.07}_{} 14.50 \text{ Top of Seal}$
Setting Time: <u>15 min</u>	<u>614.67</u> <u>15.90</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	<u></u> <u></u> <u></u> Top of Screen
Installation Method: <u>Gravity</u>	608 20 22 28 Bottom of Saraan
Type of Backfill Material:	
Installation Method: n/a	606.57 24.00 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTFE PVC OTHER:	Kiser Pipe Length (feet) 20.08 Steel Bottom of Screen to End Can 0.52
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4 54
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 25.15
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:	ntgomery Well #: <u>G210</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: G210
State Plane Coordinate: X 2,515,299.0 Y 875,359.7 (or) Latitude:	<u>39° 4' 5.1"</u> Longitude: <u>-89° 23' 46.8</u> "
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: Layne-Western Co	Driller: D. Mahurin
Consulting Firm: Hanson Professional Services Inc.	Geologist: _ Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By:Suzanna L. Simpson	Date Started:
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/8/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u></u>
	<u></u>
Type of Surface Seal:	<u>630.48</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	<u>627.48</u> <u></u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time: _>24 hr.	$\underline{\nabla} \qquad \underline{615.38} \qquad \underline{15.10} \qquad \text{Static Water Level} \\ (After Completion) 11/(5/2010)$
Type of Bentonite Seal Granular Pellet Shurny	(Alter Completion) 11/15/2010
(choose one)	614.02 16.45 Tax of Sec.
Setting Time: 15 hrs	$\frac{-014.05}{10.45} = 10.45$ Top of Seat
	<u>612.98</u> <u>17.50</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u>	611.00 10.20 Tax of Survey
Grain Size: <u>10/20</u> (sieve size)	<u></u> 19.39 Top of Screen
Installation Method: <u>Gravity</u>	Bottom of Screen
Type of Backfill Material:	<u>606.02</u> <u>24.46</u> Bottom of Well
Installation Method:	<u>605.48</u> <u>25.00</u> Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0 ID of Pisor Pipo 2.0
(Choose one type of material for each area)	Protective Casing Length (feet) 5.0
	Riser Pipe Length (feet) 21.90
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Cap (feet) 0.53
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.54
Kiser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Saraan SS204 SS214 DTFE DVC OTHER:	Total Length of Casing (feet) 26.97
Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Solution Sol	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: _Mor	ntgomery Well #: <u>G211</u>
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #: G211
State Plane Coordinate: X 2,515,449.1 Y 875,424.5 (or) Latitude:	<u>39° 4' 5.7"</u> Longitude: <u>-89° 23' 44.9</u> "
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: _ Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: Suzanna L. Simpson	Date Started: <u>10/11/2010</u> Date Finished: <u>10/11/2010</u>
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/15/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
	<u>632.83</u> <u>-2.52</u> Top of Protective Casing
	<u>632.64</u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>630.31</u> <u>0.00</u> Ground Surface
	<u>627.31</u> <u>3.00</u> Top of Annular Sealant
lype of Annular Sealant: <u>High-solids bentonite</u>	
Setting Time: >24 hr	∇ 616.17 14.14 Static Water Level
	(After Completion) 11/15/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	
Setting Time: <u>15 min</u>	<u>614.91</u> <u>15.40</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	$\underbrace{\underline{612.97}}_{17.34} \text{Top of Screen}$
Installation Method: <u>Gravity</u>	608 43 21 88 Bottom of Screen
Type of Backfill Material: <u>n/a</u>	<u>607.90</u> <u>22.41</u> Bottom of Well
Installation Method: <u>n/a</u>	<u>_606.31</u> <u>_24.00</u> Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0 Riser Pipe Length (feet) 10.67
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Cap (feet) 19.07
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.54
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 24.74
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agenc	y Well Completion Report
Site #: County:	ontgomery Well #: G212
Site Name:AEG Coffeen Power Station CCB Management Facili	ty Borehole #:G212
State Plane Coordinate: X 2,515,583.0 Y 875,486.5 (or) Latitude	e: <u>39° 4' 6.3"</u> Longitude: <u>-89° 23' 43.1"</u>
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: Hanson Professional Services Inc.	Geologist: _ Rhonald W. Hasenyager, LPG #196-000246
Drilling Method:Hollow stem auger	_ Drilling Fluid (Type):
Logged By:Suzanna L. Simpson	Date Started:10/11/2010 Date Finished:10/11/2010
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
	<u>633.12</u> Top of Protective Casing
	<u>632.89</u> <u></u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>630.59</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	<u>627.59</u> <u>3.00</u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time:24 hr.	$\underline{\nabla} \qquad \underline{616.10} \qquad \underline{14.49} \qquad \text{Static Water Level} \\ (After Completion) 11/15/2010$
Type of Bentonite Seal Granular Pellet Slurry	
(choose one) Installation Method: Gravity	616.89 13.70 Top of Seal
Setting Time: <u>17 min</u>	615.79 <u>14.80</u> Top of Sand Pack
Type of Sand Pack: Owerte and	
Grain Size: 10/20 (sieve size)	<u> 613.85</u> <u></u> Top of Screen
Installation Method: <u>Gravity</u>	
Type of Backfill Material: <u>n/a</u>	$ \begin{array}{c c} \hline \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ $
(if applicable)	606 59 24 00 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTFE PVC OTHER	Riser Pipe Length (feet) 19.04
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER	Screen Lenoth (1st slot to 1st slot) (feet) 4.55
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER	Total Length of Casing (feet) 24.11
Screen SS304 SS316 PTFE PVC OTHER	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: _ Mon	ntgomery Well #: <u>G213</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: G213
State Plane Coordinate: X 2,515,723.5 Y 875,544.4 (or) Latitude:	<u>39° 4' 6.9"</u> Longitude: <u>-89° 23' 41.4"</u>
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: Suzanna L. Simpson	Date Started:
Report Form Completed By: _ Suzanna L. Simpson	Date:10/19/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
Type of Surface Seal: Concrete	<u>630.34</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	<u>627.64</u> <u>2.70</u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time: >24 hr.	$\underline{\nabla} \qquad \underline{615.01} \qquad \underline{15.33} \qquad \text{Static Water Level} \\ (After Completion) 11/15/2010$
Type of Bentonite Seal Granular Pellet Slurry	
(choose one) Installation Method: Gravity	<u>616.54</u> <u>13.80</u> Top of Seal
Setting Time: <u>10 min</u>	
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	$\begin{array}{c c} & \underline{-613.59} & \underline{-16.75} & \text{Top of Screen} \\ \hline \end{array}$
Installation Method: <u>Gravity</u>	<u>609.05</u> <u>21.29</u> Bottom of Screen
Type of Backfill Material: <u>n/a</u> (if applicable)	Bottom of Well
Installation Method: <u>n/a</u>	<u>606.34</u> <u>24.00</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Destasting Cosing 00204 00214 NEED NVG 00000	Riser Pipe Length (feet) 19.22
Flotecurve clasing SS304 SS316 PTFE PVC OTHER: Riser Pine Above W T SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.53
Riser Pipe Below W.T. SS304 SS316 PTFE (PVC) OTHER:	Screen Length (1st slot to last slot) (feet) 4.54 Total Length of Casing (feet) 24.20
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Site #: County: Montgomery Well #: G214 Site Name: AEG Coffeen Power Station CCB Management Facility Borehole #: G214 State State State State
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u> Borehole #: <u>G214</u>
State
Plane Coordinate: X 2,515,960.8 Y 875,668.0 (or) Latitude: 39° 4' 8.1" Longitude: -89° 23' 38.3
Surveyed By: <u>Jeffrey D. Emrick</u> IL Registration #: <u>035-003507</u>
Drilling Contractor: Layne-Western Co Driller: D. Mahurin
Consulting Firm: <u>Hanson Professional Services Inc.</u> Geologist: <u>Rhonald W. Hasenyager, LPG #196-000246</u>
Drilling Method: <u>Hollow stem auger</u> Drilling Fluid (Type): <u>n/a</u>
Logged By: Suzanna L. Simpson Date Started: 10/14/2010 Date Finished: 10/14/2010
Report Form Completed By: <u>Suzanna L. Simpson</u> Date: <u>10/19/2010</u>
ANNULAR SPACE DETAILS Elevations Depths (0.01 ft.) (MSL)* (BGS) (0.01 ft.)
<u>633.08</u> <u>-2.69</u> Top of Protective Casing
<u>632.85</u> <u>632.85</u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u> <u>630.39</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite
Installation Method: Tremie
Setting Time: $>24 \text{ hr.}$ \searrow $\underline{609.48}$ 20.91 Static Water Level (After Completion) 11/15/2010
Type of Bentonite Seal Granular Pellet Slurry
(choose one) Installation Method: <u>Gravity</u> <u>615.39</u> <u>15.00</u> Top of Seal
Setting Time: 14 min 614.34 16.05 Top of Sand Pack
Type of Sand Pack:
Grain Size: $10/20$ (sieve size) 612.64 17.75 Top of Screen
Installation Method: <u>Gravity</u> 608.25 22.14 Bottom of Screen
Type of Backfill Material: $\underline{n/a}$ $\underline{(if applicable)}$ $\underline{607.74}$ $\underline{22.65}$ Bottom of Well
Installation Method: <u>n/a</u> <u>606.39</u> <u>24.00</u> Bottom of Borehole
* Referenced to a National Geodetic Datum
CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Chocea and two of motorial for each area) ID of Riser Pipe (inches) 2.0
Protective Casing Length (feet) 5.0
Riser Pipe Length (feet) 20.21
Protective Casing SS304 SS316 PTFE PVC OTHER: Steel Bottom of Screen to End Cap (feet) 0.51
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Screen Length (1st slot to last slot) (feet) 4.39 Biogr Bing Below W.T. SS204 SS216 PTFE PVC OTHER: Screen Length (1st slot to last slot) (feet) 4.39
Niser ripe below w.r. South SS300 Fire rive other. Total Length of Casing (feet) 25.11 Screen SS304 SS316 PTFE PVC OTHER: Screen Slot Size ** (inclusio) 0.010

Illinois Environmental Protection Agence	У	Well Completion Report			
Site #: County: <u>N</u>	Site #: County: Montgomery				
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	ty	Borehole #: G215			
State Plane Coordinate: X_2,515,971.6 Y_875,810.2 (or) Latitud	e: <u>39° 4' 9.5"</u>	Longitude: <u>-89°</u> <u>23 ' 38.2 "</u>			
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-00	3507			
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin				
Consulting Firm: Hanson Professional Services Inc.	_ Geologist: _ Rhonald W. H	Hasenyager, LPG #196-000246			
Drilling Method: <u>Hollow stem auger</u>	_ Drilling Fluid (Type): <u>n/a</u>	a			
Logged By: <u>Suzanna L. Simpson</u>	Date Started:10/13/20	10 Date Finished: 10/13/2010			
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010				
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (0.01 ft.)			
		<u>-2.82</u> Top of Protective Casing			
	633.06_	<u>-2.58</u> Top of Riser Pipe			
Type of Surface Seal: <u>Concrete</u>	630.48	Ground Surface			
Type of Annular Sealant: High-solids bentonite	627.58_	2.90 Top of Annular Sealant			
Installation Method: Tremie					
Setting Time:24 hr.	⊻	22.84 Static Water Level			
		(After Completion) 11/15/2010			
Type of Bentonite Seal Granular Pellet Slurry (choose one)	Ť				
Installation Method: <u>Gravity</u>	614.08	<u>16.40</u> Top of Seal			
Setting Time: 20 min	612.98	17.50 Top of Sand Pack			
Type of Sand Pack: <u>Quartz sand</u>					
Grain Size: 10/20 (sieve size)		<u>19.41</u> Top of Screen			
Installation Method: <u>Gravity</u>		22.90 D.4 CO			
Type of Backfill Material:	<u></u>	24.31 Bottom of Well			
(if applicable)	606.17	24.31 Bottom of Borehole			
	* Referenced to a N	Lational Geodetic Datum			
	CASI	NG MEASUREMENTS			
	Diameter of Borehol	e (inches) 8.0			
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	(inches) 2.0			
	Protective Casing Le	ength (feet) 5.0			
Protective Casing SS304 SS316 PTEE PVC OTHER	Riser Pipe Length	(feet) 21.99			
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER	Bottom of Screen to Screen Length (1st	slot to last slot) (feet) U.51			
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER	Total Length of Casi	ing (feet) 26.89			
Screen SS304 SS316 PTFE PVC OTHER	Screen Slot Size **	(inches) 0.010			

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:	ntgomery Well #: <u>G216</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: G216
State Plane Coordinate: X 2,515,968.5 Y 875,976.1 (or) Latitude:	<u>39° 4' 11.2"</u> Longitude: <u>-89° 23' 38.2</u> "
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: D. Mahurin
Consulting Firm: Hanson Professional Services Inc.	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: <u>Suzanna L. Simpson</u>	Date Started: <u>10/13/2010</u> Date Finished: <u>10/13/2010</u>
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
	<u>633.02</u> <u>-2.74</u> Top of Protective Casing
	<u></u>
Type of Surface Seal: Concrete	<u>630.28</u> <u>0.00</u> Ground Surface
Time of Annulus Caslant	627.78 2.50 Top of Annular Sealant
Installation Method: Tremie	
Setting Time: >24 hr.	\bigtriangledown 607.52 22.76 Static Water Level
	(After Completion) 11/15/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)	YTT
Installation Method: <u>Gravity</u>	<u>613.28</u> <u>17.00</u> Top of Seal
Setting Time: <u>15 min</u>	612.08 <u>18.20</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	$\underline{\underline{610.24}} \underline{20.04} \text{Top of Screen}$
Installation Method: <u>Gravity</u>	
Type of Backfill Material:	<u>605.86</u> <u>24.42</u> Bottom of Screen <u>605.35</u> <u>24.93</u> Bottom of Well
(if applicable)	604.28 26.00 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTEF PVC OTHER	Riser Pipe Length (feet) 22.52 Steel Dattern of Screen to Find Com 0.51
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of screen to End Cap (feet) 0.51 Screen Length (fest) 4.38
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 27 41
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:	ntgomery Well #: G217
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: G217
State Plane Coordinate: X 2,515,963.0 Y 876,185.6 (or) Latitude:	<u> </u>
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: Layne-Western Co	Driller: D. Mahurin
Consulting Firm: Hanson Professional Services Inc.	Geologist: _ Rhonald W. Hasenyager, LPG #196-000246
Drilling Method:Hollow stem auger	Drilling Fluid (Type):
Logged By:Suzanna L. Simpson	Date Started: <u>10/12/2010</u> Date Finished: <u>10/12/2010</u>
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>633.34</u> <u>-2.67</u> Top of Protective Casing
	<u>633.10</u> <u>2.43</u> Top of Riser Pipe
Type of Surface Seal: Concrete	<u>630.67</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	<u>628.27</u> <u>2.40</u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time:24 hr.	$\underline{\nabla}$ <u>609.28</u> <u>21.39</u> Static Water Level (After Completion) 11/15/2010
Type of Bentonite Seal Granular Pellet Slurry	
(choose one) Installation Method: Gravity	612.82 17.85 Top of Seal
Setting Time:	611.82 18.85 Top of Sand Pack
Turn of Sand Darlan o	
Grain Size: 10/20 (sieve size)	<u>610.18</u> <u>20.49</u> Top of Screen
Installation Method: Gravity	
	<u>605.79</u> <u>24.88</u> Bottom of Screen
(if applicable)	
Installation Method:	604.67 26.00 Bottom of Borehole * Referenced to a National Geodetic Datum
	Diameter of Borehole (inches) 8 0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Cacing SC204 SC214 DTEE DVG OTHER	Riser Pipe Length (feet) 22.92
Riser Pine Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.50
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (fact) 27.81
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Illinois Environmental Protection Ag	gency			Well	Completio	n Report
Site #: County	Site #: County: Montgomery		W	/ell #:	i218	
Site Name: <u>AEG Coffeen Power Station CCB Management</u>	Facility			В	orehole #:	G218
State Plane Coordinate: X 2,515,962.2 Y 876,380.9 (or)	Latitude:	<u>39°</u>	4 ' 15.2 "	Longitud	e: <u>-89</u> °	<u>23 ' 38.2 "</u>
Surveyed By: <u>Jeffrey D. Emrick</u>	IL	Registra	tion #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-Western Co</u>	Dr	iller:	D. Mahurin			
Consulting Firm: Hanson Professional Services Inc.	Ge	ologist:	Rhonald W.	Hasenyage	r, LPG #196-00	0246
Drilling Method: <u>Hollow stem auger</u>	Dr	illing Flu	uid (Type): <u>n/a</u>	ı		
Logged By: <u>Suzanna L. Simpson</u>	Da	te Starte	d: <u>10/12/20</u>	010 Dat	e Finished: <u>1(</u>)/12/2010
Report Form Completed By: <u>Suzanna L. Simpson</u>	Da	.te:	10/19/2010			
ANNULAR SPACE DETAILS			Elevations (MSL)*	Depths (BGS)	(0.01 ft	.)
		_	633.34	-2.70	Top of Protectiv	e Casing
			633.11	2.47	Top of Riser Pip	be
Type of Surface Seal: <u>Concrete</u>			630.64	0.00	Ground Surface	
			627.14	3.50	Top of Annular	Sealant
Installation Mathed: Tramia		8				
Setting Time: >24 hr.			609.89	20.75	Static Water Le	vel
·					(After Completion) 11/15/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)	TY					
Installation Method: <u>Gravity</u>		¥	613.14	17.50	Top of Seal	
Setting Time: <u>17 min</u>		Ř	612.14	18.50	Top of Sand Pa	ck
Type of Sand Pack:Quartz sand						
Grain Size: <u>10/20</u> (sieve size)			610.31	20.33	Top of Screen	
Installation Method: <u>Gravity</u>			(05.97	24 77		
Type of Backfill Material:n/a			605.87	24.77	Bottom of Scree Bottom of Well	n
(if applicable)			604 64	26.00	Bottom of Bore	nole
			* Referenced to a	National Geodeti	c Datum	
			CAS	ING MEA	SUREMENTS	
		Di	ameter of Boreho	ole	(inches	8.0
(Choose one type of material for each area)		ID	of Riser Pipe		(inches) 2.0
		Pr	otective Casing L	ength	(feet	5.0
Protective Casing SS304 SS316 PTFE PVC	OTHER: Stee		ser Pipe Length	End Car	(feet	0.50
Riser Pipe Above W.T. SS304 SS316 PTFE PVC	OTHER:		reen Lenoth (14	st slot to last slo	t) (feet	4 44
Riser Pipe Below W.T. SS304 SS316 PTFE PVC	OTHER:		tal Length of Ca	sing	(feet	27.74
Screen SS304 SS316 PTFE PVC	OTHER:	Sc	reen Slot Size **		(inches	0.010

Illinois Environmental Protection Agency	y Well Completion Report
Site #: County: Mon	ntgomery Well #:G270
Site Name: AEG Coffeen Power Station CCB Management Facilit	ty Borehole #: G270
State- Plant PlaneCoordinate: X 874,801.9 Y 2,514,996.8 (or) Latitude:	°°' Longitude:°'
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):
Logged By: Suzanna L Simpson	Date Started: <u>2/26/2008</u> Date Finished: <u>2/26/2008</u>
Report Form Completed By: Suzanna L Simpson	Date: <u>2/29/2008</u>
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	626.413.49 Top of Protective Casing
Type of Surface Seal: Concrete	<u>622.92</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: Bentonite chips	619.92 3.00 Top of Annular Sealant
Installation Method: Gravity	
Setting Time: _>24 hr.	$ \underline{\nabla} \qquad \underline{617.30} \qquad \underline{5.62} \qquad \underline{\text{Static Water Level}} \\ (After Completion) \frac{3}{12}/2008 $
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	<u>619.92</u> <u>3.00</u> Top of Seal
Setting Time: <u>>24 hr.</u>	610.92 <u>12.00</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	
Grain Size: 10/20 (sieve size)	<u>609.79</u> <u>13.13</u> Top of Screen
Installation Method: <u>Gravity</u>	
Type of Backfill Material: n/a	
(if applicable)	
Installation Method: <u>n/a</u>	604.65 18.27 Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
(1.1.1.1.21)	Protective Casing Length (feet) 5.0
	Riser Pipe Length (feet) 16.18
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Cap (feet) 0.35
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.79
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 21.32

Screen

SS304 SS316 PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Environm	Well Completion Report							
Site #:	1	W	Vell #: G	280				
Site Name: AEG Coffeen Power Station CCR Management Facility						orehole #	G280	
State- Plant Plane Coordinate: X875,045.1Y_2,515,679.5 (or) Latitude:°'								
Surveyed By: Jeffrey D. Emrick			IL Registration #:035-003507					
Drilling Contractor: <u>Testing Service Corporation</u>				Driller: <u>B. Williamson</u>				
Consulting Firm: <u>Hanson Professional Services Inc.</u>			Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246					
Drilling Method: Hollow stem auger			Drilling Fluid (Type):					
Logged By: <u>Suzanna L Simpson</u>			Date Started: <u>2/26/2008</u> Date Finished: <u>2/26/2008</u>					
Report Form Completed By: Suzanna L Simpson				Date: <u>2/29/2008</u>				
ANNULAR SPACE DETAILS				Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
				625.79	-2.84	Top of Protectiv	e Casing	
				625.30	-2.35	Top of Riser Pip	e	
Type of Surface Seal: Concrete				<u> 622.95 </u>	0.00	Ground Surface		
Type of Appular Seclent: Bontonite ching				620.85	2.10	Top of Annular	Sealant	
Installation Mathadi Cravity	11103	- 9						
Setting Time: _>24 hr.			<u>z</u>	618.61	4.34	Static Water Lev	vel	
						(After Completion) 3/12/2008	
Type of Bentonite Seal Granular	ŤŢ							
Installation Method: <u>Gravity</u>			×	620.85	2.10	Top of Seal		
Setting Time: _>24 hr.			×	<u>611.75</u> <u>11.20</u> Top of Sand Pack				
Type of Sand Pack: <u>Quartz sand</u>		_						
Grain Size: 10/20 (sieve size)				610.16	12.79	Top of Screen		
Installation Method: Gravity		-						
Type of Backfill Material: n/a				<u>605.32</u> 604.97	<u>17.63</u> 17.98	Bottom of Scree Bottom of Well	n	
	(if applicable)	_						
Installation Method:				Bottom of Borehole * Referenced to a National Geodetic Datum				
				CAS		CLIDEMENTS		
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)				Diameter of Borehole		SUKEMEN IS	8.0	
				ID of Riser Pipe		(inches)	2.0	
(Protective Casing Length		(feet)	5.0	
				Riser Pipe Length	(feet)	15.14		
Protective Casing SS3	04 SS316 PTFE P	VC OTHER: (Steel	Bottom of Screen	to End Cap	(feet)	0.35	
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:				Screen Length (1	ot) (feet)	4.84		
Riser Pipe Below W.T. SS3	04 SS316 PTFE P	VC OTHER:		Total Length of Ca	asing	(feet)	20.33	

Screen

SS304 SS316 PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Illinois Environmental Protection Agency	Well Completion Report					
Site #: County:	ntgomery Well #:					
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #:T202					
State Plane Coordinate: X 2,514,895.0 Y 876,699.4 (or) Latitude:	<u>39°</u> <u>4'</u> <u>18.4"</u> Longitude: <u>-89°</u> <u>23'</u> <u>51.7"</u>					
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507					
Drilling Contractor: Layne-Western Co	Driller: D. Mahurin					
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246					
Drilling Method:Hollow stem auger	Drilling Fluid (Type):					
Logged By:Suzanna L. Simpson	Date Started: 10/15/2010 Date Finished: 10/15/2010					
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010					
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)					
Type of Surface Seal:	<u>626.22</u> <u>0.00</u> Ground Surface					
Type of Annular Sector: High solids bentonite	<u>623.42</u> <u>2.80</u> Top of Annular Sealant					
Installation Method: Tremie						
Setting Time: >24 hr.	∇ <u>613.72</u> <u>12.50</u> Static Water Level					
	(After Completion) 11/15/2010					
Type of Bentonite Seal Granular Pellet Slurry (choose one)	Ϋ́Τ					
Installation Method: <u>Gravity</u>	<u>616.50</u> <u></u> Top of Seal					
Setting Time: <u>15 min</u>	<u>615.27</u> <u>10.95</u> Top of Sand Pack					
Type of Sand Pack:Quartz sand						
Grain Size: 10/20 (sieve size)	<u></u>					
Installation Method: <u>Gravity</u>						
Type of Backfill Material:	<u>609.01</u> <u>17.21</u> Bottom of Well					
(if applicable)	608 22 18 00 Bottom of Borehole					
	* Referenced to a National Geodetic Datum					
	CASING MEASUREMENTS					
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0					
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0					
	Protective Casing Length (feet) 5.0					
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Can (feet) 0.56					
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.38					
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 19.62					
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010					
APPENDIX A4

LANDFILL

F	[EL]	DI	BOR	IN	G	L(DG				H	ANSON
	CLIEN Sit Locatio	T: A te: C n: C	EG Coffe CB Mana offeen, Il	een P agem linois	ower ent F	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers		во	REHOLE ID: Well ID:	G101 G101
WF	Projec DATE EATHE	st: S: Si Fir R: O	tart: 2/2/ nish: 2/2/ vercast, c	/2010 /2010 cold () (10-30	D's)		FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	ıby		Surface Elev: Completion: Station:	625.27 ft. MSL 21.92 ft. BGS 876,551.76N 2,514,214.31E
5	SAMPL	Е	Т	EST	ING	Ĵ	TOPOGR	APHIC MAP INFORMATION:	WAT	ER LEVEL	INFORMATI	ON:
	1 (in)				/ft³)		Quadra	ngle: Coffeen, IL	Ţ	= 15.50 - 12.28	While drilling	•
	Tota ⁄ery		6 in 1e	e (%	л. (lb	Type	Section	up: East Fork 10, Tier 7N; Range 3W		$\frac{1}{2} = 12.38 - \frac{1}{2} = 7.31 - \frac{1}{2}$	3/1/2010	ion
nber	00 /	e	ws/ Valı	istur	Der	(tsf) lure	Denth	Lithologic		Borehole	Elevation	
Νur	Rec % H	Typ	Blo N - RQ	Mo	Dry	Qu Fai	ft. BGS	Description		Detail	ft. MSL	Remarks
		\backslash	, ,					TOPSOIL - Brown (10YR5/3), moist, soft, silty CLA with slight trace sand and gravel, roots.	АY 			
1A	18/24 75%	ss	2-3 N=3	24				Dark grayish brown (10YR4/2), moist, soft, silty CL/ with slight trace sand, trace roots.	AY		624	
		\forall					2	Dark grayish brown (10YR4/2) with 15% yellowish br	rown			
2A	18/24	V ss	<i>1-3</i> 3-5	30				(101 K)/4 modes, moles, modelin, sity CLA1, sig	/			
	75%	Λ	N=6					(10YR5/4) mottles, moist, medium, silty CLAY, slig	ght		622	
		(4				£ I	
2 4	19/24	V	2-3	26								
JA	79%	Å ss	4-4 N=7	20				Gray (10YR5/1) with 25% yellowish brown (10YR5, mottles, moist, medium, silty CLAY, slight trace roo	/6) ots.		- 620	
		Δ					6				E I	
		V	13									
4A	19/24 79%	ss	4-3	21							618	
		\wedge	IN-/					Gray (10YR5/1) with 10% yellowish brown (10YR5, mottles, moist, medium, silty CLAY with slight trace s	/6) sand.			
		∇					8				£ I	
5A	22/24	V ss	1-3 3-4	23								
	9270	\wedge	N=6					mottles, moist, medium, silty CLAY with trace sand a	and		616	
		$\left(\right)$					10				£ I	
	20/24	V	1-2									
6A	83%	\bigwedge	N=4	24				Gray (10YR6/1) with 35% yellowish brown (10YR5, mottles, moist, soft, silty CLAY with trace sand and silts	/6) light		614	
		()					12	trace gravel.	iigiit			
	22/24	V	1-2				I¥ ≣					
74	92%	ss	3-2 N=5	17				Gray (10YR5/1) with 25% yellowish brown (10YR5,	/6)		612	
//		Δ						gravel.	trace			
8A		\backslash		15				Brown (10YR5/3), very moist, medium, silty, clayey, fine- to coarse-grained SAND with slight trace grave	very el.			
ðВ	14/24 58%	ss	5-8	15							<u> </u>	
		\wedge	N=/				₽	Brown (10YR5/3), very moist, loose, silty, very fine-	- to		610	
		\forall						coarse-grained SAND with slight trace gravel.				
9A	16/24	ss	2-5	16]					
	67%	Λ	N=20					Brown (10YR5/3), very moist, medium dense, silty, v	very		608	
		H					18					
10.4	17/24	V	19-20	14				Brown (10VR5/3) very mojet dense silty very fine	- to			
10A 10B	71%	ss	22-18 N=42	8				coarse-grained SAND with slight trace gravel.	10		606	
	.	\square										
NC	DTE(S):	G10	1 installe	d in l	boreł	nole.						

F	ELI) F	BORI	NC	G L() G		HANSON
	CLIENT Site	: Al	EG Coffeer	1 Powe	er Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G101
	Location	: Co	offeen, Illin	iois	i uomoj		Drilling Method: 4¼" HSA w/SS samplers	Well ID: G101 Surface Elev: 625.27 ft MSL
	DATES	: St Fin	art: 2/2/2(010 010			FIELD STAFF: Driller: T. List Helper: M. Herbet/S. Hamb	Completion: 21.92 ft. BGS
WI	EATHER	: 0	vercast, col	d (lo-3	0's)	1	Eng/Geo: S. Simpson	2,514,214.31E
5	SAMPLE	2	TE	STING	<u> </u>	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION: $\mathbf{\nabla} = 15.50$. While drilling
	otal (i		in .	(%) (lb/ft ³	ype	Town	Iship: East Fork	$\underline{\Psi} = 12.38$ - Upon Completion $\overline{\Psi} = 12.38 - 24/2010$
mber	соv / Т <i>Recove</i>	þe	ws/6 Value D	y Den.	(tsf) ilure T	Depth	Ithologic	$\underline{\underline{Y}} = \frac{1.31 - 3/1/2010}{\text{Borehole}}$
Nu	Re %	L I	BIC N N	N D	Qu	ft. BGS	Description	Detail ft. MSL Remarks
11A	16/23 70%	SS	2-16 42-60/5" N=58	8			Brown (10YR5/3), slightly moist, hard, clayey SILT wit 	h / 604
							End of Boring = 21.9 ft. BGS	

F	[EL]	DI	BOR	IN	G	L	DG			H	ANSON
	CLIEN Si Locatio Proje	T: Al te: Co on: Co ct:	EG Coffe CB Mana offeen, Il	een P agem linois	ower ent F	Statio	n	CONTRACTOR: TSC Rig mfg/model: CME-650 Track Drill Drilling Method: 4¼" HSA (blind drill)	E	BOREHOLE ID: 5 Well ID: 6 Surface Elev:	SB-03a G102 (MW3S) 625.70 ft. MSL
WF	DATE	S: St Fin R: Pa	tart: 4/2 hish: 4/2 artly cloud	8/200 8/200 dy, n)6)6 nild (1	mid-60)'s)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager		Completion: Station:	17.15 ft. BGS 876,554.77N 2,514,531.48E
5	SAMPL	E	Г	EST	ING	ſ	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVE	L INFORMATIO	N:
	al (in)				o/ft³)	0)	Quad	rangle: Coffeen, IL	$\mathbf{Y} = 14.00$ $\mathbf{V} = 7.03$	- While drilling	
5	/ Tota wery		<i>6 in</i> lue	re (%	n. (It	T_{YPe}	Section	on 11, Tier 7N; Range 3W	$\underline{\underline{\Psi}} = 7.05$	- 0/1/00	
Numbe	Recov % Reco	Type	Blows / N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Boreho Detail	le Elevation ft. MSL	Remarks
1A		V	2_2	17		2.07		soil ery dark grayish brown (10YR3/2), lean CLAY			
1B	20/24 83%	ss	4-6 N=6	20		SP		Grayish brown (10YR5/2), lean CLAY, trace sand			
1C		$\left(\right)$		26		3.30 B	2	Yellowish brown (10YR5/6), lean CLAY, trace sand		624	
	17/24	V ss	2-3 4-5					Yellowish brown (10YR5/6) with 40% gray (N5/1)			
2A	/1%	Λ	N=7	25		3.05 B		mottles, lean CLAY, trace sand		622	
		$\left(\right)$				D	4	loess			
	20/24	Vss	2-3					Yellowish brown (10YR5/6) with 40% gray (N5/1)			
3A	83%	\bigwedge	N=6	16		1.96		mottles, sandy SILT, trace gravel			
		()				В	6-			620	
	24/24	V	4-3					Yellowish brown (10YR5/6) with 50% grav (N5/1)			
4A	100%	\int_{0}^{ss}	5-6 N=8	21		2.27	<u>*</u>	mottles, lean CLAY, trace sand and gravel			
		4				В	8-		//	618	
	21/24	V	1-3					Yellowish brown (10YR5/6) lean CLAY little sand tra	ice		
5A	88%	Å ss	3-4 N=6	20		2.18		gravel			
5B		Δ		19		B	10 -	Dark gray (10YR4/1), sandy CLAY, trace gravel		616	
		V	1.2								
	18/24 75%	ss	2-4 N=4	24		0.07					
6A		\square	1, 1	24		0.87 B	12	Gray (10YR6/1) with 50% yellowish brown (10YR5/8)	614	
		\mathbf{N}						mottles, lean CLAY, little sand, trace gravel			
7A	23/24 96%	X SH SS	$\begin{array}{c} 3-2\\ 2-4\\ \end{array}$	19							
7B	16/24 67%	Λ	N=4	12						612	
		∇					<u></u>	hagarstown			
8A	23/24	ss	3-12 29-50	13				Yellowish brown (10YR5/8), silty, fine SAND, little medium sand, trace gravel, wet			
8B 8C	10/0	\wedge	N=41	13				Yellowish brown (10YR5/8), silty, fine SAND, little cla		610	
80	12/12	\forall	0.03	12			16	wet Yellowish brown (10YR5/6). lean CLAY. little sand. tra			
9A	100%	$\int_{-\infty}^{\infty}$	8-82	10		6.98 B		gravel Yellowish brown (10YR5/6), silty SAND, trace gravel			
	I		I	1			. –	Vellowish brown (10YR5/4), clavev SILT, trace sand ar	nd		
								End of Boring = 17.15 ft. BGS	/		
NO	NOTE(S) , $C(10)$ (MW02S) installed in blind drilled barehold within 10 \oplus -SCD 02										
	/1E(3):	010	2 (IVI W U	, с) п	istall		iniu-ui illeo	i borenoie within 10 it of 5D-05.			

F	EL	DI	BOR	IN	IG	L()G						ANSON
	CLIEN Sit Locatio Projec DATE	T: Al te: Co on: Co ct: XS: St Fin	EG Coffe CB Mana offeen, II tart: 2/1: nish: 2/1	een P agem linois 5/20 5/20	Power ent F s 10 10	Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Han	ıby		BOF	REHOLE II Well II Surface Ele Completion Station	b: G103 b: G103 v: 627.94 ft. MSL n: 18.03 ft. BGS n: 876,199.48N
WE	EATHE SAMDI	R: Co	old, snow	ry (lo	-20's)		Eng/Geo: S. Simpson	1				2,514,501.19E
er	/ Total (in)		/ 6 in alue	re (%)	en. (lb/ft ³)	f) e Type	TOPOGRA Quadra Townsh Section	APHIC MAP INFORMATION: ngle: Coffeen, IL ip: East Fork 10, Tier 7N; Range 3W	WA	FER LE $\mathbf{\Psi} = \mathbf{I}$ $\mathbf{\Psi} = \mathbf{I}$ $\mathbf{\Psi} = \mathbf{I}$	VEL I Dry - N .99 - 3	NFORMA While drilling 3/1/2010	r ion: ^g
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		Bore De	hole tail	Elevation ft. MSL	Remarks
1A	20/24 83%	ss	5-4 5-7 N=9	16			2	FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a	and				
2A	16/24 67%	ss	3-4 6-7 N=10	25			4	5% dark yellowish brown (10YR4/6) mottles, moist, s silty CLAY with trace sand and slight trace gravel.	stiff,				
3A	17/24 71%	ss	3-5 7-8 N=12	21			6	Grayish brown (10YR5/2) with 25% yellowish brow (10YR5/6) mottles, moist, stiff, silty CLAY with slig trace sand, slight trace roots.	vn ght				
4A	16/24 67%	ss	2-3 5-6 N=8	24			8	Grayish brown (10YR5/2) with 25% yellowish brow (10YR5/6) mottles, moist, medium, silty CLAY with s trace sand.	vn slight			 	
5A	24/24 100%	ss	3-4 4-4 N=8	22			10	Gray (10YR5/1) with 15% yellowish brown (10YR5 mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	5/6) and	-			
6A	20/24 83%	ss	<i>1-1</i> <i>3-3</i> N=4	23			12					 616	
7A	22/24 92%	ss	1-2 2-4 N=4	26			14 -	Gray (10YR5/1) with 15% yellowish brown (10YR5 mottles, soft, medium, silty CLAY with trace sand a slight trace gravel.	5/6) ind			 614	
8A	20/24 83%	ss	1-2 3-3 N=5	22			16					612	
9A 9B	22/24 92%	ss	1-10 21-33 N=31	14 10			⊻ _	Brown (10YR5/3) with 15% gray (10YR6/1) mottles, moist, medium dense, silty, very fine- to medium-grai SAND. Brown (10YR5/3), slightly moist, hard, clayey SILT v trace sand and gravel. End of Boring = 18.0 ft. BGS	very ined with				
NO)TE(S):	Well	complet	ed pi	rior to	o const	ruction of ber	m road. Boring surface elevation is as of the well install	l date ar	nd not the	e final o	constructed	elevation. Page 1 of 1

FI WE	CLIENT: AEG Coffeen Power Sta Site: CCB Management Facil Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 10/8/2010 Finish: 10/8/2010 WEATHER: Sunny, mid) G	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA (blind drill) FIELD STAFF: Driller: D. Mahurin Helper: J. Litsch/D. Smail Eng/Geo: R. Hasenyager	BO	DREHOLE II Well II Surface Elev Completion Station	ANSON 2: R104 2: R104 4: 629.03 ft. MSL 4: 19.85 ft. BGS 4: 875,857.80N 2,514,503.41E
S	AMPL	E		EST	INC	;	TOPOGR	APHIC MAP INFORMATION:			
	al (in				(,ff ³)	(tsf)	Quadra	ingle: Coffeen, IL			
<u>ь</u>	' Tot: wery		6 in ue	re (%	n. (It	$Q_{\rm T}^{\rm D}$	Section	10, Tier 7N; Range 3W			
mbe	cov / Reco	pe	D 2D	oistur	y De	i (tsf ilure	Depth	Lithologic	Borehole	e Elevation	
Ŋ	Re %	Ty	R N Blo	Ă	<u>D</u>	-Qu Fa	ft. BGS	Description	Detail	ft. MSL	Remarks
							2	FILL - Brown (10YR5/3) with 5% gray (10YR5/1) and 5% dark yellowish brown (10YR4/6) mottles, moist, very stiff silty CLAY with slight trace sand and gravel.	۵. <u>۲. و. و. و. و. ار ار ار ار ار ار ار ار ار ار ار ار ار </u>	628	
							6	Grayish brown (10YR5/2) with 15% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY with slight trace sand, slight trace roots.		624	
								Gray (10YR5/1) with 20% yellowish brown (10YR5/6) mottles, moist, medium, silty CLAY with slight trace sand		622	
							14	Gray (10YR5/1) with 20% yellowish brown (10YR5/6) mottles, moist, soft, silty CLAY with trace sand and slight trace gravel.		616	
								Gray (10YR6/1) with 10% yellowish brown (10YR5/6) mottles, moist, soft, silty CLAY with trace sand and slight trace gravel.		614	
								Gray (10YR6/1), moist, very soft, clayey, very fine- to coarse-grained SAND with trace gravel			
							18	Brown (10YR5/3), slightly moist, hard, very silty CLAY with trace sand and gravel.		612	
								Gray (10YR4/1), slightly moist, hard, very silty CLAY with sand and gravel.		610	
			I				·	End of Boring = 19.85 ft. BGS			
NC	TE(S):	R104	4 blind dr	illed	l in b	orehole	approximate	ely 8 ft. north of G104. Lithology taken from G104.			D 1 . 6 1

F	EL	DI	BOR	IN	JG	L)G		HANSON
WE	CLIEN Si Locatio Proje DATE	T: A te: C on: C ct: (S: St Fin R: O	EG Coffe CB Mana offeen, Ill art: 2/1: ish: 2/1: vercast, c	een P igem linois 5/201 5/201 sold,	ower ent F 3 10 10 wind	Y Statio acility y (lo-2	n 0's)	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	BOREHOLE ID: G104 Well ID: G104 Surface Elev: 627.96 ft. MSL Completion: 20.00 ft. BGS Station: 875,849.26N 2,514,504.98E
S	SAMPL	E	Т	EST	ING	r	TOPOGE	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	1 (in)				(ft ³)		Quadı	angle: Coffeen, IL	\mathbf{Y} = Dry - While drilling
	Tota very		6 in Le	e (%	J. (]p	Type	Section	n 10, Tier 7N; Range 3W	$\underline{\Psi} = $ $\underline{\nabla} = 15.40 - 3/1/2010$
mbeı	cov /	be	vws / Vali	istur	y Dei	(tsf) ilure	Depth	Lithologic	Borehole Elevation
Nu	Re %	L _V	Blc N- R(Ŭ	Dr.	Qu_{Fa}	ft. BGS	Description	Detail ft. MSL Remarks
1A	16/24 67%	ss	9-11 12-14 N=23	16			2	FILL - Brown (10YR5/3) with 5% gray (10YR5/1) and dark yellowish brown (10YR4/6) mottles, moist, very s	5% tiff,626
2A	20/24 83%	ss	3-5 9-12 N=14	21			4	silty CLAY with slight trace sand and gravel.	624
3A	19/24 79%	ss	3-0 8-9 N=14	21			6	Grayish brown (10YR5/2) with 15% yellowish brow (10YR5/6) mottles, moist, stiff, silty CLAY with slig trace sand, slight trace roots.	n ht 622
4A	17/24 71%	ss	2-3 4-7 N=7	25			8		620
5A	22/24 92%	ss	1-2 5-8 N=7	24			10	Gray (10YR5/1) with 20% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with slight trace s	6) and
6A	18/24 75%	ss	1-2 3-5 N=5	21			12		
7A	23/24 96%	ss	woh-2 3-4 N=5	23			14 —	Gray (10YR5/1) with 20% yellowish brown (10YR5/ mottles, moist, soft, silty CLAY with trace sand and sli trace gravel.	6) ght 614
8A	22/24 92%	ss	<i>1-3</i> <i>3-3</i> N=6	24			⊻	Gray (10YR6/1) with 10% yellowish brown (10YR5/ mottles, moist, soft, silty CLAY with trace sand and sh trace gravel.	6) ght 612
9A		\backslash	1.6	16				Gray (10YR6/1), moist, very soft, clayey, very fine- coarse-grained SAND with trace gravel.	
9B	17/24 71%	ss	won-o 27-40 N=33	13			18	Brown (10YR5/3), slightly moist, hard, clayey SILT w trace sand and gravel.	$\begin{array}{c c c c c c c c c c c c c c c c c c c $
10A	22/24 92%	ss	10-24 44-66 N=68	7			20	Gray (10YR4/1), slightly moist, hard, clayey SILT wi sand and gravel.	th
NO)TE(S):	Well	complete	ed pr	ior to	o consti	ruction of b	End of Boring = 20.0 ft. BGS erm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.

F	ELIEN CLIEN Sit Locatio Projec DATE	D I F : Al f : Co n : Co e t: S : St F in R : Su	BOR EG Coffe CB Mana offeen, II cart: 2/10 ish: 2/11 inny, colo	een P agem linois 6/201 6/201 d, wi	NG owen ent F s 10 10 ndy (Statio Facility	DG n D's)	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ /4" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	BOREHOLE ID: G105 Well ID: G105 Surface Elev: 626.86 ft. MSL Completion: 19.83 ft. BGS by Station: 875,499.70N 2,514,509.15E WATER LEVEL INFORMATION:			
L	Total (in)	E	6 in ue	re (%) a	n. (lb/ft ³)	Type	TOPOGR Quadr Towns Section	EAPHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork n 10, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi =$ Dry - While drilling $\Psi =$ $\overline{\Psi} = 16.08 - 3/1/2010$			
Numbe	Recov / % Reco	Type	Blows / N - Val RQD	Moistur	Dry De	Qu (tsf. Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	18/24 75%	ss	4-4 4-7 N=8	20			2					
2A	19/24 79%	ss	2-4 6-10 N=10	19				FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a 5% dark yellowish brown (10YR4/6) mottles, moist medium, silty CLAY with slight trace sand and grave	and st, rel. 624			
3A	19/24 79%	ss	2-4 5-6 N=9	29			6	Grayish brown (10YR5/2) with 40% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY, slight tra	rown ace622			
4A	20/24 83%	ss	<i>l-4</i> <i>4-5</i> N=8	29			8	Gray (10YR5/1) with 30% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with slight trace s	5/6) sand.			
5A	22/24 92%	ss	<i>1-3</i> <i>3-3</i> N=6	20			10	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with slight trace s	5/6) sand.			
6A	24/24 100%	ss	1-2 3-4 N=5	25			12	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles moist medium silty CLAY with trace sand a	5/6) and			
7A	24/24 100%	ss	1-2 3-5 N=5	22			14 –	slight trace gravel.				
8A	19/24 79%	ss	1-2 2-2 N=4	18				Gray (10YR6/1) with 15% yellowish brown (10YR5/ mottles, moist, soft, silty CLAY with trace sand and gravel.	5/6) d			
9A	($\overline{\mathbf{V}}$	1.2	18			⊻ 16-	Gray (10YR5/1), very moist to wet, loose, silty, fine- coarse-grained SAND with slight trace gravel.	- to			
9B	18/24 75%	ss	8-10 N=10	14				Brown (10YR5/3), very moist to wet, loose, silty, ver fine- to coarse-grained SAND with slight trace grave	ery el.			
10.4	19/22	∇	11-40	. 7			18	Brown (10YR5/3), slightly moist, hard, clayey SILT w				
IUA	86%	A 35	N=93					Gray (10 r K4/1), sugnty moist, hard, clayey SIL1 wi sand and gravel.				
NC)TE(S):	Well	complet	ed pr	rior to	o const	ruction of be	erm road. Boring surface elevation is as of the well install	l date and not the final constructed elevation.			
									Page 1 of 1			

FI	EL	DI	BOR	IN	JG	L(DG		HANSON
	CLIEN Sit Locatio Projec DATE	T: Al te: Co on: Co ct: S: St Fin	EG Coffe CB Mana offeen, Il art: 2/10 ish: 2/1	een P agem linois 6/201	ower ent F s 10	Statio Cacility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Har	BOREHOLE ID: G106 Well ID: G106 Surface Elev: 625.96 ft. MSL Completion: 18.00 ft. BGS station: 875,149.76N
WF	CATHE	R: O	vercast, c	old,	wind	ly (mid	-20's)	Eng/Geo: S. Simpson	2,514,512.79E
er	/ Total (in) [WS	E	/6 in alue	EST (%) and	en. (lb/ft ³) NI	f) e Type	TOPOGR Quadr Towns Section	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork h 10, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = $ Dry - While drilling $\Psi = $ $\overline{\Psi} = $ $\overline{\Psi} = $ 12.62 - 3/1/2010
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	17/24 71%	ss	7-8 13-16 N=21	18			2	FILL - Brown (10YR5/3) with 15% gray (10YR5/1)	and
2A	16/24 67%	ss	2-4 5-7 N=9	16			4	5% dark yellowish brown (10 y K4/6) mottles, moist, silty CLAY with trace sand and slight trace gravel	
3A	17/24 71%	ss	2-4 6-7 N=10	21			6	Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY with sli trace sand, slight trace roots.	ght 620
4A	20/24 83%	ss	2-3 5-6 N=8	24			8	Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, medium, silty CLAY with trace sand.	slight 618
5A	22/24 92%	ss	<i>1-3</i> <i>3-5</i> N=6	21			10	Gray (10YR5/1) with 10% yellowish brown (10YR5 mottles moist medium silty CLAY with trace sand	5/6) and
6A	22/24 92%	ss	1-2 4-4 N=6	24			12 -	slight trace gravel.	- 614
7A	23/24 96%	ss	1-3 4-4 N=7	22			⊻	Yellowish brown (10YR5/6) with 10% gray (10YR6 mottles, moist, medium, silty CLAY with trace sand slight trace gravel.	5/1) and 612
8A 8B	19/24 79%	ss	<i>1-4</i> 7-7 N=11	20 14				Gray (10YR6/1) with 10% yellowish brown (10YR5/ mottles, moist, soft, silty CLAY with trace sand. Brown (10YR5/3), moist, medium dense, clayey, vo fine- to medium-grained SAND with slight trace gra Yellowish brown (10YR5/6) with 5% gray (10YR6 mottles, moist, medium, clayey SILT with trace sand	5/6)
9A	23/24 96%	ss	7-15 22-22 N=37	8			18	Gray (10YR5/3), slightly moist, hard, clayey SILT trace sand and gravel. Gray (10YR4/1), slightly moist, hard, clayey SILT v trace sand and gravel. End of Boring = 18.0 ft. BCS	with //
NO)TE(S):	Well	complet	ed pr	ior to	o const	ruction of b	erm road. Boring surface elevation is as of the well instal	l date and not the final constructed elevation. Page 1 of 1

F	EL clien	D] T: A	BOR EG Coffe	een P	lower ower	Statio	DG	CONTRACTOR: Layne-Western Co Big mfr/model: CME 750 ATV Drill		
	Locatio	n: C	offeen, Il	linoi	ent r S	aciiity		Drilling Method: 4/4" HSA w/SS samplers	DU	Well ID: G107
WI	Proje DATE EATHE	ct: S: Si Fir R: O	tart: 2/1 nish: 2/1 vercast, c	7/20 7/20 cold,	10 10 wind	y (mid	-20's)	FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Hamt Eng/Geo: S. Simpson	уy	Surface Elev: 627.11 ft. MSL Completion: 20.00 ft. BGS Station: 874,994.33N 2,514,358.25E
5	SAMPL	Æ	1	TEST	ING	r	TOPOGE	RAPHIC MAP INFORMATION:	WATER LEVEL	INFORMATION:
	otal (in)		in	(%)	(lb/ft^3)	ədi	Quadı Towns	rangle: Coffeen, IL ship: East Fork	$\underline{\Psi} = 16.80 - $ $\underline{\Psi} = 11.56 - $	While drilling Upon Completion
ber	v / T		vs / 6 Value	sture	Den.	tsf) tre T_{J}	Section	n 10, Tier 7N; Range 3W	$\underline{\vee} = 10.40$ -	3/1/2010
Num	Recc % Re	Type	Blow N - V RQI	Mois	Dry	Qu (Faili	Depth ft. BGS	Lithologic Description	Borehole Detail	Elevation ft. MSL Remarks
1.4	0/18 0% 5/6	BD Xss	2	12				GRAVEL FILL		626
2A	14/24 58%	ss	3-3 6-8 N=9	12			2	FILL - Brown (10YR5/3) with 10% gray (10YR5/1) at 5% dark yellowish brown (10YR4/6) mottles, moist, st	nd iff,	624
	16/04	$\left(\right)$	3-5				4	Silty CLAY with trace sand and slight and gravel.		
3A	16/24 67%	ss	5-7 N=10	20			6	Grayish brown (10YR5/2) with 30% yellowish brown		622
4A	18/24 75%	ss	2-3 6-7 N=9	27			8	(10ÝR5/6) mottles, moist, stiff, silty ĆLAY with slig trace sand and gravel.	nt	620
5A	22/24 92%	ss	2-3 4-5 N=7	18						618
6A	22/24 92%	ss	1-3 3-4 N=6	23			¥ ¥ 12	Gray (10YR5/1) with 10% yellowish brown (10YR5// mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	b) nd	616
7A	24/24 100%	ss	1-2 3-4 N=5	27						614
8A	2/24	↓ ↓ ss	1-3 2-3	23			14	Gray (10YR6/1) with 35% yellowish brown (10YR5/0 mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	5) nd	612
9A		$\left(\right)$	N=5	22			16 -	Brown (10YR5/3), very moist to wet, very loose, claye very fine- to coarse-grained SAND.	y,	
9B	20/24 83%	ss	woh-3 8-12 N=11	10			⊻	Brown (10YR5/3), slightly moist, stiff, clayey SILT w trace sand and gravel.	th	610
10.4	20/24	\forall	8-25	11			18	Brown (10YR5/3), slightly moist, hard, clayey SILT w	 ith +	
10A	83%	ss	26-58 N=51					Gray (10YR4/1), slightly moist, hard, clayey SILT with trace sand and gravel.	h	608
	I		1	I	1	I	20 -	End of Boring = 20.0 ft. BGS		
NC)TE(S):	Well	l complet	ed pi	ior to	o const	ruction of b	erm road. Boring surface elevation is as of the well install of	late and not the final	constructed elevation.

FIELD BORING I CLIENT: AEG Coffeen Power Sta Site: CCB Management Facil Location: Coffeen, Illinois Project: DATES: Start: 2/12/2010 Finish: 2/12/2010 WEATHER: Overcast, cold ~25F SAMPLE TESTING							n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Hamt Eng/Geo: D. Lamb	BOREHOLE ID: G108 Well ID: G108 Surface Elev: 625.58 ft. MSL Completion: 20.00 ft. BGS Station: 874,948.81N 2,514,248.25E
5	SAMPL	Е	Т	EST	INC	3	TOPOGI	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
er	- / Total (ii :overy		/ 6 in alue	ure (%)	en. (lb/ft ³)	f) 'e Type	Quadi Towns Sectio	rangle: Coffeen, IL ship: East Fork n 10, Tier 7N; Range 3W	$\underline{\Psi}$ = 14.50 - While drilling $\underline{\Psi}$ = 19.00 - Upon completion $\underline{\nabla}$ = 8.93 - 3/1/2010
quin	Recov % Rec	Type	Blows N - V; RQD	Moisti	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	23/24 96%	ss	24-25 13-13 N=38	16			2	FILL - Grayish brown (10YR5/2) with 10% yellowis brown (10YR5/6) mottles, moist, hard, silty CLAY wi slight trace gravel.	h ith
2A	13/24 54%	ss	4-5 8-11 N=13	26			4	FILL - Grayish brown (10YR5/2) with 10% yellowisl brown (10YR5/6) mottles, moist, stiff, silty CLAY wi trace sand and slight trace gravel.	h 622
3A	20/24 83%	ss	2-2 5-7 N=7	28			6	Grayish brown (10YR5/2) with 25% yellowish brown (10YR5/8) and 5% very dark brown (10YR2/2) mottle moist, medium, clayey SILT with trace sand and sligh trace gravel.	n
4A	24/24 100%	ss	2-3 5-6 N=8	18			8		618
5A	23/24 96%	ss	1-2 3-4 N=5	20			∑ 10	Gray (10YR5/1) with 10% brownish yellow (10YR6/3 mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	8) nd 616
6A	24/24 100%	ss	<i>1-3</i> <i>3-5</i> N=6	19			12	Grayish brown (10YR5/2) with brownish yellow (10YR6/8) mottles, moist, soft, sandy CLAY with trac gravel	614
7A	19/24 79%	ss	<i>1-1</i> <i>1-2</i> N=2	19			14	Brownish yellow (10YR6/8), very moist, soft, sandy CLAY with trace gravel. Brownish yellow (10YR6/8), very moist, soft, sandy SI	
8A 8B	23/24 96%	ss	2-4 7-10 N=11	19 13			Ţ	with trace gravel. Brownish yellow (10YR6/6), wet, medium, SILT.	
9A	22/24 92%	ss	10-24 25-10 N=49	11			16	Light yellowish brown (10YR6/4) with 10% yellowis brown (10YR5/6) mottles, moist, stiff, clayey SILT wi trace sand and gravel.	h = - -
9B		()		8			18-	Grav (10YR5/1), slightly moist hard SILT with grave	
10A	24/24	V	10-25	10				Gray (10YR5/1), wet, hard, SILT with sand and grave	= + + a.
	100%	ss ss	40-40 N=65				20	Gray (10YR5/1), very moist, hard, SILT with gravel End of Boring = 20.0 ft. BGS	· 606

NOTE(S): Well completed prior to construction of berm road. Boring surface elevation is as of the well install date and not the final constructed elevation.

F	[EL]	DI	BOR	I	NG	L C	DG		C.	
	CLIEN Sit Locatio Projec DATE	T: A te: C on: C ct: S: St E:-	EG Coffe CB Mana offeen, II tart: 2/1	een P agem linois 1/20	Power nent F s 10	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List	BOREHO Surfa Con	DLE ID: G109 Well ID: G109 ace Elev: 624.79 ft. MSL apletion: 18.00 ft. BGS Station: 874 070 10N
WE	EATHE	rii R: Si	unny, col	1/20 d ~32	10 2F			Eng/Geo: D. Lamb	lby	2,514,137.84E
er 55	/Total (in) Very	E	/6 in lue	rest (%) er	en. (lb/ft ³)	Type	TOPOG Quad Town Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 10, Tier 7N; Range 3W	WATER LEVEL INFO $\Psi = 14.20$ - While $\Psi = 11.50$ - Upon $\overline{\Psi} = 8.85 - 3/1/20$	RMATION: drilling completion 110
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (tsi Failur	Depth ft. BGS	Lithologic Description	Borehole Elev Detail ft. 1	vation MSL Remarks
1A	24/24 100%	ss	6-7 7-8 N=14	22			2	FILL - Grayish brown (10YR5/2) with 40% yellowi brown (10YR5/6) mottles, moist, stiff, silty CLAY w slight trace gravel.	sh /ith	24
2A	19/24 79%	ss	3-5 5-6 N=10	27						22
3A	20/24 83%	ss	2-5 6-8 N=11	24			4	Light yellowish brown (10YR6/4) with 50% browni yellow (10YR6/8) mottles, moist, medium, silty CLA slight trace roots.	sh .Y, 6	20
4A	24/24 100%	ss	2-4 5-6 N=9	19			6 	Light brownish gray (10YR6/2) with 10% brownish ye (10YR6/6) mottles, moist, medium, silty CLAY with s trace sand and gravel.	ellow light	18
5A	22/24 92%	ss	2-3 4-5 N=7	20			∑ 10	Light brownish gray (10YR6/2) with 10% brownish ye (10YR6/6) and 2% very dark gray (10YR3/1) mottle moist, medium, silty CLAY with slight trace sand a	sllow	16
6A	24/24 100%	ss	1-3 3-4 N=6	19			Ā	gravel.		14
7A	23/24 96%	ss	1-1 2-2 N=3	19			12 — — — — —	Light brownish gray (10YR6/2) with 30% brownish ye (10YR6/8) mottles, moist, medium, sandy CLAY w slight trace gravel.	ellow ith 6	12
8A	22/24 92%	ss	8-15 15-21 N=30	14				Brownish yellow (10YR6/6), wet, dense, silty SAND trace gravel.	with 6	10
	24/24	↓ ss	12-29 44-45				16	Brownish yellow (10YR6/8), wet, dense, SAND with gravel.	trace	08
9A	100%	\land	N=73	7			18	Grayish brown (10YR5/2), slightly moist, hard, grave SILT with sand.		
NC)TE(S):	Well	complet	ed pi	rior to	o const	ruction of l	erm road. Boring surface elevation is as of the well install	date and not the final const	ructed elevation.

FIEI	D]	BOR	IN	JG	L(DG		
CLIE S Locat Proj DAT	NT: A Site: C ion: C ject: `ES: S	EG Coffe CB Mana offeen, Il tart: 2/1	een P agem linois 1/201	Power ent F s	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List	BOREHOLE ID: G110 Well ID: G110 Surface Elev: 624.81 ft. MSL Completion: 18.00 ft. BGS
WEATH	Fu ER: S	unny, col	1/20 d 10-	10 •20F			Helper: M. Herbst/S. Ham Eng/Geo: D. Lamb	by Station: 8/5,015.42N 2,514,057.73E
SAMP	PLE	T	TEST	TING	3	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
al (in			(0)	b/ft ³)	e	Quad Town	rangle: Coffeen, IL ship: East Fork	$\mathbf{Y} = 15.00$ - While drilling $\mathbf{Y} =$
er / Tot overv		/ 6 in Ilue	ure (%	en. (1	f) e Typ	Sectio	n 10, Tier 7N; Range 3W	$\overline{\underline{\nabla}}$ = 9.50 - 3/1/2010
Numb Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A 24/24 1005 2A 14/24 589	4 56 4 58 4 6 58 58 58 58 58 58 58 58 58 58	8-6 6-8 N=12 2-4 5-7 N=9	17 25			2	FILL - Yellowish brown (10YR5/6) with 20% light brownish gray (10YR6/2) mottles, moist, stiff, silty CL with slight trace sand.	AY
22/24 929 3A	4 ss	1-4 5-9 N=9	22			4		620
4A	4 Ss	3-6 8-9 N=14	18			6	Grayish brown (10YR5/2) with 20% yellowish brow (10YR5/6) and 5% very dark brown (10YR2/2) mottl moist, medium, silty CLAY.	n es, 618
5A	ss	1-3 4-6 N=7	21			∑ 10	Grayish brown (10YR5/2) with 20% yellowish brow (10YR5/6) and 5% very dark brown (10YR2/2) mottl moist, medium, silty CLAY with slight trace sand an gravel.	n es, d
6A 24/24 1009	ss	2-4 4-6 N=8	21			12	Light brownish gray (10YR6/2) with 30% brownish ye (10YR6/8) mottles, moist, medium, clayey SILT with t sand and gravel.	llow race
7A 24/24 1009	4 2% ss	1-2 3-3 N=5	22				Light brownish gray (10YR6/2) with 30% brownish ye (10YR6/8) mottles, very moist, medium, sandy CLA with trace gravel.	llow Y 612
8A 19/24		1-2 2-1	24			14 -	Light brownish gray (10YR6/2) with 30% brownish ye (10YR6/8) mottles, moist, soft, clayey SILT with trac sand and slight trace gravel.	llow ce 610
/9%		N=4				16 -	Gray (10YR6/1) with 30% brownish yellow (10YR6/ mottles, wet, soft, sandy CLAY.	8)
9A 24/24 1009	ss	7-26 49-60 N=75	6				Grayish brown (10YR5/2), slightly moist, hard, claye SILT with trace sand and gravel.	≥y = = 608
		I	1	I	I	18 -	End of Boring = 18.0 ft. BGS	
NOTE(S): Wel	l complet	ed pr	ior to	o const	ruction of l	perm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.

FIELD BORING	LOG	CAR HANSON
CLIENT: AEG Coffeen Powe Site: CCB Management I Location: Coffeen, Illinois Project: DATES: Start: 2/10/2010 Einich: 2/11/2010	r Station CONTRACTOR: Layne-Western Co Facility Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helmon: M. Herbert/S. Hembri	BOREHOLE ID: G111 Well ID: G111 Surface Elev: 625.28 ft. MSL Completion: 18.00 ft. BGS Station: 875.058 70N
WEATHER: Sunny, breezy ~25F	Eng/Geo: D. Lamb	2,513,981.72E
$\begin{array}{c c} \text{cr} \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ \hline T \\ T \\$	TOPOGRAPHIC MAP INFORMATION: V Quadrangle: Coffeen, IL Quadrangle: East Fork Section 10, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\mathbf{\nabla} = 14.50$ - While drilling $\mathbf{\nabla} = \mathbf{\nabla} = 10.50 - 3/1/2010$
Numb _n Recov % Rec 7 ye Blows N - V _z RQD Moistt Dry D	$ \begin{array}{c} \underline{s} \\ \underline{s} \\ \overline{o} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline{s} \\ \underline$	Borehole Elevation Detail ft. MSL Remarks
1A 20/24 ss 7-7 83% N=13 18	2	624
$2A \begin{vmatrix} 13/24 \\ 54\% \end{vmatrix} \begin{vmatrix} ss \\ N = 12 \end{vmatrix} ss \begin{vmatrix} 3-5 \\ 7-8 \\ N = 12 \end{vmatrix} 20$	FILL - Grayish brown (10YR5/2) with 10% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY with slight trace sand and gravel.	622
$3A \begin{vmatrix} 18/24 \\ 75\% \\ 18/26 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 100 \\ 1$		620
$4A \begin{vmatrix} 16/24 \\ 67\% \\ 4B \end{vmatrix} ss \begin{vmatrix} 4-12 \\ 20-17 \\ N=32 \end{vmatrix} 18$	Grayish brown (10YR5/2) with 20% yellowish brown (10YR5/6) and 5% dark brown (10YR3/3) mottles, moist medium, clayey SILT with slight trace sand, slight trace roots.	t, 618
$5A \begin{vmatrix} 22/24 \\ 92\% \end{vmatrix} \\ ss \begin{vmatrix} 2-3 \\ 4-5 \\ N=7 \end{vmatrix} \\ 21$	Grayish brown (10YR5/2) with 5% yellowish brown (10YR5/6) mottles, moist, medium, clayey SILT with trac sand.	xe
$6A \begin{array}{c} 24/24 \\ 100\% \end{array} \right ss \begin{array}{c} 2-3 \\ 6-6 \\ N=9 \end{array} 23$	Grayish brown (10YR5/2) with 5% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY with slight trace sand.	614
7A $24/24$ ss $1-4$ 5-6 N=9 20	Crasich berur (10VD5/2) with 200/ willweich brown	612
	$ \begin{array}{c} \text{Grayish brown} \\ 14 \\ \blacksquare \end{array} \begin{array}{c} \text{Grayish brown} \\ (10YR5/8) \text{ mottles, moist, stiff, clayey SILT with trace} \\ \text{sand and gravel.} \end{array} $	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Yellowish brown (10YR5/8), wet, soft, clayey SAND wit trace gravel.	h 610
9A $18/18$ 100% ss $12-50$ 66 N=116 7	Grayish brown (10YR5/2) with 20% brownish yellow (10YR6/6) mottles, slightly moist, hard, clayey SILT with gravel.	h
0% П _{вр}	18 <u>=</u> End of Boring = 18.0 ft. BGS	
NOTE(S): Well completed prior t	o construction of berm road. Boring surface elevation is as of the well install da	te and not the final constructed elevation. Page 1 of 1

F	[EL]	DI	BOR	IN	JG	L	DG				ANSON
WE	CLIEN Sit Locatio Projec DATE EATHEI	Γ: Α α: Ο n: Ο ct: S: St Fir R: Ο	EG Coffe CB Mana offeen, II tart: 2/9, nish: 2/9 old, snow	een P agem linois /2010 /2010 /2010	ower ent F s))) ndy (Statio acility	n)	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	E	OREHOLE II Well II Surface Ele Completion Station	D: G119 D: G119 v: 626.57 ft. MSL n: 20.10 ft. BGS n: 875,675.04N 2,513,907.73E
5	SAMPL	E	ſ	EST	ING	ì	TOPOGR	APHIC MAP INFORMATION.	WATER LEVE	L INFORMA	
er	/Total (in) overy		/ 6 in alue	ıre (%)	en. (lb/ft ³)	f) e Type	Quadr Towns Section	angle: Coffeen, IL hip: East Fork 1 10, Tier 7N; Range 3W		- While drilling	g
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moisti	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Boreho Detail	le Elevation ft. MSL	Remarks
1A	16/24 67%	ss	3-2 3-3 N=5	21			2			-626	
2A	16/24 67%	ss	2-4 5-5 N=9	20				FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a 5% yellowish brown (10YR5/6) mottles, moist, mediu silty CLAY with trace sand and slight trace gravel.	nd im,	624	
3A	10/24 42%	ss	2-5 5-6 N=10	25						-622	
4A	17/24 71%	ss	2-5 4-5 N=9	18			8	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with trace sand	6) L	- 620	
5A	20/24 83%	ss	2-3 4-5 N=7	19			10	· · · · · · · · · · · · · · · · · · ·		- 618	
6A	19/24 79%	ss V	1-3 4-6 N=7	21			∑ 12	Gray (10YR5/1) with 10% yellowish brown (10YR5, mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	6) nd	616	
7A	20/24 83%	ss	7-3 3-5 N=6	20			14	Gray (10YR6/1) with 30% yellowish brown (10YR5. mottles, moist, medium, silty CLAY with trace sand a gravel.	6) nd		
8A	18/24 75%	ss	1-2 2-2 N=4	18			16	Gray (10YR6/1) with 30% yellowish brown (10YR5/ mottles, moist, soft, sandy CLAY with slight trace gra	6) vel.	- 612	
9A 9B	22/24 92%	ss	3-9 13-16 N=22	13 12			10 mmfmpfmmmp 18 pm	Brown (10YR5/3), moist, medium, silty CLAY with the sand and slight trace gravel. Brown (10YR5/3), moist, medium dense, silty, very fine-grained SAND. Brown (10YR5/3), slightly moist, hard, clayey SILT we trace sand and gravel.	ith	610	
10A	17/24 71%	ss	6-25 33-39 N=58	8			20	Dark gray (10YR4/1), slightly moist, hard, clayey SII with trace sand and gravel.	Л	608	
								End of Boring = 20.1 ft. BGS			
NC)TE(S):	Well	complet	ed pr	ior to	o const	ruction of b	erm road. Boring surface elevation is as of the well install	date and not the fir	nal constructed of	elevation.

ſ

l

F	EL	DI	BOR	IN	١G	L(DG						ANSON		
	CLIEN Si Locatio	T: A te: C on: C	EG Coffe CB Mana offeen, II	en P gem linois	'ower ient F s	station acility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers			BC	DREHOLE II Well II	D: G120 D: G120		
WF	Proje DATE EATHE	et: S: St Fir R: C	art: 2/8/ nish: 2/8/ old, snow	/2010 /2010 v, wii) 0 ndy (mid-20	l's)	FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	by			Surface Ele Completion Station	v: 627.21 ft. MSL n: 20.00 ft. BGS n: 875,854.43N 2,513,905.84E		
5	SAMPL	Æ	T	EST	ING	, ,	TOPOGRA	PHIC MAP INFORMATION:	WAT	VATER LEVEL INFORMATION:					
	ul (in)				/ft ³)		Quadran	ngle: Coffeen, IL	7	<u> </u>	Dry -	While drillin	g		
H	/ Tota wery		/ 6 in lue	re (%	en. (Ib) Type	Section 1	10, Tier 7N; Range 3W	7	$\underline{\underline{V}} = 1$	3.85 -	3/1/2010			
Numbe	Recov % Reco	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description		Boi D	rehole etail	Elevation ft. MSL	Remarks		
1A	20/24 83%	ss	3-3 4-5 N=7	16								626			
		$\left(\right)$	3_4				2-	FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a 5% yellowish brown (10YR5/6) mottles, moist, mediu	and um,						
2A	19/24 79%	ss	7-9 N=11	20			4	slity CLAY with trace sand and slight trace gravel.				624			
3A	16/24	V	2-3	26											
	67%	$\left(\begin{array}{c} \\ \\ \\ \end{array} \right)^{33}$	N=7				6-	Gray (10YR5/1) with15% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY with slig trace sand and gravel.	,ht			- 622			
4A	17/24 71%	ss	3-5 6-6 N=11	24				Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY with slig trace sand and gravel, slight trace roots.				620			
5A	14/24 58%	ss	2-4 4-6 N=8	22			8	Gray (10YR5/1) with15% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY with slig trace sand.		-		618			
	22/24	$\left(\right)$	1-3				10			-					
6A	92%		3-4 N=6	23			12 -	Gray (10YR5/1) with brownish yellow (10YR6/6) mot moist, medium, silty CLAY with trace sand and sligh trace gravel.	tles, ht			616			
7A	22/24 92%	ss	1-2 3-5 N=5	21						-		614			
8A	19/24 79%	↓ ss	woh-2 3-3	25			14	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	/6) and						
		Δ	N=5				16-								
94	24/24	V	woh-3	21				Brown (10YR5/3), very moist, soft, clayey SAND wi slight trace gravel.	ith						
9B	100%		N=15	10			18	Brown (10YR5/3), slightly moist, stiff, clayey SILT w trace sand and slight trace gravel.	vith		-	610			
10A	24/24 100%	ss	13-36 46-70 N=82	7				Dark gray (10YR4/1), slightly moist, hard, clayey SII with trace sand and gravel.	LT			608			
	I		I	I	I	I	20	End of Boring = 20.0 ft. BGS				⊥J– I			
NC)TE(S):	Well	complete	ed pr	rior to	o consti	ruction of berr	n road. Boring surface elevation is as of the well install	date an	d not th	ne fina	l constructed	elevation.		

F	(EL)	D I F: A	BOR EG Coffe	een P	ower	LC Statio	DG n	CONTRACTOR: Layne-Western Co	HANSON
	Sit Locatio	e: C0 n: C0	CB Mana offeen, II	agem linois	ent F 3	acility		Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	BOREHOLE ID: G121 Well ID: G121
WB	Projec DATE EATHEI	et: S:St Fin R:O	tart: 2/4 hish: 2/4 vercast, c	/2010 /2010 cold ()) [lo-3()'s)		FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Han Eng/Geo: S. Simpson	Surface Elev: 627.94 ft. MSL Completion: 22.00 ft. BGS station: 875,964.59N 2,513,904.35E
5	SAMPL	E	T	EST	ING	Ì	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	al (in)				5/ft ³)	0	Quadr Towns	angle: Coffeen, IL hin: East Fork	Ψ = Dry - While drilling Ψ =
r	/ Toti overy		/ 6 in lue	re (%	en. (It	(T_{yp})	Section	10, Tier 7N; Range 3W	$\underline{\nabla} = 14.44 - 3/1/2010$
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry Do	Qu (ts1 Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	22/24 92%	ss	4-4 7-8 N=11	15			2		626
2A	20/24 83%	ss	3-5 8-12 N=13	16			4	 FILL - Brown (10YR4/3) with 10% gray (10YR5/1) a 5% dark yellowish brown (10YR4/6) mottles, moist, s silty CLAY with slight trace sand and gravel. 	and tiff,
3A	18/24 75%	ss	1-4 5-6 N=9	27			6		-622
4A	18/24 75%	ss	2-3 5-6 N=8	25			8	Dark yellowish brown (10YR4/4), moist, medium, si CLAY with slight trace sand. Gray (10YR5/1) with 25% dark yellowish brown (10YR4/6) mottles, moist, medium, clayey SILT with sand.	trace 620
5A	18/24 75%	ss	2-2 3-4 N=5	24			10	Gray (10YR6/1) with 10% yellowish brown (10YR5 mottles majet medium silty CLAY with slight traces	(6) and 618
6A	13/24 54%	ss	2-2 4-4 N=6	23			12		616
7A	19/24 79%	ss	<i>woh-2</i> <i>3-4</i> N=5	23			14	Gray (10YR6/1) with 10% yellowish brown (10YR5 mottles moist medium silty CLAY with trace sand	(6) and 614
8A	18/24 75%	ss	<i>1-2</i> <i>2-2</i> N=4	23				slight trace gravel.	-612
9A	23/24 96%	ss	woh-woi 1-2	h 21			18	Brown, (10YR5/3), very moist, very soft, clayey SAM with slight trace gravel.	ND 610
10A	22/24 92%	ss	4-12 26-30 N=38	8			20	Brown (10YR5/3), slightly moist, hard, very silty CL with trace sand and gravel.	AY 608
NC	DTE(S):	Well	complet	ed pr	ior to	o const	ruction of be	erm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.
1									Page 1 of 2

F	[EL]) I	3OR	I	١G	L(DG		HANS	
	CLIEN Sit Location Projec	Г: АІ е: С(n: С(t:	EG Coffe CB Mana offeen, Ill	een F igem linoi	'ower ient F s	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G121 Well ID: G121 Surface Elev: 627.94 ft.	MSL
W	DATE	S: St Fin R: O	art: 2/4/ ish: 2/4/ vercast, c	/201 /201 cold (0 0 (lo-3)	0's)		FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	Completion: 22.00 ft. by Station: 875,964. 2,513,904 2,513,904	BGS .59N .35E
	SAMPL	E	T s	EST	IP/fl ₃)	ed ad	TOPOGR Quadra Towns	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork	WATER LEVEL INFORMATION: $\Psi = $ Dry - While drilling $\Psi = $	
Number	kecov / Tc 6 Recover	Sype	<u>slows / 6 i</u> V - Value ROD	Aoisture (Den. (Qu (tsf) Gailure Ty	Section Depth ft BGS	Lithologic	$\overline{\nabla} = 14.44 - 3/1/2010$ Borehole Elevation Detail ft MSL Remar	
11A	24/24	ss	13-23 31-48 N=54	7			22	Dark gray (10YR4/1), slightly moist, hard, clayey SI with sand and gravel. End of Boring = 22.0 ft. BGS		

NOTE(S): Well completed prior to construction of berm road. Boring surface elevation is as of the well install date and not the final constructed elevation.

F	[EL]	DI	BOR	IN	JG	L	DG		
	CLIEN Sit Locatio Proiec	Γ: Α] α: C0 n: C0 x: C	EG Coffe CB Mana offeen, II	een P agem linois	ower ent F	Statio acility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G122 Well ID: G122 Surface Elev: 628.05 ft. MSL
WI	DATE	S: St Fin R: O	art: 2/4/ iish: 2/4/ vercast, c	/2010 /2010 cold ()) [lo-3()'s)		FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: D. Lamb	Completion: 20.00 ft. BGS by Station: 876,080.14N 2,513,902.82E
5	SAMPL	Е	Т	EST	ING	r I	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	l (in)				(ff ³)	0)	Quad Towr	rangle: Coffeen, IL	$\mathbf{Y} = 17.00$ - While drilling
	. Tota very		6 in ue	e (%	n. (lb	Type	Sectio	on 10, Tier 7N; Range 3W	$\nabla = 12.84 - 3/1/2010$
Numbe	Recov / % Reco	Type	Blows / N - Val RQD	Moistur	Dry De	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
	23/24	V	5-5						
1A	96%	ss	<i>6-11</i> N=11	16					
							2-	FILL - Gravish brown (10YR5/2) with 15% gray (10YR5/1) and 5% yellowish brown (10YR5/6) motth	es, 626
	20/24	ss	3-3 5-5					moist, stiff, silty CLAY with trace sand and slight tra gravel.	
2A	0370	\wedge	N=8	22					
		$\overline{\mathbf{V}}$					4-		
	20/24 83%	ss	3-4 5-6						
3A		\wedge	N=9	25					
		$\overline{\mathbf{V}}$					6-	Dark yellowish brown (10YR4/4), moist, medium, sil CLAY.	fty 622
4A	19/24 79%	ss	<i>I-5</i> <i>5-6</i>	29					
		\square	N=10				•		
		\bigvee	1.2					Gray (10YR5/1) with 20% dark yellowish brown (10YR4/6) mottles moiet medium silty CLAY	620
5A	20/24 83%	ss	3-3 N=6	21				(101 K4/0) motics, most, menuin, sity CLAT.	
		Δ	11 0				10		618
	10/24	\bigvee	1-2					Grayish brown (10YR5/2), moist, medium, clayey SII	
6A	79%	ss	3-3 N=5	22				witti saitti.	
		$\left(\right)$					12	Yellowish brown (10YR5/6) mottles, moist, medium, s CLAY with trace gravel.	silty 616
7A	16/24	ss	1-2 3-4	21			¥ _	Gray (10YR5/1), moist, medium, clayey SILT with fi	ne liii liii
	0/%	\wedge	N=5					sand.	
		$\overline{\mathbf{V}}$	1_1				14-	Brown (10YR4/3), moist, medium, silty CLAY with tr gravel.	ace 614
8A	19/24 79%	ss	2-2 N=3	21				Grayish brown (10YR5/2), moist, soft, sandy SILT.	
	20/24	V ss	1-1 4 16				16 —	Yellowish brown (10YR5/4) with 40% dark yellowis brown (10YR4/6) mottles, moist, medium, silty CLA with slight trace sand.	h Y
9A	83%	\bigwedge	N=5	14			10	Dark yellowish brown (10YR4/6), wet, medium dens	ie,
		\bigvee	10-38					silty SAND.	
10A	24/24 100%	ss	51-58 N=89	6			20	Brownish yellow (10YR6/6), slightly moist, hard, clay SILT with trace sand and gravel. Gray (10YR5/1), slightly moist, hard, clayey SILT wi gravel.	
							20	End of Boring = 20.0 ft. BGS	
NC)TE(S):	Well	complete	ed pr	ior to	o const	ruction of	perm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.

F	[EL]	DI	BOR	IN	IG	L	DG		
	CLIEN Sit Locatio Proje	T: Al te: Co on: Co ct:	EG Coffe CB Mana offeen, II	een P igem linois	Power ient F s	r Statio Facility	n	CONTRACTOR: Layne-Western CoRig mfg/model: CME-750 ATV DrillDrilling Method: 4¹/₄" HSA w/SS samplers	BOREHOLE ID: G123 Well ID: G123 Surface Elev: 628.12 ft. MSL
WI	DATE	S: St Fin R: O	art: 2/3/ nish: 2/4/ vercast, c	/2010 /2010 cold (0 0 (lo-30)'s)		FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Hamt Eng/Geo: D. Lamb	Completion: 24.00 ft. BGS by Station: 876,189.60N 2,513,901.46E
5	SAMPL	E	Т	EST	TING	ř	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
r	/ Total (in) wery		6 in lue	re (%)	:n. (lb/ft ³)) ; Type	Quadr Towns Section	angle: Coffeen, IL hip: East Fork 10, Tier 7N; Range 3W	\mathbf{Y} = 16.50 - While drilling \mathbf{Y} = \mathbf{Y} = 15.98 - 3/1/2010
Numbe	Recov % Reco	Type	Blows / N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	22/24 92%	ss	6-8 9-11 N=17	16			2	FILL - Gravish brown (10YR5/2) with 10% gray	628
2A	19/24 79%	ss	4-5 6-7 N=11	18			4	(10YR5/1) and 5% yellowish brown (10YR5/6) mottle moist, stiff, silty CLAY with trace sand and slight trac gravel.	²⁸ , ²⁰ - 624
3A	17/24 71%	ss	2-3 5-6 N=8	24			6	Dark grayish brown (10YR4/2) with 20% yellowish bro (10YR5/6) mottles, moist, medium, silty CLAY with sli trace sand, slight trace roots.	wn ight 622
4A	18/24 75%	ss	1-3 5-8 N=8	24			8	Gray (10YR5/1) with 30% yellowish brown (10YR5/0 mottles, moist, medium, silty CLAY with slight trace sa	5) ind.
5A	20/24 83%	ss	2-3 4-5 N=7	18			10 -	Gray (10YR5/1) with 10% yellowish brown (10YR5/t mottles, moist, medium, silty CLAY with trace sand a slight trace gravel	5) nd 618
6A	19/24 79%	ss	1-2 3-5 N=5	21			12		
7A	23/24 96%	ss	<i>1-3</i> <i>4-4</i> N=7	19				Gray (10YR5/1) with 10% yellowish brown (10YR5/0 mottles, moist, medium, silty CLAY with sand and slig trace gravel.	5) ght
84	22/24 92%	ss	1-2 3-3 N=5	17			14	Dark yellowish brown (10YR3/6) with dark yellowish brown (10YR4/6) mottles, moist, medium, silty CLA ³ with trace gravel.	h Y 614
0/1		$\left(\right)$	1_1				⊻ ₁₆	Gray (10YR6/1) with 30% yellowish brown (10YR5/0 mottles, moist, medium, sandy SILT.	
9A	22/24 92%	ss	2-2 N=3	19			18	Dark yellowish brown (10YR4/6), wet, soft, clayey SAN	ND. 610
10A	12/24 50%	ss	3-3 3-4 N=6	16				Yellowish brown (10YR5/6), wet, medium, sandy SIL with trace gravel.	
NC)TE(S):	Well	complete	ed pr	rior to	o const	$20 \rightarrow 20$	rm road. Boring surface elevation is as of the well install of	late and not the final constructed elevation.

I

FI	[EL]	DI	3OR	I	JG	¦ L(JG		HANSON
(CLIEN	Г: А	EG Coffe	en P	owe	r Statio	n	CONTRACTOR: Layne-Western Co	
	Sit	e: C	CB Mana	ıgem	ent F	Facility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G123
	Locatio	n: Co	offeen, Ill	linois	5			Drilling Method: 41/4" HSA w/SS samplers	Well ID: G123
	Projec	t:							Surface Elev: 628.12 ft. MSL
	DATES	S: St	art: 2/3/	2010)			FIELD STAFF: Driller: T. List	Completion: 24.00 ft. BGS
		Fin	nish: 2/4/	/2010	0			Helper: M. Herbst/S. Han	by Station: 876,189.60N
WE	LATHE	R: 0	vercast, c	old ((lo-30	0's)		Eng/Geo: D. Lamb	2,513,901.46E
S	SAMPL	E	Т	EST	INC	j	TOPOG	ΓΑΡΗΙC ΜΑΡ ΙΝΕΟΡΜΑΤΙΟΝ	WATER LEVEL INFORMATION.
er	/ Total (in) overy		/ 6 in Ilue	ıre (%)	en. (lb/ft ³)	f) e Type	Quad Towr Section	rangle: Coffeen, IL ship: East Fork on 10, Tier 7N; Range 3W	$\mathbf{V} = 16.50 - \text{While drilling}$ $\mathbf{V} =$ $\mathbf{V} = 15.98 - 3/1/2010$
Numb6	Recov % Rec	Type	Blows N - Va RQD	Moistr	Dry D	Qu (ts1 Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
11.4	20/24 83%	ss	6-12 12-12 N=24	11				Yellowish brown (10YR5/6), wet, medium, sandy Sl with trace gravel. [Continued from previous page]	
117			2.0	11			22	Gray (10YR5/1), moist, very stiff, clayey SILT with t gravel.	
12A	19/24 79%	ss	3-8 13-9 N=21	9				Dark gray (10YR4/1), moist, very stiff, sandy SILT v trace clay and gravel.	vith
'	1 1		1	1	1	1	24 —	End of Boring = 24.0 ft. BGS	

F	[EL]	DI	BOR	IN	IG	L(DG		HANSON
	CLIEN Sit Locatio	Г: А се: О п: О	EG Coffe CB Mana offeen, Il	een P agem linois	Powe lent H s	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G124 Well ID: G124
WI	DATE	S: St Fir R: St	art: 2/3/ nish: 2/3/	/2010 /2010 d (m	0 0 id-30)'s)		FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Han Eng/Geo: S. Simpson	by Station: 876,304.85N 2 513 900 34E
	SAMPL	E	T	EST	TINC	; ;	TOPOCP		WATED I EVEL INCOMATION.
	otal (in) Y		и	(%	[lb/ft ³]	pe	Quadra Townsł	ngle: Coffeen, IL ip: East Fork	Ψ = Dry - While drilling Ψ =
ber	v / Tc coven		s/6i alue	ture (Jen. ($re T_{y}$	Section	10, Tier 7N; Range 3W	$\underline{\nabla} = 10.99 - 3/1/2010$
Numl	Reco % Re	Type	Blow N - V RQD	Moist	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	24/24 100%	ss	6-4 5-6 N=9	17					628
2A	22/24 92%	ss	4-5 7-8 N=12	21				FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a 5% yellowish brown (10YR5/6) mottles, moist, stiff, s CLAY with slight trace sand and gravel.	and silty 626
3A	19/24 79%	ss	2-4 6-7 N=10	25				Dark grayish brown (10YR4/2) with 5% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY.	
4A	13/24 54%	ss	3-3 6-6 N=9	28				Dark grayish brown (10YR4/2) with 5% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY, slight transition roots.	own ace 622
5A	20/24 83%	ss	<i>1-3</i> 5-6 N=8	21				Gray (10YR5/1) with 15% yellowish brown (10YR5 mottles, moist, medium, silty CLAY with slight trace and gravel.	620 sand
6A	17/24 71%	ss	<i>1-3</i> <i>3-4</i> N=6	22			⊻ _		
7A	18/24 75%	ss	<i>1-3</i> <i>3-4</i> N=6	23				Gray (10YR5/1) with 15% yellowish brown (10YR5 mottles, moist, medium, silty CLAY with trace sand slight trace gravel.	(6) ind 616
8A 8B	20/24 83%	ss	woh-woh 2-3	28 22				Gray (10YR5/1) with 30% yellowish brown (10YR5 mottles, moist, very soft, silty CLAY with trace sand	614
9A 9B	19/24 79%	ss	<i>1-2</i> <i>3-3</i> N=5	22				Gray (10YR6/1) with 5% yellowish brown (10YR5/ mottles, moist, medium, sandy CLAY with slight tra gravel. Gray (10YR6/1) with 20% vellowish brown (10YR5/	60 - 612
10A	20/24 83%	ss	4-10 17-23 N=27	12			20	mottles, moist, medium, sandy CLAY with slight trac gravel. Yellowish brown (10YR5/4), moist, medium, clayey S with trace sand and slight trace gravel. Yellowish brown (10YR5/4), slightly moist, stiff, cla SILT with trace sand and slight trace gravel. Gray (10YR5/1), slightly moist, stiff, clayey SILT w trace sand and slight trace gravel. End of Boring = 20.0 ft. BGS	ce ILT $-610rey$ ith -1 -1 -1 -1 -610
NC)TE(S):	Well	complete	ed pi	rior t	o const	ruction of be	rm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.

FI	ELIEN Sit Locatio Projec DATE	D I T: A te: Co on: Co ct: (S: St Fir R: St	BOR EG Coffe CB Mana offeen, II tart: 2/2 hish: 2/3 unny, colu	een P agem linois /2010 /2010 d (m	NG owen ent F s 0 0 id-30	s Statio Facility	DG ⁿ	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Haml Eng/Geo: S. Simpson	BOREHOLE ID: G125 Well ID: G125 Surface Elev: 628.85 ft. MSL Completion: 20.13 ft. BGS Station: 876,409.47N 2,513,899.12E
5	otal (in)	E	Г ц	EST	(lb/ft ³)	vpe	TOPOGR Quadr Towns	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork	WATER LEVEL INFORMATION: $\Psi = 17.00$ - While drilling $\Psi = Dry$ - Upon completion
Number	Recov / T % Recove	Type	<i>Blows / 6</i> N - Value RQD	Moisture	Dry Den.	Qu (tsf) Failure T	Depth ft. BGS	Lithologic Description	$\underline{\nabla} = 8.58 - 3/1/2010$ Borehole Elevation Detail ft. MSL Remarks
1A	18/24 75%	ss	5-6 7-9 N=13	17			2	FILL - Brown (10YR5/3) with 5% gray (10YR5/1) and	5%
2A	16/24 67%	ss	3-4 6-9 N=10	17			4	yellowish brown (10YR5/6) mottles, moist, stiff, silt CLAY with slight trace sand and gravel.	y 626
3A	20/24 83%	ss	2-5 5-7 N=10	25					624
4A	19/24 79%	ss	2-4 5-5 N=9	23			8	(10YR5/2) with 15% yellowish brow (10YR5/6) mottles, moist, stiff, silty CLAY with slig trace sand and gravel, slight trace roots.	n ht 622
5A	14/24 58%	ss	3-4 4-6 N=8	25			⊻ 10	Gray (10YR5/1) with 25% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with slight trace s and gravel, slight trace roots.	6) and
6A	18/24 75%	ss	2-3 3-5 N=6	22			12	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with slight trace sa	618 6) and.
7A	20/24 83%	ss	3-4 3-4 N=7	23			14	Gray (10YR5/1) with 10% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	6) nd
8A	24/24 100%	ss	3-3 3-3 N=6	23			16	Gray (10YR6/1) with 10% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with trace sand a	6) nd
9A	24/24 100%	ss	woh-1 2-1 N=3	25			⊻		
9B 10A	19/24	$\left(\right)$	woh-1	18 22			18	Brown (10YR5/3), very moist, very soft, clayey SAN with slight trace gravel.	D 610
10B	79%	^{ss} ^{ss}	N=8	11			20-	Brown (10YR5/3), slightly moist, medium, clayey SII with trace sand and slight trace gravel.	
NC)TE(S):	Well	complet	ed pi	ior to	o consti	ruction of b	End of Boring = 20.1 ft . BGS erm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.

F	ELI CLIEN Sit	D I T: A a: C	BOR EG Coffe CB Mana	een P	IG owe	F LC r Statio Facility	DG	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G126
	Locatio Projec DATE	n: Co et: S: St	offeen, Il	linoi: 0/20	s 10	-		Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers FIELD STAFF: Driller: T. List	Well ID: G126 Surface Elev: 622.96 ft. MSL Completion: 18.00 ft. BGS
WF	EATHEI	Fin R: Pa	nish: 2/1 artly clou	0/20 dy, v	10 vind	10 mpł	ı, ~25F	Helper: M. Herbst/S. Ham Eng/Geo: D. Lamb	by Station: 875,062.44N 2,513,895.37E
	tal (in)	E	1	TEST	INC (July)	с т әс	TOPOGR Quadr Towns	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork	WATER LEVEL INFORMATION: $\underline{\Psi} = 13.20$ - While drilling $\underline{\Psi} = 13.00$ - Upon completion
umber	scov / To Recover	/pe	<i>ows / 6 ii</i> - Value Q D	oisture (⁹	ry Den. (u (tsf) <i>uilure Ty</i> _l	Section Depth	10, Tier 7N; Range 3W	$\underline{\nabla} = 8.82 - 3/1/2010$ Borehole Elevation
Ź	20/24	<u>F</u>	9-7 12-10	M	D	P ₆	ft. BGS	Description FILL - Dark yellowish brown (10YR4/6), moist, very s sandy CLAY with silt and trace gravel.	tiff,
1A	83%	$\bigwedge_{i=1}^{\infty}$	N=19	15			2	Light grayish brown (10YR6/2), slightly moist, stiff, s 	
2A	19/24 79%	ss	2-4 6-9 N=10	26			4	Light grayish brown (10YR6/2) with 20% yellowish br (10YR5/6) mottles, moist, stiff, silty CLAY, slight tra roots.	own ice 620
3A	22/24 92%	ss	2-4 5-7 N=9	20			6		618
4A	24/24 100%	ss	2-3 4-5 N=7	22			8	Dark brown (10YR3/3) with 10% dark yellowish bro (10YR4/6) mottles, moist, stiff, silty CLAY.	wn 616
5A	24/24 100%	ss	1-3 4-5 N=7	22			⊻	Light gravish brown (10YR6/2) with 30% vellowish br	614
6A	22/24 92%	ss	1-3 4-5 N=7	18				(IÕYŘ5/6) mottles, moist, medium, silty ĆLAY with t sand and slight trace gravel.	race 612
7A	20/24 83%	ss	<i>1-2</i> <i>3-4</i> N=5	19			12	Grayish brown (10YR5/2) with dark yellowish brow (10YR4/6) mottles, moist, medium, clayey SILT with s and slight trace gravel.	n sand
8A	24/24	ss	4-15 12-29	20			14	Gray (10YR6/1). very moist, medium, sandy CLAY v trace gravel. Yellowish brown (10YR5/8) with 40% dark yellowis	h 608
	10070	$\left(\right)$	N=27				16	brown (10YR4/6) mottles, slightly moist, very stiff, cla SILT with trace gravel. Brownish yellow (10YR6/8), moist, very stiff, sandy S with gravel.	
9A	23/24 96%	ss	29-39 39-29 N=78	8				Grayish brown (10YR5/2), moist, hard, sandy SILT w gravel.	rith
							18	End of Boring = 18.0 ft. BGS	
NC)TE(S):	G120	5 installe	d in l	boreł	nole.			

WE	CLIEN Sit Locatio Projec DATE ATHEI	T: A te: C n: C xt: S: St Fir R: O E	EG Coffe CB Mana offeen, Il tart: 2/10 hish: 2/1 vercast, v	een P agem linois 0/20 0/20 vind	Power ent F s 10 10 15m	r Statio Facility ph, ~10	n)-20F	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Hamb Eng/Geo: D. Lamb	BOREHOLE ID: T127 Well ID: T127 Surface Elev: 625.53 ft. MSL Completion: 19.80 ft. BGS Station: 875,359.21N 2,513,911.02E
er	/ Total (in)		/ 6 in alue	ure (%)	en. (lb/ft ³)	t) 'e Type	TOPOGR Quadr Towns Sectior	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork 1 10, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 15.30$ - While drilling $\Psi = 18.00$ - Upon completion $\overline{\Psi} = 11.26 - 3/1/10$
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1	24/24 100%	ss	8-5 5-8 N=10				2	FILL - yellowish brown (10YR5/6) with 20% grayish brown (10YR5/2) mottles moist stiff, silty CLAY wi	h th
2	10/24 42%	ss Ss	5-6 7-11 N=13				4	trace sand and gravel.	622
3	24/24	ss Ss Ss	5-7 N=9				6	Dark yellowish brown (10YR4/6) with 30% brownisl yellow (10YR6/6) and 10% very dark gray (10YR3/1 mottles, moist, medium, clayey SILT with trace sand	n 620
4	100%		N=9				8-	Grayish brown (10YR5/2) with 20% dark yellowish bro (10YR4/6) mottles, moist, medium, clayey SILT with sand.	wn 1 618
5	24/24 100%	ss	2-4 5-6 N=9				10	Grayish brown (10YR5/2) with 20% dark yellowish bro (10YR4/6) and 5% very dark brown (10YR2/2) mottle moist, medium, clayey SILT with sand and slight trac	wn s, e
6	24/24 100%	ss	5-7 N=8				∑ 12	Dark yellowish brown (10YR4/6) with 5% light brown	
7	24/24 100%	ss	1-2 3-3 N=5				14	gray (10YR6/2) mottles, moist, medium, sandy SILT w gravel. Light brownish gray (10YR6/2) with 30% yellowish broc (10YR5/6) mottles, very moist, medium, clayey SANI	wn
8	24/24 100%	ss	1-1 2-1 N=3				Y		610
0	24/24 100%	ss	8-12 28-30 N=40					(10YR6/8) mottles, wet, very losse, silty SAND trace w gravel.	/ith
9	22/22 100%	ss	8-44 65-60/4'	,			⊻ 18	Grayish brown (10YR5/2), slightly moist, hard, clave	y 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
10	10070	\square	N=109					SILT with sand and gravel.	606
NO	TE(S):	Well	complet	ed pi	rior to	o const	ruction of be	erm road. Boring surface elevation is as of the well install of	date and not the final constructed elevation.

F	[EL]	DI	BOR	IN	IG	L	DG							
	CLIEN Sit Locatio	T: A te: C n: C	EG Coffe CB Mana offeen, II	een P igem linois	Power ient F s	r Statio Facility	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: T128 Well ID: T128					
WE	Proje DATE EATHE	st: S: Si Fir R: C	t art: 2/9/ nish: 2/9/ old, wind	/2010 /2010 y, sn	0 0 10w (lo-20's))	FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	Surface Llev: 626.27 ft. MSL Completion: 22.00 ft. BGS by Station: 875,509.70N 2,513,909.45E 2513,909.45E					
5	SAMPL	E	Т	EST	TING	7	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:					
	[(in)				(ft ³)		Quadra	angle: Coffeen, IL	$\mathbf{\Psi} = 16.84$ - While drilling					
	[ota] ery		6 e	(%)	[])	ype	Towns	hip: East Fork	$\underline{\Psi} = \nabla = 12.35 - 3/1/10$					
lber	/ vc	0	vs/(Valu	sture	Den	tsf) ure 1	D							
Num	Reco %R	Type	Blow N - N RQI	Moi	Dry	Qu (Faili	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks					
1A	12/24 50%	ss	37-17 15-18 N=32	14			2	FILL - Brown (10YR5/3) with 5% gray (10YR5/1) and yellowish brown (10YR5/6) mottles, moist, hard (froz 	15% en), 					
2A	7/24 29%	ss	4-7 7-13 N=14				4	FILL - Brown (10YR5/3) with 10% gray (10YR5/1) a 5% yellowish brown (10YR5/6) mottles, moist, stiff, s CLAY with slight trace sand and gravel.	ind ility					
3A	12/24 50%	ss	3-4 6-8 N=10	26			6	Gray (10YR5/1) with 20% yellowish brown (10YR5/ mottles, moist, stiff, silty CLAY with slight trace sam slight trace roots.	6) d,					
4A	19/24 79%	ss	2-4 5-7 N=9	24			8		620					
5A	22/24 92%	ss	1-3 4-5 N=7	20			10	Gray (10YR5/1) with 20% yellowish brown (10YR5, mottles, moist, stiff, silty CLAY with trace sand and sl trace gravel.	(6) ight 616					
6A	22/24 92%	ss	<i>1-3</i> 5-5 N=8	20			√ 12 -							
7A	22/24 92%	ss	2-3 4-5 N=7	19			14-							
8A	20/24 83%	ss	2-2 3-3 N=5	18				Gray (10YR6/1) with 30% yellowish brown (10YR5/ mottles, moist, medium, silty CLAY with trace sand a slight trace gravel.	(6) and					
9A 9R	22/24 92%	ss	woh-1 8-7 N=9	19			¥ (Brown (10YR5/3), moist, soft, clayey SAND with sli <u>trace gravel.</u> Gray (10YR6/1), wet, loose, silty, very fine- to <u>coarse-grained SAND.</u>	ght 610					
10.4		$\left(\right)$					18	Brown (10YR5/3), wet, loose, silty, very fine- to fine-grained SAND.						
10A	18/24 75%	ss	3-10 13-11 N=23	17				Brown (10YR5/3), slightly moist, very stiff, clayey Sl with trace sand and gravel.						
NC)TE(S):	Well	complet	ed pr	ior to	o const	20 - 20	rm road. Boring surface elevation is as of the well install	date and not the final constructed elevation.					

F	TEL	DI	BOR	IN	JG	G L() G			<		ANSON	
v	CLIEN Si Locatio Proje DATH	T: A te: C on: C ct: CS: S(Fir R: C	EG Coffe CB Mana offeen, Ill tart: 2/9/ nish: 2/9/ old, wind	en P gem linois /2010 /2010 ly, sn	'owen ent F 3 0 0 0	r Statio Facility (lo-20's)	n	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4¼" HSA w/SS samplers FIELD STAFF: Driller: T. List Helper: M. Herbst/S. Ham Eng/Geo: S. Simpson	by	во	REHOLE ID Well ID Surface Elev Completion Station	 : T128 : T128 : 626.27 ft. MSL : 22.00 ft. BGS : 875,509.70N 2,513,909.45E 	
	SAMPI	E	Т	EST	INC	Ĵ	TOPOGR	APHIC MAP INFORMATION:	WATER	R LEVEL INFORMATION:			
	al (in))/ft ³)	0	Quadr Towns	angle: Coffeen, IL shin: Fast Fork	= <u>\</u> \ =	$\mathbf{Y} = 16.84$ - While drilling			
er	/ Toti overy		/ 6 in due	ure (%	en. (ll	f) e Typ	Section	n 10, Tier 7N; Range 3W	=	12.35 -	3/1/10		
Numb	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	В	orehole Detail	Elevation ft. MSL	Remarks	
11.	A 24/24 100%	ss	8-16 18-25 N=34	16			22	Dark gray (10YR4/1), slightly moist, hard, clayey SII with trace sand and gravel. End of Boring = 22.0 ft. BGS	LT				

NOTE(S): Well completed prior to construction of berm road. Boring surface elevation is as of the well install date and not the final constructed elevation.

F	EL	DI	BOR	IN	NG	G L(DG			<	Ka h	ANSON		
	CLIENT Sit Location Projec	F: III e: Co n: Co n: 14	inois Pov offeen Po offeen, M 1F0078	wer H wer S lontg	Ioldii Statio omer	ngs on ty Cour	nty, Illinois	CONTRACTOR: Ramsey Rig mfg/model: D-50 Turbo Tracked MST 8 Drilling Method: 4¼" Hollow Stem Auger wit	00ATV h Split spoon	BO	REHOLE ID Well ID Surface Elev	: TA31 : TA31 : 623 89 ft MSL		
WE	DATE	S: St Fin R: O	art: 10/ nish: 10/ vercast, r	28/2 /28/2 nild -	014 014 - mid	50's		FIELD STAFF: Driller: B. Williamson Helper: D. Crump Eng/Geo: R. Hasenyager			Completion Station	 20.19 ft. BGS 876,542.25N 2,513,856.77E 		
	SAMPL	E	Т	EST	TING	;	TOPOGRAPHIC MAP INFORMATION: WATH			ATER LEVEL INFORMATION:				
	tal (ii <i>y</i>		u	(%)	lb/ft ³)	<i>p</i> (tsf be	Quadra Towns	angle: Coffeen, IL hip: East Fork	$\underline{\Psi} = 8$ $\underline{\Psi} = 8$	$\underline{\Psi} = 8.11 - 10/30/2014$ $\underline{\Psi} =$				
er	//To cover:		s / 6 i alue	ure (en. ($^{\rm sf)}_{\rm e Tyr}$	Section	10, Tier 7N; Range 3W	<u> </u>					
Numł	Recov % Re	Type	Blows N - V RQD	Moist	Dry I	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Bore De	ehole tail	Elevation ft. MSL	Remarks		
1	24/24 100%	ss	1-2 2-2 N=4	26			2				622			
2	22/24 92%	ss	2-3 3-4 N=6	28		1.44 B	4-	Yellowish brown (10YR5/4), moist, soft, SILT with lit clay and trace very fine sand. Brown (10YR4/3), moist, soft, SILT with little clay ar trace very fine sand.	tle	×//63 63 63 6	620			
3	23/24 96%	ss	2-3 4-4 N=7	20		1.71 Sh	6		ر _م ر _م ر _م ر	1,1,1,1,1	618			
4	24/24 100%	ss	6-7 6-6 N=13	22		1.55 Sh	▼ 8			2525				
5	22/24 92%	ss	2-2 2-4 N=4	22		1.55 Sh	10	Gray (10YR5/1) with 20% yellowish brown (10YR5/ mottles, moist, soft, CLAY with few silt and trace very t to fine sand.	8) ine		614			
6	24/24 100%	ss	2-1 3-2 N=4	20		0.93 B	12	Brownish yellow (10YR6/6) with 20% gray (10YR6/ mottles, moist, soft, CLAY with few silt and little very f to fine sand and trace small gravel.	l) ine		612			
7A	24/24	W	2-3	19										
	100%	∬ ⁵⁵	5-5 N=6					Brownish yellow (10YR6/6), wet, slightly dense, SAN (very fine to coarse) with little sand and silt and trace cl	D ay					
7B 8A	24/24	V ss	4-5 8-7	22 12			14	and small gravel. Brownish yellow (10YR6/6), moist, soft, CLAY with finds silt and trace very fine to fine sand and small gravel. Brownish yellow (10YR6/6) wet, loose, SAND (very finds)	ew		610			
8B	100%	Λ	N=13	13				to very coarse) and trace small gravel. Brownish yellow (10YR6/6) very moist, soft, SAND (v fine to medium) with few clay and silt. Yellowish brown (10YR5/8), moist, soft, CLAY with f	ery		608			
9	22/24 92%	ss	5-4 4-4 N=8	18			18-	silt and trace sand (very fine to very coarse) and smal gravel. Yellowish brown (10YR5/8), wet, loose, very fine to ve coarse SAND.	l		606			
10	16/24 67%	ss	2-4 3-5 N=7	12			20	Gray (10YR5/1), moist, hard, CLAY with few silt an (very fine to very coarse) sand and small gravel	d		604			
	I		I	1	I	I	⊥ <u> </u>	End of Boring at 20.19 ft.			14-			
	TE(C)	м	., .	.11 T	4.2.1	·	. 1 1							

NOTE(S): Monitoring well TA31 installed in borehole.

Г

FI WE	ELIENT Site Location Projec DATES	D I F: III e: Co n: Co n: Co t: 14 S: St Fin R: Pa	BOR inois Pow offeen Po offeen, M E0078 art: 10/ ish: 10/ irtly sunn	ver F wer F lontg 27/2 27/2 y, m	Holdii Statio comer 014 014 014	ngs on ry Cour mid 70	00ATV 1 Split spoon	BOI	REHOLE II Well II Surface Elev Completion Station	ANSON 2: TA32 2: TA32 2: 618.93 ft. MSL 2: 16.47 ft. BGS 2: 877,532.57N 2,513,605.19E			
		E	Т	EST	FINC	;	TOPOGI	APHIC MAP INFORMATION: WATER LE			EVEL INFORMATION:		
ber	v / Total (i covery		s / 6 in alue	ture (%)	Den. (lb/ft ³	sf) <i>Qp</i> (tsf re Type	Quadi Towns Sectio	angle: Concentration $\underline{\Psi}$ hip: East Fork $\underline{\Psi}$ n 3, Tier 7N; Range 3W $\underline{\nabla}$.05 - 10/50/2014		
Numl	Reco ^r % Re	Type	<i>Вlow.</i> N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Bore De	ehole tail	Elevation ft. MSL	Remarks	
1	18/24 75%	ss	2-1 2-3 N=3	21			2	Dark yellowish brown (10YR4/4), moist, soft, clayey SII with trace sand (very fine).	LT		618		
	21/24 88%	ss	2-1 1-1 N=2							2//2	616		
2 3A 3B	14/24 58%	ss	<i>1-1</i> 3-3 N=4	28 24 22		1.15 Sh	4	Black (10YR2/1), moist, soft, clayey SILT with trace sate (very fine).	nd e	<u>, , , , , , , , , , , , , , , , , , , </u>	614		
4	20/24 83%	ss	4-5 4-7 N=9	22		0.96 B		Silt and trace sand (very fine to fine). Yellowish brown (10YR5/6) with 30% gray (10YR5/1 mottles, moist, medium, CLAY with little silt and trace sa) ind		612		
5	21/24 88%	ss	4-7 12-22 N=19	9		8.45 Sh		(very fine to very coarse) and gravel (small).			610		
6	21/24 88%	ss	6-16 27-42 N=43	8		10.47 Sh	12	Yellowish brown (10YR5/8) with 30% Gray (10YR5/1 mottles, slightly moist, very hard, CLAY with silt and litt sand (very fine to very coarse) trace gravel (small).) tle		608		
7	12/24 50%	ss	36-55 50/4"	6			¥ 14 -	Yellowish brown (10YR5/8), slightly moist, loose, SAN (very fine to very coarse) and few gravel (small to medium Gray (10YR5/1), slightly moist, very hard, CLAY with s and few gravel (small to medium)	D		606		
8	16/24 67%	ss	16-35 69 N=104	8			16-1	Gray (10YR5/1), moist, dense, SAND (very fine to coars Gray (10YR5/1), slightly moist, very hard, CLAY with s and few gravel (small to medium).	se).		604		
							1	End of Boring at 16.47 ft.					

F	FIELD BORING LOG CLIENT: Illinois Power Generating Company CONTRACTOR: Ramsey													
	CLIEN Sit Locatio	T: 111 te: C n: C	linois Pov offeen W offeen, II	wer (Tell S linoi:	Gener ealin s	rating C g & Ass	Company smt Well Inst	tall CONTRACTOR: Ramsey Rig mfg/model: Diedrich D-50 Drilling Method: 3 ¼" HSA, split spoon sample	BOREHOLE ID: TA33a Well ID: n/a					
WE	Proje DATE CATHE	ct: 12 S: S1 Fir R: S1	tart: 6/2 nish: 6/2 nny, cali	/201 2/201 m, hi	5 .5 i-60s			FIELD STAFF: Driller: B. Williamson Helper: D. Crump Eng/Geo: S. Keim	Surface Elev: 622.5 ft. MSL Completion: 30.0 ft. BGS Station: 876,610.00N 2,513,248.00E					
5	SAMPL	Æ	Г	EST	ΓINO	J	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:					
	l (in				/ft ³)	(tsf)	Quadra	angle: Coffeen, IL	$\mathbf{Y} = 13.80$ - during drilling $\mathbf{V} =$					
	Tota very		6 in ue	e (%	n. (lb	Q	Section	a 3, Tier 7N; Range 3W	$\mathbf{Y} = \mathbf{\nabla}$					
Imber	cov / Reco	be	2D 20	Distur	y Dei	i (tsf) ilure	Depth	Lithologic	Borehole Elevation					
ñ	% Re	Ê	R N	Ž	D	Fa	ft. BGS	Description Vellowish brown (10VR5/4) moist_stiff_clavey_SILT	Detail ft. MSL Remarks					
		М	2_3					trace very fine-grained sand, roots.	,					
1A	88%	ss	5-5 N=8					Light brownish gray (10YR6/2) with 35% yellowish brownish	wn					
		$\langle \rangle$						(10YR5/6) mottles, moist, stiff to very stiff, clayey SIL. trace very fine- to fine-grained sand.						
		\square							620					
	20/24	ss	3-4 4-5											
2A	05/0	Λ	N=8					Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, very stiff, clayey SILT, little	•					
		$\left(\right)$					4	fine- to medium-grained sand, trace coarse-grained sand	1.					
	23/24	M	5-6						618					
34	96%	Å ss	6-9 N=12											
511		\square					6	Gray (10YR5/1) with 5-10% dark yellowish brown						
		M						(10YR4/6) mottles, moist, very stiff, silty CLAY, little fit to medium-grained sand, trace coarse-grained sand.	ne- 616					
	24/24	ss	10-10											
4A	10070	Λ	N=20											
		H					8	Dark yellowish brown (10YR4/6) with 3% gray (10YR6 mottles moist, very stiff, silty CLAY, little fine- to						
5 1	21/24	M	2-4					coarse-grained sand, trace small gravel	614					
57	88%	Å ^{ss}	6-6 N=10					Gray (10YR6/1) with 30% dark yellowish brown						
		\square						coarse-grained sand, trace small gravel.						
		М							612					
	24/24	ss	0-3 4-8					Gray (10YR6/1) moist, medium, fine- to coarse-grained	d					
6A		Λ	N=7					sandy CLAY, little silt, trace small to large gravel.						
		H					12	Brown (10YR5/3) with 5% dark vellowish brown						
7A	24/24	$\mathbb{N}_{\mathbb{R}}$	15-27					(10YR4/6) mottles, slightly moist, very stiff, very fine-t	to					
,,,,	100%	A 55	32-41 N=59					sand, trace small to large gravel, friable.						
7B		Д					▼ 14 -	Dark yellowish brown (10YR4/6), very moist, dense,						
		М	5.12					sity, very line to inclum-granica system.						
8A	20/24 83%	ss	18-22 N=20											
		Λ	N-30											
		\square					16	Brown (10YR5/3), slightly moist, hard, clayey SILT, fe fine- to coarse-grained sand, trace small gravel.						
9A	16/24	\mathbb{N}_{ss}	23-44											
	67%	\mathbb{N}^{3}	50/5"											
		()					18							
		М	12_12						604					
10A	22/24 92%	ss	22-25 N=34					Dark gray (10YR4/1), slightly moist, hard, clayey SIL7 few fine- to coarse-grained sand, trace small to large grav	r,					
		$\langle \rangle$	11 34											
NC)TE(S):	Bore	hole seal	led w	/ith h	igh-soli	ids bentonite	grout.						

F	FIELD BORING LOG											
WE	CLIEN Sit Location Projec DATE CATHEI	Γ: 11 e: C n: C t: 14 S: S Fir R: S	linois Pov offeen Wo offeen, Ill 4E0078A tart: 6/2, nish: 6/2 unny, cali	ver C ell So linois /201 /201 m, hi	Gener ealing 5 5 -60s	rating C g & As	Company smt Well Inst	CONTRACTOR: Ramsey Rig mfg/model: Diedrich D-50 Drilling Method: 3 ¼" HSA, split spoon samp FIELD STAFF: Driller: B. Williamson Helper: D. Crump Eng/Geo: S. Keim	bler	BOREHOLE ID: TA33a Well ID: n/a Surface Elev: 622.5 ft. MS Completion: 30.0 ft. BG Station: 876,610.00N 2,513,248.00H		
er <i>Overy</i> <i>ine</i> (%) <i>ine</i> (%)				en. (lb/ft ³) N	$\frac{D}{Dp} \frac{Dp}{(tsf)}$	- TOPOGRAPHIC MAP INFORMATION: Quadrangle: Coffeen, IL Township: East Fork Section 3, Tier 7N; Range 3W			WATER LEVEL INFORMATION: $\underline{\Psi} = 13.80$ - during drilling $\underline{\Psi} = $ $\underline{\nabla} = $			
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D(Qu (tsf Failure	Depth ft. BGS	Lithologic Description		Borehole Detail	Elevation ft. MSL	Remarks
11A	24/24 100%	ss	11-17 25-32 N=42				22			ار گروگر گروگر از گروگر گروگر		
12A	23/24 96%	ss	17-23 35-44 N=58				24			کر کر کر کر کر کر کر کر کر کر	600	
13A	21/24 88%	ss	26-30 30-37 N=60				26-	Dark gray (10YR4/1), slightly moist, hard, clayey SIL few fine- to coarse-grained sand, trace small to large gra [Continued from previous page]	LT, avel.	ا را را را را را ا را را را را را را را را را را را	598	
14A	18/24 75%	ss	19-35 47-50 N=82				28			د از از از از از د از از از از از د از از از از از	596	
15A	$5A \begin{vmatrix} 24/24 \\ 100\% \\ N=61 \end{vmatrix} SS \begin{vmatrix} 18-27 \\ 34-35 \\ N=61 \end{vmatrix} = 30$							End of boring = 30.0 feet		ر ال ال ال ال ال ال ال ال ال ال ال ال ال		

F	FIELD BORING LOG											
WF	CLIEN Sit Locatio Projec DATE	T: 111 te: Co n: Co ct: 14 S: St Fin R: St	inois Pow offeen We offeen, Ill E0078A art: 6/2, ish: 6/2,	ver C ell Se inois /201 /201	Gener ealin 5 5 5	rating C g & As	Company smt Well Insta	CONTRACTOR: Ramsey Rig mfg/model: Diedrich D-50 Drilling Method: 3 ¼" HSA, blind drill FIELD STAFF: Driller: B. Williamson Helper: D. Crump Eng/Geo: S. Keim	BOREHOLE ID: TA33b Well ID: TA33 Surface Elev: 622.51 ft. MSL Completion: 17.44 ft. BGS Station: 876,605.45N 2 513 248 73E			
WE S	SAMPL	к. э. Е	T	EST	-005 []]N(j	TODOCD		2,515,240.751			
nber	ov / Total (in) ecovery	e	<i>vs / 6 in</i> Value D	sture (%)	Den. (lb/ft ³)	(tsf) <i>Qp</i> (tsf) ure Type	Quadra Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 3, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\underline{\Psi} = Dry - during drilling$ $\underline{\Psi} = 7.00 - 6/16/15$ $\underline{\nabla} =$			
Nun	Rec % R	Typ	Blor N - RQ	Moi	Dry	Qu Fail	ft. BGS	Description	Detail ft. MSL Remarks			
							2 2	Yellowish brown (10YR5/4), moist, stiff, clayey SILT trace very fine-grained sand, roots. Light brownish gray (10YR6/2) with 35% yellowish bro (10YR5/6) mottles, moist, stiff to very stiff, clayey SIL' trace very fine- to fine-grained sand. Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, very stiff, clayey SILT, little	e 622			
							4	fine- to medium-grained sand, trace coarse-grained sand Gray (10YR5/1) with 5-10% dark yellowish brown (10YR4/6) mottles, moist, very stiff, silty CLAY, little fi to medium-grained sand, trace coarse-grained sand.	d. 618 ne- 616			
							8	Dark yellowish brown (10YR4/6) with 3% gray (10YR6 mottles, moist, very stiff, silty CLAY, little fine- to coarse-grained sand, trace small gravel Gray (10YR6/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY, little fine- coarse-grained sand, trace small gravel.	5/1) - 614 to			
								Gray (10YR6/1) moist, medium, fine- to coarse-graine sandy CLAY, little silt, trace small to large gravel.	612 d			
								Brown (10YR5/3) with 5% dark yellowish brown (10YR4/6) mottles, slightly moist, very stiff, very fine- fine-grained sandy SILT, trace medium- to coarse-grain- sand, trace small to large gravel, friable.	to ed			
							14	Dark yellowish brown (10YR4/6), very moist, dense, silty, very fine- to medium-grained SAND.				
							16	Brown (10YR5/3), slightly moist, hard, clayey SILT, fe fine- to coarse-grained sand, trace small gravel.	2W 608			
			-					End of boring = 17.44 feet				

F	FIELD BORING LOG CRE HANSON											
	CLIEN Sit Location	F : III e: Co n: Co	inois Pov offeen We	ver C ell S inois	Gener ealin s	rating C g & Ass	Company CONTRACTOR: Ramsey Smt Well Install Rig mfg/model: Diedrich D-50 Drilling Method: 3 ¼" HSA, split spoon sampler Well ID: TA34					
	Projec DATE:	t: 14 S: St Fin	E0078A art: 6/3 ish: 6/3	/201 /201	5 5		FIELD STAFF: Driller: B. Williamson Completion: 16.10 f Helper: D. Crump Station: 875,90	tt. MSL ft. BGS 6.10N				
WE	ATHE	R: M	ostly clou	ıdy,	wind	y, mid-	60s Eng/Geo: S. Keim 2,513,46	6.73E				
	SAMPL	E	Т	EST	FINC	; 	TOPOGRAPHIC MAP INFORMATION: WATER LEVEL INFORMATION:					
	al (ir		~	()	b/ft ³)	e (tsf	Quadrangle: Coffeen, IL $\Psi = 13.00$ - during drillingTownshin: East Fork $\Psi = 810 - 6/16/15$					
н	/ Tot		/ 6 in lue	Ire (%	sn. (]	$Q_{\rm T}$	Section 10, Tier 7N; Range 3W $\overline{\Sigma} =$	$\overline{\Sigma} =$				
Numbe	Recov % Reco	Type	Blows , N - Va RQD	Moistu	Dry De	Qu (tsf Failure	DepthLithologicBoreholeElevationft. BGSDescriptionDetailft. MSLRemain	arks				
1 4	21/24 88%	ss	3-3 3-3 N=6				Gray (10YR5/1) with 10% dark yellowish brown (10YR3/6) mottles, moist, stiff, SILT, some clay, trace very fine-grained, roots. Grayish brown (10YR5/2) with 10% yellowish brown (10YR5/6) mottles moist very stiff silty (I AY trace					
171	17/04	$\langle \rangle$	2-2				2 Light gravish brown (10YR6/2) with 15% vellowish brown					
2A	71%	ss	4-6 N=6				(10YR5/6) mottles, slightly moist, very stiff, silty CLAY, trace fine-grained sand.					
3A	22/24 92%	ss	3-3 3-4 N=6				Gravish brown (10YB5/2) with 10% vellowish brown					
	24/24 100%	ss	3-4 6-7 N=10				6 (10YR5/6) mottles, moist, stiff, silty CLAY, little fine-grained sand, trace medium- to coarse-grained sand.					
4A	24/24		2-2									
5A	100%	ss	5-5 N=7				Grayish brown (10YR5/2) with 25% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY, little fine-grained sand trace medium- to coarse-grained sand					
6A	22/24 92%	ss	3-3 4-5 N=7				trace small gravel.					
7A	21/24	ss (3-3 4-5				12 612 Dark yellowish brown (10YR4/6) with 15% light gray (10YR7/1) mottles, very moist, very soft, fine- to medium grained sandy CLAX trace coarse grained sand					
7B	88%	Ą	N=7				Gray (10YR6/1), very moist, loose, clayey, silty, fine- to 					
8В	24/24 100%	ss	18-28 25-32 N=53				Coarse-grained SAND and small to large GRAVEL. Yellowish brown (10YR5/6), slightly moist, hard, clayey SILT, little fine- to coarse-grained sand, trace small gravel.					
8C							$16 = \underbrace{-\frac{\text{fine-to coarse-grained sand, trace small gravel.}}_{\text{Dark gray (10YR4/1), slightly moist, hard, clayey SILT, }}_{608}$					
							End of boring = 16.1 feet					

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: _Mor	tigomery Well #: G101
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #: G101
State- Plant Plane Coordinate: X_2,514,214.3 Y_876,551.8 (or) Latitude:	°' Longitude:° '
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: T. List
Consulting Firm: Hanson Professional Services Inc.	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: <u>Suzanna L. Simpson</u>	Date Started: <u>2/2/2010</u> Date Finished: <u>2/2/2010</u>
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:2/4/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>627.89</u> <u>-2.62</u> Top of Protective Casing
	<u>627.60</u> <u></u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>625.27</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>High-solids bentonite</u>	<u></u>
Installation Method:	
Setting Time: >24 hr	<u>∠</u> <u>617.96</u> <u>7.31</u> Static Water Level (After Completion) 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	<u>614.27</u> <u></u> Top of Seal
Setting Time: 20 min	612.1413.13 Top of Sand Pack
Type of Sand Pack:Quartz sand	600 50 15 68 Top of Sereen
Grain Size: 10/20 (sieve size)	
	<u>604.95</u> <u>20.32</u> Bottom of Screen
Type of Backfill Material: <u>Quartz sand</u> (if applicable)	Bottom of Well
Installation Method:	<u>603.35</u> <u>21.92</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
	Riser Pipe Length (feet) 18.01
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Cap (feet) 0.57
Riser Pine Below W T SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.64 Total Length af Casing (a) 22.22
Screen SS304 SS316 PTFE PVC OTHER:	Iotal Length of Casing (Teet) 25.22 Screen Slot Size ** (inches) 0.010

Well Completion Form (revised 02/06/02)

**Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agency	Well Completion	on Report
Site #: County: _ Mon	ntgomery Well #:G10	2 (MW3S)
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #:	SB-03a
State- Plant Plane Coordinate: X_2,514,531.5 Y_876,554.8 (or) Latitude:	°'" Longitude:°	""
Surveyed By: <u>Darren E. Forgy</u>	IL Registration #:035-003637	
Drilling Contractor:	Driller: <u>B. Williamson</u>	
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:	000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>	
Logged By:Testing Services Corp	Date Started: Date Finished:	4/28/2006
Report Form Completed By: <u>Rhonald W. Hasenyager</u>	Date:6/7/2006	
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ± (MSL)* (BGS)	ît.)
	<u>629.45</u> <u>-3.75</u> Top of Protect	ive Casing
		ipe
Type of Surface Seal: <u>Concrete</u>	<u>625.70</u> <u>0.00</u> Ground Surfac	e
Type of Annular Sealant: Bentonite chins	<u>623.70</u> <u></u> Top of Annula	r Sealant
Installation Method: Gravity		
Setting Time: +24 hr	\bigtriangledown 618.67 7.03 Static Water L	evel
	(After Completion	on) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry (choose one)		
Installation Method: <u>Gravity</u>	<u>623.70</u> <u>2.00</u> Top of Seal	
Setting Time: <u>+24 hr.</u>	<u>616.20</u> <u>9.50</u> Top of Sand P	ack
Type of Sand Pack: <u>Quartz sand</u>		
Grain Size: #5 (sieve size)	$\underline{\underline{613.68}} \underline{12.02} \text{Top of Screen}$	
Installation Method: <u>Gravity</u>	(00.02 16.70 D	
Type of Backfill Material:	<u>608.52</u> <u>16.78</u> Bottom of Scr <u>608.55</u> <u>17.15</u> Bottom of We	een Il
(if applicable)		
Installation Method:	<u>608.55</u> <u>17.15</u> Bottom of Bor * Referenced to a National Geodetic Datum	ehole
		,
	Diameter of Borehole (inch	, s) 80
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inch	es) 2.0
	Protective Casing Length (fe	et) 5.0
Protective Cacing SC204 SC214 PTEE DVG OTHER	Riser Pipe Length (fe	et) 15.28
Riser Pine Above W.T. SS304 SS316 PTFE PVC OTHER	Bottom of Screen to End Cap (fe	(0.37)
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (15	$\frac{4.70}{20.41}$
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inch	es) 0.010

Well Completion Form (revised 02/06/02)

**Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agency	y		Well	Completion	Report
Site #:	County:Mo	ontgomery	7	W	/ell #:G	103
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	у		В	orehole #:	G103
State- Plant Plane Coordinate: X_2,514,50	1.2 Y <u>876,199.5</u> (or) Latitude	:°	· · · · ·	<u> Longitud</u>	e:°	
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regi	stration #:035-0	003507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Prot</u>	fessional Services Inc.	Geologi	ist: <u>Rhonald W</u>	. Hasenyage	r, LPG #196-000	0246
Drilling Method: <u>Hollow sterr</u>	n auger	Drilling	g Fluid (Type): <u>n</u>	/a		
Logged By: <u>Suzanna L. Simp</u>	son	_ Date St	arted: <u>2/15/2</u>	010 Dat	e Finished: 2/	15/2010
Report Form Completed By:	izanna L. Simpson	_ Date: _	2/18/2010			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			634.07	3.08	Top of Protective	Casing
	F		633.80	-2.81	Top of Riser Pipe	2
Type of Surface Seal: <u>Concrete</u>			630.99	0.00	Ground Surface	
Type of Annular Sealant: High-	solids bentonite		624.93	6.06	Top of Annular S	lealant
Installation Method: Tremie	e ()					
Setting Time:24 hr.		Σ	614.00	16.99	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry				(i iidi compision)	5, 1, 2010
Installation Method: <u>Gravit</u>	(choose one) y		618.51	12.48	Top of Seal	
Setting Time: <u>8 min</u>	X	X	617.35	13.64	Top of Sand Pack	x
Type of Sand Pack: Quartz sand	1					
Grain Size: 10/20 (sie	eve size)		615.11	15.88	Top of Screen	
Installation Method: <u>Gravit</u>	<u>y</u>		610 32	20.67	Bottom of Screen	
Type of Backfill Material: <u>n/a</u>	(if applicable)		609.90	21.09	Bottom of Well	L.
Installation Method: <u>n/a</u>	(1 49)104010)		609.90	21.09	Bottom of Boreho	ole
			* Referenced to	a National Geodet	ic Datum	
		1	CA	SING MEA	SUREMENTS]
WELL CONS	STRUCTION MATERIALS		Diameter of Borel	nole	(inches)	8.0
(Choose or	ne type of material for each area)		ID OI KISET Pipe	Length	(inches)	2.0
			Riser Pipe Length	Length	(Teet)	18.69
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Bottom of Screen	to End Cap	(feet)	0.42
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	:	Screen Length (1st slot to last slo	t) (feet)	4.79
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	:	Total Length of C	asing	(feet)	23.90
Screen	SS304 SS316 PTFE PVC OTHER	:	Screen Slot Size *	*	(inches)	0.010

Well Completion Form (revised 02/06/02)

^{**}Hand-Slotted Well Screens Are Unacceptable
Illinois Environmental Protection Agency	Well Completion Rep	ort
Site #: County: _ Mo	ontgomery Well #: <u>R104</u>	
Site Name: AEG Coffeen Power Station CCB Management Facility	y Borehole #: R104	
State Plane Coordinate: X 2,514,503.4 Y 875,857.8 (or) Latitude:	e: <u>39° 4' 10"</u> Longitude: <u>-89° 23' 5</u>	56.8 "
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-003507	
Drilling Contractor: Layne-Western Co	Driller: D. Mahurin	
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W. Hasenyager, LPG #196-000246	
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>	
Logged By: <u>Rhonald W. Hasenyager</u>	Date Started: Date Finished: 10/8/201	0
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:10/19/2010	
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)	_
	(MSL)* (BGS)	
	<u>632.02</u> <u>-2.99</u> Top of Protective Casing	5
	<u>631.84</u> <u></u> Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>	<u>629.03</u> <u>0.00</u> Ground Surface	
	<u>627.03</u> <u>2.00</u> Top of Annular Sealant	
Type of Annular Sealant: High-solids bentonite	4	
Setting Time: 224 kg	√ 612.85 16.18 Static Water Level	
Setung Time. <u>24 m.</u>	V 012.85 10.16 State water Level (After Completion) 11/15/20	010
Type of Bentonite Seal Granular Pellet Slurry		
Installation Method: <u>Gravity</u>	<u>617.03</u> <u></u> Top of Seal	
Setting Time: <u>15 min</u>	616.01 <u>13.02</u> Top of Sand Pack	
Type of Sand Pack: Quartz sand		
Grain Size: 10/20 (sieve size)	<u></u>	
Installation Method:		
Turna of Baalefill Matarial: n/a	$= \underbrace{\begin{array}{c} 609.71 \\ 609.18 \end{array}} \underbrace{\begin{array}{c} 19.32 \\ 19.85 \end{array}}_{\text{Bottom of Screen}} \operatorname{Bottom of Well}$	
(if applicable)	<u>009.16</u> Bottom of wen	
Installation Method:	609.18 19.85 Bottom of Borehole * Referenced to a National Geodetic Datum	
	CASING MEASUREMENTS	
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.()
(Choose one type of material for each area)	Protective Casing Length (feet) 5.0)
	Riser Pipe Length (feet) 17.6	50
Protective Casing SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.4	53
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Discreption Discreption	Screen Length (1st slot to last slot) (feet) 4.4	53
Kisei Pipe Below W. I. SS304 SS316 PIFE PVC OTHER: Screen SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 22.	56 010
	Screen Slot Size ** (inches) 0.0	010

Illinois Environmental Protection Agency	Well Completion Report
Site #: County:	ontgomery Well #: G104
Site Name: AEG Coffeen Power Station CCB Management Facility	y Borehole #: G104
State- Plant Plane Coordinate: X_2,514,505.0 Y_875,849.3 (or) Latitude:	:°' Longitude:°'
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type): <u>n/a</u>
Logged By: Suzanna L. Simpson	Date Started: <u>2/15/2010</u> Date Finished: <u>2/15/2010</u>
Report Form Completed By:Suzanna L. Simpson	Date:2/18/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>633.29</u> <u>-3.17</u> Top of Protective Casing
	<u>632.94</u> 2.82 Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>630.12</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: High-solids bentonite	<u>624.92</u> <u>5.20</u> Top of Annular Sealant
Installation Method: Tremie	
Setting Time: >24 hr.	$\underline{\nabla}$ <u>614.72</u> <u>15.40</u> Static Water Level (After Completion) $\frac{3}{1/2010}$
Type of Bentonite Seal Granular Pellet Slurry	
(choose one) Installation Method: Gravity	<u></u>
Setting Time: <u>10 min</u>	617.42 <u>12.70</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)	<u></u>
Type of Backfill Material: <u>Quartz sand</u> (if applicable)	$\underbrace{\begin{array}{c} \underline{610.51} \\ \underline{610.04} \end{array}}_{\underline{20.08}} \underbrace{\begin{array}{c} 19.61 \\ \underline{8} \end{array}}_{\underline{8} } Bottom of Screen}$
Installation Method: <u>Gravity</u>	<u>607.92</u> <u>22.20</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTFE PVC OTHER:	Riser Pipe Length (feet) 17.73 Steel Bottom of Screen to End Can 0.47
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4 70
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 22.90
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Screen Slot Size ** **Hand-Slotted Well Screens Are Unacceptable

Illinois Enviror	mental Protection Agency	y		Well	Completion	Report
Site #:	County: <u>M</u>	ontgomery		W	/ell #:G	105
Site Name: <u>AEG Coffeen Pow</u>	ver Station CCB Management Facilit	у		В	orehole #:	<u>G105</u>
State- Plant Plane Coordinate: X_2,514,509	<u>9.2</u> Y <u>875,499.7</u> (or) Latitude	e:°		Longitud	e:°	
Surveyed By:Jeffrey D. Emric	ж	IL Registra	tion #:035-0	03507		
Drilling Contractor: <u>Layne-Wes</u>	stern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.	_ Geologist:	Rhonald W.	Hasenyage	r, LPG #196-000	0246
Drilling Method: <u>Hollow stem</u>	auger	_ Drilling Fl	uid (Type): <u>n/a</u>	1		
Logged By: <u>Suzanna L. Simps</u>	son	_ Date Starte	ed: <u>2/16/20</u>	10 Dat	e Finished:/	16/2010
Report Form Completed By:	zanna L. Simpson	Date:	2/18/2010			
ANNULAR SPAC	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			632.40	3.14	Top of Protective	Casing
	Ę		632.08	-2.82	Top of Riser Pipe	•
Type of Surface Seal: <u>Concrete</u>			629.26	0.00	Ground Surface	
Type of Annular Sealant: <u>High-sealant</u>	olids bentonite		623.80	5.46	Top of Annular S	lealant
Installation Method: <u>Tremie</u>						
Setting Time: <u>>24 hr.</u>		$\overline{\Delta}$	613.18	16.08	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Gram	Ilar Pellet Slurry (choose one)					
Installation Method: <u>Gravity</u>	 	x x	616.55	12.71	Top of Seal	
Setting Time: <u>10 min</u>	×	×	615.50	13.76	Top of Sand Pack	ζ.
Type of Sand Pack: <u>Quartz sand</u>						
Grain Size: <u>10/20</u> (siew	ve size)		613.15		Top of Screen	
Installation Method: <u>Gravity</u>	·		608 36	20.90	Bottom of Screen	
Type of Backfill Material:Quart	z sand		607.89	21.37	Bottom of Well	L.
Installation Method: <u>Gravity</u>	(ii appreade)		606.80 * Referenced to a	22.46 National Geodeti	Bottom of Boreh	ble
			CAS	SING MEA	SUREMENTS	8.0
WELL CONS (Choose ond	TRUCTION MATERIALS e type of material for each area)		of Riser Pipe		(inches)	2.0
	<i>,</i>	Pr	otective Casing L	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTEE PVC OTHER	· Steel	ser Pipe Length	- End C	(feet)	18.93
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER		ereen Length (1)	D End Cap	(feet)	<u>0.4/</u> <u>4</u> 70
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	: To	otal Length of Cas	sing	(feet)	24.19
Screen	SS304 SS316 PTFE PVC OTHER	:	reen Slot Size **		(inches)	0.010

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Environmental Protection Agency		Well Completion Report
Site #: County:	tgomery	Well #: G106
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>		Borehole #: G106
State- Plant Plane Coordinate: X_2,514,512.8 Y_875,149.8 (or) Latitude:	· ''	Longitude:°'"
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003	507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>	
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W. Ha</u>	senyager, LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type): <u>n/a</u>	
Logged By: <u>Suzanna L. Simpson</u>	Date Started:2/16/2010	Date Finished:2/16/2010
Report Form Completed By:Suzanna L. Simpson	Date:2/18/2010	_
ANNULAR SPACE DETAILS	Elevations E (MSL)*	Depths (0.01 ft.) (BGS)
		<u>-3.06</u> Top of Protective Casing
		-2.76 Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>628.39</u>	0.00 Ground Surface
Type of Annular Sealant: <u>High-solids bentonite</u>	622.94	5.45 Top of Annular Sealant
Installation Method:		
Setting Time: >24 hr	<u>615.77</u>	12.62 Static Water Level (After Completion) 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)		
Installation Method: <u>Gravity</u>	<u>617.44</u>	10.95 Top of Seal
Setting Time: <u>10 min</u>	<u> </u>	12.29 Top of Sand Pack
Type of Sand Pack:Quartz sand	614.02	14.37 Top of Screen
Grain Size: <u>10/20</u> (sieve size)		
Installation Method: <u>Gravity</u>	609.43	18.96 Bottom of Screen
Type of Backfill Material:		19.44 Bottom of Well
Installation Method: Gravity		20.45 Bottom of Borehole
	CASIN Diameter of Borehole	G MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	(inches) 2.0
	Protective Casing Leng	gth (feet) 5.0
Protective Casing SS304 SS316 DTEE DVC OTHER	Steel Riser Pipe Length	(feet) 17.13
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to E	na Lap (feet) 0.48 of to last slot) (feet) 4.60
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing	g (feet) 22.30
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **	(inches) 0.010

**Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agency	у		Well	Completion	Report
Site #:	County:	ontgomery		W	Vell #: G	107
Site Name: AEG Coffeen Por	wer Station CCB Management Facilit	y		В	orehole #: 0	G107
State- Plant Plane Coordinate: X_2,514,355	8.3 Y 874,994.3 (or) Latitude	e:°	"	Longitud	e:°	
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regis	stration #:035-00	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	_ Driller:	T. List			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	_ Geologis	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	0246
Drilling Method: <u>Hollow stem</u>	auger	_ Drilling	Fluid (Type): <u>n/a</u>	l		
Logged By: <u>Suzanna L. Simp</u>	son	_ Date Sta	arted: <u>2/17/20</u>	10 Dat	e Finished:/	17/2010
Report Form Completed By:Su	zanna L. Simpson	_ Date: _	2/18/2010			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
			630.60	-2.81	Top of Protective	Casing
			630.23	2.44	Top of Riser Pipe	2
Type of Surface Seal: <u>Concrete</u>			627.79	0.00	Ground Surface	
			624.08	3.71	Top of Annular S	ealant
Type of Annular Sealant: <u>High-s</u>	solids bentonite	<u> </u>				
Installation Method: <u>Tremie</u>	2		(17.20	10.40		1
Setting Time: <u>>24 hr.</u>		⊻	017.39	10.40	(After Completion)	3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method: <u>Gravit</u>	y	K X	617.41	10.38	Top of Seal	
Setting Time: <u>8 min</u>	Ž	×	616.24		Top of Sand Pack	ζ.
Type of Sand Pack: <u>Quartz sand</u>	1					
Grain Size: 10/20 (sig	eve size)		613.92	13.87	Top of Screen	
Installation Method: <u>Gravit</u>	y		609 29	18 50	Bottom of Screer	
Type of Backfill Material:Quar	tz sand		608.82	18.97	Bottom of Well	
Installation Method: Gravit	(ii applicatic)		607.08	20.71	Bottom of Boreho	ole
			* Referenced to a l	National Geodet	ic Datum	
			CAS	ING MEA	SUREMENTS	
WELL CONS	TDUCTION MATERIALS	-	Diameter of Boreho	le	(inches)	8.0
(Choose or	ne type of material for each area)	-	ID of Riser Pipe		(inches)	2.0
		-	Protective Casing L Riser Pine Length	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTFE PVC OTHER	: Steel	Bottom of Screen to	End Cap	(feet)	0.47
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	:	Screen Length (1s	t slot to last slo	t) (feet)	4.63
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	:	Total Length of Cas	sing	(feet)	21.41
Screen	SS304 SS316 PTFE PVC OTHER	:	Screen Slot Size **		(inches)	0.010

Screen Slot Size ** **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agency		Well	Completion	Report
Site #: County:	ntgomery	W	Tell #: G1	08
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>		В	orehole #:	G108
State- Plant Plane Coordinate: X_2,514,248.3 Y_874,948.8 (or) Latitude:	<u> </u>	Longitude	e:°	
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-00)3507		
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W. I</u>	Hasenyager	r, LPG #196-000	246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>			
Logged By: Diane M. Lamb	Date Started:2/12/201	10 Date	e Finished: <u>2/1</u>	2/2010
Report Form Completed By: Diane M. Lamb	Date:2/19/2010			
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
	630.52	3.02	Top of Protective	Casing
		2.72	Top of Riser Pipe	
Type of Surface Seal: Concrete	<u>627.50</u>	0.00	Ground Surface	
Type of Annular Sealant: High-solids bentonite	622.50	5.00	Top of Annular S	ealant
Installation Method:				
Setting Time: _>24 hr	<u>618.57_</u>	8.93	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry				
Installation Method: <u>Granular</u>	614.00	13.50	Top of Seal	
Setting Time: <u>10 min</u>	<u>_612.80</u>	14.70	Top of Sand Pack	
Type of Sand Pack:Quartz sand				
Grain Size: 10/20 (sieve size)		16.82	Top of Screen	
Installation Method: <u>Gravity</u>	606.00	21 50	Bottom of Screen	
Type of Backfill Material: <u>n/a</u>	605.50	22.00	Bottom of Well	
Installation Method:	605.50	22.00	Bottom of Boreho	ble
	* Referenced to a N	National Geodetic	c Datum	
	CAS	ING MEAS	SUREMENTS	
WELL CONSTRUCTION MATERIALS	Diameter of Boreho	le	(inches)	8.0
(Choose one type of material for each area)	Protective Casing L	ength	(inches)	5.0
	Riser Pipe Length		(feet)	19.54
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to	End Cap	(feet)	0.50
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st	t slot to last slot	t) (feet)	4.68
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Cas	ing	(feet)	24.72
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **		(inches)	0.010

Illinois Enviro	nmental Protection Agency	y			Well	Completion	n Report
Site #:	County:M	ontgomery	7		W	/ell #:G	109
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	у			В	orehole #:	G109
State- Plant Plane Coordinate: X 2,514,13	7.8 Y <u>874,970.1</u> (or) Latitude	:	·		Longitud	e:°	
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regi	stration #: _	035-0	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	_ Driller:	T. List				
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geolog	st: <u>Rhona</u>	ald W.	Hasenyage	r, LPG #196-00	0246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	Fluid (Type): <u>n/a</u>	a		
Logged By: <u>Diane M. Lamb</u>		_ Date St	arted:2	2/11/20	0 <u>10</u> Date	e Finished: 2/	11/2010
Report Form Completed By:	ane M. Lamb	Date:	2/19/2	2010			
ANNULAR SPA	CE DETAILS		Eleva (MS	tions L)*	Depths (BGS)	(0.01 ft.)
			630.	08	-2.88	Top of Protective	e Casing
	F		_629.	76	-2.56	Top of Riser Pip	e
Type of Surface Seal: <u>Concrete</u>			<u> </u>	20	0.00	Ground Surface	
Type of Annular Sealant: High-	solids bentonite		<u>621.</u>	70	5.50	Top of Annular	Sealant
Installation Method: Tremie							
Setting Time: <u>>24 hr.</u>		Ā	618.	35	8.85	Static Water Lev (After Completion)	el 3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry						
Installation Method: <u>Granul</u>	(choose one) ar		_614.	90	12.30	Top of Seal	
Setting Time: <u>10 min</u>	X	×	_614.	60	12.60	Top of Sand Pac	k
Type of Sand Pack: <u>Quartz sanc</u>	1						
Grain Size: <u>10/20</u> (sid	eve size)		611.	81	15.39	Top of Screen	
Installation Method: <u>Gravit</u>	<u>y</u>		607.	27	19.93	Bottom of Screen	1
Type of Backfill Material: <u>n/a</u>	(if applicable)		606.	70	20.50	Bottom of Well	
Installation Method: <u>n/a</u>			<u>606.</u> * Pafar	70	20.50	Bottom of Boreh	ole
			Kelen	enced to a	National Geoden		
			D: /	CAS	SING MEAS	SUREMENTS	8.0
WELL CONS	STRUCTION MATERIALS		ID of Riser	Boreh Pine	ole	(inches)	<u>8.0</u> 2.0
(Choose of	the the or material for each area)		Protective C	Casing I	ength	(feet)	5.0
			Riser Pipe I	ength		(feet)	17.95
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Bottom of S	Screen t	o End Cap	(feet)	0.57
Riser Pipe Above W.T.	SS304 SS316 PTFE (PVC) OTHER		Screen Leng	gth (1	st slot to last slo	t) (feet)	4.54
Screen	SS304 SS316 PTFE (PVC) OTHER	· · · · · · · · · · · · · · · · · · ·	Total Lengt	h of Ca Size **	sing	(feet)	23.06
			Scieen Siot	SIZE **		(incnes)	0.010

Illinois Enviro	nmental Protection Agency	y		Well	Completion	Report
Site #:	County:Mo	ontgomery	,	W	/ell #:G	10
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	у		В	orehole #:	G110
State- Plant Plane Coordinate: X_2,514,05	7.7 Y <u>875,015.4</u> (or) Latitude	:°		Longitud	e:°	
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regi	stration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologi	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	Fluid (Type): <u>n/a</u>	1		
Logged By: <u>Diane M. Lamb</u>		Date Sta	arted: <u>2/11/20</u>	<u>10</u> Dat	e Finished: 2/	1/2010
Report Form Completed By:	ane M. Lamb	Date:	2/19/2010			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			629.96	2.94	Top of Protective	Casing
	Ę		629.65	-2.63	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>			<u> </u>	0.00	Ground Surface	
Type of Annular Sealant: <u>High-</u>	solids bentonite		621.86	5.16	Top of Annular S	ealant
Installation Method: Tremie						
Setting Time: <u>>24 hr.</u>		∑	617.52	9.50	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method: <u>Granul</u>	(choose one) ar		615.46	11.56	Top of Seal	
Setting Time: <u>10 min</u>			614.06	12.96	Top of Sand Pacl	ζ.
Type of Sand Pack: <u>Quartz sanc</u>	1					
Grain Size: <u>10/20</u> (sig	eve size)		611.97	15.05	Top of Screen	
Installation Method: <u>Gravit</u>	<u>y</u>		607.43	19.59	Bottom of Screen	L
Type of Backfill Material: <u>n/a</u>	(if applicable)		606.86	20.16	Bottom of Well	
Installation Method: <u>n/a</u>			606.86 * Referenced to a	20.16	Bottom of Boreh	ole
			Referenced to a	National Geodet		
			CAS	ING MEA	SUREMENTS	0.0
WELL CONS	STRUCTION MATERIALS		Diameter of Boreho	ble	(inches)	8.0
(Choose of	ie type of material for each afea)		Protective Casing L	ength	(feet)	5.0
			Riser Pipe Length		(feet)	17.68
Protective Casing	SS304 SS316 PTFE PVC OTHER:	Steel	Bottom of Screen to	End Cap	(feet)	0.57
Riser Pipe Above W.T.	SS304 SS316 PTFE (PVC) OTHER:		Screen Length (19	st slot to last slo	t) (feet)	4.54
Screen	SS304 SS316 PTFE PVC OTHER		Total Length of Cas	sing	(feet)	22.79
Serven	SSSOT SSSTO THE CITC OTHER.		Screen Slot Size **		(inches)	0.010

^{**}Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agency	y			Well	Completion	Report
Site #:	County:M	ontgomery	7		W	/ell #:G	11
Site Name: <u>AEG Coffeen Pov</u>	wer Station CCB Management Facilit	у			В	orehole #:	G111
State- Plant Plane Coordinate: X 2,513,98	1.7 Y 875,058.7 (or) Latitude	:		_'"	Longitud	e:°	"
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regi	stration	#:	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	<u> </u>	ist			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologi	ist: <u>R</u>	honald W.	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	; Fluid (Type): <u>n/a</u>	a		
Logged By: Diane M. Lamb		_ Date St	arted: _	2/10/20	0 <u>10</u> Date	e Finished:/	1/2010
Report Form Completed By:	ane M. Lamb	Date:	2	/19/2010			
ANNULAR SPA	CE DETAILS		El	evations (MSL)*	Depths (BGS)	(0.01 ft.)	
			_	630.19	2.95	Top of Protective	Casing
	T If		_	<u>629.90</u>	-2.66	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>			» — —	627.24	0.00	Ground Surface	
Type of Annular Sealant: High-s	solids bentonite		/ _	622.52	4.72	Top of Annular S	ealant
Installation Method:							
Setting Time: <u>>24 hr.</u>		⊻	_	616.74	10.50	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry						
Installation Method: <u>Granul</u>	(choose one) ar	v v	_	616.41	10.83	Top of Seal	
Setting Time: <u>10 min</u>	X	X	_	614.52	12.72	Top of Sand Pacl	ς.
Type of Sand Pack: <u>Quartz sanc</u>	<u>I</u>						
Grain Size: <u>10/20</u> (sig	eve size)		-	612.63	14.61	Top of Screen	
Installation Method: <u>Gravit</u>	<u>y</u>			608.09	19.15	Bottom of Screen	L
Type of Backfill Material: <u>n/a</u>	(if applicable)		_	607.52	19.72	Bottom of Well	
Installation Method: <u>n/a</u>				607.52	<u>19.72</u>	Bottom of Boreh	ole
			Diama	CAS	SING MEAS	SUREMENTS	8.0
WELL CONS	TRUCTION MATERIALS		ID of H	Riser Pipe	JIE	(inches)	2.0
	a gro or material for each area)		Protect	tive Casing L	ength	(feet)	5.0
		1	Riser F	Pipe Length		(feet)	17.27
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Bottom	n of Screen to	o End Cap	(feet)	0.57
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER		Screen	Length (1	st slot to last slo	t) (feet)	4.54
Screen	SS304 SS316 PTFE PVC OTHER	· · · · · · · · · · · · · · · · · · ·	1 otal L Screen	ength of Cas Slot Size **	sing	(inches)	0.010
	1			5101 5120		(menes)	0.010

Illinois Enviro	nmental Protection Agency	y		Well	Completion	Report
Site #:	County:Mo	ontgomery	,	W	/ell #:G	19
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	у		В	orehole #:	G119
State- Plant Plane Coordinate: X 2,513,90	7.7 Y <u>875,675.0</u> (or) Latitude	:	'	Longitud	e:°	"
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regi	stration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Prot</u>	fessional Services Inc.	Geologi	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow sterr</u>	auger	Drilling	Fluid (Type): <u>n/a</u>	1		
Logged By: <u>Suzanna L. Simp</u>	son	_ Date St	arted: <u>2/9/201</u>	1 <u>0</u> Dat	e Finished:/	9/2010
Report Form Completed By: <u>Su</u>	izanna L. Simpson	_ Date: _	2/18/2010			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			631.85	3.00	Top of Protective	Casing
	F		631.55	2.70	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>			628.85	0.00	Ground Surface	
Type of Appular Scalant High	rolida hontonita		623.57	5.28	Top of Annular S	ealant
Installation Method: Tremi						
Setting Time: >24 hr.	<u> </u>	∇	617.59	11.26	Static Water Leve	el
					(After Completion)	3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry (choose one)	YT.				
Installation Method: <u>Gravit</u>	<u>y</u>	X	614.87	13.98	Top of Seal	
Setting Time: <u>15 min</u>	X	×	613.57	15.28	Top of Sand Pack	2
Type of Sand Pack: <u>Quartz sand</u>	1					
Grain Size: <u>10/20</u> (si	eve size)		611.56	17.29	Top of Screen	
Installation Method: <u>Gravit</u>	<u>y</u>		607.02	21.83	Bottom of Screer	
Type of Backfill Material: <u>n/a</u>	(if applicable)		606.47	22.38	Bottom of Well	·
Installation Method: <u>n/a</u>			606.47	22.38	Bottom of Boreho	ole
			* Referenced to a	National Geodet	ic Datum	
		1	CAS	ING MEA	SUREMENTS	1
WELL CONS	STRUCTION MATERIALS		Diameter of Boreho	ole	(inches)	8.0
(Choose or	ne type of material for each area)		ID of Riser Pipe		(inches)	2.0
			Protective Casing L	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Rottom of Screen to	Find Can	(feet)	0.55
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER		Screen Length (14	st slot to last slo	t) (feet)	4.54
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER		Total Length of Cas	sing	(feet)	25.08
Screen	SS304 SS316 PTFE PVC OTHER		Screen Slot Size **		(inches)	0.010

Illinois Environmental Protection Agency		Well	Completion	Report
Site #: County: _Mor	ntgomery	W	fell #:G1	.20
Site Name: AEG Coffeen Power Station CCB Management Facility		Во	orehole #: 0	6120
State- Plant Plane Coordinate: X_2,513,905.8 Y_875,854.4 (or) Latitude:	<u> </u>	Longitude	o:o	
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-00)3507		
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W.</u>	Hasenyager	<u>, LPG #196-000</u>	246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>			
Logged By: <u>Suzanna L. Simpson</u>	Date Started:2/8/201	0 Date	Finished:2/	8/2010
Report Form Completed By:Suzanna L. Simpson	Date: 2/18/2010			
ANNULAR SPACE DETAILS	Elevations	Depths	(0.01 ft.)	
	(MSL) ¹ 632.18	-2.88	Top of Protective	Casing
		-2.57	Top of Riser Pipe	e doning
Type of Surface Seal: <u>Concrete</u>	<u>629.30</u>	0.00	Ground Surface	
Type of Annular Sealant: High-solids bentonite	624.22	5.08	Top of Annular S	ealant
Installation Method: Tremie				
Setting Time: _>24 hr.	<u>615.45</u>	13.85	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry				
Installation Method: <u>Gravity</u>	618.14	11.16	Top of Seal	
Setting Time: <u>8 min</u>	<u>616.22</u>	13.08	Top of Sand Pack	C.
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)		15.10	Top of Screen	
Type of Backfill Material: Quartz sand	<u>609.68</u>	<u>19.62</u> 20.21	Bottom of Screen	
(if applicable)			Dottoin of wen	
Installation Method: <u>Gravity</u>	607.22 * Referenced to a 1	22.08 National Geodetic	Bottom of Boreho	ble
	CAS	ING MEAS	SUREMENTS	
	Diameter of Boreho	le	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe		(inches)	2.0
	Protective Casing L	ength	(feet)	5.0
Protective Casing SS304 SS316 DTEE DVC OTHER	Steel	E LC	(feet)	17.67
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to	End Cap	(feet)	0.59
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Cas	ing	(feet)	22.78
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **	-0	(inches)	0.010

Screen Slot Size ** **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: Mon	ntgomery Well #: <u>G121</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facility</u>	Borehole #: G121
State- Plant Plane Coordinate: X_2,513,904.4 Y_875,964.6 (or) Latitude:	°' Longitude:°'
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: <u>Suzanna L. Simpson</u>	Date Started: <u>2/4/2010</u> Date Finished: <u>2/4/2010</u>
Report Form Completed By:Suzanna L. Simpson	Date:2/18/2010
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	633.14 Top of Protective Casing
	<u>632.83</u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>629.57</u> <u>0.00</u> Ground Surface
Type of Annular Sealant:High-solids bentonite	<u>624.41</u> <u>5.16</u> Top of Annular Sealant
Installation Method:	
Setting Time: <u></u>	
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	616.8112.76 Top of Seal
Setting Time: <u>7 min</u>	<u>615.49</u> <u>14.08</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u>	
Grain Size: 10/20 (sieve size)	$\underbrace{\underline{612.78}}_{16.79} \text{Top of Screen}$
Installation Method: <u>Gravity</u>	608 10 21 47 Bottom of Screen
Type of Backfill Material: Quartz sand	$\underbrace{\begin{array}{c} \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline \hline$
Installation Method: <u>Gravity</u>	<u>605.41</u> <u>24.16</u> Bottom of Borehole
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	Diameter of Borehole(inches)8.0ID of Riser Pipe(inches)2.0
Contraction of the second second second second second second second second second second second second second s	Protective Casing Length (feet) 5.0
Protostivo Cosing S2204 S214 PEEE DVG OTHER	Riser Pipe Length (feet) 20.05
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.48
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 25.21
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

Screen Slot Size **

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Well Completion Form (revised 02/06/02)

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: _Mor	ntgomery Well #: <u>G122</u>
Site Name AEG Coffeen Power Station CCB Management Facility	Borehole # G122
State- Plant Plane Coordinate: X_2,513,902.8 Y_876,080.1 (or) Latitude:	°' Longitude:°'"
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-003507
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W. Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>
Logged By: Diane M. Lamb	Date Started:2/4/2010 Date Finished:2/4/2010
Report Form Completed By: Diane M. Lamb	Date: <u>2/9/2010</u>
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	632.98 Top of Protective Casing
Type of Surface Seal: <u>Concrete</u>	<u>629.86</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>High-solids bentonite</u>	<u>625.01</u> <u>4.85</u> Top of Annular Sealant
Installation Method:	
Setting Time: <u>>24 hr.</u>	$\underline{\frown} \qquad \underline{617.02} \qquad \underline{12.84} \qquad \text{Static Water Level} \\ \text{(After Completion) } 3/1/2010$
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	<u>617.01</u> <u>12.85</u> Top of Seal
Setting Time: <u>10 min</u>	<u>615.41</u> <u>14.45</u> Top of Sand Pack
Type of Sand Pack:Quartz sand	613.35 16.51 Top of Screen
Grain Size: <u>10/20</u> (sieve size)	
Installation Method: <u>Gravity</u>	<u>_608.81</u> <u>_21.05</u> Bottom of Screen
Type of Backfill Material: <u>Quartz sand</u> (if applicable)	<u>608.20</u> <u>21.66</u> Bottom of Well
Installation Method:	<u>608.01</u> <u>21.85</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protoctive Cocing SCIA4 SCIA6 NEED NAC OFFICE	Riser Pipe Length (feet) 19.34
Flotecurve clasing SS304 SS316 PTFE PVC OTHER: Riser Pine Above W T SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to End Cap (feet) 0.61
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.54 Total Length of Casing (fast) 24.40
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size ** (inches) 0.010

**Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agency	y		Well	Completion	Report
Site #:	County:M	ontgomery		W	Vell #: G1	23
Site Name: <u>AEG Coffeen Pov</u>	ver Station CCB Management Facilit	У		В	orehole #:	G123
State- Plant Plane Coordinate: X 2,513,901	1.5 Y <u>876,189.6</u> (or) Latitude	:°		Longitud	e:°	
Surveyed By:Jeffrey D. Emri	ck	IL Regis	stration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologi	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	Fluid (Type): <u>n/a</u>	a		
Logged By: <u>Diane M. Lamb</u>		Date Sta	arted: <u>2/3/201</u>	10 Dat	e Finished:/	4/2010
Report Form Completed By:	ane M. Lamb	Date:	2/9/2010			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			633.29	-3.16	Top of Protective	Casing
	F		632.96	-2.83	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>				0.00	Ground Surface	
Type of Annular Sealant: High-s	solids bentonite		625.06	5.07	Top of Annular S	ealant
Installation Method: Tremie	, 4					
Setting Time: <u>>24 hr.</u>		Σ	614.15	15.98	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method: <u>Gravity</u>	(cnoose one)		612.31	_17.82_	Top of Seal	
Setting Time: <u>10 min</u>		X	611.14	18.99	Top of Sand Pack	c.
Type of Sand Pack: <u>Quartz sand</u>	L					
Grain Size: <u>10/20</u> (sie	eve size)		609.19	20.94	Top of Screen	
Installation Method: <u>Gravity</u>	/		604.67	25.46	Bottom of Screen	
Type of Backfill Material: <u>n/a</u>	(if applicable)		604.06	26.07	Bottom of Well	
Installation Method: <u>n/a</u>	(11		604.06		Bottom of Boreho	ole
			* Referenced to a	National Geodet	ic Datum	
		Г	CAS	SING MEA	SUREMENTS	
WELL CONS	TRUCTION MATERIALS	-	Diameter of Boreho	ole	(inches)	8.0
(Choose or	e type of material for each area)	-	Protective Casing I	ength	(inches)	5.0
			Riser Pipe Length		(feet)	23.77
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Bottom of Screen to	o End Cap	(feet)	0.61
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	:	Screen Length (1	st slot to last slo	ot) (feet)	4.52
Riser Pipe Below W.T.	SS304 SS316 PTFE (PVC) OTHER		Total Length of Ca	sing	(feet)	28.90
Screen	SS304 SS316 PIFE PVC OTHER		Screen Slot Size **	:	(inches)	0.010

Illinois Environmental Protection Agency		Well Co	mpletion	Report
Site #: County: _ Mon	tgomery	Well #	4: <u>G1</u>	24
Site Name: AEG Coffeen Power Station CCB Management Facility		Boreho	ole#: C	6124
State- Plant Plane Coordinate: X 2,513,900.3 Y 876,304.9 (or) Latitude:		Longitude:	o	
Surveyed By: Jeffrey D. Emrick	IL Registration #:035-0	03507		
Drilling Contractor: <u>Layne-Western Co</u>	Driller: <u>T. List</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W.</u>	Hasenyager, LH	PG #196-000	246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>n/a</u>	1		
Logged By: <u>Suzanna L. Simpson</u>	Date Started: 2/3/202	10 Date Fin	ished: <u>2/</u> .	3/2010
Report Form Completed By: <u>Suzanna L. Simpson</u>	Date:2/5/2010			
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
	633.70	<u>-3.28</u> Top	p of Protective	Casing
	633.39	<u>-2.97</u> Top	p of Riser Pipe	
Type of Surface Seal: Concrete	<u>630.42</u>	<u>0.00</u> Gro	ound Surface	
Type of Annular Sealant: High-solids bentonite	625.45	<u>4.97</u> Top	p of Annular S	ealant
Installation Method:				
Setting Time: _>24 hr.	<u>Z</u> <u>619.43</u>	<u>10.99</u> Sta	tic Water Leve fter Completion)	l 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry				
Installation Method: <u>Gravity</u>	618.34	<u>12.08</u> Top	p of Seal	
Setting Time: <u>10 min</u>	<u>616.50</u>	<u>13.92</u> Top	p of Sand Pack	
Type of Sand Pack:Quartz sand				
Grain Size: 10/20 (sieve size)	<u>614.44</u>	<u>15.98</u> Top	p of Screen	
Installation Method: <u>Gravity</u>	609.91	20.51 Por	ttom of Saraan	
Type of Backfill Material:	609.36	<u>20.31</u> Bot <u>21.06</u> Bot	ttom of Well	
Installation Method:		21.97 Bot	ttom of Boreho	le
	CAS	SING MEASUR	REMENTS	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	ne	(inches)	<u> </u>
(Protective Casing I	ength	(feet)	5.0
Protective Cacing SC204 SC216 PTEE DVC OTHED.	Steel Riser Pipe Length	E 10	(feet)	18.95
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bottom of Screen to	D End Cap	(feet)	0.55
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Ca	Sing	(feet)	24.03
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **		(inches)	0.010

Screen Slot Size **

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Well Completion Form (revised 02/06/02)

Illinois Enviro	nmental Protection Agency	y		Well	Completion	Report
Site #:	County:Mo	ontgomery	7	W	/ell #:G	125
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facilit	у		В	orehole #:	G125
State- Plant Plane Coordinate: X 2,513,89	9.1 Y 876,409.5 (or) Latitude	:	·	Longitud	e:°	
Surveyed By: <u>Jeffrey D. Emri</u>	ck	IL Regi	stration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Prot</u>	fessional Services Inc.	Geologi	ist: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	0246
Drilling Method: <u>Hollow sterr</u>	auger	Drilling	g Fluid (Type): <u>n/a</u>	a		
Logged By: <u>Suzanna L. Simp</u>	son	_ Date St	arted: <u>2/2/201</u>	10 Dat	e Finished:/	3/2010
Report Form Completed By:	izanna L. Simpson	Date:	2/5/2010			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			633.82	3.14	Top of Protective	Casing
	F		633.51	-2.83	Top of Riser Pipe	2
Type of Surface Seal: <u>Concrete</u>			<u>630.68</u>	0.00	Ground Surface	
Type of Annular Sealant: High-	solids bentonite		625.77	4.91	Top of Annular S	ealant
Installation Method: Tremie	e ()					
Setting Time:24 hr.		⊻	622.10	8.58	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method:	(choose one) y		617.15	13.53	Top of Seal	
Setting Time: <u>10 min</u>	X	X	615.95	14.73	Top of Sand Pack	x
Type of Sand Pack: <u>Quartz sand</u>	1					
Grain Size: <u>10/20</u> (si	eve size)		613.65	17.03	Top of Screen	
Installation Method: <u>Gravit</u>	<u>y</u>		609.12	21.56	Bottom of Screer	
Type of Backfill Material: <u>n/a</u>	(if applicable)	==	608.64	22.04	Bottom of Well	L
Installation Method: <u>n/a</u>	(1 497.1000)		_608.64	22.04	Bottom of Boreh	ole
			* Referenced to a	National Geodeti	ic Datum	
			CAS	SING MEA	SUREMENTS	
WELL CONS	STRUCTION MATERIALS		Diameter of Boreho	ole	(inches)	8.0
(Choose or	ne type of material for each area)		Protective Casing I	enoth	(inches)	5.0
			Riser Pipe Length	Juigui	(feet)	19.86
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Bottom of Screen to	o End Cap	(feet)	0.58
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	:	Screen Length (1	st slot to last slo	t) (feet)	4.53
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	:	Total Length of Ca	sing	(feet)	24.97
Screen	SS304 SS316 PTFE PVC OTHER		Screen Slot Size **		(inches)	0.010

^{**}Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Ag	ency		Well	Completion	Report
Site #: County:	Montgomery	ý	W	/ell #:G1	126
Site Name: <u>AEG Coffeen Power Station CCB Management F</u>	Facility		В	orehole #:	G126
State- Plant Plane Coordinate: X_2,513,895.4 Y_875,062.4 (or) La	atitude:	- ' "	Longitud	e:°	
Surveyed By: <u>Jeffrey D. Emrick</u>	IL Reg	istration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Layne-Western Co</u>	Driller:	T. List			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geolog	ist: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000)246
Drilling Method: <u>Hollow stem auger</u>	Drilling	g Fluid (Type): <u>n/a</u>	1		
Logged By: Diane M. Lamb	Date St	tarted: <u>2/10/20</u>	<u>10</u> Date	e Finished:2/2	10/2010
Report Form Completed By: Diane M. Lamb	Date: _	2/19/2010			
ANNULAR SPACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
		625.69	-2.73	Top of Protective	Casing
		625.39	-2.43	Top of Riser Pipe	2
Type of Surface Seal: <u>Concrete</u>		622.96	0.00	Ground Surface	
Type of Annular Sealant: <u>High-solids bentonite</u>		619.96	3.00	Top of Annular S	Sealant
Installation Method:					
Setting Time: <u>>24 hr.</u>	⊥ ⊻	614.14	8.82	Static Water Leve (After Completion)	el 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry					
Installation Method: <u>Granular</u>		613.96	9.00	Top of Seal	
Setting Time: <u>10 min</u>		612.86	10.10	Top of Sand Pack	X
Type of Sand Pack: <u>Quartz sand</u>		610.07	12.89	Top of Screen	
Grain Size: <u>10/20</u> (sieve size)				1	
Installation Method: <u>Gravity</u>		605.53	17.43	Bottom of Screen	I
Type of Backfill Material: <u>n/a</u> (if applicable)		604.96	18.00	Bottom of Well	
Installation Method:		604.96 * Referenced to a	<u>18.00</u> National Geodeti	Bottom of Boreho	ole
		CAS	ING MEAS	SUREMENTS	80
WELL CONSTRUCTION MATERIALS		ID of Riser Pipe	ne	(inches)	2.0
(choose one type of material tot cach area)		Protective Casing L	ength	(feet)	5.0
	1	Riser Pipe Length		(feet)	15.32
Protective Casing SS304 SS316 PTFE PVC (OTHER: Steel	Bottom of Screen to	End Cap	(feet)	0.57
Riser Pipe Adove W. I. SS304 SS316 PIPE PVC Riser Pine Below W.T. SS304 SS316 PTEE PVC 0	OTHER:	Screen Length (1s	st slot to last slo	t) (feet)	4.54
Screen SS304 SS316 PTFE PVC	OTHER:	Screen Slot Size **	sing	(feet)	0.010

Illinois Environmental Protection Agency	Well Compl	etion Report
Site #: County: _Mor	ntgomery Well #:	T127
Site Name: AEG Coffeen Power Station CCB Management Facility	Borehole #:	T127
State- Plant Plane Coordinate: X_2,513,911.0 Y_875,359.2 (or) Latitude:	°'" Longitude:	°
Surveyed By:	IL Registration #:	
Drilling Contractor: Layne-Western Co	Driller: <u>T. List</u>	
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W. Hasenyager, LPG #19</u>	96-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type): <u>n/a</u>	
Logged By: Diane M. Lamb	Date Started: <u>2/10/2010</u> Date Finished:	2/10/2010
Report Form Completed By:	Date:2/19/2010	
ANNULAR SPACE DETAILS	Elevations Depths (0. (MSL)* (BGS)	01 ft.)
	<u>631.29</u> 3.22 Top of Pro	otective Casing
		ser Pipe
Type of Surface Seal: <u>Concrete</u>	<u>628.07</u> <u>0.00</u> Ground St	urface
Type of Annular Sealant: <u>High-solids bentonite</u>	<u>622.43</u> <u>5.64</u> Top of Ar	nnular Sealant
Installation Method:		
Setting Time: _>24 hr.	∑ <u>616.81</u> <u>11.26</u> Static Wat (After Con	ter Level npletion) 3/1/2010
Type of Bentonite Seal Granular Pellet Slurry (choose one)		
Installation Method: <u>Granular</u>	<u>613.43</u> <u>14.64</u> Top of Se	al
Setting Time: <u>10 min</u>	<u>612.32</u> <u>15.75</u> Top of Sa	nd Pack
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)	<u>610.54</u> <u>17.53</u> Top of Sc	reen
Type of Backfill Material: n/a	$\frac{606.00}{605.43} = \frac{22.07}{22.64}$ Bottom of	Screen
(if applicable)	605.43 22.64 Bottom of	Borehole
	* Referenced to a National Geodetic Datum	
	CASING MEASUREME	NTS
	Diameter of Borehole	(inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe	(inches) 2.0
	Protective Casing Length	(feet) 5.0
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Riser Pipe Length	(feet) 20.42 (feet) 0.57
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot)	(feet) 0.57 (feet) 4.54
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing	(feet) 25.53
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **	(inches) 0.010

Illinois Environ	mental Protection Agency	r		Well	Completion	Report
Site #:	County: <u>Mo</u>	ntgomery		W	Vell #:	28
Site Name: AEG Coffeen Pow	er Station CCB Management Facility	7		В	orehole #:	128
State- Plant Plane Coordinate: X_2,513,909.	.5 Y 875,509.7 (or) Latitude:	0		Longitude	e:°	
Surveyed By:		IL Registra	ution #:			
Drilling Contractor: <u>Layne-Wes</u>	tern Co	Driller:	T. List			
Consulting Firm: <u>Hanson Profe</u>	ssional Services Inc.	Geologist:	Rhonald W.	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem a</u>	auger	Drilling Flu	uid (Type): <u>n/a</u>	1		
Logged By: <u>Suzanna L. Simps</u>	on	Date Starte	ed: <u>2/9/201</u>	10 Date	e Finished:2/	9/2010
Report Form Completed By: <u>Suz</u>	anna L. Simpson	Date:	2/18/2010			
ANNULAR SPAC	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			631.23	-2.79	Top of Protective	Casing
			630.93	-2.49	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>			628.44	0.00	Ground Surface	
Turne of Annular Scalant: High so	slide hortonite		623.38	5.06	Top of Annular S	ealant
Installation Mathada Transia						
Setting Time: >24 hr			616.09	12.35	Static Water Leve	1
50000 mile. <u></u>		<u>×</u>			(After Completion)	3/1/2010
Type of Bentonite Seal Granul	Iar Pellet Slurry (choose one) Image: Choose one one one one one one one one one on					
Installation Method: <u>Granula</u>	<u>r</u>	x x	615.53	12.91	Top of Seal	
Setting Time: <u>10 min</u>	————	×	614.13	14.31	Top of Sand Pack	
Type of Sand Pack: <u>Quartz sand</u>						
Grain Size: 10/20 (siew	e size)		611.91	16.53	Top of Screen	
Installation Method: <u>Gravity</u>			607.40	21.04	D-#	
Type of Backfill Material:Quartz	z sand	==	606.80	21.64	Bottom of Well	
Installation Method: Gravity	(if applicable)		604 38	24.06	Bottom of Boreho	le
instantation metrica. <u>Gravity</u>			* Referenced to a	National Geodeti	c Datum	inc.
			CAS	SING MEAS	SUREMENTS	
		Di	ameter of Boreho	ole	(inches)	8.0
WELL CONST (Choose one	TRUCTION MATERIALS type of material for each area)	ID	of Riser Pipe		(inches)	2.0
		Pr	otective Casing L	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTEE DVC OTHED.	Steel Ri	ser Pipe Length	F 10	(feet)	20.02
Riser Pipe Above W.T.	SS304 SS316 PTFE (PVC) OTHER:		ottom of Screen to	o End Cap	(feet)	0.60
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER:		tal Length of Cas	sing	(feet)	25.13
Screen	SS304 SS316 PTFE PVC OTHER:	Sc	creen Slot Size **		(inches)	0.010

Screen Slot Size ** **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agency		Well	Completion	Report
Site #: County:		We	ell#: TA	.31
Site Name: Coffeen Power Station		— Bor	rehole #:	TA31
State Plane Coordinate: X 2,513,856.8 Y 876,542.2 (or) Latitude:	<u>39° 4' 16.930"</u>	Longitude:	<u>89°</u> 24	4.920"
Surveyed By: <u>Gary C. Rogers</u>	IL Registration #:035-0	02957		
Drilling Contractor: Ramsey	Driller: <u>B. Williamson</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W.</u>	Hasenyager,	LPG #196-0002	246
Drilling Method: <u>Hollow Stem Auger</u>	Drilling Fluid (Type): <u>no</u>	ne		
Logged By:Rhonald W. Hasenyager	Date Started:10/28/20	014 Date	Finished: <u>10/</u>	28/2014
Report Form Completed By: <u>Rhonald W. Hasenyager</u>	Date:11/5/2014			
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
	626.90	3.01	Top of Protective	Casing
	626.55	-2.66	Top of Riser Pipe	-
Type of Surface Seal: Concrete	623.89	0.00	Ground Surface	
Type of Annular Sectorite	620.89	3.00	Top of Annular S	ealant
Installation Method: tremie				
Setting Time: _ 45 min	<u>615.78</u>	8.11	Static Water Leve (After Completion)	el 10/30/2014
Type of Bentonite Seal Granular Pellet Slurry			(
(choose one)	613.56	10.33	Top of Seal	
Setting Time:30 min.	610.81	12.00	T-n -f C-n d Dl	_
			Top of Sand Pack	L.
Type of Sand Pack: Quartz Sand	608.80	15.09	Top of Screen	
Grain Size: <u>10/20</u> (sieve size)			1	
	604.32	19.57	Bottom of Screen	
Type of Backfill Material:			Bottom of Well	
Installation Method:		20.19 National Geodetic	Bottom of Boreho	ble
	CAS Diameter of Boroks	SING MEAS	UREMENTS	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	ne	(inches)	2.0
(choose one type of material for each area)	Protective Casing L	ength	(feet)	5.0
	Riser Pipe Length		(feet)	17.82
Protective Casing SS304 SS316 PTFE PVC OTHER: S Ricer Pine Above W T SS304 SS316 PTFE (DVC) OTHER: S	Bottom of Screen to	o End Cap	(feet)	0.55
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length (1s	st slot to last slot)	(feet)	4.48
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **	5mig	(inches)	0.010

**Hand-Slotted Well Screens Are Unacceptable

Well Completion Form (revised 02/06/02)

Illinois Environmental Protection Agency		Well (Completion	Report
Site #: County:		Well	I#:TA	.32
Site Name: Coffeen Power Station		Bore	ehole #:	TA32
State Plane Coordinate: X_2,513,605.2 Y_877,532.6 (or) Latitude:	<u>39° 4' 26.730"</u>	Longitude:		<u>8.000"</u>
Surveyed By: <u>Gary C. Rogers</u>	IL Registration #:035-0	02957		
Drilling Contractor: <u>Ramsey</u>	Driller: <u>B. Williamson</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W.</u>	Hasenyager, I	LPG #196-000	246
Drilling Method: <u>Hollow Stem Auger</u>	Drilling Fluid (Type): <u>no</u>	ne		
Logged By:Rhonald W. Hasenyager	Date Started:10/27/20	014 Date F	inished: <u>10/</u>	27/2014
Report Form Completed By:	Date:11/5/2014			
ANNULAR SPACE DETAILS	Elevations	Depths	(0.01 ft.)	
	(iVISL) 621.76	(DG3) -2.83 т	op of Protective	Casing
	621.42	-2.49 T	op of Riser Pipe	
Type of Surface Seal:	<u>618.93</u>	<u>0.00</u>	Fround Surface	
Type of Annular Sealant: Bentonite	616.03	<u>2.90</u> T	op of Annular S	ealant
Installation Method: tremie				
Setting Time: <u>30 min.</u>	₫	<u>14.05</u> S	tatic Water Leve	
			(After Completion)	10/30/2014
Type of Bentonite Seal Granular Pellet Slurry (choose one)	Ť			
Installation Method: <u>Gravity</u>	611.09	<u>7.84</u> T	op of Seal	
Setting Time: <u>35 min.</u>	609.18	<u>9.75</u> T	op of Sand Pack	(
Type of Sand Pack:Quartz Sand				
Grain Size: 10/20 (sieve size)		<u>11.31</u> T	op of Screen	
Installation Method: <u>Gravity</u>		15 (0)		
Type of Backfill Material:	603.25	<u>15.68</u> E <u>16.47</u> E	Bottom of Screen Bottom of Well	
(if applicable)	602.46	16 <i>4</i> 7 p	Pottom of Porch	Na
	* Referenced to a	National Geodetic E	Datum	JIC .
	CAS	SING MEASU	IREMENTS	
	Diameter of Boreho	ole	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe		(inches)	2.0
	Protective Casing L	ength	(feet)	5.0
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Riser Pipe Length	End Con	(feet)	0.65
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1)	st slot to last slot)	(Ieet)	4.37
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Ca	sing	(feet)	18.96
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **		(inches)	0.010

**Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agency			Well	Completion	Report
Site #: County: _ Mon	tgomery		W	/ell #:TA	<u>A33</u>
Site Name: Coffeen Well Sealing & Assmt Well Install			В	orehole #:	A33b
State Plane Coordinate: X_2,513,248.7 Y_876,605.4 (or) Latitude:	39°4'	17.500"	Longitud	e: <u>89°</u> 24	<u>12.700"</u>
Surveyed By:Gary C. Rogers	IL Registration #:	035-002	2957		
Drilling Contractor: <u>Ramsey</u>	Driller: <u>B. Wil</u>	liamson			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhc</u>	onald W. H	asenyager	c, LPG #196-000	246
Drilling Method: Hollow Stem Auger	Drilling Fluid (Ty	pe): <u>Non</u>	e		
Logged By: <u>Suzanna L. Keim</u>	Date Started:	6/2/2015	5 Date	e Finished: <u>6</u> /	2/2015
Report Form Completed By:Suzanna L. Keim	Date:6/4	/2015			
ANNULAR SPACE DETAILS	Elev	v ations	Depths	(0.01 ft.)	
	62	<u>25.05</u>	-2.54	Top of Protective	Casing
	62	.5.27	-2.76	Top of Riser Pipe	2
Type of Surface Seal:		2.51	0.00	Ground Surface	
	62	.0.51	2.00	Top of Annular S	ealant
lype of Annular Sealant: <u>High-solids bentonite</u>					
Setting Time: >48 hrs	7 61	5 51	7.00	Static Water I ev	_]
			1.00	(After Completion)	6/16/2015
Type of Bentonite Seal Granular Pellet Slurry (choose one)	YTT I				
Installation Method: <u>Gravity</u>	61	4.51	8.00	Top of Seal	
Setting Time: <u>30 minutes</u>	61	2.11	10.40	Top of Sand Pacl	ζ.
Type of Sand Pack:Quartz sand			10.00		
Grain Size: <u>10-20</u> (sieve size)		0.28	12.23	Top of Screen	
Installation Method: <u>Gravity</u>		15 62	16 89	Bottom of Screen	
Type of Backfill Material: <u>n/a</u> (if applicable)		05.07	17.44	Bottom of Well	
Installation Method:	60	5.07	17.44	Bottom of Boreh	ole
	* Re	ferenced to a Na	ational Geodet	ic Datum	
		CASI	NG MEAS	SUREMENTS	
WELL CONSTRUCTION MATERIALS	Diameter	of Borehole	2	(inches)	7.5
(Choose one type of material for each area)	ID of Ris	er Pipe		(inches)	2.0
	Riser Pin	e Lenoth	ngtn	(feet)	<u> </u>
Protective Casing SS304 SS316 PTFE PVC OTHER: S	teel Bottom o	f Screen to 1	End Cap	(feet)	0.55
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Le	ength (1st s	slot to last slo	t) (feet)	4.66
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Ler	ngth of Casin	ng	(feet)	20.20
Screen SS304 SS316 PTFE PVC OTHER:	Screen SI	ot Size **		(inches)	0.010

Illinois Environmental Protection Agency		Well	Completion	Report
Site #: County: _ Mon	W	Vell #:TA	34	
Site Name: Coffeen Well Sealing & Assmt Well Install		В	orehole #:	ГА34
State Plane Coordinate: X_2,513,466.7 Y_875,906.1 (or) Latitude:	<u>39° 4' 10.500"</u>	Longitude	e: <u>89°</u> 24	<u>10.000"</u>
Surveyed By:Gary C. Rogers	IL Registration #:035-0	02957		
Drilling Contractor: Ramsey	Driller: <u>B. Williamson</u>			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W.</u>	Hasenyager	, LPG #196-000	246
Drilling Method: Hollow Stem Auger	Drilling Fluid (Type): <u>No</u>	one		
Logged By:Suzanna L. Keim	Date Started:6/3/201	5 Date	e Finished: <u>6/</u>	3/2015
Report Form Completed By: Suzanna L. Keim	Date:6/4/2015			
ANNULAR SPACE DETAILS	Elevations	Depths	(0.01 ft.)	
		2.67	Top of Protective	Casing
	626.52	-2.42	Top of Riser Pipe	
Type of Surface Seal: Concrete	<u>624.10</u>	0.00	Ground Surface	
	623.10	1.00	Top of Annular S	ealant
Type of Annular Sealant: Bentonite chips				
Installation Method: <u>Gravity</u>		0.10	C W I	,
Setting Time: <u>I hour</u>		8.10	(After Completion)	6/16/2015
Type of Bentonite Seal Granular Pellet Slurry				
Installation Method:	<u> </u>	n/a	Top of Seal	
Setting Time: <u>n/a</u>	<u>615.10</u>	9.00	Top of Sand Pack	Σ.
Type of Sand Pack:Quartz sand				
Grain Size: <u>10-20</u> (sieve size)	<u></u>	10.92	Top of Screen	
Installation Method:Gravity				
Type of Backfill Material:	$\equiv \underbrace{\begin{array}{c} 608.69 \\ 608.00 \end{array}}$	15.41 16.10	Bottom of Screen Bottom of Well	
(if applicable)		16.10		
Installation Method:	* Referenced to a	16.10 National Geodeti	Bottom of Boreho	ble
	CAS	ING MEAS	SURFMENTS	
	Diameter of Boreho	ole	(inches)	7.5
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe		(inches)	2.0
	Protective Casing L	ength	(feet)	5.0
Protective Casing SS304 SS316 PTEE PV/C OTHER R	Riser Pipe Length	E LO	(feet)	13.34
Riser Pipe Above W.T. SS304 SS316 PTFE (PVC) OTHER:	Bottom of Screen to	• End Cap	(feet)	0.69
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Cas	a siot to tast slo sing	(feet)	18.52
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size **		(inches)	0.010

APPENDIX A5

SOUTHWEST POND

F	EL	D]	BOR	I	NG	L C	DG		
I	CLIEN Sit Locatio Proje	T: A te: C on: C	EG Coffe CB Mana offeen, II	een P agem linois	'ower ient F s	Station Facility	1	CONTRACTOR: Testing Service Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 41/4" HSA w/SS samplers	BOREHOLE ID: G151 Well ID: G151 Surface Elev: 622 82 ft MSI
WE	DATE	S: Si Fin R: C	tart: 12/ nish: 12/ loudy, rai	, 19/2 19/2 in (m	011 011 nid-6	0's)		FIELD STAFF: Driller: B. Williamson Helper: R. McCuan Eng/Geo: R. Fiorito	Completion: 20.46 ft. BGS Station: 875,023.67N 2,513,805.93E
S	SAMPL	Æ	T	EST	TINC	3	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
L	' Total (in) wery		<i>6 in</i> ue	re (%)	n. (lb/ft ³)	$\begin{array}{c} Qp \ (tsf) \\ Type \end{array}$	Quadr Towns Section	rangle: Coffeen, IL hip: East Fork n 10, Tier 7N; Range 3W	$\Psi = 16.00 - \text{While drilling}$ $\Psi = 11.50 - \text{Upon completion}$ $\overline{\Psi} = 7.79 - 12/21/2011$
Numbe	Recov	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	21/24 88%	ss	2-3 4-5 N=7				2	FILL - Light yellowish brown (10YR6/4), silty CLAY some sand and gravel.	with622
2A	19/24 79%	ss	2-3 6-9 N=9				4	FILL - Brown (10YR5/3), silty CLAY with some sand slight trace gravel.	and620
3A	19/24 79%	ss	2-4 6-8 N=10					FILL - Brown (10YR5/3), silty CLAY with some sand slight trace gravel.	and 618
4A	19/24 79%	ss	15-15 18-11 N=33				⊻ 8	clayey SILT with trace sand and slight trace gravel.	616
5A	18/24 75%	ss	1-3 4-4 N=7				10	Brown (10YR5/3) with 30% gray (10YR6/1) mottles, CLAY with sand and gravel.	silty 614
6A	16/24 67%	ss	3-3 4-6 N=7				¥ 12-	Light gray (10YR5/1) with 40% yellowish brown (10YR5/6) mottles, sandy CLAY with some silt and sl	ght 612
7A	22/24 92%	ss	9-9 10-9 N=19				14	Light brown (10VD5/2) gilty SAND with slight trac	610
	10/24	М	15-15						
	19/24 79%	ss	35-70 N=50					Light gray (10YR6/1), silty SAND with slight trace gra	vel. 608
8A 9A	4/4 100%	🗙 ss	50/4"				I 16 _	Light brown (10YR6/4), silty SAND with some grav	
10A	10/23 43%	ss	12-18 70-99/5 N=88				18	Dark gray (10YR5/1), very silty CLAY with sand and t gravel. End of Boring = 20.5 ft BCS	race 606
		<u> </u>	1		1.	11		End of Boring - 20.5 it. DOS	
	71E(S):	G15	1 Install	ea in	DORE	mole.			

F	FIELD BORING LOG												
	CLIEN Si Locatio Proie	T: A te: C on: C	EG Coffe CB Mana offeen, II	een P agem linois	'ower ient F s	Station Facility	1	CONTRACTOR: Testing Service Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G152 Well ID: G152 Surface Eley: 623.06 ft MSL				
WE	DATES: Start: 12/20/2011 Finish: 12/20/2011 WEATHER: Cloudy, (mid-40's), rainy la							FIELD STAFF: Driller: B. Williamson Helper: R. McCuan Eng/Geo: R. Fiorito	Completion: 18.57 ft. BGS Station: 874,687.53N 2,513,894.46E				
	SAMPL	LE	Г	EST	FINC	3	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:				
	tal (ii <i>v</i>		u	(%)	(p/ft ³)	e (tsf	Quac Towi	rangle: Coffeen, IL ship: East Fork	$\Psi = 14.50$ - While drilling $\Psi = 3.05$ - Upon completion				
er	· / To		√6 ii alue	alue ture (en. ()	$^{(f)}_{\rm D}Q$	Secti	n 10, Tier 7N; Range 3W	$\overline{\Sigma} = 17.00 - 12/21/2011$				
qumN	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
		М	2.2					Dark brown (10YR4/2), silty CLAY with trace sand	and				
1A	12/24 50%	ss	2-2 2-3 N=4				2-	Light yellowish brown (10YR6/4), silty CLAY with sand and slight trace gravel.	trace				
2A 2B	22/24 92%	ss	2-4 9-7 N=13				⊻ _	Gray (10YR6/1), silty CLAY with trace sand and sl trace gravel.	ight 620				
3A	15/24 63%	ss	3-4 5-6 N=9					Gray (10YR6/1) and 25% brown (10YR5/6) mottles, CLAY with trace sand and slight trace gravel.	silty618				
3B 5A	19/24 79%	ss	10-10 12-10 N=22				6	Gray (10YR5/1), clayey SILT silty CLAY with sand slight trace gravel.	land				
5B	20/24 83%	ss	2-4 6-7 N=10				10-	Gray (10YR5/1) with 25% brown (10YR5/6) mottles CLAY with trace sand and slight trace gravel.	614				
6A	21/24 88%	ss	2-3 5-7 N=8				12	Gray (10YR6/1) with 25% yellowish brown (10YR mottles, silty CLAY with trace sand and slight trace gr	5/6) ravel.				
7A	20/24 83%	ss	5-7 9-10 N=16				14	Gray (10YR6/1) with 25% yellowish brown (10YR mottles, clayey SAND with some silt and slight trace g	5/6) gravel.				
,11		\square					⊥ ¹⁴	Yellowish brown (10YR5/6) with 20% gray (10YR6 mottles, clayey SAND with some silt and slight trace g	5/1),				
	21/24	ss	3-4 7-11				_	Yellowish brown (10YR5/6), silty SAND.					
8A		\mathbb{N}	N=11					Yellowish brown (10YR5/6), silty SAND with slight	trace				
8B	22/24	$\left \right $	14-35					Dark grayish brown (10YR4/2), very silty CLAY with	n sand				
9A	96%	ss s	60-70 N=95				I¥	Grayish brown (10YR5/2) silty CLAY with sand and gravel.	trace				
							=	End of Boring = 18.6 ft. BGS					
NC	End of Boring = 18.6 ft. BGS												
) I E(3)	. 013	∠ install	eu in	oore	noie.							

F	FIELD BORING LOG HANSON											
CLIENT: AEG Coffeen Power Static Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A							l	CONTRACTOR: Testing Service Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 41/4" HSA w/SS samplers	BOREHOLE ID: G153 Well ID: G153 Surface Elev: 623.30 ft. MSL			
DATES: Start: 12/15/2011 Finish: 12/15/2011 WEATHER: Cloudy, windy, (mid-high 3							0's)	FIELD STAFF: Driller: B. Williamson Helper: R. McCuan Eng/Geo: R. Fiorito	Completion: 20.76 ft. BGS Station: 874,532.71N 2,513,532.68E			
S	SAMPL	E	Т	EST	FINC		TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	otal (i) y		и	(%)	lb/ft ³)	<i>p</i> (tsf pe	Quadr Towns	angle: Coffeen, IL hip: East Fork	$\underline{\Psi} = 16.00$ - While drilling $\underline{\Psi} = 15.70$ - Upon completion			
ber	v / To cover		s/6i alue	ture ('	Den. ($f_{\rm T} O$	Section	10, Tier 7N; Range 3W	$\underline{\nabla} = 17.55 - 12/21/2011$			
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	16/24 67%	ss						FILL - Light yellowish brown (10YR6/4), silty CLAY sand and slight trace gravel.	with			
1B	0770	\wedge					2					
	17/24	\mathbb{N}	woh-3				2	FILL - Gray (10YR6/1), silty CLAY with sand and sl trace gravel.	ight			
	71%	ss	4-6 N=7					C C	620			
2A		$\left(\right)$					4	Grav (10VP6/1) with 50% brownich vellow (10VP6				
3A	18/24	ss	3-4 5-5					mottles, silty CLAY with trace sand.				
	/ 5/ 0	Λ	N=9						618			
	21/24	\mathbb{N}	4-5				0					
4A	21/24 88%	ss	6-8 N=11					Gray (10YR5/1), silty CLAY with trace sand.	616			
		$\left(\right)$					8					
	20/24	V ss	1-4 4-5									
5A	03%	N	N=8					Gray (10YR6/1), silty CLAY with sand and slight transvel.	ace 614			
		Π	14				10	Gray (10YR6/1), silty CLAY with sand and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slight transformed and slig				
	20/24 83%	ss	4-6 N=8					Gray (10YR6/1) with 50% brownish yellow (10YR6	612			
6A		Ĥ					12	motiles, sing CLAT with sand and sight trace grave				
	23/24	W _{ss}	7-6					Gray (10YR6/1) with 25% brownish yellow (10YR6	/8)			
7.	90%	N	N=12					mottles, clayey SAND with trace silt.	610			
/A		Π	15.22				14					
8A	19/24 79%	ss	37-50 N=60				_	Yellowish brown (10YR5/8), silty SAND with slight t	race 608			
	0/10	$\left(\right)$					¥ 16	gravel.				
9A	9/12 75%	X ss	50-99									
							⊻	Brownish yellow (10YR6/8), silty CLAY with sand a trace gravel.	and 606			
	20/24	\bigvee	23-50				18					
	20/24 83%	ss	58-109 N=108					Dark gray (10YR4/1), silty CLAY with sand and tra	ice 604			
10A	6/7	/ 	75-99/1	7			20-	gravel.				
11A	86%	Ш 33						End of Boring = 20.8 ft. BGS				
NO)TE(S):	G15	3 install	ed in	bore	hole.						

F	FIELD BORING LOG											
	CLIENT: AEG Coffeen Power Statio Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A							CONTRACTOR: Testing Service Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4 ¹ / ₄ " HSA w/SS samplers	BOREHOLE ID: G154 Well ID: G154 Surface Eley: 623 52 ft MSI			
DATES: Start: 12/16/2011 Finish: 12/16/2011 WEATHER: Ptly. cloudy (mid-30's))'s)		FIELD STAFF: Driller: B. Williamson Helper: R. McCuan Eng/Geo: R. Fiorito	Completion: 20:00 ft. BGS Station: 874,978.38N 2,513,243.10E			
	SAMPL	E	Т	EST		3	TOPOGI	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	y y		и	(%)	lb/ft ³	<i>p</i> (tsf pe	Quadi Towns	hip: East Fork	$\underline{\Psi} = 13.45$ - while drilling $\underline{\Psi} =$			
ber	v / Tc cover		s/6i alue	ture (Jen. ($f_{\rm re} T_{\rm y}$	Sectio	n 10, Tier 7N; Range 3W	$\underline{\nabla}$ = 11.10 - 12/21/2011			
Numl	Reco % Re	Type	Blow. N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
1A	14/24 58%	ss	3-4 4-5 N=8				2	FILL - Light yellowish brown (10YR6/4), silty CLAY trace sand and slight trace gravel.	with 622			
2A	10/24 42%	ss	woh-1 4-5 N=5					Brownish yellow (10YR6/6) with 25% gray (10YR6 mottles, silty CLAY with trace sand.	/1) 620			
3A	14/24 58%	ss	2-4 6-7 N=10					Gray (10YR5/1) with 25% yellowish brown (10YR5 mottles, silty CLAY with trace sand.	/4) 618			
4A	21/24 88%	ss	6-8 10-10 N=18				8		616			
5A	16/24 67%	ss	1-4 6-8 N=10				10	Gray (10YR6/1) with 25% brownish yellow (10YR6// mottles, silty CLAY with trace sand and slight trace grav	/6) avel. 614			
6A	13/24 54%	ss	<i>l-4</i> 6-8 N=10				⊻		612			
7A	20/24	W _{ss}	4-6					Gray (10YR6/1) with 10% brownish yellow (10YR6 mottles, silty CLAY with trace sand and slight trace gr	/8) avel			
7B	83%	N ³	N=14				⊻	Gray (10YR6/1), silty CLAY with trace sand and sli trace gravel.	ght 610			
8A		\mathbb{N}	1 10					Gray (10YR6/1), clayey SAND with trace silt.				
8B 8C	23/24 96%	ss	30-37 N=48					Gray (10YR6/1) with 50% yellowish brown (10YR5 mottles, silty SAND with slight trace gravel. Dark yellowish brown (10YR4/4), silty CLAY with s and trace gravel.	/6) and 			
9A	21/24 88%	ss	40-75 86-84 N=161					Brown (10YR4/3), silty CLAY with sand and trace gr	avel.			
10A	20/24 83%	ss	28-28 30-34 N=58					Dark gray (10YR4/1), silty CLAY with sand and tra gravel.	ice 604			
	ı 1		I	I	1	I	20 -	End of Boring = 20.0 ft. BGS				
NC	DTE(S):	G15	4 install	ed in	bore	hole.						

FI	FIELD BORING LOG											
CLIENT: AEG Coffeen Power Statio Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A							1	CONTRACTOR: Testing Service Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4¼" HSA w/SS samplers	BOREHOLE ID: G155 Well ID: G155 Surface Elev: 622.89 ft. MSL			
DATES: Start: 12/19/2011 Finish: 12/19/2011 WEATHER: Cloudy, rainy, (mid-40's)								FIELD STAFF: Driller: B. Williamson Helper: R. McCuan Eng/Geo: R. Fiorito	Completion: 20.23 ft. BGS Station: 875,127.65N 2,513,501.75E			
S	SAMPL	Е	Т	EST	TINC	Ĵ	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	al (in				/ft ³)	(tsf)	Quadra	ingle: Coffeen, IL	$\mathbf{Y} = 14.30$ - While drilling			
	Tota very		6 in ue	e (%	J. (lp	Q_p^{Γ}	Section	10, Tier 7N; Range 3W	$\Sigma = 7.90 - 12/21/2011$			
Jumber	ecov / 6 Reco	ype	lows/ I - Val tQD	Ioistur	Iry Dei	bu (tsf) ailure	Depth	Lithologic	Borehole Elevation			
Z	8%	$\overline{\mathbf{T}}$	a z B	2		0 ^H		FILL - Yellow (10YR7/6), silty CLAY with trace sand				
1.4	20/24	M	4-6					<u>slight trace gravel</u> FILL - Brown (10YR5/3), silty CLAY with sand and t	race 622			
IA	83%	Å ss	8-6 N=14									
		Δ						slight trace gravel.				
2A		M						FILL - Brownish yellow (10YR6/6), silty CLAY with t sand and slight trace gravel	trace			
	20/24	ss	2-4 6-7						620			
		Λ	N=10					FILL - Dark grayish brown (10YR4/2), silty CLAY w	vith			
		H					4-	Gray (10YR5/1), silty CLAY with trace sand and slig trace gravel.	ght			
	17/24	$\mathbb{N}_{\mathbb{R}}$	2-4						618			
	71%	1 55	5-8 N=9					Brownish vellow (10VP6/6) with 10% grav (10VP6				
3A		Д					6-	mottles, silty CLAY with trace sand and slight trace gra	avel.			
		M	67									
	24/24 100%	ss	11-11						616			
4A		Λ	N=18									
		Π					* 8-	Dark grav (10YR4/1) with 10% brown (10YR4/3) mot	ttles			
5A	22/24	\mathbb{V}_{ss}	woh-3					silty CLAY with trace sand and slight trace gravel.	614			
	92%	Λ^{33}	N=9									
		4					10					
6A		M	2_4					Dark yellowish brown (10YR4/6) with 30% grayish br (10YR5/2) mottles, silty CLAY with trace sand and sli	own ight			
	17/24	ss	6-6					trace gravel.	-612			
		Λ	N-10									
		\square					12	Grav (10VR6/1) with 50% vellowish brown (10VR5	16 1			
7A	24/24	V ss	6-8				Ā 🗍	mottles, clayey SAND with trace silt.	-610			
	100%	Λ	N=18									
		(▼ 14	Vellowish brown (10VR6/4) clayey SAND with trace				
8A	22/24	M	woh-wo	h				Tenowish brown (101100-1), edgeg 5514D with date				
	96%	ss	4-15					Yellowish brown (10YR5/6), silty SAND with slight to	race 608			
8B		$\langle \rangle$					16	Etaroi.				
02	16/16	M	15-50					Yellowish brown (10YR5/6), silty SAND with slight to				
	100%	Å ss	99/4"					gravel.	606			
9A								trace gravel				
		\Box					18-					
10.4	15/17	X ss	24-68					Light gray (10YR6/2), silty CLAY with sand and tra				
10A	0070	\square	19/5					gravel.	004			
							20					
	End of Boring = 20.2 ft. BGS											
NO	TEAD	C17	E 1000 11	. J ·	1	h . 1 .						
NO	71E(S):	615	5 install	ed in	bore	enole.						

Illinois Environmental Protection Agency		Well Completion Report
Site #: County:	ontgomery	Well #: G151
Site Name: CCB Management Facility		Borehole #: G151
State Plane Coordinate: X 2,513,805.9 Y 875,023.7 (or) Latitude	2:	Longitude:
Surveyed By: Jeffrey D Emrick	IL Registration #: 035-00	3507
Drilling Contractor: Testing Service Corp	Driller: B Williamson	
Consulting Firm: Hanson Professional Services Inc	Geologist: Rhonald W H	asenvager I PG #196_000246
	Drilling Eheid (Terra) n/2	
Honow stem auger		
Logged By: <u>Ryne M. Fiorito</u>	_ Date Started:12/19/201	1 Date Finished: <u>12/19/2011</u>
Report Form Completed By: Rhonald W. Hasenyager	Date:12/27/2011	
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (0.01 ft.) (BGS)
	626.24	<u>-3.42</u> Top of Protective Casing
T [<u>625.93</u>	-3.11 Top of Riser Pipe
Type of Surface Seal:	622.82	0.00 Ground Surface
	620.49	2.33 Top of Annular Sealant
Type of Annular Sealant: <u>High-solids bentonite</u>		I
Installation Method:		
Setting Time: <u>>24 hr.</u>	$\underline{\nabla}$ <u>615.03</u>	7.79 Static Water Level (After Completion) 12/21/2011
Type of Bentonite Seal Granular Pellet Slurry		
Installation Method: <u>Gravity</u>	611.82	
Setting Time: <u>24 min</u>	600.07	12.75 Tax of Cand Dada
\sim		<u>13.75</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u>	607.48	15.24 Top of Sereen
Grain Size: <u>10/20</u> (sieve size)		<u>15.54</u> Top of Screen
Installation Method: <u>Gravity</u>	602.98	19.84 Bottom of Screen
Type of Backfill Material: <u>n/a</u>	602.36	20.46Bottom of Well
Installation Method: n/a	602.36	20.46 Bottom of Borehole
	* Referenced to a N	ational Geodetic Datum
	CASI	NG MEASUREMENTS
	Diameter of Borehol	e (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	(inches) 2.0
,	Protective Casing Le	ngth (feet) 5.0
	Riser Pipe Length	(feet) 18.45
Protective Casing SS304 SS316 PTFE PVC OTHER	Bottom of Screen to	End Cap (feet) 0.62
Kiser Pipe Above W. I. SS304 SS316 PIFE PVC OTHER Piper Pipe Above W. I. SS304 SS316 PIFE PVC OTHER	Screen Length (1st	slot to last slot) (feet) 4.50
Screen SS304 SS316 PTFE (PVC) OTHER	Total Length of Casi	ng (feet) 23.57
	Screen Slot Size **	(inches) 0.010

Illinois Environ	mental Protection Agency	7		Wel	Completion	n Report
Site #:	County: <u>N</u>	Iontgomery		V	Vell #: G	152
Site Name: <u>CCB Management</u>	t Facility			B	orehole #:	G152
State Plane Coordinate: X 2,513,894	1.5 Y <u>874,687.5</u> (or) Latitud	le:		Longitud	le:	
Surveyed By:Jeffrey D. Emric	k	IL Regis	tration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Testing Se</u>	rvice Corp.	_ Driller:	B. Williamson			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.	Geologis	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	Fluid (Type): <u>n/a</u>	a		
Logged By: <u>Ryne M. Fiorito</u>		Date Sta	rted: <u>12/20/20</u>	011 Dat	e Finished: <u>12</u>	/20/2011
Report Form Completed By:R	ionald W. Hasenyager	_ Date: _	12/27/2011			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
			626.67	3.61	Top of Protective	e Casing
			626.52	3.46	Top of Riser Pip	e
Type of Surface Seal: <u>Concrete</u>			623.06	0.00	Ground Surface	
Type of Annular Seclant: Seclar	standad to pear surface		621.06	2.00	Top of Annular S	Sealant
Installation Mathedu	Rended to hear surface					
Setting Time:		⊻	_606.06_	17.00	Static Water Lev	el
Type of Pentonite Seel	ular Dollat Slurry				(Alter Completion)	12/21/2011
Type of Bentonne Seal (Gran	(choose one)					
Installation Method: <u>Gravit</u>	<u>y</u>		621.06		Top of Seal	
Setting Time: <u>>24 hr.</u>	X	X	611.56	11.50	Top of Sand Pac	k
Type of Sand Pack: <u>Quartz sand</u>	1					
Grain Size: 10/20 (sie	ve size)		609.47	13.59	Top of Screen	
Installation Method:Gravit	У		(0 1 0 7	10.00		
Type of Backfill Material:n/a			<u>604.97</u> <u>604.49</u>	<u>18.09</u> <u>18.57</u>	Bottom of Screen Bottom of Well	1
	(if applicable)		<u> </u>	10.55		_
Installation Method: <u>n/a</u>			604.49 * Referenced to a	18.57 National Geode	Bottom of Boreh	ole
			CAS	NG MEA	SUREMENTS	
		Γ	Diameter of Boreho	ole	(inches)	8.0
WELL CONS (Choose on	TRUCTION MATERIALS e type of material for each area)	-	ID of Riser Pipe		(inches)	2.0
	/		Protective Casing L	ength	(feet)	5.0
			Riser Pipe Length		(feet)	17.05
Protective Casing Riser Pine Above W.T.	SS304 SS316 PIFE PVC OTHE		Bottom of Screen to	o End Cap	(feet)	0.48
Riser Pipe Below W.T.	SS304 SS316 PTFE (PVC) OTHE	R:	Screen Length (1s	st slot to last slo	(feet)	4.50
Screen	SS304 SS316 PTFE PVC OTHE	R:	Screen Slot Size **	SILIS	(inches)	0.010

Illinois Environmental Protection	Agency				Wel	l Comple	tion Rej	port
Site #: C	ounty: <u>Mont</u>	gomery	1		V	Vell #:	G153	
Site Name: <u>CCB Management Facility</u>					E	orehole #:	G153	
State Plane Coordinate: X_2,513,532.7 Y_874,532.7 (o	or) Latitude: _				Longitud	le:		
Surveyed By:Jeffrey D. Emrick		IL Regi	istrati	on #: <u>035-0</u>	03507			
Drilling Contractor: <u>Testing Service Corp.</u>		Driller:	В	. Williamson				
Consulting Firm: <u>Hanson Professional Services Inc.</u>		Geologi	ist: _	Rhonald W.	Hasenyage	r, LPG #196	-000246	
Drilling Method: <u>Hollow stem auger</u>		Drilling	g Flui	d (Type): <u>n/a</u>	a			
Logged By: <u>Ryne M. Fiorito</u>		Date St	tarted	:12/15/20	011 Dat	e Finished: _	12/15/20	11
Report Form Completed By: <u>Rhonald W. Hasenyager</u>		Date: _		12/27/2011				
ANNULAR SPACE DETAILS				Elevations (MSL)*	Depths (BGS)	(0.0	1 ft.)	
				626.59	3.29	Top of Prot	ective Casing	g
				626.35	3.05	Top of Rise	r Pipe	
Type of Surface Seal: Concrete			>	623.30	0.00	Ground Sur	face	
Type of Annular Sealant: High-solids bentonite				622.63	0.67	Top of Ann	ular Sealant	
Installation Method: Tremie								
Setting Time:24 hr.		-		605.75	17.55	Static Wate (After Compl	r Level etion) 12/21/20	011
Type of Bentonite Seal Granular Pellet Slurry (choose one)		ŤT						
Installation Method: <u>Gravity</u>	- xx	x x		610.30	13.00	Top of Seal		
Setting Time: <u>>24 hr.</u>	- 🕅	×		608.30	15.00	Top of Sand	d Pack	
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)				607.40		Top of Scre	en	
Installation Method: <u>Gravity</u>				602.96	_20.34	Bottom of S	Screen	
Type of Backfill Material: <u>n/a</u> (if applicable)	_ L			_602.54_	20.76	Bottom of V	Well	
Installation Method: <u>n/a</u>				602.54 * Referenced to a	20.76 National Geode	Bottom of H tic Datum	Borehole	
							TO	
			Dia	CAS	ole	SUREMEN	1S	0
WELL CONSTRUCTION MATERIAL (Choose one type of material for each area)	S		ID c	of Riser Pipe		(ii	nches) 2.	0
			Prot	ective Casing L	ength		(feet) 5.	0
Protective Casing SS204 SS214 DTEE D]	Rise	er Pipe Length			(feet) 18.	95
Flotective Casing 55304 55310 PTFE P Riser Pine Above W T \$\$3304 \$\$316 PTFE P	VC OTHER		Bott	tom of Screen to	o End Cap		(feet) 0.	42
Riser Pipe Below W.T. SS304 SS316 PTFE	VC OTHER:		Tot	en Length (ls	st slot to last slo	ot)	(feet) 22	81
Screen SS304 SS316 PTFE P	VC OTHER:		Scre	en Slot Size **	5115	(ii	$\frac{1}{23}$ nches) 0.	.010

Illinois Environmental Protection Agency			Well	Completion	Report
Site #: County:	ntgomery		W	Vell #: G1	.54
Site Name: CCB Management Facility			В	orehole #:	G154
State Plane Coordinate: X 2,513,243.1 Y 874,978.4 (or) Latitude:			Longitud	e:	
Surveyed By:Jeffrey D. Emrick	IL Registrati	ion #: <u>035-00</u>	03507		
Drilling Contractor:Testing Service Corp	Driller: <u>B</u>	. Williamson			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: _	Rhonald W.	Hasenyager	r, LPG #196-0002	246
Drilling Method: Hollow stem auger	Drilling Flui	d (Type): <u>n/a</u>	l		
Logged By: <u>Ryne M. Fiorito</u>	Date Started	:12/16/20)11 Date	e Finished: <u>12/</u>	16/2011
Report Form Completed By: <u>Rhonald W. Hasenyager</u>	Date:	12/27/2011			
ANNULAR SPACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
		626.55	-3.03	Top of Protective	Casing
		626.35	-2.83	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>		623.52	0.00	Ground Surface	
Type of Annular Sealant: High solids bentonite		_622.44	1.08	Top of Annular S	ealant
Installation Method: Tremie					
Setting Time: _>24 hr.	☑	612.42	11.10	Static Water Leve (After Completion)	el 12/21/2011
Type of Bentonite Seal Granular Pellet Slurry (choose one)					
Installation Method: <u>Gravity</u>	××	613.02	10.50	Top of Seal	
Setting Time: 24 min	×	_610.02_	13.50	Top of Sand Pack	1
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)		609.26	14.26	Top of Screen	
Installation Method: Gravity		604.76	18.76	Bottom of Screen	
Type of Backfill Material:		_604.42_	19.10	Bottom of Well	
Installation Method:		603.52 * Referenced to a		Bottom of Boreho	ble
		CAS		SUDEMENITS	
	Dia	meter of Boreho	le	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID o	of Riser Pipe		(inches)	2.0
	Prot	tective Casing L	ength	(feet)	5.0
Destastiva Casing SC204 SC214 DEEL DVC SCEUD	Rise	er Pipe Length		(feet)	17.09
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Bot	tom of Screen to	End Cap	(feet)	0.34
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:		al Length of Cas	t siot to last slo	(feet)	21.93
Screen SS304 SS316 PTFE PVC OTHER:	Scre	en Slot Size **	- D	(inches)	0.010

Illinois Environn	nental Protection Agency	y		Wel	Completion	Report
Site #:	County:!	Montgomery		W	Vell #: G1	55
Site Name: <u>CCB Management F</u>	Facility			В	orehole #:	G155
State Plane Coordinate: X 2,513,501.8	<u>8</u> Y <u>875,127.7</u> (or) Latitu	ıde:		Longitud	e:	
Surveyed By:Jeffrey D. Emrick		IL Registr	ration #: <u>035-0</u>	03507		
Drilling Contractor: <u>Testing Serv</u>	ice Corp.	Driller:	B. Williamson			
Consulting Firm: <u>Hanson Profess</u>	sional Services Inc.	Geologist:	Rhonald W.	Hasenyage	r, LPG #196-0002	246
Drilling Method: <u>Hollow stem au</u>	ıger	Drilling F	luid (Type): <u>n/a</u>	a		
Logged By: <u>Ryne M. Fiorito</u>		Date Start	ted: <u>12/19/20</u>	011 Dat	e Finished: <u>12/</u>	19/2011
Report Form Completed By: <u>Rhor</u>	nald W. Hasenyager	Date:	12/27/2011			
ANNULAR SPAC	E DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			626.07	3.18_	Top of Protective	Casing
			625.86	-2.97	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>			622.89	0.00	Ground Surface	
Type of Annular Sealant: High col	ids bentonite		621.39	1.50	Top of Annular S	ealant
Installation Method: Tremie						
Setting Time: _>24 hr.		Σ	614.99	7.90	Static Water Leve (After Completion)	el 12/21/2011
Type of Bentonite Seal Granula	r Pellet Slurry (choose one)					
Installation Method: <u>Gravity</u>	×	* **	611.89	11.00	Top of Seal	
Setting Time: <u>27 min</u>			609.14	13.75	Top of Sand Pack	
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve s	size)		607.80	15.09	Top of Screen	
Installation Method: <u>Gravity</u>			603.31	19.58	Bottom of Screen	
Type of Backfill Material: <u>n/a</u>	(if applicable)		602.66	20.23	Bottom of Well	
Installation Method: <u>n/a</u>			602.66 * Referenced to a		Bottom of Boreho	ble
			CAS		SUDEMENTS	
		Г	Diameter of Boreho	ole	(inches)	8.0
WELL CONST (Choose one ty	RUCTION MATERIALS (pe of material for each area)	п	D of Riser Pipe		(inches)	2.0
		P	rotective Casing L	ength	(feet)	5.0
Protective Casing	SS304 SS316 DTEE DVC OTTO		Liser Pipe Length		(feet)	18.28
Riser Pipe Above W.T	SS304 SS316 PTFE PVC OTH		Bottom of Screen to	o End Cap	(feet)	0.43
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTH	ER: T	otal Length of Ca	sing	(feet)	23 20
Screen	SS304 SS316 PTFE PVC OTH	ER:	creen Slot Size **		(inches)	0.010

APPENDIX A6

ASH POND 1

F	CLIENT: Natural Resource Technology, Inc. CONTRACTOR: Ramsey Geotechnical Engineering, LLC Site: Coffeen Energy Center Rig mfg/model: D-50 Turbo Tracked MST 800ATV Location: Coffeen, Illinois Drilling Method: Hollow Stem Auger (3¼"overdrill/4¼") Project: 15E0030 Surface Elev: 623.82 ft. MSL													
WE	DATE	S: St Fir R: St	tart: 9/8 nish: 9/8 nny, hi 7	/201 /201 /0's	5 5			FIELD STAFF: Driller: D. Crump Completion: 20.29 ft. BGS Helper: D. Groves Station: 2,514,455.48N Eng/Geo: K. Theesfeld 874,375.37E				a: 20.29 ft. BGS a: 2,514,455.48N 874,375.37E		
S	SAMPL	E	Т	EST	TINC	j o	TOPOGRAPHIC MAP INFORMATION: WATER LI				L INFORMA	TION:		
	tal (ir		ı	()	b/ft ³)	o (tsf) e	Quadra Townsł	ngle: Coffeen, IL ip: East Fork	L L	L = 14.00 - L = 14.00	During Drilli	ng		
er	Tol /		/6 <i>ii</i> ılue	ure (⁰	en. (l	$f_{\rm Typ}^{\rm (f)}$	Section	10, Tier 7N; Range 3W	Ī	$\overline{\underline{\nabla}}$ =				
qunN	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		Borehole Detail	Elevation ft. MSL	Remarks		
1A	17/24	M_{co}	15-10	14				Light gray (10YR7/2), dry, very stiff, SILT with little of and trace gravel.	clay					
111	71%		/-0 N=17				2	Yellowish brown (10YR5/4) with 5% dark brown (10YR3/3) mottles, dry, very stiff, SILT with few clay	and		622			
	19/24	Mss	2-4					Yellowish brown (10YR5/4) with 15% dark yellowish brown (10YR4/6) and 5% dark brown (10YR3/3) mot	sh tles,					
2A	79%	\wedge	N=9	25		1.50		moist, stiff, SILT with few clay.			620			
	22/24	M	2-2					Yellowish brown (10YR5/4) with 15% dark yellowish brown (10YR4/6) mottles moist medium CLAY wi	sh ith					
3A	92%	\bigwedge	N=5	23		0.40		some silt and trace fine-grained sand and small grave	el.		618			
	24/24	\overline{M}	5-5				6	Vellowish brown (10VR5/4) with 15% dark vellowi	- — — —					
4A	100%	Å ss	6-6 N=11	19		1.20		brown (10YR4/6) and 5% dark brown (10YR3/3) mot moist, stiff, CLAY with some silt and trace fine-grain	tles, ed					
		Ħ					8	sand and small gravel.			616			
5A	20/24 83%	ss	3-4 N=5	21		1.40		Yellowish brown (10YR5/4) with 30% dark yellowis brown (10YR4/6) and 5% dark brown (10YR3/3) mot moist, stiff, SILT with some clay and trace very fine-	sh tles, to					
		$\left(\right)$					10	fine-grained sand and small gravel			614			
6A	22/24 92%	ss	2-2 3-3 N=5	2 3 5 18		0.50		Dark yellowish brown (01YR4/6) with 30% yellowis brown (10YR5/4) mottles, moist, soft, SILT with few of	sh clay					
011		$\left(\right)$		10		0.00	12	and little fine- to coarse-grained sand and small gravel, t wood fragments.	trace		612			
7A	17/24 71%	ss	3-4 5-5	19		0.30		Dark yellowish brown (01YR4/6) with 15% yellowis	- <u>— —</u> — sh		+ <u>-</u> -			
		A	11-9				⊻ 14	brown (10YR5/4) mottles, moist, soft, SILT with few c and very fine- to fine-grained sand and trace small grav	clay vel.		610			
	19/24 79%	ss	3-11 21-28					Dark yellowish brown (10YR4/4), wet, dense, very fine- fine-grained SAND with some silt, few clay and trace si	e- to mall					
	,,,,,	Д	N=32				16	gravel.		-	608			
	24/24 100%	ss	21-36 39-50 N=75					Dark yellowish brown (10YR4/4), wet, dense, very fine fine-grained SAND with few silt, little clay and trace sr gravel.	e- to mall					
8A		$\left(\right)$	1 75	7		4.50	18	Yellowish brown (10YR5/6) with 5% strong brown (7.5YR5/6) mottles, moist, hard, SILT with few clay a little fine-grained sand and small gravel.	n and		606			
	$\left \begin{array}{c}11/24\\46\%\end{array}\right \left \begin{array}{c}16.9\\88\\N=39\end{array}\right \left \begin{array}{c}7\\7\\N=39\end{array}\right \left \begin{array}{c}7.56\\7\\10\\N=39\end{array}\right \left \left \begin{array}{c}7.56\\7\\10\\N=39\end{array}\right \left \left \begin{array}{c}7.56\\7\\10\\N=39\end{array}\right \left \left \begin{array}{c}7.56\\7\\10\\N=39\end{array}\right \left \left \begin{array}{c}7.56\\7\\10\\N=39\end{array}\right \left \left \left \begin{array}{c}7.56\\7\\10\\N=39\end{array}\right \left \left \left \left \left \left \left \left \left \left \left \left \left \left \left \left \left \left \left$							Dark grayish brown (10YR4/2) with 5% strong brow (7.5YR5/6) mottles, moist, hard, SILT with few clay a little fine-grained sand and small gravel.	vn and					
$\begin{bmatrix} 0.73 \\ 0\% \end{bmatrix} BD \begin{bmatrix} 0.74 \\ 0\% \end{bmatrix} = \begin{bmatrix} 20 \\ 0\% \end{bmatrix}$ End of boring = 20 29 feet														
	End of boring = 20.29 leet													

ſ
F	FIELD BORING LOGG CLIENT: Natural Resource Technology, Inc. Site: Coffeen Energy Center Location: Coffeen, Illinois Project: 15E0030 DATES: Start: 9/4/2015 Finish: 9/4/2015 FIELD STAFF: Driller: D. Crump Helper: D. Groves Station: 2,515,582.97N												
WE	ATHEF	R: Su	inny, hi 7	/201 /0's	5		Eng/Geo: K. Theesfeld	872,234.82E					
	SAMPL	E	T	EST		G G	TOPOGRAPHIC MAP INFORMATION: WAT	FER LEVEL INFORMATION: $\mathbf{V} = 12.00$ - During Drilling					
er -	/ / Total (covery		<i>i / 6 in</i> alue	ure (%)	en. (lb/ft	if) Qp (ts e Type	Township: East ForkImage: Section 11, Tier 7N; Range 3W	$\mathbf{Y} = \mathbf{V}$					
Numb	Recov % Re	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth Lithologic ft. BGS Description	Borehole Elevation Detail ft. MSL Remarks					
1A	19/24 79%	ss	4-4 3-5 N=7	18		1.80	Brown (10YR4/3), moist, stiff, SILT with few clay, trace organics.						
2A	22/24 92%	ss	2-4 4-6 N=8	26		2.40	2mottles, moist, stiff, SILT with few clay, trace organics Brown (10YR5/3) with 30% yellowish brown (10YR5/6) mottles, moist, stiff, SILT with few clay, trace very fine- to medium-grained sand.	618					
3A	21/24 88%	ss	2-2 3-4 N=5	22		1.30	Brown (10YR5/3) with 15% yellowish brown (10YR5/6) mottles, moist, stiff, CLAY with some silt and trace very fine- to medium-grained sand.						
4A	24/24 100%	ss	6-4 6-6 N=10	19		1.10	Brown (10YR5/3) with 10% yellowish brown (10YR5/6) mottles, moist, medium to stiff, CLAY with some silt and little very fine- to coarse-grained sand and small gravel.						
5A	21/24 88%	ss	<i>1-2</i> <i>3-4</i> N=5	21			Brown (10YR5/3) with 5% yellowish brown (10YR5/6) mottles, moist, stiff, SILT with some clay and little very fine- to coarse-grained sand and small gravel.						
6A	24/24 100%	ss	3-2 3-3 N=5	19		0.80	10 - Brown (10YR5/3) with 20% yellowish brown (10YR5/6) mottles, moist, stiff, SILT with some clay and little very fine- to coarse-grained sand and small gravel. ▼ 12						
7A	24/24 100%	ss	2-4 6-21 N=10	13		1.60	Yellowish brown (10YR5/6) with 10% grayish brown (10YR5/2) and 5% yellowish brown (10YR5/4) mottles, wet, stiff, SILT with few clay and little fine- to coarse-grained sand and small gravel.						
8A	21/24 88%	SS BD	20-27 50 N=77	7		4.50	Grayish brown (10YR5/2) with 5% brown (10YR5/3) mottles, dry, hard, SILT with few clay, very fine- to coarse-grained sand and small gravel.						
	0%		-		-		End of boring = 16.21 feet						
1													

WE	Sit Location Projec DATES ATHEF	e: Co n: Co t: 15 S: St Fin R: Su	offeen En offeen, Ill 5E0030 cart: 9/3/ hish: 9/4, hinny, hi 7	ergy inois /201: /201 0's	Cent ; 5 5	er		Rig mfg/model: D-50 Turbo Tracked MST & Drilling Method: Hollow Stem Auger (3¼"ove FIELD STAFF: Driller: D. Crump Helper: D. Groves Eng/Geo: K. Theesfeld	800. verdi	ATV ill / 4	4 ¹ ⁄4")	B	OR S	REHOLE II Well II Surface Elev Completion Station	 b: G302 b: G302 c: 617.95 ft. MSL i: 18.39 ft. BGS i: 2,516,214.19N 872,252.95E 		
S	AMPL	E	Т	EST		L L	- TOPOGRAPHIC MAP INFORMATION: WATER LI					LE [*]	LEVEL INFORMATION:					
	otal (i <i>ry</i>		in	(%)	(lb/ff ³	<i>Jp</i> (ts. /pe	Townsl	hip: East Fork		$\underline{\mathbf{Y}} = 14.00$ - During Drining $\underline{\mathbf{Y}} = \mathbf{\overline{Y}}$					lig			
lber	T / VC	n	vs / 6 Value D	sture (Den.	tsf) (ure Ty	Section	11, Tier 7N; Range 3W	<u> </u>									
Nun	Reco % R	Typ	Blov N - J	Moi	Dry	Qu (Fail	ft. BGS	Description			E	Bore Det	hol	le	ft. MSL	Remarks		
1A	23/24 96%	ss	5-5 5-6 N=10	17		1.50		Yellowish brown (10YR5/6) to dark yellowish brown (10YR4/6), moist, hard, SILT with some clay and trac- fine-grained sand and small gravel.	n ce									
2A	20/24	ss	2-3 4-4	27		2.60	2	Very dark brown (10YR3/1), moist, hard, SILT with sor clay and trace fine-grained sand and small gravel.	ome			C 2 2 2 K	2 2 5 CN		616			
	83%	Λ	N=7				4	Dark grayish brown (10YR4/2) with very dark gray (10YR3/1) mottles, moist, stiff, CLAY with some silt ar trace sand.	and						614			
3A	21/24 88%	ss	1-2 3-4 N=5	26		1.80	6	Brown (10YR5/3) with dark yellowish brown (10YR4/ mottles, moist, stiff, CLAY with some silt and trace san	1/6) nd.						612			
4A	24/24 100%	ss	4-7 8-8 N=15	18		1.60		Brown (10YR5/3) with dark vellowish brown (10YR4/	/6)									
5A	24/24 100%	ss	2-3 5-5 N=8	17		1.80	8	mottles, moist, stiff, CLAY with some silt and few very fine- to fine-grained sand.	rý						610			
64	18/24 75%	ss	2-2 4-5 N=6	19				Brown (10YR4/3) with dark yellowish brown (10YR4/ mottles, moist, stiff, SILT with some clay and few sand	/6) id.									
UA	24/24		7-7	17			12	Dark gray (10YR4/1), moist, stiff, SILT with some cla and few sand. Yellowish brown (10YR5/8) with reddish brown (5YR4 inclusions, moist, stiff, SILT with some clay and few sar	ay 4/4) and.						606 			
7A	100%		N=15	16		1.70	¥ 14-	Dark gray (10YR4/1), moist, stiff, SILT with some clay few sand and trace small gravel.	ay,						604			
8A 8B	22/24 92%	ss	2-5 25-32 N=30	18 12		0.80 4.50		Grayish brown (10YR5/2), wet, hard, SILT with little cl and very fine-grained sand and trace small gravel.	clay									
	24/24		7-24				16	Brown (10YR5/3), wet, hard, SILT with little clay and v fine-grained sand and trace small gravel.	very	,					602			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $						4.50	18	Brown (10YR5/3) grading to yellowish brown (10YR5/4), moist, hard, SILT with some clay, few small gravel.							600			
	0%		I	I	I	i I		End of boring = 18.39 feet					<u>fiii</u>		J			

	Surfa	ace Elevation: <u>619.10</u> Datum <u>msl</u>	Completion Date: No	Impletion Date: 8/26/10 Impletion Date: 0 Northing: 871382.45 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 0 1 0 0 1 0 0 1 0 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <				N	WELL DIAGRAM				
	DEPTH IN FEET	DESCR	IPTION OF MA	TERIAL	GRAPHIC	DRY UNIT WE SPT BLOW CORE RECOV	SAMP	Diam	Stickup eter: 6 in	ches	-2.6	E. C.	
		FILL: brown, slity c	ay	·		1-2-3	SS1			Concrete	1.0	618.1	
	- 5-	Medium stiff, grayis	h-brown, silty CLAY w	ith lignite - CL		1-3-3	SS2	2" sch 40 PVC		Bentonite			
s						1-3-4	SS3		1705 1501	-	ð .0	611.1	
N SOIL TYPE SES ONLY.	- 10-	Soft, brown, silty Cl Stiff to hard, brown	AY - CL to grayish-brown, silty	CLAY - CL (TILL)		2-3-3	SS4	-			10.0	609,1	
INDARIES BETWEE						27-44 -50/1"	SS5	2" sch 40 PVC 0,10 slotted		Filter sand			
THE APPROXIMATE B GRAPHIC LOG FOR I	20	Boring terminated a	it 20 feet.			44-50/2"	SS6	Bottom cap			20.0 20.4	599.1 598.7	
ON LINES REPRESENT	- 25-												
OTE: STRATELICATIC	— 30- — -	-											
538301.GPJ 12/13/10	- 35-	- - - - -											
PJ GTINC 06		GROUNDWATER D	ATA	DRILLING		<u> </u>		Drawn by: KSA Date: 9/10/10	Checked t	у :DaTK Ар	p'vd. by	MBC 4/11	
OI-COFFEEN.CI	EI	NCOUNTERED AT <u>9</u> F	FEET 🗜	AUGER <u>4 1/4</u> WASHBORING FR <u>MVU</u> DRILLER <u>S</u>	HOLLO OM <u>SWG_</u> L	W STEM FEET OGGER		C	GEOTE	CHNOL FROM T	OGY		
WL J017150				<u>CME 55TRK</u> [HAMMER TYF	PE <u>Aut</u>	81G <u>0</u>		Ameren-Co	offeen Ast	n Pond Ev	aluati	ion	
JF BORING 2002	RE	MARKS:						LOC	S OF BOR	ING: APV G303	V-3		
1000								Pro	ject No.	J017150.	.01		

	Surfa	ace Elevation: <u>623.46</u> Datum <u>msl</u>	Completion Date: <u>8/26/10</u> Northing: 871397.48 Easting: 2515520.23	8/26/10 porthing: 871397.48 asting: 2515520.23 U			WELL DIAGRAM				
	DEPTH IN FEET	DESCR	IPTION OF MATERIAL	GRAPHIC	DRY UNIT WE SPT BLOW (CORE RECOV	SAMPI	Stickup Dlameter-6 Inches -33 526.8				
		Medium stiff, grayis	h-brown, silty CLAY - CL		0.1.1	881	Concrete 1.0 622.5				
					0-1-1	551	2" sch 40 PVC				
	- 5-				1-2-5	SS2	Bentonite				
					1-3-4	SS3					
ES							8.0 615,5				
OIL TYP 5 ONLY.	- 10-				1-2-3	S\$4	10.0 613.5				
VEEN S											
S BETV ON PUF		Medium to fine SAN	ID - SP		1-2-1	SS5	Filter sand				
NDARIE	15-	Hard, gray, silty CL	AY - CL (TILL)				0.10 slotted				
DR ILLU											
DXIMA					14-47	SS6					
E APPR RAPHIC	- 20-	Boring terminated a	t 20 feet.				Bottom cap <u>1:::{</u> 20.0 503.5 20.4 603.1				
ENT TH UAL. G											
EPRES	- 25-										
.INES R MAY BE											
ATION I											
RATIFIC E TRAN	- 30-										
TE: STF AND TH											
10 NO.											
J 12/13	- 35-										
301.GP											
NC 0631											
зРJ GTI		GROUNDWATER D	ATA DRILLING	<u>G DATA</u>		_	Drawn by: KSA Checked by: Drk App'vd. by: R's/ Date: 9/10/10 Date: 1-4-11 Date: 1/4/11				
FEENC	EN		AUGER 4 1/4'		W STEM		GEOTECHNOLOGYZ				
01- COF			<u>MVU</u> DRILLER	<u></u>	OGGER		FROM THE GROUND UP				
L J017150			<u>CME 55TRK</u> HAMMER T	DRILL F YPE <u>Aut</u>	RIG <u>0</u>		Ameren-Coffeen Ash Pond Evaluation				
2002 WI	REI	MARKS:									
ORING							LOG OF BORING: APW-4				
JG OF E							Project No. J017150.01				
ΞL											

F	(EL)								
	Sit Locatio	e: Co n: Co	offeen Po offeen, Il	wer linois	Stati s	on		Rig mfg/model: D-50 Turbo Tracked MST 80 Drilling Method: 4 ¼" HSA, split spoon sample	0ATV BOREHOLE ID: G305 r Well ID: G305
WE	Projec DATE ATHEI	et: 15 S: St Fin R: Cl	5E0030 cart: 5/3 nish: 5/3 loudy, br	/201 /201 eezy,	6 6 , war	m, lo 60	0s	FIELD STAFF: Driller: B. Williamson Helper: D. Crump Eng/Geo: S. Keim	Surface Elev: 622.54 ft. MSL Completion: 18.45 ft. BGS Station: 2,515,199.36N 871,156.33E
5	SAMPL	E	T	EST	TINC	J	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	$\begin{array}{c c} \begin{array}{c} & & \\ & & \\ & & \\ \end{array} \end{array} = \begin{array}{c} \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \end{array} \\ \begin{array}{c} & & \\ & & \\ \end{array} \\ \end{array} \\ \begin{array}{c} & & \\ \end{array} \\ \end{array} \\ \begin{array}{c} & & \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array}$						Quadra Towns	angle: Coffeen, IL hin: East Fork	$\mathbf{\Psi} = 3.00$ - During Drilling $\mathbf{\Psi} =$
er	/ Tot		/6 in alue	ure (%	en. (l)	$(1) \frac{Q_{f}}{D_{f}}$	Section	14, Tier 7N; Range 3W	$\overline{\underline{\nabla}}$ =
qumN	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	14/24	N _{ss}	2-2	12				FILL - Grayish brown (10YR5/2), moist, soft, silty CLA with few small to coarse sand and few small to large grave	Y el
	58%		N=7				2	FILL - Black (10YR2/1), moist, loose, silty, fine- to coarse-grained SAND with little coal fragments.	
2A	17/24 71%	ss	6-7 7-6 N=14	19			¥ _	Gray (10YR6/1) and light yellowish brown (10YR6/4), moist, very stiff, SILT with little clay.	
3A	18/24 75%	ss	3-3 6-4 N=9	28				Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, very stiff, silty CLAY with tra- very fine-grained sand.	ce 618
4A	16/24 67%	ss	6-6 7-8 N=13	24			8	Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY with trace ve fine- to coarse-grained sand.	ry
5A	23/24 96%	ss	1-3 5-6 N=8	21			10	Dark gray (10YR4/1) with 25% dark yellowish brown (10YR4/6) mottles, moist, very stiff, silty CLAY with few fine- to coarse-grained sand and trace small gravel.	N - 614
6A	21/24 88%	ss	3-4 6-6 N=10	21			12	Gray (10YR5/1) with 35% dark yellowish brown (10YR4/6) mottles, moist, very stiff, silty CLAY with few fine- to coarse-grained sand and trace small gravel.	612 N
7A	24/24 100%	ss	8-8 9-9 N=17	18				Yellowish brown (10YR5/8) with 15% gray (10YR5/1) mottles, moist, very stiff, silty CLAY with few fine- to coarse-grained sand and trace small gravel.	610
0 4	19/24 79%	ss	3-3 4-4 N=7	10				Gray (10YR5/1) with 15% yellowish brown (10YR5/6) mottles, moist, stiff, silty CLAY with little fine- to coarse-grained sand and trace small gravel.	608
oA				10			16	mottles, moist, stiff, silty CLAY with some fine- to coarse-grained sand and trace small gravel.	
9A	22/24 92%	ss	<i>1-3</i> <i>7-14</i> N=10	19				Brown (10YR5/3), wet, loose, very silty, very fine- to coarse-grained SAND with trace small gravel.	
9B	0/5 0%	BD		15			18	Brown (10YR5/3) with 40% yellowish brown (10YR5/8 mottles, moist, very stiff, SILT with little clay and trace fine- to coarse-grained sand. End of boring = 18.45 feet	

FI () WE	FIELD BORRING LOG CLIENT: Natural Resource Technology, Inc. Site: Coffeen Power Station Location: Coffeen, Illinois Project: 15E0030 DATES: Start: 5/3/2016 Finish: 5/3/2016 WEATHER: Sunny, calm, warm, lo 60s SAMPLE TESTING CONTRACTOR: Ramsey Geotechnical Engineering LLC Rig mfg/model: D-50 Turbo Tracked MST 800ATV Drilling Method: 4 ¼" HSA, split spoon sampler FIELD STAFF: Driller: B. Williamson Helper: D. Crump WEATHER: Sunny, calm, warm, lo 60s Eng/Geo: S. Keim											Event Hanson BOREHOLE ID: G306 Well ID: G306 Surface Elev: 622.84 ft. MSL Completion: 18.00 ft. BGS Station: 2,516,120.41N 871,140.98E 6000000000000000000000000000000000000			
S	SAMPL (ii)	E	T	EST		(tsf)	TOPOGRAPHIC MAP INFORMATION: Quadrangle: Coffeen, ILWATER LEV $\Psi = 5.5$					EVEL INFORMATION: 5.50 - During Drilling			
er	/ Total		/ <i>6 in</i> alue	ure (%)	en. (lb/	f) Qp (e Type	Towns Section	hip: East Fork n 14, Tier 7N; Range 3W	$\bar{\mathbf{\Lambda}} =$						
Numb	Recov % Rec	Type	Blows N - V; RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	orehole Detail	Elevation ft. MSL	Remarks				
1A	12/24 50%	ss	<i>1-3</i> <i>3-4</i> N=6	14				Very dark brown (10YR2/2), moist, medium, SILT v little clay and few very fine- to medium-grained sand, r trace coal fragments.	vith poots,						
2A			5 /	21			2	Dark gray (10YR4/1) with 5% dark yellowish brow (10YR3/6) mottles, moist, stiff, SILT with little clay a	n						
2B	24/24 100%	ss	5-4 N=9	19				Gray (10YR6/1) with 10% yellowish brown (10YR5 mottles, moist, very stiff, SILT with little clay and tra very fine-grained sand.	5/6) ace		620				
3A	22/24 92%	ss	2-2 3-3 N=5	30			¥	Gray (10YR6/1) with 20% yellowish brown (10YR5	5/6)		618				
4A	20/24 83%	ss	3-4 6-6 N=10	26			8	very fine-grained sand.			616				
5A	24/24 100%	ss	2-2 3-3 N=5	23			10	Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, very stiff, silty CLAY with	trace		614				
6A	22/24 92%	ss	1-2 3-4 N=5	20			12	very fine- to coarse-grained sand.			612				
7A	20/24 83%	ss	5-6 6-6 N=12	21				Gray (10YR5/1) with 30% dark yellowish brown (10YR4/6) mottles, moist, stiff, silty CLAY with few fine- to coarse-grained sand.	very		610				
8A	20/24	N	2-2	15				Yellowish brown (10YR5/6), wet, soft, very fine-to coarse-grained sandy CLAY with little silt.	0		608				
8B	83%		N=10	12			16	Yellowish brown (10YR5/6), wet, medium dense, sil very fine- to medium-grained SAND with trace coarse-grained sand.	lty,						
9A	23/24	ss	14-17 28-50/5'	10				coarse-grained SAND with little silt, little very fine-gra sand, and trace small gravel.	ained		606				
9B	2070	/\	N=45	13			18	(10YR4/6) mottles, moist, hard, SILT with little clay, very fine- to coarse-grained sand, and trace small grav End of boring = 18.0 feet	few vel.						

F	FIELD BORING LOG CLIENT: Illinois Power Generating Co. CONTRACTOR: Bulldog Drilling, Inc.													
	Sit Location Project	e: Co n: Co	offeen Po offeen, II	ower linois	Stati s	on Ash	Pond 1	Rig mfg/model: CME 55LC Track Drill Drilling Method: 4 ¹ / ₄ " Hollow Stem Auger v Spoon	/Continuous Split	BOREHOLE ID: G307 Well ID: G307 Surface Floy: 622.08 ft MSI				
WF	DATE	S: St Fin	art: 07/2 ish: 07/2 vercast x	26/20 27/20	016 016 0 & h	umid (i	mid-80s)	FIELD STAFF: Driller: J. Gates Helper: C. Clines Fng/Geo: R. Hasenvager		Completion: Station:	18.22 ft. BGS 871,398.55N 2 515 553 26E			
	SAMPL	к. О Е	T	TEST	TING	iunnu (i i	TOPOCO				2,515,555.20E			
	(in)				t3)	sf)	Quadr	angle: Coffeen	$\mathbf{\Psi} = 14.0$	ER LEVEL INFORMATION: Z = 14.00 - during drilling				
	otal		in	(%)	(lb/f	<i>Qp</i> (t ype	Towns	hip: East Fork	$\underline{\Psi} = -1.7$	6 - 7/27/2016 @ 0	7:30			
lber	р <i>лоза</i> Г / лс	d)	vs/6 Value	sture	Den.	tsf) ure T	Section		<u> </u>	1 51				
Nun	Rec % R	Typ	Blov N - J	Moi	Dry	Qu (Failt	ft. BGS	Description	Detai	il ft. MSL	Remarks			
	18/24	M	1-3					Brown (10YR5/3), moist, stiff, SILT with few clay, t sand, gravel and roots.	race					
1A	75%	ss	3-2 N=6	22										
							2	Gray (10YR5/1), moist, soft, CLAY, with some silt, t	race	620				
	22/24	M	2-1					suite, graver and roots.						
24	92%	ss	3-3 N=4	28										
2/1		/ \		20			4			- (19				
		\mathbb{N}	1.2					Gray (10YR5/1) with 30% yellowish brown (10YR5 mottles, moist, soft, CLAY with some silt, trace sar	5/8) nd,					
	24/24 100%	ss	1-2 3-4 N=5					gravel and roots.						
3A		$\langle \rangle$	IN-3	26										
							6			616				
	24/24	ss	1-3 3-3											
4A	10070	\mathbb{N}	N=6	18										
							8			614				
	24/24	ss	3-3 4-5					Yellowish brown (10YR5/6) with 20% gray (10YR5 mottles, moist, medium CLAY, with some silt, few v	ery					
5A	100%	A	N=7	19				tine- to medium-grained sand, and trace gravel.						
							10			612				
	24/24	$\mathbb{V}_{\mathbb{R}}$	3-3											
6A	100%	N ³³	N=7	20										
							12	Gray (10YR5/1), moist, medium, CLAY with little sil	t and	610				
7.	24/24	V	woh-2	20				Gray (10YR5/1) with 25% yellowish brown (10YR5 mottles, moist, soft, CLAY with some silt, trace sand	and					
/A	100%	ss	5-13 N=7	20				gravel.						
7B	-			11			⊻ 14	Gray (10YR6/1), wet, medium dense, very fine- to v coarse-grained SAND with few silt and trace clay	ery	••••• ••••• •••••				
	24/24	M	12-9						• • • • • • •	· · · · · · · · · · · · · · · · · · ·				
84	100%	ss	6-9 N=15	20				Yellow brown (10YR5/6), wet. medium dense, very to very coarse-grained SAND, with little silt and tra	ine-	**** <u>-</u>				
04		/ \		20			16	gravel. Vallowish brown (10VP5/8) moist hard SII T with						
	18/18	\bigvee_{\sim}	8-30					clay, little very fine- to very coarse-grained sand and gravel.	race	- 606				
0.4	100%	$\int_{-\infty}^{\infty}$	N=80	0				Gray (10YR5/1), moist, hard, SILT with some clay, I	ittle					
ЭA	0/9	BD		Ó			18	very mile- to very coarse-grained sand and trace grav	·CI.					
	End of boring = 18.2 feet													
NC	DTE(S):													
1														

Illinois Environn	nental Protectio	on Agency				Well	Completion	Report
Site #:		County: <u>Mon</u>	tgomery	7		W	/ell #: G2	.81
Site Name: <u>Coffeen Power Stati</u>	on					В	orehole #:	G281
State Plane Coordinate: X 874,375.4	4 Y_2,514,455.5	(or) Latitude:				Longitud	e:	
Surveyed By: <u>Gary C. Rogers</u>			IL Regi	istrati	on #: <u>035-0</u>	02957		
Drilling Contractor: <u>Ramsey</u>			Driller:	D	. Crump			
Consulting Firm: <u>Hanson Profes</u>	sional Services Inc.		Geolog	ist: _	Rhonald W.	Hasenyage	r, LPG #196-0002	246
Drilling Method: <u>Hollow stem at</u>	uger		Drilling	g Flui	d (Type): <u>no</u>	ne		
Logged By: <u>Kristen L. Theesfel</u>	d		Date St	arted	:9/8/201	1 <u>5</u> Dat	e Finished:9/	8/2015
Report Form Completed By: <u>Suza</u>	anna L. Keim		Date:		10/6/2015			
ANNULAR SPAC	E DETAILS				Elevations	Depths	(0.01 ft.)	
					626.64	2.82	Top of Protective	Casing
					626.36	-2.54	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>				~	623.82	0.00	Ground Surface	
				/	622.82	1.00	Top of Annular S	ealant
Type of Annular Sealant: <u>High-sol</u>	lids bentonite							
Setting Time: >24 hours			7				Static Water Leve	1
							(After Completion)	
Type of Bentonite Seal Granula	ar Pellet Slurry (choose one)		VT.					
Installation Method: <u>Gravity</u>		x x	××		612.59	11.23	Top of Seal	
Setting Time: <u>25 minutes</u>		— X	×		610.37	13.45	Top of Sand Pack	
Type of Sand Pack: <u>Quartz Sand</u>								
Grain Size: 10-20 (sieve	size)				608.31	15.51	Top of Screen	
Installation Method: <u>Gravity</u>					60 2 66			
Type of Backfill Material: <u>n/a</u>					<u>603.66</u> <u>603.53</u>	20.16	Bottom of Screen Bottom of Well	
Installation Mothod:	(if applicable)				602 52	20.20	Pottom of Porch	Ja
					* Referenced to a	National Geodet	ic Datum	ne
					CAS	SING MEA	SUREMENTS	
				Dia	meter of Boreho	ole	(inches)	8.0
WELL CONST (Choose one ty	RUCTION MATERIA	ALS		ID	of Riser Pipe		(inches)	2.0
				Prot	ective Casing L	ength	(feet)	5.0
		DI/C		Rise	er Pipe Length		(feet)	17.80
Protective Casing	88304 88316 PTFE	PVC OTHER: S	teel	Bot	tom of Screen to	o End Cap	(feet)	0.38
Riser Pipe Below W T	SS304 SS316 PTFF	PVC OTHER		Scre	en Length (1s	st slot to last slo	t) (feet)	4.65
Screen	SS304 SS316 PTFE	PVC OTHER:		Scr	an Length of Cas	sing	(inches)	0.010
				Louit			(inches)	0.010

Illinois Environmental Protection Agency		Wel	l Completion	Report
Site #: County:	ntgomery	V	Vell #: G3	01
Site Name: Coffeen Power Station		В	orehole #:	301
State Plane Coordinate: X 872,234.8 Y 2,515,583.0 (or) Latitude:		_ Longitud	le:	
Surveyed By: Gary C. Rogers	IL Registration #:035-	002957		
Drilling Contractor: Ramsey	Driller: <u>D. Crump</u>			
Consulting Firm: Hanson Professional Services Inc.	Geologist: <u>Rhonald W</u>	. Hasenyage	r, LPG #196-0002	246
Drilling Method:Hollow stem auger	Drilling Fluid (Type): <u>n</u>	one		
Logged By: Kristen L. Theesfeld	Date Started:9/4/20	<u>)15</u> Dat	e Finished:9/4	4/2015
Report Form Completed By: <u>Suzanna L. Keim</u>	Date: <u>10/6/2015</u>			
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
	<u></u>		Top of Protective	Casing
	622.65	-2.38	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>	<u>620.27</u>	0.00	Ground Surface	
Type of Annular Sectorite Ching	618.20	2.07	Top of Annular Se	ealant
Installation Method: Gravity				
Setting Time: _>24 hours	<u>v</u>		Static Water Leve (After Completion)	l
Type of Bentonite Seal Granular Pellet Slurry (choose one)	TT .			
Installation Method: <u>Gravity</u>		n/a	Top of Seal	
Setting Time: 25 minutes	612.75	7.52	Top of Sand Pack	
Type of Sand Pack: <u>Quartz Sand</u> Grain Size: <u>10-20</u> (sieve size)	608.96	11.31	Top of Screen	
Installation Method: <u>Gravity</u>	604.31	15.96	Bottom of Screen	
Type of Backfill Material: <u>n/a</u> (if applicable)		16.21	Bottom of Well	
Installation Method:	604.06 * Referenced to	16.21 a National Geoder	Bottom of Boreho tic Datum	le
	CA		CLID EMENITS	
	Diameter of Borel	nole	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe		(inches)	2.0
/	Protective Casing	Length	(feet)	5.0
	Riser Pipe Length	l	(feet)	13.56
Protective Casing SS304 SS316 PTFE PVC OTHER: Riser Pine Above W T SS304 SS316 PTFE PVC OTHER:	Bottom of Screen	to End Cap	(feet)	0.38
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of C	1st slot to last slo	(feet)	4.05
Screen SS304 SS316 PTFE PVC OTHER:	Screen Slot Size *	*	(inches)	0.010

Illinois Environmental Protection Agency			Wel	l Completion	Report
Site #: County: Mon	tgomery		W	Vell #: G3	02
Site Name: Coffeen Power Station			В	orehole #:	G302
State Plane Coordinate: X 872,253.0 Y 2,516,214.2 (or) Latitude:			Longitud	e:	
Surveyed By: <u>Gary C. Rogers</u>	IL Registrat	ion #: <u>035-0</u>	02957		
Drilling Contractor: Ramsey	Driller: <u> </u>	D. Crump			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:	Rhonald W.	Hasenyage	r, LPG #196-0002	246
Drilling Method: Hollow stem auger	Drilling Flui	id (Type): <u>no</u>	ne		
Logged By: Kristen L. Theesfeld	Date Started	1: <u>9/3/201</u>	1 <u>5</u> Dat	e Finished:9/	4/2015
Report Form Completed By: <u>Suzanna L. Keim</u>	Date:	10/7/2015			
ANNULAR SPACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
		620.34	-2.39	Top of Protective	Casing
	=	620.04	-2.09	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>		617.95	0.00	Ground Surface	
Type of Annular Sealant: High solids bentonite		615.95	2.00	Top of Annular S	ealant
Installation Method: Tremie					
Setting Time: _>24 hours	Z			Static Water Leve (After Completion)	l
Type of Bentonite Seal Granular Pellet Slurry					
Installation Method: <u>Gravity</u>	X X	607.78	10.17	Top of Seal	
Setting Time: 25 minutes	×	605.88	12.07	Top of Sand Pack	[
Type of Sand Pack: <u>Quartz Sand</u> Grain Size: <u>10-20</u> (sieve size)		604.74	13.21	Top of Screen	
Installation Method: <u>Gravity</u>		600.09	17.86	Bottom of Screen	
Type of Backfill Material:		599.56	18.39	Bottom of Well	
Installation Method:		599.56 * Referenced to a	18.39 National Geodet	Bottom of Boreho	ble
		CAS	SING MEA	SURFMENTS	
	Dia	meter of Boreho	ole	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID	of Riser Pipe		(inches)	2.0
	Pro	tective Casing L	ength	(feet)	5.0
Protective Casing SS204 SS216 PTEE DVC OTHER. S	Rise	er Pipe Length		(feet)	15.30
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: S	Bot	tom of Screen to	<u>o End Cap</u>	(feet)	0.53
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Tot	al Length of Cas	sing	(feet)	20.48
Screen SS304 SS316 PTFE (PVC) OTHER:	Scr	een Slot Size **	0	(inches)	0.010

Illinois Environmental Protection Agency			Wel	Completior	n Report
Site #: County: Mon	itgomery		W	Vell #:G	305
Site Name: Coffeen Power Station			В	orehole #:	G305
State Plane Coordinate: X 871,156.3 Y 2,515,199.4 (or) Latitude:			Longitud	e:	
Surveyed By: <u>Gary C. Rogers</u>	IL Registrat	ion #: <u>035-0</u> 0	02957		
Drilling Contractor: Ramsey	Driller: <u>B</u>	. Williamson			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: _	Rhonald W.	Hasenyage	r, LPG #196-000	246
Drilling Method: Hollow stem auger	Drilling Flui	d (Type): <u>no</u>	ne		
Logged By:Suzanna L. Keim	Date Started	1:5/3/201	1 <u>6</u> Dat	e Finished: 5	/3/2016
Report Form Completed By:Suzanna L. Keim	Date:	5/19/2016			
ANNULAR SPACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
		625.88	3.34	Top of Protective	e Casing
	_	625.55	-3.01	Top of Riser Pipe	e
Type of Surface Seal:		622.54	0.00	Ground Surface	
		621.54	1.00	Top of Annular S	Sealant
Instellation Method. Crewity					
Setting Time: >12 hours	7			Static Water Lev	el
	<u>×</u>			(After Completion)	•
Type of Bentonite Seal Granular Pellet Slurry	VT I				
Installation Method: <u>n/a</u>	××	n/a	n/a	Top of Seal	
Setting Time:	×	611.04		Top of Sand Pac	k
Type of Sand Pack:Quartz Sand					
Grain Size: 10-40 (sieve size)		609.10	13.44	Top of Screen	
Installation Method:		(04.27	10.27	D.4. (C	
Type of Backfill Material: <u>n/a</u>		604.27	18.27	Bottom of Screen Bottom of Well	1
(if applicable)		604 09	18.45	Bottom of Boreh	ole
		* Referenced to a	National Geodet	tic Datum	
		CAS	ING MEA	SUREMENTS	
	Dia	meter of Boreho	ole	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID	of Riser Pipe		(inches)	2.0
	Pro	tective Casing L	ength	(feet)	5.0
Protective Casing SS204 SS216 DTEE DV/C OTHER S	Rise	er Pipe Length		(feet)	16.45
Riser Pipe Above W.T. SS304 SS316 PTFE (PVC) OTHER	Bot	tom of Screen to	<u>o End Cap</u>	(feet)	0.18
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:		al Lenoth of Cas	sing	(feet)	21.46
Screen SS304 SS316 PTFE PVC OTHER:	Scr	een Slot Size **		(inches)	0.010

Site # County Montgomery Well # G306 Site A Borehole # G306 Site A Borehole # G306 Site A Site A G306 Site A Site A G306 Site A Site A G306 Site A Site A G306 Site A Site A G306 Site A Site A G306 Site A Site A G306 Site A Site A Gande Site A Site A Gande Site A Site A Site A	Illinois Environmental Protection Agency			Wel	l Completion	Report
Site Name Collegen Power Station	Site #: County:	ntgomery		V	Vell #:G3	306
State Plane Coordinate: X. \$71,141.0. Y. 2.516,120.4. (or) Lanitude:	Site Name: Coffeen Power Station			B	orehole #:	G306
Surveyed By: Gary C. Rogers II. Registration #: <u>035-002957</u> Drilling Contractor <u>Ramsey</u> Drilling Contractor <u>Ramsey</u> Drilling Contractor <u>Ramsey</u> Drilling Muhdi <u>Halson Professional Services Inc.</u> Geologier: <u>Rhonald W. Hasenyager, I.PC(#196-000246</u> Drilling Huid (Type): <u>none</u> Logged By: <u>Suzana L. Keim</u> Date Started: <u>5/3/2016</u> Date Finished: <u>5/3/2016</u> Report Form Completed By: <u>Suzana L. Keim</u> Date: <u>5/19/2016</u> ANNULAR SPACE DETAILS Flevations <u>Depths</u> (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (0.01 ft.) (MSI <i>y</i> ¹) (BGS) (DGS) (DG (Front Surface Casaing <u>625.72</u> , <u>-2.88</u> . Top of Riser Pipe <u>621.84</u> , <u>1.00</u> . Top of Annular Sealant Installation Method: <u>nia</u> Static Water Level (<i>Uhe Completion</i>) Type of Banckfill Material: <u>nia</u> (<i>dome conj</i>) Installation Method: <u>nia</u> (<i>dome conj</i>) (<i>dome co</i>	State Plane Coordinate: X 871,141.0 Y 2,516,120.4 (or) Latitude:			Longitud	e:	
Drilling Contractor: Ransey Driller B. Williamson Consulting Firm: Llasson Professional Services Inc. Geologist: Rhonald W. Hasenyager, LPG #196-000246. Drilling Muhci: Hollow stem auger Drilling Fluid (Type):	Surveyed By: Gary C. Rogers	IL Registrati	on #: 035-00	02957		
Cursuling Firm: <u>Hanson Professional Services Inc.</u> Geologist <u>Khonald W. Hasenyager, LPG #196-000246</u> Drilling Fluid (Type): <u>none</u> Logged By: <u>Suzanna L. Keim</u> Date Statter <u>5/3/2016</u> Date Finished: <u>5/3/2016</u> ANNULAR SPACE DETAILS <u>ANNULAR SPACE DETAILS</u> <u>Concrete</u> Type of Surface Sea! <u>Concrete</u> Type of Surface Sea! <u>Concrete</u> Type of Surface Sea! <u>Concrete</u> Type of Surface Sea! <u>Concrete</u> Type of Annular Sealant: <u>Beatomite Chips</u> Installation Method: <u>na</u> Setting Time: <u>n/a</u> Type of Sard Pack: <u>Out 25 hours</u> Type of Sard Pack: <u>Infa</u> Type of Sard Pack: <u>Out 25 hours</u> Type of Sard Pack: <u>Out 25 hours</u> Type of Sard Pack: <u>Infa</u> <u>Concrete</u> <u>1786</u> Bottom of Screen <u>609.77</u> <u>13.07</u> Top of Sard Pack <u>609.77</u> <u>13.07</u> Top of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>605.16</u> <u>17.68</u> Bottom of Screen <u>604.98</u> <u>17.86</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> Bottom of Screen <u>604.98</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>17.86</u> <u>18.87</u> <u>17.86</u> <u>18.00</u> <u>18.87</u> <u>17.86</u> <u>18.00</u> <u>18.87</u> <u>17.86</u> <u>18.00</u> <u>18.87</u> <u>17.86</u> <u>18.00</u> <u>18.87</u> <u>17.86</u> <u>18.00</u> <u>18.87</u> <u>18.90</u> <u>19.10</u> <u>19.10</u> <u>19.10</u> <u>19.10</u> <u>19.10</u> <u>19.10</u> <u>19.10</u>	Drilling Contractor: Ramsey	Driller: <u>B</u>	. Williamson			
Drilling Method: Hollow stem auger Drilling Fluid (Type): none Logged By: Suzanna L. Keim Date: 5/3/2016 Date Finished: 5/3/2016 Report Form Completed By: Suzanna L. Keim Date: 5/19/2016 ANNULAR SPACE DETAILS Elevations Depths (0.01 ft.) Gala (MSL)* Depths (0.01 ft.) Gala (Signa) Surface Seat: Concrete Gala (Signa) Gitting Fine: Surface Seat: Concrete Gala (Signa) Surface Seat: Concrete	Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: _	Rhonald W.	Hasenyage	r, LPG #196-000	246
Logged By: Suzanna L. Keim Date Startiet: 5/3/2016 Date Finished: 5/3/2016 Report Form Completed By: Suzanna L. Keim Date: 5/19/2016 Elevations Depths (0.01 ft.) ANNULAR SPACE DETAILS Elevations Depths (0.01 ft.) yee of Surface Scal: Concrete 622.572 -2.88 Top of Protective Casing type of Annular Sealant: Betroante Chaps 621.84 1.00 Top of Annular Sealant Installation Method: n/a Static Water Lovel Outer Sand Static Water Lovel type of Bentonite Seal - Grandur Pellet Static Static Water Lovel type of Sand Pack: Quartz Sand 609.77 13.07 Top of Sand Pack type of Backfill Material: n/a (dropokabe) 609.16 17.68 Bottom of Screen installation Method: Gravity	Drilling Method:Hollow stem auger	Drilling Flui	d (Type): <u>no</u>	ne		
Report Form Completed By: Suzanna L, Keim Date: 5/19/2016 ANNULAR SPACE DETAILS Flevations Depths (0.01 ft.) (MSL)* (BGS) (Dot of Riser Pipe (Marc Completion) (Marc Completion) (Marc Completion) Type of Bentonite Scal - Granular Pellet (Marc Completion) (Marc Completion) (Marc Completion) (Marc Completion) (Marc Completi	Logged By: <u>Suzanna L. Keim</u>	Date Started	:5/3/201	<u>6</u> Dat	e Finished: <u>5</u> /	3/2016
ANNULAR SPACE DETAILS Elevations Depths (0.01 ft.) (MSL)* (BGS) (Dot ft.) (MSL)* (BGS) (Dot ft.) (MSL)* (BGS) (Dot ft.) (MSL)* (BGS) (Dot ft.) (MSL)* (BGS) (Dot ft.) (MSL)* (BGS) (Dot ft.) (MSL)* (BGS) (BGS) (Dot ft.) (MSL)* (BGS) (Dot ft.) (Dot ft.) (MSL)* (BGS) (Dot ft.) (Dot ft.) (MSL)* (BGS) (Dot ft.) (Dot ft.) (MSL)* (Bot ft.)	Report Form Completed By:Suzanna L. Keim	Date:	5/19/2016			
(MSL)* (BGS) G26.12 -3.28 Top of Protective Casing G26.12 -3.28 Top of Protective Casing G25.72 -2.88 Top of Riser Pipe Type of Annular Sealant: Bentonite Chips	ANNULAR SPACE DETAILS		Elevations	Depths	(0.01 ft.)	
100012 -5.20 Fight Fluctive Casing Type of Surface Seal: Concrete 625.72 -2.88 Top of Riser Pipe Type of Annular Sealant: Bentonite Chips 621.84 1.00 Top of Annular Sealant Installation Method: Gravity			(MSL)*	(BGS)	Top of Protective	Casing
Type of Surface Seal: Concrete Type of Annular Scalant: Bentonite Chips Installation Method: Gravity Setting Time: >12 hours Type of Bentonite Seal – Granular Pellet Static Water Level (Alter Completion) Type of Bentonite Seal – Granular Pellet Static Water Level (Alter Completion) Type of Bentonite Seal – Granular Pellet Static Water Level (Alter Completion) Type of Sand Pack: Quartz Sand (Gravity Type of Sand Pack: Quartz Sand (Gravity Type of Backfill Material: n/a Type of Backfill Material: n/a (#appleable) (#appleable) Installation Method: Gravity Type of Backfill Material: n/a (Backfill Material: n/a (Backfill Material: n/a (Choose one type of material for each area) Protective Casing \$204 \$316 PTTE Protective Casing \$204 \$316 PTTE Riser Pipe Above W.T. \$3204 \$3316 PTTE Riser Pipe Blow W.T. \$3204 \$3316 PTTE Riser Pipe Blow W.T.			020.12	-5.28	Top of Flotective	Casing
Type of Surface Seal: Concrete Type of Annular Sealant: Bentonite Chips Installation Method: Gravity Setting Time: >12 hours Type of Bentonite Seal – Graular Pellet Stury Installation Method: n/a Type of Bentonite Seal – Graular Setting Time: n/a n/a n/a Grain Size: 10.40 (above one) n/a Installation Method: n/a Grain Size: 10.40 (above size) 602.77 Installation Method: n/a (fapplicable) 605.16 Installation Method: $n'a$ (fapplicable) 604.98 Installation Method: $n'a$ (choose one type of material for each area) 604.98 (Choose one type of material for each area) 604.98 (Is dot to hat slot) 600.18 Streen 600.188 Netter Construction MATERIALS Diameter of Borchole (Indee) 50.0 (Choose one type of mate		=	625.72	-2.88	Top of Riser Pipe	•
Type of Annular Sealant: Bentonite Chips Installation Method: Gravity Setting Time: >12 hours Type of Bentonite Seal Granular Pellet Slurry (ebose ore) Installation Method: n/a Setting Time: n/a Installation Method: n/a Setting Time: n/a Installation Method: m/a Grain Size: 10-40 (arev size) 609.77 Installation Method: Gravity Type of Backfill Material: n/a (frapplicable) (frapplicable) Installation Method: Gravity Type of Backfill Material: n/a (frapplicable) (frapplicable) Installation Method: Gravity Installation Method: Gravity Static Water Level (frapplicable) Installation Method: Gravity Installation Method: Gravity Installation Method: Gravity Installation Method: Charge Explanation of Well Installation Method: Gravity In	Type of Surface Seal: <u>Concrete</u>		622.84	0.00	Ground Surface	
Type of Annual's Sealaht: Definition (Chips) Installation Method: Gravity Setting Time: >12 hours Type of Bentonite Seal - Granular Pellet Slurry (choose one) Installation Method: n/a main n/a Type of Bentonite Seal - Granular Pellet Slurry (choose one) Installation Method: n/a Type of Sand Pack: Quartz Sand Grain Size: 10-40 Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (forgenicable) Installation Method: (forgenicable) Installation Method: Casing Streen Streen Length (forgenicable) (forgenicable) MELL CONSTRUCTION MATERIALS Diameter of Borchole (forgenicable) The (Choose one type of material for each area) Casing Length (feet 15.95 Diameter of Borchole (forgenicable) (free of 15.95 Bottom of Screen to End Cap (feet 0.18 Screen Length of Casing <td>Turne of Annulus Statistics Destantia China</td> <td></td> <td>621.84</td> <td>1.00</td> <td>Top of Annular S</td> <td>ealant</td>	Turne of Annulus Statistics Destantia China		621.84	1.00	Top of Annular S	ealant
Installation Method:	Type of Annular Sealant: <u>Bentonite Chips</u>					
Setting Time: >12 hours Statte Water Level (After Completion) Type of Bentonite Seal Granular Pellet Slurry (choose one) Statte Water Level (After Completion) Installation Method: n/a Top of Seal Setting Time: n/a Top of Seal Setting Time: n/a Top of Seal Grain Size: 10-40 (sieve size) Installation Method: Gravity Type of Backfill Material: n/a Top of Screen 609.77 13.07 Top of Screen 605.16 17.68 Bottom of Screen 604.98 17.86 Bottom of Screen 604.98 18.00 Bottom of Screen 604.98 17.86 Bottom of Screen 604.98 17.86 Bottom of Borehole * Referenced to a National Geodetic Datum CASING MEASUREMENTS Casing Ength (see) 5.0 Riser Pipe Above W.T. S304 S316 PTFE Riser Pipe Length (sec) 5.0 Riser Pipe Length (sec) 5.0 Riser Pipe Below W.T. S304 S316 PTFE PVC Streen S304 <t< td=""><td>installation Method: <u>Gravity</u></td><td>_ </td><td></td><td></td><td></td><td>1</td></t<>	installation Method: <u>Gravity</u>	_				1
Type of Bentonite Seal Granular Pellet Slurry Installation Method: n/a Top of Seal Setting Time: n/a n/a Top of Seal Type of Sand Pack: Quartz Sand 609.77 13.07 Top of Screen Grain Size: 10.40 (siew size) 609.77 13.07 Top of Screen Installation Method: Gravity 605.16 17.68 Bottom of Screen Type of Backfill Material: n/a (if applicable) 604.84 18.00 Bottom of Screen Installation Method:	Setting Time: _>12 hours	⊻			(After Completion)	el
Installation Method: n/a Top of Seal Setting Time: n/a Top of Seal Setting Time: n/a Top of Seal Type of Sand Pack: Quartz Sand 609.77 13.07 Top of Screen Installation Method: Gravity 609.77 13.07 Top of Screen Type of Backfill Material: n/a	Type of Bentonite Seal Granular Pellet Slurry					
Setting Time:n/a	Installation Method:		n/a	n/a	Top of Seal	
Type of Sand Pack: Quartz Sand Grain Size: 10-40 Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (if applicable) Installation Method: (if applicable)	Setting Time: $\underline{n/a}$		611.24	11.60	Ton of Sond Dool	-
Type of Sand Pack: Quartz Sand Grain Size: 10-40 (sieve size) Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (if applicable) Installation Method: (if applicable) </td <td></td> <td></td> <td>_011.24_</td> <td></td> <td>Top of Sand Pack</td> <td>2</td>			_011.24_		Top of Sand Pack	2
Grain Size: 10-40 (sive size) Installation Method: Gravity Type of Backfill Material: n/a (if applicable) (if applicable) Installation Method: (if applicab	Type of Sand Pack:Quartz Sand		609 77	13.07	Top of Screen	
Installation Method: <u>Gravity</u> Type of Backfill Material: <u>n/a</u> (if applicable) Installation Method: <u>(if applicable)</u> Installation Method: <u>(if applicable)</u> Installation Method: <u>(if applicable)</u> MELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Protective Casing <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Above W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: Steel Riser Pipe Below W.T. <u>S5304</u> <u>S5316</u> PTFE <u>PVC</u> OTHER: <u>Steel</u> <u>Screen Length</u> (1st slot to last slot) (feet) 4.61 Total Length of Casing (feet) 20.74 Screen Slot Size ** (incluse) 0.010	Grain Size: <u>10-40</u> (sieve size)		009.11		Top of Screen	
Type of Backfill Material: n/a Installation Method: Installati	Installation Method: <u>Gravity</u>		605 16	17.68	Bottom of Screen	
Installation Method: 604.84 18.00 Bottom of Borehole * Referenced to a National Geodetic Datum CASING MEASUREMENTS WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Protective Casing \$\$304 \$\$316 PTFE PVC OTHER: Screen Length (feet) 0.18 Screen \$\$304 \$\$316 PTFE PVC OTHER: Screen Slot Size ** (inches) 0.010	Type of Backfill Material: <u>n/a</u>		604.98	17.86	Bottom of Well	
Installation Method:	(if applicable)		(04.94	10.00		
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Diameter of Borehole (inches) 8.0 Diameter of Borehole (inches) 2.0 Protective Casing \$\$304 \$\$316 PTFE PVC OTHER: Screen Length (feet) 5.0 Riser Pipe Above W.T. \$\$304 \$\$316 PTFE PVC OTHER: Screen Length (feet) 0.18 Screen \$\$304 \$\$316 PTFE PVC OTHER: Screen Slot Size ** (inches) 0.010			* Referenced to a	National Geode	tic Datum	ble
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) Diameter of Borehole (inches) 8.0 ID of Riser Pipe (inches) 2.0 Protective Casing \$\$304 \$\$316 PTFE PVC OTHER: Steel Riser Pipe Above W.T. \$\$304 \$\$316 PTFE PVC OTHER: Riser Pipe Below W.T. \$\$304 \$\$316 PTFE PVC OTHER: Screen \$\$304 \$\$316 PTFE PVC OTHER: Screen \$\$304 \$\$316 PTFE PVC OTHER: Screen Slot Size ** (inches) \$\$0.0			CAS		SUDEMENITS	
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area) ID of Riser Pipe (inches) 2.0 Protective Casing SS304 SS316 PTFE PVC OTHER: Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER: Screen Length (feet) 0.18 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Screen Length (feet) 4.61 Total Length of Casing (feet) Screen Slot Size ** (inches) 20.74		Dia	meter of Boreho	le	(inches)	8.0
Protective CasingSS304SS316PTFEPVCOTHER:SteelProtective CasingSS304SS316PTFEPVCOTHER:Bottom of Screen to End Cap(feet)0.18Riser Pipe Above W.T.SS304SS316PTFEPVCOTHER:Screen Length(1st slot to last slot)(feet)4.61Riser Pipe Below W.T.SS304SS316PTFEPVCOTHER:Total Length of Casing(feet)20.74ScreenSS304SS316PTFEPVCOTHER:Screen Slot Size **(inches)0.010	WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID o	of Riser Pipe		(inches)	2.0
Protective CasingSS304SS316PTFEPVCOTHER: SteelRiser Pipe Length(feet)15.95Riser Pipe Above W.T.SS304SS316PTFEPVCOTHER:Bottom of Screen to End Cap(feet)0.18Riser Pipe Below W.T.SS304SS316PTFEPVCOTHER:Screen Length(1st slot to last slot)(feet)4.61ScreenSS304SS316PTFEPVCOTHER:Total Length of Casing(feet)20.74ScreenSS304SS316PTFEPVCOTHER:Screen Slot Size **(inches)0.010		Prot	tective Casing L	ength	(feet)	5.0
Protective CasingSS304SS316P1FEPVCOTHER: SteelBottom of Screen to End Cap(feet)0.18Riser Pipe Above W.T.SS304SS316PTFEPVCOTHER:Screen Length (1st slot to last slot)(feet)4.61Riser Pipe Below W.T.SS304SS316PTFEPVCOTHER:Total Length of Casing(feet)20.74ScreenSS304SS316PTFEPVCOTHER:Screen Slot Size **(inches)0.010		Rise	er Pipe Length		(feet)	15.95
Nisci ripe Above w.1. SS304 SS310 FTFE LVC OTHER: Screen Length (1st slot to last slot) (feet) 4.61 Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: Total Length of Casing (feet) 20.74 Screen SS304 SS316 PTFE PVC OTHER: Screen Slot Size ** (inches) 0.010	Protective Casing SS304 SS316 PTFE PVC OTHER: S	Bott	tom of Screen to	End Cap	(feet)	0.18
Reset tipe below w.t. Source of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the cive of the ci	Riser Pine Below W T SS304 SS316 PTFE PVC OTHER:	Scre	en Length (1s	t slot to last slo	ot) (feet)	4.61
the national states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and the states and t	Source Source The left The left <th< td=""><td> Tota</td><td>ai Length of Cas</td><td>sing</td><td>(inches)</td><td>0.010</td></th<>	Tota	ai Length of Cas	sing	(inches)	0.010

Illinois Environmental Protection Agency			Well	Completion	Report
Site #: County: Mon	itgomery		W	/ell #:G	307
Site Name: Coffeen Power Station Ash Pond 1			В	orehole #:	G307
State Plane Coordinate: X_2,515,553.3 Y_871,398.6 (or) Latitude:			Longitud	e:	
Surveyed By:Gary C. Rogers	IL Registra	tion #: <u>035-0</u>	02957		
Drilling Contractor: Bulldog Drilling, Inc.	Driller:	J. Gates			
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:	Rhonald W.	Hasenyager	r, LPG #196-000	246
Drilling Method: Hollow Stem Auger	Drilling Flu	uid (Type): <u>no</u>	ne		
Logged By:Rhonald W. Hasenyager	Date Starte	ed: <u>7/26/20</u>	<u>16</u> Dat	e Finished:7/2	27/2016
Report Form Completed By: <u>Rhonald W. Hasenyager</u>	Date:	7/28/2016			
ANNULAR SPACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
		624.72	2.64	Top of Protective	Casing
		624.47	-2.39	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>		622.08	0.00	Ground Surface	
Type of Annular Sealant Bentonite Chins		620.08	2.00	Top of Annular S	ealant
Installation Method: Gravity					
Setting Time: <u>18 hrs.</u>	Z	623.84	1.76	Static Water Leve (After Completion)	el 7/27/2016
Type of Bentonite Seal Granular Pellet Slurry (choose one)					
Installation Method:		n/a	n/a	Top of Seal	
Setting Time:	×	610.10	11.98	Top of Sand Pacl	C
Type of Sand Pack: Quartz sand Grain Size: (sieve size)		609.12	12.96	Top of Screen	
Installation Method: <u>Gravity</u>		604.28	17.80	Bottom of Screen	
Type of Backfill Material:		603.86	18.22	Bottom of Well	
Installation Method:		603.86 * Referenced to a	18.22 National Geodet	Bottom of Boreho ic Datum	ble
		CAS	SING MEA	SUREMENTS	
	Di	ameter of Boreho	ole	(inches)	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID	of Riser Pipe		(inches)	2.0
	Pr	otective Casing L	ength	(feet)	5.0
Destasting Cosing CO204 CO214 DTEL DIVE STORES	Ri	ser Pipe Length		(feet)	15.37
Protective Casing 55304 \$5316 PTPE PVC OTHER: S Riser Pine Above W T \$\$5304 \$\$5316 PTEE PVC OTHER: S	BC	ottom of Screen to	o End Cap	(feet)	0.40
Riser Pipe Below W.T. SS304 SS316 PTFE (PVC) OTHER.		tel Length (1s	st slot to last slo	t) (feet)	4.84
Screen SS304 SS316 PTFE PVC OTHER:		reen Slot Size **	<u>эш</u>	(inches)	0.010

APPENDIX A7

HISTORICAL BORING LOGS

F		D]	BOR	RI	NC	5 L	OG						C		ANSON
WI	CLIEN Sit Locatio Projec DATE	T: A te: C on: C on: C c t: 05 S: St Fin R: O	EG Coffe CB Mana offeen, II 5S3004A cart: 5/3, hish: 5/3, vercast, 1	een I agen llinoi /200 /200 mild	Powe nent l is 6 6 (mid	er Statio Facility I-60's)	ion CONTRACTOR: Testing Service Corporation y Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager TOPOGRAPHIC MAP INFORMATION: Ouadrangle: Coffeen, IL					I	BOI	REHOLE ID Well ID Surface Elev Completion Station	b: SB-01 b: MW1D c: 607 ft. MSL i: 40 ft. BGS i: 874,972.6N 2,513,478.0E
5	SAMPL	Е	Т	EST	TING	Ĵ	TOPOGE	APHIC MAP INFORMATION:	W	ATE	RL	EVI	ELI	INFORMAT	FION:
er	' / Total (in) sovery		/6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	Quadr Towns Section	angle: Coffeen, IL hip: East Fork n 10, Tier 7N; Range 3W		₹ ₹ ∑	= 3 = = 3	34.00 36.28) - V 3 - N	While drilling	g 5/1/06
Numb	Recov % Rec	Type	Blows N - V: RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description			Bo D	oreho Detail	le I	Elevation ft. MSL	Remarks
1A	16/24 67%	ss	1-1 1-2 N=2	22		0.78 B		Very dark gray (10YR3/1), clayey SILT, trace sa	and			<u> </u>		606	
2A	20/24 83%	ss	2-3 5-5 N=8	13		3.71 BSh		Light gray (10YR7/1) with 40% yellowish brow (10YR5/8) mottles, clayey SILT, trace sand and gr	vn ravel					 	
3A	22/24 92%	ss	2-2 3-6 N=5	14		2.62 BSh		Gray (10YR5/1) with 25% yellowish brown (10YF mottles, clayey SILT, trace sand and gravel	R5/6)			נ, נ, נ, נ, נ,		602	
4A	23/24 96%	ss	8-12 19-19 N=31	13		3.30 BSh		Yellowish brown (10YR5/6) with 20% black (10YI mottles, clayey SILT, little sand and gravel	R2/1)		ינ, נ, נ, נ, נ, נ, נ, נ, נ, נ,		600	
5A	24/24 100%	ss	4-9 13-19 N=22	13		4.80 BSh	10	Yellowish brown (10YR5/6) with 40% gray (N5. mottles, clayey SILT, trace sand and gravel	5/1)			נ, נ, נ, נ, נ, נ, נ, ה, נ, נ, נ, נ, נ,		 598	
6A	22/24 92%	ss	3-6 12-15 N=18	12		8.73 <i>B</i>	12							596	
7A	24/24 100%	ss	14-19 23-30 N=42	12		7.86 B	14-	Dark gray (N4/1) with 25% yellowish brown (10Y1 mottles, clayey SILT, trace sand and gravel	R5/6))		נינינינינינינ		 594 	
8A	24/24 100%	ss	4-8 12-14 N=20	13		7.56 B	16				<u> </u>	נ, נ, נ, נ, נ, נ		592	
9A	24/24 100%	ss	16-16 20-21 N=36	14		7.01 B	18-	Dark gray (N4/1), clayey SILT, trace sand and gra	avel					590	
$10A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} ss \begin{vmatrix} 3-5 \\ 8-11 \\ N=13 \end{vmatrix} 14 \begin{vmatrix} 5.24 \\ B \end{vmatrix}$						5.24 B	20							588	
	NOTE(S): N	1W01D i	insta	lled i	in SB-()1.								

F	EL	D]	BOF	RI	NC	5 L	OG						ANSON
	CLIEN Sit Location Projec	T: Al e: C0 n: C0 :t: 05	EG Coffe CB Mana offeen, Il S3004A	een F agem Ilinoi	Powe nent l is	er Statio Facility	on ,	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler			BOI	REHOLE I Well I Surface Ele	D: SB-01 D: MW1D ey: 607 ft. MSL
WE	DATE	S: St Fin R: O	art: 5/3/ iish: 5/3/ vercast, 1	/2000 /2000 mild	6 6 (mid	l-60's)		FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager				Completio Statio	40 ft. BGS n: 874,972.6N 2,513,478.0E
5	SAMPL	E	Т	TEST	TINO 	47	TOPOGR	APHIC MAP INFORMATION:	WATE	R LEV	EL I	INFORMA	TION:
	al (in				b/ft ³)	в	Quadra Townsl	nngle: Coffeen, IL nin: East Fork	Ţ Ţ	= 34.0 =	0 - 1	While drillir	ıg
u.	/ Tot overy		/ 6 in lue	ire (%	en. (Il) e Typ	Section	10, Tier 7N; Range 3W	Ţ	= 36.2	28 - 1	MW01D on	6/1/06
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry Do	Qu (tsf Failur	Depth ft. BGS	Lithologic Description		Boreh Deta	ole il	Elevation ft. MSL	Remarks
11A	24/24 100%	SH		15		3.69 B	22					586	Shelby tube taken from shallow well borehole at indicated depth.
12A	24/24 100%	ss	10-11 12-16 N=23	14		5.24 B	24 –						Shelby tube taken from shallow well
13A	6/24 25%	SH		14		3.69 B	26					582	borehole at indicated depth.
14A	24/24 100%	ss	10-12 18-18 N=30	15		4.27 B	28	Dark gray (N4/1), clayey SILT, trace sand and gra [Continued from previous page]	vel			580	
15A	24/24 100%	ss	5-9 11-16 N=20	14		4.27 B	30					578	
16A	24/24 100%	ss	5-8 10-12 N=18	17		2.72 BSh	32					576	
17A	24/24 100%	ss	8-14 14-16 N=28	14		5.62 B	¥ 34					574	
18A	23/24 96%	ss	8-28 40-65 N=68	14			√ 36	Gray (N4/1), silty, fine to medium SAND, little coa sand, trace gravel, wet	ırse			572	
19B	24/24 100%	ss	24-14 17-16 N=31	19		5.43 B	38	Very dark gray (10YR3/1), silty CLAY				570	
20A	24/24 100%	ss	3-5 6-10 N=11	24		3.50 BSh	40	Dark gray (N4/1) with 30% dark yellowish brow (10YR4/6) mottles, silty CLAY End of Boring = 40.0 ft. BGS	n			568	
	NOTE(S): N	1W01D i	instal	lled i	in SB-0)1.						

F		D]	BOF	RII	NC	5 L	OG				K	ANSON
	CLIEN Sit Locatio	T: A te: C on: C	EG Coff CB Mana offeen, II	een H agen Ilinoi	Powe nent l is	er Statio Facility	on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS & CME samplers		В	OREHOLE ID Well ID:	SB-02 n/a
wi	DATE DATE	S: St Fin R: Pa	art: 5/5 art: 5/5 artly clou	/2000 /2000 /2000	6 6 mild	(high-:	50's)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager			Completion: Station:	50 ft. BGS 876,410.0N 2,513,210.0E
5	SAMPL	E	1	TEST	TING	Ĵ	TOPOGE	APHIC MAP INFORMATION:	WATER	R LEVE	L INFORMAT	ION:
	al (in				5/ft ³)	e	Quadr Towns	angle: Coffeen, IL hin: East Fork	$\mathbf{\underline{V}} = \mathbf{\underline{V}}$	= 12.80 = 7.42	- While drilling - MW02S on 6/	1/06
н	/ Tot əvery		/ 6 in lue	re (%	en. (Il	Typ	Section	10, Tier 7N; Range 3W	<u> </u>	= 7.36	- MW02D on 6/	1/06
Numbe	Recov % Reco	Type	Blows , N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description		Borehol Detail	e Elevation ft. MSL	Remarks
1A	24/24	V	3-3	23				soil grayish brown (10YR4/2), clayey SILT, trace	sand	╧╎┓╶╕ ╷┼┨╼ _┙ ╺┽		
1B	100%	ss	4-5 N=7	18		1.96		loess Gray (10YR6/1), clayey SILT, trace sand				
1C		()	,	29		В 1.94		Yellowish brown (10YR5/4) with 20% gray (10YR mottles, lean CLAY, trace sand	86/1)			
		∇				B				<u> </u>	622	
	24/24	ss	3-4 4-6					Yellowish brown (10YR5/8) with 15% gray (10YR mottles lean CLAY trace sand	86/1)	<u></u>		
2A	10070	Λ	N=8	25		2.89 B		moties, ican CEATT, trace sand				
	(()					4			<u></u>	620	
	24/24	V	3-5					Yellowish brown (10YR5/8) with 40% gray				
3A	100%	1 33	N=10	20		2.91		(10YR6/1)mottles, lean CLAY, trace sand		<u></u>		
	(Д				В	6		//	<u> </u>	618	
		V	10-8									
	24/24 100%	ss	8-10 N-16					Gray (10YR5/1) with 20% yellowish brown (10YR mottles, lean CLAY, little sand	85/6)			
4A		()	N=10	17		2.91 B	1			<u></u>		
	1	∇					8			<u></u>	616	
	24/24	ss	3-3 4-5					C (10VD5/1) '(1.50% II '1.1.1. (10VD		<u></u>	E I	
5A	10070	Λ	N=7	19		1.94 B		mottles, lean CLAY, little sand	(3/6)			
	($\left(\right)$		10		2.12	10-				614	
0A	24/24	V	2-3	18		2.15 B				<u> </u>		
6B	100%	1	N=9	17				Gray (10YR5/1) with 25% yellowish brown (10YR mottles, clayey SAND, trace gravel	35/6)	A-1		
	(4					12			A- 1	612	
7A	24/24	V	4-4	14		2.06		Gray (10 Y R6/1) with 30% white (10 Y R8/1) moth sandy CLAY, trace gravel	ies,			
70	100%	ss	7-10 N=11	17		В		Yellowish brown (10YR5/4), silty, fine SAND, lin	ttle			
/В		()		1/			14	medium sand, wet				
		∇					14				610	
	24/24	ss	15-23 33-68									
8A	10070	Λ	N=56	10						1		
	10/10	\forall	10 62/1				16-	Pale brown (10YR6/3), silty, fine SAND, trace grave	el, wet		- 608	
9A	100%	Δ^{ss}	48-02/4	10		3.92 Sh				1-7-		
	,	\Box					18				606	
10A	12/12 100%	X ss	15-45	9		8.07						
		\square		Ĺ		BSh		Dark gray (10YR4/1), clayey SILT, little sand and g	gravel			
	NOTE(S): B	orehole	aban	done	ed usin	g bentonite	grout pumped from bottom of borehole.	- <u> </u>	14	1	
1												



F	[EL]	D]	BOF	115	NC	J L	OG		<u> </u>		
WI	CLIENT Site Location Projec DATE EATHE	Γ: Α e: C n: C s: C t: 0: Σ: S Γ Γ R: P	EG Coffe CB Mana offeen, II 5S3004A tart: 5/5, nish: 5/5 artly clou	een F igem linoi /200 /200 idy,	Powe lent H is 6 6 6 mild	r Static Facility (high-	on 7 50's)	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS & CME samplers FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	BO	REHOLE ID: S Well ID: n Surface Elev: Completion: Station:	B-02 /a 624 ft. MSL 50 ft. BGS 876,410.0N 2,513,210.0E
	SAMPL	E		EST	TING	<u>;</u>	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL	INFORMATIO	N:
er	/Total (in)		/6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 10, Tier 7N; Range 3W		While drilling MW02S on 6/1/0 MW02D on 6/1/0)6 06
Numb	Recov % Rec	Type	Blows N - V _č RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Detail	Elevation ft. MSL	Remarks
15A	60/60	CS		14			42 -			584	
16A	60/60 100%	cs		13			46	Very dark gray (10Y3/1), lean CLAY, trace sand a gravel [Continued from previous page] End of Boring = 50.0 ft. BGS	nd	578	

NOTE(S): Borehole abandoned using bentonite grout pumped from bottom of borehole.

FI	EL	DI	BOR	RI	NC	d L	OG		
1	CLIEN Sit Locatio	T: Al ce: C(n: C(EG Coffe CB Mana offeen, Il	een F igem linoi	Powe lent I s	er Statio Facility	on V	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA (blind drill)	BOREHOLE ID: SB-02a Well ID: MW2S
	DATE	s: 05 S: St	33004A art: 5/5/	2006	5			FIELD STAFF: Driller: B. Williamson	Completion: 16 ft. BGS
		Fin	ish: 5/5/	2000	6			Helper: R. Keedy	Station: 876,408.9N
WE	ATHE	R: Pa	rtly clou	dy, r	nild	(high-:	50's)	Eng/Geo: R. Hasenyager	2,513,210.0E
2		E	1	ESI		J	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	al (i				b/ft ³	e	Quac Towr	shin: East Fork	$\underline{\Psi}$ = 12.80 - while drilling $\overline{\Psi}$ = 7.42 - MW02S on 6/1/06
г	/ Tot wery		6 in ue	re (%	n. (]]	T_{yp}	Section	n 10, Tier 7N; Range 3W	$\underline{\nabla}$ = 7.36 - MW02D on 6/1/06
Numbe	Recov . % Reco	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
								soil grayish brown (10YR4/2), clayey SILT, trace sa	and 4.4
								loess Gray (10YR6/1), clayey SILT, trace sand	
								Yellowish brown (10YR5/4) with 20% gray (10YR6/ mottles, lean CLAY, trace sand	(1)
									- 622
								Yellowish brown (10YR5/8) with 15% gray (10YR6/ mottles, lean CLAY, trace sand	(1)
							4-		
								Yellowish brown (10YR5/8) with 40% gray (10YR6/1)mottles, lean CLAY, trace sand	
							6	Gray (10YR5/1) with 20% vellowish brown (10YR5/	
							¥	mottles, lean CLAY, little sand	
							8-		- 616
								Gray (10YR5/1) with 50% yellowish brown (10YR5/ mottles, lean CLAY, little sand	/6)
							10		614
								Gray (10YR5/1) with 25% yellowish brown (10YR5/ mottles, clayey SAND, trace gravel	/6)
							12- ▼	Gray (10YR6/1) with 30% white (10YR8/1) mottles sandy CLAY, trace gravel	612
								Yellowish brown (10YR5/4), silty, fine SAND, little medium sand, wet	e
								Pale brown (10YR6/3), silty, fine SAND, trace gravel,	wet 610
					l		I	End of Boring = 15.51 ft. BGS	

See SB-02 for sample & testing details

NOTE(S): MW02S installed in blind-drilled borehole within 10 ft of SB-02.

WE	CLIENT Site Location Project DATES EATHER	": Al : Co : Co : 05 : St Fin & Pa	EG Coffe CB Mana offeen, III S3004A art: 5/5/2 ish: 5/5/2 rtly cloud	en P gem linoi 2006 2006 dy, r	Powe ent H s 5 5 nild	r Statio Facility (high-4	on 50's)	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 4¼" HSA (blind drill) FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	BOREHOLE ID: SB-02b Well ID: MW2D Surface Elev: 624 ft. MSL Completion: 27 ft. BGS Station: 876,414.0N 2,513,209.7E				
er	/ Total (in)	2	/6 in alue	rre (%)	en. (lb/ft ³)	f) e Type	TOPOGI Quadu Towns Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 10, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\underline{\Psi} = 12.80$ - While drilling $\underline{\Psi} = 7.42$ - MW02S on 6/1/06 $\underline{\nabla} = 7.36$ - MW02D on 6/1/06				
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
							2 4 6 10	soil grayish brown (10YR4/2), clayey SILT, trace s loess Gray (10YR6/1), clayey SILT, trace sand Yellowish brown (10YR5/4) with 20% gray (10YR6 mottles, lean CLAY, trace sand Yellowish brown (10YR5/8) with 15% gray (10YR6 mottles, lean CLAY, trace sand Yellowish brown (10YR5/8) with 15% gray (10YR6 mottles, lean CLAY, trace sand Yellowish brown (10YR5/8) with 40% gray (10YR6/1)mottles, lean CLAY, trace sand Gray (10YR5/1) with 20% yellowish brown (10YR5 mottles, lean CLAY, little sand Gray (10YR5/1) with 50% yellowish brown (10YR5 mottles, lean CLAY, little sand	and 5/1) 622 622 618 5/6) 616 614				
							12	Gray (10YR5/1) with 25% yellowish brown (10YR5 mottles, clayey SAND, trace gravel Gray (10YR6/1) with 30% white (10YR8/1) mottle sandy CLAY trace gravel	5/6) 58, 612				
							¥ 1	Yellowish brown (10YR5/4), silty, fine SAND, litt medium sand, wet	le				
							14	Pale brown (10YR6/3), silty, fine SAND, trace gravel	, wet 608				
								Dark gray (10YR4/1), clayey SILT, little sand and gra	avel 606				

F	[EL]	DI	BOR	I	NC	5 L	OG		
WI	CLIEN Sit Location Projec DATE EATHE	Γ: ΑΙ e: CC n: Cc zt: 05 S: St Fin R: Pa	EG Coffe CB Mana offeen, III S3004A art: 5/5/ iish: 5/5/ urtly clou	en P gem linoi 2000 2000 dy, 1	owe ent I s 6 6 mild	r Statio Facility (high-:	on 1/ 50's)	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 4¼" HSA (blind drill) FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	BOREHOLE ID: SB-02b Well ID: MW2D Surface Elev: 624 ft. MSL Completion: 27 ft. BGS Station: 876,414.0N 2,513,209.7E
5	SAMPL	E	T	EST	ING	t	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION: $\nabla = 12.80$ While drilling
ər	Number Recov / Total (ir % Recovery Type Blows / 6 in No - Value RQD Moisture (%) Dry Den. (lb/ft ³)						Town Section	rangie: Colleen, iL ship: East Fork n 10, Tier 7N; Range 3W	$\Psi = 12.80$ - while drilling $\Psi = 7.42$ - MW02S on 6/1/06 $\overline{\Psi} = 7.36$ - MW02D on 6/1/06
Numb	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (tsf Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
							22	Brown (10YR5/3), clayey SILT, little sand, trace gr	avel
							26-	Blueish gray (5B5/1), with 30% dark yellowish bro (10YR4/6) mottles, lean CLAY, trace coal	own 598
1						1		End of Boring - 27 22 ft BCS	

End of Boring = 27.22 ft. BGS See SB-02 for sample & testing details

NOTE(S): MW02D installed in blind-drilled borehole within 10 ft of SB-02.

F	IELD BORING LOG CLIENT: AEG Coffeen Power Station CONTRACTOR: Testing Service Corporation											
	CLIEN Sit Locatio Projec DATE	T: A n: C n: C t: 05 S: St	EG Coffe CB Mana offeen, II S3004A art: 4/2	een F agem llinoi 7/200	Powe nent l is 06	er Statio Facility	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS sampler & 4¼" HSA overdrill FIELD STAFF: Driller: B. Williamson Helmen B. Kanda	BOREHOLE I Well I Surface Ele Completio	D: SB-03 D: MW3D ev: 626 ft. MSL m: 58 ft. BGS			
WI	EATHE	ԲՈ R: Տւ	u sn: 4/2 unny, mil	//200 ld (hi	06 igh-5	50's)	Eng/Geo: R. Hasenyager	Statio	2,514,535.3E			
5	SAMPL	E	T	EST	TING	ī	TOPOGRAPHIC MAP INFORMATION: WATER LE	VEL INFORMA	TION:			
	ll (in)				/ft³)		Quadrangle: Coffeen, IL $\underline{\Psi} = 14.$ Township: Fort Fort $\overline{\Psi} = -7.$.00 - While drillin	1g 6/1/06			
	' Tota wery		6 in ue	re (%	n. (lb	Type	I ownsmip: East Polk $\underline{\mathbf{y}} = 7$ Section 11, Tier 7N; Range 3W $\overline{\mathbf{y}} = 55$.40 - MW03D on	6/1/06			
Numbe	Recov	Type	Blows / N - Val RQD	Moistur	Dry De	Qu (tsf) Failure	Depth Lithologic Bore ft. BGS Description Det	hole Elevation tail ft. MSL	Remarks			
1A	20/24	\bigvee	2-2	17		2.07	soil Very dark grayish brown (10YR3/2), lean CLAY					
1B	83%	ss	4-6 N=6	20		SP	Grayish brown (10YR5/2), lean CLAY, trace sand					
1C	(26		3.30 B	$2 - \frac{1}{2}$ Yellowish brown (10YR5/6), lean CLAY, trace sand	624				
2A	17/24 71%	ss	2-3 4-5 N=7	25		3.05	Yellowish brown (10YR5/6) with 40% gray (N5/1) mottles, lean CLAY, trace sand					
	(В	4	622				
3A	20/24 83%	ss	2-3 3-5 N=6	16		1.96 B	Yellowish brown (10YR5/6) with 40% gray (N5/1) mottles, sandy SILT, trace gravel					
4A	24/24 100%	ss	4-3 5-6 N=8	21		2.27 B	Yellowish brown (10YR5/6) with 50% gray (N5/1) mottles, lean CLAY, trace sand and gravel	618				
5A 5B	21/24 88%	ss	1-3 3-4 N=6	20 19		2.18 <i>B</i>	Yellowish brown (10YR5/6), lean CLAY, little sand, trace gravel 10 Dark gray (10YR4/1), sandy CLAY, trace gravel	616				
6A	18/24 75%	ss	1-2 2-4 N=4	24		0.87 B	12 – Gray (10YR6/1) with 50% yellowish brown (10YR5/8) mottles, lean CLAY, little sand, trace gravel	614	Shelby tube taken from shallow well			
7A 7B	23/24 96% 16/24 67%	SH SS	3-2 2-4 N=4	19 12				612	borehole at indicated depth.			
8A	23/24 96%	ss	3-12 29-50 N-41	13			Imagarstown Yellowish brown (10YR5/8), silty, fine SAND, little medium sand, trace gravel, wet					
8B 8C		\square	1,	13 12			Yellowish brown (10YR5/8), silty, fine SAND, little clay,	610				
9A	14/24	V ss	8-82 85-72	10		6.98 B	Yellowish brown (10YR5/6), lean CLAY, little sand, trace gravel Yellowish brown (10YR5/6) silty SAND_trace gravel					
9B	58%	$\left(\right)$	N=167	13		6.18 BSh	18 vandalia brown (10YR5/4), clayey SILT, trace sand and gravel	608				
10A	18/24 75%	ss ss	6-21 32-49 N=53	8		11.95 Sh	Gray (10YR5/1), sandy SILT, trace gravel	606				
	NOTE(S): N	1W03D i	Instal	Iled i	ın SB-(33.					

F]	[EL]	DI	BOR	RI	NC	G L	OG					ANSON
WI	CLIEN Sit Location Projec DATE	Γ: Α] e: C n: C t: 05 S: St Fin R: Su	EG Coffe CB Mana offeen, II 5S3004A art: 4/2 aish: 4/2 unny, mil	een F agem linoi 7/200 7/200 Id (hi	Powe nent l is 06 06 igh-5	er Statio Facility 50's)	on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler & 4 ¹ / ₄ " HS FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	SA overdrill	BO	REHOLE I Well II Surface Ele Completio Statio	D: SB-03 D: MW3D ev: 626 ft. MSL n: 58 ft. BGS n: 876,554.5N 2,514,535.3E
	SAMPL	E	T	EST	TINC	3	TOPOCI	A DILIC MAD INFORMATION.	WATED	EVEL		TION
er	//Total (in) covery		alue / 6 in	ure (%)	ben. (lb/ft ³)	sf) re Type	Quadı Towns Section	rangle: Coffeen, IL ship: East Fork n 11, Tier 7N; Range 3W	VATER <u>V</u> = <u><u>V</u></u> = <u><u>V</u></u> =	14.00 - ` 7.03 - ` 55.40 - `	While drillin MW03S on MW03D on	ng 6/1/06 6/1/06
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failui	Depth ft. BGS	Lithologic Description	B	orehole Detail	Elevation ft. MSL	Remarks
11A	19/24 79%	ss	3-20 35-29 N=55	10		5.36 Sh	22					Wood fragments
12A	20/24 83%	ss	15-25 89-69 N=114	8		13.00 Sh	24	Gray (10YR5/1), sandy SILT, trace gravel [Continued from previous page]			602	
13A	23/24 96%	ss	14-19 24-22 N=43	9				Yellowish brown (10YR5/4), lean CLAY, trace sand	l and			
13B	{	\rightarrow		13		6.98 <i>BSh</i>	26	gravel			- 600	
14A	24/24 100%	ss	19-21 26-32 N=47	15		7.01 <i>BSh</i>	28	Gray (10YR4/1), clayey SILT, trace sand and grav	/el		598	
15A	21/24 88%	ss	10-25 25-23 N=50	12				Light gray (10YR6/1), SILT, trace sand				
15B	24/24	$\overline{)}$	7-12	13		8.53 BSh	30-				- 596	
16A	100%	ss	19-30 N=31	13		9.16 BSh	32				594	
17A			29-35	14		6.59 B		Grav $(10 V P A/1)$ lean CLAV trace sand and grave	el			
17B	100%	ss	39-42 N=74	14		3.49 Sh	34				592	
18A	23/24 96%	ss	6-8 11-17 N=19	13		7.21 BSh	36	mithboro			590	
19A	24/24 100%	ss	19-21 31-31 N=52	13		6.98 BSh	38	Gray (10YR4/1), clayey SILT, trace sand and grav	vel		588	
20A	24/24 100%	ss	6-10 15-24 N=25	14		6.76 <i>BSh</i>		Gray (10YR4/1), lean CLAY, trace sand and gravel			586	
	NOTE(S): N	1W03D i	nstal	lled i	in SB-()3.					Page 2 of 3

F		D]	BOR	RI		5 L	OG			<	€ ≱H/	ANSON
	CLIEN Sit Locatio Projec	T: Al ae: C0 n: C0	EG Coffe CB Mana offeen, Il	een l agen lino	Powe nent l is	er Statio Facility	on V	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler & 4 ¹ / ₄ " HS.	A overdrill	во	REHOLE ID: S Well ID: I	SB-03 MW3D 626 ft MSL
WI	DATE	S: St Fin R: Su	art: 4/2' ush: 4/2' unny, mil	7/20 7/20 ld (h	06 06 igh-5	50's)	1	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager			Completion: Station:	58 ft. BGS 876,554.5N 2,514,535.3E
	Fotal (in)	E	e e	(%)	. (lb/ft ³) UI	iype	TOPOG Quad Town	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 11 Tier 7N: Range 3W	WATER LI $\underline{\Psi} = 14$ $\underline{\Psi} = 7$ $\nabla = 5$	E VEL 4.00 - 7.03 - 5 40 -	INFORMATIC While drilling MW03S on 6/1/	DN: 06
Number	Recov / 7 % Recov	Type	Blows/6 N - Valu RQD	Moisture	Dry Den	Qu (tsf) Failure 1	Depth ft. BGS	Lithologic Description	Bor D	ehole etail	Elevation ft. MSL	Remarks
21A	19/24 79%	ss	4-10 16-23 N=26	13		8.04 <i>BSh</i>	42	Gray (10YR4/1), lean CLAY, trace sand and grave [Continued from previous page]	1		- 584	
22A	21/24 88%	ss	19-27 28-32 N=55	13		7.56 BSh	44				582	
23A	23/24 96%	ss	4-9 14-18 N=23	14		6.98 BSh	46-	Gray (10YR4/1), clayey SILT, trace sand and grave	el		580	
24A	24/24 100%	ss	20-26 30-33 N=56	14		6.59 BSh	48				578	
25A	24/24 100%	ss	6-10 13-12 N=23	20		3.30 BSh	50-				576	
26A	24/24 100%	ss	4-7 6-7 N=13	22		2.91 BSh	52	Gray (10YR4/1), lean CLAY, trace sand and grave	1		574	
27A 27B	24/24 100%	ss	7-18 37-85 N=55	16 13		4.05 BSh	54	yarmouth (10Y4/1), silty, fine to medium SAND, wet			572	
28A 28B	23/24 96%	ss	15-34 34-19 N=68	11 13		3.22 SP	⊻ 56-	Gray (10YR4/1), silty, fine SAND, trace clay, wet			570	
29A	24/24 100%	ss	19-22 28-18 N=50	20		5.82 B 4.36 BSh	50	Very dark gray (10YR3/1), lean CLAY, trace sand a gravel	nd		568	
		_					30	End of Boring = 58.0 ft. BGS				
	NOTE(S): N	1W03D i	insta	lled i	in SB-(03.					





F wi	ELLI CLIEN: Sit Location Projec DATE: EATHEI SAMPLI	D] F: A e: C n: C n: C fit: 05 S: St Fir R: Pa E	BOR EG Coffe CB Mana offeen, III 533004A tart: 5/11 artly sunn T	een F agem linoi 1/200 ay, co EST	Powe nent l is 06 06 06 06 06 06	GL er Statie Facility mid-50	OG on Vs) TOPOGR	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ /4" HSA w/SS & CME samplers FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager APHIC MAP INFORMATION:	EVALUATER LEVEL INFORMATION: WATER LEVEL INFORMATION:			
mber	cov / Total (i <i>Recovery</i>	be	ws / 6 in - Value D	bisture (%)	y Den. (lb/ft ³	(tsf) ilure Type	Townsl Section	hip: East Fork 2, Tier 7N; Range 3W Lithologic	$\underline{\Psi} = 5.67 - \underline{\Psi} = 5.67$ Borehole	MW04S on 6/1/0	06	
<u>щ</u> 11А 12А	60/60 100% 60/60 100% - 60/60 100%	CS CS		Nois 8	Dry	Qu (1 Fail	Depth ft. BGS 22 24 26 28 30 30 32 34 34 36 38	Lithologic Description		Elevation ft. MSL - 602 - 600 - 598 - 598 - 596 - 596 - 594 - 599 - 599 - 599 - 599 - 599 - 599 - 598 - 588 - 588 - 588 - 588 - 588	Remarks	
	 NOTE(\$	[5): E	 Borehole a	aban	done	d usin	$\begin{vmatrix} & \\ & \\ & \\ & \\ g \text{ bentonite } g \end{vmatrix}$	grout pumped from bottom of borehole.			Page 2 of 3	



NOTE(S): Borehole abandoned using bentonite grout pumped from bottom of borehole.



FIELD BORING LOG														
	CLIEN Sit Locatio Projec DATE	T: A te: C on: C ct: 05 S: St	EG Coffe CB Mana offeen, Il 5S3004A cart: 5/12	een H agen linoi 2/20	Powe nent l is 06	er Statio Facility	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS sampler & 4¼" HSA overdrill FIELD STAFF: Driller: B. Williamson			BOREHOLE ID: SB-05 Well ID: MW5D Surface Elev: 623 ft. MS Completion: 54 ft. BG				
WF	CATHE	Fin R: Si	nish: 5/1 [°] unny mil	7/20 d (m	06 nid-6	0's)		Helper: R. Keedy Eng/Geo: R. Hasenvager		Statio	n: 878,174.8N 2 513 290 3E			
5	SAMPL	E E	T	ES7		;	TODOCDA DILIC MA DINEODMATION.			EVEL INFORMATION:				
	(in)				ft ³)		Quadu	rangle: Coffeen, IL	$\underline{\Psi} = 10.$	10.00 - While drilling				
	Total ⁄ery		6 in Ie	(%) e	r. (lb/	Type	Towns Section	ship: East Fork n 3. Tier 7N: Range 3W	$\underline{\Psi} = 6.$ $\overline{\nabla} = 50.$.74 - MW05S on .44 - MW05D on	6/1/06 6/1/06			
mber	cov / Reco	be	D D D	oisture	y Der	(tsf) ilure	Depth	Lithologic	- Bore	hole Elevation				
'nZ	Re %	L Å	R N.	Ŭ	Ā	Qu Fa	ft. BGS	Description	Det	tail ft. MSL	Remarks			
1A	10/24	V	0-2	29				soil grayish brown (10YR4/2), clayey SILT, trace	sand	622				
1B	79%	Å ss	3-4 N=5	28				loess		S IIE				
		Δ					2	Gray (10YR5/1 with 50% yellowish brown (10YR5 mottles_clayey_SILT	5/6)					
		V	2-5							620				
2A	22/24 92%	ss	5-7 N=10	27		2.47 BSh								
2B		Δ		22		2.13	4	(10YR4/1) mottles, clayey SILT, trace sand						
		V	2_2			В				618				
2.4	24/24 100%	ss	3-6 N=5	21		2.22		Dark gray (10YR4/1) with 30% light gray (10YR7 mottles, lean CLAY	/1)					
3A		()		21		2.33 BSP	6							
		∇								616				
	24/24 100%	ss	7-0 6-8				* =	Gray (10YR6/1), lean CLAY, trace sand						
4A		(N=12	21		1.90 BSh								
		∇					8		-	614				
	18/24 75%	ss	1-3 4-5					Gray (10YR6/1) with 50% yellowish brown (10YR mottles, lean CLAY, trace sand	5/6)					
5A		\bigwedge		22		1.78 B	▼							
6A		∇		16				Yellowish brown (10YR5/6), clayey SAND, trace gr wet	avel,	612				
	20/24 83%	ss	0-1 3-4					Yellowish brown (10YR5/6) with 50% gray 10YR5	5/1)					
6B		\wedge	N=4	22		0.70 BSh		mottles, sandy CLAY						
7.		∇		10			12	Gray 10YR6/1), clayey, fine to medium SAND, tra	ace					
/A	24/24 100%	ss	3-6 17-20	19				Brownish vellow (10VR6/6) silty fine SAND tra						
7B		\bigwedge	N=23	16				medium sand						
		∇					14							
81	20/24 83%	ss	4-16 25-25	20				Yellowish brown (10YR5/6), silty, fine SAND, w	et	- 608				
		\wedge	N=41	11										
о Б 9А		\forall		11			16-	Brown (101K5/3), silty SAND and GRAVEL, w	et					
	24/24	ss	14-18 38-62							606				
0.5	10070	\wedge	N=56							NIE				
9B		\forall		8			18	Dark gray (10YR4/1), sandy SILT, trace clay and gr	avel	NE				
	18/24	ss	14-39 77							604				
10A	1370	\wedge	N=116	9		3.27 Sh								
	 NOTE(∐ (S): N	 1W05D i	 nsta	 lled i	n SB-0	20 <u>⊣</u>			k (FFR)	l			
	NOTE(5): MWU5D installed in SB-05.													

F	EL	D 1	BOR	I	NC	L	OG			<		NSON
	CLIENT Site Location Projec DATES	Г: А е: С n: С t: 05 S: St Fir	EG Coffe CB Mana offeen, Ill S3004A cart: 5/12	en F gem linoi 2/200	Powe lient I s 06	r Statio Facility	on ,	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler & 4 ¹ / ₄ " HSA overdrill FIELD STAFF: Driller: B. Williamson Helperg P. Koody			REHOLE ID: S Well ID: M Surface Elev: Completion: Station:	B-05 1W5D 623 ft. MSL 54 ft. BGS 878 174 8N
WI	EATHEI	R: Si	unny, mil	d (m	id-6	0's)		Eng/Geo: R. Hasenyager			Station.	2,513,290.3E
ber 5	v / Total (in)	DFF LESLING alue alue between the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second					TOPOGR Quadra Townsł Section	APHIC MAP INFORMATION: ngle: Coffeen, IL nip: East Fork 3, Tier 7N; Range 3W	EVEL INFORMATION: 0.00 - While drilling 6.74 - MW05S on 6/1/06 0.44 - MW05D on 6/1/06			
Numł	Recor % Re	Type	Blow: N - V RQD	Moist	Dry I	Qu (ts Failu	Depth ft. BGS	Lithologic Description	Bor D	rehole etail	Elevation ft. MSL	Remarks
							22	Dark gray (10YR4/1), sandy SILT, trace clay and grace <i>[Continued from previous page]</i>	avel		602	
11A	60/60 100%	CS		7				Dark gray (10YR4/1), silty, fine SAND, trace medi sand	um		600	
11B	-			7			24					
12A	60/60 100%	cs		7			26				596 596 594	
	60/60 100%	CS	cs		30	30	Dark gray (10YR4/1), sandy SILT, trace clay and grading of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se	avel		592 592 590		
13A	_	+					8	34			588	
14A	48/60 80%	CS		7			36	Gray (10YR6/1), fine to medium SAND				
14B				14			38	Dark gray (10YR4/1), sandy SILT, trace clay and gradient of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second sec	avel			
	$ \square \square \square \square \square \square \square \square \square \square \square \square \square \square \square \square \square \square \square$											

L



FIELD BORING LOG									
WI	CLIEN Si Locatio Proje DATH	T: A te: C on: C ct: 05 S: St Fin CR: Su	EG Coffe CB Mana offeen, II S3004A art: 5/17 iish: 5/17 unny, mil	een H agerr linoi 7/20 7/20 d (rr	Powe nent l s 06 06 nid-6	er Stati Facility 0's)	on y	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 4 ¹ /4" HSA (blind drill) FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	BOREHOLE ID: SB-05a Well ID: MW5S Surface Elev: 623 ft. MSL Completion: 18 ft. BGS Station: 878,175.6N 2,513,285.5E
5	SAMPL	Æ	Т	EST	TIN(Ĵ	торо	GRAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	l (in)				/ft ³)		Qu	drangle: Coffeen, IL	$\mathbf{Y} = 10.00$ - While drilling
	Tota very		6 in ue	e (%	n. (lb	Type	Sec	ion 3, Tier 7N; Range 3W	$\underline{\Psi} = 6.74$ - MW053 of 6/1/06 $\underline{\nabla} = 50.44$ - MW05D on 6/1/06
Numbei	Recov / % Reco	Type	Blows / N - Val RQD	Moistur	Dry De	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
								soil grayish brown (10YR4/2), clayey SILT, trace sa	and $\frac{\sqrt{3}}{2}$
							2 -	loess Gray (10YR5/1 with 50% yellowish brown (10YR5/ mottles, clayey SILT	/6) 622
							4-	Very dark brown (10YR2/2) with 20% dark gray (10YR4/1) mottles, clayey SILT, trace sand	
							-	Dark gray (10YR4/1) with 30% light gray (10YR7/1 mottles, lean CLAY	1) 618
							<u>▼</u> 	Gray (10YR6/1), lean CLAY, trace sand	
							-	Gray (10YR6/1) with 50% yellowish brown (10YR5/ mottles, lean CLAY, trace sand	(6)
							≖ 10 -	Yellowish brown (10YR5/6), clayey SAND, trace grav	ivel,
							-	Yellowish brown (10YR5/6) with 50% gray 10YR5/ mottles, sandy CLAY	
							12 -	Gray 10YR6/1), clayey, fine to medium SAND, trac gravel, wet	ce 610
							- 14 -	Brownish yellow (10YR6/6), silty, fine SAND, trace medium sand	
							-	Yellowish brown (10YR5/6), silty, fine SAND, we	t 608
							16-	Brown (10YR5/3), silty SAND and GRAVEL, wet	t 0.00
							-	Dark gray (10YR4/1), sandy SILT, trace clay and grav	ivel
								End of Boring = 17.71 ft. BGS See SB-05 for sample & testing details	
								see 52 65 for sample & lesting defaus	
	NOTE	(5)• •	W059 :-	neto1	1ad :	n hline	Idrillad	orabola within 10 ft of SP 05	

F]	[EL]	D 1	BOR	RI	NC	5 L	OG			ON			
	CLIEN Sit Location	Г: А) е: С(n: С(EG Coffe CB Mana offeen, Il	een I agen llinoi	Powe nent l is	er Stati Facility	on y	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler	BOREHOLE ID: SB-06 Well ID: n/a	BOREHOLE ID: SB-06 Well ID: n/a			
WI	Projec DATE EATHEI	t: 05 S: St Fin R: Pa	S3004A art: 5/4, iish: 5/4, urtly sunr	/200 /200 ny, c	6 6 ool (:	mid-50)'s)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	Surface Elev: 623 f Completion: 60 f Station: 879 2,513	t. MSL t. BGS ,015.0N ,190.0E			
5	SAMPL	E	Г	EST	rine I	÷	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:				
	al (in				5/ft ³)	e	Quad	rangle: Coffeen, IL shin: East Fork	$\Psi = 9.30$ - While drilling $\Psi = 6.21$ - MW06S on 6/1/06				
ы	/ Tot		/ 6 in lue	re (%	an. (It	Typ	Sectio	n 3, Tier 7N; Range 3W	$\underline{\nabla} =$				
Numbe	Recov % Rec	Type	Blows , N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Rema	urks			
1A			2.2	25		1.22		soil grayish brown (10YR4/2), clayey SILT, trac	e sand $4 \frac{1}{12} = 4 \frac{1}{12}$				
1B	24/24 100%	ss	2-5 3-5 N=6	28		SP 1.96 B		loess Light brownish gray (10YR6/2) with 20% yellov brown (10YR5/8) mottles, clayey SILT, trace s	wish and 622				
	†	7					2						
2A	24/24	ss	2-4 5-6	27		1.94 BSh		Gray (10YR5/1) with 15% yellowish brown (10Y mottles lean CLAY trace sand	(R5/6) 620				
2B		(N=9	19		2.52 BSh							
	†	7					4						
	24/24	ss	1-3 4						618				
3A	10070	(N=7	21		1.36 <i>B</i>							
	†						₹ 6	Gray (10YR5/1) with 15% yellowish brown (10Y mottles, clayey SILT, trace sand	(R5/6)				
	24/24	ss	4-6 6						616				
4A	100%	(N=12	17		1.78 B							
54	{			18		0.85	8-						
511	22/24	ss	1-2 3-3			BSh		Gray (10YR6/1) with 20% yellowish brown (10Y mottles clavey fine SAND	R5/8)				
5B	9270	(N=5	17		0.81 None		Yellowish brown (10YR5/6), silty, fine SAND, trac	ce clay,				
5C	(21		0.31 None	10	<u>wci</u>					
	20/24	ss	0-1 2-4					Dark yellowish brown (10YR4/6), silty, fine SAN	D. wet				
6A	0570	(N=3	25									
	{						12						
	24/24	ss	2-4 7-8					Brownish yellow (10YR6/6), silty, fine to medium little coarse sand, trace gravel, wet	SAND,				
7A	100%	Ą	N=11	15				Yellowish brown (10YR5/4), sandy SILT, trace g	ravel,				
7B	{			12		1.48 BSh	14	Brown (10YR5/3), silty, fine SAND, little medium	n sand,				
	22/24	ss	11-33 57-35					trace graver, wet					
	92%	Ą	N=90					Gray (N5/1), clayey SILT, little sand, trace gra	vel				
8B	{			8		7.18 Sh	16						
	22/24	ss	30-39 46-55										
9A 84	92%	Λ	N=85	9		11.35							
5/1	{					5.0	18	Dark gray (N4/1), clayey SILT, trace sand and g	ravel				
	24/24	ss	4-23 50-51										
10A	100%	Λ	N=73	7		11.64 Sh							
	<u> </u> NOTE(_] S): B	orehole	 aban	 done	d usin	20 ⊣∃ g bentonite	grout pumped from bottom of borehole.					
1	NOTE(5): Dotenote abandoned using benconte grout pumped from bottom of borenote.												

F	[EL]	D]	BOF	RI	NC	5 L	OG				<	S H/	ANSON	
	CLIEN Sit Locatio	T: A ce: C n: C	EG Coff CB Mana offeen, Il	een I agen Ilinoi	Powe nent l is	er Statio Facility	on ⁄	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler			BOREHOLE ID: SB-06 Well ID: n/a			
WI	Projec DATE EATHE	ct: 05 S: St Fin R: Pa	5S3004A t art: 5/4 nish: 5/4 artly sum	/200 /200 ny, c	6 6 ool (mid-50)'s)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager				Surface Elev: Completion: Station:	623 ft. MSL 60 ft. BGS 879,015.0N 2,513,190.0E	
5	SAMPL	E	Т	TEST	TINC	5	TOPOGR	APHIC MAP INFORMATION:	WAT	TER LE	EVEL	INFORMATIO	DN:	
	mber cov / Total (in <i>Recovery</i> pe pe po vy Den. (lb/ft ³) (tsf) (tsf) (tsf) gd gg gg gg gg gg gg gg gg gg						Quadr Towns	angle: Coffeen, IL hip: East Fork	$\underline{\Psi}$ = 9.30 - while drilling $\underline{\Psi}$ = 6.21 - MW06S on 6/1/06					
lber							Section	13, Tier 7N; Range 3W	-	<u>Z</u> =				
Nun	Recc % R	Type	Blov N - J	Moi	Dry	Qu (Fail	ft. BGS	Lithologic Description		Bor De	ehole etail	ft. MSL	Remarks	
11A	20/24 83%	ss	6-16 33-58 N=49	11		3.49 <i>BSh</i>	22 -					602		
12A	20/24 83%	ss	45-56 54-50/3 N=110	9		11.64 Sh	24-					600		
13A	12/24 50%	ss	26-78	12		2.84 Sh	26					598		
14A	8/24 <i>33%</i>	ss	52-48/2	9		5.43 BSh	28-	Dark gray (N4/1), clayey SILT, trace sand and grave [Continued from previous page]				596		
15A	24/24 100%	ss	10-24 30-40 N=54	13		4.95 BSh	30-		ravel			594		
16A	21/24 88%	ss	10-16 37-38 N=53	8		10.91 BSh	32					592		
17A	17/24 71%	ss	36-47 61/5"	9			34					590		
18A	22/24 92%	ss	11-36 45-60 N=81	9		10.04 Sh	36-					588		
19A 19B	22/24 92%	ss	40-35 34-29 N=69	10 13		9.60 Sh 8.92	38-					586		
20A	24/24 100%	ss	3-8 12-15 N=20	14		B 8.53 BSh		Very dark greenish gray (10Y3/1), lean CLAY, trace	e sand			584		
	NOTE(S): B	orehole	aban	done	ed usin	g bentonite	grout pumped from bottom of borehole.						


F	[EL]	D] T: Al	BOR EG Coffe	en F	NC Powe	F L	OG	CONTRACTOR: Testing Service Corporation	Get Hanson
	Sit Locatio Projo	e: C(n: C(CB Mana offeen, Il	igen linoi	ient I	Facility	,	Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA (blind drill)	BOREHOLE ID: SB-06a Well ID: MW6S Surface Elay: 623 ft MSL
WF	DATE	S: St Fin R: Pa	art: 5/4/ ish: 5/4/ rtly sunr	200 200	6 6 ool (1	mid-50	's)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	Completion: 16 ft. BGS Station: 879,021.2N 2,513,189.4E
S	SAMPL	E	T	EST	ING	ŗ	TOPOCI	A DILIC MAD INFORMATION.	WATED I EVEL INCODMATION.
	al (in)			()	5/ft ³)	в	Quadi	angle: Coffeen, IL bin: East Fork	$\Psi = 9.30 - \text{While drilling}$ $\Psi = 6.21 - \text{MW06S on } 6/1/06$
÷	/ Tot		6 in lue	re (%	n. (]	T_{yp}	Sectio	n 3, Tier 7N; Range 3W	$\overline{\underline{\nabla}}$ =
Numbe	Recov . % Reco	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
								soil grayish brown (10YR4/2), clayey SILT, trace s	sand $\left \frac{\sqrt{h_2}}{1} \right = \left \frac{\sqrt{h_2}}{1} \right $
								loess Light brownish gray (10YR6/2) with 20% yellowis brown (10YR5/8) mottles, clayey SILT, trace sand	sh d
							2	Gray (10YR5/1) with 15% yellowish brown (10YR5 mottles, lean CLAY, trace sand	5/6) 620
							¥ 6	Gray (10YR5/1) with 15% yellowish brown (10YR5 mottles, clayey SILT, trace sand	5/6)
							¥	Gray (10YR6/1) with 20% yellowish brown (10YR5 mottles, clayey, fine SAND	5/8)614
							10	Dark yellowish brown (10YR4/6), silty, fine SAND,	wet 612
								Brownish yellow (10YR6/6), silty, fine to medium SA little coarse sand, trace gravel, wet	AND,610
							14	Yellowish brown (10YR5/4), sandy SILT, trace grav wet Brown (10YP5/3) silty fine SAND little medium s	vel,
								trace gravel, wet	
							1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Gray (N5/1), clayey SILT, little sand, trace gravel	
							16-4	Dark gray (N4/1), clayey SILT, trace sand and grav	
								See SB-06 for sample & testing details	

NOTE(S): MW06S installed in blind-drilled borehole within 10 ft of SB-06.

W	CLIENT Site Location Projec DATES EATHEI	e: Co n: Co t: 05 S: St Fin R: Pa	CB Mana offeen, Il offeen, 06 2006 2006 2006 2006 2006	owe ent H s 5 5 5 nild	(mid-7	on 7 Os)	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers FIELD STAFF: Driller: P. McIntire Helper: S. McCartney Eng/Geo: R. Hasenyager			BOF	REHOLE II Well ID Surface Elev Completion Station	b: SB-07 b: n/a c: 625 ft. MSL a: 54 ft. BGS a: 879,180.0N 2,514,390.0E	
er	V Total (in)	2	/ 6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	TOPOG Quad Town Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork on 2, Tier 7N; Range 3W	WATI ⊈ ⊈ ∑	$\mathbf{ER} \mathbf{L} \mathbf{I}$ $\mathbf{L} = 10$ $\mathbf{L} = 4$ $\mathbf{L} = 4$	EVEL I 0.80 - V 4.90 - N	NFORMA While drilling AW07S on 6	FION: 3 /1/06
Numb	Recov % Rec	Type	Blows N - V2 RQD	Moistu	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		Bor De	ehole etail	Elevation ft. MSL	Remarks
A			1-1	23		2.40 B		soil lark gray (10YR3/1), moist, soft, clayey SILT w trace sand and trace gravel.	vith			- 624	
В	18/24 75%	ss	1-1 N=2	24		D	2	Dark gray (10YR4/1), moist, soft, silty CLAY with tra and and trace gravel.	ace				
Δ	24/24 100%	ss	0-0 1-1 N=1	26		2 33		Gray (10YR5/1) with 20% yellowish brown mottles moist, soft, silty CLAY with trace sand and trace grav	s, vel.			622	ion ukotko tkin in one t
Л				20		B	4					ash ba	ised on how soft this is + c
A B C	24/24 100%	ss	0-1 1-2 N=2	25 25 22		4.33 BSh 2.52 BSh 3.05	Ţ	Gray (10YR2/1), very moist, soft, clayey SILT. Gray (10YR6/1) with 30% yellowish brown (10YR5/ mottles, moist soft, silty CLAY with trace sand and tra- gravel. Black (10YR2/1), moist, soft, silty CLAY with trace sa	/8) ace and				
4	24/24 100%	ss	0-1 1-2 N=2	22		в 1.75 В	6 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Dark gray (10YR4/1), very moist, soft, silty CLAY w some sand and trace gravel.	/ - '			618	
A	24/24 100%	ss	0-0 1-1 N=1	22		1.24 B							
A	24/24 100%	ss	0-0 0-1	24		0.54 B	10- Y	With 50% yellowish brown (101R5/ mottles, moist, soft, silty CLAY with sand and grave	(8) el.			614 614	
B			N=0	20 24		1.65 B	12	 Yellowish brown (10YR5/8) wet, very soit, clayey, very fine- to fine-grained, SAND with trace gravel. Yellowish brown (10YR5/8) wet, soft, clayey, very fin to fine-grained SAND with trace gravel 	ne-				
в	24/24 100%	ss	0-0 1-2 N=1	13		2.89 B	14	to the galled, of the will date gravel.					
A	24/24 100%	ss	2-5 7-9 N=12	8		5.04 BSh	16	Gray (10YR4/1) moist very hard candy clavery SII	Т			610	
A	24/24 100%	ss	3-6 6-8 N=12	9		9.27 BSh	18	with gravel.					
)A	24/24 100%	ss	3-6 8-11 N=14	9		11.13 BSh							

F	[EL]	D]	BOR	RI		G L	OG			N
	CLIEN Sit Locatio Projec	Γ: Α e: C n: C st: 05	EG Coffe CB Mana offeen, Il 5S3004A	een F agem linoi	Powe nent l	er Statio Facility	on y	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA w/SS & CME samplers	BOREHOLE ID: SB-07 Well ID: n/a Surface Eley: 625 ft. M	SL
wi	DATE EATHE	S: S Fi R: Pa	tart: 5/5/ nish: 5/8/ artly clou	/2000 /2000 dy, 1	6 6 mild	(mid-7	70s)	FIELD STAFF: Driller: P. McIntire Helper: S. McCartney Eng/Geo: R. Hasenyager	Completion: 54 ft. B0 Station: 879,180 2,514,390	3S).0N).0E
	SAMPL	E	Т	EST	FINC	3	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:	
	y y		u	(%	lb/ft ³	эd	Townsl	nip: East Fork	$\mathbf{\underline{Y}} = 10.80$ - while drilling $\mathbf{\underline{Y}} = 4.90$ - MW07S on 6/1/06	
ber	v / Td		s/6i /alue	ture (Den. ($re T_y$	Section	2, Tier 7N; Range 3W	<u> </u>	
Num	Recc %Rt	Type	Blow N - V RQI	Mois	Dry	Qu (Faili	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks	
11A				8				Gray (10YR4/1), moist, very hard, sandy, clayey S with gravel. [Continued from previous page]	SILT 604	
11B	35/48 73%	CS		14			22-	Gray (10YR6/1), wet, loose, fine- to medium-grai SAND.	ined	
12.4	50/60 83%	cs					24	Gray (10YR5/1), moist, very hard, sandy, clayey S with gravel.	SILT	
12A 13A	56/60	CS		6			30			
13B 13C	95%			6 8			32	Gray (10YR5/1), wet, loose, medium- to very coarse-grained SAND and GRAVEL. Gray (10YR5/1), wet, dense, very fine- to fine-gra SAND.	ined	
14A	60/60 100%	cs		14			34 36 38 38 38	Gray (10YR5/1), moist, very hard, sandy, clayey S with gravel.	SILT 590	
	NOTE(S): E (orehole a ME-105	aban 0 ha	done d 28	ed usin 0# harr	g bentonite g mer for SPT	rout pumped from bottom of borehole.	Do co 1	.£ 7



CLIENT: A BC COFFee Power Station CONTRACTOR: Reynols Dolling Cop. DORELOLE ID: Sta-07a; Market Coffee, Illinos Coffee, Illinos Big mfg/model: CAR-1050 ATV Rig DORELOLE ID: Sta-07a; Project: 053004A Starface Elev: (62 fs. MSI); DORELOLE ID: Sta-07a; Well D: SMV75 Project: 053004A FIELD STAFF: Driller: P. McIntire Starface Elev: (62 fs. MSI); DORESCHEMENT: 141.DGS FIELD STAFF: Driller: P. McIntire Starface Elev: (62 fs. MSI); DOROGRAPHIC MAPPINFORMATION: Vel TERLE/STAFF; VETTIRE: Coverast, mid mid-d050 Englese: R. Hasensyger 2544-397.SE Vertractore Interview Station: Vertractore Station: Station: Station: Station: Vertractore Station: Station: Station: Station: Vertractore Station: Station: Station: Station: Vertractore Station: Station: Station: Station: Vertractore Station: Station: Station: Station: Vertractore Station: Station: Station: Station: Vertractore Station: Station: Station: Station: <tr< th=""><th>F</th><th>[EL]</th><th>DI</th><th>BOR</th><th>I</th><th>NG</th><th>d L</th><th>OG</th><th></th><th>Ć</th><th></th><th>NSON</th></tr<>	F	[EL]	DI	BOR	I	NG	d L	OG		Ć		NSON
Price Price: P. Melnice: Completion: 14 n. BGS WEATHER: Uncertain under de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de la service de		CLIEN Sit Locatio Projec	T: Al e: CC n: Cc n: Cc	EG Coffe CB Mana offeen, III S3004A	en P gem linoi	owe ent F s	r Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill)	BOREHO	DLE ID: S Well ID: M ce Elev:	B-07a 1W7S 625 ft. MSL
SAMPLE TESTING Quarangle: Coffeen. II. Quarangle: Coffeen. II. Quarangle: Coffeen. II. Quarangle: Coffeen. II. Quarangle: Coffeen. II. Testing: Testing Sector 2. WATER LEVEL INFORMATION: Quarangle: Coffeen. II. Testing: Testing Sector 2. agg	wi	DATE	S: Sta Fin R: Ov	art: 5/9/ ish: 5/9/ /ercast, n	2006 2006 nild	5 5 (mid	-60's)		FIELD STAFF: Driller: P. McIntire Helper: S. McCartney Eng/Geo: R. Hasenyager	Con	pletion: Station:	14 ft. BGS 879,181.1N 2,514,397.5E
and and the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the constraint of the const	,	SAMPL	E	Т	EST	ING	ł	TOPOC	ADHIC MAD INFORMATION:	WATER LEVEL INFO	PMATIO	N.
The second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se)er	/ / Total (in) covery		a∕6 in alue	ure (%)	Den. (lb/ft ³)	sf) re Type	Quad Town Sectio	angle: Coffeen, IL hip: East Fork 1 2, Tier 7N; Range 3W	$\Psi = 10.80 - While$ $\Psi = 4.90 - MW0$ $\Psi =$	drilling 7S on 6/1/0	06
Image: Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Start Star	Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failu	Depth ft. BGS	Lithologic Description	Borehole Elev Detail ft. 1	ation MSL	Remarks
Gray (10YR4/1), moist, very hard, sandy, clayey SILT with gravel. End of Boring = 14.39 ft. BGS								2 4 10 12 12	Soil lark gray (10YR3/1), moist, soft, clayey SIL? Trace sand and trace gravel. Dark gray (10YR4/1), moist, soft, silty CLAY with and and trace gravel. Dark gray (10YR5/1) with 20% yellowish brown mott moist, soft, silty CLAY with trace sand and trace gravel. 10ess Black (10YR2/1), very moist, soft, clayey SIL? Gray (10YR6/1) with 30% yellowish brown (10YI mottles, moist soft, silty CLAY with trace sand and gravel. Black (10YR2/1), moist, soft, silty CLAY with trace gravel. Dark gray (10YR4/1), very moist, soft, silty CLAY with trace gravel. Dark gray (10YR4/1), very moist, soft, silty CLAY with trace gravel. Gray (10YR5/1) with 30% yellowish brown (10YI mottles, moist, soft, silty CLAY with sand and gravel. Gray (10YR5/1) with 30% yellowish brown (10YI mottles, moist, soft, silty CLAY with sand and gravel. Gray (10YR5/1) with 30% yellowish brown (10YI mottles, moist, soft, silty CLAY with sand and gravel. Yellowish brown (10YR5/8) wet, very soft, clayey fine- to fine-grained, SAND with trace gravel. Yellowish brown (10YR5/8) wet, soft, clayey, very to fine-grained, SAND with trace gravel.	f with 1 / 1 / 1 / 1 / 1 / 1 / 1 / 1 / 1 / 1 /	24 22 20 18 16 14	
								14	Gray (10YR4/1), moist, very hard, sandy, clayey s with gravel. End of Boring = 14.39 ft. BGS	SILT		

See SB-07 for sample & testing details

F]	[EL]	D 1	BOF	RI	NC	5 L	OG				ANSON
	CLIEN Sit Locatio Projec	Γ: Α΄ e: C n: C n: C	EG Coff CB Man offeen, II 5S3004A	een F agem llinoi	Powe nent l is	er Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers	В	OREHOLE ID: Surface Elev:	SB-08 1/a 625 ft. MSL
wi	DATE EATHE	S: St Fir R: Fo	t art: 5/9 nish: 5/1 oggy to p	/2000 0/200 partly	6 06 ⁷ suni	ny, mil	d (hi-60's)	FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager		Completion: Station:	59 ft. BGS 879,770.0N 2,514,480.0E
5	SAMPL	E	Т	TEST	ING	J	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVE	LINFORMATIO)N:
	Fotal (in) ery		in	(%)	(lb/ft ³)	Jype	Quad Town	rangle: Coffeen, IL ship: East Fork	$\underline{\Psi} = 12.70$ $\underline{\Psi} = 5.33$ $\overline{\nabla} = -$	- While drilling - MW08S on 6/1/	06
Number	Recov /] % Recov	Type	<i>Blows / 6</i> N - Valu RQD	Moisture	Dry Den.	Qu (tsf) Failure 1	Depth ft. BGS	Lithologic Description	<u> </u>	e Elevation ft. MSL	Remarks
	24/24 100%	ss	1-1 1 N=2					soil Very dark gray (10YR3/1), moist, soft, clayey SILT trace sand and trace gravel.	with $\frac{\sqrt{b}}{\sqrt{b}}$	<u>624</u>	
	24/24		1-1				2	Gray (10YR5/1) with 20% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand	.5/6)	622	
2A	100%		N=2	25		1.59 B	4-	Gray (10YR5/1) with 20% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand	.5/6)		
3A	24/24 100%	ss	1-1 1 N=2	20		1.86 <i>B</i>	Ā	Gray (10YR5/1) with 40% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand	.5/6)	620	
3B	24/24	V se	0-1	21		1.20 B	6-	loess Gray (10YR5/1), moist, soft, clayey SILTwith little and trace gravel.	sand	618	
4A	100%		N=2	25		1.01 None	8-	Yellowish brown (10YR5/6) with 20% gray (10YR mottles moist soft sandy CLAY with sl trace gra	25/1)		
5A	24/24 100%	ss	1 N=2	21			10-				
6A 6B	24/24 100%	ss	0-1 1 N=2	21 31		0.70 BSh		Yellowish brown (10YR5/6) with 10% gray (10YR mottles, moist, soft, sandy CLAY with sl. trace gra	15/1) ivel.	- 614 	
7A	20/24 83%	ss	0-1 3 N=4	13			12- ▼	Grav (10YR5/1) wet soft verv silty verv fine-	to	612	
7B				15			14	coarse-grained SAND.			
8A	23/24 96%	ss	7-11 16 N=27	9		5.45 Sh		Light gray (10YR6/1) with 50% yellowish brow (10YR5/6) mottles, wet, very dense, very fine- t fine-grained SAND. Gray (10YR5/1) with 30% yellowish brown (10YR	n o	610	
8B				8			16	mottles, wet, dense, fine- to very coarse-grained SA	ND.	608	
	24/30 80%	CS					18-	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel	ILT		
9A	-	ł		5				with trace graver.		606	
	 NOTE	∥ S): ₽	 Borehole	 ahan	 done	d usin	₂₀ ∃	grout numped from bottom of borehole			
	(Č	CME-105	50 ha	d 280	0# han	imer for SF	Ť.			Page 1 of 3

F]	[EL]	D]	BOF	RI	NC	5 L	OG			<	S H	ANSON
	CLIEN Sit Location	Г: А е: С п: С	EG Coffe CB Mana offeen, Il	een F agem Ilinoi	Powe nent l is	er Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers		во	REHOLE ID: S Well ID: 1	5B-08
wi	Projec DATE EATHE	t: 03 S: S Fin R: Fo	53004A tart: 5/9/ nish: 5/1/ oggy to p	/2000 0/200 artly	6 06 7 sun	ny, mil	d (hi-60's)	FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager			Surface Elev: Completion: Station:	625 ft. MSL 59 ft. BGS 879,770.0N 2,514,480.0E
5	SAMPL	E	Г	EST	TINC	Ĵ	TOPOGR	APHIC MAP INFORMATION:	WATER	LEVEL	INFORMATIO	DN:
ber	v / Total (in covery		<i>s / 6 in</i> <i>l</i> alue	ture (%)	Den. (lb/ft ³)	sf) tre Type	Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 2, Tier 7N; Range 3W	⊻ = ⊻ = ∑ =	12.70 - 5.33 -	While drilling MW08S on 6/1/	06
Num	Recc % Re	Type	Blow N - V RQI	Mois	Dry	Qu (1 Failt	Depth ft. BGS	Lithologic Description	E	Borehole Detail	Elevation ft. MSL	Remarks
10A	60/60 100%	CS		7			22			د از از از از از از از از از از از از از		
11A	36/60 60%	CS		10			24					
12A	60/60 100%	CS		6			30	Gray (10YR5/1), moist, very hard, clayey, sandy SI with trace gravel. [Continued from previous page]	LT			
13A	60/60 100%	cs		7			34				590	
	NOTE(S): Borehole abandoned using bentonite grout pumped from bottom of borehole. CME-1050 had 280# hammer for SPT.											



Fl	EL	D]	BOR EG Coffe	RII en l	NC	J L	OG	CONTRACTOR: Reynolds Drilling Corn	Get Hanson
	Si Locatio Proje	te: C on: C ot: 05	CB Mana offeen, II 5S3004A	igen linoi	nent l	Facility	y	Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill)	BOREHOLE ID: SB-08a Well ID: MW8S Surface Elev: 625 ft. MSL
WI	DATI EATHE	2 S: S (Fir 2 R: F(art: 5/10 nish: 5/10 oggy to pa)/20 0/20 artly	06 06 7 sun	ny, mil	ld (hi-60	's) FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	Completion: 1/ ft. BGS Station: 879,776.6N 2,514,478.8E
5	SAMPI	LE	T	ES1	TINC	; ;	торо	GRAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	Total (i <i>very</i>		6 in ue	e (%)	1. (lb/ft ³	Type	To Se	vnship: East Fork tion 2, Tier 7N; Range 3W	$\underline{\Psi} = 12.70$ - while drilling $\underline{\Psi} = 5.33$ - MW08S on 6/1/06 $\underline{\nabla} =$
Numbeı	Recov / % Reco	Type	Blows / N - Valı RQD	Moistur	Dry Dei	Qu (tsf) Failure	Depti ft. BG	Lithologic S Description	Borehole Elevation Detail ft. MSL Remarks
		1						very dark gray (10YR3/1), moist, soft, clayey SILT trace sand and trace gravel.	$\Gamma \text{ with } \begin{array}{ c c } \hline 1 & 1 & 1 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2 & 2 \\ \hline 1 & 2$
							2	Gray (10YR5/1) with 20% yellowish brown (10YI mottles, moist, soft, silty CLAY with little sand	R5/6) d622
							4	Gray (10YR5/1) with 20% yellowish brown (10YI mottles, moist, soft, silty CLAY with little sand	R5/6) d.
							Ā	Gray (10YR5/1) with 40% yellowish brown (10YI mottles, moist, soft, silty CLAY with little sand	R5/6) d.
							6	loess Gray (10YR5/1), moist, soft, clayey SILTwith little and trace gravel.	e sand
							8	Yellowish brown (10YR5/6) with 20% gray (10YI mottles, moist, soft, sandy CLAY with sl. trace gr	R5/1) avel.
							10	Yellowish brown (10YR5/6) with 10% gray (10YI mottles, moist, soft, sandy CLAY with sl. trace gr	R5/1) avel.
							⊻ 14	Gray (10YR5/1), wet, soft, very silty, very fine- coarse-grained SAND.	to
								Light gray (10YR6/1) with 50% yellowish brow (10YR5/6) mottles, wet, very dense, very fine- fine-grained SAND. Gray (10YR5/1) with 30% yellowish brown (10YR	$ \begin{array}{c c} & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ & \bullet & \bullet \\ $
							16	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel.	AND. SILT 608
								End of Boring = 17.08 ft. BGS See SB-08 for sample & testing details	
	NOTE	(S): N	4W08S ii	nstal	lled i	n blind	l-drilled	borehole within 10 ft of SB-08.	
									Page 1 of 1

F J	CLIEN Sit Location Projec DATE EATHEI	D F : Al e : CO n : Co e t: 05 S : St F in R : Or F	EG Coffe CB Mana offeen, II S3004A art: 5/3, iish: 5/3, vercast, 1	een Fagem llinoi /2000 /2000 mild	NC Powe nent H is 6 6 (mid	-60's)	OG on	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ /4" HSA w/SS & CME samplers FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	BOREHOLE ID: SB-09 Well ID: MW9D Surface Elev: 625 ft. MSL Completion: 54 ft. BGS Station: 879,679.7N 2,515,666.3E
er	/ Total (in)	E.	/ 6 in Ilue	ES (%) arr	en. (lb/ft ³)	f) e Type	TOPOGI Quadı Towns Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 2, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 14.00$ - While drilling $\Psi = 5.23$ - MW09S on 6/1/06 $\overline{\Psi} = 52.46$ - MW09D on 6/1/06
quuN	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	22/24 92%	ss	0-1 1-1 N=2	24		1.65 B	2	soil lark gray (10YR3/1), moist, soft, clayey SILT trace sand and trace gravel.	with
2A	20/24 83%	ss	1-2 1-1 N=3	27		2.06 <i>B</i>	4-	Yellowish brown (10YR5/6), moist, soft, silty CLAY little sand.	d with
3A	24/24 100%	ss	1-1 1-1 N=2	24		1.65 <i>B</i>	<u>▼</u>	Gray (10YR5/1) with 20% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand.	5/6)
4A	24/24 100%	ss	1-1 1-2 N=2	23		1.57 B	8	Gray (10YR5/1) with 50% yellowish brown (10YR mottles, moist, soft, silty CLAY with sand.	5/6)
5A	24/24 100%	ss	0-1 1-1 N=2	27		1.85 B	10-	Yellowish brown (10YR5/6) with 10% Gray (10YR mottles, moist, soft, silty CLAY with sand.	25/1)
6A	24/24 100%	ss	0-0 0-0 N=0	22		1.44 B	12-	Yellowish brown (10YR5/8), very moist, soft, san CLAY with trace gravel.	dy 614
7A 7B 7C	24/24 100%	ss	0-0 1-2 N=1	21 18 15		0.93 BSh 0.97 BSh	¥ 14	Yellowish brown (10YR5/6) with 30% gray (10YR mottles, moist to very moist, clayey SILT with sand trace gravel. Gray (10YR5/1), wet, soft, clayey very fine- to fine-grained SAND	5/1) l and
8A 8B	24/24 100%	ss	1-3 6-6 N=9	18 18				Gray (10YR5/1), wet, loose, fine- to very coarse-gra SAND with trace gravel. Gray (10YR5/1), wet, dense, silty very fine- to fine-grained SAND.	ined 610
9A	24/24 100%	ss	7-13 19-25 N=32	8		7.86 Sh	16	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel.	ILT
	12/12 100%	Cs S): M C	IW09D i ME-105	instal 0 hae	lled i d 280	n SB-()# ham	20 20 99. umer for SP	Т.	606

FI	EL	D]	BOR	RII	NC	5 L	OG					<		ANSON
	CLIEN Sit Location	Г: А е: С п: С	EG Coffe CB Mana offeen, Ill	en H Igen linoi	Powe nent I is	er Statio Facility	on V	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA w/SS & CME samplers	8			во	REHOLE I Well I	D: SB-09 D: MW9D
WF	Projec DATE EATHEI	t: 0: S: S(Fir R: 0	533004A tart: 5/3/ nish: 5/3/ vercast, n	200 200 nild	6 6 (mid	l-60's)		FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager					Completio Statio	ev: 625 ft. MSL on: 54 ft. BGS on: 879,679.7N 2,515,666.3E
2 L	Very (in) Total (in)	E	t f in T	ES] (%) a	n. (lb/ft ³) NI	Type	TOPOGR Quadra Towns Section	APHIC MAP INFORMATION: ingle: Coffeen, IL iip: East Fork 2, Tier 7N; Range 3W	W	ATER ⊻ = ⊻ = ∑ =	LEV 14.0 5.2 52.4	7 EL 00 - 23 - 46 -	INFORMA While drillin MW09S on MW09D on	ng 6/1/06 6/1/06
Numbe	Recov	Type	Blows / N - Val RQD	Moistur	Dry De	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description		E	Boreh Deta	nole ail	Elevation ft. MSL	Remarks
11A	60/60 100%	CS		2			22						604	
12A	49/60 82%	cs		3			26	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel. [Continued from previous page]	SILT				598	2" Sand stringer
13A	60/60 100%	cs		3			30 32 34						592	
14A	60/60 100%	cs		10			36-	Yellowish brown (10YR5/6) with 40% gray (10YR mottles, moist, very hard, clayey SILT with little sar occasional dry, silt stringers (<1").	R5/1) nd and				588	
14B				11			38-	Gray (10YR5/1), moist, very hard, clayey SILT with sand and trace gravel.	h little		2 5 5 5 5 5		586	DBII I ED NOTE.
								Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel.	SILT					Appears more plastic
	NOTE(S): MW09D installed in SB-09. CME-1050 had 280# hammer for SPT.													

F	FIELD BORING LOG CLIENT: AEG Coffeen Power Station CONTRACTOR: Reynolds Drilling Corp.												
	CLIENT Site Location Projec	Γ: Α] e: C(n: C(t: 05	EG Coffe CB Mana offeen, Ill S3004A	en F gem linoi	Powe lent l s	r Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers	BOREHOLE ID: SB-09 Well ID: MW9D Surface Elev: 625 ft. MSL				
WI	DATES EATHER	S: St Fin R: O [,]	art: 5/3/2 iish: 5/3/ vercast, n	2000 2000 nild	5 5 (mid	-60's)		FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	Completion: 54 ft. BGS Station: 879,679.7N 2,515,666.3E				
5	SAMPLI	E	Т	EST	ING	ŗ	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:				
er	' / Total (in) :overy		/6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	Quad Towr Sectio	rangle: Coffeen, IL ship: East Fork on 2, Tier 7N; Range 3W	Y = 14.00 - While drilling Y = 5.23 - MW09S on 6/1/06 ∇ = 52.46 - MW09D on 6/1/06				
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks				
15A	60/60 100%	CS		12			42	Gray (10YR5/1), moist, very hard, clayey, sandy SII with trace gravel. [Continued from previous page]	_T				
	_						44	Very dark grayish brown (10YR3/2), moist, hard, sil <u>CLAY with trace sand and trace organic matter.</u> Very dark grayish brown (10YR3/2), moist, firm, PE/	Ity AT. 580				
19	60/60 100%	CS		22			46	Dark greenish gray (10BG4/1), moist, firm, silty CLA with little sand and trace gravel.	AY 578				
16A				17			48	Gray (10YR5/1), moist, very hard, clayey, sandy SII with trace gravel. Dark gray (10YR3/1), moist, hard, clayey SILT.	.T				
17A	60/60 100%	cs					50- 52- ⊊	Dark greenish gray (10BG4/1), moist, firm, silty CLA with little sand and trace gravel.	AY				
							54 -=	End of Boring = 54.0 ft.					

F	[EL]	DI	BOR	RI	NC	5 L	OG				H	ANSON
	CLIEN Sit Locatio Proioc	T: Al e: C(n: C(EG Coffe CB Mana offeen, II	een H igerr linoi	Powe nent l is	er Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA (blind drill)		В	OREHOLE ID: Well ID: Surface Eleve	SB-09a MW9S 625 ft MSL
WF	DATE	S: St Fin R: O	art: 5/3/ ish: 5/3/ vercast, r	/200 /200 nild	6 6 (mic	l-60's)		FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager			Completion: Station:	16 ft. BGS 879,684.9N 2,515,666.2E
5	SAMPL	E	Т	EST	ΓINC	ť	TOPOGE	RAPHIC MAP INFORMATION:	WATE	R LEVE	L INFORMATI	ON:
ler	/ / Total (in) covery	Recov / Total (in % Recovery Type Blows / 6 in N - Value RQD Moisture (%) Dry Den. (lb/ft ³) Qu (tsf)			f) re Type	Quadr Towns Section	rangle: Coffeen, IL hip: East Fork n 2, Tier 7N; Range 3W	₹ ₹ ∑	$= 14.00 \\ = 5.23 \\ = 52.46$	- While drilling - MW09S on 6/1/ - MW09D on 6/1	/06 /06	
Numb	Recov % Rec	Type	Blows N - V: RQD	Moist	Dry D	Qu (ts Failun	Depth ft. BGS	Lithologic Description		Borehole Detail	e Elevation ft. MSL	Remarks
							2	soil lark gray (10YR3/1), moist, soft, clayey SILT trace sand and trace gravel. Yellowish brown (10YR5/6), moist, soft, silty CLAY little sand.	with		624 622 622	
							₽ 6	Gray (10YR5/1) with 20% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand.	.5/6)		-618	
							8	Gray (10YR5/1) with 50% yellowish brown (10YR5 mottles, moist, soft, silty CLAY with sand.	.5/6)		616	
							10	Yellowish brown (10YR5/6) with 10% Gray (10YR mottles, moist, soft, silty CLAY with sand.	25/1)		614	
							12	Yellowish brown (10YR5/8), very moist, soft, sand CLAY with trace gravel.	ıdy			
							¥ 14 1111111	Yellowish brown (10YR5/6) with 30% gray (10YR5 mottles, moist to very moist, clayey SILT with sand trace gravel. Gray (10YR5/1), wet, soft, clayey very fine- to fine-grained SAND. Gray (10YR5/1), wet, loose, fine- to very coarse-gra SAND with trace gravel. Gray (10YR5/1), wet, dense, silty very fine- to fine-grained SAND.	5/1) I and		612	
							16-	with trace gravel. End of Boring = 16.20 ft. BGS See SB-09 for sample & testing details	ĺ		444	

NOTE(S): MW09S installed in blind-drilled borehole within 10 ft of SB-09.

F	[EL]	D]	BOF	RI	NC	G L	OG		CR HANSON
	CLIEN Sit Locatio Projec DATE	T: A ie: C n: C it: 05 S: St E:	EG Coff CB Man offeen, I 5S3004A tart: 5/1	een I agen Ilinoi /200	Powe nent l is 6	er Statio Facility	on V	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers FIELD STAFF: Driller: K. Doetzel Halson S. McCataga	BOREHOLE ID: SB-10 Well ID: MW10D Surface Elev: 621 ft. MSL Completion: 49 ft. BGS
WI	EATHE	ги R: О	vercast,	/200 mild	o (mić	1-60's)		Eng/Geo: R. Hasenyager	2,515,914.0E
5	SAMPL	Е	1	TEST	FINC	3	TOPOGI	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
Jer	v / Total (in) covery		s / 6 in alue	ture (%)	Den. (lb/ft ³)	sf) re Type	Quadi Town Sectio	rangle: Coffeen, IL ship: East Fork n 2, Tier 7N; Range 3W	
Numl	Reco ^r % Re	Type	Blow: N - V RQD	Moist	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
	17/24	V ss	1-1 1-1					soil tark gray (10YR3/1), moist, soft, clayey SILT trace sand and trace gravel.	with $\begin{pmatrix} x & y \\ y & y \end{pmatrix}$ $\begin{pmatrix} x & y \\ y & y \end{pmatrix}$ $\begin{pmatrix} x & y \\ y & y \end{pmatrix}$
1A	71%		N=2	23		2.33 B	2	Gray (10YR5/1) with 30% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand	5/6) - 620
2A	24/24 100%	ss	1-1 2-2 N=3	25		2.47 BSh	4-	Gray (10YR5/1) with 10% yellowish brown (10YR	5/6)
3A	24/24 100%	ss	1-2 1-2 N=3	24		2.33 B	Ā	motiles, moisi, soit, sity CLAT with fittle said	616
4A	24/24	V se	1-1	23		1.55 B	6	Gray (10YR5/1), moist, soft, clayey, very fine- t fine-grained SAND.	
4B	100%		N=2				8-	Gray (10YR5/1) with 10% yellowish brown (10YR mottles, moist, soft, silty CLAY with little sand and gravel	5/6) trace
5A	17/24 71%	ss	1-1 1-2 N=2	23		2.06 B	10		-612
6A	24/24 100%	ss	0-1 1-1 N=2	17		1.16 B	⊻ 12	Gray (10YR5/1) with 10% yellowish brown (10YR mottles, clayey, very fine- to medium-grained SANE trace gravel.	5/8) 9 with 610
7A	23/24 96%	ss	1-1 3-3 N=4	17		2.84 <i>B</i>		Conside based (10ND5/2) and base succession	608
7B	-	$\left(\right)$		18			14	medium-grained SAND with trace coarse- to ver coarse-grained sand.	
8A	24/24 100%	ss	3-5 9-9 N=14	10		7.64 BSP	16-	Yellowish brown (10YR5/6), moist, firm, silty CL with little sand and trace gravel.	AY 606
10A	12/33 36%	CS		16			18	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel.	ILT
	NOTE(S): N	/ /W10D	insta	 led i	in SB-1	20 ⁻¹		602
		C	CME-105	50 ha	d 28	0# harr	mer for SP	Т.	Page 1 of 3

F]		D]	BOR	RI	NC	5 L	OG		HANSON
	CLIEN Sit Locatio	T: A e: C n: C	EG Coffe CB Mana offeen, Ill	een F Igem linoi	Powe nent l is	er Stati Facility	on V	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers	BOREHOLE ID: SB-10 Well ID: MW10D
WI	Projec DATE EATHE	21: 05 S: S1 Fir R: O	5S3004A tart: 5/1/ nish: 5/1/ vercast, n	2000 2000 nild	6 6 (mid	l-60's)		FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	Surface Elev: 621 ft. MSL Completion: 49 ft. BGS Station: 878,245.1N 2,515,914.0E
	otal (in) otal	E	T 	EST (%)	(lb/ft ³)	spe adv	TOPOGRA Quadran Townshi	PHIC MAP INFORMATION: ngle: Coffeen, IL p: East Fork	WATER LEVEL INFORMATION: $\Psi = 11.80$ - While drilling $\Psi = 4.91$ - MW10S on 6/1/06
lumber	ecov / T 6 Recove	ype	<i>lows / 6</i> I - Value QD	Ioisture	bry Den.	u (tsf) ailure T	Depth	Lithologic	$\underline{\nabla} = 47.48 - MW I0D \text{ on } 6/1/06$ Borehole Elevation Detail ft MSL Remarks
111A	56/60 93%	CS CS	B N	12			22	Gray (10YR5/1), moist, very hard, clayey, sandy Sl with trace gravel. [Continued from previous page]	ILT
11B				13				Dark grayish brown (10YR4/2), very moist, soft, PE	AT.
11C				196			28		592
12A	60/60 100%	CS		13			30	Gray (10YR4/1), moist, very hard, clayey, sandy Sl with trace gravel.	
13A	60/60 100%	cs S): M	1W10D in 'MF-1050	12 nstal	lled i	in SB-	36 38 38 40 10.		-586
1									Page 2 of 3

F	ELI	D]	BOR	RI		G L	OG		(ANSON
WI	CLIENT Site Location Projec DATES EATHEF	f: Al e: Co n: Co t: 05 S: St Fin R: O	EG Coffe CB Mana offeen, II 5S3004A art: 5/1/ iish: 5/1/ vercast, r	een I Igem linoi 200 200 nild	Powe nent l is 6 6 (mic	er Stati Facility 1-60's)	on /	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA w/SS & CME samplers FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	ВС	DREHOLE I Well I Surface Ele Completio Statio	D: SB-10 D: MW10D ev: 621 ft. MSL on: 49 ft. BGS n: 878,245.1N 2,515,914.0E
5	SAMPLI	£	Т	EST	FINC	3	торос	RAPHIC MAP INFORMATION:	WATER LEVEL	INFORMA	TION:
er	lumber ecov / Total (in é <i>Recovery</i> ype <i>lows / 6 in</i> 1- Value QD foisture (%) ry Den. (lb/ft ³)					f) e Type	Quad Town Secti	rangle: Coffeen, IL ship: East Fork on 2, Tier 7N; Range 3W		While drillin MW10S on MW10D on	ng 6/1/06 6/1/06
Numb	Recov % Rec	Type	Blows N - V2 RQD	Moistı	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Detail	Elevation ft. MSL	Remarks
14A 14B	60/60 100%	CS		14 23			42	Gray (10YR4/1), moist, very hard, clayey, sandy S with trace gravel. [Continued from previous page] Greenish gray (5G5/1), moist, firm, silty CLAY with sand and sl. trace gravel.	ILT	- 580	1" Gravel stringer
15A	-			13			44	Greenish gray (5G5/1) with 30% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY with 1 sand and sl. trace gravel.	/n ittle	576	
15B	00/60 100%	CS		25			 ⊻ 48	Greenish gray (5G5/1), moist, firm, silty CLAY with sand and sl. trace gravel.	n little	574	
								End of Boring $= 48.75$ ft.			

FI	EL	DI	BOR	RII	NC	L	OG		HANSON
WE	CLIEN Sit Locatio Projec DATE	Γ: Α] e: C0 n: C0 ct: 05 S: St Fin P: Su	EG Coffe CB Mana offeen, II S3004A art: 5/2, ish: 5/2,	een H agem Ilinoi /2000 /2000	Powe nent I is 6 6	er Statio Facility	5n 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA (blind drill) FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Con: P. Hasenvager	BOREHOLE ID: SB-10a Well ID: MW10S Surface Elev: 621 ft. MSL Completion: 16 ft. BGS Station: 878,250.5N 2 515 914 4E
5	AMPL	E.	т. Т	TEST		2			2,515,714.41
ber	v / Total (in) covery		s / 6 in ⁷ alue	ture (%)	Den. (lb/ft ³)	sf) tre Type	TOPOO Qua Tow Secti	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork on 2, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 11.80$ - While drilling $\Psi = 4.91$ - MW10S on 6/1/06 $\overline{\Psi} = 47.48$ - MW10D on 6/1/06
Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
							2	Gray (10YR5/1) with 10% yellowish brown (10YR5/1) with 10% yellowish brown (10YR5/1) with 10% yellowish brown (10YR5/1) with 10% soft, silty CLAY with little sand.	with 5/6) 620 618 616 616
							8	Gray (10YR5/1) with 10% yellowish brown (10YR5 mottles, moist, soft, silty CLAY with little sand and t gravel.	5/6) trace 5/8) - 612
							¥ 12 14 14	Grayish brown (10YR5/2), wet, loose, very fine- to medium-grained SAND trace gravel. Grayish brown (10YR5/2), wet, loose, very fine- t medium-grained SAND with trace coarse- to very coarse-grained sand. Yellowish brown (10YR5/6), moist, firm, silty CLA with little sand and trace gravel. Gray (10YR5/1), moist, very hard, clayey, sandy SI with trace gravel.	with 608 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
		I		1	I	I	=	End of Boring = 16.30 ft. BGS See SB-10 for sample & testing details	

F]	(EL) CLIEN' Sit Locatio Projec DATE CATHE	D] T: A ^T e: C ⁰ n: C ⁰ r: C ⁰ r: C ⁰ s: St S: St Fin R: Pa	BOR EG Coffe CB Mana offeen, II S3004A sart: 4/2 hish: 4/2 artly clou	een I agen Ilinoi 7/20 8/20 Idy, 1	Powe nent l is 06 06 mild	Facility (mid-6	OG on y	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4¼" HSA w/SS & CME samplers FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	BOREHOLE ID: SB-11 Well ID: MW11D Surface Elev: 622 ft. MSL Completion: 36 ft. BGS Station: 876,749.6N 2,515,976.7E
er	/ Total (in)	E	/ 6 in Jue	rre (%)	en. (lb/ft ³)	e Type	TOPOG Quad Town Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 2, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 11.70$ - While drilling $\Psi = 5.42$ - MW11S on 6/1/06 $\overline{\Psi} = 6.03$ - MW11D on 6/1/06
Numb	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (ts) Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	24/24	\bigvee	3-3	16				soil very dark grey (10YR3/1), moist, soft, clayey SI	$\mathbf{T}. \qquad \underbrace{\begin{smallmatrix} \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} \\ \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & \underline{\lambda} & $
1B	100%	ss	4-4 N=7	34		1.09 B	2	Yellowish brown (10YR5/6) with 20% gray (10YF mottles, moist , soft, silty CLAY.	5/1) 620
2A	24/24 100%	ss	4-7 14-21 N=21	27		2.13 BSh		Yellowish brown (10YR5/6) with 40% gray (10YF mottles, moist , soft, silty CLAY.	5/1) 618
3A	24/24	\bigvee	4-6	24		2.27 BSh		Gray (10YR5/1), moist, soft, clayey SILT.	
3B	100%	ss	7-8 N=13	19		2.33 B	⊻		616
4A	24/24 100%	ss	/-8 <i>13-14</i> N=21	21		2.13 None	8-	Gray (10YR4/1), moist, firm, silty CLAY with little	sand 614
5A	24/24 100%	ss	3-4 4-4 N=8	20		1.55 Sh	10		612
6A	24/24 100%	ss	3-2 2-3 N=4	19		0.78 BSh	▼		
6B		$\overline{\langle \cdot \rangle}$	3-6	18			12	Gray (10YR4/1), wet, soft, sandy CLAY. Yellowish brown (10YR5/6) with 50% gray (10YR mottles wet soft sandy CL AV with trace grave	
7A	24/24 100%	ss	14-21 N=20	16				Gray (10YR5/1), moist, very hard, sandy, clayey S	
7B	11/12	ss	41-61/5	9		5.45 BSP	14	with gravel. Gray (10YR5/1), moist, hard, silty CLAY with little	sand 608
8A				7			16	and trace gravel. Light gray (10YR7/1) with 10% yellowish brow (10YR5/8) mottles, moist, hard, silty CLAY with 1 sand and trace gravel.	n ittle 606
9A	48/54 89%	CS		8			18	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel.	ILT
	NOTE(S): N C	/W11D i XME-105	 insta 0 ha	 11ed i d 280	in SB-1 0# ham	$20^{-11.}$	Т.	$\begin{array}{ c c }\hline \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline$



F	EL]	DI	BOR	RI	NG	L	00	Ĵ			•		ANSON
	CLIEN Sit Locatio Proiec	T: Al ce: C(n: C(ct: 05	EG Coffe CB Mana offeen, II S3004A	een F igem linoi	Powe lent H s	r Statio Facility	on 7		CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill)		В	OREHOLE II Well II Surface Ele	D: SB-11a D: MW11S y: 622 ft. MSL
WI	DATE	S: St Fin R: Pa	art: 4/28 ish: 4/28 rtly clou	8/200 8/200 dv. 1)6)6 nild	(mid-6	i0's)		FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenvager			Completion Station	n: 14 ft. BGS n: 876,749.4N 2.515.971.2E
	SAMPL	E	T	EST	ING	t i	ТО	DOCT					FION
er	/ Total (in) overy		/ 6 in Ilue	ire (%)	en. (lb/ft ³)	f) e Type		POGE Quadr Fowns Section	CAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 2, Tier 7N; Range 3W	wan ⊈ ⊻ ∑	$\frac{1}{2} = 11.70 + \frac{1}{2} = 5.42 + \frac{1}{2} = 6.03 + \frac{1}{2} = 6.03 + \frac{1}{2} = 6.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + \frac{1}{2} = 0.03 + 1$	- While drillin - MW11S on 6 - MW11D on 6	g 5/1/06 5/1/06
Numb	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry D	Qu (tsi Failur	De ft. E	pth 3GS	Lithologic Description		Borehole Detail	e Elevation ft. MSL	Remarks
							Ţ Ţ	2	soil very dark grey (10YR3/1), moist, soft, clayey SII Yellowish brown (10YR5/6) with 20% gray (10YR mottles, moist, soft, silty CLAY. Yellowish brown (10YR5/6) with 40% gray (10YR mottles, moist, soft, silty CLAY. Gray (10YR5/1), moist, soft, clayey SILT. Gray (10YR4/1), moist, firm, silty CLAY with little and trace gravel.	LT. (5/1) R5/1) e sand		620 620 618 616 614 614	
							⊻ 1		Gray (10YR4/1), wet, soft, sandy CLAY.			610	
									Y ellowish brown (10YR5/6) with 50% gray (10YR mottles, wet, soft, sandy CLAY with trace grave	x5/1) el.			
							1		Gray (10YR5/1), moist, very hard, sandy, clayey S with gravel.	SILT		- 608	
								–	End of Boring = 14.08 ft. BGS See SB-11 for sample & testing details				

$ \mathbf{F} $	[EL]	D]	BOF	RII		5 L	OG		C HANSON
	CLIEN Sit Locatio Projec DATE	T: A te: C on: C ct: 05	EG Coff CB Mana offeen, II 5S3004A art: 5/1	een H agen Ilinoi	Powe nent l is 06	er Statio Facility	on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS & CME sampler	BOREHOLE ID: SB-12 Well ID: MW12D Surface Elev: 622 ft. MSL Completion: 50 ft BGS
		Fir	ish: 5/1	0/20	06		1 (1-: 60'-)	Helper: R. Keedy	Station: 875,515.1N
	EATHE SAMPL	к: го Е		EST		ny, mi G			2,515,900.0E
	otal (in) ry		'n	(%)	(lb/ft ³)	pe	TOPOG Quad Town	rangle: Coffeen, IL ship: East Fork	$\Psi = 12.00 - \text{While drilling}$ $\Psi = 6.76 - \text{MW12S on } 6/1/06$
lber	ov / To	0	$\sqrt{3}/6$	sture (Den.	tsf) ure T_{j}	Sectio	n 11, Tier 7N; Range 3W	$\nabla = 46.90 - MW12D \text{ on } 6/1/06$
Num	Recc %R	Type	Blow N - N	Mois	Dry	Qu (Faili	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	24"/24	ss	2-3 4-5 N=7	22		1.27 Sh		soil Very dark gray (10YR3/1), clayey SILT, trace s	sand $\frac{\sqrt{2}}{\sqrt{2}}$ $\frac{\sqrt{2}}{\sqrt{2}}$ 622
1B		Д		27		2.33 B	2	(10YR5/6) mottles, lean CLAY	
2A	19"/24	ss	2-4 5-7 N=9	24		2.91 B		Yellowish brown (10YR5/8) with 40% grayish b (10YR5/2) mottles, lean CLAY	brown
3A	20"/24	ss	2-2 3-4 N=5	21		2.13 B	6	Gray (10YR5/1), lean CLAY, trace sand and gra	ravel
4A	24/24 100%	ss	4-5 5-6 N=10	21		1.36 BSh	⊻ 8	Gray (10YR5/1) with 10% yellowish brown (10Y mottles, lean CLAY, trace sand	YR5/6)
5A	24"/24	ss	1-2 2-5 N=4	20		1.47 BSh	10	Yellowish brown (10YR5/8) with 20% gray (10Y mottles, lean CLAY, trace sand and gravel	YR6/1)
6A	20"/24	ss	0-1 3-3 N=4	21		0.62 B	¥ 12	hagarstown Yellowish brown (10YR 5/8) with 25% gray (10Y mottles, clayey SAND, trace gravel	YR6/1)
7A	21"/24	V	2-2	22		0.19 B		Gray (10YR6/1), clayey SAND, trace gravel, v	wet 610
7B 7C			N=5	22 15		3.71	14	Light yellowish brown (10 YR4/0), elayey SAND, gravel, wet	wnish which which which we have a second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second sec
8A	24"/24	ss	4-13 18-29 N=31	9		B 5.15 BSh	16	vandalia	
9A	24"/24	ss	26-32 46-50 N=78	9		6.59 Sh	10	Dark greenish gray (N4/1), clayey SILT, trace sar gravel	nd and 606
10A	24"/24	ss	21-31 63-71 N=94	11		6.39 Sh	20		
	NOTE(S): N	1W12D i	insta	lled i	in SB-1	2.		



F	[EL]	D 1	BOR	RI	NC	5 L	OG		•		ANSON
WI	CLIEN Sit Location Projec DATE EATHEI	T: A e: C n: C et: 05 s: St S: St Fin R: Fo	EG Coffe CB Mana offeen, Il S3004A cart: 5/10 ish: 5/10 oggy to p	een l agen lino 0/20 0/20 artly	Powe nent 1 is 06 06 y sun	er Stati Facility ny, mil	on 7 d (hi-60's)	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¼" HSA w/SS & CME samplers FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	во	REHOLE II Well II Surface Ele Completion Station	D: SB-12 D: MW12D v: 622 ft. MSL n: 50 ft. BGS n: 875,515.1N 2,515,900.6E
5	SAMPL	Е	Т	EST	FIN	i i	TOPOG	RAPHIC MAP INFORMATION:	WATER LEVEL	INFORMA	TION:
er	/ Total (in) overy		/ 6 in lue	ıre (%)	en. (lb/ft ³)	f) e Type	Quad Town Sectio	rangle: Coffeen, IL ship: East Fork on 11, Tier 7N; Range 3W	$\underline{\Psi} = 12.00 - $ $\underline{\Psi} = 6.76 - $ $\underline{\nabla} = 46.90 - $	While drillin MW12S on 6 MW12D on	g 5/1/06 6/1/06
Numb	Recov % Rec	Type	Blows N - V2 RQD	Moistu	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Detail	Elevation ft. MSL	Remarks
15A	60/60 100%	CS		14			42	Very dark gray (N3/1), clayey SILT, trace sand and g [Continued from previous page]	ravel		
	-	Γ						Very dark gray (N3/1), PEAT		<u> </u>	
	60/60 100%	CS					46	Gray (N5/1) with 30% yellowish brown (10YR5/ mottles, lean CLAY	5)	- 576	
16A				45			50	End of Boring = 50.0 ft. BGS			

FI	EL	D 1	BOR	RI	NG	L	OG			<	Ka h	
	CLIEN Sit Locatio Projec	T: Al ae: CO n: Co at: 05	EG Coffe CB Mana offeen, II S3004A	en F gem linoi	Powe lent H s	r Statio Facility	on ,	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ /4" HSA (blind drill)		BC	OREHOLE ID Well ID Surface Elev	: SB-12a : MW12S : 622 ft. MSL
	DATE	S: St Fin	art: 5/1()/20()/20()6)6			FIELD STAFF: Driller: B. Williamson Helper: P. Keedy			Completion	: 16 ft. BGS
WF	ATHE	R: Fo	ggy to p	artly	suni	ny, mil	d (hi-60's)	Eng/Geo: R. Hasenyager			Station	2,515,900.5E
S	SAMPL	E	Т	EST	ING	r	TOPOG	RAPHIC MAP INFORMATION:				
	l (in)			_	ʻft ³)		Quad	rangle: Coffeen, IL				
	Tota ery		6 in Ie	(%) e	1. (Jb/	Type	Town Sectio	ship: East Fork n 11. Tier 7N: Range 3W				
nber	00 / 1003	е	ws/alt	isture	Den	(tsf) lure	Denth	Lithologic	Bore	hole	Flevation	
INU	Rec % H	Typ	Blo N- RQ	Mo	Dry	Qu Fai	ft. BGS	Description	De	tail	ft. MSL	Remarks
								soil Very dark gray (10YR3/1), clayey SILT, trace sand	$1 \qquad \underbrace{\frac{\sqrt{1}}{\sqrt{1}}}_{\sqrt{1}\sqrt{1}}$		622	
							2	Dark gray (10YR4/1) with 15% yellowish brown (10YR5/6) mottles, lean CLAY				
								Yellowish brown (10YR5/8) with 40% grayish brow (10YR5/2) mottles, lean CLAY	vn		- 620	
							4	Gray (10YR5/1), lean CLAY, trace sand and grave	1		618	
							0	Gray (10YR5/1) with 10% yellowish brown (10YR5 mottles, lean CLAY, trace sand	/6)		616	
							8	Yellowish brown (10YR5/8) with 20% gray (10YR6 mottles, lean CLAY, trace sand and gravel	/1)		614	
								Yellowish brown (10YR 5/8) with 25% gray (10YR6 mottles, clayey SAND, trace gravel	5/1)		612	
							12	Gray (10YR6/1), clayey SAND, trace gravel, wet	///		610	
								Dark yellowish brown (10YR4/6), clayey SAND, tra gravel, wet	ice			
							14	Light yellowish brown (10YR6/4) with 30% browni yellow (10YR6/6) mottles, clayey SILT, trace sand a gravel Dark greenish gray (N4/1), clayey SILT, trace sand a	sh ind ind		608	
								End of Boring = 15.61 ft. BGS				

See SB-12 for sample & testing details

F]	(EL) CLIEN Sit Locatio Projec DATE EATHE	D] T: A ^T te: C on: C on: C ct: 05 S: St Fin R: O	EG Coffe CB Mana offeen, II 5S3004A tart: 5/9 hish: 5/9 vercast, 1	een H agem llinoi /2000 /2000 mild	Powe nent l is 6 6 (mid	Facility	OG on	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ /4" HSA w/SS sampler & 4 ¹ /4" HS FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	SA overdrill	BOREHOLE ID: Well ID: Surface Elev: Completion: Station:	ANSON SB-13 MW13D 623 ft. MSL 55 ft. BGS 874,694.3N 2,513,929.9E
er	/ Total (in)	<u>ac</u>	/6 in alue	ure (%)	en. (lb/ft ³)	e Type	TOPOG Quad Town Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork on 10, Tier 7N; Range 3W	WATER LEV $\underline{\Psi} = 12.4$ $\underline{\Psi} = 8.2$ $\underline{\nabla} = 56.0$	EL INFORMATI 0 - While drilling 4 - MW12S on 6/1 3 - MW13D on 6/1	ON: /06 1/06
Numb	Recov % Rec	Type	Blows N - V: RQD	Moist	Dry D	Qu (ts Failun	Depth ft. BGS	Lithologic Description	Boreho	ble Elevation il ft. MSL	Remarks
1A 1B 1C	22/24 92%	ss	4-5 4-4 N=9	21 15 28		2.13 B	2	soil yish brown (10YR5/2), clayey SILT, trace sar Light gray (10YR7/2), clayey SILT, trace sand Light brownish gray (10YR6/2) with 15% yellowish b (10YR5/6) mottles, lean CLAY, trace sand	d	622	
2A	21/24 88%	ss	3-4 5-8 N=9	25		2.13 BSh	4	Gray (10YR6/1) with 40% yellowish brown (10YR mottles, lean CLAY, trace sand	5/6)	620	
3A 3B	24/24 100%	ss	4-5 6-8 N=11	22 21		2.84 Sh 2.91 BSh	6	Dark gray (10YR4/1), lean CLAY, trace sand and gr	avel	618	
4A	24/24 100%	ss	9-12 10-10 N=22	23		2.33 B	Ā 8 =			616	
5A	24/24 100%	ss	3-4 6-7 N=10	21		2.72 Sh	10	Gray (10YR5/1) with 25% yellowish brown (10YR mottles, lean CLAY, trace sand and gravel	5/6)	614	
6A	19/24 79%	ss	1-3 6-8 N=9	23		1.94 Sh	▼ ¹²			612	
7A 7B	21/24 88%	ss	6-8 10-12 N=18	18		1.94 Sh	14-	Yellowish brown (10YR5/6), silty SAND, trace gra wet	vel,	- 610 	
8A	22/24 92%	ss	7-21 29-30 N=50	11		BSN		Yellowish brown (10YR5/8) with 30% light brown	ish	608	
8B		$\left(\right)$		9			16	gray (10YR6/2) mottles, sandy SILT, trace grave	1		
9A	23/24 96%	ss	25-28 28-45 N=56	9		9.16 Sh	18	Dark gray (10YR4/1), sandy SILT, trace gravel		606	
10A	24/24 100%	ss Ss	18-27 31-36 N=58	8	1	12.00 Sh	20			604	
	NOTE(5): N	1 W 13D 1	msta.	ned 1	ш 3В -1	13.				Page 1 of 3

F]	ELIENT Site Location Projec DATES EATHEI	D] F: A e: C n: C f: 05 S: St Fir R: O E	BOR EG Coffe CB Mana offeen, III 533004A tart: 5/9/ vercast, n T	en F gem linoi 2006 2006 nild EST	Powe nent l is 6 6 (mid	F L (or Static Facility	OG on CONTRACTOR: Testing Service Corporation 7 Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler & 4 ¹ / ₄ " HSA overdrill FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager TOPOGRAPHIC MAP INFORMATION: Quadrangle: Coffeen, IL Township: East Fork V = V r>V = V = V = V = V = V = V = V = V = V = V = V =				BOREHOLE ID: SB-13 Nell ID: MW13D Surface Elev: 623 ft. MSL Completion: 55 ft. BGS Station: 874,694.3N 2,513,929.9E		
ıber	ov / Total (in) ecovery	0	<i>vs / 6 in</i> Value	sture (%)	Den. (lb/ft ³)	tsf) ure Type	Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 10, Tier 7N; Range 3W	$\mathbf{\underline{V}} = 12$ $\mathbf{\underline{V}} = 8$ $\mathbf{\underline{V}} = 50$	2.40 - 3.24 - 5.03 -	While drilling MW12S on 6/1 MW13D on 6/2	/06	
Nun	Recc % R	Type	Blov N - '	Moi	Dry	Qu (Fail	Depth ft. BGS	Lithologic Description		ehole etail	ft. MSL	Remarks	
11A	60/60 100%	CS		13			22				- 602 - 602 		
12A	60/60 100%	cs					26	Dark gray (10YR4/1), sandy SILT, trace gravel [Continued from previous page]					
13A	60/60 100%	CS		15			32				592		
14A	60/60 100%	CS	JW13D :	15 neta1		n SP 1	36 38 38 40	Dark gray (10YR4/1), lean CLAY, trace sand and gr	avel		586		
	TOLE(2	э): N	1 10 10 1	nstäl	neu l	ш Э Б-1						Page 2 of 3	

F	ELI	D]	BOR	RI	NC	F L	OG			C	
	CLIEN Sit	Г: А е: С	EG Coffe CB Mana	een H	Powe lent I	r Statio Facility	on ⁷	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig		BOREHOLE ID:	SB-13
	Location Projec	n: C 1: 0:	offeen, II 5S3004A	linoi	s	uoiiiij		Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler & 4 ¹ / ₄ " HSA	A overdrill	Well ID: 1 Surface Elev:	4W13D 623 ft. MSL
	DATE	S: St Fir	tart: 5/9/	200	5			FIELD STAFF: Driller: B. Williamson Helper: R. Keedy		Completion:	55 ft. BGS 874 694 3N
WI	EATHE	R: O	vercast, n	nild	(mid	-60's)		Eng/Geo: R. Hasenyager			2,513,929.9E
	E E	£	1	ESI		ŕ	TOPOGR Quadr	APHIC MAP INFORMATION: angle: Coffeen, IL	WATER LE $\mathbf{V} = 12$	VEL INFORMATIO .40 - While drilling	DN:
	Total (very		6 in Ie	(%) e	l). (Jb/ff	Type	Towns Section	hip: East Fork 10. Tier 7N: Range 3W	$\overline{\Psi} = 8$ $\overline{\nabla} = 56$.24 - MW12S on 6/1/ .03 - MW13D on 6/1	06 /06
Number	Recov /' % Recov	Type	<i>Blows/(</i> N - Valu RQD	Moisture	Dry Den	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	Bore	chole Elevation tail ft. MSL	Remarks
15A	60/60 100% 60/60 100%	CS CS		15		P F	42 44 44 44 44 44 44 44 44 44 44 44 44 4	Description Dark gray (10YR4/1), lean CLAY, trace sand and gra [Continued from previous page]	avel		
17A 17B 17C	60/60 100%	CS		14 20 14			50	Gray (10YR4/1), silty, fine to medium SAND, we Gray (10YR4/1), sandy SILT		572	
17D				22			54-	Dark greenish gray (5GY4/1) with 25% yellowish bro (10YR5/6) mottles, lean CLAY End of Boring = 55.0 ft. BGS	own	568	
	NOTE(S	S): N	/W13D i	nsta	lled i	n SB-1	13.				

F	EL	DI	BOR	RI	NC	5 L	OG		
	CLIEN Si Locatio Proje DATE	T: A te: C on: C ct: 05 CS: St	EG Coffe CB Mana offeen, III S3004A art: 5/9/	en F gem linoi 2000	Powe nent l is 6	er Stati Facility	on V	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA (blind drill) FIELD STAFF: Driller: B. Williamson	BOREHOLE ID: SB-13a Well ID: MW13S Surface Elev: 623 ft. MSL Completion: 17 ft. BGS
wi	EATHE	Fin R: O	ish: 5/9/ vercast, n	200 nild	6 (mić	l-60's)		Helper: R. Keedy Eng/Geo: R. Hasenvager	Station: 874,695.7N 2.513,925.3E
	SAMPI	E.	Т	EST	TINC	; ;	TOPOG	RAPHIC MAP INFORMATION.	WATER LEVEL INFORMATION.
er	/Total (in) :overy		∕6 in alue	ure (%)	en. (lb/ft ³)	f) e Type	Quad Towr Sectio	rangle: Coffeen, IL ship: East Fork on 10, Tier 7N; Range 3W	$\Psi = 12.00 - While drilling$ $\Psi = 6.76 - MW13S \text{ on } 6/1/06$ $\Psi = 46.90 - MW12D \text{ on } 6/1/06$
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
		•						soil yish brown (10YR5/2), clayey SILT, trace sand	
								Light gray (10YR7/2), clayey SILT, trace sand	
							2	Light brownish gray (10YR6/2) with 15% yellowish bro (10YR5/6) mottles, lean CLAY, trace sand	own
							4	Gray (10YR6/1) with 40% yellowish brown (10YR5/6 mottles, lean CLAY, trace sand	6)
								Dark gray (10YR4/1), lean CLAY, trace sand and grav	vel
							6 ▼ 10 10	Gray (10YR5/1) with 25% yellowish brown (10YR5/ mottles, lean CLAY, trace sand and gravel	(6) -614 -612
							14	Yellowish brown (10YR5/6), silty SAND, trace grave wet	el,
								Yellowish brown (10YR5/8) with 30% light brownish gray (10YR6/2) mottles, sandy SILT, trace gravel	h
								Dark gray (10YR4/1), sandy SILT, trace gravel	
	I		ı		I	I	. =	End of Boring = 16.62 ft. BGS See SB-13 for sample & testing details	

NOTE(S): MW13S installed in blind-drilled borehole within 10 ft of SB-13.

F	EL	D]	BOF	RII	NC	F L	OG		C		ANSON			
	CLIEN Sit Locatio	T: A te: C on: C on: C	EG Coff CB Mana offeen, Il 5S3004A	een I agen linoi	Powe nent I is	er Statio Facility	on CONTRACTOR: Testing Service Corporation y Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler			BOREHOLE ID: SB-14 Well ID: n/a				
WI	DATE	S: Si Fir R: Su	art: 5/1 ish: 5/2 inny, mil	/200 /200 d (n	6 6 nid-6	0's)		FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager		Surface Elev: Completion: Station:				
5	SAMPL	.Е	Т	ES1		.	TOPOGE	RAPHIC MAP INFORMATION:	WATER LEVEL I	INFORMATIO	DN:			
	otal (i ry		in	(%)	(lb/ft ³	əde	Towns	ship: East Fork	$\underline{\Psi} = 14.00 - N$ $\underline{\Psi} = 4.49 - N$	MW14S on 6/1/0)6			
lber	ov / To	0	$\sqrt{3}/6$	sture (Den.	tsf) ure $T_{\mathcal{I}}$	Section	n 10, Tier 7N; Range 3W						
Num	Reco % R	Type	Blov N - J	Moi	Dry	Qu (Fail	Depth ft. BGS	Lithologic Description	Borehole Detail	ft. MSL	Remarks			
1A	23/24 96%	ss	2-3 2-3 N-5	16				Grayish brown (10YR5/2), clayey SILT, trace sa	ind	624				
1B		Δ	11-5	26		2.33	2	Gray (10YR6/1) with 30% yellowish brown (10Y) mottles, lean CLAY, trace sand	R6/8)					
2A	24/24 100%	ss	3-4 5-7 N=9	23		2.35 B 3.10		Light gray (10YR7/1) with 30% yellowish brow (10YR6/8) mottles, lean CLAY, trace sand	/n					
		$\left(\right)$				В	4							
3A	23/24 96%	ss	3-3 5-5 N=8	19		2.33 B	<u>₹</u>							
4A	24/24 100%	ss	5-6 5-7 N=11	23		2.68 BSh	8	Light gray (10YR7/1) with 15% yellowish brow (10YR6/8) mottles, lean CLAY, trace sand	/n	618				
5A	24/24 100%	ss	2-2 3-4 N=5	26		1.83 B	10			616				
6A	19/24 79%	ss	2-2 3-5 N=5	17		2.18 B	12-	Yellowish brown (10YR5/8) with 50% light gr (10YR7/1) mottles, sandy CLAY	IY	614				
7A	20/24 83%	ss	2-3 3-3 N=6	22					1.16 B				612	
8A		\forall	- 14	16		1.36	I4 <u>−</u>	Yellowish brown (10YR5/6), silty, fine SAND, t medium sand and gravel, wet	ace	610				
	24/24 100%	ss	5-14 14-20 N=28			В		Yellowish brown (10YR5/6), sandy SILT, trace g	ravel					
8B		()		11		5.77 RSh	16							
9A	12/24 50%	ss	57-65	10		2011		Yellowsh brown (10YR5/6) with 40% gray (10YI mottles, sandy SILT, trace gravel	26/1)					
10A	24/24 100%	ss	6-8 16-18 N=24	12		5.04 BSh		Dark gray (10YR4/1), clayey SILT, trace sand and	gravel					
	NOTE((S): E	orehole	aban	done	ed usin	g bentonite	grout pumped from bottom of borehole.						

WE	Site Site Location Projec DATES	e: C0 n: C0 t: 05 S: St Fin R: Su	CB Mana offeen, Il S3004A art: 5/1/ ish: 5/2/ inny, mil	2000/2000/2000/2000/2000/2000/2000/200	Powe nent I is 6 6 6 nid-6	o's)	n CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ /4" HSA w/SS sampler FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager					BOREHOLE ID: SB-14 Well ID: n/a Surface Elev: 625 ft. MSI Completion: 60 ft. BGS Station: 875,740.0 2,514,130.0		
ber	OV / Total (in)		/ 6 in Jue	ture (%)	Den. (lb/ft ³)	sf) re Type	TOPOGR Quadra Townsł Section	APHIC MAP INFORMATION: ingle: Coffeen, IL iip: East Fork 10, Tier 7N; Range 3W	W	/ATER ⊻ = ⊻ = ∑ =	LEVE 14.00 4.49	L INFORMATI - While drilling - MW14S on 6/1	ON: /06	
[unn]	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description		17	Borehole Detail	e Elevation ft. MSL	Remarks	
11A	24/24 100%	ss	2-7 13-30 N=20	13		9.70 B	22					604		
12A	24/24 100%	ss	26-40 36-40 N=76	9		13.09 BSP	24					602		
13A	24/24 100%	ss	8-18 28-34 N=46	9		8.73 BSP	26-					- 600		
14A	22/24 92%	ss	20-18 24-30 N=42	9		7.42 BSP	28	Dark gray (10YR4/1), clayey SILT, trace sand and g [Continued from previous page]	grave	el (
15A	19/24 79%	ss	8-27 33-67 N=60	9			30							
16A	24/24 100%	ss	8-25 27-33 N=52	10		9.60 BSh	32					594		
17A	20/24 83%	ss	11-15 20-24 N=35	14		6.80 B	34					- 592		
18A	24/24 100%	ss	3-4 7-9 N=11	16		3.88 B	36					- 590		
19A	24/24 100%	ss	8-12 13-15 N=25	16		6.18 B	38	Dark gray (N4/1), lean CLAY, trace sand and gra	avel			588		
20A	24/24 100%	ss	3-7 10-13 N=17	14		3.10 B						- 586		

F]		D]	BOF	NI		F L	OG		Generation Hanson
	CLIEN Sit Locatio Projec DATE	T: Al n: Co n: Co ct: 05 S: St Fin	EG Coff CB Man offeen, II 5S3004A cart: 5/1 hish: 5/2	een I agen llinoi /2000	Powe nent l is 6	er Statio	on V	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler FIELD STAFF: Driller: B. Williamson Helper: R. Keedy	BOREHOLE ID: SB-14 Well ID: n/a Surface Elev: 625 ft. MSL Completion: 60 ft. BGS Station: 875,740.0N
WI	WEATHER: Sunny, mild (mid-60's)							Eng/Geo: R. Hasenyager	2,514,130.0E
nber	xov / Total (in)	<u>E</u>	<i>ws / 6 in</i> Value D	isture (%)	/ Den. (lb/ft ³)	(tsf) lure Type	TOPOGR Quadra Towns Section	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork 10, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 14.00$ - While drilling $\Psi = 4.49$ - MW14S on 6/1/06 $\overline{\Sigma} =$ Borehole Elevation
INN	Rec % H	Tyr	$\frac{Blo}{N}$ -	Mo	Dr)	Qu Fai	ft. BGS	Description	Detail ft. MSL Remarks
21A	23/24 96%	ss	3-6 8-13 N=14	15		4.80 B	42		- 584
22A	24/24 100%	ss	13-15 16-18 N=31	14		5.62 B	44	Dark gray (N4/1), lean CLAY, trace sand and grav	vel
23A	24/24 100%	ss	4-8 11-13 N=19	15		4.65 B	46	[Continuea from previous page]	
24A	24/24 100%	ss	18-18 20-20 N=38	15		4.65 <i>B</i>	48-		
25A	24/24 100%	ss	4-7 9-11 N=16	19		2.13 BSh		Dark gray (N4/1), clayey SILT, trace sand and gra Gray (N4/1), wet, loose, fine- to medium-grained SA	vel
26A	22/24 92%	ss	3-5 6-8 N=11	22		3.30 BSh	52	Greenish gray (5BG5/1), lean CLAY	-574
27A	24/24 100%	ss	3-5 5-7 N=10	25		2.89 BSh	54	Greenish gray (5BG5/1) with 15% yellowish brow (10YR5/6) mottles, lean CLAY	vn
28A	21/24 88% 0/24 0%	SS SH	4-6 7-8 N=13	22		3.71 BSh	56	Greenish gray (5BG5/1) with 25% yellowish brow (10YR5/6) mottles, lean CLAY	vn
29A	14/24 58%	ss	0-0 0-0 N=0	22		3.09 BSh	58	Greenish gray (5BG5/1) with 50% yellowish brow (10YR5/6) mottles, lean CLAY	vn
30A	22/24 92%	ss	5-6 8-12 N=14	19		4.46 BSh	60	Yellowish brown (10YR4/6) with 10% greenish g (5BG5/1) mottles, lean CLAY End of Boring = 60.0 ft. BGS	ray 566
	NOTE(S): B	orehole	aban	done	ed usin	g bentonite	grout pumped from bottom of borehole.	



F	FIELD BORING LOG COR HANSON												
	CLIEN Sit Locatio Projec	T: A te: C n: C ct: 05	EG Coff CB Mana offeen, Il 5S3004A	een H agem Ilinoi	Powe nent l is	er Statio Facility	on CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ /4" HSA w/SS sampler	F	ID: SB-15 ID: n/a ev: 624 ft. MSL				
WF	DATE	S: St Fin R: O	art: 4/2 iish: 4/2 vercast, o	4/20 5/20 cool	06 06 (lo-5	60's)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager		Completion: 84 ft. B0 Station: 875,970 2,515,080				
5	SAMPL	E	T	EST	rine I	Ĵ	TOPOGRAPHIC MAP INFORMATION: WAT	ER LEVI	EL INFORMA	ATION:			
	al (in)				0/ft ³)	0	Quadrangle: Coffeen, IL	= 13.40	40 - While drilling				
н	/ Tot: overy		/ 6 in lue	re (%	en. (It	Typ	Section 11, Tier 7N; Range 3W	L = 5.24	- MW15D on	n 6/1/06			
Numbe	Recov % Rec	Type	Blows N - Va RQD	Moistu	Dry Do	Qu (tsf Failur	Depth Lithologic ft. BGS Description	Boreho Detail	le Elevation ft. MSL	Remarks			
		\bigvee	2.2				soilDark brown (10YR3/3), clayey SILT						
1A	24/24 100%	ss	2-2 3-4 N=5	19			Dark grayish brown (10YR3/2), clayey SILT, trace sand						
1B		$\left(\right)$		27		1.94 B	2 Grayish brown (10YR5/2), with 50% very dark gray (10YR3/1) mottles, lean CLAY, trace sand		622				
2A	24/24 100%	ss	2-2 4-6 N=6	25		3.10	Gray (10YR6/1), lean CLAY, trace sand						
		$\left(\right)$				В	4 10ess		620				
	20/24 83%	ss	2-3 3-5 N=6				Yellowish brown (10YR5/6) with 40% gray (10YR6/1) mottles, lean CLAY, trace sand						
3A		Д	N=0	29		2.10 B	6-		618				
	24/24	V ss	4-6 5-5				Gray (10YR6/1), lean CLAY, trace sand						
4A	10070	\square	N=11	24		1.75 B	8		616				
	22/24	V	1-2										
5A	92%	\bigwedge 33	N=5	26		1.55 B			614				
	22/24	$\overline{\mathbf{V}}$	2-3				Gray (10YR6/1) with 50% yellowish brown (10YR5/6)						
6A	92%	ss .	3-4 N=6	22		1.85 B	inotties, real CLA 1, intre sand, trace graver						
	19/24 79%	SH	1 1				12		612	Shelby tube taken from shallow well borehole at			
7A	24/24 100%	ss 🕺	5-5 N=9	23		1.22	\mathbf{Y}			indicated depth.			
7B		$\left(\right)$		17		В	14		610				
84	21/24 88%	ss	2-6 15-19 N=21	11		3.22	Pale brown (10YR6/3), clayey SILT, little sand, trace gravel						
071		$\left(\right)$				BSP	16 Yellowish brown (10YR5/4), silty, fine to coarse SAND,		608				
9A	24/24	V ss	18-29 40-50	20			Yellowish brown (10YR5/4) silty fine SAND wet						
9B 9C	100%	\int	N=69	21			Gray (10YR6/1), sandy SILT, trace medium to coarse sand and trace gravel		 606				
		\bigvee					vandalia						
10.4	17/24 71%	ss	11-43 59/5"	_		7 40	Dark gray (10YR4/1), clayey SILT, little sand, trace gravel						
10A			ough 1	/	dei	$B^{7.42}$			604				
	NOTE(э): В	orenole	avan	uone	u usin	g bentomte grout pumpet from bottom of borenole.						

FIELD BORING LOG										CR HANSON																	
	CLIEN Sit Locatio Project	T: A ce: C n: C n: C	EG Coffe CB Mana offeen, Il	een H agen linoi	Powe nent I is	er Statio Facility	on 7	 CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3¹/₄" HSA w/SS sampler 				BOREHOLE ID: SB-15 Well ID: n/a Surface Elev: 624															
WI	DATE	S: Si Fir R: O	tart: 4/24 nish: 4/22 vercast, c	4/20 5/20 cool	06 06 (10-5	0's)	FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager					84 ft. BGS 875,970.0N 2,515,080.0E															
5	SAMPLE TESTING							APHIC MAP INFORMATION:	WATI	ERI	LEVEL	INFORMATI	ON:														
	y y		u	(%	[] [] [] [] [] [] [] [] [] [] [] [] [] [ьe	Quadra Townsl	ingle: Coffeen, IL ip: East Fork	₹ Ţ	=	13.40 - 4.99 -	MW15S on 6/1	/1/06 /1/06														
ber	v / Tc		s/6i /alue	ture (Den. (sf) ure Ty	Section	11, Tier 7N; Range 3W	Į	=	5.24 -	MW15D on 6/	1/06														
Num	Recc % Re	Type	Blow N - N	Mois	Dry	Qu (Failt	Depth ft. BGS	Lithologic Description		B	orehole Detail	Elevation ft. MSL	Remarks														
11A	14/24 58%	ss	14-55 45/2"	8			22 -	Dark gray (10YR4/1), clayey SILT, little sand, trace gravel				602															
12A	8/24 33%	ss	100/8"	8		6.76 SP	24					600															
13A	23/24 96%	ss	12-28 43-57/5' N=71	5			26					- 598															
14A	8/24 <i>33%</i>	ss	59-41/2'	6		7.95 BSh	28		ce																		
15A	16/24 67%	ss	11-26 74/4"	12		4.74 BSh	30	[Continued from previous page]				 594															
16A	12/24 50%	ss	39-61	7			32-																				
17A	10/24 42%	ss	49-51/4"	9		5.43 B	34					590															
18A	11/24 46%	ss	100-95	11																	36					588	
19A	8/24 <i>33%</i>	ss	61-39/2'	10			38	Dark gray (10YR4/1), silty, fine to medium SAND, t coarse sand and gravel, wet	race			586															
20A	24/24 100%	ss	21-41 21-24 N=62	12		16.00 None		Very dark gray (10YR3/1), clayey SILT, little sand, t gravel Very dark gray (10YR3/1) with 20% dark gravish br	race																		
20B	NOTE	∐ €)• ₽	orehola	13	 den:	9.38	$ _{40} \equiv$	(10YR4/2) mottles, clayey SILT, trace sand and gra	vel		<u>[</u>	584															
	NOTE(5); E	orenote	avan	uone	a using	s venionite g	rou pumped from bottom of borenole.																			


F	[EL]	D]	BOF	RI	NC	5 L	OG				S?H	ANSON	
	CLIEN Sit Location Projec	T: A e: C n: C n: C	EG Coff CB Mana offeen, II 5S3004A	een I agen llinoi	Powe nent l is	er Statio Facility	ty Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS sampler				BOREHOLE ID: SB-15 Well ID: n/a Surface Elev: 624 ft. M		
WI	DATE; EATHEI	S: St Fir R: O	tart: 4/2 nish: 4/2 vercast, o	4/20 5/20 cool	06 06 (10-5	60's)		FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager			Completion: 84 ft. Station: 875,9 2,515,0		
5	SAMPL	E	Т	EST	ΓINC	3	TOPOGR	APHIC MAP INFORMATION:	WA	TER LEVEL	INFORMATIO	DN:	
Ŀ.	/ Total (in) wery		6 in lue	re (%)	n. (lb/ft ³)) Type	Quadr Towns Sectior	angle: Coffeen, IL hip: East Fork 11, Tier 7N; Range 3W		$\underline{\Psi} = 13.40 - $ $\underline{\Psi} = 4.99 - $ $\underline{\nabla} = 5.24 - $	While drilling MW15S on 6/1/ MW15D on 6/1/	06 ⁄06	
Numbe	Recov % Reco	Type	Blows / N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	•	Borehole Detail	Elevation ft. MSL	Remarks	
31A	24/24 100%	ss	4-7 13-15 N=20	18		5.42 BSh	62	Olive (5Y4/3) with 15% greenish gray (10GY5/1) n	nottles,		562		
32A	24/24 100%	ss	10-15 11-16 N=26	20		4.74 BSh	64	lean CLAY, trace sand and gravel [Continued from previous page]			560		
33A	24/24 100%	ss	6-10 11-13 N=21	16		6.98 BSh	66	Greenish gray (10Y5), lean CLAY, trace sand and	gravel		558		
34A	24/24 100%	ss	11-14 18-31 N=32	18		6.98 BSh	68 -				556		
35A	23/24 96%	ss	9-18 27-40 N=45	15		11.95 BSh	70-				- 554		
36A	24/24 100%	ss	4-12 18-24 N=30	16		7.15 BSh	72 -				552		
37A	24/24 100%	ss	17-29 36-47 N=65	17		8.24 <i>BSh</i>	74	Dark yellowish brown (10YR4/4), lean CLAY, trac and gravel	ce sand		- 550		
38A	20/24 83%	ss	12-18 23-28 N=41	17		6.59 BSh	76-				- 548		
39A	9/24 38%	ss	29-39 48-66 N=87	16			78-				546		
40A	24/24 100%	ss	5-9 13-18 N=22	18		6.21 <i>B</i>	80				544		
	NOTE(S): E	Sorehole	aban	idone	ed usin	g bentonite	grout pumped from bottom of borehole.					

FIELD BC	RI	NC	5 L	OG		C	NSON		
CLIENT: AEG (offeen	Powe	er Statio	on	CONTRACTOR: Testing Service Corporation				
Site: CCB M	anagen	nent]	Facility	r	Rig mfg/model: CME-650 Track Rig	BOREHOLE ID: SB-15			
Location: Coffee	, Illino	is			Drilling Method: 31/4" HSA w/SS sampler	Well ID: n/a			
Project: 05S30	4A					Surface Elev:	624 ft. MSL		
DATES: Start:	/24/20	06			FIELD STAFF: Driller: B. Williamson	Completion:	84 ft. BGS		
Finish:	4/25/20	006			Helper: R. Keedy	Station:	875,970.0N		
WEATHER: Overca	t, cool	(10-5	0's)		Eng/Geo: R. Hasenyager		2,515,080.0E		
SAMPLE	TES	FIN (ť	TOPOGRA	PHIC MAP INFORMATION:	WATER I EVEL INFORMATION	1.		
(ni))		Ouadrar	Ougdrandle: Coffeen II $\nabla = 13.40$. While drilling				
al),ft	в	Townshi	n: East Fork	$\nabla = 4.99 - MW15S \text{ on } 6/1/06$			
Tot very 6 in	8		T_{yp}	Section 1	1. Tier 7N: Range 3W	$\overline{\nabla} = 5.24 - MW15D \text{ on } 6/1/06$	5		
ber vv / scor		Der	tsf) tre		, ,	_			
lum % Recc Уре	<u>ð</u> ig	Ly.	ou (ailt	Depth ft BCS	Lithologic	Borehole Elevation	Pamarks		
			94	II. DOS	Description		Remarks		
41A 24/24 100% ss 6. 13. N=	8 6 21 17		5.82 B	82	Dark yellowish brown (10YR4/4), lean CLAY, trace	e sand			
$42A \begin{vmatrix} 24/24 \\ 100\% \end{vmatrix} ss \begin{vmatrix} 18 \\ 25 \\ N = 100\% \end{vmatrix}$	28 25 ³ 18		5.82 BSh	84	End of Boring = 84.0 ft BCS				

F	EL	D]	BOR	RI	NC	L	OG		
	CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A						on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill)	BOREHOLE ID: SB-15a Well ID: MW15S Surface Elev: 624 ft. MSL
	DATI	ES: St Fir	art: 4/25 hish: 4/25	5/20 5/20	06 06			FIELD STAFF: Driller: B. Williamson Helper: R. Keedy	Completion: 20 ft. BGS Station: 875.971.1N
WI	EATHF	E R: O	vercast, c	cool	(lo-5	0's)		Eng/Geo: R. Hasenyager	2,515,076.3E
5	SAMPI		Т	EST	TING	}	TOPOGE	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	al (in			()	b/ft ³)	0	Quadr Towns	angle: Coffeen, IL hin: East Fork	$\Psi = 13.40$ - While drilling $\Psi = 4.99$ - MW15S on 6/1/06
r	/ Tot overy		/6 in lue	re (%	en. (Il	Typ	Section	n 11, Tier 7N; Range 3W	$\underline{\nabla}$ = 5.24 - MW15D on 6/1/06
Numbe	Recov % Reco	Type	Blows , N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
								soil <u>Dark brown (10YR3/3), clayey SILT</u>	
								Dark grayish brown (10YR3/2), clayey SILT, trace	sand
							2	Grayish brown (10YR5/2), with 50% very dark gr (10YR3/1) mottles, lean CLAY, trace sand	ay 622
								Gray (10YR6/1), lean CLAY, trace sand	620
							4	·····	620
							Ϋ́	Yellowish brown (10YR5/6) with 40% gray (10YR mottles, lean CLAY, trace sand	6/1)
							6		
								Gray (10YR6/1), lean CLAY, trace sand	
							8 10 12	Gray (10YR6/1) with 50% yellowish brown (10YR mottles, lean CLAY, little sand, trace gravel	5/6)
							14	Gray (10YR6/1) with 30% yellowish brown (10YR mottles, silty, fine to medium SAND, trace coarses and gravel, wet	5/6) sand 610
								Pale brown (10YR6/3), clayey SILT, little sand, tr gravel	
							16	Yellowish brown (10YR5/4), silty, fine to coarse SA wet	AND,
								Yellowish brown (10YR5/4) silty fine SAND, w	et $\left \begin{array}{c} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $
								Gray (10YR6/1), sandy SILT, trace medium to coa sand and trace gravel	urse 606
								Dark gray (10YR4/1), clayey SILT, little sand, tra gravel See SB-15 for sample & testing details	ice
			 			1.1.	। <u> </u>	End of Boring = 19.62 ft. BGS	



F]	EL	D]	BOR	RII	NC	5 L	OG		C HANSON			
CLIENT: AEG Coffeen Power Stati Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A						er Statio Facility	on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill)	BOREHOLE ID: SB-15b Well ID: MW15D Surface Elev: 624 ft. MSL			
w	DATI EATHE	ES: St Fin ER: O	art: 4/24 hish: 4/25 vercast, c	4/20 5/20 cool	06 06 (10-5	0's)		FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager	Completion: 39 ft. BGS Station: 875,970.5N 2,515,080.7E			
;	SAMPI	LE	Т	EST		÷	TOPOGRA Quadra	APHIC MAP INFORMATION: ngle: Coffeen, IL	WATER LEVEL INFORMATION: $\underline{\Psi} = 13.40$ - While drilling			
ıber	ov / Total ecovery	0	<i>vs / 6 in</i> Value)	sture (%)	Den. (lb/	tsf) ure Type	Townsh Section	ip: East Fork 11, Tier 7N; Range 3W	Ψ = 4.99 - MW15S on 6/1/06 $\overline{\nabla}$ = 5.24 - MW15D on 6/1/06			
Nun	Reco % R	Type	Blov N - J	Moi	Dry	Qu (Fail	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
								Dark gray (10YR4/1), clayey SILT, little sand, trace gravel [Continued from previous page]	e - 602 - 600 - 598 - 598 - 594 - 592 - 592 - 590 - 588			
							20	Dark gray (10YR4/1), silty, fine to medium SAND, tr coarse sand and gravel, wet	ace			
							30	Very dark gray (10YR3/1), clayey SILT, little sand, tr gravel End of Boring = 38.80 ft. BGS				
								See SB-15 for sample & testing details				
	NOTE	(S): N	1W15D i	nsta	lled i	n blinc	l-drilled bore	hole within 10 ft of SB-15.	D			

L

CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Offeen, Illinois Project: 0553004A DATES: Start: 4/21/2006CONTRACTOR: Reynolds Drilling Corp. Rig mf/model: CME-830 Track Rig Drilling Method: 4/4" HSA w/SS & CME samplersBOREHU Location: Coffeen, Illinois Surfa DATES: Start: 4/21/2006FIELD STAFF: Driller: K. Doetzel Com Finist: 4252/2006WEATHER: Overcast, cool (mid-40)s)FIELD STAFF: Driller: K. Doetzel Com Tomsship: East ForkOPOGRAPHIC MAP DNFORMATION: Quadrangle: Coffeen, IL Tomsship: East Fork Section 2. Tier 7N: Range 3WWATER LEVEL INFO Image Section 2. Tier 7N: Range 3W1A $21.04 \\ 40 \\ 2024 \\ 400 \\ 50 \\ 2024 \\ 40 \\ 40 \\ 40 \\ 40 \\ 40 \\ 40 \\$	FIELD BORING LOG												
Drinner of the with the with the with the coll can be an plot if the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the period of the pe	OLE ID: SB-16												
SAMPLETESTINGTOPOGRAPHIC MAP INFORMATION: Quadrangle: Coffeen, IL Township: East Fork Section 2, Tier 7N: Range 3WWATER LEVEL INFO $I = 12.80$. While $I = 2.80$. While $I = 5.74$ - MWI($I = 65$ 1A 21/24 1B21/24 SS 4.4 4.7 22 $I = 5.74$ 22 $I = 5.74$ 100°F $I = 12.80$ Borehole Elev $I = 0.90$ 1A 24/24 2A21/24 $I = 5.74$ 22 $I = 5.74$ 22 $I = 12.83$ B 21/100°F $I = 12.80$ Borehole Elev $I = 0.90$ 1A 24/24 2A21/24 $I = 5.74$ 22 $I = 12.83$ B 22 $I = 2.13$ B 21/100°F $I = 12.80$ Brown (107R2/1), sl. moist, firm, silty CLAY with trace sand.24/24 $I = 0.96$ 23.4 $I = 5.75$ 2.13 $I = 12.83$ $I = 13.86$ 4 $I = 0.96$ Gray (107R5/1) with 15% dark yellowish brown (107R4/6) mottles, moist, very soft, very silty CLAY with trace sand.4A $I = 0.96$ 3.4 $I = 0.96$ 2.13 $I = 0.96$ 4 $I = 0.96$ 62 $I = 0.96$ 24/24 $I = 0.96$ 2.33 $I = 0.96$ 3.4 $I = 0.96$ 2.33 $I = 0.96$ 4 $I = 0.96$ 24/24 $I = 0.96$ 3.4 $I = 0.96$ 2.13 $I = 0.96$ 4 $I = 0.96$ 4 $I = 0.96$ 24/24 $I = 0.96$ 3.4 $I = 0.96$ <th>ace Elev: 626 ft. MSL npletion: 92 ft. BGS Station: 877,355.0N 2,515,080.0E</th>	ace Elev: 626 ft. MSL npletion: 92 ft. BGS Station: 877,355.0N 2,515,080.0E												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	DRMATION:												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	6S on 6/1/06												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	6D on 6/1/06												
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	vation MSL Remarks												
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	120												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	524												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $													
$4A \begin{bmatrix} 24/24\\ 100\% \end{bmatrix} \text{ ss } \begin{bmatrix} 2-3\\ 4-6\\ N=7 \end{bmatrix} 25 \begin{bmatrix} 2.13\\ B\\ 8 \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 15\% \text{ dark yellowish brown} \\ (10YR4/6) \text{ mottles, moist, very soft, very silty CLAY with } \\ \text{trace sand.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR4/6) \text{ mottles, moist, very soft, very silty CLAY with } \\ \text{trace sand.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 15\% \text{ dark yellowish brown} \\ (10YR4/6) \text{ mottles, moist, very soft, very silty CLAY with } \\ \text{trace sand.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{mottles, moist, firm, silty CLAY with some sand and trace } \\ \text{gravel.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{mottles, moist, firm, silty CLAY with some sand and trace } \\ \text{gravel.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{mottles, moist, firm, silty CLAY with some sand and trace } \\ \text{gravel.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{mottles, moist, firm, silty CLAY with some sand and trace } \\ \text{gravel.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel.} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown} \\ \text{for a gravel} \end{bmatrix} \end{bmatrix} \end{bmatrix} \begin{bmatrix} 6\\ Gray (10YR$	522												
$\begin{bmatrix} 24/24\\ 100\% \end{bmatrix}$ ss $\begin{bmatrix} 3-4\\ 5-5\\ N=9 \end{bmatrix}$ 24 $\begin{bmatrix} 2.33\\ B\\ 10 \end{bmatrix}$ $\begin{bmatrix} Gray (10YR5/1) \text{ with } 10\% \text{ yellowish brown } (10YR5/8) \\ mottles, moist, firm, silty CLAY with some sand and trace gravel. \\ \end{bmatrix}$	20												
gravel.	518												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Shelby tube taken from shallow well borehole at												
$\begin{bmatrix} 7A \\ 24/24 \\ 24/24 \end{bmatrix} \xrightarrow{22} \begin{bmatrix} 1.94 \\ BSh \\ \blacksquare \end{bmatrix} \xrightarrow{12} = 61$	indicated depth.												
7B 100% 55 1-7 N=14 18 18 14 Dark yellowish brown (10YR4/6), wet, sl. dense, silty, very fine- to fine-grained SAND.	12												
21/24 ss 1-2 88/4 ss 2-4 N=4 20 CLAY with sand and trace gravel. Dark yellowish brown (10YR4/6), wet, loose, silty, very fine- to fine-grained SAND. Dark yellowish brown (10YR4/6), wet, soft, silty CLAY	012												
with sand and trace gravel. 16 Yellowish brown (10YR5/6), wet, loose, very fine- to very Coarse-orained SAND	510												
$\begin{array}{c c c c c c c c c c c c c c c c c c c $													
9B 15 18 Gray (10YR5/1), moist, hard, clayey SILT with sand and trace gravel. 60	508												
$10B \begin{vmatrix} 20/24 \\ 83\% \end{vmatrix} \times \begin{vmatrix} 27.54 \\ 59.59 \\ N=113 \end{vmatrix} 17 \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad $													
Image: Construction of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second													

F		D]	BOF	RII	N(G L	OG			(ANSON	
	CLIEN Sit Locatio	T: A ce: C n: C	EG Coff CB Mana offeen, II	een I agen llinoi	Powe nent 1 is	er Stati Facility	on V	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers	BOREHOLE ID: SB-16 Well ID: n/a				
wi	DATES: Start: 4/21/2006 Finish: 4/25/2006 WEATHER: Overcast, cool (mid-40's)							FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	Completion: 92 ft. BGS Station: 877,355.0 2,515,080.0			926 ft. BGS 91 ft. BGS 92 ft. BGS 92 ft. BGS 92 ft. BGS 92 ft. BGS 92 ft. BGS 92 ft. BGS	
r.	/Total (in) WE	E	/ 6 in lue	res (%) ar	an. (lb/ft ³) N	Type	TOPOGR Quadr Towns Section	APHIC MAP INFORMATION: angle: Coffeen, IL nip: East Fork 2, Tier 7N; Range 3W	WATER <u>▼</u> = <u>▼</u> = <u>▼</u> =	LEVEL 12.80 - 5.74 - 51.37 -	INFORMA While drillin MW16S on MW16D on	ng 6/1/06 6/1/06	
Numbe	Recov % Reco	Type	Blows , N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description]	Borehole Detail	Elevation ft. MSL	Remarks	
11A	10/24 42%	ss	10-96	8			22				606 		
12A	14/24 58%	ss	84-132	10		3.10 BSh	24				602		
13A	20/24 83%	ss	41-68 82 N=150	10		7.56 B	26						
14A	12/24 50%	ss	58-119	10		9.89 B	20				600	Dusky red (7.5YR3/4) staining.	
15A	24/24 100%	ss	30-48 70-71 N=118	9		5.62 B	30	Gray (10YR4/1), moist, very hard, clayey, sandy SI	LT				
16A	24/24 100%	ss	50-54 68-93 N=122	9			32	[Continued from previous page]					
17A	35/36 97%	CS		17			34				592		
18A	60/60 100%	cs		10			36						
	NOTE	S): F	 Borehole	 aban	 Idone	ed usin	$\begin{vmatrix} \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\$	rout pumped from bottom of borehole.				Wood fragments.	
										_		Page 2 of 5	



F	[EL]	D]	BOF	RI	NC	5 L	OG			6		HANSON
	CLIEN Sit Locatio	T: A e: C n: C	EG Coff CB Mana offeen, II	een F agem Ilinoi	Powe nent] is	er Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers	8	во	REHOLE I Well I	D: SB-16 D: n/a
wi	DATE	S: St Fir R: O	tart: 4/2 nish: 4/2 vercast, o	1/200 5/200 cool	06 06 (mid	-40's)		FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager			Completio Statio	on: 92 ft. BGS on: 877,355.0N 2,515,080.0E
	Total (in)	E	6 in ue	(%) a	n. (lb/ft ³) NI	Type	TOPOGR Quadr Towns Sectior	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork .2, Tier 7N; Range 3W	WAT	FER LEVEL $\underline{\Psi} = 12.80 - $ $\underline{\Psi} = 5.74 - $ $\underline{\nabla} = 51.37 - $	INFORMA While drilli MW16S on MW16D on	ATION: ng 6/1/06 6/1/06
Number	Recov / % Reco	Type	Blows/ N - Valı RQD	Moistur	Dry Dei	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	1	Borehole Detail	Elevation ft. MSL	Remarks
	0/48 <i>0%</i>	RC					62	Greenish gray (5G6/1) with 40% yellowish brov (10YR5/6) mottles, moist, hard, silty CLAY with sl sand. [Continued from previous page]	vn . trace		566	
26A	24/24 100%	ss	32-34 42-51 N=76	25		2.72 BSh	64				- 562	
	0/24 <i>0%</i>	RC					66	Yellowish brown (10YR5/6) with 20% greenish g (5G6/1) mottles, moist, hard, silty CLAY with trace and trace coal fragments.	gray e sand		560	
	24/24 100%	ss	15-21 21-21 N=42				68 -				558	
28A		$\overline{\mathbf{V}}$	14-17	18		2.72 BSh						
29A	100%	ss	21-25 N=38	20		2.91 BSh	70				556	70' to 79.5' - possible oxidation rinds.
30A	24/24 100%	ss	12-21 34-35 N=55	18		5.04 BSh	72			د از از از از از از از از از از از از از از از از از از از از از از	554	
31A	24/24 100%	ss	16-21 27-35 N=48	16		8.15 <i>BSh</i>	74	Yellowish brown (10YR5/6) with zones of gra (10YR4/1) mottles, moist, hard, clayey SILT with sand and trace gravel.	y some	ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ال ا	552	
32A	60/60 100%	cs		19			76			ار از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از از ا		
	NOTE(э): В	orenole	adan	uone	a usin	g bentonite ;	grout pumped from bottom of borehole.				Da es 4 ef 5



NOTE(S): Borehole abandoned using bentonite grout pumped from bottom of borehole.

F	[EL	D]	BOR	RI	N(G L	OG		C HANSON
	CLIENT: AEG Coffeen Power Sta Site: CCB Management Facili Location: Coffeen, Illinois Project: 05S3004A						on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4¼" HSA (blind drill)	BOREHOLE ID: SB-16a Well ID: MW16S Surface Elev: 626 ft. MSL
	DATE	ES: St Fir	art: 4/2: hish: 4/2	5/20 5/20	06 06			FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney	Completion: 20 ft. BGS Station: 877.355.1N
w	EATHE	R : 0	vercast, o	cool	(mid	l-40's)		Eng/Geo: R. Hasenyager	2,515,088.0E
5	SAMPL	LE	Г	EST	FINC	G I	TOPOGR	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	al (in				5/ft ³)	0	Quadr Towns	angle: Coffeen, IL hin: East Fork	$\Psi = 12.80 - \text{While drilling}$ $\Psi = 5.74 - \text{MW16S on } 6/1/06$
н	/ Tot		/ 6 in lue	re (%	an. (It	Typ	Section	1 2, Tier 7N; Range 3W	$\underline{\nabla} = 51.37$ - MW16D on 6/1/06
Numbe	Recov % Reco	Type	Blows , N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
								soil (10YR2/1), sl. moist, firm, clayey SILT with sand and trace gravel.	trace $V = V$
							2	Brown (10YR4/3), sl. moist, firm, silty CLAY with sand.	trace -624
							4	Gray (10YR5/1) with 25% yellowish brown (10YR mottles, moist, very soft, very silty CLAY with trace	25/8) e sand.
							6	Gray (10YR5/1) with 15% dark yellowish brown (10YR4/6) mottles, moist, very soft, very silty CLAY trace sand.	n Y with
							8 10 12	Gray (10YR5/1) with 10% yellowish brown (10YR mottles, moist, firm, silty CLAY with some sand and gravel.	25/8) d trace 616 614
							<u>₹</u>	Dark yellowish brown (10YR4/6), wet, sl. dense, si very fine- to fine-grained SAND.	ilty,
							14	Dark yellowish brown (10YR4/6), moist, firm, sil CLAY with sand and trace gravel.	lty 612
								Dark yellowish brown (10YR4/6), wet, loose, silty, fine- to fine-grained SAND.	very
							16	Dark yellowish brown (10YR4/6), wet, soft, silty Cl with sand and trace gravel.	LAY
								Yellowish brown (10YR5/6), wet, loose, very fine-to	$\begin{array}{c c} \hline \\ \hline \\ o \text{ very } \\ \hline \\ \circ \\ \circ \\ \circ \\ \circ \\ \circ \\ \circ \\ \circ \\ \circ \\ \circ$
								Grav (10VR5/1) wat loose fine to medium and	
								SAND.	
								Gray (101K5/1), moist, hard, clayey SILT with sand trace gravel.	d and 608
								Gray (10YR5/1), wet, loose, very fine- to fine-grai SAND.	ined
								Gray (10YR4/1), moist, very hard, clayey, sandy S with trace gravel.	
								End of Boring = 19.90 ft. BGS See SB-16 for sample & testing details	
	NOTE	(S): N	4W16S i	nstal	led i	n blind	-drilled bor	ehole within 10 ft of SB-16.	

\mathbf{F}		D]	BOR	RI	NC	G L	OG		C HANSON
CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A						er Statio Facility	on V	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill)	BOREHOLE ID: SB-16b Well ID: MW16D Surface Elev: 626 ft. MSL
	DATE	S: St Fir	art: 4/2	1/200 5/200	06 06			FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney	Completion: 51 ft. BGS Station: 877,354.9N
W	EATHE SAMPL	R: 0 E	vercast, c	cool	(mid	-40's)		Eng/Geo: R. Hasenyager	2,515,079.4E
r	/Total (in)	L	6 in ue	re (%)	n. (lb/ft ³)	Type	TOPOGI Quadı Towns Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 2, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 12.80$ - While drilling $\Psi = 5.74$ - MW16S on 6/1/06 $\overline{\Psi} = 51.37$ - MW16D on 6/1/06
Numbe	Recov % Reco	Type	Blows / N - Val RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
								soil (10YR2/1), sl. moist, firm, clayey SILT with sand and trace gravel.	trace $4 \frac{1}{4}$ $4 \frac{1}{4}$ 626
							2	Brown (10YR4/3), sl. moist, firm, silty CLAY with sand.	trace624
							4	Gray (10YR5/1) with 25% yellowish brown (10YF mottles, moist, very soft, very silty CLAY with trace	25/8) e sand.
							6	Gray (10YR5/1) with 15% dark yellowish brow (10YR4/6) mottles, moist, very soft, very silty CLA trace sand.	n Y with
							8-	Gray (10YR5/1) with 10% yellowish brown (10YF	5/8)
							10	mottles, moist, firm, silty CLAY with some sand and gravel.	d trace 616
							⊻ _	Dark yellowish brown (10YR4/6), wet, sl. dense, s very fine- to fine-grained SAND.	ilty,
							14-	Dark yellowish brown (10YR4/6), moist, firm, si CLAY with sand and trace gravel. Dark yellowish brown (10YR4/6), wet, loose, silty,	lty very 612
							- fuun	fine- to fine-grained SAND. Dark yellowish brown (10YR4/6), wet, soft, silty C with sand and trace gravel	
							16-	Yellowish brown (10YR5/6), wet, loose, very fine- t	o very
								Gray (10YR5/1), wet, loose, fine- to medium-grai	ned
							18	Gray (10YR5/1), moist, hard, clayey SILT with san trace gravel.	d and 608
								Gray (10YR5/1), wet, loose, very fine- to fine-gras SAND.	ined
								Gray (10YR4/1), moist, very hard, clayey, sandy S with trace gravel.	
	NOTE(S): N	1W16D i	nstal	lled i	in blind	d-drilled bo	rehole within 10 ft of SB-16.	

\mathbf{F}	IEL	D]	BOR	RI	NC	5 L	OG			<	S H	NSON
	CLIENT: AEG Coffeen Power Statio Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 4/21/2006 Finish: 4/25/2006					er Statio Facility	on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-850 Track Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill) FIELD STAFF: Driller: K. Doetzel	BOREHOLE ID: SB-16b Well ID: MW16D Surface Elev: 626 ft. MSL Completion: 51 ft. BGS			
W	EATHE	R: 0	vercast, c	:00l	06 (mid	-40's)		Eng/Geo: R. Hasenyager			Station:	877,354.9N 2,515,079.4E
nber	ov / Total (in) ecovery	E ع	vs / 6 in Value D	sture (%)	Den. (lb/ft ³) ZI	tsf) ure Type	TOPOGR Quadra Townsh Section	APHIC MAP INFORMATION: ingle: Coffeen, IL ip: East Fork 2, Tier 7N; Range 3W	WATER LEV $\underline{\Psi} = 12.3$ $\underline{\Psi} = 5.3$ $\underline{\nabla} = 51.3$ Benefit	VEL 80 - 74 - 37 -	INFORMATIC While drilling MW16S on 6/1/ MW16D on 6/1/	DN: 06 06
Nun	Rec %R	Typ	Blov N - N	Moi	Dry	Qu (Fail	ft. BGS	Description	Deta	nole ail	ft. MSL	Remarks
	NOTE	(S): M	1W16D i	nsta	lled i	in blind	22 24 24 26 30 30 32 34 34 34 34 34 34 34 34 34 34 34 34 34	Gray (10YR4/1), moist, very hard, clayey, sandy SI with trace gravel. [Continued from previous page]	Γ		-604 -602 -602 -598 -598 -594 -592 -590 -588	

L



See SB-16 for sample & testing details

NOTE(S): MW16D installed in blind-drilled borehole within 10 ft of SB-16.

FIELD BORING LOG												
BOREHOLE ID: SB-17 Well ID: MW17D Surface Elev: 627 ft. MSL												
Completion: 54 ft. BGS Station: 878,659.0N												
2,515,090.4E												
WATER LEVEL INFORMATION:												
$\Psi = 11.70$ - While drilling $\Psi = 6.89$ - MW17S on 6/1/06												
$\overline{\underline{\nabla}}$ = 54.45 - MW17D on 6/1/06												
Borehole Elevation Detail ft. MSL Remarks												
Y. 626												
/6) race 624												
/1) 622												
/1)												
618												
y, 616												
ery ine- 614												
very 612												
to 610												





F J	CLIEN Sit Locatio Projec DATE	D J T: Al e: Co n: Co c: C	EG Coffe CB Mana offeen, II 5S3004A fart: 5/4/ hish: 5/4/ artly sunn	een H agem linoi 2000 2000 y, co	Powe nent l is 6 6 6 001 (GL er Statie Facility mid-50	OG on 7	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA (blind drill) FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager	BOREHOLE ID: SB-17a Well ID: MW17S Surface Elev: 627 ft. MSL Completion: 24 ft. BGS Station: 878,658.5N 2,515,084.8E			
	Total (in)	E	E Zin	ES (%) :	. (lb/ft ³)	Type	TOPOGF Quadr Towns Section	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 2. Tier 7N: Range 3W	WATER LEVEL INFORMATION: $\Psi = 11.70$ - While drilling $\Psi = 6.89$ - MW17S on 6/1/06 $\nabla = 54.45$ - MW17D on 6/1/06			
Number	Recov /' % Recov	Type	<i>Blows / (</i> N - Valu RQD	Moisture	Dry Den	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks			
								soil grayish brown (10YR4/2), moist, soft, clayey with trace sand and trace gravel.				
							2	Yellowish brown (10YR5/8), moist, soft, silty CL	AY626			
								Gray (10YR5/1) with 40% yellowish brown (10YR mottles, moist, firm, silty CLAY with little sand and gravel.	5/6) trace 624			
							6	Yellowish brown (10YR5/6) with 10% gray (10YR mottles, moist, firm sandy, clayey SILT.	.5/1) 622			
							₹	Yellowish brown (10YR5/6) with 30% gray (10YR mottles, moist, firm sandy, clayey SILT.	.5/1) 620			
							10	Yellowish brown (10YR5/6), moist, soft, very san clayey SILT.	dy,			
							¥ 12	Gray (10YR5/1) with 30% yellowish brown (10YR mottles, very moist, soft, very sandy, clayey SIL Moderate yellowish brown (10YR5/4), wet, loose, <u>fine- to fine-grained SAND</u> . Yellowish brown (10YR5/6), wet, dense, silty, very to fine-grained SAND	5/6) r. very fine-			
								Yellowish brown (10YR5/6), very moist, dense, s very fine-grained SAND. Yellowish brown (10YR5/6) wet, dense, fine- to medium-grained SAND.	ilty, 614			
							16	Moderate yellowish brown (10YR5/4), wet, dense, and very fine-grained SAND. Yellowish brown (10YR5/6), moist, clayey SILT and fine-grained SAND with trace gravel.	SILT i very			
							18	Yellowish brown (10YR5/6), wet, loose, very fine medium-grained SAND.	- to			
								Yellowish brown (10YR5/6), wet, loose, very fine fine-grained SAND. Yellowish brown (10YR5/6), wet, sl. dense, SILT 	- to with ery - 608			



See SB-17 for sample & testing details

FIELD BORING LO CLIENT: AEG Coffeen Power Static Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 5/11/2006 Finish: 5/11/2006 WEATHER: Partly sunny, cool (mid-50				Powe nent l is 06 06 001 (:	GL er Statio Facility mid-50	OG on /	DG n CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4 ¹ / ₄ " HSA w/SS & CME samplers FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney s) Eng/Geo: R. Hasenyager			BOREHOLE ID: SB-18 Well ID: n/a Surface Elev: 626 ft. MSL Completion: 54 ft. BGS Station: 878,605.0N 2,513,750.0E		
er	v/Total (in)	Ľ	/ 6 in alue	ire (%)	en. (lb/ft ³)	e Type	TOPOGI Quadi Town Sectio	RAPHIC MAP INFORMATION: rangle: Coffeen, IL ship: East Fork n 3, Tier 7N; Range 3W	WATER \mathbf{I} $\mathbf{\Psi} =$ $\mathbf{\Psi} =$ $\mathbf{\Psi} =$ $\mathbf{\Psi} =$	LEVEL 12.20 - 6.87 -	INFORMATIC While drilling MW18S on 6/1/	DN: 06
Numb	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		orehole Detail	Elevation ft. MSL	Remarks
1A	18/24 75%	ss	0-1 1-1 N=2	24		1.31 BSh	2	Yellowish brown (10YR5/6) with 50% gray (10YR mottles, moist, silty CLAY with little sand.	4/1)		624	
2A	24/24 100%	ss	1-2 2-2 N=4	28		1.78 Sh	4	Yellowish brown (10YR5/6) with 10% gray (10YR mottles, moist, silty CLAY with little sand.	4/1)		622	
3A	24/24 100%	ss	1-2 2-2 N=4	23		1.32 BSh	6				620	
4A	24/24 100%	ss	1-1 1-2 N=2	24		1.09 B	¥ 8	Gray (10YR4/1)with 10% Yellowish brown (10YR mottles, moist, silty CLAY with little sand.	25/6)		618	
5A	24/24 100%	ss	1-1 1-2 N=2	28		0.54 <i>BSh</i>	10				616	
6A	24/24 100%	ss	0-0 1-2 N=1	21		0.39 B	¥ ¹²	Light gray (10YR6/1) moist, soft, clayey, very fine fine-grained SAND.	- to		614	
7A 7B	24/24 100%	ss	3-5 9-15 N=14	17				Light gray (10YR6/1) moist, soft, silty, very fine- fine-grained SAND Yellowish brown (10YR5/6), very moist, soft, silty, fine- to fine-grained, SAND with trace gravel.	very		612	
8A 8B	24/24 100%	ss	8-9 9-10 N=18	14 11				Yellowish brown (10YR5/6), wet, loose, fine- to medium-grained SAND. Yellowish brown (10YR5/6), very moist, soft, sandy fine- to fine-grained) SILT.	o (very		610	
8C 9A	24/36 67%	cs		9			16	Gray (10YR5/1), moist, very hard, clayey, sandy S with trace gravel.	ILT			
	NOTE(В): В С	orehole ME-105	aban 0 ha	done d 280	ed using 0# ham	g bentonite	grout pumped from bottom of borehole. T.			606	Page 1 of 3

F]	ELIEN CLIEN Sit Locatio Projec DATE	D] T: A ae: C n: C n: C ct: 05 ct: 05 S: St Fir R: Pa	EG Coffe CB Mana offeen, II 533004A tart: 5/1 hish: 5/1 artly sum	RI een H agem llinoi 1/20 1/20 ny, co	Powe nent l is 06 06 06 001 (Facility	OG on y'	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA w/SS & CME samplers FIELD STAFF: Driller: K. Doetzel Helper: S. McCartney Eng/Geo: R. Hasenyager		во	REHOLE I Well I Surface Ele Completio Statio	HANSON D: SB-18 D: n/a ev: 626 ft. MSL on: 54 ft. BGS on: 878,605.0N 2,513,750.0E
ler .	/ Total (in)	E	1/6 in alue	ure (%)	ben. (lb/ft ³)	t) re Type	TOPOGR Quadra Towns Section	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork 3, Tier 7N; Range 3W	WATE ⊈ ⊻ ∑	R LEVEL = 12.20 - = 6.87 - =	INFORMA While drillin MW18S on	ng 6/1/06
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		Borehole Detail	Elevation ft. MSL	Remarks
10A	60/60 100%	cs					22			د	604	
11A	60/60 100%	cs		5			26	Gray (10YR5/1), moist, very hard, clayey, sandy SI with trace gravel. [Continued from previous page]	LT		- 600 - 598 - 598	
12A	60/60 100%	CS		11			30			د و د و د و و و د و و و و و و و و و و و	590	Coal fragment seam
13A	60/60 100%	CS		10			36	Gray (10YR5/1), moist, dense, SILT.		لور ور و و و و و و و و و و و و و و و و و	590	
13B	-			12			40	Gray (10YR5/1), moist, very hard, clayey, sandy SI with trace gravel.	LT		586	
	NOTE(S): E C	Sorehole CME-105	aban 0 ha	done d 28	ed using 0# ham	g bentonite g imer for SPT	grout pumped from bottom of borehole.				Page 2 of 3



F	EL CLIEN Sin Locatio Projec DATE	D] T: Al te: Co on: Co on: Co ct: 05 C: St Fin R: Pa	BOR EG Coffe CB Mana offeen, II S3004A art: 5/1 ish: 5/1 rtly sunr	een F agem linoi 1/200 1/200 ay, co	Powe nent l is 06 06 06	Facility	OG on 's)	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-1050 ATV Rig Drilling Method: 4¼" HSA (blind drill) FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: R. Hasenyager		BOREHOLE ID: Surface Elev: Completion: Station:	ANSON SB-18a MW18S 626 ft. MSL 16 ft. BGS 878,604.7N 2,513,745.2E
5		E	Т	EST	INC	;	TOPOG	RAPHIC MAP INFORMATION:	WATER LEV	EL INFORMATIO	DN:
er -	/ / Total (in covery		a/6 in alue	ure (%)	ben. (lb/ft ³)	sf) re Type	Quad Towr Sectio	rangle: Coffeen, IL ship: East Fork on 3, Tier 7N; Range 3W		0 - While drilling 7 - MW18S on 6/1/	06
Numb	Recov % Re	Type	Blows N - V RQD	Moist	Dry I	Qu (ts Failu	Depth ft. BGS	Lithologic Description	Boreh Deta	ole Elevation il ft. MSL	Remarks
							2 4 4 8 10	Soil grayish brown (10YR4/2), moist, soft, clayey SII with trace sand and trace gravel. Yellowish brown (10YR5/6) with 50% gray (10YR4/1 mottles, moist, silty CLAY with little sand. Yellowish brown (10YR5/6) with 10% gray (10YR4/1 mottles, moist, silty CLAY with little sand. Yellowish brown (10YR5/6) with 10% gray (10YR4/1 mottles, moist, silty CLAY with little sand. Gray (10YR4/1)with 10% Yellowish brown (10YR5/6 mottles, moist, silty CLAY with little sand.		- 624 - 622 - 620 - 618 - 616	
							¥ 12-	Light gray (10YR6/1) moist, soft, clayey, very fine- to fine-grained SAND.		614	
							14	Yellowish brown (10YR5/6), very moist, sort, sinty, very fine- to fine- grained, SAND with trace gravel. Yellowish brown (10YR5/6), wery noist, soft, sandy (v fine- to fine-grained) SILT. Gray (10YR5/1), moist, very hard, clayey, sandy SIL' with trace gravel.	ery III	610	
	End of Boring = 16.40 ft. BGS See SB-18 for sample & testing details										

NOTE(S): MW18S installed in blind-drilled borehole within 10 ft of SB-18.







Page 3 of 3

APPENDIX 3-A-4 Well Completion Reports

Illinois Enviro	nmental Protection Agen	cy		Well C	Completion	Report
Site #:	County: M	ontgomery	,	Wel	11 #:MW	/1D
Site Name: AEG Coffeen Po	itv		Bor	ehole #: S	B-01	
State Plane Coordinate: X 874,97	2.6 Y 2,513,478.0 (or) Latitude	e:°	<u> </u>	Longitude:	o	
Surveyed By: <u>Darren E. Forg</u>	ΣΥ.	IL Regis	stration #: <u>035-0</u>	03637		
Drilling Contractor: <u>Testing S</u>	ervice Corporation	_ Driller:	B. Williamson			
Consulting Firm: <u>Hanson Pro</u>	fessional Services Inc.	_ Geologi	st: <u>Rhonald W</u>	Hasenyager,	LPG #196-00	0246
Drilling Method: <u>Hollow ster</u>	n auger	_ Drilling	Fluid (Type): <u>Po</u>	table water		
Logged By: <u>Testing Services</u>	s Corp.	_ Date Sta	arted: <u>5/3/200</u>) <u>6</u> Date I	Finished: <u>5/</u>	3/2006
Report Form Completed By: <u>R</u>	honald W Hasenyager	_ Date: _	6/7/2006			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			609.29	1.897	Гор of Protective	Casing
	ſ		609.01	1.617	Гор of Riser Pipe	•
Type of Surface Seal: <u>Concrete</u>					Ground Surface	
			604.80		Гор of Annular S	ealant
Type of Annular Sealant: <u>Bento</u>	onite grout					
Setting Time: +24 hr			571.12	36.28	Static Water Leve	4
Setting Time. <u>-+2+m</u> .		×			(After Completion)	6/1/2006
Type of Bentonite Seal Gran	ular Pellet Slurry (choose one)					
Installation Method: <u>Gravi</u>	t <u>y</u>	x x	578.90	28.50	Гор of Seal	
Setting Time: <u>21 min.</u>	×	×	575.82	31.58	Fop of Sand Pack	5
Type of Sand Pack: <u>Quartz san</u>	d					
Grain Size: #5 (si	eve size)		574.11	33.29	Гор of Screen	
Installation Method: <u>Gravi</u>	ty		5 (0.25	20.05		
Type of Backfill Material: <u>n/a</u>			<u> </u>	<u></u>	Bottom of Screen Bottom of Well	
Installation Method: n/a	(if applicable)		567.40	40.00 F	Bottom of Boreho	ole
			* Referenced to a	National Geodetic	2 Datum	
			CASI	ING MEASU	UREMENTS	
WELL CONS	TRUCTION MATERIALS		Diameter of Boreho	ole	(inches)	7.3
(Choose or	ne type of material for each area)	ŀ	ID of Riser Pipe		(inches)	2.0
		F	Protective Casing I	Length	(feet)	5.0
Protective Casing	SS304 SS316 PTFE PVC OTHER	: Steel	Rottom of Screen to	o End Cap	(feet)	0.36
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	:	Screen Length (1st	t slot to last slot)	(feet)	4.76
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	:	Total Length of Ca	sing	(feet)	40.29

PTFE PVC OTHER:

SS316

 Screen
 SS304

 Well Completion Form (revised 02/06/02)

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Enviro	nmental Protection Agend	ey		Well C	Completion	Report
Site #:	County: Mo	ontgomery	1	Wel	11 #:MV	V2S
Site Name: AEG Coffeen Po	wer Station CCB Management Facil	itv		Bor	ehole #: S	B-02a
State Plane Coordinate: X 876,408	8.9 Y 2,513,210.0 (or) Latitude	»: °	· "	Longitude:	°	1 11
Surveyed By: Darren E. Forg	y	IL Regi	stration #:035-00)3637		
Drilling Contractor: <u>Testing S</u>	ervice Corporation	Driller:	B. Williamson			
Consulting Firm: <u>Hanson Prop</u>	fessional Services Inc.	Geolog	ist: <u>Rhonald W I</u>	Hasenyager,	LPG #196-00	0246
Drilling Method: <u>Hollow sten</u>	n auger	Drilling	Fluid (Type): <u>Po</u>	table water		
Logged By: <u>Testing Services</u>	Corp.	Date St	arted:5/5/200	6 Date I	Finished: <u>5/</u>	5/2006
Report Form Completed By:	honald W Hasenyager	Date:	6/7/2006			
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)	
			(MSL)*	(BGS)	For of Drotostive	Casing
				3.451	rop of Protective	Casing
			627.07	2.977	Fop of Riser Pipe	2
Type of Surface Seal: <u>Concrete</u>		V III	624.10	0.00 0	Ground Surface	
Type of Annular Sealant: Bento	unite chins		619.72	4.387	Гор of Annular S	ealant
Installation Mathod: Gravit						
Setting Time: +24 hr			616.68	7.42	Static Water Lev	el
Setting Time. <u>124 m.</u>		¥			(After Completion)	6/1/2006
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method:	ty	××	619.72	4.387	Fop of Seal	
Setting Time: <u>+24 hr.</u>	X	×	614.60	9.507	Гор of Sand Pacl	X
Type of Sand Packs						
Grain Size: #5 (sie	d		613.76	_10.34]	Гор of Screen	
Installation Method: Gravit	IV SILCY					
			608.98	<u>15.12</u> H	Bottom of Screer	1
Type of Backfill Material: <u>n/a</u>	(if applicable)		608.59	<u> 15.51 </u> 1	Bottom of Well	
Installation Method: <u>n/a</u>			<u>608.59</u>	<u>15.51</u> I	Bottom of Boreh	ole
			* Referenced to a	National Geodetic	Datum	
		ſ	CASI	NG MEASU	UREMENTS	
WELL CONS	TRUCTION MATERIALS	-	Diameter of Boreho	ole	(inches)	7.3
(Choose on	the type of material for each area)	-	ID of Riser Pipe		(inches)	2.0
		-	Protective Casing L	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTFE PVC OTHER	Steel	Riser Fipe Length	o End Can	(Ieet)	0 39
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER		Screen Length (1st	slot to last slot)	(feet)	4.78
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER		Total Length of Cas	sing	(feet)	18.48

Well Completion Form (revised 02/06/02)

Screen

SS304 SS316 PTFE PVC OTHER:

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Environmental Protection Agence	y Well Completion Report
Site #: County:Mc	ntgomery Well #:MW2D
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	ty Borehole #: SB-02b
State Plane Coordinate: X 876,414.0 Y 2,513,209.7 (or) Latitude	•°' Longitude:°'
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Testing Services Corp.</u>	Date Started: <u>5/5/2006</u> Date Finished: <u>5/5/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u></u>
	<u>627.07</u> <u>7</u> Top of Riser Pipe
Type of Surface Seal: Concrete	<u>624.10</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite grout</u>	<u>621.00</u> <u></u> Top of Annular Sealant
Installation Method:	
Setting Time: <u>+24 hr.</u>	$ \underline{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ } \underbrace{ \ }$
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	$\underbrace{606.00}_{18.10}$ Top of Seal
Setting Time:20 min.	603.92 Top of Sand Pack
Type of Sand Pack: Quartz sand	
Grain Size: #5 (sieve size)	$\phantom{00000000000000000000000000000000000$
Installation Method: <u>Gravity</u>	597.27 26.83 Bottom of Screen
Type of Backfill Material: <u>n/a</u> (if applicable)	<u>596.88</u> <u>27.22</u> Bottom of Well
Installation Method: Re-drill borehole	<u>596.88</u> <u>27.22</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel	
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Screen	SS304	SS316	PTFE	PVC	OTHER:	
Well Completion Form (revised 02/06/02)						

25.00 Riser Pipe Length (feet) 0.39 Bottom of Screen to End Cap (feet) 4.80 Screen Length (1st slot to last slot) (feet) 30.19 Total Length of Casing (feet) 0.010 Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agenc	y Well Completion Report
Site #: County: _Mo	ntgomery Well #:MW3D
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	tyBorehole #:SB-03
State Plane Coordinate: X 876,554.5 Y 2,514,535.3 (or) Latitude:	<u> </u>
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Testing Services Corp.</u>	Date Started: <u>4/27/2006</u> Date Finished: <u>4/27/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>629.37</u> <u></u> Top of Protective Casing
	<u>628.94</u> <u></u>
Type of Surface Seal: Concrete	<u>625.70</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite grout</u>	<u>623.00</u> <u></u> Top of Annular Sealant
Installation Method:	
Setting Time: <u>+24 hr.</u>	570.30 55.40 Static Water Level (After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	
Setting Time:25 min.	<u>575.60</u> <u>50.10</u> Top of Sand Pack
Type of Sand Pack: Quartz sand	572 41 52 20 Tan of Server
Grain Size: #5 (sieve size)	<u></u>
Installation Method: <u>Gravity</u>	568.64 57.06 Bottom of Screen
Type of Backfill Material: <u>Cuttings</u> (if applicable)	<u>568.30</u> <u>57.40</u> Bottom of Well
Installation Method:Over-drill borehole	<u>567.70</u> <u>58.00</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS

Diameter of Borehole	(inches)	8.0
ID of Riser Pipe	(inches)	2.0
Protective Casing Length	(feet)	5.0
Riser Pipe Length	(feet)	55.51
Bottom of Screen to End Cap	(feet)	0.36
Screen Length (1st slot to last slot)	(feet)	4.77
Total Length of Casing	(feet)	60.64
Screen Slot Size **	(inches)	0.010
**Hand-Slotted Well Screens Are Unacceptable		

WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel	
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Screen	SS304	SS316	PTFE	PVC	OTHER:	
Well Completion Form (revised 02/06/02)						

Illinois Environmental Protection Agency	Well Completion Report
Site #: County: Mor	ntgomery Well #: MW3S
Site Name: AEG Coffeen Power Station CCB Management Facility	v Borehole #· SB-03a
State Consider V 876 554 8 V 2514 521 5 (cm) Latitude	o ' " Longitudo: ° ' "
France Coordinate: $\underline{A} = 0.70, 504, \overline{0} = 1, 2, 514, 551, \overline{5}$ (6) Latitude.	Longitude
Surveyed By: Darren E. Forgy	IL Registration #:
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):Potable water
Logged By: <u>Testing Services Corp.</u>	Date Started: <u>4/28/2006</u> Date Finished: <u>4/28/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>629.45</u> <u>-3.75</u> Top of Protective Casing
	<u>628.96</u>
Type of Surface Seal: Concrete	<u>625.70</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: Bentonite chips	<u>623.70</u> <u></u> Top of Annular Sealant
Installation Mathods Cravity	
Setting Time: +24 hr	7 618.67 7.03 Static Water Level
	(After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	<u>623.70</u> <u></u> Top of Seal
Setting Time:24 hr.	616.20 9.50 Top of Sand Pack
Type of Sand Pack:Quartz sand	613 68 12 02 Top of Screen
Grain Size:#5 (sieve size)	
Installation Method: <u>Gravity</u>	608.92 16.78 Bottom of Screen
Type of Backfill Material:	<u>608.55</u> <u>17.15</u> Bottom of Well
Installation Method: n/a	608 55 17.15 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTEE DVC OTHER (Riser Pipe Length (feet) 15.28
Riser Pipe Above W.T. SS304 SS316 PTFE (PVC) OTHER:	Bottom of Screen to End Cap (feet) 0.37
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Science Length (1st slot to last slot) (feet) 4.70 Total Length of Casing (feet) 20.41

Well Completion Form (revised 02/06/02)

Screen

SS304 SS316 PTFE PVC OTHER:

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Environmental Protection Agenc	y Well Completion Report
Site #: County:Mo	ntgomery Well #: MW4S
Site Name: AEG Coffeen Power Station CCB Management Facilit	ty Borehole #: SB-04a
State Plane Coordinate: X 877,999.7 Y 2,514,450.6 (or) Latitude:	•°' Longitude:°'
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: B. Williamson
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): Potable water
Logged By: Testing Services Corp.	Date Started:5/11/2006 Date Finished:5/11/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>626.07</u> Top of Protective Casing
	<u>625.60</u> <u></u> Top of Riser Pipe
Type of Surface Seal: Concrete	<u>622.40</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: Bentonite chins	619.57 2.83 Top of Annular Sealant
Installation Method: Gravity	
Setting Time: <u>+24 hr.</u>	<u> 616.73</u> <u> 5.67</u> Static Water Level (After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry	
(choose one) Installation Method: <u>Gravity</u>	<u>619.57</u> <u>2.83</u> Top of Seal
Setting Time: <u>+24 hr.</u>	614.15 <u>8.25</u> Top of Sand Pack
Type of Sand Pack: Ouartz sand	
Grain Size: #5 (sieve size)	<u>612.57</u> <u>9.83</u> Top of Screen
Installation Method: <u>Gravity</u>	
Type of Backfill Material: <u>n/a</u>	608.14 14.26 Bottom of Screen 607.63 14.77 Bottom of Well
Installation Method:	607.63 Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 7.3
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTEE PVC OTHER.	Riser Pipe Length (feet) 14.25
Riser Pipe Above W.T. SS304 SS316 PTFE (PVC) OTHER:	Bottom of Screen to End Cap (feet) 0.51
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 19 19

Well Completion Form (revised 02/06/02)

Screen

SS304 SS316 PTFE PVC OTHER:

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Environmental Protection Agency				Well Completion Report				
Site #: County: Montgomery				Well	1#:MW	/5D		
Site Name: AEG Coffeen Power Station CCB Management Eacility				Bore	ehole #: S	B-05		
State Plane Coordinate: X 878,174.8 Y 2,513,290.3 (or) Latitude:			''' Dorenote ".'DD 00					
Surveyed By: Darren E. Forgy IL Re			egistration #:035-003637					
Drilling Contractor: <u>Testing S</u>	ervice Corporation	Driller:	ler: B. Williamson					
Consulting Firm: <u>Hanson Prop</u>	fessional Services Inc.	_ Geologis	eologist: <u>Rhonald W Hasenyager, LPG #196-000246</u>					
Drilling Method: <u>Hollow sten</u>	1 auger	_ Drilling]	lling Fluid (Type): Potable water					
Logged By:Testing Services	Corp.	Date Star	Date Started: 5/12/2006 Date Finished: 5/17/2006					
Report Form Completed By: RI	nonald W Hasenyager	Date:	6/7/2006					
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)			
			(MSL)*	(BGS)	on of Protostive	Casing		
			020.21	3.011	op of Protective	Casing		
	ļ		625.78	<u>-3.18</u> T	op of Riser Pipe	2		
Type of Surface Seal: Concrete			622.60	0.00 0	Fround Surface			
Tune of Annular Scelent: Rontonite grout			620.77	<u>1.83</u> T	op of Annular S	ealant		
Installation Method: Tremi	e							
Setting Time: +24 hr			572.16	50.44 S	tatic Water Leve	el		
					(After Completion)	6/1/2006		
Type of Bentonite Seal Gran	ular Pellet Slurry (choose one)							
Installation Method: <u>Gravity</u>			582.29	<u>40.31</u> T	op of Seal			
Setting Time: <u>18 min.</u>		×	<u>579.14</u> <u>43.46</u> Top of Sand Pack			(
Type of Sand Pack: Quartz same	1							
Grain Size: #5 (sieve size)			577.03	<u>45.57</u> T	op of Screen			
Installation Method:	<u>y</u>							
Turne of Backfill Material: n/a			<u> </u>	<u>50.33</u> B	Bottom of Screen	l		
Type of Dackin Matchai.	(if applicable)			<u></u>				
Installation Method: <u>n/a</u>			<u>568.60</u> * Referenced to a N	<u>54.00</u> B National Geodetic	Bottom of Boreho Datum	ole		
			G + G					
		[CASI	NG MEASU	JREMENTS	8.0		
WELL CONSTRUCTION MATERIALS			ID of Riser Pipe		(inches)	2.0		
(Cnoose one type of material for each area)			Protective Casing L	ength	(feet)	5.0		
	Γ		Riser Pipe Length		(feet)	48.74		
Protective Casing	SS304 SS316 PTFE PVC OTHEF	t: Steel	Bottom of Screen to	o End Cap	(feet)	0.39		
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	<u>e</u>	Screen Length (1st	slot to last slot)	(feet)	4.76		
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	2:	Fotal Length of Cas	sing	(feet)	53.89		

PTFE PVC OTHER:

Well Completion Form (revised 02/06/02)

SS304

SS316

Screen

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Environmental Protection Agene	cy Well Completion Report					
Site #: County:	ontgomery Well #:MW5S					
Site Name: <u>AEG Coffeen Power Station CCB Management Facil</u>	ity Borehole #:SB-05a					
State Plane Coordinate: X 878,175.6 Y 2,513,285.5 (or) Latitude	:°' Longitude:°'					
Surveyed By: _ Darren E. Forgy	IL Registration #:035-003637					
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>					
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246					
Drilling Method: Hollow stem auger	Drilling Fluid (Type):Potable water					
Logged By: Testing Services Corp.	Date Started: <u>5/17/2006</u> Date Finished: <u>5/17/2006</u>					
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006					
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)					
	<u>626.14</u> <u>-3.54</u> Top of Protective Casing					
	<u></u>					
Type of Surface Seal: Concrete	<u>622.60</u> <u>0.00</u> Ground Surface					
Type of Annular Sealant: Bentonite chips	<u>620.49</u> <u>2.11</u> Top of Annular Sealant					
Installation Method: <u>Gravity</u>						
Setting Time: <u>+24 hr.</u>	$\underline{\nabla}$ <u>615.86</u> <u>6.74</u> Static Water Level (After Completion) 6/1/2006					
Type of Bentonite Seal Granular Pellet Slurry						
Installation Method: <u>Gravity</u>	$\underline{620.49} \underline{2.11} \text{Top of Seal}$					
Setting Time: <u>+24 hr.</u>						
Type of Sand Pack: Quartz sand	609.94 12.66 Top of Screen					
Grain Size: #5 (sieve size)						
	$\underline{605.19} \qquad \underline{17.41} \qquad \text{Bottom of Screen}$					
Type of Backfill Material: <u>Cuttings</u> (if applicable)	604.89 <u>17.71</u> Bottom of Well					
Installation Method:Over-drill borehole	<u>604.89</u> <u>17.71</u> Bottom of Borehole * Referenced to a National Geodetic Datum					
CASING MEASUREMENTS						

Diameter of Borehole	(inches)	8.0
ID of Riser Pipe	(inches)	2.0
Protective Casing Length	(feet)	5.0
Riser Pipe Length	(feet)	15.69
Bottom of Screen to End Cap	(feet)	0.40
Screen Length (1st slot to last slot)	(feet)	4.75
Total Length of Casing	(feet)	20.84
Screen Slot Size **	(inches)	0.010
**Hand-Slotted Well Screens Are Unacceptable		

WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel	
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Screen	SS304	SS316	PTFE	PVC	OTHER:	
Well Completion Form (revised 02/06/02)						

Illinois Environmental Protection Agency				Well Completion Report				
Site #: County: Montgomery				Well #	#: MW	6S		
Site Name: AEG Coffeen Power Station CCB Management Facility				Boreh	ole#: SB	-06a		
State Plane Coordinate: X 879.02	21.2 Y 2.513.189.4 (or) Latitude	. •	, "	Longitude:	°	1 "		
France Coordinate: $\underline{X} = \underline{575, 021.2} = \underline{1} = \underline{2, 515, 169.4}$ (61) Latitude:			Longnude					
Drilling Contractor: Testing	Service Corporation	Driller	Williamson					
Drilling Contractor: <u>Testing Service Corporation</u> Dr			C. L. M. Dharald W.Haammaar, LDC #106 000246					
Consulting Firm: <u>Hanson Professional Services Inc.</u> Ge			Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246					
Drilling Method: <u>Hollow ste</u>	m auger	Drilling	Fluid (Type): <u>Po</u>	table water				
Logged By: <u>Testing Service</u>	es Corp.	Date Sta	rted: <u>5/4/200</u>	06 Date Fin	nished: <u>5/4</u>	/2006		
Report Form Completed By:	Rhonald W Hasenyager	Date:	6/7/2006					
ANNULAR SPA	ACE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)			
			626.67	<u>-3.57</u> Toj	p of Protective	Casing		
			626.21	<u>-3.11</u> Top	p of Riser Pipe			
Type of Surface Seal: Concrete			623.10	0.00 Gro	ound Surface			
			620.10 3.00 Top of Appular			alant		
Type of Annular Sealant: <u>Bentonite chips</u>		TP/		10		ulullt		
Installation Method: Grav	rity							
Setting Time: ± 24 hr.		$\overline{\Delta}$	616.89	<u>6.21</u> Static Water Level (After Completion) 6/1/2006				
Type of Bentonite Seal Gra	nular Pellet Slurry							
(choose one)			620.10	3.00 Tot	n of Seal			
Setting Times + 24 kg			X					
Setting time: ± 24 m.		×	<u>613.34</u> <u>9.76</u> Top of Sand Pack					
Type of Sand Pack: <u>Quartz sa</u>	nd							
Grain Size: #5 (sieve size)			612.06	<u>11.04</u> Toj	p of Screen			
Installation Method: <u>Grav</u>	/ity							
Type of Backfill Material: n/a			<u>607.48</u> 607.02	<u>15.62</u> Bot 16.08 Bot	ttom of Screen ttom of Well			
	(if applicable)							
Installation Method: <u>n/a</u>			 <u>607.02</u> <u>16.08</u> Bottom of Borehole * Referenced to a National Geodetic Datum 					
			CASI	ING MEASUF	REMENTS			
			Diameter of Boreho	ole	(inches)	7.3		
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)			ID of Riser Pipe		(inches)	2.0		
			Protective Casing L	_ength	(feet)	5.0		
Protostive Cosing	\$\$204 \$\$216 PTEE BUC OTTER	(Steel)	Riser Pipe Length		(feet)	14.15		
Riser Pine Above W T	SS304 SS316 PTEE PVC OTHER:		Bottom of Screen to	o End Cap	(feet)	0.46		
Riser Pipe Below W T	SS304 SS316 PTFE PVC OTHER.		Screen Length (1st	t slot to last slot)	(feet)	4.58		
		-	Total Length of Cas	sing	(teet)	19.19		

Screen SS304 SS316 PTFE PVC OTHER: Well Completion Form (revised 02/06/02)

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010
Illinois Environmental Protection Agence	cy Well Completion Report					
Site #: County:Mo	ontgomery Well #:MW7S					
Site Name: AEG Coffeen Power Station CCB Management Facility Borehole #: SB-07a						
State Plane Coordinate: X 879,181.1 Y 2,514,397.5 (or) Latitude	:°' Longitude:°'					
Surveyed By: Darren E. Forgy	IL Registration #:035-003637					
Drilling Contractor: <u>Reynolds Drilling Corp.</u>	Driller: P. McIntire					
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W Hasenyager, LPG #196-000246					
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): Potable water					
Logged By:Rhonald W Hasenyager	Date Started:5/9/2006 Date Finished:5/9/2006					
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006					
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)					
	627.71 Top of Protective Casing					
	<u>627.56</u> <u></u> Top of Riser Pipe					
Type of Surface Seal: Concrete	<u>624.50</u> <u>0.00</u> Ground Surface					
Type of Annular Sealant: <u>Bentonite chips</u>	<u>621.70</u> <u></u> Top of Annular Sealant					
Installation Method: <u>Gravity</u> Setting Time: <u>+24 hr.</u>	∑					
Installation Method: <u>Gravity</u>	<u>621.70</u> <u>2.80</u> Top of Seal					
Setting Time: <u>+24 hr.</u>	616.23 <u>8.27</u> Top of Sand Pack					
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>#JC50FS</u> (sieve size)	<u>614.59</u> <u>9.91</u> Top of Screen					
Installation Method: <u>Gravity</u> Type of Backfill Material: <u>n/a</u> (if applicable)	<u>610.71</u> <u>13.79</u> Bottom of Screen <u>610.11</u> <u>14.39</u> Bottom of Well					
Installation Method: <u>n/a</u>	610.11 14.39 Bottom of Borehole * Referenced to a National Geodetic Datum					
	CASING MEASUREMENTS					
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0					
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0 Protective Casing Length (feet) 5.0					
	Riser Pipe Length (feet) 12.37					
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Can (fact) 0.60					

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: (Stee
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Screen	SS304	SS316	PTFE	PVC	OTHER:	
Well Completion Form (revised 02/06/02)						

ID of Riser Pipe (inches) 2.0 Protective Casing Length (feet) 5.0 Riser Pipe Length (feet) 12.37 Bottom of Screen to End Cap (feet) 0.60 Screen Length (1st slot to last slot) (feet) 4.48 Total Length of Casing (feet) 17.45 Screen Slot Size ** (inches) 0.010 **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agence	y Well Completion Report
Site #: County:Mo	ontgomery Well #:MW8S
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	tyBorehole #:SB-08a
State Plane Coordinate: X 879,776.6 Y 2,514,478.8 (or) Latitude	:°' Longitude:°'"
Surveyed By: _ Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Reynolds Drilling Corp.</u>	Driller: <u>K. Doetzel</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started:5/10/2006 Date Finished:5/10/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	628.26 Top of Protective Casing
Ę	<u></u>
Type of Surface Seal: Concrete	<u>624.70</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite chips</u>	<u>622.20</u> <u>2.50</u> Top of Annular Sealant
Installation Method: <u>Gravity</u>	
Setting Time: <u>+24 hr.</u>	∑ <u>619.37</u> <u>5.33</u> Static Water Level (After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	$\underline{-622.20} \underline{-2.50} \text{Top of Seal}$
Setting Time: <u>+24 hr.</u>	<u>614.72</u> <u>9.98</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u>	
Grain Size: <u>#JC50FS</u> (sieve size)	<u></u>
Installation Method: <u>Gravity</u>	608.70 16.00 Bottom of Screen
Type of Backfill Material: <u>Quartz sand</u>	<u>608.10</u> <u>16.60</u> Bottom of Well
Installation Method: <u>Gravity</u>	<u>607.62</u> <u>17.08</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0 Protecting Coging Lyngth 5.0
	Protective Casing Length (feet) 5.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel		
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:		
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:		
Screen	SS304	SS316	PTFE	PVC	OTHER:		
Well Completion Form (revised 02/06/02)							

14.73 Riser Pipe Length (feet) 0.60 Bottom of Screen to End Cap (feet) 4.49 Screen Length (1st slot to last slot) (feet) 19.82 Total Length of Casing (feet) 0.010 Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agenc	y Well Completion Report
Site #: County: _Mo	ntgomery Well #:MW9D
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	y Borehole #:SB-09
State Plane Coordinate: X 879,679.7 Y 2,515,666.3 (or) Latitude:	°' Longitude:°' "
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Reynolds Drilling Corp.</u>	Driller: <u>K. Doetzel</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started: <u>5/3/2006</u> Date Finished: <u>5/3/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u></u> <u></u>
	<u></u>
Type of Surface Seal: Concrete	<u>624.60</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite grout</u>	<u>621.70</u> <u>2.90</u> Top of Annular Sealant
Installation Method: <u>Tremie</u> Setting Time: <u>+24 hr.</u>	<u></u>
Type of Bentonite Seal Granular Pellet Slurry (choose one)	582.60 42.00 Top of Seal
Setting Time: <u>15 min.</u>	<u>580.80</u> <u>43.80</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>#JC50FS</u> (sieve size)	<u>578.79</u> <u>45.81</u> Top of Screen
Installation Method: <u>Gravity</u> Type of Backfill Material: <u>Quartz sand</u> (if applicable)	
Installation Method: <u>Gravity</u>	570.60 54.00 Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel	
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Screen	SS304	SS316	PTFE	PVC	OTHER:	
Well Completion Form (revised 02/06/02)						

52.25 Riser Pipe Length (feet) 0.43 Bottom of Screen to End Cap (feet) 4.76 Screen Length (1st slot to last slot) (feet) 57.44 Total Length of Casing (feet) 0.010 Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agen	cy	Well Completion Report				
Site #:	County: _ M	ontgomery	Well #: MW9S				
Site Name: AEG Coffeen Po	wer Station CCB Management Facil	ity	Borehole #: SB-09a				
State Plane Coordinate: X 879.684	4.9 Y 2,515,666.2 (or) Latitude	. • • • •	Longitude: ° ' "				
Surveyed By: Darren E. Forg	<u>ــــــــــــــــــــــــــــــــــــ</u>	IL Registration #:	03637				
Drilling Contractor: Reynolds	Drilling Corp.	Driller: K. Doetzel					
Consulting Firm: Hanson Prof	fessional Services Inc.	Geologist: Rhonald W	Hasenvager, LPG #196-000246				
Drilling Method: Hollow ster	n auger	Drilling Eluid (Type): Po	otable water				
L D D D D D D D D D D D D D D D D D D D		Diffining Fluid (Type). $\underline{-10}$	$\frac{1}{2} = \frac{1}{2}	Logged By: <u>Reynolds Drillin</u>	<u>ig Corp.</u>	Date Started:5/3/200	<u>J6</u> Date Finished: $5/3/2006$
Report Form Completed By: <u>Rl</u>	honald W Hasenyager	Date: <u>6/7/2006</u>					
ANNULAR SPA	CE DETAILS	Elevations (MSL)*	Depths (0.01 ft.) (BGS)				
	_	627.84	<u>-3.24</u> Top of Protective Casing				
	F	627.51	<u>-2.91</u> Top of Riser Pipe				
Type of Surface Seal: <u>Concrete</u>		624.60	0.00 Ground Surface				
		621.35	3.25 Top of Annular Sealant				
Type of Annular Sealant: <u>Bento</u>	onite chips						
Installation Method: <u>Gravit</u>	ty						
Setting Time: <u>+24 hr.</u>		⊻619.37	<u>5.23</u> Static Water Level (After Completion) 6/1/2006				
Type of Bentonite Seal Gran	ular Pellet Slurry						
	(choose one)		2.05 5 6 1				
Installation Method: <u>Gravit</u>		<u>621.35</u>	<u>3.25</u> Top of Seal				
Setting Time: <u>+24 hr.</u>	——————————————————————————————————————	615.49	Top of Sand Pack				
Type of Sand Pack:Quartz sand	d						
Grain Size: #JC50FS (sie	eve size)		<u>11.21</u> Top of Screen				
Installation Method: <u>Gravit</u>	ty						
Ture of Dockfill Material		<u>608.98</u>	<u>15.62</u> Bottom of Screen				
Type of Backfill Material: <u>I/a</u>	(if applicable)		<u>10.20</u> Bottom of wen				
Installation Method: <u>n/a</u>		<u>608.40</u> * Referenced to a	16.20 Bottom of Borehole National Geodetic Datum				
		CAS	ING MEASUREMENTS				
		Diameter of Boreh	ole (inches) 8.0				
WELL CONS (Choose on	TRUCTION MATERIALS	ID of Riser Pipe	(inches) 2.0				
		Protective Casing I	Length (feet) 5.0				
		Riser Pipe Length	(feet) 14.12				
Protective Casing	SS304 SS316 PTFE PVC OTHER	Bottom of Screen t	to End Cap (feet) 0.58				
Riser Pipe Above w.1.	SS304 SS316 PIFE PVC OTHER	Screen Length (1s	st slot to last slot) (feet) 4.41				

SS304

SS304

SS316

SS316

PTFE PVC OTHER:

PTFE PVC OTHER:

Riser Pipe Below W.T.

Screen

**Hand-Slotted Well Screens Are Unacceptable

Total Length of Casing

Screen Slot Size **

19.11

0.010

(feet)

Illinois Environmental Protection Agenc	cy Well Completion Repor
Site #: County:Mo	ontgomery Well #:MW10D
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	ity Borehole #: SB-10
State Plane Coordinate: X 878,245.1 Y 2,515,914.0 (or) Latitude:	:°' Longitude:°'
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Reynolds Drilling Corp.</u>	Driller: <u>K. Doetzel</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W Hasenyager, LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type): _ Potable water
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started:5/1/2006 Date Finished:5/1/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>624.72</u> Top of Protective Casing
	<u></u>
Type of Surface Seal: Concrete	<u>621.20</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite grout</u>	<u></u>
Installation Method: <u>Tremie</u> Setting Time: <u>+24 hr.</u>	∑ <u>47.48</u> Static Water Level (After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry (choose one) Installation Method: <u>Tremie</u>	<u>619.77</u> <u>1.43</u> Top of Seal
	<u>581.65</u> <u>39.55</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>#JC50FS</u> (sieve size)	<u>579.46</u> <u>41.74</u> Top of Screen
Installation Method: <u>Gravity</u> Type of Backfill Material: <u>Quartz sand</u> (if applicable)	<u>574.63</u> <u>46.57</u> Bottom of Screen <u>574.18</u> <u>47.02</u> Bottom of Well
Installation Method: <u>Gravity</u>	<u>572.45</u> <u>48.75</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (fact) 5.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel		
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:		
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:		
Screen	SS304	SS316	PTFE	PVC	OTHER:		
Well Completion Form (revised 02/06/02)							

Protective Casing Length (feet) 45.06 Riser Pipe Length (feet) 0.45 Bottom of Screen to End Cap (feet) 4.73 Screen Length (1st slot to last slot) (feet) 50.24 Total Length of Casing (feet) 0.010 Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agenc	с у	Well Completion Report
Site #:	County:Mo	ontgomery	Well #: MW10S
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facili	ity	Borehole #: SB-10a
State Plane Coordinate: X 878,250).5 Y 2,515,914.4 (or) Latitude	:''"	Longitude:°'"
Surveyed By: <u>Darren E. Forg</u>	У	IL Registration #:035-0	03637
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.	Driller: <u>K. Doetzel</u>	
Consulting Firm: <u>Hanson Prot</u>	essional Services Inc.	Geologist: <u>Rhonald W</u>	Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow sten</u>	1 auger	Drilling Fluid (Type): Po	otable water
Logged By: <u>Reynolds Drillin</u>	g Corp.	Date Started:5/2/200	06 Date Finished:5/2/2006
Report Form Completed By: <u>R</u>	ionald W Hasenyager	Date: 6/7/2006	
ANNULAR SPA	CE DETAILS	Elevations	Depths (0.01 ft.)
		624.55	-3.35 Top of Protective Casing
	F		<u>-3.04</u> Top of Riser Pipe
Type of Surface Seal: Concrete		621.20	Ground Surface
Type of Annular Sealant: <u>Bento</u>	nite chips	618.83	<u>2.37</u> Top of Annular Sealant
Installation Method: <u>Gravit</u>	<u>y</u>		
Setting Time: <u>+24 hr.</u>		⊻616.29	<u>4.91</u> Static Water Level (After Completion) 6/1/2006
Type of Bentonite Seal Gran	ular Pellet Slurry		
	(choose one)	(10.02	
Installation Method: <u>Gravit</u>	<u>y</u>		<u>2.37</u> Top of Seal
Setting 11me: ± 24 nr.	——————————————————————————————————————	<u> </u>	9.30 Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u>	<u>i</u>		
Grain Size: <u>#JC50FS</u> (sie	ve size)		<u>11.28</u> Top of Screen
Installation Method: <u>Gravit</u>	<u>y</u>	605 44	15.76 Pottom of Screen
Type of Backfill Material: <u>n/a</u>	(if applicable)	<u>604.90</u>	<u>16.30</u> Bottom of Well
Installation Method: <u>n/a</u>			16.30 Bottom of Borehole
		CAS	ING MEASUREMENTS
WELL CONS	TRUCTION MATERIALS	Diameter of Boreh	ole (inches) 8.0 (inches) 2.0
(Choose on	e type of material for each area)	Protective Casing	Length (feet) 5.0
		Riser Pipe Length	(feet) 14.32
Protective Casing	SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen	to End Cap (feet) 0.54
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER:	Screen Length (1	st slot to last slot) (feet) 4.48

SS304

SS304

SS316

SS316

PTFE PVC OTHER:

PTFE PVC OTHER:

Riser Pipe Below W.T.

Screen

**Hand-Slotted Well Screens Are Unacceptable

Total Length of Casing

Screen Slot Size **

(feet)

(inches)

19.34

0.010

Illinois Enviro	nmental Protection Agend	cy	Well Completion Report
Site #:	County:M	ontgomery	Well #: MW11D
Site Name: AEG Coffeen Por	wer Station CCB Management Facil	ity	Borehole #: SB-11
State Plane Coordinate: X 876,749	9.6 Y_2,515,976.7 (or) Latitude	<u></u>	Longitude:°''
Surveyed By: <u>Darren E. Forg</u>	у	IL Registration #:035-0	003637
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.	Driller: <u>K. Doetzel</u>	
Consulting Firm: <u>Hanson Prot</u>	fessional Services Inc.	Geologist: <u>Rhonald W</u>	Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem</u>	1 auger	Drilling Fluid (Type): P	otable water
Logged By: Reynolds Drillin	g Corp.	Date Started: $4/27/20$	006 Date Finished: 4/28/2006
Report Form Completed By: RI	nonald W Hasenvager	Date: 6/7/2006	
	CE DETAILS		Denths (0.01 ft)
		(MSL)*	(BGS)
		625.70	<u>-3.70</u> Top of Protective Casing
	E	625.36	<u>-3.36</u> Top of Riser Pipe
Type of Surface Seal: <u>Concrete</u>		<u>622.00</u>	0.00 Ground Surface
Type of Annular Scalents - Ponto	nite grout	618.89	<u>3.11</u> Top of Annular Sealant
Installation Mathada Trami			
Setting Time: +24 hr	<u></u>	∇ 615.97	6.03 Static Water Level
Setting Time. <u>+24 iii.</u>			(After Completion) 6/1/2006
Type of Bentonite Seal Gran	ular Pellet Slurry		
Installation Method:Gravit	y	597.20	24.80 Top of Seal
Setting Time: <u>18 min.</u>		595.59	26.41 Top of Sand Pack
			I
Type of Sand Pack: <u>Quartz sand</u>	<u>1</u>	593.69	28.31 Top of Screen
Installation Mathedu Cravit	ve size)		
Instantion Method. <u>Oravit</u>	<u>y</u>	588.96	33.04 Bottom of Screen
Type of Backfill Material: <u>n/a</u>	(if applicable)	588.50	33.50 Bottom of Well
Installation Method: <u>n/a</u>			36.33 Bottom of Borehole
		CAS	SING MEASUREMENTS
		Diameter of Boreh	nole (inches) 8.0
WELL CONS (Choose on	TRUCTION MATERIALS e type of material for each area)	ID of Riser Pipe	(inches) 2.0
		Protective Casing	Length (feet) 5.0
		Riser Pipe Length	(feet) 31.67
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER: SS304 SS316 PTFE PVC OTHER	Bottom of Screen	to End Cap (feet) 0.46
1			si sioi to fast sioi) (feet) 4.73

Protective Casing	SS304	SS316	PTFE	PVC	OTHER:		
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:		
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:		
Screen	SS304	SS316	PTFE	PVC	OTHER:		
Well Completion Form (revised 02/06/02)							

Total Length of Casing (feet) Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

36.86

0.010

Illinois Environmental Protection Agency			Well Completion Report
Site #: County: _ Montgomery			Well #:MW11S
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management Facili	ty	Borehole #: SB-11a
State Plane Coordinate: X 876,749	9.4 Y 2,515,971.2 (or) Latitude		°'''
Surveyed By: <u>Darren E. Forg</u>	У	IL Registration #:035-	003637
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.	Driller: <u>K. Doetzel</u>	
Consulting Firm: <u>Hanson Prot</u>	Sessional Services Inc.	Geologist: <u>Rhonald W</u>	⁷ Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow sten</u>	auger	Drilling Fluid (Type): <u>F</u>	Potable water
Logged By: <u>Reynolds Drillin</u>	g Corp.	Date Started:	006 Date Finished: 4/28/2006
Report Form Completed By: RI	nonald W Hasenyager	Date: 6/7/2006	
ANNULAR SPA	CE DETAILS	Elevations	Denths (0.01 ft.)
		(MSL)*	(BGS)
			<u>-3.47</u> Top of Protective Casing
	L.	625.16	<u>-3.16</u> Top of Riser Pipe
Type of Surface Seal: Concrete		<u>622.00</u>	0.00 Ground Surface
Tune of Annular Scalenti – Ponto	nite chine	620.00	2.00 Top of Annular Sealant
Installation Mathod: Gravit			
Setting Time: ± 24 hr	<u>y</u>	√ 616.58	5.42 Static Water Level
50000 Finite. <u>- + 2 + 111.</u>		<u> </u>	(After Completion) 6/1/2006
Type of Bentonite Seal Gran	<u>ilar</u> Pellet Slurry (choose one)		
Installation Method: <u>Gravit</u>	y	620.00	<u>2.00</u> Top of Seal
Setting Time: <u>26 min.</u>	X	615.25	<u>6.75</u> Top of Sand Pack
Type of Sand Pack: Ouerta com	4		
Grain Size: #JC50FS (sie	ve size)		<u>8.89</u> Top of Screen
Installation Method:	y		
Type of Packfill Material: n/a		<u>_608.37</u>	13.63 Bottom of Screen
Type of Backfill Material. <u>Il/a</u>	(if applicable)		<u>14.08</u> Bouom of wen
Installation Method: <u>n/a</u>			<u>14.08</u> Bottom of Borehole a National Geodetic Datum
		CA	SINC MEASUDEMENTS
		Diameter of Bore	hole (inches) 80
WELL CONS	TRUCTION MATERIALS	ID of Riser Pine	(inches) 2.0
(Unoose on	e type of material for each area)	Protective Casing	Length (feet) 5.0
		Riser Pipe Length	n (feet) 12.04
Protective Casing	SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen	to End Cap (feet) 0.46
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.74

(feet)

(inches)

17.24

0.010

Total Length of Casing

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

Protective Casing	SS304	SS316	PTFE	PVC	OTHER:
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:
Screen	SS304	SS316	PTFE	PVC	OTHER:
Well Completion Form (revised 02/06/02))				

Illinois Environmental Protection Agency	y		Well	Completion Report
Site #: County:Montgomery				Yell #: MW12D
Site Name: <u>AEG Coffeen Power Station CCB Management Facilit</u>	у		Be	orehole #:SB-12
State Plane Coordinate: X 875,515.1 Y 2,515,900.6 (or) Latitude:	0		Longitud	e:''"
Surveyed By: Darren E. Forgy	IL Registration #	#:035-00)3637	
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B.</u> W	illiamson		
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rh</u>	nonald W H	Hasenyage	r, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (T	Гуре): <u>Pot</u>	table wate	r
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started:	5/10/200	<u>)6</u> Date	e Finished: <u>5/10/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/	/7/2006		
ANNULAR SPACE DETAILS	Ele (N	evations MSL)*	Depths (BGS)	(0.01 ft.)
	6	525.49	-3.29	Top of Protective Casing
	6	525.03	-2.83	Top of Riser Pipe
Type of Surface Seal: Concrete	6	522.20	0.00	Ground Surface
Type of Annular Sealant: <u>Bentonite grout</u>	6	519.43	2.77	Top of Annular Sealant
Installation Method: <u>Tremie</u>	7 5	575.30	46.90	Static Water Level
				(After Completion) 6/1/2006
Type of Bentonite Seal (<u>Granular</u>) Pellet Slurry (choose one)	Ť			
Installation Method: <u>Gravity</u>		85.50	36.70	Top of Seal
Setting Time: <u>+24 hr.</u>	<u>× _5</u>	81.69	40.51	Top of Sand Pack
Type of Sand Pack:Quartz sand	5	79 74	42 46	Top of Screen
Grain Size: #5 (sieve size)			-12.10	Top of Scient
Installation Method: <u>Gravity</u>	5	575.21	46.99	Bottom of Screen
Type of Backfill Material:	5	574.73	47.47	Bottom of Well
Installation Method:Over-drill borehole	<u>5</u> * R	72.20 Referenced to a N		Bottom of Borehole tic Datum
		CASI	NG MEA	SUREMENTS
	Diamete	er of Boreho	ole	(inches) 8.0
(Choose one type of material for each area)	ID of R	iser Pipe		(inches) 2.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER:	Steel
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:	
Screen	SS304	SS316	PTFE	PVC	OTHER:	
Well Completion Form (revised 02/06/02)						

Protective Casing Length (feet) 5.0 Riser Pipe Length 45.29 (feet) 0.48 Bottom of Screen to End Cap (feet) 4.53 Screen Length (1st slot to last slot) (feet) 50.30 Total Length of Casing (feet) Screen Slot Size ** 0.010 (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agenc	y Well Completion Report
Site #: County:Mo	ntgomery Well #: MW12S
Site Name: AEG Coffeen Power Station CCB Management Facili	ty Borehole #: SB-12a
State Plane Coordinate: X 875,520.1 Y 2,515,900.5 (or) Latitude:	°' Longitude:°'"
Surveyed By: <u>Darren E. Forgy</u>	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: Rhonald W Hasenyager, LPG #196-000246
Drilling Method: Hollow stem auger	Drilling Fluid (Type):Potable water
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started:5/10/2006 Date Finished:5/10/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.)
Type of Surface Seal: Concrete	622.20 0.00 Ground Surface
	619.20 3.00 Top of Annular Sealant
Type of Annular Sealant: <u>Bentonite chips</u>	
Installation Method: <u>Gravity</u>	7 615 44 6 76 Statis Water Lough
Setting Time: <u>+24 nr.</u>	∑ <u>013.44</u> <u>0.70</u> Static water Lever (After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	<u>619.20</u> <u>3.00</u> Top of Seal
Setting Time: <u>18 min.</u>	613.95 8.25 Top of Sand Pack
Turne of Sond Books on the	
Grain Size: #5 (ciava cize)	<u>611.59</u> 10.61Top of Screen
Installation Method: Gravity	
	<u>607.02</u> <u>15.18</u> Bottom of Screen
Type of Backfill Material: <u>n/a</u> (if applicable)	<u>606.59</u> <u>15.61</u> Bottom of Well
Installation Method:	<u>* Referenced to a National Geodetic Datum</u>
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 7.3
(Choose one type of material for each area)	Protective Casing Length (fact) 5.0
	Riser Pipe Length (feet) 13.51
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Cap (feet) 0.43
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.57
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 18.51

Screen

SS304 SS316 PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Environmental Protection Agency				Well Com	pletion	Report
Site #: County: Montgomery				Well #:	MW	13D
Site Name: AEG Coffeen Po	wer Station CCB Management Facil	itv		Borehole	:#: S	B-13
State Plane Coordinate: X 874,69	4.3 Y 2,513,929.9 (or) Latitude	:°	· · · · · ·	Longitude:		
Surveyed By: <u>Darren E. Forg</u>	.y	IL Registi	ration #: <u>035-003</u>	3637		
Drilling Contractor: <u>Testing S</u>	ervice Corporation	Driller:	B. Williamson			
Consulting Firm: <u>Hanson Pro</u>	fessional Services Inc.	Geologist	: <u>Rhonald W Ha</u>	asenyager, LP	G #196-00	0246
Drilling Method: <u>Hollow ster</u>	n auger	Drilling F	luid (Type): <u>Pota</u>	ble water		
Logged By: <u>Reynolds Drillir</u>	ig Corp.	Date Start	ted: <u>5/9/2006</u>	Date Finish	ned: <u>5/9</u>	9/2006
Report Form Completed By: <u>R</u>	honald W Hasenyager	Date:	6/7/2006			
ANNULAR SPA	CE DETAILS		Elevations I	Depths (BGS)	(0.01 ft.)	
			<u>626.33</u>	<u>-3.63</u> Top o	f Protective	Casing
			625.87	<u>-3.17</u> Top o	f Riser Pipe	:
Type of Surface Seal: Concrete			622.70	<u>0.00</u> Groun	nd Surface	
Type of Annular Sealant: Bento	onite grout		619.64	<u>3.06</u> Top o	f Annular S	ealant
Installation Method: Tremi	e					
Setting Time: <u>+24 hr.</u>		∑	566.67	56.03 Static (After	Water Leve Completion)	el 6/1/2006
Type of Bentonite Seal Gran	ular Pellet Slurry					
Installation Method: <u>Gravi</u>	ty X	K X	577.48	<u>45.22</u> Top o	f Seal	
Setting Time: <u>+24 hr.</u>	Ž	×	574.76	<u>47.94</u> Top o	f Sand Pack	
Type of Sand Pack: <u>Quartz san</u>	d					
Grain Size: #5 (sie	eve size)		572.89	<u>49.81</u> Top o	f Screen	
Installation Method: <u>Gravi</u>	ty		568 10	54 60 Botto	m of Screen	
Type of Backfill Material: <u>n/a</u>	(if applicable)		567.70	<u>55.00</u> Botto	m of Well	
Installation Method: <u>n/a</u>			<u>567.70</u>	55.00 Botto	m of Boreho	ole
			CASIN	IG MEASURE	MENTS	0.0
WELL CONS	TRUCTION MATERIALS		D of Riser Pipe	e	(inches)	8.0
(Choose or	te type of material for each area)	P	rotective Casing Lei	ngth	(feet)	5.0
		R	iser Pipe Length		(feet)	52.98
Protective Casing	SS304 SS316 PTFE PVC OTHER:	Steel	ottom of Screen to I	End Cap	(feet)	0.40
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER:	s	creen Length (1st sl	lot to last slot)	(feet)	4.79
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER:	T	otal Length of Casir	ng	(feet)	58.17

Screen

SS316

SS304

PTFE PVC OTHER:

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Environmental Protection Agency	y Well Completion Report
Site #: County: _Mon	ntgomery Well #: MW13S
Site Name: AEG Coffeen Power Station CCB Management Facilit	ty Borehole #: SB-13a
State Y 2,513,925.3 (or) Latitude:	°' Longitude:°'"
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started: <u>5/9/2006</u> Date Finished: <u>5/9/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>626.42</u> <u>-3.72</u> Top of Protective Casing
	<u>625.92</u> Top of Riser Pipe
Type of Surface Seal: Concrete	<u>622.70</u> <u>0.00</u> Ground Surface
Type of Appular Scalapt: Poptonite chine	619.35 3.35 Top of Annular Sealant
Installation Methods Crowity	
Setting Time: _ +24 hr	$\underline{\nabla}$ <u>614.46</u> <u>8.24</u> Static Water Level (After Completion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry	
(choose one)	619.35 3.35 Top of Seal
Setting Time: 21 min.	
	$\underline{\times}$ <u>612.65</u> <u>10.05</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u>	611 27 11 43 Top of Screen
Grain Size: #5 (sieve size)	
Installation Method: <u>Gravity</u>	606.47 16.23 Bottom of Screen
Type of Backfill Material: <u>n/a</u> (if applicable)	606.08 16.62 Bottom of Well
Installation Method:	<u>606.08</u> <u>16.62</u> Bottom of Borehole
	* Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 7.3
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Protective Casing SS304 SS316 PTFE PVC OTHER:	Kiser Pipe Length (feet) 14.65 Steel Bottom of Screen to End Can 0.20
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.80
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 19.84

Screen

SS304 SS316 PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Environmental Protection Agency	y Well Completion Report
Site #: County: _Mon	ntgomery Well #: MW14S
Site Name: AEG Coffeen Power Station CCB Management Facilit	v Borehole #: Sb-14a
State Plane Coordinate: X 875,737.8 Y 2,514,125.9 (or) Latitude:	°'" Longitude:°'"
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):Potable water
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started:5/2/2006 Date Finished:5/2/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	627.232.63 Top of Protective Casing
Type of Surface Seal: Concrete	<u>624.60</u> <u>0.00</u> Ground Surface
Ture of Armular Socients Destorite shine	<u>622.20</u> <u>2.40</u> Top of Annular Sealant
Let their Mithele Conic	
Setting Time: +24 hr	$\overline{7}$ 620.11 4.49 Static Water Level
	(After Completion) 6//2006
Type of Bentonite Seal Granular Pellet Slurry	
Installation Method: <u>Gravity</u>	<u>622.20</u> <u>2.40</u> Top of Seal
Setting Time: 23 min.	
Type of Sand Pack: Owerta cond	
Grain Size: #5 (sieve size)	<u>612.34</u> <u>12.26</u> Top of Screen
Installation Method: Gravity	
	607.58 <u>17.02</u> Bottom of Screen
Type of Backfill Material: <u>n/a</u> (if applicable)	<u>607.22</u> <u>17.38</u> Bottom of Well
Installation Method:	607.22 17.38 Bottom of Borehole
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	Protective Casing Length (feat) 5.0
	Riser Pipe Length (feet) 14.48
Protective Casing SS304 SS316 PTFE PVC OTHER:	Steel Bottom of Screen to End Cap (feet) 0.36
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTHER:	Screen Length (1st slot to last slot) (feet) 4.76
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER:	Total Length of Casing (feet) 19.60

Screen

SS304 SS316 PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Environmental Protection Ag	ency		Well Comp	letion Report
Site #: County: Montgomery			Well #:	MW15S
Site Name: AEG Coffeen Power Station CCB Management F	Facility		Borehole #:	SB-15a
State Plane Coordinate: X 875,971.1 Y 2,515,076.3 (or) Lat	itude:°	'" L	ongitude:	
Surveyed By: <u>Darren E. Forgy</u>	IL Registrat	ion #:035-0036	37	
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B</u>	8. Williamson		
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: _	Rhonald W Hase	enyager, LPG #	\$196-000246
Drilling Method: _ Hollow stem auger	Drilling Flui	id (Type): <u>Potabl</u>	le water	
Logged By:Reynolds Drilling Corp.	Date Started	1: <u>4/25/2006</u>	Date Finished	:4/25/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:	6/7/2006		
ANNULAR SPACE DETAILS		Elevations De (MSL)* (E	epths (0 BGS)	.01 ft.)
		627.06 -3	3.26 Top of P	rotective Casing
		626.60	2.80 Top of R	iser Pipe
Type of Surface Seal: Concrete		623.80 0).00 Ground S	Surface
Turn of Annulus Contents Destantion bins		623.30 0).50 Top of A	nnular Sealant
Type of Annular Searant: Bentonite crips				
Installation Method: <u>Gravity</u>		618.81	1 99 Static W	atar Laval
Setting Time. <u>+24 m.</u>			(After Co	mpletion) 6/1/2006
Type of Bentonite Seal Granular Pellet Slurry				
Installation Method: <u>Gravity</u>	x x	614.30 9	9.50 Top of S	eal
Setting Time: 20 min.		611.60 1	2.20 Top of S	and Pack
Type of Sand Pack: Quartz sand				
Grain Size: #5 (sieve size)		609.39 14	4.41 Top of S	creen
Installation Method: <u>Gravity</u>				
Type of Backfill Material: n/a		<u>604.64</u> <u>1</u>	9.16 Bottom of 9.62 Bottom of 9.16	of Screen
(if applicable)			<u></u> Bottom C	ji wen
Installation Method: <u>n/a</u>		604.18 19 * Referenced to a Natio	9.62 Bottom of onal Geodetic Datum	of Borehole
		CASING	MEASUREM	ENTS
	Dia	meter of Borehole	INIEASUREINI	(inches) 8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID o	of Riser Pipe		(inches) 2.0
(2 · · · · · · · · · · · · · · · · · · ·	Prot	tective Casing Leng	th	(feet) 5.0
	Rise	er Pipe Length		(feet) 17.28
Protective Casing SS304 SS316 PTFE PVC OT	THER: Steel Bot	tom of Screen to En	nd Cap	(feet) 19.62
Riser Pipe Above W.T. SS304 SS316 PTFE PVC 01	THER: Screen	een Length (1st slot	to last slot)	(feet) 4.77
Riser Pipe Below W.T. SS304 SS316 PTFE PVC 01	THER: Tota	al Length of Casing		(feet) 41.67

Screen SS304 SS316 PTFE PVC OTHER: Well Completion Form (revised 02/06/02)

 Total Length of Casing
 (feet)

 Screen Slot Size **
 (inches)

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Environmental Protection Agenc	y Well Completion Report
Site #: County: _Mo	ntgomery Well #:MW15D
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	ty Borehole #: SB-15b
State Plane Coordinate: X 875,970.5 Y 2,515,080.7 (or) Latitude:	°' Longitude:°'
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started: <u>4/24/2006</u> Date Finished: <u>4/25/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u> Top of Protective Casing
	<u>626.45</u> <u></u>
Type of Surface Seal: Concrete	<u>623.80</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite grout</u>	<u>620.55</u> <u>3.25</u> Top of Annular Sealant
Installation Method:	
Setting Time: <u>+24 hr.</u>	$\underline{\nabla} \qquad \underline{618.56} \qquad \underline{5.24} \qquad \underline{\text{Static Water Level}}_{(After Completion) 6/1/2006}$
Type of Bentonite Seal Granular Pellet Slurry (choose one)	
Installation Method: <u>Gravity</u>	
Setting Time: <u>+24 hr.</u>	<u>591.80</u> <u>32.00</u> Top of Sand Pack
Type of Sand Pack: Quartz sand	500.12 22.69 Tan of Series
Grain Size: #5 (sieve size)	<u></u>
Installation Method: <u>Gravity</u>	<u></u>
Type of Backfill Material:	<u>585.00</u> <u>38.80</u> Bottom of Well
Installation Method: <u>Re-drill borehole</u>	585.00 38.80 Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:
Riser Pipe Below W.T.	SS304	SS316	PTFE	(PVC)	OTHER:
Screen	SS304	SS316	PTFE	PVC	OTHER:
Well Completion Form (revised 02/06/02)				

36.32 Riser Pipe Length (feet) 0.36 Bottom of Screen to End Cap (feet) 4.77 Screen Length (1st slot to last slot) (feet) 41.45 Total Length of Casing (feet) 0.010 Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agene	cy Well Completion Report
Site #: County:	ontgomery Well #: <u>MW16S</u>
Site Name: <u>AEG Coffeen Power Station CCB Management Facil</u>	ity Borehole #:SB-16a
State Plane Coordinate: X 877,355.1 Y 2,515,088.0 (or) Latitude	:°' Longitude:°'
Surveyed By: Darren E. Forgy	IL Registration #:035-003637
Drilling Contractor: _ Reynolds Drilling Corp.	Driller: K. Doetzel
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist:Rhonald W Hasenyager, LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Rhonald W Hasenyager</u>	Date Started: <u>4/25/2006</u> Date Finished: <u>4/25/2006</u>
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	<u>629.62</u> -3.52 Top of Protective Casing
	629.28 Top of Riser Pipe
Type of Surface Seal: Concrete	<u>626.10</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: <u>Bentonite chips</u>	<u>624.66</u> <u></u> Top of Annular Sealant
Setting Time: <u>+24 hr.</u>	$ \underline{\nabla} \qquad \underline{620.36} \qquad \underline{5.74} \qquad \text{Static Water Level} \\ (After Completion) 6/1/2006 $
Type of Bentonite Seal <u>Granular</u> Pellet Slurry (choose one) Installation Method: <u>Gravity</u>	624.661.44 Top of Seal
Setting Time: <u>17 min.</u>	612.40 <u>13.70</u> Top of Sand Pack
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>#JC50FS</u> (sieve size)	<u>611.51 14.59</u> Top of Screen
Installation Method: <u>Gravity</u>	$\underline{\underline{606.69}}$ $\underline{\underline{19.41}}$ Bottom of Screen $\underline{606.24}$ $\underline{19.76}$ Bottom of Well
(if applicable)	<u>000.34</u> <u>17.70</u> Doctorn of wen
Installation Method: <u>Gravity</u>	606.20 19.90 Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
	Diameter of Borehole (inches) 8.0
(Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel			
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:			
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:			
Screen	SS304	SS316	PTFE	PVC	OTHER:			
Well Completion Form (revised 02/06/02)								

ID of Riser Pipe(inches)2.0Protective Casing Length(feet)5.0Riser Pipe Length(feet)17.74Bottom of Screen to End Cap(feet)0.38Screen Length(1st slot to last slot)(feet)4.82Total Length of Casing(feet)22.94Screen Slot Size **(inches)0.010**Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agency	Well Completion Report				
Site #: County:	ntgomery	Well #:MW16D			
Site Name: <u>AEG Coffeen Power Station CCB Management Facilit</u>	у	Borehole #: SB-16b			
State Plane Coordinate: X 877,354.9 Y 2,515,079.4 (or) Latitude:		Longitude:°'"			
Surveyed By: <u>Darren E. Forgy</u>	IL Registration #:035-0	003637			
Drilling Contractor: <u>Reynolds Drilling Corp.</u>	Driller: <u>K. Doetzel</u>				
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W</u>	Hasenyager, LPG #196-000246			
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): Po	otable water			
Logged By: <u>Rhonald W Hasenyager</u>	Date Started:4/21/20	006 Date Finished: <u>4/25/2006</u>			
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006				
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (0.01 ft.) (BGS)			
	629.68	<u>-3.58</u> Top of Protective Casing			
	629.33	<u>-3.23</u> Top of Riser Pipe			
Type of Surface Seal: Concrete	626.10	<u>0.00</u> Ground Surface			
Type of Annular Sealant: <u>Bentonite grout</u>	<u>623.77</u>	<u>2.33</u> Top of Annular Sealant			
Installation Method: <u>Tremie</u> Setting Time: <u>+24 hr.</u>	<u></u> <u>574.73</u>	<u>51.37</u> Static Water Level (After Completion) 6/1/2006			
Type of Bentonite Seal Granular Pellet Slurry Installation Method: Gravity Setting Time: +24 hr.	<u>_584.65</u> <u>_582.65</u>	<u>41.45</u> Top of Seal <u>43.45</u> Top of Sand Pack			
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>#JC50FS</u> (sieve size)		45.90 Top of Screen			
Installation Method: <u>Gravity</u> Type of Backfill Material: <u>n/a</u> (if applicable)	<u>575.76</u> <u>575.32</u>	50.34Bottom of Screen50.78Bottom of Well			
Installation Method: <u>Re-drill borehole</u>	<u>575.10</u> * Referenced to a	51.00 Bottom of Borehole			
	CAS	ING MEASUREMENTS			
	Diameter of Boreh	ole (inches) 8.0			
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe	(inches) 2.0			

Protective Casing SS304 SS316 PTFE PVC OTHER: Steel Riser Pipe Above W.T. PVC OTHER: SS304 SS316 PTFE Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTHER: PVC OTHER: Screen SS304 SS316 PTFE Well Completion Form (revised 02/06/02)

ID of Riser Pipe (inches) 2.05.0 Protective Casing Length (feet) Riser Pipe Length 48.83 (feet) Bottom of Screen to End Cap 0.44 (feet) Screen Length (1st slot to last slot) (feet) 4.74 Total Length of Casing 54.01 (feet) 0.010 Screen Slot Size ** (inches) **Hand-Slotted Well Screens Are Unacceptable

Illinois Environmental Protection Agence	cy Well Completion Report
Site #: County: Mo	ontgomery Well #:MW17D
Site Name: <u>AEG Coffeen Power Station CCB Management Facili</u>	ty Borehole #:SB-17
State Plane Coordinate: X 878,659.0 Y 2,515,090.4 (or) Latitude	:°' Longitude:°'"
Surveyed By: <u>Darren E. Forgy</u>	IL Registration #:035-003637
Drilling Contractor: <u>Reynolds Drilling Corp.</u>	Driller: <u>K. Doetzel</u>
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type): <u>Potable water</u>
Logged By: <u>Reynolds Drilling Corp.</u>	Date Started:5/4/2006 Date Finished:5/4/2006
Report Form Completed By: <u>Rhonald W Hasenyager</u>	Date:6/7/2006
ANNULAR SPACE DETAILS	Elevations Depths (0.01 ft.) (MSL)* (BGS)
	630.62 Top of Protective Casing
	<u></u>
Type of Surface Seal: Concrete	<u>627.10</u> <u>0.00</u> Ground Surface
Type of Annular Sealant: Bentonite grout	<u>624.92</u> <u>2.18</u> Top of Annular Sealant
Installation Method:	
Setting Time: <u>+24 hr.</u>	$\underline{\nabla}$ 572.65 54.45 Static Water Level
Ture of Pontonite Soal Granular Pallet Shurry	
(choose one)	
Installation Method: <u>Gravity</u>	$\underbrace{581.55}_{45.55}$ Top of Seal
Setting Time: <u>+24 hr.</u>	<u>580.25</u> <u>46.85</u> Top of Sand Pack
Type of Sand Pack: Quartz sand	
Grain Size: <u>#JC50FS</u> (sieve size)	$\boxed{578.28} 48.82 \text{Top of Screen}$
Installation Method: <u>Gravity</u>	572.79 52.22 Pottom of Samon
Type of Backfill Material: <u>n/a</u> (if applicable)	
Installation Method: <u>n/a</u>	<u>573.23</u> <u>53.87</u> Bottom of Borehole * Referenced to a National Geodetic Datum
	CASING MEASUREMENTS
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser Pipe (inches) 2.0
	Protective Casing Length (feet) 5.0
Distractive Cosing SS204 SS216 DTEE DVC OTHER	Riser Pipe Length (feet) 52.01

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Steel			
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:			
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:			
Screen	SS304	SS316	PTFE	PVC	OTHER:			
Well Completion Form (revised 02/06/02)								

ID of Riser Pipe (inches) 2.0 Protective Casing Length (feet) 5.0 Riser Pipe Length (feet) 52.01 Bottom of Screen to End Cap (feet) 0.55 Screen Length (1st slot to last slot) (feet) 4.50 Total Length of Casing (feet) 57.06 Screen Slot Size ** (inches) 0.010 **Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Ag	gency	Well Completion Rep								
Site #:	County:	Montgome	nery Well #:MW17S								
Site Name: AEG Coffeen Pov	wer Station CCB Management 1	Facility		Во	orehole #: S	B-17a					
State Plane Coordinate: X 878,658	3.5 Y 2,515,084.8 (or) La	titude:	o <u>''''</u>	Longitude	:°						
Surveyed By: <u>Darren E. Forg</u>	у	IL Reg	IL Registration #:035-003637								
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.	Driller	Driller: K. Doetzel								
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geolo	Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246								
Drilling Method: <u>Hollow stem</u>	auger	Drillin	g Fluid (Type): <u>Po</u>	otable water							
Logged By: <u>Reynolds Drillin</u>	g Corp.	Date S	tarted: <u>5/4/200</u>	06 Date	Finished: <u>5/</u>	4/2006					
Report Form Completed By: <u>R</u>	ionald W Hasenyager	Date:	6/7/2006								
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.))					
			630.68	-3.58	Top of Protective	e Casing					
			630.34	3.24	Top of Riser Pip	e					
Type of Surface Seal: Concrete			<u>627.10</u>	0.00	Ground Surface						
Type of Annular Sealant: <u>Bento</u>	nite chips		626.40	0.70	Top of Annular S	Sealant					
Installation Method: <u>Gravit</u> Setting Time: <u>+24 hr.</u>	y	Ţ	620.21	6.89	Static Water Lev (After Completion)	el 6/1/2006					
Installation Method:	(choose one)		617.33	9.77	Top of Seal						
Setting Time: <u>22 min.</u>			614.80	12.30	Top of Sand Pac	k					
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>#JC50FS</u> (sie	lve size)		613.08	14.02	Top of Screen						
Installation Method: <u>Gravit</u> Type of Backfill Material: <u>n/a</u>	y(if applicable)		<u>603.54</u> <u>602.99</u>	<u>23.56</u> 24.11	Bottom of Screen Bottom of Well	1					
Installation Method: <u>n/a</u>			602.99 * Referenced to a	 National Geodeti	Bottom of Boreh	ole					
			CAS	ING MEAS	UREMENTS						
			Diameter of Boreh	ole	(inches)	8.0					
WELL CONS (Choose on	e type of material for each area)		ID of Riser Pipe		(inches)	2.0					
			Protective Casing	Length	(feet)	5.0					
Protective Casing	SS304 SS316 PTFE PVC O	THER: Steel	Riser Pipe Length	End Can	(feet)	17.26					

Protective Casing	SS304	SS316	PTFE	PVC	OTHER: Ste			
Riser Pipe Above W.T.	SS304	SS316	PTFE	PVC	OTHER:			
Riser Pipe Below W.T.	SS304	SS316	PTFE	PVC	OTHER:			
Screen	SS304	SS316	PTFE	PVC	OTHER:			
Well Completion Form (revised 02/06/02)								

ID of Riser Pipe(inches)2.0Protective Casing Length(feet)5.0Riser Pipe Length(feet)17.26Bottom of Screen to End Cap(feet)0.55Screen Length(1st slot to last slot)(feet)Total Length of Casing(feet)27.35Screen Slot Size **(inches)0.010**Hand-Slotted Well Screens Are Unacceptable

Illinois Enviro	nmental Protection Agen	cy	Well Completion Repo								
Site #:	County: <u>M</u>	ontgomery		W	/ell #:						
Site Name: AEG Coffeen Po	wer Station CCB Management Facil	ity		В	orehole #: SB-18a						
State Plane Coordinate: X 878,604	4.7 Y 2,513,745.2 (or) Latitude	e:° _	· · · · · ·	Longitud	le:''						
Surveyed By: <u>Darren E. Forg</u>	<u>.</u>	IL Regis	Registration #:035-003637								
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.	Driller:	iller: <u>B. Williamson</u>								
Consulting Firm: <u>Hanson Pro</u>	fessional Services Inc.	Geologis	eologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246								
Drilling Method: <u>Hollow sten</u>	n auger	Drilling	Fluid (Type): <u>Po</u>	otable wate	r						
Logged By: <u>Reynolds Drillin</u>	ig Corp.	_ Date Sta	rted: <u>5/11/20</u>	06 Dat	e Finished:5/11/2006	5					
Report Form Completed By: <u>R</u>	honald W Hasenyager	Date:	6/7/2006								
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	—					
			629.02	3.42	Top of Protective Casing	5					
	Ē		628.71	3.11_	Top of Riser Pipe						
Type of Surface Seal: <u>Concrete</u>			625.60	0.00	Ground Surface						
Type of Annular Sealant: <u>Bento</u>	onite chips		622.13	3.47	Top of Annular Sealant						
Installation Method:	ty										
Setting Time: <u>+24 hr.</u>		∇	618.73	6.87	Static Water Level (After Completion) 6/1/2006						
Type of Bentonite Seal Gran	ular Pellet Slurry										
Installation Method: <u>Gravit</u>	ty XX	x x	622.13	3.47	Top of Seal						
Setting Time: <u>25 min.</u>	——————————————————————————————————————	×	615.79	9.81	Top of Sand Pack						
Type of Sand Pack: <u>Quartz san</u>	d										
Grain Size: <u>#JC50FS</u> (side	eve size)		614.29	11.31	Top of Screen						
Installation Method: <u>Gravit</u>	ty		609.81	15 79	Bottom of Screen						
Type of Backfill Material: <u>n/a</u>	(if applicable)		609.20	16.40	Bottom of Well						
Installation Method: <u>n/a</u>			609.20 * Referenced to a	<u>16.40</u> National Geode	Bottom of Borehole etic Datum						
			CAS	ING MEA	SURFMENTS						
		-	Diameter of Boreh	ole	(inches) 8.0						
WELL CONS (Choose or	TRUCTION MATERIALS ne type of material for each area)		ID of Riser Pipe		(inches) 2.0						
		-	Protective Casing l	Length	(feet) 5.0	-					
Protective Casing	SS304 SS316 PTFE PVC OTHER	: Steel	Riser Pipe Length	o Fnd Can	(feet) 14.42	2					
-	+		- stron or server t	Line Cap	(100) 0.0.						

 Riser Pipe Above W.T.
 SS304
 SS316
 PTFE
 PVC
 OTHER:

 Riser Pipe Below W.T.
 SS304
 SS316
 PTFE
 PVC
 OTHER:

 Screen
 SS304
 SS316
 PTFE
 PVC
 OTHER:

 Well Completion Form (revised 02/06/02)
 V
 V
 V
 V

**Hand-Slotted Well Screens Are Unacceptable

(feet)

(feet)

(inches)

4.48

19.51

0.010

Screen Length (1st slot to last slot)

Total Length of Casing

Screen Slot Size **

APPENDIX A8

AECOM BORING LOGS AND PIEZOMETER CONSTRUCTION DOCUMENTATION

Project Number:

Project Location: Coffeen Power Station, IL

60440742

Log of Boring COF-B001

Sheet 1 of 2

Date(s) Drilled	08/07/2015 8:40 AM to 08/07/2015 4:10 PM	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	35.0 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	634.8 ft NAVD88
Borehole Backfill	Cement-Bentonite Grout (Installed COF-P000 5 ft South of COF-B001)	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 871595.4 E 2516695.5 (ft NAD83)	Groundwater Level(s)	Not Encountered		

<u>F</u>			SA	MPLES	S	_		e							
Elevation (fee	Depth (feet)	Type	Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAL DESCRIPTION evaluation tet 4.8 0.0	Natural Moistur Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
_	-		S1	3 4 6	67		Stiff, moist, brown and gray, silty and sandy CLAY, trace fine gravel, topsoil upper 2 inches, (CL) (EMBANKMENT FILL).	12.9				3.5			
 630	- 5-		S2	2 4 6	61		Stiff, moist, brown and gray, medium plasticity, silty and sandy CLAY, trace fine gravel, (CL) (EMBANKMENT FILL).	20.3		31	17	2.5			
- -	-		S3	2 5 7	72		Stiff, very moist, brown and gray, medium plastic, silty and sandy CLAY, trace gravel (CL)	15.4				1.75			
_ 625	- - 10-		54	2 4 5	83		Stiff, moist, brown and gray, sandy CLAY, trace fine gravel, with gray silt seams, (CL) (EMBANKMENT FILL).	16.1				1.5			
_	-	-					-								
_ _620	- 15–		S5		92		Very stiff, very moist, dark grayish brown with yellowish brown and dark gray, medium plastic, CLAY, with sand, (CL) (EMBANKMENT FILL).	14.7	129.4	35	20	2.5			
_	-						-								
— —615 —	- 20-		S6	1 2 2	83		5.8 Very stiff to 19' 19.0 Soft, wet, brown and gray, silty CLAY, trace fine sand and decayed organic matter, organic odor, (CL) (NATIVE).	23.2							
_	-		S7		92		Gray with yellowish brown, very plastic, CLAY, with sand, (CH) 2 1.8 23.0	23.4	125.7	66	44				
 610 	- 25-	:	S8	1 4 5	100		Stiff, very moist, brown and gray, high plasticity,	19.6		41	26	1.5			
_	-	-													
— —605	- 30-		S9	1 3 3	100		5.3 29.5	16.8				< 0.25			



Report. GEO_SOIL; FIIe N; PROJECTS/60428794_DYNEGY_CCR_RULEASMT/SUB_00/10.0_CALCULATIONS_ANALYSIS_DATA/SITE INVESTIGATION/COFFEENBORINGS/GINT TEMPLATE/DYNEGY_COFFEEN_2015.GFJ; 12/29/2015.5:42:56 PM

Project Location: Coffeen Power Station, IL

Log of Boring COF-B002

Date(s) Drilled	08/06/2015 9:30 AM to 08/06/2015 3:00 PM	Logged By	E. Drumright		Checked By	D. Swanson		
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA		Borehole Depth			
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology		Surface 635.4 ft NAVD88			
Borehole Backfill	Piezometer COF-P002	Sampling Method(s)	SS / ST		Hammer Data	Automatic		
Boring Location	N 871459.5 E 2516041.9 (ft NAD83)	Groundwater Level(s)	Groundwater Level(s) 28 ft on 8/6/2015 1:00:00 PM					
						<u> </u>		
(feet)	et)		isture		x ape			

	ect Number: 60440742														She	et 1	of 2		
Date(s)	08/0)6/2015	9:30 AM to	o 08/0	6/2015 :	3:00 PM	Logged	E	. Drumrig	ht				C	heck	ed	D. Swa	anson	
Drilling Method	Hol	low Ste	m Auger				Drill Bit Size/Type	7.	.5 inch O.	D. HSA				B	Borehole 35.5 ft				
	СМ	E 550X					Drilling	r G	eotechno	logy				S	Surface 635.4 ft NAVD88			'D88	
Borehole Backfill	Piez	zomete	r COF-P00	2			Sampling Method(s) SS / ST					H	lamm ata	ner	Autom	natic			
Boring	N 87	71459.5	5 E 251604	1.9 (ft	NAD83)	Groundwa Level(s)	ater 2	8 ft on 8/6	/2015 1:00:	:00 P	М							
		S	AMPLE	S	_							à							
Elevation (fee	Depth (feet)	Type Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MA ⁻ Elevation (feet) 635.4	TERIAL	DE	SCRI	PTION	Depth (feet) 0.0	Natural Moistur Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
		S1	1 2 4	78		Medium st CLAY, trac FILL).	iff, very mois e organics,	st, bro (CL)	own and g (EMBAN	gray, KMENT	_	25.4				2.0			
	5	S2	2 3 5	89		_ Stiff, very r plastic, sar _ (EMBANKI	noist, browr ndy CLAY, ti MENT FILL	n and race f).	gray, me îne grave	dium I, (CL)	-	25.9		35	18	1.0			
	-	S3	1 3 5	83		Medium st sandy CLA (EMBANKI	iff, very moi Y, trace fine MENT FILL)	st, bro e grav).	own and g /el, (CL)	gray,	_	25.9				1.25			
	10	S4		100		└ Very stiff, \ high plastic (EMBANK	very moist, g city, CLAY, y MENT FILL	gray w with s).	vith yellov and, (CL)	vish brown)	, _	17.8	134.5	40	25	3.5			
	- - - 15-	S5	1 1 2	78		Soft, moist fine sand, (EMBANKI Soft, very r silty CLAY	to wet, brov trace fine gr MENT FILL noist, grav v , (CL) (EMB	wn an ravel,). with bi &ANKN	id gray, C (CL) rown, Iow MENT FIL	LAY, with / plasticity, L).	-	23.3 26.7	120.2	25	7	0.75			
	- - 20-	S7	235	94		Medium sti plasticity, (seams, (Cl	iff, very moi CLAY, trace L) (NATIVE	st, bro fine s).	own and g sand, with	gray, high i brown silt	<u>- 17.0</u> -	25.4		47	29	1.5			
	- - 25- -	S8	234	100							-	18.9				0.75			
	30-	59	1 2 4			^{607.4} Loose, ver _ SAND, trad	y wet, browi ce gravel, (६	n fine SP-SC	to coarse C) (NATIV	e clayey /E).	28.0	13.6							



Report. GEO_SOIL; File N; PROJECTS(60428794_DYNEGY_CCR_RULEASMT)SUB_00/10.0_CALCULATIONS_ANALYSIS_DATASITE INVESTIGATION/COFFEEN'BORINGS(GINT TEMPLATEDYNEGY_COFFEEN_2015.GPJ; 12/29/2015.5:43:03 PM

Project Location: Coffeen Power Station, IL

Log of Boring COF-B003

Sheet 1 of 2

Project	Number:	60440742

Date(s) Drilled	08/05/2015 11:30 AM to 08/05/2015 5:30 PM	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	45.0 ft
Drill Rig Type	CME 550X (Rubber Tire ATV)	Drilling Contractor	Geotechnology	Surface Elevation	635.7 ft NAVD88
Borehole Backfill	Piezometer COF-P003	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 871786.2 E 2515149.7 (ft NAD83)	Groundwater Level(s)	Not Encountered		

j;			SA	MPLES	Ş	_	gu la la la la la la la la la la la la la	
Elevation (fee	Depth (feet)	Type	Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAT DESCLIDIN (teet) (teet) Content (%) Total Unit Watural Moistur Content (%) Total Unit Weight (pcf) Liquid Limit Plasticity Index Su (ksf) TXUU (ksf) TXUU (ksf)	ŝ
635 	-		S1	3 5 4	50		Stiff, dry, brown and gray, sandy CLAY, trace fine gravel, trace coarse ash, (CL) (EMBANKMENT FILL).	
-	- 5		S2	2 3 4	67		Medium stiff, moist, brown and gray, high _ 18.1 42 26 1.25 plasticity, sandy CLAY, trace fine gravel, trash _ coarse ash, (CL) (EMBANKMENT FILL).	
630 	-		S3	2 2 5	67		Medium stiff, moist, brown and gray, sandy CLAY, trace fine gravel, (CL) (EMBANKMENT 15.0 FILL).	
_ 625	- 10-		S4	2 4 8			Stiff, very moist, brown and gray, sandy CLAY, _ trace fine gravel, (CL) (EMBANKMENT FILL). 17.4 1.5	
_	-							
_ 620	15- -		S5		92		621.7 Stiff, very moist, dark gray trace yellowish brown, high plasticity, CLAY, (CH). 21.8 127.5 54 36 1.75	
 	- - 20-		S6	2 3 4	83		Medium stiff, very moist, brown and gray, medium plasticity, CLAY, trace fine sand, (CH).	
- - -	-			WOH			Medium stiff, very moist, brown and gray, high	
 610 	25		S7	23	100		plasticity, CLAY, with sand, with iron stained 20.8 50 34 1.75 seams, (CH).	
-	30-		S8	2 1 9	100		606.7 29.0 Stiff, very moist, brown, low plasticity, sandy silty CLAY, trace fine gravel, (CL) (TILL). 12.5 21 6 3.5	
·							A_LUM	



Project Location: Coffeen Power Station, IL

Log of Boring COF-B004

Sheet 1 of 2

Project Number: 6044074

Date(s) Drilled	08/04/2015 10:20 AM to 08/05/2015 11:00 AM	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	45.0 ft
Drill Rig Type	CME 550X (Rubber Tire ATV)	Drilling Contractor	Geotechnology	Surface Elevation	635.0 ft NAVD88
Borehole Backfill	Cement-Bentonite Grout (Installed COF-P005 5 ft West of COF-B004)	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 872197.6 E 2516082 (ft NAD83)	Groundwater Level(s)	Not Encountered		

F		5	SAN	IPLES	3	-	ω
Elevation (fee	Depth (feet)	Type	Samoling Resist	OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAL DESCRIPTION Image: Material Moistur Matural Moistur 000 Matural Moistur 000 Total Unit 1000000000000000000000000000000000000
_	-	S	1	2 3 5	67		Medium stiff, moist, brown, CLAY, with fine sand, trace fine gravel, (CL) (EMBANKMENT 18.0 FILL).
— —630	- 5-	s	2	3 4 5	67		Stiff, moist, brown and gray, sandy CLAY, trace fine gravel, (CL) (EMBANKMENT FILL). 10.9 3.5
_	-	s	3	2 4 4	72		Medium stiff, moist, brown and gray, sandy CLAY, trace fine gravel, (CL) (EMBANKMENT 11.0 FILL).
— —625	- 10-	S4	1		79		_ Medium stiff, moist, yellowish brown with gray, _ high plasticity, CLAY, with sand, (CL) _ 12.8 136.6 39 24 2.25 _ (EMBANKMENT FILL).
_	-						
 620 	- 15–	S	5	3 5 6	94		Stiff, moist, brown and gray, medium plastic,
_ _ _615	- - 20	S	6	2 3 7	83		Stiff, very moist, brown and gray, sandy CLAY, trace fine gravel, with gray silt seams, (CL) (NATIVE). 21.8 1.75
_ _ 610 	- - 25	ST	7		92		Very stiff, very moist, reddish brown, high plasticity, silty CLAY, trace fine sand, with dark gray silt seams, (CL).
_ _ _605	- - 30-	S	3	1 4 5	100		Stiff, very moist, brown and gray silt seams, high plasticity, CLAY, with fine to medium sand, (CL).

AE 22 ALC: N



Report. GEO_SOIL; FIIe N; PROJECTS/60428794_DYNEGY_CCR_RULEASMT/SUB_00/10.0_CALCULATIONS_ANALYSIS_DATA/SITE INVESTIGATION/COFFEENBORINGS/GINT TEMPLATE/DYNEGY_COFFEEN_2015.GFJ; 12/29/2015.5:43:17 PM

Project Location: Coffeen Power Station, IL

Log of Boring COF-B005

Sheet 1 of 2

Project Number:	60440742
-----------------	----------

Date(s) Drilled	08/08/2015 7:45 AM to 08/08/2015 12:00 PM	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	60.0 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	635.1 ft NAVD88
Borehole Backfill	Cement-Bentonite Grout (Installed COF-P006 5 ft South of COF-B005)	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 872109.6 E 2516693.3 (ft NAD83)	Groundwater Level(s)	Not Encountered		

<u>_</u>			SAN	IPLES	3				e							
Elevation (fee	Depth (feet)	Type	Compline Deciet	oanipiling resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAL DESCRI	PTION Depth (feet) 0.0	Natural Moistur Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
_	-	s	1	2 3 4	83		Medium stiff, moist, brown and gray (CL) (EMBANKMENT FILL).	, CLAY,	21.1				1.5			
_ _630	- 5	s	2		54		Stiff, moist, brown, low plasticity, sa (CL) (EMBANKMENT FILL).	ndy CLAY,	8.0	136.6	22	9	3.0			
-	-	s	3	1 3 4	72		Medium stiff, moist, brown, sandy C (EMBANKMENT FILL).	LAY, (CL)	13.1				0.75			
 625	- 10-	s	4	2 7 6	89		Stiff, moist, brown, low plasticity, sa CLAY, trace fine gravel, (CL) (EMB/ FILL).	ndy silty ANKMENT 	9.9		20	6	3.0			Shale in tip
-	-								-							
_ _620	- 15–	s	5	2 6 6	78		Stiff, moist, brown and gray, sandy (fine gravel, (CL) (EMBANKMENT F	CLAY, trace . ILL). –	10.3				2.0			
_	-								-							
— —615 —	- 20— -	s	6	2 3	67		CLAY, trace fine gravel, (CL) (EMB/ FILL).	, sandy ANKMENT –	9.4				1.0			
_	-						2.1		-							
— —610 —	- 25 -	s	7		92		medium plastic, CLAY, with sand, w organics, (CL) (NATIVE).	ith –	18.7	131.4	37	20	1.5			
_	-			1			6.6 Stiff, brown and gray, medium plasti		-							
	30-	s	8	4 6	89		trace fine sand, trace organics, (CH		21.9				2.0			

Project: Project Loca Project Num	Dyne tion: ber:	egy Coffeen 6044074	Powe	r Stat	ion, IL			L	.og	of	Bor She	ing et 2	COF-B005
Elevation (feet)	Type Number S	Sampling Resist. Barlawa OR Core RQD (%)	Recovery (%)	Graphic Symbol	MATERIAL DESCRIPTION	Natural Moisture Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
	59	1 1 1	100		Very stiff, low plasticity, sandy CLAY, with organics, (CL). Very loose, saturated, brown and gray, silty, clayey SAND, (SC-SM) (TILL).	26.7		22	6	2.5			35': Attempt ST refusal at 35.5'
_ _ 595 40 -	S10	8 22 27	100		Dense, wet, brown and gray, clayey SAND, trace fine gravel, (SC) (TILL).	11.6				-			
_ 	S11	4 7 11	100		Very stiff, very moist, brown and gray, medium plastic, sandy CLAY, trace gravel, iron stained vertical seams, (CL) (TILL).	12.8		32	17	3.25			
585 50	S12	2 6 8	100		Stiff, saturated, brown and gray, medium plastic, sandy CLAY, trace gravel, iron stained vertical seams, (CL) (TILL).	15.5		32	17	3.0			
 ₅₈₀ 55 - 	S13	2 5 7	100		Stiff, very moist, brown and gray, sandy CLAY, trace gravel, iron stained vertical seams, (CL) (TILL).	23.2				1.25			
575 60 - 	S14	2 3 6	100		Stiff, brown and gray, high plasticity, CLAY,	23.3		47	30	1.25			Installed Piezometer COF-P006 with 5 ft offset to the South.
65-													

25

Project Location: Coffeen Power Station, IL

Log of Boring COF-B006

Sheet 1 of 2

Project	Number:	60440742	
Date(s)	00/06/204	4.00 DM to 08/07/2015 7:20 AM	

Date(s) Drilled	08/06/2015 4:00 PM to 08/07/2015 7:30 AM	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	33.5 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	631.9 ft NAVD88
Borehole Backfill	Cement-Bentonite Grout	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 871654.5 E 2515234.7 (ft NAD83)	Groundwater Level(s)	4 ft on 8/6/2015 5:00:00 PM		

	GT)		S	AMPLES	S	_	v	
: : i	Elevation (fee	⊖Depth (feet)	Type Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAT DESCLIDEN MORTHING (teet) Total Unit Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur Noistur	IARKS
_ 6	630	-	S1	5 13 15	78	\dot{J}	Medium dense, moist, black CINDERS, well-graded sand, with silt, trace fine gravel, (ASH). 6.6	
		- 5	S2	2 2 2	100		Very loose, wet, black CINDERS, with clay, (ASH). 48.3	
—e	25	-	S3		17			
		- 10	S4	3 4 6	100	Ż	Loose, wet, black CINDERS, well-graded sand, 16.7 16.7	
—е 	20	-	55	WOH 2			617.9 14.0 24.4 57 37 1 75	
_ e	615	15— - -		3			Medium stiff, very moist, brown and gray, high plasticity CLAY, trace fine sand, (CH) (NATIVE)	
		- - 20—	S6	1 3 4	67		21.9 1.25	
_ e	510	-						
-		- 25-	S7	1 1 1	100		Very soft, very wet, brown, low plasticity, sandy CLAY, (CL) (TILL). 	
ε	605	-						
-		- 30-	S8	20 50/4"			Hard, very wet, brown and gray, sandy CLAY, (CL) (TILL).	



Report. GEO_SOIL; File N.PROJECTS(60428794_DYNEGY_CCR_RULEASMTSUB_00/10.0_CALCULATIONS_ANALYSIS_DATASITE INVESTIGATION/COFFEENBORINGS(GINT TEMPLATEDYNEGY_COFFEEN_2015, GPJ; 12/29/2015, 5:43:32 PM



Project Number:

Project Location: Coffeen Power Station, IL 60440742

Log of Boring COF-B006A

Sheet 1 of 1

Date(s) Drilled	08/19/2015 7:15 AM to 08/19/2015 10:15 AM	Logged By	A. Grossman	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	25.5 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	632 ft NAVD88
Borehole Backfill	Cement-Bentonite Grout	Sampling Method(s)	Piston	Hammer Data	-
Boring Location	N 871649.5 E 2515234.7 (ft NAD83)	Groundwater Level(s)	Not Encountered		

		S/		S					a)							
Elevation (fee	Depth (feet)	Type Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbol	MATERIA Elevation (feet) 632.0	AL DESCRIPTIO	Depth (feet) 0.0	Natural Moisture Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
— —630 —	-				?	-		-								
_	5	PS-1		75		Loose, moist, black _ , CINDERS, (ASH) 	k, medium to coarse gr).	ained - 								
625 	-				?			-								
 	10- -	PS-2		20		- - -		-								
_	-	PS-3		70		^{619.5} _ Stiff, light grayish b (CL). _	prown, silty CLAY, some	<u>12.5</u> e ash, _ -					1.25			
— —615	-	PS-4		83		-		-					1.25			
_	20-	PS-5		73				-					1.25			
—610 —	-	PS-6		93		- Stiff to very stiff, gra	ayish brown CLAY, (Cl	– L).					2.0			
-	25-					606.5 _ End of	Boring at 25.5 ft	25.5								Auger refusal at 25.5' bgs.
605 	-					-		-								
	30-					1	A =CC) - MC								

28

Project Location: Coffeen Power Station, IL

Log of Boring COF-B007

Sheet 1 of 2

Project Number: 60440742	Sheet 1 of 2				
Date(s) 08/11/2015 12:00 AM to	Logged By	A. Grossman	Checked By	D. Swanson	
Drilling Method Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	37.5 ft	
Drill Rig Type CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	638.0 ft NAVD88	
Borehole Backfill Piezometer COF-P009	Sampling Method(s)	SS / ST	Hammer Data	Automatic	
Boring Location N 873894 E 2515111.2 (ft NAD83)	Groundwater Level(s)	^r Not Encountered			

(j	SAMPLES					Ð										
Elevation (fe	Depth (feet)	Type	Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symb	MATERIAL DESCRIPTION (feet) 638.0	Depth (feet) 0.0	Natural Moistu Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Inde	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
Γ	0-						Very stiff, slightly moist, brown, silty and sand CLAY, some fine to coarse gravel, trace silt	ły								
-	-	Ø	S1	6	89		(CL) (EMBANKMENT FILL).	-	21 4							
F	-	4	01	12				-								
-635	-							-								
-	-		S2		83		CLAY, well sorted, (CL) (EMBANKMENT FIL	L)	23.2	126.8	46	27				
-	5-	E					<u> </u>	_	-							
╞	-	E	S3		100		Medium stiff to stiff, moist, gray trace yellowis	sh -	21.7							
╞	-	E					brown, medium plastic, CLAY, well sorted, w sand, (CL) (EMBANKMENT FILL).	th _	-							
-630	-							-								
F	-	8	S4	2 5	78		 Stiff, moist, brown and gray, medium plastic, CLAY, trace fine sand, well sorted, (CL) 	-	25.3							
╞	10-	R		7			(EMBANKMENT FILL).	_	-							
╞	-	$\left\{ \right\}$					1 	-	-							
╞	-	$\left \right $						-	-							
-625	-						4 4- 4	-	-							
╞	-	Ø	S5	27	83		Very stiff, moist, brown and gray, high plastic	ity, _	21.6		47	28				
F	15-	4		9			(EMBANKMENT FILL).	_				20				
F	-							-	-							
L	-						621.0	17.0	-							
-620	-						-	-	-							
	-	Ø	00	2	0.4		Medium stiff, moist, brown and gray, high	-	0.15							
	20-	8	50	4	94		 plasticity, CLAY, trace fine to coarse sand, (CH). 	_	24.5							
			S7		92			-	22.5	127 0	52	35				
	-		0.				high plasticity, CLAY, with sand, trace gravel	_								
-615	-		58		100		(CH).	_	18.3	128 5	43	27				
	-		00					-	10.0	120.0						
	25															
	23															
	-]						-]							
-610	-	1					- 	20.0								
F	•••	Ø	S9	1	72		Soft, wet to saturated, brown and gray, sandy	20.0	20.3							
F	30-	<u>r/J</u>		<u> </u>	1	1////		_	1	1			I			
<u> </u>							A ECO A	1 -								



Report. GEO_SOIL; FIIe N; PROJECTS/60428794_DYNEGY_CCR_RULEASMT/SUB_00/10.0_CALCULATIONS_ANALYSIS_DATA/SITE INVESTIGATION/COFFEENBORINGS/GINT TEMPLATE/DYNEGY_COFFEEN_2015.GFJ; 12/29/2015.5:43:43 PM
Project Location: Coffeen Power Station, IL

Log of Boring COF-B008

Project	Number:	60440742
		••••••

Date(s) Drilled	08/13/2015 8:00 AM to 08/14/2015 9:30 AM	Logged By	A. Grossman	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	38.5 ft
Drill Rig Type	СМЕ 550Х	Drilling Contractor	Geotechnology	Surface Elevation	635.7 ft NAVD88
Borehole Backfill	Cement-Bentonite Grout	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 873886.6 E 2516222.7 (ft NAD83)	Groundwater Level(s)	Not Encountered		

E F			SA	MPLE	S		φ	
Elevation (fee	Depth (feet)	Type	Number	Sampling Resist. OR Core ROD (%)	Recovery (%)	Graphic Symbo	MATERIAT DESCUENT Moistur Natural Moistur Content (%) Total Unit Weight (pcf) Liquid Limit Pocket Pen. Su (ksf) TOTANDE Su (ksf) Su (k	
—635 — —	- - -		S1	2 4 5	89		Stiff, moist, brown and gray, CLAY, trace sand, with roots, (CL) (EMBANKMENT FILL) 19.7	
 630	- 5		S2	2 1 2	67		Soft, moist, brown, high plasticity, CLAY, with roots, (CL) (EMBANKMENT FILL).	
_	-		S3	1 3 6	78		628.7 7.0 Stiff, moist, brown and gray, medium plastic, 21.7 CLAY, trace sand, with roots, (CH) 21.7 (EMBANKMENT FILL). -	
_ _625	10-		S4	2 4 8	78		21.6 54 36 2.25	
	_		S5		83		Very stiff to hard, moist to wet, dark gray with trace grayish brown, high plasticity, CLAY, with sand, with roots, (CL) (EMBANKMENT FILL).	
_	- - 15—		S6		100		Stiff, wet, light yellowish brown with light gray, medium plastic, CLAY, with sand, with roots, (CL) (EMBANKMENT FILL).	
—620 — —	-							
-	- 20-			2			615.7 20.0 Stiff moist brown and gray medium plastic	
615 	-		S7	3 6	83		CLAY, trace sand, (CH) (NATIVE).	
_ _610 _	25- -		S8	2 6 7	78		Stiff, moist, brown and gray , medium plastic, CLAY, with sand, (CH).	
_	- - 30-		S9		96		Soft, moist to wet, grayish brown, medium - plastic, CLAY, some silt, (CL) (NATIVE) 19.1 128.4 0.5	
L								



32

Report. GEO_SOIL; File N; PROJECTS(60428794_DYNEGY_CCR_RULEASMT)SUB_00/10.0_CALCULATIONS_ANALYSIS_DATASITIE INVESTIGATION/COFFEEN'BORINGS(GINT TEMPLATEDYNEGY_COFFEEN_2015.GPJ; 12/29/2015.5:43:50 PM

Project Number:

Project Location: Coffeen Power Station, IL 60440742

Log of Boring COF-B009

Sheet 1 of 2

Date(s) Drilled	08/13/2015 9:00 AM to 08/13/2015 12:15 PM	Logged By	E. Drumright/A.Grossman	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	40.0 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	635.2 ft NAVD88
Borehole Backfill	Piezometer COF-B010	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 873505 E 2516620.8 (ft NAD83)	Groundwater Level(s)	Not Encountered		

F			SA	MPLE	s	Τ_	ω	
Elevation (fee	⊖ Depth (feet)	Type	Number	Sampling Resist. OR	Recovery (%)	Graphic Symbo	MATERIAL DESCRIPTION 00 Remarks 00 Matural Moistur Matural Moistur Content (%) Total Unit National National Su (ksf) Liquid Limit Torvane Su (ksf) Su (ksf) Torvane	
	-		S1	3 4 6	83	2	Medium dense, moist, 2" sandy clay layer over 12" black, fine to coarse CINDERS, (ASH). 2.6	
— —630	5-		S2	1 2 3	44		631.7 3.5 Medium stiff, very moist, 2" ash layer over - brown and gray, silty CLAY, trace sand, (CH) 26.1 (EMBANKMENT FILL). -	
-	-		S3	2 4 5	83		Stiff, very moist, dark grayish brown, high plasticity, CLAY, (CH) (EMBANKMENT FILL). 17.1 58 40 2.0	
_ 625	- 10-		S4	2 5 6	61		Stiff, very moist, brown and gray, CLAY, trace sand, (CH) (EMBANKMENT FILL). 22.4 2.0	
-	-							
_ 620	- 15-		S5		79		Very stiff, very moist, olive gray and grayish brown, high plasticity, CLAY, with sand, (CH) (EMBANKMENT FILL). 619.7 15.5 18.4 131.5 50 35 2.25	
-	-		S6		75		Stiff, very moist, brown and gray, CLAY, (CL-CH) (EMBANKMENT FILL). 23.3 128.5 1.75	
_ 615	20-		S7	1 3 6	100		29.3 1.75	
-	-							
_ _ _610	- 25-		S8	1 3 5	89		Medium stiff, very moist, brown and gray, high plasticity, CLAY, trace sand, (CL) (EMBANKMENT FILL).	
_	-	-					<u>608.2</u> 27.0	
_	30-		S9		92		Very stiff, brown and gray, high plasticity, CLAY, (CH) (NATIVE).	

AECOM 33



Report. GEO_SOIL; FIIe N; PROJECTS/60428794_DYNEGY_CCR_RULEASMT/SUB_00/10.0_CALCULATIONS_ANALYSIS_DATA/SITE INVESTIGATION/COFFEENBORINGS/GINT TEMPLATE/DYNEGY_COFFEEN_2015.GFJ; 12/29/2015.5:43:57 PM

Project Location: Coffeen Power Station, IL

Log of Boring COF-B010

Sheet 1 of 2

Project Number: 60440742

Date(s) Drilled	08/10/2015 3:00 PM to 08/11/2015 10:00 AM	Logged By	E. Drumright/A.Grossman	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	HSA	Borehole Depth	44.0 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	642.2 ft NAVD88
Borehole Backfill	Piezometer COF-P012	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 872690.8 E 2515090.7 (ft NAD83)	Groundwater Level(s)	10 ft on 8/10/2015 3:30:00 PM		

f)		S/		5				e							
Elevation (fee	Depth (feet)	Type Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAL DESCRIPTION	epth feet) 0.0	Natural Moistul Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
640	-	S1	3 6 7	89		Stiff, moist, brown and gray, CLAY, trace fine gravel and ash, with orange-brown seams, (CL) (FILL).	-	15.1				4.5			
_	5	S2	3 7 7	72	?	 638.5 Medium dense, moist, 2" layer of well-graded SAND, over medium to coarse grained CINDERS (ASH). 	3.7	5.8							
— —635	-	S3	8 14 16	72	?	Dense, moist, black medium to coarse grained CINDERS (ASH).	-	3.8							
_	10-	S4	3 9 13	72		Medium dense, moist to wet, black medium to coarse grained, CINDERS (ASH).	_	9.6							
 630	-				<u>,</u>		-								
_	15-	S5	2 7 12	89	?	 Medium dense, wet, black medium to coarse grained CINDERS, (ASH). 	_	14.2							
— —625 —	-				2		_								
_	20-	S6	4 6 6	67		Addium dense, black, non-plastic, CINDERS, (ASH).		14.8		NP	NP				
-620 -	-						_								
_	25- _	S7	3 3 5	100		Medium stiff, wet, brown and gray, CLAY, trace fine sand, (CH) (NATIVE).	<u>24.0</u> 	3.9/24.1							
—615 —	-		-			513.7	28.5								
_	30-	S8	3 3 5	100		Medium stiff, brown and gray, high plasticity, sandy CLAY, (CL).	-	21.9		48	31				
<u> </u>						AECOM	_								



Report. GEO_SOIL; FIIe N; PROJECTS/60428794_DYNEGY_CCR_RULEASMT/SUB_00/10.0_CALCULATIONS_ANALYSIS_DATA/SITE INVESTIGATION/COFFEENBORINGS/GINT TEMPLATE/DYNEGY_COFFEEN_2015.GFJ; 12/29/2015.5:44:04 PM

Project Number: 60440742

Project Location: Coffeen Power Station, IL

Log of Boring COF-B011

Date(s) Drilled	08/11/2015 9:45 AM to 08/11/2015 1:15 PM	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hollow Stem Auger	Drill Bit Size/Type	7.5 inch O.D. HSA	Borehole Depth	40.0 ft
Drill Rig Type	CME 550X	Drilling Contractor	Geotechnology	Surface Elevation	635.1 ft NAVD88
Borehole Backfill	Piezometer COF-P014	Sampling Method(s)	SS / ST	Hammer Data	Automatic
Boring Location	N 872538.3 E 2515411.6 (ft NAD83)	Groundwater Level(s)	Not Encountered		

F		S	AMPLE	s	-		ø							
Elevation (fee	Depth (feet)	Type Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	MATERIAL DESCRIPTION Elevation (feet) 635.1 Depth (feet) 0.0	Natural Moistur Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
	-	S1	2 3 6	67		Stiff, moist, brown and gray, CLAY, with fine to medium sand, with roots, (CL-CH) (EMBANKMENT FILL).	23.9				2.5			
	-	S2	2 4 5	94		631.6	23.5		67	47	1.75			
	-	S3	2 4 6	89		Stiff, moist, brown and gray, CLAY, trace fine	26.1				2.0			
_ _ 625	- - 10	S5	30/3" 4 8 8	<u>17</u> 83		Very stiff, very moist, brown and gray, medium	21.8 10.9		32	18	4.0			S4: Concrete or boulder chips in tip of spoon. Drill through to continue.
-	-					limestone rock in middle of sample, (CL) - (EMBANKMENT FILL) 								
 620	- - 15-	S6		58		Very stiff, yellowish brown with grayish brown, medium plastic, sandy CLAY, (CL) (EMBANKMENT FILL).	11.5	139.4	34	20	4.0			Gravel in tip. Bent
_	-	S7		54		Hard, very moist, yellowish brown trace gray, medium plastic, sandy CLAY, trace gravel, (CL) (EMBANKMENT FILL).	12.0	134.1	30	16	4.5+			tip of tube.
— —615	- 20—	S8	2 3 6	83		Stiff, very moist, brown and gray, CLAY, with orange-brown silt seams, (CL-CH) (NATIVE).	27.3				1.75			
-	-													
— —610	25-	S9	2 4 5	100		Stiff, brown and gray, CLAY, trace fine to medium sand, (CL)	19.4				1.5			
-	-													
-	- 30	S10	1 2 3	100		Medium stiff, brown and gray, medium plastic, 605.6sandy CLAY, trace fine gravel, (CL). 29.5	17.3		32	17	0.25			

Log of Boring COF-B011 Project Location: Coffeen Power Station, IL Sheet 2 of 2 Project Number: 60440742 SAMPLES Elevation (feet) Natural Moisture Content (%) Graphic Symbol Plasticity Index Sampling Resist. SDepth (feet) Recovery (%) OR Core RQD (%) Total Unit Weight (pcf) Pocket Pen. -iquid Limit TXUU (ksf) MATERIAL DESCRIPTION REMARKS Type Number Torvane Su (ksf) Su (ksf) Elevation (feet) Depth (feet) -605 Soft to stiff, wet, with orange brown, sandy CLAY TILL, trace fine sand, with orange-brown S11 100 9.9 144.5 silt seams, (CL). 32.0 Brown and gray, sand CLAY, trace fine gravel, S12 8.9 02.1 (CL). 33.0 Very dense, very moist, gray, silty clayey 16 32 41 SAND, trace gravel, with orange-brown silt seams, (SC-SM) (TILL). S13 94 9.3 4.5 +600 35 38.5 Hard, wet, gray, sandy silty CLAY, trace fine grave, (CL) (TILL). 20 46 S14 83 9.8 4.5+ 50/4" 40.0 40 595 End of Boring at 40 ft Installed Piezometer COF-P014 in boring. 45 590 50 585 55 580 575 60 65

Report GEO SOIL; FIE N; PROJECTS/60428794_DYNEGY_CCR_RULEASMTISUB_00/10,0_CALCULATIONS_ANALYSIS_DATAISITE INVESTIGATION/COFFEENBORINGS/GINT TEMPLATE/DYNEGY_COFFEEN_2015.GPJ; 12/29/2015.5:44:11 PM

Project: Dynegy

Project Number:

Project Location: Coffeen Power Station, IL

60440742

Log of Boring COF-HA1

Date(s) Drilled	08/30/2015 12:00 AM to	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hand Auger	Drill Bit Size/Type	2.25" O.D Hand Auger	Borehole Depth	8.0 ft
Drill Rig Type		Drilling Contractor	AECOM	Surface Elevation	610.5 ft NAVD88
Borehole Backfill	Drill Cuttings	Sampling Method(s)	Bag	Hammer Data	None
Boring Location	N 872484.5 E 2516572.3 (ft NAD83)	Groundwater Level(s)	7.3' ft on 8/30/2015		

l (ji		SA	MPLES	<u>Ş</u>					e							
Elevation (fee	Depth (feet)	Type Number	Sampling Resist. OR Core RQD (%)	Recovery (%)	Graphic Symbo	Ele (fee		Depth (feet) 0.0	Natural Moistu Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
-610	0-	S1					Very stiff to stiff, moist to very moist, brown to brown and gray, medium plastic, CLAY	0	12.9							
_	_	S2					(CL-CH), (FILL).	-	13.3		35	14				
-	-	S3				607	5	30	21.7							
-	_	S4					Stiff to very stiff, very moist, brown and gray,		19.0							
-	_	S5						-	18.1							
-605	5-	S6				604	5	60	18.0							
-	-	S7					Stiff, very wet, brown and gray, high plasticity	/,	16.0		35	21				
-	-	S8				602	5	80	22.9							
_	-					002	End of Boring at 8 ft	0.0	1							
_	-					F		-								
-600	10-							_								
-	_					F		-	1							
_	-							-								
-	_					F		-								
_	-					F		-	-							
-595	15-															
F	-					F		-	-							
-	-					F		-	-							
_	-					F		-	-							
_	-					F		-	-							
-590	20-					F		_	-							
-	-					F		-								
_	-					F		-								
_	-					F		-	-							
_	-					F		-	-							
-585	25—					F		_	-							
L	-					F		-	-							
L	-					F		-	-							
L	-					F		-	-							
F	-					F		-	-							
	30-		I		1	1						l			<u> </u>	
<u> </u>							AECOA	A -								

Project Number:

Project Location: Coffeen Power Station, IL

60440742

Log of Boring COF-HA2

Date(s) Drilled	08/30/2015 12:00 AM to	Logged By	E. Drumright	Checked By	D. Swanson
Drilling Method	Hand Auger	Drill Bit Size/Type	2.25" O.D Hand Auger	Borehole Depth	5.5 ft
Drill Rig Type		Drilling Contractor	AECOM	Surface Elevation	594.6 ft NAVD88
Borehole Backfill	Drill Cuttings	Sampling Method(s)	Bag	Hammer Data	None
Boring Location	N 873496.7 E 2516761.2 (ft NAD83)	Groundwater Level(s)	Not Encountered		

Elevation Oepth (fe	Tvne	Number	Sampling Resist. OR	Core RQD (%)	Recovery (%)	Graphic Symbo	Ele (fe 594	MATERIAL DESCRIPTION	Depth (feet) 0.0	Natural Moisture Content (%)	Total Unit Weight (pcf)	Liquid Limit	Plasticity Index	Pocket Pen. Su (ksf)	Torvane Su (ksf)	TXUU (ksf)	REMARKS
-		S1					<u>59</u> 3	^{3.9} <u>cinders and organics, (CH) (TOPSOIL).</u>	0.7	18.6		51	25				
-		S2						Very stiff, very moist, brown, CLAY, (CL).	-	20.9		47	27				
_		S3					591	1. <u>6</u>	3 <u>.0_</u>	22.8							
-		S4						Stiff to very stiff, very moist, brown, high plasticity, CLAY, trace sand to sandy, none to) _	15.8							
-590 5-		S5						trace fine gravel, (CL) (TILL).		13.4		43	27				
- 3		S6		_		////	589	9.1 End of Boring at 5.5 ft	5.5	11.4							
_							Γ		-								
_							F		-	1							
_							F		-	-							
-585							F		-	-							
10-							F			-							
							F		-	-							
							F		-	-							
							F		-	-							
- 							ŀ		-	-							
⁻⁵⁸⁰ 15 -	-						╞			-							
-							ŀ		-	-							
-							ŀ		-	-							
-							ŀ		-	-							
-							Ļ		-	-							
-575 20 -							L			_							
							L		_								
_							L		_								
-									_								
-	1						[-]							
-570	1						ſ		-	1							
25-	1						F			1							
L							F		-	-							
							F		-	-							
	$\left \right $						F		-	-							
-565							F		-	-							
30-							<u> </u>	A=004	A								

Project Location: COFFEEN POWER STATION, ILLINOIS

Project Number: 60440742

Log of Piezometer

Piezometer Location	COF-P000	Date Installed	8/7/2015	Time
Installed By		Observed By	AECOM	Total 25 FT Depth 25
Method of Installation	HOLLOW STEM AUGER	Drilling Contractor	GEOTECHNOLOGY, INC.	Surface Elevation
Screened Interval	15-25 FT	Completion Zone	EMBANKMENT/FOUNDATION	
Remarks	OFFSET 5 FT SOUTH FROM COF-B001	Groundwater Level(s)	17.9 FT ON 8/29/2015	





Project Location: COFFEEN POWER STATION, ILLINOIS

Project Number: 60440742

Log of Piezometer

Piezometer Location	COF-P002	Date Installed	8/6/2015	Time	
Installed By		Observed By	AECOM	Total Depth	35.5 FT
Method of Installation	HOLLOW STEM AUGER	Drilling Contractor	GEOTECHNOLOGY, INC.	Surface Elevation	635.4 FT
Screened Interval	25.5-35.5 FT	Completion Zone	POND #1 FOUNDATION		
Remarks	INSTALL IN COF-B002	Groundwater Level(s)	10.5 FT ON 8/29/2015		



Project: Dynegy Project Location: COFFEEN POWER STATION, ILLINOIS Project Number: 60440742

Log of Piezometer

Piezometer Location	COF-P003	Date Installed	8/5/2015	Time	
Installed By		Observed By	AECOM	Total Depth	43.5 FT
Method of Installation	HOLLOW STEM AUGER	Drilling Contractor	GEOTECHNOLOGY, INC.	Surface Elevation	635.7 FT
Screened Interval	28.5-43.5 FT	Completion Zone	POND #1 FOUNDATION		
Remarks	INSTALL IN COF-B003	Groundwater Level(s)	15.3 FT ON 8/29/2015		



Project: Dynegy Log of Piezometer Project Location: COFFEEN POWER STATION, ILLINOIS Sheet 1 of 1 Project Number: 60440742 Piezometer Date Time COF-P005 8/5/2015 Installed Location Installed Observed Total AECOM 20 FT Βv Depth By Method of Installation Drilling Contractor Surface Elevation HOLLOW STEM AUGER GEOTECHNOLOGY, INC. 635.0 FT Completion Zone Screened 15-20 FT EMBANKMENT Interval Remarks Groundwater OFFSET 5' WEST OF COF-B004 14.4 FT ON 8/29/2015 Level(s) NUMBER OF BOLLARDS: 3 BOLLARD TYPE AND DIAMETER: 3"Ø STEEL STEEL PROTECTIVE CASING AND CAP 4" SQ. T STICK-UP ELEVATION: 2-3 FT (VARIES) Ę Ŀ 2" SCH. 40 PVC CAP CONC. PAD 1.5-2 FT SQ -× × × 44 4 4 44 4 44 4,4 *d* * 4 * 4 44 4 1₄ ġ, . i⁴ . w . w. .4 DEPTH TO BOTTOM OF CONCRETE PAD: 3-4 IN TYP. 4 ΤΥΡ. E CONCRETE 4 DEPTH TO BOTTOM OF SURFACE CASING: 2-3 FT TYP. 2 TYPE AND DIAMETER OF RISER PIPE: 2" NOM. SCH 40 PVC PRE-MIXED CEMENT-BENTONITE GROUT è DEPTH TO TOP OF SEAL: 11 FT _ BENTONITE SEAL TYPE AND THICKNESS: PELLETS (2FT NOM.) 10 DEPTH TO TOP OF FILTER PACK:13 FT SAND FILTER PACK TYPE: COMMERCIAL SCREENED MED./COARSE . DEPTH TO TOP OF SCREEN:15 FT_ TYPE OF SCREEN: MACHINE-SLOTTED PVC SCH 40

NOT TO SCALE

DEPTH TO BOTTOM OF SCREEN: 20 FT_

DEPTH OF BOTTOM TO PLUGGED_ BLANK CASING: 20.5 FT DEPTH OF BORING: 20.5 FT_

46

7.75 IN

SCREEN SLOT SIZE: 0.010 IN.

Project Location: COFFEEN POWER STATION, ILLINOIS

Project Number: 60440742

Log of Piezometer

Piezometer Location	COF-P006	Date Installed	8/10/2015	Time	
Installed By		Observed By	AECOM	Total Depth	45 FT
Method of Installation	HOLLOW STEM AUGER	Drilling Contractor	GEOTECHNOLOGY, INC.	Surface Elevation	635.1 FT
Screened Interval	30-45 FT	Completion Zone	POND #1 FOUNDATION		
Remarks	OFFSET 5' SOUTH FROM COF-B005	Groundwater Level(s)	22.2 FT ON 8/29/2015		







Project: Dynegy Log of Piezometer Project Location: COFFEEN POWER STATION, ILLINOIS Sheet 1 of 1 Project Number: 60440742 Piezometer Date Time 8/12/2015 COF-P009 Installed Location Installed Observed Total AECOM 37.5 FT Βv Depth By Method of Installation Drilling Contractor Surface Elevation HOLLOW STEM AUGER GEOTECHNOLOGY, INC. 638.0 FT Completion Zone Screened 20-30 FT POND #2 FOUNDATION Interval Remarks Groundwater INSTALL IN COF-B007 Level(s) NUMBER OF BOLLARDS: 3 BOLLARD TYPE AND DIAMETER: 3"Ø STEEL STEEL PROTECTIVE CASING AND CAP 4" SQ. T STICK-UP ELEVATION: 2-3 FT (VARIES) É Ŀ 2" SCH. 40 PVC CAP CONC. PAD 1.5-2 FT SQ -× × × 44 4 4 44 4 44 44 *d + + + + + 41 4 . Ġ. 44 . w. ra .4 DEPTH TO BOTTOM OF CONCRETE PAD: 3-4 IN TYP. à ΤΥΡ. F CONCRETE 4 4 4 DEPTH TO BOTTOM OF SURFACE CASING: 2-3 FT TYP. 2 TYPE AND DIAMETER OF RISER PIPE: 2" NOM. SCH 40 PVC PRE-MIXED CEMENT-BENTONITE GROUT DEPTH TO TOP OF SEAL:16 FT_ BENTONITE SEAL TYPE AND THICKNESS: PELLETS (2FT NOM.) 10 DEPTH TO TOP OF FILTER PACK: 18 FT SAND FILTER PACK TYPE: COMMERCIAL SCREENED MED./COARSE . DEPTH TO TOP OF SCREEN: 20 FT_ Ŧ TYPE OF SCREEN: MACHINE-SLOTTED PVC SCH 40 DEPTH TO BOTTOM OF SCREEN: 30 FT_ SCREEN SLOT SIZE: 0.010 IN. DEPTH OF BOTTOM TO PLUGGED_ BLANK CASING: 30.5 FT i. DEPTH OF BORING: 37.5 FT_ GROUT 30.5-37.5 ON 8/11/2015 SET OVERNIGHT 7.75 IN NOT TO SCALE

Log of Piezometer Project Location: COFFEEN POWER STATION, ILLINOIS Sheet 1 of 1 Project Number: 60440742 Piezometer Date Time 8/13/2015 COF-P010 Installed Location Installed Observed Total AECOM 38.5 FT Βv Depth By Method of Installation Drilling Contractor Surface Elevation HOLLOW STEM AUGER GEOTECHNOLOGY, INC. 635.2 FT Completion Zone Screened 23 5-38 5 FT POND #2 FOUNDATION Interval Remarks Groundwater INSTALL IN COF-B009 29.6 FT ON 8/29/2015 Level(s) NUMBER OF BOLLARDS: 3 BOLLARD TYPE AND DIAMETER: 3"Ø STEEL STEEL PROTECTIVE CASING AND CAP 4" SQ. T STICK-UP ELEVATION: 2-3 FT (VARIES) 11 Ē Ŀ 2" SCH. 40 PVC CAP CONC. PAD 1.5-2 FT SQ -× × × 44 4 4 44 4 44 4,4 *d* * 4 * 4 41 4 14 Č. . 1 **M** 14 .4 DEPTH TO BOTTOM OF CONCRETE PAD: 3-4 IN TYP. ۰. TYP. F CONCRETE 4 DEPTH TO BOTTOM OF SURFACE CASING: 2-3 FT TYP. 2 TYPE AND DIAMETER OF RISER PIPE: 2" NOM. SCH 40 PVC PRE-MIXED CEMENT-BENTONITE GROUT 4 DEPTH TO TOP OF SEAL: 19.5 FT_ BENTONITE SEAL TYPE AND THICKNESS: PELLETS (2FT NOM.) 10 DEPTH TO TOP OF FILTER PACK: 21.5 FT SAND FILTER PACK TYPE: COMMERCIAL SCREENED MED./COARSE . DEPTH TO TOP OF SCREEN: 23.5 FT TYPE OF SCREEN: MACHINE-SLOTTED PVC SCH 40 DEPTH TO BOTTOM OF SCREEN: 38.5 FT_ SCREEN SLOT SIZE: 0.010 IN. DEPTH OF BOTTOM TO PLUGGED_ BLANK CASING: 39 FT DEPTH OF BORING: 39 FT _ 7.75 IN

NOT TO SCALE

Project: Dynegy

Log of Piezometer Project Location: COFFEEN POWER STATION, ILLINOIS Sheet 1 of 1 Project Number: 60440742 Piezometer Date Time 8/11/2015 COF-P012 Installed Location Installed Observed Total AECOM 42.5 FT Βv Depth By Method of Installation Drilling Contractor Surface Elevation HOLLOW STEM AUGER GEOTECHNOLOGY, INC. 635.1 FT Completion Zone Screened 32 5-42 5 FT POND #2 FOUNDATION Interval Remarks Groundwater INSTALL IN COF-B010 16.8 FT ON 8/29/2015 Level(s) NUMBER OF BOLLARDS: 3 BOLLARD TYPE AND DIAMETER: 3"Ø STEEL STEEL PROTECTIVE CASING AND CAP 4" SQ. T STICK-UP ELEVATION: 2-3 FT (VARIES) 11 Ē Ŀ 2" SCH. 40 PVC CAP CONC. PAD 1.5-2 FT SQ -× × × 44 4 4 44 4 4 4,4 *d* * 4 * 4 4 4 14 Ġ. 1 w.4 DEPTH TO BOTTOM OF CONCRETE PAD: 3-4 IN TYP. ۰. TYP. F CONCRETE 4 DEPTH TO BOTTOM OF SURFACE CASING: 2-3 FT TYP. 2 TYPE AND DIAMETER OF RISER PIPE: 2" NOM. SCH 40 PVC PRE-MIXED CEMENT-BENTONITE GROUT 4 DEPTH TO TOP OF SEAL: 28.5 FT_ BENTONITE SEAL TYPE AND THICKNESS: PELLETS (2FT NOM.) 10 DEPTH TO TOP OF FILTER PACK: 30.5 FT SAND FILTER PACK TYPE: COMMERCIAL SCREENED MED./COARSE . DEPTH TO TOP OF SCREEN: 32.5 FT_ TYPE OF SCREEN: MACHINE-SLOTTED PVC SCH 40 DEPTH TO BOTTOM OF SCREEN: 42.5 FT_ SCREEN SLOT SIZE: 0.010 IN. DEPTH OF BOTTOM TO PLUGGED_ BLANK CASING: 43 FT DEPTH OF BORING: 43 FT_ 7.75 IN NOT TO SCALE

Project: Dynegy

Log of Piezometer Project Location: COFFEEN POWER STATION, ILLINOIS Sheet 1 of 1 Project Number: 60440742 Piezometer Date Time 8/11/2015 COF-P014 Installed Location Installed Observed Total AECOM 38.5 FT Βv Depth By Method of Installation Drilling Contractor Surface Elevation HOLLOW STEM AUGER GEOTECHNOLOGY, INC. 642.2 FT Completion Zone Screened 28 5-38 5 FT POND #2 FOUNDATION Interval Remarks Groundwater INSTALL IN COF-B011 21.5 FT ON 8/29/2015 Level(s) NUMBER OF BOLLARDS: 3 BOLLARD TYPE AND DIAMETER: 3"Ø STEEL STEEL PROTECTIVE CASING AND CAP 4" SQ. T STICK-UP ELEVATION: 2-3 FT (VARIES) 11 Ę Ŀ 2" SCH. 40 PVC CAP CONC. PAD 1.5-2 FT SQ -× × × 44 4 44 4 4 4,4 *d* * * * * 41 4 1₄ Č. . 14 .4 DEPTH TO BOTTOM OF CONCRETE PAD: 3-4 IN TYP. ۰. TYP. E CONCRETE 4 DEPTH TO BOTTOM OF SURFACE CASING: 2-3 FT TYP. 2 TYPE AND DIAMETER OF RISER PIPE: 2" NOM. SCH 40 PVC PRE-MIXED CEMENT-BENTONITE GROUT 4 DEPTH TO TOP OF SEAL: 24.5 FT_ BENTONITE SEAL TYPE AND THICKNESS: PELLETS (2FT NOM.) 10 DEPTH TO TOP OF FILTER PACK: 26.5 FT_ SAND FILTER PACK TYPE: COMMERCIAL SCREENED MED./COARSE . DEPTH TO TOP OF SCREEN: 28.5 FT_ TYPE OF SCREEN: MACHINE-SLOTTED PVC SCH 40 DEPTH TO BOTTOM OF SCREEN: 38.5 FT_ SCREEN SLOT SIZE: 0.010 IN. DEPTH OF BOTTOM TO PLUGGED ______ BLANK CASING: 39 FT DEPTH OF BORING: 39 FT_ 7.75 IN NOT TO SCALE

Project: Dynegy

APPENDIX B

HYDRAULIC CONDUCTIVITY TEST RESULTS

APPENDIX B1

HYDRAULIC CONDUCTIVITY ANALYSES (WELLS)



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G105fh.aqt Title: G105 @ CCR Landfill – Falling Head Test Date: 11/11/16 Time: 11:38:43

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G105

AQUIFER DATA

Saturated Thickness: 14.82 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G105

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.65 ft Static Water Column Height: 14.82 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.58 ft Total Well Penetration Depth: 14.82 ft

No. of Observations: 82

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.656	0.953	1.458	5.723	0.939
0.014	1.66	1.023	1.445	6.073	0.911
0.029	1.652	1.093	1.439	6.443	0.878
0.045	1.643	1.173	1.428	6.843	0.845
0.061	1.631	1.263	1.412	7.263	0.817
0.079	1.625	1.343	1.399	7.703	0.789
0.098	1.623	1.443	1.387	8.173	0.754
0.118	1.615	1.543	1.375	8.673	0.723
0.139	1.607	1.643	1.36	9.203	0.688
0.161	1.599	1.753	1.344	9.763	0.653
0.185	1.595	1.873	1.332	10.36	0.62
0.21	1.592	2.003	1.315	10.96	0.589
0.236	1.583	2.133	1.298	11.66	0.556

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.264	1.577	2.273	1.276	12.36	0.521
0.294	1.575	2.423	1.266	13.06	0.49
0.325	1.563	2.583	1.239	13.86	0.457
0.359	1.559	2.743	1.228	14.76	0.422
0.394	1.549	2.923	1.206	15.56	0.395
0.431	1.545	3.113	1.185	16.56	0.359
0.471	1.541	3.313	1.162	17.56	0.328
0.513	1.534	3.523	1.142	18.56	0.299
0.557	1.525	3.743	1.118	19.66	0.269
0.604	1.515	3.983	1.093	20.86	0.242
0.654	1.509	4.233	1.069	22.16	0.218
0.707	1.494	4.493	1.045	23.46	0.187
0.763	1.487	4.773	1.02	24.86	0.164
0.823	1.48	5.073	0.991		
0.883	1.463	5.383	0.97		

SOLUTION

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00015	cm/sec
Ss	6.0E-5	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.06776 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0001464	1.653E-5	+/- 3.29E-5	8.858	cm/sec
Ss	5.669E-5	3.386E-5	+/- 6.739E-5	1.674	ft ⁻¹
Kz/Kr	1.	1.041	+/- 2.071	0.961	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.06614 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.006265 ft ²
	7.93E-5 ft ²
Std. Deviation	0.008905 ft
Mean	-0.002224 ft
No. of Residuals	82
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G105rh.aqt Title: G105 @CCR Landfill – Rising Head Test Date: 11/11/16 Time: 11:43:09

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G105

AQUIFER DATA

Saturated Thickness: 14.82 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G105

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.72 ft Static Water Column Height: 14.82 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.58 ft Total Well Penetration Depth: 14.82 ft

No. of Observations: 112

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.834	0.4953	1.6	4.943	1.251
0.004167	1.767	0.5293	1.596	5.243	1.237
0.008333	1.773	0.5643	1.593	5.553	1.214
0.01505	1.77	0.6013	1.588	5.893	1.2
0.01712	1.758	0.6413	1.585	6.243	1.176
0.02083	1.753	0.6833	1.58	6.613	1.152
0.025	1.757	0.7273	1.577	7.013	1.13
0.02917	1.746	0.7746	1.567	7.433	1.107
0.03612	1.738	0.8243	1.564	7.873	1.087
0.03933	1.74	0.8773	1.561	8.343	1.059
0.04537	1.729	0.9333	1.555	8.843	1.037
0.05595	1.721	0.9933	1.546	9.373	1.012
0.05933	1.716	1.053	1.542	9.933	0.98

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06633	1.714	1.123	1.538	10.53	0.96
0.07433	1.711	1.193	1.528	11.13	0.933
0.08333	1.704	1.263	1.522	11.83	0.901
0.09133	1.698	1.343	1.512	12.53	0.873
0.1013	1.691	1.433	1.51	13.23	0.844
0.1113	1.687	1.513	1.499	14.03	0.813
0.1213	1.682	1.613	1.491	14.93	0.788
0.1323	1.679	1.713	1.482	15.73	0.757
0.1443	1.673	1.813	1.478	16.73	0.722
0.1573	1.665	1.923	1.468	17.73	0.698
0.1703	1.664	2.043	1.456	18.73	0.667
0.1843	1.657	2.173	1.443	19.83	0.636
0.1993	1.657	2.303	1.441	21.03	0.606
0.2153	1.647	2.443	1.427	22.33	0.577
0.2313	1.644	2.593	1.416	23.63	0.549
0.2493	1.644	2.753	1.404	25.03	0.521
0.2683	1.637	2.913	1.388	26.53	0.494
0.2883	1.63	3.093	1.376	28.13	0.469
0.3093	1.629	3.283	1.37	29.73	0.443
0.3313	1.624	3.483	1.349	31.53	0.422
0.3553	1.623	3.693	1.339	33.43	0.397
0.3803	1.619	3.913	1.323	35.43	0.377
0.4063	1.614	4.153	1.305	37.53	0.355
0.4343	1.608	4.403	1.286		
0.4643	1.604	4.663	1.271		

SOLUTION

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	5.7E-5	cm/sec
Ss	0.001	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.02575 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G106fh.aqt Title: G106 @ CCR Landfill – Falling Head Test Date: 12/01/16 Time: 09:04:15

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G106

AQUIFER DATA

Saturated Thickness: 12.5 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G106

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.67 ft Static Water Column Height: 12.5 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.58 ft Total Well Penetration Depth: 12.5 ft

No. of Observations: 121

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.00455	4.279	0.282	1.547	2.98	1.317
0.008333	6.825	0.298	1.542	3.16	1.316
0.0125	4.654	0.316	1.537	3.35	1.306
0.01667	2.121	0.335	1.534	3.55	1.303
0.02083	2.856	0.355	1.527	3.76	1.294
0.025	3.905	0.376	1.518	3.98	1.29
0.02917	3.468	0.398	1.516	4.22	1.283
0.03745	2.726	0.422	1.505	4.47	1.274
0.04113	2.798	0.447	1.499	4.73	1.271
0.0448	1.772	0.473	1.496	5.01	1.263
0.04847	0.514	0.501	1.487	5.31	1.257
0.05212	1.755	0.531	1.477	5.62	1.25
0.0603	2.176	0.562	1.474	5.96	1.24

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06397	2.082	0.596	1.444	6.31	1.229
0.06/6/	1.996	0.631	1.461	6.68	1.226
0.07133	1.85	0.668	1.451	7.08	1.213
0.075	1.264	0.708	1.444	/.5	1.202
0.07865	1.141	0.75	1.44	7.94	1.191
0.08232	1.388	0.794	1.432	8.41	1.183
0.08598	1.706	0.841	1.421	8.91	1.168
0.08965	1.859	0.891	1.42	9.44	1.159
0.09332	1.805	0.944	1.414	10.	1.148
0.09697	1.651	1.	1.408	10.6	1.137
0.1006	1.533	1.06	1.399	11.2	1.127
0.106	1.534	1.12	1.396	11.9	1.11
0.112	1.644	1.19	1.386	12.6	1.095
0.1208	1.639	1.26	1.382	13.3	1.085
0.126	1.59	1.33	1.378	14.1	1.068
0.133	1.594	1.41	1.374	15.	1.048
0.141	1.623	1.5	1.371	15.8	1.031
0.15	1.59	1.58	1.367	16.8	1.013
0.158	1.592	1.68	1.362	17.8	0.992
0.168	1.594	1.78	1.358	18.8	0.976
0.178	1.583	1.88	1.353	19.9	0.953
0.188	1.584	1.99	1.351	21.1	0.929
0.199	1.58	2.11	1.344	22.4	0.909
0.211	1.573	2.24	1.343	23.7	0.884
0.224	1.569	2.37	1.337	25.1	0.858
0.237	1.565	2.51	1.332	26.6	0.833
0.251	1.56	2.66	1.328		
0.266	1.552	2.82	1.327		

SOLUTION

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice In(Re/rw): 3.694

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	4.0E-5	cm/sec
y0	1.45	ft

 $T = K^*b = 0.01524 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G106rh.aqt Title: G106 @CCR Landfill – Rising Head Test Date: 11/18/16 Time: 11:53:58

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G106

AQUIFER DATA

Saturated Thickness: 12.5 ft Anisotropy Ratio (Kz/Kr): 0.001

SLUG TEST WELL DATA

Test Well: G106

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.25 ft Static Water Column Height: 12.5 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.58 ft Total Well Penetration Depth: 12.5 ft

No. of Observations: 102

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	0.763	0.2487	0.152	1.957	0.086
0.00785	0.098	0.2647	0.151	2.077	0.081
0.01152	0.02	0.2827	0.149	2.207	0.08
0.0152	0.329	0.3017	0.149	2.337	0.082
0.01887	0.187	0.3217	0.143	2.477	0.073
0.02662	0.179	0.3427	0.14	2.627	0.074
0.03028	0.248	0.3647	0.14	2.787	0.073
0.03395	0.264	0.3887	0.14	2.947	0.071
0.03762	0.175	0.4137	0.132	3.127	0.067
0.04558	0.225	0.4397	0.133	3.317	0.068
0.04927	0.226	0.4677	0.132	3.517	0.067
0.05293	0.217	0.4977	0.127	3.727	0.059
0.05662	0.202	0.5287	0.128	3.947	0.059
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	<u>Time (min)</u>	Displacement (ft)
------------	-------------------	------------	-------------------	-------------------	-------------------
0.06387	0.2	0.5627	0.125	4.187	0.059
0.06753	0.21	0.5977	0.124	4.437	0.053
0.07125	0.21	0.6347	0.122	4.697	0.058
0.07532	0.208	0.6747	0.117	4.977	0.051
0.079	0.204	0.7167	0.116	5.277	0.049
0.08268	0.202	0.7607	0.117	5.587	0.05
0.08637	0.196	0.8077	0.114	5.927	0.041
0.09267	0.198	0.8577	0.11	6.277	0.04
0.09967	0.199	0.9107	0.109	6.647	0.038
0.1077	0.196	0.9667	0.106	7.047	0.031
0.1167	0.186	1.027	0.108	7.467	0.004
0.1247	0.187	1.087	0.104	7.907	0.028
0.1347	0.178	1.157	0.104	8.377	0.027
0.1447	0.177	1.227	0.1	8.877	0.023
0.1547	0.174	1.297	0.097	9.407	0.02
0.1657	0.174	1.377	0.095	9.967	0.016
0.1777	0.168	1.467	0.093	10.57	0.012
0.1907	0.164	1.547	0.093	11.17	0.007
0.2037	0.159	1.647	0.09	11.87	0.006
0.2177	0.161	1.747	0.089	12.57	0.005
0.2327	0.155	1.847	0.087	13.27	0.001

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice In(Re/rw): 7.28

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.00074	cm/sec
уО	0.14	ft

 $T = K^*b = 0.2819 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G107fh.aqt Title: G107 @ CCR Landfill – Falling Head Test Date: 11/28/16 Time: 10:25:40

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G107

AQUIFER DATA

Saturated Thickness: 10.86 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G107

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.35 ft Static Water Column Height: 10.86 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.48 ft Total Well Penetration Depth: 10.86 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
-0.03917	2.789	0.3245	2.143	3.063	1.797	
-0.035	2.535	0.3495	2.134	3.252	1.775	
-0.03078	2.269	0.3755	2.132	3.453	1.761	
-0.02297	2.403	0.4035	2.123	3.663	1.741	
-0.01928	2.442	0.4335	2.117	3.882	1.719	
-0.01558	2.401	0.4645	2.113	4.122	1.699	
-0.0119	2.368	0.4985	2.104	4.372	1.678	
-0.003683	2.371	0.5335	2.096	4.633	1.652	
0.	2.38	0.5705	2.092	4.912	1.628	
0.003683	2.362	0.6105	2.081	5.213	1.605	
0.00735	2.355	0.6525	2.071	5.523	1.59	
0.01582	2.349	0.6965	2.071	5.862	1.55	
0.01953	2.344	0.7435	2.061	6.213	1.529	

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.02323	2.335	0.7935	2.052	6.582	1.495
0.0285	2.372	0.8465	2.046	6.982	1.468
0.03668	2.321	0.9025	2.042	7.402	1.436
0.0435	2.304	0.9625	2.029	7.843	1.402
0.05523	2.301	1.023	2.023	8.313	1.369
0.0605	2.292	1.093	2.01	8.813	1.336
0.07527	2.276	1.163	2.005	9.342	1.304
0.0805	2.27	1.232	1.998	9.902	1.267
0.09568	2.262	1.313	1.984	10.5	1.229
0.1015	2.25	1.403	1.972	11.1	1.191
0.1135	2.236	1.482	1.962	11.8	1.154
0.1265	2.226	1.583	1.95	12.5	1.11
0.1395	2.215	1.683	1.944	13.2	1.078
0.1535	2,212	1.783	1,931	14.	1.032
0.1685	2.204	1.893	1.914	14.9	0.985
0.1845	2.195	2.013	1.903	15.7	0.943
0.2005	2,182	2.143	1.888	16.7	0.897
0.2185	2.177	2.273	1.877	17.7	0.854
0.2375	2.174	2.413	1.858	18.7	0.807
0.2575	2.165	2.563	1.846	19.8	0.76
0.2785	2.155	2.723	1.83		0110
0.3005	2.148	2.882	1.813		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	6.3E-5	cm/sec
Ss	0.0015	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.02085 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	4.079E-5	3.684E-6	+/- 7.31E-6	11.07	cm/sec
Ss	0.008117	0.001889	+/- 0.003747	4.297	ft ⁻¹
Kz/Kr	1.	1.82	+/- 3.611	0.5494	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

$T = K^*b = 0.0135 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.96	-0.37
Ss	-0.96	1.00	0.19
Kz/Kr	-0.37	0.19	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.3683 ft ²
Variance	0.003683 ft ²
Std. Deviation	0.06069 ft
Mean	0.002549 ft
No. of Residuals	103
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G107rh.aqt Title: G107 @CCR Landfill – Rising Head Test Date: 11/18/16 Time: 11:55:10

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G107

AQUIFER DATA

Saturated Thickness: 10.86 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G107

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.45 ft Static Water Column Height: 10.86 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.48 ft Total Well Penetration Depth: 10.86 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.826	0.3302	1.381	2.934	1.161
0.004167	1.704	0.3522	1.385	3.114	1.151
0.008333	1.594	0.3762	1.372	3.304	1.138
0.0125	1.576	0.4012	1.377	3.504	1.121
0.01975	1.555	0.4272	1.365	3.714	1.111
0.02345	1.531	0.4552	1.365	3.934	1.097
0.02713	1.53	0.4852	1.356	4.174	1.084
0.0308	1.513	0.5162	1.355	4.424	1.068
0.03893	1.479	0.5502	1.352	4.684	1.05
0.04263	1.493	0.5852	1.348	4.964	1.036
0.04632	1.498	0.6222	1.346	5.264	1.015
0.04998	1.49	0.6622	1.34	5.574	1.
0.05843	1.479	0.7042	1.334	5.914	0.983

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06212	1.449	0.7482	1.336	6.264	0.961
0.06582	1.467	0.7952	1.334	6.634	0.941
0.06952	1.458	0.8452	1.324	7.034	0.919
0.07817	1.446	0.8982	1.318	7.454	0.896
0.08188	1.443	0.9542	1.311	7.894	0.877
0.08717	1.445	1.014	1.304	8.364	0.853
0.09723	1.441	1.074	1.302	8.864	0.831
0.1042	1.441	1.144	1.298	9.394	0.805
0.1175	1.428	1.214	1.292	9.954	0.777
0.1222	1.427	1.284	1.287	10.55	0.749
0.1322	1.419	1.364	1.284	11.15	0.727
0.1422	1.415	1.454	1.272	11.85	0.693
0.1544	1.416	1.534	1.265	12.55	0.666
0.1652	1.418	1.634	1.254	13.25	0.64
0.1782	1.413	1.734	1.248	14.05	0.607
0.1912	1.414	1.834	1.241	14.95	0.576
0.2052	1.408	1.944	1.233	15.75	0.543
0.2202	1.4	2.064	1.224	16.75	0.509
0.2362	1.402	2.194	1.213	17.75	0.475
0.2522	1.395	2.324	1.204	18.75	0.441
0.2702	1.396	2.464	1.195	19.85	0.4
0.2892	1.386	2.614	1.186		
0.3092	1.382	2.774	1.172		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	8.9E-5	cm/sec
Ss	0.0002	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.02946 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G110fh.aqt Title: G110 @ CCR Landfill – Falling Head Test Date: 11/28/16 Time: 10:27:16

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G110

AQUIFER DATA

Saturated Thickness: 10.21 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G110

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.32 ft Static Water Column Height: 10.21 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 10.21 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
-0.03917	2.789	0.3245	2.143	3.063	1.797
-0.035	2.535	0.3495	2.134	3.252	1.775
-0.03078	2.269	0.3755	2.132	3.453	1.761
-0.02297	2.403	0.4035	2.123	3.663	1.741
-0.01928	2.442	0.4335	2.117	3.882	1.719
-0.01558	2.401	0.4645	2.113	4.122	1.699
-0.0119	2.368	0.4985	2.104	4.372	1.678
-0.003683	2.371	0.5335	2.096	4.633	1.652
0.	2.38	0.5705	2.092	4.912	1.628
0.003683	2.362	0.6105	2.081	5.213	1.605
0.00735	2.355	0.6525	2.071	5.523	1.59
0.01582	2.349	0.6965	2.071	5.862	1.55
0.01953	2.344	0.7435	2.061	6.213	1.529

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.02323	2.335	0.7935	2.052	6.582	1.495
0.0285	2.372	0.8465	2.046	6.982	1.468
0.03668	2.321	0.9025	2.042	7.402	1.436
0.0435	2.304	0.9625	2.029	7.843	1.402
0.05523	2.301	1.023	2.023	8.313	1.369
0.0605	2.292	1.093	2.01	8.813	1.336
0.07527	2.276	1.163	2.005	9.342	1.304
0.0805	2.27	1.232	1.998	9.902	1.267
0.09568	2.262	1.313	1.984	10.5	1.229
0.1015	2.25	1.403	1.972	11.1	1.191
0.1135	2.236	1.482	1.962	11.8	1.154
0.1265	2.226	1.583	1.95	12.5	1.11
0.1395	2.215	1.683	1.944	13.2	1.078
0.1535	2,212	1.783	1,931	14.	1.032
0.1685	2.204	1.893	1.914	14.9	0.985
0.1845	2.195	2.013	1.903	15.7	0.943
0.2005	2,182	2.143	1.888	16.7	0.897
0.2185	2.177	2.273	1.877	17.7	0.854
0.2375	2.174	2.413	1.858	18.7	0.807
0.2575	2.165	2.563	1.846	19.8	0.76
0.2785	2.155	2.723	1.83		0110
0.3005	2.148	2.882	1.813		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.7E-5	cm/sec
Ss	0.004	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.01463 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	4.675E-5	2.299E-5	+/- 4.561E-5	2.034	cm/sec
Ss	0.004389	0.004059	+/- 0.008053	1.081	ft ⁻¹
Kz/Kr	1.	19.63	+/- 38.94	0.05095	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

$T = K^*b = 0.01455 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.98
Ss	-0.99	1.00	0.95
Kz/Kr	-0.98	0.95	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.4068 ft ²
Variance	0.004068 ft ²
Std. Deviation	0.06378 ft
Mean	0.009142 ft
No. of Residuals	103
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G110rh.aqt Title: G110 @ CCR Landfill – Rising Head Test Date: 11/18/16 Time: 11:56:09

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G110

AQUIFER DATA

Saturated Thickness: 10.21 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G110

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.66 ft Static Water Column Height: 10.21 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 10.21 ft

No. of Observations: 107

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.475	14.71	1.143	125.3	0.283
0.25	1.751	15.67	1.128	133.1	0.259
0.5	1.662	16.63	1.109	140.9	0.241
0.986	1.638	17.71	1.091	149.3	0.222
1.108	1.584	18.85	1.068	158.3	0.207
1.25	1.608	20.05	1.051	167.9	0.189
1.5	1.543	21.31	1.028	177.5	0.172
1.75	1.552	22.63	1.007	188.3	0.157
2.142	1.511	24.07	0.988	199.7	0.142
2.265	1.507	25.57	0.961	211.7	0.131
2.5	1.495	27.13	0.939	224.3	0.117
2.75	1.486	28.81	0.917	237.5	0.106
3.	1.475	30.61	0.895	251.9	0.097

1

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
3.386	1.454	32.47	0.87	266.9	0.088
3.509	1.447	34.51	0.846	282.5	0.075
3.75	1.436	36.61	0.819	299.3	0.07
4.	1.425	38.83	0.791	317.3	0.063
4.25	1.411	41.23	0.765	335.9	0.057
4.5	1.404	43.75	0.741	356.4	0.047
4.75	1.397	46.39	0.713	377.3	0.048
5.11	1.383	49.21	0.69	399.5	0.041
5.47	1.371	52.21	0.659	423.5	0.039
5.89	1.353	55.39	0.629	448.7	0.03
6.31	1.344	58.75	0.607	475.1	0.034
6.73	1.332	62.35	0.576	503.3	0.025
7.21	1.318	65.95	0.553	533.3	0.026
7.75	1.305	70.15	0.526	565.1	0.024
8.23	1.289	74.35	0.497	598.7	0.018
8.83	1.273	78.55	0.473	634.9	0.018
9.43	1.259	83.35	0.45	670.7	0.018
10.03	1.246	88.75	0.42	712.7	0.011
10.69	1.229	93.55	0.398	754.7	0.01
11.41	1.212	99.55	0.374	796.7	0.009
12.19	1.198	105.5	0.346	844.7	0.013
12.97	1.181	111.5	0.324	898.7	0.008
13.81	1.164	118.1	0.303		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	2.0E-5	cm/sec
Ss	0.0015	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.006224 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	<u>Std. Error</u>	Approx. C.I.	<u>t-Ratio</u>	cm/sec
Kr	1.835E-5	1.521E-6	+/- 3.015E-6	12.07	
Ss	0.002304	0.0005947	+/- 0.001179	3.874	ft ⁻¹
Kz/Kr	1.	1.446	+/- 2.867	0.6918	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.00571 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.98	-0.95
Ss	-0.98	1.00	0.88
Kz/Kr	-0.95	0.88	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.074 ft ²
Variance	0.0007115 ft ²
Std. Deviation	0.02667 ft
Mean	-0.002906 ft
No. of Residuals	107
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G119fh.aqt Title: G119 @ CCR Landfill – Falling Head Test Date: 11/18/16 Time: 11:57:05

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G119

AQUIFER DATA

Saturated Thickness: 9.17 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G119

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.4 ft Static Water Column Height: 9.17 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.58 ft Total Well Penetration Depth: 9.17 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	5.641	0.291	2.976	2.795	2.126
0.004167	6.01	0.31	2.963	2.955	2.094
0.008333	5.281	0.33	2.945	3.135	2.06
0.01565	4.584	0.351	2.927	3.325	2.023
0.01932	4.038	0.373	2.913	3.525	1.981
0.02303	3.534	0.397	2.899	3.735	1.944
0.02672	3.198	0.422	2.881	3.955	1.908
0.03417	3.166	0.448	2.858	4.195	1.864
0.03785	3.065	0.476	2.845	4.445	1.821
0.04152	3.094	0.506	2.826	4.705	1.78
0.0452	3.421	0.537	2.812	4.985	1.734
0.05333	3.237	0.571	2.796	5.285	1.691
0.05702	3.29	0.606	2.773	5.595	1.646

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06068	3.272	0.643	2.76	5.935	1.598
0.0644	3.251	0.683	2.739	6.285	1.557
0.07275	3.251	0.725	2.721	6.655	1.504
0.07642	3.239	0.769	2.703	7.055	1.456
0.0801	3.228	0.816	2.683	7.475	1.404
0.08383	3.218	0.866	2.66	7.915	1.354
0.08827	3.217	0.919	2.64	8.385	1.305
0.09195	3.206	0.975	2.622	8.885	1.253
0.09563	3.201	1.035	2.601	9.415	1.203
0.101	3.19	1.095	2.579	9.975	1.158
0.108	3.18	1.165	2.556	10.57	1.103
0.116	3.172	1.235	2.534	11.18	1.051
0.125	3.151	1.305	2.51	11.88	0.996
0.133	3.147	1.385	2.484	12.57	0.952
0.143	3.128	1.475	2.458	13.28	0.906
0.153	3.119	1.555	2.437	14.07	0.858
0.163	3.101	1.655	2.406	14.98	0.805
0.174	3.087	1.755	2.375	15.78	0.765
0.186	3.075	1.855	2.35	16.77	0.718
0.199	3.063	1.965	2.319	17.77	0.675
0.212	3.045	2.085	2.289	18.77	0.636
0.226	3.034	2.215	2.259	19.88	0.595
0.241	3.019	2.345	2.226	21.07	0.563
0.257	3.007	2.485	2.195	22.38	0.524
0.273	2.993	2.635	2.161	23.68	0.493

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	8.6E-5	cm/sec
Ss	0.01	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.02404 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

ParameterEstimateStd. EndApprox. C.1.t-RatioKr $8.635E-5$ $7.858E-5$ $+/-0.0001557$ 1.099 cm/sSs 0.01033 0.01841 $+/-0.03648$ 0.5609 ft^{-1}	sec
--------------------------------------------------------------------------------------------------------------------------------------------------------	-----

Kz/Kr	0.03458	7.758	+/- 15.38	0.004457

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.02413 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.96
Ss	-0.99	1.00	0.91
Kz/Kr	-0.96	0.91	1.00

Residual Statistics

for weighted residuals

Sum of Squares	18.21 ft ²
Variance	0.164 ft ²
Std. Deviation	0.405 ft
Mean	0.06266 ft
No. of Residuals	114
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G119rh.aqt Title: G119 @CCR Landfill – Rising Head Test Date: 11/28/16 Time: 10:28:38

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G119

AQUIFER DATA

Saturated Thickness: 9.17 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G119

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.2 ft Static Water Column Height: 9.17 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.58 ft Total Well Penetration Depth: 9.17 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.268	0.326	2.045	2.77	1.689
0.004167	2.258	0.348	2.043	2.93	1.675
0.008333	2.247	0.372	2.032	3.11	1.655
0.0125	2.243	0.397	2.028	3.3	1.636
0.01667	2.155	0.423	2.023	3.5	1.61
0.02452	2.201	0.451	2.015	3.71	1.594
0.0282	2.189	0.481	2.009	3.93	1.572
0.03187	2.189	0.512	2.	4.17	1.547
0.03555	2.184	0.546	1.996	4.42	1.521
0.04347	2.171	0.581	1.989	4.68	1.499
0.04715	2.168	0.618	1.983	4.96	1.47
0.05082	2.167	0.658	1.973	5.26	1.442
0.0545	2.165	0.7	1.963	5.57	1.412

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06295	2.156	0.744	1.958	5.91	1.382
0.06663	2.154	0.791	1.948	6.26	1.348
0.07033	2.155	0.841	1.942	6.63	1.318
0.076	2.134	0.894	1.931	7.03	1.288
0.08445	2.143	0.95	1.926	7.45	1.247
0.091	2.143	1.01	1.912	7.89	1.214
0.1	2.138	1.07	1.904	8.36	1,173
0.108	2.126	1.14	1.893	8.86	1.136
0.118	2.124	1.21	1.877	9.39	1.096
0.128	2.123	1.28	1.873	9.95	1.052
0.138	2.118	1.36	1.864	10.55	1.011
0.149	2.109	1.45	1.853	11.15	0.965
0.161	2.103	1.53	1.843	11.85	0.92
0.174	2.093	1.63	1.831	12.55	0.875
0.187	2.09	1.73	1.813	13.25	0.827
0.201	2.083	1.83	1.801	14.05	0.781
0.216	2.081	1.94	1.787	14.95	0.725
0.232	2.076	2.06	1,773	15.75	0.678
0.248	2.067	2.19	1,759	16.75	0.62
0.266	2 064	2.32	1 741	17 75	0.569
0.285	2.001	2 46	1 728	18 75	0.519
0.305	2.054	2.61	1.71	10.70	5.617

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	8.2E-5	cm/sec
Ss	0.001	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.02292 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G125fh.aqt Title: G125 @ CCR Landfill – Falling Head Test Date: 11/28/16 Time: 09:18:08

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G125

AQUIFER DATA

Saturated Thickness: 8.72 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G125

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.82 ft Static Water Column Height: 8.72 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.53 ft Total Well Penetration Depth: 8.67 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.869	0.691	2.554	4.58	1.98
0.008	3.276	0.741	2.544	4.86	1.949
0.018	2.542	0.794	2.528	5.16	1.919
0.02802	2.821	0.85	2.516	5.47	1.886
0.038	2.741	0.91	2.508	5.81	1.855
0.049	2.75	0.97	2.496	6.16	1.82
0.061	2.737	1.04	2.481	6.53	1.789
0.074	2.74	1.11	2.466	6.93	1.755
0.087	2.728	1.18	2.451	7.35	1.717
0.101	2.724	1.26	2.433	7.79	1.682
0.116	2.718	1.35	2.416	8.26	1.643
0.132	2.709	1.43	2.403	8.76	1.604
0.148	2.708	1.53	2.384	9.29	1.57

<u>Time (min)</u>	Displacement (ft)	$\frac{\text{Time (min)}}{1.63}$	Displacement (ft)	Time (min)	Displacement (ft)
0.185	2.693	1.73	2.35	10.45	1.331
0.205	2.683	1.84	2.33	11.05	1.45
0.226	2.675	1.96	2.312	11.75	1.402
0.248	2.672	2.09	2.291	12.45	1.359
0.272	2.665	2.22	2.27	13.15	1.316
0.297	2.655	2.36	2.254	13.95	1.267
0.323	2.652	2.51	2.229	14.85	1.223
0.351	2.639	2.67	2.207	15.65	1.182
0.381	2.632	2.83	2.184	16.65	1.134
0.412	2.62	3.01	2.161	17.65	1.081
0.446	2.614	3.2	2.135	18.65	1.037
0.481	2.608	3.4	2.11	19.75	0.989
0.518	2.595	3.61	2.086	20.95	0.946
0.558	2.585	3.83	2.058	22.25	0.903
0.6	2.573	4.07	2.029	23.55	0.852
0.644	2.564	4.32	2.001	24.95	0.81

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.8E-5	cm/sec
Ss	0.004	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.01276 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	4.918E-5	3.398E-5	+/- 6.756E-5	1.447	cm/sec
Ss	0.003764	0.005048	+/- 0.01004	0.7457	ft ⁻¹
Kz/Kr	1.	23.17	+/- 46.07	0.04315	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.01307 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.95
Kz/Kr	-0.99	0.95	1.00

Residual Statistics

for weighted residuals

Sum of Squares	1.205 ft ²
	0.01385 ft ²
Std. Deviation	0.1177 ft
Mean	-0.01063 ft
No. of Residuals	90
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G125rh.aqt Title: G125 @CCR Landfill – Rising Head Test Date: 11/28/16 Time: 09:18:43

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G125

AQUIFER DATA

Saturated Thickness: 8.72 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G125

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.84 ft Static Water Column Height: 8.72 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.53 ft Total Well Penetration Depth: 8.72 ft

No. of Observations: 113

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
-0.04167	2.799	0.3553	2.709	3.888	2.292	
-0.0375	2.992	0.3813	2.707	4.128	2.273	
-0.03098	2.871	0.4093	2.701	4.378	2.247	
-0.02893	2.922	0.4393	2.695	4.638	2.223	
-0.025	2.865	0.4703	2.692	4.918	2.196	
-0.02083	2.851	0.5043	2.685	5.218	2.169	
-0.01667	2.867	0.5393	2.681	5.528	2.141	
-0.009933	2.861	0.5763	2.677	5.868	2.113	
-0.007867	2.868	0.6163	2.673	6.218	2.081	
-0.004167	2.838	0.6583	2.668	6.588	2.054	
0.	2.846	0.7023	2.656	6.988	2.02	
0.004167	2.821	0.7493	2.651	7.408	1.987	
0.009984	2.828	0.7993	2.645	7.848	1.955	

1

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.01433	2.812	0.8523	2.639	8.318	1.917
0.02033	2.826	0.9083	2.63	8.818	1.879
0.02733	2.81	0.9683	2.621	9.348	1.843
0.03433	2.798	1.028	2.611	9.908	1.806
0.04133	2.804	1.098	2.606	10.51	1.768
0.04933	2.8	1.168	2.596	11.11	1.726
0.05833	2.789	1.238	2.585	11.81	1.683
0.06633	2.786	1.318	2.576	12.51	1.643
0.07633	2.784	1.409	2.564	13.21	1.603
0.08633	2.777	1.488	2.556	14.01	1.553
0.09633	2.777	1.588	2.541	14.91	1.508
0.1073	2.775	1.688	2.532	15.71	1.466
0.1193	2.766	1.788	2.521	16.71	1.418
0.1323	2.762	1.898	2.504	17.71	1.373
0.1453	2.759	2.018	2.489	18.71	1.327
0.1593	2.752	2.148	2.475	19.81	1.283
0.1743	2.75	2.278	2.459	21.01	1.237
0.1903	2.747	2.418	2.443	22.31	1.189
0.2063	2.744	2.568	2,429	23.61	1.142
0 2243	2 736	2 728	2 414	25.01	1 101
0.2433	2 733	2 888	2 393	26.51	1 051
0.2633	2 728	3 068	2 376	28.01	1 011
0.2843	2 724	3 258	2 357	29.71	0.964
0.2070	2.724	3 458	2 336	31 51	0 924
0.3303	2.718	3.668	2.314	01.01	5.724

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.1E-5	cm/sec
Ss	0.0015	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.0109 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G153fh.aqt Title: G153 @ SW Detention Pond – Falling Head Test Date: 11/28/16 Time: 09:19:16

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G153

AQUIFER DATA

Saturated Thickness: 3. ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G153

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.28 ft Static Water Column Height: 11.78 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.44 ft Total Well Penetration Depth: 11.78 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	2.241	0.5299	1.994	3.212	1.455	
0.0037	2.266	0.5699	1.976	3.412	1.433	
0.01193	2.246	0.6119	1.962	3.622	1.406	
0.01993	2.244	0.656	1.948	3.842	1.381	
0.02993	2.229	0.7029	1.933	4.082	1.353	
0.03995	2.225	0.7529	1.91	4.332	1.324	
0.04993	2.218	0.806	1.896	4.592	1.294	
0.06093	2.208	0.8619	1.887	4.872	1.264	
0.07293	2.203	0.9219	1.869	5.172	1.234	
0.08593	2.196	0.9819	1.847	5.482	1.2	
0.09893	2.187	1.052	1.829	5.822	1.165	
0.1129	2.178	1.122	1.818	6.172	1.136	
0.128	2.168	1.192	1.801	6.542	1.106	

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.1439	2.162	1.272	1.781	6.942	1.069
0.16	2.149	1.362	1.763	7.362	1.031
0.178	2.142	1.442	1.745	7.802	0.997
0.1969	2.153	1.542	1.725	8.272	0.962
0.2169	2.122	1.642	1.699	8.772	0.922
0.2379	2.107	1.742	1.681	9.302	0.88
0.2599	2.101	1.852	1.663	9.862	0.847
0.2839	2.088	1.972	1.643	10.46	0.806
0.309	2.077	2.102	1.62	11.06	0.767
0.335	2.069	2.232	1.598	11.76	0.718
0.3629	2.056	2.372	1.579	12.46	0.682
0.3929	2.046	2.522	1.554	13.16	0.633
0.4239	2.028	2.682	1.533	13.96	0.59
0.458	2.021	2.842	1.507	14.86	0.54
0.4929	2.006	3.022	1.481		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00025	cm/sec
Ss	0.015	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.02286 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0002498	2.214E-5	+/- 4.406E-5	11.28	cm/sec
Ss	0.0138	0.002624	+/- 0.005222	5.259	ft ⁻¹
Kz/Kr	1.	1.248	+/- 2.484	0.801	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.02284 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.98
Ss	-0.99	1.00	0.95
Kz/Kr	-0.98	0.95	1.00

Residual Statistics

for weighted residuals



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G153rh.aqt Title: G153 @ SW Detention Pond – Rising Head Test Date: 11/28/16 Time: 09:19:48

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G153

AQUIFER DATA

Saturated Thickness: 3. ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G153

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.23 ft Static Water Column Height: 11.78 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.44 ft Total Well Penetration Depth: 11.78 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.222	0.2334	1.14	1.631	0.868
0.003666	1.261	0.2494	1.13	1.731	0.852
0.007333	1.285	0.2674	1.125	1.831	0.841
0.011	1.214	0.2864	1.12	1.941	0.825
0.01467	1.288	0.3064	1.117	2.061	0.811
0.01832	1.246	0.3274	1.108	2.191	0.796
0.02222	1.224	0.3494	1.1	2.321	0.779
0.02638	1.26	0.3734	1.097	2.461	0.757
0.03055	1.219	0.3984	1.087	2.611	0.744
0.03472	1.245	0.4244	1.088	2.771	0.726
0.03888	1.225	0.4524	1.081	2.931	0.715
0.04305	1.235	0.4824	1.071	3.111	0.694
0.04722	1.203	0.5134	1.061	3.301	0.675
<u>Time (min)</u>	Displacement (ft)	<u>Time (min)</u> 0 5474	Displacement (ft)	<u>Time (min)</u> 3 503	Displacement (ft)
-------------------	-------------------	-----------------------------	-------------------	----------------------------	-------------------
0.05843	1.207	0.5824	1.046	3.711	0.639
0.06338	1.216	0.6194	1.038	3.931	0.622
0.0728	1.206	0.6594	1.031	4.171	0.601
0.07738	1.204	0.7014	1.019	4.421	0.581
0.08438	1.207	0.7454	1.01	4.681	0.557
0.09303	1.186	0.7924	1.	4.961	0.513
0.1014	1.191	0.8424	0.992	5.261	0.521
0.1124	1.178	0.8954	0.978	5.571	0.499
0.1194	1.185	0.9514	0.972	5.911	0.48
0.1325	1.171	1.011	0.958	6.261	0.452
0.1394	1.175	1.071	0.949	6.631	0.436
0.1518	1.165	1.141	0.938	7.031	0.413
0.1626	1.165	1.211	0.925	7.451	0.392
0.1754	1.159	1.281	0.919	7.891	0.368
0.1884	1.155	1.361	0.906	8.361	0.351
0.2024	1.146	1.451	0.893	8.861	0.318
0.2174	1.142	1.531	0.878	9.391	0.301

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00054	cm/sec
Ss	0.0025	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.04938 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0005399	9.804E-5	+/- 0.0001948	5.507	cm/sec
Ss	0.00256	0.001483	+/- 0.002947	1.726	ft ⁻¹
Kz/Kr	1.	1.411	+/- 2.803	0.7089	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.04937 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.97
Kz/Kr	-0.99	0.97	1.00

Residual Statistics

Sum of Squares 0.0362 ft ²	
Variance	
Std. Deviation 0.02005 ft	
Mean	
No. of Residuals 93	
No. of Estimates 3	



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G206fh.aqt Title: G206 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 10:38:25

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G206

AQUIFER DATA

Saturated Thickness: 13.15 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G206

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.01 ft Static Water Column Height: 13.15 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.41 ft Total Well Penetration Depth: 13.15 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.143	0.617	2.752	4.087	1.458
0.008	3.062	0.661	2.737	4.337	1.386
0.017	2.865	0.708	2.718	4.597	1.325
0.025	3.029	0.758	2.701	4.877	1.26
0.035	3.023	0.811	2.69	5.177	1.192
0.045	2.995	0.867	2.669	5.487	1.126
0.055	2.986	0.927	2.655	5.827	1.06
0.066	2.978	0.987	2.64	6.177	0.994
0.078	2.97	1.057	2.619	6.547	0.933
0.091	2.963	1.127	2.599	6.947	0.864
0.104	2.95	1.197	2.576	7.367	0.807
0.118	2.941	1.277	2.527	7.807	0.748
0.133	2.926	1.367	2.48	8.277	0.696

Time (min) 0.149 0.165 0.183 0.202 0.222 0.243 0.265 0.289 0.314 0.368 0.398 0.429 0.463 0.498	Displacement (ft) 2.927 2.92 2.906 2.901 2.895 2.887 2.878 2.878 2.874 2.859 2.849 2.849 2.846 2.836 2.819 2.81 2.81 2.81	Time (min) 1.447 1.547 1.647 1.747 1.857 1.977 2.107 2.237 2.377 2.527 2.687 2.847 3.027 3.217 3.417	Displacement (ft) 2.434 2.388 2.336 2.288 2.236 2.182 2.136 2.081 2.019 1.965 1.905 1.841 1.782 1.717 1.653	Time (min) 8.777 9.307 9.867 10.47 11.07 11.77 12.47 13.17 13.97 14.87 15.67 16.67 17.67 18.67 19.77	Displacement (ft) 0.634 0.604 0.536 0.483 0.442 0.399 0.356 0.322 0.284 0.257 0.224 0.19 0.17 0.148 0.123
0.463 0.498 0.535 0.575	2.81 2.8 2.784 2.77	3.217 3.417 3.627 3.847	1.717 1.653 1.588 1.524	18.67 19.77 20.97	0.148 0.123 0.112

Slug Test Aquifer Model: Confined Solution Method: Bouwer-Rice In(Re/rw): 3.715

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.0003	cm/sec
v0	3.25	ft

$T = K^*b = 0.1202 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
K	0.0002598	2.211E-6	+/- 4.395E-6	117.5	cm/sec
y0	3.047	0.007694	+/- 0.0153	396.	ft

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.1041 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Κ	y0
Κ	1.00	0.55
y0	0.55	1.00

Residual Statistics

for weighted residuals

 $\begin{array}{l} \text{Sum of Squares} \ldots \ldots \ldots 0.168 \ \text{ft}^2 \\ \text{Variance} \ldots \ldots \ldots \ldots 0.001931 \ \text{ft}^2 \\ \text{Std. Deviation} \ldots \ldots \ldots 0.04394 \ \text{ft} \\ \text{Mean.} \ldots \ldots \ldots \ldots \ldots \ldots 0.0007809 \ \text{ft} \\ \text{No. of Residuals} \ldots \ldots 89 \\ \text{No. of Estimates} \ldots 2 \end{array}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G206rh.aqt Title: G206 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:20:24

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G206

AQUIFER DATA

Saturated Thickness: 13.15 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G206

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.79 ft Static Water Column Height: 13.15 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.41 ft Total Well Penetration Depth: 13.15 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.857	0.2011	1.701	1.858	1.367
0.0042	1.813	0.2144	1.698	1.964	1.354
0.005578	1.81	0.2284	1.695	2.084	1.337
0.008333	1.806	0.2431	1.689	2.211	1.312
0.01111	1.802	0.2591	1.691	2.344	1.295
0.01389	1.798	0.2758	1.683	2.484	1.27
0.01823	1.781	0.2931	1.678	2.631	1.252
0.0196	1.783	0.3118	1.675	2.791	1.231
0.02222	1.778	0.3318	1.666	2.958	1.209
0.025	1.775	0.3524	1.662	3.131	1.183
0.02778	1.765	0.3751	1.659	3.318	1.161
0.03093	1.759	0.3984	1.653	3.518	1.133
0.03333	1.765	0.4231	1.647	3.724	1.109

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.03611	1.764	0.4498	1.639	3.951	1.083
0.03889	1.756	0.4778	1.634	4.184	1.054
0.04167	1.755	0.5071	1.627	4.431	1.019
0.04444	1.75	0.5384	1.616	4.698	0.993
0.04843	1.755	0.5718	1.611	4.978	0.967
0.05244	1.755	0.6071	1.607	5.271	0.935
0.06822	1.746	0.6444	1.597	5.584	0.904
0.0696	1.745	0.6844	1.587	5.918	0.873
0.07099	1.742	0.7244	1.578	6.271	0.843
0.07236	1.743	0.7711	1.567	6.644	0.812
0.07777	1.743	0.8178	1.559	7.044	0.781
0.0831	1.739	0.8644	1.553	7.444	0.748
0.08977	1.744	0.9178	1.536	7.911	0.715
0.09644	1.739	0.9778	1.528	8.378	0.684
0.1031	1.734	1.031	1.523	8.844	0.652
0.1104	1.733	1.098	1.502	9.378	0.621
0.1184	1.727	1.164	1.494	9.978	0.586
0.1271	1.726	1.231	1.479	10.51	0.556
0.1358	1.719	1.304	1.465	11.18	0.528
0.1451	1.718	1.384	1.454	11.84	0.502
0.1551	1.716	1.471	1.435	12.51	0.469
0.1658	1.714	1.558	1.422	13.24	0.445
0.1764	1.71	1.651	1.406	14.04	0.414
0.1884	1.707	1.751	1.385		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00016	cm/sec
Ss	0.0003	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.06413 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0001598	1.133E-5	+/- 2.245E-5	14.1	cm/sec
Ss	0.0003168	8.69E-5	+/- 0.0001722	3.645	ft ⁻¹
Kz/Kr	1.	0.8787	+/- 1.742	1.138	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.06404 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.96
Kz/Kr	-0.99	0.96	1.00

Residual Statistics

Sum of Squares	 0.01105 ft ²
Variance	 0.0001033 ft ²
Std. Deviation .	 0.01016 ft
Mean	 -7.742E-5 ft
No. of Residuals	 110
No. of Estimates	 3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G208fh.aqt Title: G208 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:21:07

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G208

AQUIFER DATA

Saturated Thickness: 18.08 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G208

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.88 ft Static Water Column Height: 18.08 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.53 ft Total Well Penetration Depth: 18.08 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	2.84	0.582	2.736	3.854	2.313	
0.007	2.857	0.624	2.724	4.095	2.288	
0.01912	2.868	0.668	2.715	4.344	2.26	
0.024	2.863	0.715	2.708	4.604	2.237	
0.032	2.851	0.765	2.703	4.884	2.2	
0.042	2.858	0.818	2.695	5.184	2.172	
0.052	2.897	0.874	2.685	5.494	2.146	
0.062	2.852	0.934	2.679	5.834	2.112	
0.073	2.855	0.994	2.668	6.184	2.078	
0.085	2.846	1.064	2.656	6.554	2.038	
0.09877	2.848	1.134	2.648	6.954	2.002	
0.111	2.836	1.204	2.635	7.374	1.968	
0.125	2.824	1.284	2.627	7.814	1.937	

Time (min) 0.14 0.156 0.172 0.19 0.209 0.229 0.25 0.272 0.296 0.321 0.347 0.347 0.375 0.405 0.436 0.47	Displacement (ft) 2.823 2.82 2.813 2.808 2.807 2.806 2.798 2.794 2.813 2.785 2.778 2.776 2.769 2.769 2.762 2.758	Time (min) 1.374 1.454 1.554 1.654 1.754 1.864 1.984 2.114 2.244 2.384 2.534 2.694 2.854 3.034 3.224	Displacement (ft) 2.615 2.605 2.59 2.576 2.565 2.553 2.535 2.516 2.5 2.48 2.462 2.444 2.425 2.405 2.386	Time (min) 8.284 8.784 9.314 9.874 10.47 11.07 11.77 12.47 13.18 13.97 14.87 15.68 16.67 17.67 18.67	Displacement (ft) 1.89 1.857 1.814 1.769 1.725 1.678 1.632 1.583 1.541 1.49 1.437 1.391 1.343 1.283 1.235
0.47 0.505 0.542	2.758 2.741 2.743	3.034 3.224 3.424 3.634	2.405 2.386 2.36 2.37	18.67 19.77	1.235 1.194

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	6.0E-5	cm/sec
Ss	0.0003	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.03306 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	<u>Approx. C.I.</u>	<u>t-Ratio</u>	cm/sec
Kr	6.04E-5	4.084E-6	+/- 8.118E-6	14,79	
Ss	0.0002752	6.697E-5	+/- 0.0001331	4.109	ft ⁻¹
Kz/Kr	1.	0.9578	+/- 1.904	1.044	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.03329 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.96
Kz/Kr	-0.99	0.96	1.00

Residual Statistics

Sum of Squares	 	0.007952 ft_2^2
Variance	 	9.246E-5 ft ²
Std. Deviation	 	0.009616 ft
Mean	 	-0.001847 ft
No. of Residuals	 	89
No. of Estimates	 	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G208rh.aqt Title: G208 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:21:28

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G208

AQUIFER DATA

Saturated Thickness: 18.08 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G208

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.95 ft Static Water Column Height: 18.08 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.53 ft Total Well Penetration Depth: 18.08 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	3.003	0.4671	2.833	3.665	2.528	
0.003667	3.002	0.5011	2.831	3.885	2.513	
0.01203	2.96	0.5361	2.827	4.125	2.492	
0.01573	2.946	0.5731	2.822	4.375	2.478	
0.01943	2.946	0.6131	2.816	4.635	2.455	
0.02312	2.947	0.6551	2.808	4.915	2.437	
0.03195	2.923	0.6991	2.805	5.215	2.41	
0.03565	2.913	0.7461	2.8	5.525	2.392	
0.03937	2.923	0.7961	2.79	5.865	2.375	
0.04613	2.914	0.8491	2.785	6.215	2.353	
0.05507	2.917	0.9051	2.78	6.585	2.33	
0.06312	2.92	0.9651	2.773	6.985	2.306	
0.07312	2.905	1.025	2.764	7.405	2.274	

Time (min) 0.08313 0.09312 0.1041 0.1161 0.1291 0.1421 0.1561 0.1711 0.1871 0.2031 0.2211 0.2401 0.2601 0.2811 0.3031 0.3271 0.3521 0.3781 0.4061	Displacement (ft) 2.906 2.91 2.903 2.89 2.89 2.896 2.878 2.873 2.873 2.873 2.873 2.873 2.873 2.869 2.869 2.869 2.869 2.869 2.851 2.853 2.85 2.849	$\frac{\text{Time (min)}}{1.095}$ $\frac{1.095}{1.165}$ $\frac{1.235}{1.315}$ $\frac{1.405}{1.485}$ $\frac{1.585}{1.685}$ $\frac{1.685}{1.785}$ $\frac{1.895}{2.015}$ $\frac{2.145}{2.275}$ $\frac{2.416}{2.565}$ $\frac{2.725}{2.885}$ $\frac{3.065}{3.255}$	Displacement (ft) 2.761 2.752 2.744 2.735 2.728 2.719 2.703 2.702 2.688 2.681 2.668 2.66 2.645 2.633 2.614 2.606 2.592 2.572 2.563	Time (min) 7.845 8.315 8.815 9.345 9.905 10.51 11.11 11.81 12.51 13.21 14.01 14.91 15.71 16.71 17.71 18.71 19.81 21.01	Displacement (ft) 2.254 2.224 2.198 2.172 2.141 2.099 2.091 2.044 2.022 1.995 1.965 1.927 1.9 1.863 1.835 1.8 1.766 1.734
0.4361	2.842	3.455	2.547		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	2.1E-5	cm/sec
Ss	0.0028	ft ⁻¹
K7/Kr	1.	

 $T = K^*b = 0.01157 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	2.111E-5	2.793E-6	+/- 5.544E-6	7.557	cm/sec
Ss	0.002783	0.000655	+/- 0.0013	4.25	ft ⁻¹
Kz/Kr	1.	5.786	+/- 11.48	0.1728	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

$T = K^*b = 0.01163 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.96
Kz/Kr	-0.99	0.96	1.00

Residual Statistics

Sum of Squares 0.012	242 ft ²
Variance	01321 ft ²
Std. Deviation 0.01	149 ft
Mean	1463 ft
No. of Residuals 97	
No. of Estimates 3	



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G209fh.aqt Title: G209 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:22:04

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G209

AQUIFER DATA

Saturated Thickness: 25.48 ft Anisotropy Ratio (Kz/Kr): 0.01

SLUG TEST WELL DATA

Test Well: G209

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.96 ft Static Water Column Height: 25.48 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 25.48 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.947	0.609	2.648	3.839	1.809
0.009	2.943	0.653	2.63	4.079	1.762
0.017	2.956	0.7	2.612	4.329	1.713
0.027	2.944	0.75	2.594	4.589	1.663
0.037	2.94	0.803	2.575	4.869	1.615
0.047	2.934	0.859	2.554	5.169	1.559
0.058	2.923	0.919	2.537	5.479	1.509
0.07	2.913	0.979	2.514	5.819	1.456
0.083	2.954	1.049	2.493	6.169	1.398
0.096	2.858	1.119	2.47	6.539	1.348
0.11	2.901	1.189	2.453	6.939	1.295
0.125	2.878	1.269	2.426	7.359	1.237
0.141	2.871	1.359	2.399	7.799	1.185

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.157	2.862	1.439	2.376	8.269	1.127
0.175	2.851	1.539	2.348	8.769	1.07
0.194	2.84	1.639	2.322	9.299	1.012
0.214	2.832	1.739	2.296	9.859	0.956
0.235	2.816	1.849	2.265	10.46	0.9
0.257	2.81	1.969	2.232	11.06	0.787
0.281	2.795	2.099	2.197	11.76	0.691
0.306	2.779	2.229	2.168	12.46	0.601
0.332	2.768	2.369	2.134	13.16	0.53
0.36	2.758	2.519	2.098	13.96	0.468
0.39	2.741	2.679	2.057	14.86	0.406
0.421	2.729	2.839	2.021	15.66	0.359
0.455	2.713	3.019	1.986	16.66	0.315
0.49	2.696	3.209	1.937	17.66	0.283
0.455 0.49 0.527 0.567	2.713 2.696 2.681 2.666	3.019 3.209 3.409 3.619	1.986 1.937 1.897 1.854	16.66 17.66 18.66	0.315 0.283 0.252

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0002	cm/sec
Ss	0.0001	ft ⁻¹
Kz/Kr	0.01	

 $T = K^*b = 0.1553 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G209rh.aqt Title: G209 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:22:32

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G209

AQUIFER DATA

Saturated Thickness: 25.48 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G209

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.6 ft Static Water Column Height: 25.48 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 25.48 ft

No. of Observations: 107

		Observ	vation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.613	0.2953	2.417	2.62	1.796
0.00205	2.603	0.3153	2.411	2.78	1.761
0.006167	2.599	0.3363	2.401	2.94	1.728
0.01033	2.606	0.3583	2.391	3.12	1.693
0.0145	2.577	0.3823	2.384	3.31	1.651
0.02102	2.561	0.4073	2.375	3.51	1.615
0.02307	2.568	0.4333	2.368	3.72	1.579
0.027	2.562	0.4613	2.359	3.94	1.538
0.03117	2.556	0.4913	2.352	4.18	1.495
0.03533	2.55	0.5223	2.339	4.43	1.453
0.0395	2.541	0.5563	2.326	4.69	1.414
0.04367	2.539	0.5913	2.319	4.97	1.37
0.04783	2.536	0.6283	2.304	5.27	1.328

1

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.052	2.532	0.6683	2.289	5.581	1.282
0.05617	2.537	0.7103	2.281	5.92	1.232
0.06033	2.526	0.7543	2.264	6.27	1.188
0.06633	2.521	0.8013	2.25	6.64	1.141
0.07233	2.517	0.8513	2.237	7.04	1.095
0.07933	2.516	0.9043	2.22	7.46	1.05
0.08633	2.506	0.9603	2.202	7.9	1.
0.09333	2.507	1.02	2.189	8.37	0.953
0.1013	2.499	1.08	2.172	8.87	0.911
0.1103	2.497	1.15	2.151	9.4	0.865
0.1183	2.495	1.22	2.131	9.96	0.815
0.1283	2.487	1.29	2.112	10.56	0.77
0.1383	2.482	1.37	2.092	11.16	0.732
0.1483	2.48	1.46	2.067	11.86	0.686
0.1593	2.473	1.54	2.048	12.56	0.647
0.1713	2.466	1.64	2.019	13.26	0.606
0.1843	2.461	1.74	1.997	14.06	0.572
0.1973	2.457	1.84	1.973	14.96	0.53
0.2113	2.447	1.95	1.945	15.76	0.498
0.2263	2.447	2.07	1.913	16.76	0.463
0.2423	2.437	2.2	1.89	17.76	0.434
0.2583	2.427	2.33	1.857	18.76	0.403
0.2763	2.425	2.47	1.824		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00016	cm/sec
Ss	0.0004	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1243 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	<u>Std. Error</u>	<u>Approx. C.I.</u>	<u>t-Ratio</u>	cm/sec
Kr	0.0001569	8.484E-6	+/- 1.682E-5	18.49	
Ss	0.0004212	8.747E-5	+/- 0.0001734	4.816	ft ⁻¹
Kz/Kr	1.	0.6918	+/- 1.372	1.446	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.1218 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.95
Kz/Kr	-0.99	0.95	1.00

Residual Statistics

Sum of Squares	0.02054 ft ²
	0.0001975 ft ²
Std. Deviation	0.01405 ft
Mean	-0.004298 ft
No. of Residuals	107
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G210fh.aqt Title: G210 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:23:01

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G210

AQUIFER DATA

Saturated Thickness: 15.7 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G210

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.95 ft Static Water Column Height: 15.7 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 15.7 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.988	0.582	1.54	3.854	0.568
0.007	1.947	0.624	1.522	4.094	0.526
0.015	1.908	0.668	1.498	4.344	0.491
0.024	1.894	0.715	1.472	4.604	0.461
0.032	1.89	0.765	1.452	4.884	0.429
0.042	1.886	0.818	1.425	5.184	0.397
0.052	1.871	0.874	1.401	5.494	0.367
0.062	1.868	0.934	1.373	5.834	0.335
0.073	1.858	0.994	1.344	6.184	0.307
0.085	1.849	1.064	1.317	6.554	0.28
0.098	2.179	1.134	1.285	6.954	0.254
0.111	1.84	1.204	1.258	7.374	0.232
0.125	1.771	1.284	1.225	7.814	0.206

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.14	1.813	1.374	1.188	8.284	0.186
0.156	1.802	1.454	1.158	8.784	0.167
0.172	1.789	1.554	1.123	9.314	0.149
0.19	1.777	1.654	1.087	9.874	0.137
0.209	1.762	1.754	1.055	10.47	0.118
0.229	1.75	1.864	1.018	11.07	0.109
0.25	1.735	1.984	0.982	11.77	0.098
0.272	1.723	2.114	0.943	12.47	0.087
0.296	1.709	2.244	0.908	13.17	0.076
0.321	1.693	2.384	0.866	13.97	0.068
0.347	1.676	2.534	0.832	14.87	0.062
0.375	1.66	2.694	0.79	15.67	0.051
0.321	1.693	2.384	0.866	13.97	0.068
0.347	1.676	2.534	0.832	14.87	0.062
0.375	1.66	2.694	0.79	15.67	0.051
0.405	1.643	2.854	0.757	16.67	0.051
0.436	1.625	3.034	0.718	17.67	0.051
0.47	1.609	3.224	0.68	18.67	0.042
0.505	1.585	3.424	0.639	19.77	0.038
0.542	1.563	3.634	0.603		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0005	cm/sec
Ss	0.0001	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.2393 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004908	0.000128	+7- 0.0002545	3.835	cm/sec
Ss	0.0001094	0.0001408	+/- 0.0002798	0.7771	ft ⁻¹
Kz/Kr	1.	2.525	+/- 5.019	0.3961	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.2349 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.98
Kz/Kr	-1.00	0.98	1.00

Residual Statistics

Sum of Squares	0.1181 ft ²
	0.001373 ft ²
Std. Deviation	0.03705 ft
Mean	0.003124 ft
No. of Residuals	89
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G210rh.aqt Title: G210 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:23:24

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G210

AQUIFER DATA

Saturated Thickness: 15.7 ft Anisotropy Ratio (Kz/Kr): 0.9532

SLUG TEST WELL DATA

Test Well: G210

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2. ft Static Water Column Height: 15.7 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 15.7 ft

No. of Observations: 104

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.058	0.326	1.714	2.77	0.778
0.004167	2.011	0.348	1.7	2.93	0.746
0.01072	1.989	0.372	1.686	3.11	0.707
0.01277	1.976	0.397	1.669	3.3	0.671
0.01667	1.977	0.423	1.655	3.5	0.636
0.02083	1.967	0.451	1.639	3.71	0.596
0.025	1.969	0.481	1.619	3.93	0.563
0.03177	1.957	0.512	1.602	4.17	0.527
0.03382	1.951	0.546	1.581	4.42	0.489
0.0375	1.945	0.581	1.563	4.68	0.455
0.04167	1.95	0.618	1.546	4.96	0.425
0.04583	1.936	0.658	1.523	5.26	0.393
0.05278	1.932	0.7	1.496	5.57	0.36

1

<u>Time (min)</u>	Displacement (ft)	<u>Time (min)</u>	Displacement (ft)	<u>Time (min)</u>	Displacement (ft)
0.056	1.929	0.744	1.48	5.91	0.33
0.06203	1.918	0.791	1.456	6.26	0.301
0.07057	1.912	0.841	1.43	6.63	0.279
0.076	1.902	0.894	1.407	7.03	0.258
0.083	1.9	0.95	1.379	7.45	0.227
0.091	1.892	1.01	1.35	7.89	0.213
0.1	1.878	1.07	1.327	8.36	0.187
0.108	1.873	1.14	1.294	8.86	0.174
0.118	1.865	1.21	1.264	9.39	0.155
0 128	1 856	1 28	1 236	9 95	0 14
0 138	1 849	1.36	1 209	10 55	0 124
0 149	1 841	1 45	1 171	11 15	0 109
0.161	1 83	1 53	1 143	11.85	0 104
0.174	1 818	1.63	1 107	12 55	0.007
0.174	1 808	1.03	1.107	12.33	0.072
0.107	1.000	1.75	1.073	14.05	0.003
0.201	1.0	1.03	1.042	14.05	0.077
0.210	1.784	1.94	1.005		0.07
0.232	1.777	2.06	0.969	15.75	0.067
0.248	1.768	2.19	0.931	16.75	0.058
0.266	1.756	2.32	0.898	17.75	0.056
0.285	1.74	2.46	0.86	18.75	0.05
0.305	1.727	2.61	0.823		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00048	cm/sec
Ss	0.0002	ft ⁻¹
Kz/Kr	0.9532	

 $T = K^*b = 0.2297 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004796	2.439E-5	+/- 4.839E-5	19.66	cm/sec
Ss	0.0002031	4.796E-5	+/- 9.515E-5	4.235	ft ⁻¹
Kz/Kr	0.9532	0.495	+/- 0.9822	1.925	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

$T = K^*b = 0.2295 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.98
Kz/Kr	-0.99	0.98	1.00

Residual Statistics

Sum of Squares	0.008583 ft^2
Variance	8.498E-5 ft ²
Std. Deviation	0.009219 ft
Mean	0.001622 ft
No. of Residuals	104
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G212fh.aqt Title: G212 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:23:53

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G212

AQUIFER DATA

Saturated Thickness: 12.71 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G212

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.95 ft Static Water Column Height: 12.71 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.55 ft Total Well Penetration Depth: 12.71 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.913	0.4676	2.534	3.456	1.624
0.00365	2.884	0.5016	2.513	3.666	1.581
0.0073	2.877	0.5366	2.5	3.886	1.541
0.01157	2.875	0.5736	2.473	4.126	1.501
0.01758	2.873	0.6136	2.45	4.376	1.455
0.02457	2.869	0.6556	2.426	4.636	1.412
0.03157	2.863	0.6996	2.405	4.916	1.362
0.03857	2.858	0.7466	2.386	5.216	1.314
0.04657	2.85	0.7966	2.36	5.526	1.27
0.05557	2.841	0.8496	2.337	5.866	1.221
0.06357	2.83	0.9056	2.31	6.216	1.175
0.07357	2.828	0.9656	2.284	6.586	1.128
0.08357	2.813	1.026	2.263	6.986	1.085

<u>Time (min)</u> 0.09357	Displacement (ft) 2.806	Time (min) 1.096	Displacement (ft)	Time (min) 7.406	Displacement (ft) 1.039
0.1046	2.799	1.166	2.209	7.846	0.987
0.1166	2.788	1.236	2.184	8.316	0.941
0.1296	2.781	1.316	2.158	8.816	0.895
0.1426	2.762	1.406	2.128	9.346	0.852
0.1566	2.753	1.486	2.104	9.906	0.81
0.1716	2.74	1.586	2.072	10.51	0.766
0.1876	2.728	1.686	2.071	11.11	0.725
0.2036	2.715	1.786	2.015	11.81	0.682
0.2216	2.699	1.896	1.986	12.51	0.644
0.2406	2.685	2.016	1.951	13.21	0.611
0.2606	2.674	2.146	1.921	14.01	0.569
0.2816	2.659	2.276	1.886	14.91	0.533
0.3036	2.642	2.416	1.852	15.71	0.505
0.3276	2.627	2.566	1.814	16.71	0.471
0.3526	2.628	2.726	1.78	17.71	0.443
0.3786	2.592	2.886	1.745	18.71	0.417
0.4066	2.582	3.066	1.704	19.81	0.39
0.4366	2.554	3.256	1.667	21.01	0.366

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00013	cm/sec
Ss	0.004	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.05036 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0001337	5.707E-6	+/- 1.133E-5	23.42	cm/sec
Ss	0.004071	0.0003972	+/- 0.0007888	10.25	ft ⁻¹
Kz/Kr	1.	1.219	+/- 2.421	0.8205	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window
$T = K^*b = 0.05178 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.98	-0.97
Ss	-0.98	1.00	0.93
Kz/Kr	-0.97	0.93	1.00

Residual Statistics

Sum of Squares \ldots \ldots \ldots 0.01956 ft ²
/ariance
Std. Deviation 0.0145 ft
Mean
No. of Residuals 96
No. of Estimates 3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G212rh.aqt Title: G212 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:24:15

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G212

AQUIFER DATA

Saturated Thickness: 12.71 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G212

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.88 ft Static Water Column Height: 12.71 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.55 ft Total Well Penetration Depth: 12.71 ft

	Observation Data						
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
0.	2.866	0.3265	2.662	2.93	1.871		
0.003683	2.881	0.3485	2.657	3.111	1.825		
0.007333	2.87	0.3725	2.647	3.301	1.786		
0.01102	2.861	0.3975	2.635	3.501	1.74		
0.01473	2.854	0.4235	2.625	3.711	1.693		
0.01843	2.846	0.4515	2.614	3.93	1.65		
0.02213	2.848	0.4815	2.602	4.17	1.603		
0.02583	2.84	0.5125	2.586	4.42	1.555		
0.0338	2.829	0.5465	2.577	4.681	1.481		
0.03748	2.822	0.5815	2.563	4.96	1.46		
0.04115	2.822	0.6185	2.549	5.261	1.409		
0.0448	2.818	0.6585	2.532	5.571	1.356		
0.04847	2.815	0.7005	2.493	5.911	1.297		

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05215	2.813	0.7445	2.501	6.261	1.254
0.0565	2.812	0.7915	2.486	6.63	1.201
0.0625	2.809	0.8415	2.467	7.031	1.15
0.0695	2.805	0.8945	2.45	7.45	1.095
0.0765	2.797	0.9505	2.431	7.891	1.046
0.0835	2.79	1.011	2.411	8.361	0.993
0.0915	2.779	1.071	2.392	8.861	0.939
0.1005	2.785	1.141	2.366	9.39	0.886
0.1085	2.779	1.21	2.341	9.95	0.841
0.1185	2.769	1.281	2.323	10.55	0.789
0.1285	2.767	1.361	2.295	11.15	0.744
0.1385	2.758	1.45	2.266	11.85	0.694
0.1495	2.755	1.531	2.243	12.55	0.648
0.1615	2.746	1.631	2.214	13.25	0.608
0.1745	2.734	1.73	2.184	14.05	0.567
0.1875	2.727	1.831	2.156	14.95	0.52
0.2015	2.725	1.94	2.123	15.75	0.487
0.2165	2.718	2.061	2.088	16.75	0.447
0.2325	2.715	2.191	2.055	17.75	0.414
0.2485	2.701	2.321	2.022	18.75	0.381
0.2665	2.695	2.461	1.987	19.85	0.354
0.2855	2.682	2.611	1.947		
0.3055	2.676	2.771	1.907		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00018	cm/sec
Ss	0.0003	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.06973 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter Kr	Estimate	Std. Error 7 1E-6	Approx. C.I. +/- 1 408E-5	t-Ratio	cm/sec
Ss Kz/Kr	0.0002962	4.965E-5 0.4616	+/- 9.845E-5 +/- 0.9153	5.966 2.166	ft ⁻¹

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.06825 \text{ cm}^2/\text{sec}$

Parameter Correlations

Kr	Ss	Kz/Kr
1.00	-0.99	-0.99
-0.99	1.00	0.96
-0.99	0.96	1.00
	<u>Kr</u> 1.00 -0.99 -0.99	<u>Kr</u> <u>Ss</u> 1.00 -0.99 -0.99 1.00 -0.99 0.96

Residual Statistics

Sum of Squares	0.01265 ft ²
Variance	0.0001228 ft ²
Std. Deviation	0.01108 ft
Mean	-0.003437 ft
No. of Residuals	106
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G215fh.aqt Title: G215 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:24:40

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G215

AQUIFER DATA

Saturated Thickness: 12.7 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G215

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.57 ft Static Water Column Height: 12.7 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.39 ft Total Well Penetration Depth: 12.7 ft

	Observation Data						
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
0.	2.543	0.49	2.112	3.244	0.927		
0.006	2.561	0.525	2.093	3.444	0.871		
0.013	2.554	0.562	2.059	3.654	0.824		
0.02	2.533	0.602	2.035	3.874	0.776		
0.027	2.518	0.644	1.999	4.114	0.726		
0.03818	2.518	0.688	1.973	4.364	0.677		
0.044	2.516	0.735	1.938	4.624	0.63		
0.052	2.518	0.785	1.905	4.904	0.586		
0.062	2.501	0.838	1.866	5.204	0.538		
0.072	2.484	0.894	1.835	5.514	0.489		
0.082	2.476	0.954	1.8	5.854	0.447		
0.093	2.472	1.014	1.769	6.204	0.406		
0.105	2.449	1.084	1.727	6.574	0.369		

<u>Time (min)</u> 0.118 0.131 0.145 0.16 0.176 0.192 0.21 0.229 0.249 0.27 0.27 0.292	Displacement (ft) 2.43 2.42 2.401 2.393 2.381 2.364 2.344 2.329 2.311 2.295 2.272	Time (min) 1.154 1.224 1.304 1.394 1.474 1.574 1.674 1.774 1.884 2.004 2.134	Displacement (ft) 1.689 1.653 1.61 1.568 1.528 1.481 1.441 1.398 1.353 1.313 1.259	Time (min) 6.974 7.394 7.834 8.304 8.804 9.339 9.894 10.49 11.09 11.79 12.49	Displacement (ft) 0.33 0.299 0.262 0.237 0.21 0.182 0.163 0.144 0.127 0.108 0.094
0 102	2.001	1 574	1 / 91	0 2 2 0	0 192
0.172	2.304	1.374	1.401	7.337	0.102
0.21	2.344	1.6/4	1.441	9.894	0.163
0.229	2.329	1.774	1.398	10.49	0.144
0.249	2.311	1.884	1.353	11.09	0.127
0.27	2.295	2.004	1.313	11.79	0.108
0.292	2.272	2.134	1.259	12.49	0.094
0.316	2.251	2.264	1.221	13.19	0.081
0.341	2.231	2.404	1.171	13.99	0.063
0.367	2.208	2.554	1.123	14.89	0.053
0.395	2.188	2.714	1.074	15.69	0.045
0.425	2.162	2.874	1.026		
0.456	2.136	3.054	0.976		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0005	cm/sec
Ss	9.0E-5	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.1935 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0005031	5.308E-5	+/- 0.0001055	9.478	cm/sec
Ss	8.802E-5	4.315E-5	+/- 8.578E-5	2.04	ft ⁻¹
Kz/Kr	1.	1.055	+/- 2.097	0.9479	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.1948 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-1.00	-1.00
Ss	-1.00	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	. 0.00943 ft ²
Variance	. 0.0001109 ft ²
Std. Deviation	. 0.01053 ft
Mean	. 3.96E-5 ft
No. of Residuals	. 88
No. of Estimates	. 3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G215rh.aqt Title: G215 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:25:02

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G215

AQUIFER DATA

Saturated Thickness: 12.7 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G215

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.66 ft Static Water Column Height: 12.7 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.39 ft Total Well Penetration Depth: 12.7 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	2.653	0.275	2.4	2.329	1.526	
0.003683	2.613	0.2939	2.389	2.469	1.482	
0.00735	2.658	0.314	2.376	2.619	1.437	
0.01102	2.688	0.335	2.364	2.779	1.39	
0.01922	2.637	0.3569	2.35	2.939	1.347	
0.02288	2.63	0.381	2.339	3.119	1.296	
0.02657	2.635	0.4059	2.323	3.309	1.245	
0.03023	2.624	0.432	2.308	3.509	1.193	
0.03862	2.615	0.46	2.29	3.719	1.144	
0.04228	2.609	0.49	2.274	3.939	1.095	
0.04595	2.603	0.521	2.258	4.179	1.042	
0.04962	2.58	0.555	2.237	4.429	0.988	
0.05535	2.573	0.5899	2.224	4.689	0.939	

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05905	2.574	0.627	2.199	4.969	0.882
0.06277	2.577	0.667	2.18	5.269	0.832
0.06643	2.569	0.7089	2.16	5.588	0.776
0.07012	2.565	0.753	2.142	5.921	0.726
0.0738	2.56	0.8	2.115	6.269	0.675
0.07795	2,559	0.8499	2.093	6.639	0.624
0.08495	2,554	0.903	2.069	7.039	0.58
0.09195	2.546	0.9589	2.041	7.459	0.502
0.09995	2.536	1.019	2.022	7.899	0.486
0.109	2.529	1.079	1.989	8.369	0.443
0.117	2.518	1.149	1.962	8.869	0.398
0.127	2.519	1.219	1.93	9.399	0.359
0.1369	2.505	1.289	1.902	9,959	0.318
0 147	2 497	1 369	1 87	10.56	0.28
0.158	2.49	1.459	1.834	11.16	0.244
0.1699	2.483	1.539	1.801	11.86	0.212
0 183	2 473	1 639	1 764	12 56	0 182
0 196	2 46	1 739	1 726	13.26	0 157
0.21	2 447	1 839	1 691	14.06	0 134
0.225	2 439	1 949	1 652	14 96	0 112
0 241	2 428	2 069	1 612	15 76	0.094
0.257	2.416	2.199	1.567	10.70	5.074

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00035	cm/sec
Ss	0.00015	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1355 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0003512	5.917E-5	+/- 0.0001174	5.935	cm/sec
Ss	0.0001143	8.861E-5	+/- 0.0001758	1.29	ft ⁻¹
Kz/Kr	1.	1.688	+/- 3.348	0.5925	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.1359 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	0.05196 ft ²
Variance	0.0005144 ft ²
Std. Deviation	0.02268 ft
Mean	-0.007737 ft
No. of Residuals	104
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G218fh.aqt Title: G218 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:25:29

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G218

AQUIFER DATA

Saturated Thickness: 14.16 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G218

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.75 ft Static Water Column Height: 14.16 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.44 ft Total Well Penetration Depth: 14.16 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.731	0.4149	2.356	3.074	1.217
0.00365	2.758	0.4449	2.335	3.264	1.168
0.007317	2.755	0.4759	2.314	3.464	1.118
0.01097	2.741	0.5099	2.29	3.674	1.068
0.01462	2.724	0.5449	2.266	3.894	1.016
0.01988	2.697	0.5819	2.237	4.134	0.964
0.02588	2.691	0.6219	2.216	4.384	0.913
0.03338	2.696	0.6639	2.19	4.644	0.87
0.03988	2.683	0.7079	2.166	4.924	0.815
0.04687	2.675	0.7549	2.139	5.224	0.76
0.05488	2.668	0.8049	2.107	5.534	0.713
0.06387	2.655	0.8579	2.073	5.874	0.662
0.07187	2.65	0.9139	2.045	6.224	0.616

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.08188	2.64	0.9739	2.011	6.594	0.571
0.09188	2.632	1.034	1.981	6.994	0.524
0.1019	2.615	1.104	1.944	7.414	0.477
0.1129	2.607	1.174	1.907	7.854	0.433
0.1249	2.593	1.244	1.878	8.324	0.39
0.1379	2.582	1.324	1.835	8.824	0.355
0.1509	2.571	1.414	1.799	9.354	0.317
0.1649	2.557	1.494	1.762	9.918	0.279
0.1799	2.546	1.594	1.712	10.51	0.248
0.1959	2.533	1.694	1.677	11.11	0.221
0.2119	2.517	1.794	1.635	11.81	0.191
0.2299	2.501	1.904	1.593	12.51	0.169
0.2489	2.482	2.024	1.548	13.21	0.147
0.2689	2.468	2.154	1.501	14.01	0.129
0.2899	2.455	2.84	1.455	14.91	0.106
0.2489 0.2689 0.2899 0.3119 0.3359 0.3609 0.3869	2.482 2.468 2.455 2.436 2.417 2.399 2.378	2.024 2.154 2.284 2.424 2.574 2.734 2.894	1.548 1.501 1.455 1.411 1.364 1.316 1.267	13.21 14.01 14.91 15.71 16.71 17.71	0.147 0.129 0.106 0.087 0.078 0.065

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00041	cm/sec
Ss	0.0001	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.177 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004038	6.674E-5	+/- 0.0001326	6.05	cm/sec
Ss	9.763E-5	7.71E-5	+/- 0.0001531	1.266	ft ⁻¹
Kz/Kr	1.	1.614	+/- 3.206	0.6195	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-1.00	-1.00
Ss	-1.00	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	0.03742 ft ²
Variance	0.0004068 ft ²
Std. Deviation	0.02017 ft
Mean	-0.006913 ft
No. of Residuals	95
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G218rh.aqt Title: G218 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:25:51

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: G218

AQUIFER DATA

Saturated Thickness: 14.16 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G218

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.87 ft Static Water Column Height: 14.16 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.44 ft Total Well Penetration Depth: 14.16 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	2.887	0.3008	2.558	2.316	1.494	
0.004167	2.872	0.3218	2.54	2.456	1.444	
0.008333	2.856	0.3438	2.519	2.606	1.391	
0.0125	2.839	0.3678	2.504	2.766	1.342	
0.01667	2.834	0.3928	2.484	2.926	1.279	
0.02083	2.826	0.4188	2.468	3.106	1.228	
0.025	2.821	0.4469	2.446	3.296	1.171	
0.02917	2.811	0.4768	2.428	3.496	1.116	
0.03333	2.806	0.5078	2.407	3.706	1.057	
0.0375	2.799	0.5418	2.382	3.926	1.001	
0.04167	2.79	0.5768	2.359	4.166	0.941	
0.04583	2.781	0.6138	2.338	4.416	0.89	
0.05183	2.782	0.6538	2.315	4.676	0.84	

Time (min) 0.05785 0.06585 0.07183 0.07883 0.08693 0.09583 0.104 0.1138 0.1238 0.1238 0.1238 0.1338 0.1338 0.1448 0.1568 0.1698 0.1828 0.1968 0.2118 0.2278	Displacement (ft) 2.772 2.753 2.749 2.738 2.732 2.73 2.715 2.686 2.704 2.698 2.684 2.673 2.662 2.639 2.647 2.627 2.614	Time (min) 0.6958 0.7398 0.7868 0.8368 0.8898 0.9458 1.006 1.066 1.136 1.206 1.276 1.356 1.446 1.526 1.626 1.726 1.826	Displacement (ft) 2.284 2.263 2.229 2.203 2.166 2.136 2.099 2.068 2.03 1.991 1.956 1.914 1.872 1.83 1.787 1.737 1.695	Time (min) 4.956 5.256 5.566 5.906 6.256 6.626 7.026 7.446 7.886 8.356 8.856 9.386 9.386 9.946 10.55 11.15 11.85 12.55	Displacement (ft) 0.785 0.732 0.676 0.624 0.581 0.534 0.488 0.449 0.406 0.366 0.335 0.296 0.266 0.241 0.216 0.195 0.172
0.2118	2.627	1.726	1.737	11.85	0.195
0.2278	2.614	1.826	1.695	12.55	0.172
0.2438	2.599	1.936	1.651	13.25	0.15
0.2618	2.588	2.056	1.602	14.05	0.139
0.2808	2.568	2.186	1.546	14.95	0.126

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00041	cm/sec
Ss	0.0001	ft ⁻¹
K7/Kr	1.	

 $T = K^*b = 0.177 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004113	3.569E-5	+/- 7.084E-5	11.53	cm/sec
Ss	0.0001095	4.481E-5	+/- 8.895E-5	2.444	ft ⁻¹
Kz/Kr	1.	0.8554	+/- 1.698	1.169	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

$T = K^*b = 0.1775 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-1.00	-1.00
Ss	-1.00	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	0.01553 ft ²
Variance	0.0001617 ft ²
Std. Deviation	0.01272 ft
Mean	-0.004094 ft
No. of Residuals	99
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G270fh.aqt Title: G270 @ GMF Recycle Pond – Falling Head Test Date: 11/28/16 Time: 09:26:15

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G270

AQUIFER DATA

Saturated Thickness: 16.41 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G270

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.15 ft Static Water Column Height: 16.41 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.79 ft Total Well Penetration Depth: 16.41 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.166	0.3988	2.549	2.558	1.129
0.003684	3.185	0.4288	2.512	2.718	1.064
0.007367	3.122	0.4598	2.478	2.878	1.016
0.01105	3.068	0.4938	2.442	3.058	0.946
0.0194	3.105	0.5288	2.406	3.248	0.888
0.02382	3.099	0.5658	2.365	3.448	0.828
0.03082	3.057	0.6059	2.327	3.658	0.77
0.05118	3.026	0.6478	2.284	3.878	0.717
0.05982	3.027	0.6918	2.241	4.118	0.659
0.06355	3.019	0.7388	2.197	4.368	0.611
0.06725	3.01	0.7888	2.155	4.628	0.561
0.07712	2.995	0.8418	2.109	4.908	0.515
0.08595	2.97	0.8978	2.065	5.208	0.468

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00055	cm/sec
Ss	0.0002	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.2751 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0005615	2.665E-5	+/- 5.3E-5	21.07	cm/sec
Ss	0.0001649	3.741E-5	+/- 7.442E-5	4.408	ft ⁻¹
Kz/Kr	1.	0.4806	+/- 0.9559	2.081	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.2809 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.98
Kz/Kr	-1.00	0.98	1.00

Residual Statistics

um of Squares 0.01042 ft ²	
'ariance	
td. Deviation	
lean0.001403 ft	
lo. of Residuals 86	
lo. of Estimates 3	



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G270rh.aqt Title: G270 @ GMF Recycle Pond – Rising Head Test Date: 11/28/16 Time: 09:26:36

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G270

AQUIFER DATA

Saturated Thickness: 16.41 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G270

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.35 ft Static Water Column Height: 16.41 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.79 ft Total Well Penetration Depth: 16.41 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.415	0.2566	3.017	2.069	1.715
0.003683	3.383	0.2747	3.	2.199	1.653
0.00735	3.366	0.2937	2.981	2.329	1.582
0.01102	3.347	0.3136	2.957	2.469	1.521
0.01857	3.338	0.3347	2.934	2.619	1.453
0.02223	3.333	0.3567	2.912	2.779	1.387
0.02595	3.325	0.3806	2.889	2.939	1.327
0.02962	3.297	0.4057	2.864	3.119	1.257
0.03803	3.298	0.4316	2.837	3.309	1.188
0.04172	3.29	0.4597	2.811	3.509	1.121
0.0454	3.291	0.4896	2.782	3.719	1.053
0.04907	3.236	0.5206	2.759	3.939	0.988
0.05722	3.262	0.5546	2.727	4.179	0.92

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06092	3.259	0.5897	2.696	4.429	0.861
0.06458	3.255	0.6267	2.662	4.689	0.799
0.06827	3.253	0.6666	2.628	4.969	0.735
0.07693	3.238	0.7087	2.594	5.269	0.682
0.08063	3.224	0.7527	2.557	5.579	0.625
0.08433	3.228	0.7996	2.52	5.919	0.57
0.08803	3.221	0.8497	2.484	6.269	0.516
0.09683	3.204	0.9026	2.443	6.639	0.467
0.1006	3.197	0.9587	2.397	7.039	0.421
0.1089	3.196	1.019	2.357	7.459	0.377
0.1167	3.184	1.079	2.315	7.899	0.339
0.1267	3.175	1.149	2.264	8.369	0.303
0.1366	3.156	1.219	2.211	8.869	0.266
0.1467	3.146	1.289	2.169	9.399	0.24
0.1577	3.13	1.369	2.113	9.959	0.204
0.1697	3.116	1.459	2.059	10.56	0.173
0.1827	3.103	1.539	2.008	11.16	0.163
0.1956	3.087	1.639	1.95	11.86	0.148
0.2097	3.067	1,739	1,889	12.56	0.132
0 2246	3 051	1 839	1 835	13.26	0 119
0.2407	3.032	1.949	1.777	14.06	0.103

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00048	cm/sec
Ss	3.0E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.2401 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004785	5.292E-5	+/- 0.000105	9.043	cm/sec
Ss	3.353E-5	1.819E-5	+/- 3.61E-5	1.843	ft ⁻¹
Kz/Kr	1.	1.096	+/- 2.175	0.912	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

 $T = K^*b = 0.2394 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	0.01528 ft ²
Variance	0.0001544 ft ²
Std. Deviation	0.01242 ft
Mean	0.001996 ft
No. of Residuals	102
No. of Estimates	3



Diagnostic Statistics

Estimation complete! Parameter change criterion (ETOL) reached.

Aquifer Model: Confined Solution Method: KGS Model

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.000162	1.925E-5	+/- 3.835E-5	8.418	cm/sec
Ss	0.0007608	0.0002807	+/- 0.0005592	2.71	ft ⁻¹
Kz/Kr	1.	1.965	+/- 3.914	0.5089	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.03635 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.97	-0.97
Ss	-0.97	1.00	0.88
Kz/Kr	-0.97	0.88	1.00

Residual Statistics

Sum of Squares	0.02832 ft ²
Variance	0.0003726 ft ²
Std. Deviation	0.0193 ft
Mean	-0.002737 ft
No. of Residuals	79
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G271rh.aqt Title: G271 @ GMF Recycle Pond – Rising Head Test Date: 11/28/16 Time: 09:27:29

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G271

AQUIFER DATA

Saturated Thickness: 7.36 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G271

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.75 ft Static Water Column Height: 7.36 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.35 ft Total Well Penetration Depth: 7.36 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.752	0.3685	1.642	2.902	1.334
0.003667	1.759	0.3945	1.635	3.082	1.318
0.00735	1.758	0.4225	1.631	3.272	1.297
0.01105	1.753	0.4525	1.625	3.471	1.28
0.01943	1.741	0.4835	1.62	3.682	1.259
0.02312	1.739	0.5175	1.613	3.902	1.241
0.02678	1.741	0.5525	1.611	4.141	1.22
0.03045	1.742	0.5895	1.603	4.391	1.199
0.03727	1.735	0.6295	1.597	4.652	1.178
0.04097	1.729	0.6715	1.589	4.931	1.154
0.04467	1.731	0.7155	1.582	5.231	1.132
0.04837	1.728	0.7625	1.578	5.542	1.104
0.0545	1.732	0.8125	1.568	5.882	1.079

Time (min) 0.06252 0.0715 0.0795 0.0895 0.09952 0.1095 0.1205 0.1325 0.1455	Displacement (ft) 1.726 1.711 1.717 1.714 1.711 1.709 1.701 1.7 1.691	Time (min) 0.8655 0.9215 0.9815 1.042 1.111 1.182 1.252 1.331 1.422	Displacement (ft) 1.561 1.555 1.546 1.54 1.524 1.521 1.511 1.502 1.492	Time (min) 6.231 6.601 7.002 7.422 7.862 8.332 8.832 9.361 9.922	Displacement (ft) 1.054 1.022 0.994 0.968 0.933 0.903 0.872 0.837 0.805
0.1325 0.1455 0.1585 0.1725 0.1875 0.2035 0.2195 0.2375 0.2565 0.2765 0.2975 0.3195	$\begin{array}{c} 1.7\\ 1.691\\ 1.691\\ 1.688\\ 1.679\\ 1.676\\ 1.671\\ 1.668\\ 1.665\\ 1.659\\ 1.656\\ 1.656\\ 1.648\end{array}$	1.331 1.422 1.502 1.601 1.702 1.802 1.912 2.031 2.162 2.292 2.432 2.582	1.502 1.492 1.48 1.475 1.458 1.446 1.434 1.42 1.405 1.392 1.377 1.365	9.361 9.922 10.52 11.13 11.82 12.52 13.22 14.02 14.92 15.72 16.72 17.72	0.837 0.805 0.77 0.737 0.702 0.661 0.626 0.584 0.552 0.52 0.52 0.479 0.441
0.3435	1.642	2.741	1.352	18.72	0.404

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00011	cm/sec
Ss	0.0003	ft ⁻¹
K7/Kr	1.	

 $T = K^*b = 0.02468 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0001097	2.066E-5	+/- 4.102E-5	5.308	cm/sec
Ss	0.0002898	0.0001997	+/- 0.0003965	1.451	ft ⁻¹
Kz/Kr	1.	2.523	+/- 5.007	0.3964	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

$T = K^*b = 0.02461 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.97
Kz/Kr	-0.99	0.97	1.00

Residual Statistics

Sum of Squares	0.02807 ft ²
Variance	0.0002924 ft ²
Std. Deviation	0.0171 ft
Mean	-0.003834 ft
No. of Residuals	99
No. of Estimates	3


Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G273fh.aqt Title: G273 @ GMF Recycle Pond – Falling Head Test Date: 11/28/16 Time: 09:27:52

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G273

AQUIFER DATA

Saturated Thickness: 6.69 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G273

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.31 ft Static Water Column Height: 6.69 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.48 ft Total Well Penetration Depth: 6.69 ft

No. of Observations: 89

		Observ	vation Data		
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
-3.499	1.386	22.86	0.973	169.8	0.252
-3.249	1.325	24.72	0.954	180.6	0.234
-2.85	1.405	26.76	0.928	192.	0.214
-2.639	1.363	28.86	0.91	204.	0.197
-2.279	1.387	31.08	0.887	216.6	0.179
-1.859	1.36	33.48	0.868	229.8	0.165
-1.439	1.343	36.	0.844	244.2	0.15
-1.02	1.331	38.64	0.82	259.2	0.136
0.	1.307	41.46	0.794	274.8	0.122
0.48	1.296	44.46	0.774	291.6	0.108
1.081	1.282	47.64	0.745	309.6	0.099
1.681	1.272	51.	0.722	328.2	0.087
2.281	1.261	54.6	0.695	348.6	0.082

1

Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
2.94	1.252	58.2	0.668	369.6	0.072
3.66	1.237	62.4	0.643	391.8	0.064
4.44	1.223	66.6	0.62	415.8	0.061
5.22	1.209	70.8	0.593	441.	0.052
6.061	1.195	75.65	0.566	467.4	0.04
6.96	1.18	81.	0.537	495.6	0.039
7.92	1.167	85.8	0.514	525.6	0.033
8.88	1.154	91.8	0.487	557.4	0.031
9.96	1.137	97.8	0.461	591.	0.021
11.1	1.12	103.8	0.439	627.	0.022
12.3	1.105	110.4	0.412	663.	0.019
12.56	1.089	117.6	0.291	705	0.012
11.1	1.12	103.8	0.439	627.	0.022
12.3	1.105	110.4	0.412	663.	0.019
13.56	1.088	117.6	0.381	705	0.012
14.88 16.32 17.82	1.065 1.049 1.032	125.4 133.2 141.6	0.363 0.337 0.318	747. 789. 837.	0.012 0.013 0.009 0.004
19.38 21.06	1.014 0.993	150.6 160.2	0.292 0.275	891.	0.004

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.001	cm/sec
Ss	9.0E-5	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.2039 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	/
KĽ	0.001024	0.0001056	+/- 0.0002099	9.693	cm/sec
Ss	8.684E-5	3.541E-5	+/- 7.039E-5	2.452	ft ⁻¹
Kz/Kr	1.	2.292	+/- 4.557	0.4362	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.2087 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.47	-0.92
Ss	-0.47	1.00	0.11
Kz/Kr	-0.92	0.11	1.00

Residual Statistics

Sum of Squares	0.02896 ft ²
	0.0003367 ft ²
Std. Deviation	0.01835 ft
Mean	0.004437 ft
No. of Residuals	89
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G273rh.aqt Title: G273 @ GMF Recycle Pond – Rising Head Test Date: 11/28/16 Time: 09:28:15

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: G273

AQUIFER DATA

Saturated Thickness: 6.69 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G273

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.64 ft Static Water Column Height: 6.69 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.48 ft Total Well Penetration Depth: 6.69 ft

	Observation Data					
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	
0.	1.692	15.73	1.245	132.3	0.405	
0.123	1.702	16.82	1.227	140.1	0.378	
0.355	1.653	17.95	1.214	148.5	0.351	
0.605	1.609	19.16	1.194	157.5	0.327	
1.131	1.573	20.41	1.175	167.1	0.301	
1.254	1.57	21.73	1.158	176.7	0.282	
1.377	1.558	23.18	1.139	187.5	0.258	
1.605	1.556	24.68	1.12	198.9	0.237	
1.855	1.541	26.23	1.101	210.9	0.218	
2.298	1.522	27.91	1.082	223.5	0.198	
2.421	1.524	29.73	1.058	236.7	0.182	
2.605	1.517	31.57	1.036	251.1	0.165	
2.855	1.506	33.62	1.012	266.1	0.15	

Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
3.105	1.496	35.72	0.994	281.7	0.14
3.355	1.487	37.94	0.97	298.5	0.119
3.605	1.482	40.34	0.944	316.5	0.107
3.855	1.473	42.85	0.92	335.1	0.099
4.215	1.468	45.49	0.897	355.5	0.089
4.575	1.451	48.31	0.869	376.5	0.082
4.995	1.441	51.31	0.843	398.7	0.069
5.415	1.428	54.49	0.816	422.7	0.059
5 835	1 42	57.85	0.79	447 9	0.054
6 315	1 411	61 45	0 765	474 3	0.049
6 855	1 397	65.06	0 734	502 5	0.041
7 335	1 387	69.25	0.708	532.5	0.038
7.035	1 37/	73 45	0.700	564 3	0.030
0 5 2 5	1.374	75.45	0.075	507.0	0.033
0.000	1.303	77.00	0.000	277.7 422.0	0.020
9.130	1.301	82.40	0.023	033.9	0.023
9.795	1.338	87.86	0.597	669.9	0.023
10.52	1.331	92.66	0.567	/11.9	0.017
11.3	1.316	98.66	0.534	753.9	0.015
12.07	1.3	104.7	0.509	795.9	0.011
12.91	1.287	110.7	0.482	843.9	0.012
13.82	1.273	117.3	0.456	897.9	0.009
14.78	1.26	124.5	0.431	945.9	0.008

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00083	cm/sec
Ss	0.0015	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1692 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0008319	7.881E-5	+/- 0.0001563	10.56	cm/sec
Ss	0.001271	0.0003781	+/- 0.0007497	3.362	ft ⁻¹
Kz/Kr	1.	1.63	+/- 3.232	0.6135	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.1696 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.98
Ss	-0.99	1.00	0.95
Kz/Kr	-0.98	0.95	1.00

Residual Statistics

Sum of Squares 0.02784 ft ²	~
Variance	2
Std. Deviation 0.01652 ft	
Vean	t
No. of Residuals 105	
No. of Estimates 3	



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G279fh.aqt Title: G279 @ GMF Recycle Pond – Falling Head Test Date: 11/28/16 Time: 09:28:43

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G279

AQUIFER DATA

Saturated Thickness: 7. ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G279

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.15 ft Static Water Column Height: 7. ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.39 ft Total Well Penetration Depth: 7. ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.068	0.2487	1.429	1.831	0.295
0.003683	2.003	0.2667	1.398	1.941	0.273
0.00735	2.023	0.2857	1.366	2.061	0.248
0.011	2.109	0.3057	1.329	2.191	0.226
0.0187	2.042	0.3267	1.297	2.321	0.206
0.02238	2.008	0.3487	1.259	2.461	0.19
0.02607	1.986	0.3727	1.223	2.611	0.172
0.02973	1.974	0.3977	1.186	2.771	0.153
0.0381	1.959	0.4237	1.149	2.931	0.137
0.0418	1.943	0.4517	1.113	3.111	0.126
0.04547	1.925	0.4817	1.071	3.301	0.116
0.04915	1.916	0.5127	1.031	3.501	0.102
0.05758	1.893	0.5467	0.992	3.711	0.094

Time (min) 0.06127 0.06497 0.06865 0.07633 0.08003 0.08372 0.09167	Displacement (ft) 1.88 1.868 1.858 1.839 1.826 1.812 1.788 1.762	Time (min) 0.5817 0.6187 0.6587 0.7007 0.7447 0.7917 0.8417	Displacement (ft) 0.952 0.913 0.879 0.833 0.797 0.76 0.72 0.602	Time (min) 3.931 4.171 4.421 4.681 4.961 5.261 5.571 5.011	Displacement (ft) 0.086 0.077 0.07 0.066 0.06 0.056 0.052 0.047
0.1087	1.742	0.9507	0.642	6.261	0.041
0.1187 0.1287	1.689	1.011 1.071	0.609 0.574	6.631 7.031	0.044 0.038
0.1387 0.1497	1.667 1.64	1.141 1.211	0.533 0.502	7.451 7.891	0.038
0.1617 0.1747	1.614	1.281	0.473	8.361 8.861	0.034
0.1877 0.2017 0.2167 0.2227	1.555 1.527 1.492	1.451 1.531 1.631 1.721	0.403 0.381 0.345	9.391 9.951 10.55	0.035 0.03 0.032 0.032

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0017	cm/sec
Ss	0.0007	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.3627 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.001719	0.0001754	+7- 0.0003483	9.798	cm/sec
Ss	0.0007021	0.0002498	+/- 0.000496	2.811	ft ⁻¹
Kz/Kr	1.	1.488	+/- 2.955	0.672	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.99
Ss	-0.99	1.00	0.97
Kz/Kr	-0.99	0.97	1.00

Residual Statistics

Sum of Squares 0.02749 ft ²
Variance
Std. Deviation 0.01719 ft
Mean
No. of Residuals 96
No. of Estimates 3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G279rh.aqt Title: G279 @ GMF Recycle Pond – Rising Head Test Date: 11/28/16 Time: 10:43:04

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: G279

AQUIFER DATA

Saturated Thickness: 7. ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G279

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.48 ft Static Water Column Height: 7. ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.39 ft Total Well Penetration Depth: 7. ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.581	0.2162	1.635	1.659	0.405
0.004167	2.516	0.2302	1.608	1.759	0.375
0.008333	2.433	0.2452	1.574	1.859	0.345
0.0125	2.383	0.2612	1.545	1.969	0.317
0.01667	2.339	0.2772	1.513	2.089	0.289
0.02083	2.308	0.2952	1.479	2.219	0.263
0.025	2.269	0.3142	1.456	2.349	0.241
0.02917	2.235	0.3342	1.415	2.489	0.213
0.03333	2.212	0.3552	1.38	2.639	0.198
0.0375	2.177	0.3772	1.345	2.799	0.18
0.04167	2.165	0.4012	1.312	2.959	0.165
0.04583	2.135	0.4262	1.279	3.139	0.146
0.05	2.1	0.4522	1.24	3.329	0.132

Time (min) 0.05417 0.05833 0.0625 0.06667 0.07083 0.075 0.07917 0.08517 0.09117 0.09817 0.1062 0.1122	Displacement (ft) 2.086 2.067 2.053 2.041 2.012 2.003 1.993 1.974 1.95 1.926 1.893 1.886	Time (min) 0.4802 0.5102 0.5412 0.5752 0.6102 0.6472 0.6872 0.7292 0.7732 0.8202 0.8702 0.9232	Displacement (ft) 1.206 1.168 1.127 1.09 1.048 1.018 0.974 0.936 0.899 0.857 0.819 0.781	Time (min) 3.529 3.739 3.959 4.199 4.449 4.709 4.989 5.291 5.599 5.939 6.289 6.659	Displacement (ft) 0.118 0.105 0.093 0.086 0.082 0.072 0.065 0.051 0.058 0.055 0.05 0.05 0.05
0.1062	1.893	0.8702	0.819	6.289	0.05
0.1122	1.886	0.9232	0.781	6.659	0.034
0.1225 0.1292	1.860 1.844 1.834	0.9232 0.9792 1.039	0.738 0.699	7.059 7.479	0.034 0.042 0.039
0.1372	1.822	1.099	0.662	7.919	0.039
0.1472	1.788	1.169	0.621	8.389	0.033
0.1572	1.758	1.239	0.579	8.889	0.035
0.1672	1.747	1.309	0.545	9.419	0.029
0.1829	1.704	1.389	0.509	9.979	0.03
0.1902	1.694	1.479	0.473	10.58	0.029
0.2032	1.666	1.559	0.44	11.18	0.027

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0015	cm/sec
Ss	0.0025	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.32 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.001475	0.0001155	+/- 0.0002292	12.77	cm/sec
Ss	0.00256	0.0005911	+/- 0.001173	4.331	ft ⁻¹
Kz/Kr	1.	1.448	+/- 2.872	0.6908	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

 $T = K^*b = 0.3147 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.98	-0.96
Ss	-0.98	1.00	0.91
Kz/Kr	-0.96	0.91	1.00

Residual Statistics

Sum of Squares 0.08946 ft ²
Variance
Std. Deviation 0.03006 ft
Mean
No. of Residuals 102
No. of Estimates 3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G280fh.aqt Title: G280 @ GMF Recycle Pond – Falling Head Test Date: 11/28/16 Time: 09:29:18

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 17 Jun 2016 Test Well: G280

AQUIFER DATA

Saturated Thickness: 15.26 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G280

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.96 ft Static Water Column Height: 15.26 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.84 ft Total Well Penetration Depth: 15.26 ft

		Observ	vation Data		
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
0.	1.974	24.31	1.378	153.8	0.263
0.417	1.933	26.11	1.349	163.4	0.231
0.61	1.928	27.97	1.312	173.1	0.205
0.971	1.913	30.01	1.277	183.8	0.18
1.607	1.763	32.11	1.24	195.3	0.155
1.81	1.964	34.33	1.203	207.3	0.135
2.23	1.879	36.73	1.164	219.8	0.117
2.789	1.866	39.25	1.124	233.1	0.102
3.25	1.849	41.89	1.087	247.4	0.087
3.73	1.837	44.71	1.046	262.4	0.069
4.33	1.825	47.71	1.006	278.	0.065
4.93	1.805	50.89	0.966	294.8	0.049
5.53	1.79	54.25	0.924	312.8	0.045

<u>Time (sec)</u> 6.19 6.91 7.69 8.47 9.31 10.21 11.17 12.13 13.21 14.35 15.55 16.81 18.13 19.57 21.07	Displacement (ft) 1.772 1.748 1.778 1.615 1.705 1.674 1.652 1.63 1.605 1.58 1.552 1.527 1.496 1.467 1.438	Time (sec) 57.85 61.45 65.65 69.85 74.05 78.85 84.25 89.05 95.05 101.1 107.1 113.6 120.8 128.7 136.4	Displacement (ft) 0.88 0.842 0.797 0.75 0.714 0.67 0.625 0.588 0.544 0.509 0.463 0.428 0.39 0.352 0.322	Time (sec) 331.4 351.8 372.8 395. 419. 444.3 470.6 499. 528.8 560.6 594.2 630.4 666.2 708.2 750.2	Displacement (ft) 0.039 0.03 0.024 0.019 0.018 0.016 0.014 0.017 0.006 0.008 0.007 0.004 0.008 0.004 0.008 0.006 0.004
22.63	1.408	144.8	0.289	750.2	0.004

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00125	cm/sec
Ss	2.0E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.5814 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G280rh.aqt Title: G280 @ GMF Recycle Pond – Rising Head Test Date: 11/28/16 Time: 09:29:41

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 17 Jun 2016 Test Well: G280

AQUIFER DATA

Saturated Thickness: 15.26 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G280

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.13 ft Static Water Column Height: 15.26 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.84 ft Total Well Penetration Depth: 15.26 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.135	0.2583	1.622	2.07	0.369
0.002083	2.133	0.2763	1.594	2.2	0.337
0.006167	2.114	0.2953	1.568	2.33	0.303
0.01033	2.097	0.3153	1.542	2.47	0.273
0.0145	2.087	0.3363	1.511	2.62	0.246
0.02105	2.06	0.3583	1.486	2.78	0.219
0.0231	2.054	0.3823	1.447	2.94	0.193
0.027	2.049	0.4073	1.424	3.12	0.163
0.03117	2.037	0.4333	1.388	3.31	0.144
0.03533	2.024	0.4613	1.351	3.51	0.125
0.04208	2.006	0.4913	1.323	3.72	0.11
0.04413	2.004	0.5223	1.283	3.94	0.093
0.04783	2.001	0.5563	1.251	4.18	0.078

Time (min) 0.052 0.05617 0.0631 0.06633 0.07235 0.08297 0.08633 0.09333	Displacement (ft) 1.986 1.979 1.96 1.955 1.943 1.92 1.916 1.899 1.924	Time (min) 0.5913 0.6283 0.6683 0.7103 0.7543 0.8013 0.8513 0.9043 0.9043	Displacement (ft) 1.214 1.173 1.135 1.097 1.06 1.019 0.978 0.932 0.901	Time (min) 4.43 4.69 4.97 5.27 5.58 5.921 6.27 6.64 7.04	Displacement (ft) 0.064 0.055 0.049 0.039 0.037 0.029 0.022 0.018 0.016
0.09333	1.899	0.9043	0.932	6.64	0.018
0.103	1.884	0.9603	0.891	7.04	0.016
0.1103	1.868	1.02	0.853	7.46	0.015
0.1191	1.849	1.08	0.809	7.9	0.008
0.1283	1.834	1.15	0.768	8.37	0.011
0.1383	1.814	1.22	0.727	8.87	0.007
0.1483	1.798	1.29	0.686	9.4	0.01
0.1593	1.779	1.37	0.642	9.96	0.011
0.1713	1.761	1.46	0.599	10.56	0.01
0.1843	1.739	1.54	0.562	11.16	0.01
0.1973	1.718	1.64	0.524	11.86	0.005
0.2113	1.696	1.74	0.479	12.56	0.005
0.2263	1.672	1.84	0.445	13.26	0.008
0.2423	1.645	1.95	0.409		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0013	cm/sec
Ss	6.0E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.6047 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.001322	0.0001102	+/- 0.0002187	11.99	cm/sec
Ss	5.714E-5	2.337E-5	+/- 4.636E-5	2.445	ft ⁻¹
Kz/Kr	1.	0.8362	+/- 1.659	1.196	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

 $T = K^*b = 0.6148 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	0.005558 ft_2^2
Variance	5.672E-5 ft ²
Std. Deviation	0.007531 ft
Mean	-0.002476 ft
No. of Residuals	101
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G281fh.aqt Title: G281 @ Ash Pond 2 – Falling Head Test Date: 11/28/16 Time: 09:30:07

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 17 Jun 2016 Test Well: G281

AQUIFER DATA

Saturated Thickness: 15.8 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G281

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.93 ft Static Water Column Height: 15.8 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.65 ft Total Well Penetration Depth: 15.8 ft

		Observ	ation Data		
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
0.	1.951	21.91	1.404	154.7	0.229
0.123	1.921	23.47	1.377	164.3	0.194
0.344	1.925	25.15	1.347	173.9	0.171
0.594	1.91	26.95	1.31	184.7	0.153
0.844	1.904	28.81	1.282	196.1	0.131
1.261	1.888	30.85	1.246	208.1	0.109
1.453	1.904	32.95	1.208	220.7	0.093
1.816	1.878	35.17	1.176	233.9	0.075
2.451	1.863	37.57	1.139	248.3	0.062
2.653	1.839	40.09	1.105	263.3	0.052
3.073	1.843	42.73	1.062	278.9	0.042
3.651	1.826	45.55	1.024	295.7	0.035
4.093	1.81	48.55	0.984	313.7	0.031

Time (sec) 4.689 5.173 5.774 6.374 7.033 7.753 8.533 9.313 10.15 11.05 12.01 12.97 14.05	Displacement (ft) 1.796 1.782 1.763 1.751 1.733 1.714 1.695 1.677 1.657 1.637 1.618 1.592 1.566	Time (sec) 51.73 55.09 58.69 62.29 66.49 70.69 74.89 79.69 85.09 89.89 95.89 101.9 107.9	Displacement (ft) 0.941 0.901 0.86 0.822 0.768 0.725 0.686 0.643 0.599 0.559 0.512 0.473 0.435	Time (sec) 332.3 352.7 373.7 395.9 419.9 445.1 471.5 499.7 529.7 561.5 595.1 631.1 667.1	Displacement (ft) 0.025 0.02 0.013 0.011 0.01 0.007 0.009 0.004 0.002 0.006 0.002 0.001 -0.001
12.01 12.97 14.05	1.618 1.592 1.566	95.89 101.9 107.9	0.512 0.473 0.435	631.1 667.1	0.002 0.001 -0.001
15.19 16.39 17.65	1.542 1.517 1.494	114.5 121.7 129.5 127.2	0.397 0.356 0.32	709.1 751.1 793.2	0. -0.003 0.005
20.41	1.437	145.7	0.255		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0021	cm/sec
Ss	2.0E-12	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 1.011 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.002112	0.0008403	+/- 0.00167	2.513	cm/sec
Ss	1.86E-12	9.133E-9	+/- 1.815E-8	0.0002036	ft ⁻¹
Kz/Kr	1.	2098.5	+/- 4169.8	0.0004765	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 1.017 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.97	0.97
Ss	-0.97	1.00	-1.00
Kz/Kr	0.97	-1.00	1.00

Residual Statistics

Sum of Squares 0.01011 ft ²	
Variance	
Std. Deviation 0.01072 ft	
Mean0.00248 ft	
No. of Residuals 91	
No. of Estimates 3	



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G281rh.aqt Title: G281 @ Ash Pond 2 – Rising Head Test Date: 11/28/16 Time: 09:31:04

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 17 Jun 2016 Test Well: G281

AQUIFER DATA

Saturated Thickness: 15.8 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G281

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.09 ft Static Water Column Height: 15.8 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.65 ft Total Well Penetration Depth: 15.8 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.116	0.2528	1.733	2.081	0.632
0.004167	2.1	0.2688	1.716	2.211	0.586
0.0105	2.067	0.2868	1.697	2.341	0.553
0.01257	2.063	0.3058	1.677	2.481	0.515
0.01667	2.054	0.3258	1.661	2.631	0.47
0.02083	2.051	0.3468	1.636	2.791	0.433
0.025	2.033	0.3688	1.619	2.951	0.398
0.03152	2.021	0.3928	1.598	3.131	0.359
0.03358	2.019	0.4178	1.574	3.321	0.325
0.0375	2.012	0.4438	1.549	3.521	0.297
0.04167	2.002	0.4718	1.527	3.731	0.259
0.04583	1.996	0.5018	1.5	3.951	0.236
0.05258	1.984	0.5328	1.477	4.191	0.208

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05465	1.977	0.5668	1.446	4.441	0.183
0.05833	1.966	0.6018	1.419	4.701	0.162
0.0625	1.962	0.6388	1.384	4.981	0.145
0.06667	1.96	0.6788	1.357	5.281	0.125
0.07362	1.952	0.7208	1.328	5.591	0.11
0.07682	1.942	0.7648	1.295	5.931	0.095
0.08287	1,935	0.8118	1.263	6.281	0.077
0.09345	1.919	0.8618	1.225	6.651	0.069
0.09682	1.917	0.9148	1.195	7.051	0.062
0.1038	1,906	0.9708	1,154	7.471	0.053
0.1135	1.889	1.031	1.115	7.911	0.046
0.1208	1.887	1.091	1.081	8.381	0.035
0.1316	1.869	1.161	1.04	8.881	0.035
0.1388	1.859	1.231	1.001	9.411	0.025
0.1488	1.85	1.301	0.966	9.971	0.02
0.1588	1.836	1.381	0.925	10.57	0.02
0.1698	1.821	1.471	0.878	11.17	0.016
0.1818	1.811	1.551	0.842	11.87	0.01
0.1948	1.797	1.651	0.797	12.57	0.015
0.2078	1,789	1.751	0.756	13.27	0.013
0.2218	1.761	1.851	0.711	14.07	0.016
0.2368	1.749	1.961	0.674		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00089	cm/sec
Ss	6.2E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.4286 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0008928	0.0001189	+/- 0.0002359	7.51	cm/sec
Ss	6.185E-5	4.021E-5	+/- 7.978E-5	1.538	ft ⁻¹
Kz/Kr	1.	1.309	+/- 2.598	0.7637	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

$T = K^*b = 0.4299 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-1.00	-1.00
Ss	-1.00	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

Sum of Squares	0.01291 ft ²
Variance	0.0001278 ft ²
Std. Deviation	0.01131 ft
Mean	-0.003426 ft
No. of Residuals	104
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G301fh.aqt Title: G301 @ Ash Pond 1 – Falling Head Test Date: 11/28/16 Time: 09:31:30

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G301

AQUIFER DATA

Saturated Thickness: 11.82 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G301

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.3 ft Static Water Column Height: 11.82 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.65 ft Total Well Penetration Depth: 11.82 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.31	0.4745	2.745	3.672	1.428
0.004167	3.255	0.5085	2.719	3.893	1.375
0.008333	3.267	0.5435	2.691	4.133	1.319
0.0125	3.263	0.5805	2.658	4.383	1.264
0.0185	3.248	0.6205	2.633	4.643	1.205
0.02452	3.236	0.6625	2.6	4.923	1.145
0.0315	3.222	0.7065	2.57	5.223	1.089
0.0385	3.23	0.7535	2.54	5.532	1.034
0.0455	3.199	0.8035	2.51	5.872	0.975
0.0535	3.198	0.8565	2.478	6.223	0.917
0.0625	3.185	0.9125	2.442	6.593	0.861
0.0705	3.174	0.9725	2.404	6.992	0.809
0.0805	3.154	1.033	2.37	7.412	0.752

Time (min) 0.0905 0.1005 0.1115 0.1235 0.1365 0.1495 0.1635 0.1785 0.1945 0.2105 0.2285 0.2475 0.2885 0.3105 0.3345 0.3595 0.3855	Displacement (ft) 3.145 3.123 3.111 3.099 3.084 3.064 3.048 3.029 3.013 2.995 2.976 2.955 2.935 2.915 2.895 2.895 2.867 2.846 2.814	Time (min) 1.103 1.173 1.242 1.323 1.413 1.492 1.593 1.692 1.793 1.903 2.023 2.152 2.283 2.422 2.572 2.732 2.893 3.072	Displacement (ft) 2.337 2.296 2.251 2.22 2.18 2.143 2.098 2.053 2.012 1.964 1.927 1.876 1.83 1.78 1.734 1.684 1.639 1.584	Time (min) 7.853 8.322 8.822 9.352 9.912 10.51 11.11 11.81 12.51 13.21 14.01 14.91 15.71 16.71 17.72 18.72 19.81 21.01	Displacement (ft) 0.697 0.647 0.593 0.544 0.5 0.45 0.405 0.326 0.292 0.258 0.227 0.204 0.179 0.151 0.132 0.114 01
0.3595 0.3855 0.4135 0.4435	2.846 2.814 2.797 2.77	2.893 3.072 3.263 3.462	1.639 1.584 1.535 1.484	19.81 21.01	0.114 0.1

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00027	cm/sec
Ss	0.0009	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.09727 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0002698	1.629E-5	+/- 3.234E-5	16.56	cm/sec
Ss	0.0008799	0.0001922	+/- 0.0003815	4.579	ft ⁻¹
Kz/Kr	1.	0.8564	+/- 1.7	1.168	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

 $T = K^*b = 0.09719 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.98	-0.97
Ss	-0.98	1.00	0.93
Kz/Kr	-0.97	0.93	1.00

Residual Statistics

Sum of Squares	0.07757 ft ²
Variance	0.0008252 ft ²
Std. Deviation	0.02873 ft
Mean	-0.006371 ft
No. of Residuals	97
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G301rh.aqt Title: G301 @ Ash Pond 1 – Rising Head Test Date: 11/28/16 Time: 09:32:12

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G301

AQUIFER DATA

Saturated Thickness: 11.82 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G301

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.37 ft Static Water Column Height: 11.82 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.65 ft Total Well Penetration Depth: 11.82 ft

Observation Data							
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
-0.02083	3.462	0.2933	2.946	2.938	1.265		
-0.01667	3.446	0.3133	2.923	3.118	1.201		
-0.0125	3.401	0.3343	2.902	3.308	1.137		
0.	3.35	0.3563	2.874	3.508	1.071		
0.004167	3.343	0.3803	2.85	3.718	1.007		
0.008333	3.341	0.4053	2.823	3.938	0.94		
0.0125	3.326	0.4313	2.801	4.178	0.884		
0.01667	3.313	0.4593	2.77	4.428	0.822		
0.02083	3.312	0.4893	2.741	4.688	0.766		
0.025	3.305	0.5203	2.714	4.968	0.705		
0.02917	3.298	0.5543	2.681	5.268	0.653		
0.03333	3.3	0.5893	2.65	5.578	0.597		
0.04102	3.275	0.6263	2.615	5.918	0.552		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
------------	-------------------	------------	-------------------	------------	-------------------		
0.04472	3.264	0.6663	2.581	6.268	0.499		
0.04838	3.256	0.7083	2.544	6.638	0.449		
0.05208	3.26	0.7523	2.507	7.038	0.407		
0.0559	3.249	0.7993	2.466	7.465	0.366		
0.05975	3.244	0.8493	2.427	7.898	0.327		
0.06433	3.236	0.9023	2.385	8.368	0.29		
0.07033	3.232	0.9583	2.338	8.868	0.252		
0.07733	3.219	1.018	2.295	9.398	0.222		
0.08433	3.208	1.078	2.25	9.958	0.19		
0.09133	3.198	1.148	2.2	10.56	0.161		
0.09933	3.187	1.218	2.151	11.16	0.141		
0.1083	3.178	1.288	2.106	11.86	0.124		
0.1163	3.16	1.368	2.051	12.56	0.102		
0.1263	3.149	1.458	1.993	13.26	0.087		
0.1363	3.132	1.538	1.941	14.06	0.07		
0.1463	3.123	1.638	1.885	14.96	0.055		
0.1573	3.108	1.738	1.821	15.76	0.033		
0.1693	3.089	1.838	1.771	16.76	0.032		
0.1823	3.074	1.948	1.711	17.76	0.028		
0.1953	3.057	2.068	1.649	18.76	0.014		
0.2093	3.043	2.198	1.585	19.86	0.012		
0.2243	3.022	2.328	1.523	21.06	0.009		
0.2403	3.004	2.468	1.459	22.36	0.005		
0.2563	2.986	2.618	1.395				
0.2743	2.963	2.778	1.33				

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0005	cm/sec
Ss	9.0E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1801 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004957	5.04E-5	+/- 9.988E-5	9.836	cm/sec
Ss	9.396E-5	4.339E-5	+/- 8.599E-5	2.166	ft ⁻¹

AQTESOLV	for	Windows	

Kz/Kr	1.	1.072	+/- 2.124	0.933
	••		.,	0.700

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.1786 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.98
Kz/Kr	-1.00	0.98	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.03615 ft ²
	0.0003317 ft ²
Std. Deviation	0.01821 ft
Mean	-0.004631 ft
No. of Residuals	112
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G302fh.aqt Title: G302 @ Ash Pond 1 – Falling Head Test Date: 11/28/16 Time: 09:32:35

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G302

AQUIFER DATA

Saturated Thickness: 11.37 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G302

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.05 ft Static Water Column Height: 11.37 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.65 ft Total Well Penetration Depth: 11.37 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.007	0.3938	2.54	2.901	1.102
0.004167	2.93	0.4218	2.519	3.081	1.044
0.008333	2.962	0.4518	2.486	3.271	0.989
0.0125	3.015	0.4828	2.468	3.471	0.932
0.01667	3.036	0.5168	2.421	3.681	0.88
0.02083	3.017	0.5518	2.387	3.901	0.83
0.02683	2.978	0.5888	2.356	4.141	0.777
0.03283	2.971	0.6288	2.321	4.391	0.731
0.03983	2.97	0.6708	2.284	4.651	0.679
0.04683	2.958	0.7148	2.243	4.931	0.632
0.05382	2.942	0.7618	2.197	5.231	0.589
0.06183	2.934	0.8118	2.161	5.541	0.542
0.07083	2.923	0.8648	2.122	5.881	0.497

Time (min) 0.07883 0.08883 0.1088 0.1198 0.1318 0.1448 0.1578 0.1718 0.1718 0.1868 0.2028 0.2188 0.2258 0.2558 0.2758 0.2968 0.3188	Displacement (ft) 2.896 2.899 2.877 2.862 2.893 2.805 2.827 2.803 2.786 2.768 2.752 2.731 2.709 2.701 2.667 2.644 2.62	Time (min) 0.9208 0.9808 1.041 1.111 1.181 1.251 1.331 1.421 1.501 1.601 1.701 1.801 1.911 2.031 2.161 2.291 2.431	Displacement (ft) 2.077 2.034 1.997 1.942 1.894 1.857 1.796 1.744 1.7 1.643 1.59 1.539 1.483 1.435 1.401 1.32 1.26	Time (min) 6.231 6.601 7.001 7.421 7.861 8.331 8.831 9.361 9.921 10.52 11.12 11.82 12.52 13.22 14.02 14.92 15.72	Displacement (ft) 0.461 0.42 0.379 0.341 0.279 0.247 0.216 0.192 0.171 0.149 0.131 0.114 0.101 0.086 0.074 0.066
0.2988 0.3188 0.3428 0.3678	2.644 2.62 2.597 2.573	2.291 2.431 2.581 2.741	1.32 1.26 1.223 1.155	14.92 15.72 16.72	0.074 0.066 0.056

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00049	cm/sec
Ss	0.0002	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1698 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0004933	4.307E-5	+/- 8.553E-5	11.45	cm/sec
Ss	0.0001891	7.368E-5	+/- 0.0001463	2.566	ft ⁻¹
Kz/Kr	1.	0.9375	+/- 1.862	1.067	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

$T = K^*b = 0.171 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.98
Kz/Kr	-1.00	0.98	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.02827 ft ²
Variance	0.0003073 ft ²
Std. Deviation	0.01753 ft
Mean	0.0006103 ft
No. of Residuals	95
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G302rh.aqt Title: G302 @ Ash Pond 1 – Rising Head Test Date: 11/28/16 Time: 09:32:58

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G302

AQUIFER DATA

Saturated Thickness: 11.37 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G302

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.97 ft Static Water Column Height: 11.37 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.65 ft Total Well Penetration Depth: 11.37 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.004617	2.935	0.224	2.609	1.88	1.297
0.0084	2.907	0.237	2.593	1.99	1.246
0.01393	2.92	0.251	2.579	2.11	1.19
0.01758	2.907	0.266	2.558	2.24	1.132
0.02123	2.908	0.282	2.534	2.37	1.079
0.02502	2.901	0.298	2.516	2.511	1.021
0.02918	2.903	0.316	2.496	2.66	0.97
0.03335	2.889	0.335	2.474	2.82	0.912
0.03752	2.882	0.355	2.451	2.98	0.863
0.04168	2.869	0.376	2.43	3.16	0.811
0.04585	2.871	0.398	2.405	3.35	0.758
0.05002	2.862	0.422	2.375	3.55	0.708
0.05418	2.857	0.447	2.346	3.763	0.658

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05835	2.848	0.473	2.32	3.98	0.608
0.06252	2.842	0.501	2.292	4.22	0.562
0.06668	2.84	0.531	2.263	4.47	0.518
0.07085	2.829	0.562	2.227	4.73	0.473
0.07502	2.821	0.596	2.193	5.015	0.432
0.07918	2.819	0.631	2.162	5.31	0.394
0.08335	2.809	0.668	2.125	5.62	0.358
0.08752	2.796	0.708	2.086	5.96	0.325
0.09168	2.793	0.75	2.052	6.31	0.286
0.09585	2.786	0.794	2.017	6.684	0.258
0.1	2.782	0.841	1.972	7.08	0.234
0.106	2.775	0.891	1.928	7.5	0.206
0.112	2.766	0.944	1.885	7.94	0.181
0.119	2.757	1.	1.847	8.41	0.159
0.126	2.745	1.06	1.798	8.91	0.14
0.133	2.731	1.12	1.752	9.44	0.127
0.141	2.721	1.19	1.705	10.	0.103
0.15	2.708	1.26	1.658	10.6	0.091
0.158	2.698	1.33	1.609	11.2	0.08
0.168	2.684	1.41	1.559	11.9	0.067
0.178	2.67	1.5	1.503	12.6	0.058
0.188	2.652	1.58	1.441	13.3	0.05
0.199	2.639	1.68	1.402	14.1	0.043
0.211	2.63	1.78	1.35		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00063	cm/sec
Ss	0.0001	ft ⁻¹
Kz/Kr	1.	

$T = K^*b = 0.2183 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.00063	2.361E-5	+/- 4.679E-5	26.69	cm/sec
Ss	0.0001057	1.773E-5	+/- 3.513E-5	5.966	ft ⁻¹
Kz/Kr	1.	0.4005	+/- 0.7939	2.497	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.2183 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-1.00
Ss	-0.99	1.00	0.99
Kz/Kr	-1.00	0.99	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.00378 ft ²
Variance	3.532E-5 ft ²
Std. Deviation	0.005943 ft
Mean	-0.0002972 ft
No. of Residuals	110
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G303fh.aqt Title: G303 @ Ash Pond 1 – Falling Head Test Date: 11/28/16 Time: 09:33:23

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G303

AQUIFER DATA

Saturated Thickness: 17.03 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G303

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.15 ft Static Water Column Height: 17.03 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 10. ft Total Well Penetration Depth: 17.03 ft

	Observation Data				
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
-0.1886	2.358	0.386	1.897	4.728	1.044
-0.185	2.423	0.426	1.881	5.028	1.012
-0.1813	2.411	0.468	1.863	5.338	0.98
-0.176	2.312	0.512	1.847	5.678	0.949
-0.17	2.203	0.559	1.828	6.028	0.916
-0.163	2.162	0.609	1.81	6.398	0.886
-0.156	2.193	0.662	1.791	6.798	0.855
-0.149	2.224	0.718	1.771	7.218	0.822
-0.141	2.219	0.778	1.754	7.658	0.793
-0.132	2.193	0.838	1.734	8.128	0.761
-0.124	2.178	0.908	1.712	8.628	0.731
-0.114	2.174	0.978	1.69	9.158	0.702
-0.104	2.174	1.048	1.669	9.718	0.672

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
-0.09402	2.163	1.128	1.646	10.32	0.638
-0.08302	2.151	1.218	1.623	10.92	0.606
-0.07102	2.147	1.298	1.604	11.62	0.575
-0.05802	2.138	1.398	1.578	12.32	0.547
-0.04502	2.124	1.498	1.555	13.02	0.517
-0.03102	2.118	1.598	1.53	13.82	0.491
-0.016	2.111	1.708	1.499	14.72	0.462
0.	2.094	1.828	1.476	15.52	0.439
0.016	2.09	1.958	1.448	16.52	0.405
0.034	2.075	2.088	1.426	17.52	0.383
0.053	2.064	2.228	1.423	18.52	0.358
0.073	2.052	2.378	1.366	19.62	0.342
0.09398	2,039	2,538	1.368	20.82	0.311
0.116	2.031	2.698	1.308	22.12	0.284
0.14	2.013	2.878	1.28	23.42	0.264
0.165	2.	3.068	1.255	24.82	0.239
0.191	1.987	3.268	1.223	26.32	0.22
0.219	1,974	3.478	1,193	27.92	0.201
0.249	1,958	3,698	1,162	29.52	0.187
0.28	1.945	3.938	1,136	31.32	0.176
0.314	1.928	4.188	1,102	002	5.170
0.349	1.911	4.448	1.072		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	5.6E-5	cm/sec
Ss	0.002	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.02907 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	5.383E-5	3.02E-6	+/- 5.993E-6	17.82	cm/sec
Ss	0.002308	0.0004271	+/- 0.0008474	5.403	ft ⁻¹
Kz/Kr	1.	1.674	+/- 3.321	0.5974	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

$T = K^*b = 0.02794 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.94	-0.51
Ss	-0.94	1.00	0.32
Kz/Kr	-0.51	0.32	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.2436 ft ²
Variance	0.002436 ft ²
Std. Deviation	0.04936 ft
Mean	0.01078 ft
No. of Residuals	103
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G303rh.aqt Title: G303 @ Ash Pond 1 – Rising Head Test Date: 11/28/16 Time: 09:33:48

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G303

AQUIFER DATA

Saturated Thickness: 17.03 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G303

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.33 ft Static Water Column Height: 17.03 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 10. ft Total Well Penetration Depth: 17.03 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
-0.0315	2.324	0.477	1.972	4.101	1.25	
-0.02733	2.297	0.512	1.961	4.351	1.227	
-0.02317	2.301	0.549	1.945	4.611	1.195	
-0.019	2.32	0.589	1.927	4.891	1.168	
-0.013	2.313	0.631	1.911	5.191	1.138	
-0.007	2.282	0.675	1.898	5.501	1.109	
0.	2.279	0.722	1.883	5.841	1.069	
0.007	2.273	0.772	1.868	6.191	1.044	
0.014	2.255	0.825	1.853	6.561	1.015	
0.022	2.256	0.881	1.835	6.961	0.985	
0.031	2.243	0.941	1.819	7.381	0.955	
0.039	2.231	1.001	1.798	7.821	0.928	
0.049	2.218	1.071	1.782	8.291	0.898	

Time (min) 0.059 0.069 0.08 0.092 0.105 0.118 0.132 0.147 0.163 0.179 0.197 0.216 0.236	Displacement (ft) 2.218 2.196 2.186 2.177 2.162 2.156 2.145 2.134 2.125 2.134 2.125 2.118 2.104 2.096 2.068	Time (min) 1.141 1.211 1.291 1.381 1.461 1.561 1.661 1.761 1.871 1.991 2.121 2.251 2.391	Displacement (ft) 1.762 1.741 1.726 1.7 1.68 1.659 1.639 1.617 1.592 1.571 1.545 1.521 1.497	Time (min) 8.791 9.321 9.881 10.48 11.08 11.78 12.48 13.18 13.98 14.88 15.68 16.68 17.68	Displacement (ft) 0.868 0.838 0.804 0.774 0.746 0.718 0.689 0.666 0.639 0.611 0.59 0.557 0.533
0.118	2.156	1.561	1.659	11.78	0.718
0.132	2.145	1.761	1.639	13.18	0.666
0.163	2.125	1.871	1.592	13.98	0.639
0.179	2.118	1.991	1.571	14.88	0.611
0.197	2.104	2.121	1.545	15.68	0.59
0.216	2.096	2.251	1.521	16.68	0.557
0.236	2.068	2.391	1.497	17.68	0.533
0.257	2.072	2.541	1.473	18.68	0.507
0.279	2.063	2.701	1.443	19.78	0.486
0.303	2.049	2.861	1.42	20.98	0.465
0.328	2.038	3.041	1.392	22.28	0.444
0.354	2.025	3.231	1.366	23.58	0.424
0.382	2.012	3.431	1.337	24.98	0.405
0.412	2.	3.641	1.31		
0.443	1.985	3.861	1.281		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	3.1E-5	cm/sec
Ss	0.0085	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.01609 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	3.152E-5	1.915E-6	+/- 3.801E-6	16.46	cm/sec
Ss	0.008559	0.0008401	+/- 0.001668	10.19	ft ⁻¹
Kz/Kr	1.	7.083	+/- 14.06	0.1412	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error

No estimation window

 $T = K^*b = 0.01636 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.99	-0.98
Ss	-0.99	1.00	0.96
Kz/Kr	-0.98	0.96	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.01115 ft ²
Variance	0.0001149 ft ²
Std. Deviation	0.01072 ft
Mean	-0.001744 ft
No. of Residuals	100
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G304fh.aqt Title: G304 @ Ash Pond 1 – Falling Head Test Date: 11/28/16 Time: 09:40:21

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 17 Jun 2016 Test Well: G304

AQUIFER DATA

Saturated Thickness: 18.81 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G304

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.75 ft Static Water Column Height: 18.81 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 10. ft Total Well Penetration Depth: 18.81 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	1.867	0.3132	1.121	2.307	0.144	
0.003917	1.786	0.3352	1.089	2.447	0.126	
0.008083	1.492	0.3592	1.065	2.597	0.108	
0.01225	1.685	0.3842	1.036	2.757	0.091	
0.01897	1.493	0.4102	1.004	2.917	0.081	
0.02103	1.755	0.4382	0.972	3.097	0.068	
0.02475	1.68	0.4682	0.947	3.287	0.053	
0.02892	1.376	0.4992	0.911	3.487	0.048	
0.03308	1.553	0.5332	0.882	3.697	0.041	
0.04002	1.691	0.5682	0.846	3.917	0.031	
0.04323	1.569	0.6052	0.814	4.157	0.026	
0.04927	1.594	0.6452	0.783	4.407	0.025	
0.05982	1.537	0.6872	0.746	4.667	0.018	

Time (min) 0.06323 0.07023 0.07825 0.08723 0.09523 0.1052 0.1152 0.1252 0.1362 0.1482 0.1612 0.1742 0.1882 0.2032 0.2192 0.2352 0.2532 0.2722	Displacement (ft) 1.549 1.518 1.5 1.481 1.465 1.46 1.408 1.407 1.39 1.362 1.344 1.3 1.297 1.272 1.249 1.227 1.2 1.174	Time (min) 0.7312 0.7782 0.8282 0.8812 0.9372 0.9972 1.057 1.127 1.197 1.267 1.347 1.437 1.517 1.617 1.717 1.817 1.927 2.047	Displacement (ft) 0.714 0.675 0.642 0.605 0.569 0.541 0.508 0.473 0.435 0.407 0.375 0.342 0.315 0.284 0.258 0.231 0.21 0.185	Time (min) 4.947 5.247 5.557 5.897 6.247 6.617 7.017 7.437 7.877 8.347 8.347 9.377 9.937 10.54 11.14 11.84 12.54 13.24	Displacement (ft) 0.017 0.012 0.012 0.011 0.007 0.007 0.009 0.006 0.001 0.002 0.004 0.002 0.003 0.001 0.001 0.001 0.002 0.002 0.002 0.002
0.2532 0.2722 0.2922	1.2 1.174 1.145	1.927 2.047 2.177	0.21 0.185 0.162	12.54 13.24	0.002 -0.002

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00089	cm/sec
Ss	7.0E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.5103 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G304rh.aqt Title: G304 @ Ash Pond 1 – Rising Head Test Date: 11/28/16 Time: 09:40:55

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 17 Jun 2016 Test Well: G304

AQUIFER DATA

Saturated Thickness: 18.81 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G304

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.85 ft Static Water Column Height: 18.81 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 10. ft Total Well Penetration Depth: 18.81 ft

	Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.	1.977	0.2563	1.372	1.948	0.257	
0.0041	1.901	0.2743	1.349	2.068	0.229	
0.008267	1.852	0.2933	1.324	2.198	0.203	
0.01243	1.83	0.3133	1.296	2.328	0.178	
0.01895	1.804	0.3343	1.273	2.468	0.157	
0.02102	1.8	0.3563	1.246	2.618	0.129	
0.02493	1.784	0.3803	1.217	2.778	0.111	
0.0291	1.769	0.4053	1.186	2.938	0.092	
0.03327	1.76	0.4313	1.158	3.118	0.077	
0.04	1.74	0.4593	1.124	3.308	0.067	
0.04207	1.733	0.4893	1.094	3.508	0.054	
0.04577	1.73	0.5203	1.06	3.718	0.042	
0.04993	1.718	0.5543	1.029	3.938	0.033	

Time (min)Displacement (ft)Time (min) 0.0541 1.711 0.5893 0.06107 1.695 0.6263 0.06427 1.688 0.6663 0.07027 1.676 0.7083 0.08085 1.653 0.7523 0.08427 1.648 0.7993 0.09127 1.632 0.8493 0.09967 1.619 0.9023 0.1083 1.604 0.9583 0.1163 1.573 1.018 0.1263 1.573 1.078 0.1363 1.554 1.148 0.1463 1.538 1.218 0.1693 1.501 1.368 0.1823 1.443 1.458 0.1953 1.459 1.538 0.2093 1.443 1.638 0.2093 1.443 1.638 0.2043 1.394 1.838	$\begin{array}{c} 0.997\\ 0.962\\ 0.924\\ 0.883\\ 0.849\\ 0.813\\ 0.773\\ 0.739\\ 0.698\\ 0.655\\ 0.619\\ 0.579\\ 0.536\\ 0.503\\ 0.466\\ 0.425\\ 0.391\\ 0.352\\ 0.32\\ 0.291\\ \end{array}$	$\begin{array}{c} 4.178\\ 4.178\\ 4.428\\ 4.688\\ 4.968\\ 5.268\\ 5.578\\ 5.918\\ 6.268\\ 6.638\\ 7.038\\ 7.458\\ 7.898\\ 8.368\\ 8.868\\ 9.398\\ 9.958\\ 10.56\\ 11.16\end{array}$	0.026 0.025 0.017 0.017 0.012 0.012 0.012 0.005 0.004 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.001 0.005 0.001 0.003 0.006
--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.001	cm/sec
Ss	1.0E-6	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.5733 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G401fh.aqt Title: G401 @ Ash Pond 2 – Falling Head Test Date: 11/28/16 Time: 10:44:18

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G401

AQUIFER DATA

Saturated Thickness: 4.92 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G401

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.6 ft Static Water Column Height: 4.92 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.63 ft Total Well Penetration Depth: 4.92 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	0.629	0.3217	0.157	3.127	0.111
0.00417	0.537	0.3427	0.157	3.317	0.106
0.00834	0.531	0.3647	0.162	3.517	0.11
0.0125	0.441	0.3887	0.161	3.727	0.104
0.01667	0.542	0.4137	0.154	3.947	0.098
0.02084	0.422	0.4397	0.151	4.187	0.095
0.025	0.44	0.4677	0.156	4.437	0.095
0.02917	0.459	0.4977	0.159	4.697	0.084
0.03334	0.375	0.5287	0.156	4.977	0.087
0.0375	0.391	0.5627	0.174	5.277	0.078
0.04167	0.329	0.5977	0.16	5.587	0.069
0.04584	0.366	0.6347	0.16	5.927	0.065
0.05	0.346	0.6747	0.155	6.277	0.057

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05417	0.441	0.7167	0.171	6.651	0.049
0.05834	0.495	0.7607	0.159	7.047	0.059
0.0625	-0.542	0.8077	0.167	7.467	0.055
0.06667	0.033	0.8577	0.153	7.907	0.052
0.07267	0.28	0.9107	0.19	8.377	0.046
0.07867	0.163	0.9667	0.154	8.877	0.039
0.08567	0.31	1.027	0.154	9.407	0.036
0.09267	0.278	1.087	0.15	9.967	0.033
0.09967	0.283	1.157	0.144	10.57	0.031
0.1077	0.179	1.227	0.153	11.17	0.034
0.1167	0.176	1.297	0.143	11.87	0.025
0.1247	0.144	1.377	0.146	12.57	0.021
0.1347	0.139	1.467	0.147	13.27	0.018
0.1447	0.146	1.547	0.144	14.07	0.021
0.1547	0.173	1.647	0.138	14.97	0.008
0.1657	0.153	1.747	0.138	15.77	0.015
0.1777	0.16	1.847	0.141	16.77	0.
0.1907	0.147	1.957	0.133	17.77	0.011
0.2037	0.132	2.077	0.13	18.77	0.007
0.2177	0.156	2.207	0.124	19.87	0.001
0.2327	0.183	2.337	0.132	21.07	0.023
0.2487	0.148	2.477	0.124	22.37	0.014
0.2647	0.14	2.627	0.117	23.67	0.012
0.2827	0.154	2.787	0.12		
0.3017	0.153	2.947	0.117		

Slug Test Aquifer Model: Confined Solution Method: Bouwer-Rice In(Re/rw): 3.118

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.00018	cm/sec
уO	0.18	ft

 $T = K^*b = 0.02699 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G401rh.aqt Title: G401 @ Ash Pond 2 – Rising Head Test Date: 11/28/16 Time: 10:45:18

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G401

AQUIFER DATA

Saturated Thickness: 4.92 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G401

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.6 ft Static Water Column Height: 4.92 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.63 ft Total Well Penetration Depth: 4.92 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.532	0.2808	0.333	2.056	0.18
0.004167	1.512	0.3008	0.33	2.187	0.173
0.008333	1.479	0.3218	0.32	2.316	0.173
0.0125	1.44	0.3438	0.316	2.456	0.169
0.01667	1.414	0.3678	0.309	2.606	0.162
0.02083	1.362	0.3928	0.302	2.766	0.158
0.025	1.322	0.4188	0.298	2.926	0.152
0.02917	1.28	0.4468	0.294	3.106	0.147
0.03333	1.237	0.4768	0.286	3.296	0.137
0.0375	1.191	0.5078	0.283	3.496	0.13
0.04167	1.182	0.5418	0.277	3.706	0.128
0.04583	1.161	0.5768	0.273	3.926	0.118
0.05183	1.085	0.6138	0.269	4.166	0.113

Time (min) 0.05783 0.06483 0.07183 0.07883 0.08683 0.09583 0.1038 0.1138 0.1238 0.1338 0.1448 0.1568 0.1698 0.1828 0.1968 0.1968 0.2118	Displacement (ft) 1.029 0.977 0.928 0.874 0.824 0.771 0.722 0.673 0.618 0.576 0.531 0.49 0.455 0.427 0.406 0.389	$\begin{array}{r} \underline{\text{Time (min)}}\\ 0.6538\\ 0.6958\\ 0.7398\\ 0.7868\\ 0.8868\\ 0.8898\\ 0.9458\\ 1.006\\ 1.066\\ 1.136\\ 1.206\\ 1.276\\ 1.356\\ 1.446\\ 1.526\\ 1.626\\ \end{array}$	Displacement (ft) 0.269 0.261 0.258 0.253 0.247 0.244 0.239 0.236 0.237 0.228 0.226 0.22 0.218 0.212 0.206 0.202	Time (min) 4.416 4.676 4.956 5.256 5.566 5.906 6.256 6.626 7.026 7.446 7.886 8.356 8.856 9.39 9.946 10.55	Displacement (ft) 0.108 0.102 0.098 0.09 0.082 0.078 0.072 0.066 0.059 0.054 0.047 0.042 0.033 0.025 0.02 0.015
0.1968	0.406	1.526	0.206	9.946	0.02
0.2118	0.389	1.626	0.202	10.55	0.015
0.2278	0.376	1.726	0.198	11.15	0.011
0.2438	0.357	1.826	0.19	11.85	0.004
0.2618	0.345	1.936	0.19	12.55	0.

Slug Test Aquifer Model: Confined Solution Method: Bouwer-Rice In(Re/rw): 3.118

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.00028	cm/sec
y0	0.3	ft

 $T = K^*b = 0.04199 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G402fh.aqt Title: G402 @ Ash Pond 2 – Falling Head Test Date: 11/28/16 Time: 09:41:35

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G402

AQUIFER DATA

Saturated Thickness: 13.25 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G402

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.12 ft Static Water Column Height: 13.25 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 10. ft Total Well Penetration Depth: 13.25 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.302	0.2327	0.776	1.747	0.246
0.004167	1.376	0.2487	0.766	1.847	0.233
0.008333	1.477	0.2647	0.749	1.957	0.217
0.0125	1.324	0.2827	0.741	2.077	0.205
0.01667	1.315	0.3017	0.731	2.207	0.189
0.02083	1.2	0.3217	0.708	2.337	0.175
0.025	1.121	0.3427	0.702	2.477	0.164
0.02917	1.109	0.3646	0.682	2.627	0.152
0.03333	1.065	0.3887	0.67	2.787	0.139
0.0375	1.044	0.4137	0.657	2.947	0.13
0.04167	1.022	0.4396	0.639	3.127	0.115
0.04583	1.005	0.4677	0.627	3.317	0.113
0.05	0.991	0.4976	0.611	3.517	0.105

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00045	cm/sec
Ss	0.002	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1817 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G402rh.aqt Title: G402 @ Ash Pond 2 – Rising Head Test Date: 11/28/16 Time: 09:41:57

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G402

AQUIFER DATA

Saturated Thickness: 13.25 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G402

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.5 ft Static Water Column Height: 13.25 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 10. ft Total Well Penetration Depth: 13.25 ft

No. of Observations: 95

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
-0.0001503	0.49	0.2658	0.412	1.94	0.225
0.004016	0.51	0.2848	0.412	2.06	0.212
0.009716	0.481	0.3049	0.406	2.19	0.206
0.01235	0.486	0.3258	0.403	2.32	0.2
0.01652	0.484	0.3478	0.399	2.46	0.19
0.02068	0.486	0.3719	0.397	2.61	0.184
0.02485	0.48	0.3968	0.391	2.77	0.175
0.02902	0.477	0.4228	0.383	2.93	0.163
0.03318	0.476	0.4508	0.382	3.11	0.162
0.03735	0.473	0.4808	0.377	3.3	0.15
0.04152	0.473	0.5119	0.373	3.5	0.145
0.04568	0.472	0.5458	0.364	3.71	0.136
0.04985	0.467	0.5808	0.363	3.93	0.126

1

Time (min) 0.05585 0.06185 0.06885 0.07585 0.09085 0.09985 0.1079 0.1179 0.1278 0.1379 0.1489 0.1489 0.1608 0.1739 0.1868 0.2009 0.2158	Displacement (ft) 0.467 0.463 0.462 0.459 0.456 0.454 0.451 0.45 0.45 0.45 0.442 0.443 0.443 0.438 0.439 0.434 0.42 0.422	Time (min) 0.6178 0.6579 0.6998 0.7438 0.7909 0.8408 0.8939 0.9498 1.01 1.07 1.157 1.21 1.28 1.36 1.45 1.53 1.63	Displacement (ft) 0.355 0.352 0.341 0.337 0.336 0.327 0.321 0.315 0.306 0.301 0.293 0.286 0.28 0.268 0.263 0.256 0.248	Time (min) 4.17 4.42 4.68 4.96 5.26 5.57 5.91 6.26 6.63 7.03 7.45 7.89 8.36 8.86 9.39 9.95 10.55	Displacement (ft) 0.12 0.111 0.107 0.104 0.098 0.085 0.081 0.075 0.068 0.067 0.062 0.055 0.054 0.051 0.048
0.2158 0.2319 0.2478	0.42 0.422 0.421 0.417	1.63 1.73 1.83	0.230 0.248 0.241 0.232	10.55 11.15	0.048 0.046

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00019	cm/sec
Ss	0.001	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.07673 \text{ cm}^2/\text{sec}$


Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G403fh.aqt Title: G403 @ Ash Pond 2 – Falling Head Test Date: 11/28/16 Time: 09:42:29

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G403

AQUIFER DATA

Saturated Thickness: 13.99 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G403

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.87 ft Static Water Column Height: 13.99 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.67 ft Total Well Penetration Depth: 13.99 ft

		Observ	vation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.798	0.832	2.493	5.792	1.841
0.01	2.842	0.892	2.478	6.142	1.798
0.02	2.804	0.952	2.462	6.512	1.762
0.031	2.799	1.022	2.451	6.912	1.726
0.043	2.8	1.092	2.436	7.332	1.683
0.056	2.784	1.162	2.422	7.772	1.648
0.069	2.78	1.242	2.41	8.242	1.604
0.083	2.77	1.332	2.389	8.742	1.564
0.098	2.764	1.412	2.378	9.272	1.525
0.114	2.768	1.512	2.36	9.832	1.481
0.13	2.745	1.612	2.343	10.43	1.436
0.148	2.739	1.712	2.328	11.03	1.391
0.167	2.726	1.822	2.31	11.73	1.346

Time (min) 0.187 0.208 0.23 0.254 0.279 0.305 0.333 0.363 0.363	Displacement (ft) 2.716 2.701 2.693 2.682 2.674 2.66 2.645 2.636 2.636	Time (min) 1.942 2.072 2.202 2.342 2.492 2.652 2.812 2.992 2.182	Displacement (ft) 2.29 2.272 2.255 2.236 2.223 2.194 2.169 2.146	Time (min) 12.43 13.13 13.93 14.83 15.63 16.63 17.63 18.63 19.72	Displacement (ft) 1.295 1.25 1.202 1.147 1.102 1.053 0.999 0.948 0.901
0.463 0.5 0.54 0.582 0.626 0.673 0.723 0.776	2.597 2.586 2.57 2.56 2.542 2.533 2.518 2.507	3.592 3.812 4.052 4.302 4.562 4.842 5.142 5.452	2.073 2.061 2.016 1.991 1.96 1.929 1.896 1.867	22.23 23.53 24.93 26.43 28.03 29.63	0.794 0.741 0.685 0.634 0.576 0.526

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.3E-5	cm/sec
Ss	0.0075	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.01834 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G403rh.aqt Title: G403 @ Ash Pond 2 – Rising Head Test Date: 11/28/16 Time: 09:42:50

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G403

AQUIFER DATA

Saturated Thickness: 13.99 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G403

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.18 ft Static Water Column Height: 13.99 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.67 ft Total Well Penetration Depth: 13.99 ft

		Observ	vation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	2.159	0.3607	2.032	3.289	1.596
0.001117	2.168	0.3857	2.026	3.489	1.572
0.004817	2.16	0.4117	2.02	3.699	1.55
0.008684	2.157	0.4397	2.018	3.919	1.527
0.0126	2.154	0.4697	2.009	4.159	1.508
0.01625	2.153	0.5007	1.998	4.409	1.483
0.0199	2.153	0.5347	1.992	4.669	1.46
0.02357	2.148	0.5697	1.984	4.949	1.43
0.02722	2.144	0.6067	1.975	5.249	1.401
0.03087	2.143	0.6467	1.969	5.559	1.376
0.03452	2.142	0.6887	1.961	5.899	1.342
0.03867	2.14	0.7327	1.952	6.249	1.311
0.04467	2.136	0.7797	1.942	6.619	1.285

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05067	2.135	0.8297	1.932	7.019	1.255
0.05767	2.132	0.8827	1.929	7.439	1.223
0.06467	2.128	0.9387	1.917	7.879	1.194
0.07167	2.13	0.9987	1.904	8.349	1.156
0.07967	2.123	1.059	1.893	8.849	1.128
0.08867	2.114	1.129	1.881	9.379	1.094
0.09667	2.116	1.199	1.869	9.939	1.056
0.1067	2.11	1.269	1.861	10.54	1.023
0.1167	2.107	1.349	1.847	11.14	0.988
0.1267	2.107	1.439	1.834	11.84	0.955
0.1377	2.1	1.519	1.82	12.54	0.915
0.1497	2.095	1.619	1.806	13.24	0.876
0.1627	2.094	1.719	1.787	14.04	0.837
0.1757	2.084	1.819	1.779	14.94	0.796
0.1897	2.079	1.931	1.752	15.74	0.761
0.2047	2.075	2.049	1.747	16.74	0.719
0.2207	2.074	2.179	1.728	17.74	0.684
0.2367	2.066	2.309	1.709	18.74	0.647
0.2547	2.064	2.449	1.697	19.84	0.608
0.2737	2.056	2.599	1.67	21.04	0.568
0.2937	2.049	2.759	1.653	22.34	0.528
0.3147	2.045	2.919	1.639	23.64	0.477
0.3367	2.04	3.099	1.61	25.04	0.45

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	7.2E-5	cm/sec
Ss	0.0015	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.0307 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter Kr	Estimate	Std. Error	Approx. C.I. + $\frac{1007E_{5}}{5}$	t-Ratio	cm/sec
Ss Kz/Kr	0.001636	0.0003018 1.686	+/- 0.0005986 +/- 3.343	5.42 0.5931	ft ⁻¹

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.03066 \text{ cm}^2/\text{sec}$

Parameter Correlations

Kr	Ss	Kz/Kr
1.00	-0.99	-0.98
-0.99	1.00	0.94
-0.98	0.94	1.00
	<u>Kr</u> 1.00 -0.99 -0.98	<u>Kr</u> <u>Ss</u> 1.00 -0.99 -0.99 1.00 -0.98 0.94

Residual Statistics

for weighted residuals

Sum of Squares	0.01972 ft ²
	0.0001878 ft ²
Std. Deviation	0.01371 ft
Mean	-0.004 ft
No. of Residuals	108
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G404fh.aqt Title: G404 @ Ash Pond 2 – Falling Head Test Date: 11/28/16 Time: 09:43:12

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G404

AQUIFER DATA

Saturated Thickness: 8.95 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G404

X Location: 1. ft Y Location: 0. ft

Initial Displacement: 3.17 ft Static Water Column Height: 8.95 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.75 ft Total Well Penetration Depth: 8.95 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.136	0.4477	2.579	3.467	1.094
0.007967	3.119	0.4787	2.552	3.677	1.021
0.01163	3.099	0.5127	2.52	3.897	0.976
0.0153	3.1	0.5476	2.488	4.137	0.909
0.01895	3.103	0.5847	2.453	4.387	0.862
0.02685	3.088	0.6247	2.418	4.647	0.811
0.03055	3.087	0.6667	2.382	4.927	0.762
0.03567	3.072	0.7107	2.337	5.227	0.715
0.04267	3.064	0.7577	2.311	5.537	0.667
0.04967	3.047	0.8077	2.268	5.877	0.62
0.05767	3.041	0.8607	2.226	6.227	0.579
0.06667	3.022	0.9166	2.187	6.597	0.538
0.07467	3.01	0.9767	2.142	6.997	0.498

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.08467	2.998	1.037	2.101	7.417	0.456
0.09467	2.985	1.107	2.051	7.857	0.421
0.1047	2.973	1.177	2.008	8.327	0.384
0.1157	2.958	1.247	1.962	8.827	0.342
0.1277	2.941	1.327	1.914	9.357	0.314
0.1407	2.922	1.417	1.864	9.917	0.281
0.1537	2.909	1.497	1.818	10.52	0.24
0.1677	2.889	1.597	1.765	11.12	0.196
0.1827	2.872	1.697	1.712	11.82	0.157
0.1986	2.865	1.797	1.664	12.52	0.13
0.2147	2.834	1.907	1.61	13.22	0.112
0.2327	2.814	2.027	1.558	14.02	0.093
0.2517	2.785	2.157	1.501	14.93	0.081
0.2717	2.765	2.287	1.452	15.72	0.068
0.2927	2.754	2.427	1.397	16.72	0.06
0.2717	2.765	2.287	1.452	15.72	0.068
0.2927	2.754	2.427	1.397	16.72	0.06
0.3147	2.716	2.577	1.34	17.72	0.047
0.3387	2.695	2.737	1.285	18.72	0.039
0.3637	2.666	2.897	1.232	19.82	0.04
0.3897	2.639	3.077	1.176	21.02	0.031
0.4177	2.61	3.267	1.123	22.32	0.025

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00042	cm/sec
Ss	0.00033	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1146 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G404rh.aqt Title: G404 @ Ash Pond 2 – Rising Head Test Date: 11/28/16 Time: 09:43:34

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G404

AQUIFER DATA

Saturated Thickness: 8.95 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G404

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.1 ft Static Water Column Height: 8.95 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.75 ft Total Well Penetration Depth: 8.95 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.022	0.2695	2.663	2.648	1.302
0.004167	3.042	0.2855	2.646	2.807	1.252
0.008333	3.052	0.3035	2.624	2.967	1.199
0.0125	3.03	0.3225	2.608	3.147	1.147
0.01667	3.038	0.3425	2.588	3.337	1.092
0.02083	3.027	0.3635	2.57	3.537	1.038
0.025	3.002	0.3855	2.545	3.747	0.989
0.02917	3.002	0.4095	2.526	3.967	0.939
0.03333	2.991	0.4345	2.5	4.207	0.885
0.0375	2.992	0.4605	2.476	4.457	0.837
0.04167	2.973	0.4885	2.453	4.717	0.788
0.04583	2.965	0.5185	2.431	4.997	0.742
0.05	2.953	0.5495	2.404	5.297	0.692

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.05417	2.949	0.5835	2.374	5.607	0.649
0.05833	2.942	0.6185	2.345	5.947	0.602
0.0625	2.93	0.6555	2.315	6.297	0.556
0.06667	2.926	0.6955	2.287	6.667	0.517
0.07083	2.918	0.7375	2.257	7.067	0.48
0.075	2.906	0.7815	2.226	7.487	0.44
0.07917	2.902	0.8285	2.187	7.927	0.406
0.08333	2.894	0.8785	2.155	8.397	0.369
0.0875	2.887	0.9315	2.119	8.897	0.336
0.0935	2.883	0.9875	2.083	9.428	0.304
0.0995	2.873	1.047	2.044	9.987	0.275
0.1075	2.854	1.107	2.007	10.59	0.251
0.1135	2.851	1.178	1.958	11.19	0.223
0.1205	2.832	1.247	1.924	11.89	0.204
0.1286	2.824	1.317	1.882	12.59	0.184
0.1375	2.818	1.397	1.839	13.29	0.163
0.1455	2.807	1.488	1.787	14.09	0.145
0.1555	2.793	1.567	1.751	14.99	0.125
0.1655	2.781	1.668	1.698	15.79	0.115
0.1755	2.771	1.767	1.653	16.79	0.097
0.1865	2.76	1.867	1.603	17.79	0.087
0.1985	2.745	1.978	1.556	18.79	0.078
0.2115	2.728	2.097	1.509	19.89	0.067
0.2245	2.713	2.227	1.457	21.09	0.061
0.2385	2.695	2.357	1.408	22.39	0.044
0.2535	2.68	2.498	1.357		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter Kr	Estimate	cm/sec
Ss	0.0007	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1037 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G405fh2.aqt Title: G405 @ Ash Pond 2 – Falling Head Test Date: 11/28/16 Time: 09:43:55

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G405

AQUIFER DATA

Saturated Thickness: 11.12 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G405

X Location: 1. ft Y Location: 0. ft

Initial Displacement: 0.114 ft Static Water Column Height: 11.12 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.75 ft Total Well Penetration Depth: 11.12 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.0223	0.1	1.799	0.031	10.84	0.001
0.0335	0.127	1.904	0.03	11.01	0.001
0.0447	0.107	2.016	0.028	11.17	0.001
0.0558	0.106	2.134	0.026	11.34	0.001
0.067	0.104	2.259	0.024	11.51	0.001
0.0782	0.095	2.392	0.023	11.67	0.001
0.0893	0.087	2.532	0.021	11.84	0.001
0.1005	0.106	2.681	0.019	12.01	0.001
0.1117	0.087	2.838	0.018	12.17	0.001
0.1228	0.12	3.005	0.016	12.34	0.001
0.134	0.099	3.172	0.015	12.51	0.001
0.1452	0.1	3.338	0.014	12.67	0.001
0.1563	0.096	3.505	0.013	12.84	0.001

Timo (min)	Displacement (ft)	Timo (min)	Displacement (ft)	Timo (min)	Displacement (ft)
$\frac{11110}{0.1675}$		3 672	$\frac{\text{Displacement (it)}}{0.012}$	<u>13 01</u>	
0.1075	0.075	2 9 2 9	0.012	12 17	0.001
0.1707	0.074	4.005	0.01	12.17	0.001
0.1090	0.093	4.005	0.01	13.34	0.001
0.201	0.091	4.172	0.01	13.31	0.001
0.2122	0.091	4.330	0.009	12.07	0.001
0.2233	0.09	4.303	0.008	13.04	0.001
0.235	0.089	4.0/2	0.008	14.01	0.001
0.2473	0.000	4.030 E.00E	0.007	14.17	0.001
0.2007	0.088	5.005	0.007	14.34	0.001
0.2747	0.086	5.1/2	0.006	14.51	0.001
0.2895	0.085	5.338	0.006	14.07	0.001
0.3052	0.084	5.505	0.006	14.84	0.001
0.3218	0.083	5.672	0.005	15.01	0.001
0.3395	0.082	5.838	0.005	15.17	0.001
0.3582	0.081	6.005	0.005	15.34	0.001
0.378	0.08	6.172	0.004	15.51	0.001
0.399	0.078	6.338	0.004	15.67	0.001
0.4212	0.077	6.505	0.004	15.84	0.001
0.4447	0.076	6.672	0.004	16.	0.001
0.4695	0.074	6.838	0.004	16.17	0.001
0.4958	0.074	7.005	0.003	16.34	0.001
0.5238	0.071	7.172	0.003	16.5	0.
0.5535	0.07	7.338	0.003	16.67	0.
0.5848	0.068	7.505	0.003	16.84	0.
0.618	0.066	7.672	0.003	17.	0.
0.6532	0.065	7.838	0.003	17.17	0.
0.6905	0.063	8.005	0.003	17.34	0.
0.73	0.061	8.172	0.002	17.5	0.
0.7718	0.06	8.338	0.002	17.67	0.
0.8162	0.058	8.505	0.002	17.84	0.
0.8632	0.056	8.672	0.002	18.	0.
0.913	0.054	8.838	0.002	18.17	0.
0.9657	0.052	9.005	0.002	18.34	0.
1.022	0.05	9.172	0.002	18.5	0.
1.081	0.048	9.338	0.002	18.67	0.
1.143	0.047	9.505	0.002	18.84	0.
1.21	0.045	9.672	0.002	19.	0.
1.28	0.043	9.838	0.002	19.17	0.
1.355	0.041	10.01	0.002	19.34	0.
1.434	0.039	10.17	0.002	19.5	0.
1.517	0.037	10.34	0.002	19.67	0.
1 606	0.035	10 51	0.001		0.
1.7	0.033	10.67	0.001		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00098	cm/sec
Ss	0.00037	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.3322 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0009828	0.0002457	+/- 0.0004853	4.	cm/sec
Ss	0.0003762	0.0004077	+/- 0.0008052	0.9227	ft ⁻¹
Kz/Kr	1.	2.807	+/- 5.544	0.3562	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.3331 \text{ cm}^2/\text{sec}$

Parameter Correlations

Kr	Ss	Kz/Kr
1.00	-0.99	-0.99
-0.99	1.00	0.97
-0.99	0.97	1.00
	<u>Kr</u> 1.00 -0.99 -0.99	<u>Kr</u> <u>Ss</u> 1.00 -0.99 -0.99 1.00 -0.99 0.97

Residual Statistics

for weighted residuals

Sum of Squares	0.001412 ft ²
	8.661E-6 ft ²
Std. Deviation	0.002943 ft
Mean	-3.208E-5 ft
No. of Residuals	166
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\G405rh2.aqt Title: G405 @ Ash Pond 2 – Rising Head Test Date: 11/28/16 Time: 09:44:19

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 13 Jun 2016 Test Well: G405

AQUIFER DATA

Saturated Thickness: 11.12 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G405

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.125 ft Static Water Column Height: 11.12 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.75 ft Total Well Penetration Depth: 11.12 ft

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.066	0.112	1.441	0.039	9.013	0.002
0.077	0.11	1.525	0.037	9.179	0.002
0.088	0.108	1.613	0.035	9.346	0.002
0.099	0.107	1.707	0.033	9.513	0.002
0.11	0.105	1.807	0.031	9.679	0.002
0.121	0.104	1.912	0.029	9.846	0.002
0.132	0.102	2.023	0.027	10.01	0.002
0.143	0.101	2.142	0.026	10.18	0.002
0.154	0.099	2.267	0.024	10.35	0.002
0.165	0.098	2.399	0.022	10.51	0.001
0.176	0.097	2.54	0.021	10.68	0.001
0.187	0.096	2.688	0.019	10.85	0.001
0.198	0.095	2.846	0.018	11.01	0.001

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	0.094	3.013	0.016	11.18	0.001
0.22	0.093	3.179	0.015	11.35	0.001
0.231	0.092	3.346	0.014	11.51	0.001
0.2427	0.091	3.513	0.013	11.68	0.001
0.2552	0.09	3.679	0.012	11.85	0.001
0.2683	0.089	3.846	0.011	12.01	0.001
0.2823	0.088	4.013	0.01	12.18	0.001
0.2972	0.087	4.179	0.01	12.35	0.001
0.3128	0.086	4.346	0.009	12.51	0.001
0.3295	0.085	4.513	0.008	12.68	0.001
0.3472	0.083	4.679	0.008	12.85	0.001
0.3658	0.082	4.846	0.007	13.01	0.001
0.3857	0.081	5.013	0.007	13.18	0.001
0.4067	0.08	5.179	0.006	13.35	0.001
0.4288	0.078	5.346	0.006	13.51	0.001
0.4523	0.077	5.513	0.006	13.68	0.001
0.4772	0.075	5.679	0.005	13.85	0.001
0.5035	0.075	5.846	0.005	14.01	0.001
0.5315	0.072	6.013	0.005	14.18	0.001
0.5612	0.071	6.179	0.004	14.35	0.001
0.5925	0.069	6.346	0.004	14.51	0.001
0.6257	0.067	6.513	0.004	14.68	0.001
0.6608	0.066	6.679	0.004	14.85	0.001
0.6982	0.064	6.846	0.003	15.01	0.001
0.7377	0.062	7.013	0.003	15.18	0.001
0.7795	0.06	7.179	0.003	15.35	0.001
0.8238	0.058	7.346	0.003	15.51	0.001
0.8708	0.056	7.513	0.003	15.68	0.
0.9207	0.054	7.679	0.003	15.85	0.
0.9733	0.052	7.846	0.003	16.01	0.
1.029	0.051	8.013	0.003	16.18	0.
1.088	0.049	8.179	0.002	16.35	0.
1.151	0.047	8.346	0.002	16.51	0.
1.217	0.045	8.513	0.002	16.68	0.
1.288	0.043	8.679	0.002	16.85	0.
1.362	0.041	8.846	0.002	17.01	0.

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00097	cm/sec
Ss	0.0012	ft ⁻¹

AQTESOLV for Windows

Kz/Kr

$T = K^*b = 0.3288 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0009567	3.216E-5	+/- 6.357E-5	29.75	cm/sec
Ss	0.001278	0.0001631	+/- 0.0003224	7.835	ft ⁻¹
Kz/Kr	1.	0.4379	+/- 0.8657	2.284	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

1.

 $T = K^*b = 0.3243 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.98	-0.96
Ss	-0.98	1.00	0.92
Kz/Kr	-0.96	0.92	1.00

Residual Statistics

for weighted residuals

Sum of Squares	9.007E-5 ft ²
Variance	6.255E-7 ft ²
Std. Deviation	0.0007909 ft
Mean	-0.0003511 ft
No. of Residuals	147
No. of Estimates	3



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\R104fh.aqt Title: R104 @ CCR Landfill – Falling Head Test Date: 11/28/16 Time: 10:46:31

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: R104

AQUIFER DATA

Saturated Thickness: 14.17 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: R104

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.96 ft Static Water Column Height: 14.17 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.43 ft Total Well Penetration Depth: 14.17 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
-0.03432	1.004	0.405	0.783	3.034	0.502
-0.03015	0.97	0.436	0.777	3.224	0.495
-0.02598	0.932	0.47	0.763	3.424	0.484
-0.01998	0.905	0.505	0.76	3.634	0.478
-0.01398	0.897	0.542	0.751	3.854	0.472
-0.006983	0.913	0.582	0.743	4.094	0.457
0.	0.909	0.624	0.736	4.344	0.452
0.007	0.907	0.668	0.723	4.604	0.445
0.01902	0.907	0.715	0.716	4.884	0.434
0.024	0.907	0.765	0.704	5.184	0.428
0.032	0.932	0.818	0.699	5.494	0.415
0.04202	0.901	0.874	0.686	5.834	0.403
0.05202	0.906	0.934	0.68	6.184	0.395

Time (min) 0.062	Displacement (ft) 0.893	<u>Time (min)</u> 0.994	Displacement (ft) 0.664	<u>Time (min)</u> 6.554	Displacement (ft) 0.382
0.07762	0.889	1.064 1.13 <i>1</i>	0.626	0.954 7.37 <i>1</i>	0.372
0.09885	0.870	1.204	0.634	7.814	0.349
0.111	0.868	1.284	0.628	8.284	0.335
0.125	0.862	1.374	0.619	8.784	0.316
0.14	0.861	1.454	0.604	9.314	0.303
0.156	0.854	1.554	0.596	9.874	0.284
0.172	0.849	1.654	0.587	10.47	0.271
0.19	0.844	1.754	0.579	11.07	0.263
0.209	0.838	1.864	0.572	11.77	0.236
0.229	0.833	1.984	0.563	12.47	0.216
0.25	0.824	2.114	0.553	13.17	0.195
0.2722	0.816	2.244	0.545	13.97	0.176
0.296	0.812	2.384	0.533	14.87	0.15
0.321	0.805	2.534	0.528	15.67	0.134
0.347	0.798	2.694	0.52		
0.375	0.788	2.854	0.511		

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	7.0E-5	cm/sec
Ss	0.045	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.03023 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	6.961E-5	2.932E-5	+/- 5.826E-5	2.374	cm/sec
Ss	0.04481	0.0257	+/- 0.05106	1.744	ft ⁻¹
Kz/Kr	1.	31.19	+/- 61.98	0.03206	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.03006 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-1.00	-0.99
Ss	-1.00	1.00	0.97
Kz/Kr	-0.99	0.97	1.00

Residual Statistics

for weighted residuals

Sum of Squares 0.04042 ft ²	~
Variance	t۷
Std. Deviation 0.02143 ft	
Mean0.002289 ft	i
No. of Residuals 91	
No. of Estimates 3	



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\R104rh.aqt Title: R104 @CCR Landfill – Rising Head Test Date: 12/01/16 Time: 09:05:00

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 15 Jun 2016 Test Well: R104

AQUIFER DATA

Saturated Thickness: 14.17 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: R104

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.23 ft Static Water Column Height: 14.17 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.43 ft Total Well Penetration Depth: 14.17 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	0.239	0.2745	0.18	1.929	0.133
0.00365	0.257	0.2945	0.179	2.049	0.131
0.007317	0.246	0.3155	0.173	2.179	0.13
0.01098	0.237	0.3375	0.174	2.309	0.128
0.01465	0.231	0.3615	0.174	2.449	0.126
0.01863	0.229	0.3865	0.175	2.599	0.125
0.0228	0.227	0.4125	0.172	2.759	0.123
0.02697	0.229	0.4405	0.169	2.919	0.122
0.03113	0.218	0.4705	0.167	3.099	0.119
0.0353	0.22	0.5015	0.168	3.289	0.12
0.03947	0.215	0.5355	0.171	3.489	0.118
0.04867	0.21	0.5705	0.161	3.699	0.114
0.05237	0.208	0.6075	0.165	3.919	0.11

<u>Time (min)</u> 0.05847	Displacement (ft) 0.216	Time (min) 0.6475	Displacement (ft) 0.164	Time (min) 4.159	Displacement (ft) 0.11
0.06835	0.202	0.6895	0.163	4.409	0.11
0.07247	0.211	0.7335	0.16	4.669	0.104
0.08047	0.216	0.7805	0.156	4.949	0.101
0.08947	0.194	0.8305	0.156	5.249	0.1
0.09748	0.206	0.8835	0.155	5.559	0.1
0.108	0.194	0.9395	0.153	5.899	0.096
0.1175	0.199	0.9995	0.15	6.249	0.093
0.1275	0.192	1.059	0.152	6.619	0.089
0.1385	0.197	1.129	0.146	7.019	0.086
0.1505	0.192	1.199	0.147	7.439	0.085
0.1635	0.191	1.269	0.142	7.879	0.076
0.1765	0.189	1.349	0.144	8.349	0.073
0.1905	0.187	1.439	0.138	8.849	0.072
0.2055	0.185	1.519	0.141	9.379	0.066
0.2215	0.184	1.619	0.139	9.939	0.059
0.2375	0.182	1.719	0.138	10.54	0.06
0.2555	0.179	1.819	0.137	11.14	0.051

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice In(Re/rw): 3.761

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.00028	cm/sec
у0	0.185	ft

 $T = K^*b = 0.1209 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\T127fh.aqt Title: T127 @ CCR Landfill – Falling Head Test Date: 11/28/16 Time: 09:44:57

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: T127

AQUIFER DATA

Saturated Thickness: 10.28 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T127

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 2.85 ft Static Water Column Height: 10.28 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 10.28 ft

		Observ	vation Data		
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
0.	4.12	27.72	1.734	183.6	0.295
0.36	3.789	29.76	1.684	195.	0.268
0.721	3.285	31.86	1.634	207.	0.242
1.141	2.277	34.08	1.561	219.6	0.222
1.56	2.588	36.48	1.527	232.8	0.196
1.98	3.079	39.	1.471	247.2	0.179
2.461	2.796	41.64	1.418	262.2	0.157
3.	2.517	44.46	1.362	278.	0.147
3.48	2.611	47.46	1.307	294.6	0.129
4.08	2.556	50.64	1.251	312.6	0.117
4.681	2.528	54.	1.194	331.2	0.102
5.28	2.507	57.6	1.141	351.6	0.088
5.94	2.473	61.2	1.086	372.6	0.08

<u>Time (sec)</u> 6.66 7.44 8.22 9.06 9.96 10.92 11.88 12.96 14.1 15.3 16.56 17.88 19.32 20.82 22.38	Displacement (ft) 2.442 2.404 2.379 2.327 2.308 2.267 2.214 2.188 2.151 2.093 2.066 2.02 1.973 1.928 1.878 4.804	Time (sec) 65.4 69.6 73.8 78.6 84. 88.8 94.8 100.8 106.8 113.4 120.6 128.4 136.2 144.6 153.6	Displacement (ft) 1.03 0.975 0.924 0.871 0.815 0.773 0.717 0.668 0.628 0.588 0.541 0.501 0.461 0.422 0.39 0.555	Time (sec) 394.8 418.8 444. 470.4 498.6 528.6 560.4 594. 630. 666. 708. 750. 792. 840. 894.2	Displacement (ft) 0.069 0.065 0.056 0.053 0.042 0.037 0.036 0.031 0.024 0.022 0.017 0.013 0.015 0.01 0.007
22.38 24.06	1.878 1.834	153.6 163.2	0.39 0.356	894.2 942.	0.007
25.86	1.786	172.8	0.326	, i z .	0.007

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.0012	cm/sec
Ss	0.0005	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.376 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\T127rh.aqt Title: T127 @ CCR Landfill – Rising Head Test Date: 11/28/16 Time: 09:45:21

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 14 Jun 2016 Test Well: T127

AQUIFER DATA

Saturated Thickness: 10.28 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T127

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 3.15 ft Static Water Column Height: 10.28 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.54 ft Total Well Penetration Depth: 10.28 ft

No. of Observations: 105

		Observ	ation Data		
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	3.236	19.56	2.077	166.2	0.4
0.25	3.096	20.88	2.035	175.8	0.365
0.641	3.006	22.32	1.994	186.6	0.333
0.764	2.978	23.82	1.948	198.	0.304
1.	2.958	25.38	1.91	210.	0.278
1.25	2.928	27.06	1.859	222.6	0.253
1.5	2.907	28.86	1.813	235.8	0.227
1.903	2.874	30.72	1.771	250.2	0.206
2.027	2.868	32.76	1.723	265.2	0.188
2.25	2.826	34.86	1.672	280.8	0.168
2.5	2.812	37.08	1.625	297.6	0.152
2.75	2.793	39.48	1.576	315.6	0.14
3.	2.775	42.	1.525	334.2	0.121

1

<u>Time (min)</u>	Displacement (ft)	<u>Time (min)</u>	Displacement (ft)	Time (min)	Displacement (ft)
3.36	2.758	44.64	1.475	354.8	0.113
3.72	2.73	47.46	1.42	375.6	0.105
4.14	2.705	50.46	1.369	397.8	0.097
4.56	2.683	53.64	1.311	421.8	0.087
4.98	2.644	57.	1.258	447.	0.083
5.46	2.629	60.6	1.208	473.4	0.075
6.	2.6	64.2	1.154	501.6	0.066
6.48	2.577	68.4	1.095	531.6	0.065
7.08	2.549	72.6	1.043	563.4	0.062
7.68	2.518	76.8	0.992	597.	0.058
8.28	2.493	81.6	0.938	633.	0.052
8.94	2.457	87.	0.889	669.	0.052
9.66	2.427	91.8	0.839	711.	0.045
10.44	2.399	97.8	0.781	753.	0.044
11.22	2.368	103.8	0.73	795.	0.041
12.06	2.334	109.8	0.689	843.	0.038
12.96	2.298	116.4	0.641	897.	0.035
13.92	2.266	123.6	0.596	945.	0.035
14.88	2.23	131.4	0.552	1005.	0.032
15.96	2.195	139.2	0.51	1065.	0.028
17.1	2.156	147.6	0.471	1125.	0.027
18.3	2.115	156.6	0.434	1191.	0.021

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	1.7E-5	cm/sec
Ss	0.0025	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.005327 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\T202fh.aqt Title: T202 @ GMF Gypsum Pond – Falling Head Test Date: 11/28/16 Time: 09:45:43

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: T202

AQUIFER DATA

Saturated Thickness: 13.29 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T202

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.8 ft Static Water Column Height: 13.29 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.38 ft Total Well Penetration Depth: 13.29 ft

		Observ	ation Data		
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
0.	1.742	34.92	1.295	231.2	0.445
0.42	1.694	37.44	1.269	245.6	0.417
0.9	1.722	40.08	1.245	260.6	0.393
1.44	1.74	42.9	1.218	276.2	0.368
1.92	1.717	45.9	1.196	293.	0.347
2.52	1.716	49.08	1.168	311.	0.322
3.12	1.705	52.44	1.143	329.6	0.301
3.72	1.693	56.04	1.114	350.	0.278
4.38	1.68	59.64	1.089	371.	0.262
5.1	1.667	63.84	1.056	393.2	0.245
5.88	1.645	68.04	1.032	417.2	0.223
6.66	1.645	72.24	1.006	442.4	0.21
7.5	1.625	77.04	0.975	468.8	0.197
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
---------------	-------------------	---------------	-------------------	------------	-------------------
8.4	1.608	82.44	0.946	497.	0.18
9.36	1.602	87.24	0.92	527.	0.169
10.32	1.586	93.24	0.888	558.8	0.156
11.4	1.566	99.24	0.851	592.4	0.145
12.54	1.552	105.2	0.83	628.4	0.137
13.74	1.537	111.8	0.8	664.4	0.126
15.	1.518	119.	0.769	706.4	0.118
16.32	1.5	126.8	0.74	748.4	0.107
17.76	1.485	134.6	0.706	790.4	0.104
20.82	1.445	152.	0.644	892.4	0.097
22.5	1.424	161.6	0.617	940.6	0.088
24.3	1.401	171.2	0.59	1000.4	0.079
26.16	1.385	182.	0.56	1060.4	0.077
28.2	1.359	193.4	0.53	1120.4	0.07
30.3 32.52	1.336 1.313	205.4 218.	0.5 0.472	1186.4	0.069

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00045	cm/sec
Ss	0.002	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.1823 \text{ cm}^2/\text{sec}$



Data Set: C:\Users\ktheesfeld\AppData\Local\Temp\Temp1_CoffeenAqtesolvFiles.zip\T202rh.aqt Title: T202 @ GMF Gypsum Pond – Rising Head Test Date: 11/28/16 Time: 09:46:07

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 16 Jun 2016 Test Well: T202

AQUIFER DATA

Saturated Thickness: 13.29 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T202

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.9 ft Static Water Column Height: 13.29 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.3333 ft Screen Length: 4.38 ft Total Well Penetration Depth: 13.29 ft

		Observ	ation Data		
Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
0.	1.832	19.56	1.473	166.2	0.554
0.25	1.855	20.88	1.454	175.8	0.528
0.642	1.857	22.32	1.438	186.6	0.494
0.765	1.853	23.82	1.417	198.	0.466
1.	1.838	25.38	1.397	210.	0.438
1.25	1.83	27.06	1.378	222.6	0.406
1.5	1.821	28.86	1.352	235.8	0.385
1.903	1.805	30.72	1.33	250.2	0.359
2.027	1.796	32.76	1.308	265.2	0.327
2.25	1.797	34.86	1.288	280.8	0.306
2.5	1.788	37.08	1.265	297.6	0.283
2.75	1.783	39.48	1.24	315.6	0.259
3.167	1.768	42.	1.214	334.2	0.243

Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)	Time (sec)	Displacement (ft)
3.36	1.765	44.64	1.196	354.6	0.216
3.721	1.755	47.46	1.172	375.6	0.2
4.338	1.738	50.46	1.144	397.8	0.18
4.56	1.736	53.64	1.115	421.8	0.168
4.98	1.728	57.	1.089	447.	0.146
5 46	1 714	60.6	1 065	473.4	0 137
6	1 705	64 2	1 041	501.6	0 123
6.48	1 692	68.4	1 008	531.6	0.123
7 08	1 681	72.6	0.98	563 1	0.107
7.00	1.001	72.0	0.70	503.4	0.101
7.00	1.000	70.0 01.4	0.949	577. 422	0.093
8.28	1.004	81.0	0.918	033.	0.077
8.94	1.643	87.	0.886	669.	0.067
9.66	1.628	91.8	0.861	711.	0.061
10.44	1.615	97.8	0.832	753.	0.054
11.22	1.602	103.8	0.799	795.	0.051
12.06	1.588	109.8	0.773	843	0.046
12.96	1 574	116.4	0 741	897	0.037
13 92	1 556	123.6	0 707	945	0.037
14.00	1.550	123.0	0.707	1005	0.037
14.00	1.000	131.4	0.074	1005.	0.031
15.96	1.525	139.2	0.647	1065.	0.026
17.1	1.508	147.6	0.621	1125.1	0.023
18.3	1.49	156.6	0.583	1191.	0.025

Slug Test Aquifer Model: Unconfined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	0.00055	cm/sec
Ss	0.002	ft ⁻¹
Kz/Kr	1.	

 $T = K^*b = 0.2228 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	0.0005307	2.017E-5	+/- 4.0E-5	26.31	cm/sec
Ss	0.002349	0.0002737	+/- 0.0005428	8.581	ft ⁻¹
Kz/Kr	1.	0.6606	+/- 1.31	1.514	

C.I. is approximate 95% confidence interval for parameter

t-ratio = estimate/std. error No estimation window

 $T = K^*b = 0.215 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss	Kz/Kr
Kr	1.00	-0.98	-0.95
Ss	-0.98	1.00	0.89
Kz/Kr	-0.95	0.89	1.00

Residual Statistics

for weighted residuals

Sum of Squares	0.01824 ft ²
Variance	0.0001788 ft ²
Std. Deviation	0.01337 ft
Mean	-0.005597 ft
No. of Residuals	105
No. of Estimates	3



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\G45D_fh1.aqt Title: G45D Falling Head #1 Date: 09/27/16 Time: 14:28:20

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 29 Aug 2016 Test Well: G45D

AQUIFER DATA

Saturated Thickness: 5.64 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G45D

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.2 ft Static Water Column Height: 5.64 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 9.64 ft Total Well Penetration Depth: 9.64 ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
0.33	1.18	36.	0.8		
0.5	1.16	37.13	0.78		
0.67	1.14	38.27	0.78		
0.83	1.13	39.2	0.78		
1.	1.12	40.02	0.77		
1.17	1.1	41.	0.77		
1.33	1.1	42.	0.77		
1.5	1.1	43.37	0.76		
1.67	1.1	44.2	0.76		
1.83	1.1	45.	0.76		
2.	1.1	46.	0.76		
2.5	1.09	47.	0.75		
3.	1.09	48.	0.75		
3.5	1.07	49.	0.75		
4.	1.05	50.	0.75		
4.5	1.03	51.23	0.75		
5.	1.03	52.	0.74		
5.5	1.01	53.07	0.74		
6.	1.01	54.03	0.73		

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
6.5	1.01	55.	0.73
7.	1.	56.05	0.73
7.5	0.99	57.32	0.72
8.	0.98	58.	0.72
8.5	0.98	59.	0.72
9.	0.97	60.	0.72
9.5	0.97	70.	0.71
10.	0.96	80.	0.67
11.	0.94	90.	0.66
12.	0.93	108.6	0.63
13.	0.93	123.8	0.61
14.	0.91	149.	0.61
15.	0.9	170.4	0.58
16.	0.9	204.4	0.58
17.13	0.89	226.	0.54
18.	0.88	270.9	0.52
19.	0.88	331.8	0.51
20.	0.87	1506.	0,36
21.	0.86	1722.	0.33
22.	0.85	1960.	0.31
23.03	0.85	2879.	0.23
24.23	0.85	3117.	0.2
25.	0.84	3380.	0.2
26.	0.84	3427.	0.19
27.	0.83	3464.	0.19
28.12	0.82	4328.	0.13
29.	0.81	4405.	0.13
30.	0.81	4431.	0.13
31.	0.81	4456.	0.13
32.	0.8	4499.	0.12
33.1	0.8	4775.	0.08
34.05	0.8	4899.	0.08
35.	0.8	5848.	0.01

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.9E-7	cm/sec
Ss	0.2	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 8.423E-5 \text{ cm}^2/\text{sec}$



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\G46D_fh1.aqt Title: G46D Falling Head #1 Date: 09/27/16 Time: 14:28:21

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 29 Aug 2016 Test Well: G46D

AQUIFER DATA

Saturated Thickness: 19.34 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: G46D

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.35 ft Static Water Column Height: 19.34 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 9.65 ft Total Well Penetration Depth: 18.95 ft

Observation Data				
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.33	1.35	21.07	1.17	
0.5	1.35	22.	1.16	
0.67	1.33	23.	1.15	
0.83	1.32	24.3	1.15	
1.	1.31	25.35	1.15	
1.17	1.3	26.47	1.14	
1.33	1.3	27.33	1.14	
1.5	1.3	28.28	1.14	
1.67	1.3	29.13	1.13	
1.83	1.3	30.25	1.12	
2.	1.3	31.9	1.12	
2.25	1.29	41.25	1.11	
2.5	1.29	42.63	1.11	
2.75	1.29	43.55	1.1	
3.	1.29	44.42	1.1	
3.25	1.29	45.33	1.09	
3.5	1.28	46.18	1.09	
3.75	1.28	47.12	1.09	
4.	1.28	48.	1.09	

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
4.25	1.28	49.	1.09
4.5	1.27	50.	1.09
4.75	1.27	51.	1.09
5.	1.26	52.05	1.08
5.25	1.26	53.23	1.07
5.5	1.25	54.	1.07
5.75	1.25	55.35	1.07
6.	1.25	56.57	1.07
6.25	1.25	57.48	1.06
6.5	1.25	58.28	1.06
6.75	1.24	59.	1.05
7.	1.24	60.	1.05
7.25	1.24	70.	1.04
7.5	1.24	80.	1.01
7.75	1.23	90.	1.01
8.	1.23	100.	1.
8.25	1.23	1143.	0.55
8.5	1.23	1332.	0.5
8.75	1.22	1607.	0.46
9.	1.23	2397.	1.34
9.25	1.22	2507.	1.31
9.5	1.22	2778.	1.3
9.75	1.22	2962.	0.24
10.	1.22	3001.	0.24
11.	1.21	3056.	1.25
12.	1.21	3856.	0.16
13.	1.2	3930.	0.15
14.	1.2	3975.	0.15
15.05	1.2	4012.	0.15
16.	1.19	4057.	0.14
17.	1.18	4337.	0.1
18.13	1.18	4445.	0.12
19.07	1.17	5325.	0.11
20.	1.17		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.0E-8	cm/sec
Ss	0.2	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 2.358E-5 \text{ cm}^2/\text{sec}$



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T408_fh1.aqt Title: T408 Falling Head #1 Date: 09/27/16 Time: 14:28:22

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 1 Sept 2016 Test Well: T408

AQUIFER DATA

Saturated Thickness: 22.87 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T408

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.69 ft Static Water Column Height: 22.87 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.83 ft Total Well Penetration Depth: 22.44 ft

Observation Data				
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.188	0.829	9.44	0.67	
0.199	0.804	10.	0.669	
0.211	0.782	10.6	0.666	
0.224	0.702	11.2	0.666	
0.237	0.693	11.9	0.66	
0.251	0.715	12.6	0.662	
0.266	0.856	13.3	0.66	
0.282	0.593	14.1	0.66	
0.298	0.667	15.	0.657	
0.316	0.718	15.8	0.654	
0.335	0.686	16.8	0.657	
0.355	0.669	17.8	0.65	
0.376	0.777	18.8	0.65	
0.398	0.71	19.9	0.648	
0.422	0.692	21.1	0.645	
0.447	0.699	22.4	0.643	
0.473	0.701	23.7	0.641	
0.501	0.693	25.1	0.635	
0.531	0.697	26.6	0.636	

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.562	0.696	28.2	0.632
0.596	0.688	29.8	0.628
0.631	0.693	31.6	0.626
0.668	0.697	33.5	0.623
0.708	0.696	35.5	0.616
0.75	0.694	37.6	0.616
0.794	0.696	39.8	0.612
0.841	0.696	42.2	0.606
0.891	0.706	44.7	0.603
0.944	0.699	47.3	0.599
1.	0.689	50.1	0.595
1.06	0.687	53.1	0.592
1.12	0.689	56.2	0.586
1.19	0.689	59.6	0.58
1.26	0.69	63.1	0.571
1.33	0.691	66.8	0 569
1.41	0.685	70.8	0 562
1.5	0.69	75	0.557
1 58	0.696	79.4	0.546
1.68	0.691	84 1	0.54
1 78	0.688	89.1	0.54
1.88	0.686	94.4	0.534
1.00	0.685	100	0.524
2 11	0.603	106	0.514
2.11	0.691	112	0.304
2.27	0.69	112.	0.495
2.57	0.09	119.	0.487
2.51	0.000	120.	0.404
2.00	0.005	135.	0.474
2.02	0.000	141.	0.459
2.90	0.000	150.	0.45
3.10	0.000	158.	0.441
3.33	0.684	108.	0.427
3.33	0.684	178.	0.415
3.70	0.682	188.	0.403
3.98	0.681	199.	0.387
4.22	0.683	211.	0.368
4.47	0.682	224.	0.356
4.73	0.68	237.	0.339
5.01	0.683	251.	0.326
5.31	0.68	266.	0.307
5.62	0.681	282.	0.292
5.96	0.678	298.	0.272
6.31	0.676	316.	0.256
6.68	0.673	335.	0.233
7.08	0.674	355.	0.215
7.5	0.675	376.	0.188
7.94	0.672	398.	0.178
8.41	0.67	422.	0.146
8.91	0.668	447.	0.123

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.7E-6	cm/sec
Ss	1.0E-7	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 0.003276 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	6.277E-6	7.768E-6	+/- 1.537E-5	0.8081	cm/sec
Ss	4.0E-12	2.586E-10	+/- 5.116E-10	0.01547	ft ⁻¹
Kz/Kr	1.	not estimated			

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

 $T = K*b = 0.004376 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss
Kr	1.00	-1.00
Ss	-1.00	1.00

Residual Statistics

for weighted residuals

Sum of Squares 0	.103 ft ²
Variance	0.0007689 ft^2
Std. Deviation0	.02773 ft
Mean0	.006273 ft
No. of Residuals 1	36
No. of Estimates 2	



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T408_fh2.aqt Title: T408 Falling Head #2 Date: 09/27/16 Time: 14:28:23

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 1 Sept 2016 Test Well: T408

AQUIFER DATA

Saturated Thickness: 22.74 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T408

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.36 ft Static Water Column Height: 22.74 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.83 ft Total Well Penetration Depth: 22.31 ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.105	2.278	13.18	1.334
0.118	1.901	13.98	1.334
0.132	1.781	14.88	1.334
0.147	1.557	15.68	1.332
0.163	1.058	16.68	1.336
0.179	1.627	17.68	1.334
0.197	1.213	18.68	1.333
0.216	1.478	19.78	1.332
0.236	1.294	20.98	1.34
0.257	1.342	22.28	1.329
0.279	1.508	23.58	1.327
0.303	1.612	24.98	1.327
0.328	1.498	26.48	1.318
0.354	1.505	28.08	1.32
0.382	1.359	29.68	1.319
0.412	1.354	31.48	1.317
0.443	1.361	33.38	1.318
0.477	1.368	35.38	1.313
0.512	1.363	37.48	1.307

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.549	1.363	39.68	1.311
0.589	1.361	42.08	1.309
0.631	1.354	44.58	1,306
0.675	1.362	47 18	1 303
0 722	1 363	40.08	1.305
0.772	1 364	F2 09	1.07
0.772	1.307	52.90	1.297
0.025	1.303	56.08	1.294
0.881	1.363	59.48	1.294
0.941	1.361	62.98	1.289
1.001	1.362	70.68	0.908
1.071	1.355	74.88	1.061
1.141	1.363	79.28	1.148
1.211	1.361	83.98	1.267
1.291	1.354	88.98	1.327
1.381	1.37	94.28	1 308
1.461	1.353	99.88	1 3
1 561	1 356	105.9	1 216
1 661	1 355	111.0	1.510
1.001	1.355	111.9	1.354
1.701	1.355	118.9	1.351
1.871	1.363	125.9	1.351
1.991	1.355	132.9	1.365
2.121	1.354	140.9	1.367
2.251	1.352	149.9	1.399
2.391	1.35	157.9	1.418
2.541	1.352	167.9	1.428
2.701	1.358	177.9	1.425
2.861	1.358	187.9	1 416
3.041	1.35	198.9	1 39
3 231	1 347	210.9	1.35
3 431	1 348	222.0	1.377
3 641	1 246	223.9	1.343
2 961	1.340	230.9	1.314
3.001	1.349	250.9	1.291
4.101	1.347	265.9	1.262
4.351	1.349	281.9	1.24
4.611	1.341	297.9	1.213
4.891	1.344	315.9	1.198
5.191	1.348	334.9	1.183
5.501	1.349	354.9	1.157
5.841	1.344	375.9	1.141
6.191	1.345	397.9	1.131
6.561	1.34	421.9	1,109
6.961	1.343	446.9	1 088
7.381	1.343	472.9	1.065
7 821	1 342	500.9	1.005
8 201	1 346	520.0	1.034
8 701	1 220	JJU.9 EC1 0	1.032
0.791	1.222	201.9	1.005
9.321	1.342	595.9	0.98
9.881	1.343	630.9	0.957
10.48	1.34	667.9	0.935
11.08	1.342	707.9	0.914
11.78	1.341	749.9	0.885
12.48	1.335	793.9	0.867

Slug Test

Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	9.8E-7	cm/sec
Ss	1.0E-5	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 0.0006793 \text{ cm}^2/\text{sec}$



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T408_rh1.aqt Title: T408 Rising Head #1 Date: 09/27/16 Time: 14:28:24

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 1 Sept 2016 Test Well: T408

AQUIFER DATA

Saturated Thickness: 24.46 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T408

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.75 ft Static Water Column Height: 24.46 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.83 ft Total Well Penetration Depth: 24.03 ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.06667	1.47	4.73	1.722
0.07083	1.296	5.01	1.719
0.075	2.597	5.31	1.72
0.07917	1.239	5.62	1.716
0.08333	1.865	5.96	1.716
0.0875	2.161	6.31	1.717
0.09167	1.215	6.68	1.71
0.09583	2.203	7.08	1.716
0.1	1.678	7.5	1.713
0.106	1.974	7.94	1.714
0.112	1.513	8.41	1.716
0.119	2.062	8.91	1.713
0.126	1.617	9.44	1.715
0.133	1.711	10.	1.712
0.141	1.932	10.6	1.711
0.15	1.834	11.2	1.709
0.158	1.643	11.9	1.709
0.168	1.658	12.6	1.705
0.178	1.693	13.3	1.705

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.188	1./3	14.1	1.707
0.199	1./33	15.	1.704
0.211	1.72	15.8	1.703
0.224	1./25	16.8	1.704
0.237	1.75	17.8	1.697
0.251	1.759	18.8	1.701
0.266	1.734	19.9	1.7
0.282	1.756	21.1	1.7
0.298	1.741	22.4	1.701
0.316	1.745	23.7	1.696
0.335	1.745	25.1	1.697
0.355	1.742	26.6	1.691
0.376	1.738	28.2	1.685
0.398	1.74	29.8	1.689
0.422	1.74	31.6	1.69
0.447	1.742	33.5	1.682
0.473	1.738	35.5	1.686
0.501	1.738	37.6	1.681
0.531	1.738	39.8	1.681
0.562	1.743	42.2	1.681
0.596	1.736	44.7	1.676
0.631	1.735	47.3	1.675
0.668	1.733	50.1	1.674
0.708	1.733	53.1	1.67
0.75	1.734	56.2	1.665
0.794	1.73	59.6	1.663
0.841	1.731	63.1	1.662
0.891	1.728	66.8	1.659
0.944	1.73	70.8	1.659
1.	1.731	75.	1.652
1.06	1.732	79.4	1.65
1.12	1.729	84.1	1.646
1.19	1.728	89.1	1.646
1.26	1.725	94.4	1.636
1.33	1.734	100.	1.636
1.41	1.727	106.	1.634
1.5	1.734	112.	1.624
1.58	1.724	119.	1.624
1.68	1.728	126.	1.616
1.78	1.722	133.	1.614
1.88	1.723	141.	1.605
1.99	1.727	150.	1.605
2.11	1.72	158.	1.6
2.24	1.727	168.	1.597
2.37	1.724	178.	1.587
2.51	1.726	188.	1.584
2.66	1.72	198.	1.578
2.82	1.726	208.	1.572
2.98	1.721	218.	1,566
3.16	1.722	228.	1.562
3.35	1.722	238.	1 552
3.55	1.722	248.	1.551
3.76	1.72	258	1 547
3.98	1.723	268	1 541
4.22	1.721	278	1 538
4.47	1.719	=, 01	1.000

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	7.5E-8	cm/sec
Ss	0.015	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 5.592E-5 \text{ cm}^2/\text{sec}$



Kz/Kr = 1.

Kr

 $= 0.00075 \text{ ft}^{-1}$

Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T409_fh1.aqt Title: T409 Falling Head #1 Date: 09/27/16 Time: 14:28:25

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 31 Aug 2016 Test Well: T409

AQUIFER DATA

Saturated Thickness: 21.11 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T409

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.68 ft Static Water Column Height: 21.11 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.8 ft Total Well Penetration Depth: 20.71 ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.	1.591	1.505	1.526
0.004167	1.749	1.605	1.517
0.008333	2.437	1.705	1.513
0.0125	1.849	1.805	1.503
0.01667	2.03	1.915	1.496
0.02083	2.071	2.035	1.488
0.025	0.829	2.165	1.48
0.031	2.267	2.295	1.471
0.037	1.763	2.435	1.462
0.044	1.835	2.585	1.453
0.051	1.709	2.745	1.438
0.058	1.529	2.905	1.429
0.066	1.481	3.085	1.421
0.075	1.922	3.275	1.408
0.083	1.392	3.475	1.777
0.093	1.549	3.685	1.368
0.103	1.756	3.905	1.357
0.113	1.836	4.145	1.345
0.124	1.226	4.395	1.33

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.136	1.106	4.655	1.264
0.149	1.008	4.935	1.302
0.162	1.574	5.235	1.286
0.176	1.718	5.545	1.265
0.191	1.534	5.885	1.254
0.207	1.511	6.235	1.233
0.223	1.601	6.605	1.212
0.241	1.876	7.005	1.195
0.26	1.826	7.425	1.172
0.28	1.595	7.865	1.153
0.301	1.632	8.335	1.132
0.323	1.628	8.835	1.109
0.347	1.627	9.365	1.087
0.372	1.624	9.925	1.066
0.398	1.626	10.53	1.038
0.426	1.626	11.13	1.017
0.456	1.62	11.82	0.989
0.487	1.612	12.53	0.967
0.521	1.613	13.23	0.943
0.556	1.611	14.03	0.918
0.593	1.605	14.93	0.886
0.633	1.597	15.73	0.863
0.675	1.598	16.73	0.835
0.719	1.596	17.73	0.807
0.766	1.589	18.73	0.78
0.816	1.58	19.82	0.756
0.869	1.582	21.02	0.721
0.925	1.571	22.32	0.693
0.985	1.575	23.63	0.664
1.045	1.563	25.02	0.64
1.115	1.559	26.52	0.602
1.185	1.553	28.13	0.582
1.255	1.548	29.73	0.553
1.335	1.541	31.52	0.525
1.425	1.535		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	4.0E-5	cm/sec
Ss	0.00075	ft ⁻¹
Kz/Kr	1.	

$T = K*b = 0.02574 \text{ cm}^2/\text{sec}$



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T409_fh2.aqt Title: T409 Falling Head #2 Date: 09/27/16 Time: 14:28:26

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 31 Aug 2016 Test Well: T409

AQUIFER DATA

Saturated Thickness: 21.12 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T409

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.58 ft Static Water Column Height: 21.12 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.8 ft Total Well Penetration Depth: 20.72 ft

No. of Observations: 147

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.0375	2.629	2.98	1.331
0.04167	2.264	3.16	1.327
0.0474	2.589	3.35	1.313
0.05112	2.601	3.55	1.303
0.0548	2.488	3.76	1.291
0.05848	2.419	3.98	1.28
0.0625	2.243	4.22	1.264
0.06667	2.278	4.47	1.248
0.07083	2.602	4.73	1.235
0.075	1.82	5.01	1.222
0.07917	1.69	5.31	1.205
0.08333	1.954	5.62	1.192
0.0875	1.648	5.96	1.178
0.09167	1.93	6.31	1.16
0.09583	1.407	6.68	1.146
0.1	1.144	7.08	1.126
0.106	1.537	7.5	1.111
0.112	1.777	7.94	1.089
0.119	1.365	8.41	1.067

1

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.126	1.584	8.91	1.051
0.133	1.535	9.44	1.026
0.141	1.523	10.	1.004
0.15	1.541	10.6	0.982
0.158	1.538	11.2	0.961
0.168	1.535	11.9	0.94
0.178	1.538	12.6	0.918
0.188	1.533	13.3	0.893
0.199	1.537	14.1	0.87
0.211	1.539	15.	0.843
0.224	1.523	15.8	0.818
0.237	1.533	16.8	0.795
0.251	1.534	17.8	0.766
0.266	1.528	18.8	0.742
0.282	1.534	19.9	0.711
0.298	1.531	21.1	0.683
0.316	1.523	22.4	0.655
0.335	1.523	23.7	0.633
0.355	1.521	25.1	0.604
0.376	1.518	26.6	0.579
0.398	1.517	28.2	0.552
0.422	1.514	29.8	0.528
0.447	1.513	31.6	0.507
0.473	1.508	33.5	0.477
0.501	1.51	35.5	0.453
0.531	1.505	37.6	0.43
0.562	1.504	39.8	0.406
0.596	1.499	42.2	0.382
0.631	1.494	44.7	0.357
0.668	1.493	47.3	0.341
0.708	1.491	50.1	0.32
0.75	1.482	53.1	0.298
0.794	1.483	56.2	0.28
0.841	1.481	59.6	0.263
0.891	1.47	63.1	0.247
0.944	1.469	66.8	0.229
1.	1.462	70.8	0.211
1.06	1.457	75.	0.198
1.12	1.454	79.4	0.182
1.19	1.447	84.1	0.174
1.26	1.447	89.1	0.16
1.33	1.442	94.4	0.146
1.41	1.435	100.	0.137
1.5	1.427	106.	0.126
1.58	1.426	112.	0.119
1.68	1.394	119.	0.106
1.78	1.408	126.	0.096
1.88	1.403	133.	0.087
1.99	1.393	141.	0.082
2.11	1.387	150.	0.074
2.24	1.38	158.	0.072
2.37	1.371	168.	0.064
2.51	1.364	178.	0.06
2.66	1.354	188.	0.058
2.82	1.341		

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	3.2E-5	cm/sec
Ss	0.0025	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 0.0206 \text{ cm}^2/\text{sec}$



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T409_rh1.aqt Title: T409 Rising Head #1 Date: 09/27/16 Time: 14:28:27

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 31 Aug 2016 Test Well: T409

AQUIFER DATA

Saturated Thickness: 21.19 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T409

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.59 ft Static Water Column Height: 21.19 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.8 ft Total Well Penetration Depth: 20.79 ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.04167	2.261	2.66	1.311
0.04623	0.099	2.82	1.303
0.04998	0.796	2.98	1.288
0.05417	2.26	3.16	1.281
0.05833	1.248	3.35	1.265
0.0625	0.907	3.55	1.257
0.06667	2.035	3.76	1.245
0.07083	1.886	3.98	1.232
0.075	1.307	4.22	1.216
0.07917	1.65	4.47	1.203
0.08333	1.554	4.73	1.184
0.0875	1.548	5.01	1.172
0.09167	1.555	5.31	1.159
0.09583	1.548	5.62	1.137
0.1	1.548	5.96	1.124
0.106	1.545	6.31	1.101
0.112	1.548	6.68	1.085
0.119	1.54	7.08	1.066
0.126	1.544	7.5	1.052

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.133	1.537	7.94	1.027
0.141	1.544	8.41	1.005
0.15	1.537	8.91	0.988
0.158	1.534	9.44	0.966
0.168	1.535	10.	0.946
0.178	1.533	10.6	0.919
0.188	1.527	11.2	0.905
0.199	1.526	11.9	0.872
0.211	1.526	12.6	0.849
0.224	1.521	13.3	0.828
0.237	1.518	14.1	0.805
0.251	1.516	15.	0.78
0.266	1.517	15.8	0.753
0.282	1.514	16.8	0.73
0.298	1.514	17.8	0.705
0.316	1.507	18.8	0.682
0.335	1.506	19.9	0.652
0.355	1.505	21.1	0.633
0.376	1.5	22.4	0.603
0.398	1.493	23.7	0.58
0.422	1.491	25.1	0.548
0.447	1.488	26.6	0.526
0.473	1.488	28.2	0.506
0.501	1.482	29.8	0.481
0.531	1.476	31.6	0.45
0.562	1.473	33.5	0.433
0.596	1.471	35.5	0.403
0.631	1.467	37.6	0.384
0.668	1.461	39.8	0.363
0.708	1.456	42.2	0.345
0.75	1.455	44.7	0.329
0.794	1.449	47.3	0.309
0.841	1.43	50.1	0.285
0.891	1.55	53.1	0.269
0.944	1.461	56.2	0.255
1.	1.437	59.6	0.237
1.06	1.421	63.1	0.222
1.12	1.423	66.8	0.211
1.19	1.41	70.8	0.203
1.26	1.408	75.	0.185
1.33	1.404	79.4	0.173
1.41	1.398	84.1	0.164
1.5	1.391	89.1	0.156
1.58	1.385	94.4	0.146
1.68	1.374	100.	0.139
1.78	1.369	106.	0.128
1.88	1.36	112.	0.123
1.99	1.348	119.	0.11
2.11	1.347	126.	0.107
2.24	1.334	133.	0.097
2.37	1.328	141.	0.09
2.51	1.321		0.00

Slug Test

Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	3.2E-5	cm/sec
Ss	0.005	ft ⁻¹
Kz/Kr	1.	

 $T = K*b = 0.02067 \text{ cm}^2/\text{sec}$



Data Set: I:\16jobs\16E0080\Admin\15-Field-Laboratory Data\SlugTest2\T409_rh2.aqt Title: T409 Rising Head #2 Date: 09/27/16 Time: 14:28:28

PROJECT INFORMATION

Company: Hanson Professional Services Client: Natural Resources Technology, Project: 16E0080 Location: Coffeen Power Station Test Date: 31 Aug 2016 Test Well: T409

AQUIFER DATA

Saturated Thickness: 21.18 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: T409

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.65 ft Static Water Column Height: 21.18 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.25 ft Screen Length: 4.8 ft Total Well Penetration Depth: 20.78 ft

Observation Data				
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
0.02502	1.747	2.37	1.363	
0.02918	1.749	2.512	1.346	
0.03335	0.77	2.66	1.347	
0.03752	0.899	2.82	1.33	
0.04168	1.329	2.98	1.321	
0.04823	1.536	3.16	1.308	
0.05192	1.787	3.35	1.261	
0.05568	1.583	3.55	1.288	
0.05937	1.624	3.76	1.275	
0.06303	1.634	3.98	1.263	
0.0667	1.611	4.22	1,246	
0.07085	1.608	4.47	1.234	
0.07502	1.607	4.73	1.221	
0.07918	1.607	5.01	1.201	
0.08335	1.601	5.31	1.182	
0.08752	1.594	5.62	1.168	
0.09168	1.596	5.96	1.149	
0.09585	1.6	6.31	1 138	
0.1	1.594	6.68	1.116	
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)	
------------	-------------------	--------------------------	-------------------	
0.100	1.592	7.08	1.1	
0.112	1.589	7.5	1.076	
0.119	1.588	7.94	1.06	
0.126	1.582	8.41	1.039	
0.133	1.58	8.91	1.015	
0.141	1.585	9.44	0.995	
0.15	1.58	10.	0.971	
0.158	1.579	10.6	0 944	
0.168	1.575	11.2	0.926	
0.178	1.573	11.9	0.920	
0.188	1.566	12.6	0.874	
0.199	1.565	13.3	0.851	
0.211	1.562	14.1	0.831	
0.224	1.56	15	0.831	
0.237	1.558	15.8	0.795	
0.251	1,558	16.8	0.774	
0.266	1 559	17.0	0.747	
0.282	1 553	10 0	0.724	
0.298	1 549	10.0	0.699	
0.316	1 549	19.9	0.673	
0.335	1.540	21.1	0.642	
0.355	1.544	22.4	0.615	
0.335	1.544	23.7	0.589	
0.370	1.538	25.1	0.565	
0.398	1.534	26.6	0.538	
0.422	1.529	28.2	0.517	
0.447	1.527	29.8	0.493	
0.4/3	1.523	31.6	0.464	
0.501	1.512	33.5	0.44	
0.531	1.513	35.5	0.415	
0.564	1.498	37.6	0.396	
0.596	1.497	39.8	0.37	
0.631	1.504	42.2	0.351	
0.668	1.502	44.7	0.326	
0.708	1.501	47.3	0 311	
0.75	1.491	50.1	0.288	
0.794	1.494	53.1	0.273	
0.841	1.486	56.2	0.255	
0.891	1.48	59.6	0.24	
0.944	1.474	63.1	0.24	
1.	1.469	66.8	0.222	
1.06	1.468	70.8	0.102	
1.12	1.461	75	0.193	
1.19	1 453	70.4	0.181	
1.26	1 448	79. 4 94 1	0.1/1	
1.33	1 44	84.1	0.161	
1 41	1.436	89.1	0.15	
15	1 429	94.4	0.147	
1.5	1.420	100.	0.136	
1.50	1.423	106.	0.128	
1 70	1.413	112.	0.117	
1.70	1.405	119.	0.107	
1.00	1.398	126.	0.097	
7.33	1.391	133.	0.089	
2.11	1.385	141.	0.084	
2.24	1.373	150.	0.077	

Slug Test Aquifer Model: Confined Solution Method: KGS Model

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
Kr	3.3E-5	cm/sec
Ss	0.006	ft ⁻¹
Kz/Kr	1.	

$T = K*b = 0.0213 \text{ cm}^2/\text{sec}$

AUTOMATIC ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	Std. Error	Approx. C.I.	t-Ratio	
Kr	3.2E-5	3.191E-6	+/- 6.309E-6	10.03	cm/sec
Ss	0.006332	0.002124	+/- 0.0042	2 981	ft-1
Kz/Kr	1.	not estimated	,	2.501	

C.I. is approximate 95% confidence interval for parameter t-ratio = estimate/std. error No estimation window

$T = K*b = 0.02066 \text{ cm}^2/\text{sec}$

Parameter Correlations

	Kr	Ss
Kr	1.00	-0.93
Ss	-0.93	1.00

Residual Statistics

for weighted residuals

Sum of Squares		÷				1.445 ft ²
Variance						0.01003 ft ²
Std. Deviation .						0.1002 ft
Mean	×			x.		-0.01263 ft
No. of Residuals						146
No. of Estimates						2

T409 Rising Head #2

APPENDIX B2

HYDRAULIC CONDUCTIVITY ANALYSES (ASH POND PIEZOMETERS)



AQTESOLV for Windows

Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW1_FH.aqt Title: Falling Head Test Date: 11/11/16 Time: 13:53:32

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW1

AQUIFER DATA

Saturated Thickness: 16.8 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW1

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.15 ft Static Water Column Height: 16.8 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 13. ft Total Well Penetration Depth: 20. ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
0.011	1.082	0.8238	0.972		
0.022	1.057	0.8708	0.972		
0.033	1.006	0.9207	0.972		
0.044	0.969	0.9733	0.972		
0.055	0.948	1.029	0.973		
0.066	0.948	1.088	0.973		
0.077	0.955	1.151	0.972		
0.088	0.963	1.217	0.973		
0.099	0.967	1.288	0.973		
0.11	0.969	1.362	0.973		
0.121	0.969	1.441	0.973		
0.132	0.969	1.525	0.973		
0.143	0.967	1.613	0.975		
0.154	0.967	1.707	0.973		
0.165	0.969	1.807	0.973		
0.176	0.969	1.912	0.973		
0.187	0.967	2.023	0.973		
0.198	0.969	2.142	0.973		

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	0.969	2.267	0.973
0.22	0.969	2.399	0.973
0.231	0.969	2.54	0.973
0.2427	0.97	2.688	0.973
0.2552	0.969	2.846	0.975
0.2683	0.969	3.013	0.975
0.2823	0.969	3.179	0.975
0.2972	0.969	3.346	0.975
0.3128	0.969	3.513	0.976
0.3295	0.97	3.679	0.975
0.3472	0.97	3.846	0.976
0.3658	0.97	4.013	0.975
0.3857	0.97	4.179	0.975
0.4067	0.97	4.346	0.975
0.4288	0.97	4.513	0.975
0.4523	0.97	4.679	0.975
0.4772	0.97	4.846	0.976
0.5035	0.972	5.013	0.976
0.5315	0.972	5.179	0.975
0.5612	0.972	5.346	0.975
0.5925	0.972	5.513	0.975
0.6257	0.972	5.679	0.976
0.6608	0.972	5.846	0.975
0.6982	0.972	6.013	0.976
0.7377	0.972	6.179	0.975
0.7795	0.972		

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.192

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.00225	cm/sec
y 0	1.15	ft

 $T = K*b = 1.152 \text{ cm}^2/\text{sec}$



Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW1_RH.aqt Title: Rising Head Test Date: 11/11/16 Time: 13:53:33

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW1

AQUIFER DATA

Saturated Thickness: 16.8 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW1

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.1 ft Static Water Column Height: 16.8 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 13. ft Total Well Penetration Depth: 20. ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
0.011	0.493	0.8238	0.278		
0.022	0.308	0.8708	0.278		
0.033	0.278	0.9207	0.278		
0.044	0.272	0.9733	0.278		
0.055	0.283	1.029	0.278		
0.066	0.287	1.088	0.278		
0.077	0.284	1.151	0.28		
0.088	0.281	1.217	0.28		
0.099	0.281	1.288	0.28		
0.11	0.28	1.362	0.28		
0.121	0.278	1.441	0.278		
0.132	0.278	1.525	0.28		
0.143	0.278	1.613	0.28		
0.154	0.278	1.707	0.278		
0.165	0.278	1.807	0.28		
0.176	0.278	1.912	0.281		
0.187	0.278	2.023	0.28		
0.198	0.278	2.142	0.28		

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	0.278	2.267	0.28
0.22	0.28	2.399	0.28
0.231	0.278	2.54	0.28
0.2427	0.278	2.688	0.281
0.2552	0.278	2.846	0.281
0.2683	0.278	3.013	0.281
0.2823	0.278	3.179	0.281
0.2972	0.278	3.346	0.281
0.3128	0.278	3.513	0.283
0.3295	0.278	3.679	0.281
0.3472	0.278	3.846	0.283
0.3658	0.278	4.013	0.281
0.3857	0.278	4.179	0.281
0.4067	0.278	4.346	0.281
0.4288	0.278	4.513	0.283
0.4523	0.278	4.679	0.281
0.4772	0.278	4.846	0.281
0.5035	0.28	5.013	0.283
0.5315	0.28	5.179	0.281
0.5612	0.28	5.346	0.283
0.5925	0.281	5.513	0.283
0.6257	0.28	5.679	0.281
0.6608	0.28	5.846	0.283
0.6982	0.28	6.013	0.281
0.7377	0.278	6.179	0.283
0.7795	0.277	6.346	0.283

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.192

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.017	cm/sec
y0	0.7	ft

 $T = K*b = 8.705 \text{ cm}^2/\text{sec}$



AQTESOLV for Windows

Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW2_FH.aqt Title: Falling Head Test Date: 11/11/16 Time: 13:53:34

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW2

AQUIFER DATA

Saturated Thickness: 22.7 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW2

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.75 ft Static Water Column Height: 22.7 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 13. ft Total Well Penetration Depth: 26. ft

Observation Data					
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)		
0.011	0.475	0.6608	0.567		
0.022	0.639	0.6982	0.567		
0.033	0.581	0.7377	0.567		
0.044	0.521	0.7795	0.567		
0.055	0.53	0.8238	0.567		
0.066	0.566	0.8708	0.567		
0.077	0.575	0.9207	0.567		
0.088	0.563	0.9733	0.567		
0.099	0.555	1.029	0.567		
0.11	0.56	1.088	0.567		
0.121	0.564	1.151	0.567		
0.132	0.566	1.217	0.567		
0.143	0.564	1.288	0.567		
0.154	0.564	1.362	0.567		
0.165	0.566	1.441	0.566		
0.176	0.566	1.525	0.566		
0.187	0.566	1.613	0.567		
0.198	0.564	1.707	0.566		

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	0.566	1.807	0.567
0.22	0.564	1.912	0.567
0.231	0.566	2.023	0.567
0.2427	0.566	2.142	0.567
0.2552	0.566	2.267	0.567
0.2683	0.566	2.399	0.567
0.2823	0.567	2.54	0.567
0.2972	0.567	2.688	0.567
0.3128	0.567	2.846	0.567
0.3295	0.567	3.013	0.567
0.3472	0.567	3.179	0.567
0.3658	0.566	3.346	0.567
0.3857	0.567	3.513	0.567
0.4067	0.567	3.679	0.567
0.4288	0.567	3.846	0.567
0.4523	0.566	4.013	0.569
0.4772	0.566	4.179	0.569
0.5035	0.567	4.346	0.567
0.5315	0.567	4.513	0.567
0.5612	0.567	4.679	0.567
0.5925	0.567	4.846	0.569
0.6257	0.567	5.013	0.569

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.359

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.0045	cm/sec
y0	0.75	ft

 $T = K*b = 3.114 \text{ cm}^2/\text{sec}$



AQTESOLV for Windows

Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW2_RH.aqt Title: Rising Head Test Date: 11/11/16 Time: 13:53:35

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW2

AQUIFER DATA

Saturated Thickness: 22.7 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW2

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.1 ft Static Water Column Height: 22.7 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 13. ft Total Well Penetration Depth: 26. ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.011	0.601	0.7795	1.005
0.022	1.081	0.8238	1.005
0.033	1.212	0.8708	1.005
0.044	1.093	0.9207	1.005
0.055	0.959	0.9733	1.005
0.066	0.929	1.029	1.005
0.077	0.969	1.088	1.005
0.088	1.011	1.151	1.005
0.099	1.017	1.217	1.005
0.11	1.009	1.288	1.005
0.121	1.	1.362	1.005
0.132	0.997	1.441	1.003
0.143	1.	1.525	1.005
0.154	1.002	1.613	1.005
0.165	1.003	1.707	1.005
0.176	1.003	1.807	1.005
0.187	1.003	1.912	1.005
0.198	1.003	2.023	1.005

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	1.003	2.142	1.003
0.22	1.003	2.267	1.003
0.231	1.003	2.399	1.005
0.2427	1.003	2.54	1.003
0.2552	1.003	2.688	1.005
0.2683	1.003	2.846	1.003
0.2823	1.003	3.013	1.005
0.2972	1.005	3.179	1.003
0.3128	1.005	3.346	1.005
0.3295	1.005	3.513	1.008
0.3472	1.006	3.679	1.008
0.3658	1.005	3.846	1.008
0.3857	1.005	4.013	1.006
0.4067	1.006	4.179	1.005
0.4288	1.005	4.346	1.005
0.4523	1.005	4.513	1.005
0.4772	1.005	4.679	1.005
0.5035	1.005	4.846	1.003
0.5315	1.005	5.013	1.003
0.5612	1.003	5.179	1.003
0.5925	1.005	5.346	1.003
0.6257	1.005	5.513	1.003
0.6608	1.005	5.679	1.003
0.6982	1.005	5.846	1.005
0.7377	1.005	6.013	1.005

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.359

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.0055	cm/sec
y 0	1.65	ft

 $T = K*b = 3.805 \text{ cm}^2/\text{sec}$



Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW3_FH.aqt Title: Falling Head Test Date: 11/11/16 Time: 13:53:36

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW3

AQUIFER DATA

Saturated Thickness: 21.9 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW3

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.15 ft Static Water Column Height: 21.9 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 13. ft Total Well Penetration Depth: 26. ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.0112	1.095	0.618	0.911
0.0223	1.054	0.6532	0.911
0.0335	0.994	0.6905	0.912
0.0447	0.937	0.73	0.911
0.0558	0.897	0.7718	0.912
0.067	0.879	0.8162	0.912
0.0782	0.882	0.8632	0.911
0.0893	0.894	0.913	0.912
0.1005	0.909	0.9657	0.912
0.1117	0.917	1.022	0.912
0.1228	0.92	1.081	0.912
0.134	0.918	1.143	0.911
0.1452	0.914	1.21	0.912
0.1563	0.911	1.28	0.912
0.1675	0.908	1.355	0.912
0.1787	0.908	1.434	0.912
0.1898	0.909	1.517	0.912
0.201	0.911	1.606	0.912

Time (min)	Displacement (ft)	<u>Time (min)</u>	Displacement (ft)
0.2122	0.912	1.7	0.912
0.2233	0.912	1.799	0.914
0.235	0.912	1.904	0.914
0.2475	0.911	2.016	0.914
0.2607	0.911	2.134	0.914
0.2747	0.911	2.259	0.914
0.2895 0.3052 0.3218 0.2205	0.911 0.912 0.911	2.392 2.532 2.681	0.912 0.914 0.914
0.3395	0.911	2.838	0.914
0.3582	0.912	3.005	0.914
0.378	0.912	3.172	0.914
0.399	0.911	3.338	0.914
0.4212 0.4447 0.4695	0.911 0.911 0.911 0.911	3.505 3.672 3.838	0.914 0.914 0.915 0.914
0.4958 0.5238 0.5535 0.5848	0.912 0.911 0.912 0.912	4.005 4.172 4.338	0.914 0.914 0.915

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.359

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.003	cm/sec
y0	1.2	ft

 $T = K*b = 2.003 \text{ cm}^2/\text{sec}$



Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW3_RH.aqt Title: Rising Head Test Date: 11/11/16 Time: 13:53:37

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW3

AQUIFER DATA

Saturated Thickness: 21.9 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW3

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1.1 ft Static Water Column Height: 21.9 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 13. ft Total Well Penetration Depth: 26. ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.011	0.729	0.5315	1.127
0.022	1.192	0.5612	1.127
0.033	1.32	0.5925	1.127
0.044	1.223	0.6257	1.128
0.055	1.088	0.6608	1.128
0.066	1.054	0.6982	1.127
0.077	1.098	0.7377	1.127
0.088	1.144	0.7795	1.127
0.099	1.153	0.8238	1.127
0.11	1.137	0.8708	1.127
0.121	1.121	0.9207	1.127
0.132	1.118	0.9733	1.128
0.143	1.125	1.029	1.127
0.154	1.133	1.088	1.127
0.165	1.133	1.151	1.127
0.176	1.128	1.217	1.128
0.187	1.127	1.288	1.127
0.198	1.127	1.362	1.127

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	1.128	1.441	1.127
0.22	1.128	1.525	1.127
0.231	1.128	1.613	1.127
0.2427	1.128	1.707	1.127
0.2552	1.128	1.807	1,127
0.2683	1.128	1.912	1.127
0.2823	1.128	2.023	1.128
0.2972	1.128	2.142	1.128
0.3128	1.128	2.267	1.127
0.3295	1.128	2.399	1.127
0.3472	1.128	2.54	1.128
0.3658	1.128	2.688	1.128
0.3857	1.128	2.846	1.128
0.4067	1.128	3.013	1.128
0.4288	1.128	3.179	1 137
0.4523	1.128	3.346	1 137
0.4772	1.128	3.513	1 137
0.5035	1.127	3.679	1.137

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.359

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	0.00425	cm/sec
y 0	1.67	ft

 $T = K*b = 2.837 \text{ cm}^2/\text{sec}$



AQTESOLV for Windows

Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW5_FH.aqt Title: Falling Head Test Date: 11/11/16 Time: 13:53:38

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW5

AQUIFER DATA

Saturated Thickness: 26.89 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW5

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 1. ft Static Water Column Height: 26.89 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 20. ft Total Well Penetration Depth: 30. ft

Observation Data			
Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.011	1.516	4.013	0.532
0.022	1.474	4.179	0.514
0.033	1.459	4.346	0.502
0.044	0.895	4.513	0.49
0.055	0.962	4.679	0.479
0.066	0.914	4.846	0.465
0.077	0.938	5.013	0.456
0.088	0.944	5.179	0.446
0.099	0.937	5.346	0.435
0.11	0.932	5.513	0.426
0.121	0.931	5.679	0.414
0.132	0.941	5.846	0.405
0.143	0.941	6.013	0.395
0.154	0.992	6.179	0.386
0.165	0.938	6.346	0.376
0.176	0.946	6.513	0.365
0.187	0.928	6.679	0.358
0.198	0.932	6.846	0.35
$\begin{array}{c} 0.099\\ 0.11\\ 0.121\\ 0.132\\ 0.143\\ 0.154\\ 0.165\\ 0.176\\ 0.187\\ 0.198\\ \end{array}$	0.937 0.932 0.931 0.941 0.941 0.992 0.938 0.946 0.928 0.932	5.346 5.513 5.679 5.846 6.013 6.179 6.346 6.513 6.679 6.846	$\begin{array}{c} 0.435\\ 0.426\\ 0.414\\ 0.405\\ 0.395\\ 0.386\\ 0.376\\ 0.365\\ 0.358\\ 0.35\end{array}$

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	0.935	7.013	$\frac{1}{0.341}$
0.22	0.937	7.179	0.332
0.231	0.925	7.346	0.323
0.2427	0.932	7.513	0.317
0.2552	0.932	7.679	0.307
0.2683	0.928	7.846	0.298
0.2823	0.925	8.013	0.291
0.2972	0.923	8.179	0.283
0.3128	0.922	8.346	0.276
0.3295	0.919	8.513	0.268
0.3472	0.917	8.679	0.262
0.3658	0.914	8.846	0.255
0.3857	0.911	9.013	0.247
0.4067	0.908	9.179	0.241
0.4288	0.905	9.346	0.234
0.4523	0.904	9.513	0.226
0.4772	0.901	9.679	0.219
0.5035	0.898	9.846	0.213
0.5315	0.89	10.01	0.215
0.5612	0.884	10.18	0.200
0.5925	0.88	10.35	0.194
0.6257	0.877	10.55	0.124
0.6608	0.871	10.51	0.183
0.6982	0.866	10.85	0.174
0.7377	0.86	11.01	0.174
0.7795	0.86	11.01	0.108
0.8238	0.852	11.10	0.105
0.8708	0.843	11.55	0.157
0.9207	0.838	11.51	0.132
0.9733	0.831	11.85	0.141
1.029	0.825	12.01	0.135
1.088	0.816	12.01	0.135
1.151	0.808	12.10	0.125
1 217	0.303	12.55	0.125
1 288	0.79	12.51	0.121
1 362	0.783	12.00	0.110
1.502	0.774	12.03	0.112
1.525	0.774	13.01	0.100
1.525	0.764	13.10	0.103
1.015	0.733	13.55	0.095
1.807	0.732	13.51	0.091
1.007	0.732	13.08	0.080
2 023	0.725	13.03	0.08
2.023	0.608	14.01	0.077
2.142	0.698	14.10	0.073
2.207	0.67	14.55	0.068
2.599	0.656	14.31	0.064
2.34	0.643	14.08	0.059
2.000	0.043	14.65	0.053
2.040	0.020	15.01	0.05
3.013	0.015	15.18	0.044
2 2 1 6	0.598	13.33	0.041
3.340 2.512	0.38	15.51	0.037
3.515	0.50/	15.08	0.034
J.U/Y 2 Q14	0.535	13.83	0.03
3.840	0.538		

Slug Test Aquifer Model: Confined Solution Method: Bouwer-Rice ln(Re/rw): 4.57

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	5.5E-5	cm/sec
y0	0.95	ft

 $T = K*b = 0.04508 \text{ cm}^2/\text{sec}$



Data Set: I:\09jobs\09E0045\Admin\15-Field-Laboratory Data\AshPond2_Slug_Tests\CPS_OW5_RH.aqt Title: Rising Head Test Date: 11/11/16 Time: 13:53:38

PROJECT INFORMATION

Company: Hanson Professional Services Client: AEG Coffeen Power Station Project: 09E0045/3001 Location: Coffeen, IL Test Date: 9 Jun 2009 Test Well: OW5

AQUIFER DATA

Saturated Thickness: 26.89 ft Anisotropy Ratio (Kz/Kr): 1.

SLUG TEST WELL DATA

Test Well: OW5

X Location: 0. ft Y Location: 0. ft

Initial Displacement: 0.35 ft Static Water Column Height: 26.89 ft Casing Radius: 0.08333 ft Well Radius: 0.08333 ft Well Skin Radius: 0.333 ft Screen Length: 20. ft Total Well Penetration Depth: 30. ft

T'(1) $D'(1)$ $T'(2)$ $T'(1)$ $D'(1)$	
<u>1 ime (min)</u> Displacement (ft) 1 ime (min) Displacement	nent (ft)
0.011 0.327 2.142 0.20	64
0.022 0.327 2.267 0.20	61
0.033 0.327 2.399 0.23	57
0.044 0.324 2.54 0.23	54
0.055 0.324 2.688 0.24	48
0.066 0.324 2.846 0.24	44
0.077 0.324 3.013 0.23	39
0.088 0.323 3.179 0.23	35
0.099 0.323 3.346 0.2	3
0.11 0.323 3.513 0.22	26
0.121 0.321 3.679 0.22	21
0.132 0.321 3.846 0.21	17
0.143 0.321 4.013 0.21	12
0.154 0.321 4.179 0.20)9
0.165 0.321 4.346 0.20)5
0.176 0.321 4.513 0.2	2
0.187 0.32 4.679 0.19	96
0.198 0.32 4.846 0.1	9

Time (min)	Displacement (ft)	Time (min)	Displacement (ft)
0.209	0.32	5.013	0.185
0.22	0.32	5.179	0.181
0.231	0.318	5.346	0.178
0.2427	0.318	5.513	0.173
0.2552	0.318	5.679	0.169
0.2683	0.317	5.846	0.165
0.2823	0.317	6.013	0.162
0.2972	0.317	6.179	0.157
0.3128	0.317	6.346	0.153
0.3295	0.315	6.513	0.147
0.3472	0.315	6.679	0.142
0.3658	0.315	6.846	0.139
0.3857	0.314	7.013	0.135
0.4067	0.312	7.179	0.132
0.4288	0.312	7.346	0.127
0.4523	0.311	7.513	0.123
0.4772	0.311	7.679	0.118
0.5035	0.312	7.846	0.115
0.5315	0.311	8.013	0.112
0.5612	0.311	8.179	0.108
0.5925	0.309	8.346	0.105
0.6257	0.309	8.513	0.099
0.6608	0.308	8.679	0.096
0.6982	0.306	8.846	0.091
0.7377	0.305	9.013	0.088
0.7795	0.305	9.179	0.084
0.8238	0.303	9.346	0.08
0.8708	0.3	9.513	0.077
0.9207	0.3	9.679	0.074
0.9733	0.299	9.846	0.071
1.029	0.297	10.01	0.066
1.088	0.296	10.18	0.062
1.151	0.293	10.35	0.059
1.217	0.291	10.51	0.054
1.288	0.29	10.68	0.05
1.362	0.288	10.85	0.047
1.441	0.285	11.01	0.044
1.525	0.281	11.18	0.041
1.613	0.279	11.35	0.038
1.707	0.275	11.51	0.033
1.807	0.273	11.68	0.03
1.912	0.27	11.85	0.026
2.023	0.267	12.01	0.023

Slug Test Aquifer Model: Unconfined Solution Method: Bouwer-Rice ln(Re/rw): 4.57

VISUAL ESTIMATION RESULTS

Estimated Parameters

Parameter	Estimate	
K	4.5E-5	cm/sec
y0	0.33	ft

 $T = K*b = 0.03688 \text{ cm}^2/\text{sec}$

...

APPENDIX B3

LABORATORY HYDRAULIC CONDUCTIVITY TESTS

HISTORICAL LABORATORY RESULTS



CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE:	7/10/2006
CLIENT:	AEG Coffeen Power Station	BORING #:	SB-7
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #:	17B
SAMPLE DESCRIPTION:	Gray vf. sandy silty clay.	DEPTH (FT):	53.0-53.5
		FILE NAME:	05S3004A

WATER CONTENT OF TRIMMINGS

SPECIMEN WEIGHT (G)	693.83		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.442		TEST	TEST
DIAMETER (IN)	2.766	TARE + WET SOIL (G)	74.53	752.87
AREA (SQ IN)	6.009	TARE + DRY SOIL (G)	60.94	613.65
VOLUME (CU IN)	20.683	TARE (G)	3.68	50.10
WET DENSITY (PCF)	127.80	WATER (G)	13.59	139.22
DRY DENSITY (PCF)	103.28	DRY SOIL (G)	57.26	563.55
WT. DRY SOIL (G)	560.74	WATER CONTENT (%)	23.73	24.70
VOLUME DRY SOIL (CU IN)	12.674			
SP.GR. ASSUMED	2.70			
POROSITY (%)	38.72	STD. MAX. DEN.((LBS/CU.FT.)	112.00
HEIGHT OF HEAD (PSI)	3.70	OPTIMUM MC	DISTURE (%)	14.70
HYDRAULIC GRADIANT	29.8	% C0	OMPACTION	92.22
1/4 PORE VOLUME	32.81	PRESSURE HEA	D (CM H2O)	260.17
		PAN	EL NUMBER	6
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED:	TAP WATER	



CONSTANT HEAD PERMEABILITY TEST

JOB NUMBER: 05S3004A					TEST DATE:	7/10/2006			
CLIENT: AEG Coffeen Power Station					BORING #:	SB-7			
	JOB I	DESCRIPTION:	CCB Manageme	ent Facility			SAMPLE #:	17B	
	SAMPLE I	DESCRIPTION:	Gray vf. sandy s	silty clay.		-	DEPTH (FT):	53.0-53.5	
SPECIMEN HEIGHT (IN) 3.442				HEIGHT	OF HEAD (PSI)	3.70			
		D	IAMETER (IN)	2.766		PRESSURE H	EAD (CM H2O)	260.17	
			AREA (SQ IN)	6.009		PA	NEL NUMBER	6	
		1/4 P	ORE VOLUME	32.81			FILE NAME:	05S3004A	
		-	-	INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	II (CREWER(17)E	GEOMEAN
7/17/2006	10:56:30	7/18/2006	8:03:00	0.20	0.2000	1266.50	1266.50	2.28E-09	2.28E-09
7/18/2006	8:03:00	7/19/2006	8:02:00	0.20	0.4000	1439.00	2705.50	2.01E-09	2.14E-09
7/19/2006	8:02:00	7/19/2006	16:41:00	0.10	0.5000	519.00	3224.50	2.78E-09	2.34E-09
7/19/2006	16:41:00	7/20/2006	7:49:00	0.10	0.6000	908.00	4132.50	1.59E-09	2.12E-09
7/20/2006	7:49:00	7/21/2006	9:00:45	0.20	0.8000	1511.75	5644.25	1.91E-09	2.03E-09
7/31/2006	9:20:35	8/1/2006	9:48:00	0.20	1.0000	1467.42	7111.67	1.97E-09	2.02E-09
8/1/2006	9:48:00	8/4/2006	16:36:15	0.20	1.2000	4728.25	11839.92	6.11E-10	1.38E-09
8/4/2006	16:36:15	8/7/2006	8:56:38	0.30	1.5000	3860.38	15700.30	1.12E-09	1.27E-09
8/7/2006	8:56:38	8/16/2006	7:41:00	0.80	2.3000	12884.37	28584.67	8.97E-10	1.05E-09
8/16/2006	7:41:00	8/23/2006	8:12:52	0.60	2.9000	10111.87	38696.53	8.57E-10	8.52E-10





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: <u>7/7/2006</u>
CLIENT:	AEG Coffeen Power Station	BORING #: SB-9
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: 14
SAMPLE DESCRIPTION:	Gray vf. sandy silt / so. clay (tr. c. sand	DEPTH (FT): 26.5-27.0
	& sm. gravel).	FILE NAME: 05S3004A

WATER CONTENT OF TRIMMINGS

SPECIMEN WEIGHT (G)	438.57		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	1.951		TEST	TEST
DIAMETER (IN)	2.775	TARE + WET SOIL (G)	112.90	56.90
AREA (SQ IN)	6.048	TARE + DRY SOIL (G)	104.63	50.64
VOLUME (CU IN)	11.800	TARE (G)	3.69	3.67
WET DENSITY (PCF)	141.59	WATER (G)	8.27	6.26
DRY DENSITY (PCF)	130.87	DRY SOIL (G)	100.94	46.97
WT. DRY SOIL (G)	405.36	WATER CONTENT (%)	8.19	13.33
VOLUME DRY SOIL (CU IN)	9.162			
SP.GR. ASSUMED	2.70			
POROSITY (%)	22.36	STD. MAX. DEN.(LBS/CU.FT.)	N/A
HEIGHT OF HEAD (PSI)	2.10	OPTIMUM MC	DISTURE (%)	N/A
HYDRAULIC GRADIANT	29.8	% CC	OMPACTION	N/A
1/4 PORE VOLUME	10.81	PRESSURE HEA	D (CM H2O)	147.67
		PAN	EL NUMBER	5
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED:	TAP WATER	



CONSTANT HEAD PERMEABILITY TEST

JOB NUMBER: 05S3004A					TEST DATE:	8/30/2007			
CLIENT: AEG Coffeen Power Station					-	BORING #:	SB-9		
	JOB	DESCRIPTION:	CCB Manageme	ent Facility			SAMPLE #:	14	
	SAMPLE	DESCRIPTION:	Gray vf. sandy s	silt / so. clay (tr. c	. sand	-	DEPTH (FT):	26.5-27.0	
			& sm. gravel).			-			
		SPECIME	N HEIGHT (IN)	1.951		HEIGHT	OF HEAD (PSI)	2.10	
		D	DIAMETER (IN)	2.775		PRESSURE H	EAD (CM H2O)	147.67	
			AREA (SQ IN)	6.048		PA	NEL NUMBER	5	
		1/4 P	ORE VOLUME	10.81			FILE NAME:	05S3004A	
	[1	1	INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	in (Creek) in (Creek)	GEOMEAN
7/17/2006	10:39:00	7/17/2006	11:36:15	15.70	15.7000	57.25	57.25	3.93E-06	3.93E-06
7/17/2006	11:39:00	7/17/2006	13:13:30	22.50	38.2000	94.50	151.75	3.41E-06	3.66E-06
7/17/2006	13:16:30	7/17/2006	14:55:00	23.20	61.4000	98.50	250.25	3.38E-06	3.56E-06
7/17/2006	14:57:30	7/17/2006	16:10:00	17.50	78.9000	72.50	322.75	3.46E-06	3.54E-06
7/18/2006	8:02:00	7/18/2006	9:57:00	24.00	102.9000	115.00	437.75	2.99E-06	3.30E-06
7/18/2006	9:58:40	7/18/2006	12:38:30	24.30	127.2000	159.83	597.58	2.18E-06	2.95E-06
7/18/2006	12:41:00	7/18/2006	15:23:00	23.40	150.6000	162.00	759.58	2.07E-06	2.61E-06
7/18/2006	15:27:00	7/18/2006	16:36:00	10.80	161.4000	69.00	828.58	2.24E-06	2.35E-06
7/19/2006	8:01:00	7/19/2006	10:52:00	23.60	185.0000	171.00	999.58	1.98E-06	2.12E-06
7/19/2006	10:53:30	7/19/2006	13:55:30	24.50	209.5000	182.00	1181.58	1.93E-06	2.05E-06
7/19/2006	13:59:00	7/19/2006	16:39:00	21.70	231.2000	160.00	1341.58	1.94E-06	2.02E-06
7/20/2006	12:57:00	7/20/2006	16:04:00	24.00	255.2000	187.00	1528.58	1.84E-06	1.92E-06




CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: 4/3/2006
CLIENT:	AEG Coffeen Power Station	BORING #: SB-11
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: BAG
SAMPLE DESCRIPTION:	Dk. brn. vf. sandy silty clay.	DEPTH (FT): 4.0-8.0
		FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	349.45		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.998		TEST	TEST
DIAMETER (IN)	1.879	TARE + WET SOIL (G)	219.92	369.93
AREA (SQ IN)	2.773	TARE + DRY SOIL (G)	208.47	305.36
VOLUME (CU IN)	11.086	TARE (G)	138.74	3.72
WET DENSITY (PCF)	120.08	WATER (G)	11.45	64.57
DRY DENSITY (PCF)	103.14	DRY SOIL (G)	69.73	301.64
WT. DRY SOIL (G)	300.16	WATER CONTENT (%)	16.42	21.41
VOLUME DRY SOIL (CU IN)	6.784			
SP.GR. ASSUMED	2.70			
POROSITY (%)	38.81	STD. MAX. DEN.(108.80	
HEIGHT OF HEAD (PSI)	4.30	OPTIMUM MO	ISTURE (%)	16.10
HYDRAULIC GRADIANT	29.8	% CC	MPACTION	94.80
1/4 PORE VOLUME	17.62	PRESSURE HEA	D (CM H2O)	302.36
		PANE	EL NUMBER	7
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED: 7		



JOB NUMBER: 05S3004A						TEST DATE:	4/3/2006		
CLIENT: AEG Coffeen Power Station					-	BORING #:	SB-11		
	JOB I	DESCRIPTION:	CCB Manageme	ent Facility		-	SAMPLE #:	BAG	
							DEPTH (FT):	4.0-8.0	_
SPECIMEN HEIGHT (IN) 3.998 DIAMETER (IN) 1.879 AREA (SQ IN) 2.773 1/4 PORE VOLUME 17.62						HEIGHT (PRESSURE HI PA	OF HEAD (PSI) EAD (CM H2O) NEL NUMBER FILE NAME:	4.30 302.36 7 05S3004A	
			r	INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)		GEOMEAN
5/10/2006	13:14:25	5/11/2006	8:22:00	0.80	0.8000	1147.58	1147.58	2.18E-08	2.18E-08
5/11/2006	8:22:00	5/12/2006	8:11:35	1.90	2.7000	1429.58	2577.17	4.16E-08	3.01E-08
5/12/2006	8:11:35	5/15/2006	8:53:00	3.30	6.0000	4361.42	6938.58	2.37E-08	2.78E-08
5/15/2006	8:53:00	5/16/2006	10:31:25	1.00	7.0000	1538.42	8477.00	2.03E-08	2.57E-08
5/16/2006	10:31:25	5/17/2006	7:53:35	0.90	7.9000	1282.17	9759.17	2.20E-08	2.58E-08
5/17/2006	7:53:35	5/18/2006	16:41:40	1.20	9.1000	1968.08	11727.25	1.91E-08	2.12E-08
5/18/2006	16:41:40	5/22/2006	8:25:20	2.80	11.9000	5263.67	16990.92	1.66E-08	1.94E-08
5/22/2006	8:25:20	5/24/2006	9:06:25	1.30	13.2000	2921.08	19912.00	1.39E-08	1.77E-08
5/24/2006	9:06:25	5/25/2006	9:36:40	0.60	13.8000	1470.25	21382.25	1.28E-08	1.54E-08
5/25/2006	9:36:40	5/26/2006	9:44:45	0.70	14.5000	1448.08	22830.33	1.51E-08	1.45E-08
5/26/2006	9:44:45	5/27/2006	9:31:35	0.50	15.0000	1426.83	24257.17	1.10E-08	1.31E-08
5/27/2006	9:31:35	5/28/2006	9:24:15	0.60	15.6000	1432.67	25689.83	1.31E-08	1.29E-08
5/28/2006	9:24:15	5/29/2006	8:21:20	0.40	16.0000	1377.08	27066.92	9.09E-09	1.19E-08
5/29/2006	8:21:20	5/30/2006	12:56:30	0.70	16.7000	1715.17	28782.08	1.28E-08	1.14E-08
5/30/2006	12:56:30	5/31/2006	14:36:35	0.60	17.3000	1540.08	30322.17	1.22E-08	1.17E-08
5/31/2006	14:36:35	6/1/2006	13:16:30	0.40	17.7000	1359.92	31682.08	9.20E-09	1.07E-08
6/1/2006	13:16:30	6/2/2006	10:24:00	0.60	18.3000	1267.50	32949.58	1.48E-08	1.21E-08





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: 7/7/2006
CLIENT:	AEG Coffeen Power Station	BORING #: SB-12
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: 6B
SAMPLE DESCRIPTION:	Gray vf. sandy silty clay (tr. c. sand).	DEPTH (FT): 45.0-50.0
		FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	459.74		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.419		TEST	TEST
DIAMETER (IN)	2.347	TARE + WET SOIL (G)	65.87	513.41
AREA (SQ IN)	4.326	TARE + DRY SOIL (G)	54.60	427.45
VOLUME (CU IN)	14.792	TARE (G)	3.70	50.03
WET DENSITY (PCF)	118.40	WATER (G)	11.27	85.96
DRY DENSITY (PCF)	96.94	DRY SOIL (G)	50.90	377.42
WT. DRY SOIL (G)	376.40	WATER CONTENT (%)	22.14	22.78
VOLUME DRY SOIL (CU IN)	8.507			
SP.GR. ASSUMED	2.70			
POROSITY (%)	42.49	STD. MAX. DEN.(I	112.00	
HEIGHT OF HEAD (PSI)	3.60	OPTIMUM MO	ISTURE (%)	14.70
HYDRAULIC GRADIANT	29.1	% CO	MPACTION	86.55
1/4 PORE VOLUME	25.75	PRESSURE HEA	D (CM H2O)	253.14
		PANE	L NUMBER	8
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED: 7	TAP WATER	



JOB NUMBER: 05S3004A					_	TEST DATE:	7/7/2006		
		CLIENT:	AEG Coffeen P	ower Station			BORING #:	SB-12	
	JOB I	DESCRIPTION:	CCB Manageme	ent Facility			SAMPLE #:	6B	
	SAMPLE I	DESCRIPTION:	Gray vf. sandy s	silty clay (tr. c. sa	nd).	-	DEPTH (FT):	45.0-50.0	
						-			
		SPECIME	N HEIGHT (IN)	3.419		HEIGHT	OF HEAD (PSI)	3.60	
		D	DIAMETER (IN)	2.347		PRESSURE HI	EAD (CM H2O)	253.14	
			AREA (SQ IN)	4.326		PA	NEL NUMBER	8	
		1/4 P	ORE VOLUME	25.75			FILE NAME:	05S3004A	
						1		1	
				INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	INCREMENTAL	GEOMEAN
7/17/2006	13:26:30	7/18/2006	8:11:00	0.40	0.4000	1124.50	1124.50	7.29E-09	7.29E-09
7/18/2006	8:11:00	7/19/2006	8:09:30	0.30	0.7000	1438.50	2563.00	4.27E-09	5.58E-09
7/19/2006	8:09:30	7/20/2006	7:57:30	0.20	0.9000	1428.00	3991.00	2.87E-09	4.47E-09
7/20/2006	7:57:30	7/21/2006	9:07:45	0.20	1.1000	1510.25	5501.25	2.71E-09	3.95E-09
7/24/2006	13:38:15	7/25/2006	9:05:25	0.80	1.9000	1167.17	6668.42	1.40E-08	4.65E-09
7/25/2006	9:05:25	7/27/2006	13:38:00	1.40	3.3000	3152.58	9821.00	9.10E-09	5.62E-09
7/27/2006	13:38:00	7/28/2006	19:03:30	0.80	4.1000	1765.50	11586.50	9.28E-09	7.53E-09
7/28/2006	19:03:30	7/31/2006	8:43:30	1.60	5.7000	3700.00	15286.50	8.86E-09	1.01E-08
7/31/2006	8:43:30	8/1/2006	9:52:15	0.80	6.5000	1508.75	16795.25	1.09E-08	9.49E-09
8/1/2006	9:52:15	8/4/2006	16:38:10	1.90	8.4000	4725.92	21521.17	8.24E-09	9.26E-09
8/4/2006	16:38:10	8/7/2006	8:58:40	1.50	9.9000	3860.50	25381.67	7.96E-09	8.91E-09
8/7/2006	8:58:40	8/16/2006	7:49:00	4.50	14.4000	12890.33	38272.00	7.15E-09	8.45E-09
8/16/2006	7:49:00	8/23/2006	8:14:00	3.10	17.5000	10105.00	48377.00	6.28E-09	7.37E-09
8/23/2006	8:14:00	8/28/2006	8:02:30	2.10	19.6000	7188.50	55565.50	5.98E-09	6.80E-09





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: <u>7/10/2006</u>
CLIENT:	AEG Coffeen Power Station	BORING #: SB-13
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: 12
SAMPLE DESCRIPTION:	Gray vf. sandy silty clay (tr. c. sand	DEPTH (FT): 25.0-30.0
	& sm. gravel).	FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	483.73		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.259		TEST	TEST
DIAMETER (IN)	2.279	TARE + WET SOIL (G)	125.78	534.14
AREA (SQ IN)	4.079	TARE + DRY SOIL (G)	112.15	475.18
VOLUME (CU IN)	13.294	TARE (G)	3.75	50.14
WET DENSITY (PCF)	138.62	WATER (G)	13.63	58.96
DRY DENSITY (PCF)	123.13	DRY SOIL (G)	108.40	425.04
WT. DRY SOIL (G)	429.70	WATER CONTENT (%)	12.57	13.87
VOLUME DRY SOIL (CU IN)	9.712			
SP.GR. ASSUMED	2.70			
POROSITY (%)	26.95	STD. MAX. DEN.(LBS/CU.FT.)		N/A
HEIGHT OF HEAD (PSI)	3.50	OPTIMUM MO	DISTURE (%)	N/A
HYDRAULIC GRADIANT	29.7	% CC	OMPACTION	
1/4 PORE VOLUME	14.68	PRESSURE HEA	D (CM H2O)	246.11
		PAN	EL NUMBER	1
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED:	TAP WATER	
-		-		



JOB NUMBER: 05S3004A						TEST DATE:	7/10/2006		
CLIENT: AEG Coffeen Power Station					•	BORING #:	SB-13		
	JOB	DESCRIPTION:	CCB Manageme	ent Facility		-	SAMPLE #:	12	
	SAMPLE	DESCRIPTION:	Gray vf. sandy s	ilty clay (tr. c. sa	nd		DEPTH (FT):	25.0-30.0	
			& sm. gravel).						
		SPECIME	N HEIGHT (IN)	3.259		HEIGHT	OF HEAD (PSI)	3.50	
		D	DIAMETER (IN)	2.279		PRESSURE H	EAD (CM H2O)	246.11	
			AREA (SQ IN)	4.079		PA	NEL NUMBER	1	
		1/4 P	ORE VOLUME	14.68			FILE NAME:	05S3004A	
								-	
				INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABL	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	II (CREWER(17)E	GEOMEAN
7/19/2006	8:38:00	7/20/2006	8:11:10	6.20	6.2000	1413.17	1413.17	9.35E-08	9.35E-08
7/20/2006	8:11:10	7/21/2006	7:58:00	4.20	10.4000	1426.83	2840.00	6.27E-08	7.65E-08
7/21/2006	7:58:00	7/21/2006	16:13:00	1.20	11.6000	495.00	3335.00	5.16E-08	6.71E-08
7/24/2006	13:37:50	7/25/2006	9:06:10	0.90	12.5000	1168.33	4503.33	1.64E-08	4.72E-08
7/25/2006	9:06:10	7/27/2006	13:38:40	1.30	13.8000	3152.50	7655.83	8.78E-09	2.61E-08
7/27/2006	13:38:40	7/28/2006	19:01:35	0.70	14.5000	1762.92	9418.75	8.46E-09	1.58E-08
7/28/2006	19:01:35	7/31/2006	8:44:10	1.40	15.9000	3702.58	13121.33	8.05E-09	9.95E-09
7/31/2006	8:44:10	8/1/2006	9:53:15	0.60	16.5000	1509.08	14630.42	8.47E-09	8.44E-09
8/1/2006	9:53:15	8/4/2006	16:38:45	1.60	18.1000	4725.50	19355.92	7.21E-09	8.03E-09
8/4/2006	16:38:45	8/7/2006	8:59:19	1.20	19.3000	3860.57	23216.48	6.62E-09	7.55E-09
8/7/2006	8:59:19	8/16/2006	7:49:30	3.80	23.1000	12890.18	36106.67	6.28E-09	7.10E-09





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: 5/5/2006
CLIENT:	AEG Coffeen Power Station	BORING #: SB-14
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: BAG
SAMPLE DESCRIPTION:	Brn. vf. sandy silty clay (tr. sm. gravel).	DEPTH (FT):5.0-10.0
		FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	356.11		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.998		TEST	TEST
DIAMETER (IN)	1.879	TARE + WET SOIL (G)	214.34	378.18
AREA (SQ IN)	2.773	TARE + DRY SOIL (G)	204.66	316.21
VOLUME (CU IN)	11.086	TARE (G)	138.73	3.78
WET DENSITY (PCF)	122.37	WATER (G)	9.68	61.97
DRY DENSITY (PCF)	106.70	DRY SOIL (G)	65.93	312.43
WT. DRY SOIL (G)	310.52	WATER CONTENT (%)	14.68	19.83
VOLUME DRY SOIL (CU IN)	7.018			
SP.GR. ASSUMED	2.70			
POROSITY (%)	36.69	STD. MAX. DEN.(LBS/CU.FT.)	111.80
HEIGHT OF HEAD (PSI)	4.30	OPTIMUM MO	ISTURE (%)	14.90
HYDRAULIC GRADIANT	29.8	% CC	MPACTION	95.44
1/4 PORE VOLUME	16.67	PRESSURE HEA	D (CM H2O)	302.36
		PANE	EL NUMBER	3
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED: 7	TAP WATER	



JOB NUMBER: 05S3004A					_	TEST DATE:	5/5/2006		
CLIENT: AEG Coffeen Power Station						-	BORING #:	SB-14	
	JOB DESCRIPTION: CCB Management Facility					-	SAMPLE #:	BAG	
							DEPTH (FT):	5.0-10.0	
		SPECIME	N HEIGHT (IN)	3,998		HEIGHT	OF HEAD (PSI)	4.30	
		D	DIAMETER (IN)	1.879		PRESSURE H	EAD (CM H2O)	302.36	
			AREA (SQ IN)	2.773		PA	NEL NUMBER	3	•
		1/4 P	ORE VOLUME	16.67			FILE NAME:	05S3004A	
				INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABII	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	ITCREMENTAL	GEOMEAN
5/11/2006	13:24:00	5/12/2006	8:13:00	2.00	2.0000	1129.00	1129.00	5.54E-08	5.54E-08
5/12/2006	8:13:00	5/15/2006	8:54:20	6.30	8.3000	4361.33	5490.33	4.52E-08	5.01E-08
5/15/2006	8:54:20	5/16/2006	10:33:00	1.90	10.2000	1538.67	7029.00	3.86E-08	4.59E-08
5/16/2006	10:33:00	5/17/2006	7:54:30	1.60	11.8000	1281.50	8310.50	3.91E-08	4.41E-08
5/17/2006	7:54:30	5/18/2006	16:42:10	2.00	13.8000	1967.67	10278.17	3.18E-08	3.84E-08
5/18/2006	16:42:10	5/22/2006	8:26:15	5.00	18.8000	5264.08	15542.25	2.97E-08	3.46E-08
5/22/2006	8:27:45	5/24/2006	9:07:00	2.30	21.1000	2919.25	18461.50	2.47E-08	3.09E-08
5/24/2006	9:07:00	5/25/2006	9:37:25	1.30	22.4000	1470.42	19931.92	2.77E-08	2.83E-08
5/25/2006	9:37:25	5/26/2006	9:45:15	1.20	23.6000	1447.83	21379.75	2.59E-08	2.69E-08
5/26/2006	9:45:15	5/27/2006	9:30:50	1.20	24.8000	1425.58	22805.33	2.63E-08	2.61E-08
5/27/2006	9:30:50	5/28/2006	9:22:30	1.20	26.0000	1431.67	24237.00	2.62E-08	2.65E-08
5/28/2006	9:22:30	5/29/2006	8:22:00	1.30	27.3000	1379.50	25616.50	2.95E-08	2.70E-08
5/29/2006	8:22:00	5/30/2006	13:01:00	1.40	28.7000	1719.00	27335.50	2.55E-08	2.68E-08
5/30/2006	13:01:00	5/31/2006	14:38:00	1.20	29.9000	1537.00	28872.50	2.44E-08	2.63E-08
5/31/2006	14:38:00	6/1/2006	13:19:00	1.00	30.9000	1361.00	30233.50	2.30E-08	2.55E-08
6/1/2006	13:19:00	6/2/2006	10:27:00	1.00	31.9000	1268.00	31501.50	2.47E-08	2.44E-08







CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: <u>7/7/2006</u>
CLIENT:	AEG Coffeen Power Station	BORING #: SB-16
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: 18A
SAMPLE DESCRIPTION:	Brn. & gray f. sandy silt (tr. clay) / so.	DEPTH (FT): 37.0-37.5
	c. sand & sm. gravel.	FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	855.13		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.811		TEST	TEST
DIAMETER (IN)	2.735	TARE + WET SOIL (G)	128.66	124.70
AREA (SQ IN)	5.875	TARE + DRY SOIL (G)	117.35	113.65
VOLUME (CU IN)	22.389	TARE (G)	3.72	3.71
WET DENSITY (PCF)	145.50	WATER (G)	11.31	11.05
DRY DENSITY (PCF)	132.33	DRY SOIL (G)	113.63	109.94
WT. DRY SOIL (G)	777.72	WATER CONTENT (%)	9.95	10.05
VOLUME DRY SOIL (CU IN)	17.578			
SP.GR. ASSUMED	2.70			
POROSITY (%)	21.49	STD. MAX. DEN.(LBS/CU.FT.)	112.00
HEIGHT OF HEAD (PSI)	4.10	OPTIMUM MC	DISTURE (%)	14.70
HYDRAULIC GRADIANT	29.8	% CC	OMPACTION	118.15
1/4 PORE VOLUME	19.71	PRESSURE HEA	D (CM H2O)	288.30
		PAN	EL NUMBER	3
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED:	TAP WATER	



JOB NUMBER: 05S3004A				_	TEST DATE:	7/7/2006	_		
CLIENT: AEG Coffeen Power Station						BORING #:	SB-16		
	JOB	DESCRIPTION:	CCB Manageme	ent Facility			SAMPLE #:	18A	
	SAMPLE	DESCRIPTION:	Brn. & gray f. s	andy silt (tr. clay)	/ so.		DEPTH (FT):	37.0-37.5	
			c. sand & sm. gr	ravel.		-			
		SPECIME	N HEIGHT (IN)	3.811		HEIGHT	OF HEAD (PSI)	4.10	
		D	IAMETER (IN)	2.735		PRESSURE H	EAD (CM H2O)	288.30	
			AREA (SQ IN)	5.875		PA	NEL NUMBER	3	
		1/4 P	ORE VOLUME	19.71			FILE NAME:	05S3004A	
				INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABII	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	INCREMENTAL	GEOMEAN
7/17/2006	11:36:30	7/17/2006	13:01:30	24.30	24.3000	85.00	85.00	4.22E-06	4.22E-06
7/17/2006	13:04:30	7/17/2006	14:52:00	22.60	46.9000	107.50	192.50	3.10E-06	3.62E-06
7/17/2006	14:54:00	7/17/2006	16:12:15	15.60	62.5000	78.25	270.75	2.94E-06	3.38E-06
7/18/2006	8:08:00	7/18/2006	10:14:00	23.80	86.3000	126.00	396.75	2.79E-06	3.22E-06
7/18/2006	10:21:30	7/18/2006	13:47:00	23.10	109.4000	205.50	602.25	1.66E-06	2.55E-06
7/18/2006	13:48:30	7/18/2006	16:34:00	18.50	127.9000	165.50	767.75	1.65E-06	2.18E-06
7/19/2006	8:08:00	7/19/2006	11:38:00	22.60	150.5000	210.00	977.75	1.59E-06	1.87E-06
7/19/2006	11:40:00	7/19/2006	15:28:30	23.80	174.3000	228.50	1206.25	1.54E-06	1.61E-06
7/19/2006	15:30:30	7/19/2006	16:43:30	8.10	182.4000	73.00	1279.25	1.64E-06	1.60E-06
7/20/2006	7:55:00	7/20/2006	11:32:30	23.10	205.5000	217.50	1496.75	1.57E-06	1.58E-06
7/20/2006	11:35:30	7/20/2006	15:23:50	22.90	228.4000	228.33	1725.08	1.48E-06	1.56E-06
7/20/2006	15:27:00	7/20/2006	16:26:00	6.30	234.7000	59.00	1784.08	1.58E-06	1.56E-06





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: <u>7/10/2006</u>
CLIENT:	AEG Coffeen Power Station	BORING #: SB-16
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #:
SAMPLE DESCRIPTION:	Yel. brn. vff. sandy silty clay / so. c.	DEPTH (FT): 78.0-78.5
	sand & sm. gravel & cobbles.	FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	203.47		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	1.873		TEST	TEST
DIAMETER (IN)	1.949	TARE + WET SOIL (G)	159.86	258.48
AREA (SQ IN)	2.983	TARE + DRY SOIL (G)	140.04	227.76
VOLUME (CU IN)	5.588	TARE (G)	3.71	50.15
WET DENSITY (PCF)	138.71	WATER (G)	19.82	30.72
DRY DENSITY (PCF)	121.11	DRY SOIL (G)	136.33	177.61
WT. DRY SOIL (G)	177.64	WATER CONTENT (%)	14.54	17.30
VOLUME DRY SOIL (CU IN)	4.015			
SP.GR. ASSUMED	2.70			
POROSITY (%)	28.15	STD. MAX. DEN.(LBS/CU.FT.)	112.00
HEIGHT OF HEAD (PSI)	2.00	OPTIMUM MC	DISTURE (%)	14.70
HYDRAULIC GRADIANT	29.6	% CC	OMPACTION	108.13
1/4 PORE VOLUME	6.44	PRESSURE HEA	D (CM H2O)	140.63
		PAN	EL NUMBER	4
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED:	TAP WATER	



JOB NUMBER: 05S3004A					TEST DATE:	7/10/2006			
CLIENT: AEG Coffeen Power Station						BORING #:	SB-16		
	JOB	DESCRIPTION:	CCB Manageme	ent Facility			SAMPLE #:	32A	
	SAMPLE	DESCRIPTION:	Yel. brn. vff. s	andy silty clay / s	0. C.	-	DEPTH (FT):	78.0-78.5	
			sand & sm. grav	el & cobbles.		<u>.</u>			
		SPECIME	N HEIGHT (IN)	1.873		HEIGHT	OF HEAD (PSI)	2.00	
		Ľ	DIAMETER (IN)	1.949		PRESSURE HI	EAD (CM H2O)	140.63	
			AREA (SQ IN)	2.983		PA	NEL NUMBER	4	
		1/4 P	ORE VOLUME	6.444			FILE NAME:	05S3004A	
						1			
				INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	INCREMENTAL	GEOMEAN
7/17/2006	11:33:00	7/18/2006	8:09:00	0.80	0.8000	1236.00	1236.00	1.90E-08	1.90E-08
7/18/2006	8:09:00	7/19/2006	8:09:00	0.70	1.5000	1440.00	2676.00	1.42E-08	1.64E-08
7/19/2006	8:09:00	7/20/2006	7:56:00	0.80	2.3000	1427.00	4103.00	1.64E-08	1.64E-08
7/20/2006	7:56:00	7/21/2006	9:06:10	0.60	2.9000	1510.17	5613.17	1.16E-08	1.51E-08
7/24/2006	13:37:15	7/25/2006	9:04:50	0.70	3.6000	1167.58	6780.75	1.76E-08	1.48E-08
7/25/2006	9:04:50	7/27/2006	13:37:20	1.80	5.4000	3152.50	9933.25	1.67E-08	1.54E-08
7/27/2006	13:37:20	7/28/2006	19:03:10	1.00	6.4000	1765.83	11699.08	1.66E-08	1.54E-08
7/28/2006	19:03:10	7/31/2006	8:43:10	2.00	8.4000	3700.00	15399.08	1.58E-08	1.67E-08
7/31/2006	8:43:10	8/1/2006	9:51:30	0.80	9.2000	1508.33	16907.42	1.55E-08	1.62E-08
8/1/2006	9:51:30	8/4/2006	16:37:45	2.20	11.4000	4726.25	21633.67	1.36E-08	1.54E-08
8/4/2006	16:37:45	8/7/2006	8:58:05	1.80	13.2000	3860.33	25494.00	1.37E-08	1.46E-08
8/7/2006	8:58:05	8/16/2006	7:45:30	5.10	18.3000	12887.42	38381.42	1.16E-08	1.35E-08
8/16/2006	7:45:30	8/23/2006	8:13:36	4.60	22.9000	10108.10	48489.52	1.33E-08	1.30E-08





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE: <u>7/7/2006</u>
CLIENT:	AEG Coffeen Power Station	BORING #: SB-18
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #: 12
SAMPLE DESCRIPTION:	Gray f. sandy silt / so. c. sand	DEPTH (FT): 22.5-23.0
	& sm. gravel.	FILE NAME: 05S3004A

SPECIMEN WEIGHT (G)	733.77		BEFORE	AFTER
SPECIMEN HEIGHT (IN)	3.163		TEST	TEST
DIAMETER (IN)	2.746	TARE + WET SOIL (G)	87.91	784.05
AREA (SQ IN)	5.922	TARE + DRY SOIL (G)	82.64	736.02
VOLUME (CU IN)	18.732	TARE (G)	3.73	49.33
WET DENSITY (PCF)	149.22	WATER (G)	3.69	48.03
DRY DENSITY (PCF)	142.56	DRY SOIL (G)	78.91	686.69
WT. DRY SOIL (G)	700.99	WATER CONTENT (%)	4.68	6.99
VOLUME DRY SOIL (CU IN)	15.843			
SP.GR. ASSUMED	2.70			
POROSITY (%)	15.42	STD. MAX. DEN.(112.00	
HEIGHT OF HEAD (PSI)	3.40	OPTIMUM MC	14.70	
HYDRAULIC GRADIANT	29.8	% CC	MPACTION	127.28
1/4 PORE VOLUME	11.83	PRESSURE HEA	D (CM H2O)	239.08
		PANI	EL NUMBER	7
TEST METHOD USED:	IEPA ASTM D5084	PERMEANT USED: _		



JOB NUMBER: 05S3004A				_	TEST DATE:	7/7/2006			
		CLIENT:	AEG Coffeen P	ower Station		-	BORING #:	SB-18	
	JOB I	DESCRIPTION:	CCB Manageme	ent Facility		-	SAMPLE #:	12	
	SAMPLE I	DESCRIPTION:	Gray f. sandy si	lt / so. c. sand		_	DEPTH (FT):	22.5-23.0	
			& sm. gravel.			-			
		SPECIME	N HEIGHT (IN)	3.163		HEIGHT	OF HEAD (PSI)	3.40	
		D	IAMETER (IN)	2.746		PRESSURE HI	EAD (CM H2O)	239.08	
			AREA (SQ IN)	5.922		PA	NEL NUMBER	7	
		1/4 P	ORE VOLUME	11.83			FILE NAME:	05S3004A	
				INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)		GEOMEAN
7/17/2006	11:18:00	7/18/2006	8:05:00	8.50	8.5000	1247.00	1247.00	9.99E-08	9.99E-08
7/18/2006	8:05:00	7/19/2006	8:03:00	7.20	15.7000	1438.00	2685.00	7.34E-08	8.56E-08
7/19/2006	8:03:00	7/20/2006	7:50:00	6.20	21.9000	1427.00	4112.00	6.37E-08	7.76E-08
7/20/2006	7:50:00	7/20/2006	11:29:30	0.90	22.8000	219.50	4331.50	6.01E-08	7.28E-08
7/20/2006	11:31:30	7/20/2006	16:07:00	1.40	24.2000	275.50	4607.00	7.45E-08	6.76E-08
7/20/2006	16:07:00	7/21/2006	9:03:20	4.60	28.8000	1016.33	5623.33	6.63E-08	6.59E-08
7/24/2006	13:35:30	7/25/2006	9:04:20	1.30	30.1000	1168.83	6792.17	1.63E-08	4.69E-08
7/25/2006	9:04:20	7/27/2006	13:36:50	1.40	31.5000	3152.50	9944.67	6.51E-09	2.69E-08
7/27/2006	13:36:50	7/28/2006	19:02:50	1.20	32.7000	1766.00	11710.67	9.96E-09	1.63E-08
7/28/2006	19:02:50	7/31/2006	8:42:30	2.50	35.2000	3699.67	15410.33	9.91E-09	1.01E-08
7/31/2006	8:42:30	8/1/2006	9:50:45	1.00	36.2000	1508.25	16918.58	9.72E-09	8.89E-09
8/1/2006	9:50:45	8/4/2006	16:37:20	2.90	39.1000	4726.58	21645.17	8.99E-09	9.64E-09
8/4/2006	16:37:20	8/7/2006	8:57:34	2.30	41.4000	3860.23	25505.40	8.73E-09	9.33E-09
8/7/2006	8:57:34	8/16/2006	7:43:30	6.80	48.2000	12885.93	38391.33	7.74E-09	8.77E-09





CONSTANT HEAD PERMEABILITY TEST

ASTM D5084

JOB NUMBER:	05S3004A	TEST DATE:	7/10/2006
CLIENT:	AEG Coffeen Power Station	BORING #:	SB-19
JOB DESCRIPTION:	CCB Management Facility	SAMPLE #:	18
SAMPLE DESCRIPTION:	Brn. & gray vf. sandy silty clay.	DEPTH (FT):	55.0-60.0
		FILE NAME:	05S3004A

TEST 491.35 406.17
491.35 406.17
406.17
49.88
85.18
356.29
23.91
N/A
N/A
274.24
2



JOB NUMBER: 05S3004A				_	TEST DATE:	7/10/2006			
CLIENT: AEG Coffeen Power Station					-	BORING #:	SB-19		
	JOB	DESCRIPTION:	CCB Manageme	ent Facility		-	SAMPLE #:	18	
	SAMPLE	DESCRIPTION:	Brn. & gray vf.	sandy silty clay.		<u>.</u>	DEPTH (FT):	55.0-60.0	
						-			
		SPECIME	N HEIGHT (IN)	3.586		HEIGHT	OF HEAD (PSI)	3.90	
		D	DIAMETER (IN)	2.165		PRESSURE HI	EAD (CM H2O)	274.24	
			AREA (SQ IN)	3.681		PA	NEL NUMBER	2	1
		1/4 P	ORE VOLUME	21.142			FILE NAME:	05S3004A	1
	1	1		INCREMENT.	TOTAL	INCREMENT.	TOTAL	PERMEABI	LITY (cm/sec)
START	START	STOP	STOP	FLOW	FLOW	TIME	TIME	INCREMENTAL	MOVING AVG.
DATE	TIME	DATE	TIME	(CC)	(CC)	(MIN)	(MIN)	in (Creek) in (Creek)	GEOMEAN
7/17/2006	13:19:45	7/18/2006	8:13:00	0.70	0.7000	1133.25	1133.25	1.44E-08	1.44E-08
7/18/2006	8:13:00	7/19/2006	8:12:30	0.30	1.0000	1439.50	2572.75	4.86E-09	8.36E-09
7/19/2006	8:12:30	7/20/2006	7:59:00	0.20	1.2000	1426.50	3999.25	3.27E-09	6.11E-09
7/20/2006	7:59:00	7/21/2006	9:12:50	0.20	1.4000	1513.83	5513.08	3.08E-09	5.15E-09
7/25/2006	9:06:45	7/27/2006	13:39:04	0.40	1.8000	3152.32	8665.40	2.96E-09	3.47E-09
7/27/2006	13:39:04	7/28/2006	19:02:15	0.20	2.0000	1763.18	10428.58	2.64E-09	2.98E-09
7/28/2006	19:02:15	7/31/2006	8:44:30	0.60	2.6000	3702.25	14130.83	3.78E-09	3.09E-09
7/31/2006	8:44:30	8/1/2006	9:54:00	0.20	2.8000	1509.50	15640.33	3.09E-09	3.09E-09
8/1/2006	9:54:00	8/4/2006	16:39:20	0.80	3.6000	4725.33	20365.67	3.95E-09	3.32E-09
8/4/2006	16:39:20	8/7/2006	8:59:50	0.60	4.2000	3860.50	24226.17	3.62E-09	3.59E-09
8/7/2006	8:59:50	8/16/2006	7:51:30	2.00	6.2000	12891.67	37117.83	3.62E-09	3.55E-09
8/16/2006	7:51:30	8/23/2006	8:15:00	1.50	7.7000	10103.50	47221.33	3.46E-09	3.66E-09
8/23/2006	8:15:00	8/28/2006	8:03:30	0.90	8.6000	7188.50	54409.83	2.92E-09	3.39E-09



AECOM 2015 LABORATORY RESULTS



(MEA	SUREME	<u>ENT OF HYD</u> U	RAULIC COND	UCTIVITY	OF SATU	JRATED POF METER	ROUS MATE	RIALS			
			<u>ASTM</u>	D 5084 - 03 FLUID	METHOD C TE	<u>ST WITH I</u> P WATER	NCREAS	NG TAILWA 05 N CaSO4	TER LEVEL				
PROJE	CT NAME:	DYNEGY COFFEEN, IL	LINOIS					PROJE	CT NUMBER: DATE:	151511 7/30/20	22 15		
SAMPLI SAMPLI	E ID: E DESCR.:	COF-B002 LEAN CLAY V	S4 8. WITH SAN	5 - 10.5 feet ID, GRAY WI	TH YELLOWISH	BROWN		PANEL IDEN BURETT	ITIFICATION: BURE E INCREMEN	Ler ETTE AREA: IT LENGTH:	iexa Perm E 0.312 1.000	Board cm^2 cm	
		IN	ITIAL					VO	ADDITION	ICREMENT: IAL DATA	0.312	cm^3	
MOISTI W&T D&T T,g MOIS URE,	<u>URE%</u> , g 143.03 , g 126.58 33.92 T- % 17.8	DENSITY WET WT, g DIA, in HT, in AREA DENSITY: DENSITY:	495.1 2.861 2.182 134.5 11 4.2	7.27 5.54 41.48 PCF WET PCF DRY	cm cm cm^2		SPECI SPECI F SAT	FIC GRAVITY FIC GRAVITY POROSITY, % URATION, % VOID RATIO	2.70 ASSUMED 32.3 100.7 0.48	RECOM PRO OF COMPA OVER OP	IPACTED?: CTOR, pcf: TIMUM, %: ACTION, %: TIMUM, %:	NO NA NA NA	
				101.0						100.0	nai		
	ATION:		DRESS.	104.0	psi		100.0			100.0	nei		
TEST:			<u>r NL00</u>	104.0	psi	(12.	100.0	BIAS PRESSU	IRE (=H1-H2)	0.0	psi		
		ELAPSED	DELTA			<u>OUT</u>	<u>IN</u>						_
H1 VALU	H2 IE VALUE	<u>TIME,</u> min	<u>H</u> cm	<u>Ln</u> <u>H1/H2</u>	<u>HYD CON</u> k, cm/sec	FLOW cm^3	<u>FLOW</u> cm^3	<u>OUT/IN</u> <u>RATIO</u>	<u>HYD</u> <u>GRAD</u>	<u>% FROM</u> <u>MEAN k</u>	<u>TEMP.:</u> <u>C</u>	<u>TEMI</u> CORF	<u>P.</u> <u>R.:</u>
7.4 7.5 7.6 7.9	67.7 67.6 67.5 67.2	0.00 281.00 467.00 1292.00	60.3 60.1 59.9 59.3	0.003322 0.003333 0.010067	3.83E-09 5.76E-09 3.93E-09	0.03 0.03 0.09	0.03 0.03 0.09	1.00 1.00 1.00	10.8 10.8 10.7	7 39 5	23.0 23.3 23.2	0.93 0.92 0.92	1 5 7
8.0	67.1	1651.00	59.1	0.003378	3.01E-09	0.03	0.03	1.00	10.7	27	23.6	0.91	8
		HYL GF	DRAULIC RADIENT	1.0E-04 TC 1.0E-05 TC 1.0E-06 TC less than	0 1.0E-05 0 1.0E-06 0 1.0E-07 1.0E-07	5 10 20 30		<1.25	MAX HYDRAULIC GRADIENT ALLOWED	> 1.0E-8 OR % < 50 AT < 1.0E-8]	
	1.0E-03											3	
	1.0E-04												
, cm/sec	1.0E-05												
VUCTIVITY	1.0E-06												
ULIC CON	1.0E-07												
НҮДКА	1.0E-08								•				
	1.0E-09 0.0	0 200.	00	400.00	600.00 8	00.00	1000.00	1200.00) 1400.0	0 1600	.00 18	00.00	
L							20			- <u></u>			
		ATA\00 Projects in P	rogress\2015	Projects in Progres	ss\15151122 ah Data\{1	5151122 Parma	20 ability D5084-/	C COF-R002-94-8 5	Jler	гас	on		
<u> </u>	n.iowicAb_C		108100012010		LOU DAID I LE LOU DAID	- IVI IEE FOIIR		DUV2-04-0.0	······································				

(MEA	SUREME	ENT OF HYD	RAULIC CONE	DUCTIVITY	OF SATL PERMEA	JRATED POR METER	OUS MATEI	RIALS		
			<u>ASTM</u>	<u>D 5084 - 03</u> FLUID	METHOD C TE : DEAIRED TA	ST WITH I P WATER	NCREAS	NG TAILWAT 05 N CaSO4	ER LEVEL			
PROJECT	NAME:	DYNEGY COFFEEN, IL	LINOIS					PROJEC	T NUMBER: DATE:	151511 9/9/201	22 5	
SAMPLE ID: SAMPLE DESCR.:		COF-B003 S9 30.0 - 30.0 feet SILTY SAND, DARK YELLOWISH BROWN WITH GRAYISH BROWN						PANEL IDEN BURETTI	TIFICATION: BURE E INCREMEN	Ler ETTE AREA: IT LENGTH:	exa Perm E 0.312 1.000	Joard cm^2 cm
		IN	ITIAL							IAL DATA	0.312	cm ² 3
MOISTUR W & T, g D & T, g T, g MOIST- URE, %	RE% 170.46 155.96 32.90 11.8	DENSITY WET WT, g DIA, in HT, in AREA DENSITY: DENSITY:	482.1 2.858 2.022 141.6 126.7	7.26 5.14 41.39 PCF WET PCF DRY	cm cm cm^2		SPECI SPECI F SAT	FIC GRAVITY: FIC GRAVITY: POROSITY, %: URATION, %: VOID RATIO:	2.70 ASSUMED 24.9 96.2 0.33	RECOM PRC OF COMPA OVER OF	IPACTED?: CTOR, pcf: TIMUM, %: ACTION, %: TIMUM, %:	NO NA NA NA
SATURAT	FION:	LATERAL	PRESS.:	104.0	psi	BA	CK PRES	SURE (=UPPE	R=LOWER):	100.0	psi	
DURING		LATERAL	PRESS.:	104.0	psi	H2:	100.0	psi	H1:	100.0	psi	
TEST:								BIAS PRESSURE (=H1-H2)		0.0 psi		
<u>H1</u> VALUE	<u>H2</u> VALUE	ELAPSED <u>TIME,</u> 	DELTA H cm	<u>Ln</u> <u>H1/H2</u>	<u>HYD CON</u> <u>k, cm/sec</u>	OUT FLOW cm^3	IN FLOW cm^3	<u>OUT/IN</u> RATIO	<u>HYD</u> <u>GRAD</u>	<u>% FROM</u> <u>MEAN k</u>	<u>TEMP.:</u> <u>C</u>	<u>TEMP.</u> CORR.:
11.8 12.1 12.3 12.5 12.7	64.6 64.3 64.1 63.9 63.7	0.00 13.00 24.00 35.00 47.00	52.8 52.2 51.8 51.4 51.0	0.011429 0.007692 0.007752 0.007813	2.63E-07 2.09E-07 2.10E-07 1.94E-07	0.09 0.06 0.06 0.06	0.09 0.06 0.06 0.06	1.00 1.00 1.00 1.00	10.2 10.1 10.0 9.9	20 5 4 11	23.3 23.3 23.3 23.3 23.3	0.925 0.925 0.925 0.925
	1.05.03	HYE GF	DRAULIC RADIENT	1.0E-04 TO 1.0E-05 TO 1.0E-06 TO less than	1.0E-05 1.0E-06 1.0E-07 1.0E-07	5 10 20 30		RATIO <1.25	MAX HYDRAULIC GRADIENT ALLOWED	> 1.0E-8 OR % < 50 AT < 1.0E-8]
	1.02-03											
	1.0E-04											
Υ, cm/sec	1.0E-05											
DUCTIVIT	1.0E-06											
nric con	1.0E-07											
НҮDRA	1.0E-08 -											
	1.0E-09	0 5.00	1(D.00 15	.00 20.00) 25.0	00 3	0.00 35	00 40	00 45	.00 5	∃ 0.00
	N-10441 AD D	14T4100 Projecto in O	1001000\0045	Projects in Prosent	s)15151122 sh Dor-V	TIME, minute	estility DE024	C COE-B003 50 30 5	Ter	rac	on	,
<u> </u>				. rejusts in ritugles	LET CONTRACTOR			2 23, 2000-08-00.0				

		<u>ME</u> /	ASUREMI ASTM	<u>ENT OF HYD</u> <u>U</u> D 5084 - 03	RAULIC CON SING A FLEXI METHOD C TI	DUCTIVITY BLE WALL EST WITH I	OF SATU PERMEA	URATED POI	ROUS MATE	RIALS		Ň
PROJE	ECT NAME:	DYNEGY		FLUIC	: DEAIRED TA	P WATER	WITH 0.0	05 N CaSO4 PROJE	CT NUMBER:	151511	22	
LOCAT SAMPL SAMPL	LE ID: LE DESCR.:	COFFEEN, II COF-B004 SANDY FAT	S7 20 CLAY, RE	3.5.0 - 25.5 fe DDISH BRO\	et WN WITH DARK	GRAY		PANEL IDEN BURETT	DATE: ITIFICATION: BURE E INCREMEN	9/4/201 Ler ETTE AREA: IT LENGTH: ICREMENT:	5 nexa Perm E 0.312 1.000 0.312	Board cm^2 cm cm^3
MOIST W & 1 D & T T, g MOIS URE,	<u>FURE%</u> T, g 140.98 T, g 122.69 g 33.88 ST- , % 20.6	II <u>DENSITY</u> WET WT, g DIA, in HT, in AREA DENSITY: DENSITY:	NITIAL 438.0 2.857 2.010 129.5 107.4	7.26 5.11 41.36 PCF WET PCF DRY	cm cm cm^2		SPECI SPECI F SAT	FIC GRAVITY FIC GRAVITY POROSITY, % FURATION, % VOID RATIO	ADDITION : 2.70 : ASSUMED : 36.3 : 97.6 : 0.57	IAL DATA RECOM PRC OF COMPA OVER OF	IPACTED?: CTOR, pcf: TIMUM, %: ACTION, %: TIMUM, %:	NO NA NA NA NA
SATUF	RATION:	LATERAL	PRESS.:	104.0	psi	BA	CK PRES	SURE (=UPPI	ER=LOWER):	100.0	psi	
DURIN	IG	LATERAL	PRESS.:	104.0	psi	H2:	100.0	psi	H1:	100.0	psi	
IEST:								BIAS PRESSL	JRE (=H1-H2)	0.0	psi	
<u>H1</u> VALU	<u>H2</u> <u>JE VALUE</u>	ELAPSED TIME, 	DELTA H cm	<u>Ln</u> <u>H1/H2</u>	<u>HYD CON</u> <u>k, cm/sec</u>	OUT FLOW cm^3	IN FLOW cm^3	<u>OUT/IN</u> RATIO	<u>HYD</u> <u>GRAD</u>	<u>% FROM</u> MEAN k	<u>TEMP.:</u> <u>C</u>	<u>TEMP.</u> <u>CORR.:</u>
16.0 16.4 16.8 17.2	0 58.8 4 58.4 8 58.0 2 57.6	11.00 22.00 33.00 45.00	42.8 42.0 41.2 40.4	0.018519 0.018868 0.019231 0.019608	5.00E-07 5.10E-07 5.19E-07 4.85E-07	0.12 0.12 0.12 0.12 0.12	0.12 0.12 0.12 0.12 0.12	1.00 1.00 1.00 1.00	8.4 8.2 8.1 7.9	1 1 3 4	23.3 23.3 23.3 23.3 23.3	0.925 0.925 0.925 0.925
Г		M HYI Gl	IAXIMUM DRAULIC RADIENT	1.0E-03 TO 1.0E-04 TO 1.0E-05 TO 1.0E-06 TO less than	1.0E-04 1.0E-05 1.0E-06 1.0E-07 1.0E-07	2 5 10 20 30		0.75< RATIO <1.25	20 MAX HYDRAULIC GRADIENT ALLOWED	% < 25 AT > 1.0E-8 OR % < 50 AT < 1.0E-8]
	1.0E-03											
/ITY, cm/sec	1.0E-05											
NDUCTI	1.0E-06			*		•						
AULIC COI	1.0E-07											
НУDR	1.0E-08											
	1.0E-09 0.0	0 5.00) 10	.00 15	.00 20.00) 25.0	0 30	0.00 35	.00 40.0	00 45.	00 50]).00
	N:\CM\LAB_D	ATA\00 Projects in F	Progress\2015	Projects in Progres	s\15151122 Lab Data\(FIME, minute	'S ability D5084-C	C COF-B004-S7-23.	<u>][er</u>	гас	on	
						335						





		<u>ME</u> /	ASUREMI ASTM	<u>ENT OF HYD</u> <u>U</u> D 5084 - 03	RAULIC CON SING A FLEXI METHOD C TI	DUCTIVITY BLE WALL EST WITH I	OF SATU PERMEA	URATED POI	ROUS MATE	RIALS		Ň
PROJE	ECT NAME:	DYNEGY		FLUIC	: DEAIRED TA	P WATER	WITH 0.0	05 N CaSO4 PROJE	CT NUMBER:	151511	22	
LOCAT SAMPL SAMPL	LE ID: LE DESCR.:	COFFEEN, II COF-B004 SANDY FAT	S7 20 CLAY, RE	3.5.0 - 25.5 fe DDISH BRO\	et WN WITH DARK	GRAY		PANEL IDEN BURETT	DATE: ITIFICATION: BURE E INCREMEN	9/4/201 Ler ETTE AREA: IT LENGTH: ICREMENT:	5 nexa Perm E 0.312 1.000 0.312	Board cm^2 cm cm^3
MOIST W & 1 D & T T, g MOIS URE,	<u>FURE%</u> T, g 140.98 T, g 122.69 g 33.88 ST- , % 20.6	II <u>DENSITY</u> WET WT, g DIA, in HT, in AREA DENSITY: DENSITY:	NITIAL 438.0 2.857 2.010 129.5 107.4	7.26 5.11 41.36 PCF WET PCF DRY	cm cm cm^2		SPECI SPECI F SAT	FIC GRAVITY FIC GRAVITY POROSITY, % FURATION, % VOID RATIO	ADDITION : 2.70 : ASSUMED : 36.3 : 97.6 : 0.57	IAL DATA RECOM PRC OF COMPA OVER OF	IPACTED?: CTOR, pcf: TIMUM, %: ACTION, %: TIMUM, %:	NO NA NA NA NA
SATUF	RATION:	LATERAL	PRESS.:	104.0	psi	BA	CK PRES	SURE (=UPPI	ER=LOWER):	100.0	psi	
DURIN	IG	LATERAL	PRESS.:	104.0	psi	H2:	100.0	psi	H1:	100.0	psi	
IEST:								BIAS PRESSL	JRE (=H1-H2)	0.0	psi	
<u>H1</u> VALU	<u>H2</u> <u>JE VALUE</u>	ELAPSED TIME, 	DELTA H cm	<u>Ln</u> <u>H1/H2</u>	<u>HYD CON</u> <u>k, cm/sec</u>	OUT FLOW cm^3	IN FLOW cm^3	<u>OUT/IN</u> RATIO	<u>HYD</u> <u>GRAD</u>	<u>% FROM</u> MEAN k	<u>TEMP.:</u> <u>C</u>	<u>TEMP.</u> <u>CORR.:</u>
16.0 16.4 16.8 17.2	0 58.8 4 58.4 8 58.0 2 57.6	11.00 22.00 33.00 45.00	42.8 42.0 41.2 40.4	0.018519 0.018868 0.019231 0.019608	5.00E-07 5.10E-07 5.19E-07 4.85E-07	0.12 0.12 0.12 0.12 0.12	0.12 0.12 0.12 0.12 0.12	1.00 1.00 1.00 1.00	8.4 8.2 8.1 7.9	1 1 3 4	23.3 23.3 23.3 23.3 23.3	0.925 0.925 0.925 0.925
Г		M HYI Gl	IAXIMUM DRAULIC RADIENT	1.0E-03 TO 1.0E-04 TO 1.0E-05 TO 1.0E-06 TO less than	1.0E-04 1.0E-05 1.0E-06 1.0E-07 1.0E-07	2 5 10 20 30		0.75< RATIO <1.25	20 MAX HYDRAULIC GRADIENT ALLOWED	% < 25 AT > 1.0E-8 OR % < 50 AT < 1.0E-8]
	1.0E-03											
/ITY, cm/sec	1.0E-05											
NDUCTI	1.0E-06			*		•						
AULIC COI	1.0E-07											
НУDR	1.0E-08											
	1.0E-09 0.0	0 5.00	10	.00 15	.00 20.00) 25.0	0 30	0.00 35	.00 40.0	00 45.	00 50]).00
	N:\CM\LAB_D	ATA\00 Projects in F	Progress\2015	Projects in Progres	s\15151122 Lab Data\(FIME, minute	'S ability D5084-C	C COF-B004-S7-23.	<u>][er</u>	гас	on	
						335						





(MEA			DRAULIC COND		OF SATU			RIALS		
PPOIECT		DYNECY	<u>A31M</u>	<u>5084 - 03</u> <u>FLUIC</u>	DEAIRED TAI	P WATER	WITH 0.0	05 N CaSO4		45454	100	
LOCATIO	NAME: N:	COFFEEN, IL	LINOIS					PROJEC	DATE:	9/10/20	015	
SAMPLE I SAMPLE [D: DESCR.:	COF-B008 LEAN CLAY V BROWN	S5 11 WITH SAN	I.0 - 13.0 feet ID, DARK GF	AY WITH GRAY	TRACE GI	RAYISH	PANEL IDEN BURETT VO	TIFICATION: BURE E INCREMEN LUME PER IN	Le ETTE AREA NT LENGTH NCREMENT	nexa Perm E : 0.312 : 1.000 : 0.312	Board cm^2 cm cm^3
MOISTUF W & T, g D & T, g T, g MOIST- URE, %	RE% 125.65 112.54 33.72 16.6	DENSITY WET WT, g DIA, in HT, in AREA DENSITY: DENSITY:	468.5 2.865 2.105 131.5 112.8	7.28 5.35 41.59 PCF WET PCF DRY	cm cm cm^2		SPECI SPECI F SAT	FIC GRAVITY: FIC GRAVITY: POROSITY, %: FURATION, %: VOID RATIO:	2.70 ASSUMED 33.1 90.8 0.49	IAL DATA RECOM PRO OF COMP OVER OF	MPACTED?: DCTOR, pcf: PTIMUM, %: ACTION, %: PTIMUM, %:	NO NA NA NA
SATURA	TION:	LATERAL	PRESS.:	104.0	psi	ВА	CK PRES	SURE (=UPPE	R=LOWER):	100.0	psi	
DURING		LATERAL	PRESS.:	104.0	psi	H2:	100.0	psi	H1:	100.0	psi	
								BIAS PRESSU	RE (=H1-H2)	0.0	psi	
H1 VALUE	<u>H2</u> VALUE	<u>TIME,</u> <u>min</u>		<u>Ln</u> <u>H1/H2</u>	<u>HYD CON</u> <u>k, cm/sec</u>	<u>FLOW</u> <u>cm^3</u>	<u>FLOW</u> <u>cm^3</u>	<u>OUT/IN</u> <u>RATIO</u>	<u>HYD</u> <u>GRAD</u>	<u>% FROM</u> <u>MEAN k</u>	<u>TEMP.:</u> <u>C</u>	<u>TEMP.</u> CORR.:
7.4 7.5 7.7 8.3 8.5	69.0 68.9 68.7 68.1 67.9	0.00 123.00 368.00 1317.00 1560.00	61.6 61.4 61.0 59.8 59.4	0.003252 0.006536 0.019868 0.006711	8.22E-09 8.29E-09 6.43E-09 8.55E-09	0.03 0.06 0.19 0.06	0.03 0.06 0.19 0.06	1.00 1.00 1.00 1.00	11.5 11.4 11.2 11.1	4 5 18 9	23.1 23.1 23.6 23.3	0.929 0.929 0.918 0.925
		M HYE GF		1.0E-03 TO 1.0E-04 TO 1.0E-05 TO 1.0E-06 TO less than	1.0E-04 1.0E-05 1.0E-06 1.0E-07 1.0E-07	2 5 10 20 30		0.75< RATIO <1.25	30 MAX HYDRAULIC GRADIENT ALLOWED	% < 25 AT > 1.0E-8 OR % < 50 AT < 1.0E-8		
	1.0E-03 -											
	1.0E-04											
ΓY, cm/sec	1.0E-05											
	1.0E-06											
	1.0E-07											
	I.0E-08			.								
1	L.OE-09 0.0	0 200.4	00 4	400.00	600.00 80	00.00	1000.00	1200.00	1400.00) 1600	.00 180	
	N:\CM\LAB_D/	TA\00 Projects in Pr	rogress\2015 F	Projects in Progress	TI s\15151122 Lab Data\[15	ME, minute	S ability D5084-C	: COF-B008-S5-11.0	ller	rac	on	
						384						


HANSON 2016 LABORATORY RESULTS



CONSTANT HEAD PERMEABILITY TEST

JOB NUMBER:	16E0080			TEST DATE:	08/23/2016
CLIENT:	Natural Reso	urce T	Fechnology, Inc.	BORING #:	G405D
JOB DESCRIPTION:	Coffeen Ash I	Pond	2	SAMPLE #:	11-2
SAMPLE DESCRIPTION:	Gray silty CLA	AY, tra	ace coarse sand and	DEPTH (FT):	34.5-35.0
	small gravel.				
			WATER CONTE	ENT OF TRIMMI	<u>NGS</u>
FOR ESTIMATING PORE	E VOLUME ON	LY	_	BEFORE	AFTER
SPECIMEN WEIGHT (G)	868.38			TEST	TEST
SPECIMEN HEIGHT (IN)	3.748		TARE + WET SOIL (G)	280.51	924.55
DIAMETER (IN)	2.861		TARE + DRY SOIL (G)	245.00	806.34
AREA (SQ IN)	6.429		TARE (G)	3.75	50.12
VOLUME (CU IN)	24.095		WET SOIL(G)	n/a	874.43
WET DENSITY (PCF)	137.30		WATER (G)	35.51	118.21
DRY DENSITY (PCF)	119.68	*	DRY SOIL (G)	241.25	756.22
WT. DRY SOIL (G)	756.96	*	WATER CONTENT (%)	14.72	15.63
WT. DRY SOIL (G)	756.22		INITIAL DEGREE OF	SATURATION	102.03
VOLUME DRY SOIL (CU IN)	17.431	*	FINAL DEGREE OF	SATURATION	107.98
SP.GR. ASSUMED	2.65				
POROSITY (%)	27.66	*	STD. MAX. DEI	N.(LBS/CU.FT.)	N/A
HEIGHT OF HEAD (PSI)	4.00			MOISTURE (%)	N/A
HYDRAULIC GRADIANT	29.5	*	%	COMPACTION	N/A
PRESSURE HEAD (CM H2O)	281.27		SAMPLE F	REPARATION	Tube
			PA		5
TEST METHOD USED	ASTM D5	084	PERMEANT USED:	Tap Water	
* Estimates Only					
Remarks:					







CONSTANT HEAD PERMEABILITY TEST

JOB NUMBER:	16E0080			TEST DATE:	08/23/2016	
CLIENT:	Natural Reso	urce ⁻	Technology, Inc.	BORING #:	G406D	
JOB DESCRIPTION:	Coffeen Ash I	Pond	2	SAMPLE #:	23-3	
SAMPLE DESCRIPTION:	Gray silty CLA	<mark>۹</mark> Υ, tr	ace coarse sand and	DEPTH (FT):	43.0-43.5	
	small gravel					
			WATER CONTE	ENT OF TRIMMI	<u>NGS</u>	
FOR ESTIMATING PORE	E VOLUME ON	LY		BEFORE	AFTEF	
SPECIMEN WEIGHT (G)	827.45	-		TEST	TES	
SPECIMEN HEIGHT (IN)	3.592	-	TARE + WET SOIL (G)	333.17	880.57	
DIAMETER (IN)	2.861	-	TARE + DRY SOIL (G)	287.51	763.54	
AREA (SQ IN)	6.429		TARE (G)	3.71	49.99	
VOLUME (CU IN)	23.092	-	WET SOIL(G)	329.46	830.58	
WET DENSITY (PCF)) 136.51		WATER (G)	45.66	117.03	
DRY DENSITY (PCF)	117.59	*	DRY SOIL (G)	283.80	713.55	
WT. DRY SOIL (G)	712.77	*	WATER CONTENT (%)	16.09	16.40	
WT. DRY SOIL (G)	713.55	-	INITIAL DEGREE OF	SATURATION	104.7	
VOLUME DRY SOIL (CU IN)	16.414	*	FINAL DEGREE OF	SATURATION	107.2	
SP.GR. ASSUMED	2.65					
POROSITY (%)	28.92	*	STD. MAX. DEM	N.(LBS/CU.FT.)	N/A	
HEIGHT OF HEAD (PSI)	3.80	-		MOISTURE (%)	N/A	
HYDRAULIC GRADIANT	29.3	*	%	COMPACTION	N/A	
PRESSURE HEAD (CM H2O)	267.21	-	SAMPLE F	REPARATION	Tube	
			PA		6	
TEST METHOD USED:	ASTM D5	084	PERMEANT USED:	Tap Water		
* Estimates Only						
,						

					Hanson			N _®					
					nunson	1010331011	ui seivices	inc.					
					CONSTAN	T HEAD PE	RMEABILIT	Y TEST					
	JC JOB DE	DB NUMBER: CLIENT: SCRIPTION:	16E0080 Natural Res Coffeen Ast	ource Tec Pond 2	hnology, Inc.				TE E S DE	EST DATE: BORING #: BAMPLE #: EPTH (FT):	8/23/2016 G406D 23-3 43.0-43.5		
	SPECIMEN HEIGHT (IN) 3.592 DIAMETER (IN) 2.861 DIAMETER (IN) 2.861 AREA (SQ IN) 6.429 INITIAL DEGREE OF SATURATION 104.79 * 4 Readings - Ratio of Outflow to Inflow is between .75 to 1.25 or Steady. ** Mean Ratio is Steady if 4 or more incremental readings are ±25% or better of the mean, or within ± 50% or better for k < 1 x 10 ⁻⁸ cm/sec.												
				INC.	INC.		AVG, INC.	OUTFLOW	INC.	TOTAL	INC.	AVG.	INC.
START	START	STOP	STOP	INFLOW	OUTFLOW	TEMP.	FLOW	INFLOW	TIME	TIME	PERM.	PERM.	MEAN
DATE	TIME	DATE	TIME	(CC)	(CC)	(°C)	(CC)	(RATIO) *	(MIN)	(MIN)	(CM/SEC)	(CM/SEC)	RATIO **
08/25/2016	9:27:00	08/25/2016	15:25:00	3.20	5.4	23.4	4.3000	1.688	358.	358.	1.52E-07	1.52E-07	1.000
08/25/2016	16:18:00	08/26/2016	8:20:00	3.90	2.1	24.2	3.0000	0.538	962.	1320.	3.88E-08	9.54E-08	0.406
08/26/2016	8:20:00	08/26/2016	14:00:00	1.10	1.0	23.4	1.0500	0.909	340.	1660.	3.91E-08	7.66E-08	0.510
08/29/2016	8:03:00	08/29/2016	16:22:00	1.30	1.0	23.4	1.1500	0.769	499.	2159.	2.92E-08	6.48E-08	0.450
08/29/2016	16:22:00	08/30/2016	7:54:00	1.90	1.9	24.2	1.9000	1.000	932.	3091.	2.53E-08	3.31E-08	0.766
08/30/2016	7:54:00	08/31/2016	8:22:00	2.60	2.7	24.2	2.6500	1.038	1468.	4559.	2.24E-08	2.90E-08	0.773
08/31/2016	8:22:00	09/01/2016	8:08:00	2.40	2.3	23.8	2.3500	0.958	1426.	5985.	2.07E-08	2.44E-08	0.847
09/01/2016	8:08:00	09/02/2016	8:04:00	2.10	2.1	23.8	2.1000	1.000	1436.	7421.	1.83E-08	2.17E-08	0.845
09/02/2016	8:04:00	09/02/2016	15:00:00	0.60	0.5	23.4	0.5500	0.833	416.	7837.	1.67E-08	1.95E-08	0.856
09/06/2016	8:16:00	09/07/2016	8:25:00	2.00	2.0	24.6	2.0000	1.000	1449.	9286.	1.70E-08	1.82E-08	0.935





CONSTANT HEAD PERMEABILITY TEST

JOB NUMBER:	16E0080			TEST DATE:	08/30/2016
CLIENT:	Natural Resou	urce	Technology, Inc.	BORING #:	G406D
JOB DESCRIPTION:	Coffeen Ash F	Pond	2	SAMPLE #:	13-2
SAMPLE DESCRIPTION:	Brn & gray fin	e sa	ndy SILT, some	DEPTH (FT):	22.75-23.25
	clay, coarse s	and	and small gravel.		
			WATER CONT	ENT OF TRIMM	INGS
FOR ESTIMATING PORE	VOLUME ON	LY		BEFORE	AFTER
SPECIMEN WEIGHT (G)	560.94	-		TEST	TEST
SPECIMEN HEIGHT (IN)	3.605	-	TARE + WET SOIL (G)	95.16	615.02
DIAMETER (IN)	2.301	-	TARE + DRY SOIL (G)	88.98	568.27
AREA (SQ IN)	4.158	-	TARE (G)	14.31	49.99
VOLUME (CU IN)	14.991	-	WET SOIL(G)	80.85	565.03
WET DENSITY (PCF)	142.55	-	WATER (G)	6.18	46.75
DRY DENSITY (PCF)	131.65	*	DRY SOIL (G)	74.67	518.28
WT. DRY SOIL (G)	518.06	*	WATER CONTENT (%)	8.28	9.02
WT. DRY SOIL (G)	518.28	-	INITIAL DEGREE OF	SATURATION	79.72
VOLUME DRY SOIL (CU IN)	11.709	*	FINAL DEGREE OF	SATURATION	87.06
SP.GR. ASSUMED	2.70	-			
POROSITY (%)	21.89	*	STD. MAX. DEI	N.(LBS/CU.FT.)	N/A
HEIGHT OF HEAD (PSI)	3.90	-	OPTIMUM I	MOISTURE (%)	N/A
HYDRAULIC GRADIANT	29.9	*	%	COMPACTION	N/A
PRESSURE HEAD (CM H2O)	274.24	-	SAMPLE F	PREPARATION	Tube
			PA	ANEL NUMBER	8
TEST METHOD USED	ASTM D5	084	PERMEANT USED:	Tap Water	
* Estimates Only					
Remarks:					

					Hanson			N _®									
					nunson	1010331011	al services	inc.									
					CONSTAN	T HEAD PE	RMEABILIT	Y TEST									
	JOB NUMBER:16E0080TEST DATE:08/30/16CLIENT:Natural Resource Technology, Inc.BORING #:G406DJOB DESCRIPTION:Coffeen Ash Pond 2SAMPLE #:13-2DEPTH (FT):22.75-23.25																
	SPECIMEN HEIGHT (IN) 3.605 DIAMETER (IN) 2.301 AREA (SQ IN) 4.158 INITIAL DEGREE OF SATURATION 79.72 * 4 Readings - Ratio of Outflow to Inflow is between .75 to 1.25 or Steady. ** Mean Ratio is Steady if 4 or more incremental readings are ±25% or better of the mean, or within ± 50% or better for k < 1 x 10 ⁻⁸ cm/sec.																
				INC.	INC.		AVG, INC.	OUTFLOW	INC.	TOTAL	INC.	AVG.	INC.				
START	START	STOP	STOP	INFLOW	OUTFLOW	TEMP.	FLOW	INFLOW	TIME	TIME	PERM.	PERM.	MEAN				
DATE	TIME	DATE	TIME	(CC)	(CC)	(°C)	(CC)	(RATIO) *	(MIN)	(MIN)	(CM/SEC)	(CM/SEC)	RATIO **				
09/01/2016	9:10:00	09/01/2016	9:33:00	24.00	24.7	24.2	24.3500	1.029	23.	23.	1.99E-05	1.99E-05	1.000				
09/01/2016	13:39:00	09/01/2016	16:17:00	66.78	65.2	24.0	65.9900	0.976	158.	181.	7.88E-06	1.39E-05	0.568				
09/02/2016	8:21:00	09/02/2016	15:18:00	121.80	116.8	23.8	119.3000	0.959	417.	598.	5.42E-06	1.11E-05	0.490				
09/06/2016	8:27:00	09/06/2016	13:56:00	91.14	90.0	25.6	90.5700	0.987	329.	927.	5.01E-06	9.55E-06	0.524				
09/06/2016	14:04:00	09/06/2016	16:25:00	32.34	31.6	24.2	31.9700	0.977	141.	1068.	4.26E-06	5.64E-06	0.755				
09/07/2016	8:54:00	09/07/2016	16:06:00	81.48	79.6	25.0	80.5400	0.977	432.	1500.	3.44E-06	4.53E-06	0.759				
09/08/2016	8:02:00	09/08/2016	16:19:00	61.32	60.0	25.2	60.6600	0.978	497.	1997.	2.24E-06	3.74E-06	0.600				
09/09/2016	8:23:00	09/09/2016	14:35:00	41.16	41.4	23.8	41.2800	1.006	372.	2369.	2.10E-06	3.01E-06	0.699				
09/14/2016	9:45:00	09/15/2016	8:30:00	17.22	17.2	24.2	17.2100	0.999	1365.	3734.	2.37E-07	2.01E-06	0.118				
09/15/2016	8:30:00	09/15/2016	16:10:00	4.62	4.8	25.2	4.7100	1.039	460.	4194.	1.88E-07	1.19E-06	0.158				
09/15/2016	16:10:00	09/16/2016	8:29:00	9.66	9.2	24.2	9.4300	0.952	979.	5173.	1.81E-07	6.77E-07	0.267				





CONSTANT HEAD PERMEABILITY TEST

JOB NUMBER:	16E0080			TEST DATE:	09/02/2016
CLIENT:	Natural Reso	urce	Technology, Inc.	BORING #:	T408
JOB DESCRIPTION:	Coffeen Ash I	Ponc	12	SAMPLE #:	7-2A
SAMPLE DESCRIPTION:	Brn & gray fin	e sa	ndy SILT, some	DEPTH (FT):	23.5-24.0
	clay, coarse s	and	and small gravel.		
			WATER CONTE	ENT OF TRIMMI	NGS
FOR ESTIMATING PORE	E VOLUME ON	LY		BEFORE	AFTER
SPECIMEN WEIGHT (G)	474.47			TEST	TEST
SPECIMEN HEIGHT (IN)	3.491	-	TARE + WET SOIL (G)	99.67	523.99
DIAMETER (IN)	2.165	-	TARE + DRY SOIL (G)	92.29	487.75
AREA (SQ IN)	3.681		TARE (G)	3.77	50.16
VOLUME (CU IN)	12.852		WET SOIL(G)	95.90	473.83
WET DENSITY (PCF)	140.64		WATER (G)	7.38	36.24
DRY DENSITY (PCF)	129.82	*	DRY SOIL (G)	88.52	437.59
WT. DRY SOIL (G)	437.96	*	WATER CONTENT (%)	8.34	8.28
WT. DRY SOIL (G)	437.59	-	INITIAL DEGREE OF	SATURATION	75.45
VOLUME DRY SOIL (CU IN)	9.898	*	FINAL DEGREE OF	SATURATION	74.68
SP.GR. ASSUMED	2.70	-			
POROSITY (%)	22.98	*	STD. MAX. DEI	N.(LBS/CU.FT.)	N/A
HEIGHT OF HEAD (PSI)	3.70	-	OPTIMUM I	MOISTURE (%)	N/A
HYDRAULIC GRADIANT	29.3	*	%	COMPACTION	N/A
PRESSURE HEAD (CM H2O)	260.17		SAMPLE F	PREPARATION	Tube
			PA		7
TEST METHOD USED	ASTM D5	084	PERMEANT USED:	Tap Water	
* Estimates Only					
Remarks:					

						~								
						Y HA	NSO	N®						
					Hanson I	 Profession	al Services	Inc.						
					CONSTAN	T HEAD PE	RMEABILIT	Y TEST						
	JC	OB NUMBER:	16E0080						TE	ST DATE:	9/2/2016			
		CLIENT:	Natural Res	ource Tec	hnology, Inc.				E	BORING #:	T408			
	JOB DE	SCRIPTION:	Coffeen Ask	n Pond 2					5	SAMPLE #:	7-2A			
									DE	EPTH (FT):	23.5-24.0			
SPECIMEN HEIGHT (IN) <u>3.491</u> HEIGHT OF HEAD (PSI) <u>3.70</u>														
		DIAI	METER (IN)	2.165				PRESSU	IRE HEAD	(CM H2O)	260.17			
	AREA (SQ IN) 3.681 PANEL NUMBER 7													
INITIAL DEGREE OF SATURATION 79.72 * 4 Readings - Ratio of Outflow to Inflow is between .75 to 1.25 or Steady.														
** Mean Ratio is Steady if 4 or more incremental readings are ±25% or better of the														
							mean, or within	$1 \pm 50\%$ or bette	$r \text{ for } k < 1 \text{ x}^2$			A) (O		
OTADT	OTADT	OTOD	OTOD			TEMP	AVG, INC.		INC.		INC.	AVG.		
START	START	STOP	STOP	INFLOW	OUTFLOW		FLOW							
DATE		DATE				(°C)		(RATIO) *	(MIN)	(MIN)		(CM/SEC)	RATIO **	
09/06/2016	8:53:00	09/06/2016	16:18:00	19.70	20.2	24.2	19.9500	1.025	445.	445.	9.71E-07	9.71E-07	1.000	
09/06/2016	16:30:00	09/07/2016	8:33:00	32.76	29.52	20.2	31.1400	0.901	963.	1408.	0.09E-07	8.20E-07	0.816	
09/07/2016	16:29:00	09/08/2016	7:58:00	4.29	4.32	25.6	4.3050	1.007	929.	2337.	9.72E-08	5.79E-07	0.168	
09/08/2016	8:24:00	09/09/2016	8:20:00	7.80	8.0	25.2	7.9000	1.026	1436.	3773.	1.16E-07	4.64E-07	0.251	
09/09/2016	8:20:00	09/09/2016	14:31:00	9.80	10.0	24.2	9.9000	1.020	371.	4144. 5550	5.78E-07	3.05E-07	1.582	
09/12/2016	8:38:00	09/13/2016	8:12:00	7.20	7.5	23.8	7.3500	1.042	1414.	5558.	1.14E-07	2.26E-07	0.502	
09/14/2016	8:58:00	09/15/2016	8:21:00	10.60	10.8	24.4	10.7000	1.019	1403.	0901. 7405	1.64E-07	2.43E-07	0.076	
09/15/2016	8:21:00	09/15/2016	16:05:00	3.60	3.7	25.2	3.0500	1.028	464.	7425.	1.07E-07	2.50E-07	0.051	
09/15/2016	16:05:00	09/16/2016	8:16:00	7.40	1.3	24.2	7.3500	0.986	971.	8396.	1.64E-07	1.52E-07	1.078	



APPENDIX C

MONITORING WELL HYDROGRAPHS (2015-2016)





APPENDIX D

DISCHARGE TO SURFACE WATER CALCULATIONS

	CALCULATION SHEET	By: <u>NRK</u>		Date:_12/19/1		
NATURAL	Flux into Unnamed Creek	Chkd by: <u>EJT/SJC</u>		Date:_ <u>12/20/16</u>		
Resource Technology						
	Coffeen Ach Dand No. 2	Revision:	Date:	By:	App'd:	
Client: <u>Dynegy</u>	Closure Plan					
NRT Project #: 2380						

Problem Statement:

Calculate the flux of impacted groundwater into the unnamed creek located east of Coffeen Ash Pond No. 2.

Assumptions:

- 1. Groundwater can enter the creek through the eastern berm as shown in conceptualized cross-sections.
- 2. Coal Combustion Residuals (CCR) are in contact with Hagarstown Formation within Ash Pond No. 2.
- 3. Groundwater flow is through Hagarstown Formation only:
 - a. Hydraulic conductivity of Hagarstown Formation = 1×10^{-4} cm/s (Subsection 2.2.3)
 - b. Hydraulic conductivity of Ash Pond No. 2 embankment fill = 5.4x10⁻⁸ cm/s (AECOM, 2015; from samples of embankment fill from Ash Pond No. 1)
- Hagarstown Formation discharges directly to unnamed creek throughout 75% of eastern edge of Ash Pond No. 2 (as shown in conceptualized cross-section C-C') and Hagarstown Formation daylights to bottom of Ash Pond No. 2 with CCR in contact with Hagarstown Formation (absence of clay layer between bottom of Ash Pond No. 2 and Hagarstown Formation).

Calculations:

$$Conc_{(creek)} = \frac{Q_{(GW)} \times Conc_{(GW)} + (Q_{(creek)} - Q_{(GW)}) \times Conc_{(creek)}}{Q_{(creek)}}$$

Where:

 $Q_{(GW)} = KIA$

K = Hydraulic Conductivity of Hagarstown Formation (ft/min) I = Hydraulic Gradient $I = \frac{Ash Pond No. 2 water elev - elev of top of Hagarstown Formation$ $width of berm
<math display="block">I = \frac{631 \text{ ft} - 601 \text{ ft}}{30 \text{ ft}}$ I = 1 $A = Area of Flow (ft^2)$ A = Lx t L = length (ft) t = thickness (ft) A = 1,400 ft x 0.75 x 2 ft $A = 2,100 \text{ ft}^2$ $Q_{(GW)} = 5x10^{-4} \text{ cm/sec } x 1 x 2,100 \text{ ft}^2$ $Q_{(GW)} = 0.000016 \text{ ft/s } x 2,100 \text{ ft}^2$ $Q_{(GW)} = 0.0336 \text{ cfs or } 0.95 \text{ L/sec}$

P:\2300\2380\Deliverables\Reports\Hydrogeo Report\Appendix D\Flux into Unnamed Creek Calculation.doc

	CALCULATION SHEET	By: <u>NRK</u>	Date:_	12/19/16	
NATURAL	Flux into Unnamed Creek	Chkd by: <u>EJT/SJC</u>	-	Date:_	<u>12/20/16</u>
RESOURCE TECHNOLOGY					
	Coffeen Ash Dand No. 2	Revision:	Date:	By:	App'd:
Client: <u>Dynegy</u>	Closure Plan				
NRT Project #:					

$$Q_{(creek)} = \frac{1.49}{n} A R^{\frac{2}{3}} \sqrt{S}$$

 $\begin{array}{l} A = Flow Area \left(ft^2\right) \\ n = Manning's Roughness Coefficient (~0.05) \\ R = Hydraulic Radius (ft) \\ S = Channel Slope (ft/ft) (see attached Google Earth calculation) \end{array}$

$$Q_{(creek)} = \frac{1.49}{0.05} \times 18 \, ft^2 \, \times 1.44 \, ft^{\frac{2}{3}} \sqrt{0.004}$$

 $Q_{(creek)} = 29.8 \ x \ 18 \ x \ 1.28 \ x \ 0.063$

$$Q_{(creek)} = 43.3 \ cfs \ or \ 1,226 \ L/sec$$

$$R = \frac{A}{P}$$

A = Flow Area of creek channel (ft²) P = Wetted Perimeter of creek channel (ft)

$$R = \frac{18 ft^2}{12.5 ft}$$

$$R = 1.44 \, ft$$

$$P = b + 2\sqrt{(\frac{T-b}{2})^2 + h^2}$$

b = channel width at base of creek (ft) (see assumed geometry at discharge to lake) T = channel width at top of creek (ft) (see assumed geometry at discharge to lake)h = depth of creek channel (ft) (see assumed geometry at discharge to lake)

$$P = 8 ft + 2 \sqrt{(\frac{10 ft - 8 ft}{2})^2 + 2 ft^2}$$
$$P = 8 ft + 2\sqrt{5 ft^2}$$
$$P = 12.5 ft$$



APPENDIX D – ASH POND 2 EAST BERM CONCEPTUAL CROSS SECTIONS



С

Loess/ Clay

Till



C'

Embankment Fill Hagarstown Member V Unnamed Vandalia Creek

	CALCULATION SHEET	By: NRK	Date:	12/19/16	
NATURAL	Flux into Condenser Cooling	Chkd by: EJT/SJC	Date:	12/20/16	
Resource	Water Discharge Flume				
TECHNOLOGY			_	_	
		Revision:	Date:	By:	App'd:
Client: <u>Dynegy</u>	Coffeen Ash Pond No. 2				
	Closure Plan				
NRT Project #:2380					

Problem Statement:

Calculate the flux of impacted groundwater into the Condenser Cooling Water Discharge Flume located south of Ash Pond No. 2.

Assumptions:

- 1. Average flow in flume per NPDES permit is ~500 MGD or approximately 21,900 L/sec.
- 2. Groundwater can enter the flume through the southern berm as shown in cross-section A-A'.
- 3. Coal Combustion Residuals (CCR) are in contact with Hagarstown Formation within Ash Pond No. 2.
- 4. Groundwater flow is through Hagarstown Formation only:
 - a. Hydraulic conductivity of Hagarstown Formation = 1×10^{-4} cm/s (Subsection 2.2.3, this report)
 - b. Hydraulic conductivity of Ash Pond No. 2 embankment fill = 5.4×10^{-8} cm/s (AECOM, 2015; from samples of embankment fill from Ash Pond No. 1)
- Hagarstown Formation discharges directly to flume throughout entire southern edge of Ash Pond No. 2 (as shown in conceptualized cross-section C-C') and Hagarstown Formation daylights to bottom of Ash Pond No. 2 with CCR in contact with Hagarstown Formation (absence of clay layer between bottom of Ash Pond No. 2 and Hagarstown Formation).

Calculations:

$$Conc_{(flume)} = \frac{Q_{(GW)} \times Conc_{(GW)} + (Q_{(flume)} - Q_{(GW)}) \times Conc_{(flume)}}{Q_{(flume)}}$$

Where:

$$Q_{(GW)} = KIA$$

K = Hydraulic Conductivity of Hagarstown Formation (ft/min) I = Hydraulic Gradient $I = \frac{Ash Pond No.2 water elev - elev of top of Hagarstown Formation}{width of berm}$ $I = \frac{610 \text{ ft} - 600 \text{ ft}}{20 \text{ ft}}$ I = 0.5 $A = Area of Flow (ft^2)$ A = L x t L = length (ft) t = thickness (ft) A = 1,950 ft x 2 ft $A = 3,900 \text{ ft}^2$

P:\2300\2380\Deliverables\Reports\Hydrogeo Report\Appendix D\Flux into Condenser Cooling Water Discharge Flume Calculation.doc

	CALCULATION SHEET	By: <u>NRK</u>	Date:_	12/19/16	
NATURAL	Flux into Condenser Cooling	Chkd by: EJT/SJC		Date:_12/20/16	
Resource	Water Discharge Flume				
TECHNOLOGY					
		Revision:	Date:	By:	App'd:
Client: <u>Dynegy</u>	Coffeen Ash Pond No. 2				
NRT Project # 2380	Closure Plan				

$$Q_{(GW)} = 5x10^{-4} \ cm/\sec x \ 0.5 \ x \ 3,900 \ ft^{2}$$

$$Q_{(GW)} = 1.6x10^{-5} \ ft/\sec x \ 1,950 \ ft^{2}$$

$$Q_{(GW)} = 0.03 \ ft/sec \ or \ 0.88 \ L/sec$$

$$Q_{(flume)} = 21,900 \ L/sec$$

$$SO_{4} \ Conc_{(flume)} = \frac{\frac{0.88L}{sec} x \ \frac{3,000 \ mg}{L} + \left(\frac{21,900L}{sec} - \frac{0.88L}{sec}\right) x \ \frac{5}{21,900} \ \frac{L}{sec}$$

$$= \frac{2640 \ mg/\sec + \ 1.1 \ x \ 10^{6} \ mg/sec}{21,900 \ \frac{L}{sec}}$$

$$= 50.1 mg/L$$

Summary and Conclusions:

Since $Q_{(flume)}$ is so much larger in magnitude than $Q_{(GW)}$, the flux of impacted groundwater into the flume is negligible.

0mg

APPENDIX E

WATER WELL LOCATIONS AND RECORDS WITHIN 2,500-FOOT RADIUS OF COFFEEN ASH POND 2

E WELL SEARCH

E.1 Well Search Overview

The following sources of information were utilized in order to determine community water source and water well locations:

- Illinois State Geological Survey's Illinois Water Well (ILWATER) Internet Map Service
- Illinois State Water Survey Domestic Well Database
- Illinois EPA web-based Geographic Information System (GIS) files
- Illinois Department of Public Health
- Montgomery County Health Department

E.2 Illinois State Geological Survey (ISGS)

The ISGS website provided an ArcIMS View Map as well as a database query for water wells. ISGS database information including any boring logs and well construction information is provided in this Appendix.

E.3 Illinois State Water Survey (ISWS)

All of the wells found on-line through the ISWS Domestic Well Database were previously identified on the ISGS website. Hard copy records contained within the ISWS database, consisting of public, industrial, and commercial water wells, were not all received as of the date of this report. Since the ISWS database generally contains the same well information as the ISGS and Illinois EPA databases, some ISWS well entries on the Appendix E-1 Table were marked as pending. Should any new information be acquired from the ISWS including additional water wells not previously identified from the on-line sources of well information, it will be provided as an addendum to this report. Table E-2 lists wells located by RAPPS (2009) that were not located and identified in the on-line search for this report.

E.4 Illinois Environmental Protection Agency (IEPA)

The Illinois EPA database website provided ArcIMS Viewer Maps showing information on community, non-community, and public water supply wells as defined on the Illinois EPA website:

 Community Water Supply: a public water supply that serves or is intended to serve at least 15 service connections used by residents or regularly serves at least 25 residents.



- Non-Community Water Supply: a public water supply that is not a community water supply.
- Public Water Supply: all mains, pipes and structures through which water is obtained and distributed to the public, including wells and well structures, intakes and cribs, pumping stations, treatment plants, reservoirs, storage tanks and appurtenances, collectively or severally, actually used or intended for use for the purpose of furnishing water for drinking or general domestic use and which serve at least 15 service connections or which regularly serve at least 25 persons at least 60 days per year. A public water supply is either a community water supply or a non-community water supply.

Based on the IEPA maps, two non-CWS wells are located within Sections 10 and 15. Both non-CWS were identified in the ISGS records.

E.5 Montgomery County Health Department

Personnel from the Montgomery County Health Department confirmed the two non-CWS well systems were present within the area and noted that they were used at a campground and wildlife preserve. No additional information was provided about the area.





Table E-1. Well Search ResultsHydrogeologic Characterization ReportCoffeen Energy Center

Мар	Source	of Well Info	ormation		Location Name	Well		Location			Year	Aquifer		Well	
Well #	ISGS	ISWS***	IEPA	Other	at Time of Well Completion	Depth	County	Township	Range	Section	Subsection	Drilled	Туре	Formation	Use*
1	121352182400	115230	21824		Hueitt, Bill	32	Montgomery	7N	3W	14	NE,NE,NE	1974	Unconsolidated	clay	FD
2	121352182500	115229	21825		Stahl, Louis	32	Montgomery	7N	3W	14	NE,NE,NE	1974	Unconsolidated	clay	FD
3	121350164400	115213	1644		Flori, Eugene	20	Montgomery	7N	3W	10		1969	Unconsolidated	sand	FD
4	121350171700	115228	1717		Marfield, Mac	29	Montgomery	7N	3W	14	NE,NW,NE	1970	Unconsolidated	clay	FD
5	121350172600	115224	1726		Schuler, Paul	32	Montgomery	7N	3W	14	NW,NE,NE	1971	Unconsolidated	sand	FD
6	121352300600	115226	23006		Jump, James	41	Montgomery	7N	3W	14	SE,SE,SE	1986	Unconsolidated	ground-clay	FD
7	121352310800	***	23108		Dept. of Conservation	70	Montgomery	7N	3W	15	NE,NW,NW	1987	Unconsolidated	sandy clay	IC
8	121352221300	115222	22213		Gadshlen, Clarence	156	Montgomery	7N	3W	14	SE,NE,NE	1977	Bedrock	sandstone	FD
9	121352221400	115223	22214		Warfield, William	151	Montgomery	7N	3W	14	NE	1978	Bedrock	sandstone	FD
10	121352334900	243174	23349		Monk, Lawrence & Anita	382	Montgomery	7N	3W	14	SE,SE,NE	1993	Bedrock	gray sandstone	FD
11	121352361400	***	23614		White & Brewer	40	Montgomery	7N	3W	11	SE,NE,NE	1993	Unconsolidated	silt	MW
12	121352361500	***	23615		White & Brewer	35	Montgomery	7N	3W	11	NE	1993	Unconsolidated	silt	MW
13	121352361600	***	23616		White & Brewer	17	Montgomery	7N	3W	11			Unconsolidated	silt	MW
14	121352361700	***	23617		White & Brewer	25	Montgomery	7N	3W	11			Unconsolidated	silty clay	MW
15	121352361800	***	23618		White & Brewer	23	Montgomery	7N	3W	11			Unconsolidated	silty clay	MW
16	121352361900	***	23619		White & Brewer	40	Montgomery	7N	3W	11			Unconsolidated	sandy silt	MW
17	121352362000	***	23620		White & Brewer	20	Montgomery	7N	3W	11			Unconsolidated	sandy silt	MW
18	121352362100	***	23621		White & Brewer	33	Montgomery	7N	3W	11			Unconsolidated	sandy silt	MW
19	121352362300	***	23623		White & Brewer	48	Montgomery	7N	3W	12			Unconsolidated	clay and silt	MW
20	121352362400	***			White & Brewer	24	Montgomery	7N	3W	12			Unconsolidated	sandy silt	MW
21	121352283100	115350	22831		Sidner, Joe	50	Montgomery	7N	3W	10	NW,SW,SW	1984	Unconsolidated	gravel	FD
22	121352283200	115215	22832		Wibel, William	39	Montgomery	7N	3W	11	SE,SE,NW (A)	1981	Unconsolidated	sand and gravel	FD
23	121352380200	290232	23802		O'Dell, Kenneth & Chong	363	Montgomery	7N	3W	11	NW,SE,SE	1996	Bedrock	light gray sandstone	FD
24	121352380300	290231	23803		Childers, Joe	401	Montgomery	7N	3W	14	SW,NE,NE	1996	Bedrock	light gray sandstone	FD
25	121352396900	***	13500061		Coffeen Lake Fish & Wildlife		Montgomery	7N	3W	15	NW,NW,SE				NCWS
26	121352400700	***	13500012		Indian Grove Campground		Montgomery	7N	3W	10	SW,SW,SW				NCWS

Sources of Information

IEPA Illinois Environmental Protection Agency

ISGS Illinois State Geological Survey

- ISWS Illinois State Water Survey (Private Well Database)
- SWA IEPA Source Water Assessment

*Well Use

- FD Farm and/or Domestic Water Well
- IC Industrial/Commercial Water Well
- CWS Community Water Supply
- NCWS Non-Community Water Supply
- MW Monitoring well

Notes

Not applicable or no information available

*** ISWS data pending

(A) Well is mislocated in ISGS and/or IEPA databases



Table E-2. Other Water Wells, Precise Location Not AvailableHydrogeologic Characterization ReportCoffeen Energy Center

			Location		Well		Date
Well ID	Depth	Township	Range	Section	Use	Driller	Drilled
400397		7N	3W	10	IC		//
43308	16	7N	3W	10	IC	DAN KOHNEN	//
433123	20	7N	3W	10	IC	DAN KOHNEN	//
115214	500	7N	3W	11	IC		7/14/1996
250603	40	7N	3W	11	MO	FOX DRILLING INC.	11/17/1993
250604	35	7N	3W	11	MO	FOX DRILLING INC.	11/18/1993
250605	17	7N	3W	11	MO	FOX DRILLING INC.	1/28/1994
250610	25	7N	3W	11	MO	FOX DRILLING INC.	1/28/1994
250611	23	7N	3W	11	MO	FOX DRILLING INC.	1/28/1994
250612	40	7N	3W	11	MO	FOX DRILLING INC.	1/28/1994
250613	20	7N	3W	11	MO	FOX DRILLING INC.	1/28/1994
250614	33	7N	3W	11	MO	FOX DRILLING INC.	2/3/1994
433009	15	7N	3W	11	MO	DAN KOHNEN	//
290231	401	7N	3W	14	IC	KOHEN CONCR.	8/5/1996
377373	483	7N	3W	14	IC	SCWHARTZ	1997
377374	504	7N	3W	14	IC	SCWHARTZ	1997
377375	490	7N	3W	14	IC	SCWHARTZ	1997
377376	408	7N	3W	14	IC	SCWHARTZ	1997
377377	417	7N	3W	14	IC	SCWHARTZ	1997
377378	418	7N	3W	14	IC	SCWHARTZ	1997
377380	416	7N	3W	14	IC	SCWHARTZ	1997
403162		7N	3W	14			//
403163		7N	3W	14			//
115231	70	7N	3W	15	ST	H LINK	6/23/1987

Well Use

DO Domestic MO not specified

These wells are listed in RAPPS (2009). NRT has ordered but not yet received these records from the ISWS Domestic Wells Database.

IC not specified ST not specified



APPENDIX F

GROUNDWATER QUALITY DATA

Summary of Groundwater Data Units: mg/L

Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number Number<																								
net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net net																								Total
Nim Date Date <th< th=""><th></th><th></th><th>Arsenic,</th><th>Arsenic,</th><th>Barium,</th><th>Beryllium,</th><th>Boron,</th><th>Boron,</th><th>Cadmium,</th><th>Calcium,</th><th>Chloride,</th><th>Chromium,</th><th>Cobalt,</th><th>Fluoride,</th><th>Lead,</th><th>Lead,</th><th></th><th>Manganese,</th><th>Mercury,</th><th>Molybdenum,</th><th>Selenium,</th><th>Sulfate,</th><th>Thallium,</th><th>Dissolved</th></th<>			Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name Name <th< th=""><th>Well ID</th><th>Date</th><th>dissolved</th><th>total</th><th>total</th><th>total</th><th>dissolved</th><th>total</th><th>total</th><th>total L</th><th>total</th><th>total</th><th>total</th><th>total</th><th>dissolved</th><th>total</th><th>Lithium, total</th><th>dissolved</th><th>total</th><th>total</th><th>total</th><th>total</th><th>total</th><th>Solids</th></th<>	Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit bit <td>G151 G151</td> <td>12/21/2011</td> <td>< 0.003</td> <td>0.0012</td> <td>0.057</td> <td>< 0.001</td> <td>0.0218</td> <td>0.026</td> <td>< 0.001</td> <td></td> <td>50</td> <td>< 0.004</td> <td>< 0.002</td> <td>0.601</td> <td>< 0.002</td> <td>< 0.001</td> <td></td> <td>0.257</td> <td>< 0.0002</td> <td></td> <td>0.0021</td> <td>05</td> <td>< 0.001</td> <td>510</td>	G151 G151	12/21/2011	< 0.003	0.0012	0.057	< 0.001	0.0218	0.026	< 0.001		50	< 0.004	< 0.002	0.601	< 0.002	< 0.001		0.257	< 0.0002		0.0021	05	< 0.001	510
Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial Dial <th< td=""><td>G151 G151</td><td>01/23/2012</td><td></td><td>< 0.0012</td><td>0.037</td><td>< 0.001</td><td></td><td>0.020</td><td>< 0.001</td><td></td><td>67</td><td>< 0.004</td><td>< 0.002</td><td>0.091</td><td></td><td>< 0.001</td><td></td><td></td><td>< 0.0002</td><td></td><td>0.0021</td><td>120</td><td>< 0.001</td><td>580</td></th<>	G151 G151	01/23/2012		< 0.0012	0.037	< 0.001		0.020	< 0.001		67	< 0.004	< 0.002	0.091		< 0.001			< 0.0002		0.0021	120	< 0.001	580
bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits <th< td=""><td>G151 G151</td><td>05/22/2012</td><td>0.0014</td><td>× 0.001</td><td>0.005</td><td>0.001</td><td>0.052</td><td>0.015</td><td>× 0.001</td><td></td><td>07</td><td>× 0.004</td><td>× 0.002</td><td>0.400</td><td>< 0.001</td><td>× 0.001</td><td></td><td>0.11</td><td>< 0.0002</td><td></td><td>0.0025</td><td>120</td><td>< 0.001</td><td>520</td></th<>	G151 G151	05/22/2012	0.0014	× 0.001	0.005	0.001	0.052	0.015	× 0.001		07	× 0.004	× 0.002	0.400	< 0.001	× 0.001		0.11	< 0.0002		0.0025	120	< 0.001	520
GAI IntrAdvanta Form And nd And <th< td=""><td>G151 G151</td><td>07/23/2012</td><td>0.001</td><td>< 0.001</td><td>0.067</td><td>< 0.001</td><td>0.001</td><td>0.01</td><td>< 0.001</td><td></td><td>58</td><td>< 0.004</td><td>< 0.002</td><td>0.649</td><td></td><td>< 0.001</td><td></td><td>0.111</td><td>< 0.0002</td><td></td><td>0.0024</td><td>98</td><td>< 0.001</td><td>570</td></th<>	G151 G151	07/23/2012	0.001	< 0.001	0.067	< 0.001	0.001	0.01	< 0.001		58	< 0.004	< 0.002	0.649		< 0.001		0.111	< 0.0002		0.0024	98	< 0.001	570
bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits bits <th< td=""><td>G151</td><td>11/14/2012</td><td>< 0.001</td><td></td><td></td><td></td><td>0.013</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>< 0.001</td><td></td><td></td><td>0.091</td><td></td><td></td><td></td><td></td><td></td><td>660</td></th<>	G151	11/14/2012	< 0.001				0.013								< 0.001			0.091						660
1011 93/79/010 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 6.001 <	G151	01/30/2013	< 0.001				< 0.02								< 0.001			0.071						560
0151 0077000 <000	G151	05/20/2013	< 0.001				< 0.005								< 0.001			0.031						520
1011 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	G151	07/22/2013	< 0.001				0.02								< 0.001			0.12						500
	G151	02/19/2014	< 0.001				0.014								< 0.001			0.072						
Gin Min Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4 and Min 4	G151	05/12/2014	< 0.001				0.024								< 0.001			0.069						
GE11 OlyA2004 <	G151	08/11/2014	< 0.001				< 0.01								< 0.001			0.14						560
Ging Dig Dig <td>G151</td> <td>10/14/2014</td> <td>< 0.001</td> <td></td> <td></td> <td></td> <td>< 0.01</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>< 0.001</td> <td></td> <td></td> <td>0.031</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>570</td>	G151	10/14/2014	< 0.001				< 0.01								< 0.001			0.031						570
Ch31 OUNDADIS C-COUZ C C-COUZ -COUZ C-COUZ	G151	01/21/2015	< 0.001				< 0.01								< 0.001		-	0.066						500
Gis1 00/23/031 - 0.01 - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - - -	G151	04/08/2015	< 0.002				< 0.02								< 0.002			0.19						600
Gili 1 1000/2013 - 0001 0 0.003 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 <	G151	07/23/2015	< 0.001				0.018								< 0.001			0.093						550
G151 01/09/014	G151	10/06/2015	< 0.001				0.036								< 0.001			0.56						600
0.11 0.01///201 0.0024 0.013 0 0 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0024 0.0014 0.0034 0.0014 0.0034 0.0014 0.0034 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014 0.0014	G151	02/09/2016	< 0.001				< 0.01								< 0.001			0.18						560
C12 D/D/D/D C0003 D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D D <	G151 C152	05/11/2016	< 0.002				< 0.02								< 0.002			0.011						500
C12 01/25/012 00.001 0.013 < 0.001 0.014 < 0.001 0.001 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003	G152	12/21/2011	< 0.003	0.0022	0.12	10.001	0.0513	0.047	10.001		40	0.000	0.0000	0.642	< 0.002	0.0026		0.927	4.0.0000		0.000	110	10.001	494
01/1/1/102 0.001 0.004 0.004 0.004 0.002 0.002 0.001 0.001 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 1.00 0.0003 0.001 0.014 0.014 0.003 0.003 0.001 0.014 0.014 0.003 0.003 0.001 0.014 0.014 0.014 0.003 0.003 0.001 0.014 0.014 0.0034 0.003 0.001 0.014 0.014 0.003 0.001 0.013 0.001 0.014 0.014 0.014 0.014 0.014 0.014 0.014 0.014 0.014 0.014 0.014 0.011 0.014 0.011 0.014 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011	G152	01/25/2012		0.0023	0.13	< 0.001		0.047	< 0.001		40	0.008	0.0023	0.642		0.0026			< 0.0002		0.003	110	< 0.001	520
000000000000000000000000000000000000	G152 G152	05/13/2012	0.0011	0.0014	0.13	< 0.001	0.044	0.036	< 0.001		42	< 0.004	< 0.002	0.422	< 0.001	< 0.001		0.21	< 0.0002		0.0043	130	< 0.001	580
Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint Constraint <thconstraint< th=""> Constraint Constrai</thconstraint<>	G152 G152	03/22/2012	0.0011	0.0011	0.1	< 0.001	0.044	0.032	< 0.001		45	< 0.004	< 0.002	0 587	< 0.001	< 0.001		0.21	< 0.0002		0.0035	120	< 0.001	670
C123 01/30/2013 <	G152 G152	11/14/2012	< 0.001	0.0011	0.1	0.001	0.031	0.032	× 0.001			× 0.004	× 0.002	0.507	< 0.001	× 0.001		0.14	< 0.0002		0.0035	120	×0.001	620
GL12 OF/20/2013 < 0.001 0.015 0 0 < 0.001 0.034 0 0 570 GL12 OF/22/2013 < 0.001 0.024 0 0.0014 0.014 0.01 0.034 0 570 GL12 OF/22/2013 < 0.001 0.024 0 0.014 0.21 0 0.034 0 570 GL12 OF/22/2013 < 0.001 0.038 0 0.034 0.014 0.21 0.034 0 0.034 GL12 OF/12/2014 < 0.001 0.034 0.034 0 0.034 0.035 0 0 0.035 0 0 0.035 0 0.035 0 0.035 0 0.035 0 0.035 0 0.035 0 0.035 0 0.035 0 0.035 0 0.035 0 0 0.035 0 0.035 0 0.035 0 0 0.035 0 0.057 0 0.035 <td>G152 G152</td> <td>01/30/2013</td> <td>< 0.001</td> <td></td> <td></td> <td></td> <td>0.031</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>< 0.001</td> <td></td> <td></td> <td>0.14</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>600</td>	G152 G152	01/30/2013	< 0.001				0.031								< 0.001			0.14						600
Give OV/2/2013 < 0.001 I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I I	G152	05/20/2013	< 0.001				0.015								< 0.001			0.034						570
G152 0/1/9/2014 < 0.001 0.038 < 0.011 <th< td=""><td>G152</td><td>07/22/2013</td><td>< 0.001</td><td></td><td></td><td></td><td>0.024</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>0.0014</td><td></td><td></td><td>0.21</td><td></td><td></td><td></td><td></td><td></td><td>580</td></th<>	G152	07/22/2013	< 0.001				0.024								0.0014			0.21						580
G152 05/12/2014 < 0.001 < 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.005 0.	G152	02/19/2014	< 0.001				0.038								< 0.001			0.12						
G152 09/11/2014 < < < < 0.031 820 G152 01/14/2014 < 0.034 0.034 0.031 0.03 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.034 0.035 0.015 0.015 0.015 0.015 0.015 0.017 0.025 0.025 0.025 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.027 0.017 0.007 0.017 0.017 0.017 0.017 0.017 0.017 0.016	G152	05/12/2014	< 0.001				0.044								< 0.001			0.05						
G152 01/4/2014 <0.001 < < <	G152	08/11/2014	< 0.001				0.044								< 0.001			0.31						820
G152 0/2/1/2015 0.001 <	G152	10/14/2014	< 0.001				0.034								< 0.001			0.15						620
G152 04/08/2015 <	G152	01/21/2015	< 0.001				0.03								< 0.001			0.15						620
G152 07/23/2015 0.0078	G152	04/08/2015	< 0.002				0.025								< 0.002			0.07						680
G152 01/06/2015 0.0034	G152	07/23/2015	0.0078				0.079								< 0.001			0.89						730
G152 02/09/C16 0.0034 0.082 0.001 0.047 0.047 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <td>G152</td> <td>10/06/2015</td> <td>0.0054</td> <td></td> <td></td> <td></td> <td>0.11</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>< 0.001</td> <td></td> <td></td> <td>0.53</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>1100</td>	G152	10/06/2015	0.0054				0.11								< 0.001			0.53						1100
G152 05/11/2016 < 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <	G152	02/09/2016	0.0034				0.082								< 0.001			0.47						920
6152 11/19/2016 0 0.073 0 0.017 0 51 0 0.059 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	G152	05/11/2016	< 0.002		0.072		0.066	0.017			F1			0.500	< 0.002			0.31				120		800
G153 12/2/2011 </td <td>G152</td> <td>11/19/2016</td> <td>0.000</td> <td></td> <td>0.073</td> <td></td> <td>0.0575</td> <td>0.017</td> <td></td> <td></td> <td>51</td> <td></td> <td></td> <td>0.509</td> <td></td> <td></td> <td></td> <td>1.00</td> <td></td> <td></td> <td></td> <td>130</td> <td></td> <td>080</td>	G152	11/19/2016	0.000		0.073		0.0575	0.017			51			0.509				1.00				130		080
6153 01/25/2012 0.0056 0.0056 0.0056 0.0056 0.0056 0.0056 0.0056 0.0056 0.0057 0.001 0.0002 0.0012 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.0021 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 <t< td=""><td>G153</td><td>12/21/2011</td><td>< 0.003</td><td>0.005.0</td><td>0.020</td><td>10.001</td><td>0.0575</td><td>0.004</td><td>10.001</td><td></td><td>120</td><td>0.0052</td><td>0.0000</td><td>0.200</td><td>< 0.002</td><td>10.001</td><td></td><td>1.02</td><td>(0.0002</td><td></td><td>0.021</td><td>1500</td><td>10.001</td><td>3150</td></t<>	G153	12/21/2011	< 0.003	0.005.0	0.020	10.001	0.0575	0.004	10.001		120	0.0052	0.0000	0.200	< 0.002	10.001		1.02	(0.0002		0.021	1500	10.001	3150
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	G153	01/25/2012		0.0056	0.036	< 0.001		0.064	< 0.001		130	0.0053	0.0068	0.369		< 0.001			< 0.0002		0.021	1500	< 0.001	3000
ALSSOS/22/212O.004O.004O.002O.001OOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOO </td <td>G152</td> <td>05/22/2012</td> <td>0.004</td> <td>0.0059</td> <td>0.022</td> <td>< 0.001</td> <td>0.022</td> <td>0.030</td> <td>< 0.001</td> <td> </td> <td>001</td> <td>0.0041</td> <td>0.0034</td> <td>0.293</td> <td>< 0.001</td> <td>< 0.001</td> <td></td> <td>0.42</td> <td>< 0.0002</td> <td><u> </u></td> <td>0.027</td> <td>1900</td> <td>< 0.001</td> <td>3200</td>	G152	05/22/2012	0.004	0.0059	0.022	< 0.001	0.022	0.030	< 0.001		001	0.0041	0.0034	0.293	< 0.001	< 0.001		0.42	< 0.0002	<u> </u>	0.027	1900	< 0.001	3200
G133 07/23/2012 0.0032 0.003 0.001 0.003 0.001 0.003 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001	G155 G152	03/22/2012	0.004	0.0052	0.016	< 0.001	0.052	0.021	< 0.001		120	0.0041	< 0.002	0.241	< 0.001	< 0.001		0.42	< 0.0002		0.021	1500	< 0.001	2200
G133 01/30/2013 0.0022 0.0039 0.039 0 0 0 0.001 0.19 0.19 0 0 300 G153 01/30/2013 0.0022 0.002 0.012 0 0 0 0.013 0 0 3000 G153 05/20/2013 <0.001	G153	11/14/2012	0 0023	0.0032	0.010	< 0.001	0.04	0.031	< 0.001		130	0.0041	< 0.00Z	0.341	< 0 001	< 0.001		0.19	< 0.000Z		0.021	1300	< 0.001	3200
G133 05/20/2013 < 0.001	G153	01/30/2012	0.0023				0.039								< 0.001			0.19						3000
G153 07/22/2013 0.0027 0.028 </td <td>G153</td> <td>05/20/2013</td> <td>< 0.001</td> <td></td> <td></td> <td></td> <td>0.012</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>< 0.001</td> <td></td> <td>1</td> <td>0.13</td> <td></td> <td> </td> <td></td> <td></td> <td></td> <td>2800</td>	G153	05/20/2013	< 0.001				0.012								< 0.001		1	0.13						2800
G153 02/19/2014 0.042 0.4 0.16 G153 05/12/2014 0.021 0.41 <0.001	G153	07/22/2013	0.0027				0.028								< 0.001			0.1						3000
G153 05/12/2014 < 0.001 0.021 0.021 0.01 0.041 < 0.001 0.05 0.05	G153	02/19/2014	< 0.001				0.042							0.4	< 0.001			0.16						
	G153	05/12/2014	< 0.001				0.021							0.41	< 0.001			0.05						

		.	• • • • • •	De di u		D a second	Deves	C. J. J.		Chile Sile	Characteristics	Calcali	EL AND	11							C 15-1-	T I II ¹	Total
	Data	Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,	Lithium total	Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
G153	08/11/2014		lUldi	lolai	lulai	0.026	lotai	lulai	LULAI L	lotai	lolai	lolai	lotai		lotai	Litiliulii, totai	0.24	lotai	lotai	loldi	lotai	lotai	3500
G153	10/14/2014	0.0016				0.028								< 0.001			0.51						3400
G153	01/21/2015	< 0.001				0.02								< 0.001			0.2						3600
G153	04/08/2015	0.003				0.02								< 0.002			0.024						3700
G153	07/23/2015	< 0.001				0.015								< 0.001			0.16						3900
G153	10/06/2015	< 0.001				0.03								< 0.001			0.19						3700
G153	02/09/2016	< 0.001				0.016								< 0.001			0.046						3800
G153	05/11/2016	0.0035				0.026								< 0.002			0.037						3800
G153	11/19/2016			0.017			0.013			230			0.491								2500		4000
G154	12/21/2011	< 0.003				0.0452								< 0.002			0.462						624
G154	01/25/2012		0.0014	0.037	< 0.001		0.031	< 0.001		10	< 0.004	< 0.002	1.01		< 0.001			< 0.0002		0.0042	93	< 0.001	410
G154	03/13/2012		< 0.001	0.032	< 0.001		0.021	< 0.001		4.6	< 0.004	< 0.002	0.971		< 0.001			< 0.0002		0.0062	93	< 0.001	440
G154	05/22/2012	0.0017	.0.001	0.022	.0.001	0.025	0.020	.0.001		10		. 0.002	1.00	< 0.001	.0.001		0.038	. 0. 0002		0.0050	110	.0.001	470
G154	07/23/2012	10.001	< 0.001	0.033	< 0.001	0.04	0.038	< 0.001		4.9	< 0.004	< 0.002	1.06	10.001	< 0.001		0.042	< 0.0002		0.0058	110	< 0.001	490
G154 G154	11/14/2012	< 0.001				0.04					-			< 0.001			0.043						500
G154	01/30/2013	< 0.001				0.043								< 0.001			0.032						440
G154	07/22/2013	< 0.001				0.021								< 0.001			0.040						500
G154	02/19/2014	< 0.001				0.043								< 0.001			0.018						500
G154	05/12/2014	< 0.001				0.044								< 0.001			0.014						
G154	08/11/2014	< 0.001				0.035								< 0.001			0.081						480
G154	10/14/2014	< 0.001				0.035								< 0.001			0.0077						440
G154	01/21/2015	< 0.001				0.037								< 0.001			0.0092						450
G154	04/08/2015	< 0.002				0.023								< 0.002			0.003						440
G154	07/23/2015	< 0.001				0.024								< 0.001			0.14						430
G154	10/06/2015	< 0.001				0.044								< 0.001			0.01						500
G154	02/09/2016	< 0.001				0.041								< 0.001			0.041						560
G154	05/11/2016	< 0.002				0.037								< 0.002			0.0085						460
G270	03/11/2008	< 0.005	< 0.005	0.076	< 0.005	< 0.25	< 0.25	< 0.0025	77	9.5	< 0.025	< 0.005	0.36	< 0.0025	< 0.0025		0.19	< 0.0002		< 0.012	2.3	< 0.002	440
G270	04/21/2008	< 0.001	0.0016	0.076	< 0.001	0.078	0.071	< 0.0005	72	11	< 0.005	0.0017	0.36	< 0.0005	0.0021		0.21	< 0.0002		< 0.0025	2.4	< 0.002	420
G270	06/11/2008	< 0.001	0.0012	0.072	< 0.001	< 0.05	< 0.05	< 0.0005	70	< 10	< 0.005	0.001	0.33	< 0.0005	0.0021		0.21	< 0.0002		< 0.0025	2.8	< 0.002	430
G270	08/13/2008	< 0.001	< 0.001	0.069	< 0.001	0.052	< 0.05	< 0.0005	6/	8.9	< 0.005	< 0.001	0.38	< 0.0005	< 0.0005		0.26	< 0.0002		< 0.0025	4.2	< 0.002	440
G270	10/14/2008	< 0.001	< 0.001	0.068	< 0.001	0.073	< 0.05	< 0.0005	/1	9.4	< 0.005	< 0.001	0.35	< 0.0005	< 0.0005 4		0.22	< 0.0002		< 0.0025	4.7	< 0.002	440
G270	12/02/2008	< 0.001	< 0.001	0.068	< 0.001	0.052	0.051	< 0.0005	69 77	9.6	< 0.005	< 0.001	0.31	< 0.0005	0.00054		0.22	< 0.0002		< 0.0025	4.4	< 0.002	400
G270	09/21/2009	< 0.001	0.001			0.054	0.058	< 0.0005	75	9.4				< 0.0005	0.0011		0.20				5.2		450
G270	01/28/2010	< 0.001	< 0.001			0.054	0.057	< 0.0005	73	19				< 0.0005	0.0011		0.23				8.2		430
G270 G270	02/11/2010	0.001	× 0.001	0.056	< 0.001	0.055	0.032	× 0.0005	74	15	< 0.005	< 0.001	0.3	0.0005	0.0051		0.14	< 0.0002		< 0.0025	0.2	< 0.002	460
G270	06/09/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	61	39				< 0.001	< 0.001		0.058		1		6.8		470
G270	07/27/2010	< 0.001	< 0.001			0.011	< 0.01	< 0.001	58	40			1	< 0.001	< 0.001		0.062	1			9.1		480
G270	11/15/2010	< 0.001	0.0011			0.013	0.016	< 0.001	60	40				< 0.001	0.0021		0.069				7.6		490
G270	01/28/2011	< 0.001	< 0.001	0.064	< 0.001	< 0.01	0.012	< 0.001	80	45	< 0.004	0.002	0.36	< 0.001	0.0041		0.032	< 0.0002		0.002	9.6	< 0.001	470
G270	05/03/2011	< 0.001	< 0.001			0.01	< 0.01	< 0.001	57	57				< 0.001	< 0.001		0.023				13		430
G270	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	59	57				< 0.001	< 0.001		0.046				18		480
G270	11/11/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	61	64				< 0.001	< 0.001		0.05				23		440
G270	01/26/2012	< 0.001	< 0.001	0.049	< 0.001	< 0.01	< 0.01	< 0.001	64	73	< 0.004	< 0.002	0.363	< 0.001	< 0.001		0.0094	< 0.0002		0.0046	40	< 0.001	420
G270	05/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	67	83				< 0.001	< 0.001		0.04				57		460
G270	07/24/2012	< 0.001	< 0.001			0.092	0.018	< 0.001	73	66			<u> </u>	< 0.001	< 0.001		0.045				43		480
G270	11/14/2012	< 0.001	< 0.001	0.000	10.001	< 0.01	< 0.01	< 0.001	65	76	10.004	10.000	0.070	< 0.001	< 0.001		0.017	10,0000		0.0070	77	10.001	500
6270	01/30/2013	< 0.001	0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	65	91	< 0.004	< 0.002	0.372	< 0.001	< 0.001		0.011	< 0.0002		0.0078	96	< 0.001	540
G270	05/20/2013	0.0011 < 0.001	< 0.001			< 0.01	< 0.01	< 0.001	72	85 70			<u> </u>	< 0.001	< 0.001		0.025				120		48U 500
G270	10/14/2012	< 0.001	< 0.001			0.01	< 0.01	< 0.001	70 Q1	79				< 0.001	< 0.001		0.045				12U QC		500
5270	10/ 17/ 2013	× 0.001	× 0.001			0.014	× 0.01	× 0.001	01	12	L		1	- 0.001	× 0.001	L	0.041	I	1	1	05		520

																							Total
Well ID	Date	Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,	Lithium total	Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
G270	02/19/2014	< 0.001	< 0.001	0.041	< 0.001	< 0.01	< 0.01	< 0.001	67	56	< 0.004	< 0.002	0 279	< 0.001	< 0.001	Litiliun, totai	0.0029	< 0.0002	totai	0.0022	140	< 0.001	501103
G270	05/13/2014	< 0.001	0.0013	0.041	0.001	0.026	0.025	< 0.001	68	30	× 0.004	< 0.002	0.275	< 0.001	< 0.001		0.016	< 0.0002		0.0022	140	< 0.001	
G270	08/11/2014	< 0.001	0.0034			< 0.01	< 0.01	< 0.001	62	34				< 0.001	< 0.001		0.04				130		500
G270	10/14/2014	< 0.001	0.011			0.12	0.057	0.008	67	21				< 0.001	0.0073		0.04				140		500
G270	01/20/2015	< 0.001	< 0.001			0.011	< 0.01	< 0.001	70	18				< 0.001	< 0.001		0.011				140		500
G270	04/13/2015	< 0.002	< 0.001	0.05	< 0.001	0.025	0.047	< 0.001	70	20	< 0.004	< 0.002	0.334	< 0.002	< 0.001		0.0055	< 0.0002		0.0016	120	< 0.001	540
G270	07/22/2015	< 0.001	< 0.001	0.049	< 0.001	< 0.01	< 0.01	< 0.001		15	< 0.004	< 0.002	0.427	< 0.001	0.0018		0.47	< 0.0002	0.0011	< 0.001	110	< 0.001	550
G270	10/05/2015	< 0.001	< 0.001	0.037	< 0.001	0.013	< 0.01	< 0.001		11	< 0.004	< 0.002	0.411	< 0.001	< 0.001		0.056	< 0.0002	< 0.001	< 0.001	82	< 0.001	480
G270	11/20/2015		0.001	0.045	< 0.001		< 0.01	< 0.001	59	12	< 0.004	< 0.002	0.362		0.0015	< 0.01		< 0.0002	0.001	< 0.001	89	< 0.001	400
G270	02/10/2016	< 0.001	< 0.001	0.037	< 0.001	0.02	< 0.01	< 0.001	49	16	< 0.004	< 0.002	0.472	< 0.001	< 0.001	< 0.01	0.012	< 0.0002	< 0.001	0.0013	77	< 0.001	400
G270	05/12/2016	< 0.002	< 0.001	0.034	< 0.001	< 0.02	< 0.01	< 0.001	57	15	< 0.004	< 0.002	0.504	< 0.002	< 0.001	< 0.01	0.029	< 0.0002	< 0.001	< 0.001	77	< 0.001	370
G270	08/01/2016		< 0.001	0.037	< 0.001		< 0.01	< 0.001	50	15	< 0.004	< 0.002	0.397		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	76	< 0.001	360
G271	09/22/2009		0.003	0.08	< 0.001		0.33	< 0.0005	110	37	0.0064	0.0023	0.46		0.0054			< 0.0002		0.0038	230	< 0.002	770
G271	11/09/2009		0.0049	0.11	< 0.001		0.33	< 0.0005	140	34	0.014	0.0047	0.41		0.011			< 0.0002		0.0034	290	< 0.002	770
G271	11/10/2009	< 0.001				0.34								< 0.0005			0.11						
G271	01/19/2010	< 0.001	0.001	0.055	< 0.001	0.18	0.24	< 0.0005	110	33	< 0.005	< 0.001	0.34	< 0.0005	0.0018		0.007	< 0.0002		0.0035	320	< 0.002	860
G271	03/08/2010	< 0.001	< 0.001	0.055	< 0.001	0.25	0.23	< 0.0005	120	36	< 0.005	< 0.001	0.39	0.00061	0.0023		< 0.0025	< 0.0002		0.0053	300	< 0.002	880
G271	05/27/2010	< 0.001	0.0016	0.053	< 0.001	0.26	0.16	< 0.001	130	40	0.0056	< 0.002	0.42	< 0.001	0.0029		0.028	< 0.0002		0.0051	330	< 0.001	870
G271	07/27/2010	< 0.001	< 0.001	0.041	< 0.001	0.24	0.15	< 0.001	120	40	< 0.004	< 0.002	0.34	< 0.001	< 0.001		0.016	< 0.0002		0.005	350	< 0.001	840
G271	09/20/2010	< 0.001				0.16								< 0.001			0.0015						
G271	11/16/2010	< 0.001	< 0.001			0.14	0.15	< 0.001	110	44				< 0.001	< 0.001		< 0.001				280		830
G271	01/28/2011	< 0.001	0.0015	0.05	< 0.001	0.17	0.14	< 0.001	110	44	< 0.004	< 0.002	0.48	< 0.001	0.0035		0.0013	< 0.0002		0.0079	270	< 0.001	730
G271	05/04/2011	< 0.001	< 0.001			0.16	0.11	< 0.001	100	53				< 0.001	< 0.001		< 0.001				240		750
G271	0//2//2011	< 0.001	< 0.001			0.13	0.17	< 0.001	96	49				< 0.001	< 0.001		< 0.001				240		800
G271	11/14/2011	< 0.001	0.0013	0.024	10.001	0.15	0.21	< 0.001	110	50	10.004	(0.002	0.202	< 0.001	< 0.001		< 0.001	10,0000		0.0000	250	10.001	710
G271	01/26/2012	< 0.001	0.0013	0.034	< 0.001	0.19	0.18	< 0.001	110	44 F 1	< 0.004	< 0.002	0.393	< 0.001	< 0.001		< 0.001	< 0.0002		0.0099	240	< 0.001	750
G271 C271	05/22/2012	< 0.001	< 0.001			0.10	0.14	< 0.001	110	5.1				< 0.001	< 0.001		< 0.001				240		710
G271	07/24/2012	< 0.001	< 0.001			0.18	0.15	< 0.001	110	45 50				< 0.001	< 0.001		< 0.001				200		040
G271 G271	01/31/2013	< 0.001	0.0014	0.063	< 0.001	0.28	0.19	< 0.001	150	58	0.0066	0.0026	0.458	< 0.001	0.0012		0.0021	< 0.0002		0.005	300	< 0.001	880
G271 G271	01/31/2013	0.001	< 0.0031	0.005	< 0.001	0.32	0.32	< 0.001	130	47	0.0000	0.0020	0.450	< 0.001	< 0.003		< 0.024	< 0.000Z		0.005	350	< 0.001	790
G271 G271	07/22/2013	< 0.0013	< 0.001			0.17	0.15	< 0.001	99	47				< 0.001	< 0.001		< 0.001				360		800
G271 G271	10/14/2013	< 0.001	< 0.001			0.10	0.15	< 0.001	120	45				< 0.001	< 0.001		< 0.001				390		840
G271	02/19/2014	< 0.001	0.0028	0.062	< 0.001	0.26	0.24	< 0.001	150	51	0.0087	0.0023	0.298	< 0.001	0.0056		< 0.001	< 0.0002		0.0045	420	< 0.001	010
G271	05/13/2014	0.16	0.0017			0.45	0.33	< 0.001	140	47				0.15	0.0018		0.22				440		
G271	08/11/2014	< 0.001	0.0027			0.39	0.44	< 0.001	140	42				< 0.001	0.0061		< 0.001				500		1000
G271	10/14/2014	< 0.001	0.0019			0.44	0.5	< 0.001	150	45				< 0.001	0.0062		0.0021				480		940
G271	01/21/2015	< 0.001	< 0.001			0.42	0.51	< 0.001	120	39				< 0.001	0.0014		< 0.001				490		870
G271	04/10/2015	< 0.002	< 0.001	0.029	< 0.001	0.37	0.31	< 0.001	130	45	< 0.004	< 0.002	0.406	< 0.002	< 0.001		< 0.002	< 0.0002		0.0035	440	< 0.001	1000
G271	07/22/2015	< 0.001	< 0.001	0.028	< 0.001	0.32	0.24	< 0.001		35	< 0.004	< 0.002	0.406	0.0017	0.0036		0.001	< 0.0002	< 0.001	0.0026	350	< 0.001	1000
G271	10/08/2015	< 0.001	< 0.001	0.03	< 0.001	0.44	0.33	< 0.001		38	< 0.004	< 0.002	0.402	< 0.001	< 0.001		< 0.001	< 0.0002	0.0036	0.0035	400	< 0.001	1000
G271	11/23/2015		< 0.001	0.031	< 0.001		0.5	< 0.001	130	38	< 0.004	< 0.002	0.347		0.0012	< 0.01		< 0.0002	0.0012	0.0024	420	< 0.001	860
G271	02/16/2016	< 0.001	< 0.001	0.029	< 0.001	0.51	0.61	< 0.001	130	38	< 0.004	< 0.002	0.481	< 0.001	< 0.001	< 0.01	< 0.001	< 0.0002	< 0.001	0.0032	440	< 0.001	980
G271	05/12/2016	< 0.002	< 0.001	0.028	< 0.001	0.61	0.98	< 0.001	170	39	< 0.004	< 0.002	0.562	< 0.002	< 0.001	< 0.01	< 0.002	< 0.0002	< 0.001	0.0021	540	< 0.001	940
G271	08/05/2016		< 0.001	0.032	< 0.001		0.63	< 0.001	110	37	< 0.004	< 0.002	0.414		0.0027	< 0.01		< 0.0002	< 0.001	0.0022	440	< 0.001	840
G271	11/21/2016		< 0.001	0.031	< 0.001		0.4	< 0.001		29	< 0.004	< 0.002	0.484		< 0.001	< 0.01		< 0.0002	< 0.001	0.0029	400	< 0.001	910
G272	09/22/2009		0.0012	0.079	< 0.001		0.06	< 0.0005	84	53	< 0.005	0.0017	0.48		0.0039			< 0.0002	<u>_</u>	< 0.0025	120	< 0.002	570
G272	11/10/2009	< 0.001	< 0.001	0.073	< 0.001	0.057	< 0.05	< 0.0005	88	46	< 0.005	0.0012	0.5	< 0.0005	0.0025		0.056	< 0.0002		< 0.0025	130	< 0.002	610
G272	01/19/2010	< 0.001	< 0.001	0.068	< 0.001	< 0.05	0.051	< 0.0005	85	45	< 0.005	< 0.001	0.42	< 0.0005	0.002		0.006	< 0.0002		< 0.0025	160	< 0.002	610
G272	03/04/2010	< 0.001	< 0.001	0.061	< 0.001	< 0.05	0.05	< 0.0005	93	44	< 0.005	< 0.001	0.41	< 0.0005	0.00058		< 0.0025	< 0.0002		< 0.0025	160	< 0.002	630
G272	05/27/2010	< 0.001	< 0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	92	46	0.0049	< 0.002	0.41	< 0.001	0.0011		0.0029	< 0.0002		0.0021	190	< 0.001	670
G272	07/27/2010	< 0.001	< 0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	92	44	< 0.004	< 0.002	0.43	< 0.001	< 0.001		0.0019	< 0.0002		0.0039	210	< 0.001	690

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G272	09/20/2010	< 0.001	. 0.001			< 0.01	10.01	. 0.001	0.4	42				< 0.001	.0.001		< 0.001				200		670
G272	11/16/2010	< 0.001	< 0.001	0.07	< 0.001	< 0.01	< 0.01	< 0.001	84	42	< 0.004	< 0.002	0.49	< 0.001	< 0.001		< 0.001	< 0.0002		0.0042	200	< 0.001	670
G272 G272	01/31/2011	< 0.001	< 0.001	0.07	< 0.001	< 0.01	< 0.013	< 0.001	90	45	< 0.004	< 0.002	0.46	< 0.001	< 0.001		< 0.001	< 0.0002		0.0042	230	< 0.001	690
G272	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	99	< 100				< 0.001	< 0.001		< 0.001				250		740
G272	11/14/2011	< 0.001	< 0.001			< 0.01	0.01	< 0.001	98	41				< 0.001	< 0.001		< 0.001				230		670
G272	01/26/2012	< 0.001	0.001	0.068	< 0.001	< 0.01	< 0.01	< 0.001	98	43	< 0.004	< 0.002	0.459	< 0.001	< 0.001		< 0.001	< 0.0002		0.0021	240	< 0.001	660
G272	05/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	82	42				< 0.001	< 0.001		< 0.001				190		750
G272	07/24/2012	< 0.001	< 0.001			0.032	< 0.01	< 0.001	100	43				< 0.001	< 0.001		< 0.001				220		710
G272	11/14/2012	< 0.001	0.0011			< 0.01	< 0.01	< 0.001	110	40				< 0.001	0.0011		0.0012				220		780
G272	01/31/2013	< 0.001	< 0.001	0.072	< 0.001	< 0.01	< 0.01	< 0.001	110	44	< 0.004	< 0.002	0.461	< 0.001	< 0.001		< 0.001	< 0.0002		0.0029	330	< 0.001	760
G272	05/20/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	98	37				< 0.001	< 0.001		< 0.001				280		680
G272	07/22/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	100	35				< 0.001	< 0.001		< 0.001				260		680
G272	10/14/2013	< 0.001	< 0.001	0.00	< 0.001	< 0.01	< 0.01	< 0.001	100	32	< 0.004	< 0.002	0.255	< 0.001	< 0.001		< 0.001	< 0.0000		< 0.001	300	< 0.001	/40
G272	02/19/2014	< 0.001	< 0.001	0.06	< 0.001	< 0.01	< 0.01	< 0.001	02	40	< 0.004	< 0.002	0.355	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	210	< 0.001	
G272	03/13/2014	0.17	< 0.001			0.27	0.015	< 0.001	95 110	30				< 0.001	0.001		0.23				330		740
G272	10/14/2014	< 0.001	< 0.001			0.036	0.022	< 0.001	110	34				< 0.001	0.0014		< 0.001				310		840
G272 G272	01/21/2015	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	99	31				< 0.001	0.0020		< 0.001				380		790
G272	04/10/2015	< 0.002	< 0.001	0.059	< 0.001	< 0.02	< 0.01	< 0.001	110	37	< 0.004	< 0.002	0.399	< 0.002	< 0.001		< 0.002	< 0.0002		0.0012	340	< 0.001	800
G272	07/23/2015	< 0.001	< 0.001	0.06	< 0.001	< 0.01	0.031	< 0.001		29	< 0.004	< 0.002	0.493	< 0.001	< 0.001		< 0.001	< 0.0002	< 0.001	< 0.001	270	< 0.001	840
G272	10/08/2015	< 0.001	< 0.001	0.058	< 0.001	< 0.01	0.046	< 0.001		33	< 0.004	< 0.002	0.361	< 0.001	< 0.001		< 0.001	< 0.0002	0.0024	0.0016	340	< 0.001	660
G272	02/09/2016	< 0.001	0.0013	0.099	< 0.001	< 0.01	< 0.01	< 0.001		29	0.011	0.0022	0.516	< 0.001	0.0049		0.0013	< 0.0002	0.0013	< 0.001	290	< 0.001	660
G272	05/12/2016	< 0.001	< 0.001	0.059	< 0.001	< 0.01	< 0.01	< 0.001		29	< 0.004	< 0.002	0.561	< 0.001	< 0.001		< 0.001	< 0.0002	< 0.001	< 0.001	310	< 0.001	680
G273	09/23/2009		< 0.001	0.094	< 0.001		< 0.05	< 0.0005	120	35	< 0.005	< 0.001	0.38		< 0.0005			< 0.0002		< 0.0025	340	< 0.002	890
G273	11/10/2009	< 0.001	< 0.001	0.09	< 0.001	< 0.05	0.051	< 0.0005	140	28	< 0.005	< 0.001	0.33	< 0.0005	0.00093		0.097	< 0.0002		< 0.0025	400	< 0.002	980
G273	01/21/2010	< 0.001	< 0.001	0.085	< 0.001	0.054	< 0.05	< 0.0005	150	30	< 0.005	< 0.001	0.29	< 0.0005	< 0.0005		0.063	< 0.0002		< 0.0025	560	< 0.002	1200
G273	03/04/2010	< 0.001	< 0.001	0.079	< 0.001	< 0.05	< 0.05	< 0.0005	190	25	< 0.005	< 0.001	0.26	< 0.0005	< 0.0005		0.055	< 0.0002		< 0.0025	570	< 0.002	1300
G273	05/27/2010	< 0.001	0.0011	0.055	< 0.001	0.019	0.016	< 0.001	180	31	0.034	< 0.002	0.33	< 0.001	< 0.001		0.041	< 0.0002		0.0016	620	< 0.001	1300
G273	07/27/2010	< 0.001	< 0.001	0.048	< 0.001	0.023	0.023	< 0.001	160	30	< 0.004	< 0.002	0.37	< 0.001	< 0.001		0.048	< 0.0002		< 0.001	490	< 0.001	1100
G273	09/20/2010	< 0.001				0.024								< 0.001			0.052						
G273	11/16/2010	< 0.001	< 0.001	0.05	. 0.001	0.035	0.039	< 0.001	130	27		. 0. 002	0.20	< 0.001	< 0.001		0.058	. 0. 0002		.0.001	420	.0.001	960
G273	01/31/2011	< 0.001	0.001	0.05	< 0.001	0.16	0.21	< 0.001	1/0	33	< 0.004	< 0.002	0.38	< 0.001	< 0.001		0.047	< 0.0002		< 0.001	520	< 0.001	1100
G273	05/03/2011	< 0.001	< 0.001			0.10	0.14	< 0.001	150	20				< 0.001	< 0.001		0.042				510		1200
G273	11/14/2011	< 0.001	0.001			0.097	0.12	< 0.001	150	29				< 0.001	< 0.001		0.023				510		000
G273	01/26/2012	< 0.001	0.0013	0.043	< 0.001	0.13	0.15	< 0.001	130	23	< 0.004	< 0.002	0 359	< 0.001	< 0.001		0.041	< 0.0002		0.0012	750	< 0.001	1300
G273	05/22/2012	< 0.001	< 0.001	0.015		0.21	0.2	< 0.001	160	27	10.001	10.002	0.555	< 0.001	< 0.001		0.028			0.0012	470	.0.001	1100
G273	07/24/2012	< 0.001	< 0.001			0.12	0.094	< 0.001	140	32				< 0.001	< 0.001		0.022				360		910
G273	11/14/2012	0.0016	0.0034			0.27	0.2	< 0.001	160	33				< 0.001	0.0037		0.026				630		1100
G273	01/31/2013	< 0.001	< 0.001	0.046	< 0.001	0.48	0.46	< 0.001	180	37	< 0.004	< 0.002	0.33	< 0.001	< 0.001		0.03	< 0.0002		< 0.002	740	< 0.001	1300
G273	05/20/2013	0.0014	< 0.001			0.18	0.29	< 0.001	180	4				< 0.001	< 0.001		0.013				670		1100
G273	07/22/2013	< 0.001	< 0.001			0.21	0.25	< 0.001	160	35				< 0.001	< 0.001		0.022				510		980
G273	10/14/2013	< 0.001	< 0.001			0.18	0.18	< 0.001	140	37				< 0.001	< 0.001		0.015				450		900
G273	02/19/2014	< 0.001	< 0.001	0.039	< 0.001	0.52	0.36	< 0.001	150	38	< 0.004	< 0.002	0.286	< 0.001	< 0.001		0.015	< 0.0002	ļ	< 0.001	570	< 0.001	
G273	05/13/2014	0.24	< 0.001			0.62	0.35	< 0.001	160	47	ļ		ļ	0.22	< 0.001		0.33	ļ			620		<u> </u>
G273	08/11/2014	< 0.001	< 0.001			0.25	0.26	< 0.001	140	37	-		ļ	< 0.001	< 0.001		0.01	-	ļ		530		1000
G273	10/14/2014	< 0.001	0.0011			0.32	0.29	< 0.001	150	37			<u> </u>	< 0.001	0.0011		0.013				500		1100
G273	01/21/2015	< 0.001	< 0.001	0.020	10.001	0.58	0.45	< 0.001	150	46	10.004	10.000	0.22	< 0.001	< 0.001		0.018	10.0000		10.001	650	10.001	1200
G273	07/22/2015	< 0.002	< 0.001	0.028	< 0.001	0.29	0.48	< 0.001	200	41	< 0.004	< 0.002	0.32	< 0.002	< 0.001		0.019	< 0.0002	< 0.001	< 0.001	690 200	< 0.001	1300
G273	10/08/2015	< 0.001	< 0.001	0.044	< 0.001	0.4	0.12	< 0.001		39	< 0.004		0.382	< 0.001	< 0.001		0.0011		0.001	< 0.001	390	< 0.001	1200
G273	11/24/2015	< 0.001	< 0.001	0.059	< 0.001	0.1	0.12	< 0.001	1/0	40	< 0.004		< 0.25	< 0.001	0.001	< 0.01	0.014		< 0.0019	< 0.001	430		920
J2/3	11/24/2013		< 0.001	0.049	< 0.001	I	0.2	< 0.001	140	41	< 0.004	< 0.00Z	< 0.25		0.0011	< 0.01	L	< 0.000Z	< 0.001	< 0.001	420	< 0.001	050

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G273	02/16/2016	< 0.001	< 0.001	0.043	< 0.001	0.43	0.42	< 0.001	150	45	< 0.004	< 0.002	0.401	< 0.001	< 0.001	< 0.01	0.01	< 0.0002	< 0.001	< 0.001	550	< 0.001	1200
G273	05/12/2016	< 0.001	< 0.001	0.031	< 0.001	0.31	0.29	< 0.001	170	44	< 0.004	< 0.002	0.537	< 0.001	< 0.001	< 0.01	0.012	< 0.0002	< 0.001	< 0.001	520	< 0.001	980
G273	08/05/2016		< 0.001	0.032	< 0.001		0.17	< 0.001	120	46	< 0.004	< 0.002	0.294		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	400	< 0.001	840
G273	11/21/2016		< 0.001	0.036	< 0.001		0.15	< 0.001		48	< 0.004	< 0.002	0.39		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	440	< 0.001	900
G274	09/24/2009		0.0024	0.12	< 0.001		0.053	< 0.0005	100	55	0.0059	0.0028	0.34		0.0091			< 0.0002		< 0.0025	230	< 0.002	830
G274	11/11/2009	< 0.001	< 0.001	0.092	< 0.001	0.057	< 0.05	< 0.0005	100	54	< 0.005	< 0.001	0.35	< 0.0005	0.0012		0.007	< 0.0002		< 0.0025	250	< 0.002	820
G274	01/2//2010	< 0.001	< 0.001	0.09	< 0.001	< 0.05	< 0.05	< 0.0005	100	50	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.0039	< 0.0002		< 0.0025	260	< 0.002	850
G274	03/08/2010	< 0.001	< 0.001	0.091	< 0.001	< 0.05	< 0.05	< 0.0005	110	49	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.0056	< 0.0002		< 0.0025	270	< 0.002	8/0
G274	03/27/2010	< 0.001	< 0.001	0.09	< 0.001	< 0.01	< 0.01	< 0.001	110	45	< 0.004	< 0.002	0.52	< 0.001	0.0015		< 0.001	< 0.0002		0.0022	220	< 0.001	910
G274	09/20/2010	< 0.001	< 0.001	0.08	< 0.001	< 0.01	< 0.01	< 0.001	110	44	< 0.004	< 0.002	0.55	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	320	< 0.001	300
G274	11/16/2010	< 0.001	< 0.001			< 0.01	0.012	< 0.001	120	38				< 0.001	< 0.001		< 0.001				360		940
G274	01/31/2011	< 0.001	0.0023	0.077	0.0018	< 0.01	0.012	0.0014	110	39	< 0.004	< 0.002	0.36	< 0.001	0.0018		< 0.001	< 0.0002		< 0.001	370	0.0013	950
G274	05/03/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	120	36				< 0.001	< 0.001		< 0.001				400		980
G274	07/27/2011	< 0.001	< 0.001			0.013	< 0.01	< 0.001	120	37				< 0.001	< 0.001		< 0.001				370		980
G274	11/14/2011	< 0.001	< 0.001			0.035	0.039	< 0.001	120	37				< 0.001	< 0.001		< 0.001				370		900
G274	01/26/2012	< 0.001	< 0.001	0.068	< 0.001	0.076	0.052	< 0.001	110	34	< 0.004	< 0.002	0.379	< 0.001	< 0.001		0.001	< 0.0002		0.0014	370	< 0.001	880
G274	05/22/2012	< 0.001	< 0.001			0.12	0.11	< 0.001	120	36				< 0.001	< 0.001		< 0.001				330		920
G274	07/24/2012	< 0.001	< 0.001			0.15	0.13	< 0.001	120	40				< 0.001	< 0.001		< 0.001				300		880
G274	11/14/2012	< 0.001	0.0022			0.18	0.16	< 0.001	120	37				< 0.001	0.0031		< 0.001				420		910
G274	01/31/2013	< 0.001	< 0.001	0.059	< 0.001	0.26	0.25	< 0.001	130	36	< 0.004	< 0.002	0.382	< 0.001	< 0.001		< 0.001	< 0.0002		0.0015	460	< 0.001	870
G274	05/20/2013	0.001	< 0.001			0.25	0.27	< 0.001	120	3.6				< 0.001	< 0.001		0.0073				350		800
G274	07/22/2013	< 0.001	< 0.001			0.31	0.31	< 0.001	110	32				< 0.001	0.0015		< 0.001				330		820
G274	10/14/2013	< 0.001	< 0.001			0.52	0.46	< 0.001	110	31				< 0.001	< 0.001		< 0.001				380		840
G274	02/19/2014	< 0.001	< 0.001	0.063	< 0.001	0.79	0.47	< 0.001	110	32	< 0.004	< 0.002	0.262	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	300	< 0.001	
G274	05/13/2014	0.0055	< 0.001			0.52	0.52	< 0.001	98	27				0.0034	< 0.001		0.0059				370		000
G274	10/14/2014	< 0.001	< 0.001			0.57	0.61	< 0.001	120	25				< 0.001	< 0.001		< 0.001				220		880 770
G274	01/21/2014	< 0.001	0.004			0.02	0.55	< 0.001	110	10				< 0.001	0.01		0.001				260		770
G274	04/13/2015	< 0.001	< 0.0011	0.052	< 0.001	0.57	0.53	< 0.001	110	24	< 0.004	< 0.002	0.341	< 0.001	< 0.001		< 0.002	< 0.0002		< 0.001	390	< 0.001	770
G274	07/23/2015	< 0.001	< 0.001	0.068	< 0.001	0.48	0.49	< 0.001		24	< 0.004	< 0.002	0.403	< 0.001	0.0018		0.0015	< 0.0002	< 0.001	0.0014	320	< 0.001	890
G274	10/08/2015	< 0.001	< 0.001	0.061	< 0.001	0.43	0.74	< 0.001		22	< 0.004	< 0.002	0.265	< 0.001	< 0.001		< 0.001	< 0.0002	0.0012	< 0.001	320	< 0.001	770
G274	02/09/2016	< 0.001	< 0.001	0.07	< 0.001	0.87	0.65	< 0.001		22	< 0.004	< 0.002	0.455	< 0.001	0.001		< 0.001	< 0.0002	< 0.001	< 0.001	290	< 0.001	820
G274	05/12/2016	< 0.001	< 0.001	0.057	< 0.001	0.97	0.63	< 0.001		22	< 0.004	< 0.002	0.417	< 0.001	< 0.001		0.046	< 0.0002	< 0.001	< 0.001	350	< 0.001	770
G275	09/22/2009		< 0.005	0.12	< 0.001		4.5	< 0.0005	300	14	< 0.025	< 0.005	0.22		0.01			< 0.0002		< 0.012	990	< 0.002	2000
G275	11/11/2009	< 0.001	< 0.001	0.032	< 0.001	2.4	2.5	< 0.0005	140	12	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.032	< 0.0002		< 0.0025	350	< 0.002	910
G275	01/21/2010	< 0.001	< 0.001	0.032	< 0.001	1.2	1.2	< 0.0005	120	13	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.023	< 0.0002		< 0.0025	390	< 0.002	870
G275	03/08/2010	< 0.001	< 0.001	0.035	< 0.001	1.3	1.3	< 0.0005	160	23	< 0.005	< 0.001	0.27	< 0.0005	0.00053		0.029	< 0.0002		< 0.0025	460	< 0.002	1100
G275	05/28/2010	< 0.001	0.0016	0.047	< 0.001	2.9	2.9	< 0.001	180	18	0.0087	< 0.002	0.33	< 0.001	0.0017		0.018	< 0.0002		0.0014	540	< 0.001	1200
G275	07/26/2010	< 0.001	< 0.001	0.038	< 0.001	2.9	2.8	< 0.001	180	9.9	< 0.004	< 0.002	0.4	< 0.001	< 0.001		0.019	< 0.0002		< 0.001	550	< 0.001	1200
G275	09/20/2010	< 0.001	0.0000			2.6		0.004	200	0.7				< 0.001	0.0045		0.026				070		4700
G275	11/16/2010	< 0.001	0.0028	0.046	10.001	3./	4.1	< 0.001	260	9.7	10.004	10.000	0.24	< 0.001	0.0045		0.16	(0.0002		10.001	970	10.001	1/00
6275	01/31/2011	< 0.001	0.0015	0.046	< 0.001	3.6	3.b 2.0	< 0.001	230	11	< 0.004	< 0.002	0.34	< 0.001	< 0.001		0.016	< 0.0002		< 0.001	840	< 0.001	1500
G275	05/03/2011	< 0.001	< 0.001			3.0	3.8 17	< 0.001	240	13 0 2			+	< 0.001	< 0.001		0.04	<u> </u>			790		1200
G275	11/1//2011	0.001	0.001			4.5	4.Z	< 0.001	200	5.2 17				< 0.001	0.001		0.05				820		1500
G275	01/31/2011	< 0.0013	0.0011	0.039	< 0.001	3.8	3.6	< 0.001	310	15	0.0047	< 0.002	0.281	< 0.001	< 0.001		0.0081	< 0.0002		0.0015	370	< 0.001	1300
G275	05/22/2012	< 0.001	0.0017	0.035	. 0.001	3.7	3.4	< 0.001	240	11	0.0047	. 0.002	0.201	< 0.001	0.0015		0.01	. 0.0002		0.0010	670		1500
G275	07/24/2012	< 0.001	0.0018			4.6	4.2	< 0.001	260	13				< 0.001	0.0024		0.09		1		900		1600
G275	11/14/2012	< 0.001	0.0025		1	3.7	3.9	< 0.001	270	19			1	< 0.001	0.0037		0.28	1			950		1600
G275	05/20/2013	0.0013	0.0025			3.4	3.8	< 0.001	250	24				< 0.001	0.0015		< 0.001	1			840		1400
G275	07/22/2013	< 0.001	< 0.001			3.5	3.1	< 0.001	210	19				< 0.001	0.0013		0.0017				700		1400
G275	05/13/2014	0.25	0.0019			2.8	3.4	< 0.001	210	20				0.2	0.0037		0.34				750		
																							Total
--------------	------------	-----------	----------	---------	------------	-----------	--------	----------	-----------	-----------	-----------	---------	-----------	-----------	----------	----------------	------------	----------	-------------	-----------	----------	-----------	-----------
		Arsenic.	Arsenic.	Barium.	Bervllium.	Boron.	Boron.	Cadmium.	Calcium.	Chloride.	Chromium.	Cobalt.	Fluoride.	Lead.	Lead.		Manganese.	Mercurv.	Molvbdenum.	Selenium.	Sulfate.	Thallium.	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G275	08/11/2014	< 0.001	0.0043			4.2	4.1	< 0.001	240	20				< 0.001	0.0078		< 0.001				880		1500
G275	10/14/2014	< 0.001	0.0011			1.9	3.5	< 0.001	200	16				< 0.001	0.0012		0.003				500		840
G275	01/21/2015	< 0.001	0.0043	0.050	0.001	4.6	4.6	< 0.001	230	20	0.004	0.000	0.05	< 0.001	0.0079		0.025	0.0000		0.0040	940	0.004	1500
G275	04/13/2015	< 0.002	< 0.001	0.056	< 0.001	0.91	1.8	< 0.001	180	22	< 0.004	< 0.002	< 0.25	< 0.002	< 0.001		0.0024	< 0.0002	0.0014	0.0012	650	< 0.001	1500
G275	07/23/2015	< 0.001	< 0.001	0.035	< 0.001	3	4	< 0.001		30	< 0.004	< 0.002	0.307	< 0.001	0.001		0.012	< 0.0002	0.0014	0.0014	/50	< 0.001	1500
G275 G275	05/12/2016	< 0.001	< 0.001	0.042	< 0.001	2.5	2.3	< 0.001		14	< 0.004	< 0.002	0.452	< 0.0044	< 0.0010		0.0036	< 0.0002	< 0.001	< 0.001	310	< 0.001	1300
G276	09/23/2009		< 0.005	0.16	0.0026		0.087	< 0.0025	94	38	< 0.025	< 0.005	0.75		0.035			< 0.0002		< 0.012	170	< 0.002	620
G276	11/11/2009	< 0.001	< 0.001	0.079	< 0.001	0.057	0.062	< 0.0005	91	40	0.0059	< 0.001	0.57	< 0.0005	0.0037		0.03	< 0.0002		< 0.0025	170	< 0.002	670
G276	01/21/2010	< 0.001	< 0.001	0.075	< 0.001	0.057	0.052	< 0.0005	91	36	< 0.005	< 0.001	0.53	< 0.0005	0.00079		0.0093	< 0.0002		< 0.0025	190	< 0.002	660
G276	03/09/2010									38			0.51								180		650
G276	03/10/2010	< 0.001	< 0.001	0.083	< 0.001	0.057	0.058	< 0.0005	110		< 0.005	< 0.001		< 0.0005	0.0016		0.01	< 0.0002		< 0.0025		< 0.002	
G276	05/28/2010	< 0.001	< 0.001	0.067	< 0.001	0.035	0.033	< 0.001	100	40	0.0051	< 0.002	0.49	< 0.001	< 0.001		0.012	< 0.0002		0.0014	190	< 0.001	720
G276	07/26/2010	< 0.001	< 0.001	0.063	< 0.001	0.023	0.022	< 0.001	93	40	< 0.004	< 0.002	0.54	< 0.001	< 0.001		0.0054	< 0.0002		< 0.001	230	< 0.001	710
G276	09/20/2010	< 0.001	< 0.001			0.024	0.044	< 0.001	06	25				< 0.001	< 0.001		0.005				200		670
G276	01/31/2011	< 0.001	0.001	0.078	< 0.001	0.013	0.044	< 0.001	90	36	< 0.004	< 0.002	0.53	< 0.001	0.001		0.0018	< 0.0002		< 0.001	200	< 0.001	710
G276	05/03/2011	< 0.001	< 0.001	0.070		0.043	0.032	< 0.001	95	36	10.001	0.002	0.55	< 0.001	< 0.001		0.0016	10.0002			200	10.001	650
G276	07/27/2011	< 0.001	< 0.001			0.011	< 0.01	< 0.001	96	37				< 0.001	< 0.001		0.001				170		670
G276	11/14/2011	< 0.001	0.0013			0.081	0.08	< 0.001	94	35				< 0.001	< 0.001		< 0.001				180		620
G276	01/31/2012	< 0.001	< 0.001	0.075	< 0.001	< 0.01	0.06	< 0.001	130	32	0.0052	< 0.002	0.501	< 0.001	< 0.001		< 0.001	< 0.0002		0.0022	190	< 0.001	650
G276	05/22/2012	< 0.001	0.0011			0.066	0.073	< 0.001	97	38				< 0.001	< 0.001		0.0014				160		660
G276	07/24/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	100	38				< 0.001	< 0.001		< 0.001				190		680
G276	11/14/2012	< 0.001	0.0015	0.004	0.001	0.012	0.083	< 0.001	100	36	0.004	0.000	0.54	< 0.001	< 0.001		< 0.001	0.0000		0.0001	190	0.004	680
G276	01/31/2013	< 0.001	< 0.001	0.081	< 0.001	0.014	0.023	< 0.001	110	35	< 0.004	< 0.002	0.51	< 0.001	0.0011		< 0.001	< 0.0002		0.0031	250	< 0.001	680
G276	05/20/2013	< 0.001	< 0.001			0.030	0.031	< 0.001	99 100	3.8				< 0.001	< 0.001		< 0.001				220		670
G276	05/13/2014	< 0.001	0.0013			< 0.022	0.021	< 0.001	130	31				< 0.001	< 0.001		< 0.001				220		070
G276	08/12/2014	< 0.001	0.0013			0.018	0.027	< 0.001	120	29				< 0.001	0.006		< 0.001				220		640
G276	10/14/2014	< 0.001	< 0.001			0.028	0.019	< 0.001	100	28				< 0.001	0.0024		< 0.001				220		700
G276	01/21/2015	< 0.001	< 0.001			0.015	0.021	< 0.001	100	30				< 0.001	< 0.001		< 0.001				260		700
G276	04/13/2015	< 0.002	0.0057	0.34	0.0016	< 0.02	0.036	< 0.001	170	34	0.043	0.0047	0.486	< 0.002	0.022		< 0.002	< 0.0002		0.0034	310	< 0.001	780
G276	07/23/2015	< 0.001	< 0.001	0.096	< 0.001	0.037	0.015	< 0.001		26	< 0.004	< 0.002	0.377	< 0.001	0.0012		< 0.001	< 0.0002	0.0012	0.001	180	< 0.001	800
G276	11/24/2015		< 0.001	0.077	< 0.001		0.043	< 0.001	120	28	< 0.004	< 0.002	0.345		< 0.001	0.013		< 0.0002	0.0017	< 0.001	190	< 0.001	710
G276	02/16/2016	< 0.001	< 0.001	0.09	< 0.001	0.027	0.021	< 0.001	120	23	< 0.004	< 0.002	0.456	< 0.001	0.0014	0.015	< 0.001	< 0.0002	0.0013	0.0018	230	< 0.001	760
G276	02/17/2016	< 0.001	< 0.001	0.089	< 0.001	0.027	0.029	< 0.001	120	28	< 0.004	< 0.002	0.450	< 0.001	< 0.001	0.012	< 0.001	< 0.0002	< 0.001	< 0.001	230	< 0.001	700
G276	08/03/2016	< 0.001	< 0.001	0.075	< 0.001	0.015	0.019	< 0.001	110	22	< 0.004	< 0.002	0.443	< 0.001	< 0.001	< 0.012	< 0.001	< 0.0002	< 0.001	0.0017	1900	< 0.001	680
G276	11/21/2016		< 0.001	0.081	< 0.001		< 0.01	< 0.001		23	< 0.004	< 0.002	0.445		< 0.001	0.011		< 0.0002	< 0.001	0.002	210	< 0.001	720
G277	09/23/2009		0.027	0.61	0.0027		0.11	< 0.0025	190	41	0.052	0.04	0.79		0.072			0.00023		< 0.012	79	< 0.002	
G277	09/24/2009																						430
G277	11/11/2009	0.021	0.02	0.22	0.0011	< 0.05	0.074	< 0.0005	110	43	0.037	0.017	0.47	0.028	0.029		1.6	< 0.0002		< 0.0025	49	< 0.002	680
G277	01/19/2010	< 0.001	0.0019	0.093	< 0.001	< 0.05	0.062	< 0.0005	92	39	0.0094	0.0019	0.41	< 0.0005	0.0037		0.02	< 0.0002		< 0.0025	63	< 0.002	
G277	01/20/2010																						550
G277	03/08/2010		< 0.001	0.075	< 0.001	ļ	0.065	< 0.0005	93	43	< 0.005	< 0.001	0.4		0.00092			< 0.0002		< 0.0025	49	< 0.002	540
G277	03/09/2010	< 0.001	0.0000	0.000		0.073	0.021	10.001	400	4.5	0.000		0.1	< 0.0005	0.0001		0.0037			0.0000	50	10.001	500
62//	05/28/2010	< 0.001	0.0023	0.092	< 0.001	0.02	0.021	< 0.001	100	46	0.033	< 0.002	0.4	< 0.001	0.0031		0.0011	< 0.0002		0.0032	58	< 0.001	580
G277	07/20/2010	< 0.001	0.0018	0.085	< 0.001	0.015	0.019	< 0.001	85	40	0.0092	< 0.002	0.42	< 0.001	0.0029			< 0.0002		0.0021	58	< 0.001	790
G277	11/16/2010	< 0.001	< 0.001			< 0.012	< 0.01	< 0.001	86					< 0.001	< 0.001		0.0021						
G277	01/31/2011	< 0.001	0.0015	0.085	< 0.001	0.016	0.025	< 0.001	97	40	< 0.004	< 0.002	0.38	< 0.001	0.0024		0.0031	< 0.0002		0.003	52	< 0.001	610
G277	05/03/2011	< 0.001	< 0.001			0.02	0.02	< 0.001	100	42				< 0.001	< 0.001		< 0.001				56		600
G277	07/27/2011	< 0.001	< 0.001			0.012	< 0.01	< 0.001	110	49				< 0.001	< 0.001		< 0.001				61		650

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G277 11,	1/14/2011	< 0.001	0.0039			0.031	0.044	< 0.001	110	36				< 0.001	0.0056		0.0054				51		580
G277 01,	1/31/2012		0.003	0.12	< 0.001		0.027	< 0.001	160	37	0.012	0.0022	0.326		0.0035			< 0.0002		0.0014	45	< 0.001	
G277 05,	5/22/2012	< 0.001	< 0.001			0.021	0.026	< 0.001	110	30				< 0.001	< 0.001		0.0052				49		580
G277 07/	7/24/2012									34											52		580
G277 07/	7/25/2012	< 0.001	0.0011			< 0.01	< 0.01	< 0.001	110				-	< 0.001	0.0011		0.013						
G277 11/	1/14/2012		0.001				< 0.01	< 0.001	120	11					0.0012						42		
G277 05/	5/20/2013	< 0.001	< 0.001			0.022	0.018	< 0.001	110	3.3				< 0.001	< 0.001		< 0.001				5.7		520
G277 07/	//22/2013	< 0.001	< 0.001			0.02	0.021	< 0.001	93	18				< 0.001	< 0.001		< 0.001				45		500
G277 05/	D/15/2014	< 0.001	0.0021			0.012	0.053	< 0.001	130	30				< 0.001	0.0019		< 0.001				10		
G_{277} 10/	0/15/2014 0/16/2014	< 0.001	0.0025			0.032	0.024	< 0.001	04	5.0				< 0.001	0.0055		0.003				10		360
G277 05	5/12/2014	< 0.001	0.0011	0.09	< 0.001	0.047	0.025	< 0.001		1.2	< 0.004	< 0.002	0.495	< 0.001	0.0013		0.0017	< 0.0002	0.001	< 0.001	15	< 0.001	460
G278 05	5/26/2010		0.0011	0.078	< 0.001		0.026	< 0.001	93	82	0.016	< 0.002	0.41		< 0.001			< 0.0002		0.0058	87	< 0.001	640
G278 05,	5/28/2010	< 0.001	0.0011	0.070	.0.001	0.012	0.020	.0.001	55	02	0.010	4 0.00L	0.11	< 0.001	.0.001		0.052	10.0002		0.0000	0,	.0.001	010
G278 03	3/23/2011	< 0.001	0.0026	0.069	< 0.001	0.014	0.014	< 0.001	90	55	0.0063	< 0.002	0.55	< 0.001	0.0025		0.1	< 0.0002		0.0019	91	< 0.001	630
G278 05	5/03/2011	< 0.001	0.0056	0.15	< 0.001	< 0.01	0.026	< 0.001	97	60	0.022	0.0055	0.43	< 0.001	0.013		0.027	< 0.0002		0.0032	110	< 0.001	670
G278 07	7/25/2011	< 0.001	0.016	0.34	0.0018	< 0.01	0.023	< 0.001	170	62	0.034	0.018	0.42	< 0.001	0.033		0.032	< 0.0002		0.0029	100	< 0.001	690
G278 09,	9/19/2011	< 0.001	0.0018	0.086	< 0.001	0.093	0.023	< 0.001	100	85	0.048	< 0.002	0.51	< 0.001	0.0031		0.04	< 0.0002		0.012	120	< 0.001	640
G278 05,	5/22/2012	< 0.001	0.0032	0.098	< 0.001	0.017	0.017	< 0.001	110	82	0.007	0.0028	0.359	< 0.001	0.0059		0.46	< 0.0002		0.0025	120	< 0.001	730
G278 02,	2/09/2016	< 0.001	0.004	0.11	< 0.001	0.54	0.54	< 0.001		310	0.0092	0.0034	0.336	< 0.001	0.008		0.14	< 0.0002	0.0014	0.041	680	< 0.001	1800
G278 05/	5/13/2016	< 0.001	0.0067	0.19	< 0.001	0.5	0.2	< 0.001		180	0.017	0.0066	0.441	< 0.001	0.015		0.39	< 0.0002	0.0026	0.014	450	< 0.001	1500
G279 09;	9/23/2009		0.0064	0.095	< 0.001		0.062	< 0.0005	90	59	0.01	0.0059	0.5		0.0092			< 0.0002		< 0.0025	99	< 0.002	620
G279 11,	1/09/2009	< 0.001	< 0.001	0.065	< 0.001	0.062	0.056	< 0.0005	81	50	< 0.005	< 0.001	0.5	< 0.0005	0.00068		0.26	< 0.0002		< 0.0025	92	< 0.002	620
G279 01/	1/27/2010	< 0.001	< 0.001	0.067	< 0.001	< 0.05	0.057	< 0.0005	80	54	< 0.005	< 0.001	0.43	< 0.0005	0.0013		0.094	< 0.0002		< 0.0025	88	< 0.002	630
G279 03/	3/04/2010	< 0.001	< 0.001	0.065	< 0.001	< 0.05	< 0.05	< 0.0005	85	57	< 0.005	< 0.001	0.42	< 0.0005	0.00068		0.059	< 0.0002		< 0.0025	83	< 0.002	610
G279 05/	5/26/2010	< 0.001	< 0.001	0.068	< 0.001	< 0.01	0.01	< 0.001	89	69	0.011	< 0.002	0.42	< 0.001	< 0.001		0.022	< 0.0002		0.005	100	< 0.001	680
G279 07/	//26/2010	< 0.001	< 0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	91	64	< 0.004	< 0.002	0.43	< 0.001	< 0.001		0.019	< 0.0002		0.0015	88	< 0.001	670
$G_{279} = 09/$	1/16/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	<u>00</u>	52				< 0.001	< 0.001		0.014				06		600
G279 01	1/28/2010	< 0.001	0.0026	0 098	< 0.001	< 0.01	0.016	< 0.001	100	50	< 0.004	0 0029	0.44	< 0.001	0.001		0.0088	< 0.0002		0.0038	90	< 0.001	600
G279 01,	5/04/2011	< 0.001	< 0.001	0.050	× 0.001	< 0.01	< 0.010	< 0.001	130	49	× 0.00+	0.0025	0.44	< 0.001	< 0.0052		0.0033	< 0.0002		0.0050	100	×0.001	630
G279 07	7/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	83	51				< 0.001	< 0.001		0.0071				95		600
G279 11	1/14/2011	< 0.001	< 0.001			0.017	0.019	< 0.001	84	53				< 0.001	< 0.001		0.0035				95		580
G279 01	1/30/2012	< 0.001	< 0.001	0.05	< 0.001	< 0.01	< 0.01	< 0.001	120	51	< 0.004	< 0.002	0.54	< 0.001	< 0.001		0.0049	< 0.0002		0.0041	160	< 0.001	630
G279 05,	5/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	69	54				< 0.001	< 0.001		0.042				96		590
G279 07,	7/24/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	98	71				< 0.001	< 0.001		0.002				130		660
G279 11,	1/14/2012	0.0019	0.0018			< 0.01	< 0.01	< 0.001	97	62				< 0.001	0.001		0.019				140		750
G279 01,	1/31/2013	0.0014	< 0.001	0.06	< 0.001	< 0.01	< 0.01	< 0.001	120	60	< 0.004	< 0.002	0.418	< 0.001	< 0.001		0.0019	< 0.0002		0.0034	390	< 0.001	1200
G279 05/	5/20/2013	< 0.001	0.0012			0.01	0.011	< 0.001	87	56				< 0.001	< 0.001		0.0011				180		600
G279 07/	7/22/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	88	55				< 0.001	< 0.001		0.0078				140		560
G279 10/	0/14/2013	< 0.001	< 0.001			0.015	0.011	< 0.001	97	55				< 0.001	< 0.001		0.078				120		640
G279 02/	2/19/2014	< 0.001	< 0.001	0.058	< 0.001	0.022	0.015	< 0.001	86	54	< 0.004	< 0.002	0.331	< 0.001	< 0.001		0.029	< 0.0002		< 0.001	110	< 0.001	
G279 05/	5/13/2014	0.017	< 0.001			0.038	0.016	< 0.001	94	50				0.022	< 0.001		0.034				110		
G279 08/	0/12/2014	< 0.001	< 0.001			0.015	0.11	< 0.001	97	50				< 0.001	< 0.001		0.024				140		600
G279 10/	1/21/2014	< 0.001	< 0.001			0.025	0.012	< 0.001	92	03 7/				< 0.001	< 0.001		0.020				220		050 910
$G_{279} = 01/01/01/01$	1/21/2013	< 0.001	0.0013	0.029	< 0.001	0.032	0.031	< 0.001	100	46	< 0.004	< 0.002	0 518	< 0.001	< 0.001		0.017	0.00024		0.0056	470	< 0.001	800
G279 04/	7/23/2015	< 0.001	0.002	0.11	< 0.001	0.031	0.065	< 0.001	170	96	0.0042	0.0025	0.361	< 0.001	0.0041		0.013	< 0.00024	0.0015	0.02	470	< 0.001	1200
G279 10	0/08/2015	< 0.001	0.0015	0.096	< 0.001	1.3	1.4	< 0.001		120	0.0047	0.0033	< 0.25	< 0.001	0.0025		0.032	< 0.0002	0.0015	0.017	810	< 0.001	1700
G279 11	1/24/2015	5.001	< 0.001	0.053	< 0.001		0.63	< 0.001	140	61	< 0.004	< 0.002	0.334	0.001	0.0015	0.014	0.002	< 0.0002	< 0.001	0.0041	520	< 0.001	1100
G279 02	2/16/2016	< 0.001	< 0.001	0.082	< 0.001	0.29	0.26	< 0.001	180	130	< 0.004	< 0.002	0.392	< 0.001	< 0.001	0.012	0.011	< 0.0002	0.043	0.017	610	< 0.001	1500
G279 05	5/13/2016	< 0.001	< 0.001	0.055	< 0.001	0.11	0.073	< 0.001	120	31	< 0.004	< 0.002	0.608	< 0.001	< 0.001	< 0.01	0.0061	< 0.0002	0.024	0.0043	270	< 0.001	700
G279 08,	8/03/2016		< 0.001	0.069	< 0.001		0.24	< 0.001	210	110	< 0.004	< 0.002	0.394		< 0.001	< 0.01		< 0.0002	< 0.001	0.02	570	< 0.001	1300

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G279	11/22/2016		< 0.001	0.057	< 0.001		0.49	< 0.001		130	< 0.004	< 0.002	0.272		< 0.001	0.011		< 0.0002	< 0.001	0.017	720	< 0.001	1300
G280	03/11/2008	< 0.01	< 0.01	0.049	< 0.01	< 0.5	< 0.5	< 0.005	63	47	< 0.05	< 0.01	0.37	< 0.005	< 0.005		0.11	< 0.0002		< 0.025	60	< 0.002	420
G280	04/21/2008	< 0.001	< 0.001	0.058	< 0.001	0.06	0.062	< 0.0005	59	51	< 0.005	< 0.001	0.37	< 0.0005	0.00055		0.18	< 0.0002		0.0028	58	< 0.002	400
G280	06/11/2008	< 0.001	< 0.001	0.052	< 0.001	< 0.05	< 0.05	< 0.0005	66	43	< 0.005	< 0.001	0.33	< 0.0005	< 0.0005		0.13	< 0.0002		0.0027	62	< 0.002	430
G280	08/13/2008	< 0.001	< 0.001	0.053	< 0.001	0.05	< 0.05	< 0.0005	63	44	< 0.005	< 0.001	0.39	< 0.0005	< 0.0005		0.13	< 0.0002		0.0026	59	< 0.002	410
G280	10/13/2008	< 0.001	< 0.001	0.05	< 0.001	< 0.05	< 0.05	< 0.0005	120	45	< 0.005	< 0.001	0.35	< 0.0005	< 0.0005		0.078	< 0.0002		< 0.0025	50	< 0.002	450
G280	12/03/2008	< 0.001	< 0.001	0.11	< 0.001	< 0.05	< 0.05	< 0.0005	59	40	< 0.005	< 0.001	0.20	< 0.0005	< 0.0007		0.24	< 0.0002		< 0.0025	230 //3	< 0.002	380
G280	11/12/2009	< 0.001	< 0.001			< 0.05	< 0.05	< 0.0005	61	40				< 0.0005	< 0.0005		0.012				42		370
G280	01/28/2010	< 0.001	< 0.001			< 0.05	< 0.05	< 0.0005	60	41				< 0.0005	0.0006		0.0069				47		400
G280	02/11/2010			0.039	< 0.001						< 0.005	< 0.001	0.32					< 0.0002		0.0042		< 0.002	440
G280	06/09/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	66	65				< 0.001	< 0.001		0.0031				82		510
G280	07/27/2010	< 0.001	< 0.001			0.01	< 0.01	< 0.001	60	60				< 0.001	< 0.001		< 0.001				80		520
G280	11/16/2010	< 0.001	0.001			< 0.01	< 0.01	< 0.001	58	35				< 0.001	< 0.001		0.018				43		370
G280	01/28/2011	< 0.001	0.0012	0.082	< 0.001	< 0.01	0.012	< 0.001	100	57	< 0.004	0.0024	0.36	< 0.001	0.0055		< 0.001	< 0.0002		0.0051	76	< 0.001	500
G280	05/04/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	< 0.1	87				< 0.001	< 0.001		< 0.001				98		560
G280	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	63	45				< 0.001	< 0.001		0.015				54		420
G280	11/11/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	60	51				< 0.001	< 0.001		0.0033				59		390
G280	01/30/2012	< 0.001	< 0.001	0.047	< 0.001	< 0.01	< 0.01	< 0.001	81	54	< 0.004	< 0.002	0.44	< 0.001	< 0.001		< 0.001	< 0.0002		0.0055	68	< 0.001	440
G280	05/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	65	67				< 0.001	< 0.001		< 0.001				93		470
G280	07/24/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	72	48				< 0.001	0.0026		0.0073				51		360
G280	11/14/2012	0.0013	0.0021	0.026	< 0.001	< 0.01	< 0.01	< 0.001	63	40	< 0.004	< 0.002	0 422	< 0.001	0.0014		0.0081	< 0.0002		0.0042	48	< 0.001	280
G280	01/31/2013	< 0.001	< 0.001	0.050	< 0.001	< 0.01	< 0.01	< 0.001	67	40 50	< 0.004	< 0.00Z	0.425	< 0.001	< 0.001		< 0.001	< 0.0002		0.0042	54 71	< 0.001	560 /10
G280	07/22/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	73	52				< 0.001	< 0.001		< 0.001				67		400
G280	10/14/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	72	51				< 0.001	< 0.001		< 0.001				65		480
G280	02/19/2014	< 0.001	< 0.001	0.041	< 0.001	< 0.01	< 0.01	< 0.001	69	56	< 0.004	< 0.002	0.338	< 0.001	< 0.001		< 0.001	< 0.0002		0.0031	74	< 0.001	100
G280	05/13/2014	< 0.001	0.0014			0.017	0.029	< 0.001	73	55				< 0.001	0.001		< 0.001				78		
G280	08/12/2014	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	75	55				< 0.001	< 0.001		< 0.001				76		490
G280	10/14/2014	< 0.001	0.0012			0.013	< 0.01	< 0.001	76	60				< 0.001	0.002		< 0.001				83		480
G280	01/21/2015	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	74	62				< 0.001	< 0.001		< 0.001				87		540
G280	04/13/2015	< 0.002	< 0.001	0.045	< 0.001	0.025	< 0.01	< 0.001	71	67	< 0.004	< 0.002	0.358	< 0.002	< 0.001		0.026	< 0.0002		0.0037	86	< 0.001	480
G280	07/23/2015	< 0.001	< 0.001	0.049	< 0.001	0.01	< 0.01	< 0.001		53	< 0.004	< 0.002	0.415	< 0.001	< 0.001		0.074	< 0.0002	0.0019	0.0033	74	< 0.001	480
G280	10/08/2015	< 0.001	< 0.001	0.056	< 0.001	0.025	0.024	< 0.001		54	< 0.004	< 0.002	0.318	< 0.001	< 0.001		0.0035	< 0.0002	0.0074	0.0017	92	< 0.001	450
G280	11/24/2015		0.0066	0.11	< 0.001		0.029	< 0.001	120	54	0.019	0.0059	0.343		0.012	0.019		< 0.0002	0.0045	0.0032	94	< 0.001	460
G280	02/10/2016	< 0.001	< 0.001	0.048	< 0.001	0.012	< 0.01	< 0.001	60	55	< 0.004	< 0.002	0.466	< 0.001	0.0019	< 0.01	< 0.001	< 0.0002	0.0016	0.0033	84	< 0.001	410
G280	05/10/2016	< 0.001	< 0.001	0.045	< 0.001	< 0.01	< 0.01	< 0.001	63	50	< 0.004	< 0.002	0.429	< 0.001	< 0.001	< 0.01	< 0.001	< 0.0002	0.0014	0.0044	80	< 0.001	350
G280 G280	05/13/2016	< 0.001	< 0.001	0.044	< 0.001	< 0.01	< 0.01	< 0.001	65	45	< 0.004	< 0.002	0.497	< 0.001	0.0014	< 0.01	< 0.001	< 0.0002	0.0014	0.0035	75	< 0.001	410 350
G280 G280	11/20/2016		< 0.001	0.043	< 0.001		< 0.01	< 0.001	03	40	< 0.004	< 0.002	0.337		< 0.0014	< 0.01		< 0.0002	0.0010	0.0048	67	< 0.001	430
G281	11/20/2015		0.0043	0.14	< 0.001		< 0.01	< 0.001	150	74	0.011	0.0056	0 349		0.0063	0.013		< 0.0002	0.0015	< 0.001	300	< 0.001	820
G281 G281	02/11/2016		< 0.001	0.067	< 0.001		0.01	< 0.001	120	55	< 0.001	< 0.002	0.411		< 0.001	< 0.015		< 0.0002	< 0.0015	< 0.001	340	< 0.001	740
G281	05/10/2016		< 0.001	0.072	< 0.001		< 0.01	< 0.001	130	72	< 0.004	< 0.002	0.405		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	370	< 0.001	740
G281	08/01/2016		< 0.001	0.078	< 0.001		0.012	< 0.001	140	70	< 0.004	< 0.002	0.368		0.0011	< 0.01		< 0.0002	< 0.001	< 0.001	310	< 0.001	780
G281	08/31/2016	< 0.001				< 0.01								< 0.001			0.48						800
G401	11/21/2015		0.0046	0.037	< 0.001		3.3	< 0.001	440	3.6	0.0053	0.25	< 0.25		0.0031	0.055		< 0.0002	< 0.001	0.002	2300	< 0.001	3000
G401	02/22/2016		< 0.002	0.015	< 0.001		3.4	< 0.001	330	6	< 0.004	0.24	< 0.25		< 0.001	0.05		< 0.0002	< 0.001	< 0.002	2500	< 0.001	3000
G401	05/19/2016		0.0015	0.014	< 0.001		3.5	< 0.001	380	3	< 0.004	0.27	0.758		< 0.001	0.046		< 0.0002	< 0.001	< 0.001	2200	< 0.001	2800
G401	08/01/2016		0.004	0.053	< 0.001		4.1	0.0012	450	5.3	0.0096	0.28	< 0.25		0.0048	0.051		0.00093	< 0.001	0.0055	2100	< 0.001	2900
G401	11/17/2016		0.0027	0.021	< 0.001		4	0.0013		< 5	< 0.004	0.27	< 0.25		0.0029	0.054		< 0.0002	< 0.001	0.001	3400	< 0.001	3200
G402	11/21/2015		0.024	0.082	< 0.001		6.6	0.001	270	2.8	0.019	0.014	< 0.25		0.015	0.054		< 0.0002	0.006	0.002	1200	< 0.001	1700
G402	02/22/2016		0.027	0.11	< 0.001		5.7	< 0.001	220	2.8	0.031	0.015	0.355		0.018	0.057		< 0.0002	0.0049	0.0033	1000	< 0.001	1700
G402	05/19/2016		0.023	0.085	< 0.001		6.3	< 0.001	270	1.5	0.022	0.019	0.367		0.015	0.036		< 0.0002	0.0044	0.0016	960	< 0.001	1500

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G402	08/02/2016		0.01	0.047	< 0.001		7.4	< 0.001	240	2.2	0.0085	0.0074	0.33		0.0072	0.033		< 0.0002	0.0033	< 0.001	890	< 0.001	1500
G402	11/17/2016		0.012	0.054	< 0.001		6.9	< 0.001		2.6	0.011	0.007	0.463		0.0079	0.047		< 0.0002	0.0034	0.0012	1100	< 0.001	1700
G403	11/23/2015		0.0017	0.14	< 0.001		0.039	< 0.001	78	6.8	0.0062	< 0.002	0.442		0.0021	< 0.01		< 0.0002	0.004	< 0.001	35	< 0.001	320
G403	02/22/2016		< 0.001	0.13	< 0.001		0.064	< 0.001	69	4.1	< 0.004	< 0.002	0.518		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	17	< 0.001	340
G403	05/18/2016		< 0.001	0.16	< 0.001		0.014	< 0.001	71	2.7	< 0.004	< 0.002	0.478		< 0.001	< 0.01		< 0.0002	0.0012	< 0.001	11	< 0.001	320
G403	08/01/2016		0.0033	0.24	< 0.001		0.027	< 0.001	140	4.5	0.0073	0.0029	0.485		0.0021	< 0.01		< 0.0002	0.0055	0.0068	9.9	< 0.001	320
G403	11/17/2016		0.0026	0.2	< 0.001		0.042	< 0.001		4	< 0.004	0.0024	0.539		< 0.001	< 0.01		< 0.0002	0.001	< 0.001	8.9	< 0.001	350
G404	11/21/2015		< 0.001	0.05	< 0.001		2.1	< 0.001	110	53	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	180	< 0.001	580
G404	02/15/2016		< 0.001	0.043	< 0.001		1.6	< 0.001	110	49	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	150	< 0.001	560
G404	05/19/2016		< 0.001	0.041	< 0.001		1.4	< 0.001	89	46	< 0.004	< 0.002	0.287		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	140	< 0.001	460
G404	08/02/2016		< 0.001	0.055	< 0.001		3.2	< 0.001	120	62	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	190	< 0.001	620
G404	11/22/2016		< 0.001	0.052	< 0.001		3.4	< 0.001		61	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	380	< 0.001	880
G405	11/21/2015		0.014	0.051	< 0.001		17	0.0012	330	14	0.0051	0.0034	0.454		0.0085	< 0.01		< 0.0002	0.001	< 0.001	1700	< 0.001	2400
G405	02/15/2016		0.0028	0.018	< 0.001		16	< 0.001	320	11	< 0.004	< 0.002	0.459		0.0023	< 0.01		< 0.0002	< 0.001	< 0.001	1700	< 0.001	2500
G405	05/18/2016		0.0025	0.02	< 0.001		15	< 0.001	320	9.3	< 0.004	< 0.002	0.544		0.0011	< 0.01		< 0.0002	< 0.001	< 0.001	1800	< 0.001	2200
G405	08/02/2016		0.0063	0.028	< 0.001		17	< 0.001	280	3.4	< 0.004	0.0038	< 0.25		0.0018	< 0.01		< 0.0002	0.0016	< 0.001	1600	< 0.001	2200
G405	11/22/2016		0.0024	0.022	< 0.001		13	< 0.001		19	< 0.004	0.0021	0.525		0.0023	< 0.01		< 0.0002	0.0011	< 0.001	1400	< 0.001	2100
G406	08/30/2016	< 0.001				1.6								< 0.001			5.5						1300
G406	11/18/2016			0.024			1.9			4.4			0.345								910		1400
G407	08/30/2016	< 0.001				0.052								< 0.001			0.21						1400
G407	11/18/2016			0.024			0.078			14			0.289								830		1400
G45D	09/02/2016	0.0014				0.35								< 0.001			0.035						370
G46D	09/02/2016	0.0028				0.14								< 0.001			0.094						460

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring	Wells

User Supplied Information															
Date Range: 03/1	ate Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 ocations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405														
Locations:	APW-2,G151,G152,G15	53,G154,G27	'0,G271,G2'	72,G273,G27	4,G275,G281	,G401,G402,G	403,G404,G	405							
Location APW-2	Туре	Class													
								Sen Slope	Normal /	% of					
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Units/yr	Log Normal	Non-Detects					
B, diss	mg/L	8	7.245	7.180	8.100	6.300	0.626	0.27	Yes / Yes	0.00					
Mn, diss	mg/L	8	0.432	0.414	0.730	0.130	0.166	0.01	Yes / No	0.00					
SO4, diss	mg/L	8	805.375	840.000	1,100.000	500.000	173.105	5.17	Yes / Yes	12.50					
TDS	mg/L	8	1,651.250	1,600.000	1,810.000	1,600.000	78.456	0.00	No / No	0.00					

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

	User Supplied Information													
Date Range: 03	8/11/2008 to 08/16/2016 APW-2 G151 G152 G1	53 G154 G27() G271 G27	2 G273 G274	1 G275 G281	Option : G401 G402 G	for LT Pts: 2403 G404 G	x 0.5						
Location G151	Туре	Class	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	2,0273,027	,0273,0201	,0101,0102,0	1103,0101,0	102						
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects				
As, tot	mg/L	3	0.0007	0.0005	0.0012	0.0005	0.0004	0.00	No / No	66.67				
B, diss	mg/L	16	0.016	0.012	0.052	0.003	0.013	0.00	No / Yes	50.00				
B, tot	mg/L	3	0.0183	0.0190	0.0260	0.0100	0.0080	-0.03	Yes / Yes	0.00				
Ba, tot	mg/L	3	0.0643	0.0670	0.0690	0.0570	0.0064	0.02	Yes / Yes	0.00				
Be, tot	mg/L	3	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00				
Cd,tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00				
Cl, tot	mg/L	3	58.333	58.000	67.000	50.000	8.505	16.23	Yes / Yes	0.00				
Co, tot	mg/L	3	0.0010	0.0010	0.0010	0.0010	0.0000	0.00	No / No	100.00				
Cr, tot	mg/L	3	0.0020	0.0020	0.0020	0.0020	0.0000	0.00	No / No	100.00				
F, tot	mg/L	3	0.609	0.649	0.691	0.486	0.108	-0.09	Yes / Yes	0.00				
Hg, tot	mg/L	3	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00				
Mn, diss	mg/L	16	0.131	0.092	0.560	0.011	0.131	0.00	No / Yes	0.00				
Mn, tot	mg/L	3	0.225	0.230	0.360	0.086	0.137	0.56	Yes / Yes	0.00				
Pb, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00				
Sb, tot	mg/L	3	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00				
Se, tot	mg/L	3	0.0025	0.0024	0.0029	0.0021	0.0004	0.00	Yes / Yes	0.00				
SO4, diss	mg/L	16	111.563	110.000	140.000	100.000	11.213	4.79	No / No	0.00				
SO4, tot	mg/L	3	104.333	98.000	120.000	95.000	13.650	6.09	Yes / Yes	0.00				
TDS	mg/L	17	551.176	560.000	660.000	500.000	44.424	0.00	Yes / Yes	0.00				
Tl, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00				

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

User Supplied Information													
Date Range: 03 Locations:	3/11/2008 to 08/16/2016 APW-2,G151,G152,G1	53,G154,G27(),G271,G27	2,G273,G274	4,G275,G281	Option : ,G401,G402,G	for LT Pts: 403,G404,G	x 0.5 405					
Location G152	Туре	Class											
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects			
As, tot	mg/L	3	0.0016	0.0014	0.0023	0.0011	0.0006	0.00	Yes / Yes	0.00			
B, diss	mg/L	16	0.047	0.041	0.110	0.015	0.025	0.01	No / Yes	0.00			
B, tot	mg/L	3	0.0383	0.0360	0.0470	0.0320	0.0078	-0.03	Yes / Yes	0.00			
Ba, tot	mg/L	3	0.1200	0.1300	0.1300	0.1000	0.0173	-0.06	No / No	0.00			
Be, tot	mg/L	3	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00			
Cd,tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00			
Cl, tot	mg/L	3	42.333	42.000	45.000	40.000	2.517	10.15	Yes / Yes	0.00			
Co, tot	mg/L	3	0.0014	0.0010	0.0023	0.0010	0.0008	0.00	No / No	66.67			
Cr, tot	mg/L	3	0.0040	0.0020	0.0080	0.0020	0.0035	-0.01	No / No	66.67			
F, tot	mg/L	3	0.550	0.587	0.642	0.422	0.114	-0.11	Yes / Yes	0.00			
Hg, tot	mg/L	3	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00			
Mn, diss	mg/L	16	0.294	0.180	0.927	0.034	0.277	0.07	No / Yes	0.00			
Mn, tot	mg/L	3	0.317	0.310	0.370	0.270	0.050	0.12	Yes / Yes	0.00			
Pb, tot	mg/L	3	0.0012	0.0005	0.0026	0.0005	0.0012	0.00	No / No	66.67			
Sb, tot	mg/L	3	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00			
Se, tot	mg/L	3	0.0036	0.0035	0.0043	0.0030	0.0007	0.00	Yes / Yes	0.00			
SO4, diss	mg/L	16	161.063	135.000	300.000	107.000	61.151	26.31	No / No	0.00			
SO4, tot	mg/L	3	120.000	120.000	130.000	110.000	10.000	20.29	Yes / Yes	0.00			
TDS	mg/L	17	676.706	620.000	1,100.000	494.000	156.107	72.77	No / Yes	0.00			
Tl, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00			

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

User Supplied Information													
Date Range: 03 Locations:	3/11/2008 to 08/16/2016 APW-2,G151,G152,G1	153,G154,G27	0,G271,G2	72,G273,G27	4,G275,G281	Option 1,G401,G402,0	for LT Pts: 6403,G404,G	x 0.5					
Location G153	Туре	Class	, ,	, ,	, ,	, , ,	, ,						
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects			
As, tot	mg/L	3	0.0056	0.0056	0.0059	0.0052	0.0004	0.00	Yes / Yes	0.00			
B, diss	mg/L	16	0.028	0.027	0.058	0.012	0.012	-0.01	Yes / Yes	0.00			
B, tot	mg/L	3	0.0437	0.0360	0.0640	0.0310	0.0178	-0.07	Yes / Yes	0.00			
Ba, tot	mg/L	3	0.0247	0.0220	0.0360	0.0160	0.0103	-0.04	Yes / Yes	0.00			
Be, tot	mg/L	3	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00			
Cd,tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00			
Cl, tot	mg/L	3	140.000	130.000	160.000	130.000	17.321	0.00	No / No	0.00			
Co, tot	mg/L	3	0.0037	0.0034	0.0068	0.0010	0.0029	-0.01	Yes / Yes	33.33			
Cr, tot	mg/L	3	0.0045	0.0041	0.0053	0.0041	0.0007	0.00	No / No	0.00			
F, tot	mg/L	5	0.363	0.369	0.410	0.293	0.047	-0.01	Yes / Yes	0.00			
Hg, tot	mg/L	3	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00			
Mn, diss	mg/L	16	0.235	0.175	1.020	0.024	0.249	-0.12	No / Yes	0.00			
Mn, tot	mg/L	3	0.673	0.510	1.300	0.210	0.563	-2.21	Yes / Yes	0.00			
Pb, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00			
Sb, tot	mg/L	3	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00			
Se, tot	mg/L	3	0.0230	0.0210	0.0270	0.0210	0.0035	0.00	No / No	0.00			
SO4, diss	mg/L	16	1,823.750	1,800.000	2,100.000	1,500.000	206.458	184.47	Yes / Yes	0.00			
SO4, tot	mg/L	3	1,633.333	1,500.000	1,900.000	1,500.000	230.940	0.00	No / No	0.00			
TDS	mg/L	17	3,361.765	3,300.000	3,900.000	2,800.000	340.739	42.47	Yes / Yes	0.00			
Tl, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00			

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

			User	Supplied In	formation						
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405											
Location G154	Туре	Class									
Donomotor	Linita	Count	Maan	Madian	Manimum	Minimum	Std Day	Sen Slope	Normal /	% of Non Detects	
Parameter	Units	Count	Iviean	Wiedian	Maximum	wiininuni	Stu Dev	Units/yr	Log Normai	Non-Detects	
As, tot	mg/L	3	0.0008	0.0005	0.0014	0.0005	0.0005	0.00	No / No	66.67	
B, diss	mg/L	16	0.036	0.037	0.045	0.021	0.008	0.00	No / No	0.00	
B, tot	mg/L	3	0.0300	0.0310	0.0380	0.0210	0.0085	0.01	Yes / Yes	0.00	
Ba, tot	mg/L	3	0.0340	0.0330	0.0370	0.0320	0.0026	-0.01	Yes / Yes	0.00	
Be, tot	mg/L	3	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00	
Cd,tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00	
Cl, tot	mg/L	3	6.500	4.900	10.000	4.600	3.035	-10.35	Yes / Yes	0.00	
Co, tot	mg/L	3	0.0010	0.0010	0.0010	0.0010	0.0000	0.00	No / No	100.00	
Cr, tot	mg/L	3	0.0020	0.0020	0.0020	0.0020	0.0000	0.00	No / No	100.00	
F, tot	mg/L	3	1.014	1.010	1.060	0.971	0.045	0.10	Yes / Yes	0.00	
Hg, tot	mg/L	3	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00	
Mn, diss	mg/L	16	0.062	0.035	0.462	0.003	0.112	0.00	No / Yes	0.00	
Mn, tot	mg/L	3	0.157	0.160	0.160	0.150	0.006	0.02	No / No	0.00	
Pb, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00	
Sb, tot	mg/L	3	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00	
Se, tot	mg/L	3	0.0054	0.0058	0.0062	0.0042	0.0011	0.00	Yes / Yes	0.00	
SO4, diss	mg/L	16	110.250	105.000	175.000	82.000	23.601	-11.96	No / Yes	0.00	
SO4, tot	mg/L	3	98.667	93.000	110.000	93.000	9.815	34.50	No / No	0.00	
TDS	mg/L	17	476.118	460.000	624.000	410.000	52.309	17.11	No / No	0.00	
Tl, tot	mg/L	3	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00	

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring Web	lls

User Supplied Information										
Date Range: 03/11 Locations:	Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Cocations: APW-2 G151 G152 G153 G154 G270 G271 G272 G273 G274 G275 G281 G401 G402 G403 G404 G405									
Location G270	Туре	Class	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	_,0_/0,0_/	.,	, 0 101, 0 102, 0	,,,,			
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	38	0.0010	0.0005	0.0110	0.0005	0.0018	0.00	No / No	73.68
B, diss	mg/L	34	0.029	0.012	0.125	0.005	0.034	0.00	No / No	47.06
B, tot	mg/L	38	0.0207	0.0050	0.1250	0.0050	0.0264	0.00	No / No	68.42
Ba, tot	mg/L	20	0.0527	0.0495	0.0760	0.0320	0.0155	0.00	Yes / Yes	0.00
Be, tot	mg/L	20	0.00060	0.00050	0.00250	0.00050	0.00045	0.00	No / No	100.00
Ca, tot	mg/L	34	66.588	67.000	81.000	49.000	7.785	-0.34	Yes / Yes	0.00
Cd,tot	mg/L	38	0.0007	0.0005	0.0080	0.0003	0.0012	0.00	No / No	97.37
Cl, tot	mg/L	38	35.521	20.500	91.000	5.000	27.343	2.03	No / No	2.63
Co, tot	mg/L	20	0.0011	0.0010	0.0025	0.0005	0.0005	0.00	No / No	85.00
Cr, tot	mg/L	20	0.0027	0.0020	0.0125	0.0020	0.0023	0.00	No / No	100.00
F, tot	mg/L	20	0.373	0.361	0.504	0.279	0.058	0.00	Yes / Yes	0.00
Hg, tot	mg/L	20	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0050	0.0050	0.0050	0.0050	0.0000	0.00	No / No	100.00
Mn, diss	mg/L	34	0.094	0.045	0.470	0.003	0.108	-0.02	No / Yes	0.00
Mn, tot	mg/L	34	0.146	0.083	0.620	0.021	0.134	-0.03	No / No	0.00
Mo, tot	mg/L	8	0.0007	0.0005	0.0011	0.0005	0.0003	0.00	No / No	62.50
Pb, tot	mg/L	38	0.0011	0.0005	0.0073	0.0003	0.0013	0.00	No / No	68.42
Ra-226,228, tot	pCi/L	4	0.5815	0.5340	0.9970	0.2610	0.3415	0.82	Yes / Yes	25.00
Sb, tot	mg/L	20	0.0013	0.0015	0.0015	0.0010	0.0002	0.00	No / No	100.00
Se, tot	mg/L	20	0.0019	0.0013	0.0078	0.0005	0.0020	0.00	No / No	60.00
SO4, diss	mg/L	34	54.312	42.500	140.000	2.100	50.372	15.70	No / No	0.00
SO4, tot	mg/L	38	58.747	65.000	140.000	2.300	50.239	15.58	No / No	0.00
TDS	mg/L	37	453.514	460.000	550.000	340.000	52.769	10.59	Yes / Yes	0.00
Tl, tot	mg/L	20	0.0007	0.0005	0.0010	0.0005	0.0002	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary f	for Select Monitoring Wells

User Supplied Information										
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405										
Location G271	Туре	Class								
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	33	0.0012	0.0005	0.0049	0.0005	0.0011	0.00	No / No	60.61
B, diss	mg/L	29	0.281	0.260	0.610	0.130	0.128	0.05	No / Yes	0.00
B, tot	mg/L	33	0.3088	0.2400	0.9800	0.1100	0.1904	0.03	No / Yes	0.00
Ba, tot	mg/L	19	0.0456	0.0340	0.1100	0.0280	0.0220	0.00	No / No	0.00
Be, tot	mg/L	19	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	29	123.276	120.000	170.000	96.000	17.834	2.68	Yes / Yes	0.00
Cd,tot	mg/L	33	0.0005	0.0005	0.0005	0.0003	0.0001	0.00	No / No	100.00
Cl, tot	mg/L	33	41.427	42.000	58.000	5.100	9.067	0.00	No / No	0.00
Co, tot	mg/L	19	0.0014	0.0010	0.0047	0.0005	0.0010	0.00	No / No	78.95
Cr, tot	mg/L	19	0.0037	0.0020	0.0140	0.0020	0.0032	0.00	No / No	73.68
F, tot	mg/L	19	0.415	0.410	0.562	0.298	0.062	0.01	Yes / Yes	0.00
Hg, tot	mg/L	19	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0050	0.0050	0.0050	0.0050	0.0000	0.00	No / No	100.00
Mn, diss	mg/L	29	0.015	0.001	0.220	0.001	0.045	0.00	No / No	62.07
Mn, tot	mg/L	29	0.058	0.021	0.270	0.001	0.075	-0.01	No / Yes	0.00
Mo, tot	mg/L	8	0.0010	0.0005	0.0036	0.0005	0.0011	0.00	No / No	62.50
Pb, tot	mg/L	33	0.0022	0.0012	0.0110	0.0005	0.0024	0.00	No / No	45.45
Ra-226,228, tot	pCi/L	4	0.5905	0.5785	1.0200	0.1850	0.4251	-1.19	Yes / Yes	0.00
Sb, tot	mg/L	19	0.0014	0.0015	0.0015	0.0010	0.0002	0.00	No / No	100.00
Se, tot	mg/L	19	0.0040	0.0035	0.0099	0.0018	0.0021	0.00	No / Yes	0.00
SO4, diss	mg/L	29	354.483	360.000	520.000	220.000	87.447	35.77	Yes / Yes	0.00
SO4, tot	mg/L	33	356.970	350.000	540.000	230.000	87.446	28.25	Yes / Yes	0.00
TDS	mg/L	31	861.935	860.000	1,000.000	710.000	96.035	29.19	No / No	0.00
Tl, tot	mg/L	19	0.0006	0.0005	0.0010	0.0005	0.0002	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring V	Vells

			User	Supplied In	formation					
Date Range: 03	/11/2008 to 08/16/2016					Option	for LT Pts:	x 0.5		
Locations:	APW-2,G151,G152,G1	53,G154,G27),G271,G27	2,G273,G274	4,G275,G281	,G401,G402,G	403,G404,G	405		
Location	Туре	Class								
G272	Groundwater									
								Sen Slope	Normal /	% of
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Units/yr	Log Normal	Non-Detects
As, tot	mg/L	29	0.0006	0.0005	0.0013	0.0005	0.0002	0.00	No / No	86.21
B, diss	mg/L	29	0.019	0.005	0.270	0.005	0.050	0.00	No / No	86.21
B, tot	mg/L	29	0.0144	0.0050	0.0600	0.0050	0.0167	0.00	No / No	68.97
Ba, tot	mg/L	15	0.0673	0.0620	0.0990	0.0580	0.0108	0.00	No / No	0.00
Be, tot	mg/L	15	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	25	97.400	98.000	110.000	82.000	9.000	3.90	No / No	0.00
Cd,tot	mg/L	29	0.0005	0.0005	0.0005	0.0003	0.0001	0.00	No / No	100.00
Cl, tot	mg/L	29	39.552	41.000	53.000	29.000	6.294	-2.64	Yes / Yes	3.45
Co, tot	mg/L	15	0.0011	0.0010	0.0022	0.0005	0.0004	0.00	No / No	80.00
Cr, tot	mg/L	15	0.0029	0.0020	0.0110	0.0020	0.0024	0.00	No / No	86.67
F, tot	mg/L	15	0.449	0.459	0.561	0.355	0.058	0.00	Yes / Yes	0.00
Hg, tot	mg/L	15	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Mn, diss	mg/L	29	0.011	0.001	0.230	0.001	0.043	0.00	No / No	75.86
Mn, tot	mg/L	29	0.029	0.011	0.150	0.002	0.040	0.00	No / Yes	0.00
Mo, tot	mg/L	4	0.0012	0.0009	0.0024	0.0005	0.0009	0.00	Yes / Yes	50.00
Pb, tot	mg/L	29	0.0011	0.0005	0.0049	0.0005	0.0011	0.00	No / No	65.52
Sb, tot	mg/L	15	0.0014	0.0015	0.0015	0.0010	0.0002	0.00	No / No	100.00
Se, tot	mg/L	15	0.0017	0.0013	0.0042	0.0005	0.0012	0.00	No / Yes	53.33
SO4, diss	mg/L	29	264.483	270.000	380.000	140.000	62.996	28.10	Yes / Yes	0.00
SO4, tot	mg/L	29	253.448	250.000	380.000	120.000	68.935	30.52	Yes / Yes	0.00
TDS	mg/L	27	702.963	680.000	840.000	570.000	68.770	25.54	Yes / Yes	0.00
Tl, tot	mg/L	15	0.0006	0.0005	0.0010	0.0005	0.0002	0.00	No / No	100.00

TT. . C

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

User Supplied Information										
Date Range: 03/11 Locations:	1/2008 to 08/16/2016 APW-2.G151.G152.G1	53.G154.G27	′0.G271.G2′	72.G273.G27	4.G275.G281	Option .G401.G402.0	for LT Pts: 6403.G404.G	x 0.5		
Location G273	Туре	Class	- , - , -	, , -))	,,,-				
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	33	0.0008	0.0005	0.0045	0.0005	0.0009	0.00	No / No	75.76
B, diss	mg/L	29	0.224	0.180	0.620	0.019	0.176	0.06	No / No	6.90
B, tot	mg/L	33	0.2084	0.2000	0.4800	0.0160	0.1368	0.05	Yes / No	9.09
Ba, tot	mg/L	19	0.0503	0.0440	0.0940	0.0280	0.0210	-0.01	No / Yes	0.00
Be, tot	mg/L	19	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	29	156.207	150.000	200.000	120.000	19.533	0.00	Yes / Yes	0.00
Cd,tot	mg/L	33	0.0005	0.0005	0.0005	0.0003	0.0001	0.00	No / No	100.00
Cl, tot	mg/L	33	35.788	37.000	59.000	4.000	9.591	2.49	Yes / No	0.00
Co, tot	mg/L	19	0.0009	0.0010	0.0010	0.0005	0.0002	0.00	No / No	100.00
Cr, tot	mg/L	19	0.0038	0.0020	0.0340	0.0020	0.0073	0.00	No / No	94.74
F, tot	mg/L	19	0.335	0.330	0.537	0.125	0.099	0.01	Yes / No	10.53
Hg, tot	mg/L	19	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0050	0.0050	0.0050	0.0050	0.0000	0.00	No / No	100.00
Mn, diss	mg/L	29	0.042	0.028	0.330	0.001	0.059	-0.01	No / No	0.00
Mn, tot	mg/L	29	0.057	0.043	0.190	0.019	0.043	-0.01	No / Yes	0.00
Mo, tot	mg/L	8	0.0007	0.0005	0.0019	0.0005	0.0005	0.00	No / No	87.50
Pb, tot	mg/L	33	0.0006	0.0005	0.0037	0.0003	0.0006	0.00	No / No	87.88
Ra-226,228, tot	pCi/L	4	1.2503	1.1420	2.0600	0.6570	0.6588	-2.20	Yes / Yes	0.00
Sb, tot	mg/L	19	0.0014	0.0015	0.0015	0.0010	0.0002	0.00	No / No	100.00
Se, tot	mg/L	19	0.0010	0.0005	0.0051	0.0005	0.0011	0.00	No / No	84.21
SO4, diss	mg/L	29	524.000	520.000	700.000	46.000	131.055	8.97	No / No	0.00
SO4, tot	mg/L	33	529.697	520.000	750.000	340.000	105.874	6.45	Yes / Yes	0.00
TDS	mg/L	31	1,085.484	1,100.000	1,300.000	840.000	140.139	0.00	No / No	0.00
Tl, tot	mg/L	19	0.0006	0.0005	0.0010	0.0005	0.0002	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring V	Vells

User Supplied Information										
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2.G151.G152.G153.G154.G270.G271.G272.G273.G274.G275.G281.G401.G402.G403.G404.G405										
Location G274	Type Groundwater	Class	, ,	, ,	, ,	, , ,	, ,			
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	29	0.0008	0.0005	0.0040	0.0005	0.0008	0.00	No / No	82.76
B, diss	mg/L	29	0.295	0.180	0.970	0.005	0.302	0.13	No / No	27.59
B, tot	mg/L	29	0.2636	0.1600	0.7400	0.0050	0.2533	0.11	No / No	24.14
Ba, tot	mg/L	15	0.0759	0.0700	0.1200	0.0520	0.0181	-0.01	Yes / Yes	0.00
Be, tot	mg/L	15	0.00059	0.00050	0.00180	0.00050	0.00034	0.00	No / No	93.33
Ca, tot	mg/L	25	112.320	110.000	130.000	98.000	8.440	0.00	No / No	0.00
Cd,tot	mg/L	29	0.0005	0.0005	0.0014	0.0003	0.0002	0.00	No / No	96.55
Cl, tot	mg/L	29	33.572	36.000	55.000	3.600	11.500	-4.55	Yes / No	0.00
Co, tot	mg/L	15	0.0010	0.0010	0.0028	0.0005	0.0005	0.00	No / No	93.33
Cr, tot	mg/L	15	0.0024	0.0020	0.0059	0.0020	0.0010	0.00	No / No	93.33
F, tot	mg/L	15	0.351	0.350	0.455	0.262	0.052	0.01	Yes / Yes	0.00
Hg, tot	mg/L	15	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Mn, diss	mg/L	29	0.003	0.001	0.046	0.001	0.009	0.00	No / No	68.97
Mn, tot	mg/L	29	0.049	0.022	0.370	0.003	0.087	0.00	No / Yes	0.00
Mo, tot	mg/L	4	0.0007	0.0005	0.0012	0.0005	0.0003	0.00	No / No	75.00
Pb, tot	mg/L	29	0.0015	0.0005	0.0100	0.0003	0.0024	0.00	No / No	65.52
Sb, tot	mg/L	15	0.0014	0.0015	0.0015	0.0010	0.0002	0.00	No / No	100.00
Se, tot	mg/L	15	0.0010	0.0013	0.0022	0.0005	0.0005	0.00	No / No	73.33
SO4, diss	mg/L	29	340.000	350.000	400.000	240.000	41.748	0.00	No / No	0.00
SO4, tot	mg/L	29	336.552	330.000	460.000	230.000	54.725	6.69	Yes / Yes	0.00
TDS	mg/L	27	862.593	870.000	980.000	770.000	64.307	-16.05	Yes / Yes	0.00
Tl, tot	mg/L	15	0.0007	0.0005	0.0013	0.0005	0.0003	0.00	No / No	93.33

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring Wel	ls

			User	Supplied In	nformation					
Date Range: 03	/11/2008 to 08/16/2016					Option	for LT Pts:	x 0.5		
Locations:	APW-2,G151,G152,G1	53,G154,G27	0,G271,G2′	72,G273,G27	4,G275,G281	,G401,G402,G	403,G404,G	405		
Location	Туре	Class								
G275	Groundwater									
								Sen Slope	Normal /	% of
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Units/yr	Log Normal	Non-Detects
As, tot	mg/L	25	0.0016	0.0011	0.0043	0.0005	0.0013	0.00	No / No	48.00
B, diss	mg/L	25	3.176	3.500	4.600	0.910	1.040	0.10	Yes / No	0.00
B, tot	mg/L	25	3.3360	3.6000	4.6000	1.2000	0.9652	0.04	No / No	0.00
Ba, tot	mg/L	12	0.0468	0.0390	0.1200	0.0320	0.0241	0.00	No / No	0.00
Be, tot	mg/L	12	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	22	221.364	230.000	310.000	120.000	48.726	5.01	Yes / Yes	0.00
Cd,tot	mg/L	25	0.0005	0.0005	0.0005	0.0003	0.0001	0.00	No / No	100.00
Cl, tot	mg/L	25	16.752	16.000	30.000	9.200	5.504	1.57	Yes / Yes	0.00
Co, tot	mg/L	12	0.0010	0.0010	0.0025	0.0005	0.0005	0.00	No / No	100.00
Cr, tot	mg/L	12	0.0038	0.0023	0.0125	0.0020	0.0034	0.00	No / No	83.33
F, tot	mg/L	12	0.310	0.320	0.452	0.125	0.083	0.01	Yes / No	8.33
Hg, tot	mg/L	12	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Mn, diss	mg/L	25	0.051	0.019	0.340	0.001	0.086	0.00	No / Yes	8.00
Mn, tot	mg/L	25	0.168	0.085	0.780	0.012	0.203	0.01	No / Yes	0.00
Mo, tot	mg/L	3	0.0008	0.0005	0.0014	0.0005	0.0005	0.00	No / No	66.67
Pb, tot	mg/L	25	0.0024	0.0013	0.0100	0.0003	0.0028	0.00	No / Yes	36.00
Sb, tot	mg/L	12	0.0014	0.0015	0.0015	0.0010	0.0002	0.00	No / No	100.00
Se, tot	mg/L	12	0.0014	0.0013	0.0060	0.0005	0.0015	0.00	No / No	66.67
SO4, diss	mg/L	25	677.600	720.000	920.000	280.000	183.627	30.33	No / No	0.00
SO4, tot	mg/L	25	684.000	720.000	990.000	310.000	213.288	0.84	Yes / No	0.00
TDS	mg/L	24	1,388.333	1,500.000	2,000.000	840.000	270.373	0.00	Yes / No	0.00
Tl, tot	mg/L	12	0.0007	0.0005	0.0010	0.0005	0.0002	0.00	No / No	100.00

TT. . C ...

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

			User	Supplied In	formation					
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405										
Location G281	Туре	Class								
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	4	0.0015	0.0005	0.0043	0.0005	0.0019	0.00	No / No	75.00
B, tot	mg/L	4	0.0080	0.0075	0.0120	0.0050	0.0036	0.01	Yes / Yes	50.00
Ba, tot	mg/L	4	0.0893	0.0750	0.1400	0.0670	0.0341	-0.03	Yes / Yes	0.00
Be, tot	mg/L	4	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	4	135.000	135.000	150.000	120.000	12.910	13.36	Yes / Yes	0.00
Cd,tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
Cl, tot	mg/L	4	67.750	71.000	74.000	55.000	8.655	-4.99	Yes / Yes	0.00
Co, tot	mg/L	4	0.0022	0.0010	0.0056	0.0010	0.0023	0.00	No / No	75.00
Cr, tot	mg/L	4	0.0043	0.0020	0.0110	0.0020	0.0045	-0.01	No / No	75.00
F, tot	mg/L	4	0.383	0.387	0.411	0.349	0.030	0.00	Yes / Yes	0.00
Hg, tot	mg/L	4	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0070	0.0050	0.0130	0.0050	0.0040	-0.01	No / No	75.00
Mo, tot	mg/L	4	0.0008	0.0005	0.0015	0.0005	0.0005	0.00	No / No	75.00
Pb, tot	mg/L	4	0.0021	0.0008	0.0063	0.0005	0.0028	0.00	No / Yes	50.00
Ra-226,228, tot	pCi/L	4	0.7225	0.5910	1.4900	0.2180	0.5416	0.80	Yes / Yes	25.00
Sb, tot	mg/L	4	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00
Se, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
SO4, tot	mg/L	4	330.000	325.000	370.000	300.000	31.623	68.72	Yes / Yes	0.00
TDS	mg/L	4	770.000	760.000	820.000	740.000	38.297	-28.65	Yes / Yes	0.00
Tl, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring Web	lls

			User	Supplied In	nformation					
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405										
Location G401	Туре	Class								
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	4	0.0028	0.0028	0.0046	0.0010	0.0018	0.00	Yes / Yes	25.00
B, tot	mg/L	4	3.5750	3.4500	4.1000	3.3000	0.3594	0.79	Yes / Yes	0.00
Ba, tot	mg/L	4	0.0298	0.0260	0.0530	0.0140	0.0188	0.01	Yes / Yes	0.00
Be, tot	mg/L	4	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	4	400.000	410.000	450.000	330.000	55.976	112.15	Yes / Yes	0.00
Cd,tot	mg/L	4	0.0007	0.0005	0.0012	0.0005	0.0003	0.00	No / No	75.00
Cl, tot	mg/L	4	4.475	4.450	6.000	3.000	1.408	0.61	Yes / Yes	0.00
Co, tot	mg/L	4	0.2600	0.2600	0.2800	0.2400	0.0183	0.05	Yes / Yes	0.00
Cr, tot	mg/L	4	0.0047	0.0037	0.0096	0.0020	0.0036	0.00	Yes / Yes	50.00
F, tot	mg/L	4	0.283	0.125	0.758	0.125	0.317	0.00	No / No	75.00
Hg, tot	mg/L	4	0.00031	0.00010	0.00093	0.00010	0.00042	0.00	No / No	75.00
Li, tot	mg/L	4	0.0505	0.0505	0.0550	0.0460	0.0037	-0.01	Yes / Yes	0.00
Mo, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
Pb, tot	mg/L	4	0.0022	0.0018	0.0048	0.0005	0.0021	0.00	Yes / Yes	50.00
Ra-226,228, tot	pCi/L	4	1.3010	1.3150	1.6300	0.9440	0.3593	-0.17	Yes / Yes	0.00
Sb, tot	mg/L	4	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00
Se, tot	mg/L	4	0.0023	0.0015	0.0055	0.0005	0.0023	0.00	Yes / Yes	50.00
SO4, tot	mg/L	4	2,275.000	2,250.000	2,500.000	2,100.000	170.783	-390.59	Yes / Yes	0.00
TDS	mg/L	4	2,925.000	2,950.000	3,000.000	2,800.000	95.743	-185.33	Yes / Yes	0.00
Tl, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring Web	lls

			User	Supplied In	nformation					
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5										
Location G402	Туре	Class	0,0271,027	12,0213,021	4,6275,6201	,0401,0402,0	1403,0404,0	405		
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As. tot	mg/L	4	0.0210	0.0235	0.0270	0.0100	0.0075	-0.02	Yes / Yes	0.00
B. tot	mg/L	4	6.5000	6.4500	7.4000	5.7000	0.7071	1.83	Yes / Yes	0.00
Ba. tot	mg/L	4	0.0810	0.0835	0.1100	0.0470	0.0259	-0.08	Yes / Yes	0.00
Be, tot	mg/L	4	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	4	250.000	255.000	270.000	220.000	24.495	-21.49	Yes / Yes	0.00
Cd.tot	mg/L	4	0.0006	0.0005	0.0010	0.0005	0.0003	0.00	No / No	75.00
Cl, tot	mg/L	4	2.325	2.500	2.800	1.500	0.618	-1.11	Yes / Yes	0.00
Co, tot	mg/L	4	0.0139	0.0145	0.0190	0.0074	0.0048	0.00	Yes / Yes	0.00
Cr, tot	mg/L	4	0.0201	0.0205	0.0310	0.0085	0.0093	-0.03	Yes / Yes	0.00
F, tot	mg/L	4	0.294	0.343	0.367	0.125	0.114	0.17	No / No	25.00
Hg, tot	mg/L	4	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0450	0.0450	0.0570	0.0330	0.0122	-0.03	Yes / Yes	0.00
Mo, tot	mg/L	4	0.0047	0.0047	0.0060	0.0033	0.0011	0.00	Yes / Yes	0.00
Pb, tot	mg/L	4	0.0138	0.0150	0.0180	0.0072	0.0046	-0.01	Yes / Yes	0.00
Ra-226,228, tot	pCi/L	4	3.1500	3.5500	4.4600	1.0400	1.5553	-2.49	Yes / Yes	0.00
Sb, tot	mg/L	4	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00
Se, tot	mg/L	4	0.0019	0.0018	0.0033	0.0005	0.0012	0.00	Yes / Yes	25.00
SO4, tot	mg/L	4	1,012.500	980.000	1,200.000	890.000	133.010	-392.46	Yes / Yes	0.00
TDS	mg/L	4	1,600.000	1,600.000	1,700.000	1,500.000	115.470	-346.15	No / No	0.00
Tl, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

			User	Supplied In	formation					
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5										
Locations: Location G403	АР W-2,G151,G152,G1 Туре	Class	0,G271,G27	2,G275,G274	i,G275,G281	,G401,G402,G	403,6404,6	405		
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	4	0.0015	0.0011	0.0033	0.0005	0.0013	0.00	Yes / Yes	50.00
B, tot	mg/L	4	0.0360	0.0330	0.0640	0.0140	0.0213	-0.03	Yes / Yes	0.00
Ba, tot	mg/L	4	0.1675	0.1500	0.2400	0.1300	0.0499	0.14	Yes / Yes	0.00
Be, tot	mg/L	4	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	4	89.500	74.500	140.000	69.000	33.887	49.18	No / Yes	0.00
Cd,tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
Cl, tot	mg/L	4	4.525	4.300	6.800	2.700	1.702	-4.64	Yes / Yes	0.00
Co, tot	mg/L	4	0.0015	0.0010	0.0029	0.0010	0.0010	0.00	No / No	75.00
Cr, tot	mg/L	4	0.0044	0.0041	0.0073	0.0020	0.0028	0.00	Yes / Yes	50.00
F, tot	mg/L	4	0.481	0.482	0.518	0.442	0.031	0.05	Yes / Yes	0.00
Hg, tot	mg/L	4	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0050	0.0050	0.0050	0.0050	0.0000	0.00	No / No	100.00
Mo, tot	mg/L	4	0.0028	0.0026	0.0055	0.0005	0.0024	0.00	Yes / Yes	25.00
Pb, tot	mg/L	4	0.0013	0.0013	0.0021	0.0005	0.0009	0.00	No / No	50.00
Ra-226,228, tot	pCi/L	4	0.3674	0.1973	0.9680	0.1070	0.4057	0.85	Yes / Yes	25.00
Sb, tot	mg/L	4	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00
Se, tot	mg/L	4	0.0021	0.0005	0.0068	0.0005	0.0032	0.00	No / No	75.00
SO4, tot	mg/L	4	18.225	14.000	35.000	9.900	11.610	-30.93	Yes / Yes	0.00
TDS	mg/L	4	325.000	320.000	340.000	320.000	10.000	0.00	No / No	0.00
Tl, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring We	ells

			User	Supplied In	formation					
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405										
Location G404	Туре	Class								
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects
As, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
B, tot	mg/L	4	2.0750	1.8500	3.2000	1.4000	0.8057	0.40	Yes / Yes	0.00
Ba, tot	mg/L	4	0.0473	0.0465	0.0550	0.0410	0.0064	0.00	Yes / Yes	0.00
Be, tot	mg/L	4	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00
Ca, tot	mg/L	4	107.250	110.000	120.000	89.000	13.048	7.16	Yes / Yes	0.00
Cd,tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
Cl, tot	mg/L	4	52.500	51.000	62.000	46.000	6.952	0.62	Yes / Yes	0.00
Co, tot	mg/L	4	0.0010	0.0010	0.0010	0.0010	0.0000	0.00	No / No	100.00
Cr, tot	mg/L	4	0.0020	0.0020	0.0020	0.0020	0.0000	0.00	No / No	100.00
F, tot	mg/L	4	0.166	0.125	0.287	0.125	0.081	0.00	No / No	75.00
Hg, tot	mg/L	4	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00
Li, tot	mg/L	4	0.0050	0.0050	0.0050	0.0050	0.0000	0.00	No / No	100.00
Mo, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
Pb, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
Ra-226,228, tot	pCi/L	4	1.1275	1.0655	2.0700	0.3090	0.8526	2.14	Yes / Yes	25.00
Sb, tot	mg/L	4	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00
Se, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00
SO4, tot	mg/L	4	165.000	165.000	190.000	140.000	23.805	-12.27	Yes / Yes	0.00
TDS	mg/L	4	555.000	570.000	620.000	460.000	68.069	-13.82	Yes / Yes	0.00
Tl, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00

Coffeen	
Ash Pond 2 - Statistical Summary for Select Monitoring Web	lls

User Supplied Information												
Date Range: 03/11/2008 to 08/16/2016 Option for LT Pts: x 0.5 Locations: APW-2,G151,G152,G153,G154,G270,G271,G272,G273,G274,G275,G281,G401,G402,G403,G404,G405												
Location G405	Туре	Class	, ,	, ,	, ,	, , , ,	, ,					
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Sen Slope Units/yr	Normal / Log Normal	% of Non-Detects		
As, tot	mg/L	4	0.0064	0.0046	0.0140	0.0025	0.0054	-0.01	Yes / Yes	0.00		
B, tot	mg/L	4	16.2500	16.5000	17.0000	15.0000	0.9574	-1.96	Yes / Yes	0.00		
Ba, tot	mg/L	4	0.0293	0.0240	0.0510	0.0180	0.0151	-0.01	Yes / Yes	0.00		
Be, tot	mg/L	4	0.00050	0.00050	0.00050	0.00050	0.00000	0.00	No / No	100.00		
Ca, tot	mg/L	4	312.500	320.000	330.000	280.000	22.174	-57.04	Yes / Yes	0.00		
Cd,tot	mg/L	4	0.0007	0.0005	0.0012	0.0005	0.0003	0.00	No / No	75.00		
Cl, tot	mg/L	4	9.425	10.150	14.000	3.400	4.462	-13.96	Yes / Yes	0.00		
Co, tot	mg/L	4	0.0023	0.0022	0.0038	0.0010	0.0015	0.00	Yes / Yes	50.00		
Cr, tot	mg/L	4	0.0028	0.0020	0.0051	0.0020	0.0016	0.00	No / No	75.00		
F, tot	mg/L	4	0.396	0.457	0.544	0.125	0.185	-0.23	Yes / No	25.00		
Hg, tot	mg/L	4	0.00010	0.00010	0.00010	0.00010	0.00000	0.00	No / No	100.00		
Li, tot	mg/L	4	0.0050	0.0050	0.0050	0.0050	0.0000	0.00	No / No	100.00		
Mo, tot	mg/L	4	0.0009	0.0008	0.0016	0.0005	0.0005	0.00	Yes / Yes	50.00		
Pb, tot	mg/L	4	0.0034	0.0021	0.0085	0.0011	0.0034	-0.01	Yes / Yes	0.00		
Ra-226,228, tot	pCi/L	4	0.6429	0.4678	1.3700	0.2660	0.4976	1.01	Yes / Yes	25.00		
Sb, tot	mg/L	4	0.0015	0.0015	0.0015	0.0015	0.0000	0.00	No / No	100.00		
Se, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00		
SO4, tot	mg/L	4	1,700.000	1,700.000	1,800.000	1,600.000	81.650	-71.62	Yes / Yes	0.00		
TDS	mg/L	4	2,325.000	2,300.000	2,500.000	2,200.000	150.000	-347.29	Yes / Yes	0.00		
Tl, tot	mg/L	4	0.0005	0.0005	0.0005	0.0005	0.0000	0.00	No / No	100.00		

APPENDIX G

WATER QUALITY TREND GRAPHS

























ATTACHMENT I



Illinois Power Generating Company 134 CIPS Lane Coffeen, IL 62017

Groundwater Monitoring Plan Addendum for Ash Pond No. 2 Coffeen Power Plant, Coffeen, IL

Ramboll Americas Engineering Solutions, Inc. (Ramboll) is providing the attached Addendum to the Groundwater Monitoring Plan (GMP) for inclusion in the Operating Permit Applications as required under Title 35 of the Illinois Administrative Code (35 I.A.C.), § 845.230 and allowed under 35 I.A.C. § 845.210(d)(1). The GMP was previously submitted to and approved by Illinois Environmental Protection Agency (IEPA) as part of the *Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2* (Closure Plan; AECOM, 2017) submitted for Coffeen Power Plant (CPP) Ash Pond Number (No.) 2 (AP2), Vistra Identification (ID) No. 102, IEPA ID No. W1350150004-02, National Inventory of Dams (NID) No. IL50723.

In addition to the historical documents, Ramboll is providing an Addendum to the CPP AP2 GMP (Attachment 1), that modifies the existing monitoring program and network to align with Part 845. Upon issuance of the Operating Permit, groundwater monitoring will be performed as specified in the Addendum.

BACKGROUND

AECOM submitted the Closure Plan, dated January 26, 2017. Included in Appendix A and B of the Closure Plan were the Hydrogeologic Site Characterization Report and GMP. IEPA provided comments on the Closure Plan in a letter dated October 27, 2017. A response to comment letter submitted by Illinois Power Generating Company (IPGC) dated November 27, 2017 included revisions to the GMP, which defined groundwater monitoring for AP2 following approval of the Closure Plan (Attachment 2). The Closure Plan was approved by IEPA in a letter dated January 30, 2018. Closure of AP2 was completed on November 17, 2020.

On April 21, 2021, 35 I.A.C. § 845 became effective, and 35 I.A.C. § 845.100(i) provides the following with respect to certain CCR units closed prior to the effective date:

i) If a CCR surface impoundment has completed an Agency-approved closure before April 21, 2021, this Part does not require the owner or operator of the CCR surface impoundment to resubmit to the Agency any closure plan, closure report, or closure certification for that completed closure.

SUBMITTALS

The attached documents are being provided to address requirements of 35 I.A.C. § 845.230 as follows:

October 25, 2021

Ramboll 234 W. Florida Street Fifth Floor Milwaukee, WI 53204 USA

T 414-837-3607 F 414-837-3608 www.ramboll.com

Ref. 1940100806-002


- Addendum to the CPP AP2 Groundwater Monitoring Plan (new submittal, Attachment 1). This Addendum includes revisions to the monitoring well network, analytical parameters, and statistical procedures included in the previously submitted Groundwater Monitoring Plan (Natural Resource Technology, Inc. [NRT], 2016). These modifications are proposed to meet and fulfill the requirements in 35 I.A.C. § 845.630 and 35 I.A.C. § 845.640 (Groundwater Monitoring Systems and Statistical Procedures); and 35 I.A.C. § 845.650(b) (background samples). The proposed modifications were identified and developed using existing and previously approved documents, but additional information has been provided where necessary.
- Groundwater Monitoring Plan, dated January 24, 2017, and subsequent Response to Comments (Attachment 2). This attachment provides a copy of the existing groundwater monitoring plan as it was approved by IEPA.

Sincerely,

Eric J. Tlachac, PE // Senior Managing Engineer

D +1 414 837 3541 M +1 262 719 4526 eric.tlachac@ramboll.com

ATTACHMENTS:

Attachment 1 Addendum to the Groundwater Monitoring Plan

Attachment 2 Groundwater Monitoring Plan, dated January 24, 2017 and Response to Comment Letter dated November 27, 2017

Brian G. Hennings, PG Senior Managing Hydrogeologist

D +1 414 837 3524 M +1 262 719 4512 brian.hennings@ramboll.com ATTACHMENT 1 ADDENDUM TO THE GROUNDWATER MONITORING PLAN Intended for Illinois Power Generating Company

Date **October 25, 2021**

Project No. **1940100806-002**

ADDENDUM TO THE GROUNDWATER MONITORING PLAN



ADDENDUM TO THE GROUNDWATER MONITORING PLAN COFFEEN POWER PLANT ASH POND NO. 2

Coffeen Power Plant Ash Pond No. 2
1940100806-002
Illinois Power Generating Company
Addendum to the Groundwater Monitoring Plan
FINAL
October 25, 2021
Kristen L. Theesfeld

Ramboll 234 W. Florida Street Fifth Floor Milwaukee, WI 53204 USA

T 414-837-3607 F 414-837-3608 https://ramboll.com

Brian G. Hennings, PG Senior Managing Hydrogeologist Eric J. Tlachac, PE Senior Managing Engineer

Nathaniel R. Keller Senior Hydrogeologist Kristen L. Theesfeld Hydrogeologist

LICENSED PROFESSIONAL CERTIFICATIONS

35 I.A.C. § 845.630 Groundwater Monitoring Systems (PE)

I, Eric J. Tlachac, a qualified professional engineer in good standing in the State of Illinois, certify that the groundwater monitoring system described in this document (Addendum to the Groundwater Monitoring Plan, Coffeen Power Plant Ash Pond No. 2), meets the intent of 35 I.A.C. § 845.630. The monitoring system was developed based on information included in the IEPA approved Hydrogeologic Site Characterization Report submitted with the IEPA approved Closure and Post Closure Care Plan.

Eric J. Tlachac Qualified Professional Engineer 062-063091 Illinois Date: October 25, 2021

35 I.A.C. § 845.630 Groundwater Monitoring Systems (PG)

I, Brian G. Hennings, a qualified professional geologist in good standing in the State of Illinois, certify that the groundwater monitoring system described in this document (Addendum to the Groundwater Monitoring Plan, Coffeen Power Plant Ash Pond No. 2), meets the intent of 35 I.A.C. § 845.630. The monitoring system was developed based on information included in the IEPA approved Hydrogeologic Site Characterization Report submitted with the IEPA approved Closure and Post Closure Care Plan.

Brian G. Hennings Professional Geologist 196.001482 Illinois Date: October 25, 2021

CONTENTS

Licensed F	Professional Certifications	1
1.	Introduction	5
1.1	Overview	5
1.2	Purpose and Scope	5
2.	Groundwater Monitoring Systems	7
2.1	IEPA Closure Plan Monitoring Program	7
2.2	40 C.F.R. § 257 Monitoring Program	8
2.3	Proposed Part 845 Monitoring Well Network	9
2.3.1	Proposed Additional Wells to the Current IEPA Closure Plan	
	Monitoring	9
2.3.2	Proposed Reduction of Wells from Current IEPA Closure Plan	
	Monitoring	9
2.3.3	Proposed Monitoring Network	10
2.4	Well Abandonment	11
3.	Applicable Groundwater Quality Standards	12
3.1	Groundwater Classification	12
3.2	Statistical Evaluation of Background Groundwater Data	12
3.3	Applicable Groundwater Protection Standards	12
4.	Groundwater Monitoring Plan	14
4.1	Monitoring Networks and Parameters	14
4.1.1	IEPA Closure Plan Monitoring	14
4.1.2	40 C.F.R. § 257 Monitoring	14
4.1.3	Part 845 Groundwater Monitoring	14
4.2	Sampling Schedule	15
4.3	Groundwater Sample Collection	16
4.4	Laboratory Analysis	16
4.5	Quality Assurance Program	16
4.6	Groundwater Monitoring System Maintenance Plan	17
4.7	Statistical Analysis	17
4.8	Data Reporting	17
4.9	Compliance with Applicable On-site Groundwater Quality Standards	18
4.10	Alternate Source Demonstrations	18
5.	References	19

TABLES (IN TEXT)

- Table A
 IEPA Closure Plan Groundwater Monitoring Program Parameters
- Table B
 40 C.F.R. § 257 Groundwater Monitoring Program Parameters
- Table CSummary of Sulfate Concentrations for Wells G154, G407, G410, and G411 from March2018 through August 2021
- Table D
 Proposed Part 845 Monitoring Well Network
- Table E
 Part 845 Groundwater Monitoring Program Parameters
- Table FPart 845 Sampling Schedule

TABLES (ATTACHED)

- Table 2-1
 Monitoring Well Locations and Construction Details
- Table 3-1 Background Groundwater Quality and Standards
- Table 4-1 Sampling and Analysis Summary
- Table 4-2
 Detection and Reporting Limits for Part 845 Parameters

FIGURES (ATTACHED)

- Figure 1-1 Site Location Map
- Figure 1-2 Site Map
- Figure 2-1 IEPA Monitoring Well Locations and Wells G1001 and G1003
- Figure 2-2 Uppermost Aquifer Groundwater Elevation Contours April 20, 2021
- Figure 2-3 Proposed Part 845 Groundwater Monitoring Well Network

APPENDICES

- Appendix A G1001 and G1003 Boring and Well Construction Logs
- Appendix B Statistical Analysis Plan

ACRONYMS AND ABBREVIATIONS

§	Section
35 I.A.C.	Title 35 of the Illinois Administrative Code
77 I.A.C.	Title 77 of the Illinois Administrative Code
40 C.F.R.	Title 40 of the Code of Federal Regulations
AP2	Ash Pond No. 2
ASD	Alternate Source Demonstration
bgs	below ground surface
CCR	coal combustion residuals
Closure Plan	Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2
cm/s	centimeters per second
CPP	Coffeen Power Plant
GMF	Gypsum Management Facility
GMP	Groundwater Monitoring Plan
GWPS	Groundwater Protection Standard
ID	Identification
IEPA	Illinois Environmental Protection Agency
IPGC	Illinois Power Generating Company
LCU	lower confining unit
mg/L	milligrams per liter
NA	not applicable
NID	National Inventory of Dams
No.	Number
NRT	Natural Resource Technology, Inc.
Part 845	Residuals in Surface Impoundments: Title 35 of the Illinois Administrative Code § 845
QA/QC	quality assurance/quality control
Ramboll	Ramboll Americas Engineering Solutions, Inc.
RL	Reporting Limit
SI	Surface Impoundment
SW	surface water
TDS	total dissolved solids
UA	uppermost aquifer
USEPA	United States Environmental Protection Agency
WLO	water level only

1. INTRODUCTION

1.1 Overview

In accordance with requirements of the Standards for the Disposal of Coal Combustion Residuals (CCR) in Surface Impoundments (SIs): Title 35 of the Illinois Administrative Code (35 I.A.C.) Section (§) 845 (Part 845) (Illinois Environmental Protection Agency [IEPA], April 15 2021), Ramboll Americas Engineering Solutions, Inc. (Ramboll) has prepared this Addendum to the Groundwater Monitoring Plan (GMP) on behalf of Coffeen Power Plant (CPP) (**Figure 1-1**), operated by Illinois Power Generating Company (IPGC). This Addendum applies specifically to the CCR Unit referred to as Ash Pond Number (No.) 2 (AP2; Vistra identification [ID] No. 102, IEPA ID No. W1350150004-02, and National Inventory of Dams [NID] No. IL50723) (**Figure 1-2**). AP2 is a closed, unlined CCR SI that was previously used to manage CCR and non-CCR waste streams at the CPP.

AECOM submitted the *Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2* (Closure Plan) dated January 26, 2017, which was approved by the IEPA on January 30, 2018. The Groundwater Monitoring Plan (Natural Resource Technology, Inc. [NRT], 2017a) defined groundwater monitoring for AP2 following approval of the Closure Plan. Closure of AP2 was completed on November 17, 2020.

On April 21, 2021, Part 845 became effective, and for CCR units closed prior to the effective date the following section was included (35 I.A.C. § 845.100(i)):

If a CCR surface impoundment has completed an Agency-approved closure before April 21, 2021, this Part does not require the owner or operator of the CCR surface impoundment to resubmit to the Agency any closure plan, closure report, or closure certification for that completed closure.

This Addendum includes modifications to the previously approved GMP to provide content required by 35 I.A.C. § 845.630 (Groundwater Monitoring System), 35 I.A.C. 845.640 (Groundwater Sampling and Analysis), and 35 I.A.C. 845.650 (Groundwater Monitoring Program) for AP2. Specifically, this Addendum incorporates additional monitoring wells and monitoring parameters specified in 35 I.A.C. § 845.600.

1.2 Purpose and Scope

The purpose of this Addendum is to provide updated GMP text, tables, and figures to incorporate modifications made to the existing monitoring program to comply with Part 845. Following issuance of the Part 845 Operating Permit, the application for which this Addendum is attached, groundwater monitoring at AP2 will include the following:

- Monitoring required by Title 40 of the Code of Federal Regulations (40 C.F.R.) § 257 Subpart D (pre-existing with no modifications)
- Part 845 Monitoring (proposed)

Details of the monitoring programs (schedules and parameters), monitoring well networks, and analysis (statistical methods) are included in this Addendum. No changes are proposed to the monitoring networks utilized for the current 40 C.F.R. § 257 Subpart D monitoring; however, those details have been included for completeness. Additional information regarding the

hydrogeology and groundwater quality were included with the Closure Plan and are not reproduced in this Addendum.

2. GROUNDWATER MONITORING SYSTEMS

The Part 845 groundwater monitoring network for AP2 was developed to monitor post-closure groundwater quality and trends and demonstrate compliance with the applicable groundwater quality standards identified in **Section 3**. The existing and proposed groundwater monitoring well networks consist of a sufficient number of wells, installed at appropriate locations and depths, to monitor post-closure compliance with groundwater quality standards for 40 C.F.R. § 257 and 35 I.A.C. § 845.600.

The monitoring wells are designed and constructed in a manner consistent with the standards of 40 C.F.R. § 257 and Title 77 of the Illinois Administrative Code (77 I.A.C.) § 920.170, as required by 35 I.A.C. § 845.630(e), including the following:

- All monitoring wells are cased in a manner that maintains the integrity of the boreholes.
- Wells are screened to allow sampling only at the specified interval.
- All wells are covered with vented caps, unless located in flood-prone areas, and equipped with devices to protect against tampering and damage.

Consistent with applicable standards, the monitoring well networks described below fulfill the following goals:

- Enable the collection of groundwater samples that represent the quality of background water that has not been affected by AP2.
- Enable the collection of groundwater samples that represent the quality of downgradient groundwater.
- Include wells that are located within the stratigraphic unit(s) that may serve as potential chemical migration pathways.

Groundwater monitoring at AP2 is currently being performed in accordance with the GMP that was approved in the Closure Plan. The Closure Plan specified one monitoring network, that complies with both the IEPA Closure Plan monitoring requirements and the 40 C.F.R. § 257 monitoring requirements. This Addendum proposes additions to that network which were developed to comply with the requirements of Part 845. It is anticipated that upon acceptance and approval of the Operating Permit application (and by extension the GMP) for the CPP and upon acceptance and approval of Part 845 by the United States Environmental Protection Agency (USEPA) as a State CCR Permit Program, the Part 845 monitoring program will supersede the Closure Plan and 40 C.F.R. § 257 monitoring program.

2.1 IEPA Closure Plan Monitoring Program

The approved IEPA Closure Plan monitoring well network consists of fourteen groundwater monitoring wells used to monitor the uppermost aquifer, including three background wells (G270, G280, and G281) and eleven compliance wells (G154, G279, G401, G402, G403, G404, G405, G406, G407, G410, and G411). These wells are monitored in accordance with Water Pollution Control Permit 2020-EA-65027-1 Special Condition No. 6. The well locations are shown on **Figure 2-1**.

The GMP established a monitoring program that meets the requirements of 35 I.A.C. § 620.410 and groundwater samples are collected quarterly and analyzed for the parameters listed in **Table A** below.

Field Parameters		
pH	Temperature	Turbidity
Dissolved Oxygen	Oxidation/Reduction Potential	Specific Conductance
Static Water Elevation	Depth of Well	Depth to Water
Elevation of Groundwater Surfac	9	
Inorganics (Total, except Tot	al Dissolved Solids [TDS])	
Chloride	Fluoride	Sulfate
Cyanide	Nitrate-Nitrogen	TDS
Metals (Total)		
Antimony	Cadmium	Lithium
Arsenic	Calcium	Mercury
Barium	Chromium	Molybdenum
Beryllium	Cobalt	Selenium
Boron	Lead	Thallium
Metals (Dissolved)		
Aluminum	Iron	Silver
Arsenic	Lead	Vanadium
Boron	Manganese	Zinc
Copper	Nickel	
Other (Total)		
Radium 226 and 228 combined		

Table A. IEPA Closure Plan Groundwater Monitoring Program Parameters

2.2 40 C.F.R. § 257 Monitoring Program

The 40 C.F.R. § 257 monitoring well network consists of seven of the same groundwater monitoring wells used to monitor the uppermost aquifer, including two background wells (G270 and G281) and five compliance wells (G401, G402, G403, G404, and G405).

Assessment monitoring in accordance with 40 C.F.R. § 257.95 was initiated on April 9, 2018. Details on the procedures and techniques used to fulfill the groundwater sampling and analysis program requirements are found in the Sampling and Analysis Plan for AP2 (NRT, 2017b).

Groundwater samples are collected semi-annually and analyzed for the following laboratory and field parameters from Appendix III and Appendix IV of 40 C.F.R. § 257, summarized in **Table B** below.

Field Parameters ¹			
Groundwater Elevation	рН		
Appendix III Paramete	ers (Total, except TDS)		
Boron	Chloride	Sulfate	
Calcium	Fluoride	TDS	
Appendix IV Paramete	ers (Total)		
Antimony	Cadmium	Lithium	Selenium
Arsenic	Chromium	Mercury	Thallium
Barium	Cobalt	Molybdenum	Radium 226 and 228 combined
Beryllium	Lead		

Table B.	40 C.F.R.	8 257	Groundwater	Monitorina	Program	Parameters
Table D.	40 0.1 .1.	3 23/	Groundwater	Plotticoring	riogram	i arameters

¹Dissolved oxygen, temperature, specific conductance, oxidation/reduction potential, and turbidity are recorded during sample collection.

2.3 Proposed Part 845 Monitoring Well Network

2.3.1 Proposed Additional Wells to the Current IEPA Closure Plan Monitoring

The land between AP2 and the Unnamed Tributary underwent significant improvement during closure construction which made areas, that were previously inaccessible, available for monitoring well construction. In 2021, additional monitoring wells G1001 and G1003 were completed between the eastern edge of AP2 and the Unnamed Tributary (**Figure 2-1**) targeting the uppermost aquifer (Hagarstown Member), which is classified as primarily sandy to gravelly silts and clays with thin beds of sands. No sand layers were observed in the soil boring at G1001 (**Appendix A**); however, a well was screened from 6 to 11 feet below ground surface (bgs) within the LCU and yields water. Sand layers were observed in the soil boring at G1003 (**Appendix A**) and the well was screened across sandy deposits from 8 to 12 feet bgs; however, this well has not yielded water (no measurable groundwater) since construction in May of 2021. Well G1001 is proposed as an additional well to the network for monitoring groundwater downgradient of AP2. Well G1003 will continue to be monitored for the presence of groundwater.

2.3.2 Proposed Reduction of Wells from Current IEPA Closure Plan Monitoring

Monitoring wells G154, G279, G407, G410, and G411 were included in the IEPA groundwater monitoring plan to monitor sulfate in groundwater that could be attributed to AP2. With the exception of well G407, these wells are proposed for removal from the AP2 monitoring well network as described below:

Monitoring wells G154, G410, and G411. Groundwater elevation measurements collected from these wells confirm the presence of a groundwater divide located west of AP2 (Figure 2-2). Groundwater on the east side of the divide flows toward the Unnamed Tributary while groundwater on the west side of the divide flows toward the western lobe of Coffeen Lake. AP2 is located on the eastern side of the divide; therefore, these wells on the west side of the divide are not downgradient of AP2. Additionally, minimum, maximum, and median sulfate concentrations observed from March 2018 to August 2021 in wells (G154, G410, and G411) have not had concentrations greater than the groundwater protection standard (GWPS) of 400 milligrams per liter (mg/L) as presented in Table C below.

 Monitoring well G279. This well is located along the eastern edge of another CCR unit, the Gypsum Management Facility [GMF] Recycle Pond (Vistra ID No. 104, IEPA ID No. W1350150004-04, and NID No. IL50578). G279 is part of the 40 C.F.R. § 257 monitoring well network and the proposed Part 845 monitoring well network for the GMF Recycle Pond; therefore, this well will be removed from the AP2 monitoring network.

Location	Count	Minimum (mg/L)	Maximum (mg/L)	Median (mg/L)
G154	16	74	160	79
G407	16	400	1600	1100
G410	16	31	55	41
G411	16	110	300	250

Table C. Summary of Sulfate Concentrations for Wells G154	, G407, G410,	and G411 f	rom March
2018 through August 2021.			

Groundwater monitoring well G407 is also located on the east side of the groundwater divide and sulfate concentrations have been steadily decreasing since the well was installed; however, this well will be retained in the proposed monitoring network to continue to monitor potential sulfate exceedances consistent with the existing permit.

2.3.3 Proposed Monitoring Network

The groundwater monitoring network proposed in this plan will include eleven wells screened in the uppermost aquifer, or the LCU where the uppermost aquifer is absent (G270, G280, G281, G401, G402, G403, G404, G405, G406, G407, and G1001). The proposed network is summarized in **Table D** below and displayed on **Figure 2-3**. Eleven wells (three background and eight compliance) will be used to monitor groundwater concentrations within the uppermost aquifer, or the LCU where the uppermost aquifer is absent. Surface water elevations will be monitored at three staff gauges: SG02, monitoring the former discharge flume; SG03, monitoring the eastern lobe of Coffeen Lake; and SG04, monitoring the Unnamed Tributary.

The groundwater samples collected from the eleven wells will be used to monitor and evaluate groundwater quality and demonstrate compliance with the groundwater quality standards listed in 35 I.A.C. § 845.600(a). The proposed monitoring wells will yield groundwater samples that represent the quality of downgradient groundwater at the CCR boundary (as required in 35 I.A.C. § 845.630(a)(2)). Monitoring well depths and construction details are listed in **Table 2-1** and summarized in **Table D** below.

Well ID	Monitored Unit	Well Screen Interval (feet bgs)	Well Type ¹
G270	UA	13.1-17.9	Background
G280	UA	12.8-17.6	Background
G281	UA	15.5-20.2	Background
G401	UA	14.4-18.8	Compliance
G402	UA	10-20	Compliance
G403	UA	13.1-17.8	Compliance
G404	UA	6.4-11.2	Compliance
G405	UA	9.0-13.8	Compliance
G406	UA	13.6-18.4	Compliance
G407	UA	13.8-18.6	Compliance
G1001	LCU	6-11	Compliance
SG02	SW	NA	WLO
SG03	SW	NA	WLO
SG04	SW	NA	WLO

Table D. Proposed Part 845 Monitoring Well Network

 $^{\rm 1}\,{\rm Well}$ type refers to the role of the well in the monitoring network.

bgs = below ground surface

LCU = lower confining unit

NA = not applicable

SW = surface water

UA = uppermost aquifer

WLO = water level only

2.4 Well Abandonment

No wells are currently proposed for abandonment.

3. APPLICABLE GROUNDWATER QUALITY STANDARDS

3.1 Groundwater Classification

The 35 I.A.C. § 620 groundwater classification at AP2 was presented in the GMP (NRT, 2017a) and is summarized here. Groundwater at AP2 meets the definition of Class I - Potable Resource Groundwater (35 I.A.C. § 620.210) based on the following criteria:

- Groundwater in the uppermost aquifer is located 10 feet or more below the land surface and within,
- A geologic material which is capable of a hydraulic conductivity of 1 x 10⁻⁴ centimeters per second (cm/s) or greater using a slug test.

3.2 Statistical Evaluation of Background Groundwater Data

A Statistical Analysis Plan (**Appendix B**) has been developed to describe procedures that will be used to establish background conditions and implement compliance monitoring as necessary and required by 35 I.A.C. § 845.640 and 35 I.A.C. § 845.650. The Statistical Analysis Plan was prepared in accordance with the requirements of 35 I.A.C. § 845.640(f), with reference to the acceptable statistical procedures provided in USEPA's *Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities, Unified Guidance (Unified Guidance, March 2009)*, and is intended to provide a logical process and framework for conducting the statistical analysis of the data obtained during groundwater monitoring.

In accordance with 35 I.A.C. § 845.640(f)(1), the statistical method chosen for analysis of background groundwater quality was either the tolerance interval or the prediction interval procedure for each constituent listed in 35 I.A.C. § 845.600(a)(1) at this CCR unit per 35 I.A.C. § 845.640(f)(1)(C). A comparison of the statistical background concentrations and groundwater quality standards listed in 35 I.A.C. § 845.600(a)(1) and the resulting GWPSs are summarized in **Table 3-1**.

3.3 Applicable Groundwater Protection Standards

The applicable GWPS will be established in accordance with 35 I.A.C. § 845.600(a) (greater of the background concentration or numerical limit specified in 35 I.A.C. § 845.600(a)(1)). The results of the statistical analysis of background groundwater data (**Table 3-1**) indicate that background concentrations in the uppermost aquifer are less than the groundwater quality standards listed in 35 I.A.C. § 845.600(a)(1) with the exception of lead. Therefore, for these parameters, the groundwater quality standards listed in 35 I.A.C. § 845.600(a)(1) will be applied to the results from the proposed groundwater monitoring network. The background concentration for lead will be applied to the lead results from the proposed groundwater monitoring network.

Under most circumstances, the GWPS will be compared to the lower confidence limit for the observed concentrations for each constituent in each compliance well. Exceptions are when there are high percentages (greater than 50 percent) of non-detects in compliance well data, for which a future mean (for 50 to 70 percent non-detects) or median (for greater than 70 percent non-detects) will be compared to the GWPS. Consistent with the *Unified Guidance*, the same general statistical method of confidence interval testing against a fixed GWPS is recommended in compliance and corrective action programs. Confidence intervals provide a flexible and statistically accurate method to test how a parameter estimated from a single sample compares

to a fixed numerical limit. Confidence intervals explicitly account for variation and uncertainty in the sample data used to construct them.

Evaluation of the applicable standards will occur in conjunction with the analysis of groundwater quality results. Background calculations and the resulting concentrations may be updated as appropriate, in accordance with the Statistical Analysis Plan included in **Appendix B**.

4. GROUNDWATER MONITORING PLAN

The groundwater monitoring plan will monitor and evaluate groundwater quality to demonstrate compliance with the groundwater quality standards included in 40 C.F.R. § 257.94(e), 40 C.F.R. § 257.95(h), and 35 I.A.C. § 845.600(a). The groundwater monitoring program will include sampling and analysis procedures that are consistent and provide an accurate representation of groundwater quality at the background and compliance wells as required by 35 I.A.C. § 845.630. As discussed within **Section 2**, three monitoring programs specific to AP2 exist, the Closure Plan monitoring program, the 40 C.F.R. § 257 monitoring program, and the proposed Part 845 monitoring program. It is expected that upon acceptance and approval of the Operating Permit applications (and by extension the GMPs) for the CPP and upon acceptance and approval of Part 845 by the USEPA as a State CCR Permit Program, the proposed Part 845 monitoring program will supersede the Closure Plan and 40 C.F.R. § 257 monitoring programs.

4.1 Monitoring Networks and Parameters

4.1.1 IEPA Closure Plan Monitoring

The existing IEPA-approved Closure Plan monitoring program was discussed in detail in **Section 2.1**. Fourteen groundwater monitoring wells are used to monitor the uppermost aquifer, including three background wells (G270, G280, and G281) and eleven compliance wells (G154, G279, G401, G402, G403, G404, G405, G406, G407, G410, and G411), and are sampled on a quarterly frequency for the parameters listed in the GMP (NRT, 2017a). Well locations and parameters will continue to be monitored and reported as required by the Closure Plan until IEPA approves the proposed Part 845 monitoring network.

4.1.2 40 C.F.R. § 257 Monitoring

The existing 40 C.F.R. § 257 monitoring program was discussed in detail in **Section 2.2**. Seven groundwater monitoring wells, including two background wells (G270 and G281) and five compliance wells (G401, G402, G403, G404, and G405) are sampled for Appendix III and Appendix IV parameters on a semi-annual frequency. Well locations and parameters will continue to be monitored and reported as required by 40 C.F.R. § 257 until USEPA approves Part 845.

4.1.3 Part 845 Groundwater Monitoring

The proposed Part 845 Monitoring Network will consist of eleven monitoring wells, three background monitoring wells (G270, G280, and G281), eight compliance wells (G401, G402, G403, G404, G405, G406, G407, and G1001), to monitor potential impacts from AP2 (**Figure 2-2**). These monitoring wells are screened within the uppermost aquifer, or the LCU where the uppermost aquifer is absent, along the perimeter of or near AP2 and include G407, located west of AP2, which is retained to continue monitoring sulfate as described in **Sections 2.3.1** and **2.3.2**. Three staff gauges (SG02, SG03, and SG04) will be used to monitor surface water elevations adjacent to AP2. Groundwater samples will be collected and analyzed for the laboratory and field parameters summarized in **Table E** below.

Field Parameters ¹			
Groundwater Elevation	рН	Turbidity	
Metals (Total)			
Antimony	Boron	Cobalt	Molybdenum
Arsenic	Cadmium	Lead	Selenium
Barium	Calcium	Lithium	Thallium
Beryllium	Chromium	Mercury	
Inorganics (Total)			
Chloride	Fluoride	Sulfate	TDS
Other (Total)			

Table E.	Part 845	Groundwater	Monitoring	Program	Parameters
	1 410 45	Groundwater	Plomeoring	riogram	i arameters

Radium 226 and 228 combined

 1 Dissolved oxygen, temperature, specific conductance, and oxidation/reduction potential will be recorded during sample collection.

4.2 Sampling Schedule

Groundwater sampling for the approved Closure Plan will be maintained until IEPA approval of the Part 845 GMP. Groundwater sampling for the Part 845 monitoring well network will initially be performed quarterly according to the following schedule:

Frequency	Duration
Monthly	Begins: the quarter following approval of this plan and issuance of the Operating Permit.
(groundwater elevations only)	Ends: Following the 30-year post closure care period and following IEPA approval of documentation that groundwater concentrations are below standards in 35 I.A.C. § 845.600 and concentrations exceeding background are not increasing and meet requirements in 35 I.A.C. § 845.780 (c)(2)(B)(i) and (ii).
Quarterly	Begins: the quarter following approval of this plan and issuance of the Operating Permit.
(groundwater quality)	Ends: Following the 30-year post closure care period and following IEPA approval of documentation that groundwater concentrations are below standards in 35 I.A.C. § 845.600 and concentrations exceeding background are not increasing and meet requirements in 35 I.A.C. § 845.780 (c)(2)(B)(i) and (ii), or upon IEPA approval of an alternate schedule as allowed by 35 I.A.C. § 845.650(b)(4).
Semi-annual (groundwater quality)	Begins: Following 5 years of quarterly groundwater monitoring and IEPA approval of a demonstration that groundwater concentrations are below standards in 35 I.A.C. § 845.600 and not exhibiting statistically-significant increasing trends, monitoring effectiveness is not compromised by a semi-annual schedule, and sufficient data has been collected to characterize groundwater.
	Ends: Following detection of a statistically-significant increasing trend in groundwater concentrations or an exceedance of the standards in 35 I.A.C. § 845.600 (quarterly monitoring shall be resumed in these circumstances), or following the 30-year post closure care period and following IEPA approval of documentation that groundwater concentrations are below standards in 35 I.A.C. § 845.600 and concentrations exceeding background are not increasing and meet requirements in 35 I.A.C. § 845.780 (c)(2)(B)(i) and (ii).

Table F. Part 845 Sampling Schedule

Groundwater monitoring for the 40 C.F.R. § 257.94 well network will continue to follow a schedule in accordance with the requirements of 40 C.F.R. § 257.94 and 40 C.F.R. § 257.95. Upon USEPA approval of Part 845 as a State CCR Permit Program, the 40 C.F.R. § 257.94 monitoring will be discontinued, and replaced by the Part 845 monitoring.

4.3 Groundwater Sample Collection

Groundwater sampling procedures have been developed and the collection of groundwater samples is being implemented to meet the requirements of 35 I.A.C. § 845.640. In addition to groundwater well samples, quality assurance samples will be collected as described in **Section 4.5 (Table 4-1)**.

4.4 Laboratory Analysis

Laboratory analysis will be performed consistent with the requirements of 35 I.A.C. § 845.640(j) by a state-certified laboratory using methods approved by IEPA and USEPA. Laboratory methods may be modified based on laboratory equipment availability or procedures, but the Reporting Limit (RL) for all parameters analyzed, regardless of method, will be lower than the applicable groundwater quality standard. RLs for the applicable parameters are summarized in **Table 4-2**. Concentrations lower than the RL will be reported as less than the RL.

4.5 Quality Assurance Program

Consistent with the requirements of 35 I.A.C. § 845.640(a)(5), the sampling and analysis program includes procedures and techniques for quality assurance/quality control (QA/QC). Additional quality assurance samples to be collected will include the following:

- Field duplicates will be collected at a frequency of one per group of ten or fewer investigative water samples.
- One equipment blank sample will be collected and analyzed for each day of sampling. If dedicated sampling equipment is used, then equipment blank samples will not be collected.
- The duplicate and equipment blank quality assurance samples will be supplemented by the laboratory QA/QC program, which typically includes:
 - Regular generation of instrument calibration curves to assure instrument reliability
 - Laboratory control samples and/or quality control check standards that have been spiked, and analyses to monitor the performance of the analytical method
 - Matrix spike/matrix spike duplicate analyses to determine percent recoveries and relative percent differences for each of the parameters detected
 - Analysis of replicate samples to check the precision of the instrumentation and/or methodology employed for all analytical methods
 - Analysis of method blanks to assure that the system is free of contamination

Water quality meters used to measure pH and turbidity will be calibrated according to manufacturer's specifications. At a minimum, it is recommended that calibration of pH occur daily prior to sampling and checked for accuracy at the end of each day. Unusual or suspect pH measurements during sampling events will be flagged, evaluated, and additional calibration may be performed throughout the sampling events. Turbidity meters will be checked daily, prior to

and following sampling. Unusual measurements or erratic meter performance will be flagged and evaluated for overall effects on the data prior to reporting.

4.6 Groundwater Monitoring System Maintenance Plan

Consistent with the requirements of 35 I.A.C. § 845.630(e)(2), maintenance will be performed as needed to assure that the monitoring wells provide representative groundwater samples. Monitoring wells will be inspected during each groundwater sampling event; inspections will consist of the following:

- Visual inspection, clearing of vegetation, replacement of markers, and painting of protective casings as needed to assure that monitoring wells are clearly marked and accessible.
- Visual inspection and repair or replacement of well aprons as needed to assure that they are intact, drain water away from the well, and have not heaved.
- Visual inspection and repair or replacement of protective casings as needed to assure that they are undamaged, and that locks are present and functional.
- Checks to assure that well caps are intact and vented, unless in flood-prone areas in which case caps will not be vented.
- Annual measurement of monitoring well depths to determine the degree of siltation within the wells. Wells will be redeveloped as needed to remove siltation from the screened interval if it impedes flow of water into the well.
- Checks to assure that wells are clear of internal obstructions, and flow freely.

If maintenance of a monitoring well cannot address an identified deficiency, a replacement well will be installed.

4.7 Statistical Analysis

Statistical analysis will be consistent with procedures listed in 35 I.A.C. § 845.640(f). A Statistical Analysis Plan, provided in **Appendix B**, has been developed to summarize the statistical procedures that will be used to evaluate the groundwater results.

4.8 Data Reporting

Data reporting for the 40 C.F.R. § 257 monitoring program will be consistent with recordkeeping, notification, and internet posting requirements described in 40 C.F.R. § 257.105 through 40 C.F.R. § 257.107.

Groundwater monitoring and analysis completed as part of the Part 845 monitoring under an approved monitoring program will be reported to IEPA within 60 days after completion of sampling and place the data in the facility's operating record as required by 35 I.A.C. § 845.610(b)(3)(D). Within 14 days of posting to the operating record, information will be posted to the publicly accessible internet site "Illinois CCR Rule Compliance Data and Information" as required by 35 I.A.C. § 845.810(d). Information will also be provided to IEPA annually, by January 31 for data collected the preceding year. The report will include the status of the groundwater monitoring and corrective action plan for the CPP AP2 in addition to other requirements detailed in 35 I.A.C. § 845.610(e).

4.9 Compliance with Applicable On-site Groundwater Quality Standards

In accordance with 35 I.A.C. § 845.600(a)(1), the groundwater protection standard at the waste boundary will be the higher of either the 35 I.A.C. § 845.600 standard or the concentration determined by background groundwater monitoring.

As provided in 35 I.A.C. § 845.780(c)(2), at the end of the 30-year post-closure care period, groundwater monitoring will continue to be conducted in post-closure care until the groundwater results show the concentrations are:

- Below the GWPS in 35 I.A.C. § 845.600; and
- Not increasing for those constituents over background, using the statistical procedures and performance standards in 35 I.A.C. § 845.640(f) and (g), provided that:
 - Concentrations have been reduced to the maximum extent feasible; and
 - Concentrations are protective of human health and the environment.

Following detection of an exceedance of the GWPS, an Alternate Source Demonstration (ASD) will be evaluated as described in **Section 4.10**.

4.10 Alternate Source Demonstrations

As allowed in 35 I.A.C. § 845.650(e), following detection of an exceedance of the GWPS, an ASD will be evaluated and, if completed, submitted to IEPA within 60 days. The ASD will provide lines of evidence that a source other than AP2 caused the contamination and AP2 did not contribute to the contamination, or that the exceedance of the GWPS resulted from error in sampling, analysis, statistical evaluation, natural variation in groundwater quality, or a change in the potentiometric surface and groundwater flow direction.

The ASD will include information and analysis that supports the conclusions and a certification of accuracy by a qualified professional engineer. Once the ASD is approved by IEPA, the Part 845 groundwater monitoring will continue as defined in **Section 4.1.3**.

If an ASD is not completed and submitted, or IEPA does not approve the ASD, a notification of the exceedance will be provided to IEPA and placed in the operating record. Additional actions will also be completed as required by 35 I.A.C § 845.650(d)(1) through (3), including initiation of an assessment of corrective measures under 35 I.A.C § 845.660. As allowed in 35 I.A.C § 845.650(e)(7), a petition for review of IEPA's non-concurrence under 35 I.A.C. § 105 may also be filed.

5. REFERENCES

AECOM, 2017. Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2 at Illinois Power Generating Company, Coffeen Power Station, 134 Cips Lane, Coffeen, IL 62107, January 26, 2017.

Illinois Environmental Protection Agency, 2021. *Standards for the Disposal of Coal Combustion Residuals in Surface Impoundments: Title 35 of the Illinois Administrative Code § 845*, April 15, 2021.

Natural Resource Technology, Inc. (NRT), 2017a. *Groundwater Monitoring Plan, Ash Pond No. 2, Coffeen Power Station*, January 24, 2017.

Natural Resource Technology, Inc. (NRT), 2017b. *Sampling and Analysis Plan, Final, Coffeen Ash Pond No. 2, Coffeen Power Station, Coffeen, Illinois, Project No. 2285, Revision 0*. October 17, 2017

United States Environmental Protection Agency (USEPA), March 2009. Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities – Unified Guidance. Office of Resource Conservation and Recovery, Program Implementation and Information Division, United States Environmental Protection Agency, Washington D.C. EPA/530/R-09/007.

TABLES

TABLE 2-1. MONITORING WELL LOCATIONS AND CONSTRUCTION DETAILS

GROUNDWATER MONITORING PLAN COFFEEN POWER PLANT ASH POND NO. 2 COFFEEN, ILLINOIS

Well Number	Туре	HSU	Date Constructed	Top of PVC Elevation (ft)	Measuring Point Elevation (ft)	Measuring Point Description	Ground Elevation (ft)	Screen Top Depth (ft BGS)	Screen Bottom Depth (ft BGS)	Screen Top Elevation (ft)	Screen Bottom Elevation (ft)	Well Depth (ft BGS)	Bottom of Boring Elevation (ft)	Screen Length (ft)	Screen Diameter (inches)	Latitude (Decimal Degrees)	Longitude (Decimal Degrees)
G1001	С	UA	04/05/2021	597.61		Top of PVC	594.82	6.00	11.00	588.82	583.82	11.00	562.82	5	4	39.063324	-89.391236
G270	В	UA	02/26/2008		625.86	Top of Disk	623.73	13.13	17.92	610.60	605.81	18.27	605.50	4.8	2	39.066564	-89.397403
G280	В	UA	02/26/2008	625.35	625.35	Top of Riser	623.11	12.79	17.63	610.32	605.48	17.98	605.10	4.8	2	39.067216	-89.394992
G281	В	UA	09/08/2015		626.36	Top of Disk	623.82	15.51	20.16	608.31	603.66	20.30	603.50	4.7	2	39.065405	-89.399322
G401	С	UA	09/14/2015		625.57	Top of Disk	623.03	14.36	18.79	608.67	604.24	19.29	603.70	4.4	2	39.060259	-89.395295
G402	С	UA	08/27/2010		613.37	Top of Disk	610.36	10.00	20.00	600.36	590.36	20.40	590.00	10	2	39.060207	-89.391712
G403	С	UA	09/11/2015		626.47	Top of Disk	623.81	13.11	17.78	610.70	606.03	18.15	605.70	4.7	2	39.063167	-89.398779
G404	С	UA	05/01/2007		615.67	Top of Disk	613.57	6.42	11.17	607.15	602.40	11.62	601.60	4.8	2	39.064329	-89.392493
G405	С	UA	05/01/2007		623.63	Top of Disk	621.40	9.01	13.76	612.39	607.64	14.21	607.20	4.8	2	39.064345	-89.396234
G406	С	UA	08/19/2016	625.36	625.36	Top of PVC	621.86	13.56	18.37	608.30	603.49	18.75	603.10	4.8	2	39.060309	-89.398508
G407	С	UA	08/16/2016	621.32	621.32	Top of PVC	618.35	13.78	18.61	604.57	599.74	19.04	598.40	4.8	2	39.061574	-89.402004
SG-02	WLO	SW			605.87	Top of Prot Casing	605.87									39.059695	-89.391429
SG-03	WLO	SW			594.94	Top of Prot Casing	594.94									39.059092	-89.390342
SG-04	WLO	SW			599.52	Top of Prot Casing	599.52									39.064146	-89.390504

Notes:

All elevation data are presented relative to the North American Vertical Datum 1988 (NAVD88), GEOID 12A Type refers to the role of the well in the monitoring network: background (B), compliance (C), or water level measurements only (WLO) WLO wells are temporary pending implementation of impoundment closure per an approved Construction Permit application

-- = data not available

BGS = below ground surface

ft = foot or feet HSU = Hydrostratigraphic Unit PVC = polyvinyl chloride

SW = surface water

UA = uppermost aquifer

generated 10/05/2021, 3:12:20 PM CDT

TABLE 3-1. BACKGROUND GROUNDWATER QUALITY AND STANDARDS

GROUNDWATER MONITORING PLAN COFFEEN POWER PLANT ASH POND NO. 2 COFFEEN, ILLINOIS

Parameter	Background Concentration	845 Limit	Groundwater Protection Standard	Unit
Antimony, total	0.003	0.006	0.006	mg/L
Arsenic, total	0.0066	0.010	0.010	mg/L
Barium, total	0.14	2.0	2.0	mg/L
Beryllium, total	0.001	0.004	0.004	mg/L
Boron, total	0.029	2	2	mg/L
Cadmium, total	0.001	0.005	0.005	mg/L
Chloride, total	75	200	200	mg/L
Chromium, total	0.019	0.1	0.1	mg/L
Cobalt, total	0.0059	0.006	0.006	mg/L
Fluoride, total	0.513	4.0	4.0	mg/L
Lead, total	0.012	0.0075	0.012	mg/L
Lithium, total	0.019	0.04	0.04	mg/L
Mercury, total	0.0002	0.002	0.002	mg/L
Molybdenum, total	0.0045	0.1	0.1	mg/L
pH (field)	7.5 / 6.6	9.0 / 6.5	9.0 / 6.5	SU
Radium 226 and 228 combined	1.89	5	5	pCi/L
Selenium, total	0.0048	0.05	0.05	mg/L
Sulfate, total	370	400	400	mg/L
Thallium, total	0.001	0.002	0.002	mg/L
Total Dissolved Solids	840	1200	1200	mg/L

Notes:

For pH, the values presented are the upper / lower limits GWPS for calcium and turbidity do not apply per 35 I.A.C. § 845.600(b) mg/L = milligrams per liter SU = standard unitspCi/L = picocuries per litergenerated 10/06/2021, 4:21:50 PM CDT

TABLE 4-1. SAMPLING AND ANALYSIS SUMMARY

ADDENDUM TO THE GROUNDWATER MONITORING PLAN COFFEEN POWER PLANT ASH POND NO. 2 COFFEEN, ILLINOIS

Parameter	Analytical Method ¹	Number of Samples	Field Duplicates ²	Field Blanks ³	Equipment Blanks ³	MS/MSD ⁴	Total	Container Type	Minimum Volume ⁵	Preservation (Cool to 4 °C for all samples)	Sample Hold Time from Collection Date
Metals											
Metals ⁶	6020, Li - EPA 200.7	11	2	0	0	1	14	plastic	600 mL	HNO ₃ to pH<2	6 months
Mercury	7470A or 6020	11	2	0	0	1	14	plastic	400 mL	HNO ₃ to pH<2	28 days
Inorganic Parameters											
Fluoride	9214 or EPA 300	11	2	0	0	1	14	plastic	300 mL	Cool to 4 °C	28 days
Chloride	9251 or EPA 300	11	2	0	0	1	14	plastic	100 mL	Cool to 4 °C	28 days
Sulfate	9036 or EPA 300	11	2	0	0	1	14	plastic	50 mL	Cool to 4 °C	28 days
Total Dissolved Solids	SM 2540 C	11	2	0	0	1	14	plastic	200 mL	Cool to 4 °C	7 days
Radium											
Radium 226	9315 or EPA 903	11	0	0	0	0	11	plastic	1000 mL	HNO_3 to $pH<2$	6 months
Radium 228	9320 or EPA 904	11	0	0	0	0	11	plastic	1000 mL	HNO ₃ to pH<2	6 months
Field Parameters											
рН	SM 4500-H+ B	11	NA	NA	NA	NA	11	flow-through cell	NA	none	immediately
Dissolved Oxygen ⁸	SM 4500-0/405.1	11	NA	NA	NA	NA	11	flow-through cell	NA	none	immediately
Temperature ⁸	SM 2550	11	NA	NA	NA	NA	11	flow-through cell	NA	none	immediately
Oxidation/Reduction Potential ⁸	SM 2580 B	11	NA	NA	NA	NA	11	flow-through cell	NA	none	immediately
Specific Conductance ⁸	SM 2510 B	11	NA	NA	NA	NA	11	flow-through cell	NA	none	immediately
Turbidity ⁷	SM 2130 B	11	NA	NA	NA	NA	11	flow-through cell or hand-held turbidity meter	NA	none	immediately

Notes:

¹ Analytical method numbers are from SW-846 unless otherwise indicated. Analytical methods may be updated with more recent versions as appropriate.

² Field duplicates will be collected at a frequency of one per group of 10 or fewer investigative water samples. Field duplicates will not be collected for radium analysis.

³ Field blanks will be collected at the discretion of the project manager; Equipment blanks will be collected at a rate of 1 per sampling event if non-dedicated equipment is used.

⁴ Matrix Spike/Matrix Spike Duplicate (MS/MSD) samples will be collected at a frequency of one per group of 20 or fewer investigative water samples per CCR unit/multi-unit. Additional volume to be determined by laboratory.

⁵ Sample volume is estimated and will be determined by the laboratory.

⁶ Metals = antimony, arsenic, barium, beryllium, boron, cadmium, calcium, chromium, cobalt, lead, lithium, molybdenum, selenium, thallium. Metals may be analyzed via ICP/ ICP-MS USEPA methods 6010 or 6020 depending on laboratory instrument availability.

⁷ If turbidity exceeds 10 NTUs, a duplicate sample filtered through a .45 micron filter may be collected for metals analysis in addition to the unfiltered sample. Both samples would be submitted for analysis.

⁸ Parameter collected for quality assurance and quality control for field sampling purposes only; not required to be collected or reported under Part 845; collection of parameter may be discontinued without notification. < = less than

^oC = degrees Celsius

 $HNO_3 = nitric acid$

mL = milliliter

NA = not applicable

NTU = nephelometric turbidity unit

[O: CJC 08/18/21; C: LDC 08/31/21][U:KLT 10/4/21; C: CJC 10/4/21]



TABLE 4-2. DETECTION AND REPORTING LIMITS FOR PART 845 PARAMETERS

ADDENDUM TO THE GROUNDWATER MONITORING PLAN

COFFEEN POWER PLANT

ASH POND NO. 2

COFFEEN, ILLINOIS

Constituent	CAS	Unit	Analytical Methods ¹	USEPA MCL ²	35 I.A.C. § 845.600	RL ^{4, 5}	MDL ⁵
Metals							
Antimony	7440-36-0	mg/L	6020	0.006	0.006	0.003	0.00036
Arsenic	7440-38-2	mg/L	6020	0.01	0.01	0.001	0.00013
Barium	7440-39-3	mg/L	6020	2	2	0.001	0.00028
Beryllium	7440-41-7	mg/L	6020	0.004	0.004	0.001	0.000017
Boron	7440-42-8	mg/L	6020	NS	2	0.01	0.0023
Cadmium	7440-43-9	mg/L	6020	0.005	0.005	0.001	0.000042
Calcium	7440-70-2	mg/L	6020	NS	NS	0.15	0.15
Chromium	7440-47-3	mg/L	6020	0.1	0.1	0.004	0.00027
Cobalt	7440-48-4	mg/L	6020	0.006	0.006	0.002	0.000017
Lead	7439-92-1	mg/L	6020	0.015	0.0075	0.001	0.000025
Lithium	7439-93-2	mg/L	6020 or EPA 200.7	0.04	0.04	0.02	0.0001
Mercury	7439-97-6	mg/L	6020 or 7470A	0.002	0.002	0.0002	0.000078
Molybdenum	7439-98-7	mg/L	6020	0.1	0.1	0.001	0.000063
Selenium	7782-49-2	mg/L	6020	0.05	0.05	0.001	0.00032
Thallium	7440-28-0	mg/L	6020	0.002	0.002	0.001	0.000062
Inorganics							
Fluoride	7681	mg/L	9214 or EPA 300	4	4	0.25	0.065
Chloride	16887-00-6	mg/L	9251 or EPA 300	250 ³	200	1	0.15
Sulfate	18785-72-3	mg/L	9036 or EPA 300	250 ³	400	1	0.24
Total Dissolved Solids	10052	mg/L	SM 2540C	500 ³	1200	17	
Other							
Radium 226 and 226 combined	7440-14-4	pCi/L	9315/9320 or EPA 903/904	5	5	6	7
Field							
рН	NA	SU	SM 4500-H+ B	NS	6.5-9.0	NA	NA
Oxidation/Reduction Potential	NA	mV	SM 2580 B	NS	NS	NA	NA
Dissolved Oxygen	NA	mg/L	SM 4500-0/405.1	NS	NS	NA	NA
Temperature	NA	°C	SM 2550	NS	NS	NA	NA
Specific Conductivity	NA	µS/cm	SM 2510 B	NS	NS	NA	NA



TABLE 4-2. DETECTION AND REPORTING LIMITS FOR PART 845 PARAMETERS

ADDENDUM TO THE GROUNDWATER MONITORING PLAN

COFFEEN POWER PLANT

ASH POND NO. 2

COFFEEN, ILLINOIS

Constituent	CAS	Unit	Analytical Methods ¹	USEPA MCL ²	35 I.A.C. § 845.600	RL ^{4, 5}	MDL ⁵
Turbidity	NA	NTU	SM 2130 B	NS	NS	NA	NA

[O: CJC 08/18/21; C: LDC 08/31/21],[U:KLT 9/23/21; C: CJC 10/4/21]

Notes:

¹ Analytical method numbers are from SW-846 unless otherwise indicated. Metals will be analyzed via Method 6020 or 6010 depending on laboratory equipment availability. Selected method will ensure reporting limits (RL) are below Title 35 of the Illinois Administrative Code (35 I.A.C.) § 845.600 groundwater protection standards.

² USEPA MCL = United States Environmental Protection Agency Maximum Contaminant Level.

³ USEPA SMCL = United States Environmental Protection Agency Secondary Maximum Contaminant Level.

⁴ RLs will be less than the 35 I.A.C. § 845.600 groundwater protection standards.

⁵ RLs and method detection limits (MDL) will vary depending on the laboratory performing the work.

⁶ All radium results will be reported (values may be positive or negative) and will include uncertainty and the calculated MDC.

⁷ Laboratories calculate a minimum detectable concentration (MDC) based on the sample.

°C = degrees Celsius

 μ S/cm = microSiemens per centimeter

CAS = Chemical Abstract Number

MDL = Method detection limit as established by the laboratory

mg/L = milligrams per liter

mV = millivolts

NA = Not applicable

NS = No standard

NTU = nephelometric turbidity unit

pCi/L = picoCuries per liter

RL = Reporting limit as established by the laboratory

SM = Standard Methods for the Examination of Water and Wastewater

SU = standard units



FIGURES



SITE	LOCAT	MAP

FIGURE 1-1

RAMBOLL AMERICAS ENGINEERING SOLUTIONS, INC.



ADDENDUM TO THE GROUNDWATER MONITORING PLAN ASH POND NO.2 COFFEEN POWER PLANT COFFEEN, ILLINOIS



0 1,000 2,000





PART 845 REGULATED UNIT (SUBJECT UNIT) SITE FEATURE LIMITS OF FINAL COVER	SITE MAP	FIGURE 1-2
	ADDENDUM TO THE GROUNDWATER MONITORING PLAN	RAMBOLL AMERICAS ENGINEERING SOLUTIONS, INC.
0 275 550	COFFEEN POWER PLANT COFFEEN, ILLINOIS	RAMBOLL

, CNES/Airbus DS, USDA, USGS, AeroGRID, IGN, and the GIS Us



EPA MONITORING WELL

HONITORING WELL

PART 845 REGULATED UNIT (SUBJECT UNIT) SITE FEATURE LIMITS OF FINAL COVER PROPERTY BOUNDARY

IEPA MONITORING WELL LOCATIONS AND

ADDENDUM TO THE GROUNDWATER MONITORING PLAN

FIGURE 2-1

RAMBOLL AMERICAS ENGINEERING SOLUTIONS, INC.





PROJECT: 169000XXXX | DATED: 10/6/2021 | DESIGNER: STOLZSD

Y:\Mapping\Projects\22\2285\MXD\845_Operating_Permit\Coffeen\AP2_GMP\Figure 1-3_GWE Contours 20210419 AP2.mxd



BACKGROUND WELL

- HONITORING WELL
- GROUNDWATER ELEVATION CONTOUR (2-FT CONTOUR INTERVAL, NAVD88) INFERRED GROUNDWATER ELEVATION CONTOUR

0	275	550
		Feet

PART 845 REGULATED UNIT (SUBJECT UNIT)

LIMITS OF FINAL COVER

PROPERTY BOUNDARY

NOTE:

FOR CONTOURING.

NM = NOT MEASURED

--

ELEVATIONS IN PARENTHESES WERE NOT USED

UPPERMOST AQUIFER GROUNDWATER ELEVATION CONTOURS APRIL 20, 2021

DWATER FIGURE 2-2 NTOURS

> RAMBOLL AMERICAS ENGINEERING SOLUTIONS, INC.



ADDENDUM TO THE GROUNDWATER MONITORING PLAN



COMPLIANCE WELL PART 845 REGULATED UNIT (SUBJECT UNIT) SITE FEATURE 🖶 BACKGROUND WELL LIMITS OF FINAL COVER HONITORING WELL PROPERTY BOUNDARY General Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Gamma → Ga

PROPOSED 845 GROUNDWATER MONITORING WELL NETWORK

ADDENDUM TO THE GROUNDWATER MONITORING PLAN

FIGURE 2-3

RAMBOLL AMERICAS ENGINEERING SOLUTIONS, INC.



APPENDICES
APPENDIX A G1001 & G1003 BORING AND WELL CONSTRUCTION LOGS

Geosyntec [▷]						Client: Dynegy WELI				L LOG		
	C	onsulta	nts			Proje	ct: GLP8005, Coffeen Power Station		Well No.	G1	001 f 2	
engine	eers sci	entists inno	vators			Addr	ess: Coffeen, IL 62017		raye.		1 2	
Drilling Start Date: 04/05/2021							Boring Depth (ft): 32	Well [Depth (ft):	12		
Drilling En	d Date	e: 04/0	5/202 orte i	21 Drillin	a		Boring Diameter (in): 6	Scree	Diameter (in):	1	In	
Drilling Me	ethod:	y. Rob Dire	ct Pu	ish	9		DTW After Drilling (ft):	Riser	Material:	Sch	40 PVC	
Drilling Eq	uipme	ent: Geo	prob	e			Top of Casing Elev. (ft)	Scree	n Material:	Sch	40 PVC Slotte	ed
Driller:							Ground Elev. (ft):	Seal I	Material(s):	Ben	tonite Chips	
Logged By	y:	A . T	oye				Northing, Easting (NAD83):	Filter	Pack:	San	d	
		7		COLI	EC	Т					MEASURE	
DEPTH (ft)	ATER LEVE	WELL	ample Type	covery (in)	ow Counts	l Value RQD (%)	SOIL/ROCK VISUAL DESC	CRIPT	ON		ab Sample	DEPTH (ft)
	3	0	လိ	Re	Ē						Га	
0			DP	24/48			(0') SILT (ML); very dark grayish brown (2.5Y	3/2), m	edium soft, m	oist,		Γ 0
			DP	48/48			(4') CLAY (CL); with trace sand, yellowish brown (2.5' trace gravel 2-2.5' bgs. (4') CLAY (CL); with trace sand, yellowish brown on the same sand, yellowish brown (10YR 4/3) (10YR 5/6) mottling.	wn (10'	YR 5/6), stiff, ellowish browr	1	Chem (6-11')	- - - - - - - - - - - - - - - - - - -
			DP	48/48			(12') CLAY (CL); dark gray (10YR 4/1), mediun (16') As above: dark gray (2.5Y 4/1).	m stiff,	moist.			- 15 - 15 - 20
	.o: I	vo groun	uwat		Joun	lei ea.						

Geosyntec ^D clia							Clien	t: Dynegy WELL LO ct: GL P8005 Coffeen Power Station Well No. G1001				L LOG 001	
	engineer	CC	nsulta	vators			Addr	ess: Coffeen, IL 62017		Page:	2 0	f 2	
Drillir Drillir Drillir Drillir Drillir Drille Logg	ig Start ig End I ig Com ig Meth ig Equi ir: ed By:	Date Date pany od:	e: 04/0 : 04/0 : Rob Dire nt: Geo A. T	5/202 5/202 erts l ct Pu prob	21 21 Drillir Jsh	ıg		Boring Depth (ft): 32 Boring Diameter (in): 6 DTW During Drilling (ft): DTW After Drilling (ft): Top of Casing Elev. (ft) Ground Elev. (ft): Northing, Easting (NAD83):	Well I Well I Scree Riser Scree Seal N Filter	Well Depth (ft):12Well Diameter (in):1Screen Slot (in):0.010Riser Material:Sch 40 PVCScreen Material:Sch 40 PVC SlottedSeal Material(s):Bentonite ChipsFilter Pack:Sand			ed
F	7	VEL	NO	Z COLLECT MEASU					MEASURE				
DEPTH (fi	ΓΙΣΗΟΓΟ(WATER LE'	WELL COMPLET	Sample Typ	Recovery (in	Blow Counts	N Value RQD (%)	SOIL/ROCK VISUAL DESC	JROCK VISUAL DESCRIPTION				
20-				DP	48/48		 	(20') As above: gravel at 1 ft.					- 20
25 - 30 -				DP	48/48			(24') As above: dry pocket at 2.8 ft. (28') As above: trace gravel and silt at 2 ft.					- 25 - 25 - 30
- 35								(32') End of Boring.					- 35
м	IOTES:	: N	lo groun	dwat	er en:	coun	tered.						

Geosyntec⊳

SOIL BORING LOG

Page of]

consultants

	E								BORING NUMBER: 5 61002-2					
	PROJE	ст: (729	800	5	PROJECT NO .: Coffeen MWA	E N							
	LOCAT	ION;	688	een,	, TL	PHASE NO.: 2	G. S. EL	EVATIO	*·X					
	CLIENT	:	<u>V:5-</u>	ten		TASK NO.:	BORING	DRING DEPTH: 21.0						
	DRILLI	IG CO	: Ro	bert	rs (Eny DRILLER:	DRILLIN	IG METH	od: D	P				
	LOGGE	D BY:	R	To	<u>ye</u>	STARTED: 575121	COMPL	έτεο: S	152	121				
	Depth (ft)	Elevation (ft)	Sample No.	Recovery / Attempted	Strata Change	Soil/Rock Description		Plew Count N	Un it Weight (pcf)-	Q _P (tsf)	Qu (tsf)	wc (%)		
234 6- 8- 1114 18- 222			Cherts Santa Lenter Jante Janta Cherty Janta	三日の こう いち いち いち いち いう こう こう ひん ひん しょう しん ひん しょう しょう しょう しょう しょう しょう しょう しょう しょう しょう		1.3' of <u>STLTY CLAY</u> , med groy, both, moith 24'' of <u>CLAYEY SAND</u> , orange brown to 1 3rey, sith, most 12" of <u>CLAY</u> w Some <u>SAND</u> , orange brown to 13" of <u>CLAY</u> w Some <u>SAND</u> , orange brown to 13" of <u>CLAY</u> w Some <u>SAND</u> , orange brown to 13" of <u>SAND</u> works fire grained, slightly 3.7' of <u>SAND</u> w Some <u>STLT</u> , greyish brow 0.7' of <u>SAND</u> , orange, slightly rwith 1.5' of <u>STLTY SAND</u> , orange w grey mothed draw NO RECOVERY 1.8' <u>STLT w LAND</u> w <u>CLAY</u> , grey of <u>CLAY</u> , grey, med Plattic, rwith 2' <u>CLAY</u> , grey, med Plattic, rwith EOB 21'	ther minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist minist mini	AT SOLAN		5 IZ	Qu (t	wc (\$		
L					-									

Geosyntec[>] consultants **MONITORING WELL** 1420 Kensington Rd., Suite 103 **CONSTRUCTION DETAIL** Oak Brook, Illinois 60523 (630) 203-3340 Coffeen, IL G1001 Well ID Site Location **Coffeen Power Station** A Toye, C Luttrell **Field Personnel Project Name** GLP8005 A Toye Project Number **Recorded By** Permit Number 4-05-2021 ☑ above ground protective casing Installation Date(s) flush mount protective casing □ other Direct Push **Drilling Method** 48 6.00" **Borehole Diameter** inches **Roberts Environmental Drilling Contractor** Driller ground surface elevation_ □ surveyed estimated N/A **Drilling Fluid** □ surface seal backfill N/A Gallons □ surface seal grout Fluid Loss During Drilling feet* Materials Used **Riser Pipe:** Diameter _ inches drilled hole inches diameter Construction 40 ☑ PVC schedule _ well casing_ _ inches diameter □ Stainless Steel □ Other _ 5.00 Length_ 🗆 backfill Slotted Area: feet O grout _ Diameter _____4.00_ inches density of grout _ .010 Slot Size _ _ inches Construction schedule 40 🛛 PVC □ Stainless Steel feet* \Box Other _ Silt Trap Used □ Yes □ No □ bentonite slurry ☑ bentonite pellets Bottom End Cap: □ Male □ Female □ Slip □ PVC feet* 4 □ Stainless Steel feet* 6 □ Other ____ feet* Top Cap: □ Male □ Female □ Slip □ J Plug □ PVC □ Stainless Steel well screen $\frac{4}{.010}$ inches diameter slot □ Other ___ O Protective . • 5 Length _ Casing: feet . . 6 Diameter inches • . • □ Cast Aluminum Construction • Cast Steel □ gravel pack □ Other _ ☑ sand pack Casing . 11 Installation: ___feet □ formation collapse Length _ 4 _____ inches Diameter ____ PVC Material -Sandpack: Filter Sil Coarse Sand: <u>bags</u> of ____ lb per bag Size - feet* Fine Sand: <u>4</u> bags of <u>50</u> lb per bag Size Seal: Bentonite Pellets: 2 bags of 50 lb per bag Type Bentonite Plug feet* Bentonite Slurry: <u>bags of</u> ber bag Type Measuring Point is Top of Well Casing Grout: Unless Otherwise Noted Cement: ____ bags of ___ lb per bag Type

* Depth Below Ground Surface

Bentonite: <u>bags of</u> bper bag Type

Geosyntec⊳

SOIL BORING LOG

Page of]

consultants

									BORING NUMBER: #- G1002-2						
	PROJECT: (JLP8005 PROJECT NO .: (Sffeen MAA								E N						
	LOCAT	ION: (Coff	een,	T1	PHASE NO.:	G. S. EL	EVATIO	"×						
	CLIENT	÷ \	<u>): 5-</u>	ten		TASK NO.:	BORING	RING DEPTH: 21,0							
	DRILLIN	IG CO:	Ro	bert	rs (Eny DRILLER:	DRILLIN	G METH	od: D	P					
	LOGGE	O BY:	<u>R</u>	To	ye	STARTED: 5725121	COMPL	етер: 5	125	121					
2346-8-12-12-12-12-12-12-12-12-12-12-12-12-12-	ELLIENT DRILLIN LOGGE		2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	20 212 Recovery 10 By or Recovery 14 7	Strata Change	TASK NO.: Env DRILLER: Started: S [25]2) Soil/Rock Description 1.3' of STLTY CLAY, med grey, bett, moist 24'' of CLAYEY SAND, orange brown to light 3rey, sitt, must 12'' of CLAY & Some SAND, orange brown to light 3rey, sitt, must 12'' of CLAY & Some SAND, orange brown to light 3rey, sitt, must 12'' of CLAY & Some SAND, orange brown to light 0.6 & Somb, orange fine graind, slightly 3.7 of SAND & Some SILT, pregish brown 0.71' of SAND & orange, slightly rightly 1.5' of SILTY SAND, orange watcher MO RECOVERY	BORING DRILLIN COMPLI	B DEPTH: IG METH ETED: S V Stel V Stel V Stel Soly Soly	Unitweighter	0° (21) (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151) ° (151)	Qu (tsf)	wc (%)			
18-19-22-22-22-22-22-22-22-22-22-22-22-22-22				20 24	-	1.8' STUT WIAND & CLAY, 9 9, the green low PHSHisday 2' CLAY, grey, med Plashic, muit EOB 21'	J,								

Luminant ILLINOIS POWER GENERATING COMPANY COFFEEN POWER STATION DECEMBER 2020 TOPOGRAPHY COFFEEN, ILLINOIS



PROJECT LOCATION MAP N.T.S.

LIST OF DRAWINGS

- 1 COVER SHEET
- 2 EXISTING TOPOGRAPHY NORTH
- B EXISTING TOPOGRAPHY SOUTH
- AERIAL TOPOGRAPHY NORTH
- 5 AERIAL TOPOGRAPHY SOUTH



ADDRESS 134 CIPS LANE COFFEEN, ILLINOIS 62017





VICINITY MAP

PREPARED BY:



502 Earth City Plaza, Suite 120 Earth City, MO 63045 www.ingenae.com



LE	GEND ————————————————————————————————————	(ISTING CONTC	OUR (2' INTERVA	J 5' 350 ' 		502 Earth City Plaza, Suite 120 Earth City, MO 63045 www.ingenae.com
_	— — — — LI <i>I</i>	MITS OF BATHY	METRIC SURVEY	/		
	PII	PING				
		ANHOLE I FT				
NOT 1. 2.	es: This exhib Intended Existing (Dragoni Topogra 12/03/2020	IT REPRESENTS TO BE A BOUN CONTOURS SHO LY AEROSOLU PHIC/BATHYM 0 & 12/04/2020	A TOPOGRAPH IDARY SURVEY. OWN ARE FROM TIONS DATED 1 .ETRIC SURVEYS). ASH POND 1	IIC FIELD SURVE A AERIAL SURV 2/03/2020 ANE COMPLETED E BATHYMETRIC	EY AND IS NOT EY COMPLETED BY D BY INGENAE DATED SURVEY WAS	
	COMPLET	ed by Ingenai	E ON 04/15/202	20.		Submissions / Revisions: Date:
3.	NO UNDEF SURVEY.	RGROUND OR	OVERHEAD UTI	LITIES WERE LO	CATED DURING THIS	1 2
4.	THE ACCU TRUTHING TOPOGRA TOPOGRA PLANE CC DATUM.	IMULATED DAT FIELD DATA CO PHIC/BATHYM PHIC DRAWIN ORDINATE-ZO	3 4 5 6 7			
		TAF	RGET POI	NT TABLE		
	POINT #	NORTHING	EASTING	ELEVATION	DESCRIPTION	10
	502	876457.9	2513942.9	630.43	TARGET NW	11
	503	876570.4	2515939.4	630.34	TARGET NE	12
	504	871464.3	2515211.1	637.22	TARGET SW	13

TARGET SE

TARGET MW

TARGET ME

505

507

508

871534.1 2516666.1 635.99

873944.5 2514681.3 623.70

873867.4 2516612.2 631.43



Project Name & Location:



134 CIPS Lane Coffeen, IL 62017

DO NOT SCALE PLANS Copying, Printing, Software and other processes required to produce these prints can stretch or shrink the actual paper or layout. Therefore, scaling of this drawing may be inaccurate. Contact IngenAE with any need for additional dimensions or clarifications.

Drawing Name: existing TOPOGRAPHY

Project No.

Drawing No.

Approved By:

Scale:

AS NOTED

SITE

СВ

ΒH

Type:

Drawn By:

NORTH Date: 5/28/2021





LEGEND -EXISTING CONTOUR (2' INTERVAL) -EXISTING CONTOUR (10' INTERVAL) -LIMITS OF BATHYMETRIC SURVEY 175' PING CONCRETE MANHOLE 502 Earth City Plaza, Suite 120 MH Earth City, MO 63045 INLET www.ingenae.com WELL SOIL BEARING MNA WELL NOTES: 1. EXISTING AERIAL SHOWN ARE FROM AERIAL SURVEY COMPLETED BY DRAGONFLY AEROSOLUTIONS DATED 12/03/2020. 2. THE WELL LOCATIONS AND ELEVATIONS, BORING LOCATIONS AND BENCHMARK DATA WAS RETRIEVED IN A SUBSEQUENT SURVEY DONE BY INGENAE DATED MARCH 24, 2021. TARGET POINT TABLE POINT # NORTHING EASTING ELEVATION DESCRIPTION 502 876457.9 2513942.9 630.43 TARGET NW 503 876570.4 2515939.4 630.34 TARGET NE 504 871464.3 2515211.1 637.22 TARGET SW 505 871534.1 2516666.1 635.99 TARGET SE 507 873944.5 2514681.3 623.70 TARGET MW 508 873867.4 2516612.2 631.43 TARGET ME SOIL BORING TABLE Submissions / Revisions: Date: POINT # NORTHING EASTING ELEVATION DESCRIPTION 1075 876525.9 2515953.9 630.82 SB-289 GROUND WELL LOCATION POINT TABLE POINT # NORTHING EASTING ELEVATION DESCRIPTION 1071 875111.4 2514684.6 631.41 G-206D GROUND 1072 875111.4 2514684.6 634.35 G-206D TOP METAL CASING 1074 875111.4 2514684.6 634.14 G-206D TOP PVC WELL 1082 874285.3 2516366.5 617.52 G-275D GROUND 1083 874285.3 2516366.5 620.69 G-275D TOP METAL CASING 1084 874285.3 2516366.5 620.31 G-275D TOP PVC WELL 1105 874113.0 2516503.0 608.30 G-283 GROUND 1106 874113.0 2516503.0 611.07 G-283 TOP METAL CASING 1107 874113.0 2516503.0 610.75 G-283 TOP PVC WELL 1086 874423.6 2516922.9 615.33 G-284 GROUND 1087 874423.6 2516922.9 618.66 G-284 TOP METAL CASING 1088 874423.6 2516922.9 618.42 G-284 TOP PVC WELL 1094 874795.0 2516680.4 610.54 G-285 GROUND 1095 874795.0 2516680.4 613.90 G-285 TOP METAL CASING 1097 874795.0 2516680.4 613.52 G-285 TOP PVC WELL 1098 875072.2 2516561.8 609.97 G-286 GROUND 1099 875072.2 2516561.8 613.57 G-286 TOP METAL CASING 1100 875072.2 2516561.8 613.13 G-286 TOP PVC WELL 1078 875442.8 2516415.5 614.34 G-287 GROUND 1079 875442.8 2516415.5 617.88 G-287 TOP METAL CASING 1080 875442.8 2516415.5 617.45 G-287 TOP PVC WELL 1090 875279.6 2517071.4 617.08 G-288 GROUND 1091 875279.6 2517071.4 620.37 G-288 TOP METAL CASING 1092 875279.6 2517071.4 620.07 G-288 TOP PVC WELL 1009 872235.1 2515583.1 620.88 G-301 GROUND 1010 872235.1 2515583.1 623.18 G-301 TOP METAL CASING 1011 872235.0 2515583.0 622.99 G-301 TOP PVC COUPLING WELL 1003 872252.9 2516214.1 618.52 G-302 GROUND 1017 872252.9 2516214.1 620.53 G-302 TOP METAL CASING 1018 872252.9 2516214.1 620.34 G-302 TOP PVC COUPLING WELL Luminant 1047 871382.1 2516639.5 619.33 G-303 GROUND 1048 871382.1 2516639.5 622.38 G-303 TOP METAL CASING 1049 871382.1 2516639.5 622.18 G-303 TOP PVC WELL COUPLING 1068 871156.4 2515199.5 623.23 G-305 GROUND 1069 871156.4 2515199.5 626.06 G-305 TOP METAL CASING Project Name & Location: 1070 871156.4 2515199.5 625.67 G-305 TOP PVC WELL 1056 871140.9 2516120.4 623.57 G-306 GROUND 1057 871140.9 2516120.4 626.38 G-306 TOP METAL CASING 1058 871140.9 2516120.4 625.91 G-306 TOP PVC WELL COFFEEN 1067 871398.6 2515553.4 624.73 G-307 GROUND 1066 871398.6 2515553.4 624.91 G-307 TOP METAL CASING POWER STATION 1065 871398.6 2515553.4 624.60 G-307 TOP PVC WELL G-307D GROUND 1062 871397.2 2515560.3 622.51 1063 871397.2 2515560.3 625.29 G-307D TOP METAL CASING 1064 871397.2 2515560.3 624.88 G-307D TOP PVC WELL 134 CIPS Lane 1027 871454.7 2515101.4 621.59 G-308 GROUND 1028 871454.7 2515101.4 624.96 G-308 TOP METAL CASING Coffeen, IL 62017 1029 871454.7 2515101.4 624.59 G-308 TOP PVC WELL 1030 871865.8 2515067.1 622.77 G-309 GROUND 1031 871865.8 2515067.1 626.20 G-309 TOP METAL CASING 1032 871865.8 2515067.1 625.88 G-309 TOP PVC WELL 1006 872239.4 2515159.4 619.89 G-310 GROUND _____1007 872239.4 2515159.4 623.32 G-310 TOP METAL CASING 1008 872239.4 2515159.4 622.87 G-310 TOP PVC WELL 1012 872238.7 2515881.8 618.32 G-311 GROUND 1015 872238.7 2515881.8 618.39 G-311 GROUND Copyright © 2021 1013 872238.7 2515881.8 621.55 G-311 TOP METAL CASING IngenAE, LLC www.ingenae.com 1014 872238.7 2515881.8 621.04 G-311 TOP PVC WELL 1016 872238.8 2515896.2 621.75 G-311D TOP METAL CASING 1002 872238.8 2515896.2 621.24 G-311D TOP PVC WELL do not scale plans Copying, Printing, Software and other processes 1019 872260.9 2516557.4 616.92 G-312 GROUND required to produce these prints can stretch or shrink 1020 872260.9 2516557.4 620.11 G-312 TOP METAL CASING the actual paper or layout. Therefore, scaling of this 1021 872260.9 2516557.4 619.78 G-312 TOP PVC WELL drawing may be inaccurate. Contact IngenAE with 1038 871976.8 2516803.7 611.51 G-313 GROUND any need for additional dimensions or clarifications. 1039 871976.8 2516803.7 614.62 G-313 TOP METAL CASING 1040 871976.8 2516803.7 614.30 G-313 TOP PVC WELL Drawing Name: 2015 871630.2 2516852.1 611.11 G-314 GROUND 2014 871630.2 2516852.1 614.28 G-314 TOP METAL CASING AERIAL 2013 871630.2 2516852.1 613.88 G-314 TOP PVC WELL 2010 871642.0 2516853.9 610.87 G-314D GROUND PHOTOGRAPHY

2011 871642.0 2516853.9 614.10 G-314D TOP METAL CASING 2012 871642.0 2516853.9 613.70 G-314D TOP PVC WELL

1051 871643.1 2517211.6 603.06 G-316 TOP METAL CASING

1052 871643.1 2517211.6 602.59 G-316 TOP PVC WELL

1054 871234.2 2517087.4 642.26 G-317 TOP METAL CASING

1055 871234.2 2517087.4 641.93 G-317 TOP PVC WELL

1023 872312.0 2516714.4 605.87 SG-02 TOP METAL WELL

1024 871638.7 2515366.3 631.85 XPW-01 GROUND

1025 871638.7 2515366.3 634.92 XPW-01 TOP METAL CASING

1026 871638.7 2515366.3 634.57 XPW-01 TOP PVC WELL

1001 871987.1 2515627.3 636.64 XPW-02 GROUND 1101 871987.1 2515627.3 640.02 XPW-02 TOP METAL

 1102
 871987.1
 2515627.3
 639.69
 XPW-02 TOP PVC WELL

 1004
 872095.3
 2515211.8
 635.52
 XSG-01 TOP PVC WELL

1022 872095.1 2517025.0 594.94 SG-03 TOP METAL WELL 1085 873935.3 2516963.3 599.52 SG-04 TOP METAL WELL

1053 871234.2 2517087.4 638.85 G-317 GROUND

G-315 TOP PVC WELL

G-316 GROUND

1059 871385.0 2516086.6 620.94 G-315 GROUND 1060 871385.0 2516086.6 623.89 G-315 TOP METAL CASING

1061 871385.0 2516086.6 623.52

1050 871643.1 2517211.6 599.64

NORTH Project No. 5/28/2021

Drawing No.

 $\mathbf{\mathcal{I}}$

SITE

Approved By:

Scale:

Date:

Type:

Drawn By: СВ

ΒH

AS NOTED



LEGEND ————————————————————————————————————	502 Earth City Plaza, Suite 120 Earth City, MO 63045 www.ingenae.com		
 PIPING CONCRETE MANHOLE INLET PROPOSED MNA IOTES: THIS EXHIBIT REPRESENTS INTENDED TO BE A BOUNT 2. EXISTING CONTOURS SH DRAGONFLY AEROSOLUTOPOGRAPHIC/BATHYA 12/03/2020 & 12/04/202 COMPLETED BY INGENA 3. NO UNDERGROUND OR THIS SURVEY. 4. THE ACCUMULATED DATA CONTOPOGRAPHIC/BATHYA TOPOGRAPHIC/BATHYA TOPOGRAPHIC/BATHYA TOPOGRAPHIC DRAWIN STATE PLANE COORDIN. ELEVATION DATUM. 5. ADDITIONAL MNA SAMI PERFORMED ON 04/22/3 503 876570.4 503 876570.4 503 876570.4 503 871534.1 507 873944.5 508 873867.4	A SAMPLE SITE A TOPOGRAPHIC FIELD SURV NDARY SURVEY. OWN ARE FROM AERIAL SURV VIIONS DATED 12/03/2020 AN METRIC SURVEYS COMPLETED D. ASH POND 1 BATHYMETRIC E ON 04/15/2020. OVERHEAD UTILITIES WERE LO TA FROM AERIAL DRONE SUR OLLECTION SURVEYS AND METRIC SURVEYS USED TO PRO IG AS SHOWN HEREON IS BAS ATE-ZONE WEST NAD 1983 AN PLING SITES GROUND BATHYM 2021. RGET POINT TABLE EASTING ELEVATION 2513942.9 630.43 2515939.4 630.34 2515211.1 637.22 2516666.1 635.99 2514681.3 623.70 2516612.2 631.43	VEY AND IS NOT VEY COMPLETED BY D BY INGENAE DATED SURVEY WAS OCATED DURING VEYS, GROUND DUCE THE SE ON ILLINOIS ID NAVD 88 AETRIC SURVEY DESCRIPTION TARGET NW TARGET NW TARGET SW TARGET SW TARGET SW TARGET MW TARGET MW	Submissions / Revisions: Date: 1
			Copyright © 2021 WWW.ingenac.com DO NOT SCALE PLANS Copying, Printing, Software and other processes required to produce these prints can stretch or shrink the actual paper or layout. Therefore, scaling of this drawing may be inaccurate. Contact IngenAE with any need for additional dimensions or clarifications. Drawing Name: Drawing Name: Drawing Name: Date: Project No. 5/28/2021 Type: Drawing No. SITE Drawn By: CB Approved By: BH Scale: AS NOTED



LEGEND ————————————————————————————————————	502 Earth City Plaza, Suite 120 Earth City, MO 63045 www.ingenae.com			
 LIMITS OF BATHY PIPING CONCRETE MANHOLE INLET WELL MNA WELL 	METRIC SURVEY N ARE FROM AEF JTIONS DATED 12 ON OF OUTFALL F ND ELEVATIONS S RETRIEVED IN A H 24, 2021.	RIAL SURVEY C 2/03/2020. PIPE LOCATED , BORING LOC SUBSEQUENT	COMPLETED BY BY OTHERS. CATIONS AND SURVEY DONE BY	Submissions / Revisions: Date: 1 2 3 - 4 - 5 - 6 - 7
ТА	RGET POIN	NT TABLE		7 8 0
POINT # NORTHING	EASTING E	ELEVATION	DESCRIPTION	10
502 876457.9 503 876570.4	2513942.9	630.43		11 12
504 871464.3	2515939.4	637.22	TARGET SW	13
505 871534.1	2516666.1	635.99	TARGET SE	
507 873944.5	2514681.3	623.70	TARGET MW	
SU8 873807.4 POINT NO. NORTHIN 1113 873636.9 1112 873636.9 1111 873636.9 1111 873636.9 1111 873636.9 1112 872676.8 1119 872676.8 1120 872676.8 1116 872510.8 1101 874781.0 AS SURVEY BY INGENA	MINA WEII MINA WEII G EASTING ELEV/I 1 2516757.78 597 1 2516757.78 597 2 2516757.78 597 3 2516735.78 607 3 2516735.78 609 4 2515601.43 625 5 2515020.93 623 6 4/20/21 1	051.45 LLS ATION DEs 4.82 G-100 7.95 G-1001 TC 7.61 G-1001 7.02 G-100 0.47 G-1002 TC 9.97 G-1002 5.77 SB-40 3.53 SB-27	SCRIPTION 01 GROUND 0P METAL CASING TOP PVC WELL 02 GROUND 0P METAL CASING TOP PVC WELL 01A GROUND 70A GROUND	Luminant Project Name & Location: COFFEEN POWER STATION 134 CIPS Lane Coffeen, IL 62017
				Copyright © 2021
				DO NOT SCALE PLANS Copying, Printing, Software and other processes required to produce these prints can stretch or shrink the actual paper or layout. Therefore, scaling of this drawing may be inaccurate. Contact IngenAE with any need for additional dimensions or clarifications. Drawing Name: AERIAL PHOTOGRAPHY SOUTH Date: SITE Drawn By: CB Approved By: BH Scale: AS NOTED

APPENDIX B STATISTICAL ANALYSIS PLAN

Prepared for Illinois Power Generating Company

Date **October 25, 2021**

Project No. 1940100806-002

STATISTICAL ANALYSIS PLAN ASH POND NO. 2 COFFEEN POWER PLANT COFFEEN, ILLINOIS



STATISTICAL ANALYSIS PLAN COFFEEN POWER PLANT ASH POND NO. 2

Project Name	Coffeen Power Plant Ash Pond No. 2
Project No.	1940100806-005
Recipient	Illinois Power Generating Company
Document type	Statistical Analysis Plan
Version	FINAL
Date	October 25, 2021

Ramboll 234 W. Florida Street Fifth Floor Milwaukee, WI 53204 USA

T 414-837-3607 F 414-837-3608 https://ramboll.com

Brian G. Hennings, PG Senior Managing Hydrogeologist

S

Eric J. Tlachac, PE Senior Managing Engineer

Rachel A. Banoff, EIT Project Statistician

LICENSED PROFESSIONAL CERTIFICATIONS

This certification is based on the description of the statistical methods selected to evaluate groundwater as presented in the following Statistical Analysis Plan; Coffeen Power Plant Ash Pond No. 2. The procedures described in the plan will be used to establish background conditions and implement compliance monitoring as necessary and required by 35 I.A.C. § 845.640 and 35 I.A.C. § 845.650. The Statistical Analysis Plan was prepared in accordance with the requirements of 35 I.A.C. § 845.640(f), with reference to the acceptable statistical procedures provided in the United States Environmental Protection Agency (USEPA)'s Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities, Unified Guidance (Unified Guidance, March 2009), and is intended to provide a logical process and framework for conducting the statistical analysis of the data obtained during groundwater monitoring. In accordance with 35 I.A.C. § 845.640(f)(1), the statistical method chosen for analysis of background groundwater quality will be either the tolerance interval or the prediction interval procedure for each constituent listed in 35 I.A.C. § 845.600(a)(1) at this CCR unit per 35 I.A.C. § 845.640(f)(1)(C). Groundwater Protection Standards (GWPS) will be established in accordance with 35 I.A.C. § 845.600(a) (areater of the background concentration or numerical limit specified in 35 I.A.C. § 845.600(a)(1)). The GWPS will be compared to the lower confidence limit for the observed concentrations for each constituent in each compliance well. Consistent with the Unified Guidance, the same general statistical method of confidence interval testing against a fixed GWPS is recommended in compliance and corrective action programs. Confidence intervals provide a flexible and statistically accurate method to test how a parameter estimated from a single sample compares to a fixed numerical limit. Confidence intervals explicitly account for variation and uncertainty in the sample data used to construct them.

Description of the statistical methods chosen for analysis of groundwater monitoring data and application of these methods for determining exceedances of the GWPS identified in 35 I.A.C. § 845.600(a) is provided in this Statistical Analysis Plan.

35 I.A.C. § 845.640 Statistical Analysis (PE)

I, Eric J. Tlachac, a qualified professional engineer in good standing in the State of Illinois, certify that the statistical methods summarized above and described in this document (Statistical Analysis Plan; Coffeen Power Plant Ash Pond No. 2) are appropriate for evaluating the groundwater monitoring data collected as described in the attached document and are in substantial compliance with 35 I.A.C. § 845.640.

On the

Eric J. Tlachac Qualified Professional Engineer 062-063091 Illinois Date: October 25, 2021



35 I.A.C. § 845.640 Statistical Analysis (PG)

I, Brian G. Hennings, a qualified professional geologist in good standing in the State of Illinois, certify that the statistical methods described in this document (Statistical Analysis Plan; Coffeen Power Plant Ash Pond No. 2) are appropriate for evaluating the groundwater monitoring data collected as described in the attached document and are in substantial compliance with 35 I.A.C. § 845.640.

Brian G. Hennings Professional Geologist 196.001482 Illinois Date: October 25, 2021



35 I.A.C. § 845.640 Statistical Analysis

I, Rachel A. Banoff, a qualified professional, certify that the statistical methods described in this document (Statistical Analysis Plan; Coffeen Power Plant Ash Pond No. 2), are appropriate for evaluating the groundwater monitoring data collected as described in the attached document and are in substantial compliance with 35 I.A.C. § 845.640.

Rachel A. Banoff, EIT Project Statistician Date: October 25, 2021

CONTENTS

Licensed Professional Certifications							
1.	Introduction	6					
1.1	Statistical Analysis Objectives	6					
1.2	Statistical Analysis Plan Approach	6					
2.	Background Monitoring and Data Preparation	8					
2.1	Sample Independence	8					
2.2	Non-Detect Data Processing	9					
2.3	Testing for Normality	9					
2.4	Testing for Outliers	9					
2.5	Trend Analysis	10					
2.6	Spatial Variation	10					
2.7	Temporal Variation	10					
2.8	Updating Background	11					
3.	Compliance Monitoring	13					
3.1	GWPS Establishment and Exceedance Determination	13					
3.1.1	The Upper Tolerance Limit	14					
3.1.2	Parametric Confidence Intervals around a Mean	16					
3.1.3	Non-Parametric Confidence Intervals around a Median	16					
3.1.4	The Upper Prediction Limit for a Future Mean	17					
3.1.5	The Non-Parametric Upper Prediction Limit for a Future Median	17					
3.1.6	Parametric Linear Regression and Confidence Band	18					
3.1.7	Non-Parametric Thiel-Sen Trend Line and Confidence Band	20					
3.2	Determination of Statistically Significant Increases over						
	Background	21					
4.	References	22					

TABLES (IN TEXT)

 Table A
 Statistical Calculations Used in Compliance Monitoring Procedures

ACRONYMS AND ABBREVIATIONS

§	Section
35 I.A.C.	Title 35 of the Illinois Administrative Code
ANOVA	analysis of variance
CCR	coal combustion residuals
COC	constituents of concern
GWPS	groundwater protection standard
IEPA	Illinois Environmental Protection Agency
LCL	lower confidence limit
LTL	lower tolerance limit
MSE	mean squared error
Ρ	probability
Part 845	Residuals in Surface Impoundments: Title 35 of the Illinois Administrative Code
	§ 845
RCRA	Resource Conservation and Recovery Act
RL	reporting limit
ROS	regression on order statistics
SI	surface impoundment
SSI	statistically significant increase
SWFPR	site-wide false positive rate
Unified Guidance	Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities,
	Unified Guidance (USEPA, 2009)
UPL	upper prediction limit
USEPA	United States Environmental Protection Agency
UTL	upper tolerance limit

1. INTRODUCTION

In April 2021, the Illinois Environmental Protection Agency (IEPA) issued a final rule for the regulation and management of Coal Combustion Residuals (CCR) in surface impoundments (SIs) under the Standards for the Disposal of CCR in Surface Impoundments: Title 35 of the Illinois Administrative Code (35 I.A.C.) § 845 (Part 845). Facilities regulated under Part 845 are required to develop and sample a groundwater monitoring well network to evaluate whether impounded CCR materials are impacting downgradient groundwater quality. The groundwater quality evaluation must include selection and certification by a qualified professional engineer of the statistical procedures to be used. The procedures described in the evaluation will be used to establish background conditions and implement compliance and corrective action monitoring as necessary and required by 35 I.A.C. § 845.640 and 35 I.A.C. § 845.650. This Statistical Analysis Plan was prepared in accordance with the requirements of 35 I.A.C. § 845.640(f), with reference to the acceptable statistical procedures provided in United States Environmental Protection Agency's (USEPA's) *Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities, Unified Guidance* (March 2009).

This Statistical Analysis Plan does not include procedures for groundwater sample collection and analysis, as these activities are conducted in accordance with the Sampling and Analysis Plan prepared for each CCR unit in accordance with 35 I.A.C. § 845.640. This Statistical Analysis Plan will be used as the primary reference for evaluating groundwater quality during operation and post-closure care.

1.1 Statistical Analysis Objectives

This Statistical Analysis Plan is intended to provide a logical process and framework for conducting the statistical analyses of data obtained during groundwater monitoring conducted in accordance with the Sampling and Analysis Plan for each CCR unit. The Statistical Analysis Plan will enable a qualified professional engineer to certify that the selected statistical methods are appropriate for evaluating the groundwater monitoring data for the applicable CCR unit(s).

1.2 Statistical Analysis Plan Approach

The main sections of this Statistical Analysis Plan should be viewed as a "generic" outline of statistical methods utilized for each CCR unit and constituent required to be monitored. The statistical analysis of the groundwater monitoring data, however, will be conducted on an individual-constituent or well basis, and may involve the use of appropriate statistical procedures depending on multiple factors such as detection frequency and normality distributions.

The CCR Rule outlines two phases of groundwater monitoring:

- Background Monitoring in accordance with 35 I.A.C. § 845.650(b)(1)
- Compliance Monitoring in accordance with 35 I.A.C. § 845.650

Each phase of the groundwater monitoring program requires specific statistical procedures to accomplish the intended purpose. During the background monitoring phase, background groundwater quality will be established utilizing upgradient and background wells and downgradient groundwater quality data will be collected to facilitate statistics in subsequent phases. Compliance Monitoring is then initiated through the evaluation of the downgradient

groundwater monitoring data for exceedances of the groundwater protection standard (GWPS) established by Part 845 (concentration specified in 35 I.A.C. § 845.600 or an IEPA-approved background concentration). The developed statistical analysis plan will be implemented for each monitoring phase and in accordance with the statistical procedures.

2. BACKGROUND MONITORING AND DATA PREPARATION

The background and compliance monitoring wells were sampled and analyzed for constituents, as listed in Part 845 (antimony, arsenic, barium, beryllium, boron, cadmium, calcium, chloride, chromium, cobalt, fluoride, lead, lithium, mercury, molybdenum, pH, radium 226 and 228 combined, selenium, sulfate, thallium, total dissolved solids, and turbidity), during the baseline phase of the groundwater monitoring program.

The background monitoring well(s) were placed upgradient of the CCR unit, or at an alternative background location, where they are not affected by potential leakage from the CCR unit. Compliance monitoring wells were placed at the waste boundary of the CCR unit, along the same groundwater flow path. As 35 I.A.C. § 845.630(a) specifies, the location of these wells ensures that background accurately represents the quality of unaffected groundwater, while compliance wells accurately represent groundwater quality at the waste boundary and monitor all potential contaminant pathways.

As required by 35 I.A.C. § 845.650(a)(1), eight sampling events were completed within 180 days of April 21, 2021. As outlined, groundwater sampling procedures included sampling of the background and compliance wells using low-flow sampling methods, collection of one field quality control sample per event, and groundwater samples were not field filtered before laboratory analysis of total recoverable metals.

Following completion of the eight sampling events, background groundwater quality was established for Part 845 constituents. Groundwater monitoring will be conducted quarterly for at least the first five years. In accordance with 35 I.A.C. § 845.650(b)(4), after the first five years, a request to reduce the monitoring frequency to semiannual may be submitted to IEPA if all of the following can be demonstrated:

- Groundwater monitoring effectiveness will not be compromised by the reduced frequency
- Sufficient data has been collected to characterize groundwater
- Monitoring to date does not show any statistically significant increasing trends
- The concentrations of monitored constituents at the compliance monitoring wells are below the applicable GWPSs established in 35 I.A.C. § 845.600

The following subsections outline the statistical tests and procedures (methods) that will be utilized to evaluate data collected for each constituent in both background and compliance wells for Background and Compliance Monitoring. When necessary and contingent upon equivalent statistical power, an alternative test not included in this Statistical Analysis Plan may be chosen due to site-specific data requirements.

2.1 Sample Independence

Independence of sample results is a major assumption for most statistical analyses. To ensure physical independence of groundwater sampling results, the minimum time between sampling events must be longer than the time required for groundwater to move through the monitoring well. The sampling schedules for both the baseline and compliance monitoring periods are specified in 35 I.A.C. § 845.650(b) and may conflict with the statistical assumption of independence of sample results.

2.2 Non-Detect Data Processing

The reporting limit (RL) will be used as the lower level for the reporting of non-detected groundwater quality data. For all summary statistics (box plots, timeseries, etc.), the RL will be substituted for concentrations reported below the RL, including non-detects. With professional judgement, analytical results between the RL and the method detection limit, *i.e.*, estimated values, typically identified with a "J" flag, may be utilized if provided by the laboratory.

For all statistical test procedures:

- If the frequency of non-detect data are less than or equal to 15 percent, half of the RL will be substituted for these data
- If the non-detect frequency is between 15 percent and 50 percent, either the Kaplan-Meier or robust regression on order statistics (ROS) will be used to estimate the mean and standard deviation adjusted for the presence of left-censored values
- If the non-detect frequency is greater than 50 percent, a non-parametric test will be used
- If only one background result is detected that value will be used as the non-parametric upper prediction limit (UPL)

2.3 Testing for Normality

Many statistical analyses assume that sample data are normally distributed (parametric). However, environmental data are frequently not normally distributed (nonparametric). 35 I.A.C. § 845.640(g) requires the knowledge of the background data distribution for comparison to compliance results. The *Unified Guidance* document recommends the Shapiro-Wilk normality test for sample sizes of 50 or less, and the Shapiro-Francia normality test for sample sizes greater than 50.

When possible, transformation of datasets to achieve normal distributions is preferred.

2.4 Testing for Outliers

Part 845 constituents will be screened for the existence of outliers using a method described by the *Unified Guidance*. Outliers are extreme data points that may represent an anomaly or erroneous data point. To test for outliers, one or more of the following outlier tests will be utilized:

- Dixon's test, for well-constituent pairs with less than 25 samples, assumes normally distributed data.
- Rosner's test, for well-constituent pairs with more than 20 samples, assumes normally distributed data.
- Grubb's test for well-constituent pairs with seven or more samples, assumes normally distributed data.
- Time series, box-whisker plots, and probability plots provide visual tools to identify potential outliers, and evaluation of seasonal, spatial, or temporal variability for both normally and non-normally distributed data.

Data quality control, groundwater geochemistry, and sampling procedures will be evaluated as potential sources of error leading to an outlier result. The outlier tests cannot be used alone to determine whether a value is a true outlier that should be excluded from future statistical

analysis. Corroborating evidence needed to exclude values includes a discrete data reporting or analytical error, or potential laboratory bias. Absent corroborating evidence, the flagged values are considered true, but extreme, values in the data set. Professional judgement will be used to exclude extreme outliers from further statistical analyses. Outliers will be retained in the database.

With professional judgement, a confirmatory sample may be collected to allow for the distinction between an outlier and a true representation of groundwater quality at the monitoring point. If re-sampling is conducted, this sample will be collected within 90 days following outlier identification. If the confirmatory sample indicates the original result as an outlier, it will be reported as such.

2.5 Trend Analysis

Statistical analyses supporting the lack of trend are a fundamental step to confirm the assumption that groundwater quality values are stationary or constant over time at a CCR unit. These analyses allow for evaluation of variation in the background and compliance data for each constituent over time. A statistically significant increasing trend in background data could indicate an existing release from the CCR unit or alternate source, requiring further investigation. In addition, statistically significant trending background data can result in increased standard deviation and, therefore, greater prediction or control limits. Consequently, the increased prediction or control limit will have less power or ability to identify a release from the CCR unit.

A linear regression, coupled with a t-test for slope significance at a 95 percent confidence level (0.05 significance level), may be used on datasets for each constituent with few non-detects and a normally distributed variance of the mean to evaluate time trends. The Theil-Sen trend line, coupled with the Mann-Kendall test for slope significance at a 95 percent confidence level (0.05 significance level), will be used for datasets with frequent non-detects or non-normal variance. Similarly, trend analyses could also be used on compliance data to evaluate a possible release from the CCR unit.

2.6 Spatial Variation

Spatial trends and/or variation between background wells could indicate an existing release from a CCR unit. If the spatial variability is not due to an existing release, intrawell comparisons in compliance wells may be used to account for spatial variability and monitor for a future release. However, the CCR unit being monitored was placed into service prior to the start of groundwater monitoring and it is unknown whether a previous release has occurred. Accordingly, intrawell comparisons in compliance wells cannot be used to determine the occurrence of a future release. Interwell comparisons between compliance wells and background wells will be used.

2.7 Temporal Variation

Time series plots can be used to identify temporal dependence. Potentially significant temporal components of variability can be identified by graphing single constituent data from multiple wells together on a time series plot. With temporal dependence, the time series plot as a pattern of parallel traces, in which the individual wells will tend to rise and fall together across the sequence of sampling dates. Time series plots can be helpful by plotting multiple constituents over time for the same well, or averaging values for each constituent across wells on each sampling event and then plotting the averages over time. In either case, the plots can signify whether the general concentration pattern over time is simultaneously observed for different

constituents. If so, it may indicate that a group of constituents is highly correlated in groundwater or that the same artifacts of sampling and/or lab analysis impacted the results of several monitoring parameters.

Hydrologic factors such as drought, recharge patterns or regular (*e.g.*, seasonal) water table fluctuations may be responsible for the temporal variation. In these cases, it may be useful to test for the presence of a significant temporal effect by first constructing a parallel time series plot and then running a formal one-way analysis of variance (ANOVA) ($\alpha = 0.05$) for temporal effects. A one-way ANOVA for temporal effects considers multiple well data sets for individual sampling events or seasons as the relevant statistical factor. If event-specific analytical differences or seasonality appear to be an important temporal factor, the one-way ANOVA for temporal effects. The one-way ANOVA for temporal effects. The one-way ANOVA for temporal effects assumes that affect other temporal effects. The one-way ANOVA for temporal effects assumes that the data groups are normally distributed with constant variance. It is also assumed that for each of a series of background wells, measurements are collected at each well on sampling events or dates common to all the wells. Results of the ANOVA can also be used to create temporally stationary residuals, where the temporal effect has been 'subtracted from' the original measurements. These stationary residuals may be used to replace the original data in subsequent statistical testing.

If the data cannot be normalized, a similar test for a temporal or seasonal effect can be performed using the Kruskal-Wallis test ($\alpha = 0.05$). Each sampling event should be treated as a separate `well,' while each well is treated as a separate `sampling event.' In this case, no residuals can be computed since the Kruskal-Wallis test employs ranks of the data rather than the measurements themselves.

Where both spatial and temporal variation occur, two-way ANOVA can be considered where both well location and sampling event/season are treated as statistical factors. This procedure is described in Davis (1994).

2.8 Updating Background

Updating the background dataset periodically by adding recent results to an existing background dataset can improve the statistical power and accuracy of the statistical analysis, especially for non-parametric prediction intervals. The Unified Guidance recommends updating statistical limits (background) when at least four to eight new measurements (every 1 to 2 years under a quarterly monitoring program), are available for comparison to historical data. Professional judgement will be used to evaluate whether any background data appear to be affected by a release and need to be excluded from a background update. A t-test for equal means (if normal data distribution) or appropriate non-parametric test (if non-normal data distribution) such as a Mann-Whitney (or Wilcoxon) rank-sum or box-whisker plots, will be conducted to evaluate whether the two groups of background sample populations are statistically different prior to updating any background datasets. A 0.05 significance level will be utilized when evaluating the two populations, with the null hypothesis that they are equivalent. In addition, time series graphs or other trend evaluation statistics will be conducted on the new background dataset to verify the absence of a release or changing groundwater quality. If the tests indicate that there are no statistical differences between the two background populations, the new data will be combined with the existing dataset. If the two populations are found to be different, the data will be reviewed to evaluate the cause of the difference. If the differences appear to be caused by a

release (if the new data are significantly higher, or lower for pH), then the previous background dataset may continue to be used. Furthermore, verified outliers will not be added to an existing background dataset. In accordance with the *Unified Guidance*, continual background updates will not be conducted due to the lack of sufficient samples for a statistical comparison.

3. COMPLIANCE MONITORING

Compliance monitoring is designed to monitor groundwater for evidence of a release by comparing Part 845 constituents in compliance wells to both background concentrations and the GWPS. Compliance Monitoring will begin the quarter following approval of this groundwater monitoring plan and issuance of the Operating Permit. The selected Compliance Monitoring statistical method used to compare compliance groundwater quality data for each constituent to the GWPS will provide for adequate statistical power, error levels and individual test false positive rates, and be appropriate for the distribution and detection frequency of the background dataset. Statistical power is the ability of a statistical test to detect a true exceedance.

In accordance with 35 I.A.C. § 845.610(b)(3)(D), compliance monitoring statistical analyses will be completed and submitted to IEPA within 60 days after completion of sampling.

3.1 GWPS Establishment and Exceedance Determination

In accordance with 35 I.A.C. § 845.600(a), the GWPS will be the constituent concentrations specified in 35 I.A.C. § 845.600(a)(1) except for when the background concentration is greater, or no concentration is specified (*i.e.*, for calcium and turbidity), in which case the GWPS will be the background concentration. The GWPS based on background concentration will be calculated using a parametric upper tolerance limit (UTL), a parametric UPL for a future mean, or a non-parametric UPL for a future median.

Statistical calculations that will be utilized in Compliance Monitoring procedures are summarized in **Table A** below and listed in **Sections 3.1.1** through **3.1.7**. Depending on the distribution of the data and the percentage of non-detects, it may be more appropriate to use a parametric model over a non-parametric model. As necessary, other techniques as mentioned in the *Unified Guidance* and/or new methods will be implemented.

			Compliance M	lonitoring				
		Background	Data	Compliance Data				
Significant Trend?	Percent Non- Detects	Distribution	GWPS Determination	Percent Non-Detects	Distribution	Method to Determine Exceedance		
				≤75	Normal	Parametric Lower Confidence Limit around a Normal Mean		
	0 ≤ 50	Normal	35 I.A.C § 845.600(a)(1) constituent concentration or The Upper	≤75	Log-Normal	Parametric Lower Confidence Limit around a Lognormal Geometric Mean		
			Tolerance Limit	NA	Non-Normal	Non-Parametric Lower		
No				>75	Unknown/ Cannot be determined	Confidence Limit around a Median		
	50 ≤ 70	Normal	The Upper Prediction Limit for a Future Mean	NA	NA	Future mean		
	>70	Non-Normal	Upper Prediction Limit for a Future Median	NA	NA	Future median		
	100	Non-Normal	Double Quantification Rule	NA	NA	Individual Retesting Values		
Yes	0 ≤ 50	Normal	UCL of Confidence Band around Linear Regression	≤75	Residuals after subtracting trend are normal, equal variance	Lower Limit from Confidence Band around Linear Regression		
	50 ≤ 100 Non-Normal		UCL of Confidence Band around Thiel-Sen trend line	≤75	Residuals not normal	Lower Limit from Confidence Band around Thiel-Sen		

Table A. Statistical Calculations Used in Compliance Monitoring Procedures

3.1.1 The Upper Tolerance Limit

The UTL will be used to calculate the GWPS when pooled background data are normally distributed, with a non-detect frequency of 50 percent or less. When non-detect frequency is 15 percent or less, half the RL will be substituted for non-detects. The *Unified Guidance* recommends 95 percent confidence level and 95 percent coverage (95/95 tolerance interval).

• When non-detect frequency is 15 percent or less, half the RL will be substituted for nondetects (simple substitution), and the normal mean and standard deviation will be calculated.

- The Kaplan-Meier or the ROS method will be used when the detection frequency is between 15 percent and 50 percent. The Kaplan-Meier method assesses the linearity of a censored probability plot to determine whether the background sample can be approximately normalized. If so, then the Kaplan-Meier method will be used to compute estimates of the mean and standard deviation adjusted for the presence of left-censored values. The Kaplan-Meier or ROS estimate of the mean and standard deviation.
- If background normality cannot be achieved, non-parametric UTLs will not be calculated until a minimum of 60 background samples have been collected (to achieve 95 percent coverage).

The parametric UTL on a future mean will be calculated from the background dataset as follows:

$$UTL = \overline{x} + \kappa (n, \gamma, \alpha - 1) \cdot s$$

 \overline{x} = background sample mean

s = background sample standard deviation

 κ (*n*, γ , $\alpha - 1$) = one-sided normal tolerance factor based on the chosen coverage (γ) and confidence level (α -1) and the size of the background dataset (n). Values are tabulated in Table 17-3 in Appendix D of the *Unified Guidance*. If exact values are not provided, then κ values can be estimated by linear interpolation.

If the UTL is constructed on the logarithms of original observations to achieve normality, where \overline{y} and s_y are the log-mean and log-standard deviation, the limit will be exponentiated for back-transformation to the concentration scale as follows:

$$UTL = \exp\left[\overline{y} + \kappa (n, \gamma, \alpha - 1) \cdot s_y\right]$$

 \overline{y} = background sample log-mean

 s_v = background sample log-standard deviation

When the GWPS is based on the 35 I.A.C. § 845.600(a)(1) constituent concentrations or a UTL derived from the background dataset, an exceedance in compliance wells relative to the GWPS will be evaluated using confidence intervals. A confidence interval defines the upper and lower bound of the true mean of a constituent concentration in groundwater within a specified confidence range.

- Non-detects in compliance data will be handled similarly to upgradient analyses, with half the RL substituted for non-detects when the frequency is 15 percent or less.
- The Kaplan-Meier, or the ROS method, will be used when the detection frequency is between 15 percent and 50 percent to compute estimates of the mean and standard deviation adjusted for the presence of left-censored values. These estimates will then be substituted for the sample mean and standard deviation.

Once the GWPS is established for background data using the UTL, either parametric or non-parametric confidence intervals will be computed for each constituent in compliance wells to identify GWPS exceedances.

3.1.2 Parametric Confidence Intervals around a Mean

If compliance data are approximately normal, one-sided parametric confidence intervals around a sample mean will be constructed for each constituent and well pair. The lower confidence limit (LCL) will be calculated as:

$$LCL_{1-\alpha} = \overline{x} - t_{1-\alpha,n-1} \cdot \frac{s}{\sqrt{n}}$$

 \overline{x} = compliance sample mean

s = compliance sample standard deviation

n =compliance sample size

 $t_{1-\alpha,n-1}$ = obtained from a Student's t-table with (n-1) degrees of freedom (Table 16-1 in Appendix D of the *Unified Guidance*)

The chosen t value will aim to achieve both a low false-positive rate, and high statistical power. Minimum a values are tabulated in Table 22-2 of Appendix D of the *Unified Guidance*. The selected minimum a value, from which the t value will be derived, will have at least 80 percent power $(1-\beta = 0.8)$ when the underlying mean concentration is twice the GWPS.

If compliance data are distributed lognormally, the LCL will be computed around the lognormal geometric mean as:

$$LCL_{1-\alpha} = \exp\left(\overline{y} - t_{1-\alpha,n-1} \cdot \frac{s_y}{\sqrt{n}}\right)$$

 \overline{y} = compliance sample log-mean

 s_y = compliance sample log-standard deviation

3.1.3 Non-Parametric Confidence Intervals around a Median

Non-parametric confidence intervals around the median will be computed if the compliance data contain greater than 50 percent non-detects or are not normally distributed. The mathematical algorithm used to construct non-parametric confidence intervals is based on the probability (*P*) that any randomly selected measurement in a sample of n concentration measurements will be less than an unknown *P* x 100th percentile of interest (where *P* is between 0 and 1). Then the probability that the measurement will exceed the *P* x 100th percentile is (1-P). The number of sample values falling below the *P* x 100th percentile out of a set of n should follow a binomial distribution with parameters n and success probability *P*, where 'success' is defined as the event that a sample measurement is below the *P* x 100th percentile. The probability that the interval formed by a given pair of order statistics will contain the percentile of interest will then be determined by a cumulative binomial distribution Bin(x;n,p), representing the probability of *x* or fewer successes occurring in *n* trials with success probability p. *P* will be set to 0.50 for an interval around the median.

The sample size n will be ordered from least to greatest. Given P = 0.50, candidate interval endpoints will be chosen by ordered data values with ranks close to the product of $(n+1) \times 0.50$. If the result of $(n+1) \times 0.50$ is a fraction (for even-numbered sample sizes), the rank values immediately above and below will be selected as possible candidate endpoints. If the result of $(n+1) \times 0.50$ is an integer (for odd-numbered sample sizes), one will be added to and subtracted

from the result to get the upper and lower candidate endpoints. The ranks of the endpoints will be denoted L^* and U^* . For a one-sided LCL, the confidence level associated with endpoint L^* will be computed as:

$$1 - \alpha = Bin(L^* - 1; n, 0.50) = \sum_{x=L^*}^n \binom{n}{x} \left(\frac{1}{2}\right)^n$$

If the candidate endpoint(s) do not achieve the desired confidence level, new candidate endpoints $(L^{*}-1)$ and $(U^{*}+1)$ and achieved confidence levels will be calculated. If one candidate endpoint equals the data minimum or maximum, only the rank of the other endpoint will be changed. Achievable confidence levels are tabulated using these equations in Table 21-11 in Appendix D of the *Unified Guidance*.

Both parametric and non-parametric confidence limits will then be compared to the GWPS. The CCR unit is considered to be in compliance if the LCL is equal to or lower than the GWPS for all detected constituents at all compliance monitoring wells. A GWPS exceedance is determined if the LCL exceeds the GWPS.

3.1.4 The Upper Prediction Limit for a Future Mean

The parametric UPL for a future mean will be used to calculate the GWPS if the pooled background data contain 50 to 70 percent non-detects and normality can be achieved. The Kaplan-Meier or ROS methods will be used to estimate the mean and standard deviation. The non-parametric UPL for a future median will be calculated as the GWPS if background samples cannot be normalized or contain greater than 70 percent non-detects. The parametric UPL for a future mean will be calculated at a topological dataset at follows:

$$UPL_{1-\alpha} = \overline{x} + \kappa s$$

 \overline{x} = background sample mean

s = background standard deviation

 κ = multiplier based on the order (p) of the future mean to be predicted, the number of compliance wells to be tested (w), the background sample size (n) the number (c) of constituents of concern (COCs), the "1-of-m" retesting scheme, and the evaluation schedule (annual, semi-annual, quarterly). Values are tabulated in 19-5 to 19-9 in Appendix D of the *Unified Guidance*.

The mean of order p will be computed for each well and compared against the UPL. For any compliance point mean that exceeds the limit, p additional resamples may be collected at that well for a 1-of-2 retesting scheme. Resample means will then be compared to the UPL. A GWPS exceedance has been deemed to occur at a compliance well when the initial mean and all resample means exceed the UPL.

3.1.5 The Non-Parametric Upper Prediction Limit for a Future Median

The non-parametric UPL for a future median will be used to calculate the GWPS if the pooled background data contain greater than 70 percent non-detects and normality cannot be achieved. Non-parametric methods assume that the data does not have an underlying distribution. To calculate the non-parametric UPL on a future value, the target per-constituent false positive rate (α_{const}) will be determined as follows:

 $\alpha_{const} = 1 - (1 - \alpha)^{1/c}$

 α = the site-wide false positive rate (SWFPR) of 0.10 recommended by the *Unified Guidance*

c = the number of monitoring constituents

The number of yearly statistical evaluation (nE) will be multiplied by the number of compliance wells (w) to determine the look-up table entry, w*. The background sample size (n) and w* will be used to select an achievable per-constituent false positive rate value in Table 19-24 of Appendix D in the *Unified Guidance*. The chosen achievable per-constituent false positive rate value will determine the type of non-parametric prediction limit (maximum or 2nd highest value in background) and a retesting scheme for a future median. The background data will be sorted in ascending order, and the upper prediction limit will be set to the appropriate order statistic previously determined by the achievable per-constituent false positive rate value in Table 19-24. If all constituent measurements in a background sample are non-detect, the Double Quantification rule will be used. The use of the Double Quantification rule in Compliance Monitoring will only be applicable if the RL is above the 35 I.A.C. § 845.600(a)(1) constituent concentration is not specified in § 845.600(a)(1). This scenario is highly unlikely. The constituent will also be removed from calculations identifying the target false positive rate.

Two initial measurements per compliance well will be collected. If both do not exceed the upper prediction limit, a third initial measurement will not be collected since the median of order 3 will also not exceed the limit. If both exceed the prediction limit, a third initial measurement will not be collected since the median will also exceed the limit. If one initial measurement is above and one below the limit, a third initial observation may be collected to determine the position of the median relative to the UPL. Up to three resamples will be collected in order to assess the resample median. In all cases, if two or more of the compliance point observations are non-detect, the median will be set equal to the RL. The median value for each compliance well will be compared to the UPL. For the 1-of-2 retesting scheme, if any compliance point median exceeds the limit, up to three additional resamples will may be collected from that well. The resample median will be compared to the UPL. A GWPS exceedance has been deemed to occur at a compliance well when either the initial median, or both the initial median and resample median exceed the UPL.

If the concentrations of detected constituents are below the established GWPS, Compliance Monitoring will continue.

3.1.6 Parametric Linear Regression and Confidence Band

If the t-test detects a significant trend in the parametric linear regression line using either background or compliance data for a particular constituent, confidence bands accounting for trends will be constructed to account for the trend-induced variation. If this is not accounted for, a wider confidence interval will inevitably be calculated for a given confidence level and sample size (n). A wider confidence interval will result in less statistical power, or ability to demonstrate an exceedance or return to compliance. When a linear trend line has been estimated, a series of confidence intervals is estimated at each point along the trend. This creates a simultaneous confidence band that follows the trend line. As the underlying population mean increases or decreases, the confidence band does also to reflect this change at that point in time.

Linear regression will be used when background or compliance data are approximately normally distributed, with a constant sample variance around the mean, and the frequency of non-detects is low. The linear regression of concentration against sampling date (time) will be computed as follows:

$$\hat{b} = \sum_{i=1}^{n} (t_i - \overline{t}) \cdot x_i / (n-1) \cdot s_t^2$$

 $x_i = i^{\text{th}}$ concentration value and

 $t_i = i^{\text{th}}$ sampling date

 \overline{t} = sampling mean date

 s_t^2 = variance of the sampling dates

This estimate leads to the following regression equation:

 $\hat{x} = \overline{x} + \hat{b} \cdot (t - \overline{t})$

 \overline{x} = mean concentration level

 \hat{x} = estimated mean concentration at time t

The regression residuals will also be computed at each sampling event to ensure uniformity and lack of significant skewness. Regression residuals will be computed at each sampling event as follows:

$$r_i = x_i - \hat{x}_i$$

The estimated variance around the regression line, or mean squared error (MSE) will be computed as follows:

$$s_e^2 = \frac{1}{n-2} \sum_{i=1}^n r_i^2$$

The confidence intervals around a linear regression trend line given confidence level (1-a) and a point in time (t_0) , will be computed as follows:

$$LCL_{1-\alpha} = \hat{x}_{0} - \sqrt{2s_{e}^{2} \cdot F_{1-2\alpha,2,n-1} \cdot \left[\frac{1}{n} + \frac{(t_{0} - \overline{t})^{2}}{(n-1) \cdot s_{t}^{2}}\right]}$$
$$UCL_{1-\alpha} = \hat{x}_{0} - \sqrt{2s_{e}^{2} \cdot F_{1-2\alpha,2,n-2} \cdot \left[\frac{1}{n} + \frac{(t_{0} - \overline{t})^{2}}{(n-1) \cdot s_{t}^{2}}\right]}$$

 \hat{x}_0 = estimated mean concentration from the regression equation at time t₀

 $F_{1-2\alpha,2,n-2}$ = upper (1-2 α)th percentage point from an F-distribution with 2 and (n-2) degrees of freedom

For background data, the UCL around the linear regression line will be used as the GWPS for the trending constituent. For compliance data, confidence bands around the linear regression line will be compared to the GWPS. The CCR unit is considered to be in compliance if the LCL is equal to or lower than the GWPS for all detected constituents at all compliance wells. A GWPS exceedance is determined when the LCL based on the trend line first exceeds the GWPS.

3.1.7 Non-Parametric Thiel-Sen Trend Line and Confidence Band

If the Mann-Kendall test detects a significant trend in the non-parametric Thiel-Sen line using either background or compliance data for a particular constituent, confidence bands accounting for trends will be constructed to account for the trend-induced variation. The Thiel-Sen trend line will be used as a non-parametric alternative to linear regression when trend residuals cannot be normalized or if there are a higher percentage of non-detects in either background or compliance data. The Thiel-Sen trend line estimates the median concentration over time by combining the median pairwise slope with the median concentration value and the median sample date. To compute the Thiel-Sen line, the data will first be ordered by sampling event x1, x2, xn. All possible distinct pairs of measurements (x_i , x_j) for j > i will be considered and the simple pairwise slope estimate will be computed for each pair as follows:

$$m_{ij} = (x_j - x_i)/(j - i)$$

With a sample size of n, there will be a total of N = n(n-1)/2 pairwise estimates (m_{ij}) . If a given observation is a non-detect, half the RL will be substituted. The N pairwise slope estimates (m_{ij}) will be ordered from least to greatest (renamed m(1), m(2),..m(N)). The Thiel-Sen estimate of slope (Q) will be calculated as the median value of the list depending on whether N is even or odd as follows:

$$Q = \begin{cases} m_{([N+1]/2)} \text{ if } N \text{ is odd} \\ (m_{(N/2)} + m_{([N+2]/2)})/2 \text{ if } N \text{ is even} \end{cases}$$

The sample concentration magnitude will be ordered from least to greatest, x(1), x(2), to x(n) and the median concentration will be calculated as follows:

$$\tilde{x} = \begin{cases} x_{([n+1]/2)} \text{ if } n \text{ is odd} \\ (x_{(n/2)} + x_{([n+2]/2)})/2 \text{ if } n \text{ is even} \end{cases}$$

The median sampling date (\tilde{t}) with ordered times (t(1), t(2), to t(n)) will also be determined in this way. The Thiel-Sen trend line will then be computed for an estimate at any time (t) of the expected median concentration (x) as follows:

$$x = \tilde{x} + Q \cdot (t - \tilde{t}) = (\tilde{x} - Q \cdot \tilde{t}) + Q \cdot t$$

To construct a confidence band around the Thiel-Sen line, sample pairs (ti, xi) will be formed with a sample date (ti) and the concentration measurement from that date (xi). Bootstrap samples (B) will be formed by repeatedly sampling n pairs at random with replacement from the original sample pairs. This will be repeated 500 times. For each bootstrap sample, a Thiel-Sen trend line will be constructed using the equation above. A series of equally spaced time points (tj) will be identified along the range of sampling dates represented in the original sample, j =1 to m. The Thiel-Sen trend line associated with each bootstrap replicate will be used to compute an estimated concentration (\hat{x}_j^B). An LCL will be constructed for the lower α^{th} percentile $\hat{x}_j^{[\alpha]}$ from the distribution of estimated concentrations at each time point (tj). For a UCL, compute the upper (1- α)th percentile, $\hat{x}_j^{[1-\alpha]}$ at each time point (tj).

For background data, the UCL around the Thiel-Sen trend line will be used as the GWPS for the trending constituent. For compliance data, confidence bands around the Thiel-Sen trend line will be compared to the GWPS. The CCR unit is considered to be in compliance if the LCL is equal to or lower than the GWPS for all detected constituents at all compliance wells. A GWPS exceedance is confirmed when the LCL based on the trend line first exceeds the GWPS.

3.2 Determination of Statistically Significant Increases over Background

In accordance with 35 I.A.C. §§ 845.610(b)(3)(B) and 845.640(h), individual monitoring event concentrations for each constituent detected in the compliance monitoring wells during compliance monitoring sampling events will be compared to the background concentration as determined by the methods described above. An exceedance of the background concentration for any constituent measured at any compliance monitoring well, or constituent detection if not detected in the background samples, constitutes a Statistically Significant Increase (SSI). An exception to this method is pH, where two-sided (upper and lower) tolerance limits are established from the distribution of the background groundwater quality data. An exceedance of either the UTL or lower tolerance limit (LTL) would constitute an SSI for pH.

4. **REFERENCES**

Davis, C.B., 1994. *Environmental Regulatory Statistics*. In GP Patil & CR Rao (Eds.) *Handbook of Statistics, Volume 12: Environmental Statistics*, Chapter 26. New York: Elsevier Science B.V.

United States Environmental Protection Agency (USEPA), 2009. *Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities: Unified Guidance*. EPA 530-R-09-007. March 2009.

ATTACHMENT 2 GROUNDWATER MONITORING PLAN DATED JANUARY 24, 2017; AND RESPONSE TO COMMENT LETTER DATED NOVEMBER 27, 2017


By UPS

November 27, 2017

William E. Buscher, P.G. Manager, Hydrogeology and Compliance Unit Groundwater Section, Division of Public Water Supplies, Bureau of Water Illinois Environmental Protection Agency 1021 North Grand Avenue East Springfield, IL 62794-9276

Re: Coffeen Power Station Inactive Ash Pond No. 2 Responses to IEPA's October 27, 2017 Comments on IPGC's February 1, 2017 Closure and Post-Closure Care Plans

Dear Mr. Buscher:

The enclosed document has been prepared to respond to the eight comments in your October 27, 2017 letter regarding our February 1, 2017 Closure and Post-Closure Care plans for inactive ash pond no. 2. The comments are based in part upon the telephone conference call discussion with you and your staff and Natural Resources Technology staff on November 7, 2017.

Please let us know if you would like to meet with us, or have another telephone conference call to discuss these responses. My telephone no. is 618-343-7761.

Sincerely,

Rick Diericx Managing Director – Environmental Compliance Group

Enclosures

bcc: M. Ballance / J. Frierdich / V. Modeer – Collinsville T. Davis – Collinsville Stu Cravens – NRT (Bloomington, IL) Rick Diericx Reading File - Collinsville





OBG | There's a Way

November 21, 2017

Mr. Rick Diericx Managing Director – Environmental Compliance Group Dynegy Operating Company 1500 Eastport Plaza Drive Collinsville, IL 62234-6135

Subject:Response to IEPA Comments – Coffeen Station Inactive Ash Pond No. 2
Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2
NRT Project No. 2380

Dear Mr. Diericx:

Natural Resource Technology, Inc., an OBG Company (NRT) is providing this letter to Dynegy Operating Company (Dynegy) in response to comments received from the Illinois Environmental Protection Agency (IEPA) dated October 27, 2017 regarding the *Closure and Post-Closure Care Plan for the Coffeen Ash Pond No. 2* (Closure Plan; AECOM, January, 2017) at Illinois Power Generating Company Coffeen Power Station, in Coffeen, IL.

This Response to Comments will serve as Addendum 1 to the Closure Plan dated January 2017. For ease of review, IEPA comments are presented below in italics, followed by responses. Supplemental information to support the responses, when required, is included as Attachments 1-3. This document provides responses to all IEPA comments numbered 1 - 8.

Comment 1

It is mentioned multiple times in the closure and post-closure care plans that there are coal mines in the vicinity at depth. Please provide additional information on the locations and depths of the coal mines in the vicinity of Ash Pond No. 2.

<u>Response:</u> The Truax-Traer Coal Company and the Consolidation Coal Company extracted coal underlying Ash Pond No. 2 from 1964 to 1983 (Attachment 1 – Mine Index 871). The mine was originally known as the "Hillsboro" Mine and after Consolidation Coal Company took over the mining operation it was renamed "Consolidation No. 63, Hillsboro". Herrin No. 6 Coal was mined from depths of 500-510 feet below ground surface. The coal seam was 5.8-7.1 feet thick and an estimated 26,800,000 tons of coal were removed from the mine during the operational period (see Reference 1) with an extraction ratio of approximately 25% based on an estimate of volume removed from the recorded mine maps. A Mine Workings Map dated 1969 is overlaid on an aerial of Ash Pond No. 2 and included in Attachment 1 (Figure 1 – Overlay of Historic Mine). Comparison of the mine extents from 1969 with the extent of the mine included in Reference 1 and shown on the Coal Mines in Illinois, Coffeen Quadrangle map included in Attachment 1 indicates there was no additional mining below Ash Pond No. 2 between 1969 and 1983 when the mine closed. As stated in Mine Index 871 (Attachment 1) the Coal Section of the Illinois State Geologic Survey (ISGS) has been assured that the extents included in Reference and Attachment 1 are final and complete.

 Image: Constraint of the street, sifth Floor
 p 414-837-3607
 NRT | AN OBG COMPANY obg.com/nrt

 Milwaukee, WI 53204
 f 414-837-3608
 obg.com/nrt

Comment 2

The Agency requests the addition of a groundwater monitoring well on the east side of Inactive Ash Pond No. 2.

<u>Response:</u> NRT has evaluated the site conditions in the area east of Ash Pond No. 2 and has determined that there is no feasible access to install a monitoring well outside of the embankment because of steep slopes, heavy vegetation, and the presence of wetlands. In addition, several borings along the east side of Ash Pond No. 2 (B403A, P010, and G402) did not encounter the uppermost aquifer during drilling. Given significant access concerns and the limited nature of the aquifer NRT does not recommend installation of a monitoring well.

However, if required or deemed necessary in the future by the IEPA, a boring could be advanced through the berm following completion of closure construction activities to determine if the uppermost aquifer (Hagarstown) is present. Assuming the uppermost aquifer is present, a well could be installed through the berm if requested.

Comment 3

The two new groundwater monitoring wells to the west of the Inactive Ash Pond No. 2 are approved.

<u>Response:</u> A schedule for installation of the wells will be developed following approval of this Closure Plan. Appropriate documentation will be submitted to the IEPA upon completion of installation.

Comment 4

Will changes need to be made to the NPDES permit as part of the implementation of the closure plan?

<u>Response:</u> Yes, an application to modify the Coffeen NPDES permit (IL0000108) is being prepared to authorize the discharge of "dewatering" wastewaters from Ash Pond No. 2 to Coffeen Lake. The application will include an Antidegradation Assessment for Coffeen Lake in support of that modification request.

Comment 5

Calibration of the MODFLOW model was completed using only November 2016 groundwater elevations. Further calibration using more groundwater elevation data over a larger span of time should be completed.

<u>Response</u>: November 2016 groundwater elevations were used to calibrate the MODFLOW model because this was the first and only complete data set available for model development that included groundwater elevations from monitoring wells set in the uppermost aquifer, including wells G406 and G407 installed in August 2016. In order to address IEPA concerns about the efficacy of the groundwater elevation data range used for the model calibration, a comparison of observed versus predicted groundwater elevation values and groundwater flow directions will be provided in post-closure annual reports to assess model performance. If the predictive model does not adequately represent groundwater elevation and flow directions, the model will be recalibrated using available groundwater elevation data collected after November 2016. Similarly, post-closure groundwater quality data will be compared to transport model predictions to assess model performance in post-closure annual reports.



Comment 6

In the Application for a GMZ, it is noted that the problem with groundwater was identified via sampling in 2015. The Agency notes that the Coffeen Power Station received a Violation Notice in 2012 for groundwater standards violations in groundwater around Inactive Ash Pond No. 2.

<u>Response:</u> The GMZ application was revised to include reference to Violation Notice W-2012-00064 (Attachment 2).

Comment 7

Also in the GMZ application, no other remedies to groundwater violations are considered other than to state they are not deemed practical or cost effective. Please discuss other remedies considered and why they are not practical for mitigation relative to the nature of the subsurface or cost-effective.

Response: The GMZ Application was revised to include the following text (Attachment 2):

"Previous experience at similar sites developing and evaluating alternative remedial options and determining costs indicates capping is often the most cost-effective and cost-efficient remedy. Therefore, dewatering and capping were initially evaluated. Based on the results of the evaluation and predictive modeling, the selected remedy successfully mitigates groundwater impacts. Groundwater monitoring will continue to be performed to evaluate the effectiveness of the remedy. If the selected remedy is not demonstrated as successful through collection of data and comparison to predictive values and applicable groundwater quality standards, then other remedial options will be evaluated."

Comment 8

All monitoring wells must be sampled for the parameters listed in 35 IAC 620.410 (a) and (d), with the exception of perchlorate. Statistical analysis for each well's parameters using approved methods listed in 40 CFR 257 should be included.

<u>Response:</u> Tables 2 - 5 of the Groundwater Monitoring Plan, which was included as Appendix B of the Closure Plan, have been revised to include all parameters of 35 IAC 620.410 (a) and (d) with the exception of perchlorate. Revised tables are included in Attachment 3. Note these tables also include aluminum and proposed changes to groundwater standards which are included in IEPA's Proposed Changes to 35 IAC Part 620.

Please don't hesitate to contact us if you have any questions regarding these responses to comments and associated attachments provided herein.

Sincerely, NRT | An OBG Company

stand R Kellen

Nathaniel R. Keller, PG Hydrogeologist

tuit ! hade

Stu J. Cravens, PG Principal Hydrogeologist



Attachments:

Attachment 1: Comment 1 – Historic Mine Documentation
Attachment 2: Comment 2 – Revised GMZ Application
Attachment 3: Comment 8 – Revised Tables 2,3,4, and 5 from Appendix B: Groundwater Monitoring Plan

cc: Ms. Amy Zimmer – IEPA, Hydrogeology and Compliance Unit Mr. Tom Davis, PE - Dynegy, Inc.
Mr. Jason Frierdich, PE – Dynegy, Inc.
Mr. Matt Ballance, PE – Dynegy, Inc.
Mr. Vic Modeer, PE – Dynegy, Inc.
Mr. John Romang, - Illinois Power Generating Company

References:

1 Obrad, Jennifer M., 2011. Directory of Coal Mines in 7.5-Minute Quadrangle Series, Coffeen Quadrangle Montgomery and Bond Counties. The Board of Trustees of the University of Illinois.





Attachment 1

Comment 1-Historic Mine Documentation



Mine Index 871 Consolidation Coal Company, Hillsboro Mine (Consolidation No. 63 Mine)

Type: Underground Total mined-out acreage shown: 4,841

SHAFT, SLOPE, DRIFT or TIPPLE LOCATIONS

Туре	County	Township-Range	Section	Quarters-Footage	
Man shaft	Montgomery	7N 3W	14	SW NE NW	
Air shaft	Montgomery	7N 2W	18	SE NE NE	
Hoist & air shaft	Montgomery	7N 3W	14	NE NW NW	

GEOLOGY

		Th	ickness (1	ft)	Mining	
Seam(s) Mined	Depth (ft)	Min	Max	Ave	Method	
Herrin	500-510			5.83-7.17	RPP	

<u>Geologic Problems Reported</u>: Roof problems were widespread, the sites characterized by slickensided fault planes that cut irregularly through the roof shales and claystones. Small clay dikes were also associated with this small-scale faulting. Floor heaving was slight, but had been a larger problem in the past.

PRODUCTION HISTORY

Company	Mine Name	Years	(tons)	
Truax-Traer Coal Company	Hillsboro	1964-1970	5,605,812	
Consolidation Coal Company	Consolidation No. 63, Hillsboro	1971-1983	21,173,542	
			26,779,354	

. .

Last reported production: July 1983

SOURCES OF DATA

		Original	Digitized		
Source Map	Date	Scale	Scale	Мар Туре	
Company, Coal Section files	2-1-1983	1:12000	1:2170	Final *	

* The map date is before mine closure, but the Coal Section has been assured that the workings shown on the map are indeed final. The mined area shown on the accompanying map is the approximate size expected for the reported production. This suggests that the mine outline is complete.

Annotated Bibliography (data source, brief description of information)

Coal Reports - Production, ownership, years of operation, mine type, depth, thickness. Directory of Illinois Coal Mines (Montgomery County) - Mine names, mine index, ownership, years of operation. Mine notes (Montgomery County) - Shaft location, seam, depth, thickness, geologic problems. Company map, Coal Section files, 1983 Line Project - Shaft locations, mine outline, mining method.





COFFEEN ASH POND NO. 2 | RESPONSE TO IEPA COMMENTS



Attachment 2

Comment 2 – Revised GMZ Application



Title 35, Illinois Admin. Code, Part 620 – APPENDIX D Confirmation of an Adequate Corrective Action Pursuant to 35 Ill. Adm. Code 620.250(a)(2)

Pursuant to 35 Ill. Adm. Code 620.250(a) if an owner or operator provides a written confirmation to the Agency that an adequate corrective action, equivalent to a corrective action process approved by the Agency, is being undertaken in a timely and appropriate manner, then a groundwater management zone may be established as a threedimensional region containing groundwater being managed to mitigate impairment caused by the release of contaminants from a site. This document provides the form in which the written confirmation is to be submitted to the Agency.

- Note 1.Parts I and II are to be submitted to IEPA at the time that the facility claims the alternative
groundwater standards. Part III is to be submitted at the completion of the site investigation.
At the completion of the corrective process, a final report is to be filed which includes the
confirmation statement included in Part IV.
- Note 2. The issuance of a permit by IEPA's Division of Air Pollution Control or Water Pollution Control for a treatment system does not imply that the Agency has approved the corrective action process.
- Note 3. If the facility is conducting a cleanup of a unit which is subject to the requirements of the Resource Conservation and Recovery Act (RCRA) or the 35 Ill. Adm. Code 731 regulations for Underground Storage Tanks, this confirmation process is not applicable and cannot be used.
- Note 4. If the answers to any of these questions require explanation or clarification, provide such in an attachment to this document.

Information provided in the following technical documents is referenced within this form:

- AECOM, 2016c. Revised 30% Closure Design Package for Coffeen Power Station Ash Pond No. 2. April 16, 2016.
- Natural Resource Technology, Inc., 2016a. Hydrogeologic Characterization Report. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.
- Natural Resource Technology, Inc., 2016b. Groundwater Monitoring Plan. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.
- Natural Resource Technology, Inc., 2016c. Groundwater Model Report. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.
- Natural Resource Technology, Inc., 2016d. Hydrostatic Modeling Report. Coffeen Ash Pond No. 2, Illinois Power Generating Company Coffeen Power Station, Coffeen, IL.

A legal description and map of the proposed GMZ is provided in Appendix A of this GMZ Application. The GMZ will extend vertically through all water-bearing strata through the Hagarstown Member and upper weathered portions of the Vandalia Till Member at an estimated elevation ranging from approximately 604 to 608 ft MSL.



Part I. Facility Information

Facility Name	Coffeen Power Station
Facility Address	134 CIPS Lane, Coffeen, IL 62017
County Montg	omery
Standard Industri	al Code (SIC) 4911

- 1. Provide a general description of the type of industry, products manufactured, raw materials used, location and size of the facility. *Electric power generation and coal combustion residual (CCR) disposal. Ash Pond 2 is located within the Coffeen Power Station which encompasses approximately 4,000 acres including a 1,100-acre lake.*
- 2. What specific units (operating or closed) are present at the facility which are or were used to manage waste, hazardous waste, hazardous substances or petroleum?

	YES	<u>NO</u>
Landfill	Х	
Surface Impoundment	X	
Land Treatment		Х
Spray Irrigation		Х
Waste Pile		Х
Incinerator		Х
Storage Tank (above ground)	X	
Storage Tank (underground)	X	
Container Storage Area	X	0
Injection Well	X	
Water Treatment Units	X	
Septic Tanks	X	
French Drains	X	
Transfer Station		Х
Other Units (please describe)		2

Provide an extract from a USGS topographic or county map showing the location of the site and a more detailed scaled map of the facility with each waste management unit identified in Question 2 or known/suspected source clearly identified. Map scale must be specified and the location of the facility must be provided with respect to Township, Range and Section. *Facility is located in Sections 10 and 11, Tier 7 N, Range 3 W, of the 3rd PM. Figure 1 has the facility located on a USGS topographic map (7½ minute).*



- 4. Has the facility ever conducted operations which involved the generation, manufacture, processing, transportation, treatment, storage or handling of "hazardous substances" as defined by the Illinois Environmental Protection Act? Yes ⊠ No □
 If the answer to this question is "yes" generally describe these operations. *Storage and handling of anhydrous ammonia, sulfuric acid, 50% sodium hydroxide, and chlorine gas.*
- Has the facility generated, stored or treated hazardous waste as defined by the Resource Conservation and Recovery Act? Yes ⊠ No □
 If the answer to this question is "yes" generally describe these operations. *Small quantity TSD.*
- 6. Has the facility conducted operations which involved the processing, storage or handling of petroleum? Yes ⊠ No □ If the answer to this question is "yes" generally describe these operations. *Store, load, and unload diesel fuel and kerosene.*
- 7. Has the facility ever held any of the following permits?
 - a. Permits for any waste storage, waste treatment or waste disposal operation. Yes ⊠ No □ If the answer to this question is "yes", identify the IEPA permit numbers. *IL0000108 and 1998-289- UIC.*
 - b. Interim Status under the Resources Conservation and Recovery Act (filing of a RCRA Part A application).
 Yes □ No ⊠
 If the answer to this question is "yes", attach a copy of the last approved Part A application.
 - c. RCRA Part B Permits. Yes □ No ⊠
 If the answer to this question is "yes", identify the permit log number.
- 8. Has the facility ever conducted the closure of a RCRA hazardous waste management unit? Yes 🗆 No 🗵
- 9. Have any of the following State or federal government actions taken place for a release at the facility?
 - a. Written notification regarding known, suspected or alleged contamination on or emanating from the property (e.g., a Notice pursuant to Section 4(q) of the Environment Protection Act)? Yes ⊠ No □ If the to this question is "yes", identify the caption and date of issuance. *Violation Notice No. W-2012-00064 was issued on June 27, 2012 for boron, manganese, sulfate, and total dissolved solids concentrations which exceeded Class I GW Standards at APW-2/G402.*
 - b. Consent Decree or Order under RCRA, CERCLA, EPAct Section 22.2 (State Superfund), or EPAct Section 21(f) (State RCRA). Yes □ No ⊠
 - c. If either of Items a. or b. were answered by checking "yes", is the notice, order or decree still in effect? Yes ⊠ No □ *Concentrations remain above Class I GW Standards which is why this GMZ is being requested.*
- 10. What groundwater classification will the facility be subject to at the completion of the remediation?

Class I 🛛 Class II 🗆 Class III 🗆 Class IV 🗆

If more than one Class applies, please explain.

11. Describe the circumstances which the release to groundwater was identified. Groundwater sampling at Ash Pond 2 was initiated in 2015. Exceedances of Class I groundwater quality standards in monitoring wells associated with Ash Pond 2 include the parameters arsenic, boron, lead, manganese, sulfate, and total dissolved solids.



Based on my inquiry of those persons directly responsible for gathering the information, I certify that the information submitted is, to the best of my knowledge and belief, true and accurate.

Coffeen Power Station

Facility Name

134 CIPS Lane, Coffeen, IL 62017

Location of Facility

1358030005

Illinois EPA Identification Number

Signature of Owner/Operator

Illinois Power Generating Company Name of Owner/Operator

Date



PART II: Release Information

1. Identify the chemical constituents release to the groundwater. Attach additional documents as necessary.

Chemical Description	Chemical Abstract No.
Arsenic	7440-38-2
Boron	7440-42-8
Lead	7439-92-1
Manganese	7439-96-5
Sulfate	14808-79-8
Total Dissolved Solids	10052

- 2. Describe how the site will be investigated to determine the source or sources of the release. Ash Pond 2 has been investigated as described in the Hydrogeologic Characterization Report (Natural Resource Technology, Inc. [NRT], 2016a).
- 3. Describe how groundwater will be monitored to determine the rate and extent of the release. The monitoring network to monitor the rate and extent of the release is described in the Groundwater Monitoring Plan (NRT, 2016b).
- 4. Has the release been contained on-site at the facility? *The release is contained within the facility boundary. Migration of CCR constituents is limited by Coffeen Lake, which acts as a groundwater discharge area and hydraulic barrier.*
- 5. Describe the groundwater monitoring network and groundwater and soil sampling protocols in place at the facility. *The groundwater monitoring network and sampling protocols are described in the Groundwater Monitoring Plan (NRT, 2016b).*
- 6. Provide the schedule for investigation and monitoring. *The site investigation is complete and groundwater monitoring will continue for the required/permitted frequency and monitoring period as described in the Groundwater Monitoring Plan Section 4.2: Sampling Schedule (NRT, 2016b).*
- 7. Describe the laboratory quality assurance program utilized for the investigation. Laboratory quality assurance is described in the Groundwater Monitoring Plan Sections 4.4: Laboratory Analysis and 4.5: Quality Assurance (NRT, 2016b). The quality assurance/quality control procedures described in the Groundwater Monitoring Plan will be supplemented by the selected Illinois EPA-approved laboratory's QA Manual.
- 8. Provide a summary of the results of available soil testing and groundwater monitoring associated with the release at the facility. The summary or results should provide the following information: dates of sampling; types of samples taken (soil or water); locations and depths of samples; sampling and analytical methods; analytical laboratories used; chemical constituents for which analyses were performed; analytical detection limits; and concentrations of chemical constituents in ppm (levels below detection should be identified as "ND"). A narrative summary of the results of groundwater monitoring is discussed in the Hydrogeologic Characterization Report Section 3: Groundwater Quality (NRT, 2016a). Analytical data summary tables and graphs are available in the Hydrogeologic Characterization Report Appendix F: Groundwater Quality Data and Appendix G: Water Quality Trend Graphs (NRT, 2016a). Lab reports for all monitoring events have previously been submitted to the Agency.





PART II: Release Information (Continued)

Based on my inquiry of those persons directly responsible for gathering the information, I certify that the information submitted is, to the best of knowledge and belief, true and accurate and confirm that the actions identified herein will be undertaken in accordance with the schedule set forth herein.

Coffeen Power Station		
Facility Name	Signature of Owner/Operator	
134 CIPS Lane, Coffeen, IL 62017	Illinois Power Generating Company	
Location of Facility	Name of Owner/Operator	
1358030005		
Illinois EPA Identification Number	Date	





Part III: Remedy Selection Information

- 1. Describe the selected remedy. The remedy includes ash dewatering, relocating/reshaping the CCR within Ash Pond 2 to achieve acceptable grades, construction of a geomembrane cover system and establishing a vegetative cover to minimize long-term erosion (AECOM, 2016).
- 2. Describe other remedies which were considered and why they were rejected. *Previous experience at similar* sites developing and evaluating remedial alternatives and costs indicate capping is often the most cost effective and cost-efficient. Therefore, dewatering and capping were initially evaluated. Based on the results of the evaluation and modeling, the selected remedy successfully mitigates groundwater impacts. If the selected remedy is not shown successful through collection of data and comparison to predictive values, then other remedial options will be evaluated.
- Will waste, contaminated soil or contaminated groundwater be removed from the site in the course of this remediation? Yes □ No ⊠
 If the answer to this question is "yes", where will the contaminated material be taken?
- 4. Describe how the selected remedy will accomplish the maximum practical restoration of beneficial use of groundwater. The dewatering and installation of a geomembrane cover system will control the potential for water infiltration into the closed CCR unit and will allow drainage of surface water off of the cover system. These actions will reduce leachate generation and migration and groundwater quality will improve over time, as described in the Groundwater Model Report (NRT, 2016c).
- 5. Describe how the selected remedy will minimize any threat to public health or the environment. The currently defined extent of the release does not threaten public health. As discussed in the Hydrogeologic Characterization Report Section 2.5 (NRT, 2016a), there are currently no impairments to groundwater usage on the Coffeen Power Station property or surrounding properties associated with Ash Pond 2. No impairments to groundwater usage resulting from establishment of the proposed GMZ are anticipated. CCR dewatering and the geomembrane cover system will reduce leachate generation and migration from Ash Pond 2 and minimize CCR constituents entering the environment, as described in the Groundwater Model Report (NRT, 2016c).
- 6. Describe how the selected remedy will result in compliance with the applicable groundwater standards. *The in place closure of Ash Pond 2, as proposed, will result in a reduction of leachate production, decreasing CCR constituent concentrations and contraction of the groundwater plume. A Groundwater Model Report (NRT, 2016c), included in Appendix D of AECOM 2016, suggests that the geosynthetic cover system will control recharge and subsequent leachate generation within the limits of the Site and reduce concentrations of boron below Class I standards. Concentration reductions are expected to begin approximately one year after completion of the cover system.*
- 7. Provide a schedule for design, construction and operation of the remedy, including dates for the start and completion. *A schedule for implementing the remedies is included in Section 1.3 in AECOM, 2016.*
- 8. Describe how the remedy will be operated and maintained. *The operation and maintenance of the remedy is described in Section 3: Post-Closure Care Plan (AECOM, 2016).*
- 9. Have any of the following permits been issued for the remediation?
 - a. Construction or Operating permit from the Division of Water Pollution Control. Yes \Box No igtiade
 - b. Land treatment permit from the Division of Water Pollution Control. Yes □ No ⊠ If the answer to this question is "yes", identify the permit number.
 - c. Construction or Operating permit from the Division of Air Pollution Control. Yes \Box No \boxtimes



If the answer to this question is "yes", identify the permit number.

10. How will groundwater at the facility be monitored following completion of the remedy to ensure that the groundwater standards have been attained? *Groundwater monitoring procedures are described in Section 4 of the Groundwater Monitoring Plan (NRT, 2016b).*

Based on my inquiry of those persons directly responsible for gathering the information, I certify that the information submitted is, to the best of my knowledge and belief, true and accurate and confirm that the actions identified herein will be undertaken in accordance with the schedule set forth herein.

Coffeen Power Station	
Facility Name	Signature of Owner/Operator
134 CIPS Lane, Coffeen, IL 62017	Illinois Power Generating Company
Location of Facility	Name of Owner/Operator
1358030005	
Illinois EPA Identification Number	Date



COFFEEN ASH POND NO. 2 | RESPONSE TO IEPA COMMENTS



Attachment 3

Comment 8 – Revised Tables 2,3,4, and 5 from Appendix B: Groundwater Monitoring Plan



Table 2. Proposed Monitoring Well Network and Analyses Coffeen Power Station - Ash Pond 2 **Groundwater Monitoring Plan**

	Other Analyses (USEPA	CCR Rule or IEPA)		40 CER 257 - Annendix III	and Amendia IV	Barameters Groundwater		LIEVAUOI			None			IEPA Approved Parameters	40 CFR 257 - Appendix III and Appendix IV	Parameters, Groundwater Elevation
	Additional Monitoring	Programs Performed at Well	Ash Pond 2 & GRP - CCR								None	BION		SW Pond - IEPA		GRP - IEPA and CCR
Proposed Analyses for	IEPA	Monitoring								IEPA 620.410	(a), and (d), no	nerchlorate ³				
	Screen	Bottom	20.9	22.7	21.3	22.8	20.4	13.7	16.5	18.4	21.6	TBD	TBD	21.6	29.6	20.5
	Screen	Тор	16.1	18.1	16.9	12.8	15.8	9.0	11.7	13.6	16.8	TBD	TBD	17.1	25.3	15.7
	Screen	length	5	5	4	10	5	5	5	5	5	TBD	TBD	5	4	5
Bottom of	Screen	Elevation	605	603.66	604.24	590.6	606.03	601.93	607.14	603.49	599.74	TBD	TBD	604.76	602.40	605.32
Top of	Screen	Elevation	609.79	608.31	608.67	600.60	610.70	606.68	611.89	608.30	604.57	TBD	TBD	609.26	606.79	610.16
Measuring Point	Elevation	(2015)	625.92	626.36	625.57	613.37	626.47	615.67	623.63	621.86	621.32	TBD	TBD	626.35	632.04	625.85
Ground Surface	at Time	of Install	622.92	623.82	623.03	610.56	623.81	613.10	620.90	621.86	618.35	TBD	TBD	623.52	629.19	622.95
	Boring/	Well ID	G270	G281	G401	G402	G403	G404	G405	G406	G407	G410 ¹	G411 ¹	G154	G279	G280

Notes:

¹ Proposed wells to be installed upon approval of Closure Plan and GMZ application

² Field parameters include: pH, oxidation -reduction potential, specific conductance, temperature, and dissolved oxygen

³ Groundwater samples collected for metals analyses will be field filtered. Groundwater quality analyses including methods and sampling details are included in Table 4.



Table 3. Background Groundwater Quality and Applicable Groundwater Quality Standards Groundwater Monitoring Plan

Coffeen Power Station - Ash Pond 2

			Backaround	Applicable Groundwater		
	Sampling	IL Class I Std ¹	Concentration ²	Standard ³ for	Maximum ⁵	Minimum ⁵
Parameters (totals)	Program	(mg/L)	tor IEPA (mg/L)	IEPA (mg/L)	(mg/L)	(mg/L)
Aluminum (d)	IEPA	3.5	toq	tDQ	1.0	c00.0>
Antimony	CCR, IEPA	0.006	tbd	tbd	<0.003	<0.003
Arsenic	CCR, IEPA	0.01	tbd	tbd	0.25	<0.001
Barium	CCR, IEPA	2.0	tbd	tbd	0.24	0.014
Beryllium	CCR, IEPA	0.004	tbd	tbd	0.0018	<0.001
Boron	CCR, IEPA	2.0	tbd	tbd	11	<0.01
Calcium	CCR	NS	tbd	tbd	450	<0.1
Cadmium	CCR, IEPA	0.005	tbd	tbd	0.008	<0.001
Chloride	CCR, IEPA	200	tbd	tbd	160	1.5
Chromium	CCR, IEPA	0.1	tbd	tbd	0.034	<0.004
Cobalt	CCR, IEPA	1, 0.002 ⁴	tbd	tbd	0.28	<0.001
Copper (d)	IEPA	0.65, 0.2 ⁴	tbd	tbd	0.021	<0.001
Cyanide	IEPA	0.2	tbd	tbd	<0.005	<0.003
Fluoride	CCR, IEPA	4	tbd	tbd	1.06	0.031
Iron (d)	IEPA	5	tbd	tbd	13	<0.005
Lead	CCR, IEPA	0.0075	tbd	tbd	0.220	<0.001
Lithium	CCR	NS	tbd	tbd	0.057	<0.01
Manganese(d)	IEPA	0.15	tbd	tbd	1.02	<0.018
Mercury	CCR, IEPA	0.002	tbd	tbd	0.00093	<0.0002
Molybdenum	CCR, IEPA	NS	tbd	tbd	0.043	<0.001
Nickel (d)	IEPA	0.1	tbd	tbd	0.035	<0.003
Nitrate-N	IEPA	10	tbd	tbd	8.8	<0.01
Selenium	CCR, IEPA	0.05	tbd	tbd	0.027	<0.001
Silver (d)	IEPA	0.05	tbd	tbd	<0.005	<0.003
Sulfate	CCR, IEPA	400	tbd	tbd	2,500	2.3
Thallium	CCR, IEPA	0.002	tbd	tbd	0.0013	<0.001
TDS (d)	CCR, IEPA	1,200	tbd	tbd	3,900	320
Vanadium (d)	IEPA	0.049, 0.00049 ⁴	tbd	tbd	0.025	<0.003
Zinc (d)	IEPA	5	tbd	tbd	0.59	<0.002
Field pH	CCR, IEPA	6.5 - 9.0	tbd	tbd	8.03	5.80
Radium 226/228	CCR. IEPA	20/20.54	tbd	tbd	4.46	0.185

Notes:

All parameters are totals unless noted. Standards apply to dissolved or total concentrations (d) Dissolved **tbd** = To Be Determined for Illinois EPA monitoring program; CCR Appendix III and IV parameters based on future monitoring , started in November 2015

Bold = Background Concentration exceeds Class I Groundwater Standard

Red = Exceeds Applicable Groundwater Standard NS = No Class II Groundwater Standard

USEPA (t) = background concentration for parameter [total] required under USEPA program (40 CFR Part 257)

² Background Concentration to be calculated following 8 sampling events at all wells in accordance with 40 CFR 257 ¹ IPCB 620 Class I: Potable Resource Groundwater Standard

³ Applicable Groundwater Standard is the higher of the Background Concentration and the Class I Groundwater Standard (or the lower if compared to the pH lower limit)

⁴ Standards listed are proposed changes to 35 III. Adm. Code Part 620 by Illinois EPA

⁵ Groundwater concentrations based on historical results for wells in the proposed sampling program ³ Groundwater samples collected for metals analysis as required by IL620.410 will be field filtered



Table 4. Sampling and Analysis Summary Coffeen Power Station - Ash Pond 2 **Groundwater Monitoring Plan**

Parameter	Analytical Method ⁵	Number of Samples	Field Duplicates ¹	Field Blanks ²	Equipment Blanks ²	MS/MSD ³	Total	Container Type	Minimum Volume ⁴	Preservation (Cool to 4 °C for all samples)	Sample Hold Time from Collection Date
Metals - Appendix III ⁽¹⁾											
Boron (total and dissolved)	6020	14	2	0	0	-	17	plastic	600 mL	HNO ₃ to pH<2	6 months
Calcium	6020	7	1	0	0	1	9	plastic	600 mL	HNO ₃ to pH<3	6 months
Metals - Appendix IV ⁽²⁾ and Add	litional Metals										
Other Metals ⁽³⁾ (total and discolved)	6020	14	2	0	0	١	17	plastic	600 mL	HNO3 to pH<2	6 months
Manganese (d)	6020	14	2	0	0	-	17	plastic	600 mL	HNO3 to pH<2	6 months
Lithium	6020	7	-	0	0	۲	თ	plastic	600 mL	HNO3 to pH<2	6 months
Mercury	7470A or 6020	14	2	0	0	-	17	plastic	400 mL	HNO ₃ to pH<2	28 days
norganic Parameters - Append	ix III (1) and Othe	er Inorganic P	arameters								
	SM 4500-CN										
Cyanide	or C – EPA	14	2	0	0	-	17	amber	50 mL	NaOH, Cool to 4 °C	14 days
Elinorido	335.4 0214	11	ç	c	c	•	17	nlactic	300 ml	Cont to 4 °C	28 dave
Chloride	9251	4	10	0	0 0	• •	17	plastic	100 mL	Cool to 4 °C	28 days
Nitrate-N	EPA 300.0	14	0	0	0	-	17	plastic	10 mL	Cool to 4 °C	48 hours
Sulfate	9036	14	2	0	0	-	17	plastic	50 mL	Cool to 4 °C	28 days
Total Dissolved Solids	SM 2540 C	14	2	0	0	1	17	plastic	200 mL	Cool to 4 °C	7 days
Radium - Appendix IV ⁽²⁾											
Radium 226	9315 or EPA 903	14	0	0	0	Ļ	15	plastic	1000 mL	HNO3 to pH<2	6 months
Radium 228	9320 or EPA 904	14	0	0	0	۲	15	plastic	1000 mL	HNO ₃ to pH<2	6 months
Field Parameters											
(1) Hd	SM 4500-H+ B	14	NA	NA	NA	AN	14	flow-through cell	NA	none	immediately
Dissolved Oxygen	SM 4500- 0/405.1	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Temperature	SM 2550	14	NA	AN	NA	NA	14	flow-through cell	NA	none	immediately
Oxidation/Reduction	SM 2580 B	14	AN	NA	NA	NA	14	flow-through cell	AN	none	immediately
Specific Conductivity	SM 2510 B	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Turbidity ⁽⁴⁾	SM 2130 B	14	NA	NA	NA	NA	14	flow-through cell or hand- held turbidity meter	NA	none	immediately

Notes: ⁽¹⁾ USEPA Appendix III Parameters (boron, calcium, chloride, fluoride, pH, sulfate, total dissolved solids (TDS))

⁽²⁾ USEPA Appendix IV Parameters

(antimony, arsenic, barium, beryllium, cadmium, chromium, cobalt, fluoride, lead, lithium, mercury, molybdenum, selenium, thallium, radium 226 and 228 combined)

(3) Other Metals = aluminum, antimony, arsenic, barium, beryllium, cadmium, chromium, cobalt, copper, iron, lead, molybdenum, nickel, silver, selenium, thallium, vanadium, zinc

(4) If turbidity exceeds 10 NTUs, a duplicate sample filtered through a .45 micron filter may be collected for metals analysis in addition to the unfiltered sample. Both samples would be submitted for analysis. NA = not applicable HNO₃ = nitric acid

°C = degrees Celsius

mL = milliliter

1. Field duplicates will be collected at a frequency of one per group of 10 or fewer investigative water sample. Field duplicates will not be collected for radium analysis.

2. Field blanks will be collected at the discretion of the project manager; Equipment blanks will be collected at a rate of 1 per sampling event if non-dedicated equipment is used.

3. Matrix Spike/Matrix Spike Duplicate (MS/MSD) samples will be collected at a frequency of one per group of 20 or fewer investigative water samples per CCR unit/multi-unit. Additional volume to be determined by laboratory.

Sample volume is estimated and will be determined by the laboratory.
 Analytical method numbers are from SW-846 unless otherwise indicated. Analytical methods may be updated with more recent versions as apropriate.



Table 5. Summary of Detection Limits for Proposed Monitoring Program Class I Groundwater StandardsGroundwater Monitoring PlanCoffeen Power Station - Ash Pond 2

Constituent	Unit	Analytical Methods ¹	USEPA MCL ² (ug/L)	IL Class I Std ⁷ (ug/L)	RL ⁴ (ug/L)	MDL ⁴ (ug/L)
Metals ⁸						
Aluminum (d)	µg/L	6020	200 ³	3500	10	0.85
Antimony	µg/L	6020	6	6	1	0.25
Arsenic	µg/L	6020	10	10	1	0.25
Barium	µg/L	6020	2000	2000	1	0.4
Beryllium	µg/L	6020	4	4	1	0.5
Boron	µg/L	6020	NS	2000	25	10
Boron(d)	µg/L	6020	NS	2000	2.3	10
Cadmium	µg/L	6020	5	5	1	0.25
Calcium	µg/L	6020	NS	NS	125	100
Chromium	µg/L	6020	100	100	1	0.3
Cobalt	µg/L	6020	NS	1000, 2	1	0.25
Copper (d)	µg/L	6020	NS	650, 200	3	0.025
Cyanide	µg/L	4500	200	200	5	0.85
Iron (d)	µg/L	6010	300 ³	5,000	10	0.88
Lead	µg/L	6020	NS	7.5	1	0.25
Lithium	µg/L	6020	NS	NS	1	0.5
Manganese (d)	µg/L	6020	50 ³	150	1	0.055
Mercury	µg/L	6020 or 7470A	2	2	0.2	0.051
Molybdenum	µg/L	6020	NS	NS	1	0.25
Nickel (d)	µg/L	6020	NS	100	5	0.075
Nitrate- N	µg/L	300	10000	10000	30	8
Selenium	µg/L	6020	50	50	1	0.9
Silver (d)	µg/L	6020	100 ³	50	55	0.028
Thallium	µg/L	6020	2	2	1	0.25
Vanadium (d)	µg/L	6020	NS	49, 0.49	5	0.27
Zinc (d)	µg/L	6020	5000 ³	5	6	0.495
Inorganics	States and	the state of the				
Fluoride	mg/L	9214	4	4	0.1	0.05
Chloride	mg/L	9251	250 ³	200	5	1
Sulfate	mg/L	9036	250 ³	400	10	5
Total Dissolved Solids	mg/L	SM 2540 C	500 ³	1200	20	10
Other			- I State		The Contraction	
Combined Radium 226/228	pCi/L	9315/9320 or EPA 903/904	5	20/20, 5	5	6
Field			120123			
pН	SU	SM 4500-H+ B	NS	6.5-9.0	NA	NA
Oxidation/Reduction Potential	mV	SM 2580 B	NS	NS	NA	NA
Dissolved Oxygen	mg/L	SM 4500-0/405.1	NS	NS	NA	NA
Temperature	°C	SM 2550	NS	NS	NA	NA
Specific Conductivity	µS/cm	SM 2510 B	NS	NS	NA	NA
Turbidity	NTU	SM 2130 B	NS	NS	NA	NA

Notes:

NS = No standard

RL = Reporting limit as established by the laboratory MDL = Method detection limit as established by the laboratory

SM = Standard Methods for the Examination of Water and Wastewater

ug/L = micrograms per liter

mg/L = milligrams per liter

pCi/L = picoCuries per liter

µS/cm = microSiemens per centimeter NTU = nephelometric turbidity unit

(d) = dissolved analysis

1. Analytical method numbers are from SW-846 unless otherwise indicated.

2. USEPA MCL = United States Environmental Protection Agency Maximum Contaminant Level.

3. USEPA SMCL = United States Environmental Protection Agency Secondary Maximum Contaminant Level.

4. Reporting limits and method detection limits will vary depending on the laboratory performing the work.

5. All radium results will be reported (values may be positive or negative) and will include uncertainty and the calculated MDC.

6. Laboratories calculate a minimum detectable concentration (MDC) based on the sample.

7. 35 IAC 620.410 standards are listed including proposed changes submitted to the IL Pollution Control Board



SMARTER SOLUTIONS

EXCEPTIONAL SERVICE

VALUE

GROUNDWATER MONITORING PLAN

Ash Pond 2 **Coffeen Power Station** Coffeen, Illinois

January 24, 2017



TECHNOLOGY

ENVIRONMENTAL CONSULTANTS



ENVIRONMENTAL CONSULTANTS

234 W. Florida Street, Fifth Floor Milwaukee, Wisconsin 53204 (P) 414.837.3607 (F) 414.837.3608

GROUNDWATER MONITORING PLAN

ASH POND 2 COFFEEN POWER STATION COFFEEN, ILLINOIS

Project No. 2380

Prepared For:

Illinois Power Generating Company Coffeen Power Station 134 Cips Lane Coffeen, IL 62107

Prepared By:

Natural Resource Technology, Inc. 234 W. Florida Street, Fifth Floor Milwaukee, Wisconsin 53204

January 24, 2017

hanth

Stuart J. Cravens, PG Principal Hydrogeologist



al R Kellen

Nathaniel R. Keller, PG Hydrogeologist

WWW.NATURALRT.COM

TABLE OF CONTENTS

1	INTRO	DDUCTION	1-1
	1.1	Overview	1-1
	1.2	Site Location and Background	1-1
	1.3	Conceptual Site Model	
2	GRO	JNDWATER MONITORING	2-1
	2.1	Existing Well Network and Monitoring Programs	2-1
		2.1.1 IEPA Monitoring	2-2
		2.1.2 40 CFR Part 257 Monitoring	2-2
	2.2	Proposed Monitoring Network	2-3
	2.3	Abandoned Wells	2-4
3	APPL	ICABLE GROUNDWATER QUALITY STANDARDS	3-1
	3.1	Groundwater Classification	3-1
	3.2	Applicable Groundwater Quality Standards	3-1
	3.3	Statistical Evaluation of Background Groundwater Data	
	3.4	Statistical Analysis Results	
	3.5	Proposed Exceptions to the Groundwater Monitoring Parameters	
4	GRO	JNDWATER MONITORING PLAN	
	4.1	Groundwater Monitoring	4-1
	4.2	Sampling Schedule	
	4.3	Groundwater Sample Collection	
	4.4	Laboratory Analysis	
	4.5	Quality Assurance Program	
	4.6	Groundwater Monitoring System Maintenance Plan	
	4.7	Annual Statistical Analysis	
	4.8	Data Reporting	
	4.9	Compliance with Applicable On-Site Groundwater Quality Standards	
	4.10	Corrective Action	
5	REFE	RENCES	5-1



FIGURES

- Figure 1 Site Location Map
- Figure 2 Monitoring Well Location Map
- Figure 3 Potentiometric Contour Map May 2016
- Figure 4Potentiometric Contour Map November 2016
- Figure 5 Proposed Monitoring Well Network and Abandonments

TABLES

Table 1	Summary of Existing Monitoring Well Network
Table 2	Proposed Well Network and Analysis
Table 3	Background Groundwater Quality and Applicable Groundwater Quality Standards
Table 4	Sampling and Analysis Summary
Table 5	Summary of Detection Limits for Proposed Monitoring Program Class I Groundwater Standards

APPENDICES

- Appendix A Monitoring Well Network Boring Logs and Monitoring Well Construction Reports
- Appendix B Statistical Procedure for Background
- Appendix C Groundwater Monitoring Data
- Appendix D Groundwater Sampling Protocol



1 INTRODUCTION

1.1 Overview

This Groundwater Monitoring Plan was prepared by Natural Resource Technology, Inc. (NRT) in support of a Closure Plan for Ash Pond 2 located at the Coffeen Power Station (CPS) which is owned by Illinois Power Generating Company (IPGC).. This plan and the Closure Plan will apply specifically Ash Pond 2, one of the five Coal Combustion Residuals (CCR) units associated with the CPS.

This plan describes the groundwater monitoring and reporting to be completed in support of the Closure Plan for the Ash Pond 2. In addition to this groundwater monitoring plan, a Groundwater Management Zone Application is being submitted under separate cover.

1.2 Site Location and Background

Ash Pond 2 is located approximately 2 miles south of the City of Coffeen in Montgomery County, Illinois (Figure 1). The power plant and the CCR Units are situated on a peninsula between two lobes of Coffeen Lake which was created in 1963 by damming a portion of the East Fork of Shoal Creek (IDNR, 2014). The lake covers approximately 1,100 acres and provides cooling water for the CPS.

Ash Pond 2 is located within Section 11 Township 7 North and Range 7 East. The city of Coffeen is approximately 2 miles north of the CPS and the city of Hillsboro, IL is about 8 miles to the northwest. The CPS is located in an agricultural area. Historically, however, several coal mines were operated at depth in the vicinity of the site as well as a US Minerals processing facility located to the north. The CPS property is bordered by Coffeen Lake on the west, east, and south, and by agricultural land to the north.

Ash Pond 2 was first investigated in 2010, as requested by the Illinois Environmental Protection Agency (IEPA). Results of the investigation (NRT, 2013) indicated impacts to groundwater in the vicinity of Ash Pond 1 and 2, including exceedances of Class I Groundwater Standards for boron, sulfate, manganese, sulfate and total dissolved solids. Additional wells were installed in 2015 to comply with the Federal CCR Rule (40 CFR Part 257), and define the extent of exceedances associated with Ash Pond 2. Based on the groundwater results of wells installed in 2015, Hanson submitted a Corrective Action Plan (CAP, Hanson, 2016) to define the proposed remedy. Ash Pond 2 will be closed by leaving CCR in place using an alternative geomembrane cover system, following partial dewatering of the pond. This design will control the potential for slope failure and water infiltration into the closed CCR unit and will allow for drainage of surface water off of the cover system (AECOM, 2016).



IEPA responded to Dynegy regarding the CAP with a draft letter including comments. Dynegy and IEPA met on July 20, 2016 to discuss the CAP and proposed remedy. Following the discussion, IEPA in a letter dated August 9, 2016 provided the following comments (summarized):

- 1. Investigation may be required to define the source and extent of exceedances from Ash Pond 2.
- 2. A vertical component is required for the GMZ.
- 3. The GMZ contains portions of CCR units not proposed to close; GMZ must be revised to include only areas where CAP for Ash Pond 2 will mitigate impacts.
- 4. Hydraulic conductivity of foundation layer needs to be evaluated, to determine if CAP is appropriate.
- 5. A groundwater monitoring plan, specific to Ash Pond 2 must be submitted with CAP.
- 6. Modeling is required to demonstrate corrective action will be successful, and at what point in the future.
- 7. Evaluate the impacts of the corrective action on Coffeen Lake, with respect to applicable surface water standards.

This GW monitoring plan is provided as requested in Comment #5. The Closure Plan, to which this Groundwater Monitoring Plan, as well as the Hydrogeologic Characterization Report, Groundwater Model Report, and Hydrostatic Model Report are attached, provides the additional information necessary to address these comments and justify the selected corrective action.

1.3 Conceptual Site Model

The geology and hydrogeology described above is summarized and grouped into the following hydrostratigraphic units for the remaining discussion in this report:

- Ash Unit Saturated CCR within the various CCR Units
- Upper Confining Unit Low permeability clays and silts, including the Loess Unit and the upper clayey till portion of the Hagarstown Member
- Uppermost Aquifer (Groundwater Monitoring Zone) Thin (generally less than 3 feet), moderate permeability sand, silty sand, and sandy silt/clay units which include the Hagarstown Member and the upper Vandalia Member (where weathered)
- Lower Confining Unit Thick (generally greater that 15 feet), very low permeability sandy silt till or clay till that includes the unweathered lower Vandalia Member, Mulberry Grove Member (discontinous), and Smithboro Member
- Bedrock Pennsylvanian-age Bond Formation characterized by limestone and calcareous clays and shales. Bedrock was not encountered in any borings advanced onsite

Mounding of water within saturated ash in the impoundment creates a component of radial flow. The extent of this groundwater movement appears to be limited, as the elevated heads overlying the Upper



Confining Unit dissipate across the Ash Pond 2 berms. Potentially impacted water from the seeps observed along the berms may partially infiltrate through the Upper Confining Unit and/or run off toward the Lake or Unnamed Creek.

The Uppermost Aquifer underlying Ash Pond 2 consists of the Hagarstown Member and the weathered (upper) portions of the Vandalia Member, which is being monitored to define the extent of CCR constituents derived from Ash Pond 2. The Uppermost Aquifer is confined except where the Hagarstown Member is exposed along the eastern side of the impoundment within the former ravine (Figure 4). Based on hydraulic conductivity values (10⁻³ to 10⁻⁴ cm/sec) measured in the monitoring wells screened in the Hagarstown Member, groundwater at CPS has previously been classified as Class I in accordance with 35 IAC 620 (Hanson, 2009).

CCR within Ash Pond 2 is underlain by the Upper Confining Unit beneath the majority of the impoundment footprint. However, in former drainage features present prior to construction the saturated ash is in contact with the Hagarstown Beds and underlain by the Vandalia Member. Given the relatively high permeability of the Hagarstown Beds, leachate from Ash Pond 2 infiltrates into this unit, migrates through/under the eastern berm (where the Hagarstown Member is continuous below the berm) and discharges along the slope, as evidenced by the observation of seeps (Figure 4).

Groundwater within the Hagarstown Beds beyond the boundary of Ash Pond 2 flows predominantly to the east and south. Both the southern discharge flume and the Unnamed Creek intersect and cutoff the Hagarstown unit, eliminating further migration of potentially impacted groundwater. Impacted groundwater may also migrate to the north and northwest in the Hagarstown, potentially under the influence of the passive (gravity drain) underdrain system associated with the Recycle Pond and active underdrain system associated with the Landfill.



2 GROUNDWATER MONITORING

A groundwater investigation around Ash Pond 2 was initially completed in 2010-2012, A monitoring program was initiated in November 2015 to comply with the requirements of 40 CFR Part 257 (CCR Rule). Upon approval of the Closure Plan in which this document is included, the monitoring network will consist of the current CCR monitoring program and additional wells associated with other CPS units to comply with both IEPA and CCR Rule requirements. The proposed groundwater monitoring well network consists of a sufficient number of wells, installed at appropriate locations and depths, to monitor post-closure compliance with groundwater quality standards for Class I: Potable Resource Groundwater.

The proposed groundwater monitoring program included in this document is consistent with the requirements of 35 IAC Part 620 with Illinois EPA-approved modifications as well as 40 CFR Part 257. The wells will monitor the Hagarstown unit in the vicinity of Ash Pond 2 which has been designated the uppermost aquifer and has been monitored for the IEPA approved programs associated with the Landfill, Gypsum Stack Pond and Gypsum Recycle Pond.

The monitoring wells are designed and constructed in accordance with applicable standards, including the following:

- All monitoring wells are cased in a manner that maintains the integrity of the boreholes
- Wells are screened to allow sampling only at the specified interval
- All wells are covered with vented caps, unless located in flood-prone areas, and equipped with devices to protect against tampering and damage

The monitoring well network described below fulfills the following goals:

- Enable the collection of groundwater samples that represent the quality of background water that has not been affected by the Ash Pond 2
- Enable the collection of groundwater samples that represent the quality of downgradient groundwater
- Include wells that are located within the stratigraphic unit that may serve as potential contaminant migration pathways

2.1 Existing Well Network and Monitoring Programs

The existing monitoring wells present around the five units is summarized in Table 1 and shown on Figure 2. This subsection discusses the monitoring wells associated with Ash Pond 2, and Subsections 2.1.1 and 2.1.2 describe the monitoring performed around Ash Pond 2. Monitoring wells specifically associated with Ash Pond 2 include five leachate head wells (OW1 – OW-5), 7 monitoring wells



(G401 - G407), 8 piezometers (B-1(d), B-2(s), G45D, G46D, P009, P010, P012, and P014), and 2 temporary piezometers (T408 and T409). Monitoring wells around Ash Pond 2 were installed beginning in 2010 (G402), and the remaining wells and piezometers were installed in 2015 and 2016. The recent addition of wells and the piezometers, one of which is located within the ash impoundment, was completed in 2015 and 2016 to comply with the CCR Rule and define the vertical extent of CCR impacts. Monitoring wells around other CCR units were installed between 2008 and 2011 to monitor groundwater around the Landfill, Gypsum Stack Pond, and Recycle Pond as required by the IEPA.

2.1.1 IEPA Monitoring

Currently, no IEPA required groundwater monitoring is performed around Ash Pond 2. Following closure of Ash Pond 2 groundwater monitoring will be initiated to meet requirements of 35 IAC Part 620.410.

2.1.2 40 CFR Part 257 Monitoring

CCR Rule monitoring commenced in November 2015 and currently consists of quarterly groundwater elevation measurements and water quality samples collected at background monitoring wells G270 and G281, and downgradient wells G401, G402, G403, G404, and G405. The groundwater will be analyzed for Appendix III and Appendix IV parameters (see below) for eight quarterly sampling events. Piezometers (P009, P010, P012, and P014) are measured monthly for groundwater elevation only.

Appendix III Parameters (total, except TDS)						
Boron	Chloride		рН		Total Dissolved Solids (TDS)	
Calcium	Fluoride		Sulfate			
Appendix IV Parameters (total)						
Antimony	Cadmium		Lead		Selenium	
Arsenic	Chromium		Lithium		Thallium	
Barium	Cobalt		Mercury		Radium 226/228	
Beryllium	Fluoride		Molybdenum			
Field Parameters						
Dissolved Oxygen		Specific Conductivity		Temperature		Turbidity
Oxidation/Reduction Potential						

Following the completion of eight quarterly sampling events, the monitoring program will be modified so that the analytical parameters and sampling frequency are appropriate to the objectives and requirements of the CCR Rule.



2.2 Proposed Monitoring Network

This Closure Plan proposes a groundwater monitoring program to meet the requirements of 35 IAC Part 620.410. Wells to be sampled are summarized in Table 2 and below, and the proposed monitoring network is shown on Figure 5.

Boring logs and monitoring well construction reports for the groundwater monitoring system are provided in Appendix A. The well depths well screen intervals, depth to groundwater, and position relative to Ash Pond 2 are summarized below:

Well Number	Well Screen Interval (ft bgs)	Depth To Groundwater ¹ (ft bgs)	Well Position in Hagarstown Unit	Additional Units Monitored	
G154	14.3-18.8	11.67	Upgradient / Background	SW Pond	
G270	13.1-17.9	7.61		Gypsum Recycle Pond	
G279	22.4-26.8	25.11	Upgradient /		
G280	12.8-17.6	7.39	Background		
G281	15.5-20.2	6.35			
G401	14.4-18.8	17.73	Downgradient / Side gradient	None	
G402	10-20	9.13	Downgradient		
G403	13.1-17.8	4.67	Downgradient		
G404	6.4-11.2	5.09	Downgradient /		
G405	9.0-13.8	5.15	Side gradient		
G406	13.6-18.4	9.35	Downgradient		
G407	13.8-18.6	8.05			
G410 ²	TBD	TBD	Background /		
G411 ²	TBD	TBD	Domigradion		

¹Groundwater depth elevations shown are from November 2016

² Proposed wells, to be installed upon approval of Closure Plan

Additional sampling is completed, as required by IEPA, at monitoring wells surrounding the Landfill, Gypsum Stack Pond, and Gypsum Recycling Pond. These sampling programs are not included or discussed in this Plan. However, groundwater elevations from these monitoring wells will be used for data analysis when they are collected in conjunction with the Ash Pond 2 monitoring.

Wells G410 and G411 are proposed to be installed, upon approval of the Closure Plan, to define the extent of sulfate exceedances above Class I standards in the Hagarstown Unit to the west of G407. The sulfate exceedances at well G407 have not been attiributed to Ash Pond 2 and will be further evaluated following completion of proposed wells G410 and G411 to the west.



2.3 Abandoned Wells

Monitoring wells and piezometers not included in the proposed network will be abandoned upon approval of this Groundwater Monitoring Plan. Piezometers G45D, north of Ash Pond 2, and G46D located on the southwest corner are no longer required because no CCR constituents were detected at concentrations of concern at these locations. Temporary wells T408 (nested with G45D), and T409 (nested with G46D), will also be abandoned because they were installed only to obtain horizontal hydraulic conductivities in the Vandalia Till Unit.

The leachate wells OW-1 through OW-5 and piezometers (B-1(d), B-2(s), P009, P010, P012, and P014) will be abandoned prior to capping of Ash Pond 2. The locations of the leachate well, monitoring wells, and piezometers that will be abandoned are shown on Figure 5.



3 APPLICABLE GROUNDWATER QUALITY STANDARDS

3.1 Groundwater Classification

Based on the detailed geologic information provided for the Hagarstown Unit at the Site, groundwater can be classified as Class I - Potable Resource Groundwater. Although the thickness of the sand and gravel is generally less than 5 feet, the field hydraulic conductivity tests performed on the Hagarstown had geometric mean hydraulic conductivities of approximately 2.9×10^{-4} cm/s, and in some locations it was measured as high as 10^{-3} cm/sec. Sands and gravels with a hydraulic conductivity of greater than 1×10^{-4} cm/s meet the provisions of Section 620.210 for Class I Potable Resource Groundwater.

3.2 Applicable Groundwater Quality Standards

The groundwater quality standards for the proposed monitoring well network screened in the Hagarstown Unit are Class I Potable Resource Groundwater [35 IAC 620.410] standards or background concentrations, based on statistical analyses (Section 3.3). For determination of background concentrations, only parameters included in the proposed analyte list (Table 2) and from wells within the proposed network were used in the evaluation. Background concentrations were calculated for arsenic, boron, manganese, pH, sulfate and TDS from background wells G270, G280, and G281.

Based on the statistical evaluation of the background groundwater data (Table 3), background concentrations exceed groundwater quality standards for Class I Potable Resource Groundwater for manganese. Therefore, the background groundwater concentration for manganese will apply to the proposed monitoring well network. Since background concentrations are lower than the Class I Potable Resource Groundwater standard for arsenic, boron, pH, sulfate and TDS, the Class I Potable Resource Groundwater will be applied.

Background groundwater quality for additional parameters subject to the USEPA CCR Rule (i.e., Appendix III and IV) will be established through statistical evaluation following completion of 8 quarters of groundwater sampling of background wells that commenced in November 2015. The groundwater quality standard for these parameters at the proposed monitoring well network for Ash Pond 2 will be the greater of either the background concentration or the groundwater quality standard for Class I Potable Resource Groundwater. Parameters with no Class I standard will not be evaluated for the IEPA monitoring program but will be evaluated separately for the USEPA CCR Rule with the methods outlined in 40 CFR Part 257. The current list of applicable groundwater quality standards for the monitoring well network is shown on Table 3.
3.3 Statistical Evaluation of Background Groundwater Data

A statistical evaluation was performed to determine the maximum background concentrations likely to occur upgradient of Ash Pond 2 at the CPS. The groundwater quality data collected from upgradient monitoring wells G270, G280, and G281 was evaluated using the Electric Power Research Institute (EPRI, March 2014) computer database and analysis program, MANAGES[™] (Version 3.4.49). The statistical analysis procedures used here are consistent with procedures described in the document: 2009 Unified Guidance. "Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities—Unified Guidance," March 2009, EPA 530/R-09-2007 (USEPA, 2009).

The statistical methodology is provided in Appendix B. Establishing the tolerance interval(s) for the groundwater constituents was accomplished by using either a parametric or non-parametric procedure, based on the percentage of non-detects in the data sets and the distribution of the sample population. If the statistical data for a constituent had less than 50 percent non-detects and was normally or log-normally distributed, a parametric procedure was used. If the data was not normally or log-normally distributed or had more than 50 percent non-detects, a non-parametric procedure was used. Appendix B, Figure B-1 is a flow chart which illustrates the processes followed to determine the appropriate statistical procedure used for each constituent, based on its statistical characteristics.

3.4 Statistical Analysis Results

The results of the statistical analyses for the groundwater in Hagarstown Unit are located in Appendix B. Appendix B includes a statistical summary of the background water quality data from Wells G270, G280, and G281 and the statistical analysis procedure inputs and results.

Calculated background values (upper and lower limits) for the tested inorganic constituents and pH are listed in Appendix B, Table B-1 along with the percent non-detects, normal or lognormal distribution, test method, and confidence level. The calculated background values are also shown on Table 3 and are compared to the groundwater quality standards for Class I: Potable Resource Groundwater. The higher of the two values is shown as the Applicable Groundwater Standard on Table 3 (see additional discussion provided in Section 3.2).

3.5 Proposed Exceptions to the Groundwater Monitoring Parameters

Based on the results of groundwater monitoring performed at CPS to date, the following exceptions to the above applicable Class I: Potable Resource Groundwater standards are proposed:

Analytical results (Appendix C) indicate iron has historically exceeded the groundwater quality standards of Class I Potable Resource Groundwater inorganic constituents listed in 35 IAC 620.410(a)(1). However, elevated iron concentrations are attributed to reducing



conditions and aquifer composition and are not prevalent in groundwater associated with Ash Pond 2.

- Nitrate is included in the inorganic parameters for Class I Potable Resource Groundwater. Historical monitoring has not exhibited nitrate exceedances and nitrate is not prevalent in groundwater associated with Ash Pond 2.
- Class I Potable Resource Groundwater requires monitoring the inorganic constituents copper, cyanide, nickel, silver, vanadium, and zinc listed in 35 IAC 620.410(a)(1). These constituents will not be monitored because they may be sensitive to the aquifer reducing conditions and are not associated with the chemical characteristics of CCR at Ash Pond 2.
- Perchlorate is commonly used as an oxidizer in solid propellants, munitions, fireworks, airbag initiators for vehicles, matches and signal flares. It is also used in some electroplating operations and found in some disinfectants and herbicides (USEPA, 2014). Perchlorate is an inorganic constituent listed in 35 IAC 620.410(a)(1) but has not been previously analyzed. Perchlorate will not be monitored because it is not associated with the chemical characteristics of CCR at Ash Pond 2.

The proposed groundwater monitoring parameters for the proposed monitoring well network are discussed further in Section 4.1.



4 GROUNDWATER MONITORING PLAN

The groundwater monitoring plan will monitor and evaluate groundwater quality to demonstrate compliance with the groundwater quality standards for Class I: Potable Resource Groundwater as well as USEPA MCLs or background exceedances, as appropriate. As discussed in Section 3, the proposed post-closure groundwater sampling network consists of three background monitoring wells and 11 compliance monitoring wells as shown on Figure 5.

4.1 Groundwater Monitoring

The proposed IEPA (35 IAC Part 620.410) well network for Ash Pond 2 consists of 11 monitoring wells (Table 2) installed in the Hagarstown Beds adjacent to Ash Pond 2, the Gypsum Recycle Pond and the SW Pond and 3 background monitoring wells (G270, G280, and G281). The wells included in the IEPA Closure Monitoring will be analyzed for the following laboratory and field parameters:

Laboratory Parameters									
Metals (dissolved)									
Arsenic	Lead		Manganese						
Boron									
Inorganics (totals)									
Sulfate	Total Disso	olved Solids							
Field Parameters									
рН		Temperature							
Oxidation/Reduction	Potential	Specific Conductivity							
Dissolved Oxygen		Turbidity							

In addition to the IEPA Closure Monitoring, wells G270, G279, G280, G281, and G401 – G405 will be sampled and analyzed for additional parameters to comply with the CCR Rule. These wells will be monitored for the following laboratory and field parameters shown below:

Laboratory Parameters								
Metals (totals)								
Antimony	Cadmium	Lithium						
Arsenic	Calcium	Mercury						
Barium	Chromium	Molybdenum						
Beryllium	Cobalt	Selenium						
Boron	Lead	Thallium						

Inorganics (totals)								
Sulfate								
Total Disso	Total Dissolved Solids							
Other (total)								
Radium 226 and 228 combined								
	Temperature							
Potential	Specific Conductivity							
	Turbidity							
	Sulfate Total Disso 3 combined							

As discussed in Section 3.5, other constituents listed under 35 IAC 620 will not be monitored at the proposed monitoring well network because the groundwater monitoring results to date indicate that the inorganic constituents copper, cyanide, nitrate, nickel, silver, and zinc either meet the Class I: Potable Resource Groundwater standards or are not associated with the chemical characteristics of CCR at Ash Pond 2. Iron will not be sampled because it is indicative of reducing conditions in the aquifer and is not prevalent in groundwater associated with Ash Pond 2. Following the initial eight rounds of sampling at the CCR Rule monitoring wells, the parameters to be monitored will be in accordance with the requirements of 40 CFR Part 257.94 and 257.95 and this plan.

4.2 Sampling Schedule

Groundwater sampling for the proposed monitoring well network will initially be performed quarterly according to the following schedule:

Frequency	Duration						
Quarterly	Begins: upon approval of this plan, for wells and parameters within the IEPA Closure Monitoring Program only. Wells within the CCR monitoring program will be monitored in accordance with 40 CFR 257.						
	Ends: 5 years after completion of cap and upon demonstration that monitoring effectiveness is not compromised and that there are no increasing trends attributable to Ash Pond 2						
Semiannual	Begins: after IEPA approves that quarterly monitoring requirements have been satisfied.						
	Ends: 5 years after initiation of semiannual monitoring and upon demonstration that monitoring effectiveness is not compromised and that there are no increasing trends attributable to Ash Pond 2.						
Annual	Begins: after IEPA approves that semiannual monitoring requirements have been satisfied.						
	Ends: upon IEPA approval of a certified post-closure care report, but no less than 30 years from the date of closure as specified in 40 CFR 257						

Five years after approval of the Closure Plan, a request may be made to modify the post-closure care plan to reduce the frequency of groundwater monitoring to semi-annual sampling by demonstrating all of the following:

- Monitoring effectiveness will not be compromised by the reduced frequency of monitoring
- Sufficient data has been collected to characterize groundwater
- Concentrations of constituents monitored at the downgradient boundaries show no statistically significant increasing trends that can be attributed to the former ash ponds

If concentrations of parameters of concern at the downgradient wells of the site show no statistically significant increasing trends that can be attributed to Ash Pond 2 for the five years after reducing the monitoring frequency to semi-annual, a request may be made to modify the post-closure care plan to reduce monitoring frequency to annual sampling by demonstrating the same items above as for the reduction to semi-annual monitoring.

Groundwater monitoring may be discontinued upon Illinois EPA's approval of a certified post-closure care report after a minimum 30 years of post-closure groundwater monitoring has been completed. Specifically, when no statistically significant increase is detected in the concentration of any constituent above that measured and recorded during the immediately preceding scheduled sampling for four consecutive years after changing to an annual monitoring frequency.

Groundwater monitoring for the 40 CFR Part 257 well network will follow a schedule in accordance with the requirements of 40 CFR Part 257.94 and 257.95. Post-closure care groundwater monitoring will continue for a minimum of 30 years in accordance with 40 CFR Part 257.104

4.3 Groundwater Sample Collection

Groundwater samples will be collected consistent with the requirements of 35 IAC Part 620 and 40 CFR 257.93 as described in Appendix D. In addition to groundwater well samples, quality assurance samples will be collected as described in Section 4.5 (Table 4).

4.4 Laboratory Analysis

Laboratory analysis will be performed consistent with the requirements of 35 IAC Part 620 and 40 CFR 257.93 by a state-certified laboratory using methods approved by Illinois EPA and USEPA (Table 4). The practical quantitation limit (PQL) for all parameters analyzed will be lower than the applicable groundwater quality standard. Concentrations lower than the PQL will be reported as less than the PQL. A list of these parameters and the required PQLs are summarized in Table 5.

4.5 Quality Assurance Program

Consistent with the requirements of 35 IAC Part 620 and 40 CFR 257.93, the sampling and analysis program includes procedures and techniques for quality assurance and quality control (Table 4). Additional quality assurance samples to be collected will include the following:

- Two blind duplicate groundwater samples from randomly selected monitoring wells
- One equipment blank sample will be collected and analyzed for each day of sampling. If dedicated sampling equipment is used, than equipment blank samples will not be collected.

The duplicate and equipment blank quality assurance samples will be supplemented by the laboratory QA/QC program, which typically includes:

- Regular generation of instrument calibration curves to assure instrument reliability
- Laboratory control samples and/or quality control check standards that have been spiked, and analyses to monitor the performance of the analytical method
- Matrix spike/matrix spike duplicate analyses to determine percent recoveries and relative percent differences for each of the parameters detected
- Analysis of replicate samples to check the precision of the instrumentation and/or methodology employed for all analytical methods
- Analysis of method blanks to assure that the system is free of contamination

4.6 Groundwater Monitoring System Maintenance Plan

Consistent with the requirements of 35 IAC Part 620 and 40 CFR 257.91, maintenance will be performed as needed to assure that the monitoring wells provide representative groundwater samples. Monitoring wells will be inspected during each groundwater sampling event. Monitoring well inspections will consist of the following:

- Visual inspection, clearing of vegetation, replacement of markers, and painting of protective casings as needed to assure that monitoring wells are clearly marked and accessible
- Visual inspection and repair or replacement of well aprons as needed to assure that they are intact, drain water away from the well, and have not heaved
- Visual inspection and repair or replacement of protective casings as needed to assure that they are undamaged, and that locks are present and functional
- Checks to assure that well caps are intact and vented, unless in flood-prone areas in which case caps will not be vented





- Annual measurement of monitoring well depths to determine the degree of siltation within the wells. Wells will be redeveloped as needed to remove siltation from the screened interval if it impedes flow of water into the well
- Checks that wells are clear of internal obstructions, and flow freely

If maintenance of a monitoring well cannot address an identified deficiency, a replacement well will be installed.

4.7 Annual Statistical Analysis

Trend analysis will be performed annually for each of the monitored parameters. Sen's Estimate of Slope will be applied to a minimum of four consecutive quarterly monitoring results. If there are increasing trends during closure and post-closure care periods, they will be further investigated as described below.

- If the results of sampling and analysis show an increasing trend at any compliance monitoring well, a Mann-Kendall analysis will be performed at 95 percent confidence to determine whether or not the increasing trend is statistically significant.
- If a statistically significant increasing trend occurs during post-closure care, further investigation of monitored concentrations will be performed as well as more frequent inspections of the surface of the cover system.
- If the investigation attributes a statistically significant increasing trend to a source other than Ash Pond 2, then the Illinois EPA will be notified in writing, stating the cause of the increasing trend and providing the rationale used in such a determination.
- If there is not an alternative source causing the statistically significant increasing concentration and the sampling frequency had been reduced to semi-annual or annual sampling, a quarterly sampling schedule will be reestablished. The frequency of sampling will return to either semi-annual or annual, once four consecutive quarterly samples show no statistically significant increasing trend.

Notifications concerning statistically significant increasing trends and revisions of the sampling frequency will be reported to Illinois EPA in writing within 30 days after making the determinations.

In addition as required in 40 CFR Part 257.93, statistical analysis will be performed to determine whether or not a statistically significant increase over a background value has occurred for each constituent and at each well. Appropriate statistical methods will be chosen from the list of methods provided and the test chosen will be conducted separately for each constituent in each monitoring well. In addition, each statistical method chosen will comply with the performance standards, as appropriate, based on the test method used. If a statistically significant increase over background values is determined, procedures from 40 CFR Part 257 will be followed including 1) establishing an assessment monitoring program or 2) demonstrating that a source other than Ash Pond 2 caused the increase or demonstrating another plausible reason for the increase (error in sampling, etc.).

4.8 Data Reporting

Sampling and analysis data from quarterly, semi-annual and/or annual groundwater monitoring for the monitoring well network will be reported to Illinois EPA within 60 days after completion of sampling. Statistical analysis of the laboratory analytical data will be reported to Illinois EPA with the annual report for the facility, as described in the closure plan.

Data reporting for the 40 CFR Part 257 monitoring well network will be consistent with recordkeeping, notification, and internet posting requirements described in 40 CFR 257.105 through 257.107.

4.9 Compliance with Applicable On-Site Groundwater Quality Standards

In accordance with IAC 620 Section 620.240, the compliance boundary is a lateral distance of 25 feet outward from the outermost edge of Ash Pond 2 berms. Following completion of the corrective action, the groundwater standard at the compliance boundary will be in accordance with IAC 620 Section 450(a)(4) for groundwater quality restoration such that the standard for each released chemical constituent will be the higher of either the Class I groundwater standard or the concentration determined by groundwater monitoring at the compliance boundary.

Compliance with on-site groundwater quality standards, as measured at the proposed monitoring well network, will be achieved when there are no statistically significant increasing trends that are attributed to Ash Pond 2 for parameters detected at the compliance boundary after a minimum 30 years of post-closure groundwater monitoring has been completed. Evaluation of groundwater quality data under USEPA (2015) will be consistent with 40 CFR Part 257.93 and 257.94.

4.10 Corrective Action

If a statistically significant increasing trend is observed to continue over a period of two or more years in groundwater sampled at the well network, and a subsequent hydrogeologic site investigation demonstrates that such exceedances are due to a release from Ash Pond 2 and corrective actions are appropriate to mitigate such releases, a corrective action plan will be proposed as a modification to the post-closure care plan. A corrective action plan will be submitted to Illinois EPA within 180 days after completion of the investigation activities. The plan will propose corrective actions to be undertaken to mitigate the impacts associated with the constituents of concern which exceed applicable groundwater standards.



5 REFERENCES

AECOM, 2016. Revised 30% Closure Design Package for Coffeen Power Station Ash Pond No. 2. April 16, 2016.

Natural Resource Technology, Inc. 2016. Hydrogeologic Site Characterization Report

USEPA, 2009. Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities – Unified Guidance. EPA 530/R-09/007, U.S. Environmental Protection Agency, Office of Resource Conservation and Recovery, Washington, D.C., 554 p. + 4 app.

USEPA, January 2014. Technical Fact Sheet - Perchlorate

USEPA, April 17, 2015. 40 CFR Parts 257 and 261. Hazardous and Solid Waste Management System; Disposal of Coal Combustion Residuals from Electric Utilities; Final Rule



FIGURES







Time: 12/22/2016 3:14:57





TABLES

Table 1Summary of Existing Monitoring Well NetworksGroundwater Monitoring Plan - Ash Pond 2Coffeen Energy Center - Ash Pond 2

	Surface TOC Top of B		Bottom of				
			Elevation (ft	Elevation (ft	Screen	Screen	
Well ID	Easting	Northing	MSL)	MSL)	Elevation	Elevation	Interpreted Screened Unit
G301	2,515,582.97	872,234.82	620.27	622.65	609.0	604.3	Upper Vandalia Till
G302	2,516,214.19	872,252.95	617.95	620.04	604.7	600.1	Upper Vandalia Till
	0.540.000.05	074 000 44	610.10				
G303	2,516,639.65	871,382.14	619.10	622.02	609.1	599.1	Hagarstown/Vandalia Till Contact
G304	2,515,519.74	871,397.69	623.46	626.72	613.5	603.5	Hagarstown Beds
G305	2,515,199.36	871,156.33	622.54	625.55	609.1	604.3	Hagarstown Beds
G306	2,516,120.41	871,140.98	622.84	625.72	609.8	605.2	Hagarstown Beds
G307	2,515,553.26	871,398.55	622.08	624.47	609.1	604.3	Hagarstown Beds
G281	2,514,455.48	874,375.37	623.82	626.36	608.31	603.66	Hagarstown Beds
G401	2,515,614.84	872,510.57	623.03	625.57	608.67	604.24	Hagarstown Beds
G402	2,516,632.46	872,500.49	610.56	613.37	600.6	590.6	Upper Vandalia Till
G403	2,514,616.63	873,561.34	623.81	626.47	610.7	606.03	Hagarstown Beds
G404	2,516,397.82	873,999.91	613.10	615.67	606.68	601.93	Hagarstown Beds
G405	2,515,335.67	873,996.63	620.90	623.63	611.89	607.14	Hagarstown Beds
G406	2,514,702.38	872,521.34	621.86	621.86	608.3	603.49	Hagarstown Beds
G407	2,513,705.87	872,973.39	618.35	621.32	604.57	599.74	Hagarstown Beds
T408	2,515,314.91	873,999.36	621.09	624.08	600.43	595.6	Vandalia Till
T409	2,514,693.89	872,517.79	621.85	625.01	600.06	595.26	Vandalia Till (Sand Seam)
G45D	2,515,322.23	873,998.03	620.94	623.81	589.06	579.42	Smithboro Till
G46D	2,514,697.78	872,519.70	621.91	625.24	580.3	570.65	Smithboro Till
OW-1	2,514,754.56	872,875.48	639.80	641.14	634.74	619.74	Ash
OW-2	2,515,525.87	873,761.43	639.70	641.34	629.84	614.84	Ash
OW-3	2,515,824.43	873,147.40	640.40	641.78	631.28	616.28	Ash
OW-4	2,516,277.90	873,535.97	644.70	damaged	629.7	604.7	Ash
OW-5	2,516,466.21	872,770.83	637.70	638.8	627.7	602.7	Ash
B-2(s)	2,516,615.41	873,650.85		639.44			Fill
B-1(d)	2,516,615.78	873,647.93		638.7			Fill
G101	2.514.215.00	876.575.00	625.27	627.6	609.59	604.95	Hagarstown Beds
G102	2.514.537.00	876.554.00	625.70	630.96	613.68	608.92	Hagarstown Beds
G103	2.514.500.93	876.200.00	630.99	633.8	615.11	610.32	Hagarstown Beds
R104	2,514,503.40	875,857.80	629.03	632.84	614.44	609.71	Hagarstown Beds
G105	2.514.509.03	875.500.00	629.26	632.08	613.15	608.36	Hagarstown Beds
G106	2,514,513.08	875,150.00	628.39	631.15	614.02	609.43	Hagarstown Beds
G107	2.514.358.30	874,994.07	627.79	630.23	613.92	609.29	Hagarstown Beds
G108	2.514.248.46	874.948.69	627.50	630.22	610.68	606	Hagarstown Beds (Silt)
G109	2.514.137.80	874.970.13	627.20	629.76	611.81	607.27	Hagarstown Beds
G110	2.514.057.93	875.015.38	627.02	629.65	611.97	607.43	Hagarstown Beds
G111	2.513.981.89	875.058.47	627.24	629.9	612.63	608.09	Hagarstown Beds
G119	2.513.910.00	875.675.00	628.85	631.55	611.56	607.02	Hagarstown Beds
G120	2.513.905.81	875.845.00	629.30	631.87	614.2	609.68	Hagarstown Beds
G121	2.513.910.00	875.965.00	629.57	632.83	612.78	608.1	Hagarstown Beds
G122	2 513 905 00	876 080 00	629.86	632.69	613.35	608.81	Hagarstown Beds
G123	2 513 905 00	876 190 00	630 13	632.96	609 19	604 67	Hagarstown Beds
0120	2,010,000.00		000.10	002.00	000.10	001.07	Hagarstown (Silt)/Vandalia Till
G124	2,513,900.48	876,305.00	630.42	633.39	614.44	609.91	Contact
G125	2,513,900.00	876,410.00	630.68	633.51	613.65	609.12	Hagarstown Beds
G126	2,513,878.29	875,049.31	622.96	625.39	610.07	605.53	Hagarstown Beds (Sandy clay)
T127	2,513,915.00	875,360.00	628.07	630.96	610.54	606	Hagarstown Beds
T128	2,513,915.00	875,510.00	628.44	630.93	611.91	607.4	Hagarstown Beds
TA31	2,513,856.77	876,542.25	623.89	626.55	608.8	604.32	Hagarstown Beds
TA32	2,513,605.19	877,532.57	618.93	621.42	607.62	603.25	Hagarstown Beds
TA33	2,513,248.73	876,605.45	622.51	625.27	610.28	605.62	Hagarstown Beds
TA34	2,513,466.73	875,906.10	624.10	626.52	613.18	608.69	Hagarstown Beds



G151	2,513,805.90	875,023.70	622.82	625.93	607.48	602.98	Hagarstown Beds
G152	2,513,894.50	874,687.50	623.06	626.52	609.47	604.97	Hagarstown Beds
G153	2,513,532.70	874,532.70	623.30	626.35	607.4	602.96	Hagarstown Beds
G154	2,513,243.10	874,978.40	623.52	626.35	609.26	604.76	Hagarstown Beds
G155	2,513,501.80	875,127.70	622.89	625.86	607.8	603.31	Hagarstown Beds
G200	2.515.650.00	877.930.60	624.20	625.94	612.01	607.22	Hagarstown Beds
R201	2.514.842.00	877.925.30	624.02	626.34	611.75	607.36	Hagarstown Beds
T202	2.514.895.00	876.699.40	626.22	628.63	613.95	609.57	Hagarstown Beds
G205	2.515.914.90	875.550.20	622.15	624.45	612.11	607.62	Hagarstown Beds
G206	2.514.669.20	875.103.90	630.54	632.82	613.03	608.62	Hagarstown Beds
G207	2.514.837.90	875.166.40	630.61	633.21	612.37	607.84	Hagarstown Beds
G208	2.514.993.60	875.231.50	630.57	633.16	613.04	608.51	Hagarstown Beds
G209	2.515.149.60	875.298.20	630.57	632.91	612.83	608.29	Hagarstown Beds
G210	2.515.299.00	875.359.70	630.48	632.99	611.09	606.55	Hagarstown Beds
		075 404 50	000.04				Hagarstown (Sandy
G211	2,515,449.10	875,424.50	630.31	632.64	612.97	608.43	Clay)/Vandalia Till Contact
G212	2,515,583.00	875,486.50	630.59	632.89	613.85	609.3	Hagarstown Beds
G213	2,515,723.50	875,544.40	630.34	632.81	613.59	609.05	Hagarstown Beds
G214	2,515,960.80	875,668.00	630.39	632.85	612.64	608.25	Hagarstown Beds
G215	2,515,971.60	875,810.20	630.48	633.06	611.07	606.68	Hagarstown Beds
G216	2,515,968.50	875,976.10	630.28	632.76	610.24	605.86	Hagarstown Beds
G217	2,515,963.00	876,185.60	630.67	633.1	610.18	605.79	Hagarstown Beds
G218	2,515,962.20	876,380.90	630.64	633.11	610.31	605.87	Hagarstown Beds
MW02S	2,513,210.04	876,408.86	624.10	sealed	613.76	608.98	
MW04S	2,514,450.58	877,999.73	622.40	625.79	612.57	608.14	Hagarstown Beds
MW05S	2,513,285.49	878,175.59	622.60	625.8	609.94	605.19	Hagarstown Beds
MW06S	2,513,189.40	879,021.15	623.10	626.14	612.06	607.48	Hagarstown Beds
MW07S	2,514,397.54	879,181.12	624.50	627.54	614.59	610.71	Hagarstown Beds
MW08S	2,514,478.83	879,776.62	624.70	627.9	613.19	608.7	Hagarstown Beds
MW09S	2,515,666.24	879,684.90	624.60	627.46	613.39	608.98	Hagarstown Beds
MW10S	2,515,914.37	878,250.50	621.20	624.22	609.92	605.44	Hagarstown Beds
MW11S	2,515,971.16	876,749.44	622.00	625.08	613.11	608.37	Hagarstown Beds
MW12S	2,515,900.54	875,520.08	622.20	625.21	611.59	607.02	Hagarstown Beds
MW13S	2,513,925.29	874,695.66	622.70	625.89	611.27	606.47	Hagarstown Beds
MW14S	2,514,125.95	875,737.78	624.60	sealed	612.34	607.58	Hagarstown Beds
MW15S	2,515,076.27	875,971.13	623.80	sealed	609.39	604.64	Hagarstown Beds
MW16S	2,515,087.98	877,355.14	626.10	629.37	611.51	606.69	Hagarstown Beds
MW17S	2,515,084.76	878,658.54	627.10	630.47	613.08	603.54	Hagarstown Beds
MW18S	2,513,745.20	878,604.67	625.60	628.66	614.29	609.81	Hagarstown Beds
MW20S	2,515,876.50	874,228.00		622.84			
G270	2,514,996.84	874,801.92	622.92	625.92	609.79	605	Hagarstown Beds
G271	2,515,517.12	874,239.38	622.89	625.57	612.93	608.58	Hagarstown Beds
G272	2,515,744.99	874,234.83	620.72	623.81	611.61	606.74	Hagarstown Beds (Silt)
G273	2,515,975.49	874,235.24	620.17	623.02	611.09	605.61	Hagarstown Beds
G274	2,516,195.60	874,239.25	621.67	624.04	608.77	604	Hagarstown Beds
G275	2,516,375.86	874,298.94	616.14	618.26	607.92	603.52	Hagarstown Beds
G276	2,516,358.83	874,438.60	629.14	632.	606.73	601.92	Hagarstown Beds
G277	2,516,370.51	874,581.80	620.79	623.08	606.5	602.02	Hagarstown Beds
G278	2,516,200.66	874,875.37	628.85	631.19	609.92	605.15	Hagarstown Beds
G279	2,516,245.60	875,028.06	629.19	632.04	606.79	602.4	Hagarstown Beds
G280	2,515,679.48	875,045.11	622.95	625.85	610.16	605.32	Hagarstown Beds

Notes:

Ground surface elevations based on information included on boring logs when the well was installed, and top of PVC elevations based on survey completed in Fall 2015.



Table 2 Proposed Monitoring Well Network and Analyses Groundwater Monitoring Plan - Ash Pond 2 Coffeen Energy Center - Ash Pond 2

Boring/Well ID	Ground Surface at Time of Install	Measuring Point Elevation (2015)	Top of Screen Elevation	Bottom of Screen Elevation	Screen length	Screen Top	Screen Bottom	Proposed Analyses for IEPA Monitoring	Additional Monitoring Programs Performed at Well	Other Analyses (USEPA CCR Rule or IEPA)
G270	622.92	625.92	609.79	605	5	16.1	20.9		Ash Pond 2 & GRP - CCR	
G281	623.82	626.36	608.31	603.66	5	18.1	22.7			40 CER 257 - Annendix III
G401	623.03	625.57	608.67	604.24	4	16.9	21.3			and Appendix IV
G402	610.56	613.37	600.60	590.6	10	12.8	22.8		Ach Bond 2 CCP	Baramators Croundwater
G403	623.81	626.47	610.70	606.03	5	15.8	20.4		ASII FUIIU 2 CCK	Elevation
G404	613.10	615.67	606.68	601.93	5	9.0	13.7	Arsenic		
G405	620.90	623.63	611.89	607.14	5	11.7	16.5	(a), boron (a),		
G406	621.86	621.86	608.30	603.49	5	13.6	18.4	lead (d),		
G407	618.35	621.32	604.57	599.74	5	16.8	21.6	manganese (d),	Nana	Neze
G410 ¹	TBD	TBD	TBD	TBD	TBD	TBD	TBD	sunate, and total	None	None
G411 ¹	TBD	TBD	TBD	TBD	TBD	TBD	TBD	dissolved solids,		
G154	623.52	626.35	609.26	604.76	5	17.1	21.6	(2)	SW Pond - IEPA	IEPA Approved Parameters
G279	629.19	632.04	606.79	602.40	4	25.3	29.6		GPP - IEPA and CCP	40 CFR 257 - Appendix III and Appendix IV
G280	622.95	625.85	610.16	605.32	5	15.7	20.5			Parameters, Groundwater Elevation

Notes:

¹ Proposed wells to be installed upon approval of Closure Plan and GMZ application

² Field parameters include: pH, oxidation -reduction potential, specific conductance, temperature, and dissolved oxygen Groundwater quality analyses including methods and sampling details are included in Table 4.



Table 3

Background Groundwater Quality and Applicable Groundwater Quality Standards Groundwater Monitoring Plan - Ash Pond 2 Coffeen Energy Center

			Background	Applicable Groundwater		
Parameters (totals)	Sampling Program	IL Class I Std ¹ (mg/L)	Concentration ² for IEPA (mg/L)	Standard ³ for IEPA (mg/L)	Maximum ⁴ (mg/L)	Minimum ⁴ (mg/L)
Antimony	CCR	0.006	tbd	tbd	<0.003	< 0.003
Arsenic	CCR	0.01	tbd	tbd	0.014	<0.001
Arsenic (d)	IEPA	0.01	0.001	0.01	0.25	<0.002
Barium	CCR	2.0	tbd	tbd	0.24	0.014
Beryllium	CCR	0.004	tbd	tbd	0.0018	<0.001
Boron	CCR	2.0	tbd	tbd	17	<0.02
Boron (d)	IEPA	2.0	0.12	2.0	8.1	<0.01
Calcium	CCR	NS	tbd	tbd	450	<0.1
Cadmium	CCR	0.005	tbd	tbd	0.008	<0.001
Chloride	CCR	200	tbd	tbd	160	1.5
Chromium	CCR	0.1	tbd	tbd	0.034	<0.004
Cobalt	CCR	1.0	tbd	tbd	0.28	<0.001
Fluoride	CCR	4	tbd	tbd	1.06	0.031
Lead	CCR	0.0075	tbd	tbd	0.018	<0.001
Lead (d)	IEPA	0.0075	0.001	0.0075	0.220	<0.001
Lithium	CCR	NS	tbd	tbd	0.057	<0.01
Manganese(d)	IEPA	0.150	0.47	0.47	1.02	<0.018
Mercury	CCR	0.002	tbd	tbd	0.00093	<0.0002
Molybdenum	CCR	NS	tbd	tbd	0.043	<0.001
Selenium	CCR	0.05	tbd	tbd	0.027	<0.001
Sulfate	CCR, IEPA	400	370	400	2,500	2.3
Thallium	CCR	0.002	tbd	tbd	0.0013	<0.001
TDS (d)	CCR, IEPA	1,200	780	1,200	3,900	320
Field pH	CCR, IEPA	6.5 - 9.0	6.64-7.88	6.5-9.0	8.03	5.80
Radium 226/228 [*]	CCR	20/20	tbd	tbd	4.46	0.185

Notes:

All parameters are totals unless noted. Standards apply to dissolved or total concentrations

(d) Dissolved

tbd = To Be Determined for Illinois EPA monitoring program; CCR Appendix III and IV parameters based on future monitoring , started in November 2015

Bold = Background Concentration exceeds Class I Groundwater Standard

Red = Exceeds Applicable Groundwater Standard

NS = No Class II Groundwater Standard

na = not applicable; parameter [dissolved and total] not proposed for monitoring program USEPA (t) = background concentration for parameter [total] required under USEPA program (40 CFR Part 257)

¹ IPCB 620 Class I: Potable Resource Groundwater Standard

² Background Concentration obtained from Appendix B - Statistical Procedure for Calculation of Background

(Table B-1 Tolerance Limits for Background Monitoring Wells G270, G280, and G281 using the Upper and Lower Limits)

³ Applicable Groundwater Standard is the higher of the Background Concentration and the Class I Groundwater Standard (or the lower if compared to the pH lower limit)

⁴ Groundwater concentrations based on historical results for wells in the proposed sampling program

* Radium 226 and 228 reported separately for IPCB Class I Groundwater Standard, reported combined for USEPA 40 CFR Part 257

Table 4. Sampling and Analysis Summary Groundwater Monitoring Plan - Ash Pond 2 Coffeen Energy Center

Parameter	Analytical Method⁵	Number of Samples	Field Duplicates ¹	Field Blanks ²	Equipment Blanks ²	MS/MSD ³	Total	Container Type	Minimum Volume⁴	Preservation (Cool to 4 °C for all samples)	Sample Hold Time from Collection Date
Metals - Appendix III ⁽¹⁾											
Boron (total and dissolved)	6020	14	2	0	0	1	17	plastic	600 mL	HNO ₃ to pH<2	6 months
Calcium	6020	7	1	0	0	1	9	plastic	601 mL	HNO ₃ to pH<3	6 months
Metals - Appendix IV (2) and Additional Metals											
Other Metals (3)	6020	7	1	0	0	1	9	plastic	600 mL	HNO3 to pH<2	6 months
Arsenic (dissolved)	6020	14	2	0	0	1	17	plastic	600 mL	HNO3 to pH<2	6 months
Lead (dissolved)	6020	14	2	0	0	1	17	plastic	600 mL	HNO3 to pH<2	6 months
(discolved)	6020	14	2	0	0	1	17	plastic	600 mL	HNO3 to pH<2	6 months
Mercury	7470A or 6020	7	1	0	0	1	9	plastic	400 mL	HNO_3 to pH<2	28 days
Inorganic Parameters - Appe	ndix III ⁽¹⁾										
Fluoride	9214	7	1	0	0	1	9	plastic	300 mL	Cool to 4 °C	28 days
Chloride	9251	7	1	0	0	1	9	plastic	100 mL	Cool to 4 °C	28 days
Sulfate	9036	14	2	0	0	1	17	plastic	50 mL	Cool to 4 °C	28 days
Total Dissolved Solids	SM 2540 C	14	2	0	0	1	17	plastic	200 mL	Cool to 4 °C	7 days
Radium - Appendix IV ⁽²⁾											
Radium 226	9315 or EPA 903	7	0	0	0	1	8	plastic	1000 mL	HNO3 to pH<2	6 months
Radium 228	9320 or EPA 904	7	0	0	0	1	8	plastic	1000 mL	HNO_3 to pH<2	6 months
Field Parameters											
рН ⁽¹⁾	SM 4500- H+ B	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Dissolved Oxygen	SM 4500- O/405.1	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Temperature	SM 2550	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Oxidation/Reduction Potential	SM 258O B	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Specific Conductivity	SM 2510 B	14	NA	NA	NA	NA	14	flow-through cell	NA	none	immediately
Turbidity (4)	SM 2130 B	14	NA	NA	NA	NA	14	flow-through cell or hand- held turbidity meter	NA	none	immediately

Notes:

⁽¹⁾ USEPA Appendix III Parameters (boron, calcium, chloride, fluoride, pH, sulfate, total dissolved solids (TDS))

⁽²⁾ USEPA Appendix IV Parameters

(antimony, arsenic, barium, beryllium, cadmium, chromium, cobalt, fluoride, lead, lithium, mercury, molybdenum, selenium, thallium, radium 226 and 228 combined)

⁽³⁾ Other Metals = antimony, arsenic, barium, beryllium, cadmium, chromium, cobalt, lead, lithium, molybdenum, selenium, thallium

(4) If turbidity exceeds 10 NTUs, a duplicate sample filtered through a .45 micron filter may be collected for metals analysis in addition to the unfiltered sample. Both samples would be submitted for analysis. NA = not applicable

HNO₃ = nitric acid

°C = degrees Celsius

mL = milliliter

1. Field duplicates will be collected at a frequency of one per group of 10 or fewer investigative water sample. Field duplicates will not be collected for radium analysis.

2. Field blanks will be collected at the discretion of the project manager; Equipment blanks will be collected at a rate of 1 per sampling event if non-dedicated equipment is used.

3. Matrix Spike/Matrix Spike Duplicate (MS/MSD) samples will be collected at a frequency of one per group of 20 or fewer investigative water samples per CCR unit/multi-unit. Additional volume to be determined by laboratory.

4. Sample volume is estimated and will be determined by the laboratory.

5. Analytical method numbers are from SW-846 unless otherwise indicated. Analytical methods may be updated with more recent versions as apropriate.



Table 5

Summary of Detection Limits for Proposed Monitoring Program Class I Groundwater Standards Groundwater Monitoring Plan - Ash Pond 2

Coffeen Energy Center

Constituent	Unit	Analytical Methods ¹	USEPA MCL ²	IL Class I Std ¹ (ug/L)	RL ⁴	MDL ⁴
Metals						
Antimony	μg/L	6020	6	6	1	0.25
Arsenic	μg/L	6020	10	10	1	0.25
Arsenic(d)	μg/L	6020	10	10	1	0.13
Barium	μg/L	6020	2000	2000	1	0.4
Beryllium	µg/L	6020	4	4	1	0.5
Boron	μg/L	6020	NS	2000	25	10
Boron(d)	µg/L	6020	NS	2000	2.3	10
Cadmium	µg/L	6020	5	5	1	0.25
Calcium	µg/L	6020	NS	NS	125	100
Chromium	µg/L	6020	100	100	1	0.3
Cobalt	µg/L	6020	NS	1,000	1	0.25
Lead	µg/L	6020	15	7.5	1	0.25
Lead(d)	µg/L	6020	15	7.5	1	0.025
Lithium	µg/L	6020	NS	NS	1	0.5
Manganese (d)	µg/L	6020	50 ³	150	1	0.055
Mercury	µg/L	6020 or 7470A	2	2	0.2	0.051
Molybdenum	µg/L	6020	NS	NS	1	0.25
Selenium	µg/L	6020	50	50	1	0.9
Thallium	µg/L	6020	2	2	1	0.25
Inorganics						
Fluoride	mg/L	9214	4	4	0.1	0.05
Chloride	mg/L	9251	250 ³	200	5	1
Sulfate	mg/L	9036	250 ³	400	10	5
Total Dissolved Solids	mg/L	SM 2540 C	500 ³	1200	20	10
Other						
Combined Radium 226/228	pCi/L	9315/9320 or EPA 903/904	5	20/20	5	6
Field						
pH	SU	SM 4500-H+ B	NS	6.5-9.0	NA	NA
Oxidation/Reduction Potential	mV	SM 2580 B	NS	NS	NA	NA
Dissolved Oxygen	mg/L	SM 4500-O/405.1	NS	NS	NA	NA
Temperature	°C	SM 2550	NS	NS	NA	NA
Specific Conductivity	μS/cm	SM 2510 B	NS	NS	NA	NA
Turbidity	NTU	SM 2130 B	NS	NS	NA	NA

Notes:

NS = No standard

RL = Reporting limit as established by the laboratory

MDL = Method detection limit as established by the laboratory

SM = Standard Methods for the Examination of Water and Wastewater

ug/L = micrograms per liter

mg/L = milligrams per liter

pCi/L = picoCuries per liter

 μ S/cm = microSiemens per centimeter

NTU = nephelometric turbidity unit

(d) = dissolved analysis

1. Analytical method numbers are from SW-846 unless otherwise indicated.

2. USEPA MCL = United States Environmental Protection Agency Maximum Contaminant Level.

3. USEPA SMCL = United States Environmental Protection Agency Secondary Maximum Contaminant Level.

4. Reporting limits and method detection limits will vary depending on the laboratory performing the work.

5. All radium results will be reported (values may be positive or negative) and will include uncertainty and the calculated MDC.

6. Laboratories calculate a minimum detectable concentration (MDC) based on the sample.



611

APPENDIX A

MONITORING WELL NETWORK BORING LOGS AND MONITORING WELL CONSTRUCTION REPORTS

F	[EL]	D 1	BOR	I	NG	G L(DG		
CLIENT: AEG Coffeen Power Statior Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A						Statior Facility	1	CONTRACTOR: Testing Service Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 41/4" HSA w/SS samplers	BOREHOLE ID: G154 Well ID: G154 Surface Elev: 623 52 ft MSI
WE	DATE	S: S Fin R: Pt	tart: 12/ nish: 12/ ly. cloud	16/2 16/2 y (m	011 011 id-30)'s)		FIELD STAFF: Driller: B. Williamson Helper: R. McCuan Eng/Geo: R. Fiorito	Completion: 20.00 ft. BGS Station: 874,978.38N 2,513,243.10E
5	SAMPL	E	Т	EST	TINC	G C	TOPOGE	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	tal (i) <i>y</i>		u	(%)	lb/ft ³)	<i>p</i> (tsf be	Quadr Towns	angle: Cotteen, IL hip: East Fork	$\underline{\Psi} = 13.45$ - While drilling $\underline{\Psi} =$
ber	/ To		s/6i alue	ure (')en. ($^{\rm sf)}_{\rm C} Q$	Section	10, Tier 7N; Range 3W	$\overline{\Sigma} = 11.10 - 12/21/2011$
Numł	Recov % Re	Type	Blows N - V RQD	Moist	Dry I	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	14/24 58%	ss	3-4 4-5 N=8				2	FILL - Light yellowish brown (10YR6/4), silty CLAY trace sand and slight trace gravel.	with 622
2A	10/24 42%	ss	woh-1 4-5 N=5					Brownish yellow (10YR6/6) with 25% gray (10YR6/ mottles, silty CLAY with trace sand.	/1)620
3A	14/24 58%	ss	2-4 6-7 N=10					Gray (10YR5/1) with 25% yellowish brown (10YR5, mottles, silty CLAY with trace sand.	/4) 618
4A	21/24 88%	ss	6-8 10-10 N=18						616
5A	16/24 67%	ss	1-4 6-8 N=10				10	Gray (10YR6/1) with 25% brownish yellow (10YR6, mottles, silty CLAY with trace sand and slight trace gra	/6) ivel. 614
6A	13/24 54%	ss	<i>1-4</i> 6-8 N=10				⊻		612
7A	20/24	V ss	4-6 8-12					Gray (10YR6/1) with 10% brownish yellow (10YR6, mottles, silty CLAY with trace sand and slight trace gra	/8) ivel.
7B	83%	\wedge	N=14				⊻	Gray (10YR6/1), silty CLAY with trace sand and slig trace gravel.	sht 610
8A		M	1 18					Gray (10YR6/1), clayey SAND with trace silt.	
8B 8C	23/24 96%	ss	30-37 N=48				16 T	Gray (10YR6/1) with 50% yellowish brown (10YR5, mottles, silty SAND with slight trace gravel. Dark yellowish brown (10YR4/4), silty CLAY with sa and trace gravel.	/6) = 608 608
9A	21/24 88%	ss	40-75 86-84 N=161					Brown (10YR4/3), silty CLAY with sand and trace gra	ivel.
10A	20/24 83%	ss	28-28 30-34 N=58					Dark gray (10YR4/1), silty CLAY with sand and tra gravel.	ce 604
	ı 1		I	T	1	I	20 -	End of Boring = 20.0 ft. BGS	
NC	DTE(S):	G15	4 installe	ed in	bore	hole.			

FIELD BORING LOG CLIENT: AEG Coffeen Power Station Site: CCB Management Facility Location: Coffeen, Illinois Project: 05S3004A DATES: Start: 2/26/2008 Finish: 2/26/2008 WEATHER: Overcast, cold							n CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS & CME samplers FIELD STAFF: Driller: B. Williamson Helper: R. Keedy Eng/Geo: S. Simpson	BOREHOLE ID: G270 Well ID: G270 Surface Elev: 622.9 ft. MSL Completion: 18.3 ft. BGS Station: 874,801.92N 2,514,996.84E				
2	E E	E	1	ESI		r	TOPOGRAPHIC MAP INFORMATION: WAT Quadrangle: Coffeen, IL	FER LEVEL INFORMATION: $\mathbf{V} = 16.00$ - While drilling				
	Fotal (<i>ery</i>		5 in e	(%)	fl/fl	Type	Township: East Fork					
umber	cov/' Recov	be	<i>эws/(</i> - Valu 2D	oisture	y Den	ı (tsf) ülure 1	Depth Lithologic					
ź	Re %	Ĥ	R N	Ž	Ā	Q_{I}	ft. BGS Description	Detail ft. MSL Remarks				
	20/24 83%	ss	2-2 2-4 N=4				Dark grayish brown (10YR4/2), moist, firm, clayey SILT	622				
1A	(11-1	24			2 — Dark gravish brown (10YR4/2), moist, firm, silty CLAY Dark gravish brown (10YR4/2) with 5% vellowish brown					
2A	19/24	ss	3-4 5-9	22		2.33	(10YR5/8) mottles, moist, firm, silty CLAY, slight trace	- 620				
2B	1970	\land	N=9	20		В 5.04	Gray (10YR5/1) with 70% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY, slight trace sand and gravel					
	20/24	\bigvee	14-5			Sh						
3A	83%	ss	7-8 N=12	17		2.52 Sh	Ψ $G = \begin{bmatrix} 0 & 0 \\ 0 & 0 \\ 0 & 0 \end{bmatrix}$ Dark gray (10YR4/1) with 5% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY, trace sand, slight trace gravel					
4A	24/24 100%	ss	8-6 7-5 N-12	21		1.24 BSh		616				
4B	($\overline{\langle}$	N=15	21		1.20 B	8 – Gray (10YR5/1) with 10% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY, trace sand, slight trace					
5A	22/24 92%	ss	2-3 4-4 N=7	21		1.36 B	gravel	614				
64	24/24	\bigvee	1-2	21		0.74	Gray (10YR5/1) with 60% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY, trace sand, slight trace gravel	612				
UA	100%	ss	2-3 N=4	21		BSh	Gray (10YR5/1), moist, soft, sandy CLAY					
6B	ť	$\overline{\mathbf{V}}$	2.2	24		0.78 B	12					
7A	17/24 71%	ss	2-3 N=4	21			Gray (10YR5/1), moist, soft, sandy CLAY, trace gravel	- 610				
8A	($\left(\right)$		20			14 - Gray (10YR5/1), moist, soft, fine- to coarse-grained - SAND, trace gravel					
۶D	19/24 79%	ss	1-3 5-6 N=8	17		1 16	Dark yellowish brown (10 Y R4/4), moist, sort, sandy <u>CLAY</u> <u>Gray (10 Y R5/1) with 10% yellowish brown (10 Y R5/8)</u>	608				
00	ł			17		4.40 Sh	$\Psi_{16} = \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - \frac{1}{16} - $					
9A	24/24	ss	6-8 30-35	20			Yellowish brown (10YR5/4), wet, soft, fine to coarse SAND	606				
9B	10070	\square	N=38	8			Gray (10YR5/1), moist, hard, silty CLAY, trace sand and gravel					
			I	I	I	l	End of Boring = 18.27 ft. BGS					
	NOTE(S):											

WE	LIENT Site Location Projec DATES	C: Al e: Co n: Co t: 05 S: St Fin R: St F	EG Coffe CB Mana offeen, III 5S3004A cart: 9/1 nish: 9/1 nny, wan	een Pagem linois 0/200 0/200 rm (8	ower ent F 09 09 09 00's)	Station Cacility	JG 1	CONTRACTOR: Layne-Western Co Rig mfg/model: CME-750 ATV Drill Drilling Method: 4/4" HSA w/SS & CME sar FIELD STAFF: Driller: G. Mills Helper: J. Twellman Eng/Geo: R. Hasenyager	BOREHOLE ID: G279 Well ID: G279 Surface Elev: 629.19 ft. MSL Completion: 28.00 ft. BGS Station: 875,028.06N 2,516,245.60E
er	/ Total (in)		1/6 in alue	ure (%)	en. (lb/ft ³)	f_{e} Type	TOPOGR Quadr Towns Sectior	APHIC MAP INFORMATION: angle: Coffeen, IL hip: East Fork n 11, Tier 7N; Range 3W	WATER LEVEL INFORMATION: $\Psi = 23.60$ - While drilling $\Psi = 24.68 - 9/21/09$ $\overline{\Psi} =$
Numb	Recov % Rec	Type	Blows N - V RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	24/24 100%	ss	3-3 5-6 N=8	18			2	FILL - Brown (10YR4/3) with 30% yellowish brow (10YR5/6) mottles, moist, firm, silty CLAY with sand trace gravel.	n and
2A	24/24 100%	ss	5-9 10-11 N=19	14			4		626
3A	24/24 100%	ss	5-9 9-10 N=18	17			6		624
4A	24/24 100%	ss	4-5 7-6 N=12	21					622
5A	24/24 100%	ss	3-3 5-7 N=8	19				FILL - dark gray (10YR4/1) with 10% brownish yelle (10YR6/6) mottles, moist, hard, silty CLAY with sand trace gravel.	ow and
6A	24/24 100%	ss	3-4 6-9 N=10	17			12		618
7A	23/24 96%	ss	2-5 5-6 N=10	23			14		
0 4	24/24 100%	ss	2-3 7-6 N=10	22				Brownish yellow (10YR6/8) with 30% gray (10YR5, mottles, moist, firm, silty CLAY with slight trace sand gravel.	(1) and
9A	18/24 75%	ss	4-7 8-9 N=15	25				Yellowish brown (10YR5/8) with 20% gray (10YR6, mottles, moist, firm, silty CLAY with slight trace sand gravel.	(1) and -612
10A	24/24 100%	ss	3-6 7-10 N=13	17				Gray (10YR6/1) with 25% yellowish brown (10YR5, mottles, moist, firm, silty CLAY with sand and trace gra	/8) avel.
NC)TE(S):						20		Page 1 of 2

FIELD BORING LOG												
	CLIEN	F: A	EG Coffe	en P	ower	Station	n	CONTRACTOR: Layne-Western Co				
	Sit	e: C	CB Mana	gem	ent F	acility		Rig mfg/model: CME-750 ATV Drill	BOREHOLE ID: G279			
	Locatio	n: C	offeen, Ill	inois	5			Drilling Method: 4¼" HSA w/SS & CME samp	lers Well ID: G279			
	Projec	et: 05	5S3004A						Surface Elev: 629.19 ft. MSL			
	DATE	S: St	tart: 9/10	0/20	09			FIELD STAFF: Driller: G. Mills	Completion: 28.00 ft. BGS			
		Fir	nish: 9/1	0/20	09			Helper: J. Twellman	Station: 875,028.06N			
WE	CATHE	R: Su	unny, war	m (8	30's)			Eng/Geo: R. Hasenyager	2,516,245.60E			
	SAMPL	E	Т	EST	INC	J	TOPOGI	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:			
	(in)				f3)	sf)	Quad	angle: Coffeen, IL	$\mathbf{\Psi} = 23.60$ - While drilling			
	$\frac{y}{pe} \frac{n}{(t;t)}$							hip: East Fork	$\Psi = 24.68 - 9/21/09$			
<u>ц</u>	' To		<i>6 i</i> ue	e l	j.	0 E	Sectio	n 11, Tier 7N; Range 3W	$\underline{\nabla}$ =			
mbe	cov / Recc	ь	U D	istu	/ De	(tsf lure	Depth	Lithologic	Borehole Elevation			
Nu	Re.	T_{y_j}	Blc R	Ĭ	D.	Qu Fai	ft. BGS	Description	Detail ft. MSL Remarks			
11A	23/24 96%	ss	2-4 5-7 N=9	18				Gray (10YR6/1) with 25% yellowish brown (10YR5/8) mottles, moist, firm, silty CLAY with sand and trace grave [Continued from previous page]	el. 608			
12A	19/24 79%	ss	4-9 8-9 N=17	13			 Ţ	Yellowish brown (10YR5/8), moist, firm, clayey SILT ar very fine-grained SAND with slight trace gravel.	d			
12B				12			24					
13A 14A	17/24 71%	ss	<i>1-5</i> <i>5-7</i> N=10	18 16			⊻ 26-	Light brownish gray (10YR6/2), wet, loose, very fine- to coarse-grained SAND.	, 604			
	24/24	V ss	10-10					Brownish yellow (10YR6/6), moist, hard, very silty CLA	Y			
14B	$B \begin{bmatrix} 24.2.\\ 100\% \\ 100\% \end{bmatrix} \overset{\mathrm{ss}}{\underset{N=28}{\overset{I8-I8}{\underset{N=28}{\overset{I4}{\underset{I4}}}}} \\ I4 \end{bmatrix}$							Gray (10YR6/1), moist, hard, very silty CLAY with sand and trace gravel.	n - 602			
							28	End of Boring = 28.0 ft. BGS				

F	[EL]	D 1	BOF	RII	NC	L	OG		CR HANSON
	CLIEN Sit Locatio Projec DATE	T: A a: C n: C ct: 05 S: St Fin	EG Coffe CB Mana offeen, II S3004A art: 2/2 ish: 2/2	een H agen Ilinoi 6/200 6/20	Powe lient l ls 08 08	r Statio Facility	on 7	CONTRACTOR: Testing Service Corporation Rig mfg/model: CME-650 Track Rig Drilling Method: 3 ¹ / ₄ " HSA w/SS & CME sample FIELD STAFF: Driller: B. Williamson Helper: R. Keedy	BOREHOLE ID: G280 Surface Elev: 623.0 ft. MSL Completion: 18.0 ft. BGS Station: 875,045.11N
WF	EATHE	R: O	vercast, o	cold				Eng/Geo: S. Simpson	2,515,679.48E
5	SAMPL	E	1	TEST	TING	}	TOPOGI	RAPHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	al (in				5/ft ³)	0	Quad	rangle: Coffeen, IL	$\Psi = 15.60$ - While drilling $\Psi = 4.34 - 3/12/08$
L	' Toti wery		6 in ue	re (%	n. (It	T_{yp}	Sectio	n 11, Tier 7N; Range 3W	$\overline{\underline{\nabla}} = -4.54 + 5.12700$ $\overline{\underline{\nabla}} = -2.54 + 5.12700$
umbe	ecov /	ype	lows/ - Val QD	loistu	ry De	u (tsf ailure	Depth	Lithologic	Borehole Elevation
z	8%	E	a z z	Z		0£		Description	
1A	24/24 100%	ss	5-3 4-4 N-7	23				Dark grayish brown (10YR4/2), moist, firm, clayey	SILT (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
1B		\bigwedge	IN=/	26		2 33		Brown (10YR4/3) with 20% yellowish brown (10Y mottles, moist, firm, silty CLAY	R5/8)
2.A		∇		30		B 1.28	2	Dark yellowish brown (10YR4/4), moist, firm, silty	CLAY
	24/24	ss	3-4 4-6			BSh			
20		\wedge	N=8	25				Dark gray (10YR4/1) with 40% yellowish brow (10YP5/8) mottles moist firm silty CLAY	/n
2B	($\overline{1}$		25			⊻ ⁴	(101K5/6) motics, moist, min, sity CLA1	
	19/24 79%	ss	3-4 6-6						618
3A	,,,,,	\wedge	N=10	14		3.10 Sh		Dark gray (10YR4/1) with 40% yellowish brow	n E
	(6-	(10YR5/8) mottles, moist, firm, silty CLAY, slight sand	trace
	22/24	V ss	9-11 10-8						616
4A	92%	Λ	N=21	18		1.67		Dark gray (10YR4/1) with 40% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY, trace	/n sand,
	($\left(\right)$				БSN	8		
54	19/24	\bigvee_{a}	2-2	20		1 47			614
511	79%	1 35	4-4 N=6			B		Dark gray (10YR4/1) with 40% yellowish brow (10YR5/8) mottles, moist, firm, silty CLAY, sand,	trace
5B	(21		1.28 B	10-	gravel	
	22/24	V	2-3			Б			
	92%	Å ss	<i>3-3</i> N=6					Vallowich brown (10VP5/8) with 20% light gr	-612
6A		\square		20			12	(10YR6/1) mottles, moist, soft, sandy CLAY	ly
		\backslash	3 14					Yellowish brown (10YR5/8), moist, soft, fine to co	barse /
	23/24 96%	ss	23-21 N-37					Yellowish brown (10YR5/8), moist, firm, sandy CI	\overline{LAY} , -610
7A		\wedge	11-57	13			14	trace gravel	
/11		∇					14	Yellowish brown (10YR5/4), moist, firm, clayey S	SILT, = _
	23/24	ss	12-17 24-26					duce said and graver	
8A		Λ	N=41	9			I I I		
8B	(15			16-	Yellowish brown (10YR5/4), wet, soft, fine- to coarse-grained SAND. trace gravel	
9A	24/24	ss	11-27 54-43	26					
	100%	\land	N=81					Gray (10YR5/1), moist, hard, silty CLAY, trace san gravel	id and
9B				7	I		ı = <u>1</u>	End of Boring = 17.98 ft. BGS	
	NOTE(S):							

F	(EL) CLIEN Sit Locatio Projec	D] T: N te: C on: C ct: 15	BOR atural Re offeen En offeen, Ill 5E0030	source lergy linois	NG ce Te Cent	E LC echnolo ter	DG gy, Inc.	CONTRACTOR: Ramsey Geotechnical Engineering, LLC Rig mfg/model: D-50 Turbo Tracked MST 800ATV Drilling Method: Hollow Stem Auger (3¼"overdrill/4¼") BOREHOLE ID: G281 Well ID: G281 Surface Eley: 623.82 ft, MSL							
WE	DATE	S: St Fir R: St	tart: 9/8 nish: 9/8 unny, hi 7	/201 /201 /0's	5 5			FIELD STAFF: Driller: D. Crump Helper: D. Groves Eng/Geo: K. Theesfeld			Completion Station	: 20.29 ft. BGS : 2,514,455.48N 874,375.37E			
S	SAMPL	Æ	Т	EST	TINC	Ĵ	TOPOGR	APHIC MAP INFORMATION:	WATEI	TER LEVEL INFORMATION:					
	tal (ir		ı	()	b/ft ³)	o (tsf) e	Quadrangle: Coffeen, IL Township: East Fork			$\Psi = 14.00$ - During Drilling $\Psi =$					
er	Tol /		/6in alue	ure (⁰	en. (l	$f_{\rm Typ}^{\rm (f)}$	Section	10, Tier 7N; Range 3W	<u> </u>	=					
qunN	Recov % Rec	Type	Blows N - Vi RQD	Moist	Dry D	Qu (ts Failur	Depth ft. BGS	Lithologic Description		Borehole Detail	Elevation ft. MSL	Remarks			
1A	17/24	M_{co}	15-10	14				Light gray (10YR7/2), dry, very stiff, SILT with little and trace gravel.	clay						
111	71%		N=17				2	Yellowish brown (10YR5/4) with 5% dark brown (10YR3/3) mottles, dry, very stiff, SILT with few clay	and		622				
	19/24	Mss	2-4					Yellowish brown (10YR5/4) with 15% dark yellowish brown (10YR4/6) and 5% dark brown (10YR3/3) mot	sh tles,						
2A	79%	\square	N=9	25		1.50		moist, stiff, SILT with few clay.			620				
	22/24	M_{ss}	2-2					Yellowish brown (10YR5/4) with 15% dark yellowish brown (10YR4/6) mottles, moist, medium, CLAY with the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state	sh ith						
3A	92%	$\mathbb{N}^{\mathbb{Z}}$	N=5	23		0.40		some silt and trace fine-grained sand and small grave	શે.		618				
	24/24	\square	5-5				0	Yellowish brown (10YR5/4) with 15% dark yellowis	sh						
4A	100%	\bigwedge^{ss}	0-0 N=11	19		1.20		brown (10YR4/6) and 5% dark brown (10YR3/3) mot moist, stiff, CLAY with some silt and trace fine-grain sand and small gravel.	tles, ed		616				
	20/24	\overline{M}	2-2				8	Vallaviek brown (10VD5/4) with 200/ dark vallavi	ah						
5A	83%	ss	3-4 N=5	21		1.40		brown (10YR4/6) and 5% dark brown (10YR3/3) mot moist, stiff, SILT with some clay and trace very fine- fine grained sand and graph gravel	tles, to						
	22/24	M	2_2				10				- 614				
6A	92%	ss	3-3 N=5	18		0.50		Dark yellowish brown (01YR4/6) with 30% yellowish brown (10YR5/4) mottles, moist, soft, SILT with few of and little fine- to coarse-grained sand and small gravel, i	sh clay trace						
		\square	24				12	wood fragments.			612				
7A	17/24 71%	ss	5-5 N=9	19		0.30		Dark yellowish brown (01YR4/6) with 15% yellowish brown (10YR5/4) mottles, moist, soft, SILT with few of	sh clay						
		$\left(\right)$					⊻ 14 -	and very fine- to fine-grained sand and trace small grav	vel.		610				
	19/24 79%	ss	3-11 21-28 N=32					Dark yellowish brown (10YR4/4), wet, dense, very find fine-grained SAND with some silt, few clay and trace su gravel.	e- to mall						
		$\left(\right)$					16	Dark vellowish brown (10YR4/4) wet dense verv fin	e- to		608				
	24/24 100%	ss	21-36 39-50 N=75					fine-grained SAND with few silt, little clay and trace si gravel.	nall						
8A		$\left(\right)$		7		4.50	18	Yellowish brown (10YR5/6) with 5% strong brown (7.5YR5/6) mottles, moist, hard, SILT with few clay a little fine-grained sand and small gravel	and		606				
	11/24 46%	ss	16-9 30-50 N=39					Dark grayish brown (10YR4/2) with 5% strong brow (7.5YR5/6) mottles, moist, hard, SILT with few clay a little fine-grained sand and small gravel	vn and						
	0/3 0%	BD					20	$\mathbf{F}_{nd} = \mathbf{f}_{nd} = \mathbf{f}_{nd} = \mathbf{f}_{nd} + \mathbf{f}_{nd} = \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} = \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}_{nd} + \mathbf{f}$		-	604				
								End of boring = 20.29 feet							

ſ

FI	ELIEN CLIEN Sit Locatio Projec DATE	D] T: N te: C n: C n: C s: S S: S F ir R: S	BOR atural Re offeen En offeen, III 5E0030 cart: 9/1 hish: 9/1	sourd lergy linois 4/20 4/20 50's	ce Te Cent s 15	GLC echnolo ter	DG gy, Inc.	CONTRACTOR: Ramsey Geotechnical Engine Rig mfg/model: D-50 Turbo Tracked MST 8 Drilling Method: Hollow Stem Auger (3¼"ove FIELD STAFF: Driller: D. Crump Helper: D. Groves Eng/Geo: R. Hasenvager	eering, LLC 00ATV erdrill / 4¼") BOREHOLE ID: G401 Surface Elev: 623.03 ft. MSL Completion: 19.30 ft. BGS Station: 2,515,614.84N 872 510 57E
S	SAMPL	E	T	EST	TINC	J	TOPOG	Caphic Map INFORMATION	WATER LEVEL INFORMATION:
	l (in)				(H ³)	(tsf)	Quad	rangle: Coffeen, IL	$\mathbf{\Psi} = \text{Dry} - \text{During Drilling}$
	Tota very		6 in ue	e (%)	u. (lb	$_{\rm Type}^{Qp}$	Town Sectio	ship: East Fork n 11, Tier 7N; Range 3W	$\underline{\Psi} = \underline{\nabla}$
Number	Recov / % Reco	Type	Blows / N - Vali RQD	Moistur	Dry De	Qu (tsf) Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks
1A	16/24 67%	ss	2-2 3-7 N=5	17			2	Dark grayish brown (10YR4/2), moist, soft, CLAY with little silt and trace very fine- to fine-grained sand - FIL Yellowish brown (10YR5/6) moist, medium, CLAY with some silt and trace very fine- to coarse-grained sand - FI	th
2A	21/24 88%	ss	8-11 8-9	17		1.80		Dark gray (10YR4/1), moist, stiff, SILT with little clay a trace very fine-grained sand.	and620
2B		\wedge	N=19	25				Yellowish brown (10YR5/6), moist, stiff, CLAY with so silt and trace very fine- to fine-grained sand.	me
3A	23/24 96%	ss	3-4 7-8 N=11	23		2.50		Gray (10YR5/1) with 20% yellowish brown (10YR5/6 mottles, moist, medium, CLAY with some silt and trac very fine- to fine-grained sand.	
4A	24/24 100%	ss	8-9 12-14 N=21	21		3.30		(ray (10VD5/1) with 200/ vollowish because (10VD5/	616
4B		Ĥ		19		2.80	8	mottles, moist, stiff, SILT and very fine-grained SAND v trace clay	vith
5A	24/24 100%	ss	2-3 4-5 N=7	21		1.30	10		614
6A	24/24 100%	ss	2-4 5-6 N=9	17		2.50	12	Gray (10YR5/1) with 30% yellowish brown (10YR5/6 mottles, moist, medium, CLAY with some silt and trac very fine- to fine-grained sand.	6) e - 612
7A	24/24 100%	ss	9-7 8-9 N=15	21		1.40	14		610
8A 8B	24/24 100%	ss	2-3 2-4 N=5	17 19		1.30	16	Gray (10YR6/1), moist soft, CLAY with very fine- to fine-grained sand and little silt. Yellowish brown (10YR5/6), wet, loose, very fine- to	
9A	20/24 83%	ss	5-4 5-10 N=9	21				fine-grained SAND with trace silt. Yellowish brown (10YR5/6), wet medium, SILT with so very fine-grained sand and little clay. Yellowish brown (10YR5/6), wet, loose, very fine- to	me 606
9B	12/16		23-41	16			18	Gray (10YR5/1), moist, very hard, SILT with few clay a	
10A	/3%	\wedge	50/4"	6		4.50		$E_{nd} \text{ of horing} = 19.3 \text{ feet}$	604

NOTE(S): G401 installed in borehole.

	Suri	ace Elevation: <u>610.56</u> Datum <u>msl</u>	Completion Date: N	8/27/10 lorthing: 872502.26 Easting: 2516632.59) LOG	IGHT (pct) COUNTS ERY/ROD	.ES	V	VELL DI	AGRAM		
	DEPTH IN FEET	DESCR	IPTION OF MA	TERIAL	GRAPHIC	DRY UNIT WE SPT BLOW (CORE RECOV	Idmes	Dian T	Stickup neter: 6 in	iches	-2.8	Level Contraction
		Medium stiff, brown	, silty CLAY - CL			6-3-3	SS1			Concrete	1.0	609.6
ļ	 5	Modium atiff any -				1-2-4	SS2	2" sch 40 PVC		Bentonite		
		iwediom sun, gray, s	Sing CLAF, trace sand	1- CL	⊽	1-2-5	SS3					
IL TYPES ONLY.	— 10-	Hard to stiff, brown CL (TILL)	to gray, silty CLAY, tr	ace sand seams -		9-22-28	SS4			Ť	8.0 10.0	602.6 600.6
TWEEN SO URPOSES												
DARIES BE TRATION P	- 15-					1-6-12	SS5	2° sch 40 PVC 0.10 slotted		Filter sand		
ATE BOUN FOR ILLUS		- - -										
APPROXIM APHIC LOG	- 20-	Boring terminated a	t 20 feet.			3-4-9	SS6	Bottom cap		<u> </u>	20,0 20,4	590.6 590.2
SENT THE DUAL. GRV												
AV BE GRA	- 25											
CATION LIN NSITION M												
D THE TRATIFI	- 30-	-										
310 NOTE	·											
1.GPJ 12/1	- 35-											
INC 063830												
PJ GT		GROUNDWATER DA	ATA	DRILLING	<u>DATA</u>			Drawn by: KSA Date: 9/10/10	Checked b Date: /-	M-11 Dat	o'vd. by: te: 1/0	KR8
01- COFFEEN.G	EN	COUNTERED AT 7 F	eet \$	<u>AUGER 4 1/4*</u> WASHBORING FR <u>MVU</u> DRILLER <u>S</u>	ЮLLO [;] ОМ <u>WG</u> _L(W STEM FEET DGGER		G	GEOTE	CHNOL FROM TO	OGY IE GROUN	
WL J017150.(<u>CME 55TRK</u> D HAMMER TYP	RILL R È <u>Auto</u>	IG 2		Ameren-Co	offeen Ash	Pond Ev	aluati	on	
BORING 2002	REM	MARKS:						LOC	G OF BOR	ING: APV G402	/-2	
LOG OF	. .			<u></u>			÷	Pro	ject No.	J017150.	01	

F	FIELD BORING LOG HANSON												
	CLIEN Si Locatio Proje DATE	T: N te: C on: C ct: 15 (S: St	atural Re offeen En offeen, Ill 5E0030 t art: 9/1	esour nergy linois 1/20	ce Te Cen 5	echnolo <u></u> ter	gy, Inc.	CONTRACTOR: Ramsey Geotechnical Engine Rig mfg/model: D-50 Turbo Tracked MST 8 Drilling Method: Hollow Stem Auger (3¼"ove FIELD STAFF: Driller: D. Crump	BOREHOLE ID: G403 Well ID: G403 Surface Elev: 623.81 ft. MSL Completion: 18.15 ft. BGS Station: 2.514.616.63N				
WE	ATHE	Fir R: Ra	n ish: 9/1 aining, hi	1/20 60's	15			Helper: D. Groves Eng/Geo: K. Theesfeld			Station:	2,514,616.63N 873,561.34E	
5	SAMPL	Æ	T	EST	TINC	J	TOPOG	RAPHIC MAP INFORMATION:	WATER I	EVEL	INFORMAT	ION:	
	al (in)/ft ³)	(tsf) e	Quad	rangle: Coffeen, IL	$\underline{\Psi} = 1$ $\overline{\Psi} = 1$	5.00 -	During Drilling	3	
H	/ Toti		/ 6 in lue	Ire (%	en. (lt	O_{Type}	Sectio	n 11, Tier 7N; Range 3W	<u> </u>				
Numbe	Recov % Reco	Type	Blows N - Va RQD	Moistu	Dry De	Qu (tsf Failure	Depth ft. BGS	Lithologic Description	Bc L	orehole Detail	Elevation ft. MSL	Remarks	
1A	19/24 79%	ss	2-2 2-2 N=4	25		0.80	ر الاستاليينيا	Very dark brown (10YR2/2) grading to dark grayish bro (10YR4/2), moist, medium, SILT with some clay, trac roots and grass.	wn $\frac{\sqrt{1/2}}{\sqrt{1/2}}$		<u> </u>		
	22/24	M	2-2					Very dark brown (10YR2/2) grading to dark grayish bro (10YR4/2), moist, stiff, SILT with some clay, trace woo	wn od.		-		
2A 2B	92%	ss	3-4 N=5	29 26		1.50		Yellowish brown (10YR5/4) with 10% very dark brow (10YR2/2) mottles, moist, stiff, CLAY with little silt, tra- very fine-grained sand seams (<1/16" thick).	n ice		620		
3A	8/24 <i>33%</i>	ss	2-3 4-4 N=7	25		1.50		Yellowish brown (10YR5/4), moist, stiff, CLAY with lit silt and trace very fine-grained sand.	tle				
4A	21/24 88%	ss	8-7 8-7 N=15	20		1.30		Grayish brown (10YR5/2) with 15% yellowish brown (10YR5/6) and 5% very dark grayish brown (10YR3/2 mottles, moist, stiff, CLAY with little silt and trace ver fine-grained sand.			618		
5A	20/24 83%	ss	2-2 3-3 N=5	22		0.70	10	Grayish brown (10YR5/2) with 5% yellowish brown (10YR5/6) mottles, moist, medium, CLAY with little s and trace very fine-grained sand.	lt 		614		
6A	24/24 100%	ss	2-2 3-4 N=5	23		1.40	unuhuuqu	Grayish brown (10YR5/2) with 30% yellowish brown (10YR5/6) mottles, moist, very stiff, CLAY with little s few very fine- to medium-grained sand, and trace grave	ilt, I.		612		
7A	21/24 88%	ss	5-5 6-5 N=11	20		0.90		Grayish brown (10YR5/3) with 45% yellowish brown (10YR5/6) and 5% dark brown (10YR3/3) mottles, mot stiff, SILT with some clay, few very fine- to coarse-grain sand, and trace gravel.	st, ed				
8A	24/24 100%	ss	3-2 3-6 N=5	17			14 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111 111	Grayish brown (10YR5/3) with 40% yellowish brown (10YR5/6) mottles, moist, medium, SILT with little cla few very fine- to coarse-grained sand, and trace gravel Yellowish brown (10YR5/6) with 30% grayish brown (10YR5/2) mottles, moist, medium, SILT with little cla few very fine- to coarse-grained sand, and trace gravel	y, /				
9A	19/24 79% 0/2 0%	SS BD	8-12 21-25 N=33	8		4.50		Yellowish brown (10YR5/6), wet, loose, SAND with so clay and few silt. Yellowish brown (10YR5/6) with 30% grayish brown (10YR5/2) mottles, moist, stiff, SILT with few clay, ve fine-to coarse-grained sand, and gravel. Yellowish brown (10YR5/6), moist, very stiff, SILT wi some clay and few sand and gravel. Very dark grayish brown (10YR3/2), dry, hard, SILT w little clay and few very fine- to coarse-grained sand and gravel.	me / + + + + + + + + + + + + + + + + + +		606		
								End of boring = 10.15 feet					
)TE(S)·	G40	3 installe	d in i	horel	hole							

F	[EL]	DI	BOR	I	١G	L C	DG				
WE	CLIEN Sit Locatio Projec DATE	T: N te: C n: C ct: 15 S: St S: St Fir R: Pa	atural Re offeen En offeen, Ill 5E0030 tart: 5/1 hish: 5/1 artly sunn	sour ergy inois /200 /200 ty, w	ces T Cem 3 7 7 7 7 7 7 7	[°] echnol ⁶ ter	ogy, Inc.	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4¼" Hollow stem auger with sampler FIELD STAFF: Driller: A. Rachford Helper: M. Brown Eng/Geo: R. Hasenyager	Be a split spoon	OREHOLE ID: Well ID: Surface Elev: Completion: Station:	 SB22 G404 613.10 ft. MSL 12.00 ft. BGS 873,999.77N 2,516,397.85E
	SAMPL	E	Т	EST	FINC	3	TOPOGRA	APHIC MAP INFORMATION:	WATER LEVE	L INFORMAT	TION:
ber	v / Total (in) covery		s/6 in 'alue	ture (%)	Den. (lb/ft ³)	sf) Qp (tsf) re Type	Quadra Townsh Section	ngle: Coffeen, IL ip: East Fork 11, Tier 7N; Range 3W	$ \underline{\Psi} = 2.07 $ $ \underline{\Psi} = $ $ \underline{\nabla} = $	- 05/10/2007	
Num	Reco % Re	Type	Blow N - V RQD	Mois	Dry I	Qu (t Failu	Depth ft. BGS	Lithologic Description	Borehol Detail	e Elevation ft. MSL	Remarks
1A 1B	19/24 79%	ss		26 26		0.85 B		Black (10YR2/1), moist, firm, clayey SILT (TOPSOI	$L) A \ b \ b \ c \ c \ c \ c \ c \ c \ c \ c$	612	
2A	19/24 79%	ss		16		2.47 B	₹ 2	Gray (10YR5/1) with 35% dark yellowish brown (10YR4/6) mottles, moist, firm, clayey SILT with sand trace gravel	and	610	
3A	18/24	V ss		19		2.18 B					
3B	/3%	\wedge		18		2.33 B	6	Gray (10YR6/1) with 20% yellowish brown (10YR5// mottles, moist, firm, silty CLAY with sand and trace gra	8) ivel	- 008	
4A	24/24	M		23		0.58 B		Yellowish brown (10YR5/8), moist, soft, sandy CLAY v slight trace gravel	with		
4B	100%	A ss		18			8-	Yellowish brown (10YR5/8), very moist to wet, very so clayey, very fine- to medium-grained SAND with trac gravel	oft, e	606	
5A	23/24 96%	ss		10				Yellowish brown (10YR5/4) with 30% yellowish brow (10YR5/8) mottles, moist, hard, clayey SILT with sand trace gravel	vn and	604	
5B	19/24			19			10	Yellowish brown (10YR5/6), wet, loose, very fine- to medium-grained SAND with coarse-grained sand and sl trace gravel) ight		
6B	79%	∬ ^{ss}		11			12	Dark yellowish brown (10YR4/6) with 40% yellowis brown (10YR5/8) mottles, moist, very hard, very silty CLAY with sand and gravel End of Boring = 12.0 ft. BGS	h y	602	

NOTE(S):

F	[EL]	DI	BOR	IN	JG	LC	OG			A					
WE	CLIEN Sit Locatio Projec DATE	T: Na ae: Co n: Co xt: 15 S: St Fin R: Pa	atural Re offeen En offeen, Ill E0030 cart: 5/1/ hish: 5/1/ urtly sunn	sourc ergy inois /2007 /2007 y, wa	ces T Cent 7 7 arm	echnol ter	ogy, Inc.	CONTRACTOR: Reynolds Drilling Corp. Rig mfg/model: CME-550 ATV Drill Drilling Method: 4¼" Hollow stem auger with sampler FIELD STAFF: Driller: A. Rachford Helper: M. Brown Eng/Geo: R. Hasenyager	n split spoon	BOREHOLE ID: Well ID: Surface Elev: Completion: Station:	SB21 G405 620.90 ft. MSL 14.21 ft. BGS 873,996.79N 2,515,335.70E				
5	SAMPL	E	T	EST	INC	Ĵ	TOPOGRAI	PHIC MAP INFORMATION:	WATER LE	LEVEL INFORMATION:					
ber	v / Total (in ecovery		s / 6 in /alue	ture (%)	Den. (lb/ft^3)	sf) Qp (tsf) re Type	Quadran Township Section 1	gle: Coffeen, IL p: East Fork 1, Tier 7N; Range 3W		.28 - 05/10/2007					
Num	Reco % Ré	Type	Blow N - V RQI	Mois	Dry]	Qu (1 Failu	Depth ft. BGS	Lithologic Description	Bore De	ehole Elevation tail ft. MSL	Remarks				
1A	19/24 79%	ss		43	Π	0.78 B	₹ 2-	Black (10YR2/1), moist, soft, clayey SILT (TOPSOI	L)						
1B 2A	12/12 100%	ss s		26 27		1.94 B 2.52 BSh		Gray (10YR6/1) with 30% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY	8)	618					
3A	24/24	ss		24		3.92 BSh				616					
3B	100%	A		24		2.33 BSh	6	Gray (10YR6/1), moist, firm, silty CLAY slight trace s Gray (10YR5/1), very moist, soft, clayey, very fine- t	and o						
4A	24/24 100%	ss		20		2.33 BSh		Gray (10YR6/1) with 25% yellowish brown (10YR5/ mottles, moist, firm, silty CLAY with sand and trace gra	8) ivel	614					
5A	24/24 100%	ss		24		1.55 BSh	10	Yellowish brown (10YR5/8) with 40% gray (10YR6/ mottles, moist, firm, silty CLAY with sand and trace gra	1) ivel	612					
6A 6B	24/24 100%	ss		19 18			12	Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Dark brown (10YR3/3), very moist, soft, clayey, fine- very coarse-grained SAND with slight trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to	to	610					
7A	24/24 100%	ss		9		7.42 BSh		fine-grained SAND Dark yellowish brown (10YR4/4), moist, soft, sandy SI with trace gravel Yellowish brown (10YR5/4), wet, loose, very fine- to fine-grained SAND Gray (10YR5/1), moist, very hard, very silty CLAY w sand and gravel	LT	608					
								End of Boring = 14.2 ft. BGS							

F]	EL	D]	BOR	I	NC	G L(DG		HANSON
WE	CLIEN Si Locatio Proje DATE	T: N te: C on: 13 ct: 10 (S: S) Fin R: S)	atural Resoffeen Pow 34 CIPS L 5E0080 tart: 8/19 hish: 8/19	sour wer 2ane 9/20 9/20 d-70	ces T Stati , Cot 16 16 s)	Technolo on - Asl ffeen, II	ogy, Inc. h Pond 2 L 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld	BOREHOLE ID: G406 Well ID: G406 Surface Elev: 621.86 ft. MSL Completion: 18.75 ft. BGS Station: 872,521.34N 2,514,702.38E
	SAMPL	Æ	T	EST	INC	3	TOPOGRA	APHIC MAP INFORMATION:	WATER LEVEL INFORMATION:
	l (in)				(ft ³)	(tsf)	Quadra	ngle: Coffeen	\mathbf{Y} = Dry - During Drilling
	Tota 'ery		6 in 1e	%) e	(]p	Q_p	Townsh Section	ip: East Fork 11 Tier 7 N · Range 3 W	$\underline{\Psi} = \nabla$
nber	ov /	e	ws/ Valı D	istur	Der	(tsf) ure	Denth	Lithologic	
Nur	Rec % H	Typ	Blo N - RQ	Mo	Dry	Qu Fail	ft. BGS	Description	Detail ft. MSL Remarks
								Dark brown (10YR3/3), moist, stiff, SILT with few cla and trace organics.	ay
							2	and trace organics. Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay and trace small gravel. Brown (10YR5/3) with 10% dark brown (10YR3/3) mottles, SILT with some clay. Grayish brown (10YR5/2) with 5% dark yellowish brov (10YR4/6) mottles, moist, stiff, CLAY with few silt an little fine-grained sand. Very pale brown (10YR7/4) with 25% yellowish brow (10YR5/6) mottles, moist, medium, CLAY with trace set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the	=
								and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with some fine- to medium-grained sand, little s and trace small gravel.	o) f, ilt,
								mottles, wet, loose, fine-grained SAND with some cla Gray (10YR6/1) with 5% brownish yellow (10YR6/6 mottles, moist, loose, fine-grained SAND with some cla	$\frac{y'}{y'} = -\frac{606}{600}$
							18	Gray (10YR5/1), dry, hard, SILT with few clay, few fit to coarse-grained sand and trace small gravel.	ne- $\left \right = \left \right = 604$
	I		I	I	I	I	∣ ⊐_	End of Boring = 18.75 ft. BGS	
NC)TE(S):	G40 Bor	6 installed	d in l	boriu	1g.	ent to GA06Γ		

F	EL	DI	BOR	IN	NG	G L(OG CAR HANSON	J					
1	CLIEN Sit Location Projec	Γ: Ν e: C n: 13	atural Re offeen Po 34 CIPS I 550080	sourd wer S Lane,	ces T Statio , Cof	echnolo on - Asl feen, II	logy, Inc. CONTRACTOR: Bulldog Drilling, Inc. h Pond 2 Rig mfg/model: CME-750 ATV Drill L 62017 Drilling Method: 4 1/4" Hollow Stem Auger BOREHOLE ID: G407 Surface Eley: 618 35 ft MSL	Ľ					
WE	DATE	S: St Fir R: R	art: 8/1 nish: 8/1 ain, (mid-	6/20 6/20 -70s)	16 16		FIELD STAFF: Driller: J. Dittmaier Completion: 20.00 ft. BGS Helper: M. Hill Station: 872,973.39N Eng/Geo: K. Theesfeld 2,513,705.87E	,					
	SAMPL	E	T	EST		J _	TOPOGRAPHIC MAP INFORMATION: WATER LEVEL INFORMATION:						
	tal (i		1	(%	(p/ff ³)	o (tsf oe	Quadrangle: Coffeen $\underline{\Psi}$ = 16.00 - During DrillingTownship: East Fork $\underline{\Psi}$ =						
er	' / To cover:		16 i	ure (en. ($\stackrel{(f)}{=} \stackrel{(f)}{=} O$	Section 10, Tier 7 N.; Range 3 W. $\overline{\nabla} =$						
Numb	Recov % Rec	Type	Blows N - V; RQD	Moist	Dry D	Qu (ts Failur	DepthLithologicBoreholeElevationft. BGSDescriptionDetailft. MSLRemarks						
1A	12/24 50%	ss	4-3 3-3 N=6	14		3.50	Very dark gray (10YR3/1), wet, medium, SILT with some organics. [Fill] Gray (10YR6/1), wet, loose, SAND with some gravel and						
		$\langle \rangle$					2						
2A	20/24 83%	ss	2-2 4-4 N=6	18		1.50	 brown (10YR5/6) mottles, moist, very stiff, SILT with / 						
3A	23/24 96%	ss	1-2 3-4 N=5	19		1.75	Brown (10YR5/3) with 25% yellowish brown (10YR5/6) mottles, moist, stiff, CLAY with some silt, trace fine-grained sand and trace small gravel.						
4A	24/24 100%	ss	<i>1-3</i> <i>3-5</i> N=6	19		1.50	Brown (10YR5/3) with 10% yellowish brown (10YR5/6) mottles, moist, stiff, CLAY with some silt, little fine- to coarse-grained sand and trace small gravel.						
5A	21/24 88%	ss	<i>1-2</i> <i>4-4</i> N=6	19		0.50	Yellowish brown (10YR5/6) with 25% brown (10YR5/3) mottles, moist, medium, CLAY with few silt, few fine-grained sand, and trace small gravel.						
6A	22/24 92%	ss	<i>1-2</i> <i>2-1</i> N=4	17			10 Yellowish brown (10YR5/8) with 5% gray (10YR5/1) mottles, moist, very loose, fine-grained SAND with some clay and trace small gravel. 608						
7A	24/24 100%	ss	7-29 33-17 N=62	8			12 - Gray (10YR5/1) with 25% yellowish brown (10YR5/8) mottles, moist, very dense, fine-grained SAND						
		4					Brown (10YR5/3), moist, hard, SILT with some clay and 14 little fine- to coarse-grained sand.						
8A	24/24 100%	ss	3-7 12-17 N=19	12		4.50	Yellowish brown (10YR5/4) with 5% yellowish brown (10YR5/6) and 5% black (10YR2/1) mottles, SILT with some clay and little fine- to coarse-grained sand.						
9A	24/24 100%	ss	4-9 14-20 N=23	13		4.00	Yellowish brown (10YR5/4) with 5% yellowish brown (10YR5/6), 5% dark gray (10YR4/1) and 5% black (10YR2/1) mottles, moist, hard, SILT with little fine- to coarse-grained sand and trace small gravel.						
10A	24/24 100%	ss	2-8 14-19 N=22	14		4.50	Yellowish brown (10YR5/4) with 5% yellowish brown (10YR5/6), 5% dark gray (10YR4/1) and 5% black (10YR2/1) mottles, wet, stiff, SILT with little fine- to coarse-grained sand and little small gravel. Dark grayish brown (10YR4/2) with 10% dark yellowish						
							brown (10YR3/6) mottles, moist, hard, CLAY with some silt, little fine- to coarse-grained sand and trace small gravel.						
							End of Boring = 20.0 ft. BGS						
NO)TE(S)+	G40	7 installe	d in I	horin	ıσ							
		G-10	, mount	ы III I	55111	·Ð.							
WE	CLIEN Sit Locatio Projec DATE ATHEI	T: Na te: Co n: 13 ct: 16 S: St Fin R: Cl E	atural Resou offeen Powe 44 CIPS Lar 5E0080 art: 8/19/2 oudy, (70s) TF. 5	r Stati e, Co 016 016	G	ogy, Inc. h Pond 2 . 62017	CONTRACTOR: Bulldog Drilling, Inc. Rig mfg/model: CME-750 ATV Drill Drilling Method: 4 1/4" Hollow Stem Auger FIELD STAFF: Driller: J. Dittmaier Helper: M. Hill Eng/Geo: K. Theesfeld	BOREHOLE ID: T409 Well ID: T409 Surface Elev: 621.85 ft. MSL Completion: 26.99 ft. BGS Station: 872,517.79N 2,514,693.89E					
------	----------------------------------------------------	------------------------------------------------------------------	-----------------------------------------------------------------------------------------------------------	--------------------------------	--------------------------------------------------------------------------------	----------------------------------------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	--	--	--	--	--
er ,	/ Total (in)		/ 6 in lue	en. (lb/ft ³)	$ \begin{array}{c} \text{f)} & Qp \text{ (tsf)} \\ \text{s Type} \end{array} $	TOPOGRA Quadra Townsh Section	TOPOGRAPHIC MAP INFORMATION:WATER LEVEL INFORMATION:Quadrangle: Coffeen $\mathbf{\nabla} = $ Township: East Fork $\mathbf{\nabla} = $ Section 11, Tier 7 N.; Range 3 W. $\mathbf{\nabla} = $						
Numb	Recov % Rec	Type	Blows N - Va RQD Moistr	Dry D	Qu (ts Failure	Depth ft. BGS	Lithologic Description	Borehole Elevation Detail ft. MSL Remarks					
	0/60 <i>0%</i>	BD				2 4	Dark brown (10YR3/3), moist, stiff, SILT with few claand trace organics.	e 					
	0/60 <i>0%</i>	BD				6	Gray (10YR5/1) with 10% yellowish brown (10YR5/6) Gray (10YR5/1) with 10% yellowish brown (10YR5/6) and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt and trace fine- to medium-grained sand. Gray (10YR5/1) with 10% yellowish brown (10YR5/6) and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt and trace fine- to medium-grained sand. Gray (10YR5/1) with 10% yellowish brown (10YR5/6) and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with little silt, little fine- to medium-grained sand and trace small gravel.	$ \begin{array}{c} \operatorname{Mn} & 1 \\ \operatorname{m} & - \\ \operatorname{ilt.} \\ \operatorname{m} & - \\ \operatorname{ilt.} \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m} & - \\ \operatorname{m}$					
	0/60 <i>0%</i>	BD					Gray (10YR5/1) with 10% yellowish brown (10YR5/6 and 5% very dark gray (10YR3/1) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little sil and trace small gravel. Gray (10YR6/1) with 25% brownish yellow (10YR6/8 and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with few fine- to medium-grained sand, little sil and trace small gravel. Gray (10YR6/1) with 10% brownish yellow (10YR6/8 and 5% strong brown (7.5YR4/6) mottles, moist, stiff CLAY with gravel.	$\frac{1}{5}$					
	0/60 <i>0%</i>	BD					 Gray (10YR6/1) with 5% brownish yellow (10YR6/6 mottles, wet, loose, fine-grained SAND with some clay Gray (10YR6/1) with 5% brownish yellow (10YR6/6 mottles, moist, loose, fine-grained SAND with some clay Gray (10YR6/1), dry, hard, SILT with few clay, few fir to coarse-grained sand and trace small gravel. 	he- 604 604 604					

Illinois Environmental Protection Age	ncy	Wel	l Completion	Report
Site #: County:	Montgomery		Well #: G1	54
Site Name: CCB Management Facility		I	Borehole #:C	3154
State Plane Coordinate: X 2,513,243.1 Y 874,978.4 (or) La	.titude:	Longitue	de:	
Surveyed By: Jeffrey D. Emrick	IL Registration #:	035-003507		
Drilling Contractor:Testing Service Corp.	Driller: <u>B. Willi</u>	amson		
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhon</u>	ald W. Hasenyage	er, LPG #196-0002	246
Drilling Method: Hollow stem auger	Drilling Fluid (Type	e): <u>n/a</u>		
Logged By: <u>Ryne M. Fiorito</u>	Date Started:1	<u>2/16/2011</u> Da	te Finished: <u>12/</u>	16/2011
Report Form Completed By:	Date:12/27	/2011		
ANNULAR SPACE DETAILS	Eleva (MS	tions Depths L)* (BGS)	(0.01 ft.)	
	626	.55 -3.03	Top of Protective	Casing
	<u>_626</u>	.35 -2.83	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>		.52 0.00	Ground Surface	
True of Annulas Socients - High solids hontonits	<u>622</u>	.44 1.08	Top of Annular Se	ealant
Installation Mathod: Tramia				
Setting Time: _ >24 hr.		.42 11.10	Static Water Leve (After Completion)	l 2/21/2011
Type of Bentonite Seal Granular Pellet Slurry -				
Installation Method: <u>Gravity</u>	<u>613</u>	.02 10.50	Top of Seal	
Setting Time: 24 min		.02 13.50	Top of Sand Pack	
Type of Sand Pack: <u>Quartz sand</u> Grain Size: <u>10/20</u> (sieve size)		.26 14.26	Top of Screen	
Installation Method: Gravity		.76 18.76	Bottom of Screen	
Type of Backfill Material:		.42 19.10	Bottom of Well	
Installation Method:		.52 20.00	Bottom of Boreho	le
	Diameter o	CASING MEA	SUREMENTS	8.0
WELL CONSTRUCTION MATERIALS (Choose one type of material for each area)	ID of Riser	Pipe	(inches)	2.0
	Protective	Casing Length	(feet)	5.0
	Riser Pipe	Length	(feet)	17.09
Protective Casing SS304 SS316 PTFE PVC C	DIHER: Bottom of Bottom of	Screen to End Cap	(feet)	0.34
Riser Pine Below W T SS304 SS316 PTFF PVC	THER: Screen Len	gth (1st slot to last sl	ot) (feet)	4.50
Screen SS304 SS316 PTFE PVC	OTHER: Screen Slot	: Size <u>**</u>	(inches)	0.010

Illinois Environmental Protection Agency				Well Completion			Report		
Site #:	Count	y: <u>Mon</u>	tgomer	1		W	'ell #:	<u>G2</u>	70
Site Name: AEG Coffeen Powe	er Station CCB Managemer	nt Facility	7			В	orehole #:	C	1 270
State- Plant Plane Coordinate: X <u>874,801.9</u>	9 Y 2,514,996.8 (or)	Latitude:	c	·	" L	ongitud	e:°		"
Surveyed By: <u>Jeffrey D. Emricl</u>	k		IL Registration #:035-003507						
Drilling Contractor: <u>Testing Ser</u>	vice Corporation		Driller: B. Williamson						
Consulting Firm: <u>Hanson Profes</u>	ssional Services Inc.		Geologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246						
Drilling Method: <u>Hollow stem auger</u> D				g Fluid (Type)	:				
Logged By: <u>Suzanna L Simpso</u>	n		Date St	arted: <u>2/2</u>	26/2008	Date	e Finished: _	2/2	6/2008
Report Form Completed By: Suza	anna L Simpson		Date:	2/29/20	008				
ANNULAR SPAC	CE DETAILS			Elevati (MSL	ons De)* (E	e pths 3GS)	(0.0	1 ft.)	
				626.4	<u>1 </u>	3.49	Top of Pro	tective	Casing
				_625.9	7	3.05	Top of Rise	er Pipe	;
Type of Surface Seal: Concrete				<u> </u>	2).00	Ground Su	rface	
Type of Annular Sealant: Bentonit	te chins			/ 619.9	23	3.00	Top of Ann	ular S	ealant
Installation Method: Gravity		4							
Setting Time: <u>>24 hr.</u>		Ţ	<u>7</u>	_617.3	0	5.62	Static Wate (After Comp	er Leve	el 3/12/2008
Type of Bentonite Seal Granul	ar Pellet Slurry (choose one)								
Installation Method: <u>Gravity</u>		K X	KX	619.9	23	3.00	Top of Sea	1	
Setting Time: _>24 hr.		×	Ř	610.9	2 1	2.00	Top of San	d Pack	:
Type of Sand Pack: <u>Quartz sand</u>									
Grain Size: 10/20 (sieve	e size)			609.7	9 1	3.13	Top of Scr	en	
Installation Method: <u>Gravity</u>				60 5 0	0 1		-	~	
Type of Backfill Material: <u>n/a</u>				<u> 605.0</u> <u> 604.6</u>	$\frac{0}{5}$ $\frac{1}{1}$	7.92 8.27	Bottom of Bottom of	Screen Well	
Installation Method: <u>n/a</u>	(п аррпсаоте)			_604.6	51	8.27	Bottom of	Boreho	ole
				* Referen	ced to a Natio	nal Geode	tic Datum		
					CASING	6 MEA	SUREMEN	VTS	
WELL CONST				Diameter of	Borehole		(i	nches)	8.0
(Choose one	type of material for each area)			ID of Riser H	Pipe		(i	nches)	2.0
				Protective Ca	asing Leng	th		(feet)	5.0
Protective Cosing	SS304 SS316 DTEE DVC	OTHER. (Steel	Riser Pipe L	ength	1.6		(feet)	16.18
Riser Pine Above W T	SS304 SS316 PTFF PVC	OTHER.	<u> </u>	Bottom of So	creen to En	id Cap		(feet)	0.35
Riser Pipe Below W.T.	SS304 SS316 PTFE (PVC)) OTHER:		Screen Leng	ui (1st slot	to last slo	ot)	(feet)	4.79
• · · ·				10tal Lelight	or casing			(1001)	41.34

Screen

SS304 SS316 PTFE PVC OTHER:

Screen Slot Size **
 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Enviro	nmental Protection Ag	gency		Well	Completion	Report			
Site #:	County:	Montgomer	y	W	Vell #: G2	.79			
Site Name: <u>AEG Coffeen Por</u>	wer Station CCB Management I	Facility		В	orehole #:	G279			
State- Plant Plane Coordinate: X_2,516,24	5.6 Y <u>875,028.1</u> (or) L	atitude:	<u> </u>	Longitude	e:o	_'"			
Surveyed By:Jeffrey D. Emri	ck	IL Reg	istration #: <u>035-0</u>	03507					
Drilling Contractor: <u>Layne-We</u>	estern Co	Driller	G. Mills						
Consulting Firm: <u>Hanson Prof</u>	Sessional Services Inc.	Geolog	Geologist: Rhonald W. Hasenyager, LPG #196-000246						
Drilling Method: Hollow stem auger Drilling Fluid (Type):									
Logged By: Rhonald W. Hase	enyager	Date S	tarted: 9/10/20	09 Date	e Finished: 9/1	0/2009			
Report Form Completed By: Su	zanna L. Simpson	Date:	10/7/2009						
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)				
			(NISE) 632.33	-3.14	Top of Protective	Casing			
			632.04	-2.85	Top of Riser Pipe	5			
Type of Surface Seal: <u>Concrete</u>				0.00	Ground Surface				
Type of Annular Sealant: High	colids bentonite		626.19	3.00	Top of Annular S	ealant			
Installation Method: Tremie									
Setting Time: >24 hr.			601.66	27.53	Static Water Leve	1			
					(After Completion)	9/21/2009			
Type of Bentonite Seal Gran	ular Pellet Slurry (choose one)								
Installation Method: <u>Gravit</u>	У		610.45	18.74	Top of Seal				
Setting Time: <u>18 min</u>			608.77	20.42	Top of Sand Pack	-			
Type of Sand Pack:Quartz sand	1								
Grain Size: <u>10/20</u> (sid	eve size)		606.79	22.40	Top of Screen				
Installation Method: <u>Gravit</u>	У		60 - 40						
Type of Backfill Material:Quar	tz Sand		<u>602.40</u> <u>604.51</u>	<u>26.79</u> <u>24.68</u>	Bottom of Screen Bottom of Well				
Installation Mathed: Cravit	(if applicable)		601 10	28.00	Pottom of Porcho	Ja			
Instanation Method. <u>Oravit</u>	у		* Referenced to a	National Geodetic	c Datum	ne			
			CAS	SING MEAS	SUREMENTS				
			Diameter of Boreho	ole	(inches)	8.0			
WELL CONS (Choose of	type of material for each area)		ID of Riser Pipe		(inches)	2.0			
			Protective Casing I	length	(feet)	5.0			
Protective Casing	SS304 SS316 PTFE PVC	OTHER: (Steel)	Riser Pipe Length	o End Con	(feet)	25.25			
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC	OTHER:	Screen Length (1)	st slot to last slo	t) (feet)	4.39			
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC	OTHER:	Total Length of Ca	sing	(feet)	30.17			

 Screen
 SS304

 Well Completion Form (revised 02/06/02)

SS316

PTFE

PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Environmental Protection Age	ency	Well Completion Report					
Site #: County:	Montgomery	Well #: G280					
Site Name: AEG Coffeen Power Station CCB Management Fa	acility	Borehole #: G280					
StatePlaneCoordinate: X $875,045.1$ Y $2,515,679.5$ (or)Latit	tude:°'	Defende 000					
Surveyed By:Jeffrey D. Emrick	IL Registration #:035-(L Registration #:035-003507					
Drilling Contractor: <u>Testing Service Corporation</u>	Driller: <u>B. Williamson</u>	Driller: <u>B. Williamson</u>					
Consulting Firm: <u>Hanson Professional Services Inc.</u>	Geologist: <u>Rhonald W</u>	ologist: <u>Rhonald W Hasenyager</u> , LPG #196-000246					
Drilling Method: <u>Hollow stem auger</u>	Drilling Fluid (Type):						
Logged By: <u>Suzanna L Simpson</u>	Date Started:2/26/2	008 Date Finished: <u>2/26/2008</u>					
Report Form Completed By: Suzanna L Simpson	Date: 2/29/2008						
ANNULAR SPACE DETAILS	Elevations (MSL)*	Depths (0.01 ft.) (BGS)					
	625.79	<u>-2.84</u> Top of Protective Casing					
	<u>_625.30</u>	<u>-2.35</u> Top of Riser Pipe					
Type of Surface Seal: Concrete	622.95	Ground Surface					
Type of Annular Sealant: Bentonite chins	620.85	<u>2.10</u> Top of Annular Sealant					
Installation Method: Gravity							
Setting Time: >24 hr		4.34 Static Water Level					
		(After Completion) 3/12/2008					
Type of Bentonite Seal Granular Pellet Slurry (choose one)							
Installation Method: <u>Gravity</u>	<u>620.85</u>	<u>2.10</u> Top of Seal					
Setting Time: <u>>24 hr.</u>	<u>611.75</u>	Top of Sand Pack					
Type of Sand Pack: Overte cond							
Grain Size: 10/20 (sieve size)	610.16	12.79 Top of Screen					
Installation Method: Gravity							
	<u>_605.32</u>	17.63 Bottom of Screen					
Type of Backfill Material: <u>n/a</u> (if applicable)		<u>17.98</u> Bottom of Well					
Installation Method: <u>n/a</u>		17.98 Bottom of Borehole					
	CA	SING MEASUREMENTS					
WELL CONSTRUCTION MATERIALS	Diameter of Bore	hole (inches) 8.0					
(Choose one type of material for each area)	Protective Casing	(inches) 2.0					
	Riser Pine Length	(feet) 15.14					
Protective Casing SS304 SS316 PTFE PVC OTH	HER: Steel Bottom of Screen	to End Cap (feet) 0.35					
Riser Pipe Above W.T. SS304 SS316 PTFE PVC OTH	HER: Screen Length (1st slot to last slot) (feet) 4.84					
Riser Pipe Below W.T. SS304 SS316 PTFE PVC OTH	HER: Total Length of C	Lasing (feet) 20.33					

Screen

SS304 SS316 PTFE PVC OTHER:

**Hand-Slotted Well Screens Are Unacceptable

Screen Slot Size **

0.010

Illinois Enviror	nmental Protection Agency			Wel	Completion	Report
Site #:	County:	lontgomery		V	Vell #: G2	81
Site Name: <u>Coffeen Power St</u>	tation			В	orehole #:	G281
State Plane Coordinate: X 874,37	5.4 Y <u>2,514,455.5</u> (or) Latitud	e:		Longitud	e:	
Surveyed By: Gary C. Rogers	3	IL Registra	ation #: <u>035-0</u>	02957		
Drilling Contractor: Ramsey		Driller:	D. Crump			
Consulting Firm: <u>Hanson Prot</u>	fessional Services Inc.	Geologist:	Rhonald W.	Hasenyage	r, LPG #196-0002	246
Drilling Method: Hollow stem	n auger	Drilling Fl	uid (Type):no	me		
Logged By: Kristen L. Thees	feld	Date Start	ed: 9/8/20	15 Dat	e Finished: 9/	8/2015
Report Form Completed By: St	uzanna L. Keim	– Date:	10/6/2015			
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)	
			(MSL)*	(BGS)	(
	T		626.64	-2.82	Top of Protective	Casing
			626.36	2.54	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>			623.82	0.00	Ground Surface	
			622.82	1.00	Top of Annular S	ealant
Type of Annular Sealant: <u>High-</u>	-solids bentonite				T T T T T	
Installation Method:	ie					
Setting Time: <u>>24 hours</u>		Σ			Static Water Leve (After Completion)	1
Type of Bentonite Seal Gran	nular Pellet Slurry					
Installation Method: Gravi	(choose one)		612 59	11 23	Ton of Seal	
Setting Time: 25 minutes					Top of Sear	
Setting Time. <u>23 minutes</u>	X	×	610.37	13.45	Top of Sand Pack	-
Type of Sand Pack: <u>Quartz Sar</u>	nd					
Grain Size: 10-20 (si	ieve size)		608.31		Top of Screen	
Installation Method:Gravi	ty			00.16		
Type of Backfill Material: n/a			<u>603.66</u> 603.53	20.16	Bottom of Screen Bottom of Well	
	(if applicable)					
Installation Method:			603.53 * Referenced to a	20.29 National Geoder	Bottom of Boreho	le
			CAS	SING MEA	SUREMENTS	2.0
WELL CON	STRUCTION MATERIALS		ameter of Boren	ble	(inches)	<u>8.0</u> 2.0
(Choose of	ne type of material for each area)	P	rotective Casing I	ength	(feet)	5.0
	· · · · · · · · · · · · · · · · · · ·	R	iser Pipe Length		(feet)	17.80
Protective Casing	SS304 SS316 PTFE PVC OTHER	R: Steel B	ottom of Screen to	o End Cap	(feet)	0.38
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	R: So	creen Length (19	st slot to last slo	t) (feet)	4.65
Riser Pipe Below W.T.	SS304 SS316 PTFE (PVC) OTHER		otal Length of Ca	sing	(feet)	22.83
Screen	SS304 SS316 PIFE PVC OTHE	s Se	creen Slot Size **		(inches)	0.010

Illinois Enviror	mental Protection	Agency				Wel	Completion	Report
Site #:	C	ounty: <u>Mon</u>	tgomery			W	/ell #:G4	401
Site Name: <u>Coffeen Power Sta</u>	ation					В	orehole #:	G401
State Plane Coordinate: X 872,510).6_Y_2,515,614.8_(o	r) Latitude:				Longitud	e:	
Surveyed By: Gary C. Rogers			IL Regis	stratio	on #: <u>035-0</u>	02957		
Drilling Contractor: Ramsey			Driller:	D.	Crump			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.		Geologi	ist:	Rhonald W.	Hasenyager	r, LPG #196-000	246
Drilling Method: <u>Hollow stem</u>	auger		Drilling	; Fluid	l (Type): <u>no</u>	ne		
Logged By: <u>Rhonald W. Hase</u>	nyager		Date Sta	arted:	9/14/20	<u>15</u> Dat	e Finished:9/2	14/2015
Report Form Completed By: <u>Su</u>	zanna L. Keim		Date:		10/7/2015			
ANNULAR SPA	CE DETAILS]	Elevations	Depths	(0.01 ft.)	
					(MSL)*	(BGS)	, ,	
					625.84	-2.81	Top of Protective	Casing
					625.57	-2.54	Top of Riser Pipe	;
Type of Surface Seal: <u>Concrete</u>		-		~	623.03	0.00	Ground Surface	
				/	621 33	1 70	Top of Annular S	ealant
Type of Annular Sealant: <u>Bento</u>	nite Chips	- 1			021.33		rop or runnalar o	ourunt
Installation Method: <u>Gravit</u>	у	_						
Setting Time:		_]	<u>z</u>				Static Water Leve (After Completion)	el
Type of Bentonite Seal Gran	ular Pellet Slurry							
Installation Method: Gravit	(choose one)		Ŭ.		n/a	n/a	Top of Seal	
Sotting Time: 25 minutes	y		\mathbf{X}		<u> </u>	a	Top of Sear	
Setting Time. <u>25 minutes</u>		- 🛱	×		610.12	12.91	Top of Sand Pack	C C
Type of Sand Pack:Quartz San	d	_						
Grain Size: 10-20 (sie	eve size)		╡│		608.67	14.36	Top of Screen	
Installation Method:Gravit	У	- 🗏			(0. 1.0.1	10.50		
Type of Backfill Material: n/a			₹		<u>604.24</u> 603.74	<u>18.79</u> 19.29	Bottom of Screen Bottom of Well	
	(if applicable)							
Installation Method:					603.73 * Referenced to a	19.30 National Geodet	Bottom of Boreho	ole
			[D.	CAS	SING MEA	SUREMENTS	8.0
WELL CONS	STRUCTION MATERIAL	S	-	Dian	f Riser Pipe	ne	(inches)	2.0
(Choose on	e type of material for each area)		-	Prote	ective Casing L	ength	(feet)	5.0
				Rise	r Pipe Length	0	(feet)	16.70
Protective Casing	SS304 SS316 PTFE P	VC OTHER: S	teel	Botte	om of Screen to	o End Cap	(feet)	0.50
Riser Pipe Above W.T.	SS304 SS316 PTFE P	VC OTHER:		Scre	en Length (1s	st slot to last slo	t) (feet)	4.63
Riser Pipe Below W.T.	SS304 SS316 PTFE P	VC OTHER:		Tota	l Length of Cas	sing	(feet)	21.83
Screen	58304 88316 PTFE P	VC OTHER:		Scree	en Slot Size **		(inches)	0.010

Illinois Enviror	mental Protection Agenc	сy		Wel	l Completion	Report
Site #:	County:	Montgomery		V	Vell #: G4	03
Site Name: <u>Coffeen Power Sta</u>	ation			B	orehole #:	G403
State Plane Coordinate: X 873,56	1.3 Y 2,514,616.6 (or) Latit	ude:		Longitud	le:	
Surveyed By: Gary C. Rogers		IL Regis	stration #: <u>035-0</u>	02957		
Drilling Contractor: Ramsey		Driller:	D. Crump			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	Geologis	st: <u>Rhonald W.</u>	Hasenyage	r, LPG #196-0002	246
Drilling Method: <u>Hollow stem</u>	auger	Drilling	Fluid (Type): <u>no</u>	ne		
Logged By: Kristen L. Theesf	eld	Date Sta	urted: 9/11/20	15 Dat	e Finished:9/1	1/2015
Report Form Completed By: Su	zanna L. Keim	Date:	10/7/2015			
ANNULAR SPA	CE DETAILS		Elevations	Depths	(0.01 ft.)	
			(MSL)*	(BGS)	(0000)	
			626.72	-2.91	Top of Protective	Casing
			626.47	2.66	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>			623.81	0.00	Ground Surface	
			621.81	2.00	Top of Annular S	ealant
Type of Annular Sealant: <u>Bento</u>	nite Chips					
Installation Method: <u>Gravit</u>	У					
Setting Time: <u>>24 hours</u>		Σ			Static Water Leve (After Completion)	1
Type of Bentonite Seal Gran	ular Pellet Slurry —					
Installation Method: n/a	(choose one)		n/a	n/a	Top of Seal	
Setting Time: n/a						
			612.64		Top of Sand Pack	-
Type of Sand Pack: <u>Quartz San</u>	d		(10.70	12.11	T 60	
Grain Size: 10-20 (sig	eve size)		610.70		Top of Screen	
Installation Method: <u>Gravit</u>	У		606.03	1778	Pottom of Screen	
Type of Backfill Material: <u>n/a</u>			605.66	18.15	Bottom of Well	
	(if applicable)			10.15		
Installation Method:			605.66 * Referenced to a	18.15 National Geode	Bottom of Boreho	le
			CAS	SING MEA	SUREMENTS	
		Γ	Diameter of Boreho		(inches)	8.0
WELL CONS	TRUCTION MATERIALS	-	ID of Riser Pipe		(inches)	2.0
(Choose of	_{JF} or material for each area		Protective Casing L	ength	(feet)	5.0
	Γ		Riser Pipe Length		(feet)	15.77
Protective Casing	SS304 SS316 PTFE PVC OTF	IER: Steel	Bottom of Screen to	o End Cap	(feet)	0.37
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTF	IER:	Screen Length (1s	st slot to last slo	ot) (feet)	4.67
Kisei Pipe Below W.1.	SS304 SS316 PTFE PVC OTF	IER:	Total Length of Ca	sing	(feet)	20.81
			Screen Slot Size **		(inches)	0.010

Illinois Environ	mental Protection Agency			Well	Completion	n Report
Site #:	County:	Iontgomery		W	Tell #:G4	404
Site Name: Coffeen Energy C	enter			В	orehole #:	SB22
State Plane Coordinate: X 2,516,397	7.9 Y 873,999.8 (or) Latitud	e:		Longitude	2:	
Surveyed By: <u>Darren E. Forgy</u>	,	IL Registr	ration #: <u>035-0</u>	03637		
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.	Driller:	A. Rachford			
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.	Geologist:	:, LPG #			
Drilling Method: <u>Hollow stem</u>	auger	Drilling F	luid (Type): <u>no</u>	one		
Logged By: <u>Rhonald W. Hase</u>	nyager	_ Date Start	ted:5/1/200	07 Date	e Finished: 5/	/1/2007
Report Form Completed By: <u>Rh</u>	onald W. Hasenyager	Date:	5/2/2007			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
			616.02	<u>-2.92</u>	Top of Protective	Casing
			615.77	-2.67	Top of Riser Pipe	2
Type of Surface Seal: <u>Concrete</u>			613.10	0.00	Ground Surface	
Type of Annular Sealant: Benton	nite chips		613.10	0.00	Top of Annular S	Sealant
Installation Method: gravity	, ,					
Setting Time: hours		Σ	611.03	2.07	Static Water Leve (After Completion)	el 5/10/2007
Type of Bentonite Seal Gram	ılar Pellet Slurry				(
Installation Method:	(choose one)	××	n/a	n/a	Top of Seal	
Setting Time:	X	X	608.05	5.05	Top of Sand Pack	x
Type of Sand Pack:Quartz sand	l					
Grain Size: 10/20 (sie	ve size)		_606.68_	6.42	Top of Screen	
Installation Method:gravity	,		601 93	11 17	Bottom of Screer	1
Type of Backfill Material: Form	ation sand (if applicable)		601.48	11.62	Bottom of Well	L.
Installation Method: <u>slough</u>			601.10 * Referenced to a		Bottom of Boreh	ole
			CAS	SING MEAS	SUREMENTS	80
WELL CONS	TRUCTION MATERIALS		D of Riser Pine	ne	(inches)	2.0
(Choose on	e type of material for each area)	P	rotective Casing I	ength	(incres)	5.0
		R	Liser Pipe Length		(feet)	9.09
Protective Casing	SS304 SS316 PTFE PVC OTHER	R: Steel B	Bottom of Screen to	o End Cap	(feet)	0.45
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	R: S	creen Length (1s	st slot to last slot	t) (feet)	4.75
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	т	otal Length of Ca	sing	(feet)	14.29

Screen

SS304

SS316

PTFE PVC OTHER:

 Screen Slot Size **

 **Hand-Slotted Well Screens Are Unacceptable

0.010

Illinois Enviror	nmental Protectio	n Agency			Wel	l Completion	Report
Site #:		County: <u>Mor</u>	ntgomery	7	V	Vell #: G4	05
Site Name: <u>Coffeen Energy C</u>	enter				В	orehole #:	SB21
State Plane Coordinate: X 2,515,33	5.7 Y 873,996.8	(or) Latitude:			Longitud	e:	
Surveyed By: <u>Darren E. Forg</u>	ý		IL Reg	istration #: <u>035-0</u>	03637		
Drilling Contractor: <u>Reynolds</u>	Drilling Corp.		Driller	A. Rachford			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.		Geolog	ist:, LPG #			
Drilling Method: <u>Hollow stem</u>	auger		Drilling	g Fluid (Type): <u>no</u>	one		
Logged By: <u>Rhonald W. Hase</u>	enyager		Date S	tarted: <u>5/1/20</u>	07 Dat	e Finished: <u>5/</u>	1/2007
Report Form Completed By: <u>RI</u>	nonald W. Hasenyager		Date:	5/2/2007			
ANNULAR SPA	CE DETAILS			Elevations (MSL)*	Depths (BGS)	(0.01 ft.)	
				624.04	3.14	Top of Protective	Casing
				623.78	2.88_	Top of Riser Pipe	
Type of Surface Seal: <u>Concrete</u>				620.90	0.00	Ground Surface	
				620.90	0.00	Top of Annular S	ealant
Type of Annular Sealant: <u>Bento</u>	nite chips	— 🎙	P			-	
Installation Method: <u>gravit</u>	y						
Setting Time: <u>>12 hours</u>			ℤ	619.67	1.23	Static Water Leve (After Completion)	:l 5/10/2007
Type of Bentonite Seal Gran	ular Pellet Slurry						
Installation Method:	(choose one)			n/a	n/a	Top of Seal	
Setting Time:				613.19	7.71	Top of Sand Pack	C
						1	
Type of Sand Pack: Quartz san	d	—		611.89	9.01	Top of Screen	
Installation Mathada	eve size)					-	
Installation Method: gravit	y			607.14	13.76	Bottom of Screen	
Type of Backfill Material:	(if anyliashla)			606.69	14.21	Bottom of Well	
Installation Mathod	(ii applicable)			606 60	14 21	Pottom of Porch	
]	* Referenced to a	National Geode	tic Datum	ne
					SING MEA	SUREMENTS	8.0
WELL CONS	STRUCTION MATERIA	ALS		Diameter of Boreho	ле	(inches)	2.0
(Cnoose or	ie type of material for each area)			Protective Casing I	ength	(feet)	5.0
	1			Riser Pipe Length		(feet)	11.89
Protective Casing	SS304 SS316 PTFE	PVC OTHER:	Steel	Bottom of Screen to	o End Cap	(feet)	0.45
Riser Pipe Above W.T.	SS304 SS316 PTFE	PVC OTHER:		Screen Length (1	st slot to last slo	ot) (feet)	4.75
Screen	SS304 SS310 PTFE SS304 SS316 PTFE	<u>PVC</u> OTHER:		Total Length of Ca	sing	(feet)	17.09
	1			Serven blot bize		(menes)	0.010

Illinois Environ	mental Protection Age	ncy			Well	Completion	Report	
Site #:	County:	Montgon	nery		W	/ell #:G4	106	
Site Name: <u>Coffeen Power Sta</u>	ation - Ash Pond 2				В	orehole #:	G406	
State Plane Coordinate: X 2,514,702	2.4 Y <u>872,521.3</u> (or) La	titude: <u>3</u>	<u></u>	3' 37.114"	Longitud	e: <u>-89°</u> 23	<u>54.628"</u>	
Surveyed By: <u>Gary C. Rogers</u>		IL I	IL Registration #:035-002957					
Drilling Contractor: <u>Bulldog D</u>	rilling, Inc.	Dri	iller:	J. Dittmaier				
Consulting Firm: <u>Hanson Profe</u>	essional Services Inc.	Geo	Geologist: Rhonald W. Hasenyager, LPG #196-000246					
Drilling Method: Hollow stem auger				uid (Type): <u>no</u>	ne			
Logged By: <u>Kristen L. Theesf</u>	eld	Dat	te Starte	ed: <u>8/19/20</u>	16 Date	e Finished: <u>8/</u>	9/2016	
Report Form Completed By:Su	zanna L. Keim	Dat	te:	8/24/2016				
ANNULAR SPA	CE DETAILS			Elevations	Depths	(0.01 ft.)		
			_	625.70	3.84	Top of Protective	Casing	
				625.36	3.50	Top of Riser Pipe	:	
Type of Surface Seal: <u>Concrete</u>				621.86	0.00	Ground Surface		
Type of Annular Seclent: High (colide bontonite			619.86	2.00	Top of Annular S	ealant	
Installation Method: Tremi	sonds bemonne							
Setting Time:24 hours		Σ				Static Water Leve	el	
Type of Bentonite Seal Gran	ular Pellet Slurry –					(And Completion)		
Installation Mathada Cravit	(choose one)			610 74	11.12	Top of Soci		
Satting Time: 20 minutes	y		×	_010.74_		Top of Sear		
Setting Time. <u>50 minutes</u>				609.65	12.21	Top of Sand Pack	2	
Type of Sand Pack: <u>Quartz San</u>	<u>d</u>			600 0 0	10.56			
Grain Size: 10-20 (sie	ve size)			608.30		Top of Screen		
Installation Method:Gravit	У			603 49	18 37	Bottom of Screen		
Type of Backfill Material: <u>n/a</u>	(if any limble)			603.11	18.75	Bottom of Well		
Installation Method:	(паррисаоне)			603.11	18.75	Bottom of Boreho	ole	
			_	* Referenced to a	National Geodet	ic Datum		
				CAS	SING MEAS	SUREMENTS		
			D	iameter of Boreho	ole	(inches)	8.0	
WELL CONS (Choose on	e type of material for each area)		IĽ	of Riser Pipe		(inches)	2.0	
			Pr	otective Casing L	ength	(feet)	5.0	
Protective Casing	SS304 SS316 PTFE PVC C	OTHER: Steel	$\left \frac{Ri}{Ri} \right $	ser Pipe Length	End Com	(feet)	17.06	
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC C	THER:		preen Length (1)	UENG Cap	t) (feet)	4.81	
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC C	OTHER:		otal Length of Cas	sing	(feet)	22.25	
Screen	SS304 SS316 PTFE PVC C	OTHER:	Sc	reen Slot Size **		(inches)	0.010	

Illinois Enviror	mental Protection Agency			Well	Completion	n Report
Site #:	County: <u>M</u>	ontgomery		W	Vell #:G	407
Site Name: <u>Coffeen Power Sta</u>	ation - Ash Pond 2			В	orehole #:	G407
State Plane Coordinate: X_2,513,70	5.9 Y <u>872,973.4</u> (or) Latitud	e: <u>39°</u>	3' 41.665"	Longitud	e: <u>-89°</u> <u>2</u>	4' 7.213"
Surveyed By: <u>Gary C. Rogers</u>		IL Regist	tration #: <u>035-0</u>	02957		
Drilling Contractor: <u>Bulldog D</u>	Drilling, Inc.	_ Driller: _	J. Dittmaier			
Consulting Firm: <u>Hanson Prof</u>	essional Services Inc.	_ Geologis	t: <u>Rhonald W.</u>	Hasenyager	r, LPG #196-000	0246
Drilling Method: <u>Hollow stem auger</u> Drilling Fluid (Type): <u>none</u>						
Logged By: <u>Kristen L. Theesf</u>	eld	_ Date Star	rted: <u>8/16/20</u>	<u>16</u> Dat	e Finished: <u>8</u> /	16/2016
Report Form Completed By: <u>Su</u>	izanna L. Keim	Date:	8/24/2016			
ANNULAR SPA	CE DETAILS		Elevations (MSL)*	Depths (BGS)	(0.01 ft.))
			621.70	3.35	Top of Protective	e Casing
			621.32	-2.97	Top of Riser Pip	e
Type of Surface Seal: <u>Concrete</u>			618.35	0.00	Ground Surface	
Type of Annular Sealant: High-	solids bentonite		616.35	2.00	Top of Annular	Sealant
Installation Method: Tremi	e					
Setting Time:24 hours	·				Static Water Lev	el
					(After Completion)	
Type of Bentonite Seal (<u>Gran</u>	ular Pellet Slurry (choose one)	YT.				
Installation Method: <u>Gravit</u>	y	$\overline{\mathbf{A}}$	607.50	10.85	Top of Seal	
Setting Time: <u>15 minutes</u>		×	605.50	12.85	Top of Sand Pac	k
Type of Sand Pack: <u>Quartz San</u>	d					
Grain Size: 10-20 (sig	eve size)		604.57	13.78	Top of Screen	
Installation Method:Gravit	у		500 74	10 (1		
Type of Backfill Material:Quar	tz Sand		<u> </u>	18.61	Bottom of Screen Bottom of Well	1
Installation Method: _ Gravit	(ii applicable)		_598.35_	_20.00	Bottom of Boreh	ole
			* Referenced to a	National Geodet	ic Datum	
		_	CAS	SING MEAS	SUREMENTS	
WELL CONS	STRUCTION MATERIALS	1	Diameter of Boreho	ole	(inches)	8.0
(Choose or	type of material for each area)		ID of Riser Pipe		(inches)	2.0
			Protective Casing L	ength	(feet)	5.0
Protective Casing	SS304 SS316 PTFE PVC OTHER	t: (Steel 1	Kiser Pipe Length Bottom of Screen to	End Can	(feet)	0.43
Riser Pipe Above W.T.	SS304 SS316 PTFE PVC OTHER	<u>د</u>	Screen Length (1s	st slot to last slo	t) (feet)	4.83
Riser Pipe Below W.T.	SS304 SS316 PTFE PVC OTHER	ε ·	Total Length of Cas	sina	(faat)	22.01

Screen

SS304

SS316

PTFE PVC OTHER:

**Hand-Slotted Well Screens Are Unacceptable

0.010

(inches)

Screen Slot Size **

APPENDIX B

STATISTICAL PROCEDURE FOR BACKGROUND

APPENDIX B STATISTICAL PROCEDURE FOR CALCULATION OF BACKGROUND Ash Pond 2 Groundwater Monitoring Plan

Coffeen Energy Center, Montgomery County, IL

Introduction

The purpose of the statistical calculations documented in this appendix is to determine the maximum background concentrations likely to occur upgradient of Ash Pond 2 within the Hagarstown Unit. High predicted background concentrations relative to the Illinois Class I groundwater quality standards may suggest that downgradient concentrations for those parameters in the Hagarstown are due to a background source.

The statistical analysis procedures used here are consistent with procedures described in the document: 2009 Unified Guidance. "Statistical Analysis of Groundwater Monitoring Data at RCRA Facilities—Unified Guidance," March 2009, EPA 530/R-09-2007 (USEPA, 2009).

Compliance Data Operations - Limit Calculations

The range of potential background concentrations was statistically determined using parametric and nonparametric tolerance intervals. Tolerance intervals were chosen rather than prediction intervals because a tolerance interval makes no assumption about the future number of samples, while a prediction interval assumes a finite, and known, future number of samples.

The flow diagram (Figure B-1) outlines the logic flow for calculation of limits. Background values were calculated using parametric tolerance intervals for normally distributed data, and non-parametric tolerance intervals for data with no underlying distribution or with non-detect frequencies greater than 50 percent. Parametric tolerance intervals were calculated at a 95 percent coverage rate and a Type I individual comparison error level of 0.01 (i.e., false positive rate). Parameters with 100 percent non-detects were handled with the upper tolerance limit being set to the last Reporting Limit (RL).

Statistical Data Evaluation and Results

The input dataset (attached to this summary) for background calculations were evaluated for the monitoring data from monitoring wells G270, G280, and G281 collected from all historical results, for a subset of the inorganic parameters listed in 35 IAC 620.410(a), specifically arsenic (total and dissolved), boron (total and dissolved), lead (total and dissolved), manganese (dissolved), sulfate, total dissolved solids, and pH (both upper and lower limits). Background concentrations for additional parameters will be calculated following collection of 8 rounds of data. All water quality data were stored, prepared, and statistically analyzed using MANAGES[™] Version 3.4.49 software (EPRI, March 2014).

A statistical summary of the background water quality data from G270, G280 and G281 including the mean, median, minimum, maximum, standard deviation, Sen Slope trend, normality determination, and percent non-detects for the background dataset. The statistical analysis procedure inputs and results are also provided in this Appendix.

Calculated background values for the tested inorganic constituents and pH are listed in the following Table B-1 along with the percent non-detects, normal or lognormal distribution, test method, and confidence level.





Notes

* If the option for Cohen's or Atchison's adjustment is selected and neither is appropriate, then the nonnormal comparison test will be used.

Parameter	Count of Background Results	Percent of Non Detects	Normal/ Lognormal	Test	Confidence Level	Upper Limit	Lower Limit
As (dissolved)	69	97.10	No/No	STlow2	97.10	0.001	
Boron (dissolved)	69	60.87	No/No	STlow1	97.10	0.12	
Lead (dissolved)	69	100.0	No/No	STmdl	N/A	0.001	
Manganese (diss)	69	23.19	No/No	STnon	97.10	0.48	
рН	80	0.00	Yes/Yes	STpar	99.00	7.88	6.64
Sulfate (total)	78	0.00	No/No	STnon	98.17	370	
TDS	77	0.00	No/No	STnon	98.70	820	

Table B-1. Tolerance Limits for Background Monitoring Wells G270, G280, and G281

* Key to Tests

STmdl = Comparison method if all background results are non-detect = Last MDL

STpar = Parametric Tolerance Interval on background

STIow1 = Non-Parametric Tolerance Interval on background (ND Frequency > 50%)

STnon = Non-Parametric Tolerance Interval on background

Coffeen Ash Pond 2 - Background Summary

			User S	Supplied Ir	nformation					
Date Range: 03/11/200	8 to 11/22/2016					Option	for LT Pts:	x 0.5		
Pooled Locations:	G270,G280,G281									
								Sen Slope	Normal /	% of
Parameter	Units	Count	Mean	Median	Maximum	Minimum	Std Dev	Units/yr	Log Normal	Non-Detects
As, diss	mg/L	69	0.001	0.001	0.005	0.001	0.001	0.000	No / No	97.10
B, diss	mg/L	69	0.025	0.010	0.250	0.005	0.038	-0.004	No / No	60.87
Mn, diss	mg/L	69	0.069	0.023	0.480	0.001	0.104	-0.027	No / No	23.19
Pb, diss	mg/L	69	0.001	0.001	0.003	0.000	0.000	0.000	No / No	100.00
pH (field)	STD	83	7.229	7.220	7.870	6.500	0.268	0.029	Yes / Yes	0.00
SO4, tot	mg/L	81	79.116	74.000	370.000	2.300	70.663	12.951	No / No	0.00
TDS	mg/L	80	468.750	445.000	820.000	340.000	101.612	5.988	No / No	0.00

Coffeen

Ash Pond 2- Statistical Analysis

279,G401,G402,G403,G404,G405,G406,G407
STmdl = Last MDL STpar = Parametric Tolerance Interval on Background STlow1 = Non-Parametric Tolerance Interval on background (ND Frequency > 50%) STlow2 = Non-Parametric Tolerance Interval on background (ND Frequency > 90%) STnon = Non-Parametric Tolerance Interval on background
Interwell
1 0.01
Mann-Kendall Trend Analysis
Compliance Date Range <=50% using MDL * 0.5 >50% using Aitchison's/Cohen's Decision
<=50% using MDL * 0.5 >50% using MDL * 0.5 95%

			Count	Percent								
Compliance			Of Bkg	of Non	Normal /		Confidence					
Location	Parameter	Sample Date	Results	detects	Lognormal	Test	Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	Trend

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance Trend
C151	A	02/10/2014	(0)	07.10	NT - /NT -	ST12	07.10	0.001		-0.001	N.
0151	As, diss, mg/L	02/19/2014	69	97.10	No/No	5110w2	97.10	0.001		<0.001	No
		03/12/2014	09	97.10	INO/INO		97.10	0.001		<0.001	No
		10/14/2014	69	97.10	NO/NO		97.10	0.001		<0.001	No
		10/14/2014	69	97.10	NO/NO		97.10	0.001		<0.001	No
		01/21/2015	69	97.10	No/No		97.10	0.001		<0.001	No
		04/08/2015	69	97.10	No/No		97.10	0.001		<0.002	No
		07/23/2015	69	97.10	No/No		97.10	0.001		< 0.001	No
		10/06/2015	69	97.10	No/No		97.10	0.001		< 0.001	No
		02/09/2016	69	97.10	No/No		97.10	0.001		< 0.001	No
		05/11/2016	69	97.10	No/No		97.10	0.001		< 0.002	No
G151	B, diss, mg/L	02/19/2014	69	60.87	No/No	STlow1	97.10	0.120		0.014	No
		05/12/2014	69	60.87	No/No		97.10	0.120		0.024	No
		08/11/2014	69	60.87	No/No		97.10	0.120		< 0.010	No
		10/14/2014	69	60.87	No/No		97.10	0.120		< 0.010	No
		01/21/2015	69	60.87	No/No		97.10	0.120		< 0.010	No
		04/08/2015	69	60.87	No/No		97.10	0.120		< 0.020	No
		07/23/2015	69	60.87	No/No		97.10	0.120		0.018	No
		10/06/2015	69	60.87	No/No		97.10	0.120		0.036	No
		02/09/2016	69	60.87	No/No		97.10	0.120		<0.010	No
		05/11/2016	69	60.87	No/No		97.10	0.120		< 0.020	No
G151	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		0.072	No
		05/12/2014	69	23.19	No/No		97.10	0.480		0.069	No
		08/11/2014	69	23.19	No/No		97.10	0.480		0.140	No
		10/14/2014	69	23.19	No/No		97.10	0.480		0.031	No
		01/21/2015	69	23.19	No/No		97.10	0.480		0.066	No
		04/08/2015	69	23.19	No/No		97.10	0.480		0.190	No
		07/23/2015	69	23.19	No/No		97.10	0.480		0.093	No

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
G151	Mn, diss, mg/L	10/06/2015	69	23.19	No/No	STnon	97.10	0.480		0.560	Yes	None
		02/09/2016	69	23.19	No/No		97.10	0.480		0.180	No	
		05/11/2016	69	23.19	No/No		97.10	0.480		0.011	No	
G151	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
	-	05/12/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/08/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/06/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/11/2016	69	100.00	No/No		N/A	0.001		< 0.002	No	
G151	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.38	No	
	F (),	05/12/2014	80	0.00	Yes/Yes	1	99.00	7.86	6.64	7.80	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.30	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.54	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.44	No	
		04/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.12	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.84	No	
		10/06/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.90	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.73	No	
		05/11/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.99	No	
G151	TDS mg/l	08/11/2014	77	0.00	No/No	STnor	08.07	820		560	No	
0131	TDS, IIIg/L	10/14/2014	ו ו דר	0.00	No/No	5111011	90.07	820		570	INO No	
		01/21/2014	יי רר	0.00	No/No		90.07	820		500	No	
		01/21/2013	ו ז רר	0.00	No/No		20.07	820 820		200	No	
		04/08/2015	11	0.00	10/10		90.07	820		000	INO	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance Tree	nd
G151	TDS, mg/L	07/23/2015	77	0.00	No/No	STnon	98.07	820		550	No	
		10/06/2015	77	0.00	No/No		98.07	820		600	No	
		02/09/2016	77	0.00	No/No		98.07	820		560	No	
		05/11/2016	77	0.00	No/No		98.07	820		500	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
C152	A a diag ang/I	02/10/2014	60	07.10	N. A.	STI2.	07.10	0.001		-0.001	Na	
0152	As, diss, mg/L	02/19/2014	69	97.10	No/No	5110w2	97.10	0.001		< 0.001	INO No	
		08/11/2014	69	97.10	No/No		97.10	0.001		< 0.001	INO No	
		10/14/2014	09	97.10	NO/INO		97.10	0.001		<0.001	INO N.	
		10/14/2014	69	97.10	No/No		97.10	0.001		< 0.001	INO No	
		01/21/2015	69	97.10	No/No		97.10	0.001		<0.001	INO No	
		04/08/2015	69	97.10	NO/NO		97.10	0.001		< 0.002	NO Vac	Nama
		10/06/2015	69	97.10	NO/NO		97.10	0.001		0.008	Yes	None
		10/00/2015	69	97.10	NO/NO		97.10	0.001		0.005	Yes	None
		02/09/2016	69	97.10	NO/NO		97.10	0.001		0.003	Yes	None
		05/11/2016	69	97.10	NO/INO		97.10	0.001		<0.002	No	
G152	B, diss, mg/L	02/19/2014	69	60.87	No/No	STlow1	97.10	0.120		0.038	No	
		05/12/2014	69	60.87	No/No		97.10	0.120		0.044	No	
		08/11/2014	69	60.87	No/No		97.10	0.120		0.044	No	
		10/14/2014	69	60.87	No/No		97.10	0.120		0.034	No	
		01/21/2015	69	60.87	No/No		97.10	0.120		0.030	No	
		04/08/2015	69	60.87	No/No		97.10	0.120		0.025	No	
		07/23/2015	69	60.87	No/No		97.10	0.120		0.079	No	
		10/06/2015	69	60.87	No/No		97.10	0.120		0.110	No	
		02/09/2016	69	60.87	No/No		97.10	0.120		0.082	No	
		05/11/2016	69	60.87	No/No		97.10	0.120		0.066	No	
C150		00/10/2014	60	22.10			07.10	0.400		0.100		
G152	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		0.120	No	
		05/12/2014	69	23.19	No/No		97.10	0.480		0.050	No	
		08/11/2014	69	23.19	No/No		97.10	0.480		0.310	No	
		10/14/2014	69	23.19	No/No		97.10	0.480		0.150	No	
		01/21/2015	69	23.19	No/No		97.10	0.480		0.150	No	
		04/08/2015	69	23.19	No/No		97.10	0.480		0.070	No	
		07/23/2015	69	23.19	No/No		97.10	0.480		0.890	Yes	None

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	ce Trend
G152	Mn, diss, mg/L	10/06/2015	69	23.19	No/No	STnon	97.10	0.480		0.530	Yes	None
		02/09/2016	69	23.19	No/No		97.10	0.480		0.470	No	
		05/11/2016	69	23.19	No/No		97.10	0.480		0.310	No	
G152	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		<0.001	No	
		05/12/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/08/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/06/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/11/2016	69	100.00	No/No		N/A	0.001		< 0.002	No	
G152	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.12	No	
	L	05/12/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.54	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.25	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.33	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.42	No	
		04/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.94	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.05	No	
		10/06/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.07	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.46	No	
		05/11/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.89	No	
		11/19/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.94	No	
G152	SO4, tot, mg/L	11/19/2016	78	0.00	No/No	STnon	98.17	370.0		130.0	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	ce Trend
G152	TDS, mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		820	No	
		10/14/2014	77	0.00	No/No		98.07	820		620	No	
		01/21/2015	77	0.00	No/No		98.07	820		620	No	
		04/08/2015	77	0.00	No/No		98.07	820		680	No	
		07/23/2015	77	0.00	No/No		98.07	820		730	No	
		10/06/2015	77	0.00	No/No		98.07	820		1,100	Yes	None
		02/09/2016	77	0.00	No/No		98.07	820		920	Yes	None
		05/11/2016	77	0.00	No/No		98.07	820		800	No	
		11/19/2016	77	0.00	No/No		98.07	820		680	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
C152	As disc made	02/10/2014	60	07.10	NIa /NIa	STI2002	07.10	0.001		-0.001	Na	
0155	As, diss, hig/L	02/19/2014	60	97.10	No/No	5110w2	97.10	0.001		< 0.001	No	
		08/11/2014	60	97.10	No/No		97.10	0.001		< 0.001	No	
		10/14/2014	60	97.10	No/No		97.10	0.001		< 0.001	NO	Nega
		01/21/2015	60	97.10	No/No		97.10	0.001		0.002	i es	None
		01/21/2015	69	97.10	No/No		97.10	0.001		< 0.001	NO	Nega
		04/08/2013	09	97.10	INO/INO		97.10	0.001		0.003	res	None
		10/06/2015	69	97.10	NO/NO		97.10	0.001		<0.001	NO	
		10/00/2015	69	97.10	NO/NO		97.10	0.001		<0.001	NO	
		02/09/2016	69	97.10	No/No		97.10	0.001		<0.001	No	N
		05/11/2016	69	97.10	No/No		97.10	0.001		0.004	Yes	None
G153	B, diss, mg/L	02/19/2014	69	60.87	No/No	STlow1	97.10	0.120		0.042	No	
		05/12/2014	69	60.87	No/No		97.10	0.120		0.021	No	
		08/11/2014	69	60.87	No/No		97.10	0.120		0.026	No	
		10/14/2014	69	60.87	No/No		97.10	0.120		0.028	No	
		01/21/2015	69	60.87	No/No		97.10	0.120		0.020	No	
		04/08/2015	69	60.87	No/No		97.10	0.120		0.020	No	
		07/23/2015	69	60.87	No/No		97.10	0.120		0.015	No	
		10/06/2015	69	60.87	No/No		97.10	0.120		0.030	No	
		02/09/2016	69	60.87	No/No		97.10	0.120		0.016	No	
		05/11/2016	69	60.87	No/No		97.10	0.120		0.026	No	
G153	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		0.160	No	
		05/12/2014	69	23.19	No/No		97.10	0.480		0.050	No	
		08/11/2014	69	23.19	No/No		97.10	0.480		0.240	No	
		10/14/2014	69	23.19	No/No		97.10	0.480		0.510	Yes	None
		01/21/2015	69	23.19	No/No		97.10	0.480		0.200	No	
		04/08/2015	69	23.19	No/No		97.10	0.480		0.024	No	
		07/23/2015	69	23.19	No/No		97.10	0.480		0.160	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	e Trend
G153	Mn, diss, mg/L	10/06/2015	69	23.19	No/No	STnon	97.10	0.480		0.190	No	
		02/09/2016	69	23.19	No/No		97.10	0.480		0.046	No	
		05/11/2016	69	23.19	No/No		97.10	0.480		0.037	No	
G153	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
		05/12/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/08/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/06/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/11/2016	69	100.00	No/No		N/A	0.001		< 0.002	No	
G153	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	6.91	No	
		05/12/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	6.98	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	6.95	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	6.97	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.85	No	
		04/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.52	Yes	None
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.16	No	
		10/06/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.25	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.07	No	
		05/11/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.97	No	
		11/19/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.20	No	
G153	SO4, tot, mg/L	11/19/2016	78	0.00	No/No	STnon	98.17	370.0		2,500.0	Yes	Insufficient Data Cannot Trend.

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	ce Trend
G153	TDS, mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		3,500	Yes	Upward
		10/14/2014	77	0.00	No/No		98.07	820		3,400	Yes	Upward
		01/21/2015	77	0.00	No/No		98.07	820		3,600	Yes	Upward
		04/08/2015	77	0.00	No/No		98.07	820		3,700	Yes	Upward
		07/23/2015	77	0.00	No/No		98.07	820		3,900	Yes	Upward
		10/06/2015	77	0.00	No/No		98.07	820		3,700	Yes	Upward
		02/09/2016	77	0.00	No/No		98.07	820		3,800	Yes	Upward
		05/11/2016	77	0.00	No/No		98.07	820		3,800	Yes	Upward
		11/19/2016	77	0.00	No/No		98.07	820		4,000	Yes	Upward

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance Trend
C154	As disc made	02/10/2014	60	07.10	NIa /NIa	STI2002	07.10	0.001		-0.001	Na
0154	As, diss, mg/L	02/19/2014	69	97.10	No/No	5110w2	97.10	0.001		<0.001	No
		03/12/2014	09	97.10	INO/INO		97.10	0.001		<0.001	NO
		10/14/2014	69	97.10	NO/NO		97.10	0.001		<0.001	NO
		10/14/2014	69	97.10	NO/NO		97.10	0.001		< 0.001	NO
		01/21/2015	69	97.10	No/No		97.10	0.001		<0.001	No
		04/08/2015	69	97.10	No/No		97.10	0.001		< 0.002	No
		07/23/2015	69	97.10	No/No		97.10	0.001		<0.001	No
		10/06/2015	69	97.10	No/No		97.10	0.001		< 0.001	No
		02/09/2016	69	97.10	No/No		97.10	0.001		< 0.001	No
		05/11/2016	69	97.10	No/No		97.10	0.001		< 0.002	No
G154	B, diss, mg/L	02/19/2014	69	60.87	No/No	STlow1	97.10	0.120		0.043	No
	-	05/12/2014	69	60.87	No/No		97.10	0.120		0.044	No
		08/11/2014	69	60.87	No/No		97.10	0.120		0.035	No
		10/14/2014	69	60.87	No/No		97.10	0.120		0.035	No
		01/21/2015	69	60.87	No/No		97.10	0.120		0.037	No
		04/08/2015	69	60.87	No/No		97.10	0.120		0.023	No
		07/23/2015	69	60.87	No/No		97.10	0.120		0.024	No
		10/06/2015	69	60.87	No/No		97.10	0.120		0.044	No
		02/09/2016	69	60.87	No/No		97.10	0.120		0.041	No
		05/11/2016	69	60.87	No/No		97.10	0.120		0.037	No
G154	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		0.018	No
		05/12/2014	69	23.19	No/No		97.10	0.480		0.014	No
		08/11/2014	69	23.19	No/No		97.10	0.480		0.081	No
		10/14/2014	69	23.19	No/No		97.10	0.480		0.008	No
		01/21/2015	69	23.19	No/No		97.10	0.480		0.009	No
		04/08/2015	69	23.19	No/No		97.10	0.480		0.003	No
		07/23/2015	69	23.19	No/No		97.10	0.480		0.140	No

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	e Trend
G154	Mn, diss, mg/L	10/06/2015	69	23.19	No/No	STnon	97.10	0.480		0.010	No	
		02/09/2016	69	23.19	No/No		97.10	0.480		0.041	No	
		05/11/2016	69	23.19	No/No		97.10	0.480		0.009	No	
G154	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		<0.001	No	
	, , , ,	05/12/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/08/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/06/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/11/2016	69	100.00	No/No		N/A	0.001		< 0.002	No	
G154	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.45	No	
	pii (iiiiii), 512	05/12/2014	80	0.00	Yes/Yes	1	99.00	7.86	6.64	7.91	Yes	None
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.65	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.64	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.55	No	
		04/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.25	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.62	Yes	None
		10/06/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.93	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	8.03	Yes	None
		05/11/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.00	No	
G154	TDS. mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		480	No	
	··· , G	10/14/2014	77	0.00	No/No		98.07	820		440	No	
		01/21/2015	77	0.00	No/No		98.07	820		450	No	
		04/08/2015	77	0.00	No/No		98.07	820		440	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance Tree	nd
G154	TDS, mg/L	07/23/2015	77	0.00	No/No	STnon	98.07	820		430	No	
		10/06/2015	77	0.00	No/No		98.07	820		500	No	
		02/09/2016	77	0.00	No/No		98.07	820		560	No	
		05/11/2016	77	0.00	No/No		98.07	820		460	No	

Compliance Count Percent Compliance Of Bkg of Non Normal / Confidence Location Parameter Sample Date Results detects Lognormal Test Level Upper Limit Lower Limit Analysis F	sult Exceedance Trend
G271 As, diss, mg/L 02/19/2014 69 97.10 No/No STIow2 97.10 0.001 <0	01 No
05/13/2014 69 97.10 No/No 97.10 0.001 0.	60 Yes None
08/11/2014 69 97.10 No/No 97.10 0.001 <0	01 No
10/14/2014 69 97.10 No/No 97.10 0.001 <0	01 No
01/21/2015 69 97.10 No/No 97.10 0.001 <0	01 No
04/10/2015 69 97.10 No/No 97.10 0.001 <0.	02 No
07/22/2015 69 97.10 No/No 97.10 0.001 <0	01 No
10/08/2015 69 97.10 No/No 97.10 0.001 <0	01 No
02/16/2016 69 97.10 No/No 97.10 0.001 <0	01 No
05/12/2016 69 97.10 No/No 97.10 0.001 <0.	02 No
G271 B disc mg/L $02/19/2014$ 69 60.87 No/No STlow1 07.10 0.120 0	60 Vac Nona
0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.120 0.1	50 Yes None
03/13/2014 60 60.87 No/No 97.10 0.120 0.	00 Vec None
10/14/2014 69 60.87 NO/NO 97.10 0.120 0.	40 Yes None
10/14/2014 09 00.87 NO/NO 97.10 0.120 0.	40 Yes None
01/21/2015 69 00.87 No/No 97.10 0.120 0.	20 Yes None
04/10/2015 69 60.87 No/No 97.10 0.120 0.	70 Yes None
07/22/2015 69 60.87 No/No 97.10 0.120 0.	20 Yes None
10/08/2015 69 60.87 No/No 97.10 0.120 0.	40 Yes None
02/16/2016 69 60.87 No/No 97.10 0.120 0.	10 Yes None
05/12/2016 69 60.87 No/No 97.10 0.120 0.	10 Yes None
G271 Mn, diss, mg/L 02/19/2014 69 23.19 No/No STnon 97.10 0.480 <0	001 No
05/13/2014 69 23.19 No/No 97.10 0.480 0	20 No
08/11/2014 69 23.19 No/No 97.10 0.480 <0	001 No
10/14/2014 69 23.19 No/No 97.10 0.480 0	02 No
01/21/2015 69 23.19 No/No 97.10 0.480 <0	001 No
04/10/2015 69 23.19 No/No 97.10 0.480 <0	02 No

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
G271	Mn, diss, mg/L	10/08/2015	69	23.19	No/No	STnon	97.10	0.480		< 0.001	No	
		02/16/2016	69	23.19	No/No		97.10	0.480		< 0.001	No	
		05/12/2016	69	23.19	No/No		97.10	0.480		< 0.002	No	
G271	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		<0.001	No	
	-	05/13/2014	69	100.00	No/No		N/A	0.001		0.150	Yes	None
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/10/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/22/2015	69	100.00	No/No		N/A	0.001		0.002	Yes	None
		10/08/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/16/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/12/2016	69	100.00	No/No		N/A	0.001		< 0.002	No	
G271	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.35	No	
	1	05/13/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	6.80	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.46	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.42	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.49	No	
		04/10/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.28	No	
		07/22/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.20	No	
		10/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.96	No	
		11/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.27	No	
		02/16/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.47	No	
		02/16/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.47	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.19	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.19	No	
		08/05/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.20	No	
		11/21/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.24	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	ce Trend
G271	SO4 tot mg/L	02/19/2014	78	0.00	No/No	STnon	98 17	370.0		420.0	Yes	None
	50 i, ioi, ing 1	05/13/2014	78	0.00	No/No		98.17	370.0		440.0	Yes	None
		08/11/2014	78	0.00	No/No		98.17	370.0		500.0	Yes	None
		10/14/2014	78	0.00	No/No		98.17	370.0		480.0	Yes	None
		01/21/2015	78	0.00	No/No		98.17	370.0		490.0	Yes	None
		04/10/2015	78	0.00	No/No		98.17	370.0		440.0	Yes	None
		07/22/2015	78	0.00	No/No		98.17	370.0		350.0	No	
		10/08/2015	78	0.00	No/No		98.17	370.0		400.0	Yes	None
		11/23/2015	78	0.00	No/No		98.17	370.0		420.0	Yes	None
		02/16/2016	78	0.00	No/No		98.17	370.0		440.0	Yes	None
		02/16/2016	78	0.00	No/No		98.17	370.0		360.0	No	
		05/12/2016	78	0.00	No/No		98.17	370.0		540.0	Yes	None
		05/12/2016	78	0.00	No/No		98.17	370.0		420.0	Yes	None
		08/05/2016	78	0.00	No/No		98.17	370.0		440.0	Yes	None
		11/21/2016	78	0.00	No/No		98.17	370.0		400.0	Yes	None
G271	TDS mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		1 000	Yes	None
	100, 116/1	10/14/2014	77	0.00	No/No		98.07	820		940	Yes	None
		01/21/2015	77	0.00	No/No		98.07	820		870	Yes	None
		04/10/2015	77	0.00	No/No		98.07	820		1.000	Yes	None
		07/22/2015	77	0.00	No/No		98.07	820		1.000	Yes	None
		10/08/2015	77	0.00	No/No		98.07	820		1.000	Yes	None
		11/23/2015	77	0.00	No/No		98.07	820		860	Yes	None
		02/16/2016	77	0.00	No/No		98.07	820		1,000	Yes	None
		02/16/2016	77	0.00	No/No		98.07	820		980	Yes	None
		05/12/2016	77	0.00	No/No		98.07	820		940	Yes	None
		05/12/2016	77	0.00	No/No		98.07	820		1,000	Yes	None
		08/05/2016	77	0.00	No/No		98.07	820		840	Yes	None
		11/21/2016	77	0.00	No/No		98.07	820		910	Yes	None

			Count	Percent								
Compliance			Of Bkg	of Non	Normal /		Confidence					
Location	Parameter	Sample Date	Results	detects	Lognormal	Test	Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	Trend

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	e Trend
6252	A 1' T	02/10/2014	(0)	07.10			07.10	0.001		0.001	N	
G272	As, diss, mg/L	02/19/2014	69	97.10	No/No	STIow2	97.10	0.001		< 0.001	No	
		05/13/2014	69	97.10	No/No		97.10	0.001		0.170	Yes	None
		08/11/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/14/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		01/21/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		04/10/2015	69	97.10	No/No		97.10	0.001		< 0.002	No	
		07/23/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/08/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		02/09/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
		05/12/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
G272	B diss mg/L	02/19/2014	69	60.87	No/No	STlow1	97 10	0 120		<0.010	No	
	<i>D</i> , 0100, 111 <i>g</i> , <i>D</i>	05/13/2014	69	60.87	No/No		97.10	0.120		0.270	Yes	None
		08/11/2014	69	60.87	No/No		97.10	0.120		<0.010	No	ivone
		10/14/2014	69	60.87	No/No		97.10	0.120		0.036	No	
		01/21/2015	69	60.87	No/No		97.10	0.120		<0.030	No	
		04/10/2015	69	60.87	No/No		97.10	0.120		<0.010	No	
		07/23/2015	69	60.87	No/No		97.10 97.10	0.120		<0.020	No	
		10/08/2015	69	60.87	No/No		97.10	0.120		<0.010	No	
		02/09/2016	69	60.87	No/No		97.10	0.120		<0.010	No	
		05/12/2016	69	60.87	No/No		97.10 97.10	0.120		<0.010	No	
G272	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		< 0.001	No	
		05/13/2014	69	23.19	No/No		97.10	0.480		0.230	No	
		08/11/2014	69	23.19	No/No		97.10	0.480		< 0.001	No	
		10/14/2014	69	23.19	No/No		97.10	0.480		< 0.001	No	
		01/21/2015	69	23.19	No/No		97.10	0.480		< 0.001	No	
		04/10/2015	69	23.19	No/No		97.10	0.480		< 0.002	No	
		07/23/2015	69	23.19	No/No		97.10	0.480		< 0.001	No	
Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	e Trend
------------------------	---------------------	-------------	----------------------------	------------------------------	-----------------------	-------	---------------------	-------------	-------------	-----------------	------------	---------
G272	Mn, diss, mg/L	10/08/2015	69	23.19	No/No	STnon	97.10	0.480		< 0.001	No	
		02/09/2016	69	23.19	No/No		97.10	0.480		0.001	No	
		05/12/2016	69	23.19	No/No		97.10	0.480		< 0.001	No	
G272	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
		05/13/2014	69	100.00	No/No		N/A	0.001		0.150	Yes	None
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/10/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/08/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/12/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
G272	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.75	No	
	p11 (11010); 512	05/13/2014	80	0.00	Yes/Yes	1	99.00	7.86	6.64	7.84	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.61	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.43	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.64	No	
		04/10/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.18	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.22	No	
		10/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.21	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.54	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.25	No	
G272	SOA tot $ma^{/I}$	02/19/2014	78	0.00	No/No	STron	98 17	370.0		340.0	No	
5212	504, 101, 111g/L	05/13/2014	78	0.00	No/No	51101	90.17	370.0		340.0 310.0	No	
		08/11/2014	78	0.00	No/No		90.17	370.0		310.0	No	
		10/14/2014	78	0.00	No/No		98.17	370.0		310.0	No	
		10/14/2014	78	0.00	110/110		90.17	570.0		510.0	INU	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	ce Trend
G272	SO4, tot, mg/L	01/21/2015	78	0.00	No/No	STnon	98.17	370.0		380.0	Yes	None
		04/10/2015	78	0.00	No/No		98.17	370.0		340.0	No	
		07/23/2015	78	0.00	No/No		98.17	370.0		270.0	No	
		10/08/2015	78	0.00	No/No		98.17	370.0		340.0	No	
		02/09/2016	78	0.00	No/No		98.17	370.0		290.0	No	
		05/12/2016	78	0.00	No/No		98.17	370.0		310.0	No	
G272	TDS, mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		740	No	
		10/14/2014	77	0.00	No/No		98.07	820		840	Yes	None
		01/21/2015	77	0.00	No/No		98.07	820		790	No	
		04/10/2015	77	0.00	No/No		98.07	820		800	No	
		07/23/2015	77	0.00	No/No		98.07	820		840	Yes	None
		10/08/2015	77	0.00	No/No		98.07	820		660	No	
		02/09/2016	77	0.00	No/No		98.07	820		660	No	
		05/12/2016	77	0.00	No/No		98.07	820		680	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
0072	A 11 T	02/10/2014	(0)	07.10			07.10	0.001		0.001	N	
6275	As, diss, mg/L	02/19/2014	69	97.10	NO/NO	5110W2	97.10	0.001		< 0.001	NO	N
		05/13/2014	69	97.10	NO/NO		97.10	0.001		0.240	res	None
		08/11/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/14/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		01/21/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		04/13/2015	69	97.10	No/No		97.10	0.001		< 0.002	No	
		07/23/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/08/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		02/16/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
		05/12/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
G273	B diss mg/L	02/19/2014	69	60.87	No/No	STlow1	97 10	0 120		0 520	Yes	None
	D, 0155, 119/12	05/13/2014	69	60.87	No/No		97.10	0.120		0.520	Yes	None
		08/11/2014	69	60.87	No/No		97.10	0.120		0.250	Yes	None
		10/14/2014	69	60.87	No/No		97.10	0.120		0.230	Ves	None
		01/21/2015	69	60.87	No/No		97.10 97.10	0.120		0.520	Ves	None
		04/13/2015	69	60.87	No/No		97.10 97.10	0.120		0.380	Ves	None
		07/23/2015	69	60.87	No/No		97.10 07.10	0.120		0.290	Vos	None
		10/08/2015	69	60.87	No/No		97.10	0.120		0.400	No	None
		02/16/2016	69	60.87	No/No		97.10	0.120		0.100	No	None
		05/12/2016	69	60.87 60.87	No/No		97.10 97.10	0.120		0.430	Yes	None
G273	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		0.015	No	
		05/13/2014	69	23.19	No/No		97.10	0.480		0.330	No	
		08/11/2014	69	23.19	No/No		97.10	0.480		0.010	No	
		10/14/2014	69	23.19	No/No		97.10	0.480		0.013	No	
		01/21/2015	69	23.19	No/No		97.10	0.480		0.018	No	
		04/13/2015	69	23.19	No/No		97.10	0.480		0.019	No	
		07/23/2015	69	23.19	No/No		97.10	0.480		0.001	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	e Trend
G273	Mn, diss, mg/L	10/08/2015	69	23.19	No/No	STnon	97.10	0.480		0.014	No	
		02/16/2016	69	23.19	No/No		97.10	0.480		0.010	No	
		05/12/2016	69	23.19	No/No		97.10	0.480		0.012	No	
G273	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
		05/13/2014	69	100.00	No/No		N/A	0.001		0.220	Yes	None
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/13/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/08/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/16/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/12/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
G273	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.40	No	
	• · ·	05/13/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.76	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.34	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.27	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.28	No	
		04/13/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.96	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.19	No	
		10/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.36	No	
		11/24/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	1.08	Yes	None
		02/16/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.23	No	
		02/16/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.23	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.04	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.04	No	
		08/05/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.14	No	
		11/21/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.26	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	ce Trend
G273	SO4. tot. mg/L	02/19/2014	78	0.00	No/No	STnon	98.17	370.0		570.0	Yes	None
		05/13/2014	78	0.00	No/No		98.17	370.0		620.0	Yes	None
		08/11/2014	78	0.00	No/No		98.17	370.0		530.0	Yes	None
		10/14/2014	78	0.00	No/No		98.17	370.0		500.0	Yes	None
		01/21/2015	78	0.00	No/No		98.17	370.0		650.0	Yes	None
		04/13/2015	78	0.00	No/No		98.17	370.0		690.0	Yes	None
		07/23/2015	78	0.00	No/No		98.17	370.0		390.0	Yes	None
		10/08/2015	78	0.00	No/No		98.17	370.0		450.0	Yes	None
		11/24/2015	78	0.00	No/No		98.17	370.0		420.0	Yes	None
		02/16/2016	78	0.00	No/No		98.17	370.0		550.0	Yes	None
		02/16/2016	78	0.00	No/No		98.17	370.0		530.0	Yes	None
		05/12/2016	78	0.00	No/No		98.17	370.0		520.0	Yes	None
		05/12/2016	78	0.00	No/No		98.17	370.0		500.0	Yes	None
		08/05/2016	78	0.00	No/No		98.17	370.0		400.0	Yes	None
		11/21/2016	78	0.00	No/No		98.17	370.0		440.0	Yes	None
G273	TDS, mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		1,000	Yes	None
		10/14/2014	77	0.00	No/No		98.07	820		1,100	Yes	None
		01/21/2015	77	0.00	No/No		98.07	820		1,200	Yes	None
		04/13/2015	77	0.00	No/No		98.07	820		1,300	Yes	None
		07/23/2015	77	0.00	No/No		98.07	820		1,200	Yes	None
		10/08/2015	77	0.00	No/No		98.07	820		930	Yes	None
		11/24/2015	77	0.00	No/No		98.07	820		890	Yes	None
		02/16/2016	77	0.00	No/No		98.07	820		1,100	Yes	None
		02/16/2016	77	0.00	No/No		98.07	820		1,200	Yes	None
		05/12/2016	77	0.00	No/No		98.07	820		980	Yes	None
		05/12/2016	77	0.00	No/No		98.07	820		1,100	Yes	None
		08/05/2016	77	0.00	No/No		98.07	820		840	Yes	None
		11/21/2016	77	0.00	No/No		98.07	820		900	Yes	None

			Count	Percent								
Compliance			Of Bkg	of Non	Normal /		Confidence					
Location	Parameter	Sample Date	Results	detects	Lognormal	Test	Level	Upper Limit	Lower Limit	Analysis Result	Exceedance T	Frend

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
G274	As, diss, mg/L	02/19/2014	69	97.10	No/No	STlow2	97.10	0.001		< 0.001	No	
		05/13/2014	69	97.10	No/No		97.10	0.001		0.006	Yes	None
		08/11/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/14/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		01/21/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		04/13/2015	69	97.10	No/No		97.10	0.001		< 0.002	No	
		07/23/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/08/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		02/09/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
		05/12/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
G274	P dice ma/I	02/19/2014	69	60.87	No/No	STlow1	07.10	0.120		0 700	Vas	Nono
0274	D, uiss, iiig/L	05/13/2014	69	60.87	No/No	5110w1	97.10	0.120		0.790	Vac	None
		08/11/2014	60	60.87	No/No		97.10	0.120		0.320	I es	None
		10/14/2014	60	60.87	INO/INO		97.10	0.120		0.570	res	None
		10/14/2014	09	60.87	NO/NO		97.10	0.120		0.620	res	None
		01/21/2015	69	00.87	No/No		97.10	0.120		0.680	Yes	None
		04/13/2015	69	60.87	No/No		97.10	0.120		0.570	Yes	None
		07/23/2015	69	60.87	No/No		97.10	0.120		0.480	Yes	None
		10/08/2015	69	60.87	No/No		97.10	0.120		0.430	Yes	None
		02/09/2016	69	60.87	No/No		97.10	0.120		0.870	Yes	None
		05/12/2016	69	60.87	No/No		97.10	0.120		0.970	Yes	None
G274	Mn, diss, mg/L	02/19/2014	69	23.19	No/No	STnon	97.10	0.480		< 0.001	No	
		05/13/2014	69	23.19	No/No		97.10	0.480		0.006	No	
		08/11/2014	69	23.19	No/No		97.10	0.480		< 0.001	No	
		10/14/2014	69	23.19	No/No		97.10	0.480		< 0.001	No	
		01/21/2015	69	23.19	No/No		97.10	0.480		0.004	No	
		01/21/2015 04/13/2015	69 69	23.19 23.19	No/No No/No		97.10 97.10	0.480 0.480		0.004 <0.002	No No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	e Trend
G274	Mn, diss, mg/L	10/08/2015	69	23.19	No/No	STnon	97.10	0.480		< 0.001	No	
		02/09/2016	69	23.19	No/No		97.10	0.480		< 0.001	No	
		05/12/2016	69	23.19	No/No		97.10	0.480		0.046	No	
G274	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
		05/13/2014	69	100.00	No/No		N/A	0.001		0.003	Yes	None
		08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/13/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/08/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/12/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
G274	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.53	No	
		05/13/2014	80	0.00	Yes/Yes	I.	99.00	7.86	6.64	7.61	No	
		08/11/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.40	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.23	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.33	No	
		04/13/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.84	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.30	No	
		10/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.17	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.43	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.35	No	
G274	SO4. tot. mg/L	02/19/2014	78	0.00	No/No	STnon	98.17	370.0		300.0	No	
	50 i, ioi, iiig L	05/13/2014	78	0.00	No/No		98.17	370.0		370.0	No	
		08/11/2014	78	0.00	No/No		98.17	370.0		400.0	Yes	None
		10/14/2014	78	0.00	No/No		98.17	370.0		320.0	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	e Trend
G274	SO4, tot, mg/L	01/21/2015	78	0.00	No/No	STnon	98.17	370.0		260.0	No	
		04/13/2015	78	0.00	No/No		98.17	370.0		390.0	Yes	None
		07/23/2015	78	0.00	No/No		98.17	370.0		320.0	No	
		10/08/2015	78	0.00	No/No		98.17	370.0		320.0	No	
		02/09/2016	78	0.00	No/No		98.17	370.0		290.0	No	
		05/12/2016	78	0.00	No/No		98.17	370.0		350.0	No	
G274	TDS, mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		880	Yes	None
		10/14/2014	77	0.00	No/No		98.07	820		770	No	
		01/21/2015	77	0.00	No/No		98.07	820		770	No	
		04/13/2015	77	0.00	No/No		98.07	820		770	No	
		07/23/2015	77	0.00	No/No		98.07	820		890	Yes	None
		10/08/2015	77	0.00	No/No		98.07	820		770	No	
		02/09/2016	77	0.00	No/No		98.07	820		820	No	
		05/12/2016	77	0.00	No/No		98.07	820		770	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	ce Trend
G275	As, diss, mg/L	05/13/2014	69	97.10	No/No	STlow2	97.10	0.001		0.250	Yes	None
	-	08/11/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		10/14/2014	69	97.10	No/No		97.10	0.001		< 0.001	No	
		01/21/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		04/13/2015	69	97.10	No/No		97.10	0.001		< 0.002	No	
		07/23/2015	69	97.10	No/No		97.10	0.001		< 0.001	No	
		02/09/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
		05/12/2016	69	97.10	No/No		97.10	0.001		< 0.001	No	
G275	P diss mg/	05/13/2014	69	60.87	No/No	STlow1	07.10	0.120		2 800	Vac	None
0275	D, diss, liig/L	08/11/2014	69	60.87	No/No	5110W1	97.10	0.120		2.800	Vos	None
		10/14/2014	69	60.87	No/No		97.10 97.10	0.120		4.200	Ves	None
		01/21/2015	69	60.87	No/No		97.10 97.10	0.120		1.900	Ves	None
		04/13/2015	69	60.87	No/No		97.10 97.10	0.120		4.000	Ves	None
		07/23/2015	69	60.87	No/No		97.10 97.10	0.120		3 000	Ves	None
		02/09/2016	69	60.87	No/No		97.10 97.10	0.120		4 000	Ves	None
		05/12/2016	69	60.87	No/No		97.10	0.120		2.500	Yes	None
C275	Mn dias mg/l	05/12/2014	60	22 10	No/No	STnon	07.10	0.480		0.240	No	
0275	Min, diss, ing/L	08/11/2014	60	23.19	No/No	5111011	97.10	0.480		0.540	No	
		10/14/2014	69	23.19	No/No		97.10	0.480		< 0.001	No	
		01/21/2015	69	23.17	No/No		97.10	0.480		0.005	No	
		01/21/2015	69	23.17	No/No		97.10	0.480		0.023	No	
		07/23/2015	69	23.17	No/No		97.10 97.10	0.480		0.002	No	
		02/09/2016	69	23.17	No/No		97.10 97.10	0.480		0.012	No	
		05/12/2016	69	23.19	No/No		97.10 97.10	0.480		0.000	No	
G275	Pb, diss, mg/L	05/13/2014	69	100.00	No/No	STmdl	N/A	0.001		0.200	Yes	None
	-	08/11/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	e Trend
G275	Pb, diss, mg/L	10/14/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/13/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/09/2016	69	100.00	No/No		N/A	0.001		0.004	Yes	None
		05/12/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
G275	pH (field), STD	05/13/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.72	No	
	F (), ~	08/11/2014	80	0.00	Yes/Yes	I	99.00	7.86	6.64	7.11	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	6.81	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.02	No	
		04/13/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.68	No	
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.07	No	
		02/09/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.24	No	
		05/12/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.09	No	
G275	SO4, tot, mg/L	05/13/2014	78	0.00	No/No	STnon	98.17	370.0		750.0	Yes	None
		08/11/2014	78	0.00	No/No		98.17	370.0		880.0	Yes	None
		10/14/2014	78	0.00	No/No		98.17	370.0		500.0	Yes	None
		01/21/2015	78	0.00	No/No		98.17	370.0		940.0	Yes	None
		04/13/2015	78	0.00	No/No		98.17	370.0		650.0	Yes	None
		07/23/2015	78	0.00	No/No		98.17	370.0		750.0	Yes	None
		02/09/2016	78	0.00	No/No		98.17	370.0		470.0	Yes	None
		05/12/2016	78	0.00	No/No		98.17	370.0		310.0	No	
G275	TDS. mg/L	08/11/2014	77	0.00	No/No	STnon	98.07	820		1.500	Yes	None
	120, 116, 12	10/14/2014	77	0.00	No/No		98.07	820		840	Yes	None
		01/21/2015	77	0.00	No/No		98.07	820		1 500	Yes	None
		04/13/2015	77	0.00	No/No		98.07	820		1,500	Yes	None
		07/23/2015	77	0.00	No/No		98.07	820		1,500	Yes	None
				2.50	1.0/110		20107	520		1,000	1.00	1.0110

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	e Trend
G275	TDS, mg/L	02/09/2016 05/12/2016	77 77	0.00 0.00	No/No No/No	STnon	98.07 98.07	820 820		1,500 1,300	Yes Yes	None None

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	e Trend
G279	As diss mal	02/19/2014	60	97.10	No/No	STlow?	07.10	0.001		<0.001	No	
021)	As, diss, hig/L	05/13/2014	69	97.10	No/No	5110w2	97.10	0.001		<0.001	Ves	None
		08/12/2014	69	97.10	No/No		97.10	0.001		<0.017	No	None
		10/14/2014	69	97.10	No/No		97.10	0.001		<0.001	No	
		01/21/2015	69	97.10	No/No		97.10	0.001		<0.001	No	
		04/13/2015	69	97.10	No/No		97.10	0.001		<0.001	No	
		07/23/2015	69	97.10	No/No		97.10	0.001		<0.002	No	
		10/08/2015	69	97.10	No/No		97.10	0.001		<0.001	No	
		02/16/2016	69	97.10	No/No		97.10	0.001		<0.001	No	
		05/13/2016	69	97.10	No/No		97.10	0.001		<0.001	No	
		05/15/2010	07	97.10	110/110		97.10	0.001		<0.001	110	
G279	B, diss, mg/L	02/19/2014	69	60.87	No/No	STlow1	97.10	0.120		0.022	No	
		05/13/2014	69	60.87	No/No		97.10	0.120		0.038	No	
		08/12/2014	69	60.87	No/No		97.10	0.120		0.015	No	
		10/14/2014	69	60.87	No/No		97.10	0.120		0.025	No	
		01/21/2015	69	60.87	No/No		97.10	0.120		0.032	No	
		04/13/2015	69	60.87	No/No		97.10	0.120		0.047	No	
		07/23/2015	69	60.87	No/No		97.10	0.120		0.031	No	
		10/08/2015	69	60.87	No/No		97.10	0.120		1.300	Yes	None
		02/16/2016	69	60.87	No/No		97.10	0.120		0.290	Yes	None
		05/13/2016	69	60.87	No/No		97.10	0.120		0.110	No	
G279	Mn diss ma/l	02/19/2014	69	23 19	No/No	STnon	97 10	0.480		0.029	No	
021)	Will, diss, ling/L	05/13/2014	60	23.17	No/No	5111011	97.10	0.480		0.029	No	
		08/12/2014	60	23.19	No/No		97.10	0.480		0.034	No	
		10/14/2014	60	23.19 22.10			97.10 07.10	0.480		0.024	No	
		10/14/2014 01/21/2015	60	23.19 22.10			97.10 07.10	0.480		0.020	NO	
		04/12/2015	60	23.19			97.10	0.480		0.01/	INO No	
		07/22/2015	69	23.19			97.10	0.480		0.003	INO Na	
		07/23/2015	09	23.19	1NO/1NO		97.10	0.480		0.013	INO	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
G279	Mn, diss, mg/L	10/08/2015	69	23.19	No/No	STnon	97.10	0.480		0.032	No	
		02/16/2016	69	23.19	No/No		97.10	0.480		0.011	No	
		05/13/2016	69	23.19	No/No		97.10	0.480		0.006	No	
G279	Pb, diss, mg/L	02/19/2014	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
		05/13/2014	69	100.00	No/No		N/A	0.001		0.022	Yes	None
		08/12/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/14/2014	69	100.00	No/No		N/A	0.001		< 0.001	No	
		01/21/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		04/13/2015	69	100.00	No/No		N/A	0.001		< 0.002	No	
		07/23/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		10/08/2015	69	100.00	No/No		N/A	0.001		< 0.001	No	
		02/16/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
		05/13/2016	69	100.00	No/No		N/A	0.001		< 0.001	No	
G279	pH (field), STD	02/19/2014	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.37	No	
	- · · ·	05/13/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.00	No	
		08/12/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.22	No	
		10/14/2014	80	0.00	Yes/Yes		99.00	7.86	6.64	7.09	No	
		01/21/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.14	No	
		04/13/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	6.59	Yes	None
		07/23/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.11	No	
		10/08/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.04	No	
		11/24/2015	80	0.00	Yes/Yes		99.00	7.86	6.64	7.21	No	
		02/16/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.19	No	
		02/16/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.19	No	
		05/13/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.91	No	
		05/13/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.91	No	
		08/03/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.11	No	
		11/22/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.16	No	

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	Trend
G279	SO4. tot. mg/L	02/19/2014	78	0.00	No/No	STnon	98.17	370.0		110.0	No	
	,, <u>6</u>	05/13/2014	78	0.00	No/No		98.17	370.0		110.0	No	
		08/12/2014	78	0.00	No/No		98.17	370.0		120.0	No	
		10/14/2014	78	0.00	No/No		98.17	370.0		140.0	No	
		01/21/2015	78	0.00	No/No		98.17	370.0		230.0	No	
		04/13/2015	78	0.00	No/No		98.17	370.0		470.0	Yes	Upward
		07/23/2015	78	0.00	No/No		98.17	370.0		470.0	Yes	Upward
		10/08/2015	78	0.00	No/No		98.17	370.0		810.0	Yes	Upward
		11/24/2015	78	0.00	No/No		98.17	370.0		520.0	Yes	Upward
		02/16/2016	78	0.00	No/No		98.17	370.0		610.0	Yes	Upward
		02/16/2016	78	0.00	No/No		98.17	370.0		590.0	Yes	Upward
		05/13/2016	78	0.00	No/No		98.17	370.0		230.0	No	
		05/13/2016	78	0.00	No/No		98.17	370.0		270.0	No	
		08/03/2016	78	0.00	No/No		98.17	370.0		570.0	Yes	Upward
		11/22/2016	78	0.00	No/No		98.17	370.0		720.0	Yes	Upward
G279	TDS, mg/L	08/12/2014	77	0.00	No/No	STnon	98.07	820		600	No	
		10/14/2014	77	0.00	No/No		98.07	820		650	No	
		01/21/2015	77	0.00	No/No		98.07	820		810	No	
		04/13/2015	77	0.00	No/No		98.07	820		800	No	
		07/23/2015	77	0.00	No/No		98.07	820		1,200	Yes	None
		10/08/2015	77	0.00	No/No		98.07	820		1,700	Yes	None
		11/24/2015	77	0.00	No/No		98.07	820		1,100	Yes	None
		02/16/2016	77	0.00	No/No		98.07	820		1,400	Yes	None
		02/16/2016	77	0.00	No/No		98.07	820		1,500	Yes	None
		05/13/2016	77	0.00	No/No		98.07	820		600	No	
		05/13/2016	77	0.00	No/No		98.07	820		700	No	
		08/03/2016	77	0.00	No/No		98.07	820		1,300	Yes	None
		11/22/2016	77	0.00	No/No		98.07	820		1,300	Yes	None

			Count	Percent								
Compliance			Of Bkg	of Non	Normal /		Confidence					
Location	Parameter	Sample Date	Results	detects	Lognormal	Test	Level	Upper Limit	Lower Limit	Analysis Result	Exceedance T	Frend

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	e Trend
C 401		11/21/2015	80	0.00	X7 /X7	0 T	00.00	7.04	C C 1	5.00	37	N
G401	pH (field), SID	11/21/2015	80	0.00	Yes/Yes	SIpar	99.00	/.86	6.64	5.98	Yes	None
		02/22/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	5.80	Yes	None
		05/19/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	5.96	Yes	None
		08/01/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.15	Yes	None
		11/17/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.24	Yes	None
G401	SO4. tot. mg/L	11/21/2015	78	0.00	No/No	STnon	98.17	370.0		2.300.0	Yes	None
	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	02/22/2016	78	0.00	No/No		98.17	370.0		2 500 0	Yes	None
		05/19/2016	78	0.00	No/No		98.17	370.0		2 200 0	Yes	None
		08/01/2016	78	0.00	No/No		98.17	370.0		2,100.0	Yes	None
		11/17/2016	78	0.00	No/No		98.17	370.0		3,400.0	Yes	None
G401	TDS, mg/L	11/21/2015	77	0.00	No/No	STnon	98.07	820		3,000	Yes	None
		02/22/2016	77	0.00	No/No		98.07	820		3,000	Yes	None
		05/19/2016	77	0.00	No/No		98.07	820		2,800	Yes	None
		08/01/2016	77	0.00	No/No		98.07	820		2,900	Yes	None
		11/17/2016	77	0.00	No/No		98.07	820		3,200	Yes	None

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedan	ce Trend
G402	nH (field) STD	11/21/2015	80	0.00	Ves/Ves	STpar	99.00	7 86	6 64	6.81	No	
0102	pir (neid), 51D	02/22/2016	80	0.00	Yes/Yes	Dipa	99.00	7.86	6.64	6.74	No	
		05/19/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.77	No	
		08/02/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.60	Yes	None
		11/17/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.62	Yes	None
G402	SO4, tot, mg/L	11/21/2015	78	0.00	No/No	STnon	98.17	370.0		1,200.0	Yes	None
		02/22/2016	78	0.00	No/No		98.17	370.0		1,000.0	Yes	None
		05/19/2016	78	0.00	No/No		98.17	370.0		960.0	Yes	None
		08/02/2016	78	0.00	No/No		98.17	370.0		890.0	Yes	None
		11/17/2016	78	0.00	No/No		98.17	370.0		1,100.0	Yes	None
G402	TDS_mg/L	11/21/2015	77	0.00	No/No	STnon	98.07	820		1 700	Yes	None
0.02	100, 116, 12	02/22/2016	77	0.00	No/No	5 mon	98.07	820		1,700	Yes	None
		05/19/2016	77	0.00	No/No		98.07	820		1,500	Yes	None
		08/02/2016	77	0.00	No/No		98.07	820		1,500	Yes	None
		11/17/2016	77	0.00	No/No		98.07	820		1,700	Yes	None

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance Trend
C 102		11/02/0015	00	0.00	<b>X</b> 7 ( <b>X</b> 7	C.T.	00.00			7.15	N
G403	pH (field), STD	11/23/2015	80	0.00	Yes/Yes	STpar	99.00	/.86	6.64	7.15	No
		02/22/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.96	No
		05/18/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.89	No
		08/01/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.91	No
		11/17/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.74	No
G403	SO4. tot. mg/L	11/23/2015	78	0.00	No/No	STnon	98.17	370.0		35.0	No
	~~~,·~~,·~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	02/22/2016	78	0.00	No/No		98.17	370.0		17.0	No
		05/18/2016	78	0.00	No/No		98.17	370.0		11.0	No
		08/01/2016	78	0.00	No/No		98.17	370.0		9.9	No
		11/17/2016	78	0.00	No/No		98.17	370.0		8.9	No
G403	TDS, mg/L	11/23/2015	77	0.00	No/No	STnon	98.07	820		320	No
		02/22/2016	77	0.00	No/No		98.07	820		340	No
		05/18/2016	77	0.00	No/No		98.07	820		320	No
		08/01/2016	77	0.00	No/No		98.07	820		320	No
		11/17/2016	77	0.00	No/No		98.07	820		350	No

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedanc	e Trend
G404	nH (field) STD	11/21/2015	80	0.00	Vac/Vac	STnar	99.00	7.86	6.64	672	No	
0-10-1	pri (neid), 51D	02/15/2016	80	0.00	Ves/Ves	Dipu	99.00	7.86	6.64	7.24	No	
		05/19/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.82	No	
		08/02/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.08	No	
		11/22/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	7.09	No	
G404	SO4, tot, mg/L	11/21/2015	78	0.00	No/No	STnon	98.17	370.0		180.0	No	
		02/15/2016	78	0.00	No/No		98.17	370.0		150.0	No	
		05/19/2016	78	0.00	No/No		98.17	370.0		140.0	No	
		08/02/2016	78	0.00	No/No		98.17	370.0		190.0	No	
		11/22/2016	78	0.00	No/No		98.17	370.0		380.0	Yes	None
G404	TDS, mg/L	11/21/2015	77	0.00	No/No	STnon	98.07	820		580	No	
	<i>, </i>	02/15/2016	77	0.00	No/No		98.07	820		560	No	
		05/19/2016	77	0.00	No/No		98.07	820		460	No	
		08/02/2016	77	0.00	No/No		98.07	820		620	No	
		11/22/2016	77	0.00	No/No		98.07	820		880	Yes	None

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	Trend
G405	nH (field) STD	11/21/2015	80	0.00	Vac/Vac	STnar	99.00	7.86	6.64	6.82	No	
0405	pri (neid), STD	02/15/2016	80	0.00	Ves/Ves	Sipai	99.00	7.86	6.64	7.07	No	
		05/18/2016	80	0.00	Yes/Yes		99.00 99.00	7.86	6.64	6.85	No	
		08/02/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.81	No	
		11/22/2016	80	0.00	Yes/Yes		99.00	7.86	6.64	6.90	No	
G405	SO4, tot, mg/L	11/21/2015	78	0.00	No/No	STnon	98.17	370.0		1,700.0	Yes	None
	-	02/15/2016	78	0.00	No/No		98.17	370.0		1,700.0	Yes	None
		05/18/2016	78	0.00	No/No		98.17	370.0		1,800.0	Yes	None
		08/02/2016	78	0.00	No/No		98.17	370.0		1,600.0	Yes	None
		11/22/2016	78	0.00	No/No		98.17	370.0		1,400.0	Yes	None
G405	TDS, mg/L	11/21/2015	77	0.00	No/No	STnon	98.07	820		2,400	Yes	None
		02/15/2016	77	0.00	No/No		98.07	820		2,500	Yes	None
		05/18/2016	77	0.00	No/No		98.07	820		2,200	Yes	None
		08/02/2016	77	0.00	No/No		98.07	820		2,200	Yes	None
		11/22/2016	77	0.00	No/No		98.07	820		2,100	Yes	None

Compliance Location	Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	Trend
G406	As, diss, mg/L	08/30/2016	69	97.10	No/No	STlow2	97.10	0.001		<0.001	No	
G406	B, diss, mg/L	08/30/2016	69	60.87	No/No	STlow1	97.10	0.120		1.600	Yes	Insufficient Data: Cannot Trend.
G406	Mn, diss, mg/L	08/30/2016	69	23.19	No/No	STnon	97.10	0.480		5.500	Yes	Insufficient Data: Cannot Trend.
G406	Pb, diss, mg/L	08/30/2016	69	100.00	No/No	STmdl	N/A	0.001		<0.001	No	
G406	pH (field), STD	11/18/2016	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.32	No	
G406	SO4, tot, mg/L	11/18/2016	78	0.00	No/No	STnon	98.17	370.0		910.0	Yes	Insufficient Data: Cannot Trend.
G406	TDS, mg/L	08/30/2016 11/18/2016	77 77	0.00 0.00	No/No No/No	STnon	98.07 98.07	820 820		1,300 1,400	Yes Yes	Insufficient Data: Cannot Trend. Insufficient Data: Cannot Trend.

Parameter	Sample Date	Count Of Bkg Results	Percent of Non detects	Normal / Lognormal	Test	Confidence Level	Upper Limit	Lower Limit	Analysis Result	Exceedance	Trend
As, diss, mg/L	08/30/2016	69	97.10	No/No	STlow2	97.10	0.001		<0.001	No	
B, diss, mg/L	08/30/2016	69	60.87	No/No	STlow1	97.10	0.120		0.052	No	
Mn, diss, mg/L	08/30/2016	69	23.19	No/No	STnon	97.10	0.480		0.210	No	
Pb, diss, mg/L	08/30/2016	69	100.00	No/No	STmdl	N/A	0.001		< 0.001	No	
pH (field), STD	11/18/2016	80	0.00	Yes/Yes	STpar	99.00	7.86	6.64	7.75	No	
SO4, tot, mg/L	11/18/2016	78	0.00	No/No	STnon	98.17	370.0		830.0	Yes	Insufficient Data: Cannot Trend.
TDS mg/I	08/30/2016	77	0.00	No/No	STnon	98.07	820		1,400	Yes	Insufficient Data:
	Parameter As, diss, mg/L B, diss, mg/L Mn, diss, mg/L Pb, diss, mg/L pH (field), STD SO4, tot, mg/L	Parameter Sample Date As, diss, mg/L 08/30/2016 B, diss, mg/L 08/30/2016 Mn, diss, mg/L 08/30/2016 Pb, diss, mg/L 08/30/2016 pH (field), STD 11/18/2016 SO4, tot, mg/L 08/30/2016	Parameter Sample Date Count Of Bkg Results As, diss, mg/L 08/30/2016 69 B, diss, mg/L 08/30/2016 69 Mn, diss, mg/L 08/30/2016 69 Pb, diss, mg/L 08/30/2016 69 pcH (field), STD 11/18/2016 80 SO4, tot, mg/L 08/30/2016 78	Parameter Sample Date Count Of Bkg Results Percent of Non detects As, diss, mg/L 08/30/2016 69 97.10 B, diss, mg/L 08/30/2016 69 60.87 Mn, diss, mg/L 08/30/2016 69 23.19 Pb, diss, mg/L 08/30/2016 69 100.00 pH (field), STD 11/18/2016 80 0.00 SO4, tot, mg/L 08/20/2016 78 0.00	ParameterSample DateCount Of Bkg ResultsPercent of Non detectsNormal / LognormalAs, diss, mg/L08/30/20166997.10No/NoB, diss, mg/L08/30/20166960.87No/NoMn, diss, mg/L08/30/20166923.19No/NoPb, diss, mg/L08/30/201669100.00No/NoPb, diss, mg/L08/30/201669100.00No/NoPb, diss, mg/L08/30/2016780.00Yes/YesSO4, tot, mg/L11/18/2016780.00No/No	ParameterSample DateCount Of Bkg ResultsPercent of Non detectsNormal / LognormalTestAs, diss, mg/L08/30/20166997.10No/NoSTlow2B, diss, mg/L08/30/20166960.87No/NoSTlow1Mn, diss, mg/L08/30/20166923.19No/NoSTnonPb, diss, mg/L08/30/201669100.00No/NoSTmonPb, diss, mg/L08/30/201669100.00No/NoSTmonPb, diss, mg/L08/30/2016780.00No/NoSTmonPb, diss, mg/L11/18/2016780.00No/NoSTmon	ParameterSample DateCount Of Bkg ResultsPercent of Non detectsNormal / LognormalConfidence LevelAs, diss, mg/L08/30/20166997.10No/NoSTlow297.10B, diss, mg/L08/30/20166960.87No/NoSTlow197.10Mn, diss, mg/L08/30/20166923.19No/NoSTnon97.10Pb, diss, mg/L08/30/201669100.00No/NoSTmdlN/AoH (field), STD11/18/2016800.00Yes/YesSTpar99.00SO4, tot, mg/L11/18/2016780.00No/NoSTnon98.17	Parameter Sample Date Count Of Bkg Results Percent of Non detects Normal / Lognormal Test Confidence Level Upper Limit As, diss, mg/L 08/30/2016 69 97.10 No/No STIow2 97.10 0.001 B, diss, mg/L 08/30/2016 69 60.87 No/No STIow1 97.10 0.120 Mn, diss, mg/L 08/30/2016 69 23.19 No/No STnon 97.10 0.480 Pb, diss, mg/L 08/30/2016 69 100.00 No/No STmol 97.10 0.480 Pb, diss, mg/L 08/30/2016 69 100.00 No/No STmol 97.10 0.480 pcH (field), STD 11/18/2016 80 0.00 Yes/Yes STpar 99.00 7.86 SO4, tot, mg/L 11/18/2016 78 0.00 No/No STnon 98.17 370.0	Parameter Sample Date Count Of Bkg Results Percent detects Normal / Lognormal Test Confidence Level Upper Limit Lower Limit As, diss, mg/L 08/30/2016 69 97.10 No/No STIow2 97.10 0.001 B, diss, mg/L 08/30/2016 69 60.87 No/No STIow1 97.10 0.120 Mn, diss, mg/L 08/30/2016 69 23.19 No/No STnon 97.10 0.480 Pb, diss, mg/L 08/30/2016 69 100.00 No/No STmal N/A 0.001 opH (field), STD 11/18/2016 80 0.00 Yes/Yes STpar 99.00 7.86 6.64 SO4, tot, mg/L 11/18/2016 78 0.00 No/No STnon 98.17 370.0	Parameter Sample Date Count Of Bkg Results Percent of Non Normal / Lognormal Test Confidence Level Upper Limit Lower Limit Analysis Result As, diss, mg/L 08/30/2016 69 97.10 0.001 <<0.001	Parameter Sample Date Count OF Bkg Result Percent of kest Normal / detects Test Confidence Level Upper Limit Lower Limit Analysis Result Exceedance As, diss, mg/L 08/30/2016 69 97.10 No/No STIow2 97.10 0.001 <0.001

APPENDIX C

GROUNDWATER MONITORING DATA

APPENDIX C

Summary of Groundwater Data

Units: mg/L

				- ·																	C 1 C 1		Total
Wall ID	Data	Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,	Lithium total	Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
G151	12/21/2011		lotai	lotai	lolai	0.0218	lotai	lolai	LULAI L	lolai	lolai	lotai	lotai		lotai	Litiliuili, totai	0.257	lotai	totai	lUldi	lotai	lolai	510
G151 G151	01/25/2012	< 0.003	0.0012	0.057	< 0.001	0.0218	0.026	< 0.001		50	< 0.004	< 0.002	0.691	< 0.002	< 0.001		0.257	< 0.0002		0.0021	95	< 0.001	510
G151 G151	03/13/2012		< 0.0012	0.057	< 0.001		0.020	< 0.001		67	< 0.004	< 0.002	0.051		< 0.001			< 0.0002		0.0021	120	< 0.001	580
G151 G151	05/22/2012	0.0014	× 0.001	0.005	0.001	0.052	0.015	0.001		07	× 0.004	× 0.002	0.400	< 0.001	× 0.001		0.11	< 0.0002		0.0025	120	0.001	520
G151	07/23/2012	0.001	< 0.001	0.067	< 0.001	0.002	0.01	< 0.001		58	< 0.004	< 0.002	0.649		< 0.001		0.11	< 0.0002		0.0024	98	< 0.001	570
G151	11/14/2012	< 0.001				0.013								< 0.001			0.091						660
G151	01/30/2013	< 0.001				< 0.02								< 0.001			0.071						560
G151	05/20/2013	< 0.001				< 0.005								< 0.001			0.031						520
G151	07/22/2013	< 0.001				0.02								< 0.001			0.12						500
G151	02/19/2014	< 0.001				0.014								< 0.001			0.072						
G151	05/12/2014	< 0.001				0.024								< 0.001			0.069						
G151	08/11/2014	< 0.001				< 0.01								< 0.001			0.14						560
G151	10/14/2014	< 0.001				< 0.01								< 0.001			0.031						570
G151	01/21/2015	< 0.001				< 0.01								< 0.001			0.066						500
G151	04/08/2015	< 0.002				< 0.02								< 0.002			0.19						600
G151	07/23/2015	< 0.001				0.018								< 0.001			0.093						550
G151	10/06/2015	< 0.001				0.036								< 0.001			0.56						600
G151	02/09/2016	< 0.001				< 0.01								< 0.001			0.18						560
G151	05/11/2016	< 0.002				< 0.02								< 0.002			0.011						500
G152	12/21/2011	< 0.003				0.0513								< 0.002			0.927						494
G152	01/25/2012		0.0023	0.13	< 0.001		0.047	< 0.001		40	0.008	0.0023	0.642		0.0026			< 0.0002		0.003	110	< 0.001	520
G152	03/13/2012		0.0014	0.13	< 0.001		0.036	< 0.001		42	< 0.004	< 0.002	0.422		< 0.001			< 0.0002		0.0043	130	< 0.001	580
G152	05/22/2012	0.0011				0.044								< 0.001			0.21						580
G152	07/23/2012		0.0011	0.1	< 0.001		0.032	< 0.001		45	< 0.004	< 0.002	0.587		< 0.001			< 0.0002		0.0035	120	< 0.001	670
G152	11/14/2012	< 0.001				0.031								< 0.001			0.14						620
G152	01/30/2013	< 0.001				0.033								< 0.001			0.14						600
G152	05/20/2013	< 0.001				0.015								< 0.001			0.034						570
G152	07/22/2013	< 0.001				0.024								0.0014			0.21						580
G152	02/19/2014	< 0.001				0.038								< 0.001			0.12						
G152	05/12/2014	< 0.001				0.044								< 0.001			0.05						
G152	08/11/2014	< 0.001				0.044		-			-			< 0.001			0.31						820
G152	10/14/2014	< 0.001				0.034								< 0.001			0.15						620
G152	01/21/2015	< 0.001				0.03								< 0.001			0.15						620
G152	04/08/2015	< 0.002				0.025								< 0.002			0.07						680
G152	07/23/2015	0.0078				0.079								< 0.001			0.89						/30
G152 G152	10/06/2015	0.0054				0.11								< 0.001			0.53						020
G152 G152	02/09/2010	< 0.0034				0.062								< 0.001			0.47						920
G152 G152	11/19/2016	< 0.002		0.073		0.000	0.017			51			0 509	< 0.002			0.51				130		680
G152	12/21/2011	< 0.002		0.075		0.0575	0.017			51			0.505	< 0.002			1.02				150		2150
G153	01/25/2012	< 0.005	0.0056	0.036	< 0.001	0.0373	0.064	< 0.001		120	0.0053	0 0068	0 360	< 0.002	< 0.001		1.02	< 0.0002		0.021	1500	< 0.001	3000
G153	01/23/2012		0.0050	0.030	< 0.001		0.004	< 0.001		160	0.0033	0.0000	0.303		< 0.001			< 0.0002		0.021	1900	< 0.001	3200
G153	05/22/2012	0 004	0.0039	0.022	. 0.001	0.032	0.050	\$ 0.001		100	0.0041	0.0034	0.233	< 0 001	. 0.001		0.42	\$ 0.0002		0.027	1300	. 0.001	3100
G153	07/23/2012	0.004	0.0052	0.016	< 0.001	0.052	0.031	< 0.001		130	0.0041	< 0.002	0 341	× 0.001	< 0.001		0.42	< 0.0002		0.021	1500	< 0.001	3200
G153	11/14/2012	0.0023	0.0052	0.010	. 0.001	0.04	0.001	. 0.001		150	0.0041	.0.002	0.041	< 0.001	. 0.001	1	0.19	. 0.0002		0.021	1300	. 0.001	3300
G153	01/30/2013	0.0022	1			0.039	1							< 0.001		1	0.28						3000
G153	05/20/2013	< 0.001				0.012								< 0.001		1	0.13						2800
G153	07/22/2013	0.0027				0.028	1							< 0.001	1		0.1						3000
G153	02/19/2014	< 0.001	ł	-		0.042	ł		1				0.4	< 0.001	1		0.16	1					
G153	05/12/2014	< 0.001	1			0.021							0.41	< 0.001			0.05						
	-		÷	•			÷	•			•						•	•	•	•		÷	

					- III							.	- · · ·								C 1 C 1		Total
	Data	Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	total I	chioride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,	Lithium total	Manganese,	iviercury,	total	Selenium,	Sulfate,	Inailium,	Dissolved
G153	08/11/2014		lUldi	lUldi	lulai	0.026	lotai	lolai	LULAI L	lotai	lolai	lolai	lotai		lotai	Litiliulii, totai	0.24	lotai	lotai	loldi	lotai	lotai	3500
G153	10/14/2014	0.0016				0.028								< 0.001			0.51						3400
G153	01/21/2015	< 0.001				0.02								< 0.001			0.2						3600
G153	04/08/2015	0.003				0.02								< 0.002			0.024						3700
G153	07/23/2015	< 0.001				0.015								< 0.001			0.16						3900
G153	10/06/2015	< 0.001				0.03								< 0.001			0.19						3700
G153	02/09/2016	< 0.001				0.016								< 0.001			0.046						3800
G153	05/11/2016	0.0035				0.026								< 0.002			0.037						3800
G153	11/19/2016			0.017			0.013			230			0.491								2500		4000
G154	12/21/2011	< 0.003				0.0452								< 0.002			0.462						624
G154	01/25/2012		0.0014	0.037	< 0.001		0.031	< 0.001		10	< 0.004	< 0.002	1.01		< 0.001			< 0.0002		0.0042	93	< 0.001	410
G154	03/13/2012		< 0.001	0.032	< 0.001		0.021	< 0.001		4.6	< 0.004	< 0.002	0.971		< 0.001			< 0.0002		0.0062	93	< 0.001	440
G154	05/22/2012	0.0017	.0.001	0.022	.0.001	0.025	0.020	.0.001		10		. 0.002	1.00	< 0.001	.0.001		0.038	. 0. 0002		0.0050	110	.0.001	470
G154	07/23/2012	10.001	< 0.001	0.033	< 0.001	0.04	0.038	< 0.001		4.9	< 0.004	< 0.002	1.06	10.001	< 0.001		0.042	< 0.0002		0.0058	110	< 0.001	490
G154 G154	11/14/2012	< 0.001				0.04					-			< 0.001			0.043						500
G154	01/30/2013	< 0.001				0.043								< 0.001			0.032						440
G154	07/22/2013	< 0.001				0.021								< 0.001			0.040						500
G154	02/19/2014	< 0.001				0.043								< 0.001			0.018						500
G154	05/12/2014	< 0.001				0.044								< 0.001			0.014						
G154	08/11/2014	< 0.001				0.035								< 0.001			0.081						480
G154	10/14/2014	< 0.001				0.035								< 0.001			0.0077						440
G154	01/21/2015	< 0.001				0.037								< 0.001			0.0092						450
G154	04/08/2015	< 0.002				0.023								< 0.002			0.003						440
G154	07/23/2015	< 0.001				0.024								< 0.001			0.14						430
G154	10/06/2015	< 0.001				0.044								< 0.001			0.01						500
G154	02/09/2016	< 0.001				0.041								< 0.001			0.041						560
G154	05/11/2016	< 0.002				0.037								< 0.002			0.0085						460
G270	03/11/2008	< 0.005	< 0.005	0.076	< 0.005	< 0.25	< 0.25	< 0.0025	77	9.5	< 0.025	< 0.005	0.36	< 0.0025	< 0.0025		0.19	< 0.0002		< 0.012	2.3	< 0.002	440
G270	04/21/2008	< 0.001	0.0016	0.076	< 0.001	0.078	0.071	< 0.0005	72	11	< 0.005	0.0017	0.36	< 0.0005	0.0021		0.21	< 0.0002		< 0.0025	2.4	< 0.002	420
G270	06/11/2008	< 0.001	0.0012	0.072	< 0.001	< 0.05	< 0.05	< 0.0005	/0	< 10	< 0.005	0.001	0.33	< 0.0005	0.0021		0.21	< 0.0002		< 0.0025	2.8	< 0.002	430
G270	08/13/2008	< 0.001	< 0.001	0.069	< 0.001	0.052	< 0.05	< 0.0005	6/	8.9	< 0.005	< 0.001	0.38	< 0.0005	< 0.0005		0.26	< 0.0002		< 0.0025	4.2	< 0.002	440
G270	10/14/2008	< 0.001	< 0.001	0.068	< 0.001	0.073	< 0.05	< 0.0005	/1	9.4	< 0.005	< 0.001	0.35	< 0.0005	< 0.0005		0.22	< 0.0002		< 0.0025	4.7	< 0.002	440
G270	12/02/2008	< 0.001	< 0.001	0.068	< 0.001	0.052	0.051	< 0.0005	09 77	9.6	< 0.005	< 0.001	0.31	< 0.0005	0.00054		0.22	< 0.0002		< 0.0025	4.4 2.2	< 0.002	400
G270	11/10/2009	< 0.001	0.001			0.054	0.058	< 0.0003	75	9.4 11				< 0.0005	0.0011		0.20				5.2		430
G270	01/28/2010	< 0.001	< 0.001			0.054	0.057	< 0.0005	74	19				< 0.0005	0.0011		0.23				8.2		430
G270	02/11/2010	. 0.001	. 0.001	0.056	< 0.001	0.000	0.052	. 0.0000	, -	1.5	< 0.005	< 0.001	0.3		5.0051		0.14	< 0.0002	1	< 0.0025	0.2	< 0.002	460
G270	06/09/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	61	39				< 0.001	< 0.001		0.058		1		6.8		470
G270	07/27/2010	< 0.001	< 0.001			0.011	< 0.01	< 0.001	58	40			1	< 0.001	< 0.001		0.062	1	1	1	9.1		480
G270	11/15/2010	< 0.001	0.0011			0.013	0.016	< 0.001	60	40				< 0.001	0.0021		0.069				7.6		490
G270	01/28/2011	< 0.001	< 0.001	0.064	< 0.001	< 0.01	0.012	< 0.001	80	45	< 0.004	0.002	0.36	< 0.001	0.0041		0.032	< 0.0002		0.002	9.6	< 0.001	470
G270	05/03/2011	< 0.001	< 0.001			0.01	< 0.01	< 0.001	57	57				< 0.001	< 0.001		0.023				13		430
G270	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	59	57				< 0.001	< 0.001		0.046				18		480
G270	11/11/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	61	64				< 0.001	< 0.001		0.05				23		440
G270	01/26/2012	< 0.001	< 0.001	0.049	< 0.001	< 0.01	< 0.01	< 0.001	64	73	< 0.004	< 0.002	0.363	< 0.001	< 0.001		0.0094	< 0.0002		0.0046	40	< 0.001	420
G270	05/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	67	83				< 0.001	< 0.001		0.04				57		460
G270	07/24/2012	< 0.001	< 0.001			0.092	0.018	< 0.001	73	66			<u> </u>	< 0.001	< 0.001		0.045				43		480
G270	11/14/2012	< 0.001	< 0.001	0.000	. 0.001	< 0.01	< 0.01	< 0.001	65	76		. 0.000	0.070	< 0.001	< 0.001		0.017	. 0.0005		0.0070	77	.0.001	500
6270	01/30/2013	< 0.001	0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	65	91	< 0.004	< 0.002	0.372	< 0.001	< 0.001		0.011	< 0.0002		0.0078	96	< 0.001	540
G270	05/20/2013	0.0011 < 0.001	< 0.001			< 0.01	< 0.01	< 0.001	72	85 70			<u> </u>	< 0.001	< 0.001		0.025			+	120		48U 500
G270	10/1//2013	< 0.001	< 0.001			0.01	< 0.01	< 0.001	70 Q1	79				< 0.001	< 0.001		0.045				12U Q5		500
5270	10/ 17/ 2013	× 0.001	× 0.001			0.014	× 0.01	× 0.001	01	12	L		1	- 0.001	× 0.001	L	0.041	L	1	1	05		520

																							Total
Wall ID	Data	Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,	Lithium total	Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
G270	02/19/2014								67	56			0.279			Litiliuili, totai	0.0029		totai		1/0		Solius
G270 G270	05/13/2014	< 0.001	0.0013	0.041	< 0.001	0.026	0.025	< 0.001	68	30	₹ 0.004	< 0.002	0.275	< 0.001	< 0.001		0.016	< 0.0002		0.0022	140	< 0.001	
G270	08/11/2014	< 0.001	0.0034			< 0.01	< 0.01	< 0.001	62	34				< 0.001	< 0.001		0.04				130		500
G270	10/14/2014	< 0.001	0.011			0.12	0.057	0.008	67	21				< 0.001	0.0073		0.04				140		500
G270	01/20/2015	< 0.001	< 0.001			0.011	< 0.01	< 0.001	70	18				< 0.001	< 0.001		0.011				140		500
G270	04/13/2015	< 0.002	< 0.001	0.05	< 0.001	0.025	0.047	< 0.001	70	20	< 0.004	< 0.002	0.334	< 0.002	< 0.001		0.0055	< 0.0002		0.0016	120	< 0.001	540
G270	07/22/2015	< 0.001	< 0.001	0.049	< 0.001	< 0.01	< 0.01	< 0.001		15	< 0.004	< 0.002	0.427	< 0.001	0.0018		0.47	< 0.0002	0.0011	< 0.001	110	< 0.001	550
G270	10/05/2015	< 0.001	< 0.001	0.037	< 0.001	0.013	< 0.01	< 0.001		11	< 0.004	< 0.002	0.411	< 0.001	< 0.001		0.056	< 0.0002	< 0.001	< 0.001	82	< 0.001	480
G270	11/20/2015		0.001	0.045	< 0.001		< 0.01	< 0.001	59	12	< 0.004	< 0.002	0.362		0.0015	< 0.01		< 0.0002	0.001	< 0.001	89	< 0.001	400
G270	02/10/2016	< 0.001	< 0.001	0.037	< 0.001	0.02	< 0.01	< 0.001	49	16	< 0.004	< 0.002	0.472	< 0.001	< 0.001	< 0.01	0.012	< 0.0002	< 0.001	0.0013	77	< 0.001	400
G270	05/12/2016	< 0.002	< 0.001	0.034	< 0.001	< 0.02	< 0.01	< 0.001	57	15	< 0.004	< 0.002	0.504	< 0.002	< 0.001	< 0.01	0.029	< 0.0002	< 0.001	< 0.001	77	< 0.001	370
G270	08/01/2016		< 0.001	0.037	< 0.001		< 0.01	< 0.001	50	15	< 0.004	< 0.002	0.397		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	76	< 0.001	360
G271	09/22/2009		0.003	0.08	< 0.001		0.33	< 0.0005	110	37	0.0064	0.0023	0.46		0.0054			< 0.0002		0.0038	230	< 0.002	770
G271	11/09/2009		0.0049	0.11	< 0.001		0.33	< 0.0005	140	34	0.014	0.0047	0.41		0.011			< 0.0002		0.0034	290	< 0.002	770
G271	11/10/2009	< 0.001				0.34								< 0.0005			0.11						
6271	01/19/2010	< 0.001	0.001	0.055	< 0.001	0.18	0.24	< 0.0005	110	33	< 0.005	< 0.001	0.34	< 0.0005	0.0018		0.007	< 0.0002		0.0035	320	< 0.002	860
G271	03/08/2010	< 0.001	< 0.001	0.055	< 0.001	0.25	0.23	< 0.0005	120	36	< 0.005	< 0.001	0.39	0.00061	0.0023		< 0.0025	< 0.0002		0.0053	300	< 0.002	880
G271	05/27/2010	< 0.001	0.0016	0.053	< 0.001	0.26	0.16	< 0.001	130	40	0.0056	< 0.002	0.42	< 0.001	0.0029		0.028	< 0.0002		0.0051	330	< 0.001	870
G271 C271	07/27/2010	< 0.001	< 0.001	0.041	< 0.001	0.24	0.15	< 0.001	120	40	< 0.004	< 0.002	0.34	< 0.001	< 0.001		0.016	< 0.0002		0.005	350	< 0.001	840
G271 G271	11/16/2010	< 0.001	< 0.001			0.10	0.15	< 0.001	110	11				< 0.001	< 0.001		< 0.0013				280		830
G271 G271	01/28/2011	< 0.001	0.0015	0.05	< 0.001	0.14	0.13	< 0.001	110	44	< 0.004	< 0.002	0.48	< 0.001	0.001		0.001	< 0.0002		0 0079	280	< 0.001	730
G271 G271	05/04/2011	< 0.001	< 0.0013	0.05	< 0.001	0.17	0.14	< 0.001	100	53	< 0.004	< 0.002	0.40	< 0.001	< 0.0033		< 0.0013	< 0.0002		0.0075	240	< 0.001	750
G271	07/27/2011	< 0.001	< 0.001			0.13	0.17	< 0.001	96	49				< 0.001	< 0.001		< 0.001				240		800
G271	11/14/2011	< 0.001	0.0013			0.15	0.21	< 0.001	110	50				< 0.001	< 0.001		< 0.001				250		710
G271	01/26/2012	< 0.001	0.0013	0.034	< 0.001	0.19	0.18	< 0.001	110	44	< 0.004	< 0.002	0.393	< 0.001	< 0.001		< 0.001	< 0.0002		0.0099	240	< 0.001	750
G271	05/22/2012	< 0.001	< 0.001			0.16	0.14	< 0.001	110	5.1				< 0.001	< 0.001		< 0.001				240		710
G271	07/24/2012	< 0.001	< 0.001			0.18	0.13	< 0.001	110	45				< 0.001	< 0.001		< 0.001				280		770
G271	11/14/2012	< 0.001	0.0014			0.28	0.19	< 0.001	130	50				< 0.001	0.0012		0.0021				300		940
G271	01/31/2013	< 0.001	0.0031	0.063	< 0.001	0.32	0.32	< 0.001	150	58	0.0066	0.0026	0.458	< 0.001	0.005		0.024	< 0.0002		0.005	380	< 0.001	880
G271	05/20/2013	0.0013	< 0.001			0.17	0.19	< 0.001	120	47				< 0.001	< 0.001		< 0.001				350		790
G271	07/22/2013	< 0.001	< 0.001			0.16	0.15	< 0.001	99	49				< 0.001	< 0.001		< 0.001				360		800
G271	10/14/2013	< 0.001	< 0.001			0.3	0.18	< 0.001	120	47				< 0.001	< 0.001		< 0.001				390		840
G271	02/19/2014	< 0.001	0.0028	0.062	< 0.001	0.26	0.24	< 0.001	150	51	0.0087	0.0023	0.298	< 0.001	0.0056		< 0.001	< 0.0002		0.0045	420	< 0.001	
G271	05/13/2014	0.16	0.0017			0.45	0.33	< 0.001	140	47				0.15	0.0018		0.22				440		
G271	08/11/2014	< 0.001	0.0027			0.39	0.44	< 0.001	140	42				< 0.001	0.0061		< 0.001				500		1000
G2/1	10/14/2014	< 0.001	0.0019			0.44	0.5	< 0.001	150	45				< 0.001	0.0062		0.0021				480		940
G2/1	01/21/2015	< 0.001	< 0.001	0.020	< 0.001	0.42	0.51	< 0.001	120	39	< 0.004	< 0.000	0.400	< 0.001	0.0014		< 0.001	< 0.0000		0.0025	490	< 0.001	8/0
G271	07/22/2015	< 0.002	< 0.001	0.029	< 0.001	0.37	0.31	< 0.001	130	45	< 0.004	< 0.002	0.406	< 0.002	< 0.001		< 0.002	< 0.0002	< 0.001	0.0035	250	< 0.001	1000
G271	10/08/2015	< 0.001	< 0.001	0.028	< 0.001	0.32	0.24	< 0.001		35 20	< 0.004		0.400	0.001/			0.001		0.001	0.0026	350	< 0.001	1000
G271 G271	11/23/2015	< 0.001	< 0.001	0.03	< 0.001	0.44	0.55	< 0.001	130	38	< 0.004	< 0.002	0.402	< 0.001	0.001	< 0.01	< 0.001	< 0.0002	0.0030	0.0033	400	< 0.001	860
G271 G271	02/16/2016	< 0.001	< 0.001	0.031	< 0.001	0.51	0.5	< 0.001	130	38	< 0.004	< 0.002	0.347	< 0.001	< 0.0012	< 0.01	< 0.001	< 0.0002	< 0.0012	0.0024	420	< 0.001	980
G271 G271	05/12/2016	< 0.001	< 0.001	0.025	< 0.001	0.51	0.01	< 0.001	170	39	< 0.004	< 0.002	0.401	< 0.001	< 0.001	< 0.01	< 0.001	< 0.0002	< 0.001	0.0032	540	< 0.001	940
G271	08/05/2016	. 5.002	< 0.001	0.032	< 0.001	0.01	0.63	< 0.001	110	37	< 0.004	< 0.002	0.414	0.002	0.0027	< 0.01	. 0.002	< 0.0002	< 0.001	0.0021	440	< 0.001	840
G271	11/21/2016		< 0.001	0.031	< 0.001		0.4	< 0.001		29	< 0.004	< 0.002	0.484		< 0.001	< 0.01		< 0.0002	< 0.001	0.0029	400	< 0.001	910
G272	09/22/2009		0.0012	0.079	< 0.001		0.06	< 0.0005	84	53	< 0.005	0.0017	0.48		0.0039			< 0.0002		< 0.0025	120	< 0.002	570
G272	11/10/2009	< 0.001	< 0.001	0.073	< 0.001	0.057	< 0.05	< 0.0005	88	46	< 0.005	0.0012	0.5	< 0.0005	0.0025		0.056	< 0.0002		< 0.0025	130	< 0.002	610
G272	01/19/2010	< 0.001	< 0.001	0.068	< 0.001	< 0.05	0.051	< 0.0005	85	45	< 0.005	< 0.001	0.42	< 0.0005	0.002		0.006	< 0.0002		< 0.0025	160	< 0.002	610
C272	02/04/2010	< 0.001	< 0.001	0.061	< 0.001	< 0.05	0.05	< 0.0005	0.2	лл	< 0.005	< 0.001	0.41		0.00059		< 0.0025	< 0.0003		< 0.0025	160	< 0.002	620
6272	05/04/2010	< 0.001	< 0.001	0.061	< 0.001	< 0.05	0.05	< 0.0005	93	44	< 0.005	< 0.001	0.41	< 0.0005	0.00058		< 0.0025	< 0.0002		< 0.0025	100	< 0.002	630
6272	05/2//2010	< 0.001	< 0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	92	40	0.0049	< 0.002	0.41	< 0.001	0.0011		0.0029	< 0.0002		0.0021	190	< 0.001	670
9272	07/27/2010	< 0.001	< 0.001	0.062	< 0.001	< 0.01	< 0.01	< 0.001	92	44	< 0.004	< 0.002	0.43	< 0.001	< 0.001		0.0019	< 0.0002		0.0039	210	< 0.001	690

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G272	09/20/2010	< 0.001				< 0.01								< 0.001			< 0.001						
G272	11/16/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	84	42				< 0.001	< 0.001		< 0.001				200		670
G272	01/31/2011	< 0.001	< 0.001	0.07	< 0.001	< 0.01	0.013	< 0.001	98	43	< 0.004	< 0.002	0.48	< 0.001	< 0.001		< 0.001	< 0.0002		0.0042	230	< 0.001	660
G272	05/04/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	92	42				< 0.001	< 0.001		< 0.001				210		690
G272	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	99	< 100				< 0.001	< 0.001		< 0.001				250		740 670
G272 G272	01/26/2012	< 0.001	0.001	0.068	< 0.001	< 0.01	< 0.01	< 0.001	98	41	< 0.004	< 0.002	0.459	< 0.001	< 0.001		< 0.001	< 0.0002		0.0021	230	< 0.001	660
G272 G272	05/22/2012	< 0.001	< 0.001	0.000	0.001	< 0.01	< 0.01	< 0.001	82	42	× 0.004	× 0.002	0.435	< 0.001	< 0.001		< 0.001	< 0.0002		0.0021	190	0.001	750
G272	07/24/2012	< 0.001	< 0.001			0.032	< 0.01	< 0.001	100	43				< 0.001	< 0.001		< 0.001				220		710
G272	11/14/2012	< 0.001	0.0011			< 0.01	< 0.01	< 0.001	110	40				< 0.001	0.0011		0.0012				220		780
G272	01/31/2013	< 0.001	< 0.001	0.072	< 0.001	< 0.01	< 0.01	< 0.001	110	44	< 0.004	< 0.002	0.461	< 0.001	< 0.001		< 0.001	< 0.0002		0.0029	330	< 0.001	760
G272	05/20/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	98	37				< 0.001	< 0.001		< 0.001				280		680
G272	07/22/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	100	35				< 0.001	< 0.001		< 0.001				260		680
G272	10/14/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	100	32				< 0.001	< 0.001		< 0.001				300		740
G272	02/19/2014	< 0.001	< 0.001	0.06	< 0.001	< 0.01	< 0.01	< 0.001	110	40	< 0.004	< 0.002	0.355	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	340	< 0.001	
G272	05/13/2014	0.17	< 0.001			0.27	0.015	< 0.001	93	36				0.15	< 0.001		0.23				310		
G272	08/11/2014	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	110	34				< 0.001	0.0014		< 0.001				330		740
G272	10/14/2014	< 0.001	< 0.001			0.036	0.022	< 0.001	110	37				< 0.001	0.0026		< 0.001				310		840
G272	01/21/2015	< 0.001	< 0.001	0.050	.0.001	< 0.01	< 0.01	< 0.001	99	31		. 0.002	0.200	< 0.001	0.0022		< 0.001	. 0.0002		0.0010	380	. 0.001	790
G272	04/10/2015	< 0.002	< 0.001	0.059	< 0.001	< 0.02	< 0.01	< 0.001	110	37	< 0.004	< 0.002	0.399	< 0.002	< 0.001		< 0.002	< 0.0002	10.001	0.0012	340	< 0.001	800
G272	10/08/2015	< 0.001	< 0.001	0.05	< 0.001	< 0.01	0.031	< 0.001		29	< 0.004	< 0.002	0.493	< 0.001	< 0.001		< 0.001	< 0.0002	< 0.001	< 0.001	270	< 0.001	840 660
G272	02/09/2015	< 0.001	0.001	0.038	< 0.001	< 0.01	0.040	< 0.001		20	< 0.004	0.002	0.501	< 0.001	0.001		0.001	< 0.0002	0.0024	< 0.0010	200	< 0.001	660
G272 G272	05/12/2016	< 0.001	< 0.0013	0.059	< 0.001	< 0.01	< 0.01	< 0.001		29	< 0.001	< 0.0022	0.561	< 0.001	< 0.001		< 0.0013	< 0.0002	< 0.0013	< 0.001	310	< 0.001	680
G273	09/23/2009		< 0.001	0.094	< 0.001		< 0.05	< 0.0005	120	35	< 0.005	< 0.001	0.38		< 0.0005			< 0.0002		< 0.0025	340	< 0.002	890
G273	11/10/2009	< 0.001	< 0.001	0.09	< 0.001	< 0.05	0.051	< 0.0005	140	28	< 0.005	< 0.001	0.33	< 0.0005	0.00093		0.097	< 0.0002		< 0.0025	400	< 0.002	980
G273	01/21/2010	< 0.001	< 0.001	0.085	< 0.001	0.054	< 0.05	< 0.0005	150	30	< 0.005	< 0.001	0.29	< 0.0005	< 0.0005		0.063	< 0.0002		< 0.0025	560	< 0.002	1200
G273	03/04/2010	< 0.001	< 0.001	0.079	< 0.001	< 0.05	< 0.05	< 0.0005	190	25	< 0.005	< 0.001	0.26	< 0.0005	< 0.0005		0.055	< 0.0002		< 0.0025	570	< 0.002	1300
G273	05/27/2010	< 0.001	0.0011	0.055	< 0.001	0.019	0.016	< 0.001	180	31	0.034	< 0.002	0.33	< 0.001	< 0.001		0.041	< 0.0002		0.0016	620	< 0.001	1300
G273	07/27/2010	< 0.001	< 0.001	0.048	< 0.001	0.023	0.023	< 0.001	160	30	< 0.004	< 0.002	0.37	< 0.001	< 0.001		0.048	< 0.0002		< 0.001	490	< 0.001	1100
G273	09/20/2010	< 0.001				0.024								< 0.001			0.052						
G273	11/16/2010	< 0.001	< 0.001			0.035	0.039	< 0.001	130	27				< 0.001	< 0.001		0.058				420		960
G273	01/31/2011	< 0.001	0.001	0.05	< 0.001	0.16	0.21	< 0.001	170	33	< 0.004	< 0.002	0.38	< 0.001	< 0.001		0.047	< 0.0002		< 0.001	520	< 0.001	1100
G273	05/03/2011	< 0.001	< 0.001			0.16	0.14	< 0.001	160	59				< 0.001	< 0.001		0.042				640		1200
G273	07/27/2011	< 0.001	< 0.001			0.097	0.12	< 0.001	150	29				< 0.001	< 0.001		0.029				510		1100
G273	11/14/2011	< 0.001	0.0013			0.13	0.15	< 0.001	150	29				< 0.001	< 0.001		0.041				510		990
G273	01/26/2012	< 0.001	0.001	0.043	< 0.001	0.27	0.26	< 0.001	180	27	< 0.004	< 0.002	0.359	< 0.001	< 0.001		0.033	< 0.0002		0.0012	750	< 0.001	1300
G273	05/22/2012	< 0.001	< 0.001			0.21	0.2	< 0.001	160	27				< 0.001	< 0.001		0.028				470		1100
G273	07/24/2012	< 0.001	< 0.001			0.12	0.094	< 0.001	140	32				< 0.001	< 0.001		0.022				360		910
G273	11/14/2012	0.0016	0.0034	0.046	< 0.001	0.27	0.2	< 0.001	160	33	< 0.004	< 0.002	0.22	< 0.001	0.0037		0.026	< 0.0002		< 0.002	530	< 0.001	1200
G273	01/31/2013	< 0.001	< 0.001	0.046	< 0.001	0.48	0.40	< 0.001	180	37	< 0.004	< 0.002	0.33	< 0.001	< 0.001		0.03	< 0.0002		< 0.002	670	< 0.001	1300
G273	03/20/2013	< 0.0014	< 0.001			0.18	0.29	< 0.001	160	4 25				< 0.001	< 0.001		0.013				510		080
G273	10/14/2013	< 0.001	< 0.001			0.21	0.23	< 0.001	140	35				< 0.001	< 0.001		0.022				450		900
G273	02/19/2014	< 0.001	< 0.001	0.039	< 0.001	0.10	0.36	< 0.001	150	38	< 0.004	< 0.002	0.286	< 0.001	< 0.001		0.015	< 0.0002		< 0.001	570	< 0.001	500
G273	05/13/2014	0.24	< 0.001	5.000	0.001	0.62	0.35	< 0.001	160	47	0.001	0.002	5.200	0.22	< 0.001		0.33	5.0002		0.001	620	0.001	
G273	08/11/2014	< 0.001	< 0.001	h	1	0.25	0.26	< 0.001	140	37	ł			< 0.001	< 0.001		0.01				530	1	1000
G273	10/14/2014	< 0.001	0.0011			0.32	0.29	< 0.001	150	37				< 0.001	0.0011		0.013				500		1100
G273	01/21/2015	< 0.001	< 0.001			0.58	0.45	< 0.001	150	46				< 0.001	< 0.001		0.018				650		1200
G273	04/13/2015	< 0.002	< 0.001	0.028	< 0.001	0.29	0.48	< 0.001	200	41	< 0.004	< 0.002	0.32	< 0.002	< 0.001		0.019	< 0.0002		< 0.001	690	< 0.001	1300
G273	07/23/2015	< 0.001	< 0.001	0.044	< 0.001	0.4	0.12	< 0.001		39	< 0.004	< 0.002	0.382	< 0.001	< 0.001		0.0011	< 0.0002	< 0.001	< 0.001	390	< 0.001	1200
G273	10/08/2015	< 0.001	< 0.001	0.039	< 0.001	0.1	0.15	< 0.001		46	< 0.004	< 0.002	< 0.25	< 0.001	< 0.001		0.014	< 0.0002	0.0019	< 0.001	450	< 0.001	930
G273	11/24/2015		< 0.001	0.049	< 0.001		0.2	< 0.001	140	41	< 0.004	< 0.002	< 0.25		0.0011	< 0.01		< 0.0002	< 0.001	< 0.001	420	< 0.001	890

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G273	02/16/2016	< 0.001	< 0.001	0.043	< 0.001	0.43	0.42	< 0.001	150	45	< 0.004	< 0.002	0.401	< 0.001	< 0.001	< 0.01	0.01	< 0.0002	< 0.001	< 0.001	550	< 0.001	1200
G273	05/12/2016	< 0.001	< 0.001	0.031	< 0.001	0.31	0.29	< 0.001	170	44	< 0.004	< 0.002	0.537	< 0.001	< 0.001	< 0.01	0.012	< 0.0002	< 0.001	< 0.001	520	< 0.001	980
G273	08/05/2016		< 0.001	0.032	< 0.001		0.17	< 0.001	120	46	< 0.004	< 0.002	0.294		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	400	< 0.001	840
G273	11/21/2016		< 0.001	0.036	< 0.001		0.15	< 0.001		48	< 0.004	< 0.002	0.39		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	440	< 0.001	900
G274	09/24/2009		0.0024	0.12	< 0.001		0.053	< 0.0005	100	55	0.0059	0.0028	0.34		0.0091			< 0.0002		< 0.0025	230	< 0.002	830
G274	11/11/2009	< 0.001	< 0.001	0.092	< 0.001	0.057	< 0.05	< 0.0005	100	54	< 0.005	< 0.001	0.35	< 0.0005	0.0012		0.007	< 0.0002		< 0.0025	250	< 0.002	820
G274	01/2//2010	< 0.001	< 0.001	0.09	< 0.001	< 0.05	< 0.05	< 0.0005	100	50	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.0039	< 0.0002		< 0.0025	260	< 0.002	850
G274	03/08/2010	< 0.001	< 0.001	0.091	< 0.001	< 0.05	< 0.05	< 0.0005	110	49	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.0056	< 0.0002		< 0.0025	270	< 0.002	870
G274	03/27/2010	< 0.001	< 0.001	0.09	< 0.001	< 0.01	< 0.01	< 0.001	110	43	< 0.004	< 0.002	0.32	< 0.001	< 0.0013		< 0.001	< 0.0002		< 0.0022	300	< 0.001	900
G274 G274	09/20/2010	< 0.001	< 0.001	0.08	< 0.001	< 0.01	< 0.01	< 0.001	110	44	< 0.004	< 0.002	0.55	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	320	< 0.001	300
G274	11/16/2010	< 0.001	< 0.001			< 0.01	0.012	< 0.001	120	38				< 0.001	< 0.001		< 0.001				360		940
G274	01/31/2011	< 0.001	0.0023	0.077	0.0018	< 0.01	0.012	0.0014	110	39	< 0.004	< 0.002	0.36	< 0.001	0.0018		< 0.001	< 0.0002		< 0.001	370	0.0013	950
G274	05/03/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	120	36				< 0.001	< 0.001		< 0.001				400		980
G274	07/27/2011	< 0.001	< 0.001			0.013	< 0.01	< 0.001	120	37				< 0.001	< 0.001		< 0.001				370		980
G274	11/14/2011	< 0.001	< 0.001			0.035	0.039	< 0.001	120	37				< 0.001	< 0.001		< 0.001				370		900
G274	01/26/2012	< 0.001	< 0.001	0.068	< 0.001	0.076	0.052	< 0.001	110	34	< 0.004	< 0.002	0.379	< 0.001	< 0.001		0.001	< 0.0002		0.0014	370	< 0.001	880
G274	05/22/2012	< 0.001	< 0.001			0.12	0.11	< 0.001	120	36				< 0.001	< 0.001		< 0.001				330		920
G274	07/24/2012	< 0.001	< 0.001			0.15	0.13	< 0.001	120	40				< 0.001	< 0.001		< 0.001				300		880
G274	11/14/2012	< 0.001	0.0022			0.18	0.16	< 0.001	120	37				< 0.001	0.0031		< 0.001				420		910
G274	01/31/2013	< 0.001	< 0.001	0.059	< 0.001	0.26	0.25	< 0.001	130	36	< 0.004	< 0.002	0.382	< 0.001	< 0.001		< 0.001	< 0.0002		0.0015	460	< 0.001	870
G274	05/20/2013	0.001	< 0.001			0.25	0.27	< 0.001	120	3.6				< 0.001	< 0.001		0.0073				350		800
G274	07/22/2013	< 0.001	< 0.001			0.31	0.31	< 0.001	110	32				< 0.001	0.0015		< 0.001				330		820
G274	10/14/2013	< 0.001	< 0.001	0.002	< 0.001	0.52	0.46	< 0.001	110	31	< 0.004	< 0.002	0.262	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	380	< 0.001	840
G274	02/19/2014	< 0.001	< 0.001	0.063	< 0.001	0.79	0.47	< 0.001	00	32	< 0.004	< 0.002	0.262	< 0.001	< 0.001		< 0.001	< 0.0002		< 0.001	300	< 0.001	
G274	03/13/2014	< 0.0055	< 0.001			0.52	0.52	< 0.001	120	27				< 0.0034	< 0.001		< 0.0039				370		880
G274	10/14/2014	< 0.001	0.001			0.57	0.01	< 0.001	100	23				< 0.001	0.001		< 0.001				320		770
G274	01/21/2015	< 0.001	0.0011			0.68	0.5	< 0.001	110	19				< 0.001	0.0029		0.0036				260		770
G274	04/13/2015	< 0.002	< 0.001	0.052	< 0.001	0.57	0.53	< 0.001	110	24	< 0.004	< 0.002	0.341	< 0.002	< 0.001		< 0.002	< 0.0002		< 0.001	390	< 0.001	770
G274	07/23/2015	< 0.001	< 0.001	0.068	< 0.001	0.48	0.49	< 0.001		24	< 0.004	< 0.002	0.403	< 0.001	0.0018		0.0015	< 0.0002	< 0.001	0.0014	320	< 0.001	890
G274	10/08/2015	< 0.001	< 0.001	0.061	< 0.001	0.43	0.74	< 0.001		22	< 0.004	< 0.002	0.265	< 0.001	< 0.001		< 0.001	< 0.0002	0.0012	< 0.001	320	< 0.001	770
G274	02/09/2016	< 0.001	< 0.001	0.07	< 0.001	0.87	0.65	< 0.001		22	< 0.004	< 0.002	0.455	< 0.001	0.001		< 0.001	< 0.0002	< 0.001	< 0.001	290	< 0.001	820
G274	05/12/2016	< 0.001	< 0.001	0.057	< 0.001	0.97	0.63	< 0.001		22	< 0.004	< 0.002	0.417	< 0.001	< 0.001		0.046	< 0.0002	< 0.001	< 0.001	350	< 0.001	770
G275	09/22/2009		< 0.005	0.12	< 0.001		4.5	< 0.0005	300	14	< 0.025	< 0.005	0.22		0.01			< 0.0002		< 0.012	990	< 0.002	2000
G275	11/11/2009	< 0.001	< 0.001	0.032	< 0.001	2.4	2.5	< 0.0005	140	12	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.032	< 0.0002		< 0.0025	350	< 0.002	910
G275	01/21/2010	< 0.001	< 0.001	0.032	< 0.001	1.2	1.2	< 0.0005	120	13	< 0.005	< 0.001	0.32	< 0.0005	< 0.0005		0.023	< 0.0002		< 0.0025	390	< 0.002	870
G275	03/08/2010	< 0.001	< 0.001	0.035	< 0.001	1.3	1.3	< 0.0005	160	23	< 0.005	< 0.001	0.27	< 0.0005	0.00053		0.029	< 0.0002		< 0.0025	460	< 0.002	1100
G275	05/28/2010	< 0.001	0.0016	0.047	< 0.001	2.9	2.9	< 0.001	180	18	0.0087	< 0.002	0.33	< 0.001	0.0017		0.018	< 0.0002		0.0014	540	< 0.001	1200
G275	07/26/2010	< 0.001	< 0.001	0.038	< 0.001	2.9	2.8	< 0.001	180	9.9	< 0.004	< 0.002	0.4	< 0.001	< 0.001		0.019	< 0.0002		< 0.001	550	< 0.001	1200
G275	09/20/2010	< 0.001	0.0028			2.0	4.1	< 0.001	260	0.7				< 0.001	0.0045		0.026				070		1700
G275	01/21/2010	< 0.001	0.0028	0.046	< 0.001	2.6	4.1	< 0.001	200	9.7	< 0.004	< 0.002	0.24	< 0.001	0.0045		0.16	< 0.0002		< 0.001	970	< 0.001	1700
G275	05/02/2011	< 0.001	< 0.0013	0.040	< 0.001	3.0	3.0 2.8	< 0.001	230	12	< 0.004	< 0.00Z	0.54	< 0.001	< 0.001		0.010	< 0.000Z		< 0.001	040 790	< 0.001	1600
G275	07/27/2011	< 0.001	< 0.001			4.3	4.2	< 0.001	200	9.2				< 0.001	< 0.001		0.04	1			720		1300
G275	11/14/2011	0.0015	0.0041			4.3	4.4	< 0.001	260	17				< 0.001	0.0059		0.09	1			820		1500
G275	01/31/2012	< 0.001	0.0011	0.039	< 0.001	3.8	3.6	< 0.001	310	15	0.0047	< 0.002	0.281	< 0.001	< 0.001		0.0081	< 0.0002	1	0.0015	370	< 0.001	1300
G275	05/22/2012	< 0.001	0.0017			3.7	3.4	< 0.001	240	11				< 0.001	0.0015		0.01				670		1500
G275	07/24/2012	< 0.001	0.0018			4.6	4.2	< 0.001	260	13				< 0.001	0.0024		0.09	1			900		1600
G275	11/14/2012	< 0.001	0.0025			3.7	3.9	< 0.001	270	19				< 0.001	0.0037		0.28				950		1600
G275	05/20/2013	0.0013	0.0025			3.4	3.8	< 0.001	250	24				< 0.001	0.0015		< 0.001				840		1400
G275	07/22/2013	< 0.001	< 0.001			3.5	3.1	< 0.001	210	19				< 0.001	0.0013		0.0017				700		1400
G275	05/13/2014	0.25	0.0019			2.8	3.4	< 0.001	210	20				0.2	0.0037		0.34				750		

																							Total
		Arsenic	Arsenic	Barium	Bervllium	Boron	Boron	Cadmium	Calcium	Chloride	Chromium	Cobalt	Fluoride	Lead	Lead		Manganese	Mercury	Molybdenum	Selenium	Sulfate	Thallium	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G275	08/11/2014	< 0.001	0.0043			4.2	4.1	< 0.001	240	20				< 0.001	0.0078	,	< 0.001				880		1500
G275	10/14/2014	< 0.001	0.0011			1.9	3.5	< 0.001	200	16				< 0.001	0.0012		0.003				500		840
G275	01/21/2015	< 0.001	0.0043			4.6	4.6	< 0.001	230	20				< 0.001	0.0079		0.025				940		1500
G275	04/13/2015	< 0.002	< 0.001	0.056	< 0.001	0.91	1.8	< 0.001	180	22	< 0.004	< 0.002	< 0.25	< 0.002	< 0.001		0.0024	< 0.0002		0.0012	650	< 0.001	1500
G275	07/23/2015	< 0.001	< 0.001	0.035	< 0.001	3	4	< 0.001		30	< 0.004	< 0.002	0.307	< 0.001	0.001		0.012	< 0.0002	0.0014	0.0014	750	< 0.001	1500
G275	02/09/2016	< 0.001	< 0.001	0.042	< 0.001	4	2.3	< 0.001		26	< 0.004	< 0.002	0.349	0.0044	0.0016		0.0064	< 0.0002	< 0.001	< 0.001	470	< 0.001	1500
G275	03/12/2010	< 0.001	< 0.001	0.039	0.001	2.5	2.4	< 0.001	04	20	< 0.004	< 0.002	0.432	< 0.001	0.001		0.0030	< 0.0002	< 0.001	< 0.001	170	< 0.001	620
G276	11/11/2009	< 0.001	< 0.003	0.10	< 0.0020	0.057	0.087	< 0.0025	94		0.025	< 0.003	0.75	< 0.0005	0.035		0.03	< 0.0002		< 0.012	170	< 0.002	670
G276	01/21/2000	< 0.001	< 0.001	0.075	< 0.001	0.057	0.002	< 0.0005	91	36	< 0.0055	< 0.001	0.57	< 0.0005	0.00079		0.03	< 0.0002		< 0.0025	190	< 0.002	660
G276	03/09/2010	× 0.001	× 0.001	0.075	0.001	0.037	0.032	× 0.0005	51	38	× 0.005	× 0.001	0.55	× 0.0005	0.00075		0.0055	< 0.0002		¢0.0025	130	< 0.002	650
G276	03/10/2010	< 0.001	< 0.001	0.083	< 0.001	0.057	0.058	< 0.0005	110		< 0.005	< 0.001	0.01	< 0.0005	0.0016		0.01	< 0.0002		< 0.0025	100	< 0.002	
G276	05/28/2010	< 0.001	< 0.001	0.067	< 0.001	0.035	0.033	< 0.001	100	40	0.0051	< 0.002	0.49	< 0.001	< 0.001		0.012	< 0.0002		0.0014	190	< 0.001	720
G276	07/26/2010	< 0.001	< 0.001	0.063	< 0.001	0.023	0.022	< 0.001	93	40	< 0.004	< 0.002	0.54	< 0.001	< 0.001		0.0054	< 0.0002		< 0.001	230	< 0.001	710
G276	09/20/2010	< 0.001				0.024								< 0.001			0.005						
G276	11/16/2010	< 0.001	< 0.001			0.013	0.044	< 0.001	96	35				< 0.001	< 0.001		0.0018				200		670
G276	01/31/2011	< 0.001	0.0026	0.078	< 0.001	0.036	0.052	< 0.001	96	36	< 0.004	< 0.002	0.53	< 0.001	0.001		0.0041	< 0.0002		< 0.001	200	< 0.001	710
G276	05/03/2011	< 0.001	< 0.001			0.043	0.039	< 0.001	95	36				< 0.001	< 0.001		0.0016				200		650
G276	07/27/2011	< 0.001	< 0.001			0.011	< 0.01	< 0.001	96	37				< 0.001	< 0.001		0.001				170		670
G276	11/14/2011	< 0.001	0.0013			0.081	0.08	< 0.001	94	35				< 0.001	< 0.001		< 0.001				180		620
G276	01/31/2012	< 0.001	< 0.001	0.075	< 0.001	< 0.01	0.06	< 0.001	130	32	0.0052	< 0.002	0.501	< 0.001	< 0.001		< 0.001	< 0.0002		0.0022	190	< 0.001	650
G276	05/22/2012	< 0.001	0.0011			0.066	0.073	< 0.001	97	38				< 0.001	< 0.001		0.0014				160		660
G276	07/24/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	100	38				< 0.001	< 0.001		< 0.001				190		680
G276	11/14/2012	< 0.001	0.0015	0.091	< 0.001	0.012	0.083	< 0.001	100	30	< 0.004	< 0.002	0.51	< 0.001	< 0.001		< 0.001	< 0.0002		0.0021	190	< 0.001	680
G276	01/31/2013	< 0.001	< 0.001	0.001	< 0.001	0.014	0.023	< 0.001	90	38	< 0.004	< 0.002	0.51	< 0.001	< 0.0011		< 0.001	< 0.0002		0.0031	180	< 0.001	640
G276	07/22/2013	< 0.001	< 0.001			0.030	< 0.031	< 0.001	100	32				< 0.001	< 0.001		< 0.001				220		670
G276	05/13/2014	< 0.001	0.0013			< 0.01	0.021	< 0.001	130	31				< 0.001	< 0.001		< 0.001				230		0/0
G276	08/12/2014	< 0.001	0.0013			0.018	0.027	< 0.001	120	29				< 0.001	0.006		< 0.001				220		640
G276	10/14/2014	< 0.001	< 0.001			0.028	0.019	< 0.001	100	28				< 0.001	0.0024		< 0.001				220		700
G276	01/21/2015	< 0.001	< 0.001			0.015	0.021	< 0.001	100	30				< 0.001	< 0.001		< 0.001				260		700
G276	04/13/2015	< 0.002	0.0057	0.34	0.0016	< 0.02	0.036	< 0.001	170	34	0.043	0.0047	0.486	< 0.002	0.022		< 0.002	< 0.0002		0.0034	310	< 0.001	780
G276	07/23/2015	< 0.001	< 0.001	0.096	< 0.001	0.037	0.015	< 0.001		26	< 0.004	< 0.002	0.377	< 0.001	0.0012		< 0.001	< 0.0002	0.0012	0.001	180	< 0.001	800
G276	11/24/2015		< 0.001	0.077	< 0.001		0.043	< 0.001	120	28	< 0.004	< 0.002	0.345		< 0.001	0.013		< 0.0002	0.0017	< 0.001	190	< 0.001	710
G276	02/16/2016		< 0.001	0.09	< 0.001		0.021	< 0.001	120	23	< 0.004	< 0.002	0.456		0.0014	0.015		< 0.0002	0.0013	0.0018	230	< 0.001	760
G276	02/17/2016	< 0.001	< 0.001	0.089	< 0.001	0.027	0.029	< 0.001		28	< 0.004	< 0.002	0.456	< 0.001	< 0.001		< 0.001	< 0.0002	< 0.001	< 0.001	230	< 0.001	700
G276	05/12/2016	< 0.001	< 0.001	0.079	< 0.001	0.013	< 0.01	< 0.001	130	22	< 0.004	< 0.002	0.549	< 0.001	< 0.001	0.012	< 0.001	< 0.0002	< 0.001	0.0017	240	< 0.001	720
G276	08/03/2016		< 0.001	0.085	< 0.001		0.019	< 0.001	110	23	< 0.004	< 0.002	0.443		< 0.001	< 0.01		< 0.0002	< 0.001	0.0017	1900	< 0.001	680
G276	11/21/2016		< 0.001	0.081	< 0.001		< 0.01	< 0.001	100	23	< 0.004	< 0.002	0.445		< 0.001	0.011		< 0.0002	< 0.001	0.002	210	< 0.001	720
G277	09/23/2009		0.027	0.61	0.0027		0.11	< 0.0025	190	41	0.052	0.04	0.79		0.072			0.00023		< 0.012	79	< 0.002	420
G277	11/11/2009	0.021	0.02	0.22	0.0011	< 0.05	0.074	< 0.0005	110	12	0.027	0.017	0.47	0.029	0.020		1.6	< 0.0002		< 0.0025	40	< 0.002	430 680
G277	01/19/2010	< 0.021	0.02	0.22	< 0.0011	< 0.05	0.074	< 0.0005	92	45 39	0.037	0.017	0.47	< 0.028	0.029		0.02	< 0.0002		< 0.0025	63	< 0.002	080
G277	01/20/2010	< 0.001	0.0015	0.055	₹0.001	× 0.05	0.002	< 0.0005	52	35	0.0054	0.0015	0.41	< 0.0005	0.0037		0.02	< 0.000Z		< 0.002J	05	< 0.002	550
6277	02/00/2010	ļ	< 0.001	0.075	< 0.001	+	0.065		02	10		< 0.001	0.4		0.00002			< 0.0002			40	< 0.002	E10
G277	03/00/2010	< 0.001	< 0.001	0.075	< 0.001	0.073	0.005	< 0.0005	33	43	< 0.005	< 0.001	0.4	< 0.0005	0.00092		0.0037	< 0.000Z	+	< 0.0025	49	< 0.00Z	540
G277	05/28/2010	< 0.001	0 0023	0 092	< 0.001	0.073	0.021	< 0.001	100	46	0.033	< 0.002	0.4	< 0.0003	0.0031		0.0037	< 0.0002		0 0032	58	< 0.001	580
G277	07/26/2010	< 0.001	0.0018	0.085	< 0.001	0.015	0.019	< 0.001	85	46	0.0092	< 0.002	0.42	< 0.001	0.0029		0.003	< 0.0002		0.0021	58	< 0.001	790
G277	09/20/2010	< 0.001	0.0010	0.000		0.012	0.015				0.0052		0.12	< 0.001	0.0025		0.0055	10.0002	1	0.0021	50		
G277	11/16/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	86					< 0.001	< 0.001		0.0021						
G277	01/31/2011	< 0.001	0.0015	0.085	< 0.001	0.016	0.025	< 0.001	97	40	< 0.004	< 0.002	0.38	< 0.001	0.0024		0.0031	< 0.0002		0.003	52	< 0.001	610
G277	05/03/2011	< 0.001	< 0.001			0.02	0.02	< 0.001	100	42				< 0.001	< 0.001		< 0.001				56		600
G277	07/27/2011	< 0.001	< 0.001			0.012	< 0.01	< 0.001	110	49				< 0.001	< 0.001		< 0.001				61		650

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G277	11/14/2011	< 0.001	0.0039			0.031	0.044	< 0.001	110	36				< 0.001	0.0056		0.0054				51		580
G277	01/31/2012		0.003	0.12	< 0.001		0.027	< 0.001	160	37	0.012	0.0022	0.326		0.0035			< 0.0002		0.0014	45	< 0.001	
G277	05/22/2012	< 0.001	< 0.001			0.021	0.026	< 0.001	110	30				< 0.001	< 0.001		0.0052				49		580
G277	07/24/2012	0.001	0.0011			0.01	0.01	0.001	110	34				0.001	0.0011		0.010				52		580
G277	07/25/2012	< 0.001	0.0011			< 0.01	< 0.01	< 0.001	110	11				< 0.001	0.0011		0.013				42		
G277	11/14/2012	< 0.001	0.001			0.022	< 0.01	< 0.001	120	11				< 0.001	0.0012		< 0.001				42		520
G277	03/20/2013	< 0.001	< 0.001			0.022	0.018	< 0.001	110	5.5 18				< 0.001	< 0.001		< 0.001				5.7 //5		520
G277 G277	05/13/2014	< 0.001	0.0021			0.02	0.021	< 0.001	130	30				< 0.001	0.001		< 0.001				60		500
G277 G277	10/15/2014	× 0.001	0.0021			0.012	0.024	< 0.001	84	3.6				(0.001	0.0033		0.001				18		
G277	10/16/2014	< 0.001	0.0010			0.032	0.01		0.	0.0				< 0.001	0.0000		0.003				10		360
G277	05/12/2016	< 0.001	0.0011	0.09	< 0.001	0.047	0.025	< 0.001		1.2	< 0.004	< 0.002	0.495	< 0.001	0.0013		0.0017	< 0.0002	0.001	< 0.001	15	< 0.001	460
G278	05/26/2010		0.0011	0.078	< 0.001		0.026	< 0.001	93	82	0.016	< 0.002	0.41		< 0.001			< 0.0002		0.0058	87	< 0.001	640
G278	05/28/2010	< 0.001				0.012								< 0.001			0.052						
G278	03/23/2011	< 0.001	0.0026	0.069	< 0.001	0.014	0.014	< 0.001	90	55	0.0063	< 0.002	0.55	< 0.001	0.0025		0.1	< 0.0002		0.0019	91	< 0.001	630
G278	05/03/2011	< 0.001	0.0056	0.15	< 0.001	< 0.01	0.026	< 0.001	97	60	0.022	0.0055	0.43	< 0.001	0.013		0.027	< 0.0002		0.0032	110	< 0.001	670
G278	07/25/2011	< 0.001	0.016	0.34	0.0018	< 0.01	0.023	< 0.001	170	62	0.034	0.018	0.42	< 0.001	0.033		0.032	< 0.0002		0.0029	100	< 0.001	690
G278	09/19/2011	< 0.001	0.0018	0.086	< 0.001	0.093	0.023	< 0.001	100	85	0.048	< 0.002	0.51	< 0.001	0.0031		0.04	< 0.0002		0.012	120	< 0.001	640
G278	05/22/2012	< 0.001	0.0032	0.098	< 0.001	0.017	0.017	< 0.001	110	82	0.007	0.0028	0.359	< 0.001	0.0059		0.46	< 0.0002		0.0025	120	< 0.001	730
G278	02/09/2016	< 0.001	0.004	0.11	< 0.001	0.54	0.54	< 0.001		310	0.0092	0.0034	0.336	< 0.001	0.008		0.14	< 0.0002	0.0014	0.041	680	< 0.001	1800
G278	05/13/2016	< 0.001	0.0067	0.19	< 0.001	0.5	0.2	< 0.001		180	0.017	0.0066	0.441	< 0.001	0.015		0.39	< 0.0002	0.0026	0.014	450	< 0.001	1500
G279	09/23/2009		0.0064	0.095	< 0.001		0.062	< 0.0005	90	59	0.01	0.0059	0.5		0.0092			< 0.0002		< 0.0025	99	< 0.002	620
G279	11/09/2009	< 0.001	< 0.001	0.065	< 0.001	0.062	0.056	< 0.0005	81	50	< 0.005	< 0.001	0.5	< 0.0005	0.00068		0.26	< 0.0002		< 0.0025	92	< 0.002	620
G279 G279	01/2//2010	< 0.001	< 0.001	0.067	< 0.001	< 0.05	0.057	< 0.0005	80	54	< 0.005	< 0.001	0.43	< 0.0005	0.0013		0.094	< 0.0002		< 0.0025	88	< 0.002	630
G279 G279	05/26/2010	< 0.001	< 0.001	0.003	< 0.001	< 0.03	0.03	< 0.0003	80	69	0.011	< 0.001	0.42	< 0.0003	< 0.00008		0.033	< 0.0002		0.0023	100	< 0.002	680
G279	07/26/2010	< 0.001	< 0.001	0.000	< 0.001	< 0.01	< 0.01	< 0.001	91	64	< 0.011	< 0.002	0.42	< 0.001	< 0.001		0.022	< 0.0002		0.005	88	< 0.001	670
G279	09/20/2010	< 0.001	< 0.001	0.002	0.001	< 0.01	× 0.01	0.001	51	04	× 0.004	× 0.002	0.45	< 0.001	× 0.001		0.013	< 0.000Z		0.0015	00	× 0.001	0/0
G279	11/16/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	89	52				< 0.001	< 0.001		0.0088				96		600
G279	01/28/2011	< 0.001	0.0026	0.098	< 0.001	< 0.01	0.016	< 0.001	100	50	< 0.004	0.0029	0.44	< 0.001	0.0052		0.014	< 0.0002		0.0038	93	< 0.001	600
G279	05/04/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	130	49				< 0.001	< 0.001		0.0033				100		630
G279	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	83	51				< 0.001	< 0.001		0.0071				95		600
G279	11/14/2011	< 0.001	< 0.001			0.017	0.019	< 0.001	84	53				< 0.001	< 0.001		0.0035				95		580
G279	01/30/2012	< 0.001	< 0.001	0.05	< 0.001	< 0.01	< 0.01	< 0.001	120	51	< 0.004	< 0.002	0.54	< 0.001	< 0.001		0.0049	< 0.0002		0.0041	160	< 0.001	630
G279	05/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	69	54				< 0.001	< 0.001		0.042				96		590
G279	07/24/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	98	71				< 0.001	< 0.001		0.002				130		660
G279	11/14/2012	0.0019	0.0018			< 0.01	< 0.01	< 0.001	97	62				< 0.001	0.001		0.019				140		750
G279	01/31/2013	0.0014	< 0.001	0.06	< 0.001	< 0.01	< 0.01	< 0.001	120	60	< 0.004	< 0.002	0.418	< 0.001	< 0.001		0.0019	< 0.0002		0.0034	390	< 0.001	1200
G279	05/20/2013	< 0.001	0.0012			0.01	0.011	< 0.001	8/	50				< 0.001	< 0.001		0.0011				140		500
G279	10/14/2013	< 0.001	< 0.001			0.01	0.01	< 0.001	00 07	55				< 0.001	< 0.001		0.0078				140		500 640
G279	02/19/2013	< 0.001	< 0.001	0.058	< 0.001	0.013	0.011	< 0.001	86	5/	< 0.004	< 0.002	0 331	< 0.001	< 0.001		0.078	< 0.0002		< 0.001	110	< 0.001	040
G279	05/13/2014	0.017	< 0.001	0.050	< 0.001	0.038	0.015	< 0.001	94	50	< 0.004	< 0.00Z	0.551	0.022	< 0.001		0.025	< 0.0002		< 0.001	110	< 0.001	
G279	08/12/2014	< 0.001	< 0.001			0.015	0.11	< 0.001	97	56				< 0.001	< 0.001		0.024				120		600
G279	10/14/2014	< 0.001	< 0.001			0.025	0.012	< 0.001	92	63				< 0.001	< 0.001		0.026				140		650
G279	01/21/2015	< 0.001	< 0.001		1	0.032	0.031	< 0.001	100	74		1	1	< 0.001	< 0.001		0.017	1			230		810
G279	04/13/2015	< 0.002	0.0013	0.029	< 0.001	0.047	0.016	< 0.001	170	46	< 0.004	< 0.002	0.518	< 0.002	< 0.001		0.0033	0.00024		0.0056	470	< 0.001	800
G279	07/23/2015	< 0.001	0.002	0.11	< 0.001	0.031	0.065	< 0.001		96	0.0042	0.0025	0.361	< 0.001	0.0041		0.013	< 0.0002	0.0015	0.02	470	< 0.001	1200
G279	10/08/2015	< 0.001	0.0015	0.096	< 0.001	1.3	1.4	< 0.001		120	0.0047	0.0033	< 0.25	< 0.001	0.0025		0.032	< 0.0002	0.0015	0.017	810	< 0.001	1700
G279	11/24/2015		< 0.001	0.053	< 0.001		0.63	< 0.001	140	61	< 0.004	< 0.002	0.334		0.0015	0.014		< 0.0002	< 0.001	0.0041	520	< 0.001	1100
G279	02/16/2016	< 0.001	< 0.001	0.082	< 0.001	0.29	0.26	< 0.001	180	130	< 0.004	< 0.002	0.392	< 0.001	< 0.001	0.012	0.011	< 0.0002	0.043	0.017	610	< 0.001	1500
G279	05/13/2016	< 0.001	< 0.001	0.055	< 0.001	0.11	0.073	< 0.001	120	31	< 0.004	< 0.002	0.608	< 0.001	< 0.001	< 0.01	0.0061	< 0.0002	0.024	0.0043	270	< 0.001	700
G279	08/03/2016	L	< 0.001	0.069	< 0.001	L	0.24	< 0.001	210	110	< 0.004	< 0.002	0.394		< 0.001	< 0.01		< 0.0002	< 0.001	0.02	570	< 0.001	1300

																							Total
		Arsenic.	Arsenic.	Barium.	Bervllium.	Boron.	Boron.	Cadmium.	Calcium.	Chloride.	Chromium.	Cobalt.	Fluoride.	Lead.	Lead.		Manganese.	Mercury.	Molybdenum.	Selenium.	Sulfate.	Thallium.	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G279	11/22/2016		< 0.001	0.057	< 0.001		0.49	< 0.001		130	< 0.004	< 0.002	0.272		< 0.001	0.011		< 0.0002	< 0.001	0.017	720	< 0.001	1300
G280	03/11/2008	< 0.01	< 0.01	0.049	< 0.01	< 0.5	< 0.5	< 0.005	63	47	< 0.05	< 0.01	0.37	< 0.005	< 0.005		0.11	< 0.0002		< 0.025	60	< 0.002	420
G280	04/21/2008	< 0.001	< 0.001	0.058	< 0.001	0.06	0.062	< 0.0005	59	51	< 0.005	< 0.001	0.37	< 0.0005	0.00055		0.18	< 0.0002		0.0028	58	< 0.002	400
G280	06/11/2008	< 0.001	< 0.001	0.052	< 0.001	< 0.05	< 0.05	< 0.0005	66	43	< 0.005	< 0.001	0.33	< 0.0005	< 0.0005		0.13	< 0.0002		0.0027	62	< 0.002	430
G280	08/13/2008	< 0.001	< 0.001	0.053	< 0.001	0.05	< 0.05	< 0.0005	63	44	< 0.005	< 0.001	0.39	< 0.0005	< 0.0005		0.13	< 0.0002		0.0026	59	< 0.002	410
G280	10/13/2008	< 0.001	< 0.001	0.05	< 0.001	< 0.05	< 0.05	< 0.0005	69	45	< 0.005	< 0.001	0.35	< 0.0005	< 0.0005		0.078	< 0.0002		< 0.0025	60	< 0.002	450
G280	12/03/2008	< 0.001	< 0.001	0.11	< 0.001	< 0.05	< 0.05	< 0.0005	120	110	< 0.005	< 0.001	0.26	< 0.0005	0.0007		0.24	< 0.0002		< 0.0025	230	< 0.002	760
G280	11/12/2009	< 0.001	< 0.001			< 0.05	< 0.05	< 0.0005	59 61	40				< 0.0005	< 0.0005		0.012				45		370
G280	01/28/2010	< 0.001	< 0.001			< 0.05	< 0.05	< 0.0005	60	40				< 0.0005	0.0006		0.019				42		400
G280	02/11/2010			0.039	< 0.001						< 0.005	< 0.001	0.32		0.0000		0.0000	< 0.0002		0.0042		< 0.002	440
G280	06/09/2010	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	66	65				< 0.001	< 0.001		0.0031				82		510
G280	07/27/2010	< 0.001	< 0.001			0.01	< 0.01	< 0.001	60	60				< 0.001	< 0.001		< 0.001				80		520
G280	11/16/2010	< 0.001	0.001			< 0.01	< 0.01	< 0.001	58	35				< 0.001	< 0.001		0.018				43		370
G280	01/28/2011	< 0.001	0.0012	0.082	< 0.001	< 0.01	0.012	< 0.001	100	57	< 0.004	0.0024	0.36	< 0.001	0.0055		< 0.001	< 0.0002		0.0051	76	< 0.001	500
G280	05/04/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	< 0.1	87				< 0.001	< 0.001		< 0.001				98		560
G280	07/27/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	63	45				< 0.001	< 0.001		0.015				54		420
G280	11/11/2011	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	60	51			-	< 0.001	< 0.001		0.0033				59		390
G280	01/30/2012	< 0.001	< 0.001	0.047	< 0.001	< 0.01	< 0.01	< 0.001	81	54	< 0.004	< 0.002	0.44	< 0.001	< 0.001		< 0.001	< 0.0002		0.0055	68	< 0.001	440
G280	05/22/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	65	67			-	< 0.001	< 0.001		< 0.001				93		470
G280	07/24/2012	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	/2	48				< 0.001	0.0026		0.0073				51		360
G280	11/14/2012	0.0013	0.0021	0.026	< 0.001	< 0.01	< 0.01	< 0.001	63	46	< 0.001	< 0.002	0 422	< 0.001	0.0014		0.0081	< 0.0002		0.0042	48	< 0.001	440
G280	01/31/2013	< 0.001	< 0.001	0.050	< 0.001	< 0.01	< 0.01	< 0.001	67	40 50	< 0.004	< 0.002	0.425	< 0.001	< 0.001		< 0.001	< 0.0002		0.0042	54 71	< 0.001	560 /10
G280	07/22/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	73	52				< 0.001	< 0.001		< 0.001				67		410
G280	10/14/2013	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	72	51				< 0.001	< 0.001		< 0.001				65		480
G280	02/19/2014	< 0.001	< 0.001	0.041	< 0.001	< 0.01	< 0.01	< 0.001	69	56	< 0.004	< 0.002	0.338	< 0.001	< 0.001		< 0.001	< 0.0002		0.0031	74	< 0.001	
G280	05/13/2014	< 0.001	0.0014			0.017	0.029	< 0.001	73	55				< 0.001	0.001		< 0.001				78		
G280	08/12/2014	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	75	55				< 0.001	< 0.001		< 0.001				76		490
G280	10/14/2014	< 0.001	0.0012			0.013	< 0.01	< 0.001	76	60				< 0.001	0.002		< 0.001				83		480
G280	01/21/2015	< 0.001	< 0.001			< 0.01	< 0.01	< 0.001	74	62				< 0.001	< 0.001		< 0.001				87		540
G280	04/13/2015	< 0.002	< 0.001	0.045	< 0.001	0.025	< 0.01	< 0.001	71	67	< 0.004	< 0.002	0.358	< 0.002	< 0.001		0.026	< 0.0002		0.0037	86	< 0.001	480
G280	07/23/2015	< 0.001	< 0.001	0.049	< 0.001	0.01	< 0.01	< 0.001		53	< 0.004	< 0.002	0.415	< 0.001	< 0.001		0.074	< 0.0002	0.0019	0.0033	74	< 0.001	480
G280	10/08/2015	< 0.001	< 0.001	0.056	< 0.001	0.025	0.024	< 0.001		54	< 0.004	< 0.002	0.318	< 0.001	< 0.001		0.0035	< 0.0002	0.0074	0.0017	92	< 0.001	450
G280	11/24/2015	.0.001	0.0066	0.11	< 0.001	0.012	0.029	< 0.001	120	54	0.019	0.0059	0.343	. 0. 001	0.012	0.019	.0.001	< 0.0002	0.0045	0.0032	94	< 0.001	460
G280	02/10/2016	< 0.001	< 0.001	0.048	< 0.001	0.012	< 0.01	< 0.001	60	55	< 0.004	< 0.002	0.466	< 0.001	0.0019	< 0.01	< 0.001	< 0.0002	0.0016	0.0033	84	< 0.001	410
G280	05/10/2016	< 0.001	< 0.001	0.045	< 0.001	< 0.01	< 0.01	< 0.001	63	50	< 0.004	< 0.002	0.429	< 0.001	< 0.001	< 0.01	< 0.001	< 0.0002	0.0014	0.0044	80	< 0.001	350
G280	03/13/2016	< 0.001	< 0.001	0.044	< 0.001	< U.U1	< 0.01	< 0.001	65	45 46	< 0.004	< 0.002	0.497	< 0.001	0.0014	< 0.01	< 0.001	< 0.0002	0.0014	0.0035	55	< 0.001	410 350
G280	11/20/2016		< 0.001	0.043	< 0.001		< 0.01	< 0.001	05	40	< 0.004	< 0.002	0.473		< 0.001	< 0.01		< 0.0002	0.0010	0.0034	67	< 0.001	430
G281	11/20/2015		0.0043	0.14	< 0.001		< 0.01	< 0.001	150	74	0.011	0.0056	0.349		0.0063	0.013		< 0.0002	0.0015	< 0.001	300	< 0.001	820
G281	02/11/2016		< 0.001	0.067	< 0.001		0.01	< 0.001	120	55	< 0.004	< 0.002	0.411		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	340	< 0.001	740
G281	05/10/2016		< 0.001	0.072	< 0.001		< 0.01	< 0.001	130	72	< 0.004	< 0.002	0.405		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	370	< 0.001	740
G281	08/01/2016		< 0.001	0.078	< 0.001		0.012	< 0.001	140	70	< 0.004	< 0.002	0.368		0.0011	< 0.01		< 0.0002	< 0.001	< 0.001	310	< 0.001	780
G281	08/31/2016	< 0.001				< 0.01								< 0.001			0.48						800
G401	11/21/2015		0.0046	0.037	< 0.001		3.3	< 0.001	440	3.6	0.0053	0.25	< 0.25		0.0031	0.055		< 0.0002	< 0.001	0.002	2300	< 0.001	3000
G401	02/22/2016		< 0.002	0.015	< 0.001		3.4	< 0.001	330	6	< 0.004	0.24	< 0.25		< 0.001	0.05		< 0.0002	< 0.001	< 0.002	2500	< 0.001	3000
G401	05/19/2016		0.0015	0.014	< 0.001		3.5	< 0.001	380	3	< 0.004	0.27	0.758		< 0.001	0.046		< 0.0002	< 0.001	< 0.001	2200	< 0.001	2800
G401	08/01/2016		0.004	0.053	< 0.001		4.1	0.0012	450	5.3	0.0096	0.28	< 0.25		0.0048	0.051		0.00093	< 0.001	0.0055	2100	< 0.001	2900
G401	11/17/2016		0.0027	0.021	< 0.001		4	0.0013		< 5	< 0.004	0.27	< 0.25		0.0029	0.054		< 0.0002	< 0.001	0.001	3400	< 0.001	3200
G402	11/21/2015		0.024	0.082	< 0.001		6.6	0.001	270	2.8	0.019	0.014	< 0.25		0.015	0.054		< 0.0002	0.006	0.002	1200	< 0.001	1700
G402	02/22/2016		0.027	0.11	< 0.001		5.7	< 0.001	220	2.8	0.031	0.015	0.355		0.018	0.057		< 0.0002	0.0049	0.0033	1000	< 0.001	1700
G402	05/19/2016		0.023	0.085	< 0.001		6.3	< 0.001	270	1.5	0.022	0.019	0.367		0.015	0.036		< 0.0002	0.0044	0.0016	960	< 0.001	1500

																							Total
		Arsenic,	Arsenic,	Barium,	Beryllium,	Boron,	Boron,	Cadmium,	Calcium,	Chloride,	Chromium,	Cobalt,	Fluoride,	Lead,	Lead,		Manganese,	Mercury,	Molybdenum,	Selenium,	Sulfate,	Thallium,	Dissolved
Well ID	Date	dissolved	total	total	total	dissolved	total	total	total L	total	total	total	total	dissolved	total	Lithium, total	dissolved	total	total	total	total	total	Solids
G402	08/02/2016		0.01	0.047	< 0.001		7.4	< 0.001	240	2.2	0.0085	0.0074	0.33		0.0072	0.033		< 0.0002	0.0033	< 0.001	890	< 0.001	1500
G402	11/17/2016		0.012	0.054	< 0.001		6.9	< 0.001		2.6	0.011	0.007	0.463		0.0079	0.047		< 0.0002	0.0034	0.0012	1100	< 0.001	1700
G403	11/23/2015		0.0017	0.14	< 0.001		0.039	< 0.001	78	6.8	0.0062	< 0.002	0.442		0.0021	< 0.01		< 0.0002	0.004	< 0.001	35	< 0.001	320
G403	02/22/2016		< 0.001	0.13	< 0.001		0.064	< 0.001	69	4.1	< 0.004	< 0.002	0.518		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	17	< 0.001	340
G403	05/18/2016		< 0.001	0.16	< 0.001		0.014	< 0.001	71	2.7	< 0.004	< 0.002	0.478		< 0.001	< 0.01		< 0.0002	0.0012	< 0.001	11	< 0.001	320
G403	08/01/2016		0.0033	0.24	< 0.001		0.027	< 0.001	140	4.5	0.0073	0.0029	0.485		0.0021	< 0.01		< 0.0002	0.0055	0.0068	9.9	< 0.001	320
G403	11/17/2016		0.0026	0.2	< 0.001		0.042	< 0.001		4	< 0.004	0.0024	0.539		< 0.001	< 0.01		< 0.0002	0.001	< 0.001	8.9	< 0.001	350
G404	11/21/2015		< 0.001	0.05	< 0.001		2.1	< 0.001	110	53	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	180	< 0.001	580
G404	02/15/2016		< 0.001	0.043	< 0.001		1.6	< 0.001	110	49	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	150	< 0.001	560
G404	05/19/2016		< 0.001	0.041	< 0.001		1.4	< 0.001	89	46	< 0.004	< 0.002	0.287		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	140	< 0.001	460
G404	08/02/2016		< 0.001	0.055	< 0.001		3.2	< 0.001	120	62	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	190	< 0.001	620
G404	11/22/2016		< 0.001	0.052	< 0.001		3.4	< 0.001		61	< 0.004	< 0.002	< 0.25		< 0.001	< 0.01		< 0.0002	< 0.001	< 0.001	380	< 0.001	880
G405	11/21/2015		0.014	0.051	< 0.001		17	0.0012	330	14	0.0051	0.0034	0.454		0.0085	< 0.01		< 0.0002	0.001	< 0.001	1700	< 0.001	2400
G405	02/15/2016		0.0028	0.018	< 0.001		16	< 0.001	320	11	< 0.004	< 0.002	0.459		0.0023	< 0.01		< 0.0002	< 0.001	< 0.001	1700	< 0.001	2500
G405	05/18/2016		0.0025	0.02	< 0.001		15	< 0.001	320	9.3	< 0.004	< 0.002	0.544		0.0011	< 0.01		< 0.0002	< 0.001	< 0.001	1800	< 0.001	2200
G405	08/02/2016		0.0063	0.028	< 0.001		17	< 0.001	280	3.4	< 0.004	0.0038	< 0.25		0.0018	< 0.01		< 0.0002	0.0016	< 0.001	1600	< 0.001	2200
G405	11/22/2016		0.0024	0.022	< 0.001		13	< 0.001		19	< 0.004	0.0021	0.525		0.0023	< 0.01		< 0.0002	0.0011	< 0.001	1400	< 0.001	2100
G406	08/30/2016	< 0.001				1.6								< 0.001			5.5						1300
G406	11/18/2016			0.024			1.9			4.4			0.345								910		1400
G407	08/30/2016	< 0.001				0.052								< 0.001			0.21						1400
G407	11/18/2016			0.024			0.078			14			0.289								830		1400
G45D	09/02/2016	0.0014				0.35								< 0.001			0.035						370
G46D	09/02/2016	0.0028				0.14								< 0.001			0.094						460

APPENDIX D

GROUNDWATER SAMPLING PROTOCOL

Groundwater Sampling Protocol

The following procedures shall be used in sampling groundwater at the site. This sampling protocol shall apply to the routine quarterly (or modified semi-annual or annual) sampling events. A sample collector's worksheet, comparable to the one located in Exhibit 1, may be used for noting relevant information in regard to each well.

Water Levels

Water levels shall be taken in each well prior to purging and/or sampling. Water levels should be taken as close together as practical, to prevent any time distortion of the water surface data. The following steps shall be followed to obtain accurate water level readings:

- 1. Note the general condition of the monitoring well on the worksheet. This shall include, but is not limited to the condition of the casing, the lock, evidence of tampering, condition of the pad, and any standing water.
- 2. Remove the lock and open the monitoring well. Note the condition of the interior of the casing and the condition of the well cap and riser. Open the cap, taking care not to allow dirt or foreign material into the monitoring well.
- 3. The technician shall rinse the probe and cable of the water level meter with decon water.
- 4. Slowly lower the probe into the monitoring well until the meter indicates the water surface has been reached.
- 5. Note the depth to water (to the nearest 0.01 ft) and the time on the worksheet.
- 6. Lower the probe to the bottom of well. (If a dedicated pump is installed in the well, skip this step). Note the well depth on the worksheet. The depth of the well will be measured on an annual basis, at wells that do not contain dedicated pumps. The depth of wells with dedicated pumps will be measured at least once every 5 years, or whenever the pump is removed.
- 7. Slowly remove the probe from the well. Rinse the probe and line with decon water.
- 8. Replace cap. Close and lock the well. Proceed to the next well, and repeat.

Purging of Monitoring Well – Pump Method

After all water level measurements have been taken, the monitoring wells shall be purged to provide a representative sample. Each groundwater monitoring well shall be purged by using a dedicated pump. The pump construction shall consist of inert materials consistent with the monitoring well construction (e.g., stainless steel pump bodies installed in stainless steel wells).

Purging shall be conducted utilizing a "low-flow" or minimal drawdown technique. Flow rates for this technique will typically fall below 0.5 liters/minutes, with an overall goal of not reducing the water level in the monitoring well by more than 0.3 ft during purging. Water levels should be checked frequently to ensure that the drawdown in the well does not exceed the 0.3-ft limits. Every 3 minutes to 5 minutes, readings shall be taken on the following water quality indicators to determine if a representative water sample is available.

- pH (in SU),
- Specific Conductance (in μmhos/cm or μS/cm),
- Temperature (in °F),
- And, it is suggested, at least one of the following:
 - Redox Potential (in mV);
 - Dissolved Oxygen (in mg/L); and/or
 - Turbidity (in NTU).

The water quality indicators will be considered stabilized when the following tolerances are reached after three consecutive readings:

- pH..... ±0.05 SU
- Redox Potential±10 percent
- Specific Conductance ±5 percer
- Temperature..... ±0.5°F
- Turbidity......±10 percent

Slow recovering wells require special consideration. If a well is dry, or is purged below the bottom of the pump intake, the well will be allowed to recharge for at least 12 hours. Samples shall be collected until all sample containers have been filled or the well becomes dry. Notes shall be kept on the worksheet with regard to water levels, times, volume of water removed, and any other parameters considered to be relevant.

Purging of Monitoring Well – Bailer Method

Purging and sample collection with a bailer shall be performed in the event of a non-functioning pump or from a well that does not have a dedicated pump installed. A sample shall be collected utilizing a factory packaged, clean, disposable bailer with an appropriate length of new, clean rope attached.

Calculate the number of bailer volumes of water needed to remove one (1) well volume of water.

Well Volume Calculations (2-inch well):

Schedule 40 PVC has an inside diameter of 2.067 inches.

 \therefore ((2.067 inches/12 inches/ft)/2)² • π • 1 ft of water = 0.0233 ft³/ft of water.

 $0.0233 \text{ ft}^3/\text{ft} \cdot 7.48 \text{ gallons/ft}^3 = 0.174 \text{ gallon/ft}$

Schedule 5 Stainless Steel (304 or 316) has an inside diameter of 2.245 inches.

:. $((2.245 \text{ inches/12 inches/ft})/2)^2 \cdot \pi \cdot 1 \text{ ft of water} = 0.0275 \text{ ft}^3/\text{ft of water}.$

 $0.0275 \text{ ft}^3/\text{ft} \cdot 7.48 \text{ gallons/ft}^3 = 0.206 \text{ gallon/ft}$

Volume of well (in gallons) = well type gallon/ft • (DTB - DTW); where, DTB = depth to bottom of well (from measuring point), and DTW = depth to water (from measuring point)

Bailer Volumes:

Disposable bailer volumes will vary by type and manufacturer. Volume information should be obtained before going to the site. For comparison, a 3 ft stainless steel bailer has a volume of approximately 1220 cc or 0.322 gallon and a 5 ft PVC bailer of approximately 1085 cc or 0.287 gallon.

Open monitoring well, being careful that no potential contaminant enters the well.

Remove one (1) bailer volume of water from the monitoring well. Test pH, specific conductance and temperature. Note values on worksheet. (Turbidity, redox potential and dissolved oxygen will vary considerably due to the agitation a bailer will cause in the well. Testing for these parameters is not recommended with this method.)

Remove one-half $(\frac{1}{2})$ gallon of water from the monitoring well. Test pH, specific conductance and temperature. Note values on worksheet.

Remove ½ to 1 gallon of water. Test pH, specific conductance and temperature. Record data on worksheet.

Repeat until pH, specific conductance and temperature stabilize or three (3) well volumes of water have been removed.
If the monitoring well becomes dry, or there is insufficient water to obtain all necessary samples, the monitoring well will be allowed to recharge for 24 hours. Samples shall be collected until all sample containers are filled or the well becomes dry. Notes shall be kept on the worksheet regarding water levels, times, volume of water removed, and any other parameters considered by the technician to be relevant.

If there is sufficient water volume in the monitoring well to obtain all samples, sample collection shall begin at this time.

Sample Collection Order

Samples shall be collected starting at the monitoring well with the least likelihood for contamination. Sampling shall proceed from the well with the lowest potential for contamination to the well with the highest potential for contamination.

Field Measurements

General

Upon arrival at each groundwater monitoring well, the technician shall note on the sampler's worksheet or in a field notebook the date, time, ambient air temperature, general weather conditions, and individuals present, including sample team members and any observers. (Note: Any observers shall need at a minimum, the same personal protective gear as the members of the sample team.)

Establish a "clean area" near the monitoring well where the sample containers and equipment can be stored while not in use. Every effort should be made to keep the sampling equipment and containers from contacting the ground surface. If necessary, a disposable, plastic tarp can be used as a ground cover to prevent potential contamination of the sample containers and equipment. Typically, the back of the field vehicle will be used as the "clean area".

Any non-dedicated sampling equipment (meter probes, thermometers, etc.) shall be washed in a commercial, laboratory cleaner (Alconox®, Liquinox®, or equivalent), and thoroughly rinsed in decon water before each use. Calibration shall be performed at each new monitoring location after the initial decontamination. After use, each device shall be powered down (if necessary) decontaminated, and stored in its manufacturer-approved container.

Temperature

Obtain a water sample from the well. Place the sample aliquot in a disposable container, insert the thermometer (or electronic probe), wait until the readings have stabilized, and record the temperature on the worksheet. Temperature for a glass thermometer should be noted to the nearest degree Fahrenheit (1°F). For electronic thermometers (thermocouples), temperature should be noted to the nearest tenth degree Fahrenheit (0.1°F). The thermometer or probe shall be cleaned and rinsed with decon water after use.

<u>рН</u>

Confirm calibration of the instrument by comparing with an appropriate buffer solution. Adjust for temperature compensation (if meter is not self-compensating). Rinse probe with decon water. Obtain a sample from the well and place the probe in sample aliquot. Note the pH and record on the sample worksheet. Note pH readings to the nearest tenth unit (0.1).

Specific Conductance

Confirm calibration of the instrument by comparing against an appropriate buffer solution. Adjust for temperature compensation (if meter is not self-compensating). Rinse the probe with decon water. Obtain a sample from the well and place the probe in sample aliquot. Note the specific conductance and record on the sample worksheet. Specific conductance should be noted to the nearest micromhos per centimeter (μ mhos/cm) or microSiemens per centimeter (μ S/cm).

Sample Collection Procedures

Jars and vials may ship pre-labeled from the laboratory, identifying the analysis and preservative for each type of sample. Dependent upon circumstances, sample containers may be prepared by non-laboratory personnel. If so, this should be noted on the sample worksheet or in the field notebook.

A technician shall remove a sample container from the cooler, affix a label, and in indelible, waterproof ink write the well number and/or sample I.D., the facility name, the sample collection date and time, the type of sample in the container, and the sample collector's name. A technician shall organize the containers in the following sampling order:

- Metals and Minerals (dissolved)
- Anions (dissolved)
- Total Dissolved Solids (TDS)
- Cyanides (total)

Dissolved parameters include dissolved metals and minerals, total dissolved solids (TDS), and nitrogen should be field filtered. Samples should be filtered using a 0.45-micron filter attached to the sample pump line. Other filter apparatus may be utilized as long as Illinois EPA guidelines are followed. Filters should be replaced no less frequently than at each new well, and may need to be replaced more often if flow is restricted due to particulate matter in the sample water.

Transportation of Monitoring Samples

Sample Preservation Techniques

The preservation techniques utilized in the groundwater samples will typically adhere to those listed in *Handbook for Sampling and Sample Preservation of Water and Wastewater*, U.S. EPA, EPA-600/4-82-029, September 1982 and/or *Test Methods for Evaluating Solid Wastes, Physical/Chemical Methods*,

EPA/530/SW-846, 3^{rd.} Edition, Final Update IV (January 2008).

Transportation of Samples

Samples shall be transported to the laboratory in sealed, insulated shipping containers, ice chests, or coolers. The shipping containers should be sturdy, and if samples are contained in glass bottles, dividers and/or bubble wrap should be used to restrict potential breakage. All samples will be packed in ice or a packaged refrigerant as necessary for proper preservation. Samples should be packed to maintain sample temperatures as close to 4°C (degrees Celsius) or 39°F as possible from the time the samples are collected to the time the samples are received by the laboratory. The samples should be shipped/delivered to the laboratory as soon as practical, preferably within 24 hours of sample collection.

All samples shall be accompanied by a chain-of-custody record. The sampler shall retain a copy of the record and forward the original with the samples to the analytical laboratory. Once the laboratory has received the samples, a representative from the laboratory is to complete the record, retain the original and return a copy with the chemical analysis reports to the sampler. The chain-of-custody shall contain the facility name, the wells sampled, time and date of sampling, members of the sampling party, type of samples (i.e. water, soil, leachate, etc.), number of sample bottles, requested analysis, overnight courier, etc. A sample chain-of-custody record is provided in Exhibit 2.

Attachments

Exhibit 1: Groundwater Sampling Worksheet Exhibit 2: Example Chain-of-Custody Record

ATTACHMENT J

Memorandum



Date: 25 October 2021

Subject: 35 I.A.C. Section 845.430 – Slope Maintenance Documentation for Ash Pond No. 2 at Coffeen Power Plant

Illinois Power Generating Company operates the coal fired Coffeen Power Plant (Plant) located in Montgomery County, Illinois. The Ash Pond No. 2 is a closed inactive surface impoundment storing coal combustion residuals (CCR). The requirements for the Ash Pond No. 2 are found in 35 Ill. Admin. Code 845, Standards for the Disposal of Coal Combustion Residuals in Surface Impoundments (Part 845).

Pursuant to Part 845, Section 845.230(d)(3)(C), the initial operating permit application for inactive closed CCR surface impoundments that have completed an Agency approved closure before prior to July 30, 2021, must contain documentation that the CCR surface impoundment, if not incised, will be operated, and maintained with one of the forms of slope protection specified in Section 845.430. This statement addresses the requirements of Part 845, Section 845.430 Slope Maintenance, which states:

<u>Section 845.430</u>: The slopes and pertinent surrounding areas of the CCR surface impoundment must be designed, constructed, operated, and maintained with one of the forms of slope protection specified in subsection (a) that meets all the performance standards of subsection (b).

<u>Section 845.430(a)</u>: Slope protection must consist of one of the following: 1) A vegetative cover consisting of grassy vegetation; 2) An engineered cover consisting of a single form or combination of forms of engineered slope protection measures; or 3) A combination of the forms of cover specified in subsections (a)(1) or (a)(2).

<u>Section 845.430(b):</u> Any form of cover for slope protection must meet the following performance standards: 1) The cover must be installed and maintained on the slopes and pertinent surrounding areas of the CCR surface impoundment; 2) The cover must provide protection against surface erosion, wave action, and adverse effects of rapid drawdown; 3) The cover must be maintained to allow for the observation of, and access to, the slopes and pertinent surrounding areas during routine and emergency events; 4) Woody vegetation must be removed from the slopes or pertinent surrounding areas. Any removal of woody vegetation with a diameter greater than 1/2 inch must be directed by a person familiar with the design and operation of the CCR surface impoundment and in consideration of the complexities of removal of a tree or a shrubbery, who must ensure the removal does not create a risk of destabilizing the CCR surface impoundment or otherwise adversely affect the stability and safety of the CCR surface impoundment or

Memorandum (cont'd)



35 I.A.C. Part 845 – Slope Maintenance Documentation for Ash Pond No. 2 at Coffeen Power Plant
25 October 2021
Page 2

personnel undertaking the removal; and 5) The height of vegetation must not exceed 12 inches.

Slope protection, consisting of vegetative cover, was installed on the slopes and pertinent surrounding areas of the Coffeen Ash Pond 2, and is inspected, maintained, and repaired as needed. Based on observations from weekly inspections conducted in accordance with Section 845.540(a), and the 2020 annual inspections conducted by Hanson Professional Services Inc., the vegetative cover is described to be in good working condition with a maximum vegetation height of 12 inches. The owner's Operations and Maintenance Plan (O&M Plan) provides details for maintaining grass and removing woody vegetation and addressing erosion features on the slopes. Based on a review of the documentation described above, the owner is implementing the O&M Plan, including the completion of repairs and maintenance as needed and when issues are identified during weekly and/or annual inspections. The slope maintenance portion of the O&M Plan and the Annual Inspection performed by Hanson in 2020 are included in Attachment J. The surface impoundment slope protection (vegetative cover) installed and maintained on the slopes and pertinent areas around the slopes is depicted in the aerial photograph provided below.



Excerpt from the Coffeen AP2 Operations and Maintenance Manual

1.0 Dam Safety Requirements

- 1.1 Dam Safety Inspections The plant's impoundment and flood prevention structures shall be inspected and maintained in a manner to ensure safe and environmentally responsible operations. A regular maintenance program shall be performed and shall consist of the following inspection items:
 - 1. Earth embankments: Walk the crest, side slopes, and downstream toe of the dam concentrating on surface erosion, seepage, cracks, settlement, slumps, slides, and animal burrows. Frequency of inspection: Weekly.
 - 2. Vegetation: Grass should be a thick vigorous growth to stabilize the earth embankment soils and prevent erosion form occurring. There should be NO trees on the earth embankment and none within a minimum of 20 feet of the embankment toe or other structures. Mowing frequency: Semiannually.
 - 3. Well Readings: Record level of wells on the crest and toe of the berm. Frequency: Quarterly.
 - 4. Special Inspections Special inspections of the levees and ash pond berms shall be performed after earthquakes, floods, water level exceedance in the ponds, or heavy rainfall events. Inspection and report shall be equal to an annual inspection level of detail. Water level in the pond should be noted after a heavy rainfall. Dam Safety staff shall accompany plant personnel on special inspections. Frequency: As required.

Name of Dam	Coffe	een Ash Pond #2 Dam Dam ID No None					
Permit Number	None		Class of Da	m <u>NA</u>			
Location SW 1/4	Section 1	<u>1 </u>	7N	Range	3W		
Owner Dyneg	gy Midwest Ge Name	neration, LLC	_	217-5 Telephone	534-7668 Number (Day)		
13	34 CIPS Lane			217-	534-7668		
	Street		-	Telephone	Number (Night)		
Coffeen, IL City	520 Zip 0	017Count	/	Montgomery			
Type of Dam		E	Earth Embank	ment			
Type of Spillway	Surface	Riprapped Cha	annels Around	Perimeter of Er	nbankment		
Date(s) Inspected			16-Nov-20	0			
Weather When Insp	ected		Clear				
Temperature When	Inspected		50° F				
Pool Elevation Whe	n Inspected		NA - pond	is closed and ca	pped		
Tailwater Elevation	When Inspecte	ed	NA				
WWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWW	NAL SHILL	Inspection	n Personnel:				
ENS.	NGIN	James Kr	utelski, PE	Geotechr	nical Engineer		
들러/JAMES P. KNUT 062-05420	ELSKI		Name		Title		
		Jason Ca	mpbell, PE	Dynegy Dam	Safety Manager		
HILLINO FILLINO	Summun		Name		Title		
A/1/ + 10	12012020	Gina Krar	ner	IDN	R-OWR		
4 Kul 19			Name	,	Title		
EYP 11/30/	21	John Ron	nang	Coffe	en Plant		
Professional Éngine	er's Seal		Name		Title		

Dam Inspection Report

The Department of Nautural Resources is requesting information that is necessary to accomplish the statutory purpose as outlined under the River, Lakes and Streams Act, 615 ILCS 5. Submittal of this information is REQUIRED. Failure to provide the required information could result in the initiation of non-compliance procedures as outlined in Section 3702.160 of the "Rules for Construction and Maintenance of Dams".

CONDITION CODES

- NE No evidence of a problem
- GC Good condition
- MM Item needing minor maintenance and/or repairs within the year, the safety or integrity of the item is not yet imperiled
- IM Item needing immediate maintenance to restore or ensure its safety or integrity
- EC Emergency condition which if not immediately repaired or other appropriate measures taken could lead to failure of the dam
- OB Condition requires regular observation to ensure that the condition does not become worse
- NA Not applicable to this dam
- NI Not inspected list the reason for non-inspection under deficiencies

EARTH EMBANKMENT

	CONDITION		RECOMMENDED REMEDIAL MEASURES
ITEM	CODE	DEFICIENCIES	AND IMPLEMENTATION SCHEDULE
Surface Cracks	NE		
Vertical and Horizontal Alignment of Crest	GC		Pond has been closed and re-graded.
Unusual Movement or Cracking At or Beyond Toe	NE		
Sloughing or Erosion of Embankment and Abutment Slopes	NE		
Upstream Face Slope Protection	NA		
Seepage	NA		
Filter and Filter Drains	NA		

EARTH EMBANKMENT

(Continued)

	CONDITION		RECOMMENDED REMEDIAL MEASURES
ITEM	CODE	DEFICIENCIES	AND IMPLEMENTATION SCHEDULE
Animal Damage	NE		Ash pond is under construction for capping and closure. Contractor is responsible for all items during construction.
Embankment Drainage Ditches	GC		
Vegetative Cover	NE		
Let-down Channels	GC		
Other			
Other			
Other			

PRINCIPAL SPILLWAY APPROACH CHANNEL

	CONDITION		RECOMMENDED REMEDIAL MEASURES
ITEM	CODE	DEFICIENCIES	AND IMPLEMENTATION SCHEDULE
Debris	NE		Ash pond is under construction for capping and closure. Contractor is responsible for all items during construction.
Side Slope Stability	NE		
Slope Protection	GC		
Other (Name)	NA		
Other	NA		
Other	NA		
Other	NA		

EMERGENCY SPILLWAY

Earth			Other: Name	None
ITEM	CONDITION CODE	DEFICIENCIES	RECOMMENDED REMEDIA AND IMPLEMENTATION SC	L MEASURES HEDULE
Erosion	NA			
Weeds, Logs, Other Obstructions	NA			
Side Slope Sloughing	NA			
Vegetation	NA			
Sedimentation	NA			
Riprap	NA			
Settlement of Crest	NA			
Downstream Channel	NA			
Other (Name)	NA			

SUMMARY OF MAINTENANCE DONE AND/OR

REPAIRS MADE SINCE THE LAST INSPECTION

DATE OF PRESENT INSPECTION	16-Nov-20
DATE OF LAST INSPECTION	22-Oct-19

1. <u>EARTH EMBANKMENT DAMS</u> Ash pond is under construction for capping and closure. Contractor is responsible for all items during construction.

2. CONCRETE MASONRY DAMS

NA

3. PRINCIPAL SPILLWAY

None Noted

4. OUTLET WORKS

NA

5. EMERGENCY SPILLWAY

NA

Owner's Maintenance Statement

None ance with the a	, in _ accepted n	Montgo naintenanc Signatu	omery e plan wh	_ County, hich is part of
ance with the .	accepted n	naintenanc	e plan wh	nich is part of
<u> </u>		Signatu		
		0	ire	
		Da	te	
on and Mai	ntenanc	e Plan S	tatemer	nt
	,	owner of	Coffeen #2	Ash Pond Dam,
None	_, in _	Montgo	omery	County,
naintenance p	olan includir	ng the Eme	ergency A	ction Plan (EA
	None			
ve enclosed th	e appropria	ate revisior	is or	
ve determined	that no rev	visions to th	ne plan ar	e necessary.
		Signatu	ire	
		Da	te	
	on and Mai	on and Maintenance , , in, maintenance plan includir None /e enclosed the appropria /e determined that no rev	on and Maintenance Plan S, owner of, owner of	on and Maintenance Plan Statemer Coffeen

The Department of Nautural Resources is requesting information that is necessary to accomplish the statutory purpose as outlined under the River, Lakes and Streams Act, 615 ILCS 5. Submittal of this information is REQUIRED. Failure to provide the required information could result in the initiation of non-compliance procedures as outlined in Section 3702.160 of the "Rules for Construction and Maintenance of Dams".



East slope and interior



East slope



East toe area





South slope and interior



West slope



North slope



Southwest corner - lined ditch



Let-down structure - typical

ATTACHMENT K

POST-CLOSURE PLAN FOR EXISTING CCR SURFACE IMPOUNDMENT 40 C.F.R. § 257.104 and 35 I.A.C. 845.780 REV 0 –10/30/2021

	- ((/ 40		1007		
Site Name / Address	Coffeen Power Plant / 134 Cips Lane, Coffeen, IL 61207 Illinois Power Generating Company / 6555 Sierra Drive Irving, Texas 75039					
Owner Name / Address	Illinois Power Ge	nerating	, Company / 6555 Sierra D	rive irving, Texas 75039		
CCR Unit	Ash Pond No. 2		Closure Method and	Closed In-Place		
			Final Cover Type	Vegetation Cover		
POST-CLOSURE PLAN DESCRIP	TION					
40 C.F.R. § 257.104(c)(1) and 35 I.A.C. Length of post-closure care period.	845.780(c)(1) –	Post-c by 40 provid	losure care will be conduc C.F.R. § 257.104(c)(1) and ed by 40 C.F.R. § 257.104	cted for a period of 30 years as required 35 I.A.C. 845.780(c)(1), except as (c)(2) and 35 I.A.C. 845.780(c)(2).		
40 C.F.R. § 257.104(c)(2) and 35 I.A.C. 845.780(c)(2) – Circumstances extending the post closure care period.			e end of the post-closure assessment monitoring in e care as described in this ion monitoring in accorda	care period the CCR unit is operating accordance with §257.95, the post- plan will continue until returning to nce with §257.95.		
		Under extence concer Section backgr standa have b are pro	35 I.A.C. 845.780(c)(2), the ded until groundwater mon trations are below the gr n 845.600 and are not incu- round, using the statistical ords in Section 845.640(f) a seen reduced to the maxim- pathetic of human health	e post-closure care period will be nitoring data demonstrate that oundwater protection standards in reasing for those constituents over l procedures and performance and (g), provided that concentrations num extent feasible and concentrations and the environment.		
40 C.F.R. § 257.104(d)(1)(i) and 35 I.A 845.780(d)(1)(A) – A description of th maintenance activities required in40 (257.104(b) and 35 I.A.C. 845.780(b), a at which these activities will be perfor the integrity and effectiveness of the system, maintain the groundwater mo and monitor the groundwater.	.C. e monitoring and C.F.R. § nd the frequency med, to maintain final cover onitoring system	Pursua post-ci system annua damag final co will be Noted compo mainta Vegeta includi erosio cover s causeo	ant to § 257.104(b)(1) and losure care period, period n and stormwater manage lly for evidence of settlem ge that may adversely affe over system. When praction made concurrent with ge evidence of damage, such onents, rills, surface cracks ain the integrity and effect ation will be established a ing storm drainage areas, n control. Established veg system will prevent poten d by run-on and run-off.	35 I.A.C. 845.780(b)(1), throughout the ic visual observations of the final cover ment system will be performed at least ient, subsidence, erosion, or other ct the integrity and effectiveness of the cal, visual observations of the final cover roundwater monitoring activities. In as damage to the geosynthetic is and settlement, will be repaired to tiveness of the final cover system. Ind maintained on the final cover system, where appropriate, to provide long-term getation and the slope design of the final tial erosion and damage that may be		
		Repair replac	activities may include, bu ing damaged geosyntheti	it are not limited to, repairing or c components, replacing and		

	compacting soil cover, repairing drainage channels that have been eroded, filling in depressions with soil, regrading, and reseeding areas of failed vegetation, as necessary. Pursuant to § 257.104(b)(3) and 35 I.A.C. 845.780(b)(3), the groundwater monitoring system will be maintained, and groundwater will be monitored as required by 40 C.F.R. § 257.90 through 40 C.F.R. § 257.98 and 35 I.A.C. 845.600 through 35 I.A.C. 845.680. Monitoring wells will be inspected during each groundwater sampling event. Monitoring wells and associated instrumentation will be maintained so that they perform to the design specifications throughout the life of the monitoring program. Groundwater monitoring frequency will be at least quarterly, except as provided in 40 C.F.R. § 257.94(d), 257.94(c), and 35 I.A.C. 845.650(b)(4).
40 C.F.R. § 257.104(d)(1)(ii) and 35 I.A.C. 845.780(d)(1)(B) – The name, address, telephone number and email address of the person or office to contact about the facility during the post-closure care period.	Illinois Power Generating Company 6555 Sierra Drive Irving, Texas 75039 800.633.4704 <u>ccr@dynegy.com</u>
40 C.F.R. § 257.104(d)(1)(iii) and 35 I.A.C. 845.780(d)(1)(C) – A description of the planned uses of the property during the post-closure period.	The CCR unit is located at a retired electric generation facility. Planned uses of the property during the post-closure period are currently unknown, except for post-closure care of the CCR unit. Post-closure use of the property will not disturb the integrity of the final cover system or other components of the containment system, or the function of the monitoring systems unless necessary to comply with the requirements of 40 C.F.R. Part § 257, Subpart D and 35 I.A.C. Part 845. Any other disturbance will be conducted following a demonstration that it will not increase the potential threat to human health or the environment, as required by 40 C.F.R. § 257.104(d)(1)(iii) and 35 I.A.C. 845.780 (d)(1)(C). The demonstration will be certified by a qualified professional engineer and submitted to the Illinois Environmental Protection Agency (IEPA). Per 40 C.F.R. § 257.104(d)(1)(iii) notification shall be provided to the State Director that the demonstration has been placed in the operating record and on the owners or operator's publicly accessible internet site. This CCR unit is closed. A notation on the deed to the property, or some other instrument that is normally examined during title search, will be recorded in accordance with 40 C.F.R. § 257.102(i) and 35 I.A.C. 845.760(h). The notation will notify potential purchasers of the property that the land has been used as a CCR unit and its use is restricted under the post-closure care requirements in 40 C.F.R. § 257.104(d)(1)(iii) and 35 I.A.C. 845.780(d)(1)(C) or groundwater monitoring requirements per 35 I.A.C. 845.740(b). Within 30 days of recording the deed notation, a notification stating that the notation has been recorded will be submitted to the IEPA and placed in the facility's operating record per 35 I.A.C. 845.760(h)(3). The notification will be placed on the owner or operator's publicly accessible CCR Web site in accordance with 40 C.F.R. § 257.107(i)(9) and 35 I.A.C. 845.800(d)(26) and §257.105(i)(9).

40 C.F.R. § 257.104(d)(3)(and 35 I.A.C. 845.780(d)(3) – Amendments to the initial or subsequent written post- closure plan.	Pursuant to 40 C.F.R. § 257.104(d), the initial post closure care plan for the Coffeen Ash Pond No. 2 was prepared on October 17, 2016. That plan is being amended pursuant to 40 C.F.R. § 257.104(d)(3)(i). This plan also serves as the initial post-closure care plan, prepared in accordance with 35 I.A.C. 845.780(d). Pursuant to § 257.104(d)(3) and 35 I.A.C. 845.780(d)(3), an operating
	permit modification application to amend the initial or any subsequent written post-closure care plan developed under 35 I.A.C. 845.780 (d)(1) and § 257.104(d)(1) will be submitted to IEPA. The written post-closure care plan will be amended whenever there is a change in the operation of the CCR surface impoundment that would substantially affect the written post-closure care plan in effect; or unanticipated events necessitate a revision of the written post-closure care plan, after post- closure activities have started.
	The written post-closure care plan will be amended at least 60 days before a planned change in the operation of the facility or CCR surface impoundment, or within 60 days after an unanticipated event requires the need to revise the existing plan. If the plan is revised after post- closure activities have started, a request to modify the operating permit, including an amended written post-closure care plan, will be submitted to the IEPA within 30 days following the triggering event.
40 C.F.R. § 257.104(d)(4) and 35 I.A.C. 845.780(d)(4) – Qualified professional engineering certification.	Certification by a qualified professional engineer will be appended to this plan and any amendment of this plan.
35 I.A.C. 845.780(e) – Termination of post-closure care.	Upon completion of the post-closure period, a request to terminate post-closure care will be submitted to the IEPA. The request will include a certification by a qualified professional engineer verifying that post- closure care has been completed in accordance with the post-closure care plan specified in 35 I.A.C. 845.780(d) and the requirements of 35 I.A.C. 845.780.
40 C.F.R. § 257.104(e) and 35 I.A.C. 845.780(f) – Notification of completion of the post-closure care period.	A notification of completion of post-closure care will be prepared and placed in the facility's operating record within 30 days after IEPA approval of the request to terminate post-closure care. The notification will be placed in the facility's operating record in accordance with 35 I.A.C. 845.800(d)(31) and § 257.105(i)(13).
	The notification will be placed on the owner or operator's publicly accessible CCR Internet site in accordance with the requirements of § 257.107(i)(13) and 35 I.A.C. 845.810(e). The IEPA will be notified when the notification has been placed in the operating record and on the owner or operator's publicly accessible Internet site in accordance with the requirements of § 257.106(i)(13).

Certification Statement 40 C.F.R. § 257.104 (d)(4) and 35 I.A.C. 845.780(d)(4) – Amended/Initial Written Post Closure Plan for a CCR Surface Impoundment

CCR Unit: Illinois Power Generating Company; Coffeen Power Plant; Ash Pond No. 2

I, John R. Hesemann, being a Registered Professional Engineer in good standing in the State of Illinois, do hereby certify, to the best of my knowledge, information, and belief, that the information contained in this certification has been prepared in accordance with the accepted practice of engineering. I certify, for the above referenced CCR Unit, that the information contained in the amended/initial written post closure plan, dated October 30, 2021, meets the requirements of 40 C.F.R. § 257.104 and 35 I.A.C. 845.780.

John R. Hesemann

Printed Name



9/27/2021

Date

ATTACHMENT M



HISTORY OF POTENTIAL EXCEEDANCES

This presentation of the History of Potential Exceedances, and any corrective action taken to remediate groundwater, is provided to meet the requirements of Title 35 of the Illinois Administrative Code (35 I.A.C.) § 845.230(d)(2)(M) for the Coffeen Power Plant Ash Pond No. 2, Illinois Environmental Protection Agency (IEPA) ID No. W1350150004-02.

<u>Note</u>

Groundwater concentrations observed from 2015 to 2021 in monitoring wells included in an existing groundwater monitoring program or installed in 2021 have been evaluated and summarized in the following tables. These concentrations are considered potential exceedances because the methodology used to determine them is proposed in the Statistical Analysis Plan (Appendix A to Groundwater Monitoring Plan [GMP]), which has not been reviewed or approved by IEPA at the time of submittal of the 35 I.A.C. § 845 Operating Permit application.

Alternate sources for potential exceedances as allowed by 35 I.A.C. § 845.650(e) have not yet been evaluated. These will be evaluated and presented in future submittals to IEPA as appropriate.

Table 1 summarizes how the potential exceedances were determined. Table 2 is a summary of all potential exceedances.

Results for G154, G279, G410, and G411 are provided in Tables 1 and 2 because they were included in the previously approved GMP. As discussed in the GMP Addendum, wells G154, G279, G410, and G411 have been removed from the monitoring program because these locations are not downgradient of Ash Pond No. 2.

Background Concentrations

Background monitoring wells identified in the GMP include G281, G270, and G280.

Background concentrations calculated from sampling events in 2015-2017 were compared to the standards identified in 35 I.A.C. § 845.600(a)(1). For constituents with calculated background concentrations in 2015-2017 greater than the standards in 35 I.A.C. § 845.600(a)(1), those calculated background concentrations were used as Groundwater Protection Standards (GWPSs) for comparing to statistical calculation results for each compliance well to determine potential exceedances. Compliance well statistical calculations consider concentrations from all sampling events in 2015-2021.

Corrective Action

A Corrective Measures Assessment (CMA) was completed to address statistically significant levels of total cobalt and total lithium, as required by 40 C.F.R. § 257.96. The CMA indicated the source control measure consists of closure in place with a geomembrane cover system in accordance with the Closure and Post Closure Care Plan submitted to the IEPA in January 2017 and approved on January 30, 2018. Closure construction began in July of 2019 and was completed on November 17, 2020.

Activities completed associated with the selection of a groundwater remedy include review of existing groundwater and source water data, and collection of additional samples of groundwater, source water, surface water, and aquifer solids to support analysis of natural attenuation



mechanisms, rates, and aquifer capacity. Preliminary results indicate that site-specific conditions are favorable for implementation of monitored natural attenuation (MNA) in combination with the recently completed closure referenced above.

Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G154	UA	Antimony, total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G154	UA	Arsenic, total	mg/L	10/13/2020 - 08/18/2021	CI around median	0	0.010	0.0043	0.01	Standard
G154	UA	Barium, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	0.036	2.0	0.18	2	Standard
G154	UA	Beryllium, total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G154	UA	Boron, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	0.010	2.0	0.022	2	Standard
G154	UA	Cadmium,total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G154	UA	Chloride, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	8.3	200	75	200	Standard
G154	UA	Chromium, total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.004	0.10	0.011	0.1	Standard
G154	UA	Cobalt, total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.002	0.006	0.0056	0.006	Standard
G154	UA	Fluoride, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	0.61	4.0	0.49	4	Standard
G154	UA	Lead, total	mg/L	10/13/2020 - 08/18/2021	CI around median	0	0.0075	0.0063	0.0075	Standard
G154	UA	Lithium, total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.020	0.040	0.013	0.04	Standard
G154	UA	Mercury, total	mg/L	10/13/2020 - 08/18/2021	CI around median	0	0.002	0.0002	0.002	Standard
G154	UA	Molybdenum, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	0.000919	0.10	0.0015	0.1	Standard
G154	UA	pH (field)	SU	01/21/2015 - 08/18/2021	CI around mean	6.9	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G154	UA	Radium-226 + Radium 228, tot	pCi/L	10/13/2020 - 08/18/2021	CI around mean	-0.901	5.0	2.0	5	Standard
G154	UA	Selenium, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	0.00111	0.050	0.0012	0.05	Standard
G154	UA	Sulfate, total	mg/L	10/13/2020 - 08/18/2021	CI around mean	66	400	370	400	Standard
G154	UA	Thallium, total	mg/L	10/13/2020 - 08/18/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G154	UA	Total Dissolved Solids	mg/L	01/21/2015 - 08/18/2021	CI around mean	422	1200	840	1200	Standard
G279	UA	Antimony, total	mg/L	04/13/2015 - 08/18/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G279	UA	Arsenic, total	mg/L	01/21/2015 - 08/18/2021	CI around median	0.001	0.010	0.0043	0.01	Standard
G279	UA	Barium, total	mg/L	04/13/2015 - 08/18/2021	CB around linear reg	0.020	2.0	0.18	2	Standard
G279	UA	Beryllium, total	mg/L	04/13/2015 - 08/18/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G279	UA	Boron, total	mg/L	01/21/2015 - 08/18/2021	CI around geomean	0.085	2.0	0.022	2	Standard
G279	UA	Cadmium,total	mg/L	01/21/2015 - 08/18/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard



Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G279	UA	Chloride, total	mg/L	01/21/2015 - 08/18/2021	CI around median	57	200	75	200	Standard
G279	UA	Chromium, total	mg/L	04/13/2015 - 08/18/2021	CI around median	0.004	0.10	0.011	0.1	Standard
G279	UA	Cobalt, total	mg/L	04/13/2015 - 08/18/2021	CI around median	0.002	0.006	0.0056	0.006	Standard
G279	UA	Fluoride, total	mg/L	04/13/2015 - 08/18/2021	CI around mean	0.33	4.0	0.49	4	Standard
G279	UA	Lead, total	mg/L	01/21/2015 - 08/18/2021	CI around median	0.001	0.0075	0.0063	0.0075	Standard
G279	UA	Lithium, total	mg/L	11/24/2015 - 08/18/2021	CI around median	0.010	0.040	0.013	0.04	Standard
G279	UA	Mercury, total	mg/L	04/13/2015 - 08/18/2021	CI around median	0.0002	0.002	0.0002	0.002	Standard
G279	UA	Molybdenum, total	mg/L	07/23/2015 - 08/18/2021	CB around T-S line	0.000541	0.10	0.0015	0.1	Standard
G279	UA	pH (field)	SU	01/21/2015 - 08/18/2021	CI around mean	6.9	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G279	UA	Radium-226 + Radium 228, tot	pCi/L	11/24/2015 - 08/18/2021	CI around mean	0.65	5.0	2.0	5	Standard
G279	UA	Selenium, total	mg/L	04/13/2015 - 08/18/2021	CB around linear reg	-0.0027	0.050	0.0012	0.05	Standard
G279	UA	Sulfate, total	mg/L	01/21/2015 - 08/18/2021	CI around geomean	323	400	370	400	Standard
G279	UA	Thallium, total	mg/L	04/13/2015 - 08/18/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G279	UA	Total Dissolved Solids	mg/L	01/21/2015 - 08/18/2021	CI around geomean	938	1200	840	1200	Standard
G401	UA	Antimony, total	mg/L	11/21/2015 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G401	UA	Arsenic, total	mg/L	11/21/2015 - 08/17/2021	CI around geomean	0.00191	0.010	0.0043	0.01	Standard
G401	UA	Barium, total	mg/L	11/21/2015 - 08/17/2021	CI around geomean	0.019	2.0	0.18	2	Standard
G401	UA	Beryllium, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.001	0.004	0.001	0.004	Standard
G401	UA	Boron, total	mg/L	11/21/2015 - 08/17/2021	CI around median	3.5	2.0	0.022	2	Standard
G401	UA	Cadmium,total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.001	0.005	0.001	0.005	Standard
G401	UA	Chloride, total	mg/L	11/21/2015 - 08/17/2021	CI around geomean	2.7	200	75	200	Standard
G401	UA	Chromium, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.004	0.10	0.011	0.1	Standard
G401	UA	Cobalt, total	mg/L	11/21/2015 - 08/17/2021	CI around mean	0.22	0.006	0.0056	0.006	Standard
G401	UA	Fluoride, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.25	4.0	0.49	4	Standard
G401	UA	Lead, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.001	0.0075	0.0063	0.0075	Standard
G401	UA	Lithium, total	mg/L	11/21/2015 - 08/17/2021	CI around geomean	0.039	0.040	0.013	0.04	Standard



Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G401	UA	Mercury, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.0002	0.002	0.0002	0.002	Standard
G401	UA	Molybdenum, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.001	0.10	0.0015	0.1	Standard
G401	UA	pH (field)	SU	11/21/2015 - 08/17/2021	CI around mean	6.0	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G401	UA	Radium-226 + Radium 228, tot	pCi/L	11/21/2015 - 08/17/2021	CI around geomean	0.74	5.0	2.0	5	Standard
G401	UA	Selenium, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.001	0.050	0.0012	0.05	Standard
G401	UA	Sulfate, total	mg/L	11/21/2015 - 08/17/2021	CI around median	2000	400	370	400	Standard
G401	UA	Thallium, total	mg/L	11/21/2015 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G401	UA	Total Dissolved Solids	mg/L	11/21/2015 - 08/17/2021	CI around median	2800	1200	840	1200	Standard
G402	UA	Antimony, total	mg/L	11/21/2015 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G402	UA	Arsenic, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	-0.00803	0.010	0.0043	0.01	Standard
G402	UA	Barium, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	-0.00763	2.0	0.18	2	Standard
G402	UA	Beryllium, total	mg/L	11/21/2015 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G402	UA	Boron, total	mg/L	11/21/2015 - 08/17/2021	CI around mean	5.5	2.0	0.022	2	Standard
G402	UA	Cadmium,total	mg/L	11/21/2015 - 08/17/2021	Most recent sample	0.001	0.005	0.001	0.005	Standard
G402	UA	Chloride, total	mg/L	11/21/2015 - 08/17/2021	CI around mean	1.8	200	75	200	Standard
G402	UA	Chromium, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	-0.00638	0.10	0.011	0.1	Standard
G402	UA	Cobalt, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	-0.00375	0.006	0.0056	0.006	Standard
G402	UA	Fluoride, total	mg/L	11/21/2015 - 08/17/2021	CI around median	0.30	4.0	0.49	4	Standard
G402	UA	Lead, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	-0.00606	0.0075	0.0063	0.0075	Standard
G402	UA	Lithium, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	0.011	0.040	0.013	0.04	Standard
G402	UA	Mercury, total	mg/L	11/21/2015 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G402	UA	Molybdenum, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	0.000642	0.10	0.0015	0.1	Standard
G402	UA	pH (field)	SU	11/21/2015 - 08/17/2021	CB around linear reg	6.7	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G402	UA	Radium-226 + Radium 228, tot	pCi/L	11/21/2015 - 08/17/2021	CB around linear reg	-1.04	5.0	2.0	5	Standard
G402	UA	Selenium, total	mg/L	11/21/2015 - 08/17/2021	CB around T-S line	-0.00028	0.050	0.0012	0.05	Standard
G402	UA	Sulfate, total	mg/L	11/21/2015 - 08/17/2021	CB around T-S line	421	400	370	400	Standard

Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G402	UA	Thallium, total	mg/L	11/21/2015 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G402	UA	Total Dissolved Solids	mg/L	11/21/2015 - 08/17/2021	CI around mean	1600	1200	840	1200	Standard
G403	UA	Antimony, total	mg/L	11/23/2015 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G403	UA	Arsenic, total	mg/L	11/23/2015 - 08/17/2021	CI around median	0.001	0.010	0.0043	0.01	Standard
G403	UA	Barium, total	mg/L	11/23/2015 - 08/17/2021	CB around linear reg	0.079	2.0	0.18	2	Standard
G403	UA	Beryllium, total	mg/L	11/23/2015 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G403	UA	Boron, total	mg/L	11/23/2015 - 08/17/2021	CI around mean	0.020	2.0	0.022	2	Standard
G403	UA	Cadmium,total	mg/L	11/23/2015 - 08/17/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G403	UA	Chloride, total	mg/L	11/23/2015 - 08/17/2021	CI around mean	3.7	200	75	200	Standard
G403	UA	Chromium, total	mg/L	11/23/2015 - 08/17/2021	CI around median	0.004	0.10	0.011	0.1	Standard
G403	UA	Cobalt, total	mg/L	11/23/2015 - 08/17/2021	CI around median	0.002	0.006	0.0056	0.006	Standard
G403	UA	Fluoride, total	mg/L	11/23/2015 - 08/17/2021	CI around mean	0.36	4.0	0.49	4	Standard
G403	UA	Lead, total	mg/L	11/23/2015 - 08/17/2021	CI around median	0.001	0.0075	0.0063	0.0075	Standard
G403	UA	Lithium, total	mg/L	11/23/2015 - 08/17/2021	All ND - Last	0.020	0.040	0.013	0.04	Standard
G403	UA	Mercury, total	mg/L	11/23/2015 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G403	UA	Molybdenum, total	mg/L	11/23/2015 - 08/17/2021	CI around median	0.001	0.10	0.0015	0.1	Standard
G403	UA	pH (field)	SU	11/23/2015 - 08/17/2021	CI around mean	6.9	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G403	UA	Radium-226 + Radium 228, tot	pCi/L	11/23/2015 - 08/17/2021	CI around mean	0.49	5.0	2.0	5	Standard
G403	UA	Selenium, total	mg/L	11/23/2015 - 08/17/2021	CI around median	0.001	0.050	0.0012	0.05	Standard
G403	UA	Sulfate, total	mg/L	11/23/2015 - 08/17/2021	CB around linear reg	28	400	370	400	Standard
G403	UA	Thallium, total	mg/L	11/23/2015 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G403	UA	Total Dissolved Solids	mg/L	11/23/2015 - 08/17/2021	CI around geomean	317	1200	840	1200	Standard
G404	UA	Antimony, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G404	UA	Arsenic, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.001	0.010	0.0043	0.01	Standard
G404	UA	Barium, total	mg/L	10/07/2015 - 08/17/2021	CI around mean	0.040	2.0	0.18	2	Standard
G404	UA	Beryllium, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard

Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G404	UA	Boron, total	mg/L	10/07/2015 - 08/17/2021	CI around mean	2.6	2.0	0.022	2	Standard
G404	UA	Cadmium,total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G404	UA	Chloride, total	mg/L	10/07/2015 - 08/17/2021	CB around linear reg	164	200	75	200	Standard
G404	UA	Chromium, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.004	0.10	0.011	0.1	Standard
G404	UA	Cobalt, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.002	0.006	0.0056	0.006	Standard
G404	UA	Fluoride, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.25	4.0	0.49	4	Standard
G404	UA	Lead, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.001	0.0075	0.0063	0.0075	Standard
G404	UA	Lithium, total	mg/L	11/21/2015 - 08/17/2021	CB around linear reg	0.017	0.040	0.013	0.04	Standard
G404	UA	Mercury, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G404	UA	Molybdenum, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.10	0.0015	0.1	Standard
G404	UA	pH (field)	SU	10/07/2015 - 08/17/2021	CI around mean	6.9	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G404	UA	Radium-226 + Radium 228, tot	pCi/L	11/21/2015 - 08/17/2021	CI around mean	0.57	5.0	2.0	5	Standard
G404	UA	Selenium, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.050	0.0012	0.05	Standard
G404	UA	Sulfate, total	mg/L	10/07/2015 - 08/17/2021	CI around mean	203	400	370	400	Standard
G404	UA	Thallium, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G404	UA	Total Dissolved Solids	mg/L	10/07/2015 - 08/17/2021	CB around linear reg	837	1200	840	1200	Standard
G405	UA	Antimony, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G405	UA	Arsenic, total	mg/L	10/07/2015 - 08/17/2021	CB around T-S line	-0.0105	0.010	0.0043	0.01	Standard
G405	UA	Barium, total	mg/L	10/07/2015 - 08/17/2021	CI around geomean	0.020	2.0	0.18	2	Standard
G405	UA	Beryllium, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G405	UA	Boron, total	mg/L	10/07/2015 - 08/17/2021	CB around linear reg	3.2	2.0	0.022	2	Standard
G405	UA	Cadmium,total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.001	0.005	0.001	0.005	Standard
G405	UA	Chloride, total	mg/L	10/07/2015 - 08/17/2021	CI around geomean	9.2	200	75	200	Standard
G405	UA	Chromium, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.004	0.10	0.011	0.1	Standard
G405	UA	Cobalt, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.002	0.006	0.0056	0.006	Standard
G405	UA	Fluoride, total	mg/L	10/07/2015 - 08/17/2021	CI around mean	0.42	4.0	0.49	4	Standard



Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G405	UA	Lead, total	mg/L	10/07/2015 - 08/17/2021	CB around T-S line	-0.00599	0.0075	0.0063	0.0075	Standard
G405	UA	Lithium, total	mg/L	11/21/2015 - 08/17/2021	CB around T-S line	0.010	0.040	0.013	0.04	Standard
G405	UA	Mercury, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G405	UA	Molybdenum, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.001	0.10	0.0015	0.1	Standard
G405	UA	pH (field)	SU	10/07/2015 - 08/17/2021	CI around mean	6.8	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G405	UA	Radium-226 + Radium 228, tot	pCi/L	11/21/2015 - 08/17/2021	CI around geomean	0.57	5.0	2.0	5	Standard
G405	UA	Selenium, total	mg/L	10/07/2015 - 08/17/2021	CI around median	0.001	0.050	0.0012	0.05	Standard
G405	UA	Sulfate, total	mg/L	10/07/2015 - 08/17/2021	CB around linear reg	195	400	370	400	Standard
G405	UA	Thallium, total	mg/L	10/07/2015 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G405	UA	Total Dissolved Solids	mg/L	10/07/2015 - 08/17/2021	CB around linear reg	720	1200	840	1200	Standard
G406	UA	Antimony, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G406	UA	Arsenic, total	mg/L	10/14/2020 - 08/17/2021	CI around median	0	0.010	0.0043	0.01	Standard
G406	UA	Barium, total	mg/L	10/14/2020 - 08/17/2021	CI around median	0	2.0	0.18	2	Standard
G406	UA	Beryllium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G406	UA	Boron, total	mg/L	10/14/2020 - 08/17/2021	CI around median	0	2.0	0.022	2	Standard
G406	UA	Cadmium,total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G406	UA	Chloride, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.99	200	75	200	Standard
G406	UA	Chromium, total	mg/L	10/14/2020 - 08/17/2021	CI around median	0	0.10	0.011	0.1	Standard
G406	UA	Cobalt, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.002	0.006	0.0056	0.006	Standard
G406	UA	Fluoride, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.066	4.0	0.49	4	Standard
G406	UA	Lead, total	mg/L	10/14/2020 - 08/17/2021	CI around median	0	0.0075	0.0063	0.0075	Standard
G406	UA	Lithium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.020	0.040	0.013	0.04	Standard
G406	UA	Mercury, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G406	UA	Molybdenum, total	mg/L	10/14/2020 - 08/17/2021	CI around median	0	0.10	0.0015	0.1	Standard
G406	UA	pH (field)	SU	10/14/2020 - 08/17/2021	CI around mean	6.2	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G406	UA	Radium-226 + Radium 228, tot	pCi/L	10/14/2020 - 08/17/2021	CI around mean	-0.441	5.0	2.0	5	Standard



Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G406	UA	Selenium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.050	0.0012	0.05	Standard
G406	UA	Sulfate, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	-191	400	370	400	Standard
G406	UA	Thallium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G406	UA	Total Dissolved Solids	mg/L	10/14/2020 - 08/17/2021	CI around median	0	1200	840	1200	Standard
G407	UA	Antimony, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G407	UA	Arsenic, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.010	0.0043	0.01	Standard
G407	UA	Barium, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.011	2.0	0.18	2	Standard
G407	UA	Beryllium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G407	UA	Boron, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.038	2.0	0.022	2	Standard
G407	UA	Cadmium,total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G407	UA	Chloride, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	8.1	200	75	200	Standard
G407	UA	Chromium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.004	0.10	0.011	0.1	Standard
G407	UA	Cobalt, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.002	0.006	0.0056	0.006	Standard
G407	UA	Fluoride, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.25	4.0	0.49	4	Standard
G407	UA	Lead, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.0075	0.0063	0.0075	Standard
G407	UA	Lithium, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.033	0.040	0.013	0.04	Standard
G407	UA	Mercury, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G407	UA	Molybdenum, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.00122	0.10	0.0015	0.1	Standard
G407	UA	pH (field)	SU	10/14/2020 - 08/17/2021	CI around mean	6.4	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G407	UA	Radium-226 + Radium 228, tot	pCi/L	10/14/2020 - 08/17/2021	CI around mean	-0.476	5.0	2.0	5	Standard
G407	UA	Selenium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.050	0.0012	0.05	Standard
G407	UA	Sulfate, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	177	400	370	400	Standard
G407	UA	Thallium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G407	UA	Total Dissolved Solids	mg/L	10/14/2020 - 08/17/2021	CI around mean	1520	1200	840	1200	Standard
G410	UA	Antimony, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G410	UA	Arsenic, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	-0.000536	0.010	0.0043	0.01	Standard



Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G410	UA	Barium, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.12	2.0	0.18	2	Standard
G410	UA	Beryllium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G410	UA	Boron, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.077	2.0	0.022	2	Standard
G410	UA	Cadmium,total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G410	UA	Chloride, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	304	200	75	200	Standard
G410	UA	Chromium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.004	0.10	0.011	0.1	Standard
G410	UA	Cobalt, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	0.00188	0.006	0.0056	0.006	Standard
G410	UA	Fluoride, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.25	4.0	0.49	4	Standard
G410	UA	Lead, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.0075	0.0063	0.0075	Standard
G410	UA	Lithium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.020	0.040	0.013	0.04	Standard
G410	UA	Mercury, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G410	UA	Molybdenum, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.10	0.0015	0.1	Standard
G410	UA	pH (field)	SU	10/14/2020 - 08/17/2021	CI around mean	6.2	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G410	UA	Radium-226 + Radium 228, tot	pCi/L	10/14/2020 - 08/17/2021	CI around mean	-0.742	5.0	2.0	5	Standard
G410	UA	Selenium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.050	0.0012	0.05	Standard
G410	UA	Sulfate, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	30	400	370	400	Standard
G410	UA	Thallium, total	mg/L	10/14/2020 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G410	UA	Total Dissolved Solids	mg/L	10/14/2020 - 08/17/2021	CI around mean	502	1200	840	1200	Standard
G411	UA	Antimony, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.003	0.006	0.003	0.006	Standard
G411	UA	Arsenic, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.001	0.010	0.0043	0.01	Standard
G411	UA	Barium, total	mg/L	10/13/2020 - 08/17/2021	CI around mean	0.022	2.0	0.18	2	Standard
G411	UA	Beryllium, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.001	0.004	0.001	0.004	Standard
G411	UA	Boron, total	mg/L	10/13/2020 - 08/17/2021	CI around mean	0.00782	2.0	0.022	2	Standard
G411	UA	Cadmium,total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.001	0.005	0.001	0.005	Standard
G411	UA	Chloride, total	mg/L	10/13/2020 - 08/17/2021	CI around mean	2.4	200	75	200	Standard
G411	UA	Chromium, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.004	0.10	0.011	0.1	Standard

Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G411	UA	Cobalt, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.002	0.006	0.0056	0.006	Standard
G411	UA	Fluoride, total	mg/L	10/13/2020 - 08/17/2021	CI around mean	0.56	4.0	0.49	4	Standard
G411	UA	Lead, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.001	0.0075	0.0063	0.0075	Standard
G411	UA	Lithium, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.020	0.040	0.013	0.04	Standard
G411	UA	Mercury, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.0002	0.002	0.0002	0.002	Standard
G411	UA	Molybdenum, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.001	0.10	0.0015	0.1	Standard
G411	UA	pH (field)	SU	10/13/2020 - 08/17/2021	CI around mean	6.8	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G411	UA	Radium-226 + Radium 228, tot	pCi/L	10/13/2020 - 08/17/2021	CI around mean	-0.774	5.0	2.0	5	Standard
G411	UA	Selenium, total	mg/L	10/13/2020 - 08/17/2021	CI around median	0	0.050	0.0012	0.05	Standard
G411	UA	Sulfate, total	mg/L	10/13/2020 - 08/17/2021	CI around mean	141	400	370	400	Standard
G411	UA	Thallium, total	mg/L	10/13/2020 - 08/17/2021	All ND - Last	0.001	0.002	0.001	0.002	Standard
G411	UA	Total Dissolved Solids	mg/L	10/13/2020 - 08/17/2021	CI around mean	567	1200	840	1200	Standard


TABLE 1. DETERMINATION OF POTENTIAL EXCEEDANCES

HISTORY OF POTENTIAL EXCEEDANCES COFFEEN POWER PLANT ASH POND NO. 2 COFFEEN, ILLINOIS

Notes:

Potential exceedance of GWPS HSU = hydrostratigraphic unit: UA = uppermost aquifer mg/L = milligrams per literpCi/L = picocuries per literSU = standard units Statistical Calculation = method used to calculate the statistical result: All ND - Last = All results were below the reporting limit, and the last determined reporting limit is shown CB around linear reg = Confidence band around linear regression CB around T-S line = Confidence band around Thiel-Sen line CI around geomean = Confidence interval around the geometric mean CI around mean = Confidence interval around the mean CI around median = Confidence interval around the median Most recent sample = Result for the most recently collected sample used due to insufficient data Statistical Result = calculated in accordance with Statistical Analysis Plan using constituent concentrations observed at monitoring well during all sampling events within the specified date range For pH, the values presented are the lower / upper limits GWPS = Groundwater Protection Standard GWPS Source:

Standard = standard specified in 35 I.A.C. § 845.600(a)(1)

Background = background concentration (see cover page for additional information)



TABLE 2. SUMMARY OF POTENTIAL EXCEEDANCES

HISTORY OF POTENTIAL EXCEEDANCES COFFEEN POWER PLANT ASH POND NO. 2 COFFEEN, ILLINOIS

Sample Location	HSU	Constituent	Result Unit	Sample Date Range	Statistical Calculation	Statistical Result	GWPS	Background	Part 845 Standard	GWPS Source
G401	UA	Boron, total	mg/L	11/21/2015 - 08/17/2021	CI around median	3.5	2.0	0.022	2	Standard
G401	UA	Cobalt, total	mg/L	11/21/2015 - 08/17/2021	CI around mean	0.22	0.006	0.0056	0.006	Standard
G401	UA	pH (field)	SU	11/21/2015 - 08/17/2021	CI around mean	6.0	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G401	UA	Sulfate, total	mg/L	11/21/2015 - 08/17/2021	CI around median	2000	400	370	400	Standard
G401	UA	Total Dissolved Solids	mg/L	11/21/2015 - 08/17/2021	CI around median	2800	1200	840	1200	Standard
G402	UA	Boron, total	mg/L	11/21/2015 - 08/17/2021	CI around mean	5.5	2.0	0.022	2	Standard
G402	UA	Sulfate, total	mg/L	11/21/2015 - 08/17/2021	CB around T-S line	421	400	370	400	Standard
G402	UA	Total Dissolved Solids	mg/L	11/21/2015 - 08/17/2021	CI around mean	1600	1200	840	1200	Standard
G404	UA	Boron, total	mg/L	10/07/2015 - 08/17/2021	CI around mean	2.6	2.0	0.022	2	Standard
G405	UA	Boron, total	mg/L	10/07/2015 - 08/17/2021	CB around linear reg	3.2	2.0	0.022	2	Standard
G406	UA	pH (field)	SU	10/14/2020 - 08/17/2021	CI around mean	6.2	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G407	UA	pH (field)	SU	10/14/2020 - 08/17/2021	CI around mean	6.4	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard
G407	UA	Total Dissolved Solids	mg/L	10/14/2020 - 08/17/2021	CI around mean	1520	1200	840	1200	Standard
G410	UA	Chloride, total	mg/L	10/14/2020 - 08/17/2021	CI around mean	304	200	75	200	Standard
G410	UA	pH (field)	SU	10/14/2020 - 08/17/2021	CI around mean	6.2	6.5/9.0	6.7/7.3	6.5/9	Standard/Standard

Notes:

HSU = hydrostratigraphic unit:

UA = uppermost aquifer

mg/L = milligrams per liter

pCi/L = picocuries per liter

SU = standard units

Statistical Calculation = method used to calculate the statistical result:

CB around linear reg = Confidence band around linear regression CB around T-S line = Confidence band around Thiel-Sen line

CI around mean = Confidence interval around the mean

CI around median = Confidence interval around the median

Statistical Result = calculated in accordance with Statistical Analysis Plan using constituent concentrations observed at monitoring well during all sampling events within the specified date range For pH, the values presented are the lower / upper limits

GWPS = Groundwater Protection Standard

GWPS Source:

Standard = standard specified in 35 I.A.C. § 845.600(a)(1)

Background = background concentration (see cover page for additional information)



ATTACHMENT N

Certification of Financial Assurance Requirements

On June 17, 2021, Illinois Power Generating Company provided financial assurance in the form of performance bonds to the Illinois Environmental Protection Agency in the amount of \$27,884,983 for Ash Pond 1, Ash Pond 2, and the GMF Pond System at the Coffeen Power Plant.¹

I, Matthew A. Goering, Senior Vice President of Illinois Power Generating Company, do hereby certify to the best of my knowledge for the above referenced CCR Units that the financial assurance instruments satisfy the requirements of 35 I.A.C. Part 845, Subpart I.

Matthew A. Goering Senior Vice President Illinois Power Generating Company

¹In the operating permit applications, the GMF Pond System is referred to as the GMF Gypsum Stack Pond and GMF Recycle Pond.

October